

Technical Support Document for the Final "Revised Definition of 'Waters of the United States'" Rule

U.S. Environmental Protection Agency

and

Department of the Army

December 2022



This Technical Support Document addresses in more detail the existing scientific literature and technical information in support of the final rule of the U.S. Environmental Protection Agency (EPA) and U.S. Department of the Army's ("the agencies") which revises the definition of "waters of the United States."¹ The Preamble, the 2015 report *Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence*, this Technical Support Document, and the rest of the administrative record provide the basis for the definition of "waters of the United States" established in the final rule. Where this Technical Support Document does not reflect the language in the preamble and final rule controls and should be used for purposes of understanding the scope, requirements, and basis of the final rule.

Table of Contents

Table of Contents	2
Supplementary Materials	7
List of Tables	
List of Figures	
Abbreviations	10
I. Science Report, More Recent Literature, and Other Scientific Support	12
A. Science Report: Synthesis of Peer-Reviewed Scientific Literature	12
i. Summary of Major Conclusions of the Science Report	14
ii. Discussion of Major Conclusions	
iii. Key Findings for the Science Report's Major Conclusions	19
iv. Science Report: Framework for Analysis	
B. Peer Review of Report	
C. Updates to the Literature Since Publication of the Science Report	47
i. Update Process	
ii. Results	50
1. Process	50
2. Analysis and Synthesis	
iii. Discussion	55
1. Ephemeral, Intermittent, and Perennial Streams	

¹ The *Technical Support Document for the Proposed "Revised Definition of 'Waters of the United States' Rule"* contained a legal background section and an Appendix D, "Traditional Navigable Waters." Those sections have been moved from this Technical Support Document for the final rule to the response to comments documents or to separate documents that have been added to the docket for this rule. This includes section 6 of the agencies' response to comments document and "History of the Effects of Litigation over Recent Definitions of 'Waters of the United States' (formerly contained in the legal background section) and "Waters that Qualify as 'Traditional Navigable Waters' Under Section (a)(1) of the Agencies' Regulations" (formerly contained in Appendix D of the Technical Support Document for the Proposed Rule).

	2.	Floodplain Wetlands and Open Waters	61
	3.	Non-Floodplain Wetlands and Open Waters	63
iv	. A	Abstracts Noted in the Screening Process to Disagree with the Major Conclusions of the	
S	ciend	ce Report	67
	1.	Ephemeral, Intermittent, and Perennial Stream Systems	69
	2.	Floodplain Wetland and Open Water Systems	70
	3.	Non-Floodplain Wetland and Open Water Systems	71
V	. S	creening Benefits and Limitations	72
V	i.F	Review of Additional Literature	73
V	ii. S	summary and Conclusions	74
D.	Clo	sing Comments on the Science Report and Updated Literature	75
E.	Oth	er Scientific Support	77
F.	Em	erging Science	77
G.	SA	B Review of the Proposed Rule	78
H.	Oth	er Scientific Information	79
i.	E	Ecosystem Services	79
II. E	xecu	tive Order 13990 and Review of the Navigable Waters Protection Rule	81
A.	Exe	ecutive Order 13990	81
B.	Rev	view of the 2020 NWPR	82
i.	I	mpacts of the 2020 NWPR	82
	1.	Review of Jurisdictional Determinations and Permit Data	83
ii	. S	takeholder Concerns	. 108
ii	i. S	cientific and Technical Review	. 109
	1.	2020 NWPR	. 109
	2.	White Paper	. 129
iv	7. I	mplementation Challenges	. 135
	1.	Typical Year Metric	. 136
	2.	Determining Adjacency	. 140
	3.	Ditches	. 141
	4.	Results of Regional Survey	. 142
C.	Clii	nate Change	. 145
D.	Env	vironmental Justice	. 148
III. S	cient	ific Support for the Final Rule	. 149
A.	Tril	outaries	. 151

i. Tributaries Can Provide Functions that Restore and Maintain the Physical Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
ii. Tributaries Can Provide Functions that Restore and Maintain the Chemical Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
iii. Tributaries Can Provide Functions that Restore and Maintain the Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
iv. Human-made or Human-altered Tributaries Can Provide Functions that Restore and Maintain the Chemical, Physical, and Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
v. Ephemeral and Intermittent Tributaries Can Provide Functions that Restore and Maintain the Chemical, Physical, or Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
vi. Tributary Lakes and Ponds Can Provide Functions that Restore and Maintain the Chemical, Physical, or Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, or Interstate Waters
B. Adjacent Wetlands
i. Adjacent Wetlands under the Final Rule
 Wetlands Adjacent to Traditional Navigable Waters, the Territorial Seas, or Interstate Waters 169
2. Adjacent Wetlands under the Relatively Permanent Standard
3. Adjacent Wetlands under the Significant Nexus Standard
ii. Definition of "Adjacent" Wetlands
1. Bordering, Contiguous, or Neighboring Wetlands172
2. Determination of Adjacent Wetlands
iii. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Physical Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
iv. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Chemical Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
v. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Biological Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
C. Impoundments
D. Intrastate Lakes and Ponds, Streams, or Wetlands Evaluated Under Paragraph (a)(5)202
i. Intrastate Waters Evaluated Under Paragraph (a)(5) Can Provide Functions that Restore and Maintain the Chemical, Physical, and Biological Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters
F Significant Nexus Standard 21/
i Waters Subject to the Significant Nexus Standard 214
ii "Similarly Situated"
iii "In the Region"
m. m un Region

iv. "Significantly Affect"	222
1. The significant nexus standard allows for consideration of the effects of climate change water resources consistent with the best available science	on 231
F. The Relatively Permanent Standard	232
IV. Implementation of the Final Rule	237
A. Resources for Making Jurisdictional Determinations	237
i. Available Tools	238
1. Mapping and Remote Sensing	239
2. Hydrologic Models	243
3. Advancements in Implementation Data, Tools, and Methods	244
ii. Identifying Tributaries	248
1. Identifying Tributaries That Meet the Relatively Permanent Standard	252
iii. Identifying Wetlands	257
1. Identifying Wetlands Adjacent to Traditional Navigable Waters, the Territorial Seas, Interstate Waters, Impoundments, or Tributaries	258
iv. Impoundments	261
v. Applying a Significant Nexus Standard	262
B. Case Specific Significant Nexus Analysis	265
i. Similarly Situated	265
ii. In the Region	265
Appendices Table of Contents	269
Appendices List of Tables	269
Appendix A: Glossary	270
Appendix B: References	284
Appendix C: References from the Literature Update Screening and Public Comments on Literature Published Since 2014	362
Appendix C1: References Relevant to the Conclusions of the Science Report Published Since 2014	4 362
Ephemeral, Intermittent, or Perennial Streams	362
Floodplain Wetlands and Open Waters	440
Non-floodplain Wetlands and Open Waters	492
Appendix C2: Plain-Text Language from the Abstracts of Illustrative Scientific Papers	531
Ephemeral, Intermittent, and Perennial Streams	531
Floodplain Wetlands and Open Waters	541
Non-Floodplain Wetlands and Open Waters	546
Appendix C3: Additional References	554

Appendix C3i: Reference Review Process and Findings	554
Appendix C3ii: Additional References Not Captured During Screening Process (October 2021)	560

Supplementary Materials

Supplementary Materials are separate

Supplementary Material A: Scientific Papers Selected for Forward-Citation Mapping Supplementary Material B: Papers Screened Early to Expediate Machine-Learning Model Building Supplementary Material C: Questions Answered from Each Included Scientific Paper's Abstract

List of Tables

Table 1: Jurisdictional determinations over time 91
Table 2: The 2020 NWPR compared to prior years of implementation under pre-2015 practice (AJD/PJD)
Table 3: The 2020 NWPR compared to prior years of implementation under pre-2015 practice (jurisdictional/non-jurisdictional) 93
Table 4: The 2020 NWPR compared to prior years of pre-2015 practice and 2015 Clean Water Rule implementation (jurisdictional/non-jurisdictional) 94
Table 5: Specific Resources found Non-Jurisdictional under the 2020 NWPR 98
Table 6: Arizona and New Mexico Impacts to Scope of "Waters of the United States" under the 2020 NWPR (2020-2021) Compared to Pre-2015 Practice (2016-2017, 2017-2018, 2018-2019, 2019-2020) 100
Table 7: Test of statistical significance for 2020 NWPR compared to prior years of implementation underpre-2015 practice for stream resources in Arizona and New Mexico102
Table C-1: References Relevant to the Conclusions of the Science Report (of the 37 Identified by theAgencies in October 2021 as Not Captured During the Screening Process).555
Table C-2: Peer-Reviewed References Published Since 2014 Provided During the Public CommentPeriod Relevant to the Conclusions of the Science Report

List of Figures

Figure 1: AJD and PJD (percentages)
Figure 2: AJD and PJD resources (percentages)
Figure 3: AJD findings (percentages)
Figure 4: AJD Findings by Resource (Percentages)
Figure 5: Individual aquatic resources found to be non-jurisdictional in Arizona over the past five years 103
Figure 6: Individual aquatic resources found to be non-jurisdictional in New Mexico over the past five years
Figure 7: Arizona Individual Aquatic Resources Tied to PJDs and Determination Substitutes by Year104
Figure 8: New Mexico Individual Aquatic Resources Tied to PJDs and Determination Substitutes by Year 105

Figure 9: Actions Not Requiring Permits Under Pre-2015 Practice and 2020 NWPR	
Figure 10: Proportion of questionnaire respondents from each EPA Region	
Figure 11: Issues cited by EPA Regional staff as the "biggest implementation challenge" w NWPR	rith the 2020
Figure 12: A generalized example of a stream network to illustrate stream order	

Abbreviations

2020 NWPR	Navigable Waters Protection Rule
AJD	Approved jurisdictional determination
Army	U.S. Department of the Army
CFR	Code of Federal Regulations
Corps	U.S. Army Corps of Engineers
CWA	Clean Water Act
DEM	Digital elevation model
DOC	Dissolved Organic Carbon
EA	Economic analysis
E.O.	Executive Order
EPA	U.S. Environmental Protection Agency
FEMA	Federal Emergency Management Agency
FR	Federal Register
GIS	Geographic information systems
HAWQS	Hydrologic and Water Quality System
HUC	Hydrologic unit code
HUC4	4-digit hydrologic unit code
HUC12	12-digit hydrologic unit code
JD	Jurisdictional determination
LIDAR	Light detection and ranging
NASA	National Aeronautical and Space Administration
NFW	Non-floodplain wetland
NHD	National Hydrography Dataset
NOAA	National Oceanic and Atmospheric Administration
NPDES	National Pollutant Discharge Elimination System
NRCS	Natural Resources Conservation Service
NWI	National Wetlands Inventory
OCN	Optimal climate normals
ORM2	Operation and Maintenance Business Information Link, Regulatory Module
PJD	Preliminary jurisdictional determination

Rapanos	Rapanos v. United States, 547 U.S. 715 (2006)
DIIA	Divers and Harbors A at
KHA	Rivers and Harbors Act
Riverside Bayview Homes	United States v. Riverside Bayview Homes Inc., 474 U.S. 121 (1985)
RPA	Resource and Programmatic Assessment
RPWWN	Wetlands adjacent to but not directly abutting relatively permanent waters
SAB	Science Advisory Board
Science Report	Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence
SWANCC	Solid Waste Agency of Northern Cook County v. U.S. Army Corps of Engineers, 531 U.S. 159 (2001)
SWAT	Soil and Water Assessment Tool
TAS	Treatment in a Manner Similar as a State
TMDL	Total Maximum Daily Load
TNW	Traditional navigable waters
U.S.C.	United States Code
USDA	U.S. Department of Agriculture
USFS	U.S. Department of Agriculture Forest Service
USGS	U.S. Geological Survey
U.S. FWS	U.S. Fish and Wildlife Service
WATERS	Watershed Assessment, Tracking and Environmental Results System
WMO	World Meteorological Organization
WOTUS	Waters of the United States

I. Science Report, More Recent Literature, and Other Scientific Support

EPA's 2015 report *Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence* (hereafter the Science Report) summarizes and assesses relevant scientific literature that is part of the administrative record for this final rule. In addition, the agencies considered other sources of scientific information and literature, particularly for topics that were not addressed in the Science Report. This includes peer-reviewed literature, federal and state government reports, and other relevant information. The agencies also conducted a literature search for scientific literature that had been published since the Science Report's publication, which is described in more detail in section I.C. Section I.A.i of this document provides the major conclusions of the Science Report. Section III provides additional detail of the scientific literature and the agencies' reasoning in support of the rule. The agencies' interpretation of the Clean Water Act's scope in this final rule is guided by the best available peer-reviewed science, including on the connectivity and effects that streams, wetlands, and open waters have on the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

A. Science Report: Synthesis of Peer-Reviewed Scientific Literature

The Science Report provides much of the technical support for this final rule. The Science Report is based on a review of more than 1,300 peer-reviewed publications. EPA's Office of Research and Development prepared the Science Report, a peer-reviewed synthesis of published peer-reviewed scientific literature discussing the nature of connectivity and effects of tributaries and wetlands on downstream waters. The Science Report was directly considered in the development of this final rule, as was the peer review of the Science Report led by EPA's Science Advisory Board (SAB), and is available at http://cfpub.epa.gov/ncea/cfm/recordisplay.cfm?deid=296414. The SAB's comprehensive peer review of the Science Report is discussed in detail in section I.B. The Science Report also underwent an earlier external independent peer review, and the results of both peer reviews are available in the docket for the final rule. Prior to the earlier peer review, the Science Report also underwent a peer consultation.

The Science Report reviews and synthesizes the peer-reviewed scientific literature on the connectivity or isolation of streams and wetlands relative to large water bodies such as rivers, lakes, estuaries, and oceans. The purpose of the review and synthesis is to summarize current scientific understanding about the connectivity and mechanisms by which streams and wetlands, singly or in aggregate, affect the physical, chemical, and biological integrity of downstream waters. Specific types of connections considered in the Science Report include transport of physical materials and chemicals such as water, wood, and sediment, nutrients, pesticides, and mercury; movement of organisms or their seeds or eggs; and hydrologic and biogeochemical interactions occurring in surface and groundwater flows, including hyporheic zones and alluvial aquifers. A hyporheic zone is the area next to and beneath a stream or river in which hyporheic flow (water from a stream or river channel that enters subsurface materials of the stream bed and bank and then returns to the stream or river) occurs. Science Report at A-6. An

alluvial aquifer is an aquifer with geologic materials deposited by a stream or river (alluvium) that retains a hydraulic connection with the depositing stream. *Id.* at A-1.

The final Science Report states that connectivity is a foundational concept in hydrology and freshwater ecology. Connectivity is the degree to which components of a system are joined, or connected, by various transport mechanisms and is determined by the characteristics of both the physical landscape and the biota of the specific system. Connectivity for purposes of interpreting the scope of "waters of the United States" under the Clean Water Act serves to demonstrate the "nexus" between upstream water bodies and the downstream traditional navigable water, the territorial seas, or interstate water and the strength of those connections. The scientific literature does not use the term "significant" as it is used in the context of the geographic scope of the Clean Water Act, but it does provide information on the strength of the effects on the chemical, physical, and biological functioning of traditional navigable waters, the territorial seas, and interstate waters from the connections among tributaries, adjacent wetlands, and intrastate waters that do not meet the criteria for jurisdiction under other categories of the rule and those fundamental waters. The scientific literature also does not use the terms traditional navigable waters, the territorial seas, or interstate waters. However, evidence of strong chemical, physical, and biological connections to larger rivers, estuaries, and lakes applies to that subset of rivers, estuaries, and lakes that are traditional navigable waters, the territorial seas, or interstate waters.

The agencies reiterate their previous conclusion that determining the presence of a significant nexus is not a purely scientific inquiry. This section reflects the scientific consensus on the connections and the strength of the effects that upstream tributaries, adjacent wetlands, and intrastate waters that do not meet the criteria for jurisdiction under other categories of the rule can and do have on traditional navigable waters, the territorial seas, or interstate waters. However, a significant nexus determination requires legal, technical, and policy judgment, as well as scientific considerations, for example, to assess the significance of any effects.

The Science Report presents evidence of those connections from various categories of waters, evaluated singly or in combination, which affect downstream waters and the strength of that effect. The objectives of the Science Report are (1) to provide a context for considering the evidence of connections between downstream waters and their tributary waters, and (2) to summarize current understanding about these connections, the factors that influence them, and the mechanisms by which the connections affect the function or condition of downstream waters. The connections and mechanisms discussed in the Science Report include transport of physical materials and chemicals such as water, wood, sediment, nutrients, pesticides, and mercury; functions that adjacent wetlands perform, such as storing and cleansing water; movement of organisms or their seeds and eggs; and hydrologic and biogeochemical interactions occurring in and among surface and groundwater flows, including hyporheic zones² and alluvial aquifers.

The Science Report consists of six chapters. Chapter 1 outlines the purpose, scientific context, and approach of the report. Chapter 2 describes the components of a river system and watershed; the types of physical, chemical, and biological connections that link those components; the factors that influence connectivity at various temporal and spatial scales; and methods for quantifying connectivity. Chapter 3

² The hyporheic zone is the subsurface area immediately below the bed of intermittent and ephemeral streams that remains wet even when there is no surface flow. These areas are extremely important to macro-benthic organisms critical to the biochemical integrity of streams.

reviews literature on connectivity in stream networks in terms of physical, chemical, and biological connections and their resulting effects on downstream waters. Chapter 4 reviews literature on the connectivity and effects of nontidal wetlands and certain open waters on downstream waters. Chapter 5 applies concepts and evidence from previous chapters to six case studies from published literature on Carolina and Delmarva bays, oxbow lakes, prairie potholes, prairie streams, southwestern streams, and vernal pools. Chapter 6 summarizes key findings and conclusions, identifies data gaps, and briefly discusses research approaches that could fill those gaps. A glossary of scientific terms used in the report and detailed case studies of selected systems (summarized in Chapter 5) are included in Appendix A and Appendix B of the Report, respectively.

Since its publication in 2015, the scope, findings, and conclusions of the Science Report have at times been misconstrued as inclusive of all waters and all types of interconnections, regardless of their relevance to the Clean Water Act. This is not the case. As explained in the Executive Summary and Introduction of the Report, the scope of the Science Report was clearly restricted to specific types of surface waters, and considered only those connections having clearly documented scientific effects on the integrity of larger downstream waters.

i. Summary of Major Conclusions of the Science Report

Based on the review and synthesis of more than 1,300 publications from the peer-reviewed scientific literature, the evidence supports the Science Report's five major conclusions. Citations have been omitted from the text to improve readability; please refer to the Executive Summary and individual chapters of the Science Report for supporting publications and additional information.

Conclusion 1: Streams

The scientific literature unequivocally demonstrates that streams, individually or cumulatively, exert a strong influence on the integrity of downstream waters. All tributary streams, including perennial, intermittent, and ephemeral streams, are physically, chemically, and biologically connected to downstream rivers via channels and associated alluvial deposits where water and other materials are concentrated, mixed, transformed, and transported. Streams are the dominant source of water in most rivers, and the majority of tributaries are perennial, intermittent, or ephemeral headwater streams. Headwater streams also convey water into local storage compartments such as ponds, shallow aquifers, or stream banks, and into regional and alluvial aquifers; these local storage compartments are important sources of water for maintaining baseflow in rivers. In addition to water, streams transport sediment, wood, organic matter, nutrients, chemical contaminants, and many of the organisms found in rivers. The literature provides robust evidence that streams are biologically connected to downstream waters by the dispersal and migration of aquatic and semiaquatic organisms, including fish, amphibians, plants, microorganisms, and invertebrates, that use both upstream and downstream habitats during one or more stages of their life cycles, or provide food resources to downstream communities. In addition to material transport and biological connectivity, ephemeral, intermittent, and perennial flows influence fundamental biogeochemical processes by connecting channels and shallow ground water with other landscape elements. Physical, chemical, and biological connections between streams and downstream waters interact via integrative processes such as nutrient spiraling, in which stream communities assimilate and chemically transform large quantities of nitrogen and other nutrients that otherwise would be transported

directly downstream, increasing nutrient loads and associated impairments due to excess nutrients in downstream waters.

Conclusion 2: Riparian/Floodplain Wetlands and Open Waters

The literature clearly shows that wetlands and open waters in riparian areas and floodplains are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality, including the temporary storage and deposition of channel-forming sediment and woody debris, temporary storage of local ground water that supports baseflow in rivers, and transformation and transport of stored organic matter. Riparian/floodplain wetlands and open waters improve water quality through the assimilation, transformation, or sequestration of pollutants, including excess nutrients and chemical contaminants such as pesticides and metals, that can degrade downstream water integrity. In addition to providing effective buffers to protect downstream waters from point source and nonpoint source pollution, these systems form integral components of river food webs, providing nursery habitat for breeding fish and amphibians, colonization opportunities for stream invertebrates, and maturation habitat for stream insects. Lateral expansion and contraction of the river in its floodplain result in an exchange of organic matter and organisms, including fish populations that are adapted to use floodplain habitats for feeding and spawning during high water, that are critical to river ecosystem function. Riparian/floodplain wetlands and open waters also affect the integrity of downstream waters by subsequently releasing (desynchronizing) floodwaters and retaining large volumes of stormwater, sediment, and contaminants in runoff that could otherwise negatively affect the condition or function of downstream waters.

Conclusion 3: Non-Floodplain Wetlands and Open Waters

Wetlands and open waters in non-floodplain landscape settings (hereafter called "non-floodplain wetlands") provide numerous functions that benefit downstream water integrity. These functions include storage of floodwater; recharge of ground water that sustains river baseflow; retention and transformation of nutrients, metals, and pesticides; export of organisms or reproductive propagules (*e.g.*, seeds, eggs, spores) to downstream waters; and habitats needed for stream species. This diverse group of wetlands (*e.g.*, many prairie potholes, vernal pools, playa lakes) can be connected to downstream waters through surface-water, shallow subsurface-water, and groundwater flows and through biological and chemical connections.

In general, connectivity of non-floodplain wetlands occurs along a gradient (Conclusion 4), and can be described in terms of the frequency, duration, magnitude, timing, and rate of change of water, material, and biotic fluxes to downstream waters. These descriptors are influenced by climate, geology, and terrain, which interact with factors such as the magnitudes of the various functions within wetlands (*e.g.*, amount of water storage or carbon export) and their proximity to downstream waters to determine where wetlands occur along the connectivity gradient. At one end of this gradient, the functions of non-floodplain wetlands clearly affect the condition of downstream waters if a visible (*e.g.*, channelized) surface-water or a regular shallow subsurface-water connection to the river network is present. For non-floodplain wetlands lacking a channelized surface or regular shallow subsurface connection (*i.e.*, those at intermediate points along the gradient of connectivity), generalizations about their specific effects on downstream waters from the available literature are difficult because information on both function and connectivity is needed. Although there is ample evidence that non-floodplain wetlands provide

hydrologic, chemical, and biological functions that affect material fluxes (as of publication of the Science Report in 2015), few scientific studies explicitly addressing connections between non-floodplain wetlands and river networks have been published in the peer-reviewed literature. Even fewer publications specifically focus on the frequency, duration, magnitude, timing, or rate of change of these connections. In addition, although areas that are closer to rivers and streams have a higher probability of being connected than areas farther away when conditions governing the type and quantity of flows—including soil infiltration rate, wetland storage capacity, hydraulic gradient, etc.—are similar, information to determine if this similarity holds is generally not provided in the studies we reviewed (for the Science Report). Thus, current science (as of the Report's publication in 2015) does not support evaluations of the degree of connectivity for specific groups or classes of wetlands (*e.g.*, prairie potholes or vernal pools). Evaluations of individual wetlands or groups of wetlands, however, could be possible through case-by-case analysis.

Some effects of non-floodplain wetlands on downstream waters are due to their isolation, rather than their connectivity. Wetland "sink" functions that trap materials and prevent their export to downstream waters (*e.g.*, sediment and entrained pollutant removal, water storage) result because of the wetland's ability to isolate material fluxes. To establish that such functions influence downstream waters, we also need to know that the wetland intercepts materials that otherwise would reach the downstream water. The literature reviewed does provide limited examples of direct effects of wetland isolation on downstream waters, but not for classes of wetlands (*e.g.*, vernal pools). Nevertheless, the literature reviewed supports the conclusion that sink functions of non-floodplain wetlands, which result in part from their relative isolation, will affect a downstream water when these wetlands are situated between the downstream water and known point or nonpoint sources of pollution, and thus intersect flowpaths between the pollutant source and downstream waters.

Conclusion 4: Degrees and Determinants of Connectivity

Watersheds are integrated at multiple spatial and temporal scales by flows of surface water and ground water, transport and transformation of physical and chemical materials, and movements of organisms. Although all parts of a watershed are connected to some degree—by the hydrologic cycle or dispersal of organisms, for example—the degree and downstream effects of those connections vary spatially and temporally, and are determined by characteristics of the physical, chemical, and biological environments and by human activities.

Stream and wetland connections have particularly important consequences for downstream water integrity. Most of the materials—broadly defined as any physical, chemical, or biological entity—in rivers, for example, originate from aquatic ecosystems located upstream or elsewhere in the watershed. Longitudinal flows through ephemeral, intermittent, and perennial stream channels are much more efficient for transport of water, materials, and organisms than diffuse overland flows, and areas that concentrate water provide mechanisms for the storage and transformation, as well as transport, of materials.

Connectivity of streams and wetlands to downstream waters occurs along a continuum that can be described in terms of the frequency, duration, magnitude, timing, and rate of change of water, material,

and biotic fluxes to downstream waters. These terms, which are referred to collectively as connectivity descriptors, characterize the range over which streams and wetlands vary and shift along the connectivity gradient in response to changes in natural and anthropogenic factors and, when considered in a watershed context, can be used to predict probable effects of different degrees of connectivity over time. The evidence unequivocally demonstrates that the stream channels and riparian/floodplain wetlands or open waters that together form river networks are clearly connected to downstream waters in ways that profoundly influence downstream water integrity. The connectivity and effects of non-floodplain wetlands and open waters are more variable and thus more difficult to address solely from evidence available in peer-reviewed studies.

Variations in the degree of connectivity influence the range of functions provided by streams and wetlands, and are critical to the integrity and sustainability of downstream waters. Connections with low values of one or more descriptors (*e.g.*, low-frequency, low-duration streamflows caused by flash floods) can have important downstream effects when considered in the context of other descriptors (*e.g.*, large magnitude of water transfer). At the other end of the frequency range, high-frequency, low-magnitude vertical (surface-subsurface) and lateral flows contribute to aquatic biogeochemical processes, including nutrient and contaminant transformation and organic matter accumulation. The timing of an event can alter both connectivity and the magnitude of its downstream effect. For example, when soils become saturated by previous rainfall events, even low or moderate rainfall can cause streams or wetlands to overflow, transporting water and materials to downstream waters. Fish that use nonperennial or perennial headwater stream habitats to spawn or rear young, and invertebrates that move into seasonally inundated floodplain wetlands prior to emergence, have life cycles that are synchronized with the timing of flows, temperature thresholds, and food resource availability in those habitats.

Conclusion 5: Cumulative Effects

The incremental effects of individual streams and wetlands are cumulative across entire watersheds and therefore must be evaluated in context with other streams and wetlands. Downstream waters are the time-integrated result of all waters contributing to them. For example, the amount of water or biomass contributed by a specific ephemeral stream in a given year might be small, but the aggregate contribution of that stream over multiple years, or by all ephemeral streams draining that watershed in a given year or over multiple years, can have substantial consequences on the integrity of the downstream waters. Similarly, the downstream effect of a single event, such as pollutant discharge into a single stream or wetland, might be negligible but the cumulative effect of multiple discharges could degrade the integrity of downstream waters.

In addition, when considering the effect of an individual stream or wetland, all contributions and functions of that stream or wetland should be evaluated cumulatively. For example, the same stream transports water, removes excess nutrients, mitigates flooding, and provides refuge for fish when conditions downstream are unfavorable; if any of these functions is ignored, the overall effect of that stream would be underestimated.

ii. Discussion of Major Conclusions

The Science Report synthesized a large body of scientific literature on the connectivity and mechanisms by which streams, wetlands, and open waters, singly or in aggregate, affect the physical, chemical, and biological integrity of downstream waters. The major conclusions reflect the strength of evidence available at that time in the peer-reviewed scientific literature for assessing the connectivity and downstream effects of water bodies identified in Chapter 1 of the Science Report.

The conclusions of the Science Report were corroborated by two independent peer reviews by scientists identified in the front matter of the Science Report and discussed in section I.B of this document.

The term connectivity is defined in the Science Report as the degree to which components of a watershed are joined and interact by transport mechanisms that function across multiple spatial and temporal scales. Connectivity is determined by the characteristics of both the physical landscape and the biota of the specific system. ORD's review found strong evidence supporting the central roles of the physical, chemical, and biological connectivity of streams, wetlands, and open waters—encompassing varying degrees of both connection and isolation—in maintaining the structure and function of downstream waters, including rivers, lakes, estuaries, and oceans. ORD's review also found strong evidence demonstrating the various mechanisms by which material and biological linkages from streams, wetlands, and open waters affect downstream waters, classified here into five functional categories (source, sink, refuge, lag, and transformation; discussed below), and modify the timing of transport and the quantity and quality of resources available to downstream ecosystems and communities. Thus, the literature available at the time of its publication in January 2015 provided a large body of evidence for assessing the types of connections and functions by which streams and wetlands produce the range of observed effects on the integrity of downstream waters.

ORD identified five categories of functions by which streams, wetlands, and open waters influence the timing, quantity, and quality of resources available to downstream waters:

- Source: the net export of materials, such as water and food resources;
- Sink: the net removal or storage of materials, such as sediment and contaminants;
- Refuge: the protection of materials, especially organisms;
- Transformation: the transformation of materials, especially nutrients and chemical contaminants, into different physical or chemical forms; and
- Lag: the delayed or regulated release of materials, such as stormwater.

These functions are not mutually exclusive; for example, the same stream or wetland can be both a source of organic matter and a sink for nitrogen. The presence or absence of these functions, which depend on the biota, hydrology, and environmental conditions in a watershed, can change over time; for example, the same wetland can attenuate runoff during storm events and provide groundwater recharge following storms. Further, some functions work in conjunction with others; a lag function can include transformation of materials prior to their delayed release. Finally, effects on downstream waters should

consider both actual function and potential function. A potential function represents the capacity of an ecosystem to perform that function under suitable conditions. For example, a wetland with high capacity for denitrification is a potential sink for nitrogen, a nutrient that becomes a contaminant when present in excessive concentrations. In the absence of nitrogen, this capacity represents the wetland's potential function. If nitrogen enters the wetland (*e.g.*, from fertilizer in runoff), it is removed from the water; this removal represents the wetland's actual function. Both potential and actual functions play critical roles in protecting and restoring downstream waters as environmental conditions change.

The evidence unequivocally demonstrates that the stream channels and riparian/floodplain wetlands or open waters that together form river networks are clearly connected to downstream waters in ways that profoundly influence downstream water integrity. The body of literature documenting connectivity and downstream effects was most abundant for perennial and intermittent streams, and for riparian/floodplain wetlands. Although less abundant, the evidence for connectivity and downstream effects of ephemeral streams was strong and compelling, particularly in context with the large body of evidence supporting the physical connectivity and cumulative effects of channelized flows that form and maintain stream networks.

As stated in Conclusion 3, the connectivity and effects of wetlands and open waters that lack visible surface connections to other water bodies were more difficult to address solely from evidence available in the peer-reviewed literature at the time of publication of the Science Report. The limited evidence available at the time showed that these systems have important hydrologic, water-quality, and habitat functions that can affect downstream waters where connections to them exist; the literature also provided limited examples of direct effects of non-floodplain wetland isolation on downstream water integrity. The available peer-reviewed literature, however, did not identify which types or classes of non-floodplain wetlands had or lacked the types of connections needed to convey the effects on downstream waters of functions, materials, or biota provided by those wetlands.

iii. Key Findings for the Science Report's Major Conclusions

This section summarizes key findings for each of the five major conclusions, above and in Chapter 6 of the Science Report. Citations have been omitted from the text to improve readability; please refer to individual chapters of the Science Report for supporting publications and additional information.

Conclusion 1, Streams: Key Findings

- Streams are hydrologically connected to downstream waters via channels that convey surface and subsurface water either year-round (*i.e.*, perennial flow), weekly to seasonally (*i.e.*, intermittent flow), or only in direct response to precipitation (*i.e.*, ephemeral flow). Streams are the dominant source of water in most rivers, and the majority of tributaries are perennial, intermittent, or ephemeral headwater streams. For example, headwater streams, which are the smallest channels where streamflows begin, are the cumulative source of approximately 60% of the total mean annual flow to all northeastern U.S. streams and rivers.
- In addition to downstream transport, headwaters convey water into local storage compartments such as ponds, shallow aquifers, or stream banks, and into regional and alluvial aquifers. These local storage compartments are important sources of water for maintaining baseflow in rivers.

Streamflow typically depends on the delayed (*i.e.*, lagged) release of shallow ground water from local storage, especially during dry periods and in areas with shallow groundwater tables and pervious subsurfaces. For example, in the southwestern United States, short-term shallow groundwater storage in alluvial floodplain aquifers, with gradual release into stream channels, is a major source of annual flow in rivers.

- Infrequent, high-magnitude events are especially important for transmitting materials from headwater streams in most river networks. For example, headwater streams, including ephemeral and intermittent streams, shape river channels by accumulating and gradually or episodically releasing stored materials such as sediment and large woody debris. These materials help structure stream and river channels by slowing the flow of water through channels and providing substrate and habitat for aquatic organisms.
- There is strong evidence that headwater streams function as nitrogen sources (via export) and sinks (via uptake and transformation) for river networks. For example, one study estimated that rapid nutrient cycling in small streams with no agricultural or urban impacts removed 20–40% of the nitrogen that otherwise would be delivered to downstream waters. Nutrients are necessary to support aquatic life, but excess nutrients lead to eutrophication and hypoxia, in which over-enrichment causes dissolved oxygen concentrations to fall below the level necessary to sustain most aquatic animal life in the stream and streambed. Thus, the influence of streams on nutrient loads can have significant repercussions for hypoxia in downstream waters.
- Headwaters provide habitat that is critical for completion of one or more life-cycle stages of many aquatic and semiaquatic species capable of moving throughout river networks. Evidence is strong that headwaters provide habitat for complex life-cycle completion; refuge from predators, competitors, parasites, or adverse physical conditions in rivers (*e.g.*, temperature or flow extremes, low dissolved oxygen, high sediment); and reservoirs of genetic- and species-level diversity. Use of headwater streams as habitat is especially critical for the many species that migrate between small streams and marine environments during their life cycles (*e.g.*, Pacific and Atlantic salmon, American eels, certain lamprey species). The presence of these species within river networks provides robust evidence of biological connections between headwaters and larger rivers; because these organisms also transport nutrients and other materials as they migrate, their presence also provides evidence of biologically mediated chemical connections. In prairie streams, many fishes swim upstream into tributaries to release eggs, which develop as they are transported downstream.
- Human alterations affect the frequency, duration, magnitude, timing, and rate of change of connections between headwater streams, including ephemeral and intermittent streams, and downstream waters. Human activities and built structures (*e.g.*, channelization, dams, groundwater withdrawals) can either enhance or fragment longitudinal connections between headwater streams and downstream waters, while also constraining lateral and vertical exchanges and tightly controlling the temporal dimension of connectivity. In many cases, research on human alterations has enhanced our understanding of the headwater stream-downstream water connections and their consequences. Recognition of these connections and effects has encouraged

the development of more sustainable practices and infrastructure to reestablish and manage connections, and ultimately to protect and restore the integrity of downstream waters.

Conclusion 2, Riparian/Floodplain Wetlands and Open Waters: Key Findings

- Riparian areas and floodplains connect upland and aquatic environments through both surface and subsurface hydrologic flowpaths. These areas are therefore uniquely situated in watersheds to receive and process waters that pass over densely vegetated areas and through subsurface zones before the waters reach streams and rivers. When pollutants reach a riparian or floodplain wetland, they can be sequestered in sediments, assimilated into wetland plants and animals, transformed into less harmful or mobile forms or compounds, or lost to the atmosphere. A wetland's potential for biogeochemical transformations (*e.g.*, denitrification) that can improve downstream water quality is influenced by local factors, including anoxic conditions and slow organic matter decomposition, shallow water tables, wetland plant communities, permeable soils, and complex topography.
- Riparian/floodplain wetlands can reduce flood peaks by storing and desynchronizing floodwaters. They can also maintain river baseflows by recharging alluvial aquifers. Many studies have documented the ability of riparian/floodplain wetlands to reduce flood pulses by storing excess water from streams and rivers. One review of wetland studies reported that riparian wetlands reduced or delayed floods in 23 of 28 studies. For example, peak discharges between upstream and downstream gaging stations on the Cache River in Arkansas were reduced 10–20% primarily due to floodplain water storage.
- Riparian areas and floodplains store large amounts of sediment and organic matter from upstream and from upland areas. For example, riparian areas have been shown to remove 80–90% of sediments leaving agricultural fields in North Carolina.
- Ecosystem function within a river system is driven in part by biological connectivity that links diverse biological communities with the river system. Movements of organisms that connect aquatic habitats and their populations, even across different watersheds, are important for the survival of individuals, populations, and species, and for the functioning of the river ecosystem. For example, lateral expansion and contraction of the river in its floodplain result in an exchange of matter and organisms, including fish populations that are adapted to use floodplain habitats for feeding and spawning during high water. Wetland and aquatic plants in floodplains can become important seed sources for the river network, especially if catastrophic flooding scours vegetation and seed banks in other parts of the channel. Many invertebrates exploit temporary hydrologic connections between rivers and floodplain wetland habitats, moving into these wetlands to feed, reproduce, or avoid harsh environmental conditions and then returning to the river network. Amphibians and aquatic reptiles commonly use both streams and riparian/floodplain wetlands to hunt, forage, overwinter, rest, or hide from predators. Birds can spatially integrate the watershed landscape through biological connectivity.

Conclusion 3, Non-floodplain Wetlands and Open Waters: Key Findings

- Water storage by wetlands well outside of riparian or floodplain areas can affect streamflow. Hydrologic models of prairie potholes in the Starkweather Coulee subbasin (North Dakota) that drains to Devils Lake indicate that increasing the volume of prairie pothole storage across the subbasin by approximately 60% caused simulated total annual streamflow to decrease 50% during a series of dry years and 20% during wet years. Similar simulation studies of watersheds that feed the Red River of the North in North Dakota and Minnesota demonstrated qualitatively comparable results, suggesting that the ability of prairie potholes to modulate streamflow could be widespread across eastern portions of the prairie pothole region. This work also indicates that reducing water storage capacity of wetlands by connecting formerly isolated prairie potholes through ditching or drainage to the Devils Lake and Red River basins could increase stormflow and contribute to downstream flooding. In many agricultural areas already crisscrossed by extensive drainage systems, total streamflow and baseflow are increased by directly connecting prairie potholes to stream networks. The impacts of changing streamflow are numerous, including altered flow regime, stream geomorphology, habitat, and ecology. The presence or absence of an effect of prairie pothole water storage on streamflow depends on many factors, including patterns of precipitation, topography, and degree of human alteration. For example, in parts of the prairie pothole region with low precipitation, low stream density, and little human alteration, hydrologic connectivity between prairie potholes and streams or rivers is likely to be low.
- Non-floodplain wetlands act as sinks and transformers for various pollutants, especially nutrients, which at excess levels can adversely impact human and ecosystem health and pose a serious pollution problem in the United States. In one study, sewage wastewaters were applied to forested wetlands in Florida for 4.5 years; more than 95% of the phosphorus, nitrate, ammonium, and total nitrogen were removed by the wetlands during the study period, and 66–86% of the nitrate removed was attributed to the process of denitrification (chemical and biological processes that remove nitrogen from water). In another study, sizeable phosphorus retention occurred in marshes that comprised only 7% of the lower Lake Okeechobee basin area in Florida. A non-floodplain bog in Massachusetts was reported to sequester nearly 80% of nitrogen inputs from various sources, including atmospheric deposition, and prairie pothole wetlands in the upper Midwest were found to remove >80% of the nitrate load via denitrification. A large prairie marsh was found to remove 86% of nitrate, 78% of ammonium, and 20% of phosphate through assimilation and sedimentation, sorption, and other mechanisms. Together, these and other studies indicate that onsite nutrient removal by non-floodplain wetlands is substantial and geographically widespread. The effects of this removal on rivers were generally not reported in the literature as of the 2015 publication date of the Scientific Report.
- Non-floodplain wetlands provide unique and important habitats for many species, both common and rare. Some of these species require multiple types of waters to complete their full life cycles, including downstream waters. Abundant or highly mobile species play important roles in transferring energy and materials between non-floodplain wetlands and downstream waters.
- Biological connections are likely to occur between most non-floodplain wetlands and downstream waters through either direct or stepping stone movement of amphibians, invertebrates, reptiles, mammals, and seeds of aquatic plants, including colonization by invasive species. Many species in those groups that use both stream and wetland habitats are capable of

dispersal distances equal to or greater than distances between many wetlands and river networks. Migratory birds can be an important vector of long-distance dispersal of plants and invertebrates between non-floodplain wetlands and the river network, although their influence has not been quantified. Whether those connections are of sufficient magnitude to impact downstream waters will either require estimation of the magnitude of material fluxes or evidence that these movements of organisms are required for the survival and persistence of biota that contribute to the integrity of downstream waters.

- Spatial proximity is one important determinant of the magnitude, frequency and duration of connections between wetlands and streams that will ultimately influence the fluxes of water, materials and biota between wetlands and downstream waters. However, proximity alone is not sufficient to determine connectivity, due to local variation in factors such as slope and permeability.
- The cumulative influence of many individual wetlands within watersheds can strongly affect the spatial scale, magnitude, frequency, and duration of hydrologic, biological and chemical fluxes or transfers of water and materials to downstream waters. Because of their aggregated influence, any evaluation of changes to individual wetlands should be considered in the context of past and predicted changes (*e.g.*, from climate change) to other wetlands within the same watershed.
- Non-floodplain wetlands can be hydrologically connected directly to river networks through
 natural or constructed channels, nonchannelized surface flows, or subsurface flows, the latter of
 which can travel long distances to affect downstream waters. A wetland surrounded by uplands is
 defined as "geographically isolated." Our review found that, in some cases, wetland types such as
 vernal pools and coastal depressional wetlands are collectively—and incorrectly—referred to as
 geographically isolated. Technically, the term "geographically isolated" should be applied only to
 the particular wetlands within a type or class that are completely surrounded by uplands.
 Furthermore, "geographic isolation" should not be confused with functional isolation, because
 geographically isolated wetlands can still have hydrologic, chemical, and biological connections
 to downstream waters.
- Non-floodplain wetlands occur along a gradient of hydrologic connectivity-isolation with respect to river networks, lakes, or marine/estuarine water bodies. This gradient includes, for example, wetlands that serve as origins for stream channels that have permanent surface-water connections to the river network; wetlands with outlets to stream channels that discharge to deep groundwater aquifers; geographically isolated wetlands that have local groundwater or occasional surface-water connections to downstream waters; and geographically isolated wetlands that have minimal hydrologic connection to other water bodies (but which could include surface and subsurface connections to other wetlands). This gradient can exist among wetlands of the same type or in the same geographic region.
- Caution should be used in interpreting connectivity for wetlands that have been designated as "geographically isolated" because (1) the term can be applied broadly to a heterogeneous group of wetlands, which can include wetlands that are not actually geographically isolated; (2) wetlands with permanent channels could be miscategorized as geographically isolated if the

designation is based on maps or imagery with inadequate spatial resolution, obscured views, etc.; and (3) wetland complexes could have connections to downstream waters through stream channels even if individual wetlands within the complex are geographically isolated. For example, a recent study examined hydrologic connectivity in a complex of wetlands on the Texas Coastal Plain. The wetlands in this complex have been considered to be a type of geographically isolated wetland; however, collectively they are connected both geographically and hydrologically to downstream waters in the area: During an almost 4-year study period, nearly 20% of the precipitation that fell on the wetland complex flowed out through an intermittent stream into downstream waters. Thus, wetland complexes could have connections to downstream waters through stream channels even when the individual wetland components are geographically isolated.

Conclusion 4, Degrees and Determinants of Connectivity: Key Findings

- The surface-water and groundwater flowpaths (hereafter, hydrologic flowpaths), along which water and materials are transported and transformed, determine variations in the degree of physical and chemical connectivity. These flowpaths are controlled primarily by variations in climate, geology, and terrain within and among watersheds and over time. Climate, geology, and terrain are reflected locally in factors such as rainfall and snowfall intensity, soil infiltration rates, and the direction of groundwater flows. These local factors interact with the landscape positions of streams and wetlands relative to downstream waters, and with functions (such as the removal or transformation of pollutants) performed by those streams and wetlands to determine connectivity gradients.
- Gradients of biological connectivity (*i.e.*, the active or passive movements of organisms through water or air and over land that connect populations) are determined primarily by species assemblages, and by features of the landscape (*e.g.*, climate, geology, terrain) that facilitate or impede the movement of organisms. The temporal and spatial scales at which biological pathways connect aquatic habitats depend on characteristics of both the landscape and species, and overland transport or movement can occur across watershed boundaries. Dispersal is essential for population persistence, maintenance of genetic diversity, and evolution of aquatic species. Consequently, dispersal strategies reflect aquatic species' responses and adaptations to biotic and abiotic environments, including spatial and temporal variation in resource availability and quality. Species' traits and behaviors encompass species-environment relationships over time, and provide an ecological and evolutionary context for evaluating biological connectivity in a particular watershed or group of watersheds.
- Pathways for chemical transport and transformation largely follow hydrologic flowpaths, but sometimes follow biological pathways (*e.g.*, nutrient transport from wetlands to coastal waters by migrating waterfowl, upstream transport of marine-derived nutrients by spawning of anadromous fish, uptake and removal of nutrients by emerging stream insects).
- Human activities alter naturally occurring gradients of physical, chemical, and biological connectivity by modifying the frequency, duration, magnitude, timing, and rate of change of fluxes, exchanges, and transformations. For example, connectivity can be reduced by dams,

levees, culverts, water withdrawals, and habitat destruction, and can be increased by effluent discharges, channelization, drainage ditches and tiles, and impervious surfaces.

Conclusion 5, Cumulative Effects: Key Findings

- Structurally and functionally, stream-channel networks and the watersheds they drain are fundamentally cumulative in how they are formed and maintained. Excess water from precipitation that is not evaporated, taken up by organisms, or stored in soils and geologic layers moves downgradient by gravity as overland flow or through channels carrying sediment, chemical constituents, and organisms. These channels concentrate surface-water flows and are more efficient than overland (*i.e.*, diffuse) flows in transporting water and materials, and are reinforced over time by recurrent flows.
- Connectivity between streams and rivers provides opportunities for materials, including nutrients and chemical contaminants, to be transformed chemically as they are transported downstream. Although highly efficient at the transport of water and other physical materials, streams are dynamic ecosystems with permeable beds and banks that interact with other ecosystems above and below the surface. The exchange of materials between surface and subsurface areas involves a series of complex physical, chemical, and biological alterations that occur as materials move through different parts of the river system. The amount and quality of such materials that eventually reach a river are determined by the aggregate effect of these sequential alterations that begin at the source waters, which can be at some distance from the river. The opportunity for transformation of material (e.g., biological uptake, assimilation, or beneficial transformation) in intervening stream reaches increases with distance to the river. Nutrient spiraling, the process by which nutrients entering headwater streams are transformed by various aquatic organisms and chemical reactions as they are transported downstream, is one example of an instream alteration that exhibits significant beneficial effects on downstream waters. Nutrients (in their inorganic form) that enter a headwater stream (e.g., via overland flow) are first removed from the water column by streambed algal and microbial populations. Fish or insects feeding on algae and microbes take up some of those nutrients, which are subsequently released back into the stream via excretion and decomposition (*i.e.*, in their organic form), and the cycle is repeated. In each phase of the cycling process-from dissolved inorganic nutrients in the water column, through microbial uptake, subsequent transformations through the food web, and back to dissolved nutrients in the water column-nutrients are subject to downstream transport. Stream and wetland capacities for nutrient cycling have important implications for the form and concentration of nutrients exported to downstream waters.
- Cumulative effects across a watershed must be considered when quantifying the frequency, duration, and magnitude of connectivity, to evaluate the downstream effects of streams and wetlands. For example, although the probability of a large-magnitude transfer of organisms from any given headwater stream in a given year might be low (*i.e.*, a low-frequency connection when each stream is considered individually), headwater streams are the most abundant type of stream in most watersheds. Thus, the overall probability of a large-magnitude transfer of organisms is higher when considered for all headwater streams in a watershed—that is, a high-frequency connection is present when headwaters are considered cumulatively at the watershed scale,

compared with probabilities of transport for streams individually. Similarly, a single pollutant discharge might be negligible but the cumulative effect of multiple discharges could degrade the integrity of downstream waters. Riparian open waters (*e.g.*, oxbow lakes), wetlands, and vegetated areas cumulatively can retain up to 90% of eroded clays, silts, and sands that otherwise would enter stream channels. The larger amounts of snowmelt and precipitation cumulatively held by many wetlands can reduce the potential for flooding at downstream locations. For example, wetlands in the prairie pothole region cumulatively stored about 11–20% of the precipitation in one watershed.

• The combination of diverse habitat types and abundant food resources cumulatively makes floodplains important foraging, hunting, and breeding sites for fish, aquatic life stages of amphibians, and aquatic invertebrates. The scale of these cumulative effects can be extensive; for example, coastal ibises travel up to 40 km to obtain food from freshwater floodplain wetlands for nesting chicks, which cannot tolerate salt levels in local food resources until they fledge.

iv. Science Report: Framework for Analysis

In support of the conclusions addressed above in this section, Chapter 2 of the Science Report essentially provides the framework for the analysis by describing the components of a river system and watershed; the types of physical, chemical, and biological connections that link those components; the factors that influence connectivity at various temporal and spatial scales; and methods for quantifying connectivity. In addition, Chapter 1 of the Science Report introduces the approach used for the analysis of the peer-reviewed literature.

Justice Kennedy's opinion in Rapanos v. United States, 547 U.S. 715 (2006) (Rapanos) established the framework for a significant nexus analysis that mirrors the framework through which scientists assess a river system – examining how the components of the system (e.g., wetlands), in the aggregate (in combination), in the region, contribute and connect to the river (significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters). In implementing the significant nexus standard under this rule for tributaries and adjacent wetlands, all tributaries and adjacent wetlands within the catchment area (*i.e.*, watershed) of the tributary of interest will be analyzed as part of the significant nexus analysis. The watershed scale is a scientifically valid scale for considering cumulative effects, including at the various levels of the watershed scale (e.g., small watersheds, the catchment level for an individual stream reach, the watershed that drains to the next named stream, etc.). Watershed position influences function (e.g., storage or groundwater recharge or surface outflow to other features), so it is defensible to allow for wetlands adjacent to the same tributary reach to be aggregated together with that tributary reach at the catchment level to assess the functions that work in concert to influence the traditional navigable water, the territorial seas, or the interstate water. Density is another important factor, but the effects of functions of remaining wetlands in formerly-dense wetland landscapes could become more important in light of cumulative losses.

To identify connections and effects of streams, wetlands, and other water bodies on downstream waters, the Science Report used two types of evidence from peer-reviewed, published literature: (1) direct evidence that demonstrated a connection or effect (e.g., observed transport of materials or movement of

organisms from streams or wetlands to downstream waters) and (2) indirect evidence that suggested a connection or effect (*e.g.*, presence of environmental factors known to influence connectivity, a gradient of impairment associated with cumulative loss of streams or wetlands). In some cases, an individual line of evidence demonstrated connections along the entire river network (*e.g.*, from headwaters to large rivers). In most cases, multiple sources of evidence were gathered and conclusions drawn via logical inference—for example, when one body of evidence shows that headwater streams are connected to downstream segments, another body of evidence shows those downstream segments are linked to other segments farther downstream, and so on. This approach, which borrows from weight-of-evidence approaches in causal analysis is an effective way to synthesize the diversity of evidence needed to address questions at larger spatial and longer temporal scales than are often considered in individual scientific studies. Science Report at 1-14, 1-16 (citing Suter *et al.* 2002; Suter and Cormier 2011).

A river is the time-integrated result of all waters contributing to it, and connectivity is the property that spatially integrates the individual components of the watershed. In discussions of connectivity, the watershed scale is the appropriate context for interpreting technical evidence about individual watershed components. Science Report at 2-1 (citing Newbold *et al.* 1982b; Stanford and Ward 1993; Bunn and Arthington 2002; Power and Dietrich 2002; Benda *et al.* 2004b; Naiman *et al.* 2005; Nadeau and Rains 2007; Rodriguez-Iturbe *et al.* 2009). Such interpretation requires that freshwater resources be viewed within a landscape—or systems—context. *Id.* (citing Baron *et al.* 2002). Addressing the questions asked in the Science Report, therefore, requires an integrated systems perspective that considers both the components contributing to the river and the connections between those components and the river.

Components of the River System

In the Science Report, the term river refers to a relatively large volume of flowing water within a visible channel, including subsurface water moving in the same direction as the surface water and lateral flows exchanged with associated floodplain and riparian areas. Id. at 2-2 (Naiman and Bilby 1998). Channels are natural or constructed passageways or depressions of perceptible linear extent that convey water and associated materials downgradient. They are defined by the presence of continuous bed and bank structures, or uninterrupted (but permeable) bottom and lateral boundaries. Although bed and bank structures might in places appear to be disrupted (e.g., bedrock outcrops, braided channels, flow-through wetlands), the continuation of the bed and banks downgradient from such disruptions is evidence of the surface connection with the channel that is upgradient of the perceived disruption. Such disruptions are associated with changes in the gradient and in the material over and through which the water flows. If a disruption in the bed and bank structure prevented connection, the area downgradient would lack a bed and banks, be colonized with terrestrial vegetation, and be indiscernible from the nearby land. The concentrated longitudinal movement of water and sediment through these channels lowers local elevation, prevents soil development, selectively transports and stores sediment, and hampers the colonization and persistence of terrestrial vegetation. Streams are defined in a similar manner as rivers: a relatively small volume of flowing water within a visible channel, including subsurface water moving in the same direction as the surface water and lateral flows exchanged with associated floodplain and riparian areas. Id. (citing Naiman and Bilby 1998).

A river network is a hierarchical, interconnected population of channels that drains surface and subsurface water from a watershed to a river and includes the river itself. Watershed boundaries traditionally are defined topographically, such as by ridges. These channels can convey water year-round, weekly to seasonally, or only in direct response to rainfall and snowmelt. *Id.* (citing Frissell *et al.* 1986; Benda *et al.* 2004b). The smallest of these channels, where streamflows begin, are considered headwater streams. Headwater streams are first- to third-order streams, where stream order is a classification system based on the position of the stream in the river network. *Id.* (citing Strahler 1957; Vannote *et al.* 1980; Meyer and Wallace 2001; Gomi *et al.* 2002; Fritz *et al.* 2006; Nadeau and Rains 2007). The point at which stream or river channels intersect within a river network is called a confluence. The confluence of two streams with the same order results in an increase of stream order (*i.e.*, two first-order streams join to form a second-order stream, two second-order streams join to form a third-order stream, and so on); when streams of different order join, the order of the larger stream is retained.

Terminal and lateral source streams³ typically originate at channel heads, which occur where surface-water runoff is sufficient to erode a definable channel. *Id.* at 2-3 (citing Dietrich and Dunne 1993). The channel head denotes the upstream extent of a stream's continuous bed and banks structure. Channel heads are relatively dynamic zones in river networks, as their position can advance upslope by overland or subsurface flow-driven erosion, or retreat downslope by colluvial infilling. Source streams also can originate at seeps or springs and associated wetlands.

When two streams join at a confluence, the smaller stream (*i.e.*, that with the smaller drainage area or lower mean annual discharge) is called a tributary of the larger stream, which is referred to as the mainstem. A basic way of classifying tributary contributions to a mainstem is the symmetry ratio, which describes the size of a tributary relative to the mainstem at their confluence, in terms of their respective discharges, drainage areas, or channel widths. *Id.* at 2-4 (citing Roy and Woldenberg 1986; Rhoads 1987; Benda 2008).

Surface-water hydrologic connectivity within river network channels occurs, in part, through the unidirectional movement of water from channels at higher elevations to ones at lower elevations—that is, hydrologic connectivity exists because water flows downhill. In essence, the river network represents the aboveground flow route and associated subsurface-water interactions, transporting water, energy, and materials from the surrounding watershed to downstream rivers, lakes, estuaries, and oceans (The River Continuum Concept). *Id.* (citing (Vannote *et al.* 1980).

Streamflow and the quantity and character of sediment—interacting with watershed geology, terrain, soils and vegetation—shape morphological changes in the stream channel that occur from river network headwaters to lower rivers. *Id.* (citing Montgomery 1999; Church 2002). Headwater streams are typically erosion zones in which sediment from the base of adjoining hillslopes moves directly into stream channels and is transported downstream. As stream channels increase in size and decrease in slope, a mixture of erosion and deposition processes usually is at work. At some point in the lower

³ Mock (1971) presented a classification of the streams comprising stream or river networks. He designated firstorder streams that intersect other first-order streams as sources. We refer to these as terminal source streams. Mock defined first-order streams that flow into higher order streams as tributary sources, and we refer to this class of streams as lateral source streams.

portions of river networks, sediment deposition becomes the dominant process and floodplains form. Floodplains are level areas bordering stream or river channels that are formed by sediment deposition from those channels under present climatic conditions. These natural geomorphic features are inundated during moderate to high water events. *Id.* (citing Leopold 1994; Osterkamp 2008). Floodplain and associated river channel forms (*e.g.*, meandering, braided, anastomosing) are determined by interacting fluvial factors, including sediment size and supply, channel gradient, and streamflow. *Id.* (citing Church 2002; Church 2006).

Both riparian areas and floodplains are important components of river systems. Riparian areas are transition zones between terrestrial and aquatic ecosystems that are distinguished by gradients in biophysical conditions, ecological processes, and biota. They are areas through which surface and subsurface hydrology connect water bodies with their adjoining uplands, and they include those portions of terrestrial ecosystems that significantly influence exchanges of energy and matter with aquatic ecosystems. *Id.* (citing National Research Council 2002). Riparian areas often have high biodiversity. *Id.* (citing Naiman *et al.* 2005). They occur near lakes and estuarine-marine shorelines and along river networks, where their width can vary from narrow bands along headwater streams to broad zones that encompass the floodplains of large rivers.

Floodplains are also considered riparian areas, but not all riparian areas have floodplains. All rivers and streams within river networks have riparian areas, but small streams in constrained valleys are less likely to have floodplains than larger streams and rivers in unconstrained valleys. The "100-year floodplain" is the area with a one percent annual chance of flooding. *Id.* at 2-5; USGS c. The 100-year floodplain can but need not coincide with the geomorphic floodplain.

Wetlands are transitional areas between terrestrial and aquatic ecosystems. Wetlands include areas such as swamps, bogs, fens, marshes, ponds, and pools. Science Report at 2-6 (citing Mitsch *et al.* 2009).

Many classification systems have been developed for wetlands. *Id.* (citing Mitsch and Gosselink 2007). These classifications can focus on vegetation, hydrology, hydrogeomorphic characteristics, or other factors. *Id.* (citing Cowardin *et al.* 1979; Brinson 1993; Tiner 2003a; Comer *et al.* 2005). Because the Science Report focuses on downstream connectivity, it considered two landscape settings in which wetlands occur based on directionality of hydrologic flows. Directionality of flow also is included as a component of hydrodynamic setting in the hydrogeomorphic approach and as an element of water flowpath in an enhancement of National Wetlands Inventory data (the National Wetlands Inventory is a mapping dataset of the U.S. Fish and Wildlife Service regarding the extent and types of wetlands and deepwater habitats across the country) that provides descriptors for landscape position, landform, water flow path, and waterbody type (LLLW). *Id.* (citing Brinson 1993; Smith *et al.* 1995, Tiner 2011); *see also* U.S. FWS 2010. This emphasis on directionality of flow is necessary because hydrologic connectivity plays a dominant role in determining the types of effects wetlands have on downstream waters.

A non-floodplain wetland setting is a landscape setting where a potential exists for unidirectional, lateral hydrologic flows from wetlands to the river network through surface water or ground water. Such a setting would include upgradient areas such as hillslopes or upland areas outside of the floodplain. A floodplain is the level area bordering a stream or river channel that was built by sediment deposition from the stream or river under present climatic conditions and is inundated during moderate to high flow

events. Floodplains formed under historic or prehistoric climatic conditions can be abandoned by rivers and form terraces. Any wetland setting where water could only flow from the wetland toward a river network would be considered a non-floodplain setting, regardless of the magnitude and duration of flows and of travel times. The Science Report refers to wetlands that occur in these settings as non-floodplain wetlands.

A riparian/floodplain wetland setting is a landscape setting (*e.g.*, floodplains, most riparian areas, lake and estuarine fringes) that is subject to bidirectional, lateral hydrologic flows. Wetlands in riparian/floodplain settings can have some of the same types of hydrologic connections as those in non-floodplain settings. In addition, wetlands in these settings also have bidirectional flows. For example, wetlands within a riparian area are connected to the river network through lateral movement of water between the channel and riparian area (*e.g.*, through overbank flooding, hyporheic flow). Given the Science Report's interest in addressing the effects of wetlands on downstream waters, it focused in particular on the subset of these wetlands that occur in riparian areas with and without floodplains (collectively referred to hereafter as riparian/floodplain wetlands); the Science Report generally does not address wetlands at lake and estuarine fringes. Riparian wetlands are portions of riparian areas that meet the Cowardin *et al.* (1979) three-attribute wetland criteria (*i.e.*, having wetland hydrology, hydrophytic vegetation, or hydric soils); floodplain wetlands are portions of the floodplain that meet these same criteria. *Id.* at 2-7. Given that even infrequent flooding can have profound effects on wetland development and function, the Science Report considers such a wetland to be in a riparian/floodplain setting.

Note that the scientific definition of "wetland" used in the Science Report is not the same as the longstanding Clean Water Act regulatory definition of "wetlands," retained in the final rule at paragraph (c)(1)⁴. Only aquatic resources that meet the regulatory definition of "wetlands" at paragraph (c)(1) are considered to be wetlands for Clean Water Act purposes under the final rule. The agencies are not changing their longstanding regulation that requires that an aquatic resource must meet all three parameters under normal circumstances to be considered a wetland in the regulatory sense. Cowardin wetlands do not need to have all three parameters.⁵ FGDC 2013. Conclusions in the Science Report apply to the Cowardin wetlands, and the Cowardin definition of wetlands encompasses a larger universe of wetlands than the regulatory definition. Therefore, the Science Report conclusions regarding Cowardin wetlands apply to the wetlands that meet the regulatory definition because those wetlands are a subset of the Cowardin wetlands. All wetlands that meet the regulatory definition of "wetlands" must meet all three parameters under normal circumstances, it is even more likely that they provide the many functions described in the Science Report due to the conditions in the waters that make them wetlands – that is, their hydric soils (inundated or saturated soils), hydrophytic vegetation (plants that thrive in wet

⁴ The final rule at paragraph (c)(1) states, "*Wetlands* means those areas that are inundated or saturated by surface or ground water at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include swamps, marshes, bogs, and similar areas." *See also* 33 CFR 328.3(b) (1987); 33 CFR 328.3(c)(16) (2021); 40 CFR 230.3(t) (1988); 40 CFR 120.2(3)(xvi) (2021).

⁵ The Federal Geographic Data Committee (FGDC) (2013, p. 7) noted that "three (3) indicators – hydrophytic vegetation, undrained hydric soil, and wetland hydrology; two (2) indicators—hydrophytic vegetation and wetland hydrology or undrained hydric soil and wetland hydrology; and one (1) indicator—wetland hydrology, respectively, would be used to make the identification [that a feature meets the Cowardin "wetland" definition], based on the features available at the particular site."

conditions), and wetland hydrology (inundation or saturation at the surface at some time during the growing season). In addition, many of the Cowardin wetland types are in fact open waters, as the Cowardin definition encompasses open waters like ponds, and the Science Report utilizes many references that include such open waters when discussing floodplain and non-floodplain wetlands. Thus, open waters also provide the many functions described in the Science Report and throughout this document. The Science Report acknowledges that its conclusions apply to open waters as well as wetlands, stating, "although the literature review did not address other non-floodplain water bodies to the same extent as wetlands, our overall conclusions also apply to these water bodies (*e.g.*, ponds and lakes that lack surface water inlets) because the same principles govern hydrologic connectivity between these water bodies and downstream waters." *Id.* at 4-41. Wetlands and open waters are only jurisdictional when they meet the definition of "waters of the United States."

A major consequence of the two different landscape settings (non-floodplain versus riparian/floodplain) is that waterborne materials can be transported only from the wetland to the river network for a non-floodplain wetland, whereas waterborne materials can be transported from the wetland to the river network and from the river network to the wetland for a riparian/floodplain wetland. In the latter case, there is a mutual, interacting effect on the structure and function of both the wetland and river network. In contrast, a non-floodplain wetland can affect a river through the transport of waterborne material, but the opposite is not true. Note that the Science Report limits use of riparian/floodplain and non-floodplain landscape settings to describe the direction of hydrologic flow; the terms cannot be used to describe directionality of geochemical or biological flows. For example, mobile organisms can move from a stream to a non-floodplain wetland. *Id.* at 2-8 (citing, *e.g.*, Subalusky *et al.* 2009a; Subalusky *et al.* 2009b).

Both non-floodplain and riparian/floodplain wetlands can include geographically isolated wetlands, or wetlands completely surrounded by uplands. *Id.* (citing Tiner 2003b). These wetlands have no apparent surface-water outlets, but can hydrologically connect to downstream waters through spillage or groundwater. The Science Report defines an upland as any area not meeting the Cowardin *et al.* (1979) three-attribute wetland criteria, meaning that uplands can occur in both terrestrial and riparian areas.⁶ *Id.* Thus, a wetland that is located on a floodplain but is surrounded by upland would be considered a geographically isolated, riparian/floodplain wetland that is subject to periodic inundation from the river network. Although the term "geographically isolated" could be misconstrued as implying functional isolation (*see* Mushet *et al.* 2015), the term has been defined in the peer-reviewed literature to refer specifically to wetlands surrounded by uplands. Furthermore, the literature explicitly notes that geographically isolated wetlands is essential because hydrologic gradients of connectivity in these systems support a wide range of functions that can benefit downstream waters.

River System Hydrology

⁶ Note that this definition of upland is the one that is used in the Science Report. The agencies are not promulgating a definition of upland in the rule. In addition, while the agencies consistently use the phrase "dry land" in the regulatory text to provide clarity to the public, the preamble and this Technical Support Document use the phrases "dry land" and "upland" interchangeably.

River system hydrology is controlled by hierarchical factors that result in a broad continuum of belowground and aboveground hydrologic flowpaths connecting river basins and river networks. Id. (citing Winter 2001; Wolock et al. 2004; Devito et al. 2005; Poole et al. 2006; Wagener et al. 2007; Poole 2010; Bencala et al. 2011; Jencso and McGlynn 2011). At the broadest scale, regional climate interacts with river-basin terrain and geology to shape inherent hydrologic infrastructure that bounds the nature of basin hydrologic flowpaths. Different climate-basin combinations form identifiable hydrologic landscape units with distinct hydrologic characteristics. Id. at 2-8 to 2-9 (Winter 2001; Wigington et al. 2013). Buttle (2006) posited three first-order controls of watershed streamflow generated under specific hydroclimatic conditions: (1) the ability of different landscape elements to generate runoff by surface or subsurface lateral flow of water; (2) the degree of hydrologic connectivity among landscapes by which surface and subsurface runoff can reach river networks; and (3) the capacity of the river network itself to convey runoff downstream to the river-basin outlet. Id. at 2-9. River and stream waters are influenced by not only basin-scale or larger ground-water systems, but also local-scale, vertical and lateral hydrologic exchanges between water in channels and sediments beneath and contiguous with river network channels. Id. at 2-9 (citing Ward 1989; Woessner 2000; Malard et al. 2002; Bencala 2011). The magnitude and importance of river-system hydrologic flowpaths at all spatial scales can radically change over time at hourly to yearly temporal scales. Id. (citing Junk et al. 1989; Ward 1989; Malard et al. 1999; Poole et al. 2006).

Because interactions between groundwater and surface waters are essential processes in rivers, knowledge of basic groundwater hydrology is necessary to understand the interaction between surface and subsurface water and their relationship to connectivity and effects within river systems. Subsurface water occurs in two principal zones: the unsaturated zone and the saturated zone. *Id.* (citing Winter *et al.* 1998). In the unsaturated zone, the spaces between soil, gravel, and other particles contain both air and water. In the saturated zone, these spaces are completely filled with water. Ground water refers to any water that occurs and flows (saturated groundwater flow) in the saturated zone beneath a watershed surface. *Id.* (citing Winter *et al.* 1998). Rapid flow (interflow) of water can occur through large pore spaces in the unsaturated zone. *Id.* (citing Beven and Germann 1982).

Other hydrologic flowpaths are also significant in determining the characteristics of river systems. The most obvious is the downstream water movement within stream or river channels, or openchannel flow. River water in stream and river channels can reach riparian areas and floodplains via overbank flow, which occurs when floodwaters flow over stream and river channels. *Id.* at 2-12 (citing Mertes 1997). Overland flow is the portion of streamflow derived from net precipitation that flows over the land surface to the nearest stream channel with no infiltration. *Id.* (citing Hewlett 1982). Overland flow can be generated by several mechanisms. Infiltration-excess overland flow occurs when the rainfall rates exceed the infiltration rates of land surfaces. *Id.* (citing Horton 1945). Saturation-excess overland flow occurs when precipitation inputs cause water tables to rise to land surfaces so that precipitation inputs to the land surfaces cannot infiltrate and flow overland. *Id.* (citing Dunne and Black 1970). Return flow occurs when water infiltrates, percolates through the unsaturated zones, enters saturated zones, and then returns to and flows over watershed surfaces, commonly at hillslope-floodplain transitions. *Id.* (citing Dunne and Black 1970).

Alluvium consists of deposits of clay, silt, sand, gravel, or other particulate materials that running water has deposited in a streambed, on a floodplain, on a delta, or in a fan at the base of a mountain.

These deposits occur near active river systems but also can be found in buried river valleys—the remnants of relict river systems. *Id.* (citing Lloyd and Lyke 1995). The Science Report was concerned primarily with alluvium deposited along active river networks. Commonly, alluvium is highly permeable, creating an environment conducive to groundwater flow. Alluvial groundwater (typically a mixture of river water and local, intermediate, and regional groundwater) moves through the alluvium. Together, the alluvium and alluvial ground water comprise alluvial aquifers. Alluvial aquifers are closely associated with floodplains and have high levels of hyporheic exchange. *Id.* (citing Stanford and Ward 1993; Amoros and Bornette 2002; Poole *et al.* 2006). Hyporheic exchange occurs when water moves from stream or river channels into alluvial deposits and then returns to the channels. *Id.* at 2-12, 4-8 (citing Sjodin *et al.* 2001; Bencala 2005; Gooseff *et al.* 2008; Leibowitz *et al.* 2008; Bencala 2011). Hyporheic exchange allows for the mixing of surface water and groundwater. It occurs during both high- and low-flow periods, and typically has relatively horizontal flowpaths at scales of meters to tens of meters and vertical flowpaths with depths ranging from centimeters to tens of meters. Science Report at 2-12 (citing Stanford and Ward 1988; Woessner 2000 and references therein; Bencala 2005).

Riparian areas and floodplains can have a diverse array of hydrologic inputs and outputs, which, in turn influence riparian/floodplain wetlands. Riparian areas and floodplains receive water from precipitation; overland flow from upland areas; local, intermediate, regional ground water; and hyporheic flows. *Id.* at 4-14 (National Research Council 2002; Richardson *et al.* 2005; Vidon *et al.* 2010). Water flowing over the land surface in many situations can infiltrate soils in riparian areas. *Id.* If low permeability subsoils or impervious clay layers are present, water contact with the plant root zone is increased and the water is subject to ecological functions (*e.g.*, sink or transformation) such as denitrification before it reaches the stream channel. *Id.* (citing National Research Council 2002; Naiman *et al.* 2005; Vidon *et al.* 2010).

The relative importance of the continuum of hydrologic flowpaths among river systems varies, creating streams and rivers with different flow duration (or hydrologic permanence) classes. Perennial streams or stream reaches typically flow year-round. They are maintained by local or regional groundwater discharge or streamflow from higher in the stream or river network. Intermittent streams or stream reaches flow continuously, but only at certain times of the year (e.g., during certain seasons such as spring snowmelt); drying occurs when the water table falls below the channel bed elevation. Ephemeral streams or stream reaches flow briefly (typically hours to days) during and immediately following precipitation; these channels are above the water table at all times. Streams in these flow duration classes often transition longitudinally, from ephemeral to intermittent to perennial, as drainage area increases and elevation decreases along river networks. Many headwater streams, however, originate from permanent springs and flow directly into intermittent downstream reaches. At low flows, intermittent streams can contain dry segments alternating with flowing segments. Transitions between flow duration classes can coincide with confluences or with geomorphic discontinuities within the network. Id. at 2-14 (citing May and Lee 2004; Hunter et al. 2005). Variation of streamflow within river systems occurs in response to hydrologic events resulting from rainfall or snowmelt. Stormflow is streamflow that occurs in direct response to rainfall or snowmelt, which might stem from multiple groundwater and surface-water sources. Id. (citing Dunne and Leopold 1978). Baseflow is streamflow originating from groundwater discharge or seepage (locally or from higher in the river network), which sustains water flow through the channel between hydrologic events. Perennial streams have baseflow year-round; intermittent streams have baseflow seasonally; ephemeral streams have no baseflow. All three stream types convey stormflow.

Thus, perennial streams are more common in areas receiving high precipitation, whereas intermittent and ephemeral streams are more common in the more arid portions of the United States. *Id.* (citing NHD 2008). The distribution of headwater streams (perennial, intermittent, or ephemeral) as a proportion of total stream length is similar across geographic regions and climates.

Similar to streams, the occurrence and persistence of riparian/floodplain wetland and nonfloodplain wetland hydrologic connections with river networks, via surface water (both channelized and nonchannelized) or groundwater, can be continuous, seasonal, or ephemeral, depending on the overall hydrologic conditions in the watershed. For example, a non-floodplain wetland might have a direct groundwater connection with a river network during wet conditions but an indirect regional ground-water connection (via groundwater recharge) under dry conditions. Another non-floodplain wetland might have direct surface water connections during wet periods (*e.g.*, fill and spill) and groundwater connections during drier periods. *See, e.g.*, Rains *et al.* 2008; Leibowitz and Vining 2003; Leibowitz *et al.* 2016. Geographically isolated wetlands can be hydrologically connected to the river network via nonchannelized surface flow (*e.g.*, swales or overland flow) or groundwater.

The portions of river networks with flowing water expand and contract longitudinally (in an upstream-downstream direction) and laterally (in a stream channel-floodplain direction) in response to seasonal environmental conditions and precipitation events. Id. at 2-18 (citing Hewlett and Hibbert 1967; Gregory and Walling 1968; Dunne and Black 1970; Day 1978; Junk et al. 1989; Hunter et al. 2005; Wigington et al. 2005; Rains et al. 2006; Rains et al. 2008). The longitudinal expansion of channels with flowing water in response to major precipitation events represents a transient increase in the extent of headwater streams. Intermittent and perennial streams flow during wet seasons, whereas ephemeral streams flow only in response to rainfall or snowmelt. During dry periods, flowing portions of river networks are limited to perennial streams; these perennial portions of the river network can be discontinuous or interspersed with intermittently flowing stream reaches but may be flowing in the hyporheic zone for thousands of meters before emerging. Id. (citing Stanley et al. 1997; Hunter et al. 2005; Larned et al. 2010). Thus, stream reaches can be perennial even if the entire stream channel is not. As discussed previously, perennial streams typically flow year-round, intermittent streams flow continuously only at certain times of the year (e.g., when they receive water from a spring, groundwater source, or surface snow such as melting snow), and ephemeral streams flow briefly in direct response to precipitation. In perennial streams, baseflow (the portion of flow contributed by groundwater) is typically present year-round. The definition of "perennial" allows for infrequent periods of severe drought to cause some perennial streams to not have flow year-round. Leopold 1994. Some studies have noted that perennial flow is present greater than 90% of the time, except in periods of severe drought, or greater than 80% of the time, and these definitions are consistent with the one used in the Science Report. Hedman and Osterkamp 1982; Hewlett 1982.

The dominant sources of water to a stream can shift during river network expansion and contraction. *Id.* (citing Malard *et al.* 1999; McGlynn and McDonnell 2003; McGlynn *et al.* 2004; Malard *et al.* 2006). Rainfall and snowmelt cause a river network to expand in two ways. First, local aquifers expand and water moves into dry channels, which increases the total length of the wet channel; the resulting intermittent streams will contain water during the entire wet season. *Id.* (citing Winter *et al.* 1998). Second, stormflow can cause water to enter ephemeral and intermittent streams. The larger the rainfall or snowmelt event, the greater the number of ephemeral streams and total length of flowing

channels that occur within the river network. Ephemeral flows cease within days after rainfall or snowmelt ends, causing the length of wet channels to decrease and river networks to contract. The flowing portion of river networks further shrinks as the spatial extent of aquifers with ground water in contact with streams contract and intermittent streams dry. In many river systems across the United States, stormflow comprises a major portion of annual streamflow. *Id.* (citing Hewlett *et al.* 1977; Miller *et al.* 1988; Turton *et al.* 1992; Goodrich *et al.* 1997; Vivoni *et al.* 2006). In these systems, intermittent and ephemeral streams are major sources of river water. When rainfall or snowmelt induces stormflow in headwater streams or other portions of the river network, water flows downgradient through the network to its lower reaches. As water moves downstream through a river network, the hydrograph for a typical event broadens with a lower peak. This broadening of the hydrograph shape results from transient storage of water in river network channels and nearby alluvial aquifers. *Id.* (citing Fernald *et al.* 2001).

During very large hydrologic events, aggregate flows from headwaters and other tributary streams can result in overbank flooding in river reaches with floodplains; this occurrence represents lateral expansion of the river network. *Id.* (citing Mertes 1997). Water from overbank flows can recharge alluvial aquifers, supply water to floodplain wetlands, surficially connect floodplain wetlands to rivers, and shape the geomorphic features of the floodplain. *Id.* at 2-18 to 2-19 (citing Wolman and Miller 1960; Hammersmark *et al.* 2008). Bidirectional exchanges of water between ground water and river networks, including hyporheic flow, can occur under a wide range of streamflows, from flood flows to low flows. *Id.* at 2-19 to 2-20 (citing National Research Council 2002; Naiman *et al.* 2005; Vivoni *et al.* 2006).

Many studies have documented the fact that riparian/floodplain wetlands can attenuate flood pulses of streams and rivers by storing excess water from streams and rivers. Bullock and Acreman (2003) reviewed wetland studies and reported that wetlands reduced or delayed floods in 23 of 28 studies. *Id.* at 2-21. For example, Walton *et al.* (1996) found that peak discharges between upstream and downstream gaging stations on the Cache River in Arkansas were reduced 10–20% primarily due to floodplain water storage. *Id.* Locations within floodplains and riparian areas with higher elevations likely provide flood storage less frequently than lower elevation areas.

The interactions of high flows with floodplains and associated alluvial aquifers of river networks are important determinants of hydrologic and biogeochemical conditions of rivers. *Id.* at 2-21 (citing Ward 1989; Stanford and Ward 1993; Boulton *et al.* 1998; Burkart *et al.* 1999; Malard *et al.* 1999; Amoros and Bornette 2002; Malard *et al.* 2006; Poole 2010). Bencala (1993; 2011) noted that streams and rivers are not pipes; they interact with the alluvium and geologic materials adjoining and under channels. *Id.* In streams or river reaches constrained by topography, significant floodplain and near-channel alluvial aquifer interactions are limited. In reaches with floodplains, however, stormflow commonly supplies water to alluvial aquifers during high-flow periods through the process of bank storage. *Id.* at 2-22 (citing Whiting and Pomeranets 1997; Winter *et al.* 1998; Chen and Chen 2003). As streamflow decreases after hydrologic events, the water stored in these alluvial aquifers can serve as another source of baseflow in rivers.

In summary, the extent of wetted channels is dynamic because interactions between surface water in the channel and alluvial ground water, via hyporheic exchange, determine open-channel flow. The flowing portion of river networks expands and contracts in two primary dimensions: (1) longitudinally, as intermittent and ephemeral streams wet up and dry; and (2) laterally, as floodplains and associated alluvial aquifers gain (via overbank flooding, bank storage, and hyporheic exchange) and lose (via draining of alluvial aquifers and evapotranspiration) water. Vertical ground-water exchanges between streams and rivers and underlying alluvium are also key connections, and variations in these vertical exchanges contribute to the expansion and contraction of the portions of river networks with open-channel flow. Numerous studies have documented expansion and contraction of river systems; the temporal and spatial pattern of this expansion and contraction varies in response to many factors, including interannual and long-term dry cycles, climatic conditions, and watershed characteristics. *Id.* (citing Gregory and Walling 1968; Cayan and Peterson 1989; Fleming *et al.* 2007).

Influence of Streams and Wetlands on Downstream Waters

The structure and function of rivers are highly dependent on the constituent materials stored in and transported through them. Most of these materials, broadly defined here as any physical, chemical, or biological entity, including water, heat energy, sediment, wood, organic matter, nutrients, chemical contaminants, and organisms, originate outside of the river; they originate from either the upstream river network or other components of the river system, and then are transported to the river by water movement or other mechanisms. Thus, the fundamental way in which streams and wetlands affect river structure and function is by altering fluxes of materials to the river. This alteration of material fluxes depends on two key factors: (1) functions within streams and wetlands that affect material fluxes, and (2) connectivity (or isolation) between streams and wetlands and rivers that allows (or prevents) transport of materials between the systems. *Id.*

Streams and wetlands affect the amounts and types of materials that are or are not delivered to downstream waters, ultimately contributing to the structure and function of those waters. Leibowitz *et al.* (2008) identified three functions, or general mechanisms of action, by which streams and wetlands influence material fluxes into downstream waters: source, sink, and refuge. *Id.* at 2-22 to 2-23. The Science Report expanded on this framework to include two additional functions: lag and transformation. These five functions provide a framework for understanding how physical, chemical, and biological connections between streams and wetlands and downstream waters influence river systems.

These five functions are neither static nor mutually exclusive, and often the distinctions between them are not sharp. A stream or wetland can provide different functions at the same time. These functions can vary with the material considered (*e.g.*, acting as a source of organic matter and a sink for nitrogen) and can change over time (*e.g.*, acting as a water sink when evapotranspiration is high and a water source when evapotranspiration is low). The magnitude of a given function also is likely to vary temporally; for example, streams generally are greater sources of organic matter and contaminants during high flows. *Id.* at 2-24.

Leibowitz *et al.* (2008) explicitly focused on functions that benefit downstream waters, but these functions also can have negative effects—for example, when streams and wetlands serve as sources of chemical contamination. *Id.* In fact, benefits need not be linear with respect to concentration; a beneficial material could be harmful at higher concentrations due to nonlinear and threshold effects. For example, nitrogen can be beneficial at lower concentrations but can reduce water quality at higher concentrations. Although the Science Report focused primarily on the effects of streams and wetlands on downstream waters, these same functions can describe effects of downstream waters on streams and wetlands (*e.g.*, downstream rivers can serve as sources of colonists for upstream tributaries). *Id.*
Because many of these functions depend on import of materials and energy into streams and wetlands, distinguishing between actual function and potential function is instructive. For example, a wetland with appropriate conditions (*e.g.*, a reducing environment and denitrifying bacteria) is a potential sink for nitrogen: If nitrogen applied to the land is imported into the wetland, the wetland can remove it by denitrification. The wetland will not serve this function, however, if nitrogen is not imported. Thus, even if a stream and wetland do not currently serve a function, it has the potential to provide that function under appropriate conditions (*e.g.*, when material imports or environmental conditions change). These functions can be instrumental in protecting those waters from future impacts. Ignoring potential function also can lead to the paradox that degraded streams and wetlands (*e.g.*, those receiving nonpoint-source nitrogen inputs) receive more protection than less impacted systems. *Id.* (citing Leibowitz *et al.* 2008).

Three factors influence the effect that material and energy fluxes from streams and wetlands have on downstream waters: (1) proportion of the material originating from (or reduced by) streams and wetlands relative to the importance of other system components, such as the river itself; (2) residence time of the material in the downstream water; and (3) relative importance of the material. *Id.* In many cases, the effects on downstream waters need to be considered in aggregate. For example, the contribution of material by a particular stream and wetland (*e.g.*, a specific ephemeral stream) might be small, but the aggregate contribution by an entire class of streams and wetlands (*e.g.*, all ephemeral streams in the river network) might be substantial. Similarly, the functions of a given non-floodplain wetland relative to nitrogen removal may be small, but the cumulative effects of nitrogen removal by the extant wetland on the landscape will be substantive. *See, e.g.*, Evenson *et al.* 2018; Evenson *et al.* 2021. Integrating contributions over time also might be necessary, taking into account the frequency, duration, and timing of material export and delivery. Considering the cumulative material fluxes that originate from a specific stream and wetland, rather than the individual materials separately, is essential in understanding the effects of material fluxes on downstream waters. Science Report at 2-26.

In general, the more frequently a material is delivered to the river (*i.e.*, high connectivity), the greater its effect. The effect of an infrequently supplied material, however, can also be large if the material has a long residence time in the river and wetlands. *Id.* (citing Leibowitz *et al.* 2008). For example, woody debris might be exported to downstream waters infrequently but it can persist in downstream channels. In addition, some materials are more important in defining the structure and function of a river. For example, woody debris can have a large effect on river structure and function because it affects water flow, sediment and organic matter transport, and habitat. *Id.* (citing Harmon *et al.* 1986; Gurnell *et al.* 1995). Another example is salmon migrating to a river: They can serve as a keystone species to regulate other populations and as a source of marine-derived nutrients. *Id.* (citing Schindler *et al.* 2005). For functions relating to sinks and transformations, often the less frequently a material is delivered to the river (*i.e.*, high disconnectivity), the greater the effect. More examples of disconnectivity via sink functions are provided at the end of this section.

The functions discussed above represent general mechanisms by which streams and wetlands influence downstream waters. For these altered material and energy fluxes to affect a river, however, transport mechanisms that deliver (or could deliver) these materials to the river are necessary. Connectivity describes the degree to which components of a system are connected and interact through various transport mechanisms; connectivity is determined by the characteristics of both the physical landscape and the biota of the specific system. *Id.* This definition is related to, but is distinct from,

definitions of connectivity based on the actual flow of materials between system components. *Id.* (citing, *e.g.*, Pringle 2001). That connectivity among river-system components, including streams and wetlands, plays a significant role in the structure and function of these systems is not a new concept. In fact, much of the theory developed to explain how these systems work focuses on connectivity and linkages between system components. *Id.* (citing, *e.g.*, Vannote *et al.* 1980; Newbold *et al.* 1982a; Newbold *et al.* 1982b; Junk *et al.* 1989; Ward 1989; Benda *et al.* 2004b; Thorp *et al.* 2006).

In addition to its central role in defining river systems, water movement through the river system is the primary mechanism providing physical connectivity both within river networks and between those networks and the surrounding landscape. *Id.* (citing Fullerton *et al.* 2010). Hydrologic connectivity results from the flow of water, which provides a "hydraulic highway" along which physical, chemical, and biological materials associated with the water are transported (*e.g.*, sediment, woody debris, contaminants, organisms). *Id.* (citing Fausch *et al.* 2002).

Ecosystem functions within a river system are driven by interactions between the river system's physical environment and the diverse biological communities living within it. Id. (Wiens 2002; Schroder 2006). Thus, river system structure and function also depend on biological connectivity among the system's populations of aquatic and semiaquatic organisms. Biological connectivity refers to the movement of organisms, including transport of reproductive materials (e.g., seeds, eggs, genes) and dormant stages, through river systems. Id. at 2-26 to 2-27. These movements link aquatic habitats and populations in different locations through several processes important for the survival of individuals, populations, and species. Id. at 2-27. Movements include dispersal, or movement away from an existing population or parent organism; migration, or long-distance movements occurring seasonally; localized movement over an organism's home range to find food, mates, or refuge from predators or adverse conditions; and movement to different habitats to complete life-cycle requirements. Biological connectivity can occur within aquatic ecosystems or across ecosystem or watershed boundaries, and it can be multidirectional. For example, organisms can move downstream from perennial, intermittent, and ephemeral headwaters to rivers; upstream from estuaries to rivers to headwaters; or laterally between floodplain wetlands and open waters, non-floodplain wetlands and open waters, rivers, lakes, or other water bodies.

As noted above, streams, rivers, wetlands, and open waters are not pipes; they provide opportunities for water to interact with internal components (*e.g.*, alluvium, organisms) through the five functions (*i.e.*, sink, source, lag, transformation, and refugia) by which streams, wetlands, and open waters alter material fluxes. *Id.* (citing Bencala 1993; Bencala *et al.* 2011). Connectivity between streams and wetlands provides opportunities for material fluxes to be altered sequentially by multiple streams and wetlands as the materials are transported downstream. The aggregate effect of these sequential fluxes determines the proportion of material that ultimately reaches the river. The form of the exported material can be transformed as it moves down the river network, however, making quantitative assessments of the importance of individual stream and wetland resources within the entire river system difficult. For example, organic matter can be exported from headwater streams and consumed by downstream macroinvertebrates. Those invertebrates can drift farther downstream and be eaten by juvenile fish that eventually move into the mainstem of the river, where they feed further and grow.

The assessment of stream and wetland influence on rivers also is complicated by the cumulative time lag resulting from these sequential transformations and transportations. For example, removal of nutrients by streambed algal and microbial populations, subsequent feeding by fish and insects, and release by excretion or decomposition delays the export of nutrients downstream.

Although the Science Report primarily focused on the benefits that connectivity can have on downstream systems, isolation (or more accurately, degrees of "disconnectivity") also can have important positive effects on the condition and function of downstream waters, especially when considered in the aggregate. Nevertheless, the literature reviewed supports the conclusion that sink functions of nonfloodplain wetlands, which result in part from their relative isolation, will affect a downstream water when these wetlands are situated between the downstream water and known point or nonpoint sources of pollution, and thus intersect flowpaths between the pollutant source and downstream waters. For example, waterborne contaminants that enter a non-floodplain wetland (e.g., excessive nitrate) are processed through transformations or sink functions and hence are not transported to a river system because the wetland is on the "less frequently connected" side of the aquatic connectivity continuum (*i.e.*, if the wetland is hydrologically isolated from the river, except by non-hydrologic pathways. Id. at 2-28 to 2-29). Furthermore, increased hydrologic isolation, again when considered in the aggregate, can attenuate stormflows, decoupling storm peaks and decreasing flood risks to downgradient communities (e.g., a)Golden et al. 2021b). Increased isolation can also decrease the spread of pathogens and invasive species, and increase the rate of local adaptation. Id. at 2-29 (citing, e.g., Hess 1996; Bodamer and Bossenbroek 2008; Fraser et al. 2011). Thus, both connectivity and isolation should be considered when examining material fluxes from streams and wetlands, and biological interactions should be viewed in light of the natural balance between these two factors.

Spatial and Temporal Variability of Connectivity

Connectivity is not a fixed characteristic of a system, but varies over space and time. *Id.* (citing Ward 1989; Leibowitz 2003; Leibowitz and Vining 2003). Variability in hydrologic connectivity results primarily from the longitudinal and lateral expansion and contraction of the river network and transient connection with other components of the river system. The variability of connectivity can be described in terms of frequency, duration, magnitude, timing, and rate of change. When assessing the effects of connectivity or isolation and the five general functions (sources, sinks, refuges, lags, and transformations) on downstream waters, dimensions of time and space must be considered. Id. Water or organisms transported from distant headwater streams or wetlands generally will take longer to travel to a larger river than materials transported from streams or wetlands near the river. This can introduce a lag between the time the function occurs and the time the material arrives at the river. In addition, the distribution of streams and wetlands can be a function of their distance from the mainstem channel. For example, in a classic dendritic network, there is an inverse geometric relationship between number of streams and stream order. In such a case, the aggregate level of function could be greater for terminal source streams, compared to higher order or lateral source streams. This is one reason why watersheds of terminal source streams often provide the greatest proportion of water for major rivers. Connectivity, however, results from many interacting factors. For example, the relationship between stream number and order can vary with the shape of the watershed and the configuration of the network.

The expansion and contraction of river networks affects the extent, magnitude, timing, and type of hydrologic connectivity. For example, intermittent and ephemeral streams flow only during wetter seasons or during and immediately following precipitation events. Thus, the spatial extent of connectivity between streams and wetlands and rivers increases greatly during these high-flow events because intermittent and ephemeral streams are estimated to account for 59% of the total length of streams in the contiguous United States. *Id.* (citing Nadeau and Rains 2007). Changes in the spatial extent of connectivity due to expansion and contraction are even more pronounced in the arid and semiarid Southwest, where more than 80% of all streams are intermittent or ephemeral. *Id.* at 2-29 to 2-30 (citing Levick *et al.* 2008). Expansion and contraction also affect the magnitude of connectivity because larger flows provide greater potential for material transport. *Id.* at 2-30.

Besides affecting the spatial extent and magnitude of hydrologic connectivity, expansion and contraction of the stream network also affect the duration and timing of flow in different portions of the network. Perennial streams have year-round surface water connectivity with a downstream river, while intermittent streams have seasonal connectivity. The temporal characteristics of connectivity for ephemeral streams depend on the duration and timing of storm events. Similarly, connectivity between wetlands and downstream waters can range from permanent to seasonal to episodic.

The expansion and contraction of river systems also affect the type of connectivity. For example, during wet periods when input from precipitation can exceed evapotranspiration and available storage, non-floodplain wetlands could have connectivity with other wetlands or streams through surface spillage. *Id.* (citing Leibowitz and Vining 2003; Rains *et al.* 2008). When spillage ceases due to drier conditions, hydrologic connectivity could only occur through groundwater. *Id.* (citing Rains *et al.* 2006; Rains *et al.* 2008).

When the flow of water mediates dispersal, migration, and other forms of biotic movement, biological and hydrologic connectivity can be tightly coupled. For example, seasonal flooding of riparian/floodplain wetlands creates temporary habitat that fish, aquatic insects, and other organisms use. *Id.* (citing Junk *et al.* 1989; Smock 1994; Tockner *et al.* 2000; Robinson *et al.* 2002; Tronstad *et al.* 2007). Factors other than hydrologic dynamics also can affect the temporal and spatial dynamics of biological connectivity. Such factors include movement associated with seasonal habitat use and shifts in habitat use due to life-history changes, quality or quantity of food resources, presence or absence of favorable dispersal conditions, physical differences in aquatic habitat structure, or the number and sizes of nearby populations. *Id.* (citing Moll 1990; Smock 1994; Huryn and Gibbs 1999; Lamoureux and Madison 1999; Gibbons *et al.* 2006; Gamble *et al.* 2007; Grant *et al.* 2007; Subalusky *et al.* 2009a; Schalk and Luhring 2010). For a specific river system with a given spatial configuration, variability in biological connectivity also occurs due to variation in the dispersal distance of organisms and reproductive propagules. *Id.* (citing Semlitsch and Bodie 2003).

Finally, just as connectivity from temporary or seasonal wetting of channels can affect downstream waters, temporary or seasonal drying also can affect river networks. Riverbeds or streambeds that temporarily dry up are used by aquatic organisms that are specially adapted to alternating flowing and dry conditions, and can serve as egg and seed banks for several organisms, including aquatic invertebrates and plants. *Id* at 2-30 (citing Steward *et al.* 2012). These temporary dry areas also can affect nutrient

dynamics due to reduced microbial activity, increased oxygen availability, and inputs of terrestrial sources of organic matter and nutrients. *Id.* (citing Steward *et al.* 2012).

Numerous factors affect physical, chemical, and biological connectivity within river systems. These factors operate at multiple spatial and temporal scales, and interact with each other in complex ways to determine where components of a system fall on the connectivity-isolation gradient at a given time. *Id.* at 2-30 to 2-31. The Science Report focused on five key factors: climate, watershed characteristics, spatial distribution patterns, biota, and human activities and alterations. *Id.* at 2-31. These are by no means the only factors influencing connectivity, but they illustrate how many different variables shape physical, chemical, and biological connectivity.

Climate-watershed Characteristics

The movement and storage of water in watersheds varies with climatic, geologic, physiographic, and edaphic characteristics of river systems. *Id.* (citing Winter 2001; Wigington *et al.* 2013). At the largest spatial scale, climate determines the amount, timing, and duration of water available to watersheds and river basins. Key characteristics of water availability that influence connectivity include annual water surplus (precipitation minus evapotranspiration), timing (seasonality) of water surplus during the year that is heavily influenced by precipitation timing and form (*e.g.*, rain, snow), and rainfall intensity.

Annual runoff generally reflects water surplus and varies widely across the United States. Seasonality of water surplus during the year determines when and for how long runoff and ground-water recharge occur. Precipitation and water surplus in the eastern United States is less seasonal than in the West. *Id.* (Finkelstein and Truppi 1991). The Southwest experiences summer monsoonal rains, while the West Coast and Pacific Northwest receive most precipitation during the winter season. *Id.* (citing Wigington *et al.* 2013). Throughout the West, winter precipitation in the mountains occurs as snowfall, where it accumulates in seasonal snowpack and is released during the spring and summer melt seasons to sustain streamflow during late spring and summer months. *Id.* (citing Brooks *et al.* 2012). The flowing portions of river networks tend to have their maximum extent during seasons with the highest water surplus, when conditions for flooding are most likely. Typically, the occurrence of ephemeral and intermittent streams is greatest in watersheds with low annual runoff and high water surplus seasonality but also is influenced by watershed geologic and edaphic features. *Id.* (citing Gleeson *et al.* 2011).

Rainfall intensity can affect hydrologic connectivity in localities where watershed surfaces have low infiltration capacities relative to rainfall intensities. Infiltration-excess overland flow occurs when rainfall intensity exceeds watershed surface infiltration, and it can be an important mechanism in providing water to wetlands and river networks (Goodrich *et al.* 1997; Levick *et al.* 2008). Overland flow is common at low elevations in the Southwest, due to the presence of desert soils with low infiltration capacities combined with relatively high rainfall intensities. The Pacific Northwest has low rainfall intensities, whereas many locations in the Mid-Atlantic, Southeast, and Great Plains have higher rainfall intensities. The prevalence of impermeable surfaces in urban areas can generate overland flow in virtually any setting. *Id.* (citing Booth *et al.* 2002).

River system topography and landscape form can profoundly influence river network drainage patterns, distribution of wetlands, and ground-water and surface-water flowpaths. Winter (2001) described six generalized hydrologic landscape forms common throughout the United States. *Id.*

Mountain Valleys and Plateaus and High Plains have constrained valleys through which streams and rivers flow. *Id.* at 2-31, 2-33. The Mountain Valleys form has proportionally long, steep sides with narrow to nonexistent floodplains resulting in the rapid movement of water downslope. In contrast, Riverine Valleys have extensive floodplains that promote strong surface-water, hyporheic water, and alluvial ground-water connections between wetlands and rivers. *Id.* at 2-33 to 2-34. Small changes in water table elevations can influence the water levels and hydrologic connectivity of wetlands over extensive areas in this landscape form. Local ground-water flowpaths are especially important in Hummocky Terrain. Constrained valleys, such as the Mountain Valley landform, have limited opportunities for the development of floodplains and alluvial aquifers, whereas unconstrained valleys, such as the Riverine Valley landform, provide opportunities for the establishment of floodplains. Some river basins can be contained within a single hydrologic landscape form, but larger river basins commonly comprise complexes of hydrologic landscape forms. For example, the James River in Virginia, which flows from mountains through the Piedmont to the Coastal Plain, is an example of a Mountain Valley-High Plateaus and Plains-Coastal Terrain-Riverine Valley complex.

Floodplain hydrologic connectivity to rivers and streams occurs primarily through overbank flooding, shallow ground-water flow, and hyporheic flow. Water-table depth can influence connectivity across a range of hydrologic landscape forms, but especially in floodplains. Rivers and wetlands can shift from losing reaches (or recharge wetlands) during dry conditions to gaining reaches (or discharge wetlands) during wet conditions. Wet, high water-table conditions influence both ground-water and surface-water connectivity. When water tables are near the watershed surface, they create conditions in which swales and small stream channels fill with water and flow to nearby water bodies. Id. at 2-34 (citing Wigington et al. 2003; Wigington et al. 2005). Nanson and Croke (1992) noted that a complex interaction of fluvial processes forms floodplains, but their character and evolution are essentially a product of stream power (the rate of energy dissipation against the bed and banks of a river or stream) and sediment characteristics. Id. They proposed three floodplain classes based on the stream power-sediment characteristic paradigm: (1) high-energy noncohesive, (2) medium-energy noncohesive, and (3) lowenergy cohesive. The energy term describes stream power during floodplain formation, and the cohesiveness term depicts the nature of material deposited in the floodplain. In addition, hyporheic and alluvial aquifer exchanges are more responsive to seasonal discharge changes in floodplains with complex topography. Id. (citing Poole et al. 2006).

Within hydrologic landscape forms, soil and geologic formation permeabilities are important determinants of hydrologic flowpaths. Permeable soils promote infiltration that results in ground-water hydrologic flowpaths, whereas the presence of impermeable soils with low infiltration capacities is conducive to overland flow. In situations in which ground-water outflows from watersheds or landscapes dominate, the fate of water depends in part on the permeability of deeper geologic strata. The presence of an aquiclude (a confining layer) near the watershed surface leads to shallow subsurface flows through soil or geologic materials.

These local ground-water flowpaths connect portions of watersheds to nearby wetlands or streams. *Id.* at 2-35. Alternatively, if a deep permeable geologic material (an aquifer) is present, water is likely to move farther downward within watersheds and recharge deeper aquifers. *Id.* at 2-35 to 2-36. The permeability of soils and geologic formations both can influence the range of hydrologic connectivity between non-floodplain wetlands and river networks. *Id.* at 2-36.

Climate and watershed characteristics directly affect spatial and temporal patterns of connectivity between streams and wetlands and rivers by influencing the timing and extent of river network expansion and contraction. *Id.* at 2-38. They also influence the spatial distribution of water bodies within a watershed, and in particular, the spatial relationship between those water bodies and the river. *Id.* (citing, *e.g.*, Tihansky 1999)

Hydrologic connectivity between streams and rivers can be a function of the distance between the two water bodies. Id. (citing Bracken and Croke 2007; Peterson et al. 2007). If channels functioned as pipes, this would not be the case, and any water and its constituent materials exported from a stream eventually would reach the river. Because streams and rivers are not pipes, water can be lost from the channel through evapotranspiration and bank storage and diluted through downstream inputs. Id. (Bencala 1993). Thus, material from a headwater stream that flowed directly into the river would be subject to less transformation or dilution. On the other hand, the greater the distance a material travels between a particular stream reach and the river, the greater the opportunity for that material to be altered (e.g., taken up, transformed, or assimilated) in intervening stream reaches; this alteration could reduce the material's direct effect on the river, but it could also allow for beneficial transformations. For example, organic matter exported from a headwater stream located high in a drainage network might never reach the river in its original form, instead becoming reworked and incorporated into the food chain. Similarly, higher order streams generally are located closer to rivers and, therefore, can have higher connectivity than upstream reaches of lower order. Note that although an individual low-order stream can have less connectivity than a high-order stream, a river network has many more low-order streams, which can represent a large portion of the watershed; thus, the magnitude of the cumulative effect of these low-order streams can be significant.

The relationship between streams and the river network is a function, in part, of basin shape and network configuration. Elongated basins tend to have trellis networks where relatively small streams join a larger mainstem; compact basins tend to have dendritic networks with tree-like branching, where streams gradually increase in size before joining the mainstem. This network configuration describes the incremental accumulation of drainage area along rivers, and therefore provides information about the relative contributions of streams to downstream waters. Streams in a trellis network are more likely to connect directly to a mainstem, compared with a dendritic network. The relationship between basin shape, network configuration, and connectivity, however, is complex. A mainstem in a trellis network also is more likely to have a lower stream order than one in a dendritic network. *Id.* at 2-38 to 2-39.

Distance also affects connectivity between non-floodplain and riparian/floodplain wetlands and downstream waters. *Id.* at 2-39. Riverine wetlands that serve as origins for lateral source streams that connect directly to a mainstem river have a more direct connection to that river than wetlands that serve as origins for terminal source streams high in a drainage network. This also applies to riparian/floodplain wetlands that have direct surface-water connections to streams or rivers. If geographically isolated non-floodplain wetlands have surface-water outputs (*e.g.*, depressions that experience surface-water spillage or ground-water seeps), the probability that surface water will infiltrate or be lost through evapotranspiration increases with distance. This is a beneficial function in attenuating storm flows and providing area (and time) for biogeochemical processing. Golden *et al.* 2021b; Evenson *et al.* 2018. For non-floodplain wetlands connected through ground-water flows, less distant areas are generally connected through shallower flowpaths, assuming similar soil and geologic properties. These shallower ground-

water flows have the greatest interchange with surface waters and travel between points in the shortest amount of time. Although elevation is the primary factor determining areas that are inundated through overbank flooding, connectivity with the river generally will be higher for riparian/floodplain wetlands located near the river's edge compared with riparian/floodplain wetlands occurring near the floodplain edge.

Distance from the river network also influences biological connectivity among streams and wetlands. For example, mortality of an organism due to predators and natural hazards generally increases with the distance it has to travel to reach the river network. The likelihood that organisms or propagules traveling randomly or by diffusive mechanisms such as wind will arrive at the river network decreases as distance increases.

The distribution of distances between wetlands and river networks depends on both the drainage density of the river network (the total length of stream channels per unit area) and the density of wetlands. Science Report at 2-40. Climate and watershed characteristics influence these spatial patterns, which can vary widely.

<u>Biota</u>

Biological connectivity results from the interaction of physical characteristics of the environment—especially those facilitating or restricting dispersal—and species' traits or behaviors, such as life-cycle requirements, dispersal ability, or responses to environmental cues. *Id.* Thus, the types of biota within a river system are integral in determining the river system's connectivity, and landscape features or species traits that necessitate or facilitate movement of organisms tend to increase biological connectivity among water bodies.

Diadromous fauna (*e.g.*, Pacific and Atlantic salmon, certain freshwater shrimps and snails, American eels), which require both freshwater and marine habitats over their life cycles and therefore migrate along river networks, provide one of the clearest illustrations of biological connectivity. Many of these taxa are either obligate or facultative users of headwater streams, meaning that they either require (obligate) or can take advantage of (facultative) these habitats; these taxa thereby create a biological connection along the entire length of the river network. *Id.* (citing Erman and Hawthorne 1976; Wigington *et al.* 2006). For example, many Pacific salmon species spawn in headwater streams, where their young grow for a year or more before migrating downstream, living their adult life stages in the ocean, and then migrating back upstream to spawn. Many taxa also can exploit temporary hydrologic connections between rivers and floodplain wetland habitats caused by flood pulses, moving into these wetlands to feed, reproduce, or avoid harsh environmental conditions and then returning to the river network. *Id.* at 2-40, 2-43 (citing Copp 1989; Junk *et al.* 1989; Smock 1994; Tockner *et al.* 2000; Richardson *et al.* 2005).

Biological connectivity does not solely depend on diadromy, however, as many non-diadromous organisms are capable of significant movement within river networks. *Id.* at 2-40. For example, organisms such as pelagic-spawning fish and mussels release directly into the water eggs or larvae that disperse downstream with water flow; many fish swim significant distances both upstream and downstream; and many aquatic macroinvertebrates move or drift downstream. *Id.* at 2-40 (citing, *e.g.*, Elliott 1971; Müller 1982; Gorman 1986; Brittain and Eikeland 1988; Platania and Altenbach 1998; Elliott 2003; Hitt and

Angermeier 2008; Schwalb *et al.* 2010). Taxa capable of movement over land, via either passive transport (*e.g.*, wind dispersal or attachment to animals capable of terrestrial dispersal) or active movement (*e.g.*, terrestrial dispersal or aerial dispersal of winged adult stages), can establish biotic linkages between river networks and wetlands, as well as linkages across neighboring river systems. Science Report at 2-40 (citing Hughes *et al.* 2009).

Gradients of biological connectivity (*i.e.*, the active or passive movements of organisms through water or air and over land that connect populations) are determined primarily by species assemblages, and by features of the landscape (*e.g.*, climate, geology, terrain) that facilitate or impede the movement of organisms. Science Report 6-10. For example, mass river insect migrations into headwater streams provide food subsidies to support young-of-year fish (Uno and Power 2015), including diadromous salmon (Bramblett *et al.* 2002). On the other hand, lower rates of movement between more isolated habitats can decrease the spread of pathogens (*e.g.*, Hess 1996) and invasive species (*e.g.*, Bodamer and Bossenbroek 2008) and increase regional biodiversity through adaptation to local conditions, (*e.g.*, Fraser *et al.* 2011), increasing resiliency of aquatic species to changing landuse and climate.

Human Activities and Alterations

Human activities frequently alter connectivity between headwater streams, riparian/floodplain wetlands, non-floodplain wetlands, and downgradient river networks. Id. at 2-44. In doing so, they alter the transfer and movement of materials and energy between river system components. In fact, the individual or cumulative effects of headwater streams and wetlands on river networks often become discernible only following human-mediated changes in degree of connectivity. These human-mediated changes can increase or decrease hydrologic and biological connectivity (or, alternatively, decrease or increase hydrologic and biological isolation). Id at 2-44 to 2-45. For example, activities and alterations such as dams, levees, water abstraction, piping, channelization, and burial can reduce hydrologic connectivity between streams and wetlands and rivers, whereas activities and alterations such as wetland drainage, irrigation, impervious surfaces, interbasin transfers, and channelization can enhance hydrologic connections. Id. at 2-45. Biological connectivity can be affected similarly: For example, dams and impoundments might impede biotic movement, whereas nonnative species introductions artificially increase biotic movement. Further complicating the issue is that a given activity or alteration might simultaneously increase and decrease connectivity, depending on which part of the river network is considered. For example, channelization and levee construction reduce lateral expansion of the river network (thereby reducing hydrologic connections with floodplains), but might increase this connectivity downstream due to increased frequency and magnitude of high flows.

The greatest human impact on riparian/floodplain wetlands and non-floodplain wetlands has been through wetland drainage, primarily for agricultural purposes. Estimates show that, in the conterminous United States, states have lost more than half their original wetlands (50%), with some losing more than 90%; wetland surface areas also have declined significantly. *Id.* (citing Dahl 1990).

Drainage causes a direct loss of function and connectivity in cases where wetland characteristics are completely lost. *Id.* at 2-45. In the Des Moines lobe of the prairie pothole region, where more than 90% of the wetlands have been drained, a disproportionate loss of smaller and larger wetlands has occurred. Accompanying this loss have been significant decreases in perimeter area ratios—which are associated with greater biogeochemical processing and groundwater recharge rates—and increased mean

distances between wetlands, which reduces biological connectivity. *Id.* at 2-45 to 2-46 (citing Van Meter and Basu 2015). Wetland drainage also increases hydrologic connectivity between the landscape including drained areas that retain wetland characteristics—and downstream waters. Effects of this enhanced hydrologic connectivity include (1) reduced water storage and more rapid conveyance of water to the network, with subsequent increases in total runoff, baseflows, stormflows, and flooding risk; (2) increased delivery of sediment and pollutants to downstream waters; and (3) increased transport of waterdispersing organisms. *Id.* at 2-46 to 2-47 (citing Babbitt and Tanner 2000; Baber *et al.* 2002; Mulhouse and Galatowitsch 2003; Wiskow and van der Ploeg 2003; Blann *et al.* 2009). Biological connectivity, however, also can decrease with drainage and ditching, as average distances between wetlands increase and limit the ability of organisms to disperse between systems aerially or terrestrially. *Id.* at 2-47 (citing Leibowitz 2003). Groundwater withdrawal also can affect wetland connectivity by reducing the number of wetlands. Of particular concern in the arid Southwest is that ground-water withdrawal can decrease regional and local water tables, reducing or altogether eliminating ground-water-dependent wetlands. *Id.* (citing Patten *et al.* 2008). Groundwater withdrawal, however, also can increase connectivity in areas where that ground water is applied or consumed.

B. Peer Review of Report

The process for developing the Science Report followed standard information quality guidelines for EPA. In September 2013, EPA released a draft of the Science Report for an independent SAB review and invited submissions of public comments for consideration by the SAB panel. In October 2014, after several public meetings and hearings, the SAB completed its peer review of the draft Science Report (*hereafter*, "SAB 2014a"). The SAB was highly supportive of the draft Science Report's conclusions regarding streams, riparian and floodplain wetlands, and open waters, and recommended strengthening the conclusion regarding non-floodplain waters to include a more definitive statement that reflects how numerous functions of such waters sustain the integrity of downstream waters. SAB 2014a. The final peer review report is available on the SAB website, as well as in the docket for this final rulemaking. EPA revised the draft Science Report based on comments from the public and recommendations from the SAB panel.

The SAB was established in 1978 by the Environmental Research, Development, and Demonstration Authorization Act (ERDDAA), to provide independent scientific and technical advice to the EPA Administrator on the technical basis for Agency positions and regulations. Advisory functions include peer review of EPA's technical documents, such as the Science Report. At the time the peer review was completed, the chartered SAB was comprised of more than 50 members from a variety of sectors including academia, non-profit organizations, foundations, state governments, consulting firms, and industry. To conduct the peer review, EPA's SAB staff formed an ad hoc panel based on nominations from the public to serve as the primary reviewers. The panel consisted of 27 technical experts in an array of relevant fields, including hydrology, wetland and stream ecology, biology, geomorphology, biogeochemistry, and freshwater science. Similar to the chartered SAB, the panel members represented sectors including academia, a federal government agency, non-profit organizations, and consulting firms. The chair of the panel was a member of the chartered SAB.

The SAB process is open and transparent, consistent with the Federal Advisory Committee Act, 5 U.S.C., App 2, and agency policies regarding Federal advisory committees. Consequently, the SAB has

an approved charter, which must be renewed biennially, announces its meetings in the *Federal Register*, and provides opportunities for public comment on issues before the Board. The SAB staff announced via the *Federal Register* that they sought public nominations of technical experts to serve on the expert panel: SAB Panel for the Review of the EPA Water Body Connectivity Report (via a similar process the public also is invited to nominate chartered SAB members). 78 FR 15012 (March 8, 2013). The SAB staff then invited the public to comment on the list of candidates for the panel. Once the panel was selected, the SAB staff posted a memo on its website addressing the formation of the panel and the set of determinations that were necessary for its formation (e.g., no conflicts of interest). In the public notice of the first public meetings interested members of the public were invited to submit relevant comments for the SAB Panel to consider pertaining to the review materials, including the charge to the Panel. Over 133,000 public comments were received by the Docket. Every meeting was open to the public, noticed in the *Federal Register*, and had time allotted for the public to present their views. In total, the Panel held a two-day in-person meeting in Washington, DC, in December 2013, and three four-hour public teleconferences in April, May, and June 2014. The SAB Panel also compiled four draft versions of its peer review report to inform and assist the meeting deliberations that were posted on the SAB website. In September 2014, the chartered SAB conducted a public teleconference to conduct the quality review of the Panel's final draft peer review report. The peer review report was approved at that meeting, and revisions were made to reflect the chartered SAB's review. The culmination of that public process was the release of the final peer review report in October 2014. All meeting minutes and draft reports are available on the SAB website for public access.

C. Updates to the Literature Since Publication of the Science Report

The agencies requested that EPA's Office of Research and Development (ORD) prepare an updated summary of the scientific evidence on the connectivity and downstream effects of streams (ephemeral, intermittent, and perennial) and both floodplain wetlands and open waters and non-floodplain wetlands and open waters since the publishing of the Science Report. The major conclusions of the Science Report are discussed in section I.A.i above.

The goal of this update was to analyze and synthesize the peer-reviewed scientific literature published in or after 2014, the year determined to correspond with the finalization of the Science Report, and summarize the updated "scientific understanding about the connectivity and mechanisms by which streams and wetlands, singly or in aggregate, affect the physical, chemical, and biological integrity of downstream waters." Science Report at ES-1.

As discussed in this section, since the publication of the Science Report in 2015, the published literature has supported the conclusions of the Science Report and has expanded scientific understanding and quantification of functions that ephemeral streams and non-floodplain waters perform that affect the integrity of larger downstream, particularly in the aggregate. *See also* Sullivan *et al.* 2019a, 2019b, 2020; CASS 2021. For example, the more recent literature (*i.e.*, 2014-present) has strengthened the scientific evidence underpinning the findings that non-floodplain wetlands have demonstrable hydrologic and biogeochemical downstream effects, such as decreasing peak flows, maintaining baseflows, and performing nitrate removal, particularly when considered cumulatively.

In addition to the screening process discussed in this section, the agencies also reviewed additional peer-reviewed literature published in or after 2014, including scientific references that were provided to the agencies as part of the public comment process for the proposed rule. The review of this additional literature is discussed in section I.C.vi. below. EPA's Science Advisory Board (SAB) also chose to review the proposed rule, including aspects of the Technical Support Document, for technical and scientific accuracy. The SAB's review is discussed in section I.G.

i. Update Process

The specific charge questions addressed in this update are the same as the Science Report (p. 1-1) report, namely:

A. What are the physical, chemical, and biological connections to and effects of ephemeral, intermittent, and perennial streams on downstream waters (*e.g.*, rivers, lakes, reservoirs, estuaries)?

B. What are the physical, chemical, and biological connections to and effects of riparian or floodplain wetlands and open waters (*e.g.*, riverine wetlands, oxbow lakes) on downstream waters?

C. What are the physical, chemical, and biological connections to and effects of wetlands and open waters in non-floodplain settings (*e.g.*, most prairie potholes, vernal pools) on downstream waters?

Definitions used in this analysis follow the glossary available in the Science Report (*i.e.*, at A-5 onwards) and Appendix B of this document.

ORD subject-matter scientists identified an initial database of 553 scientific peer-reviewed papers relevant to the specific charge questions (*i.e.*, questions related to the connectivity and/or effects of (a) ephemeral, intermittent, and perennial streams, (b) floodplain wetlands and open waters, or (c) non-floodplain wetlands and open waters) (Supplementary Material A). The focal paper citation database (n=553) was screened for duplicates and provided to EPA's Health and Environmental Research Online (HERO) library research service. HERO library science staff conducted a 'forward-citation mapping' analysis within the Web of Science (WoS) global citation database for the 553-forward citation-mapping papers that had identifiers within the WoS. There were 17,044 peer-reviewed scientific papers published from 2014 onwards within WoS that had cited one or more of the 553 relevant papers provided by ORD. These 17,044 papers were the set from which all three evidentiary reviews were conducted.

To analyze the updated scientific evidence within the available time frame, ORD scientists determined to include or exclude articles based on the paper's relevance to the specific charge questions through a review of the title and abstract of the scientific articles within the SWIFT-Active Screener environment (v. 1.061.0514 through v. 1.061.0527, Sciome, LLC., Research Triangle Park, NC). SWIFT-Active Screener is a software program designed to facilitate collaborative systematic reviews through application of a machine-learning algorithm that incorporates reviewer feedback (*i.e.*, include/exclude) to prioritize a population of articles for screening (Howard *et al.* 2020). Concurrently, a separate model within SWIFT-Active Screener estimates how many relevant articles remain in the screening pool; ORD established a target of 95% recall, or "...the percentage of truly relevant documents [to be] discovered

during screening." Howard *et al.* 2020 at 4. These two models within SWIFT-Active Screener provide the opportunity to rapidly identify and screen the relevant articles from a dataset, in this case 17,044 scientific papers.

ORD identified 15-20 papers from the 17,044 that were selected as an initial training set for each of the evidentiary reviews (*i.e.*, ephemeral, intermittent, and perennial streams; floodplain wetlands and open waters; and non-floodplain wetlands and open waters; Supplementary Material B). These 15-20 papers were identified through preliminary screening as likely relevant (*i.e.*, likely to be included) in the active learning model, and these papers were moved to the top of the screening prioritization (*i.e.*, screened within the 30 papers presented to reviewers). This initiated the model building within SWIFT-Active Screener; this model was retrained after every 30 papers were screened. Howard *et al.* 2020. Three ORD scientists independently screened at least 1,000 scientific papers each for the evidentiary review on the connectivity and downstream effects of ephemeral, intermittent, and perennial streams. A separate ORD screening team independently screened at least 1,000 scientific papers each focusing on floodplain wetlands and open waters; that same team of three also independently screened at least 1,000 scientific papers each focusing on floodplain wetlands and open waters.

During the screening process, ORD scientists read and reviewed the title and abstract (hereafter abstract) of each paper presented and determined if the paper was relevant to the appropriate charge question noted above. If papers were determined to be relevant, several additional questions were presented to the screener to extract further information associated with the charge questions (*e.g.*, the types of aquatic systems studied, etc.) and the ORD reviewer checked the appropriate boxes based on knowledge gleaned from the abstract. Importantly, a question specific to the major conclusions of the Science Report (*see* sections I.A.i and I.A.ii) was asked of each included paper. For instance, the screeners assessed abstracts for included papers and considered the following statements for each paper, having re-familiarized themselves with the full-text major conclusions of the Science Report. Note that five of the six screeners were co-authors of the Report, and the remaining screener has worked as a Research Ecologist on this topic since 2012. Thus, the ORD scientists who served as screeners were subject-matter experts and well-versed in the Science Report and its conclusions, as well as literature that has been published since the Report's release. The Report's conclusions were as follows:

Streams: The scientific literature unequivocally demonstrates that streams, individually or cumulatively, exert a strong influence on the integrity of downstream waters. All tributaries, regardless of size or flow duration, are physically, chemically, and biologically connected to downstream waters and strongly influence their function.

Floodplain Wetlands and Open Waters: Wetlands and open waters in riparian areas and floodplains are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality. These systems buffer downstream waters from pollution and are essential components of river food webs.

Non-Floodplain Wetlands and Open Waters: Wetlands and open waters located outside of riparian areas and floodplains, even when lacking surface water connections, provide numerous functions that could affect the integrity of downstream waters. Some benefits of these wetlands are due to their relative isolation rather than their connections.

For this particular question, screeners chose from the following answers: supports findings, refutes findings, cannot be discerned. The full list of questions asked and possible answers for selection are listed in Supplementary Material C.

ii. Results

1. Process

ORD subject-matter experts screened the abstracts of 12,659 scientific papers across the three aquatic system topic areas. Papers were assessed on connectivity and downstream effects of ephemeral, intermittent, and/or perennial streams (total papers screened = 4,194, 24.6% of the scientific literature published in or after 2014), floodplain wetlands and open waters (total screened = 4,183, 24.5% of the presented scientific literature), and non-floodplain wetlands and open waters (total screened = 4,282, 25.1% of the presented scientific literature). The total number of papers screened by each reviewer varied by aquatic system, with stream screeners ranging from 1,200-1,527 papers screened (average = 1,398), floodplain wetlands and open water screeners ranging from 1,273-1,600 (average = 1,394), and non-floodplain wetlands and open waters ranging from 1,273-1,600 (average = 1,427). ORD screeners did not reach their SWIFT-Active Screener informed discovery and review target of 95% of the relevant literature (Howard *et al.* 2020) for any of the three aquatic systems analyzed. When the SWIFT-Active Screener recursive model-rebuilding ended for stream systems, 986 of 1,225.5 papers were included for a recall value of 80.5%. Floodplain wetland and open water systems screeners included 660 papers of a predicted 813.8, a recall value of 81.1%. The calculated non-floodplain wetlands and open water systems recall value was 89.6%, with 491 papers of 547.8 included.

The vast majority of the relevant papers were found earlier in the process, with fewer and fewer relevant papers found towards the end (e.g., frequently dozens to hundreds of papers screened prior to finding a relevant paper to include as the screening process neared completion).

The ratio of included to excluded papers varied between screeners and across the three systems analyzed. The total included (n=2,137) to excluded (n=10,522) resulted in a ratio of 0.20; the include:exclude ratio was lowest for non-floodplain wetlands and open waters (0.13, indicating fewer papers were included), 0.19 for floodplain wetlands and open waters, and 0.31 for stream systems. The include:exclude ratio across screeners ranged from 0.10 to 0.34. Of the 2,137 total papers included, there were 2,022 unique papers, as 115 papers were relevant to multiple systems. The 2,022 unique papers are listed in Appendix C1.

2. Analysis and Synthesis

a. Ephemeral, Intermittent, and/or Perennial Streams

Nine hundred and eighty-six (986) scientific papers published in or after 2014 were included by the screeners. *See* Appendix C1. The stream type(s) (*e.g.*, ephemeral, intermittent, perennial, headwater [first- to third-order streams, can be ephemeral, intermittent, or perennial] for included papers was discernable in 38% of the papers (n=379; further analyzed below), with 607 papers not sufficiently descriptive to determine the stream type.

Across all screened and included papers (n=986), biological connectivity and effects was the most commonly reported type (n=344, or 35%), followed by physical (n=200, 20%) and chemical (n=163, 17%). Multiple types of connectivity and effects were noted in 265 (27%) papers, with physical and biological (n=122, 12%) and physical and chemical (n=101, 10%) more commonly found than chemical and biological (n=10, 1%). All three connectivity and effects types were reported in 32 papers (3%). Thirteen papers were noted as "connection and effects type[s]" could not be discerned, and one paper was marked as chemical connectivity and effects as well as "effect type cannot be discerned." The plurality of the papers were marked as addressing watershed-scales (e.g., cumulative connections and effects; n=379 or 38%), followed by reach-scale studies (n=360, 37%), and "scale not discernable" in 211 papers (21%). Both reach and watershed scales were marked in 32 papers (4%). The location of most studies (n=613, 62%) was not discernible from the abstract (e.g., a specific geographic location was not given in the abstract) or noted as outside of the United States. The balance was noted as having sufficient information in the abstract, such as a place name, to select occurring in the United States or parts thereof (n=373, 38%). The contributions of climate change, land use, or water use on connectivity and effects of the aquatic systems were analyzed for each paper. Most papers screened were noted as having no interacting effects (n=581, 59%), though climate interactions (n=167, 17%) and land use (n=132, 13%) were frequently noted, as was a combination of interacting effects (n=78, 8%). Water use was noted as an interacting effect in 28 papers (3%).

The screeners assessed whether the abstract contained sufficient information to support or refute the major conclusions on stream systems from the Science Report (discussed in sections I.A.i and I.A.ii). Forty-eight percent (n=471) of the papers were determined to lack sufficient information to assess their support or refutation vis-à-vis the conclusions. Of the remaining 515 papers, 506 (98%) were found to support the physical, biological, and chemical connectivity and effects of stream systems on downgradient waters. Four papers (1%) were marked as refuting the findings, while two were marked as both supports and refutes the findings. Three additional papers were marked as both supports the findings and support or refutation "cannot be discerned."

Analyzing a subset of papers where stream type was discernable to the screeners (379 of 986 papers, or 38%), the majority of the papers with indicated stream type(s) identified headwaters (n=179, 47%). Screeners noted ephemeral in 26 papers (7%), intermittent in 44 (12%), and perennial in 22 (6%), with the balance of papers a combination of types. Analyzing the 379 papers with known stream types, biological connections and/or effects were the most commonly reported (n=111, 29%), followed by physical (n=77, 20%), then chemical (n=73, 19%). Fifty-three papers (14%) were noted to have both physical and biological connections and/or effects, 38 (10%) had physical and chemical, and 21 (6%) had all three connectivity and effects types. Watershed-scale connectivity and effects were most commonly marked (n=166, 44%) followed by reach-scale (n=147, 39%) and both reach-scale and watershed-scale (n=16, 4%). The scale of the study could not be discerned in 50 of the screened papers with a known stream type denoted (13%). Forty percent of the studies (n=151) were noted as occurring in the United States or parts of the United States, whereas the location could not be discerned or did not encompass the United States in 60% (n=228) of the papers. Climate (n=80, 21%) and land use (n=53, 14%) were the most commonly occurring interactions noted, along with water use (n=11, 3%) and any combination of interacting effects (n=32, 8%).

Of the included papers with a known stream type marked (n=379), whether the conclusions supported or refuted the Science Report could be discerned in 224 of the papers (59%). Based on the abstracts read, the scientific papers reviewed with known types denoted unequivocally supported the connectivity and effects of ephemeral, intermittent, perennial, and headwater stream systems with 219 (98%) supporting the major conclusions of the Science Report and one paper noted as refuting the findings. Three papers were marked as supports and "cannot be discerned" and one paper was marked as both supports and refutes.

A further analysis was conducted on two smaller paper subsets on ephemeral streams (n=28) and both ephemeral and intermittent streams (n=70). Twenty-eight were marked as focusing on ephemeral streams (n=28, *i.e.*, marked as ephemeral [n=26], ephemeral and headwater [n=1], and ephemeral and "stream type not discernable [n=1]). Of these, 22 papers were noted as supporting the Science Report major conclusions and zero papers were marked as refuting the findings. Two additional papers noted as both supporting and "cannot be discerned." Support or refutation could not be discerned in an additional four papers.

An additional 70 papers (19%) were demarcated as ephemeral and intermittent (n=14), intermittent (n=44), headwater and intermittent (n=11), and headwater plus ephemeral and intermittent (n=1). Support or refutation was discerned in 35 of those papers. The data presented to the screeners in the abstracts were sufficient for all 35 (100%) to be denoted as supporting the major conclusions of the Science Report, namely that these systems were physically, chemically, and/or connected to and exerted a strong influence on downgradient waters.

b. Floodplain Wetlands and Open Waters

Based on the content of the abstracts read by the screeners, six hundred and sixty (660) scientific papers published in or after 2014 were included for analyses. Of these, 62% (n=406) were noted as focusing specifically on floodplain (or riparian) wetlands and open waters (further analyzed below), 31 (5%) on riverine (*i.e.*, within channel) systems, 92 (14%) addressed multiple wetland types (*e.g.*, riverine, floodplain wetlands, non-floodplain wetlands, etc.), and 33 (5%) lacked sufficient information to discern the aquatic system.

Biological connections and effects were noted as the most commonly occurring in the papers reviewed (n=195, 30%), followed by chemical (n=144, 22%) and physical (n=143, 22%). Multiple connections and effect types were frequently noted, with both physical and chemical most common (n=103, 16%), followed by physical and biological (n=34, 5%), and chemical and biological (n=10, 2%). Eighteen papers (3%) were marked as having all three types of connections and effects, and 13 papers (2%) were noted as connection and/effect type could not be discerned.

The scale of most screened and included papers were individual functions, connections, and effects (n=174, 26%) and landscape-scale analyses (n=162, 25%); 123 papers were marked as addressing both scales (19%). Fifty-three papers (8%) focused on watershed-scale analyses, while 57 (9%) noted both an individual and watershed-scale focus, 57 (9%) addressed landscape- and watershed-scales, and 14 (2%) addressed all three listed scales. Twenty-six papers (4%) lacked sufficient information for screeners to note scale, and one paper noted both landscape-scale and "scale not discernable." Thirty-nine percent (n=254) of the papers emanated from studies in the United States, whereas the balance (n=406) either

were outside of the United States or the study location could not be discerned from the abstract. Most papers did not have interacting effects noted (n=567, 86%), though 34 papers noted climate and 33 land use (5% each). Both water use and any combination of the three interacting effects were noted to occur in 13 papers (2% each).

The Science Report concludes that wetlands and open waters in riparian areas and floodplains are physically, chemically, and biologically integrated with rivers through functions that affect downstream water quality. Based on the abstracts reviewed from scientific papers published in or after 2014, 449 papers presented sufficient material in the abstract relative to this finding. Of those papers, 95% (n=427) supported the conclusion, with another 3% (n=12) marked as supports and cannot be discerned. Seven papers (2%) were noted as both supporting and refuting, two papers were marked as refutes and cannot be discerned, and one paper was marked as refuting the major conclusions.

Focusing specifically on a subset of papers with the aquatic system type noted as floodplain wetlands and open waters (n=406), these focal aquatic systems were found to mainly connect and/or affect flowing waters through biological connections and effects (n=123, 30%), followed by physical (n=85, 21%) and chemical (n=81, 20%) connections and effects. Both physical and chemical connections were noted on 76 papers (19%), followed by physical and biological (n=25, 6%), and chemical and biological (n=5, 1%). All three connectivity and effects types (n=7, 2%) were found infrequently, and connection and effect type were not discernable in four papers (1%).

Individual system-scales of analyses was the most commonly selected scale (n=121, 30%), followed by landscape-scale (n=76, 19%); both were pipped in 96 papers (24%). Watershed-scale studies were noted in 29 cases (7%), and both individual and watershed-scales in 41 papers (10%), and landscape- and watershed-scales in 22 papers (5%). All three scales of analysis were denoted in nine papers (2%), whereas no scale was discernable in twelve papers (3%).

Findings related to the major conclusions of the Science Report were discerned in 298 of the 406 papers that focused on floodplain wetlands and open waters (74%). Of those 298 papers, 98% were found to support the findings on the interrelatedness and connectivity of floodplain wetlands and open waters with rivers. Four papers (1%) were marked as both supporting and "cannot be discerned." Two papers (1%) were noted as both supporting and refuting the major conclusions.

c. Non-Floodplain Wetlands and Open Waters

Four-hundred and ninety-one (491) scientific papers published in or after 2014 were included by the screeners, and 51% (n=250, further analyzed below) were marked as explicitly focusing on known non-floodplain wetland types. A further 113 (23%) were marked as either multiple wetland types and spatial locations, and 20 (4%) had both "explicitly about known non-floodplain wetland type" and "multiple wetland types" marked. Six papers were marked as both multiple wetland types and wetlands type not discernable (1%). Wetland type was not discernable in 102 included papers (21%).

Physical connectivity and/or effects were most commonly noted (n=160, 33%), followed by biological connectivity and/or effects (n=147, 30%), and chemical connectivity and/or effects (n=90, 18%). Multiple connectivity and/or effects were frequently marked (n=82, 17%), with all but four (1% noted as chemical and biological) including physical connectivity and/or effects (*e.g.*, 48 papers marked

as addressing physical and chemical effects, 12 papers addressing physical and biological, and 18 papers including all three connectivity types). The frequency of physical connectivity and effects was likely a result of surface, shallow subsurface, or groundwater serving as the physical medium directly connecting non-floodplain wetlands to downgradient systems. Surface water, shallow subsurface water, or groundwater also serves to functionally connect non-floodplain wetlands, for instance by mobilizing dissolved or entrained materials into non-floodplain wetlands whereby lags, transformation, or sink functions affecting downgradient systems occur. Leibowitz *et al.* 2018.

The plurality of papers (n=182, 37%) addressed landscape-scale types of connections and effects (*e.g.*, connections, functions, and/or effects on other features, such as a stream). One hundred and two papers (21%) focused on watershed-scale downgradient or downstream cumulative connections, functions, and/or effects, while 75 papers (15%) were noted as addressing functions or scales associated with individual wetland and open waters. Fifty-seven (12%) addressed individual and landscape scales while 40 were marked as both landscape and watershed connections (8%). Twelve were marked as individual and watershed scales (2%), five were noted as a combination of all three scales (1%), and 18 (4%) did not have scale discernible from the abstracts. Fifty-two percent (n=257) were marked as being conducted in the US or parts thereof. Sixty-five percent (n=317) lacked interacting effects while 17% (n=81) involved climate effects and 12% (n=61) addressed land use effects. Seventeen papers (n=4%) addressed water use and fifteen papers (3%) addressed some combination of the three effects.

Two hundred and thirty papers (47%, n=230) of included scientific papers (n=491) presented sufficient information in the abstract to assess findings vis-à-vis the major conclusions from the Science Report. Of the 230 papers assessed, 97% (n=223) were marked as supporting the findings of the Science Report. Three papers (1%) were noted as refuting the findings, two papers (1%) were noted as both supporting and refuting, and two papers (1%) were noted as supporting and "cannot be discerned."

There were 250 papers (51% of 491 included papers) marked as explicitly focusing on known non-floodplain wetland and open water types. Of these papers, most were noted as having physical connectivity and effects (n=89, 36%) followed by biological (n=68, 27%), and chemical (n=53, 21%); connection and/or effect type(s) were not discernable in seven papers (3%). Twenty papers were noted as addressing physical and chemical connectivity and effects (8%), with two each of physical and chemical or biological and chemical (1% each). Nine papers (4%) addressed all three connectivity and effects types. The scale of the included studies specific to non-floodplain wetland types tended towards landscape-scale papers (n=86, 34%), followed by watershed-scale papers (n=46, 18%), and individual wetlands and waters were noted as the paper focus in 45 instances (18%); 10 papers were marked as "scale not discernable" by the screeners (4%). Thirty-six papers (14%) were noted as addressing both individual and landscape-scale analyses, while 15 (6%) were marked as both landscape- and watershedscale, and eight (3%) marked as individual and watershed-scale. Four papers (2%) were noted to be a combination of all three scales. Sixty percent (n=149) of the papers noted research was conducted in the United States or parts of the United States. Most papers (n=154, 62%) did not include an interacting effect. Forty-five papers addressed climate interactions (18%), 35 addressed land use (14%), six addressed water use (2%), and 10 papers had a combination of interacting effects.

Major conclusions from the Science Report on the connectivity and effects of non-floodplain wetlands and open waters on downstream waters state that these systems provide numerous functions that

could affect the integrity of downstream waters, and that some benefits of these wetlands emerge from their relative isolation (or disconnectivity) rather than their connections. The finding was supported when sufficient information was available in the abstract to assess. Across papers marked as explicitly focusing on known non-floodplain wetland types and sufficiently descriptive to assess the findings vis-à-vis the major conclusions (n=133 of 250), 99% of all papers (n=131) were found to support the findings, with one paper marked as both supporting and refuting the findings, and one paper noted to refute the findings.

iii. Discussion

The goal of this summary review of the scientific evidence was to analyze and synthesize the peer-reviewed scientific literature (published in or after 2014) and to provide updates on the "...scientific understanding about the connectivity and mechanisms by which streams and wetlands, singly or in aggregate, affect the physical, chemical, and biological integrity of downstream waters." Science Report at ES-1. As noted, the summarized state-of-the-science in the Science Report is given in sections I.A.i and I.A.ii; updated assessments of the state of the science were peer-reviewed and published as a Featured Collection in the *Journal of the American Water Resources Association*. Alexander *et al.* 2018; Fritz *et al.* 2018; Goodrich *et al.* 2018; Lane *et al.* 2018, Leibowitz *et al.* 2018; Schofield *et al.* 2018.

The analysis by ORD subject-matter experts summarized in this document was conducted over a six-week period and consisted of reviewing the titles and abstracts of identified peer-reviewed literature published in or after 2014. The resulting citation database, with analysis of trends and illustrative summaries of abstracts for papers published since 2014, confirm that recent research reinforces the major conclusions of the Science Report, and that this report remains the authoritative standard regarding the connections, disconnections, and resulting effects between the nation's streams, rivers, lakes, estuarine systems and seas, and the focal aquatic systems summarized here: (a) ephemeral, intermittent, and perennial streams, (b) riparian or floodplain wetlands and open waters, and (c) wetlands and open waters in non-floodplain settings.

The screening process recovered a number of papers that directly addressed data gaps or needs identified in the Science Report and identified meaningful trends in the scientific literature published in or after 2014. For instance, the scientific literature published in or after 2014 on the downgradient connectivity and effects of non-floodplain wetlands and open waters has notably and markedly expanded the total available literature on these systems. Research published in or after 2014 has seen an increase in studies on connectivity and effects across multiple spatial scales, from individual systems to landscapeand watershed-scale findings reviewed here support the findings of the Science Report as well as fill data gaps. For example, 133 papers reviewed were marked as specifically focusing on a known non-floodplain wetland and/or open water and sufficiently descriptive to address the major conclusions. Of those, 99% (n=131) were reported to support the conclusions of the Science Report, noting the nutrient removal, flood-peak attenuation, base-flow maintenance, and habitat functions of non-floodplain wetlands and open waters affecting downgradient aquatic systems. These scientific advancements now provide a solid scientific foundation that builds upon the Scientific Report's conclusions on wetlands and open waters in non-floodplain wetland settings. Based on the abstracts reviewed here, and on a recent review by Lane et al. (2018) which also updated the Science Report, the literature demonstrates that non-floodplain wetlands and open waters, particularly when evaluated in the aggregate, have substantive effects on

downstream waters through hydrological, biogeochemical, and biological connectivity pathways and gradients, performing functions affecting downstream water integrity.

The Science Report concludes that all tributaries, including ephemeral and intermittent systems, are demonstrably and conclusively connected downgradient through physical, chemical, and biological pathways. Nearly 100 scientific papers published in or after 2014 specifically addressing the connectivity and effects of ephemeral and intermittent stream systems were identified through the screening process (98 papers; 28 ephemeral stream studies plus 70 ephemeral and intermittent stream studies). When sufficient information in the abstracts from which to draw conclusions was available (57 papers), 100% of the papers were noted by abstracts reviewed here to support the finding that these stream systems, individually and cumulatively, exert a demonstrative effect on downstream waters. These additional papers further underpin while continuing to advance the scientific findings specific to ephemeral and intermittent systems.

Likewise, riparian and floodplain wetlands and open waters were found in the Science Report to be integrated with and connected to downgradient waters through processes that improve water quality, recharge groundwater, and attenuate storm flows. In this analysis, 98% of 298 scientific papers published in or after 2014 that explicitly focused on floodplain wetlands and open waters were found to support the Science Report, underlining the connections between and multiple functions performed by floodplain wetlands and open waters directly affecting downgradient rivers, streams, lakes, and other larger waters.

The abstracts reviewed for this analysis included studies conducted outside of the United States. This research was reviewed to investigate the validity of scientific principles developed from studies inside the United States by comparing them to principles developed from research on similar systems in other locations. For example, studies on the Canadian Prairie Pothole Region encompass wetlands north of the Prairie Pothole Region in the United States that are part of the same ecoregion, and provide insights into the intrinsic functions of similar wetlands in the United States that are generally more drained or filled for agriculture and other purposes. In addition, research on ephemeral streams in arid regions of Australia can be informative of the functions of similar ephemeral streams in arid portions of the United States.

The sections below provide a scientific narrative on the connectivity, functions, and effects of the three aquatic systems on downgradient rivers, streams, and other larger waters that draws on findings from the screening, as well as the available scientific literature. The screeners identified papers within each of the three focal systems analyzed wherein connectivity and/or effects were illustrative; these papers were noted and a plain-text assessment of their content provided in Appendix C2. Incorporation of additional peer-reviewed scientific literature places the summary review in the context of the existing scientific knowledge on the connectivity and effects of ephemeral, intermittent, and perennial streams, floodplains and open waters, and non-floodplain wetlands and open waters on downstream waters. After contextualizing the findings, the few papers found in the screening of the titles and abstracts marked as disagreeing with the findings of the Science Report are discussed. This section concludes with a note on the benefits and limitations of this analysis.

1. Ephemeral, Intermittent, and Perennial Streams

The Science Report, as well as the peer-reviewed updated synthesis of the scientific literature by Fritz *et al.* (2018), Schofield *et al.* (2018), and Goodrich *et al.* (2018), conclude that the scientific evidence unequivocally demonstrates that streams, including ephemeral, intermittent, and perennial streams and rivers are physically, chemically, and biologically connected to downstream rivers via channels and associated alluvial deposits. The scientific evidence in the Science Report (and Fritz *et al.* 2018, Schofield *et al.* 2018, Goodrich *et al.* 2018) notes that streams, individually or cumulatively, exert a strong influence on the integrity of downstream waters, and that all tributaries, regardless of size or flow duration, are connected to and strongly influence the functioning of downstream waters. ORD's analyses of 4,194 relevant abstracts published in or after 2014 supports these findings.

The past seven years of published, peer-reviewed scientific literature have seen an increased research focus on the extent, abundance, connectivity, and effects of these flowing water systems. For instance, though Horton (1945) established that headwater streams are the most abundant components of the fluvial network, recent estimates suggest that nearly 89% of global stream longitudinal extent is comprised of headwater stream systems. Allen *et al.* 2018. In the United States, headwater stream systems represent ~50-80% of the total currently mapped conterminous United States stream length (Nadeau and Rains 2007; Hill *et al.* 2014; Colvin *et al.* 2019), certainly an underestimation of headwater stream abundance (Fesenmyer *et al.* 2021). For instance, Fritz *et al.* (2013) analyzed nine forest watersheds and determined that the high-resolution (1:24,000-scale or better) National Hydrography Dataset (NHD) depicted only 21-33% of the actual linear stream network length in seven of the nine watersheds. Fesenmyer *et al.* (2021) recently coupled the high-resolution NHD with a contributing area threshold model, concluding that 48% of stream length in the conterminous United States is likely ephemeral (43-56%, depending on flow-area characteristics).

Ephemeral, intermittent, and perennial stream networks are hydrologically connected to downstream systems, from the source area of headwaters to the flowing waters connected downgradient, to their terminus at endorheic lakes or estuarine systems. For instance, headwater streams supply the majority of flow in most river systems. Alexander *et al.* 2007; Fritz *et al.* 2018. By providing flows to higher-order systems that comprise the full watershed network, headwaters directly connect to and affect downstream waters.

Flow response in headwater streams from precipitation varies regionally, affected by transmission, evaporation, transpiration, and groundwater recharge. The headwater streamflow response to precipitation and downgradient volume contributions can be nonlinear, with substantial increases (and decreases) in flow volumes as headwater-drained contributing areas become active on the rising limb and inactive on the falling limb of the hydrograph. McGuire and McDonnell 2010; Bergstrom *et al.* 2016. For instance, van Meerveld *et al.* (2019) noted that travel times (*e.g.*, of water, material in the stream network, etc.) changed based on stream-network extension (during periods of higher precipitation) and retraction (during periods of lower precipitation). Travel times were skewed towards shorter, faster transit of dissolved and entrained materials during wetter periods as the stream network fully reconnected and longer, more uniform travel times during drier periods as the network retracted.

The spatial and temporal distribution of river network connectivity is a primary nonlinear control on the network's precipitation-runoff response. Bachmair and Weiler 2014. When water begins to flow in

headwater tributaries and longitudinally connect with other parts of the network, their confluence with larger-order streams can have profound effects on the receiving system due to the abrupt increase in water, sediment, wood, and other entrained materials. Benda *et al.* 2004a; Xu 2016. For instance, sediments from ephemeral tributaries provided important salmonid habitat in downgradient waters in a study in the United Kingdom. Marteau *et al.* 2020. Reconnecting a tributary to downgradient receiving waters increased sediment yield to the receiving water by 65%. Marteau *et al.* 2017.

As the network contracts and disconnects during hydrograph recession, streams that were temporarily higher-order systems (*e.g.*, stream order 2+, depending on flow characteristics) revert to performing headwater stream functions. Phillips *et al.* 2011. Godsey and Kirchner (2014) demonstrated the dynamism of headwater networks, mapping a 2.6 to 7.5-fold increase in both flowing network lengths and drainage densities in four drainages between fall (dry conditions) and spring (wet conditions). Surface-flow disconnected, low-order stream reaches may also function similarly to ponded or perched wetland systems (*e.g.*, Rains *et al.* 2006) until network re-wetting and subsequent expansion. For instance, Gallo *et al.* (2020) found Arizona streams in their study area – which were noted to be important groundwater recharge systems – were highly variable in their expansion and retraction. Streams supported flowing waters for 0.6-82.4% of the time, but water was present for 4-33x longer than the stream flow (from 10-301 days) likely providing habitat, a water source, aquatic refugia, and a biogeochemically active area.

The dynamic nature of headwater streams and downgradient connectivity is well-supported in the literature – headwater streams are neither spatially nor temporally invariant but rather dynamic systems that expand, contract, fragment, and reconnect across predictable spatial and temporal scales. Hewlett and Nutter 1970. The heterogeneity of dynamic flow paths creates storage and (subsequent) flow asynchronies by ephemeral, intermittent, and perennial stream systems, by delaying and attenuating downgradient storm flows and maintaining baseflows. Saco and Kumar 2002. Similarly, the variability in vulnerable water source area expansion and contraction (*i.e.*, parts of watersheds that generate overland flow) produces subsequent variability in the timing of headwater stream connectivity at the reach-scale. The connectivity response by ephemeral, intermittent, and perennial streams thereby affects the timing and magnitude of watershed-scale hydrological fluxes in non-linear ways. The convolution of these diverse vulnerable water storages and flows across watersheds imparts hydrological stability, deepening the aquatic network's resistance to drought and deluge perturbations *See, e.g.*, Chezik *et al.* 2017; Li 2019; Rupp *et al.* 2021.

Headwater streams without apparent surface flow often have complex and abundant hyporheic flow that maintains a downgradient hydrological connection, supports characteristic surface flows (Covino 2017; Magliozzi *et al.* 2018), and maintains habitats. In fact, Stanley *et al.* (1997) noted that in arid environments, streams and rivers unconstrained by valley topography may have zero surface flow for weeks to months yet flow in the hyporheic zone for thousands of meters before reconnecting with (and hence directly affecting) the surface-flowing network. Ebersole *et al.* (2015) found that even where they do not provide direct habit for salmon themselves, ephemeral streams can contribute to the habitat needs of salmon by supplying sources of cold water that these species need to survive (*i.e.*, by providing appropriate physical conditions for cold water upwelling to occur at river confluences), transporting sediment that supports fish habitat downstream, and providing and transporting food for juveniles and

adults downstream. Similarly, Kelson and Carlson (2019) determined that groundwater discharge to tributaries in California were important in supporting steelhead trout habitat, especially during dry years.

Many organisms use and connect the entirety of the stream network, including ephemeral, intermittent, and perennial reaches. Schofield *et al.* 2018. Koizumi *et al.* (2017) found that a small tributary dried during the summer, yet four months after resuming flow, >10,000 immature fish of three species (including rainbow trout) were found using the stream. Samia and Lutscher (2017) showed that upstream habitats are important refugia maintaining organisms (*i.e.*, fish), especially those with high dispersal abilities, during hydrologic disturbances (*e.g.*, droughts, floods). Upstream systems, such as headwater streams, contribute to downstream protist and rotifer community persistence and stability in the face of disturbances. Seymour *et al.* 2015. Teachey *et al.* (2019) reported that a microbial study in Georgia showed that upgradient systems repeatedly enriched (or repopulated) down-stream systems, contributing to the stability of the aquatic network. Similarly, variation in stream networks and habitat mosaics across a large, free-flowing watershed in Alaska was shown to support Pacific salmon populations by maintaining resilient natal and juvenile rearing habitat in face of climatic and other perturbations. Brennan *et al.* 2019. In some cases, headwater stream networks and dispersal dynamics increase or decrease genetic diversity, which was found by Chiu *et al.* (2020) to also depend on metapopulation (*i.e.*, a group of spatially distributed populations) characteristics of macroinvertebrates.

Headwater streams are biogeochemical reactors within hydrologic networks, transforming and sequestering materials affecting downgradient physical and chemical characteristics and concentrations along the full aquatic network. Sanford *et al.* 2007; Creed *et al.* 2015; Fritz *et al.* 2018. The full extent of the draining river network routes significant material fluxes from the terrestrial landscape into the watershed's aquatic ecosystems via overland flow and other dispersed flow paths. Sabo and Hagen 2012; Li 2019. Headwater streams affect the variability of downgradient network exports, such as carbon (*e.g.*, Creed *et al.* 2015, Senar *et al.* 2018), with effects increasing with flow magnitude, drainage density expansion, and hydrogeochemical interactions.

Concomitant with the longitudinal expansion and contraction as well as varying lateral and hillslope connectivity (e.g., Jencso et al. 2009), headwater streams function as both sinks and sources of carbon, nitrogen, dissolved organic matter, and sediment in flowing hydrologic networks. Minshall et al. 1983, Benda et al. 2004b, Creed and Beall 2009, Phillips et al. 2011. These biogeochemical functions can markedly affect downgradient metabolism, trophic states, and integrity. Creed et al. 2015; Ali and English 2019; Fovet et al. 2020. For instance, dissolved organic material varies in its carbon lability and forms a basis for energy inputs supporting stream and river metabolism. The conveyance and sequestration of heterogenous dissolved organic material by headwater streams affects the state of downgradient systems. Lynch et al. 2019. For example, Creed et al. (2015) demonstrated high spatial heterogeneity and temporal variability in dissolved organic material in headwater streams across the United States, noting that storm events affected organic-material inputs into headwater streams. These ephemeral, intermittent, and perennial streams were sources of dissolved organic material during storm events, with mineral-soil flow paths rapidly transitioning to rich organic-soil flow paths. Mooney et al. (2020) noted that smaller tributaries connected to the Great Lakes and provided disproportionately high nutrient loads (biased towards dissolved inorganic forms) than expected for their relative area. Marcarelli et al. (2019) similarly analyzed over 2,800 tributaries to Lake Superior, finding these ephemerally, intermittently, or perennially flowing waters were detectable nutrient and organic-matter sources, contributing the bulk of these

constituents to the lake during snow-melt driven flows in the winter and spring and rain-driven flows during the other times of the year. Iowa tributaries contributed excessive nitrate to the Gulf of Mexico (Jones *et al.* 2018c). Volumetric mixing and dilution suggest the biogeochemical influence of headwater streams on downgradient systems and disturbance-mitigation capacity may wane with increasing tributary contributions and stream order in the flowing network. *See, e.g.*, Vannote *et al.* 1980; Benda *et al.* 2004a; Kellman 2004; Covino 2017.

Research has shown that headwater stream systems can readily remove nitrogen (e.g., up to 50% of dissolved inorganic nitrogen; see, e.g., Cooper 1990; Ranalli and Macalady 2010), and headwater biogeochemical processing rates are most efficient at low flows. See, e.g., Alexander et al. 2007; Preston et al. 2011. Scanlon et al. (2010) noted that low-order streams "dominate the overall removal of nitrogen, primarily due to the large portion of the network that is composed of these headwater reaches." Christensen et al. (2013) reported down-stream nitrogen concentrations were primarily a function of inputs at headwater stream reaches, with stream-buffer removal effectiveness decreasing markedly with increasing stream size. Schmadel et al. (2019) reported highest nitrogen removal in headwater stream systems, though the removal efficiency varied across river networks of the Mid-Atlantic and New England. In-channel denitrification was more efficient within first-to-third order streams as they removed ~8% of the nitrogen versus 16% in first-to-fifth order streams. Alexander et al. 2007; Wollheim et al. 2008. French et al. (2020) found that smaller headwater stream systems in Alaska played a disproportionately important role in predicting stream chemistry through the river network. Similar findings were reported in a European study by Abbott et al. (2018). Shogren et al. (2019) reported the dominant spatial scale for controlling organic carbon and inorganic nutrient stream concentrations were (headwater) watersheds of 3-30 km² in area.

Biogeochemical dynamics in headwater stream systems that are longitudinally, laterally, and vertically expanding, contracting, and mixing with groundwater or hyporheic flow for thousands of meters affects downgradient systems. Covino 2017. Nutrient inputs to headwater streams are removed through multiple pathways, including abiotic and biotic processes that either sequester or transform nutrients as they move to downgradient systems. Hedin *et al.* 1998; Bernhardt *et al.* 2005. Stanley *et al.* (1997) noted that the occurrence of nitrification in headwater streams and arid river hyporheic zones stimulated microbial and algal productivity for hundreds of meters downgradient of stream (re)emergence. Hyporheic exchange flows are predominantly a function of headwater stream systems and "play a significant role in biogeochemical cycling (*e.g.*, [carbon]) and nutrient availability and transformation, ecological food webs, and habitat for diverse organisms." Magliozzi *et al.* 2018 at 6163.

It is evident that flow variability emerging from ephemeral, intermittent, and perennial stream network storage and their source areas is asynchronously connected over time and space and maintains an adaptive downgradient system, resilient to watershed-scale perturbations. *See, e.g.*, Moore *et al.* 2015; Chezik *et al.* 2017; Rupp *et al.* 2021. Watershed properties (*e.g.*, Klaus *et al.* 2015) coupled with precipitation patterning (*e.g.*, Jencso *et al.* 2009) affect headwater stream storage and flows (Ward *et al.* 2018), with downgradient implications. Ephemeral, intermittent, and perennially flowing waters create a varied mosaic of aquatic habitats that are connected over space and time, through surface, near-surface (*e.g.*, hyporheic), and groundwaters to affect downgradient stream communities. *See, e.g.*, Ebersole *et al.* 2015; Schofield *et al.* 2018; Kelson and Carlson 2019; Chiu *et al.* 2020. Alterations affecting the synchrony of timed storage and fluxes from headwater streams at the reach-scale similarly decrease

network-scale resilience to hydrologic disturbance. Headwater streams also incontrovertibly affect downgradient systems through export of food resources (Schofield *et al.* 2018) and biogeochemical processing and functioning. Headwater streams affect the delivery, timing, and concentrations (*see, e.g.*, Enanga *et al.* 2017, Senar *et al.* 2018) of entrained materials into downgradient waterways, with changes in timing and concentration amplified by the interaction between climate, drainage areas, and the receiving headwater stream systems. Mengistu *et al.* 2013a; Mengistu *et al.* 2013b.

2. Floodplain Wetlands and Open Waters

The Science Report concludes that floodplain wetlands and open waters are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality. Among other functions, the Science Report notes that these systems buffer downstream waters from pollution through biogeochemical processing and sedimentation, desynchronize floodwaters and thereby decrease flood magnitudes, and are essential components of river food webs. Our analysis of the titles and abstracts of 4,183 papers published in or after 2014 supports the findings of the Science Report.

Floodplain wetlands and open waters are integrated with streams and rivers through surface and groundwater interactions and exchanges. Surface water interactions occur when riverbanks are overtopped and the floodplain, including wetlands and open waters, is inundated. Groundwater connectivity occurs through discharge from the floodplain system to the flowing water network, as well as recharge from the flood-induced inundation events. Covino 2017. For instance, Webb et al. (2017) reported floodplain inundation occurred during only 12% of their study period yet contributed to 72-76% of the groundwater discharge (to the river network). Inundation events often provide nutrient-laden floodwaters to floodplain wetlands and open waters wherein both physical settling and biogeochemical processing occurs. Gordon et al. (2020) reviewed the literature and found that floodplains remove an average of 200 kg-N ha⁻¹ yr⁻¹ of nitrate and 21 kg-P ha⁻¹ yr⁻¹ of total or particulate phosphorus; their synthesis reported that floodplain wetlands are most effective when located within river systems with higher nutrient loads. Dwivedi et al. (2018), who analyzed Colorado River riparian areas, demonstrated that reduced zones (*i.e.*, areas low in sediment oxygen, such as wetlands) in the floodplain had 70% greater nitrate removal capacity than non-reduced zones. Scott et al. (2014) analyzed nitrate removal in the Atchafalaya River floodplain during a 2011 major flood event, noting that ~75% nitrate reduction occurred within the floodplain, reducing total nitrate delivered downgradient by 17%. A similar result was found by Macdonald et al. (2018), who noted nitrate reduction from floodplain systems as important for protecting local drinking water supplies. Gillespie et al. (2018) analyzed sedimentation, nutrient loading, and mineralization in floodplain systems of the Valley and Ridge physiographic province in the United States and found that nutrient and sediment removal in floodplain systems improved downgradient water quality. Jensen and Ford (2019) coupled high-resolution water quality data and simulation modeling to assess the physical (e.g., hydrologic, hydraulic) and biogeochemical processes affecting nitrate cycling in a confluence floodplain wetland along the Ohio River (June 2017-June 2018). Despite the wetland comprising only 0.42% of the overall watershed drainage area, 2.6% to 58.5% of the annual nitrate loads entering the wetland were removed by the wetland. Longer water residence times in the wetland and less frequent connectivity with the river (and its oxygenated waters, which decrease denitrification rates) allowed nitrate removal to occur at higher rates. The findings by Jensen and Ford (2019, p.1545) "demonstrate the significance of [wetland] connectivity [and disconnectivity] on watershed nitrate loadings to floodplain wetland soils."

Floodplain wetlands and open waters are intimately connected to riverine food webs. Rees *et al.* (2020) studied riverine food webs through isotopic analyses, noting that floodplain-derived carbon following floodplain inundation was incorporated into the riverine food webs and was measurably found for up to four months following the flood peak. Battauz *et al.* (2017) noted that floodplain open waters were often highly vegetated with floating and emergent plants serving as zooplankton sources for downgradient waters when reconnected by flood events. Zooplankton are important food web components, and over 70 different zooplankton species were found on roots and submerged parts awaiting the proper environmental cues to emerge.

Floodplain wetlands and open waters are important habitats. Pyron *et al.* (2014, p.14) sampled and examined fish assemblages in 41 floodplain open waters and wetlands in the Ohio River Basin. Their results demonstrated "that floodplain lakes in the Ohio River basin contain high species richness and are important habitats to conserve because they have the potential to act as source pools for river fish populations." Carlson *et al.* (2016) noted that backwater habitats in the Missouri River flowing through South Dakota and Nebraska provided floodwater refugia, important for maintaining fish assemblages. Martens and Connolly (2014) found seasonally disconnected side channels (*i.e.*, floodplain open waters) resulted in improved survival for juvenile salmon during periods of disconnection. Upon reconnection with the main channel, the previous cohort would rejoin the main population while new young of year salmon would move into the side channels.

Floodplain wetlands and open waters also exert significant controls on downgradient stream temperature, impacting in-stream refugia. Dick *et al.* (2018) analyzed riparian wetlands and found that in periods of high river and riparian wetland connectivity (*i.e.*, inundation), the coupled saturation and connectivity decreased the relative importance of the riparian wetland for temperature regulation. Conversely, dry periods with less river and riparian floodplain hydrologic connectivity were found to be important periods of distinctions between river water and riparian wetland temperatures (*e.g.*, lower-temperature waters were coming from the riparian wetland to the riverine system).

High river flows can create downgradient flood hazards. Flow through floodplain wetlands and open waters slows river flows, desynchronizing floodwaters and mitigating flood magnitude effects. Quin and Destouni (2018) found that watersheds comprised of approximately 15% lakes and 0.5% floodplain wetlands decreased the streamflow variability to around 10-15%, compared to areas without lakes or floodplain wetlands, which had approximately 20-25% higher streamflow variabilities due to low landscape water storages. Similarly, Fossey *et al.* (2016) incorporated floodplain wetlands into a hydrological model, reporting that floodplain moderation of high flows (and support of base flows) occurred in proportion to the frequency of floodplain connectivity. Floodwaters significantly expand the connectivity of the stream, riparian, and non-riparian wetlands. Vanderhoof *et al.* (2016) analyzed remotely sensed data, finding that surface waters connected stream networks and wetlands from 90 m to 1,400 m, depending on the (Upper Midwestern) ecoregion. Most of the stream and wetland connectivity occurred through riparian (*i.e.*, floodplain) wetlands.

Physical, chemical, and biological connectivity and effects by floodplain wetlands and open waters were found abundantly in the screened scientific literature that was reviewed. The peer-reviewed, scientific literature strongly supports the findings that floodplain wetlands and open waters are intimately connected to and affect downstream waters.

3. Non-Floodplain Wetlands and Open Waters

The Science Report concludes that wetlands and open waters located outside of riparian areas and floodplains (*i.e.*, non-floodplain wetlands and open waters), even when lacking surface water connections, provide numerous functions that can affect the integrity of downstream waters. The report further noted that the literature at that time (circa 2014) was insufficient to provide a definitive conclusion regarding the connectivity and effects of specific classes or groups of non-floodplain wetlands, with notable exceptions (for example, the Report concludes that non-floodplain wetlands situated between a pollution source and a downstream water, intercepting the [surface or near-surface] flowpath, do affect downstream waters through sink functions). However, substantive scientific advances since the publication of the Science Report have focused on the connectivity and effects of non-floodplain wetlands (*e.g.*, Creed *et al.* 2017), which recent studies conclude comprise approximately 16% of existing total freshwater-wetland areal extent in the conterminous United States. Lane and D'Amico 2016.

Based on this analysis of 4,282 scientific peer-reviewed papers published in or after 2014, it is evident that non-floodplain wetlands – individually and in the aggregate – are connected to and can affect the physical, chemical, and biological conditions and characteristics of downgradient waters (*e.g.*, streams, rivers, and lakes). As noted in an updated 2018 analysis and synthesis on the connectivity and effects of non-floodplain wetlands, Lane *et al.* (2018) stated that peer-reviewed scientific research in hydrological modeling, remote sensing analyses, field-based observations, and coupled field and remote-sensing studies were sufficiently advanced to conclude that all non-floodplain wetlands were unequivocally interconnected with stream and river networks. *See, e.g.*, Marton *et al.* 2015, Cohen *et al.* 2016, Rains *et al.* 2016, Calhoun *et al.* 2017, Creed *et al.* 2017. They further noted that connectivity of non-floodplain wetlands and open waters occurs along a gradient (*see also* Science Report at 1-4) and:

"varies in frequency, duration, magnitude, and timing [and that this] complex landscape-scale connectivity, in turn, affects water and material fluxes — the resultants of substantial hydrological, physical, and chemical functioning in NFWs [non-floodplain wetlands] — that modify the characteristics and function of downstream waters..."

Lane et al. 2018 at 363.

The findings noted in the cited literature above plus the literature we reviewed for this evidentiary summary (published in or after 2014) demonstrate that non-floodplain wetlands, particularly when analyzed in the aggregate, are connected to and can exert a substantive and important influence on the integrity of downstream waters through notable functions affecting downgradient systems including hydrological lag and storage functions (*i.e.*, affecting baseflow and stormflows/flood-hazards in stream systems) and biogeochemical functions (*i.e.*, microbial, physical, or chemical functions transforming compounds, such as denitrification, carbon mineralization, and phosphorus sequestration).

Similarly, in a 2018 peer-reviewed review paper, Schofield *et al.* (2018) provided an updated analysis of the biological connectivity and effects of non-floodplain wetlands (*e.g.*, serving as refugia, migratory "stepping-stones," resting and feeding habitats, and breeding habitats). They concluded that biota connected streams and non-floodplain wetlands, part of the landscape-scale "freshwater ecosystem mosaic," through the lateral active or passive movements of organisms and propagules. Our analysis of the current literature supports these findings. For instance, Michelson *et al.* (2018) noted that tree

swallows (Tachycineta bicolor), an avian insectivore, specialized in feeding on emergent aquatic insects in non-floodplain wetland dominated landscapes. Kappas et al. (2017) hypothesized that avian species providing passive transportation between relatively isolated non-floodplain wetland and open waters were one potential reason fairy shrimp (Streptocephalus torvicornis) were genetically panmictic, or expressed high genetic diversity suggesting abundant genetic transmission across landscapes in a European study. Likewise, Mushet et al. (2013) noted that northern leopard frogs (Lithobates pipiens) were found to use the streams, non-floodplain wetlands, and other available habitat throughout the breadth of their 68kilometer North Dakota study area, with high genetic diversity suggesting abundant population connectivity. However, distance between available wetland (and stream) habitats is important for many species (e.g., Uden et al. 2014), such as the reticulated flatwoods salamander (Ambystoma bishopi) of the Southeastern United States. Wendt et al. (2021) reported that migration between wetland habitats >400 meters was low, limiting population interactions. However, distances from streams (*i.e.*, increased aquatic "isolation") can often limit fish presence in non-floodplain open waters and wetlands. The lack of amphibian predators can positively affect certain species. For instance, Davis et al. (2017) reported that the ornate chorus frog (*Pseudacris ornata*), an "ephemeral wetland specialist" responded positively to drought conditions as fish were excluded. Coupling the hydrological and biogeochemical functions noted above (and further discussed below), it is evident that across all three connectivity and effect types (*i.e.*, physical, chemical, and biological), some benefits of non-floodplain wetlands are due to their relative isolation rather than their connections. Creed et al. 2017.

Non-floodplain wetlands are the flow-generating origins of many downgradient systems. By providing water to downgradient systems, non-floodplain wetlands maintain and affect the physical, chemical, and biological integrity of those systems. In a chloride-tracer study across 260 North American catchments, Thorslund et al. (2018) determined that non-floodplain wetlands generated surface runoff contributing to downstream systems at ~1.2 times the catchment averages (i.e., they were watershed-scale sources of water). Nearly 90% of Florida's headwaters are sourced by non-floodplain wetlands. White and Crisman 2016; Lane et al. 2018. Brooks et al. (2018) conducted an isotopic analysis of a North Dakota watershed dominated by non-floodplain wetlands and found that surface water originating in wetlands contributed significant amounts of water to the perennial stream across high- to low-flow conditions. Vanderhoof et al. (2016) reported that surface-water expansion (e.g., increased stream flows coupled with wetland filling, merging, and spilling) resulted in increased wetland and stream connections, in some cases connecting over 90-1,400 meters. Ameli and Creed (2017) modeled non-floodplain wetland interactions with draining networks in Alberta and found quantifiable contributions from non-floodplain wetlands occurred up to 30-kilometers from the stream, further indicating non-floodplain wetlands have the potential to impart substantial flow affecting downstream systems and flow-synchrony. Similarly, at the maximum expansion of the spatially variable contributing source area, non-floodplain wetlands (*i.e.*, vernal pools and swales) in the Central Valley of California were fully surface-water connected into – and hence contributing to – an integrated and hydrologically dynamic headwater drainage network, often for months (Rains et al. 2006). Likewise, Vanderhoof et al. (2017) found that spring expansion of the hydrologic network in Maryland and Delaware connected streams and depressional wetlands, increasing hydrologic interactions (and likely material exchanges) by 12-93% by area and 12-60% by count.

In contrast to their flow-generating properties, non-floodplain wetlands can also act as flowdampening systems, attenuating surface flow through storage functions and providing watershed-scale resilience to hydrologic disturbances. Rains *et al.* 2016; Cohen *et al.* 2016. The watershed-scale effects provided by surface-water and groundwater "disconnected" non-floodplain wetlands is demonstrated throughout the literature. Shaw et al. (2012) noted 61% of a studied watershed's wetlands were disconnected from overland flow paths, hence serving as so-called flow gate-keepers (*i.e.*, holding back and thereby dampening stream flow and modifying the aquatic network until a flow-threshold is crossed and flow-connections occur; Leibowitz et al. 2016). Shook et al. (2021) reported that depressional (wetland) water storage was found to control the fraction of the watershed that contributes flow to downgradient stream systems. The effects of depressions varied – when there were few extant depressions, their size and location on the landscape was most important. In systems with greater depression abundance, depressions still controlled the relationship between water storage and the fraction of the watershed contributing surface flow downgradient, but the spatial location within the watershed decreased in importance. Nasab and Chu (2020) analyzed flows in the Red River (North Dakota, South Dakota, Minnesota), finding that non-floodplain wetlands controlled flow, most dramatically in the early spring months (e.g., preventing rain-on-snow events from creating down-stream deluges). Similarly, Shook et al. (2015) reported depressions in the Canadian Prairie Pothole Region controlled precipitation and runoff (e.g., through storage and lag functions), directly altering stream flows. Nasab et al. (2017) reported gate-keeper effects of non-floodplain wetlands predominated at low flows, whereas nonfloodplain wetlands increased streamflow during high precipitation events (e.g., as the network wetted up and the watershed became increasingly connected). Yeo et al. (2019) found that non-floodplain wetlands of coastal Maryland and Delaware functioned in the aggregate to control (attenuate) streamflow and reduce flood magnitudes. Golden et al. (2016) analyzed non-floodplain wetlands in North Carolina, concluding that increased water storage associated with non-floodplain wetlands decreased streamflows. Rajib et al. (2020) modeled the effects of 455,000 non-floodplain wetlands in the Upper Mississippi River Basin, reporting that streamflow simulations showed statistically significant changes in 70% of the basins when non-floodplain wetland storages were incorporated into the model-meaning wetland storage was an important control on downstream flows. Similar findings were reported by Mekonnen et al. (2016) in the Prairie Pothole Region. Green et al. (2019) found that drained non-floodplain wetlands and open waters in Iowa could (still) store over 900 million m³ of runoff, enough to contain a one-year, 24-hour rainfall event. Evenson et al. (2018) modeled a 1.3 to 2.8-fold increase in runoff-contributing areas affecting stream flow when non-floodplain wetlands and their cumulative water-retention capacities were lost from the landscape. This increase produced higher flood peaks and greater flow velocities in modeled downgradient systems. Similarly, modeled hydrological retention in non-floodplain wetlands was found to decrease peak stream flows by 7 to 16% (Fossey and Rousseau 2016). Wang et al. (2019) analyzed the influence of small, spatially distributed surface depressions (*i.e.*, non-floodplain systems inferred to be Prairie Pothole wetlands and open waters) in North Dakota, with model results demonstrating that depressions in the aggregate retained precipitation, demonstrably preventing excessive downgradient storm flows. Ameli and Creed (2019) reported wetlands closer to streams performed greater peak-flow attenuation than distal non-floodplain wetlands, while both types regulated baseflow (*i.e.*, dampened baseflow variance).

Non-floodplain wetlands and open waters are frequently connected to their local and regional aquifers, and hence to the stream networks, through groundwater flows (Nitzsche *et al.* 2017, Neff and Rosenberry 2018). For instance, Park *et al.* (2020) found groundwater discharge (in)to studied non-floodplain wetlands in the Southeastern United States, and that the groundwater contributing area increased during drier periods (which may in effect retard stream baseflow). Like Brooks *et al.* (2018) in

the Prairie Pothole Region, Bugna et al. (2020) used isotopic analyses in Florida and demonstrated groundwater connectivity between non-floodplain wetlands and a nearby sinkhole. Lewis and Feit (2015) reported surface-water and groundwater connectivity in a regional aquifer in Florida, noting that increased groundwater withdrawals changed biogeochemical dynamics in groundwater-connected non-floodplain wetlands. Bam et al. (2020) demonstrated that not only do non-floodplain wetlands connect to and recharge local and regional groundwaters (providing water to farm and rural communities), but that ephemerally inundated non-floodplain wetlands were the dominant recharge source. Sampath et al. (2015) demonstrated that an "isolated" Michigan fen was connected to local and regional groundwater, other regional fens, as well as a nearby pond through a groundwater "pipeline." Ameli and Creed (2017) noted that non-floodplain wetlands have "fast" surface-water and slow groundwater connections and that groundwater connected non-floodplain wetlands from throughout the watershed (while storm events connected the system via surface waters). Neff and Rosenberry (2018) noted that geologic heterogeneity can either promote or prevent groundwater connectivity and synchrony between wetlands and downstream waters because of variations in bedrock properties (e.g., composition and associated permeability). McLaughlin et al. 2014 simulated the regional hydraulic effects of non-floodplain wetlands, which were likened to a capacitor – dampening the effects of hydrologic disturbances to the aquatic network by modulating surficial aquifer dynamics and buffering stream baseflow. In their study, non-floodplain wetlands functioned as groundwater sinks during wet periods and water sources during drier periods. McLaughlin et al. (2014) further emphasized "that the role these [non-floodplain wetlands] play in buffering surficial dynamics and downstream base flow is realized even where water in these systems may never physically reach downstream systems."

Non-floodplain wetlands are bioreactors (sensu Marton et al. 2015) performing important sink and transformation functions affecting downgradient waters, which is well-supported in the literature. Bernal and Mitsch 2013; Biggs et al. 2017; Cheng and Basu 2017; Creed et al. 2017; Lane et al. 2018; Leibowitz et al. 2018; Golden et al. 2019. Non-floodplain wetland biogeochemical functions emerge from the convolution of aerobic and anaerobic microbial processes (e.g., denitrification), plant uptake, physical processes (e.g., settling, photo-degradation), and residence time in the wetland. These processes are controlled, in part, by hydrologic connectivity and isolation (*i.e.*, degrees of "disconnectivity") characteristics (Cohen et al. 2016), which affect transformation, sequestration, and transport rates. Reddy and DeLaune 2008; Baron et al. 2013; Evenson et al. 2018. For instance, Senar et al. (2018) noted watershed-scale carbon dynamics were controlled by microbial biogeochemical processing within nonfloodplain wetlands and via precipitation-mediated transport to nearby headwater streams and on to downgradient systems. Excessive carbon export from non-floodplain wetlands with increasing drought and deluge cycling (allowing for carbon build up during drought followed by rapid flushing events) and temperature-mediated microbial activity could result in "brownification" (Monteith et al. 2007) of downstream waterbodies and concomitantly change aquatic metabolism (and hence affect aquatic integrity) by blocking light affecting primary productivity. Senar et al. 2018. Enanga et al. (2016; 2017) similarly noted watershed-scale nitrogen dynamics were controlled by microbial biogeochemical processing within non-floodplain wetlands and via precipitation-mediated transport to headwater streams and on to downgradient systems.

Storage, sequestration, and processing within non-floodplain wetlands and open waters are substantive. For example, Marton *et al.* (2015) reviewed the scientific literature, estimating that non-floodplain wetlands sequestered or processed 21-317 g carbon $m^{-2} yr^{-1}$, 0.01-5.0 g phosphorus $m^{-2} yr^{-1}$,

and 0.8-2.0 g nitrogen m⁻² yr⁻¹. In a synthesis of over 600 articles, Cheng and Basu (2017) determined that the first-order reaction rate constants for nitrogen and phosphorus were inversely proportional to wetland water residence times, a result that implies that >50% of the nitrogen removal across all water bodies occurs in small wetlands ($<325 \text{ m}^2$). Cohen *et al.* (2016) found that most non-floodplain wetlands were "unambiguously small," suggesting an out-sized role in landscape nutrient dynamics. Evenson et al. (2018) modeled wetland water residence times at the watershed scale, noting a 75% decrease in residence time (and hence opportunities lost for biogeochemical processing) when smaller non-floodplain wetlands were removed from the landscape. Incorporating non-floodplain wetlands into a watershed model by Golden et al. (2019) resulted in a 7% average annual decrease in the nitrate yield to downgradient systems. A recent model analyses of the nearly 500,000 km² Upper Mississippi River Basin determined cumulative restoration of ~2% of the area to non-floodplain wetlands would result in ~12% nitrate reduction (Evenson et al. 2021). Martin et al. (2019) analyzed farmed non-floodplain wetlands, reporting that their study wetlands reduced nitrate levels in 85% of the multi-day inundation events while serving as downstream phosphorus sources from phosphorus absorbed onto agricultural soils. Denver et al. (2014) found that farmed non-floodplain wetlands in Delaware and Maryland improved water quality through nutrient processing (*i.e.*, transformation and sink functions). Similarly, Flint and McDowell (2015) reported headwater non-floodplain wetlands decreased nitrate and increased dissolved organic carbon, while seasonally affecting downstream total dissolved nitrogen concentrations.

The abstracts that were reviewed provide additional evidence that non-floodplain wetlands and open waters substantively affect downgradient streams, rivers, lakes, and other aquatic systems through variable connections which support diverse functions that improve downstream waters. Non-floodplain wetlands and open waters exist along physical, biogeochemical, and biological connectivity continuums that affect downstream waters at all points along those continuums. Many of the functions of non-floodplain wetlands and open waters are most readily discerned and quantified in the aggregate (*e.g.*, flood-magnitude attenuation, excessive nutrient mitigation, groundwater recharge, support for metapopulation dynamics). Importantly, many of the functions performed by non-floodplain wetlands and open waters that affect downstream waters result from the disconnections (often hydrological, *e.g.*, surface water storage) that create and maintain conditions conducive to the performance of beneficial functions (frequently hydrological and biogeochemical, *e.g.*, mitigation of flood peak flows, sequestration of contaminants).

iv. Abstracts Noted in the Screening Process to Disagree with the Major Conclusions of the Science Report

The Science Report concluded that the scientific evidence unequivocally demonstrates that streams, including ephemeral, intermittent, and perennial streams and rivers are physically, chemically, and biologically connected to downstream rivers via channels and associated alluvial deposits. The Science Report similarly concluded that floodplain wetlands and open waters are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality. Further, the Science Report noted non-floodplain wetlands and open waters affect the integrity of downstream waters through numerous functions, including storage of floodwater; recharge of ground water that sustains river baseflow; retention and transformation of nutrients, metals, and pesticides; export of organisms or reproductive propagules to downstream waters; and habitats needed for stream species. Despite ample evidence that non-floodplain wetlands provide hydrologic, chemical, and biological functions that affect

material fluxes to downstream waters, few scientific studies explicitly addressing connections between non-floodplain wetlands and river networks had been published in the peer-reviewed literature – with notable exceptions. For example, the Science Report concluded that non-floodplain wetlands affect downstream waters when they intercept flowpaths and runoff emanating from pollution sources.

The evidence presented in the overwhelming majority of the scientific papers addressing the connectivity and effects of the three systems analyzed led ORD subject-matter experts to confirm that recent research reinforces the findings from the Science Report, and that substantive scientific advances in or after 2014 have expanded scientific knowledge regarding the mechanisms of connectivity and quantitative effects of these systems. For instance, when sufficient information was provided in the abstract to make a determination, 100% of the scientific publications specific to ephemerally and intermittently flowing stream systems screened (n=57) were found to support the Science Report that ephemeral and intermittent streams are connected to and significantly affect downgradient systems. Nonfloodplain wetlands and open waters were found to connect to and/or perform functions substantively affecting downgradient systems in 99% of the sufficiently descriptive papers screened (131 of 133). The Science Report concluded that floodplain wetlands and open waters. The literature screened similarly incontrovertibly supports that finding, with 99% (n=292 of 298) papers with floodplain wetlands and open waters types discerned marked in support of the 2015 Science Report conclusions.

Papers noted as both supporting and refuting, or outright refuting the findings were rarely found. Screeners noted six stream papers of 515 with a refuting conclusion determination (1%); of those two were noted as both supporting and refuting (Milner *et al.* 2019; Richardson 2019) and four stream papers were noted as refuting the findings (Fryirs and Gore 2014; Laughlin *et al.* 2016; Schmidt *et al.* 2018; Anderson *et al.* 2020). However, of those papers, only Richardson (2019) and Anderson *et al.* (2020) were noted as addressing a focal stream type (headwater stream and perennial stream, respectively); the other four stream papers were marked by screeners as "stream type not discernible."

Ten screened and included floodplain wetland and open water papers of 449 (2%) were noted as both supports and refutes (Azinheira *et al.* 2014; Puttock *et al.* 2017; Kasak *et al.* 2018; Wegener *et al.* 2018; Gulbin *et al.* 2019; Redder *et al.* 2021; Leuthold *et al.* 2021; n=7) refuting and cannot be discerned (Quin *et al.* 2015, Beesley *et al.* 2020), and refuting (Painter *et al.* 2015). Riverine (within-channel) systems were noted as the focal ecosystem type for Painter *et al.* (2015), Puttock *et al.* (2017); and Kasak *et al.* (2018). Quin *et al.* (2015) was noted as wetland type not discernible, whereas Gulbin *et al.* (2019) was noted as addressing multiple wetland types and/or spatial locations (*e.g.*, floodplain, riverine, non-floodplain wetland, etc.). Azinheira *et al.* (2014), Wegener *et al.* (2018), and Beesley *et al.* (2020) were noted as conducting research on both floodplain (or riparian system) and riverine (*i.e.*, within-channel) systems, while Redder *et al.* (2021) and Leuthold *et al.* (2021) addressed floodplain wetlands and open waters.

Five non-floodplain wetland and open water papers of 230 (2%) were noted as both supporting and refuting (Sullivan *et al.* 2019a, Acreman *et al.* 2021) or refuting (Quin *et al.* 2015; Arheimer and Pers 2017; Johnston and McIntyre 2019). The focal system under study was not discernible for Arheimer and Pers (2017). Both Quin *et al.* (2015) and Acreman *et al.* (2021) were marked as addressing multiple wetland types and/or spatial locations noted (*e.g.*, non-floodplain wetlands, floodplain wetlands, streams,

etc.). Only Johnston and McIntyre (2019) and Sullivan *et al.* (2019a) were noted by the screener to be papers explicitly about a known non-floodplain wetland type (*e.g.*, geographically isolated wetland, non-floodplain wetland, etc.).

1. Ephemeral, Intermittent, and Perennial Stream Systems

Examining stream abstracts, a screener reported that Schmidt et al. (2018, p.320) found subtle population genetic structure in a common fish (Leiopotherapon unicolor) that disperses widely in arid Australian river systems. However, the same species in a tropical river system was found to have small but detectable genetic differences between upstream and downstream populations. This suggests that for this species, dispersal between tributaries and the mainstem of the river itself is more limited in the tropical system, possibly due to differences in migratory patterns. Laughlin et al. (2016, p.1808) analyzed stable isotopes from the otoliths of channel catfish (Ictalurus punctatus) and blue catfish (Ictalurus furcatus) between the middle Mississippi River and tributaries. Though differences in river water constituents were found between the main channel and tributaries, otolith characteristics suggested that channel and blue catfish were "primarily recruited from the large rivers (Missouri and Mississippi)...with minimal contributions from smaller tributaries." Fryirs and Gore (2014) analyzed sediment movement between a tributary and a mainstem river. This paper was noted as refuting the Science Report findings as the tributary in this study was found to not provide any sediment because the main stem river had created a sediment block that prevented sediments from reaching the main river. Anderson et al. (2020) assessed the effects of a high-head dam along the perennially flowing Upper Mississippi River, finding similarity in upgradient and downgradient assemblages (a "supporting" statement) while also noting certain migratory species, such as skipjack herring (Alosa chrysochloris) were not present above the dam (a "refuting" statement regarding upstream/downstream connectivity).

Two stream paper abstracts were noted as both supporting and refuting the Science Report. Richardson (2019) wrote a scientific review that was marked in the screening process as both supporting and refuting the Report's findings. The abstract notes that headwaters are the (hydrologic) source of all stream networks, a supporting statement. Conversely, the screener noted that the abstract also stated that a characteristic of some headwaters was "isolation," implying a limited of lack of connectivity to downgradient systems (*i.e.*, two ends of a connectivity gradient). Lastly, Milner *et al.* (2019) analyzed macroinvertebrate assemblages of dammed and free-flowing stream tributaries in the Sierra Nevada Mountains in California, finding that macroinvertebrate diversity was higher in the tributaries on the freeflowing river (a supporting claim) but that there were no differences in macroinvertebrate diversity downgradient of tributary junctions nor differences in diversity within the dammed river (a refuting claim).

An examination of these abstracts points to these papers, which focused on very specific kinds of connections and effects (*e.g.*, of species responses to their environments), as exceptions to the general rules summarized in Major Conclusions 1, 2, and 3 of the Science Report. Their findings are consistent with the Science Report's Major Conclusion 4, that variations in the type, degree and downstream effects of connections are determined by characteristics of the physical, chemical, and biological environments and by human activities, and that these variations support different ecosystem functions. As discussed in section I.A.i above, connectivity of streams to downstream waters occurs along a continuum that can be described in terms of the frequency, duration, magnitude, timing, and rate of change of water, material,

and biotic fluxes to downstream waters. These connectivity descriptors characterize the range over which streams and wetlands vary and shift along the connectivity gradient in response to changes in natural and anthropogenic factors and, when considered in a watershed context, can be used to predict probable effects of different degrees of connectivity over time.

2. Floodplain Wetland and Open Water Systems

A screener marked that Painter et al. (2015) refuted the floodplain wetland and open water findings of the Science Report. The refutation in this case hinged on the data in the abstract demonstrating that mercury from sediments in the systems studied (beaver ponds within a riparian zone) was transformed (methylated) in the floodplain open waters and subsequently incorporated into the aquatic food web, demonstrating both chemical and biological connection. But, in this case, the connection resulted in degrading water quality (though through transformation functions affecting downgradient waters). However, this paper also supports the Report's conclusion that floodplain wetlands and waters serve as an important part of the food web for species that also utilize downstream waters. Beesley et al. (2020) and Ouin et al. (2015) were papers marked as both refute and "cannot be discerned." Beesley et al. (2020) reported that local biofilms within floodplain habitats supported fish during inundation periods (*i.e.*, fish fed in connected floodplain habitats when inundated) but articulated that large-bodied fish (specifically) of a northern Australian river were not found to transport carbon from the floodplain and return it to the river. Quin et al. (2015) conducted a statistical modeling approach analyzing the effects of wetlands and other features on downstream pollutant retention. They found that the main effects were distance the pollutant traveled before reaching the coastal waters and the presence of major lakes; wetland contributions to pollution abatement were not detected. The screener marked this paper as both refuting and "cannot be discerned" and noted that the specific wetland type (e.g., floodplain wetland and open water) was not detailed in the abstract.

Seven papers were noted during the screening process as both supporting and refuting the Science Report's major conclusions for floodplain wetlands and open waters. Azinheira et al. (2014, p.6168) modeled floodplain solute retention, finding that modeled "inset" floodplains were inundated ~1% of the year (a supporting statement regarding connectivity), yet the flow amount residence times were too short and hence this paper was also marked as refuting the major conclusions, as there was no "substantial impact on dissolved contaminants flowing downstream." Puttock et al. (2017, p.430) found riparian beaver activity substantively increased floodplain wetland and open water storage, sedimentation, nitrogen and phosphorus retention, and flow attenuation, resulting "in lower diffuse pollutant loads in water downstream." A screener noted that these riparian open waters also increased dissolved organic carbon contributions, which was perceived by the screener as deleterious to downstream water quality. Kasak et al. (2018, p.1) reported a two-year old constructed "in-stream free surface flow" wetland removed annually 14% of phosphate (improving water quality, a supporting claim) but was a source of total nitrogen (a pollutant in excess, a refuting claim). Wegener et al. (2018) analyzed nitrate dynamics in riparian areas of mountainous stream reaches in Colorado, finding that studied riparian zones were sources of nitrate (noted as a refuting claim) while one wider zone was also a nitrate sink at high flows (marked as a supporting statement). Gulbin et al. (2019) developed a hydrologic model of the endorheic Devils Lake in North Dakota, finding that given the climate drivers affecting the region wetland restoration does not change the current flood risk (perceived as a refuting statement) but restoration would provide complimentary flood-mitigation benefits under modified climate scenarios. Both Redder et al.

(2021) and Leuthold *et al.* (2021) addressed the movement of nitrogen from riparian systems to streams, indicating connectivity between floodplain wetlands and open waters. Redder *et al.* (2021) noted that groundwater seeps in riparian areas were connecting to downgradient waters (supporting phrase), providing nutrients that may end up degrading water quality (the refuting clause). Similarly, Leuthold *et al.* (2021) found riparian wetlands connected with downgradient systems through a late-winter flush of mineralized nitrogen (a supporting point regarding connectivity), a potential pollutant (refuting point).

An examination of these abstracts suggests that the results of the studies are consistent with the Science Report's finding that the floodplain wetlands and open waters function as a source, sink, and transformer of materials, including nutrients. Regardless of the perceived benefit (*e.g.*, nutrient sink) or detriment (*e.g.*, nutrient source), the connection between the floodplain and the downstream water effect remains. The Report noted that these functions are not mutually exclusive; for example, the same wetland can be both a source of one nutrient and a sink for another nutrient. The presence or absence of these functions, which depend on the biota, hydrology, and environmental conditions in a watershed and can change over time. Some of the "refuting" papers also likely support the conclusion that connectivity occurs on a continuum, as discussed in section I.C.iv.1.

3. Non-Floodplain Wetland and Open Water Systems

Three papers screened for non-floodplain wetland and open water systems were noted as refuting the major conclusions of the Science Report. Quin et al. (2015; also noted immediately above in the floodplain wetlands and open water section (section I.C.iv.2)), determined that wetland contributions to pollution retention were not detected in their study (though specific wetland type, such as a nonfloodplain wetland or open water, was not described or noted by the screeners). Arheimer and Pers (2017) conducted a study in Sweden of over 1,574 constructed wetlands, demonstrating that their effects on nutrient reductions were minor (*i.e.*, 0.2% for nitrogen and 0.5% for phosphorus); this was marked as a refuting paper, though it should be noted that wetland type was not discernable and that the reductions though minor were nonetheless found to reduce the pollutant loads to the seas. Johnston and McIntyre (2019) conducted a study on the effects of grassland-to-cropland conversion on various geospatial metrics of non-floodplain wetland (prairie pothole) connectivity in North and South Dakota. They found that wetland area across the study system decreased by 25%, wetland size decreased by ~ 0.4 ha, and wetland density decreased by 16%, largely due to splitting of large wetlands into smaller wetlands and reduction in the area of smaller wetlands. Their analysis of landscape connectivity metrics, however, found that the geospatial connectivity of remaining wetlands remained intact, and could still support metapopulation dynamics for some species (e.g., waterfowl).

Acreman *et al.* (2021) explored the effectiveness of nature-based solutions to water issues in Africa; this paper was screened and noted as both supporting and refuting the Science Report's major conclusions for non-floodplain wetlands and open waters. This review paper addressed multiple wetland types and determined that floodplain wetlands can improve water quality and reduce flood risk, whereas headwater (non-floodplain) wetlands were only found to positively affect water quality (and not affect flood risk). Sullivan *et al.* (2019a) analyzed non-floodplain wetlands that had been ditched, drained, and converted to agricultural production until they were restored; this paper was noted as both refuting and supporting the major conclusions of the Science Report. The legacy agricultural nutrients ("agrochemicals") in the wetlands were noted to be threats to downgradient waters (a refuting statement),

with wetlands closer to streams and conveyances having a greater threat potential. Wetlands further from the stream network would be expected to better retain the agrochemicals (a supporting statement).

The abstracts described in this section focus on research pertaining to two findings in the Science Report: (1) connectivity and associated effects occur along discernable gradients and (2) non-floodplain wetland functions are not mutually exclusive (*e.g.*, a wetland can improve water quality, attenuate floods, or both (*see, e.g.*, Evenson *et al.* 2018), as further discussed in sections I.C.iv.1 and 2. The Science Report found that gradients of non-floodplain connectivity can be described in terms of the frequency, duration, magnitude, timing, and rate of change of water, material, and biotic fluxes to downstream waters. As discussed in section I.A.i, these descriptors are influenced by factors such as climate, geology, and terrain, which interact with other factors such as the magnitudes of the various functions within wetlands (*e.g.*, amount of water storage or carbon export) and their proximity to downstream waters to determine where wetlands occur along the connectivity gradient. Recognizing these complexities, the Science Report noted that evaluations of individual wetlands or groups of wetlands, however, could be possible through case-specific analysis.

v. Screening Benefits and Limitations

There are two main approaches to screening scientific literature for review, that of a "brute-force" approach wherein hundreds to thousands of papers are read and summarized, and an active-screening approach used here, with machine-learning models prioritizing papers to be screened. The brute-force approach is laborious and time-consuming. However, by reading the full paper, its context as well as specifics necessary to answer important questions on the findings can often be discerned. Conversely, the active-screening approach allows for thousands of papers to be screened expeditiously, but the screening consists only of an abstract, and important contextual details are often missing.

Furthermore, while screeners were all presented with the same instructions as to include or exclude a paper and to complete the additional information section, experiential and perceptual differences between screeners exist, nonetheless. Thus, these differences may result in different characterizations of some abstracts. The screening was also conducted such that each abstract was only screened by one person rather than two or more persons screening (and then requiring unanimity to "accept" an abstract for inclusion). There was a necessary trade-off between the number of papers to be screened and measured concordance between screeners.

Due to the available time for this review and the number (>17,000) of abstracts to review, only abstracts were read. The active-screening approach calculated that screeners reviewed between 81-90% of the relevant scientific literature. But, like any review, some relevant papers were likely missed during the screening process. The screening population is established early, and papers that end up being missed by the forward-citation mapping (*e.g.*, papers that are published subsequent to the mapping process) are not presented to the screeners as those papers are simply not part of the population of papers to be screened (*e.g.*, Crabot *et al.* 2021; Evenson *et al.* 2021; Golden *et al.* 2021a). Further limitations include some assumptions that may have been made. For instance, abstracts exploring the effects of "depressions" (*e.g.*, Wang *et al.* 2019) were typically assumed to be depressional wetlands (*e.g.*, Lewis and Feit 2015). Floodplain ecosystems include wetlands and open waters, as well as non-wetland systems.
Biogeoochemical processing, such as denitrification, often requires anoxic or reduced soil conditions typically found in wetlands and open waters. Hence, in references where biogeochemical processing was noted to occur within floodplains, it was assumed to be occurring within floodplain wetlands and open waters (unless material in the abstract characterized the study differently). A complete read of each paper would likely obviate many uncertainties but was beyond the scope of this review. However, given that the scientific papers published in or after 2014 and reviewed by the team provided overwhelming support substantiating the findings and conclusions of the Science Report, the limitations of the approach are unlikely to affect the main findings reported here.

In late October 2021, ORD reviewers additionally identified another 37 scientific papers the initial effort had not captured since their review began in mid-June 2021 (*e.g.*, scientific manuscripts accepted or published since the screening process began or missed by the original screening; *see* Appendix C3 for the 37 papers). Some of these papers were identified because the ORD reviewers were co-authors of the papers, while others were found in part by reviewing scientific journal tables of contents or by Google Scholar queries.

vi. Review of Additional Literature

As part of the notice and comment process, the agencies solicited comment on the scientific literature contained in Appendix C of the Technical Support Document for the Proposed Rule and requested from the public additional scientific literature and references relevant to the Science Report's conclusions on the connectivity and effects of streams, floodplain wetlands and open waters, and non-floodplain wetlands and open waters on the chemical, physical, and biological integrity of larger downstream waters. Several commenters provided additional literature to the agencies.

The agencies reviewed those citations submitted as part of the notice and comment process along with the 37 scientific papers noted above for relevance to the report. Some of the references provided by public commenters were already included in Appendix C of the Technical Support Document for the Proposed Rule and were not reviewed because they had already been screened by the agencies, as described further in section I.C.i. Other submitted references were determined to not meet the agencies' criteria (*i.e.*, they were not published in or after 2014, they were not peer-reviewed, or the agencies could not ascertain if the references had undergone peer review). Other references were determined to not be relevant to the conclusions of the Science Report (*i.e.*, the findings of the paper were outside the context described in section I.C). The agencies have considered such references for relevance to other aspects of this Technical Support Document and have included such relevant references where appropriate (*e.g.*, where they were relevant to implementation of the final rule).

In total, the agencies reviewed and read 80 additional peer-reviewed scientific papers published in or after 2014 to determine if their findings are relevant to the conclusions of the Science Report. This included the 37 scientific papers that the agencies had identified in October 2021 (*see* Appendix C2ii) as well as 43 peer-reviewed citations published in or after 2014 that were submitted to and assessed by the agencies as part of the public comment process. Thirty papers were found to have sufficient information to draw conclusions regarding the most up-to-date and submitted literature on the findings of the Science Report. In all cases, the conclusions of the Science Report were substantiated by these scientific references. *See* Appendix C3 for additional information.

vii. Summary and Conclusions

After analyzing the available abstracts of 12,659 scientific peer-reviewed papers published in or after 2014 and additional literature published during that period (*see* section I.C.vi), the evidence reviewed is conclusive: ephemerally, intermittently, and perennially flowing streams, floodplain wetlands and open waters, and non-floodplain wetland and open waters are hydrologically, biologically, chemically, and functionally connected to downstream systems and substantively and definitively affect down-gradient aquatic systems. This conclusion echoes the findings relative to streams and floodplain systems, while updating the non-floodplain wetland and open water findings of the Science Report. ORD's review across all three systems found overwhelming evidence from the abstracts that were screened conclusively supporting the connectivity and down-gradient effects of stream, floodplain, and non-floodplain wetland and open water systems.

Furthermore, science has substantively advanced since 2014 regarding, in particular, the downstream connectivity and effects of non-floodplain wetlands and open waters, with many examples of consequential effects in the Discussion section (section I.C.iii) above. Those advances include evidence from a study by Rains et al. (2016) titled "Geographically isolated wetlands are part of the hydrological landscape" that analyzed how non-floodplain wetlands (geographically isolated wetlands, or GIWs, in their parlance) were nodes in hydrologic networks that had aquatic network-scale effects (e.g., through lag, sink, and source functions). Non-floodplain wetlands were the dominant source of groundwater recharge, replenishing groundwaters for farm and rural communities in an isotopic study by Bam et al. (2020). Isotopic analyses, a relatively recently applied tool for hydrologic studies, have conclusively demonstrated surface-water connectivity between non-floodplain wetlands and stream systems. See, e.g., Brooks et al. 2018. An additional large spatial-scale hydrological analysis of 260 non-floodplain wetland catchments across 10 study regions throughout North America by Thorslund et al. (2018) found that nonfloodplain wetlands were watershed sources to downgradient systems. Importantly, Thorslund et al. (2018) noted there was no specific relationship between landscape position (e.g., linear distance vis-à-vis a stream network) and the hydrologic connectivity of the non-floodplain wetland catchments. They reported the following:

Significant positive correlations between GIW [geographically isolated wetland] subcatchment runoff generation and distance measures...were observed in only 2 of 53 investigated cases. We therefore conclude with regard to Research Question 3 [how well can runoff generation of GIW subcatchments be predicted from simple geographic characteristics (*e.g.*, distance and elevation relative to the stream network)?] that runoff generation is poorly predicted by simple geography. This contradicts the contention that GIWs are less hydrologically connected when further away from the stream network. The absence of distance, or indeed any consistent linear predictive associations, provides support for the explanation that runoff generation controls are specific to local topography (*e.g.*, spill elevations), vegetation (impacting ET [evapotranspiration] and infiltration), and geology (impacting groundwater conveyance).

Thorslund *et al.* 2018 at 5. Golden *et al.* (2016, p. 21) found that more distal geographically isolated wetlands may be less frequently connected to downstream waters than wetlands that are closer to the stream network, but that can still have hydrologic impacts downstream ("The further GIWs are from a

stream, the greater their capacity to increase streamflow due to the physiographic setting, hypothesized transit times, and sequencing of watershed hydrologic connectivity in the study area.")

The presence of hydrologic heterogeneity in wetland connectivity – which includes varying degrees of disconnection - across the landscape is important for biological, biogeochemical, and hydrological functions cumulatively affecting downgradient systems. According to Cohen et al. (2016, p.1978), the heterogenous hydrologic connectivity of non-floodplain wetland systems (*i.e.*, the presence of a connectivity continuum) is "precisely what enhances some GIW [geographically isolated wetland] functions and enables others." Marton et al. (2015) conducted a review of the biogeochemical functions of non-floodplain wetlands, determining that non-floodplain wetlands were "biogeochemical reactors" on the landscape due to their chemical processing rates and sequestration functions that influence and effect water quality (*i.e.*, sink and transformation functions). Marton et al. (2015, p.408) note that nonfloodplain wetlands are "integral to biogeochemical processing on the landscape and therefore [to] maintaining the integrity of US waters." Cohen et al. (2016, p.1978) conclude that "sustaining landscape functions requires conserving the entire continuum of wetland connectivity, including GIWs." This is echoed by Creed et al. (2017), who also note that "vulnerable waters" such as headwater streams and nonfloodplain wetlands provide \$15.7 trillion and \$673 billion, respectively, in ecosystem services in the conterminous United States annually. These findings and others were summarized by Lane et al. (2018) in an updated and peer-reviewed state-of-the-science on the connectivity and effects of non-floodplain wetlands and open waters. They concluded (p. 363) all non-floodplain wetlands "are interconnected with streams and river networks" and that the emergent heterogeneity of those convoluted connections and disconnections affect the hydrological, biogeochemical, and physical functions of non-floodplain wetlands, "modify[ing] the characteristics and function[s] of downstream waters."

It is evident that the conclusions of the Science Report have been bolstered by scientific advances published since 2014. The science demonstrates that the aquatic systems analyzed in the Science Report and in subsequent publications are dynamically connected laterally, longitudinally, vertically, and over time with other surface waters, with groundwater, and with the landscapes in which they function. These connections exist on gradients that vary across space and time from highly connected to highly disconnected streams, wetlands, and open waters. Similarly, the functions that affect downgradient waters also occur along connectivity gradients, from functions that predominate during highly connected periods to those that occur more so at periods of lower or no connectivity. Disconnections (*i.e.*, less connected or "isolated" conditions) such as stream-network surface-flow fragmentation (*e.g.*, as can occur with ephemeral or intermittent reaches), wetland perched on clay substrates within a floodplain, or non-floodplain wetlands embedded in uplands, often provide the optimal conditions for biogeochemical, hydrological, and biological functions of streams, wetlands, and open waters that substantively affect down-gradient waters.

D. Closing Comments on the Science Report and Updated Literature

This section updates the Closing Comments from the Executive Summary of the Science Report with information from the scientific literature published since the Report's release, including the literature reviewed and discussed in section I.C.

The structure and function of downstream waters highly depend on materials—broadly defined as any physical, chemical, or biological entity—that originate outside of the downstream waters. Most of the

constituent materials in rivers, for example, originate from aquatic ecosystems located upstream in the drainage network or elsewhere in the drainage basin, and are transported to the river through flowpaths illustrated in the introduction to this report. Thus, the effects of streams, wetlands, and open waters on rivers are determined by the presence of (1) physical, chemical, or biological pathways that enable or inhibit the transport of materials and organisms to downstream waters; and (2) functions within the streams, wetlands, and open waters that alter the quantity and quality of materials and organisms transported along those pathways to downstream waters.

The strong hydrologic connectivity of river networks is apparent in the existence of stream channels that form the physical structure of the network itself. Given the evidence reviewed in the Science Report and in more recent literature, it is clear that streams and rivers are much more than a system of physical channels for efficiently conveying water and other materials downstream. The presence of physical channels, however, is a compelling line of evidence for surface-water connections from tributaries, or water bodies of other types, to downstream waters. Physical channels are defined by continuous bed-and-banks structures, which can include apparent disruptions (such as by bedrock outcrops, braided channels, flow-through wetlands) associated with changes in the material and gradient over and through which water flows. The continuation of bed and banks downgradient from such disruptions is evidence of the surface connection with the channel that is upgradient of the perceived disruption.

Although the peer-reviewed literature available at the time of the Science Report's publication (January 2015) did not provide information to categorically identify types of non-floodplain wetlands that have the types of connections or disconnections that confer important functional effects to downstream waters, the evidence did support the conclusion that non-floodplain wetlands provide these functions and that additional information (*e.g.*, from field assessments, analysis of existing or new data, reports from local resource agencies) could be used in combination with evidence from literature in case-specific analysis. The Science Report also concluded that information from emerging research on functions occurring along the gradients of connectivity observed in non-floodplain wetlands, including studies of the types identified in Section 4.5.2 of the Science Report, could close some of the existing data gaps in the near future:

Recent scientific advances in the fields of mapping, assessment, modeling, and landscape classification indicate that increasing availability of high-resolution data sets, promising new technologies for watershed-scale analyses, and methods for classifying landscape units by hydrologic behavior can facilitate and improve the accuracy of connectivity assessments. Emerging research that expands our ability to detect and monitor ecologically relevant connections at appropriate scales, metrics to accurately measure effects on downstream integrity, and management practices that apply what we already know about ecosystem function will contribute to our ability to identify waters of national importance and maintain the long-term sustainability and resiliency of valued water resources.

Science Report at ES-15.

ORD's update of relevant literature published since 2014-2015 shows that scientific understanding of the watershed functions of non-floodplain wetlands has substantively advanced in recent years. The results of recent research confirm that functions provided by these systems support water

quality and availability in streams, rivers, lakes, reservoirs, coastal waters, and aquifers and increase the resilience of communities and ecosystems to a changing climate by mitigating for effects of floods and droughts. The research described in section I.C of this document provides new insights into the factors governing the types and degrees of connectivity that confer functional effects downstream, including evidence supporting the Science Report's finding that spatial distance must be considered in context with other factors such as topography, vegetation, soils, geology and climate, to determine the magnitude of downstream effects of non-floodplain wetland systems.

E. Other Scientific Support

In preparation for this final rule, the agencies considered scientific and technical information other than the Science Report and the literature updating the Report discussed in section I.C. This includes peer-reviewed published literature, other publications, and other information that was outside the scope of the Science Report. The agencies also reviewed and considered other data and information including jurisdictional determinations, relevant agency guidance and implementation manuals, federal and state reports, letters and commentary from the SAB on past rulemaking efforts, comment letters received on previous rulemaking efforts, consultation comment letters for this rule, letters received to the pre-proposal recommendations docket, and timely comments and associated scientific literature provided as part of the public comment period. This additional technical and scientific support is cited throughout this technical support document.

F. Emerging Science

As the agencies work to finalize and then implement the rule, they will be guided by the transparent review and application of the best available science that further informs and underpins regulatory decisions and fills data gaps on connectivity and effects across stream systems, floodplain wetlands and open waters, and non-floodplain wetlands and open waters. Examples of recent advances include quantifying the probability of perennial stream flow (*e.g.*, Jaeger *et al.* 2019), quantifying streamflow responses to shifts in future climates (*e.g.*, Jaeger *et al.* 2014), mapping stream systems (*e.g.*, Allen *et al.* 2018; Hafen *et al.* 2020), isotopically analyzing material exchange between floodplain wetlands and adjoining stream networks (*e.g.*, Tetzlaff *et al.* 2014; Sánchez-Carrillo and Álvarez-Cobelas 2018), and incorporating high temporal and high spatial resolution data into coupled floodplain and stream models (*e.g.*, Hansen *et al.* 2018; Jensen and Ford 2019).

Emerging scientific advances since 2015 have continued to inform the connectivity and effects of non-floodplain wetlands and open waters, in particular when assessed at the watershed-scale. As stated in Conclusion 3 of the Science Report, the connectivity and effects of wetlands and open waters that are not hydrologically linked to other water resources by surface water or by stream channels and their lateral surface extensions into riparian areas and floodplains are more difficult to address solely from evidence available in peer-reviewed studies at the time of the report's publication. However, as discussed in section I.C, advances have been made since then that can help inform case-specific significant nexus determinations. The currently available scientific literature on non-floodplain wetlands and open waters

shows that these systems have important hydrologic, water-quality, and habitat functions that can affect the integrity of downstream waters. The Science Report noted these effects were particularly evident where (a) connections from the non-floodplain wetlands and open waters to the downstream waters exist, (b) when non-floodplain wetlands and open waters intersect flow paths from known point or non-point sources, or (c) when considered cumulatively with other non-floodplain wetlands and open waters. The current scientific literature supports these findings, and also provides additional examples of the direct effects of non-floodplain wetland and open water "isolation" on downstream water integrity, *e.g.*, through attenuating flood peaks and mitigating excessive nutrient levels. Currently available peer-reviewed literature and scientific wetland classification systems clearly document the importance of natural variation in the types and degrees of connectivity in non-floodplain wetlands and open waters, and the effects of that natural variation on the types of ecosystem functions and services such wetlands provide.

The body of peer-reviewed scientific literature regarding the watershed functions of nonfloodplain wetlands and open waters and their cumulative effects on downstream waters continues to grow since the publication of the Science Report (e.g., Marton et al. 2015; Rains et al. 2016; Cohen et al. 2016; Creed et al. 2017; Alexander et al. 2018; Mushet et al. 2019; Mengistu et al. 2020; Evenson et al. 2021; Golden et al. 2021a). Importantly, data from ongoing and emerging research not yet published in the peer-reviewed literature could close perceived data gaps in the near future. Scientific advances in the fields of mapping (e.g., Heine et al. 2004; Tiner 2011; Lang et al. 2012; Wu et al. 2014b; Wu et al. 2015; Wu and Lane 2016; Lane and D'Amico 2016; Wu and Lane 2017; Allen et al. 2018, Colvin et al. 2019; Vanderhoof and Lane 2019; Borja et al. 2020; Fesenmyer et al. 2021), assessment (e.g., McGlynn and McDonnell 2003; Gergel 2005; McGuire et al. 2005; Ver Hoef et al. 2006; Leibowitz et al. 2008; Moreno-Mateos et al. 2008; Lane and D'Amico 2010; Ver Hoef and Peterson 2010; Shook and Pomeroy 2011; Powers et al. 2012; McDonough et al. 2015; Harvey et al. 2019; Schmadel et al. 2019; Ali and English 2019; Harvey et al. 2021), modeling (e.g., Golden et al. 2013; McLaughlin et al. 2014; Fossey and Rousseau 2016; Jones et al. 2018a; Rajib et al. 2020; Evenson et al. 2021; Golden et al. 2021a), and landscape classification (e.g., Wigington et al. 2013; White and Crisman 2016; Klammler et al. 2020) indicate that increasing availability of high-resolution data sets, promising new technologies for watershed-scale analyses, and methods for classifying landscape units by hydrologic behavior can facilitate such individual and cumulative functional characterizations by broadening their scope and improving their accuracy. Id. at 6-13. Emerging research that expands the ability to detect and monitor chemically, physically, and biologically relevant connections at appropriate scales, metrics to accurately measure effects on downstream integrity, and management practices that apply what is already known about non-floodplain wetlands and open water functioning, will contribute to advance the ability to maintain the long-term sustainability and resiliency of valued water resources. Scientific inventories of wetlands and wetland functions or ecosystem systems are likely to continue to expand the understanding of the benefits non-floodplain wetland and open water ecosystem functions and services provide to humans and the environment.

G. SAB Review of the Proposed Rule

The agencies also engaged with the SAB on several occasions during the development of this rule. As discussed in section I.B., the SAB was established in 1978 to provide independent scientific and technical advice to the EPA Administrator on the technical basis for agency positions and regulations.

On January 28, 2022, during the public comment period, the agencies met with the SAB Work Group for Review of Science Supporting EPA Decisions to explain the proposed rule, including its basis, and to address the SAB Work Group's initial questions. On February 7, 2022, the SAB Work Group signed a memorandum recommending that the Chartered SAB should review the adequacy of the science supporting the proposed rule. SAB 2022a. On March 7, 2022, during the public meeting of the Chartered SAB, the Chartered SAB unanimously voted to review the scientific and technical basis of the proposed rule. The SAB has similarly reviewed the technical and scientific basis of other past rulemakings revising the definition of "waters of the United States." *See* SAB 2014b; SAB 2020. The SAB formed a Work Group of its chartered members which issued a draft review on May 9, 2022, and the Chartered SAB held a public meeting on the matter on May 31 and June 2, 2022. The SAB issued their final review on July 5, 2022. SAB 2022b (hereafter 2022 SAB Review). All materials related to the 2022 SAB Review are available in the docket for this rule and on the SAB's website.

The SAB's review of the proposed rule was overall supportive of the science underpinning the proposed rule, including the Technical Support Document, and the discussion of shallow subsurface flow. The SAB made some recommendations on the discussion of climate change. The SAB's recommendations relevant to the final rule were also raised during the public comment period and have been considered by the agencies during their drafting of this Technical Support Document. A memorandum summarizing the agencies' interactions with the SAB and the SAB's review of the proposed rule and its supporting documents is available in the docket for this rule.

H. Other Scientific Information

i. Ecosystem Services

Streams, wetlands, lakes, ponds, and other types of aquatic resources are well-known to provide a variety of functions that translate into ecosystem services. *See, e.g.*, Creed *et al.* 2017. Ecosystem services are benefits that humans obtain from ecosystems, including provisioning, regulating, cultural, and supporting ecosystem services. Millennium Ecosystem Assessment 2005. Provisioning services relate to the food, water, timber, fiber, and other resources provided by wetlands and other aquatic resources that are consumed. *Id.* Regulating services affect climate, floods, disease, wastes, and water quality. *Id.* Cultural services include all non-material benefits obtained from ecosystems and can include recreational, aesthetic, educational, and spiritual benefits. *Id.* Supporting services are necessary for the production of other ecosystem services and include nutrient cycling, photosynthesis, and soil formation. *Id.*

Wetlands are recognized as one of the most valuable ecosystems in the planet. Costanza *et al.* 1997; Mitsch *et al.* 2015. For example, wetlands provide a wide range of ecosystem services that are directly used or appreciated by humans. *See, e.g.*, Brander *et al.* 2012a; Brander *et al.* 2012b; Chaikumbung *et al.* 2016; De Groot *et al.* 2018; Ghermandi *et al.* 2010; McLaughlin and Cohen 2013; Mitsch and Gosselink 2007; Mitsch *et al.* 2015. Provisioning services provided by wetlands include the maintenance of fisheries and wildlife for consumption, the production of rice for food, fuel sources (such as peat), medicines and pharmaceuticals derived from wetland plants and animals, ornamental resources (*e.g.*, animal and plant products used as ornaments or for landscaping), and surface and groundwater supply. *See, e.g.*, Millennium Ecosystem Assessment 2005. Regulating services include flood protection (Ameli and Creed 2019; Evenson *et al.* 2018; Lawrence *et al.* 2019; Martinez-Martinez *et al.* 2014; Tang *et al.* 2020; Taylor *et al.* 2022; Watson 2016; Wu *et al.* 2008), water purification (Ewel 1997; Ghermandi *et al.* 2010), erosion control/sediment retention (Hopkins *et al.* 2018; Richardson *et al.* 2011),

groundwater recharge (Cowdery *et al.* 2019; Harvey *et al.* 2004; Williams *et al.* 2015), carbon sequestration (Nag *et al.* 2017; Nahlik *et al.* 2016; Mitsch *et al.* 2013; Tangen *et al.* 2020), and natural hazard regulation (*e.g.*, the role coastal wetlands play in reducing the damage of hurricanes) (Millennium Ecosystem Assessment 2005). Some of these services have been monetarily valued, as discussed further in the Economic Analysis. For example, Lawrence *et al.* (2019) and Watson (2016) both provide examples of monetized flood protection benefits via avoided property damages. Additionally, Hopkins *et al.* (2018) provides an example of monetized sediment-bound nitrogen retention benefits. Cultural services provided by wetlands include cultural diversity; spiritual and religious values; educational values; inspiration for art, folklore, national symbols, architecture, and advertising; aesthetic values (*e.g.*, parks, scenic viewpoints); social relations; sense of place; cultural heritage values (including culturally significant species); and recreation and ecotourism. Millennium Ecosystem Assessment 2005. Supporting services provided by wetlands include soil formation, photosynthesis in wetland plants, primary production, nutrient cycling, and water cycling. *Id.* Some ecosystem services like erosion control can be categorized as both a supporting and a regulating service. *Id.*

Similarly, streams, including headwater streams, provide many ecosystem services to society. This includes water supply, water quality benefits (for example, via nitrogen transformation and phosphorus sequestration), and climate regulation (for example, via carbon sequestration). Creed *et al.* 2017. Streams provide of the same provisioning, regulating, cultural, and supporting services that wetlands provide to society.

Ephemeral streams and their associated wetlands, wetlands that did not meet the 2020 NWPR's revised adjacency criteria, and other aquatic resources not protected by the 2020 NWPR provide numerous and critical ecosystem services, as discussed in the Science Report. *See also* Sullivan *et al.* 2020 ("Removal of federal protection [of ephemeral streams and non-floodplain wetlands] is likely to diminish numerous ecosystem services, such as safeguarding water quality and quantity, reducing or mitigating flood risk, conserving biodiversity, and maintaining recreationally and commercially valuable fisheries"). This is echoed by Creed *et al.* (2017), who also note that "vulnerable waters" such as headwater streams and non-floodplain wetlands provide \$15.7 trillion and \$673 billion, respectively, in ecosystem services in the conterminous United States annually. These findings and others were summarized by Lane *et al.* (2018) in an updated and peer-reviewed state-of-the-science on the connectivity and effects of non-floodplain wetlands and open waters.

As discussed further in section III.E.iv, a significant nexus analysis is limited to an assessment of only those functions that have a nexus to the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. Therefore, there are some important functions provided by wetlands, tributaries, and waters evaluated under paragraph (a)(5) that translate into ecosystem services that benefit society that will not assessed in a significant nexus analysis under the final rule because they do not relate to the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, and interstate waters. There are also a wide variety of functions that streams, wetlands, and open waters provide that translate into ecosystem services that benefit society that would not be assessed in a significant nexus analysis under this rule. These include provision of areas for personal enjoyment (*e.g.*, fishing, hunting, boating, and birdwatching areas), ceremonial or religious uses, production of fuel, forage, and fibers, extraction of materials (*e.g.*, biofuels, food, such as shellfish,

vegetables, seeds, nuts, rice), plants for clothes and other materials, and medical compounds from wetland and aquatic plants or animals. While these types of ecosystem services can contribute to the economy, they are not relevant to the chemical, physical, or biological integrity of paragraph (a)(1) waters and would not be considered in a significant nexus analysis under this rule. *See also* section III.E.iv for a discussion of functions that can be considered as part of the significant nexus analysis.

The Economic Analysis for the Final Rule also discusses ecosystem services, with a focus on the monetized benefits of ecosystem services provided by wetland areas protected due to Clean Water Action section 404 mitigation requirements.

II. Executive Order 13990 and Review of the Navigable Waters Protection Rule

A. Executive Order 13990

On January 20, 2021, President Joseph R. Biden, Jr. signed Executive Order 13990, entitled "Executive Order on Protecting Public Health and the Environment and Restoring Science to Tackle the Climate Crisis," which provides that "[i]t is, therefore, the policy of my Administration to listen to the science; to improve public health and protect our environment; to ensure access to clean air and water; to limit exposure to dangerous chemicals and pesticides; to hold polluters accountable, including those who disproportionately harm communities of color and low-income communities; to reduce greenhouse gas emissions; to bolster resilience to the impacts of climate change; to restore and expand our national treasures and monuments; and to prioritize both environmental justice and the creation of the well-paying union jobs necessary to deliver on these goals." 86 FR 7037 (published January 25, 2021, signed January 20, 2021). The order "directs all executive departments and agencies (agencies) to immediately review and, as appropriate and consistent with applicable law, take action to address the promulgation of Federal regulations and other actions during the last 4 years that conflict with these important national objectives, and to immediately commence work to confront the climate crisis." Id. at section 2(a). "For any such actions identified by the agencies, the heads of agencies shall, as appropriate and consistent with applicable law, consider suspending, revising, or rescinding the agency actions." Id. The order also specifically revoked Executive Order 13778 of February 28, 2017 (Restoring the Rule of Law, Federalism, and Economic Growth by Reviewing the "Waters of the United States" Rule), which had initiated development of the Navigable Waters Protection Rule (2020 NWPR).

After completing the review mandated by the Executive Order and reconsidering the record for the 2020 NWPR, on June 9, 2021, the agencies announced their intention to revise or replace the rule. The agencies' decision was based on consideration of the text of the Clean Water Act; Congressional intent and the objective of the Clean Water Act; Supreme Court precedent; the current and future harms to the chemical, physical, and biological integrity of the nation's waters due to the 2020 NWPR; concerns raised by stakeholders about the 2020 NWPR, including implementation-related issues; the principles outlined in the Executive Order; and issues raised in ongoing litigation challenging the 2020 NWPR. EPA and the Army concluded that the 2020 NWPR did not appropriately consider the effect of the revised definition of

"waters of the United States" on the integrity of the nation's waters, and that the rule threatened the loss or degradation of waters critical to the protection of traditional navigable waters, among other concerns.

The agencies' review of the 2020 NWPR consistent with Executive Order 13990, as well as consideration some of the other directives of the Order are discussed in the sections below. The agencies also believe that they have fulfilled the Order's directive to "listen to the science," as is appropriate, and their consideration of the science is discussed in more detail in sections I and III.

B. Review of the 2020 NWPR

Pursuant to the direction in Executive Order 13990, the agencies reassessed the administrative record and basis for the 2020 NWPR and have a number of serious concerns about the 2020 NWPR. The agencies are concerned about the rule's failure to consider the statutory objective in determining the scope of "waters of the United States," including through its elimination of the significant nexus standard and the absence of any alternative standard that would protect the chemical, physical, and biological integrity of the nation's waters. The 2020 NWPR is also inconsistent with scientific information about protecting water quality, and indeed, it drastically reduced the numbers of waters protected by the Clean Water Act, including waters that significantly affect the integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. Finally, implementing the 2020 NWPR posed significant technical challenges for federal, state, and tribal agency staff as well as stakeholders because foundational concepts of the rule are confusing and not reasonably implementable, resulting in arbitrary outcomes.

i. Impacts of the 2020 NWPR

The failure of the 2020 NWPR to advance the objective of the Clean Water Act, as well as its inconsistency with science and the challenges it presents in implementation, have had real-world consequences. The agencies have found that substantially fewer waters are protected by the Clean Water Act under the 2020 NWPR compared to previous rules and practices. It is important to note that the definition of "waters of the United States" affects most Clean Water Act programs designed to restore and maintain water quality—including not only the section 402 NPDES and section 404 dredged and fill permitting programs, but water quality standards under section 303, identification of impaired waters and total maximum daily loads under section 303, section 311 oil spill prevention, preparedness, and response programs, and the section 401 Tribal and State water quality certification programs—because the Clean Water Act provisions establishing such programs use the term "navigable waters" or "waters of the United States." While the 2020 NWPR was promulgated with the expressed intent to decrease the scope of federal jurisdiction, the agencies now are concerned that the actual decrease in water resource protections was more pronounced than the qualitative predictions in the 2020 NWPR preamble and supporting documents anticipated and acknowledged to the public. These data support the agencies' conclusion that the 2020 NWPR is not a suitable alternative to the final rule.

Through an evaluation of jurisdictional determinations completed by the Corps between 2016 and 2021,⁷ EPA and the Army have identified consistent indicators of a substantial reduction in waters protected under the Clean Water Act by the 2020 NWPR. See section II.B.i for discussion on methods and results of the agencies' analyses. These indicators include an increase in the number and proportion of jurisdictional determinations completed where aquatic resources were found to be non-jurisdictional, an increase in determinations made by the Corps that no Clean Water Act section 404 permit is required for specific projects, and an increase in requests for the Corps to complete approved jurisdictional determinations (AJDs), rather than preliminary jurisdictional determinations (PJDs) which treat a feature as jurisdictional. These trends all reflect the narrow scope of jurisdiction in the 2020 NWPR's definitions. Additionally, the agencies find that these indicators likely account for only a fraction of the 2020 NWPR's impacts, because many project proponents did not seek any form of jurisdictional determination for waters that the 2020 NWPR categorically excluded, such as ephemeral streams, and the Corps would not have knowledge of or ability to track such projects. A closer look at each of these indicators will help demonstrate some of the more pronounced impacts of the 2020 NWPR on traditional navigable waters, the territorial seas, and interstate waters than were identified for the public in the 2020 NWPR and its supporting documents. As explained in detail above and in the final rule's preamble, when a water falls outside the scope of the Clean Water Act, that means, among other things, that no federal water quality standards will be established, and no federal permit will be required to control the discharge of pollutants, including dredged or fill material, into such waters unless the pollutants reach jurisdictional waters. And since many entities did not believe that they would need to seek a jurisdictional determination under the 2020 NWPR, it is impossible to fully understand the scope of degradation the 2020 NWPR's definition caused to traditional navigable waters, the territorial seas, and interstate waters.

1. Review of Jurisdictional Determinations and Permit Data

Consistent with Executive Order 13390, EPA and Army staff conducted four assessments on the effects of the 2020 NWPR on jurisdictional determinations and related individual aquatic resources using data sourced from the Corps' internal regulatory management database, Operation and Maintenance Business Information Link, Regulatory Module (referred to as the ORM2 database). *See supra* note 7. The aim of these assessments is to identify any noticeable trends relating to jurisdictional findings under

⁷ A jurisdictional determination is a written Corps determination that a water is subject to regulatory jurisdiction under section 404 of the Clean Water Act (33 U.S.C. 1344) or a written determination that a waterbody is subject to regulatory jurisdiction under section 9 or 10 of the Rivers and Harbors Act of 1899 (33 U.S.C. 401 et seq.). Jurisdictional determinations are identified as either preliminary or approved, and both types are recorded in determinations through an internal regulatory management database, called Operation and Maintenance Business Information Link, Regulatory Module (ORM2). This database documents Department of the Army authorizations under Clean Water Act section 404 and Rivers and Harbors Act section 10, including permit application processing and jurisdictional determinations. This database does not include aquatic resources that are not associated with a jurisdictional determination or that are not associated with alternatives to jurisdictional determinations (such as delineation concurrences or "No jurisdictional determination required" findings, where the Corps finds that a jurisdictional determination is not needed for a project), or permit request or resource impacts that are not associated with a Corps permit or enforcement action. An approved jurisdictional determination (AJD) is an official Corps document stating the presence or absence of "waters of the United States" on a parcel or a written statement and map identifying the limits of "waters of the United States" on a parcel. A preliminary jurisdictional determination (PJD) is a non-binding written indication that there may be "waters of the United States" on a parcel; an applicant can elect to use a PJD to voluntarily waive or set aside questions regarding Clean Water Act jurisdiction over a particular site and thus move forward assuming all waters will be treated as jurisdictional without making a formal determination.

the 2020 NWPR compared to prior regulatory practice—in particular, the implementation of 1986 definitions⁸ of "waters of the United States" consistent with relevant caselaw and associated guidance documents.⁹ This implementation will be referred to as pre-2015 practice for brevity, even though the regulatory regime was in place after 2015 (both in light of litigation surrounding the 2015 Clean Water Rule and due to finalization of the 2019 Repeal Rule, which intended to restore pre-2015 practice).

Trends in jurisdictional findings are assessed nationally and in two states within the arid West – Arizona and New Mexico. The arid West is assessed in order to determine if the 2020 NWPR is having geographically unequal impacts across the nation. A further analysis of trends in number of actions that did not require Clean Water Act section 404 permits due to definitions of "waters of the United States" is carried out at the national scale.

The four assessments use the following metrics:

- Total number of approved jurisdictional determinations (AJDs) and preliminary jurisdictional determinations (PJDs) by given time period (*see* Methods).
 - The above metric was further broken down by total number of AJDs that included jurisdictional and non-jurisdictional determinations.¹⁰
- Total number of individual aquatic resources found to be jurisdictional and non-jurisdictional within AJDs by given time period.
 - The above metric was further broken down for individual aquatic resource types found to be jurisdictional and non-jurisdictional under the 2020 NWPR in particular, wetlands and streams are enumerated.¹¹
- Total number of individual aquatic resources tied to AJDs, PJDs, delineation concurrences and findings of no JD required in two states in the arid West, Arizona and New Mexico, that were found to be jurisdictional and non-jurisdictional. This was further broken down by stream resources.¹²

⁸ EPA and the Corps have separate regulations defining the statutory term "waters of the United States," but their interpretations were substantially similar and remained largely unchanged between 1977 and 2015. *See, e.g.*, 42 FR 37122, 37144 (July 19, 1977); 44 FR 32854, 32901 (June 7, 1979). For convenience, in this document and in the preamble the agencies will generally cite the Corps' longstanding regulations and will refer to them as "the 1986 regulations," "the pre-2015 regulations," or "the regulations in place until 2015." These references are inclusive of EPA's comparable regulations that were recodified in 1988 and of the exclusion for prior converted cropland, which both agencies added in 1993.

⁹ As implemented in the time period under these assessments, the 1986 regulations were bolstered by multiple memorandums and guidance documents, including guidance related to Supreme Court decisions. This included the *Rapanos* Guidance. *See* U.S. EPA and U.S. Army Corps of Engineers. Clean Water Act Jurisdiction Following the U.S. Supreme Court's Decision in *Rapanos v. United States* and *Carabell v. United States* (Dec. 2, 2008) ("*Rapanos* Guidance"), available at https://www.epa.gov/sites/default/files/2016-

^{02/}documents/cwa_jurisdiction_following_rapanos120208.pdf.

¹⁰ The 2020 NWPR AJD data entry in ORM2 allows for and is often used to compile determinations about both jurisdictional and non-jurisdictional aquatic resources together for a single project site; under prior regulatory regimes, data entry in ORM2 restricted project managers to entering AJDs in separate entries for jurisdictional and non-jurisdictional resources on the same project site.

¹¹ Individual aquatic resources were only assessed under the 2020 NWPR because jurisdictional determinations carried out under prior regimes had less clear differentiation between types of aquatic resources. For example, a lake under prior regimes could have been classified as a tributary, an impoundment, a traditional navigable water, an interstate water, and sometimes even an adjacent water or adjacent wetland.

¹² Arizona and New Mexico were assessed because the ecosystems in these states are predominantly desert.

• Total number of projects that resulted in 'No Permit Required' closure methods.

The ORM2 database was deployed to all of the Corps' 38 districts in 2008 and has been continuously improving since that time. Because of changes to regulatory practice and tracking priorities, the data are most reliable from the year 2016 to present.¹³

a. Background

The Operation and Maintenance Business Information Link, Regulatory Module (ORM2) is the Corps' internal database that documents Clean Water Act section 404 application and permit data, including information on jurisdictional determinations (JDs).¹⁴ JDs are identified as either preliminary or approved, and both types are recorded in ORM2. An AJD is an official Corps document stating the presence or absence of "waters of the United States" on a parcel or a written statement and map identifying the limits of "waters of the United States" on a parcel. A PJD is a non-binding written indication that there may be "waters of the United States" on a parcel; an applicant can elect to use a PJD to voluntarily waive or set aside questions regarding Clean Water Act jurisdiction over a particular site and thus move forward assuming all waters will be treated as jurisdictional without making a formal determination.¹⁵

b. Methods

In the ORM2 database, an AJD can contain one or multiple aquatic resources. For this reason, the agencies assessed data on the AJD-level and at the individual aquatic resource level (*i.e.*, total number of individual aquatic resources). Similarly, PJD data was assessed at both the PJD-level and at the aquatic resource level, although information on the type of aquatic resource is not defined for PJDs. For arid west (Arizona and New Mexico) data, delineation concurrences and findings of "no JD required" were also included to give scale to the total number of resources considered in those states. Delineation concurrences and findings of "no JD required" are excluded from the national assessments due to differences in application by Corps staff at the district level and potential error in how resources were denoted when these methods were used. Additionally, a separate permitting dataset was provided by the Corps, which provided information on when AJDs and PJDs had permits associated with them and details on projects that are deemed to not require permits.

(i) Data Quality Assurance and Control

2020 NWPR AJD Data from ORM2 was refined to account for foundational differences in how AJD information is reported under the 2020 NWPR compared to prior regulatory regimes. Because a

¹³ ORM2 was not created for the purposes of assessing trends in jurisdiction; rather, it exists as a project management tool for the Corps. As such, the tool has updated over time, with changes to how data is entered into the system. Database rules associated with required field entries changed in 2016 which made data more consistent.
¹⁴ The public interface for the Corps' ORM2 Database is available at: https://permits.ops.usace.army.mil/orm-public.

¹⁵ When the Corps provides a PJD, or authorizes an activity through a general or individual permit relying on a PJD, the Corps is not making a legally binding determination of any type regarding whether jurisdiction exists over the particular aquatic resource in question even though the applicant or project proponent proceeds as though the resource were jurisdictional. A PJD is "preliminary" in the sense that a recipient of a PJD can later request and obtain an AJD if that becomes necessary or appropriate during the permit process or during the administrative appeal process. *See* 33 CFR 331.2.

single AJD in ORM under the 2020 NWPR can contain both jurisdictional and non-jurisdictional determinations, the instances of these "mixed" AJD forms had to be split in two.¹⁶ To explain, when totaling whether an AJD was for a jurisdictional or non-jurisdictional resource, if an AJD under the 2020 NWPR contained both jurisdictional and non-jurisdictional resources, it was counted in both categories (*i.e.*, a tally would be added under the jurisdictional category and the non-jurisdictional category). This refinement was made on 1,631 AJDs and thus normalized the 2020 NWPR AJDs so that it could be compared to AJDs conducted under the prior regulatory regimes. Additionally, any AJDs that were conducted on drylands/uplands or Rivers and Harbors Act section 10 waters only were excluded from this analysis, as they are either excluded from the definition of "waters of the United States" or do not fall under the joint jurisdiction of the EPA and Corps under the 2020 NWPR timeframe and a range of 1,163 to 2,479 AJDs being excluded from these analyses under the pre-2015 practice timeframes analyzed.¹⁷ If and when PJD, delineation concurrence and "no JD required" data were used, data that was found to be erroneous were removed.¹⁸

The agencies also assessed actions from 2016 to 2021 associated with the Corps' "No Permit Required" closure method within ORM2, looking specifically at closure methods for "Activities that occur in waters that are no longer WOTUS under the 2020 NWPR" and "Activities that do not occur in WOTUS." "Activity does NOT occur in WOTUS" relates to a finding of "no permit required" that is based on that activity occurring entirely outside of any "water of the United States." While there may be aquatic resources on the site, the activity will not touch those resources. "Activities that occur in waters that are no longer WOTUS under the 2020 NWPR" is a new closure method created by the Corps for the ORM2 database following implementation of the 2020 NWPR that helps track actions that were pending or in progress when the 2020 NWPR came into effect and would have required a permit prior to the 2020 NWPR, but that no longer required a permit due to the 2020 NWPR's revised definition of "waters of the United States and therefore needed to be denoted as not requiring a permit in ORM2. However, as implementation period, this closure method was not uniformly used across the Districts and by Corps project managers and thus likely undercounts the number of projects that would have required a permit prior to the 2020 NWPR but that no longer did while the 2020 NWPR was implemented.¹⁹

Time frames considered

¹⁶ Under the pre-2015 regulatory regime and the 2015 Clean Water Rule, AJDs in ORM could contain only jurisdictional features or only non-jurisdictional features.

¹⁷ June 22, 2016 – June 21, 2017: 2,479 AJDs removed; June 22, 2017 – June 21, 2018: 2,112 AJDs removed; December 23, 2019- June 21, 2020: 1,163 AJDs removed.

¹⁸ Features associated with PJDs and these delineation alternatives are presumed jurisdictional; however, there were certain aquatic resources that were denoted as being non-jurisdictional in this data; thus, they were removed. This was an insubstantial amount of data removed.

¹⁹ This closure method did not exist until July 20, 2020, a month after the 2020 NWPR was implemented. This closure method also can only be used when all resources in a given AJD are found to be non-jurisdictional. Under the 2020 NWPR, in ORM2, AJD entries can have both non-jurisdictional and jurisdictional individual aquatic resources; in those cases, the permit(s) associated with the jurisdictional resources would prevent the 'no permit required' closure method from being selected. Finally, the use of this new closure method could simply have been overlooked by staff if they had not gone through training on its use.

In comparing overall AJD and PJD trends at the national scale, the following timeframes were used:

- June 22, 2020 to June 21, 2021 for the 2020 NWPR: one year of data associated with implementation of the 2020 NWPR;
- The three most recent periods (broken into years with the same calendar dates as feasible) in which the definition of "waters of the United States" under the 1986 regulations, implemented consistent with relevant caselaw and associated guidance documents, were in use nationwide.
 - June 22, 2016 to June 21, 2017
 - o June 22, 2017 to June 21, 2018
 - December 23, 2019 to June 21, 2020^{20}

In order to assess significance of changes under the 2020 NWPR, a further assessment was carried out on data separated by years using the calendar dates of June 22 to June 21 for the years of 2016 to 2021. These significance tests were carried out using pre-2015 practice data compared to the 2020 NWPR, and using both pre-2015 practice and 2015 Clean Water Rule data compared to the 2020 NWPR (Section II.B.i.b.ii: Statistics).

In looking at trends in Arizona and New Mexico and looking at the "No Permit Required" analysis, year by year comparisons were carried out during the calendar dates of June 22 to June 21 for the years of 2016 to 2021. In Arizona and New Mexico, all determinations data were derived from either pre-2015 practice or the 2020 NWPR (no determinations were carried out under the Clean Water Rule in these states). The "No Permit Required" analysis uses data from all regulatory regimes (including pre-2015 practice and the 2015 Clean Water Rule regime) in order to capture the volume of these findings over time.

(ii) Statistics

Because data within ORM2 are imperfect in nature—due to varying regulatory regimes, spatial distribution of determinations, and economic and development trends, and general human error related to data entry—the assessment carried out is summary in nature. Cursory statistics on significance have been run through comparing the 2020 NWPR data to the 95% confidence interval calculated via annual data from 2016-2020. This was done for the following comparisons:

- Proportion of AJDs and PJDs carried out under the 2020 NWPR compared to pre-2015 practice;
- Proportion of non-jurisdictional findings for total number of individual resources via AJDs under the 2020 NWPR compared to AJDs under pre-2015 practice from 2016 to 2020 (using only pre-2015 practice data limits the assessment to the proportion of resources being found to be jurisdictional or non-jurisdictional);
- Proportion and total number of non-jurisdictional findings for total number of individual resources via AJDs under the 2020 NWPR compared to all AJDs from 2016 to 2020 (using both pre-2015 practice and 2015 Clean Water Rule data allows for assessing the 95% confidence interval based on overall volume (*i.e.*, total number by year) as well as proportion of resources being found to be jurisdictional or non-jurisdictional);

²⁰ December 23, 2019, was the day in which the 2019 Rule went into effect, which recodified the 1986 regulations.

- Total number of AJD individual aquatic resources and PJD and delineation alternative²¹ individual aquatic resources (with a further non-jurisdictional breakdown by both proportion and total number) in New Mexico and Arizona under the 2020 NWPR compared to pre-2015 practice from 2016 to 2020; and
- Total number of No Permit Required findings under the 2020 NWPR compared to pre-2015 practice.

At the national scale, rather than comparing total numbers between different time periods, it is more telling to compare percentages. This allows for a comparison between the 2020 NWPR and pre-2015 practice, even when time frames considered are not equal in length.

One test of significance based on total numbers of individual aquatic resources was carried out at the national scale which included the 2015 Clean Water Rule data in order to compare volume of resources assessed over time. For the Arizona and New Mexico analysis and the "No Permit Required" analysis, total numbers of findings are compared on a year-by-year basis. While there are external trends that can impact the overall quantity of AJDs and individual aquatic resources reviewed under AJDs, year by year comparisons of total numbers in these specific data show a substantial shift under the 2020 NWPR that the agencies believe are tied directly to definitions of "waters of the United States."

For all analyses, while exact numbers are not obtainable from the data there is more than sufficient volume and accuracy of the data to demonstrate clear trends.

c. Results and Discussion

(i) AJDs and PJDs over time

Rather than comparing jurisdictional determinations under the 2020 NWPR to all prior jurisdictional determinations, the analyses here looked specifically at the pre-2020 NWPR determinations defined by the 1986 regulations and associated guidance documents that were issued following Supreme Court decisions (*i.e.*, under pre-2015 practice).

Of the 14,143 jurisdictional determinations carried out as either AJDs or PJDs under the 2020 NWPR from June 22, 2020, to June 21, 2021, 66.5% were carried out as AJDs (Table 1, Figure 1).²² Under the various timeframes considered for pre-2015 practice, the percentage of determinations carried out as AJDs ranged from 15.6% to 23.4%. In comparing the two regimes, there has been a 183% to 326% increase in the percent of determinations carried out as AJDs under the 2020 NWPR. Proportionally fewer PJDs indicate that fewer project proponents are requesting that aquatic resources on their project site be treated as if they are jurisdictional. This has two implications: under the 2020 NWPR, project proponents were requesting AJDs rather than PJDs and/or they were simply not notifying the Corps of their activities that might result in the discharge of dredged or fill material into aquatic resources because they believed those resources were not jurisdictional under the 2020 NWPR. The lower rates of PJD requests under the

²¹ "Delineation alternatives" refers to delineation concurrences and "No JD required" findings. These are similar to PJDs in that projects on sites with such findings would move forward with permitting as if the associated resources on these sites are jurisdictional.

²² PJDs are not formal determinations and aquatic resources addressed through a PJD are treated as jurisdictional for the purposes of any associated permit action.

2020 NWPR may be the most striking metric for how trends in jurisdiction have changed. When looking at individual resources associated with jurisdictional determinations, this comparison between overall numbers considered as part of AJDs versus as part of PJDs is even more dramatic, with a 4-fold to 5-fold increase in the percent of aquatic resources assessed through AJDs under the 2020 NWPR (Table 1, Figure 2). In order to test for the significance of this shift, a further analysis was carried out, splitting pre-2015 practice data from June 22, 2016, to June 21, 2020, by calendar year, calculating the percent of AJDs and PJDs carried out for each year, and comparing the 95% confidence interval for those years to the percent found for the 2020 NWPR. The test of significance on the proportion of AJDs and PJDs carried out under the 2020 NWPR compared to pre-2015 practice shows that this shift towards more AJDs and fewer PJDs is in fact significant and outside the bounds of the 95% confidence interval for pre-2015 practice data (Table 2).

The agencies found within the AJDs completed under the 2020 NWPR, the probability of finding resources to be non-jurisdictional increased precipitously. Of the 9,399 AJDs completed by the Corps under the 2020 NWPR during the first 12 months in which that rule was in effect, ²³ the agencies found approximately 75% of AJDs completed had identified non-jurisdictional aquatic resources and approximately 25% of AJDs completed identified jurisdictional waters (Table 1, Figure 3).²⁴ Conversely, during similar one-year intervals when the 1986 regulations and applicable guidance were in effect (including following the 2019 recodification of those regulations), substantially more jurisdictional waters were identified in AJDs on average per year than compared to the first twelve months of the 2020 NWPR. During similar one-year calendar intervals when the 1986 regulations and applicable guidance were in effect, approximately 28% to 45% of AJDs completed identified non-jurisdictional aquatic resources, and 56% to 72% of AJDs completed identified jurisdictional resources.²⁵ Although the percentages varied during these periods, the change from a range of 28% to 45% non-jurisdictional AJD findings prior to the 2020 NWPR to 75% non-jurisdictional findings after issuance of the 2020 NWPR indicates that substantially fewer waters were protected by the Clean Water Act under the 2020 NWPR. This constitutes a 69% to 171% increase in the percent of AJDs resulting in non-jurisdictional findings (Table 1). Similarly, when individual aquatic resources associated with AJDs are reviewed by jurisdictional findings, there was an increase of 28% to 94% in the percent of resources being found to be nonjurisdictional under the 2020 NWPR (Table 1). Again, as commenters on the proposed rule noted, these

²³ These AJDs were completed by the Corps between the 2020 NWPR's effective date of June 22, 2020, and June 21, 2021. Data were extracted from ORM2 in September 2021.

²⁴ This excludes dry land AJDs and waters identified as jurisdictional only under section 10 of the Rivers and Harbors Act. In addition, under the 2020 NWPR, a single AJD in the Corps' database can include both affirmative and negative jurisdictional determinations. Under prior regulatory regimes, the Corps' database was structured such that a single AJD could be either affirmative, or negative, but not both. To account for this change in the structure of the database, a 2020 NWPR jurisdictional determination that includes both affirmative and negative jurisdictional resources was normalized and counted as two separate AJDs, one affirmative and one negative. The total number of AJDs considered after this process was carried out was 9,399. Prior to this normalization, the total number of AJDs considered was 7,769.

²⁵ The time periods evaluated were June 22, 2016, to June 21, 2017; June 22, 2017, to June 21, 2018; and December 23, 2019, to June 21, 2020. The date ranges here constitute periods of time when the 1986 regulations (including the 2019 Repeal Rule's recodification of those regulations) and applicable guidance were in effect nationally. Thus, 2015 Clean Water Rule determinations were not part of this analysis. Data were extracted from ORM2 in September 2021.

numbers do not account for the many entities that did not seek AJDs because they knew their features were excluded under the 2020 NWPR.

In evaluating how the 2020 NWPR's non-jurisdictional values compare to the distribution of nonjurisdictional findings from prior years, the percent of non-jurisdictional findings at the individual aquatic resource level under pre-2015 practice from 2016 to 2020 were used (separated by the June 22 – June 21 calendar year) (Table 3). The percent of non-jurisdictional findings for individual aquatic resources under the 2020 NWPR were outside the bounds of the 95% confidence intervals calculated for the 2016 to 2020 data, indicating that the Corps was significantly more likely to find projects did not require section 404 permits with these closure methods under the 2020 NWPR compared to pre-2015 practice. Within the time frame of 2016 to 2020, pre-2015 practice and the 2015 Clean Water Rule were both in effect; as such, the annual distribution of total number of non-jurisdictional resources could not be compared. A separate comparison was carried out using the total number of non-jurisdictional findings at the individual aquatic resource level as associated with all AJDs from 2016-2020 (pre-2015 practice and 2015 Clean Water Rule AJDs) (Table 4). In this case, both the percent of non-jurisdictional findings and the total number of individual resources found to be non-jurisdictional under the 2020 NWPR were outside of the bounds of the 95% confidence intervals calculated from those datasets. The proportion and scale of nonjurisdictional findings at the individual aquatic resource level under the 2020 NWPR is significantly higher than the proportion and scale of non-jurisdictional findings from prior years.

Together, there were proportionally fewer PJDs and more AJDs being carried out under the 2020 NWPR, there were fewer resources being found to be jurisdictional when AJDs were being carried out, and there were elevated findings of no section 404 permits being needed for projects based on the definition of "waters of the United States" under the 2020 NWPR.

JDs: PJDs vs AJDs				
Regulatory Regime		Pre-2015 Practice		2020 NWPR
Time period	June 22, 2016- June 21, 2017	June 22, 2017 - June 21, 2018	Dec. 23, 2019 - June 21, 2020	June 22, 2020 - June 21, 2021
PJD	21,383	13,459	4,539	4,744
PJD Resources	92,010	81,998	60,578	26,712
AJD	3,957	3,505	1,390	9,399
AJD Resources	11,978	14,267	6,603	48,313
Total JDs	25,340	16,964	5,929	14,143
% AJD	15.6%	20.7%	23.4%	66.5%
% Change in % AJD	326%	222%	183%	
Total JD Resources	103,988	96,265	67,181	75,025
% AJD Resources	11.5%	14.8%	9.8%	64.4%
% Change in % AJD Resources	459%	335%	555%	
AJDs: Jurisdictional vs. Non-j	urisdictional			
Regulatory Regime		Pre-2015 Practice		2020 NWPR
Time period	June 22, 2016- June 21, 2017	June 22, 2017 - June 21, 2018	Dec. 23, 2019 - June 21, 2020	June 22, 2020 - June 21, 2021
Jurisdictional AJDs	2,858	2,120	771	2,335
Non-jurisdictional AJDs	1,099	1,385	619	7,064
Jurisdictional AJD Resources	7,343	5,903	3,781	11,929
Non-jurisdictional AJD Resources	4,635	8,364	2,822	36,384
Total AJDs	3,957	3,505	1,390	9,399
Total AJDs % Jurisdictional AJDs	3,957 72.2%	3,505 60.5%	1,390 55.5%	9,399 24.8%

Table 1: Jurisdictional determinations over time

% Change in % Non- jurisdictional AJDs	171%	90%	69%	
Total AJD Resources	11,978	14,267	6,603	48,313
% Jurisdictional AJD				
Resources	61.3%	41.4%	57.3%	24.7%
% Non-jurisdictional AJD Resources	38.7%	58.6%	42.7%	75.1%
% Change in % non- jurisdictional AJD Resources	94%	28%	76%	

Pre-2015 Practice							
	2016-2017	2017-2018	2018-2019	2019-2020	2020-2021		
AJD	3,957	3,505	1,971	2,172	9,399		
PJD	21,383	13,459	7,024	7,478	4,744		
Percent AJD	16%	21%	22%	23%	66%		
Percent PJD	84%	79%	78%	77%	34%		
	Pr	e-2015 Practice	Summary Stat	S	·		
	Mean	Standard Deviation	95% Confide	ence Interval			
Percent AJD	20%	3%	14%	26%			
Percent PJD	80%	3%	74%	86%			

Table 2: The 2020 NWPR compared to prior years of implementation under pre-2015 practice (AJD/PJD)

Table 3: The 2020 NWPR compared to prior years of implementation under pre-2015 practice (jurisdictional/non-jurisdictional)

		2020 NWPR							
	2016-2017	2017-2018	2018-2019	2019-2020	2020-2021				
Jurisdictional Resources	7,343	5,903	5,010	5,736	12,537				
Non-jurisdictional Resources	4,635	8,364	3,486	4,161	37,799				
Percent non-jurisdictional	39%	59%	41%	42%	75%				
	Pre-2015 Practice Summary Stats								
	Mean	Standard Deviation	95% Confide	ence Interval					
Percent non-jurisdictional	45%	9%	27%	63%					

		2020 NWPR			
	2016-2017	2017-2018	2018-2019	2019-2020	2020-2021
Jurisdictional Resources	7,343	5,903	6,532	6,819	12,537
Non-jurisdictional Resources	4,635	8,364	5,277	5,754	37,799
Percent non-jurisdictional	39%	59%	45%	46%	75%
	Pre-2020 N	WPR Summai	ry Statistics		
	Mean	Standard Deviation	95% Confide	ence Interval	
Jurisdictional Resources	6,649	600	5,473	7,826	
Non-jurisdictional Resources	6,008	1,637	2,800	9,215	
Percent non-jurisdictional	47%	8%	31%	63%	

Table 4: The 2020 NWPR compared to prior years of pre-2015 practice and 2015 Clean Water Rule implementation (jurisdictional/non-jurisdictional)



Figure 1: AJD and PJD (percentages). The number of determinations being carried out under the 2020 NWPR as PJDS has decreased compared to pre-2015 practice, while the number of AJDs has increased. These data exclude both "RHA-only" AJDs and "Dry Land" AJDs as well as data from implementation of the 2015 Clean Water Rule. Data have been normalized.



Figure 2: AJD and PJD resources (percentages). The number of resources tied to determinations being carried out under the 2020 NWPR as PJDs has decreased compared to pre-2015 practice, while the number of resources tied to AJDs has increased. This data excludes both Rivers and Harbors Act only ("RHA-only") AJDs and "Dry Land" AJDs as well as data from implementation of the 2015 Clean Water Rule.



Figure 3: AJD findings (percentages). Breakdown of AJDs (by percentages) that found jurisdictional and non-jurisdictional waters for each of the four periods of record. These data exclude both "RHA-only" AJDs and "Dry Land" AJDs as well as data from implementation of the 2015 Clean Water Rule. Data have been normalized.



Figure 4: AJD Findings by Resource (Percentages). Percentage of AJD resources that were found to be jurisdictional and non-jurisdictional for each of the four periods of record. This data excludes both "RHA-only" AJDs and "Dry Land" AJDs as well as data from implementation of the Clean Water Rule. Data have been normalized.

(ii) Individual Aquatic Resources Associated with 2020 NWPR AJDs

When evaluating the effect of the 2020 NWPR on the number of jurisdictional individual aquatic resources (as opposed to the number of AJDs completed), the agencies found a similar significant reduction in protections. The Corps' ORM2 database contains AJDs that evaluated 48,313 individual aquatic resources under the 2020 NWPR between June 22, 2020, and June 21, 2021; of these individual aquatic resources, approximately 75% were found to be non-jurisdictional by the Corps (Table 1). Specifically, 70% of streams and wetlands were found to be non-jurisdictional, including 11,044 ephemeral features (mostly streams) and 15,675 wetlands that did not meet the 2020 NWPR's revised adjacency criteria (and thus are non-jurisdictional under the 2020 NWPR). Ditches were also frequently excluded (4,706 individual exclusions). When looking at the total number of aquatic resources considered under the 2020 NWPR versus under the pre-2015 regulatory regime (assessed by data split by calendar year from June 22 to June 21 for the years of 2016 to 2020), the percent of non-jurisdictional findings under the 2020 NWPR (75%) was significantly higher than the percent of non-jurisdictional findings under pre-2015 practice (average annual non-jurisdictional finding: 45%) (Table 3).

Resource Type	Wetlands and streams	Wetland	Stream
All resources considered	38,611	21,824	16,787
Jurisdictional	11,364	6,062	5,302
Non-Jurisdictional	27,247	15,762	11,485
Percent Non-jurisdictional	71%	72%	68%

Table 5: Specific Resources found Non-Jurisdictional under the 2020 NWPR

(iii) Arid West AJDs

2020 NWPR AJDs in Arizona and New Mexico were found to be dominated by nonjurisdictional ephemeral channelized features (Table 6). The number of stream features that were found non-jurisdictional in Arizona and New Mexico under the first year of 2020 NWPR implementation totaled 1,784.²⁶ Comparatively, nationally, there were 11,485 stream reaches found to be non-jurisdictional under the first year of the 2020 NWPR implementation. 15.5% of all non-jurisdictional streams across the nation in the first year of 2020 NWPR implementation were in Arizona and New Mexico.²⁷ This represents a geographic inequality in the implications of implementation of the 2020 NWPR.

The number of individual aquatic resources - in particular, streams - being considered under AJDs in the past year of data under the 2020 NWPR in Arizona and New Mexico is significantly higher than has occurred in recent years, with the total number of non-jurisdictional individual streams under the 2020 NWPR exceeding the 95% confidence interval for annual non-jurisdictional findings on individual streams from 2016-2020 data for both Arizona and New Mexico (Table 7, Figures 5 and 6). The total number of individual streams being assessed via AJDs in these states is significantly higher than the total number of individual streams that were assessed in the pre-2015 practice time frame considered here.

Looking at all types of aquatic resources and comparing non-jurisdictional AJD findings under the 2020 NWPR to prior years of implementation under pre-2015 practice (averaged over four years), there has been more than a 10-fold increase in non-jurisdictional findings on individual aquatic resources in both Arizona and New Mexico (Table 6). When individual streams considered under AJDs are inspected in this manner, there was more than a 10-fold increase in non-jurisdictional findings for individual streams in Arizona and more than a 30-fold increase in non-jurisdictional findings for individual streams in New Mexico under the 2020 NWPR (Table 5). Additionally, the number of stream resources considered under PJDs and other alternatives to AJDs (*i.e.*, Delineation Concurrences and No

²⁶ The 2020 NWPR water types included within this calculation included (a)(2) tributaries, (b)(1) surface water channel that does not contribute surface water flow directly or indirectly to an (a)(1) water in a typical year, and (b)(3) ephemeral streams. There are other water types associated with 2020 NWPR AJDs, such as (a)(1) Traditional Navigable Waters and Territorial Seas, which can contain streams but were not included in the total. Note that the numbering of the water types in this footnote are associated with the numbering in the 2020 NWPR and not with this rule.

²⁷ The next highest rate of non-jurisdictional findings (11%) for AJDs conducted under the 2020 NWPR is in the state of Utah, which similarly contains vast areas of arid West ecosystems.

JD required closure methods) decreased by a range of 75% to 100% under the 2020 NWPR (Table 6, Figures 7, and 8). In all years, the majority of resources considered were ephemeral streams. Under the 2020 NWPR, wetlands adjacent to ephemeral streams are also not jurisdictional, whereas under pre-2015 practice could have been considered for jurisdiction under a case-specific significant nexus analysis; thus, this decrease in jurisdiction and assumed jurisdiction could have cascading effects on multiple types of resources if the 2020 NWPR were reinstated.

The arid West, as exemplified by Arizona and New Mexico, experienced a comparatively higher share of non-jurisdictional findings on stream reaches than the nation as a whole did under the 2020 NWPR. The number of non-jurisdictional stream reaches is significantly elevated compared to prior years, and simultaneously, the number of PJDs and other AJD alternatives has decreased compared to prior years. The agencies thus believe that the 2020 NWPR caused disproportionate and more severe impact on aquatic resources in the arid West.

Table 6: Arizona and New Mexico Impacts to Scope of "Waters of the United States" under the 2020 NWPR (2020-2021) Compared to Pre-2015 Practice (2016-2017, 2017-2018, 2018-2019, 2019-2020)

ALL Resources								
	Arizona						lexico	
Delineation type	A Juris	JD Non- Juris	PJD	NoJD and DC*	A Juris	JD Non- Juris	PJD	NoJD and DC*
2016-2017	0	17	562	5	0	17	203	71
2017-2018	5	233	386	22	4	17	309	96
2018-2019	16	261	669	25	2	18	25	99
2019-2020	17	42	2,145	32	0	4	0	49
2016-2020 avg	10	138	941	21	2	14	134	79
2020-2021	4	1,538	52	0	2	280	35	0
% change from avg to 2020 NWPR	-58%	1,012 %	-94%	-100%	33%	1,900%	-74%	-100%
		S	Stream r	esources				
		Arizo	ona			New M	lexico	
Delineation type	A Juris	JD Non- Juris	PJD	NoJD and DC*	A Juris	JD Non- Juris	PJD	NoJD and DC*
2016-2017	0	17	547	5	0	7	193	69
2017-2018	5	231	378	16	4	15	301	83
2018-2019	16	261	658	17	2	2	24	91
						1		
2019-2020	17	42	2,113	29	0	4	0	46
2019-2020 2016-2020 avg	17 10	42 138	2,113 924	29 17	0 2	4 7	0 130	46 72

% change from avg to 2020 NWPR	-58%	1,004 %	-95%	-100%	-100%	3,657%	-75%	-100%
		Ep	hemera	l resourc	es			
		Arizo	ona			New M	lexico	
Delineation type	A. Juris	JD Non- Juris	PJD	NoJD and DC*	A Juris	JD Non- Juris	PJD	NoJD and DC*
2016-2017	0	17	547	0	0	7	193	43
2017-2018	4	231	378	15	4	15	301	32
2018-2019	16	261	658	11	1	2	24	26
2019-2020	17	42	2,113	26	0	3	0	11
2016-2020 avg	9	138	924	13	1	7	130	28
2020-2021	4	1,518	45	0	0	263	33	0
% change from avg to 2020 NWPR	-57%	1,002 %	-95%	-100%	-100%	3,796%	-75%	-100%

*NoJD and DC refer to No JD Required and Delineation Concurrence closure methods within ORM. These are equivalent to delineation substitutes. They are included here because the data associated with these categories in these specific states were not erroneous, and they show a more complete picture of how resources have been reviewed by the Corps.

А	RIZONA				
		Pre-201	5 Practice		2020 NWPR
	2016-	2017-	2018-	2019-	2020-
	2017	2018	2019	2020	2021
All AJD resources	17	236	277	59	1,525
Jurisdictional AJD resources	0	5	16	17	4
Non-jurisdictional AJD resources	17	231	261	42	1,521
PJD and Alternatives to AJDs resources	552	394	675	2,142	45
Percent non-jurisdictional resources	100.0%	97.9%	94.2%	71.2%	99.7%
Pre-2015 Pract	tice Summ	ary Statis	tics		I
	Mean	SD	95%	6 CI	
All AJD resources	147	128	-104	399	
Jurisdictional AJD resources	10	8	-7	26	
Non-jurisdictional AJD resources	138	126	-109	385	
PJD and Alternatives to AJDs resources	941	809	-645	2,526	
Percent non-jurisdictional resources	91%	13%	65%	117%	
NE	W MEXIC	20			
					2020

Table 7: Test of statistical significance for 2020 NWPR compared to prior years of implementation under pre-2015 practice for stream resources in Arizona and New Mexico. ADJZONA

2016-

All AJD resources

Jurisdictional AJD resources

Non-jurisdictional AJD resources

PJD and alternatives to AJDs resources

Pre-2015 Practice

2018-

2019-

2017-

NWPR

2020-

Percent non-jurisdictional resources	100.0%	78.9%	50.0%	100.0%	100.0%				
Pre-2015 Practice Summary Statistics									
	Mean	SD	95%						
All AJD resources	9	7	(5)	22					
Jurisdictional AJD resources	2	2	(2)	5					
Non-jurisdictional AJD resources	7	6	(4)	18					
PJD and alternatives to AJDs resources	202	151	(95)	498					
Percent non-jurisdictional resources	82.2%	23.7%	35.8%	128.6%					

Note: No 2015 Clean Water Rule determinations were carried out during these years in these states.



Figure 5: Individual aquatic resources found to be non-jurisdictional in Arizona over the past five years. Under the 2020 NWPR (2020-2021), there has been over a 10-fold increase in non-jurisdictional findings for individual aquatic resources in both Arizona and New Mexico.



Figure 6: Individual aquatic resources found to be non-jurisdictional in New Mexico over the past five years. Under the 2020 NWPR (2020-2021), there has been over a 10-fold increase in non-jurisdictional findings for individual aquatic resources in both Arizona and New Mexico.



Figure 7: Arizona Individual Aquatic Resources Tied to PJDs and Determination Substitutes by

Year. Under the 2020 NWPR (2020-2021), there has been more than a nine-fold decrease in individual aquatic resources reviewed through PJDs and determination substitutes in Arizona. Note that in 2020, there was a single transmission line project that was related to the majority of the individual resources tied to PJDs in that year.



Figure 8: New Mexico Individual Aquatic Resources Tied to PJDs and Determination Substitutes by Year. Under the 2020 NWPR (2020-2021), there has been more than a seven-fold decrease in individual aquatic resources reviewed through PJDs and determination substitutes in New Mexico. Note that in 2017, there was a single transmission line project that was related to the majority of the individual resources tied to PJDs in that year.

(iv) No Permit Required based on AJDs only

Based on an assessment of the two specific "No Permit Required" closure methods in ORM2 related to projects associated with AJDs, under the 2020 NWPR there has been more than a 150% increase in projects that do not require Clean Water Act section 404 permits as compared to what was reported under the previous regulatory regimes (Table 8, Figure 9).²⁸ Even when the new reporting field of "Activity occurs in waters that are no longer WOTUS under the 2020 NWPR" is excluded, the total number of projects that do not require Clean Water Act section 404 permits under the 2020 NWPR exceeds the 95% confidence interval for values from the 2016 to 2020 timeframe. Given that this closure method, "Activities that occur in waters that are no longer WOTUS under the 2020 NWPR," has not been used uniformly by all Corps project managers across the United States, it is likely that the overall number of projects that fit into this category are under-represented.

²⁸ This is a revised calculation that updates the prior assessment of No Permit Required data presented within the Declarations of Radhika Fox and Jaime A. Pinkham, filed in *Conservation Law Found. et al. v, EPA et al.*, 20-cv-10820-DPW (D. Mass. Jun. 9, 2021) as well as in every other district court challenge to the NWPR. The revised calculation is based on additional data spanning a longer time period and corrected underlying calculations of the percent change between the number of projects that do not require Clean Water Act section 404 permits under the NWPR compared to prior time periods.

		2020 NWPR			
	2016-	2017-	2018-	2019-	2020-
	2017	2018	2019	2020	2021
Activity does NOT occur in WOTUS	414	452	294	355	769
Activity occurs in waters that are NO longer WOTUS under the 2020 NWPR	-	-	-	-	368
Total	414	452	294	355	1,137
Percent change between given year and 2020 NWPR for Activity does NOT occur in WOTUS	86%	70%	162%	117%	
Percent change between given year and 2020 NWPR for Total	175%	152%	287%	220%	
Pre-2015 Practi	ice Summ	ary Statis	tics		
	Mean	SD	95%	6 CI	
Activity does NOT occur in WOTUS	379	69	243	514	

Table 8: No Permit Required test of significance for 2020 NWPR compared to prior years of implementation under pre-2015 practice



Figure 9: Actions Not Requiring Permits Under Pre-2015 Practice and 2020 NWPR. This figure presents the number of actions (*i.e.*, projects) with "No permit required" closure methods of "Activities that do not occur in WOTUS" and "Activities that occur in waters that are no longer WOTUS under the 2020 NWPR."

d. Data Limitations

While ORM2 contains data on individual aquatic resources that the Corps has determined are or are not jurisdictional on a site-specific basis, JDs are typically conducted at the request of the landowner or project proponent. In other words, they usually represent where landowners or project proponents want to know if jurisdictional waters are located within their properties or project sites, including but not limited to for purposes of conducting dredged or fill activities. Thus, some aquatic resource types may be over- or underrepresented in the population of PJDs and AJDs.

The agencies recognize that these PJDs and AJDs may not be uniformly distributed across the country. There may be selection bias in terms of where the Corps has available information on JDs. A landowner or applicant can decide whether they would like an AJD – meaning the Corps makes an official determination of whether an aquatic resource is jurisdictional – or whether they would prefer to voluntarily waive or set aside questions regarding jurisdiction with the use of a PJD). In addition, Corps Districts across the country vary in their receipt of requests for AJDs versus PJDs, with some Districts primarily being requested to complete PJDs, particularly prior to the 2020 NWPR. Because PJDs do not make an official determination of the jurisdictional status of an aquatic resource (*e.g.*, it cannot conclude that an aquatic resource is not a "water of the United States"), and in light of the reduction in jurisdiction under the 2020 NWPR, the use of PJDs has appeared to decrease.

The states of New Jersey and Michigan have assumed administration of the Clean Water Act section 404 permit program for certain waters within their state boundaries. On December 17, 2020,

Florida became the third state to receive approval to assume administration of the program. The Corps, however, retains administration of the section 404 permitting program for specific waters listed under the parenthetical of Clean Water Act section 404(g)(1) within states which have assumed the section 404 permitting program. Thus, the Corps conducts JDs for only a subset of waters within New Jersey, Michigan, and Florida, which have been included in the analysis of ORM2 data where available. In Florida, the number of 2020 NWPR JDs conducted by the Corps will be limited compared to the number of JDs in that state conducted under the prior regulatory regimes, as EPA's approval for the state to assume administration of the section 404 program occurred a few months after the effective date of the 2020 NWPR.

The new ORM closure method "Activities that occur in waters that are no longer WOTUS under the 2020 NWPR" was not uniformly used across the Districts and by Corps project managers and was added one month after implementation of the 2020 NWPR, and thus likely undercounts the number of projects that would have required a Clean Water Act section 404 permit prior to the 2020 NWPR but that no longer do. However, it serves as the best available indicator of projects that were tracked and no longer required a section 404 permit in light of the 2020 NWPR's reduction in Clean Water Act jurisdiction.

Despite these limitations, the agencies have concluded that assessing the ORM2 data associated with the 2020 NWPR is a reasonable way to evaluate the effects of that rule. The data represent the best national-level information on the resources that were called non-jurisdictional under the 2020 NWPR prior to its vacatur, and the agencies have concluded that it is reasonable to compare the 2020 NWPR data from 2020-2021 with data from the same time period in prior years that are associated with determinations made under the 2015 Clean Water Rule and the pre-2015 regulatory regime, which was reestablished with the 2019 Rule.

ii. Stakeholder Concerns

The agencies have heard concerns from a broad array of stakeholders, including states, tribes, scientists, and non-governmental organizations, that corroborated the agencies' data and indicated that the 2020 NWPR's reduction in the jurisdictional scope of the Clean Water Act resulted in significant environmental harms. As discussed in the Economic Analysis for the Final Rule, many Tribes and States do not regulate waters more broadly than the Clean Water Act. *See* Economic Analysis for the Final Rule, Chapter II; 2020 NWPR EA at 30-31. While some Tribes and States have authority to regulate "waters of the Tribe" or "waters of the State" more broadly than the federal government under their own laws, projects also proceeded in newly non-jurisdictional waters on Tribal lands and in States that do not and sometimes cannot regulate waters beyond those covered by the federal Clean Water Act.²⁹ Based on

²⁹ Even if a tribe has the legal authority to regulate "waters of the tribe" more broadly than the federal government, the agencies have heard from many tribes that they lack the resources and expertise to do so as a practical matter, and therefore rely on Clean Water Act protections. *See, e.g.*, 85 FR 22336-22337 ("many Tribes may lack the capacity to create a tribal water program under tribal law, to administer a program, or to expand programs that currently exist. Other tribes may rely on the Federal government for enforcement of water quality violations"). In their Tribal consultation comment letter for this rulemaking the Southern Ute Indian Tribe noted, "Unlike some states where waters that are not classified as WOTUS can be protected by state-only water quality laws, the checkerboard nature of the Reservation and the division of jurisdiction means the Tribe's water quality laws alone
available information, the agencies therefore expect that these projects could have resulted in discharges without any regulation or mitigation from federal, Tribal, or State agencies. Contrary to the predictions made in the 2020 NWPR Economic Analysis, during the year in which the 2020 NWPR was in effect, the net change made by States was deregulatory in nature. Two States which had previously protected state waters beyond the scope of "waters of the United States" removed these expansive protections, and no States that lacked these broader protections established them. *See* 2020 NWPR EA at 39-41 (estimating that some States are likely to continue their current permitting practices for dredged and fill material) and the Economic Analysis for the Final Rule Chapter II (indicating that two of those States reduced the scope of State clean water protections after the 2020 NWPR was finalized, and none of them formerly expanded protections as a direct result of the 2020 NWPR); *see also* preamble section IV.B.3.

Given the limited authority of many Tribes and States to regulate waters more broadly than the federal government, a narrowing of federal jurisdiction would mean that discharges into the newly nonjurisdictional waters would no longer be subject to regulation, including permitting processes and mitigation requirements designed to protect the chemical, physical, and biological integrity of the nation's waters. Ephemeral streams and their associated wetlands, wetlands that do not meet the 2020 NWPR's revised adjacency criteria, and other aquatic resources not protected by the 2020 NWPR provide numerous ecosystem services, as discussed in the Science Report and this document. See also Sullivan et al. 2020. The absence of protections for such resources and any subsequent unregulated and unmitigated impacts to such resources would have caused cascading, cumulative, and substantial downstream harm, including damage connected to water supplies, water quality, flooding, drought, erosion, and habitat integrity. The removal of protections from the nation's waters, and resulting detriment to the services they provide, undermines the objective of the Clean Water Act, as discussed in section IV.A.2 of the preamble to this final rule. Such effects on the chemical, physical, and biological integrity of the nation's waters were inadequately considered during the 2020 NWPR rulemaking process. See Pascua Yaqui v. EPA, no. 4:20-cv-00266-TUC-RM, slip op. at 9-10 (citing evidence that the agencies and plaintiffs provided of a "substantial reduction in waters covered under the 2020 NWPR" as demonstrating "the possibility of serious environmental harm" that weighed in favor of vacating the rule.); see also Navajo Nation v. Regan, no. 2:20-cv-00602, slip op. at 6-7 (citing the same reduction particularly "an increase in determinations by the Corps that waters are non-jurisdictional,' including excluded ephemeral resources, 'and an increase in projects for which CWA Section 404 permits are no longer required," as weighing in favor of vacatur.).

iii. Scientific and Technical Review

1. 2020 NWPR

Science plays a critical role in understanding how to protect the integrity of the nation's waters. The agencies have carefully reviewed the 2020 NWPR and its administrative record and found that the 2020 NWPR did not properly consider the extensive scientific evidence demonstrating the

might not be effective at protecting water quality within the entire Reservation. In other words, unlike states, the Tribe cannot easily enforce a definition of "tribal waters" that is broader than the EPA and Corps' definition of WOTUS and any attempt to do so could trigger lengthy and expensive jurisdictional litigation. Therefore, given the jurisdictional uncertainties of the scope of the Tribe and the State of Colorado's independent authority to protect all waters of the Reservation, the EPA and Corps current, more narrow definition of WOTUS [the 2020 NWPR] disadvantages the Reservation and leaves certain Reservation waters unprotected." Baker 2021.

interconnectedness of waters and their downstream effects, thereby undermining Congress's objective to restore and maintain the chemical, physical, and biological integrity of the nation's waters. The 2020 NWPR's definition of "waters of the United States" does not adequately consider the way pollution moves through waters or the way filling in a wetland affects downstream water resources.

The 2020 NWPR's exclusion of major categories of waters from the protections of the Act, specifically in the definitions of "tributary" and "adjacent wetlands," runs counter to the scientific record demonstrating how such waters can affect the integrity of downstream waters. Specifically, its categorical exclusion of ephemeral features and large categories of wetlands is inconsistent with the scientific record before the agencies. In addition, the 2020 NWPR's limits on the scope of protected wetlands to those that touch or demonstrate evidence of a regular surface water connection to other jurisdictional waters were counter to the ample scientific information demonstrating the effects of wetlands on downstream waters when they have other types of connections.

This section will first describe how the 2020 NWPR's treatment of tributaries and adjacent wetlands is inconsistent with the scientific understanding of these ecosystems, and then provide detailed explanations of those inconsistencies using specific examples from the 2020 NWPR preamble where the agencies indicated that they "considered scientific principles" in the definition of "waters of the United States."

The 2020 NWPR's definition of "waters of the United States" runs counter to long-standing scientific understanding of tributaries, lakes, reservoirs, coastal waters, wetlands, and other types of waters. This is most evident by the 2020 NWPR's (1) failure to recognize ecosystem functions, *i.e.*, the processes by which aquatic ecosystems support fundamental needs, including good water quality and safe, reliable supplies of fresh water for communities, industry, and agriculture; (2) failure to accommodate the cumulative effects of headwaters, non-perennial flows, and wetlands on downstream water integrity, and (3) exclusion of ephemeral streams, many floodplain wetlands, and most non-floodplain wetlands in its definition of "waters of the United States."

Variations represented as hydrological, biogeochemical, and biological connectivity gradients operate in all aquatic ecosystems, and at all spatial and temporal scales. These variations are necessary to maintain the full range of functions by which upstream waters affect (*i.e.*, have consequences for) downstream water integrity. Such functions include short- and long-term storage of water, nutrients, and sediment in wetlands (e.g., Jacques and Lorenz 1988; Vining 2002; Gleason et al. 2003; McEachern et al. 2006; Gleason et al. 2007; Fossey and Rousseau 2016; Golden et al. 2016; Mekonnen et al. 2016; Evenson et al. 2018; Green et al. 2019; Wang et al. 2019; Yeo et al. 2019; Nasab and Chu 2020; Rajib et al. 2020; Shook et al. 2021), transformation or sequestration of contaminants, removal or transformation of excess nutrients in temporary wetlands and non-perennial headwaters (e.g., Reddy et al. 1999; Kao et al. 2002; Boon 2006; Jordan et al. 2007; Mitsch and Gosselink 2007; Reddy and DeLaune 2008; Kadlec and Wallace 2009; Cheesman et al. 2010; Marton et al. 2015; Cheng and Basu 2017; Golden et al. 2019; Martin et al. 2019), provision of habitat for aquatic and semiaquatic species in wetlands and headwater streams (e.g., Scrivener et al. 1994; Curry 1997; Pires et al. 1999; Bradford et al. 2001; Meyer and Wallace 2001; Meyer et al. 2004; Cairns et al. 2005; Huryn et al. 2005; Wigington et al. 2006; Woodford and McIntosh 2010), recharge of river baseflow (e.g., Goodrich et al. 1997; Baillie et al. 2007; Callegary et al. 2007; Izbicki 2007; Brooks et al. 2018; Min et al. 2020), and provision of drinking water for humans and wildlife from variably connected surface waters of all types (e.g., Levick et al. 2008; Lohse

et al. 2020). Connections with low values for some descriptors of hydrologic connectivity (*e.g.*, lowfrequency, short-duration flooding) can have important downstream effects when values for other descriptors are high (*e.g.*, large-magnitude downstream transfer of floodwaters, sediment, large woody debris, and organisms). At the other end of the frequency gradient, high-frequency, low-magnitude vertical and lateral flows contribute to aquatic biogeochemical processes, including nutrient and contaminant transformation and organic matter accumulation. *See, e.g.*, Brunke and Gonser 1997; Karwan and Saiers 2012; Lawrence *et al.* 2013; Dwivedi *et al.* 2018; Evenson *et al.* 2018. These and other functions discussed in depth above and in the Science Report are absent from the 2020 NWPR's definition of "waters of the United States."

As the discussion of the agencies' scientific record also makes clear, the cumulative effects of upstream aquatic ecosystems on downstream water integrity should be considered in three ways:

- First, when evaluating the effect of an individual stream or wetland, all the contributions and functions that the stream or wetland provides must be considered. For example, the same stream can transport water, remove excess nutrients, mitigate flooding, and provide refuge for fish when conditions downstream are unfavorable; ignoring any of these functions would underestimate the overall effect of that stream. Note that these different functions may not all occur simultaneously, but all contribute over time. *See* Science Report.
- Second, effects of multiple streams and wetlands and the watersheds they drain are 0 fundamentally cumulative in how those watersheds are formed and maintained. Excess precipitation that is not evaporated, taken up by organisms, or stored in soils and geologic layers moves downgradient as overland or subsurface flow or through channels, which concentrates flows and carries sediment, chemical constituents, and organisms. Flows from headwater tributaries longitudinally connect with other parts of the network, and their confluence with larger-order streams can have profound effects on the receiving system due to the abrupt increase in water, sediment, wood, and other entrained materials. As flows from numerous headwater channels combine in larger channels, the volume and effects of those flows accumulate as they move downstream through the river network. As a result, the incremental contributions of individual streams and wetlands accumulate in the downstream waters. Important cumulative effects are exemplified by ephemeral flows, which are key sources of baseflow for downgradient waters in arid and semi-arid regions (Schlesinger and Jones 1984; Baillie et al. 2007; Izbicki 2007), and by the high rates of denitrification in headwater streams (Fritz et al. 2018; Science Report). The amount of nutrients removed by any one stream over multiple years or by all headwater streams in a watershed in a given year can have substantial consequences for hypoxia and eutrophication in downstream waters (Alexander et al. 2007; Alexander et al. 2009; Böhlke et al. 2009; Helton et al. 2011). Similarly, a single pollutant discharge could be negligible, but the cumulative effect of multiple unregulated discharges can and has degraded the integrity of jurisdictional waters. Thus, the overall probability of a large-magnitude transfer of materials is higher when considered for all headwater streams in a watershed—that is, there is a *high-frequency connection* when considered cumulatively at the watershed scale, compared with probabilities of transport for streams individually. See Science Report.

• Third, evaluating cumulative contributions over time is critical in streams and wetlands with variable degrees of connectivity. For example, denitrification in a single headwater stream in any given year might not affect downstream waters; over multiple years, however, this effect accumulates. Western vernal pools provide another example of cumulative effects over time. These pools typically occur as complexes in which the hydrology and ecology are tightly coupled with the local and regional geological processes that formed them. When seasonal precipitation exceeds wetland storage capacity and wetlands overflow into the river network through swales and shallow subsurface flows that generate streamflow, the vernal pool basins, swales, and seasonal streams function as a single surface-water and shallow ground-water system connected through the river network. *See* Science Report.

The definition of the term "tributary" in the 2020 NWPR categorically excluded ephemeral streams from the regulatory protections of the Act, contrary to scientific information emphasizing the vital role these streams can play in protecting the integrity of downstream waters. The 2020 NWPR's definition of the term "tributary" states that the agencies "relied on the available science to help inform where to draw the line of federal jurisdiction over tributaries," and that the "agencies' decisions in support of this final rule have been informed by science." 85 FR 22288 (April 21, 2021). The science, however, is clear that aggregate effects of ephemeral streams "can have substantial consequences on the integrity of the downstream waters" and that the evidence of such downstream effects is "strong and compelling," as discussed above. Science Report at 6-10, 6-13. The SAB in their review of the draft Science Report explains that ephemeral streams "are no less important to the integrity of the downgradient waters" than perennial or intermittent streams. SAB 2014a at 22-23, 54 fig. 3. While in the arid Southwest, features flow into downstream waters less frequently than they do in the wetter East, the Science Report emphasizes that short duration flows through ephemeral streams can transport large volumes of water to downstream rivers. Science Report at 6-10. For instance, the report notes that ephemeral streams supplied 76% of flow to the Rio Grande following a large rainstorm. Science Report at 3-8. The SAB emphasizes that the "cumulative effects" of ephemeral flows in arid landscapes can be "critical to the maintenance of the chemical, physical, and biological integrity" of downstream waters. SAB 2014a at 22.

Similarly, the 2020 NWPR's definition of "adjacent wetlands" excluded many categories of wetlands that can play a vital role in protecting the integrity of waters to which they are connected, including traditional navigable waters. In defining "adjacent wetlands," the 2020 NWPR limited the scope of wetlands protected by the Clean Water Act's regulatory programs to those that either abut or have evidence of certain surface water connections to other protected waters in a typical year. 85 FR at 22340. Specifically, the rule encompassed wetlands that (i) abut, meaning to touch, another jurisdictional water; (ii) are flooded by a jurisdictional water in a typical year; (iii) are separated from a jurisdictional water only by a natural feature, such as a berm, which provides evidence of a direct surface hydrological connection with that water; or (iv) are separated from a jurisdictional water only by an artificial structure so long as that structure allows for a direct hydrologic surface connection between the wetlands and the water in a typical year. *Id.* As with the tributary definition, the 2020 NWPR's limits on the scope of protected wetlands to those that touch or demonstrate evidence of a regular surface water connection to other jurisdictional waters were counter to the ample scientific information before the agencies demonstrating the effects of wetlands on downstream waters when they have other types of surface

connections, such as wetlands that overflow and flood jurisdictional waters or wetlands with less frequent surface water connections due to long-term drought; wetlands with shallow subsurface connections to other protected waters; or other wetlands proximate to jurisdictional waters. *See Rapanos*, 547 U.S. at 786 (Kennedy, J., concurring in the judgment) ("[g]iven the role wetlands play in pollutant filtering, flood control, and runoff storage, it may well be the absence of a hydrologic connection (in the sense of interchange of waters) that shows the wetlands' significance for the aquatic system.").

Indeed, the overwhelming scientific information before the agencies weighs decisively against repromulgating the definition of "adjacent wetlands" in the 2020 NWPR. Available scientific information demonstrates the significant effects of categories of newly excluded wetlands on the chemical, physical, and biological integrity of downstream traditional navigable waters. For example, whereas the 2020 NWPR provided that wetlands flooded by jurisdictional waters are only protected if the flooding occurs in a "typical year," the Science Report stated that wetlands that are "rarely" or "infrequently" flooded by streams and rivers can be "highly connected" to those waters and have "long-lasting effects" on them. Science Report at 4-39. The Science Report noted that effects "critical to maintaining the health of the river" result from large floods that provide "infrequent connections" with more distant wetlands. *Id.* Reflecting these concerns, the October 16, 2019 SAB Draft Commentary on the proposed 2020 NWPR stated that the narrow definition of "adjacent wetlands" in the 2020 NWPR as it was proposed "departs from established science." The agencies have weighed these statements and in light of the information about the importance of "infrequently" flooded wetlands to downstream waters, the agencies believe that the 2020 NWPR's exclusion of wetlands that lack the limited, specific types of surface water connections to other jurisdictional waters in a typical year lacked scientific support.

The SAB's assessment of the 2020 NWPR proposal recognized that the proposed rule was not consistent with the scientific information in the record, including the Draft Science Report that the SAB had previously reviewed. SAB 2020. The 2020 SAB Commentary emphasized that the proposal does not "fully incorporate the body of science on connectivity" that the SAB had reviewed in the Draft Science Report and offers "no scientific justification for disregarding the connectivity of waters accepted by current hydrological science." *Id.* at 2.

The 2020 NWPR stated that the "agencies' decisions in support of this final rule have been informed by science." 85 FR at 22288. However, the only scientific information the agencies provided in support of excluding ephemeral features and large categories of wetlands in the 2020 NWPR mischaracterized the scientific record before the agencies. For example, the scientific information that the 2020 NWPR cited as a basis for excluding ephemeral tributaries is the concept of a "connectivity gradient." *Id., citing* SAB 2014a at 3. The 2020 NWPR referred to the SAB's recommendation in their review of the draft Science Report that the agencies recognize that connectivity occurs along a gradient allowing for variation in chemical, physical, and biological connections. *Id.* at 22288, citing SAB 2014a at 3. The 2020 NWPR asserted that there is a "decreased" likelihood that waters with "less than perennial or intermittent" flow, *i.e.*, ephemeral streams, will affect the chemical, physical, and biological integrity of downstream waters. *Id.*

Upon careful review, however, the agencies have concluded the 2020 NWPR's conclusion takes the SAB's recommendation out of context and is inconsistent with the information in SAB 2014a as a whole and in the scientific record that was before the agencies. The agencies recognize that the SAB

explained that the connectivity gradient the 2020 NWPR cited was just a hypothetical example³⁰ meant to illustrate just one aspect of connectivity- duration of surface hydrological, or physical connectivityand sheds no light on the many other ways that features connect to and affect downstream waters. See, e.g., Sullivan et al. 2019a at 11559 ("The near-exclusive emphasis of the proposed rule on hydrologic connectivity contradicts the CWA's mandate to protect chemical and biological connectivity as well"). According to the SAB itself, the only scientific information the agencies provided in support of categorically excluding ephemeral features does not fully represent the discussion in the cited SAB review of the draft Science Report and runs counter to key elements of the scientific record before the agencies. SAB 2020. The SAB in their review of the draft Science Report noted that "low levels of connectivity" can have "meaningful" effects on the integrity of downstream waters. SAB 2014a at 2. In addition, members of the SAB Panel that reviewed the Draft Science Report ("SAB Panel members") noted, "The agencies improperly used the [] figure from the SAB review to support removing federal protection for ephemeral streams and non-floodplain wetlands. The conceptual figure is meant to convey that connectivity between streams and wetlands and downstream waters is more appropriately represented by a connectivity gradient (A and B); this is not a binary property. Aggregate effects and low levels of connectivity can be important." Sullivan et al. 2019a at 11560. Further, the SAB Panel members stated, "Although the connectivity gradient does suggest that certain ephemeral streams and non-floodplain wetlands may be comparably less connected to downstream waters than perennial streams and floodplain wetlands, the SAB affirmed that even low levels of connectivity can be important relative to impacts on the chemical, physical, and biological integrity of downstream waters." Id. Similarly, Sullivan et al. 2020 (at 766) noted, "such exclusions [of ephemeral streams and non-floodplain wetlands under the 2020 NWPR] are inconsistent with evidence demonstrating that these waters are functionally connected to and support the integrity of downstream waters. Removal of federal protection is likely to diminish numerous ecosystem services, such as safeguarding water quality and quantity, reducing or mitigating flood risk, conserving biodiversity, and maintaining recreationally and commercially valuable fisheries."

In addition, several SAB Panel members have expressed that the 2020 NWPR as proposed "largely ignores or misrepresents several conclusions of the [Science] Report and SAB review." Sullivan *et al.* 2019a at 11559. *See also* Sullivan *et al.* 2019b at 7 ("As members of the previous SAB panel that reviewed the Connectivity Report and the 2015 [Clean Water Rule (CWR)], we are intimately familiar with the science supporting the 2015 CWR and the critical role played by the CWA in protecting our Nation's waters. We strongly oppose the proposed Rule, which we find to be inconsistent with science, based upon flawed logic, and too ambiguous for decision-making.") Some SAB Panel members stated that the proposal was "inconsistent with the best-available science regarding scale, structural and functional connectivity, and consideration of the multiple dimensions of connectivity" and that "its exclusions are justified with information from the SAB review that has been misinterpreted or taken out of context." Sullivan *et al.* 2019a at 11559, 11560. These SAB Panel members, along with other scientists, have similarly reviewed the 2020 NWPR as finalized and have concluded that the rule disregards or misinterprets the science. Sullivan *et al.* 2020; CASS 2021. Scientists have noted that "the 2020 NWPR promotes regulations contrary to what science shows about effective water protection" (*id.*

³⁰ The figure cited is captioned in part as "*Hypothetical illustration* of connectivity gradient and potential consequences to downstream waters." SAB Review at 54 (emphasis added). Nowhere in its review does the SAB review indicate that this is the actual or only connectivity gradient.

at 767) and "conflicts with the object of the CWA: to restore and maintain the 'chemical, physical, and biological integrity of (the) Nation's waters" (Fesenmyer *et al.* 2021 at 252).

The SAB (2014, at 22) emphasized that the "cumulative effects" of ephemeral flows in arid landscapes can be "critical to the maintenance of the chemical, physical, and biological integrity" of downstream waters. Contrary to the statements in the 2020 NWPR, the Science Report made clear that the aggregate effects of ephemeral streams "can have substantial consequences on the integrity of the downstream waters" and that the evidence of such downstream effects is "strong and compelling." Science Report at 6-13; *id.* at ES-7.

The record thereby contains robust scientific information demonstrating the importance of ephemeral streams to the integrity of the Nation's waters. The SAB's assessment of the 2020 NWPR proposal recognized that the proposed rule was not consistent with the scientific information in the record, including the Draft Science Report that the SAB had previously reviewed. SAB 2020. The SAB emphasized that the proposal does not "fully incorporate the body of science on connectivity" that the SAB had reviewed in the Draft Science Report and offers "no scientific justification for disregarding the connectivity of waters accepted by current hydrological science." *Id.* at 2. According to the SAB itself, the only scientific information the agencies provided in support of categorically excluding ephemeral features does not fully represent the discussion in the cited SAB Review and runs counter to key elements of the scientific record before the agencies. *Id.* The 2020 NWPR did not explain how the agencies reconciled that information with their decision to categorically exclude ephemeral streams from the definition of "tributaries," and thus failed to articulate "a rational connection between the facts found and the choice made." *State Farm*, 463 U.S. at 43; *see also District Hosp. Partners, L.P. v. Burwell*, 786 F.3d at 59.

The 2020 NWPR also stated that the line it draws between regulated and non-regulated wetlands, which excludes large categories of wetlands previously covered by the Act, is "informed by science." 85 FR at 22314. The 2020 NWPR cited statements from the SAB's review of the Draft Science Report to the effect that wetlands situated alongside other types of waters are likely to be connected to those waters, whereas "those connections become less obvious" as the distance "increases." *Id. citing* SAB 2014a at 55; *see also id.* at 22314, *citing* SAB 2014a at 60 ("[s]patial proximity is one important determinant" influencing the connections between wetlands and downstream waters"). In addition, the 2020 NWPR cited a statement in the Science Report that explained, "areas that are closer to rivers and streams have a higher probability of being connected than areas farther away." *Id.* at 22314, *citing* the Science Report at ES-4.³¹

Despite these citations, the 2020 NWPR's definition of adjacent is not based on proximity, but instead on factors that are distinct from proximity -i.e., a "direct hydrologic connection," a "continuous surface [water] connection." *See id.* at 22340. Thus, the 2020 NWPR's definition of "adjacent wetlands" may exclude wetlands a dozen feet away from jurisdictional waters (therefore proximate under any interpretation of the term) if they are separated by a levee that does not convey flow in a typical year, but include wetlands much further away so long as they are inundated by flooding from the jurisdictional

³¹ This excerpt in the NWPR omits the end of the sentence in the Science Report, which qualifies the cited phrase: "...when conditions governing the type and quantity of flows—including soil infiltration rate, wetland storage capacity, hydraulic gradient, etc.—are similar". Science Report at ES-4.

water in a typical year. In addition, the statements the 2020 NWPR cites, addressing the importance of proximity, are taken out of context.

The 2020 NWPR preamble cites four examples where the agencies "considered scientific principles" in the definition of "waters of the United States:"

- 1. Use of streamflow duration classification in the definition of "tributary"
- 2. Definition of "adjacent" wetlands
- 3. Discussion of a hypothetical gradient of streamflow
- 4. A standard for assessing "normal" precipitation conditions ("typical year")

The first three examples use scientific terminology in their discussion rebutting public comments that the 2020 NWPR definitions and exclusions are not supported or informed by science but fail to apply the underlying scientific principles. *Id.* at 22271, 22288. The fourth example ("typical year") is informed by the traditional international standard for estimating "climate normals" but has significant flaws in its approach to evaluating "normal" precipitation conditions for assessing streamflow. The 2020 NWPR's inconsistent treatment or misapplication of long-standing scientific understanding in each of these four examples is described below.

a. Use of streamflow duration classification in the 2020 NWPR's regulatory definition of "tributary"

The 2020 NWPR includes many streams and reaches characterized by intermittent or perennial flows, but excludes all streams and reaches characterized by ephemeral flows as defined in that rule, stating that "this definition effectively furthers both the objective of the [Clean Water] Act to "restore and maintain the chemical, physical, and biological integrity of the nation's waters" and the "policy of Congress to recognize, preserve, and protect the primary responsibilities and rights of States to prevent, reduce, and eliminate pollution [and] to plan for the development and use (including restoration, preservation, and enhancement) of land and water resources . . ." 33 U.S.C. 1251(b); *see also Rapanos*, 547 U.S. at 737 (Scalia, J., plurality)." *Id.* at 22287-22288. The 2020 NWPR's assertion that limiting jurisdiction over tributaries to those that have at least intermittent flow in a "typical year" furthers the objective to "restore and maintain the chemical, physical, and biological integrity of the nation's waters" is not supported by science and represents a misuse of long-standing scientific principles.

The scientific principle of hydrologic connectivity, or the water-mediated transfer of matter, energy, and/or organisms within or between elements of the hydrologic cycle (Pringle 2003) applies to all streams. Streamflow duration is a fundamental component of hydrologic connectivity, and has been classified by the terms ephemeral, intermittent, and perennial. Hedman and Osterkamp 1982; Hewlett 1982. However, the importance of flows (*i.e.*, the functions and effects associated with flow) and the significance of their contributions to downstream waters, cannot be inferred from flow duration alone. Science Report.

Stream classification in the 2020 NWPR merely identifies three classes of flow duration – perennial, intermittent, and ephemeral. It says nothing about the functions or downstream effects of a given streamflow duration class. In applying this approach, the 2020 NWPR excludes relevant scientific information to the question of how those three classes of streamflow affect the integrity of downstream

waters that would be considered jurisdictional, and the mechanisms by which different degrees of hydrologic connectivity influence aquatic ecosystem function (further discussed above in this section).

Ward (1989) summarized the hydrologic connectivity of river ecosystems along four dimensions: longitudinal, lateral, vertical (surface-subsurface), and temporal connections. Degrees of hydrologic connectivity (*i.e.*, a hydrologic gradient) in a river network are established in each dimension by variations in the frequency, duration, magnitude, timing, and rate of change of flow. Poff *et al.* 2007. For example, functional methods for classifying streamflow consider the timing and magnitude of short-duration surface flows that transport large volumes of water and organic materials (*e.g.*, sediment) downstream (Nolan *et al.* 1987; Reid *et al.* 1995); changes in the frequency and duration of flow in response to regional climate, landscape, land use, and water use (Fritz *et al.* 2020); or the influence of water withdrawals, impoundments, and other human activities on shifting patterns of flow (*e.g.*, from perennial to intermittent or ephemeral) (Datry *et al.* 2014). For example, Gallo *et al.* (2020) quantified hydrologic discontinuities in streamflow and stream water presence in Arizona and found that, on a regional scale, stream channel density is a better predictor of streamflow and water presence than rainfall alone. In addition, their results showed that water presence as soil moisture and/or surface ponding can be 4–33 times greater than the duration of streamflow at the driest sites, which has important implications for biogeochemical processes in arid river systems. Gallo *et al.* 2020.

Hydrological connections along longitudinal (upstream-downstream), lateral (channel-tofloodplain or non-floodplain wetlands) and vertical (exchanges between surface and hyporheic flow) gradients are the physical backbone of healthy river networks. Direct evidence of hydrologic connectivity throughout river networks is apparent in the existence of stream channels that form the physical structure of the network itself. Science Report. Transitions in flow duration shape and re-form channels, distribute materials throughout the stream network, provide habitat for a wide diversity of plants and animals, and establish functional linkages between stream, wetland, and terrestrial ecosystems across space and through time. Datry et al. 2014. In addition to efficient transport of water and materials, infiltration from ephemeral flows recharges local aquifers that support riparian vegetation and animals (Science Report; Zimmer and McGlynn 2017) and regional aquifers that support baseflow in intermittent and perennial rivers (Min et al. 2020). Ephemeral flows also are important for survival and dispersal of aquatic organisms, including native fish species, during dry periods. De Jong et al. 2015. In the western U.S., 89% of streams are ephemeral or intermittent (Gallo et al. 2020) but even in forested watersheds, fieldbased surveys have shown that ephemeral channels can comprise the majority (up to 71%) of headwater stream miles (Hansen 2001; Fritz et al. 2013) based on physical channel extent. By categorically excluding an entire class of stream flows (ephemeral) from jurisdiction, the 2020 NWPR disaggregates flow-integrated stream networks, imposing arbitrary breaks in longitudinal, lateral, and vertical connections that support water quality, biodiversity, and ecosystem functioning in larger downstream waters.

Another line of evidence in the arbitrariness of the 2020 NWPR's definition of "tributary" is the inclusion of intermittent and perennial reaches upstream of ephemeral reaches when the connecting ephemeral reach flows at least once per year in a "typical year." Here the argument is that ephemeral reaches that flow at least once per year do provide a meaningful stream connection to downstream waters if the water and material could pass through a perennial or intermittent reach first, though the ephemeral reaches themselves continue to be excluded under the rule. The preamble attempts to "re-assemble"

portions of the stream network via the hydrologic connectivity established by ephemeral flows, while the rule text excludes all ephemeral streams that are instrumental in shaping and defining the river network from headwater sources.

Scientific methods for classifying flow duration (*e.g.*, as ephemeral, intermittent, perennial) assign conceptual transitions along the temporal gradient of channelized surface flows through a river network. Fritz *et al.* 2020. They support a variety of research and management objectives in river systems, which are characterized by time-varying flows that support different but equally important functions. Bracken *et al.* 2013; Science Report. The scientific principles used to develop flow classification methods do not support the notion that flows of one flow duration class are more ecologically important to downstream water integrity than flows in other classes.

There are many slight variations in the definitions of intermittent flow developed and used by scientists (*see, e.g.*, Busch *et al.* 2020), including those that specify ranges of flow duration (Fritz *et al.* 2020). None of the scientific definitions of flow duration class are based on effects on downstream water integrity. Rather, they reflect differences in application and characteristics of the location (*e.g.*, regional climate, geology, topography). Having defined durations of surface flow enables the use of USGS streamflow gage and other continuously monitored datasets for informing classifications. Because there are only three classes, most reaches will fall within specified ranges, but because flow duration itself is a gradient and the factors driving flow duration and its temporal variation are complex, there will be some reaches that fall near the interface between two flow duration classes and will be above a specified threshold one year but below it for another year, even when both years are considered "typical years." This interannual variation in streamflow permanence means that a given stream can appear to have ephemeral flow in one "typical year" and intermittent flow in a different "typical year." The 2020 NWPR does not address how to account for such interannual variation for assessing jurisdiction.

The 2020 NWPR's definition of "tributary" disproportionately excludes headwater streams, which are the major sources of water to river networks. Science Report. Intermittent and ephemeral streams conservatively account for 59% of the total length of streams in the contiguous United States, most of which are comprised of headwater networks. Nadeau and Rains 2007; *see also* Science Report at 2-29 (citing *id*.). A recent global model by Messager *et al.* (2021) estimated that 44-53% of stream reach length dries for at least 1 month per year, and that the wettest climate zone still had up to 30% of stream length being non-perennial whereas the driest climate zone had 99% of stream length being non-perennial flows activated during rainfall or snowmelt events are efficient at transporting water and materials (including permitted discharges) that accumulate in dry stream channels to downstream waters.

Changes in the spatial extent of stream networks due to expansion in response to precipitation and snowmelt and contraction during dry periods are most pronounced in the arid and semiarid Southwest, where more than 80% of all streams are intermittent or ephemeral. Levick *et al.* 2008. Godsey and Kirchner (2014) demonstrated the dynamism of headwater networks, mapping a 2.6 to 7.5-fold increase in both flowing network lengths and drainage densities in four drainages between fall (dry conditions) and spring (wet conditions). The dominant sources of water to a stream can shift during river network expansion and contraction in response to precipitation or snowmelt. Malard *et al.* 1999; McGlynn and McDonnell, 2003; McGlynn *et al.* 2004; Malard *et al.* 2006. The larger the rainfall or snowmelt event, the

more ephemeral streams flow, increasing the total length of channels contributing water and materials throughout the river network. Ephemeral flows cease within days after rainfall or snowmelt, and the flowing portion of the river network seasonally shrinks as the spatial extent of aquifers in contact with streams contract, precipitation-evapotranspiration balance shifts, and intermittent streams dry. In many river systems across the United States, stormflow comprises a major portion of annual streamflow. Hewlett *et al.* 1977; Miller *et al.* 1988; Turton *et al.* 1992; Goodrich *et al.* 1997; Vivoni *et al.* 2006. In these systems, intermittent and ephemeral streams are major sources of river water. Science Report at 2-18.

The propagation of stormflow through river networks also provides clear evidence of hydrologic connectivity between ephemeral, intermittent, and perennial headwater streams to rivers, particularly when an intense storm occurs over only the headwater portions of a river network. The contribution of tributaries to rivers during widespread floods manifests as stepped increases in discharge immediately below confluences, as water flows accumulate through a river network. Science Report at 3-7. Such propagation was recorded following a monsoonal storm event through an arid network of ephemeral channels in the watershed of the Rio Puerco River, a semiarid tributary to the Río Grande River, in the early 2000s. Vivoni et al. 2006. A storm dropped approximately 18–25% of annual rainfall on the Rio Puerco's approximately 16,000 square kilometer (approximately 6,200 square miles) drainage area over a two-day period. Discharge recorded at two USGS gages on the Rio Puerco and three USGS gages on the Río Grande downstream of the confluence illustrated sequential time lags in peak flows from the upstream gage to the downstream gage, demonstrating the downstream transfer of flows from headwaters, and increases during peak flows at least 127 km (approximately 79 miles) downstream. Id. Stormflow contributions from the ephemeral Rio Puerco accounted for 76% of flow at the Río Grande during this event, although just 3.6% of rainfall resulted in tributary runoff. A total of 49% of the flood volume recharged shallow aquifers in the mainstem of the river and the remainder entered the reservoir at the outlet. Id.; Science Report at 3-7, 3-8. However, this river, if it meets the 2020 NWPR's definition of ephemeral, would have been excluded under the 2020 NWPR despite the large quantity of flow that is provided downstream to the Río Grande during a storm event that is likely to occur annually.

The 2020 NWPR is inconsistent in its consideration of physical indicators of flow. 85 FR 22292. As noted above, physical indicators can be evidence of flows that provide important functions to larger downstream waters, including traditional navigable waters, the territorial seas, and interstate waters. The channel form and other fluvial geomorphic features are physical indicators of the recurrent, concentrated surface flow of water through the stream network to downstream waters. Without the recurrent transport of water and materials, terrestrialization (soil development, terrestrial vegetation colonization/growth) transforms channels to uplands, making them indistinguishable from the surrounding landscape. The 2020 NWPR preamble notes the value of physical indicators as a line of evidence but excludes them on the basis that they cannot be used to evaluate flow duration under "typical year" conditions (see section II.B.iv.1 for discussion of the 2020 NWPR's use of "typical year"). As noted above, flow magnitude (e.g., flooding, transfer of stored materials in high-magnitude, short-duration events) can have large effects on downstream waters. Physical indicators often reflect transport magnitude to a stronger degree than transport duration because indicators of intermediate and lower magnitude flows are often effaced by higher flows. In addition to representing responses, physical indicators can be hydrologic drivers and be represented at not only the local, reach scale, but larger scales (e.g., watershed, regional). Costigan et al. 2016.

If the scientific justification for limiting jurisdiction to intermittent and perennial reaches is that they are fed primarily by groundwater or snowpack meltwater - as the 2020 NWPR's definition of "tributary" indicates – the 2020 NWPR fails to address two essential points: (a) all three of the 2020 NWPR's streamflow duration classes convey stormflow and materials downstream, and (b) individual streams often transition longitudinally between flow duration classes, from ephemeral to intermittent to perennial, creating patchworks of ephemeral, intermittent, and perennial reaches within a single segment or tributary of a stream network. Science Report. The 2020 NWPR preamble states that the "regular and predictable" hydrologic behavior—*i.e.*, intermittent and perennial flows within a "typical year," however they are distributed within and across tributaries-qualifies them for Clean Water Act jurisdiction. 85 FR 22278. By its regulatory definition of jurisdictional streams (i.e., "tributaries") based on flow classes, the 2020 NWPR attests that the scope of the Clean Water Act is therefore determined largely by conditions that reflect snowpack and groundwater inputs to streamflow. In effect, the 2020 NWPR preferentially regulates streams in those locations having climate and landscape conditions that support persistent or seasonal flow or inundation—conditions that do not exist nationally and that are subject to predictable changes in climate patterns over the next 30+ years. It also favors protection for streams in humandominated watersheds that receive a significant proportion of their baseflow from municipal and industrial wastewater effluent discharges, stormwater detention basins, and irrigation return flow, all of which are more likely to be impaired than streams without such augmented source water and would otherwise be non-perennial (and many of which, particularly in the arid West, would otherwise by ephemeral and excluded under the 2020 NWPR). Streams whose instream flows are entirely dependent on effluent discharges are called effluent-dependent streams, whereas those that receive most, but not all, of their flow from effluent are called effluent-dominated streams. Brooks et al. 2006. About 25% of permitted effluent discharges in the United States enter streams with mean annual flows incapable of diluting effluents by more than 10-fold. This percentage of permitted effluent discharges entering streams incapable of diluting effluents by more than 10-fold increases to 60% when low-flow discharge is considered. Brooks et al. 2006; Science Report at 3-12, 3-13. While effluent-dependent or effluentdominated streams can have unique water quality characteristics (Brooks et al. 2006), there is also no scientific justification to regulate them and to not also regulate ephemeral streams, as the 2020 NWPR does.

b. Definition of "adjacent wetlands"

The 2020 NWPR preamble incorrectly cites the Science Report by quoting partial sentences out of context. For example, the 2020 NWPR preamble notes that "areas that are closer to rivers and streams have a higher probability of being connected than areas farther away" *Id.* at 22314 (citing the Science Report at ES-4). However, the complete sentence text in the Report is: "In addition, although areas that are closer to rivers and streams have a higher probability of being connected than areas farther away when conditions governing the type and quantity of flows—including soil infiltration rate, wetland storage capacity, hydraulic gradient, etc.—are similar, information to determine if this similarity holds is generally not provided in the studies we reviewed." Science Report at ES-4. The Science Report states elsewhere that, "*All factors being equal*, wetlands closer to the stream network will have greater hydrologic and biological connectivity than wetlands located farther from the same network." *Id.* at 2-40 (emphasis added). However, proximity is not a factor in the 2020 NWPR's definition of "adjacent

wetlands." Wetlands that are very close to the stream network are not considered "adjacent wetlands" under that rule unless they have the very specific hydrologic connections that the 2020 NWPR prescribes for "adjacent wetlands."

In the same section, the 2020 NWPR preamble also cites the SAB Review as saying that "[w]etlands that are situated *alongside* rivers and their tributaries are likely to be connected to those waters through the exchange of water, biota and chemicals. As the distance between a wetland and a flowing water system increases, these connections become less obvious." 85 FR 22314 (citing SAB 2014a at 55) (emphasis added in 2020 NWPR preamble). In the SAB Review, this sentence opened the section in which the SAB discusses the importance of "less obvious" connections from less-proximal (*i.e.*, non-floodplain) wetlands that influence downstream waters.

The 2020 NWPR preamble also quotes fragments of text from the Science Report and the SAB Review incorrectly and out of context in its discussion of limiting Clean Water Act jurisdiction to wetlands having a "near-permanent" hydrologic connection to a jurisdictional water in a "typical year." The 2020 NWPR defines "adjacent wetlands" as wetlands that:

(i) Abut, meaning to *touch at least at one point or side of*, a water identified in paragraph [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water];

(ii) Are *inundated by flooding* from a [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water] in a typical year;
(iii) Are physically separated from a [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water] only by a natural berm, bank, dune, or similar natural feature; or

(iv) Are physically separated from a [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water] only by an artificial dike, barrier, or similar artificial structure so long as that structure allows for a *direct hydrologic surface connection* between the wetlands and the [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water] in a typical year, such as through a culvert, flood or tide gate, pump, or similar artificial feature. An adjacent wetland is jurisdictional in its entirety when a road or similar artificial structure divides the wetland, as long as the structure allows for a direct hydrologic surface connection through or over that structure in a typical year.

Id. at 22338 (emphasis added). There is no scientific or conventional context in which proximity can be interpreted as having a direct or continuous surface water connection. When taken properly in context with scientific evidence, spatial proximity is one of multiple important physical factors influencing ecological exchanges of water, materials and biota between streams, wetlands, open waters, and downstream waters. Other factors cited in the Science Report and related publications are the individual size, number, climate, geology, terrain/slope, land use/land cover, distinctiveness, and intervening units. Fritz *et al.* 2018 at Table 2 (Key factors affecting connectivity from streams and riparian wetlands to downstream waters and its resulting effects). In addition, although the 2020 NWPR preamble cites the SAB Review as support for limiting jurisdiction to wetlands at close spatial proximity to "waters of the United States," the rule makes no reference to spatial distance - the ordinary interpretation of spatial proximity - in its definition of "adjacent wetlands." Under the 2020 NWPR, wetlands located at considerable distance from lakes or rivers could be jurisdictional, whereas more spatially proximate wetlands might not. For

example, under the 2020 NWPR, normal expansion of lakes in the Prairie Pothole Region that merge with previously disconnected wetlands via continuous surface water at distances up to 38 km (Vanderhoof and Alexander 2016) could be jurisdictional whereas coastal, riparian, and floodplain wetlands that are in fact directly spatially proximal to "waters of the United States" have been found to be non-jurisdictional.

Further, wetlands running alongside a tributary have been found to be non-jurisdictional on the basis of factors such as channel incision, build-up of bank materials from past floods that do not meet subjective criteria for a natural berm, or absence of evidence showing a single-point of "near continuous" surface connection to the tributary. This also contradicts the emphasis added in the 2020 NWPR preamble to the SAB Review's comment about wetlands that run "*alongside* of rivers and their tributaries" cited above.

The 2020 NWPR preamble states:

The final rule also provides that wetlands separated from jurisdictional waters only by a natural berm, bank, dune, or other similar natural feature are adjacent wetlands. These natural features are indicators of a sufficient hydrologic surface connection between the jurisdictional water and the wetland, and the agencies conclude that wetlands that are separated from jurisdictional waters only by such features are inseparably bound up with the adjacent jurisdictional waters and are therefore "part of those waters."

Id. at 22280 (citing *Rapanos*, 547 U.S. at 740 (Scalia, J., plurality). The preamble goes on to say that "Physically remote isolated wetlands (*i.e.*, wetlands that do not abut, are separated by more than a natural berm from, are not inundated by flooding in a typical year from, and do not have a direct hydrologic surface connection in a typical year to a jurisdictional non-wetland water) are not adjacent wetlands under the final rule." *Id.* The 2020 NWPR consequently misapplies the term "physically remote" to wetlands that are not remote in any ordinary sense of the word, and/or not "isolated" in any scientific sense from a jurisdictional water. Marton *et al.* 2015; Fossey and Rousseau 2016; Rains *et al.* 2016; Cohen et al 2016; Creed *et al.* 2017, Evenson *et al.* 2018; Lane *et al.* 2018; Golden *et al.* 2019.

Similar to streams, the occurrence and persistence of riparian/floodplain wetland and nonfloodplain wetland hydrologic connections with river networks, via surface water (both channelized and non-channelized) or shallow or deep groundwater, can be continuous, seasonal, or ephemeral, depending on the overall hydrologic conditions in the watershed. For example, a wetland might have a direct shallow subsurface connection with a river network during wet conditions but an indirect regional groundwater connection (via groundwater recharge) under dry conditions. Non-floodplain wetlands can be hydrologically connected to the river network via non-channelized surface flow (*e.g.*, swales or overland flow) (*see, e.g.*, Rains *et al.* 2006; Wilcox *et al.* 2011) or subsurface flows that support river baseflow (Science Report at 2-14, 4-2. Therefore, the assertion that they do not have a "direct hydrologic surface connection in a typical year to a jurisdictional non-wetland water" is erroneous. In fact, wetlands that outflow to streams seasonally through swales that likely would have met the definition of "waters of the United States" prior to the 2020 NWPR were no longer jurisdictional under the 2020 NWPR. However, in some cases the swales themselves – which formerly were not "waters of the United States"—were found to be jurisdictional under the 2020 NWPR. This example illustrates how jurisdiction based entirely on flow duration and flow direction and the application of 2020 NWPR categories of "adjacent wetlands" run counter to scientific understanding of stream-wetland connectivity and function in ways that subvert the objective of the Clean Water Act.

The 2020 NWPR's preamble incorrectly claimed that the rule's limits on wetlands that are jurisdictional were consistent with "longstanding practice." 85 FR at 22280. The preamble claimed consistent with longstanding practice "wetlands can be jurisdictional only if they are adjacent to territorial seas or a traditional navigable water, tributary, lake, pound, or impoundment of a jurisdictional water but the rule's definition of "waters of the United States" was not consistent with which categories of wetlands are regulated under the agencies' pre-2015 regulatory regime. For example, the 2020 NWPR did not include interstate waters as a separate category of jurisdiction even though such waters had been regulated for decades prior to the 2020 NWPR. As a result, interstate wetlands were not included as "waters of the United States" unless they met one of the rule's categories of jurisdictional waters. Similarly, wetlands adjacent to interstate waters were not be regulated under the 2020 NWPR unless they met one of the four paragraph (a) waters under that rule, which is also counter to longstanding practice. In addition, the definition of "adjacent wetlands" under the 2020 NWPR was modified to such an extent that even wetlands "running alongside" perennial streams and rivers were found to be non-jurisdictional under that rule. For example, in one approved jurisdictional determination, an area of pine flatwoods in the Mississippi Coastal Plain, the area of adjacent wetlands was reduced from 144 acres under the 2015 Clean Water Act to 9.5 acres under the 2020 NWPR. In another example, a heavy mineral sands mining project in Georgia withdrew its permit for development under the 2015 Clean Water Rule, which found 400 acres on the site near the Okefenokee NWR to be jurisdictional, and re-applied under the 2020 NWPR, which reduced the number jurisdictional wetlands acres to zero acres. At least one Corps district saw an eightfold increase in the number applications for Approved Jurisdictional Determinations (~50/year to ~400/year) in the first year of 2020 NWPR implementation, and 3,300 acres of coastal plain wetlands were determined to be non-jurisdictional over that period. Although the agencies generally have not used paragraph (a)(3) of the pre-2015 regulations to assert jurisdiction over "other waters" since the decision in Solid Waste Agency of Northern Cook County v. U.S. Army Corps of Engineers, 531 U.S. 159 (2001) (SWANCC), the Supreme Court did not invalidate that regulation. To advance the objective of the Act, and in accordance with Supreme Court precedent and the agencies' experience and expertise, the final rule provides that intrastate waters that do not meet the criteria set forth in paragraphs (a)(1) through (a)(4), (now listed under paragraph (a)(5) of the final rule) must meet either the relatively permanent standard or the significant nexus standard in order to be jurisdictional.

c. The 2020 NWPR's standard for assessing normal climate conditions ("typical year")

The 2020 NWPR's concept of "typical year" is fundamental to jurisdiction in that some regulatory definitions (*e.g.*, of a tributary, certain adjacent wetlands, and certain lakes, ponds, and impoundments of jurisdictional waters) apply only if flows are observed "when precipitation and other climatic variables are within the normal periodic range (*e.g.*, seasonally, annually) for the geographic area of the applicable aquatic resource based on a rolling thirty-year period." 85 FR 22274. Therefore, the 2020 NWPR's definition of "typical year" is a linchpin of jurisdictional decisions under the rule.

(i) Applicable Period of Record

The 2020 NWPR definition of "typical year" draws from an international standard for calculating and reporting "climate normals" – *i.e.*, a suite of statistics for temperature, precipitation, and other climatological variables from U.S. weather stations – established in the mid-1940s by the International Meteorological Organization (IMO; predecessor of the World Meteorological Organization (WMO)). A 30-year averaging period was selected by IMO because at the time, only 30 years of global climate data were available. WMO 2017. In addition, 30 observations were considered sufficient for estimating average values under relatively stable climate conditions (stationarity, *i.e.*, when temperature, precipitation, and other climate variables fluctuate seasonally or annually above or below a constant but do not show positive or negative trends) or when changes in climate patterns are small or slow-moving. WMO 2007; Wilks 2013.

One drawback of relying only on a 30-year averaging period to estimate "typical" conditions is that climate conditions in the United States are not stable. Consistent regional and national trends in temperature, precipitation, and other climatological variables are rapidly shifting climate averages (NOAA a; data available at <u>https://www.ncei.noaa.gov/access/monitoring/us-trends/</u>), and climate extremes are becoming more intense, frequent, and/or persistent (data available at <u>https://www.climate.gov/maps-data/dataset/severe-storms-and-extreme-events-data-table</u> and <u>https://www.ncei.noaa.gov/products/severe-weather</u>)</u>. These trends have been apparent in records dating back to the start of the 20th century and have become more pronounced over the past 30 years. Steinacker 2021. Both types of change – trends in temperature/precipitation and more frequent/intense climate extremes – invalidate assumptions of climate stationarity and increase error in estimation of "normals" based on 30-year averaging intervals. Arguez *et al.* 2019. While important as long-term benchmarks of climate change, in the early 2000s WMO 30-year climate normals were found to be "no longer generally useful for the design, planning, and decision-making purposes for which they were intended." Livezey *et al.* 2007 at 1759.

In response, the National Oceanic and Atmospheric Administration (NOAA) has developed several alternatives (optimal climate normals; OCN; Huang *et al.* 1996) to account for monotonic climate change. OCN include seasonal averages computed using 10-year annually-updated averaging periods for temperature and 15-year annually-updated averaging periods total precipitation. Arguez *et al.* 2019. In addition, NOAA OCN are adjusted for systematic interannual variability (*e.g.*, El Niño–Southern Oscillation; ENSO) by conditioning climate normals on the phase and intensity of ENSO events in five sets of climate normals, thereby accounting for climate anomalies, as well as background climate trends, in estimates of temperature and precipitation "normals." *Id.*

In general, shorter periods and annual updating improve the performance of climate normals in the presence of rapid trends, although determining the "best" period of record over which trends are evaluated for any given location remains challenging because optimal intervals vary by effect (*e.g.*, temperature, total precipitation) and time of year. Huang *et al.* 1996. In recent decades, NOAA has recommended that users of conventional benchmarks also consider alternative benchmarks estimated using annual updates over shorter fixed (5, 10, 15, 20-year) intervals or variable optimized intervals that are automatically adjusted to provide more accurate information for a particular time and location. NOAA 2019. Starting in May 2021, NOAA began publishing 15-year NOAA climate normals in tandem with 30-

year WMO climate normals, now updated each decade, as a standard practice. NOAA 2021; NOAA 2022b.

The 2020 NWPR improves upon the historical WMO approach by using a rolling 30-year interval updated annually. The preamble states that the agencies "considered other alternative [averaging] time periods" for assessing "normalcy" but maintained the 30-year record because it was established "[n]early a century ago." 85 FR 22274. The only justification given for retaining a 1940s-era standard is a reference to the NOAA webpage linking users to information about alternatives, including OCN, estimated over shorter averaging time periods to account for rapid changes in average climate conditions. *Id.* at 22274-22275, citing NOAA 2019. While the 2020 NWPR's approach is slightly preferable to the historical WMO approach, a more flexible approach that allows for the use of 10-year or 15-year climate normals, such as those published by NOAA, would generally be better in light of rapidly changing climatic conditions in certain parts of the country

(ii) Accuracy of Prediction Based on Historical Precipitation Quantiles

While agreeing that precipitation is just one of many important drivers of streamflow and wetland inundation, the 2020 NWPR preamble uses precipitation total as the sole basis for quantifying a "typical year" by relying on "the normal range of precipitation," estimated for given location as follows: "The agencies evaluate normal precipitation conditions based on the three 30-day periods preceding the observation date. For each period, a weighted condition value is assigned by determining whether the 30-day precipitation total falls within, above, or below the 70th and 30th percentiles for totals from the same date range over the preceding 30 years. The agencies make a determination of 'normal,' 'wetter than normal,' or 'drier than normal' based on the condition value sum." *Id.* at 22274. The preamble provides no justification for using this method to assess normalcy, except that it draws from methods developed by the National Resource Conservation Service (NRCS) and the Corps for assessments of *wetland* hydrology.

To the agencies' knowledge, the "typical year" concept had not been validated for accuracy (reliability) of predicted assessments of precipitation, or the effects of precipitation on stream hydrology for that location prior to finalization of the 2020 NWPR. Other key factors driving variations in streamflow include evapotranspiration, temperature, available water capacity in soils, snowpack conditions, groundwater conditions, and water withdrawals, all of which vary interannually but are unaccounted for in the 2020 NWPR definition of "typical year." A recent study by the Corps found that precipitation normalcy (as calculated based on the methodology described in the preamble to the 2020 NWPR) was neither a reliable predictor of streamflow normalcy, nor was it a precise predictor of streamflow percentiles, in an analysis of watersheds across the United States. Sparrow *et al.* 2022.

Further, the 2020 NWPR provides no evidence to support its assumption that "normal precipitation" across the country is accurately captured by the middle 40% of the precipitation totals recorded over the 90-day date range of interest over the 30-year period of record. Nor does it provide evidence to justify the exclusion of 60% of precipitation events because they are assumed to be "too wet" or "too dry" for jurisdiction. *Id.* The uncertainty associated with the 2020 NWPR "typical year"

guidelines is particularly high for areas currently facing rapid changes in climate variables (*e.g.*, including in the Midwest, where 100-year floods occur with more frequency in a "typical year") or climate anomalies caused by cyclic events (*e.g.*, ENOS) create conditions in which emphasis on the middle range of precipitation totals is particularly prone to over- or under-estimation of normal conditions. Given the 2020 NWPR's heavy reliance on "typical year" for Clean Water Act jurisdiction, the 2020 NWPR could have required that estimates of uncertainty associated with weighting the middle 40% of precipitation observations across the United States be provided as part of a jurisdictional determination, along with probabilities that recent events fall above, within, or below the 30th and 70th percentiles. Considering only 40% of prior events also can eliminate channel-forming events, including ephemeral stormflows that transport large volumes of water and materials to downstream waters (*e.g.*, the Rio Puerco and Río Grande example cited in section II.B.iii.1.a).

The 2020 NWPR's preamble alludes to the inadequacy of that rule's primary method for assessing "typical" climate by stating that the agencies "recognize there may be other accurate and reliable measurements of normal precipitation conditions" and places a burden of proof on those seeking more scientifically-defensible methods to assess "normal" conditions by stating that "alternative methods" must be "developed and appropriately validated, including different statistical percentiles, evaluation periods, or weighting approaches for condition values." 85 FR 22274. No such comparable support for the rolling 30-year average is provided in the preamble or elsewhere in the public record. In practice, the difficulty of finding, assembling, integrating, validating, and analyzing datasets that could provide a more accurate estimate of climate "normals" presents a high bar for implementing robust and reliable methods. As a result, most assessments of "typical year" resorted to the use of 30-year precipitation records as recommended in the 2020 NWPR preamble.

Lastly, while precipitation is an important predictor of streamflow (Eng *et al.* 2016; Jaeger *et al.* 2019; Konrad 2019), there is no support for relating the "typical year" precipitation "condition value sum" to flow behavior at the basin, tributary, or reach scale. Thus, while claiming to provide a "predictable, implementable regulatory framework" for regulating tributaries, the 2020 NWPR methods add uncertainty to assessments of flow duration class (perennial, intermittent, or ephemeral), use standards for assessing climate normals ("typical year") that have limited ability to assess current or nearfuture conditions, and in practice, depend on historical records of streamflow or wetland inundation that frequently do not exist.

d. Jurisdiction of lakes, ponds, impoundments, wetlands by oneway inundation

The 2020 NWPR preamble states that inundation by flooding of lakes (including oxbow lakes), ponds, impoundments of jurisdictional waters, and "adjacent wetlands" from jurisdictional waters in a "typical year" is "sufficient to establish jurisdiction [when it] occurs only in one direction, from the [traditional navigable water or territorial sea, tributary, or lake, pond, or impoundment of a jurisdictional water] to the lake, pond or impoundment of jurisdictional waters, rendering the feature 'itself a part of those waters' 'that are 'waters of the United States' in their own right." 85 FR 22303 (citing *Rapanos*, 547 U.S. at 740, 742 (Scalia, J., plurality). This requirement is inconsistent with the science on the

importance of flows that occur in the other direction—from a wetland or open water to the tributary network—and from other types of hydrologic connections.

Wetlands and open waters in riparian areas and floodplains are physically, chemically, and biologically integrated with rivers and other jurisdictional water via functions that can benefit from, but are not dependent upon, near-permanent, continuous surface water connections established by outflows from jurisdictional waters. Wetlands improve water quality of nearby jurisdictional waters by transformation and/or sequestration of pollutants that can degrade water integrity, intercept sediment, and store local ground water for late-season baseflow in rivers. They also provide breeding and nursery habitat for fish, amphibians, and aquatic insects that are integral components of riverine, lacustrine, and estuarine food webs. These facts are all supported by the science synthesized in the Science Report.

Long established aquatic science considers streams and rivers to have four dimensions – longitudinal connections between upstream and downstream reaches, vertical connections between the surface and subsurface, lateral connections between the river and its floodplain, and temporal connections that occur over time. Ward 1989. Lateral connections between lakes or river channels to floodplains, including wetlands and open waters, can occur through seasonal or episodic expansion and contraction of river networks or through outflows from wetlands and other surface waters to lakes and streams. These lateral connections that contribute to the chemical, physical, and biological integrity of downstream waters. For example, these wetlands can attenuate stormflow, increase baseflow, be a source of carbon and organic matter, and a sink for sediment, nitrate, and other constituents that degrade water quality. Science Report.

As discussed in the Science Report:

Riparian/floodplain wetlands can be hydrologically connected to streams and rivers through unidirectional flows (*i.e.*, from wetlands to rivers and streams, but not vice versa) of surface water and ground water from upgradient areas (*e.g.*, hillslopes and nearby uplands). In addition, riparian/floodplain wetlands have bidirectional connections to streams and rivers (*i.e.*, from wetlands to streams and rivers and vice versa) through lateral movement of surface and ground water between the channel and riparian/floodplain areas. Connections between riparian/floodplain wetlands and streams or rivers occur over a gradient of connectivity, for example, they can be permanent, can occur frequently (*e.g.*, if the wetland is located within the mean high-water mark), or can occur infrequently (*e.g.*, if the wetlands that rarely flood can have important, long-lasting effects on streams and rivers. Riparian/floodplain wetlands can reduce flood peaks by storing floodwaters, store large amounts of sediment and nutrients from upland areas, influence stream geomorphology by providing woody debris and sediment, and regulate stream temperature. Riparian/floodplain wetlands also are sources of food for stream and river invertebrates and serve as rearing habitat for fish."

Science Report at 4-1. Further, the Report states:

Wetlands in non-floodplain landscape settings lack bidirectional hydrologic connections with channels (*i.e.*, water flows from the wetland to the channel but not from the channel to the wetland). These settings, however, have the potential for unidirectional hydrologic flows from wetlands to the river network through surface water or ground water. Non-floodplain wetlands can attenuate floods through depressional storage and can recharge ground water and thereby contribute to baseflow. These wetlands can affect nutrient delivery and improve water quality by

functioning as sources (e.g., of dissolved organic carbon) and as sinks for nutrients (e.g., nitrogen), metals, and pesticides. Non-floodplain wetlands also can provide habitat or serve as sources of colonists for biological communities in downstream waters, through movement of amphibians, reptiles, birds, and mammals. The extent to which non-floodplain wetlands perform these functions depends on their hydrologic and biological connectivity with downstream waters. Non-floodplain wetlands also occur on a hydrologic gradient, from wetlands having permanent connections with perennial channels, to geographically isolated wetlands having groundwater or occasional surface-water connections, to highly isolated wetlands having minimal hydrologic connection to the river network (but which could include surface and subsurface connections to other wetlands; Section 4.4.2 [of the Science Report]). Non-floodplain wetlands that are connected to the river network through a channel (*i.e.*, wetlands that serve as stream origins) will have an effect on downstream waters, regardless of whether the outflow is permanent, intermittent, or ephemeral. For non-floodplain wetlands that do not connect to the river network through a stream channel (*i.e.*, geographically isolated wetlands and wetlands that spill into losing streams that are completely disconnected from the river network), the type and degree of connectivity with downstream waters will vary with position in the watershed and over time.

Id. at 4-1 to 4-2. The limitation in the 2020 NWPR results in wetlands and open waters that affect that chemical, physical, and biology integrity of traditional navigable waters, the territorial seas, and interstate waters being categorically excluded from jurisdiction.

Restricting jurisdiction to those wetlands with only one-way flow from jurisdictional waters to other types of waters is inconsistent with long-standing practice. For example, the *Rapanos* Guidance (p. 5) clearly states that wetlands can be adjacent if "there is an unbroken surface or shallow subsurface connection to jurisdictional waters." This criterion for a surface connection in the guidance accounts for other types of hydrologic connections than does the 2020 NWPR. In addition, the Guidance also states that wetlands are adjacent if "their proximity to a jurisdictional water is reasonably close, supporting the science-based inference that such wetlands have an ecological interconnection with jurisdictional waters." Rapanos Guidance at 5-6. Such proximity does not require inundation by flooding or other types of surface connections from the wetland to the jurisdictional water. Under the 2020 NWPR, seasonal swales (or "vernal swales") as opposed to vernal wetland pools were found to be jurisdictional as "adjacent wetlands." Vernal wetland pools that outflow to stream networks that would be jurisdictional under the pre-2015 practice as adjacent wetlands were found to not meet the definition of "adjacent wetlands" under the 2020 NWPR and thus were found to be non-jurisdictional. Swales are important features of vernal pool wetland complexes, and were found to be jurisdictional as adjacent wetlands under pre-2015 practice, as well as the vernal wetland pools. They are hydrologically connected to the pools and often support similar plant communities as vernal pools, but typically do not pond water long enough to support large ESA-listed branchiopod or amphibian species. For the same reason, they are also very dynamic systems and may not reliably delineate as three-parameter wetlands in any given year. Mitigation bankers are required to include interconnecting swales in vernal pool restoration plans, but the swales are often utilitarian (e.g., constructed to help maintain hydrology at the stie) and bankers do not request credit for them since they are inherently risky and not a highly-desired credit type. So, while vernal swales are important, it is ironic and arbitrary that they are now considered the most important wetlands in vernal pool complexes under 2020 NWPR, whereas the vernal wetland pools, which provide even more functions affecting traditional navigable waters, the territorial seas, or interstate waters, are considered non-jurisdictional under that rule.

2. White Paper

As part of the administrative record for the 2020 NWPR, the agencies added to the docket a white paper entitled "Limitations of the National Hydrography Dataset at High Resolution and the National Wetlands Inventory and their use for Determining the Scope of Waters Subject to Clean Water Act Jurisdiction." EPA and Army 2020b, hereafter "White Paper." The agencies used the White Paper in part to support their arguments at the time that the USGS National Hydrography Dataset (NHD) and the U.S. FWS National Wetlands Inventory (NWI) were inappropriate to use on a national level to estimate the 2020 NWPR's potential effect on the extent of waters that would no longer be jurisdictional under the rule, particularly as standalone datasets. While the White Paper was factual in stating that the datasets were not designed as regulatory datasets and do not explicitly depict the full geospatial scope of Clean Water Act jurisdiction, based on further analysis and interagency review the agencies have determined that the datasets, including a disproportionate focus on limitations of the datasets, but failed to adequately consider the positive value of the datasets and the breadth of the available literature surrounding both datasets.

a. Background

The NHD and the NWI are the most comprehensive and detailed hydrography and wetlands datasets for the nation and are the most accurate national datasets at the spatial scale that is relevant to Clean Water Act decision-making. Despite being the most comprehensive available datasets of their kind, however, neither the NHD or NWI were designed to be regulatory datasets, both have certain known limitations, and neither can be used as a standalone tool to determine the full scope of Clean Water Act jurisdiction. Additionally, the definitions that the datasets use may differ from regulatory definitions under the Clean Water Act (*e.g.*, the NWI's Cowardin definition of "wetlands" is broader than the regulatory definition). As Federal Geographic Data Committee (FGDC) National Geospatial Data Assets (NGDAs) that support a broad range of users and applications, it is important that these datasets as able to form the foundation of a decision support system that overlays regulatory-related information (*e.g.*, location of traditional navigable waters, modeled flow permanence and hydrologic connectivity, and approved jurisdictional determinations).

As for any rulemaking, accurately estimating the potential effects of a proposed or final action can be challenging, and a rule defining the scope of Clean Water Act jurisdiction is no exception. In a rulemaking, in the absence of precise data for all cases, the agencies typically use the best available data to estimate the direction and magnitude of potential effects of a rule. For purposes of assessing the effects of revising the definition of "waters of the United States," the agencies in their economic analysis have often relied on data from the Corps' ORM database regarding where jurisdictional determinations and Clean Water Act section 404 permits have been issued. *See* 2015 Clean Water Rule EA; 2020 NWPR EA; Economic Analysis for the Final Rule. Because the 2020 NWPR as proposed was assumed to reduce the scope of jurisdictional waters compared to the legal status quo, the agencies initially attempted to also utilize NHD and NWI to estimate the potential effects of that proposed rule. Proposed 2020 NWPR RPA;

2020 NWPR RPA. However, at that time the agencies determined that technical limitations of the datasets presented significant challenges for the purpose of determining potential effects of the proposed and final 2020 NWPR. *Id.*

b. The White Paper Presented Flawed Arguments

The limitations of the NHD and NWI noted in the White Paper can be placed into two basic categories. The first category would include limitations due to the spatial resolution. For example, streams of small length (e.g., many headwater streams) are absent from high-resolution NHD maps drawn at 1:24,000 scale. Positional accuracy is also a function of the spatial resolution of the data. The White Paper, however, contains a general lack of conceptual understanding of spatial resolution and map scale and incorrectly lists numerous examples of errors or "inaccuracies" that are in fact differences due to spatial resolution and appropriate use of spatial data sources. A map is not inaccurate if it does not include information that is more detailed than the map scale is meant to depict. The exclusion of features is purposeful. The reason for excluding features is to prevent overcrowding of the map so features at the map's scale can be clearly depicted. Again, based on this fact the number and length of streams, for example, are not underrepresented for the given map scale. In addition, it is impractical to delineate in fine detail the bends in a river for a 1:24,000 scale map. These limitations should not be characterized as inaccuracies but instead are a function of the scale at which they are drawn. The way this is written in the White Paper implies that the issue is with the dataset, when the real issue is with using the dataset for a more detailed purpose than supported by the map scale. Thus, the section on limitations is incorrectly framed and should instead have been a discussion on appropriate spatial resolutions for estimating stream and wetland location and extent in jurisdictional determinations.

The second category of limitations identified in the White Paper includes data accuracy, including data entry errors. Data entry errors include incorrect or missing attribute information such as a waterbody name, feature type, flow direction, or flow classification. The USGS manages a robust data stewardship program that empowers state and federal agencies to correct mistakes within the NHD. Similarly, the NWI data are regularly updated and maintained.

Because the White Paper fails to describe application requirements and state key definitions clearly or quantitatively, it is impossible to thoroughly assess the validity of position statements contained therein. For example, the definition of "reliable" data is not properly defined, and this presents challenges when addressing the validity of the positions taken in the paper. The expression and use of "error" and "accuracy" are used widely and improperly throughout the White Paper. A definition of these terms would be necessary to determine whether a dataset meets the degree of accuracy required for a specific application of that dataset. The comparisons made in the White Paper between the NHD and NWI and validated field data do not follow established scientific protocols for determining level of "error" or "accuracy" of remotely sensed datasets. *See, e.g.*, Congalton and Green 1990. A fundamental principle of accuracy assessment is that the reference (field) dataset must measure the same thing as the dataset being assessed (validated) and do so with a higher degree of certainty. *Id.* In the studies cited in the White Paper, a different classification system is being applied to the reference dataset and the dataset being assessed (*i.e.*, NHD or NWI). For example, the White Paper cites to Colson *et al.* (2008) as supporting that there are horizontal positional inaccuracies in the NHD dataset, but the authors used a different metric (three meters) than the metric used by the USGS (12.2 meters) and thus it is not appropriate to cite

this study in reporting inaccuracies. In addition, another analysis the White Paper cites compared NWI data from the early 1980s to field data from 2003-2010 (Wu et al. 2014a); the study did not assess NWI accuracy and thus the White Paper erred in including it in the section titled "NWI Has Certain Errors." Wu et al. (2014a, p. 27) specifically states, "Due to differences in definitions and techniques, there will be discrepancies in the determination of the wetland status of sampling sites when using the NWI and [Corps] methods. This study *did not attempt to assess the accuracy of either method* but to conduct a differential assessment of the results obtained when using these two methods" (emphasize added). Wu et al. (2014a, p. 28) also provides a more balanced perspective than the White Paper, acknowledging, "In addition, human errors from both map producers and users might also affect accuracy of the NWI map data. However, at least some of the factors above can also affect the accuracy of [Corps] field interpretations." Similarly, Matthews et al. (2016) compared areas using two different wetland definitions and therefore should not be used to determine NWI "error" or "accuracy." In addition, while errors are highlighted throughout the White Paper, there is no mention of what level of error (either of omission/commission or longitudinal accuracy) is acceptable to the agencies for estimating potential effects of rulemaking. Some degree of measurement error will occur with any dataset, independent of type (e.g., aerial photography, satellite imagery, field observations) and resolution, but that does not make such data not valuable for informing regulatory decisions. Importantly, the White Paper also fails to account for natural variability in wetland extent and stream hydrologic flow over time when describing sources of error. Hafen et al. 2020.

The White Paper also argued that since wetland definitions differ between the agencies and the NWI, NWI data should not be considered. However, those differences can largely be addressed by excluding NWI classification codes that by definition would not meet the regulatory definition of wetlands, such as codes for unvegetated NWI wetland types. This capability is not addressed in the White Paper, even though the agencies used it to exclude non-vegetated wetland types in the case study analyses for the 2020 NWPR EA.

Comparison studies between the NWI wetlands and wetlands that meet the regulatory definition often fail to exclude such NWI wetland types in their assessments, including most of those cited as supporting literature in the White Paper. Some of the references cited in the White Paper as supporting its conclusions for both the NHD and NWI were between 12 and 20 years old at the time of the White Paper's release. The NHD and NWI are continuously updated datasets, so the currency of the references cited is important. The references did not appear to be reviewed to ensure they are still pertinent to the points being made, or that current datasets were used. For example, one of the references cited information from an archived version of a webpage from the NHD website (USGS 2014) that is out of date and should not have been included in the White Paper.

c. The White Paper Contained Technical Errors

The White Paper also contains technical errors in the description of the NHD and NWI products. For example, it misrepresents characterization of NWI mapping standards (*e.g.*, the minimum threshold for wetland inclusion in the NWI is cited as being 0.05 acres, but that is not a national standard). The resolution information presented in the White Paper are only accurate for modern NWI data and not contextually relevant to legacy data, as legacy data are evaluated in the paper. Furthermore, the NWI section of the White Paper contains citations that do not address the NWI dataset (*e.g.*, Downing *et al.*

2007); were "updated" data that were not included in the NWI database (*e.g.*, Matthews *et al.* 2016³²); may be taken out of context; or are no longer current (*e.g.*, NWI disclaimer and biological wetlands definition within NWI classification standard). For example, the White Paper cites a "Special Note" found on NWI paper maps (Tiner 1997a), but the NWI no longer produces paper maps. The current relevant disclaimer on the NWI's Wetlands Mapper is: "The wetland information displayed at this site show wetland type and extent using a biological definition of wetlands. There is no attempt to define the limits of proprietary jurisdiction of any Federal, state, or local government, or to establish the geographical scope of the regulatory programs of government agencies." U.S FWS 2021.

d. The White Paper Contained Broad Generalizations and Selectively Used Citations

The White Paper also contains broad generalizations that are drawn from a singular or small number of examples that do not represent an issue fully and therefore cannot be used to prove a larger point. This type of overgeneralization is repeated throughout the paper. For example, the White Paper has a strong focus on the smallest waters that are often at or below the level of detection for the various methods of remote sensing or imagery used to construct the national maps. While it is often such features where questions of jurisdiction arise, this focus is not clearly stated, and the White Paper improperly implies that the same issues for mapping small waters are pervasive for all waters, including large rivers and wetlands. The NHD and NWI, for example, are more accurate in mapping and classifying hydrologic permanence of larger streams and wetlands but this fact is not discussed in the paper. Yamazaki et al. (2015), for example, looked at the relationship between water body size and the relative error (difference) in delineation by the NHD and G3WBM (an independent map derived from Landsat imagery). The density of points around high levels of disagreement is clearly greater for smaller waterbodies (streams and others) than for larger waterbodies. Id. at 348 (Figure 10(c)). Wu et al. (2019) compared the NHD to a digital elevation model (DEM) derived network for an entire watershed (not just headwater streams). They report high agreement (~99%) at this scale between the NHD at high resolution and DEM-derived networks for the Rogue River basin in Oregon. The White Paper does not provide information on to what degree the NHD or NWI under- or overrepresents the extent of larger-order streams and rivers, reservoirs, lakes, and coastal waters. As presented, the White Paper gives the impression that the datasets poorly represent water bodies regardless of their size and geographic location but does not provide support for such a conclusion regarding all waterbodies. For example, the White Paper cites one paper in one lowland watershed that has been heavily ditched for agriculture, making it difficult to map headwater stream networks in that geographic area (Lang et al. 2012), but the NHD contains over nine million stream segments nationwide, and more evidence from studies in a large number of geographically and hydrologically diverse watersheds would be needed to support the White Paper's broad generalizations. Furthermore, the agencies are unaware of any studies that have evaluated the accuracy of the entirety of the NHD in terms of flow permanence class.

The White Paper does not consider the breadth of literature available and only touches on a few relevant literature sources. The agencies have ascertained that the White Paper did not accurately represent the literature cited to support presumed errors or inaccuracies in the datasets, as discussed

³² The data in the study that were considered to be "updated NWI" were deemed by the NWI Program to not meet NWI standards and were therefore never included in the NWI database.

above. For example, results of the literature described in the section "NWI Has Certain Errors" actually validate the accuracy of that dataset by exhibiting good correlation between NWI data and field observations. *See* Stolt and Baker 1995; Kudray and Gale 2000; Dvorett *et al.* 2012; Sharpe *et al.* 2016. The White Paper points out perceived "errors" of the NWI from one study (Sharpe *et al.* 2016) while seemingly ignoring the study's relatively high correlations between NWI and the reference dataset. Sharpe *et al.* (2016, p. 547) in fact states, "This research *highlights the relative strength of NWI mapping* for landscape level wetland analysis, and the need to support remote sensing data by allocating field resources for accuracy assessment in specific areas based on management goals" (emphasis added).

The White Paper also selectively uses portions of cited works to support its conclusions, while ignoring other portions of the same studies that conflict with those conclusions. For example, in citing Simley (2011), the White Paper notes that in the South Platte River, some mapped stream segments in the high resolution NHD (1:24,000-scale) do not match up well with contemporary orthophotographs (1:4,000-scale), but leaves out quotations from the same citation noting good correlation at finer scale, including, "However, at 1:18,000-scale the NHD lines up quite well to the imagery," "At 1:18,000-scale the mismatch is not particularly evident," "at 1:18,000-scale the river lines up very well," etc. In addition, this study was for sections of the river with many meanders and may not be reflective of the accuracies of the NHD for the South Platte network as a whole. It is also possible that the original maps met the mapping standard, but the meanders have naturally migrated since the original map was compiled that the NHD data would not currently pass the criterion for current location. When not artificially constrained, most streams and rivers shift their position in the landscape over time. Leopold *et al.* 1964.

Some of the critiques of the NWI in the White Paper regarding use of the data for regulatory purposes (Tiner 1997a; Tiner 1997b; Kusler 2006) are statements about using the NWI data alone to determine the jurisdictional status of individual wetlands as required by law. Subsequently at the time of the rule's proposal and then finalization, the agencies were criticized by a wide variety of stakeholders for not utilizing the NHD and the NWI for estimating the extent of waters that would be no longer jurisdictional under the proposed and final rule. *See* Association of State Wetland Managers 2019; Fesenmyer *et al.* 2021. The citations in the White Paper are not directed at the broader use of NWI data along with other datasets to support the jurisdictional decision-making process. For example, the citations do not address the use of NWI data to assess policy change impacts on the overall direction and magnitude of jurisdiction under the Clean Water Act.

The review in the White Paper would have been made more complete and useful by recognizing that quantifying the uncertainty, error, and limitations of any dataset are part of the research process, and by citing appropriate applications of NHD and NWI datasets, including a discussion of cases where they have been or can be used successfully to inform questions about Clean Water Act jurisdiction (*e.g.*, as supporting data for state and federal agencies or the regulated community). For example, one of the studies cited in the White Paper noted that 90% of the sampling locations within wetland boundaries identified on NWI maps were categorized as wetlands that also met the agencies' wetland delineation criteria (Wu *et al.* 2014a), which demonstrates very high correlation between NWI mapped wetlands and wetlands that meet the agencies' regulatory definition. The White Paper could have also been more balanced by a discussion of the attributes of the databases that could be used in moving toward a more accurate dataset to inform regulatory decisions. For example, the White Paper could have included studies that apply recent technology and approaches to improve the NHD and reduce some of its past or current

limitations. *See*, *e.g.*, Poppenga *et al.* 2013; Stanislawski *et al.* 2015; Stanislawski *et al.* 2016; Stanislawski *et al.* 2017; Stanislawski *et al.* 2018; McManamay and DeRolph 2019. The White Paper mentions that LIDAR-derived stream networks can have errors of commission but does not similarly note that numerous studies have made a concerted effort to improve methods for deriving fine-scale hydrologic features from LIDAR-derived digital elevation models. *See* Goulden *et al.* 2014 and references therein; Shavers and Stanislawski 2020 and references therein. Mapping of ephemeral streams, and small streams generally, for example, is a clear research need, including from a regulatory perspective. More recent studies, like Fesenmyer *et al.* (2021), have taken approaches to better map ephemeral streams.

e. Conclusions

Decision-making is based on weighing advantages and disadvantages of the underlying datasets used in analyses. The White Paper heavily weighted some disadvantages of the NHD and NWI and therefore created an imbalanced perspective of their value to Clean Water Act decision-making as the best available national datasets for stream and wetland systems. As OMB-A-16 National Geospatial Data Assets and the hydrography and wetlands layers of the Federal Geographic Data Committee National Spatial Data Infrastructure, the NHD and NWI are the authoritative, most up-to-date, and comprehensive mapping of hydrography and wetlands for the nation. Additionally, the USGS has agreements with 41 states plus Washington DC to co-manage the NHD, enabling investment in the dataset form the local to national level. Citations and quotes are also available in the literature to support the advantages of the datasets, including high levels of dataset accuracy, but these citations (see, e.g., Nichols 1994) and quotations are not included in the White Paper (see, e.g., Stolt and Baker 1995; Kudray and Gale 2000; Simley 2011; Dvorett et al. 2012; Sharpe et al. 2016). In addition, information presented in the White Paper was generally limited geographic examples from older data rather than of the current versions of the national datasets evaluated as a whole. Because hydrologic conditions vary significantly across the country, it may not be accurate to take an analysis of the datasets in one area and assume that the results apply to the entire dataset and nation. Also, several references are out of date. The NHD framework was intended to improve accuracy over time while maintaining hydrologic connectivity and relationships within the network. For that reason, the NHD and NWI are continually updated by numerous partners engaged in their maintenance and improvement. Therefore, references that evaluate older versions of NHD (including topographic maps) and NWI are not based on the current datasets and thus should be interpreted with caution.

While the agencies agree with the White Paper's conclusion that the NHD are NWI are generally not sufficient as a sole basis for making jurisdiction determinations under the Clean Water Act and do not depict the full scope of waters that are or are not jurisdictional under the Act, the agencies now believe that it is scientifically defensible to use the datasets in national analyses to quantify the potential impact of their rulemaking efforts, so long as uncertainty and other appropriate caveats are also assessed and conveyed. In fact, the agencies used both the NHD and the NWI, in combination with programmatic data, in the 2020 NWPR EA's case study analyses. 2020 NWPR EA. In fact, one of the references that the White Paper cites, supports the use of the datasets for national analyses, stating, "hydrographical datasets, such as the NHD, are highly valuable at national, regional, and even local scales." Lang *et al.* 2012 at 462. Furthermore, as the White Paper notes, the agencies continue to agree that "NHD and NWI may be used in accordance with applicable law along with other information as part of the agencies' Clean Water Act jurisdictional determination process." White Paper at 1. In addition, the NHD serves as the underlying

framework for most water data applications throughout the country, including several Clean Water Act applications within EPA. For example, EPA uses the NHD for other EPA regulatory requirements such as tracking impaired waters under Clean Water Act section 303(d), reporting Total Maximum Daily Loads (TMDLs), tracking state-submitted drinking water quality data (SDWIS), and calculating programmatic measures, among others. The NHD and NWI were not designed to be stand-alone decision-making tools, but rather were designed to work in concert with other available data to support nationwide analyses for informing decisions. For example, both NHD and NWI are often used in conjunction with additional sources of information to support jurisdictional determinations. In addition, under section 303(d) of the Clean Water Act, states do not exclusively use the NHD to determine whether a water is impaired. Rather, they use multiple lines of evidence to make that determination. However, once the determination is made, states do use the NHD to document the spatial extent and location of those impairments. All datasets have limitations, including the NHD and NWI, but there is tremendous value in these datasets—including their use to help assess the potential impacts of a rule defining "waters of the United States."

EPA and the Army have been working with other federal agencies on improving aquatic resource mapping and modeling, including through the Advanced Water Mapping and Analytics interagency initiative with the Department of Interior (DOI) to better align EPA and the Army's regulatory needs with DOI's existing processes and national mapping capabilities. EPA, USGS, and FWS have a long history of working together to map the nation's aquatic resources. The agencies will continue to collaborate with DOI and other federal agencies to enhance the NHD, NWI, and other products to better map the nation's water resources while enhancing the utility of such geospatial products for implementation of Clean Water Act programs.

iv. Implementation Challenges

The agencies experience implementing the 2020 NWPR for over a year prior to its vacatur made clear that foundational concepts underlying much of the 2020 NWPR are confusing and difficult to implement in the way the 2020 NWPR requires. A key example is that no available tools, and certainly not the tools the 2020 NWPR preamble recommends, reliably demonstrate whether a surface water connection exists in a "typical year" in many important contexts. Surface water connection in a typical year is a necessary element of most categories of jurisdictional waters under the 2020 NWPR and demonstrating it in the way required by the rule was often difficult. While any rule that draws lines between jurisdictional waters and non-jurisdictional waters will involve some implementation challenges, the agencies have found the challenges imposed by the 2020 NWPR to be impracticable in important respects. Based on the agencies' experience, the 2020 NWPR did not achieve its stated purpose to "provide[] clarity and predictability for Federal agencies, States, Tribes, the regulated community, and the public." *See* 85 FR 22252 (April 21, 2020). The challenges that the 2020 NWPR imposed to establish jurisdiction for features that it appears to define as jurisdictional, and that significantly affect the integrity of traditional navigable waters, the territorial seas, and interstate waters, further undermine the 2020 NWPR's viability as an alternative to the final rule.

1. Typical Year Metric

The "typical year" is a concept fundamental to many of the 2020 NWPR's definitions. 85 FR 22273. Under that rule, tributaries and lakes, ponds, and impoundments of jurisdictional waters are only jurisdictional if they have certain surface water connections with a traditional navigable water or the territorial seas at least once in a typical year. 33 CFR 328.3(c)(6), (12). Two categories of wetlands only meet the adjacency test for jurisdiction if they have a surface water connection with other jurisdictional waters at least once in a typical year. 33 CFR 328.3(c)(1). As a scientific matter, the concept of "typical year conditions," including precipitation normalcy, may be relevant to ensuring that certain surface water connections in natural streams are not being observed under conditions that are unusually wet or dry. In terms of implementation, the concept of precipitation normalcy as valid in certain contexts, such as to inform determinations as to the presence of a wetland. However, in many important contexts, available tools, including the tools the 2020 NWPR preamble recommended, cannot reliably demonstrate the presence of surface water connections in a typical year, which are a necessary element of most categories of jurisdictional waters under the 2020 NWPR. For example, a recent study by the Corps found that precipitation normalcy (as calculated based on the methodology described in the preamble to the 2020 NWPR) was neither a reliable predictor of streamflow normalcy, nor was it a precise predictor of streamflow percentiles, in an analysis of watersheds across the United States. Sparrow et al. 2022. These challenges undermine the 2020 NWPR's claim that it enhances the "predictability and consistency of Clean Water Act programs." See 85 FR 22250 (April 21, 2020).

One of the significant implementation challenges of the typical year metric is that it can be difficult and sometimes impossible to identify the presence of a surface water connection in a typical year. Such connections are often not apparent from visual field observation alone. For example, on the day of a visit to an intermittent stream that flows only several months or several weeks a year, it is very unlikely that an observer would see surface water flows connecting to a downstream jurisdictional water, particularly based on the time of year during which that site visit takes place. Similarly, though many ponds or wetlands may be frequently inundated by flooding from another water, those in arid areas may be inundated only a few times every year, and sometimes the inundation occurs on a single day or within a matter of hours. While these waters satisfy the 2020 NWPR's jurisdictional test, agency staff would probably not be able to determine that they do, given how unlikely they would be to observe these infrequent connections. The difficulty of finding the direct hydrologic connections required by the typical year concept during a field visit is exacerbated by the fact that the 2020 NWPR discouraged reliance on field indicators. *See, e.g., id.* at 22292 ("The agencies . . . conclude that physical indicators of flow, absent verification of the actual occurrence of flow, may not accurately represent the flow classifications required for tributaries under this rule.").

Given the insufficiency of visual field observation to assess the presence of a surface water connection as specified in the 2020 NWPR, under that rule agency staff often needed to expend substantial time and resources to try to obtain ancillary data to determine flow conditions at a particular site in a typical year. Hydrologic modeling tools and advanced statistical analyses could be employed where sufficient flow data are available, but often data needed to conduct such an analysis is limited or lacking altogether, especially for smaller streams. Few streams across the country have hydrologic gages that continuously measure flow, as most such gages are located larger rivers with perennial flow. Flow and discharge data are rarely available for the small and/or non-perennial tributaries that collectively

constitute the largest number of stream miles in the United States; in addition, the number of long-term stream gages have declined over time and are geographically biased. Poff *et al.* 2006; Ruhi *et al.* 2018; Krabbenhoft *et al.* 2022. Moreover, "typical year conditions" are often irrelevant to the extent of flow in human-altered streams, including effluent-dependent streams. The 2020 NWPR did not explain why human-altered hydrology should be subject to the same typical year requirement as natural streams.

For the same reasons that agency staff are unlikely to witness the specific surface water connections required under the 2020 NWPR during a site visit in dry regions or during the dry season, they are also unlikely to capture evidence of a surface water connection between a stream and a downstream traditional navigable water or the territorial seas using available aerial photographs taken during typical year conditions. Available aerial photographs are often taken just once per year or once every other year, and staff have no way of ensuring that they were taken during a typical year. Highresolution satellite imagery can serve as a reliable source to demonstrate specific hydrologic connections and potentially provide additional coverage. But the availability and usability of such imagery vary across the country, depending on access, update intervals, cloud cover, and land cover (*i.e.*, vegetation or trees that obscure aerial views of stream channels), so that such tools may be unlikely to demonstrate that specific surface water connections are occuring in a typical year. High resolution satellite imagery also is not an "off the shelf" product and requires pre-processing and classification (*i.e.*, advanced tools) to identify features of interest or the presence of water. See, e.g., Mueller et al. 2016. Moreover, as the 2020 NWPR acknowledged, "characteristics of tributaries may not be visible in aerial photographs" taken during periods of "high shrub or tree cover" (85 FR 22299 (April 21, 2020)), further reducing the chances that a photograph could capture surface water connections.³³ Commenters on the proposed rule stated that Tribes and States lacked sufficient data, aerial photography, and access to other tools required to support the use of the typical year test in many locations. See, e.g., National Association of Wetland Managers 2022. They expressed concern that under-resourced communities suffer a particular lack of data necessary to support this test. Id. New satellites that are launching soon are expected to surmount some of these issues in the future (see, e.g., NASA; NASA 2019), but as this information is not yet available, regulators could not use it to inform jurisdictional decisions based on the requirements in the 2020 NWPR. Remote tools, such as aerial or satellite imagery, are often useful in implementing any definition of "waters of the United States," but the 2020 NWPR's typical year criteria made use of these resources particularly challenging.

The same difficulties created challenges in detecting surface hydrologic connections that occurred in a typical year to meet the 2020 NWPR's definition of "adjacent wetlands" or "lakes and ponds, and impoundments of jurisdictional waters." The 2020 NWPR's standard of inundation by flooding in a typical year was not tied to any commonly calculated flood interval, such as flood recurrence intervals³⁴

³³ Tree canopy cover varies across the country. The USDA Forest Service (USFS) produces percent tree canopy cover data layers for the National Land Cover Database (NLCD). USFS 2020. The most recent Tree Cover Canopy product shows the variation in tree canopy cover across the United States, with New Hampshire, West Virginia, and Maine having the highest percentage of cover in the conterminous United States, and North Dakota, Nebraska, and South Dakota have the lowest percentage. *Id.* About 24% of the land cover in the conterminous United States is forested as mapped in the 2019 NLCD (MRLC), while the USFS has estimated that about one third of the country is forested (USFS 2017).

³⁴ For example, the Federal Emergency Management Agency (FEMA) develops flood insurance rate maps based on the probability of a flood event occurring (*e.g.*, 100-year floods have a 1% probability of occurring in a given year or

(*see* USGS c; FEMA 1995), and the agencies are not aware of a tool capable of collecting the type of inundation data the 2020 NWPR required. Demonstrating that a wetland, lake, pond, or impoundment is inundated by flooding once in a typical year would require a field visit or high-resolution aerial photograph or satellite image coinciding with the exact time that the flooding occurs from a tributary to a wetland, lake, pond, or impoundment, as well as being able to demonstrate that this flooding occurred in a typical year. Determining that inundation by flooding occurs in a typical year was therefore extremely difficult, and sometimes impossible. Demonstrating that an artificial feature allows for a direct hydrologic surface connection between a wetland and a tributary in a typical year poses similar obstacles, requiring either auspiciously timed field visits, aerial photography, high-resolution satellite imagery, or data that the agencies may not be able to access, such as construction plans or operational records for an artificial levee.

The 2020 NWPR suggested the agencies "will generally use" precipitation data from NOAA to help determine the presence of a surface water connection in a typical year (see 85 FR 22274 (April 21, 2020), but the methodology described in the 2020 NWPR preamble for determining precipitation in a typical year made it difficult to use these data to inform jurisdiction. NOAA precipitation totals over the three months prior to a site observation are compared to precipitation totals observed over the preceding 30 years to determine if conditions were wetter than normal, drier than normal, or normal ("typical"). Using the methodology in the 2020 NWPR preamble, only 40% of observations over a rolling 30-year period of record are considered "normal," while 30% of observations are considered to be "wetter than normal" and 30% of observations are considered to be "drier than normal." If surface water flow was observed during normal or dry conditions, the agencies could have higher confidence that the surface water observations represented flow in a "typical year." However, if flow was observed during the 30% of conditions that are "wetter than normal," the surface water observations did not reveal whether flow would occur during a typical year. And if flow was not observed, precipitation data from the previous three months did not indicate whether flow might occur in that particular water feature under typical year conditions at a different point in the year. Therefore, if a site visit is conducted when surface water flow is not present, the agencies' suggested approach for evaluating whether a feature meets the typical year test often did not provide meaningful and relevant information for the agencies to make accurate determinations of jurisdiction. Indeed, a commenter on the proposed rule emphasized that Tribes and States have found the "typical year" requirement to require extensive hydrologic modeling and advanced statistical analyses in complex conditions. National Association of Wetland Managers 2022. Under any regulatory regime, the agencies use a weight of evidence approach to determine jurisdiction, but the 2020 NWPR typical year requirement placed onerous and, in many instances, arbitrary constraints on the data that can be used as evidence.

Use of NOAA precipitation data to assess whether surface water flow occurs in a typical year for purposes of the 2020 NWPR presents other implementation challenges. The data rely on reports from weather stations that are sometimes at a different elevation from the site in question, or far away from the site, so that their indications as to whether precipitation at a given site is normal, wetter than normal, or drier than normal can be inaccurate. Furthermore, the typical year concept as applied to the 2020 NWPR

⁵⁰⁰ year-floods have a 0.2% probability of occurring in a particular year). FEMA 2020. Flood insurance rate maps are developed by applying models and other information to identify areas that would be inundated by a flood event of a particular probability of recurring. *See, e.g.*, FEMA 1995.

does not account for the increasing number of recurrent heat waves, droughts, storms, and other extreme weather events in many parts of the country. These events can have profound impacts on local and regional hydrology, including streamflow. *See* section II.B.iv.1.c. Although the concept of "typical year" in the 2020 NWPR sought to factor in long-term climatic changes over time to some degree by considering a thirty-year rolling period of data (*see* 33 CFR 328.3(c)(13)), the 2020 NWPR did not allow the agencies flexibility to consider other time intervals when appropriate to reflect effects of a rapidly changing climate, including positive trends in temperature, increasing storm events, and extended droughts. In response to more rapid recent changes in climate, NOAA has developed alternative approaches for estimating climate normals, including seasonal averages computed using shorter, annually updated averaging periods for temperature (10-year seasonal average) and total precipitation (15-year seasonal average). The rigid rolling thirty-year approach to determining typical year in the 2020 NWPR did not allow the agencies to use these updated methods.

The 2020 NWPR noted that the agencies can look to sources of information other than site visits, aerial photographs, and precipitation data to assess whether a feature has surface water flow in a typical year. It identified the Web-based Water-Budget Interactive Modeling Program, Climate Analysis for Wetlands Tables, and the Palmer Drought Severity Index. 85 FR 22275 (April 21, 2020). These methods, which provide information useful in many other contexts, often only look at climate-related conditions generally, have well documented limitations, and often did not specifically answer the jurisdictional questions required by the 2020 NWPR. For example, the Palmer Drought Severity Index is difficult to correlate with specific water resource variables such as runoff and reservoir storage and does not account for delayed runoff from snow and ice. Further, none of these sources of information address whether surface water flow might connect a particular stream to a downstream traditional navigable water or the territorial seas, whether a particular wetland was inundated by or connected to a jurisdictional water as required under the 2020 NWPR, or how uncertainties at different locations and in different months affected the accuracy of condition estimates. While precipitation is an important factor, other information is also relevant to streamflow and surface water connections in a typical year, including the abundance of and contributions of flow from wetlands, upgradient streams, and open waters in the watershed; evapotranspiration rates; water withdrawals including groundwater pumping; and other climatic conditions. See section II.B.iii.1.c. Yet collecting this information from a variety of sources and interpreting it can be extremely time- and resource-intensive and may require special expertise, such as in climate science, remote sensing, statistical analysis, geospatial analysis, or other disciplines, that in many cases may not be feasible given available agency staff and resources. While the agencies have substantial experience using a weight of evidence approach to determine jurisdiction, for example as part of the significant nexus analysis, the "typical year" requirement of the 2020 NWPR makes it significantly more difficult to interpret available data and narrows the scope of data that can be used to determine jurisdiction.

Finally, the challenges presented by determining the presence of surface water flow in a typical year are even greater when evaluating a tributary at a distance from the downstream traditional navigable water or the territorial seas. Even streams that flow perennially or intermittently often travel many miles prior to reaching the closest traditional navigable water or the territorial seas, meaning many downstream reaches may need to be assessed. Under the 2020 NWPR, any ephemeral reaches along that pathway that did not carry surface water flow once in a typical year would render all upstream waters non-jurisdictional. 85 FR 22277 (April 21, 2020). The need to assess lengthy tributary systems imposed an

extraordinarily high burden of proof on the agencies to evaluate surface water flow in a typical year along the entire flow path from a stream of interest to a downstream traditional navigable water or the territorial seas. The longer the pathway, the more challenging the analysis. As a commenter on the 2020 NWPR's proposal noted, in adopting the test, the 2020 NWPR inserted case-specific analyses for every jurisdictional determination despite the rule's claim that it "provide[s] a predictable framework in which to establish federal jurisdiction." *Id.* at 22273–22274. The uncertainty and implementation challenges generated by the 2020 NWPR's foundational typical year test are yet another basis to replace that rule.

2. Determining Adjacency

The 2020 NWPR provided that wetlands are "adjacent" when they: (1) abut a traditional navigable water or the territorial seas; a tributary; or a lake, pond, or impoundment of a jurisdictional water; (2) are inundated by flooding from one of these waters in a typical year; (3) are physically separated from one of these waters only by a natural berm, bank, dune, or similar natural feature; or (4) are physically separated from one of these waters only by an artificial dike, barrier, or similar artificial structure so long as that structure allows for a direct hydrologic surface connection between the wetlands and the water in a typical year, such as through a culvert, flood or tide gate, pump, or similar artificial feature. *Id.* at 22338; 33 CFR 328.3(c)(1). In practice, agency staff have found several of these criteria for adjacency extremely difficult to implement in certain circumstances, in addition to the difficulties discussed above of assessing whether certain surface hydrologic connections exist in a typical year.

The artificial barrier provision led to arbitrary results. For example, under the fourth way to meet the adjacency definition, a wetland may be jurisdictional if it is separated from a jurisdictional water by an artificial structure, such as a levee, that allows for a direct hydrologic surface connection in a typical year through a culvert. However, the same wetland would not be jurisdictional if there was no levee present, even if there was a direct hydrological surface connection in a typical year through a culvert (assuming the wetland did not meet another criterion for adjacency). The 2020 NWPR therefore established that certain wetlands with a direct hydrologic surface connection to a jurisdictional water are *only* jurisdictional due to the presence of an artificial barrier. This discrepancy bears no relationship to the actual connections between the features at issue and is not supported by science or the agencies' experience.

Moreover, the provision establishing that a wetland is "adjacent" if a jurisdictional water inundates it by flooding in a typical year was extremely difficult to implement. *See* 33 CFR 328.3(c)(1)(ii). Inundation by flooding in a typical year is not a metric that is normally recorded either by implementing agencies or the regulated community. Available models generally focus on flood recurrence intervals, which do not necessarily correspond to the likelihood of inundation by flooding in a given or typical year, and the agencies would typically be unable to demonstrate that these indicators reflect typical year conditions. Indeed, the 2020 NWPR acknowledged that inundation by flooding in a typical year could correspond to a variety of flood recurrence intervals depending on location, climate, season, and other factors. 85 FR 22311. Given the absence of existing records of inundation by flooding, determining whether inundation by flooding has occurred in a typical year is challenging in many circumstances.

Compounding the challenge, the 2020 NWPR provided that wetlands can be jurisdictional if they are inundated by flooding from a jurisdictional water in a typical year—but inundation in the other

direction, *from* the wetlands *to* the jurisdictional water, is not grounds for jurisdiction. Not only is there no scientific evidence (*see* section II.B.iv.b) or legal basis (*see United States v. Riverside Bayview Homes Inc.*, 474 U.S. 121, 134 (1985), upholding the Corps' assertion of jurisdiction over "wetlands that are not flooded by adjacent waters [but] may still tend to drain into those waters") for distinguishing between inundation *of* the wetland as opposed to inundation *from* the wetland, but determining whether the limited available photographs, satellite images, or other evidence of inundation reflects flooding in one direction as opposed to another adds to the difficulty in evaluating whether this standard is met. The same challenges apply to determining whether lakes, ponds, or impoundments of jurisdictional waters are inundated by flooding in a typical year, one basis for demonstrating Clean Water Act jurisdiction over these features. 85 FR 22338-39 (April 21, 2020); 33 CFR 328.3(c)(vi). While any rule that draws lines between jurisdictional waters and non-jurisdictional waters will involve some implementation challenges, the agencies have found the challenges imposed by the 2020 NWPR to be exceptionally burdensome and impracticable.

3. Ditches

Among other requirements, the 2020 NWPR provided that a ditch³⁵ is jurisdictional as a tributary if it was originally built in a tributary or adjacent wetland, as those terms are defined in the 2020 NWPR, and emphasized that the agencies bear the burden of proof to determine that a ditch was originally constructed in a tributary or adjacent wetland. 85 FR 22299 (April 21, 2020); 33 CFR 328.3(a)(2) and (c)(12). In other words, in order to find a ditch jurisdictional, the agencies had to demonstrate that a ditch was (1) originally constructed in a stream (2) that, at the time of construction, had perennial or intermittent flow and (3) a surface water connection to a downstream traditional navigable water or the territorial seas (4) in a "typical year." Alternatively, the agencies had to show that a ditch was (1) originally constructed in a wetland (2) that either abutted or had certain surface hydrologic connections to a jurisdictional water at the time the ditch was constructed (3) in a "typical year," in order to demonstrate that the ditch is jurisdictional. Americans have been building ditches, straightening streams, and draining wetlands for hundreds of years. By contrast, to determine whether a ditch was jurisdictional under the 2020 NWPR, the agencies had to determine if it was originally built in a tributary or adjacent wetland that would have been jurisdictional under the 2020 NWPR, and therefore had to address all of the implementation challenges discussed in the preceding sections involved in determining surface water connections and wetland adjacency in a typical year-but often for ditches built twenty, fifty, one hundred, or even several hundred years ago. To the extent that sparse evidence is available to demonstrate a surface water connection in a typical year for tributaries using tools available today, evidence is even more difficult to find when looking so far back in time. States have approached the agencies seeking assistance in assessing the jurisdictional status of ditches, but the agencies were often unable to provide meaningful help given the burdens imposed by the 2020 NWPR's ditch definition.

The 2020 NWPR also provided that ditches are jurisdictional if they relocate a tributary, as that term was defined in the rule (85 FR 22341; 33 CFR 328.3(a)(2) and (c)(12)), but this standard, too, is often extremely difficult to meet. The 2020 NWPR explains that a relocated tributary is "one in which an *entire portion* of the tributary may be moved to a different location." 85 FR at 22290. In other words, the

³⁵ Ditches perform many of the same functions as natural tributaries. For example, like natural tributaries, ditches that are part of the stream network convey water that carries nutrients, pollutants, and other constituents, both good and bad, to downstream traditional navigable waters, the territorial seas, and interstate waters. *See* section III.A.iv.

2020 NWPR appears to require a ditch to divert 100% of the tributary's flow to meet the "relocate a tributary" test. While prior rules have defined relocated tributaries as jurisdictional, the requirement that the entire portion be relocated is new and has created significant implementation challenges. As a practical matter, when a tributary is relocated it often reroutes just a portion of its flow to the ditch. Assessing whether a ditch relocated 100% of a tributary's flow, as opposed to 80% or 50% of its flow, is extremely difficult and may not be possible in some circumstances. The scientific literature indicates that features like ditches that convey water continue to connect to and affect downstream waters. See section III.A.iv of this Technical Support Document. By establishing a jurisdictional standard that is extremely difficult to meet, the 2020 NWPR effectively removed from the protections of the Clean Water Act large numbers of ditches that function as tributaries and that significantly affect the integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. As is the case with tributaries, lakes and ponds, impoundments, and wetlands, the 2020 NWPR's impracticable approach to ditches made it extremely difficult to implement. In the agencies' judgment, any efficiencies the 2020 NWPR may have achieved through categorical exclusions are outweighed by the challenges the agencies encountered in implementing the rule, coupled with its failure to implement the objective of the Clean Water Act by removing protections for waters that are properly within the statute's scope. These deficiencies of the 2020 NWPR further undermine the viability of the 2020 NWPR as an alternative to the final rule.

4. Results of Regional Survey

EPA's Office of Wetlands, Oceans, and Watersheds sought information from Regional staff about potential challenges they may have encountered implementing the 2020 NWPR. A seven-question survey that included both multiple choice and open-ended questions was provided to EPA staff in all ten Regional offices. Twenty-six staff persons provided responses to the survey, including representatives from all ten EPA Regions. While the number of respondents per region varied considerably, over 60% of respondents came from EPA Regions lying mostly west of the Mississippi River (*i.e.*, Regions 6-10) (Figure 10). Although an imperfect demarcation boundary between generally more humid/temperate climatic conditions to the East and drier ones to the West, the responses provided by EPA staff west of the Mississippi River differed in some notable ways from those of staff in the eastern parts of the country.





EPA Regional field staff reported different implementation challenges with the 2020 NWPR, but some challenges were significantly more often cited than others. Respondents were asked to identify the single biggest implementation challenge with the 2020 NWPR, and then they were also asked to identify their next Top 3 biggest challenges. The same four primary issues were cited most often in response to these questions, and each of the four gathered nearly the same number of cumulative responses:

- "Documenting or determining historic conditions," (18% of responses)
- "Documenting flow regime in the dry season/dry season evaluations," (18% of responses)
- "Disproportionate burden of proof for asserting jurisdiction," (18% of responses)
- "Typical year," (17% of responses)

Despite the similarity of major implementation challenges cited cumulatively in response to the aforementioned pair of survey questions, there was greater differentiation among concerns considered by respondents as the single "biggest challenge." Almost one-quarter of respondents (23%) reported that "documenting or determining historic conditions" was the biggest implementation challenge (Figure 11). Interestingly, over two-thirds of the respondents who cited documenting historic conditions as the biggest challenge came from EPA Regions east of the Mississippi River. "Documenting flow regime during the dry season (~ dry season evaluations)" and "Typical year" were each cited as the biggest implementation challenge by 19% of respondents. In contrast to the geographic distribution of responses for historic conditions, 100% of respondents who cited dry season evaluations as the single biggest challenge of the 2020 NWPR came from EPA Regions west of the Mississippi River. "Typical year" was evenly split among EPA Regions lying in the East and West.

The only other implementation challenge cited by at least ten percent of respondents as the biggest challenge was "disproportionate burden of proof to assert jurisdiction" (15%), and three-quarters of respondents citing burden of proof as the biggest 2020 NWPR implementation challenge came from Western EPA Regions. Other aspects of 2020 NWPR for which at least one respondent reported to be the most significant implementation challenge included "ditches and water diversions," distinguishing between bridges and culverts, and "inability to evaluate flows on properties for which agency staff have no authorization to enter."





Specific reasons cited by the survey respondents explaining implementation challenges with the 2020 NPWR included many of the same issues previously described in this section. Respondents reported that a lack of historic aerial photographs or other sources of historic information rendered it difficult or impossible to document or determine historic conditions consistent with the 2020 NWPR's requirements. This affected staff's ability to assess the flow regime of ditches excavated decades ago or whether those ditches drained wetlands that would have been considered "adjacent wetlands" under the 2020 NWPR. Staff observed that even if historic imagery were available, land use, land cover, and climatic conditions prevalent at the time often bore no resemblance to more contemporary conditions.

Challenges under the 2020 NWPR associated with documenting flow regime in the dry season and other dry season evaluations centered on the timing of site visits, as one would expect. Staff noted that site visits often only occur on a single date. When this takes place during a time of year for which intermittent stream flow would not normally even be expected to occur, for example, there is little information available upon which an assessment of flow regime as defined in the 2020 NWPR can be based. Other staff noted that it is nearly impossible to differentiate intermittent flow from ephemeral flow in small streams from aerial photographs; a challenge made even more acute under a dense canopy of vegetation cover.

EPA Regional staff observed that assessing stream flow in a "typical year" had a very high likelihood for errors of omission, whereby a stream might be determined not to flow (and thereby be non-jurisdictional), when in fact it may flow on a different date during even the same typical year. For example, field conditions at the low end of the precipitation range represented by a typical year analysis provide no indication at all of what field conditions would be like at the upper end of the typical year range. Staff noted that the probability of observing a qualifying flow event (*e.g.*, stream flow, surface connection, inundation by flooding, etc.) that occurs only rarely in a "typical year" is very low.

Challenges and demands associated with proving a given feature is a jurisdictional water included a disproportionate burden of time, effort, and resources necessary for locating and researching applicable
information. Regional staff noted, for example, that in some cases, on-scene coordinators have to assess the jurisdictional status of waters quickly, often at the request of other agencies on-site of an oil spill incident. Some staff felt that the 2020 NWPR was written with no consideration of Clean Water Act section 311(c) response authorities or time frames. In addition, the regulatory agencies often do not have the time or manpower to devote to collection and assessment of historic information, so they are often compelled to accept whatever information the permit applicant provides them.

Implementation challenges of the 2020 NWPR associated with ditches encompass most other 2020 NWPR implementation challenges (*i.e.*, historic conditions, flow regime, adjacency criteria, typical year, burden of proof, etc.). Ditches excavated decades ago through wetlands have since drained those wetlands, often leaving little record that they were once wetlands and so are now impossible to prove (or disprove) that they would have been jurisdictional "adjacent wetlands" at the time of ditch excavation, as the 2020 NWPR requires. Other staff noted that ditches excavated in uplands still transport pollutants directly to jurisdictional waters, and that there was no scientific justification for 100% of flow to be redirected into a ditch before the diversion ditch itself becomes a jurisdictional water, as the 2020 NWPR stipulated in the 2020 NWPR preamble.

Other challenges cited by at least one EPA Regional staff person in response to the survey included the need to periodically access multiple properties to document wetland adjacency to the nearest tributary, which is often not feasible; and the disparate treatment of culverts and bridges under the 2020 NWPR, given that in the field there is often no clear distinction between them (*e.g.*, what is an open-bottomed culvert?).

C. Climate Change

Consistent with Executive Order 13990, the agencies, where consistent with the objective of the Clean Water Act, considered the impact of climate change on water resources to the extent those impacts would affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

The Intergovernmental Panel on Climate Change (IPCC) notes that, "Recent climate changes have had widespread impacts on human and natural systems." IPCC 2014 at 2. The IPCC explains that "Surface temperature is projected to rise over the 21st century under all assessed emission scenarios. It is very likely that heat waves will occur more often and last longer, and that extreme precipitation events will become more intense and frequent in many regions. The ocean will continue to warm and acidify, and global mean sea level to rise." *Id.* at 10. Furthermore, climate change has important implications for communities with environmental justice concerns. "Climate change will amplify existing risks and create new risks for natural and human systems. Risks are unevenly distributed and are generally greater for disadvantaged people and communities in countries at all levels of development." *Id.* at 13.

As a result of well documented climate trends, the traditional locations of rain belts and deserts are shifting. UCAR 2022. Current climate models indicate that rising temperatures will intensify the water cycle, leading to an increase in the rate of evaporation worldwide. NASA 2021; UCAR 2022. On average, this increase in evaporation is leading to more precipitation, and the impacts are likely to increase over the

century as the climate continues to warm. UCAR 2022. However, changes in evaporation and precipitation rates are not distributed evenly. *Id.* In some regions, increased evaporation rates are leading to higher levels of water vapor in the air, which in turn increases the intensity and frequency of precipitation events, including storms. NASA 2021; U.S. EPA 2022e; USGCRP 2018. As a result, stormaffected areas, such as the Northeast and the Midwest, are likely to experience increases in precipitation and increased risk of flooding. NASA 2021; UCAR 2022; U.S. EPA 2022i; USGCRP 2018. In other regions, less precipitation and increased risk of drought is associated with more frequent and intense heatwaves. Across much of the country, surface soil moisture is also anticipated to decrease as the climate warms, driven largely by increased evaporation rates. USGCRP 2018. All else being equal, this will result in increased intensity, frequency, and duration of heatwaves and droughts in these regions. *Id.* These trends are likely to be strongest in the Southwest and Southern Great Plains, where precipitation is projected to decrease in most seasons and droughts may become more frequent, prolonged, and intense. *Id.*

Climate change is expected to have a variety of impacts on water resources. Runoff from more intense and frequent storms can impair water quality as pollutants deposited on land wash into water bodies. Changes in the size and frequency of heavy precipitation events, streamflow, snowmelt timing, and snowpack accumulation may also cause river floods to become larger or more frequent than they used to be in some places. U.S. EPA 2022f. Climate change is already affecting streamflow characteristics such as the magnitude, duration, and timing of flows (U.S. EPA 2022h), in part due to changes in snowpack magnitude and seasonality. U.S. EPA 2022g. Heatwaves and associated drought can cause streams and wetlands to become drier, negatively affecting both water supplies and water quality. Heatwaves and associated drought reduce surface and soil moisture thereby increasing the extent and duration of wildfires, which can alter water quality and impact wetlands and their functions. U.S. EPA 2022b. Wetland loss and changes to stream habitat can also lead to reduced habitat for fish and other aquatic and semi-aquatic species and worsen existing shifts in species ranges. *Id.* A warming climate can also result in increased and more variable temperatures in streams, leading to fish kills and negatively affecting other aquatic species that can live only in colder water. *See, e.g.*, Mantua *et al.* 2010; Ebersole *et al.* 2020.

Climate change has been well established to contribute to sea level rise. Rising sea levels are affecting human activities in coastal areas and making coastal infrastructure more vulnerable to storm damage. Furthermore, rising sea level inundates low-lying wetlands and dry land, and further contributes to coastal flooding and erosion. U.S. EPA 2022b; U.S. EPA 2022c.; USGCRP 2018. Repeated inundation of low-lying wetlands results in the loss or migration landward of coastal wetlands. However, inland migration of wetlands can be impeded by coastal development, thereby placing these wetlands at risk of being lost. Kirwan and Megonigal 2013; Borchert *et al.* 2018; USGCRP 2018. Both tidal and non-tidal wetlands may be at risk from sea-level rise: wetland types that are vulnerable to climate change include salt marshes, bottomland hardwood swamps, freshwater marshes, mangrove swamps, and pocosins. U.S. EPA 2022b. Sea level rise can introduce saltwater into non-tidal wetlands which can lead to substantial impacts on the wetland health and distribution. Herbert *et al.* 2015. These impacts can be further exacerbated by climate-related reductions of inflows from rivers that deliver freshwater to non-tidal wetlands, illustrating that many wetlands face cumulative climate impacts from multiple sources. *Id.* Climate change can also inhibit the ability of sediment accretion in tidal wetlands leading to vegetation

"drowning." This occurs when the rate of sea level rise surpasses the ability of coastal wetlands to keep pace via sediment accretion and can lead to certain wetlands becoming submerged. Borchert *et al.* 2018.

Although water resources are vulnerable to the effects of climate change, when their interconnectedness and extent are maintained, streams and wetlands perform a variety of functions that can help restore ecologic function of other water resources in light of climate change (*i.e.*, contribute to climate resiliency) and mitigate the negative effects of climate change on other water resources, including larger downstream waters. For instance, despite their vulnerability to the effects of sea level rise, many coastal wetlands are resilient ecosystems that have the capacity to adjust to sea level rise through vertical adjustments due to feedbacks between plant growth, inundation, and sediment deposition and through the landward migration of wetlands. Id. In addition, wetlands inside and outside of floodplains store large volumes of floodwaters, thereby reducing flood peaks and protecting downstream watersheds. Science Report; U.S. EPA 2006; U.S. EPA 2022b. For example, during Hurricane Sandy in 2012, wetlands are estimated to have helped prevent \$625 million in damage by protecting properties from flooding. Narayan 2017. Coastal wetlands also help buffer storm surges and slow winds from tropical storms and hurricanes (U.S. EPA 2022b), which are increasing in intensity due to climate change (USGCRP 2017). As natural filters, wetlands also help purify and protect the quality of other waters, including drinking water sources—a function which is more important than ever as intense precipitation events spurred by a changing climate mobilize excess sediment, nutrients, and other pollutants. Additionally, small streams are particularly effective at retaining and attenuating floodwaters, a function that is increasing important in light of continued climate change. See, e.g., Science Report at ES-8. Biological communities and geomorphic processes in small streams, including ephemeral streams, and wetlands break down leaves and other organic matter, burying and sequestering a portion of that carbon that could otherwise be released into the atmosphere as either carbon dioxide or methane. Guinessey et al. 2019; Kirwan and Megonigal 2013. Carbon sequestered in soils, sediments, and vegetation is released to the atmosphere when these streams or wetlands are drained, dredged, or otherwise disturbed, contributing to greenhouse gas emissions which have negative effects on water resources. More streams are expected to become ephemeral under a changing climate. Lohse et al. 2020.

The 2020 NWPR did not appropriately acknowledge or take account of the effects of a changing climate on the chemical, physical, and biological integrity of the nation's waters. For example, its rolling thirty-year approach to determining a "typical year" did not allow the agencies flexibility to account for the effects of a rapidly changing climate, including positive trends in temperature, increasing storm events, and extended droughts (see section II.B). The 2020 NWPR also categorically excluded ephemeral streams and their adjacent wetlands from the definition of "waters of the United States." These exclusions, if in effect, would disproportionately impact the arid West. Aquatic systems comprised largely of ephemeral streams are increasingly critical to protecting and maintaining downstream integrity, for example by contributing streamflow and organic matter to larger downstream waters. This is especially true in the Southwestern United States, where the climate continues to get hotter and drier and the spatial extent of arid conditions is expanding, with increased risks of more extreme drought. Some portions of the arid West are experiencing altered monsoon seasons that have fewer but more intense storms that contribute to so-called "flashy" stream hydrology (*i.e.*, higher runoff volume, leading to more rapidly rising and falling streamflow over shorter periods of time). The agencies' use of the significant nexus standard in the final rule (see section III.E.iv.1) allows the agencies to consider the effects that water resources like ephemeral streams and wetlands have on the chemical, physical, or biological integrity of

downstream traditional navigable waters, the territorial seas, and interstate waters, including important functions that may intersect with climate change.

D. Environmental Justice

While impacts on communities with environmental justice concerns are not a basis for determining the scope of the definition of "waters of the United States," the agencies recognize that the burdens of environmental pollution and climate change often fall disproportionately on communities with environmental justice concerns (*e.g.*, minority (Indigenous peoples and/or people of color) and low-income populations, as specified in Executive Order 12898). Compared to the average populations, these communities are more likely to experience environmental and social stressors like contaminated drinking water, limited access to clean water, and inadequate water infrastructure, that increase their likelihood of being exposed to pollutants. In addition to external stressors, behavioral and cultural characteristics of these communities, like engaging in subsistence fishing and consuming higher rates of fish from polluted waters, increases their vulnerability to pollution. Taken together, these environmental, social, and behavioral factors often increase the risk of these communities experiencing negative health outcomes because of their exposure.

As the IPCC Synthesis Report notes, in addition to the existing risks that communities with environmental justice concerns face, the impacts of climate change have important additional implications for these communities. "Climate change will amplify existing risks and create new risks for natural and human systems. Risks are unevenly distributed and are generally greater for disadvantaged people and communities in countries at all levels of development." IPCC 2014 at 13. Risks to human systems like sea level rise, flooding, and drought can all have disproportionate effects on these communities. For example, prolonged droughts pose a particular threat to Indigenous populations because of their economic and cultural dependence on land and water supplies. U.S. EPA 2022d. Similarly, decreased summer streamflow, habitat loss due to increasing storm intensity and flooding, warmer stream and ocean temperatures, and seasons of reduced snowpack are impacting Pacific salmon populations in the Northwest, threatening Indigenous communities that rely culturally and economically on salmon. USGCRP 2018. Because of existing environmental and social stressors and their reliance on natural resources (*e.g.*, fish and other aquatic life for income or food) that may be negatively impacted by climate change, these communities may be less able to mitigate and adapt to the effects of climate change.

Numerous groups have raised concerns that the 2020 NWPR had disproportionate impacts on Tribes and Indigenous communities.³⁶ The 2020 NWPR decreased the scope of Clean Water Act

³⁶ See, e.g., Tribal Consultation Comment Letter from President Jonathan Nez and Vice President Myron Lizer, Navajo Nation, October 4, 2021 ("The Navajo Nation relies greatly on all its surface waters, including ephemeral, intermittent, and perennial surface waters. The Navajo Nation currently lacks the resources to implement CWA permitting and other programs necessary to maintain and protect water quality and relies on the Agencies to fill that need. Therefore, any new WOTUS rule must not reduce the scope of the waters that the Agencies can protect, or it will have 'disproportionately high and adverse human health or environmental effects' on the Navajo Nation."), and Tribal Consultation Comment Letter from Clarice Madalena, Interim Director, Natural Resources Department, Pueblo of Jemez, October 4, 2021 ("The combination of these factors—[desert] hydrology and the geographic location of Native communities—means that the Navigable Waters Rule had the effect of disparately stripping Clean Water Act protections from areas with higher Native populations. This means that the Rule disproportionately harmed Native American communities. This discriminatory impact violates the principles of environmental justice" (citations omitted). *See also* section IV.B.3.d of the preamble for the final rule.

jurisdiction across the country, including in geographic regions where regulation of waters beyond those covered by the Act is not authorized under current Tribal or State law (*see* section IV.B.3.d of the preamble for the final rule). If the 2020 NWPR were in effect, without regulations governing discharges of pollutants into previously jurisdictional waters, population groups of concern where these waters are located could experience increased water pollution and impacts from associated increases in health risk.

Further, the 2020 NWPR's categorical exclusion of ephemeral streams disproportionately impacted Tribes and communities with environmental justice concerns in the arid West. Many tribes lack the authority and resources to regulate waters within their boundaries, and they may also be affected by pollution from adjacent jurisdictions. *See supra* note 36. Therefore, if in effect, the 2020 NWPR could disproportionately expose Tribes to increased pollution and health risks.

In this final rule the agencies affirm their commitment to assessing impacts of a revised definition of "waters of the United States" on population groups of concern (*see* Chapter IV of the Economic Analysis for the Final Rule).

III. Scientific Support for the Final Rule

The objective of the Clean Water Act to protect water quality must be considered when defining "waters of the United States." The latest science supports the conclusion that the categories of waters identified in the final rule, such as impoundments (paragraph (a)(2) waters in the final rule), tributaries (paragraph (a)(3) waters in the final rule), adjacent wetlands (paragraph (a)(4) waters in the final rule), and intrastate waters that do not meet the criteria to be jurisdictional under other categories of the final rule (paragraph (a)(5) waters in the final rule), provide functions that restore and maintain the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters (paragraph (a)(1) waters in the final rule). The best available science thus confirms that the final rule provides a definition of "waters of the United States" that furthers the water quality objective of the Clean Water Act. As explained in the preamble, the final rule also establishes limitations reflecting consideration of the statute as a whole and relevant Supreme Court decisions. The agencies thus believe that the latest science informs the conclusion that the categories of waters identified in the final rule provide functions that restore and maintain the chemical, physical, and biological integrity as a whole and relevant Supreme Court decisions. The agencies thus believe that the latest science informs the conclusion that the categories of waters identified in the final rule provide functions that restore and maintain the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

In the preamble to the final rule, the agencies' reference to a "connection" to traditional navigable waters, the territorial seas, or interstate waters (when used without qualification such as "continuous surface connection" or an "unbroken surface or shallow subsurface connection") includes all the types of connections relevant to either the relatively permanent standard or the significant nexus standard that are discussed in the Science Report and in this Technical Support Document: physical (including hydrological), chemical, biological, or functional relationships (including where the water retains floodwaters or pollutants that would otherwise flow to the traditional navigable water, the territorial seas, or an interstate water). A "requisite" connection is one that satisfies either the relatively permanent or significant nexus standard.

Although the scientific conclusions of the Science Report, the research published since its release, and other literature that the agencies reviewed to support this rulemaking play a critical role in informing the agencies' interpretation of the Clean Water Act's scope, the agencies' interpretive task in this final rule – determining which waters are "waters of the United States" – requires scientific and policy judgment, as well as legal interpretation. The agencies also acknowledge that science can support other approaches to implementation, particularly regarding which waters to consider in combination with similar waters and the region in which to consider similarly situated waters but have made policy choices regarding implementation of this final rule. The agencies reiterate their previous conclusion that significant nexus is not a purely scientific determination. 80 FR 37054, 37060 (June 29, 2015). As the agencies charged with interpreting the statute, EPA and the Corps must develop the outer bounds of the scope of the Clean Water Act and science does not provide bright line boundaries for purposes of the Clean Water Act. Riverside Bayview, 474 U.S. at 132-33. This section summarizes the best available science in support of the final rule and the agencies' conclusion that the rule advances the objective of the Clean Water Act. This section reflects the scientific consensus on the strength of the effects that upstream tributaries, adjacent wetlands, and intrastate waters considered under paragraph (a)(5) of the final rule can and do have on downstream traditional navigable waters, the territorial seas, and interstate waters. However, a significant nexus determination requires legal, technical, and policy judgment, as well as scientific considerations, to assess the significance of any effects. Section IV.C of the preamble and section IV of this document discuss the agencies' approaches to making case-specific relatively permanent and significant nexus determinations under the final rule. The science demonstrates that waters fall along gradients of chemical, physical, and biological connection to traditional navigable waters, the territorial seas, and interstate waters, and it is the agencies' task to determine where along that gradient to draw lines of jurisdiction under the Clean Water Act. In making this determination, the agencies must rely, not only on the science, but also on their technical expertise and practical experience in implementing the Clean Water Act during a 50-year period.

To be clear, under the final rule no waters considered under paragraph (a)(5) are categorically jurisdictional. Rather, the agencies will assess tributaries, adjacent wetlands, and intrastate waters that do not fall with paragraphs (a)(1) through (a)(4) utilizing the relatively permanent or significant nexus jurisdictional standards. This approach advances the objective of the Clean Water Act and is consistent with the best available science because waters in these categories can have significant effects on traditional navigable waters, the territorial seas, and interstate waters. The agencies are also adding the relatively permanent and significant nexus standards to certain categories of waters based on their conclusion that together those standards are consistent with the statutory text, advance the objective and policies of the Act, and are supported by the scientific record. Indeed, the agencies are not reaching any conclusions, categorical or otherwise, about which tributaries, wetlands (other than those wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters), lakes, ponds, or other types of waters meet either the relatively permanent or the significant nexus standard. Instead, the final rule enables the agencies to make science-informed determinations of whether or not a water that falls within these categories meets either jurisdictional standard and is therefore a "water of the United States," on a case-specific basis.

A. Tributaries

Under the final rule, tributaries of traditional navigable waters, the territorial seas, interstate waters, or jurisdictional impoundments are "waters of the United States" where they meet either the relatively permanent or significant nexus standard. Asserting Clean Water Act jurisdiction over tributaries where they meet either the relatively permanent or significant nexus standards as outlined in paragraph (a)(3) of the final rule aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that tributaries, regardless of flow regime, have chemical, physical, or biological effects on downstream waters.

The scientific literature documents that tributary streams, including perennial, intermittent, and ephemeral streams, certain lakes and ponds, and certain categories of ditches are integral parts of river networks because they are directly connected to rivers via permanent surface features (channels and associated alluvial deposits) that concentrate, mix, transform, and transport water and other materials, including food resources, downstream. Alluvial deposits, or alluvium, are deposits of clay, silt, sand, gravel, or other particulate materials that have been deposited by a stream or other body of running water in a streambed, on a flood plain, on a delta, or at the base of a mountain. Science Report at A-1. Tributaries transport, and often transform, chemical elements and compounds, such as nutrients, ions, dissolved and particulate organic matter and contaminants, influencing water quality, sediment deposition, nutrient availability, and biotic functions in rivers. Streams also are biologically connected to downstream waters by dispersal and migration, processes which have critical implications for aquatic populations of organisms that use both headwater and river or open water habitats to complete their life cycles or maintain viable populations. The scientific literature clearly demonstrates that cumulatively, streams exert strong influence on the character and functioning of rivers. See id. at ES-2 and Chapter 3. In light of these well documented connections and functions, the agencies conclude that tributaries, either alone or in combination with similarly situated waters, can significantly affect the chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water. The scientific literature supports this conclusion for ephemeral tributaries, as well as for intermittent and perennial tributaries; for tributaries both near to and far from the downstream traditional navigable water, the territorial seas, or interstate water; and for natural tributaries, human-altered, or human-made tributaries, which may include certain ditches and canals. For tributaries that meet the relatively permanent standard, such tributaries, either alone or in combination with similarly situated waters in the region, will virtually always significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. For tributaries that do not meet the relatively permanent standard, under the final rule they will be evaluated, either alone or in combination with similarly situated waters in the region, on a case-specific basis to determine whether they significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

The connections that tributaries have to downstream waters and functions they provide that impact those downstream waters continue even where the tributary has a natural or human-made break in its channel or ordinary high water mark (OHWM). The presence of a channel, bed and banks, or other indicators of OHWM upstream or downstream of the break is an indication that hydrological connections still exist. *See, e.g., id.* at 2-2. The connections between a tributary and a downstream water and associated functions remain intact even where the tributary flows underground for a portion of its length,

such as in regions with karst geology or topography or lava tubes. Artificial breaks can occur, for example, when a stream has been buried (*e.g.*, diverted into pipes or other conveyances), which is common in urban watersheds. *See*, *e.g.*, *id*. at 3-3. Where the hydrologic connection still exists, chemical and biological connections mediated by the hydrologic connection can also still exist. Similarly, flow through boulder fields does not sever the hydrologic connection. When a tributary flows through a wetland enroute to another or the same tributary, connectivity and effects still exist even though the channel or ordinary high water mark is broken for the length of the wetland. Adjacent wetlands located within a tributary can provide numerous benefits downstream (*see* section III.B), and the location of the wetland in-stream can provide additional water quality benefits to the receiving waters. Flow in flat areas with very low gradients may temporarily break a tributary's channel or OHWM, but these systems continue to be connected downstream and can provide functions that benefit downstream waters. These are just illustrative examples of break in the stream channel or ordinary high water mark.

The discussion below summarizes the key points in the literature regarding the chemical, physical, and biological connections and functions of tributaries that affect downstream waters. The scientific literature does not use legal terms like "relatively permanent" to describe streamflow permanence classes. Rather, the literature uses terms like "perennial," "intermittent," or "ephemeral" to describe and classify streamflow permanence. Tributaries that meet the relatively permanent standard under the final rule are those that have flowing or standing water year-round or continuously during certain times of the year. The agencies have decided against defining the relatively permanent standard by specific stream flow classifications in the final rule (e.g., perennial, intermittent, or ephemeral). However, flow characteristics like duration and timing of flow will be considered in determining whether tributaries meet the relatively permanent or significant nexus standard. Tributaries that do not meet the relatively permanent standard include tributaries with flowing or standing water for only a short duration in direct response to precipitation. The approach to the relatively permanent standard for tributaries in the final rule would encompass tributaries considered relatively permanent under the 2020 NWPR, as well as those considered relatively permanent under the *Rapanos* Guidance, providing continuity in approach for the regulated community and other stakeholders. Tributaries that do not meet the relatively permanent standard must be assessed under the significant nexus standard. The relatively permanent standard is discussed in section III.F below and the significant nexus standard is discussed in section III.E below.

In addition, the evidence regarding human-altered and human-made tributaries and headwater streams and non-perennial streams, types of tributaries whose important functional relationships to downstream traditional navigable waters, the territorial seas, and interstate waters might not be obvious, is summarized. The evidence regarding headwater or in-stream lakes and ponds is also summarized. The scientific literature does not use legal terms like "traditional navigable water," "the territorial seas," or "interstate water." Rather, the literature assesses tributaries in terms of their connections to and effects on larger downstream waters in a watershed. Traditional navigable waters, the territorial seas, and interstate waters are simply a subset of downstream waters, and their distinction is a legal, not scientific, one; the strength of the connections and effects does not change because a river does not meet the legal standards for being traditionally navigable. While the final rule, consistent with Supreme Court case law and the Clean Water Act, addresses only those tributaries that drain (contribute flow directly or indirectly) to a traditional navigable water, the territorial seas, interstate water, or jurisdictional impoundments, the conclusions of the scientific literature with respect to the effects of tributaries on downstream waters are

applicable to the subset of downstream waters that are traditional navigable waters, the territorial seas, or interstate waters.

Based on the importance of the functions that are provided by tributaries to traditional navigable waters, the territorial seas, and interstate waters, the agencies' final rule to interpret the Clean Water Act to protect those tributaries that meet either the relatively permanent standard or the significant nexus standard reflects proper consideration of the objective of the Act and the best available science.

i. Tributaries Can Provide Functions that Restore and Maintain the Physical Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

The scientific literature unequivocally demonstrates that tributaries exert a strong influence on the physical integrity of larger downstream waters. Tributaries, even when seasonal, are the dominant source of water in most rivers, rather than direct precipitation or groundwater input to main stem river segments. See, e.g., Science Report at 3-5 (citing Winter 2007; Bukaveckas 2009). Distant headwaters with stronger connections to groundwater or consistently higher precipitation levels than downstream reaches contribute more water to downstream rivers. Id. In the northeastern United States headwater streams contribute greater than 60% of the water volume in larger tributaries, including navigable rivers. See, e.g., id. (citing Alexander et. al. 2007). The contributions of tributaries to river flows are often readily measured or observed, especially immediately below confluences, where tributary flows increase the flow volume and alter physical conditions, such as water temperature, in the main stream. The physical effects of tributaries are particularly clear after intense rainfall occurs over only the upper tributary reaches of a river network. For example, a study of ephemeral tributaries to the Río Grande in New Mexico found that after a storm event contributions of the stormflow from ephemeral tributaries accounted for 76% of the flow of the Río Grande. Id. at 3-7 to 3-8 (citing Vivoni et. al. 2006). A key effect of tributaries on the hydrologic response of river networks to storm events is dispersion, or the spreading of water output from a drainage basin over time. Geomorphic dispersion of connected tributaries influences the timing and volume of water reaching a river network outlet. See, e.g., id. at 3-10 (citing Saco and Kumar 2002). Tributaries also can reduce the amount of water that reaches downstream rivers and minimize downstream flooding, often through infiltration or seepage through channel beds and banks or through evapotranspiration. See, e.g., id. at 3-11 (citing Hamilton et al. 2005; Costelloe et.al. 2007).

One of the primary functions of tributaries is transporting sediment to downstream waters. Tributaries, particularly headwaters, shape and maintain river channels by accumulating and gradually or episodically releasing sediment and large woody debris into river channels. Sediment transport is also clearly provided by ephemeral streams. Effects of the releases of sediment and large woody debris are especially evident at tributary-river confluences, where discontinuities in flow regime and temperature clearly demonstrate physical alteration of river structure and function by headwater streams. Science Report at 3-14, 3-16, 3-18, 3-20 to 3-21. Sediment movement is critical for maintaining the river network, including rivers that are considered to be traditional navigable waters, as fluvial (produced by the action of a river or stream) sediments are eroded from some channel segments, and deposited in others downstream to form channel features, stream and riparian habitat which supports the biological communities resident downstream, and influence the river hydrodynamics. *See, e.g.*, Florsheim *et al.* 2008; Science Report at 3-13 (citing Church 2006). While essential to river systems, too much sediment

can impair ecological integrity by filling interstitial spaces, blocking sunlight transmission through the water column, and increasing contaminant and nutrient concentrations. *Id.* (citing Wood and Armitage 1997). Over-sedimentation thus can reduce photosynthesis and primary productivity within the stream network and otherwise have harmful effects on downstream biota, including on the health and abundance of fish, aquatic macrophytes (plants), and aquatic macroinvertebrates (insects) that inhabit downstream waters. *See, e.g.*, Wood and Armitage 1997. Headwater streams tend to trap and store sediments behind large structures, such as boulders and trees, that are transported downstream only during infrequent large storm events and that are the dominant means for downstream sediment transport. Science Report at 3-15 (citing Gomi and Sidle 2003; Gooderham *et al.* 2007). Similarly, large, infrequent disturbance events are the primary drivers for wood movement from headwater streams to downstream waters. *Id.* at 3-17 (citing Benda and Cundy 1990; Benda *et al.* 2005; Bigelow *et al.* 2007).

Tributaries can greatly influence water temperatures in tributary networks. This is important because water temperature is a critical factor governing the distribution and growth of aquatic life, both directly (through its effects on organisms) and indirectly (through its effects on other physiochemical properties, such as dissolved oxygen and suspended solids). Id. at 3-19 (citing Allan 1995). For instance, water temperature controls metabolism and level of activity in cold-blooded species like fish, amphibians, and aquatic invertebrates. See, e.g., Ice 2008. Temperature can also control the amount of dissolved oxygen in streams, as colder water holds more dissolved oxygen, which fish and other fauna need to breathe. Connections between tributaries and downstream rivers can affect water temperature in river networks. See, e.g., Science Report at 3-19 (citing Knispel and Castella 2003; Rice et al. 2008). In particular, tributaries provide both cold and warm water refuge habitats that are critical for protecting aquatic life in downstream traditional navigable waters, the territorial seas, and interstate waters. Id. at 3-42. Because headwater tributaries often depend on groundwater inputs, temperatures in these systems tend to be warmer in the winter (when groundwater is warmer than ambient temperatures) and colder in the summer (when groundwater is colder than ambient temperatures) relative to downstream waters. Id. (citing Power et al. 1999). Thus, tributaries provide organisms with both warm water and coldwater refuges at different times of the year. Id. (citing Curry et al. 1997; Baxter and Hauer 2000; Labbe and Fausch 2000; Bradford et al. 2001). For example, when temperature conditions in downstream waters are adverse, fish can travel upstream and use tributaries as refuge habitat, such as cold-water fish like Pacific salmon that seek important thermal refuge from warm downstream waters in small, cold tributaries. Id. (citing Curry et al. 1997; Cairns et al. 2005); Ebersole et al. 2015. Tributaries also help buffer temperatures in downstream waters. Science Report at 3-19 (citing Caissie 2006). Temperatures in tributaries affect downstream water temperature many kilometers away. Id. at 3-20 (citing Gardner and Sullivan 2004; Johnson et al. 2010).

ii. Tributaries Can Provide Functions that Restore and Maintain the Chemical Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

The scientific literature unequivocally demonstrates that tributaries exert a strong influence on the chemical integrity of downstream waters. Tributaries transform and export significant amounts of nutrients and carbon to downstream waters, serving important source functions that greatly influence the chemical integrity of downstream waters. Organic carbon, in both dissolved and particulate forms, exported from tributaries is consumed by downstream organisms. The organic carbon that is exported

downstream thus supports biological activity (including metabolism) throughout the river network. *See*, *e.g.*, Science Report at 3-30 (citing Fisher and Likens 1973; Meyer 1994; Wallace *et al.* 1997; Hall and Meyer 1998; Hall *et al.* 2000; Augspurger *et al.* 2008). Much or most of the organic carbon that is exported from tributaries has been altered either physically or chemically by ecosystem processes within the tributary streams, particularly by headwater streams. In addition to transformations associated with microbial and invertebrate activity, organic matter in streams can be transformed through other processes such as immersion and abrasion; photodegradation also can be important in ephemeral and intermittent streams where leaves accumulate in dry channels exposed to sunlight. *Id.* (citing Paul *et al.* 2006; Corti *et al.* 2011; Dieter *et al.* 2011; Fellman *et al.* 2013).

Nutrient export from tributaries has a large effect on downstream water quality, as excess nutrients from surface runoff from lawns and agricultural fields can cause algal blooms that reduce dissolved oxygen levels and increase turbidity in rivers, lakes, estuaries, and territorial seas. Water low in dissolved oxygen cannot support aquatic life; this widely-recognized phenomenon, known as hypoxia or "dead zones," occurs along coasts throughout the country, including the northern Gulf of Mexico and the Chesapeake Bay. Committee on Environment and Natural Resources 2000; Díaz and Rosenberg 2011; Murphy et al. 2011. Hypoxia threatens valuable commercial and recreational fisheries, including in the northern Gulf of Mexico, and reduces aquatic habitat quality and quantity. Committee on Environment and Natural Resources 2000; Freeman et al. 2007; Díaz and Rosenberg 2011; O'Connor and Whitall 2007; He and Xu 2015. The amount of nitrogen that is exported downstream varies depending on stream size, and how much nitrogen is present in the system. Nitrogen loss is greater in smaller, shallow streams, most likely because denitrification and settling of nitrogen particles occur at slower rates in deeper channels. Science Report at 3-23 (citing Alexander et al. 2000). At low loading rates, the biotic removal of dissolved nitrogen from water is high and occurs primarily in small tributaries, reducing the loading to larger tributaries and rivers downstream. At high nitrogen loading rates, tributaries become nitrogen saturated and are not effectively able to remove nitrogen, resulting in high nitrogen export to rivers. Id. at 3-25 to 3-26 (citing Mulholland et al. 2008). The transport of nitrogen and phosphorus downstream has also been well-documented, particularly in the cases of the Gulf of Mexico and the Chesapeake Bay. Tributary streams in the uppermost portions of the Gulf and Bay watersheds transport the majority of nutrients to the downstream waters; an estimated 85% of nitrogen arriving at the hypoxic zone in the Gulf originates in the upper Mississippi (north of Cairo, Illinois) and the Ohio River Basins. Goolsby et al. 1999. The export of nutrients from streams in the Mississippi River Basin has an effect on anoxia, or low oxygen levels, in the Gulf. Science Report at 3-24 (citing Rabalais et al. 2002). Similarly, nutrient loads from virtually the entire 64,000 square mile watershed affect water quality in the Chesapeake Bay. Simulation tools have been used to determine the nutrient and sediment load reductions that must be made at many different points throughout the entire watershed in order to achieve acceptable water quality in the mainstem of the Bay. These reductions included specific annual nitrogen caps on the upper reaches of the Susquehanna River in New York State, more than 400 miles from the mouth of the Chesapeake Bay. See e.g., Rabalais et al. 2002; U.S. Environmental Protection Agency 2003; New York Department of Environmental Conservation 2021.

Although tributaries export nutrients, carbon, and contaminants downstream, they also transform these substances. Phosphorous and nitrogen arrive at downstream waters having already been cycled, or taken up and transformed by living organisms, many times in headwater and smaller tributaries. Science Report at 1-3, 3-26 to 3-27 (citing Webster and Patten 1979; Newbold *et al.* 1981; Elwood *et al.* 1983;

Ensign and Doyle 2006). In addition, some of the nutrients taken up as readily available inorganic forms are released back to the water as organic forms that are less available for biotic uptake. Id. at 3-27 (citing Mulholland et al. 1988; Seitzinger et al. 2002). Similarly, nutrients incorporated into particulates are not entirely regenerated, but accumulate in longitudinally increasing particulate loads (*i.e.*, increases moving downstream). Id. (citing Merriam et al. 2002; Whiles and Dodds 2002; Hall et al. 2009). Headwater streams have seasonal cycles in the concentrations of phosphorous and nitrogen that are delivered downstream by accumulating nutrient derived from temporarily growing streambed biomass. Id. (citing Mulholland and Hill 1997; Mulholland 2004). Such variations have been demonstrated to affect downstream productivity. Id. (citing Mulholland et al. 1995). Nitrification, the microbial transformation of ammonium to nitrate, affects the form of downstream nutrient delivery. Nitrification occurs naturally in undisturbed headwater streams, but increases sharply in response to ammonium inputs, thereby reducing potential ammonium toxicity from pollutant inputs. Id. at 3-28 (citing Newbold et al. 1983; Chapra 1996; Bernhardt et al. 2002). Denitrification, the removal of nitrate from streamwater through transformation to atmospheric nitrogen, is widespread among headwater streams; research indicates that small, tributaries free from agricultural or urban impacts can reduce up to 40% of downstream nitrogen delivery through denitrification. Id. at 3-28 (citing Mulholland et al. 2008). Small tributaries also affect the downstream delivery of nutrients through abiotic processes. Streams can reduce phosphorus concentrations through sorption (*i.e.*, "sticking") to stream sediments. *Id.* (citing Meyer and Likens 1979). This is particularly beneficial to downstream chemical integrity where phosphorus sorbs to contaminants such as metal hydroxide precipitates. Id. (citing Simmons 2010).

Tributaries also store significant amounts of nutrients and carbon, functioning as important sinks for river networks so that they do not reach downstream traditional navigable waters, the territorial seas, or interstate waters. Small tributary streams in particular often have the greatest effect on downstream water quality, in terms of storage and reducing inputs to downstream waters. For instance, uptake and transformation of inorganic nitrogen often occurs most rapidly in the smallest tributaries. *See, e.g., id.* at 3-25 (citing Peterson *et al.* 2001). Small tributaries affect the downstream delivery of nutrients such as phosphorus through abiotic processes; such streams can reduce phosphorus concentrations by sorption to stream sediments.

Tributaries can also serve as a temporary or permanent source or sink for contaminants that adversely affect organisms when occurring at excessive or elevated concentrations, reducing the amounts of such pollutants that reach downstream traditional navigable waters, the territorial seas, or interstate waters. The transport of contaminants to downstream waters can impact water quality downstream if they are not stored in tributaries. *See, e.g., id.* at 3-34 (citing Wang *et al.* 2007). Tributaries can also serve as at least a temporary sink for contaminants that would otherwise impair downstream water quality. *See, e.g., id.* at 3-36 to 3-37 (citing Graf 1994).

The distances and extent of metal contaminant transport was shown in separate studies in the upper Arkansas River in Colorado, and Clark Fork River in Montana, where past mining activities impacted the headwater tributaries. River bed sediments showed that metals originating from the mining and smelting areas in the headwaters were reaching water bodies up to 550 km downstream. *Id.* at 3-34 (citing Axtmann and Luoma 1991; Kimball *et al.* 1995).

Military studies of the distribution, transport, and storage of radionuclides (*e.g.*, plutonium, thorium, uranium) have provided convincing evidence for distant chemical connectivity in river networks because the natural occurrence of radionuclides is extremely rare. From 1942 to 1952, prior to the full understanding of the risks of radionuclides to human health and the environment, plutonium dissolved in acid was discharged untreated into several intermittent headwater streams that flow into the Rio Grande at the Los Alamos National Laboratory, New Mexico. Id. at 3-36 (citing Graf 1994; Reneau et al. 2004). Also during this time, nuclear weapons testing occurred west of the upper Rio Grande near Socorro, New Mexico (Trinity blast site) and in Nevada, where fallout occurred on mountainous areas with thin soils that are readily transported to headwater streams in the upper Rio Grande basin. The distribution of plutonium within the Rio Grande illustrates how headwater streams transport and store contaminated sediment that has entered the basin through fallout and from direct discharge. Los Alamos Canyon, while only representing 0.4% of the drainage area at its confluence with the Rio Grande, had a mean annual bedload contribution of plutonium almost seven times that of the mainstem. Id. (citing Graf 1994). Much of the bedload contribution occurred sporadically during intense storms that were out of phase with flooding on the upper Rio Grande. Total estimated contributions of plutonium between the two sources to the Rio Grande were approximately 90% from fallout to the landscape and 10% from direct effluent discharge at Los Alamos National Laboratory. Id. at 3-36 to 3-37 (citing Graf 1994).

iii. Tributaries Can Provide Functions that Restore and Maintain theBiological Integrity of Downstream Traditional Navigable Waters, the TerritorialSeas, and Interstate Waters

Tributaries are biologically linked to downstream waters through the movement of living organisms or their reproductive propagules, such as eggs or seeds. For organisms that drift with water flow, biological connections depend on hydrological connections. However, many aquatic organisms are capable of active movement with or against water flow, and others disperse actively or passively over land by walking, flying, drifting, or "hitchhiking." All of these different types of movement form the basis of biological connectivity between headwater tributaries and downstream waters.

Headwater tributaries increase the amount and quality of habitat available to aquatic organisms. Under adverse conditions, small tributaries provide safe refuge, allowing organisms to persist and recolonize downstream areas once adverse conditions have abated. *See, e.g.*, Science Report at 3-38 (citing Meyer and Wallace 2001; Meyer *et al.* 2004; Huryn *et al.* 2005). Use of tributaries by salmon and other anadromous fish for spawning and by American eels and other catadromous fish for other life cycle needs is well-documented, but even non-migratory species can travel great distances within the river and tributary networks. *See, e.g., id.* at 3-40 (citing Gorman 1986; Sheldon 1998; Hitt and Angermeier 2008). Anadromous and catadromous fish are both types of diadromous fauna, which as explained in sections I.A.iv, require both freshwater and marine habitats over their life cycles. Anadromous fish are born in freshwater, spend most of their lives in saltwater, and return to freshwater to lay eggs. Catadromous fish breed in the ocean and spend most of their lives in freshwater. Tributaries also serve as an important source of food for biota in downstream rivers. Tributaries export plankton, vegetation, fish eggs, insects, invertebrates like worms or crayfish, smaller fish that originate in upstream tributaries and other food sources that drift downstream to be consumed by other animals. *See, e.g., id.* at 3-38 (citing Progar and Modenke 2002). For example, many fish feed on drifting insects, and numerous studies document the

downstream drift of stream invertebrates that then are eaten by fish in larger rivers. *See, e.g., id.* at 4-29 to 4-30 (citing Nakano and Murakami 2001; Wipfli and Gregovich 2002).

Biological connectivity also allows gene flow, or genetic connectivity, among tributary and river populations. Gene flow is needed to maintain genetic diversity in a species, a basic requirement for that species to be able to adapt to environmental change. Populations connected by gene flow have a larger breeding population size, making them less prone to the deleterious effects of inbreeding and more likely to retain genetic diversity or variation. *Id.* at 3-43 (citing Lande and Shannon 1996). Genetic connectivity exists at multiple scales and can extend beyond one a single river watershed, and for species capable of long distance movement (such as salmon), reveals complex interactions among spatially distant populations of aquatic organisms *Id.* (citing Hughes *et al.* 2009; Anderson 2010; Bohonak and Jenkins 2003).

Headwater streams provide unique habitat and protection for amphibians, fish, and other aquatic or semi-aquatic species living in and near the stream that may use the downstream waters for other portions of their life stages. See, e.g., Report at ES-8; Meyer et al. 2007. They also serve as migratory corridors for fish. Tributaries can improve or maintain biological integrity and can control water temperatures in the downstream waters. See, e.g., Report at 3-20 (citing Ebersole et. al. 2003; Gardner and Sullivan 2004; Johnson et al. 2010). Headwater streams also provide refuge habitat for riverine organisms seeking protection from temperature extremes, flow extremes, low dissolved oxygen, high sediment levels, or the presence of predators, parasites, and competitors. See, e.g., id. at 3-42 (citing Scrivener et al. 1994; Fraser et al. 1995; Curry 1997; Pires et al. 1999; Bradford et al. 2001; Cairns et al. 2005; Wigington et al. 2006; Woodford and McIntosh 2010). Headwater streams serve as a source of food materials such as insects, larvae, and organic matter to nourish the fish, mammals, amphibians, and other organisms in downstream streams, rivers, and lakes. See, e.g., id. at 4-22, 3-30, 3-31 (citing Fisher and Likens 1973; Meyer 1994; Wallace et al. 1997; Hall and Meyer 1998; Hall et al. 2000; Gomi et al. 2002; Augspurger et al. 2008). Disruptions in these biological processes affect the ecological functions of the entire downstream system. See, e.g., Kaplan et al. 1980; Vannote et. al. 1980. Headwater streams can help to maintain base flow in the larger rivers downstream, which is particularly important in times of drought. See, e.g., Science Report at 3-6, B-42, B-48 (citing Brooks and Lemon 2007; Tetzlaff and Soulsby 2008). At the same time, the network of headwater streams can regulate the flow of water into downstream waters, mitigating low flow and high flow extremes, reducing local and downstream flooding, and preventing excess erosion caused by flooding. See, e.g., Levick et al. 2008.

> iv. Human-made or Human-altered Tributaries Can Provide Functions that Restore and Maintain the Chemical, Physical, and Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

Under the final rule, human-made and human-altered tributaries are jurisdictional where they meet either the relatively permanent or significant nexus standard under paragraph (a)(3), except where they are excluded under paragraph (b). As discussed in the preamble, the agencies are adding an exclusion to the final rule for ditches (including roadside ditches) excavated wholly in and draining only dry land and that do not carry a relatively permanent flow of water. Under the pre-2015 regulatory regime, such ditches were considered generally non-jurisdictional. The scientific literature indicates that structures that

convey water continue to connect to and effect downstream waters, though the connectivity and effects can be different than that of natural streams. Indeed, because structures like ditches and canals can reduce water losses from evapotranspiration and seepage, such structures could enhance the extent of connectivity by more effectively conveying the water downstream. For example, ditches typically are constructed to move water off the landscape and to downstream waters more quickly, and moving that water more quickly can influence the functions that ditches perform.

Human-made and human-altered tributaries under the final rule include impoundments (which are also considered under paragraph (a)(2) of the final rule and discussed further in section III.C), ditches, canals, channelized streams, piped streams, and the like. Ditches and canals are wide-spread across the United States. Where ditches are streams that have been channelized, they would be tributaries under the final rule if they otherwise meet the provisions to be a "tributary" under paragraph (a)(3). *See* section IV.C.4 of the final rule preamble. Where ditches are constructed in natural streams, they are typically are purposely constructed to allow the hydrologic flow of the tributary to continue downstream and often straighten natural channels. Ditches can also intersect the water table, which can change the hydrological flow permanence as compared to the formerly natural stream. Human-made and human-altered tributaries, despite human manipulation, can continue to have chemical, physical, or biological connections to downstream waters supported by flow, though many of the natural functions may be lost or altered. Often-times human-made tributaries create channelized connections where they did not previously exist, such as canals that connect two rivers in different watersheds. Science Report at 1-11.

Tributary ditches and other human-made or human-altered waters can impact downstream waters individually or cumulatively with other tributaries and their adjacent wetlands. Tributary ditches and the like, as with other tributaries, can have chemical, physical, and biological connections with downstream waters that could impact those waters. Tributary ditches and canals can have perennial, intermittent, or ephemeral flow. Due to the often straightened and channelized nature of ditches, these tributaries quickly move water downstream to traditional navigable waters, the territorial seas, or interstate waters. Ditches reduce water storage on the landscape and increase conveyance, generally altering the timing of peak flows and shortening the response times to storms. Blann et al. 2009. Ditches, canals, and human-altered streams, like other tributaries, export sediment, nutrients, and other materials downstream and are effective at transporting water and these materials, including nitrogen, downstream. See, e.g., Morris et al. 2014; Schmidt et al. 2007; Sharpley et al. 2007; Strock et al. 2007. The more effective transport reduces residence time within sediments and limits nutrient cycling. Bukaveckas 2007; Needelman et al. 2007; Dollinger et al. 2015; Morris et al. 2014. The greater connection to the landscape often leads to higher nutrient loads down gradient, adding to downstream eutrophication. Alexander et al. 2008; David et al. 2010. Ditches provide habitat for fish and other aquatic organisms and can seasonally serve as a refuge for native fish species and as sheltered breeding grounds. See, e.g., Smiley Jr. et al. 2008; Colvin et al. 2009; Leslie et al. 2012. Fish and other aquatic organisms utilize canals and ditches to move to different habitats, sometimes over long distances. Rahel 2007. While ditches can have lower biodiversity than other aquatic ecosystems, they can provide habitat for species not found in larger waters. Leslie et al. 2012.

Human-made or human-altered tributaries can continue to have chemical, physical, and biological connections that affect the integrity of traditional navigable waters, the territorial seas, or interstate waters. Though the human-made or human-altered nature of such tributaries can change the nature of the

connections, it does not eliminate them. Thus, human-made and human-altered tributaries can continue to serve some of the same functions as natural tributaries, which in turn can greatly impact downstream traditional navigable waters, the territorial seas, or interstate waters, particularly when their functional contributions to the chemical, physical, and biological conditions of such downstream waters are evaluated in the aggregate.

v. Ephemeral and Intermittent Tributaries Can Provide Functions that Restore and Maintain the Chemical, Physical, or Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

Tributaries do not need to flow perennially to have a material influence on downstream traditional navigable waters, the territorial seas, or interstate waters. As described above in section I.A.iv., approximately 59% of streams across the United States (excluding Alaska) flow intermittently or ephemerally based on an analysis of National Hydrography Dataset (NHD). Science Report at 2-29 (citing Nadeau and Rains 2007). A more recent study found that ephemeral streams are undermapped or unmapped in the NHD, particularly in non-Western states, and estimated that ephemeral streams comprise 48% (range 43-56%) of stream channels by length in the conterminous United States. Fesenmyer et al. 2021. Compared with the humid regions of the country, stream and river networks in arid regions have a higher proportion of channels that flow ephemerally or intermittently. For example, ephemeral and intermittent streams are particularly prevalent in the arid and semi-arid Southwest, where they account for over 81% of streams. Levick et al. 2008; Fesenmyer et al. 2021. In Arizona, most of the stream channels—96% by length—are classified as ephemeral or intermittent. Despite their intermittent or ephemeral flow, these streams nonetheless perform many of the same ecological and hydrological functions documented in the scientific literature on perennial streams, through their movement of water, nutrients, and sediment to downstream waters. Levick et al. 2008; Science Report; Sullivan et al. 2019a, 2020; Fesenmyer et al. 2021. The importance of intermittent and ephemeral streams is documented in a 2008 peer-reviewed report by EPA's Office of Research and Development and the U.S. Department of Agriculture's Agricultural Research Service (Levick et al. 2008), which addresses the hydrological and ecological significance of ephemeral and intermittent streams in the arid and semi-arid Southwestern United States and their connections to downstream waters; the report is a state-of-the-art synthesis of current knowledge of the ecology and hydrology in these systems. Id.

Intermittent and ephemeral streams are chemically, physically, and biologically connected to downstream waters, and these connections have effects downstream. *See, e.g., id.* In some areas, stormflows channeled into alluvial floodplain aquifers by intermittent and ephemeral streams are the major source of annual streamflow in rivers. Perennial flows are not necessary for chemical connections. Periodic flows in ephemeral or intermittent tributaries can have a strong influence on biogeochemistry by connecting the channel and other landscape elements. *See, e.g.*, Report at 3-22 (citing Valett *et. al.* 2005). This episodic connection can be very important for transmitting a substantial amount of material into downstream rivers. *See, e.g., id.* (citing Nadeau and Rains 2007). Intermittent and ephemeral streams contribute to water quality in downstream traditional navigable waters, the territorial seas, and interstate waters by processing and uptaking nutrients like nitrogen. *Id.* at ES-8; Addy *et al.* 2019. Ephemeral desert streams have been shown to export particularly high sediment loadings. *See,* Science Report at 3-15 (citing Hassan 1990). Ephemeral streams can also temporarily and effectively store large amounts of

sediment that would otherwise wash downstream, contributing to the maintenance of downstream water quality and productive fish habitat. *See, e.g., id.* at 3-15 to 3-16 (citing Duncan *et al.* 1987; Trimble 1999; May and Gresswell 2003). This temporary storage of sediment thus helps maintain the chemical and biologic integrity of downstream waters.

Tributaries also need not be large rivers to have a material influence on downstream waters. As discussed above, the scientific literature supports the conclusion that tributaries, including headwater streams, provide important contributions to the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, and interstate waters. Headwater tributaries, the small streams at the uppermost reaches of the tributary network, are the most abundant streams in the United States. See, e.g., id. at 3-4 (citing Nadeau and Rains 2007). Collectively, they help shape the chemical, physical, and biological integrity of downstream waters, and provide many of the same functions as non-headwater streams. See, e.g., id. at ES-2, ES-7 to ES-9, 3-1. For example, headwater streams reduce the amount of sediment delivered to downstream waters by trapping sediment from water and runoff. See, e.g., Dieterich and Anderson 1998. Headwater streams shape river channels by accumulating and gradually or episodically releasing sediment and large woody debris into river channels. They are also responsible for most nutrient cycling and removal, and thus transforming and changing the amount of nutrients delivered to downstream waters. See, e.g., Science Report at 3-25 (citing Peterson et al. 2001). A close connection exists between the water quality of these streams and the water quality of traditional navigable waters, the territorial seas, and interstate waters. See, e.g., id.; Ohio Environmental Protection Agency 2015. Activities such as discharging a pollutant into one part of the tributary system are well-documented to affect other parts of the system, even when the point of discharge is far upstream from the water that experiences the effect of the discharge. See, e.g., National Research Council 1997; Dunnivant and Anders 2006.

The Science Report provides case studies of prairie streams and Southwest intermittent and ephemeral streams, two stream types whose jurisdictional status has been called into question post-*Rapanos*. These case studies highlight the importance of these streams to downstream waters, despite their small size and ephemeral or intermittent flow regime.

For example, the Science Report assessed the connectivity of prairie streams that drain temperate grasslands in the Great Plains physiographic region of the central United States and Canada. *Id.* at B-22 to B-37. Eventually, these streams drain into the Mississippi River or flow directly into the Gulf of Mexico or the Hudson Bay. *Id.* at 5-6, B-23. Climate in the Great Plains region ranges from semiarid to moist subhumid and intra- and interannual variation in precipitation and evapotranspiration is high. *Id.* at 5-6, B-23 to B-24 (Borchert 1950; Lauenroth *et al.* 1999; Boughton *et al.* 2010). This variation is reflected in the hydrology of prairie streams, which include ephemeral, intermittent, and perennial streamflows. *Id.* at 5-6, B-24 (citing Matthews *et al.* 1985; Matthews 1988; Brown and Matthews 1995; Sawin *et al.* 1999; Dodds *et al.* 2004; Bergey *et al.* 2008). Prairie streams are frequently subjected to the extremes of drying and flooding, and intermittent or flashy hydrology is prevalent in river networks throughout most of the Great Plains. *Id.* at B-24 (citing Matthews 1988; Zale *et al.* 1989; Poff 1996; Dodds *et al.* 2004). Prairie streams typically represent a collection of spring-fed, perennial pools and reaches, embedded within larger, intermittently flowing segments. *Id.* at B-36 (citing Labbe and Fausch 2000). Row cropping and livestock agriculture are the dominant land uses in the region, resulting in the withdrawal of water from stream channels and regional aquifers and its storage in reservoirs to support agriculture. *Id.* at 5-6, B-27

to B-28 (citing Cross and Moss 1987; Ferrington 1993; Galat *et al.* 2005; Matthews *et al.* 2005; Sophocleous 2010; Falke *et al.* 2011).

Prairie streams typically are connected to downstream waters. Like other types of streams, prairie streams present strong fluvial geomorphic evidence for connectivity to downstream waters, in that they have continuous channels (bed and banks) that make them physically contiguous with downstream waters. *Id.* at 5-6. Prairie river networks are dendritic and generally have a high drainage density, so they are particularly efficient at transferring water and materials to downstream waters. *Id.* Their pool-riffle morphology, high sinuosity, and seasonal drying, however, also enhance material storage and transformation. *Id.* The timing of connections between prairie streams and downstream waters is seasonal and therefore relatively predictable. *Id.* For example, high-magnitude floods tend to occur in late fall into later spring, although they also occur at other times during the year; this observation indicates that the magnitude of connections to downstream also varies seasonally. *Id.* at 5-6 and B-28 (citing Fausch and Bramblett 1991; Hill *et al.* 1992; Fritz and Dodds 2005).

The frequent and predictable connections between prairie streams and downstream waters have multiple physical, chemical, and biological consequences for downstream waters. Dissolved solids, sediment, and nutrients are exported from the prairie river network to downstream waters. *Id.* at 5-6. Ultimately, the expansion of the hypoxic zone in the Gulf of Mexico is a downstream consequence of cumulative nutrient loading to the Mississippi River network. *Id.* Relative to small streams and large rivers draining the moist eastern parts of the Mississippi River basin, small to midsized prairie streams deliver less than 25–50% of their nutrient load to the Gulf of Mexico. *Id.* at 5-6, B-32 (citing Alexander *et al.* 2008). Nonetheless, given the large number and spatial extent of headwater prairie streams connected to the Mississippi River, their cumulative effect contributes to downstream nutrient loading. *Id.* at 5-6, B-32.

Organisms inhabiting prairie streams have adapted to their variable hydrologic regimes and harsh physicochemical conditions via evolutionary strategies that include rapid growth, high dispersal ability, resistant life stages, fractional reproduction (*i.e.*, spawn multiple times during a reproductive season), and life cycles timed to avoid predictably harsh periods. *Id.* at 5-6, B-26 (citing Matthews 1988; Dodds *et al.* 1996b; Fausch and Bestgen 1997). Alterations in the frequency, duration, magnitude, and timing of flows —and thus hydrologic connectivity—are associated with the extinction or extirpation of species in downstream systems. *Id.* at 5-6, 3-41 (citing Morita and Yamamoto 2002; Letcher *et al.* 2007). Moreover, many fish species (*e.g.*, Arkansas River shiner, speckled chub, flathead chub) in prairie river networks require sufficient unfragmented (*i.e.*, connected) channel length with adequate discharge to keep their nonadhesive, semibuoyant eggs in suspension for incubation and early development. *Id.* at 5-6 to 5-7, B-35 (citing Cross and Moss 1987; Fausch and Bestgen 1997; Platania and Altenbach 1998; Durham and Wilde 2006; Perkin and Gido 2011). When these conditions are not met, the biological integrity of downstream waters is impaired. *Id.* at 5-7.

Human alteration of prairie river networks has affected the physical, chemical, and biological connectivity to and their consequences for downstream waters. Impoundments and water removal, through both surface flow diversions and pumping of ground-water aquifers, are common in this region. *Id.* at 5-7, B-27 to B-28 (citing Smith *et al.* 2002; Galat *et al.* 2005; Matthews *et al.* 2005; Sophocleous 2010). These activities have reduced flood magnitude and variability, altered timing, and increased

predictability of flows to downstream waters. *Id.* As a result, physical, chemical, and biological connections to downstream waters have been altered. *Id.* at B-28 (citing Cross and Moss 1987; Hadley *et al.* 1987; Galat and Lipkin 2000). In addition to the altered land uses and application of nutrients and pesticides for agriculture, human alteration of the river network itself, through channelization, levee construction, desnagging, dredging, and ditching, has enhanced longitudinal connectivity while reducing lateral and vertical connectivity with the floodplain and hyporheic zone, respectively. *Id.* at 5-7. Pumping from streams and ground water has caused historically perennial river segments to regularly dry during summer months. *Id.* at 5-7, B-27 to B-28 (citing Cross and Moss 1987; Ferrington 1993; Falke *et al.* 2011). Changes to the prairie's grazing (from bison to cattle) and burning regimes increase nutrient and suspended sediment loading to downstream waters. *Id.* at 5-7. Introduced species have extirpated endemic species and altered food web structure and processes in prairie streams, thereby affecting the biological integrity of downstream waters. *Id.*

Prairie streams have significant chemical, physical, and biological connections to downstream waters, despite extensive alteration of historical prairie regions by agriculture, water impoundment, water withdrawals, and other human activities, and the challenges these alterations create for assessing connectivity. *Id.* at B-36 to B-37 (citing Matthews and Robinson 1998; Dodds *et al.* 2004). The most notable connections are via flood propagation, contaminated sediment transport, nutrient retention and transformation, the extensive transport and movement of fish species (including eggs and larvae) throughout these networks, and refuges for prairie fishes. *Id.* at B-37 (citing Matthai 1969; Horowitz *et al.* 1988; Marron 1989; Dodds *et al.* 1996a; Fausch and Bestgen 1997; Platania and Altenbach 1998; Fritz and Dodds 2004; Fritz and Dodds 2005; Franssen *et al.* 2006; Alexander *et al.* 2008; Perkin and Gido 2011).

Similarly, southwestern intermittent and ephemeral streams exert strong influences on the structure and function of downstream waters, and the case study (included in the Science Report) echoes many of the findings of the functions of intermittent and ephemeral tributaries generally, which are described above. *See also* Goodrich *et al.* 2018. The case study focuses on the heavily studied San Pedro River, located in southeast Arizona, in particular, as a representative example of the hydrological behavior and the connectivity of rivers in the Southwest, but also examines evidence relevant to other Southwestern streams. *See* Science Report at B-37 to B-60.

Southwestern streams are predominantly ephemeral and intermittent (nonperennial) systems located in the southwestern United States. *Id.* at 5-7, B-37. Based on the NHD, 94%, 89%, 88%, and 79% of the streams in Arizona, Nevada, New Mexico, and Utah, respectively, are nonperennial. *Id.* (citing NHD 2008). Most of these streams connect to downstream waters, although 66% and 20% of the drainage basins in Nevada and New Mexico, respectively, are closed and drain into playas (dry lakes). *Id.* at 5-7. Southwestern streams generally are steep and can be divided into two main types: (1) mountainous streams that drain higher portions of basins and receive higher rates of precipitation, often as snow, compared to lower elevations; and (2) streams located in valley or plateau regions that generally flow in response to high-intensity thunderstorms. *Id.* at 5-7, B-39 (citing Blinn and Poff 2005). Headwater streams are common in both types of southwestern streams.

Nonperennial southwestern streams, excluding those that drain into playas, are periodically connected to downstream waters by low-duration, high-magnitude flows. *Id.* at 5-7. In contrast to streams

in humid regions where discharge is typically supplemented by ground water as drainage area increases, many southwestern streams lose streamflow to channel transmission losses as runoff travels downstream. *Id.* Connection of runoff and associated materials in ephemeral and intermittent streams to downstream waters is therefore a function of distance, the relative magnitude of the runoff event, and transmission losses. *Id.*

Spatial and temporal variation in frequency, duration, and timing of southwestern stream runoff is largely explained by elevation, climate, channel substrate, geology, and the presence of shallow groundwater. *Id.* at 5-8. In nonconstraining substrate, southwestern rivers are dendritic and their watersheds tend to have a high drainage density. *Id.* When high flows are present, southwestern streams are efficient at transferring water, sediment, and nutrients to downstream reaches. *Id.*; Goodrich *et al.* 2018. Ephemeral headwater streams shape larger downstream river channels by accumulating and gradually or episodically releasing stored materials such as sediment and large woody debris.³⁷ These materials help structure downstream river channels by slowing the flow of water through channels and providing substrate and habitat for aquatic organisms. Due to the episodic nature of flow in ephemeral and intermittent channels, sediment and organic matter can be deposited some distance downstream, and then moved farther downstream by subsequent precipitation events. *Id.* Over time, sediment and organic matter continue to move downstream and affect downstream waters. *Id.*

The southwestern streams case study describes the substantial connection and important consequences of runoff, nutrients, and particulate matter originating from ephemeral tributaries on the integrity and sustainability of downstream perennial streams. Channel transmission losses can be an important source of ground-water recharge that sustains downstream perennial stream and riparian systems. Science Report. For example, isotopic studies indicate that runoff from ephemeral tributaries like Walnut Gulch, Arizona supplies roughly half the San Pedro River's baseflow through shallow alluvial aquifer recharge. *Id.* Important cumulative effects of tributaries—that is the incremental contributions of individual streams in combination with similarly situated tributaries—are exemplified by ephemeral stream flows in arid landscapes, which are key sources of baseflow for downgradient waters. *Id* at 1-10 (citing Schlesinger and Jones 1984; Baillie *et al.* 2007; Izbicki 2007); Goodrich *et al.* 2018.

Human alterations to southwestern river networks affect the physical, chemical, and biological connectivity to downstream waters. Impoundments trap water, sediment, and particulate nutrients and result in downstream impacts on channel morphology and aquatic function. Science Report at 5-8. Diversion of water for consumptive can decrease downstream baseflows but typically does not affect the magnitude of peak flows. *Id.* Excessive ground-water pumping can lower ground-water tables, thereby diminishing or eliminating baseflows. *Id.* Urbanization increases runoff volume and flow velocity,

https://www.youtube.com/watch?v=qJ8JxBZt6Ws; Santa Clara Pueblo Fire/Rescue/EMS Volunteer Department, Greg Lonewolf, #4 Santa Clara Pueblo Flash Flood Event 01 Sept 2013 (April 14, 2017),

³⁷ Videos of ephemeral streams flowing after rain events in the Southwest highlight how effective ephemeral streams can be in transporting woody debris (*e.g.*, tree branches) and sediment downstream during the rainy season. *See, e.g.*, U.S. Department of Agriculture, Agricultural Research Service, *Multiflume Runoff Event August 1, 1990*, <u>https://www.tucson.ars.ag.gov/unit/WGWebcam/WalnutGulchWebcam.htm</u>; U.S. Geological Survey, *Post-fire Flash Flood in Coronado National Memorial, Arizona* (August 25, 2011),

<u>https://www.youtube.com/watch?v=nKOQzkRi4BQ;</u> Rankin Studio, *Amazing Flash Flood / Debris Flow Southern Utah HD* (July 19, 2019), <u>https://www.youtube.com/watch?v=_yCnQuILmsM</u>.

resulting in more erosive energy that can cause bank erosion, streambed down-cutting, and reduced infiltration to ground water. *Id.*

Flows from ephemeral streams are one of the major drivers of the dynamic hydrology of Southwest rivers (particularly of floods during monsoon seasons). Id. at B-42, B-49 (citing Goodrich et al. 1997; Yuan and Miyamoto 2008); Goodrich et al. Downstream river fishes and invertebrates are adapted to the variable flow regimes that are influenced strongly by ephemeral tributary systems, which provide isolated pools as refuges for fish during dry periods. Science Report at B-57 to B-58 (citing John 1964; Meffe 1984; Labbe and Fausch 2000; Rinne and Miller 2006; Lytle et al. 2008). Ephemeral tributaries in the Southwest also supply water to mainstem river alluvial aquifers, which aids in the sustaining river baseflows downstream. Id. at B-46 (citing Goodrich et al. 1997; Callegary et al. 2007); Goodrich et al. 2018. Ephemeral tributaries export sediment downstream during major hydrologic events; the sediment, in turn, influences the character of river floodplains and alluvial aquifers of downstream waters. Science Report at B-47 (citing Nanson and Croke 1992; Shaw and Cooper 2008); Goodrich et al. 2018. The nutrient and biogeochemical integrity of downstream Southwestern rivers, such as the San Pedro River, is heavily influenced by nutrient export from ephemeral tributaries after storm flow events. Science Report at 3-25, B-48 (citing Brooks and Lemon 2007; Fisher et al. 2001); Goodrich et al. 2018. Extensive downstream river riparian communities are supported by water, sediment and nutrients exported to the river from ephemeral tributaries; these riparian communities have a profound influence on the river attributes through shading, allochthonous (originating from outside of the channel) inputs of organic matter, detritus, wood, and invertebrates to the river. Science Report at B-47 to B-48 (citing Gregory et al. 1991; National Research Council 2002; Naiman et al. 2005; Stromberg et al. 2005; Baillie et al. 2007); Goodrich et al. 2018.

Ephemeral streams often have physical indicators of flow, such as an ordinary high water mark. *See, e.g.*, Lichvar and McColley 2008; Mersel and Lichvar 2014. Even discontinuous ephemeral streams, or streams characterized by alternating erosional and depositional reaches (*e.g.*, channelized flow interspersed with channel fans or other depositional areas) can exhibit OHWMs, and the Corps has developed field indicators to help field staff identify OHWM in these and other common stream types in the arid West. Lichvar and McColley 2008. In addition to discontinuous ephemeral streams, the Arid West OHWM manual also looks at alluvial fans, compound channels (streams characterized by a mosaic of terraces within a wide, active floodplain and frequently shifting low-flow channel(s)), and single-thread channels with adjacent floodplains. Lichvar and McColley 2008. These arid West stream types can exhibit an OHWM. *Id.*; Lefebvre *et al.* 2013. In arid non-perennial streams, the active floodplain represents a zone that most closely fits the concept of "ordinary" stream flow for use in delineating the OHWM. Lichvar and McColley 2008; Lichvar *et al.* 2009.

Intermittent and ephemeral tributaries are distinct from erosional features like rills and gullies that typically lack a defined channel or an ordinary high water mark. Gullies are small, relatively deep channels that are ordinarily formed on valley sides and floors where no channel previously existed. They are commonly found in areas with low-density vegetative cover or with soils that are highly erodible. *See*, *e.g.*, Brady and Weil 2002. Rills are very small incisions formed by overland water flows eroding the soil surface during rain storms. *See*, *e.g.*, Leopold 1994; Osterkamp 2008. Rills are less permanent on the landscape than streams and typically lack an ordinary high water mark, whereas gullies are younger than streams in geologic age, smaller than streams in size, and also typically lack an ordinary high water mark;

time has shaped streams into geographic features distinct from gullies and rills. See, e.g., American Society of Civil Engineers 1996; Osterkamp 2008. A rill is it is one of the first and smallest incisions to be formed as a result of concentrated flow eroding the land surface. Id. The two main processes that result in the formation of gullies are downcutting and headcutting, which are forms of longitudinal (incising) erosion. These actions ordinarily result in erosional cuts that are often deeper than they are wide, with very steep banks, often small beds, and typically only carry water during precipitation events. The principal erosional processes that modify streams are also downcutting and headcutting. In streams, however, lateral erosion is also very important. The result is that streams, except on steep slopes or where soils are highly erodible, are typically characterized by the presence of channel and an ordinary high water mark as compared to typical erosional features that are more deeply incised. It should be noted that some ephemeral streams are called "gullies" or the like in local parlance, even though they are not "gullies" in the technical sense. Such streams that meet the requirements to be jurisdictional as tributaries under paragraph (a)(3) of the final rule will be considered "waters of the United States," regardless of the name they are given locally. Similarly, a swale is a shallow trough-like depression that carries water mainly during rainstorms or snowmelt. Science Report at A-12. A swale might or might not be considered a wetland depending on whether it meets the three-factor wetland criteria, and only wetlands that meet the definition of "waters of the United States" are considered jurisdictional. A swale does not have the defined channel, including an ordinary high water mark, that a stream exhibits.

Through evidence provided throughout the report, including case studies on intermittent and ephemeral prairie streams and arid Southwestern streams, the Science Report is clear that intermittent and ephemeral streams can have important connections and impacts on downstream waters, regardless of where they are located geographically. The functions and effects of intermittent and ephemeral streams are discussed throughout the Science Report and this document.

vi. Tributary Lakes and Ponds Can Provide Functions that Restore and Maintain the Chemical, Physical, or Biological Integrity of Downstream Traditional Navigable Waters, the Territorial Seas, or Interstate Waters

Lakes and ponds can be considered tributaries under the final rule where they are directly part of the tributary system-that is, where they are in-stream or "run of the stream" (sometimes called "in-line" lakes and ponds), including lakes and ponds that are at the headwaters of the stream network. Lakes and ponds are also considered tributaries when they are connected to the tributary system via a pipe, culvert, dam, or similar structure. This is consistent with pre-2015 practice. The agencies recognize that the SAB has previously recommended that such lakes and ponds (which are considered lentic or "still water" systems) should not be combined into the same category as streams and rivers (which are considered lotic or "moving water" systems). SAB 2014b. Lakes and ponds that are considered tributaries outlet to the tributary network and contribute flow downstream at the outlet point. Thus, the agencies believe that considering such lakes and ponds to be tributaries under the final rule is consistent with information on the contributions of flow downstream of such lakes and ponds. In addition, one of the goals of this rulemaking effort is to retain the protections of the longstanding regulatory framework, and continuing to consider in-stream lakes and ponds as tributaries is consistent with that framework. Such lakes and ponds that are tributaries to traditional navigable waters, the territorial seas, interstate waters, or jurisdictional impoundments would be considered jurisdictional where they meet either the plurality or significant nexus standard under paragraph (a)(3) of the rule.

An in-stream lake or pond can be part of the headwaters (e.g., a headwater lake or pond that is directly connected to the headwater stream) or can be further downstream where, for example, a tributary flows into a lake that then flows into another tributary. Headwater and run-of-the-stream lakes and ponds serve many functions that affect the chemical, physical, and biological integrity of downstream waters. Such open waters can act as sinks, storing floodwaters, sediment, and nutrients, as these materials have the opportunity to settle out, at least temporarily, as water moves through the lake to downstream waters. See, e.g., Phillips et al. 2011; Kalinin et al. 2016. The attenuation of floodwaters can also maintain stream flows downstream. Phillips et al. 2011. Harvey and Schmadel (2021) found that present-day in-stream lakes and ponds lengthen the water transit times through watersheds by months or even years. In addition to lengthening transit times, Harvey and Schmadel (2021) also found that the in-stream lakes and ponds moderate downstream flow variability by lowering the flow peaks exacerbated by impervious surfaces and piped and tiled drainage (citing Graf 2006; Poff et al. 2006; Poff et al. 2007; Eng et al. 2013). Tributary lakes and ponds often elevate and shorten the frequency of low flows. Id. Tributary lakes and ponds can also act as sources, contributing flow, nutrients, sediment, and other materials downstream. Total Maximum Daily Loads (TMDLs) for nutrients have been established for many in-stream lakes across the country in recognition of the ability of lakes to transport nutrients downstream, contributing to downstream impairments. See, e.g., Maine Department of Environmental Protection 2006; U.S. Environmental Protection Agency 2012. Tributary lakes and ponds can also serve as habitat for species that then move downstream. For instance, brook trout that are stocked in headwater lakes in Idaho and Montana are capable of invading most downstream habitat, including through very steep channel slopes and waterfalls. Adams et al. 2001. These non-native species can then affect the biological integrity of downstream waters by impacting populations of native fish species, such as cutthroat trout, downstream. See, e.g., Dunham et al. 2002. For example, non-native trout were introduced in headwater lakes to the Little Kern River in the southern Sierra Nevada and dispersed downstream, causing the near-extinction of the native Little Kern golden trout. Knapp and Matthews 2000. These studies demonstrate the ability of organisms to travel from tributary lakes and ponds to downstream waters, which is not limited to just nonnative species; many other species can also move downstream and back again.

B. Adjacent Wetlands

Under the final rule, not all adjacent wetlands are jurisdictional. Adjacent wetlands are jurisdictional where they are wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters or where they are adjacent to jurisdictional impoundments or tributaries and meet either the relatively permanent or significant nexus standard. Adjacent wetlands meet the relatively permanent standard or with relatively permanent impoundments. Asserting Clean Water Act jurisdiction adjacent wetlands where they meet the standards in paragraph (a)(4) of the final rule aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that adjacent wetlands have chemical, physical, and biological effects on traditional navigable waters, the territorial seas, and interstate waters.

Adjacent wetlands serve many functions that directly influence the integrity of downstream waters including traditional navigable waters, the territorial seas, and interstate waters. Adjacent wetlands

store water, which can reduce flooding of downstream waters, and the loss of adjacent wetlands has been shown, in some circumstances, to increase downstream flooding. Adjacent wetlands maintain water quality and quantity, trap sediments, store and modify potential pollutants, and provide habitat for plants and animals, thereby sustaining the biological productivity of rivers, lakes, reservoirs, and estuaries, which may be traditional navigable waters, the territorial seas, or interstate waters. The scientific literature and Science Report support these conclusions, as discussed in greater detail below.

Based on the importance of the functions that are provided by adjacent wetlands to traditional navigable waters, the territorial seas, or interstate waters, the final rule's interpretation that the Clean Water Act protects adjacent wetlands where those adjacent wetlands are adjacent to traditional navigable waters, the territorial seas, or interstate waters or where those adjacent wetlands meet either the relatively permanent standard or the significant nexus standard reflects proper consideration of the objective of the Act and the best available science.

i. Adjacent Wetlands under the Final Rule

As discussed further in section III.B.ii below, in this final rule, the agencies are retaining the definition of "adjacent" unchanged from the 1986 regulations: "Adjacent means bordering, contiguous, or neighboring. Wetlands separated from other waters of the United States by man-made dikes or barriers, natural river berms, beach dunes, and the like are 'adjacent wetlands." In addition to retaining the definition of "adjacent" from the 1986 regulations, the final rule retains the adjacent wetlands provision of the 1986 regulations, with amendments to reflect the agencies' interpretation of the statutory limits on the scope of the "waters of the United States" informed by the law, the science, and agency expertise. Aquatic resources that meet this rule's definitions of "wetlands" and "adjacent" are assessed under this provision, where such wetlands are adjacent to traditional navigable waters, the territorial seas, interstate waters, jurisdictional impoundments, and tributaries. Adjacent wetlands that would be jurisdictional under the final rule include (1) wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters; (2) wetlands adjacent to and with a continuous surface connection to relatively permanent paragraph (a)(2) impoundments or jurisdictional tributaries when the jurisdictional tributaries meet the relatively permanent standard; and (3) wetlands adjacent to paragraph (a)(2) impoundments or jurisdictional tributaries when the wetlands meet the significant nexus standard. Under this rule, the relatively permanent standard (see section III.B.i.2) and the significant nexus standard (see section III.B.i.3) are independent jurisdictional standards for wetlands adjacent to impoundments of jurisdictional waters and for wetlands adjacent to tributaries. This is consistent with the pre-2015 regulatory regime.

As discussed further in sections I.A.iv and IV.A.iii, in the final rule the agencies are retaining their longstanding definition of "wetlands" from the 1986 regulations: "Wetlands means those areas that are inundated or saturated by surface or ground water at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include swamps, marshes, bogs, and similar areas." Asserting Clean Water Act jurisdiction over adjacent wetlands as outlined in paragraph (a)(4) of the final rule is supported by the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that such adjacent wetlands have chemical, physical, or biological effects on traditional navigable waters, the territorial seas, and interstate waters.

1. Wetlands Adjacent to Traditional Navigable Waters, the Territorial Seas, or Interstate Waters

Under the final, wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters are jurisdictional without need for further assessment. Wetlands must meet the definition of "adjacent' in paragraph (c)(2) of the final rule, which is described in more detail in section III.B.ii below. This is consistent with the pre-2015 regulatory regime. Asserting Clean Water Act jurisdiction over wetlands adjacent to traditional navigable waters, the territorial seas, interstate waters, or aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that such wetlands have chemical, physical, and biological effects on the waters to which they are adjacent, as discussed in sections III.B.iii, III.B.iv, III.B.v. For example, the scientific literature supports that wetlands within close proximity to traditional navigable waters, the territorial seas, or interstate waters improve water quality through assimilation, transformation, or sequestration of nutrients, sediment, and other pollutants that can affect water quality in such paragraph (a)(1) waters. These waters also provide important habitat for aquatic-associated species that utilize paragraph (a)(1) waters to forage, breed, and rest in.

2. Adjacent Wetlands under the Relatively Permanent Standard

Under the relatively permanent standard, wetlands that have a continuous surface connection with a tributary that meets that relatively permanent standard or with a relatively permanent impoundment are jurisdictional without the need for a significant nexus finding. The determination of whether a wetland is "adjacent" is distinct from whether an "adjacent" wetland meets the relatively permanent standard; however, wetlands that have a continuous surface connection to a relatively permanent water meet the definition of "adjacent" and thus are a subset of adjacent wetlands. Under the relatively permanent standard for adjacent wetlands, wetlands meet the continuous surface connection requirement if they physically abut, or touch, a relatively permanent paragraph (a)(2) impoundment or a jurisdictional tributary meets the relatively permanent standard, or if the wetlands are connected to these waters by a discrete feature like a non-jurisdictional ditch, swale, pipe, or culvert. A natural berm, bank, dune, or similar natural landform between an adjacent wetland and a relatively permanent water does not sever a continuous surface connection to the extent it provides evidence of a continuous surface connection.

A continuous surface connection does not mean a continuous *surface water* connection and does not require surface water to be continuously present between the wetland and water to which it is adjacent. The plurality opinion indicates that "continuous surface connection" is a "physical connection requirement." 547 U.S. at 751 n.13 (referring to "our physical-connect requirement" and later stating that *Riverside Bayview* does not reject "the physical-connection requirement"). The agencies' approach is consistent with science, as well as the regulatory definition of "wetlands," which does not require such aquatic resources to have water on the surface. As noted in paragraph (c)(1) of the final rule, wetlands are "those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions." Under this longstanding definition, wetlands are not required to express water at the surface—rather, wetland hydrology may be at the subsurface, for instance where soils are saturated by groundwater during the growing season. *See also* the Corps 1987 Wetlands

Delineation manual (noting that one of the key provisions of the agencies' regulatory definition is "[i]inundated or *saturated* soil conditions resulting from permanent or periodic inundation by *groundwater* or surface" (emphasis added)). Corps 1987 at p. 6. While some wetlands are permanently or semipermanently inundated, many aquatic resources that meet the regulatory definition of "wetlands" may never have surface water (*i.e.*, have saturated soils), may only have surface water during or immediately after precipitation events (*i.e.*, are irregularly inundated), or may only have water at the surface seasonally (*i.e.*, are seasonally inundated). Since wetlands frequently do not contain surface water, a requirement for continuous surface water between a relatively permanent water and adjacent wetlands would be illogical as a scientific and practical matter.

As discussed in section III.B.ii.1, scientific literature and the agencies technical expertise supporting regulating such wetlands with a continuous surface connection to tributaries that meet the relatively permanent standard and to jurisdictional relatively permanent impoundments as "waters of the United State" under the final rule.

3. Adjacent Wetlands under the Significant Nexus Standard

Under the significant standard for adjacent wetlands under paragraph (a)(4)(iii), wetlands adjacent to jurisdictional impoundments or to tributaries meet the significant nexus standard when the wetlands, either alone or in combination with similarly situated waters in the region, significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. Under the significant nexus standard in the final rule, the unit of analysis for assessing adjacent wetlands is the adjacent wetland, the tributary or impoundment to which it is adjacent, and all other adjacent wetlands that are in the tributary's catchment. The tributary includes the tributary reach of the same order, as well as any tributaries that are upstream. That portion of the tributary system, plus any adjacent wetlands, would be assessed to determine whether the adjacent wetland, in combination with similarly situated waters in the region, significantly affects the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. Wetlands adjacent to tributaries that meet the relatively permanent standard but that lack a continuous surface connect to such tributaries will also be assessed under the significant nexus standard, consistent with the pre-2015 regulatory regime. See Rapanos Guidance at 8. A determination of adjacency is based on an evaluation of the relationship between a wetland and the nearby jurisdictional water, which includes consideration of distance (proximity) and both physical and ecological connections between those waterbodies. In contrast, a determination of significantly affects is a different inquiry, which is based on evaluating whether there is a significant nexus between that adjacent wetland (in combination with similarly situated waters in the region) and a traditional navigable water, the territorial seas, or an interstate water.

As discussed in section III.B.ii, scientific literature and the agencies technical expertise supporting protecting adjacent wetlands that meet the significant nexus standard, as such wetlands, individually or in combination with similarly situated waters in the region, significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

ii. Definition of "Adjacent" Wetlands

Under the final rule, "adjacent" means bordering, contiguous, or neighboring, including wetlands separated from other "waters of the United States" by constructed dikes or barriers, natural river berms, beach dunes and the like. Further, waters that connect segments of, or are at the head of, a stream or river are "adjacent" to that stream or river. The term "adjacent" is a policy term and is not one that is used in the scientific literature. The terms "bordering," "contiguous," and "neighboring" are discussed further below. Under this definition, adjacency is focused on the distance between the wetland and the jurisdictional water. Whether the distance between the wetland and the jurisdictional water qualifies the wetland as bordering, contiguous, or neighboring (and therefore "adjacent") depends on the factual circumstances.

For purposes of adjacency, the entire wetland is adjacent if any part of the water is bordering, contiguous, or neighboring. Thus, under the relatively permanent standard, if any portion of a wetland has a continuous surface connection with a tributary that meets the relatively permanent standard or with a jurisdictional relatively permanent impoundment, the entire wetland meets the standard and thus is considered jurisdictional under paragraph (a)(4)(ii) of the final rule. The agencies' determination that an entire wetland is adjacent if any part of the water meets the definition of "adjacent" is informed by science and the agencies technical expertise and experience and is consistent with longstanding practice. It would be artificial to separate a single wetland into an adjacent and non-adjacent portion, as the entire wetland is a single functional unit, and the agencies' longstanding practice is to treat an entire adjacent wetland as one entity.

The agencies are continuing the practice outlined in the preamble to the proposed rule, consistent with the *Rapanos* Guidance, for the three well-established factors to determine adjacency. The agencies consider wetlands to be bordering, contiguous, or neighboring, and therefore "adjacent" if at least one of following three criteria is satisfied:

- (1) There is an unbroken surface or shallow sub-surface hydrologic connection to jurisdictional waters (discussed in section III.B.ii.2.a); or
- (2) They are physically separated from jurisdictional waters by "man-made dikes or barriers, natural river berms, beach dunes, and the like" (discussed in section III.B.ii.2.b); or
- (3) Where their proximity to a jurisdictional water is reasonably close such that "adjacent wetlands have significant effects on water quality and the aquatic ecosystem." *Riverside Bayview*, 474 U.S. at 135 n.9 (discussed in section III.B.ii.2.c).

If any one of the criteria is met, the wetland is "adjacent," but may require further analysis to determine if it meets the definition of "waters of the United States." *See Rapanos* Guidance at 5-8.

The agencies have determined that the longstanding regulation properly defines the term "adjacent" for purposes of the Clean Water Act because it is based on the concept of both reasonable proximity and scientific connections. Based on the scientific literature and the agencies' technical expertise and experience, as discussed further in the sections below, the agencies' longstanding definition of "adjacent" is supported by the science. This includes the terms to define "adjacent," "bordering, contiguous, or neighboring," including wetlands separated from other jurisdictional waters by man-made dikes or barriers, natural river berms, beach dunes, and the like. Wetlands that meet the definition of "adjacent' are physically proximate to and integrated with the water to which they are adjacent and, in

turn, can individually or cumulatively affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. The agencies have also determined that the longstanding implementation criteria for adjacency are supported by the science and the agencies' technical expertise and experience. The three criteria are well-established, are based on scientific principles about the relationship of wetland to the water to which it is adjacent, and complement the regulatory definition of "adjacent." Such wetlands that meet one of the three criteria can provide functions that significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

1. Bordering, Contiguous, or Neighboring Wetlands

The final rule continues to include wetlands that are bordering, contiguous, or neighboring within the definition of "adjacent," consistent with the current regulatory regime and the agencies' longstanding definition that existed prior to the 2020 NWPR. Within the definition of "adjacent," the terms bordering and contiguous are well understood. For continuity and clarity, the agencies will continue to interpret and implement the terms "bordering, contiguous, or neighboring" consistent with the current policy and practice. The science demonstrates that bordering, contiguous, or neighboring wetlands are integrated with the water to which they are adjacent and can affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

Wetlands that are bordering, contiguous, or neighboring wetlands include wetlands with an unbroken surface or shallow subsurface connection to jurisdictional waters; wetlands separated by constructed dikes or barriers, natural river berms, beach dunes, and the like; and wetlands within reasonably close proximity to other jurisdictional waters. This can include wetlands in the floodplain or the riparian area, wetlands that are outside of the floodplain or riparian area, run-of-the-stream wetlands, headwater wetlands, wetlands with a continuous surface connection to jurisdictional waters, wetlands behind a natural berm or similar natural landform, and wetlands behind artificial levees and similar artificial features, amongst others. Wetlands that are bordering or contiguous often directly abut the water which they are adjacent. (*e.g.*, they are not separated by uplands, an artificial dike, or similar artificial feature). *See, e.g.*, U.S. Army Corps of Engineers 2007a. Neighboring wetlands may or may not have a continuous surface connection to the waters to which they are adjacent, but science still demonstrates that they individually or cumulatively provide important functions that can impact on the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

As discussed further below, wetlands that are bordering, contiguous, or neighboring perform a myriad of critical chemical, physical, and biological functions that affect the integrity of traditional navigable waters, the territorial seas, or interstate waters. Such wetlands are integrally linked with the jurisdictional waters to which they are adjacent. Because of their close physical proximity to nearby jurisdictional waters, bordering, contiguous, or neighboring wetlands readily exchange their waters through the saturated soils surrounding the jurisdictional water or through surface exchange. This commingling of waters allows bordering, contiguous, or neighboring wetlands to both provide chemically transformed waters to streams and to absorb excess stream flow, which in turn can significantly affect traditional navigable waters, the territorial seas, or interstate waters. The close proximity also allows for the direct exchange of biological materials, including organic matter that serves as part of the food web of traditional navigable waters, the territorial seas, or interstate waters.

As previously discussed, "adjacent" is a policy term and not one found in the scientific literature. Similarly, "bordering, contiguous, or neighboring" are not terms found readily in the scientific literature regarding the relationship of a wetland to the tributary system. However, the agencies' technical expertise and experience support that bordering or contiguous wetlands are generally but not always found with the riparian area or floodplain. In addition, neighboring waters can also be located within the floodplain or a riparian area. Though this section addresses how bordering, contiguous, or neighboring wetlands can affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters, largely drawing from the scientific literature regarding waters in the floodplain or riparian area, the discussion of wetlands located outside the floodplain or riparian area used throughout this document, where appropriate and applicable, can also be used to support the agencies' determination that "bordering, contiguous, or neighboring" wetlands that meet the criteria to be jurisdictional under paragraph (a)(4) of the final rule should be considered "waters of the United States."

The scientific literature supports that "bordering, contiguous, or neighboring" wetlands located in riparian areas and floodplains are chemically, physically, and biologically connected to downstream traditional navigable waters, the territorial seas, or interstate waters and affect the integrity of such waters. The Science Report concludes that wetlands located in "riparian areas and floodplains are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality, including the temporary storage and deposition of channeling-forming sediment and woody debris, temporary storage of local ground water that supports baseflow in rivers, and transformation and transport of stored organic matter." Science Report at ES-2 to ES-3. Such waters act as the most effective buffer to protect downstream waters from nonpoint source pollution (such as nitrogen and phosphorus), provide habitat for breeding fish and aquatic insects that also live in streams, and retain floodwaters, sediment, nutrients, and contaminants that could otherwise negatively impact the condition or function of downstream waters.

Bordering, contiguous, or neighboring wetlands that are in the riparian area or floodplain lie within landscape settings that have bidirectional hydrological exchange with the waters to which they are adjacent. *Id.* at 2-7. Such wetlands play an integral role in the chemical, physical, and biological integrity of the waters to which they are adjacent and to traditional navigable waters, the territorial seas, or interstate waters. Riparian areas and floodplains often describe the same geographic region. *Id.* at 2-5. Therefore, the discussion of the functions of wetlands in riparian areas will typically apply to floodplains unless otherwise noted. Where connections arise specifically from the act of inundation of adjoining land during times of higher-than-normal water, the term "floodplain" is solely used to describe the area.

Riparian areas are transition zones between terrestrial and aquatic ecosystems that are distinguished by gradients in biophysical conditions, ecological processes, and biota. *Id.* at 2-4. Like riparian areas, wetlands are also transitional areas between terrestrial and aquatic ecosystems. Wetlands are often but not always found in riparian areas, but not all of the riparian area is a wetland. As noted in paragraph (c)(1) of the final rule, from a Clean Water Act regulatory perspective, wetlands are those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Only those wetlands that meet the provisions of paragraphs (a)(1) through (a)(5) of this final rule will be considered "waters of the United States." Wetlands in riparian areas

significantly influence exchanges of energy and matter with aquatic ecosystems. *See, e.g., id.* (citing National Research Council 2002).

As discussed in section I.A.iv, floodplains are low areas bordering streams, rivers, lakes, and impoundments and are inundated during moderate to high water events. *Id.* (citing Leopold 1994; Osterkamp 2008). Floodplains are also considered riparian areas, but not all riparian areas have floodplains. *Id.* at 2-5. Similar to riparian areas, wetlands are often but not always found in floodplains, but not all of the floodplain is a wetland. All rivers and streams within river networks have riparian areas, but small streams in constrained valleys are less likely to have floodplains than larger streams and rivers in unconstrained valleys. *Id.*

Wetlands, like open waters, are considered in-stream or "run-of-the-stream" where they are directly part of the tributary system. For example, an in-stream wetland can be part of the headwaters (*e.g.*, a headwater wetland) or can be further downstream where, for example, a tributary flows into a wetland that then flows into another tributary. For bordering, contiguous, or neighboring wetlands that are run-of-the-stream wetlands, the fact that such wetlands are in-stream often enhances their ability to filter pollutants and contaminants that would otherwise make it downstream; in-stream wetlands also attenuate floodwaters during wet periods and provide important sources of baseflow downstream during dry periods. *See, e.g., id.* at 4-21 (citing Morley *et al.* 2011).

One type of wetland often located in-stream are wetlands that are connected to the river network through a channel (*e.g.*, wetlands that serve as stream origins). These are wetlands from which a stream channel originates. Science Report at 4-2. Where these wetlands directly flow into jurisdictional waters, they are bordering, contiguous, or neighboring, and they meet the implementation criteria for a continuous surface connection. Because these adjacent wetlands are often located at the headwaters, the stream to which they are adjacent may not be large enough to have a floodplain (*e.g.*, they may lie at the hillslope or in high gradient areas), and thus in such circumstances they are generally non-floodplain wetlands (however, some stream-origin wetlands can be located within the floodplain or riparian area). They are part of the stream network itself, and along with first- and second-order streams, form the headwaters of the river network. Such bordering, contiguous, or neighboring wetlands have a direct hydrologic connection to the tributary network via unidirectional flow from the wetland to the headwater stream.

Wetlands that serve as stream origins connect via perennial, intermittent, or ephemeral drainages to river networks. *Id.* at 4-21 (citing Rains *et al.* 2006; Rains *et al.* 2008; Morley *et al.* 2011; McDonough *et al.* 2015). Regardless of the permanence of flow, such wetlands have an impact on downstream waters. *Id.* at 4-2, 4-40. Wetland seeps, for example, can form where groundwater discharges from breaks in slope. *Id.* at 4-20 (citing Hall *et al.* 2001; O'Driscoll and DeWalle 2010). They often have perennial connections to the stream, providing important sources of water downstream, particularly during summer baseflow. *Id.* at 4-21 (citing Morley *et al.* 2011). In Maine, for example, seeps were found to provide 40 to 80% of stream water during baseflow periods. *Id.* In other cases, surface connections between channel origin wetlands and streams are intermittent or ephemeral. In addition to surface water connections, groundwater flow can hydrologically connect wetlands that serve as stream origins with the stream network. *Id.* at 4-22.

Wetlands at the channel origin generally have chemical, physical, and biological effects on traditional navigable waters, the territorial seas, or interstate waters, including hydrologic, water quality, and habitat functions, regardless if the outflow from the wetland to the stream is perennial, intermittent, or ephemeral. *Id.* Like other wetlands, wetlands that serve as stream origins can transport channel-forming sediment and woody debris, transport stored organic matter, remove and transform pollutants and excess nutrients such as nitrogen and phosphorus, attenuate and store floodwaters, contribute to stream baseflow through groundwater recharge, and provide habitat for breeding fish, amphibians, reptiles, birds, and other aquatic and semi-aquatic species that move from the wetlands to the river network. *Id.* at 4-40, 4-42. In some cases, however, where wetlands that serve as stream origins are already saturated prior to rainfall, they can convey stormwater quickly downstream and thus actually increase flood peaks. *Id.* (citing Bullock and Acreman 2003). This is because the wetland soil, if completely saturated, cannot store any additional water, making the wetland unable to store floodwater. *Id.*

Bordering, contiguous, or neighboring wetlands, including wetlands that serve as stream origins, have important chemical, physical, and biological connections downstream that affect traditional navigable waters, the territorial seas, and interstate waters. Where they have a direct hydrologic connection to the stream network, that connection facilitates the impact they have downstream. This impact on downstream waters occurs regardless of whether their connection to the tributary network is perennial, intermittent, or ephemeral. Thus, bordering, contiguous, or neighboring wetlands serve important functions, which in turn can impact traditional navigable waters, the territorial seas, and interstate waters, particularly when their functional contributions to the chemical, physical, or biological conditions of downstream waters are combined at a catchment or watershed scale.

Bordering, contiguous, or neighboring wetlands can include those wetlands located outside the riparian area or floodplain. Such wetlands include wetlands with an unbroken surface or shallow subsurface connection to jurisdictional waters (particularly when the connection is via a discrete feature like a non-jurisdictional ditch, swale, pipe, or culvert or via an unbroken shallow subsurface connection), wetlands separated from a jurisdictional water by a natural berm or the like or by a constructed levee or the like, and wetlands within reasonably close proximity of a jurisdictional water. The science and the agencies' technical expertise and experience support implementing the definition of "adjacent" to include such wetlands, and such wetlands are integrated with the water to which they are adjacent and can provide functions that significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

Wetlands with an unbroken surface or shallow subsurface connection to other jurisdictional waters are "bordering, contiguous, or neighboring." As discussed further in section III.B.ii.2.a below, the science and the agencies' technical expertise and experience support implementing the definition of "adjacent" to include such wetlands, and such wetlands are integrated with the water to which they are adjacent and can provide functions that significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

Wetlands separated from other "waters of the United States" by man-made dikes or barriers, natural river berms, beach dunes, and the like are "bordering, contiguous, or neighboring" and thus "adjacent" under the final rule, consistent with the agencies' longstanding definition of "adjacent." As discussed further in section III.B.ii.2.b. below, the science and the agencies' technical expertise and

experience support including such wetlands in the definition of "adjacent," and such wetlands can provide functions that significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

Wetlands within reasonably close proximity of jurisdictional waters are "bordering, contiguous, or neighboring." The agencies have always recognized that adjacency is bounded by proximity. In addition, the science is clear that a wetland's proximity to downstream waters influences its impact on those waters. The Science Report states, "[s]patial proximity is one important determinant of the magnitude, frequency and duration of connections between wetlands and streams that will ultimately influence the fluxes of water, materials and biota between wetlands and downstream waters." *Id.* at ES-11. Generally, wetlands that are closer to a jurisdictional water are more likely to be connected to that water than wetlands that are farther away. As discussed further in section III.B.ii.2.c below, the science and the agencies' technical expertise and experience support implementing the definition of "adjacent" to include such wetlands. "Bordering, contiguous, or neighboring" wetlands that are within close proximity of the jurisdictional water are integrated with the water to which they are adjacent and can have significant chemical, physical, and biological connections with and effects on traditional navigable waters, the territorial seas, or interstate waters.

Based on a review of the scientific literature and the agencies' technical expertise and experience, the final rule continues the longstanding interpretation in regulations predating the 2020 NWPR that wetlands that are "bordering, contiguous, or neighboring" to other jurisdictional waters are "adjacent," including wetlands separated from other jurisdictional waters by man-made dikes or barriers, natural river berms, beach dunes, and the like. Under the final rule, adjacent wetlands are jurisdictional when they are adjacent to a traditional navigable water, the territorial seas, or an interstate water, and where they are adjacent to jurisdictional tributaries and jurisdictional impoundments and meet either the relatively permanent standard or the significant nexus standard as described in paragraph (a)(4).

2. Determination of Adjacent Wetlands

As with the proposed rule, the final rule will continue the well-established practice outlined in the *Rapanos* Guidance for the three criteria to determine adjacency. First, there is an unbroken surface or shallow subsurface connection to jurisdictional waters; this hydrologic connection maybe intermittent. Second, the wetlands are physically separated from jurisdictional waters by human-made dikes or barriers, natural river berms, beach dunes, and the like (discussed in section III.B.ii.2.b). Or third, their proximity to a jurisdictional water is reasonably close such that "adjacent wetlands have significant effects on water quality and the aquatic ecosystem." *Riverside Bayview*, 474 U.S. at 135 n.9. *See also Rapanos* Guidance at 5-6. Wetlands that adjacent will need to meet one of these under the factors under the final rule and are "bordering, contiguous, or neighboring." Such wetlands would then be assessed under paragraph (a)(4) to determine if they are "waters of the United States" under the final rule. These longstanding criteria are derived from the regulatory definition of "adjacent," and the science, along with the agencies' technical and expertise, supports including such wetlands in the definition of "adjacent." As discussed further below in this section, such wetlands can provide functions that significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

a. Wetlands with an Unbroken Surface or Shallow Subsurface Connection

Adjacent wetlands under the final rule include wetlands with an unbroken surface or shallow subsurface connection. Wetlands that meet the relatively permanent standard will always meet this criterion, as they have a continuous surface connection to the jurisdictional water to which they are adjacent. An unbroken surface or shallow subsurface connection can be established, for example, where the wetland directly abuts the jurisdictional water or by a non-jurisdictional physical feature that provides the direct connection between the wetland and a jurisdictional water, such as a pipe, culvert, non-jurisdictional ditch, or flood gate, that has at least periodic flow. Under the final rule, an unbroken surface connection includes but is not limited to surface hydrologic connections, including confined surface hydrologic connections.

An unbroken surface or shallow subsurface hydrologic connection to jurisdictional waters may be established by a physical feature or discrete conveyance that supports periodic flow between the wetland and a jurisdictional water, such as a pipe, culvert, non-jurisdictional ditch, or flood gate. Water does not have to be continuously present in this hydrologic connection and the flow between the wetland and the jurisdictional water may move in either or both directions. The hydrologic connection need not itself be a jurisdictional water.

Both confined surface and shallow subsurface connections are forms of direct hydrologic connections between adjacent wetlands and the waters to which they are adjacent. Confined surface connections consist of permanent, intermittent, or ephemeral surface connections through directional flowpaths, such as (but not limited to) swales, gullies, rills, and ditches. In some cases, these connections will be a result of "fill and spill" hydrology, but only where such a surface connection is unbroken. A directional flowpath is a path where water flows repeatedly from the wetland to the nearby jurisdictional water that at times contains water originating in the adjacent wetland as opposed to just directly from precipitation.

A shallow subsurface hydrologic connection is lateral water flow through a shallow subsurface layer, such as may be found in steeply sloping forested areas with shallow soils, soils with a restrictive horizon, or in karst systems. Devito et al. 1996; O'Driscoll and Parizek 2003; Cook and Hauer 2007. A shallow subsurface connection also exists, for example, when the adjacent wetland and the water to which it is adjacent are in contact with the same shallow aquifer (e.g., an alluvial aquifer) or with the same shallow water table which fluctuates within the soil profile, sometimes rising to or near the ground surface. In addition, water can move within confined human-made subsurface conveyance systems such as drain tiles and storm sewers (Science Report at 3-3 and 4-24 (citing Schiller et al. 2012)), and within natural subsurface conveyance systems such as karst topography or lava tubes (see, e.g., Miller et al. 2013; O'Driscoll and Parizek 2003; U.S. EPA 2002). Shallow subsurface connections may be found below the ordinary root zone (below 12 inches), where other wetland delineation factors may not be present. A combination of physical factors may reflect the presence of a shallow subsurface connection, including, position in the landscape (for example, on a slope directing flow from wetland to jurisdictional waters), stream hydrograph, and soil surveys (for example, exhibiting indicators of high transmissivity over an impermeable layer), and information indication that the water table in the stream is lower than in the shallow subsurface.

Shallow subsurface connections are distinct from deeper groundwater connections, which do not satisfy the requirement for adjacency, in that the former exhibit a direct connection to the water found on the surface in wetlands. While they may provide the connection establishing jurisdiction, these shallow subsurface flows are not "waters of the United States."

The Science Report supports that the functions of non-floodplain wetlands clearly affect the condition of downstream waters if a visible (e.g., channelized) surface-water or a regular shallow subsurface-water connection to the river network is present. Science Report at ES-3. The SAB also noted the importance of shallow subsurface connections and stated, "[t]he available science...shows that groundwater connections, particularly via shallow flow paths in unconfined aquifers, can be critical in support the hydrology and biogeochemical functions of wetlands and other waters." SAB 2014b. For nonfloodplain wetlands connected through ground-water flows, less distant areas are generally connected through shallower flowpaths, assuming similar soil and geologic properties. Id. at 2-39. These shallow subsurface flows have the greatest interchange with surface waters and travel between points in the shortest amount of time. Id. Water can move within confined man-made subsurface conveyance systems such as drain tiles and storm sewers, and in karst topography. See, e.g., Miller et al. 2013; O'Driscoll and Parizek 2003; U.S. EPA 2002. Confined subsurface systems can move water, and potential contaminants, directly to surface waters rapidly without the opportunity for nutrient or sediment reduction along the pathway. Science Report at 3-28; 4-24 (citing Royer et al. 2004). Shallow subsurface connections move quickly through the soil and impact surface water directly within hours or days rather than the years it may take long pathways to reach surface waters. The Science Report refers to local groundwater flow or shallow groundwater flow, which is a type of shallow subsurface connections. Id. at 2-11. Such shallow subsurface connections flow from the highest elevations of the water tables to nearby lowlands or surface waters. Id. (citing Winter and LaBaugh 2003). These are dynamic hydrologic connections that have the greatest interchange with surface waters. Id. The presence of a confining layer near the surface also leads to shallow subsurface flows through the soil. Id. at 2-34.

Asserting Clean Water Act jurisdiction over adjacent wetlands with an unbroken surface or shallow subsurface connection to jurisdictional waters where they are adjacent to traditional navigable waters, the territorial seas, or interstate waters, or where they meet either the relatively permanent or significant standard, aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that such wetlands have chemical, physical, and biological effects on downstream traditional navigable waters, the territorial seas, and interstate waters.

b. Wetlands Separated by Constructed Dikes or Barriers, Natural River Berms, Beach Dunes, and the Like

Wetlands separated from other "waters of the United States" by constructed dikes or barriers, natural river berms, beach dunes and the like are adjacent under the final rule, which is consistent with the pre-2015 regulatory regime and continued from the regulatory definition of "adjacent" that existed prior to the 2020 NWPR. This has been a longstanding part of the concept of adjacency under the agencies' regulations defining "waters of the United States" (except in the 2020 NWPR), and the agencies are restoring their longstanding definition of "adjacent." Including wetlands behind certain natural features or artificial barriers as adjacent wetlands is also consistent with Justice Kennedy's opinion, which noted the important functions that such wetlands can play: "In many cases, moreover, filling in wetlands separated

from another water by a berm can mean that floodwater, impurities, or runoff that would have been stored or contained in the wetlands will instead flow out to major waterways. With these concerns in mind, the Corps' definition of adjacency is a reasonable one, for it may be the absence of an interchange of waters prior to the dredge and fill activity that makes protection of the wetlands critical to the statutory scheme." *Rapanos*, 547 U.S. at 775.

If a wetland is separated from a jurisdictional water by man-made dikes or barriers, natural river berms, beach dunes, and the like, then the wetlands adjacent under this rule, consistent with the 1986 regulations. Waters separated by constructed dikes or barriers, natural river berms, beach dunes, and the like are considered bordering, contiguous, or neighboring, and the presence of the artificial barrier or natural landform does not affect their adjacency. No additional identification of a hydrologic connection between the wetland and the jurisdictional water is required for such wetlands to be considered adjacent. For example, a wetland that is separated from a jurisdictional tributary simply by a 40-foot road meets the longstanding definition of adjacent. With respect to beach dunes and similar natural landforms like river berms, more than one dune may exist between an adjacent wetland and jurisdictional water (including primary and secondary dunes), because beach dunes typically function as an interdunal system (particularly on barrier islands). For example, interdunal wetlands which are located between dune ridges would be adjacent. Similarly, more than one natural river berm may exist between an adjacent wetland and a jurisdictional water because such natural river berms were formed as a system in the floodplain due to overbank flooding of the river and typically function as a system. Thus, more than one natural river berm may exist between an adjacent wetland and the jurisdictional water.

In some cases, a wetland may be separated from a jurisdictional water by more than one humanmade dike or barrier or multiple types of barriers and landforms (*e.g.*, a wetland separated by a humanmade barrier and a natural river berm). The agencies will assess such wetlands consistent with the other adjacency criteria previously described (*i.e.*, by identifying the presence of an unbroken surface or shallow subsurface connection or determining that their proximity to a jurisdictional water is reasonably close).

Under the final rule, such wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters are jurisdictional under paragraph (a)(4)(i). Such wetlands that are adjacent to tributaries or impoundments would likely require a significant nexus analysis under paragraph (a)(4)(iii) of the final rule, with the exception of wetlands behind natural landforms where the river berm, beach dune, or the like serves as an indication of a continuous surface connection and wetlands behind artificial barriers or natural landforms that maintain a continuous surface connection to relatively permanent impoundments or tributaries, such as through a culvert, flood gate, or break in the dam, beach dune, or levee. Such wetlands with a continuous surface connection to relatively permanent impoundments or tributaries, such as through a culvert, flood gate, or break in the dam, beach dune, or levee. Such wetlands with a continuous surface connection to relatively permanent waters would be evaluated under paragraph (a)(4)(ii) of the final rule. Asserting Clean Water Act jurisdiction over adjacent wetlands separated from other jurisdictional waters by constructed dikes or barriers, natural river berms, beach dunes and the like, where the wetlands meet the standards in paragraph (a)(4) of the final rule, aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that such wetlands have chemical, physical, and biological effects on traditional navigable waters, the territorial seas, and interstate waters.

If uplands separating a wetland from jurisdictional water can reasonably be characterized as "man-made dikes or barriers, natural river berms, beach dunes, and the like," then, under the final rule, the wetlands are adjacent even if no apparent hydrologic connection exists. The terms earthen dam, dike, berm, and levee are used to describe similar structures whose primary purpose is to help control flood waters. Such structures vary in scale and size. A levee is an embankment whose primary purpose is to furnish flood protection from seasonal high water and which is therefore subject to water loading for periods of only a few days or weeks a year. Earthen embankments that are subject to water loading for prolonged periods (longer than normal flood protection requirements) are called earth dams. There are a wide variety of types of structures and an even wider set of construction methods. These range from a poorly constructed, low earthen berm pushed up by a backhoe to a well-constructed, impervious core, riprap lined levee that protects houses and cropland. Generally, levees are built to detach the floodplain from the channel, decreasing overbank flood events. Franklin *et al.* 2009. The investigation methods to determine the presence or absence of the hydrologic connection depend on the type of structure, the underlying soils, the presence of groundwater, and the depth of the water table. U.S. Army Corps of Engineers 2000 at 1-1.

Barriers between wetlands and jurisdictional waters have been constructed for centuries (*see, e.g.*, National Geographic Society 2022 ("As early as 2500 BCE, the Indus Valley Civilization, with urban centers in what is today Mohenjo Daro and Harappa, Pakistan, used levees to protect land near the Indus River" and "Since the 18th century, levees have protected Louisiana and other nearby states from flooding by the Mississippi River")). Human-made berms and the like are fairly common along streams and rivers across the United States and often accompany stream channelization. Franklin *et al.* 2009; Morrison *et al.* 2018. One study conducted in Portland, Oregon found that 42% of surveyed wetlands had dams, dikes, or berms. Kentula *et al.* 2004. Likewise, over 90% of the tidal freshwater wetlands of the Sacramento-San Joaquin Delta have been diked or leveed. Simenstad *et al.* 1999. At least 40,000 kilometers of levees, floodwalls, embankments, and dikes are estimated across the United States, with approximately 17,000 kilometers (more than 10,000 miles) of levees in the Upper Mississippi Valley alone. Gergel *et al.* 2002.

Adjacent wetlands separated from the tributary network by dikes, levees, berms, and the like typically continue to have a hydrologic connection to the jurisdictional waters to which they are adjacent. This is because berms and similar features typically do not block all water flow. Indeed, even dams, which are specifically designed and constructed to impound large amounts of water effectively and safely, do not prevent all water flow, but rather allow seepage under the foundation of the dam and through the dam itself. *See, e.g.*, International Atomic Energy Agency 2003; U.S. Bureau of Reclamation; Federal Energy Regulatory Commission 2005.

Seepage is the flow of a fluid through the soil pores. Seepage through a dam, through the embankments, foundations or abutments, or through a berm is a normal condition. Kovacic *et al.* 2000; Federal Energy Regulatory Commission 2005. This is because water seeks paths of least resistance through the berm or dam and its foundation. Association of State Dam Safety Officials 2021. All earth and rock-fill dams are subject to seepage through the embankment, foundation, and abutments. U.S. Army Corps of Engineers 1993; U.S. Army Corps of Engineers 2004. Concrete gravity and arch dams similarly are subject to seepage through the foundation and abutments. U.S. Army Corps of Engineers 1993. Levees and the like are subject to breaches and breaks during times of floods. Nilsson *et al.* 2005.
Levees are similarly subject to failure in the case of extreme events, such as the extensive levee failures caused by Hurricanes Katrina and Rita. Day *et al.* 2007. In designing levees and similar structures, seepage control is necessary to prevent possible failure caused by excessive uplift pressures, instability of the downstream slope, piping through the embankment and/or foundation, and erosion of material by migration into open joints in the foundation and abutments. *Id.*; Kovacic *et al.* 2000; U.S. Bureau of Reclamation; International Atomic Energy Agency 2003; California Division of Safety of Dams 1993.

The rate at which water moves through the embankment depends on the type of soil in the embankment, how well it is compacted, the foundation and abutment preparation, and the number and size of cracks and voids within the embankment. All but the smallest earthen dams are commonly built with internal subsurface drains to intercept water seeping from the reservoir (*i.e.*, upstream side) to the downstream side. U.S. Army Corps of Engineers 1995. Where it is not intercepted by a subsurface drain, the seepage will emerge downstream from or at the toe of the embankment. Association of State Dam Safety Officials 2021. Seepage may vary in appearance from a "soft," wet area to a flowing "spring." It may show up first as an area where the vegetation is lush and darker green. Cattails, reeds, mosses, and other marsh vegetation may grow in a seepage area. *Id*.

Engineered berms are typically designed to interfere with the seasonal pattern of water level (hydroperiod) of the area behind the berm, reducing the frequency and severity of inundation. Berms are not designed to eliminate all hydrologic connection between the channel on one side and the area behind the berm on the other. It is almost always impracticable to build a berm that will not be overtopped by a flood of maximum severity, and most berms are not designed to withstand severe floods. See, e.g., U.S. Army Corps of Engineers 1993. Levees are designed to allow seepage and are frequently situated on foundations having natural covers of relatively fine-grain impervious to semipervious soils overlying pervious sands and gravels. U.S. Army Corps of Engineers 2005a. These surface strata constitute impervious or semipervious blankets when considered in connection with seepage. Principal seepage control measures for foundation underseepage are (a) cutoff trenches, (b) riverside impervious blankets, (c) landslide berms, (d) pervious toe trenches, and (e) pressure relief wells. U.S. Army Corps of Engineers 2000. Overtopping of an embankment dam is very undesirable because the embankment materials may be eroded away. Additionally, only a small number of concrete dams have been designed to be overtopped. Water normally passes through the main spillway or outlet works; it should pass over an auxiliary spillway only during periods of high reservoir levels and high water inflow. All embankment and most concrete dams have some seepage. See, e.g., Texas Commission on Environmental Quality 2006. However, it is important to control the seepage to prevent internal erosion and instability. Proper dam construction, and maintenance and monitoring of seepage provide control.

It is important to note that natural river berms are formed by sediment deposits accumulating at or near the stream bank during flood events. Such berms vary in height from inches to feet, and also can be quite wide. Wharton *et al.* 1982. Berm-like landforms known as natural levees occur naturally and do not isolate adjacent wetlands from the streams that form them. Hydrologic connections can be bidirectional across berms or other similar features when integrated over time during and after floods when the hydraulic or hydrostatic gradient changes direction. Natural levees and the wetlands and waters behind them are part of the floodplain, including along some small streams and streams in the arid West. Johnston *et al.* 2001. Every flowing watercourse transports not only water, but sediment—eroding and rebuilding its banks and floodplains continually. Federal Interagency Stream Restoration Working Group 1998. Different deposition patterns occur under varying levels of streamflow, with higher flows having the most influence on the resulting shape of streambanks and floodplains. Id. In relatively flat landscapes drained by low-gradient streams, this natural process deposits the most sediment on the bank immediately next to the stream channel while floodplains farther from the channel are usually lower-lying wetlands ("backswamps" or "backwater wetlands") that receive less sediment. See, e.g., Johnston et al. 1997. The somewhat elevated land thus built up at streamside is called a natural levee, and this entirely natural landform is physically and hydrologically similar to narrow, human-made berms. See, e.g., Leopold et al. 1964. Natural levees are discontinuous, which allows for a hydrologic connection to the stream or river via openings in the levees and thus the periodic mixing of river water and backwater. Johnston et al. 2001. Such discontinuities in natural levees can contribute to a continuous surface connection that is sufficient to satisfy the requirements for jurisdiction of wetlands adjacent to relatively permanent impoundments or tributaries under paragraph (a)(4)(ii) of the final rule. In addition, streams with natural levees, in settings with no human interference whatsoever, retain hydrologic connection with their wetlands behind the levees by periodic flooding during high water and via seepage through and under the levee. Similarly, human-made berms are typically periodically overtopped with water from the near-by stream, and as previously mentioned, are connected via seepage.

Wetlands separated from a stream by a natural or human-made berm serve many of the same functions as those discussed above on other adjacent wetlands. Furthermore, even in cases where a hydrologic connection may not exist, there are other important considerations, such as chemical and biological factors, that result in important connections between the adjacent wetlands and the nearby "waters of the United States," that affect downstream waters.

The movement of surface and subsurface water both over berms and through soils and berms adjacent to rivers and streams is a hydrologic connection between wetlands and flowing watercourses. The intermittent connection of surface waters over the top of, or around, natural and human-made berms further strengthens the evidence of hydrologic connection between wetlands and flowing watercourses. Both natural and human-made barriers can be topped by occasional floods or storm events. *See, e.g.,* Turner *et al.* 2006; Keddy *et al.* 2007. When berms are periodically overtopped by water, wetlands and waters behind the barriers are directly connected to and interacting with the nearby stream and its downstream waters. In addition, surface waters move to and from adjacent soils (including adjacent wetland soils) continually. Along their entire length, streams alternate between effluent (water-gaining) and influent (water-losing) zones as the direction of water exchange with the streambed and banks varies. Federal Interagency Stream Restoration Working Group 1998. The adjacent areas involved in this surface water exchange with a stream or river are known as the hyporheic zone. Hyporheic zone waters are part of total surface waters temporarily moving through soil or sediment. Like within-channel waters, these waters are oxygenated and support living communities of organisms in the hyporheic zone.

Because a hydrologic connection between adjacent wetlands and downstream waters still exists despite the presence of a berm or the like, the chemical and biological connections that rely on a hydrologic connection also exist. For instance, adjacent wetlands behind berms can still serve important water quality functions, serving to filter pollutants and sediment before they reach downstream waters. Wetlands behind berms can function to filter pollutants before they enter the nearby tributary, with the water slowly released to the stream through seepage or other hydrological connections. *See, e.g.*, Osborne and Kovacic 1993; Kovacic 2000. Their ability to retain sediment and floodwaters may be enhanced by

the presence of the berm. For instance, some backwater wetlands in floodplain/riparian areas exhibit higher sedimentation rates than streamside locations. Kuenzler *et al.* 1980; Johnston *et al.* 2001. The presence of manmade levees can actually increase denitrification rates, meaning that the adjacent wetlands can more quickly transform nitrogen. Gergel *et al.* 2005. However, the presence of manmade berms does limit the ability of the river to connect with its adjacent wetlands through overbank flooding and thus limits sediment, water and nutrients transported from the river to the adjacent wetlands. *Id.*; Florsheim and Mount 2003. However, the presence of a berm does not completely eliminate the transport of sediments and water from the river to the nearby adjacent wetland, as suspended sediments and water can overflow both natural and human-made levees, though the transport is usually more pronounced in settings with natural levees. *See, e.g.*, Turner *et al.* 2006; Keddy *et al.* 2007. Sediment deposition over levees is particularly enhanced by extreme events like hurricanes. *Id.*; Reed *et al.* 2006. Wetlands behind berms, where the system is extensive, can help reduce the impacts of storm surges caused by hurricanes. Day *et al.* 2007.

Adjacent wetlands separated from water bodies by berms and the like maintain ecological connection with those water bodies. Though a berm may reduce habitat functional value and may prevent some species from moving back and forth from the wetland to the river, many major species that prefer habitats at the interface of wetland and stream ecosystems remain able to utilize both habitats despite the presence of such a berm. Additional species that are physically isolated in either stream or wetlands habitat still interact ecologically with species from the other component. Thus, adjacent wetlands with or without small berms can retain numerous similarities in ecological function. For example: wetland bird species such as wading birds are able to utilize both wetland and adjacent stream/ditch habitats; wetland amphibians would be able to bypass the berm in their adult stage; aquatic invertebrates and fish would still interact with terrestrial/wetland predators and prey in common food web relationships despite the presence of a berm. *See, e.g.*, Butcher and Zimpel 1991; Willson and Halupka 1995; Cederholm *et al.* 1999; Schwartz and Jenkins 2000; Bilton *et al.* 2001.

Beach dunes are another form of a natural barrier that does not sever adjacency. Beach dunes are formed by tidal or wave action. Multiple beach dunes may exist between a wetland and jurisdictional water (including primary and secondary dunes), because beach dunes typically function as an interdunal system (particularly on barrier islands). Adjacent wetlands behind beach dunes include intradunal and interdunal wetlands located in coastal areas, including some areas of the Great Lakes and along barrier islands. Interdunal wetlands form in swales or depressions within open dunes or between beach ridges along the coast and experience a fluctuating water table seasonally and yearly in synchrony with sea or lake level changes. Odum 1988; Albert 2000; Albert 2003; Albert 2007. For those along the ocean coast, they are typically formed as a result of oceanic processes where the wetlands establish behind relict dune ridges (dunes that were formed along a previously existing coast line). These wetlands occur in depressional areas between sand dunes or beach ridges. The waters, including wetlands, generally form when water levels of the territorial seas fall or the Great Lakes drop, creating swales that support a diverse mix of wetland vegetation and many endangered and threatened species. Odum 1988; Albert 2000; Albert 2003; Tiner 2003c; Albert 2007. Wetlands in the interdunal system are in close proximity to each other and to the surrounding traditional navigable waters or territorial seas. Their proximity to one another and to the traditional navigable waters or territorial seas indicates a close physical relationship between interdunal wetland systems and the traditional navigable waters, the territorial seas, or interstate waters. Despite the presence of the beach dunes, interdunal wetlands have chemical, physical, or biological

connections that greatly influence the integrity of the nearby traditional navigable waters or territorial seas. The wetlands are hydrologically connected to these traditional navigable waters or territorial seas through unconfined, directional flow and shallow subsurface flow during normal precipitation events and extreme events. As previously noted, they are linked to the rise and fall of the surrounding tides or water levels in the case of the Great Lakes—the water-level fluctuations of the nearby traditional navigable waters or territorial seas are important for the dynamics of the wetlands. Albert 2003. The wetlands provide floodwater storage and attenuation, retaining and slowly releasing floodwaters before they reach the nearby traditional navigable waters or territorial seas. Like other adjacent wetlands, interdunal wetlands also have important chemical connections to the nearby traditional navigable waters or territorial seas as they serve important water quality benefits. The wetlands store sediment and pollutants that would otherwise reach the surrounding traditional navigable waters or territorial seas. The wetlands are biologically connected to the surrounding traditional navigable waters or territorial seas. For instance, they provide critical habitats for species that utilize both the wetlands and the nearby traditional navigable waters or territorial seas, supporting high diversity and structure. Habitat uses include basic food, shelter, and reproductive requirements. Aquatic insects, amphibians, and resident and migratory birds all use interdunal wetlands as critical habitat, and the wetlands provide better shelter than the nearby exposed beach. Albert 2000; Smith et al. 2008. In marine coastal areas, the wetlands are often the only freshwater system in the immediate landscape, thus providing critical drinking water for the species that utilize both the wetlands and the nearby traditional navigable waters or territorial seas, although some interdunal wetlands are brackish in nature. See, e.g., Heckscher and Bartlett 2004.

Wetlands behind the extensive levee system in the Yazoo Basin are an example of adjacent wetlands behind human-made barriers. A regional hydrogeomorphic approach guidebook for the Yazoo Basin of the Lower Mississippi River Alluvial Valley assesses the functions of these wetlands. Smith and Klimas 2002. An extensive levee system was built along the river system to prevent flooding of the Mississippi River, resulting in drastic effects to the hydrology of the basin. *Id.* Despite the alteration of hydrology in the basin, extensive wetlands systems still exist behind the human-made and natural levees and maintain a hydrologic connection to the river system. These wetlands detain floodwater, detain precipitation, cycle nutrients, export organic carbon, remove elements and compounds, maintain plant communities, and provide fish and wildlife habitat. *Id.* The functions in turn provide numerous benefits to the nearby river.

c. Reasonably Close Wetlands

Adjacent wetlands under the final rule include wetlands that are within reasonably close proximity to jurisdictional waters such that "adjacent wetlands have significant effects on water quality and the aquatic ecosystem." *Riverside Bayview*, 474 U.S. at 135 n.9. The Supreme Court in *Riverside Bayview* deferred to the Corps' judgment that adjacent wetlands "that form the border of or are in reasonable proximity to" other "waters of the United States" "may be defined as waters under the Act." *Riverside Bayview*, 474 U.S. at 134. The agencies have long considered adjacent wetlands to include those within reasonably close proximity of other jurisdictional waters. *See* 42 FR 37128 (July 19, 1977) (Corps' preamble to the 1977 definition of "navigable waters," noting that "Federal jurisdiction under Section 404 must include any adjacent wetlands that form the border of or are in *reasonable proximity to other waters of the United States*, as these wetlands are part of this aquatic system") (emphasis added). The ecological relationship between jurisdictional waters and their adjacent wetlands is well documented

in the scientific literature and reflects their physical proximity as well as shared hydrological and biological characteristics. *Rapanos* Guidance at 9. The agencies conclude that close proximity between an adjacent wetland and a jurisdictional water means the wetland can modulate water quantity or water quality in the jurisdictional water, and the jurisdictional water can modulate water quantity or quality in the wetland. For example, wetlands typically help to store floodwaters, pollutants, and sediments that could otherwise reach the jurisdictional water to which they are adjacent. They can also provide flow contributions to the jurisdictional waters to which they are adjacent during high hydroperiods, where water spills from the wetland to the nearby jurisdictional water. The proximate jurisdictional waters can serve as important sources of water for adjacent wetlands, for example, through overtopping events where flow from the jurisdictional waters is stored in the wetlands.

While under this rule the agencies are not establishing distance limits for adjacency, the agencies recognize that as the distance between the wetland and jurisdictional water increases, the reasonableness of the connection between the waters will generally decrease, particularly in the absence of the type of surface or shallow subsurface connections described above, and a finding of adjacency is less likely. The distance between a jurisdictional water and its adjacent wetlands may vary by region, as well as based on site-specific factors within regions. The distance between a jurisdictional water and an adjacent wetland can vary from site to site and region to region due to differences in climate, geomorphology, landscape setting, hydrology, soils, vegetation, elevation, size of the jurisdictional water, and other site-specific variables. Because of regional variability and its effects on proximity for purposes of adjacency, wetlands in the arid West—where rainfall is generally lower, evaporation rates are higher, and riparian areas and floodplains often do not extend far from the tributary network—may often be closer to the jurisdictional water than in more humid parts of the country. On the other hand, where the jurisdictional water is wide (e.g., a wide river), topography is flat lending to larger floodplains and riparian areas, and rainfall is higher, wetlands are more likely to be determined to be reasonably close at a distance that is further from the jurisdictional water than in other circumstances because the site-specific conditions contribute to the close relationship between the wetland and the jurisdictional water, including any unbroken surface or shallow subsurface hydrologic connections between the waters.

While bright-line rules (for example, rules that state that wetlands that are more than a specific number of feet from a jurisdictional water are not "adjacent") are generally easiest to understand and implement, convenience is not the only goal the agencies must consider in administering the Clean Water Act. Because the relationship between a jurisdictional water and a proximate wetland can depend upon a number of site-specific factors, like climate, geomorphology, landscapes, hydrology, and size of the jurisdictional water (*e.g.*, the ocean compared to a headwater stream), and because the central purpose of the Act is to protect the integrity of the nation's waters, a more nuanced analysis is required. While science says that all things being equal, distance, location in a riparian area or floodplain, or discrete hydrologic connections are more likely to strengthen the relationship between a wetland and a nearby water, science does not provide bright lines on appropriate distances to determine adjacency. In implementing this provision over the years, the agencies have worked hard to balance the desire for clarity and predictability with the agencies' scientific understanding of the resources Congress has charged the agencies with protecting. The agencies have carefully considered options for nationally applicable bright lines with respect to adjacency, such as establishing that any wetland within a certain number of feet from a jurisdictional tributary is *per se* jurisdictional, in order to facilitate implementation

of the Clean Water Act and to minimize the burden on both landowners and the agencies to evaluate the scope of "waters of the United States." However, the United States is a vast country with many different types of waters, watersheds, landscapes, and hydrology. In fact, in the 2015 Clean Water Rule the agencies sought to establish a distance-based bright line for determining adjacency. As discussed in section IV.B.1 of the preamble, that rule was immediately challenged, and the distance-based limitations were a substantial factor in many of the challenges. As the Supreme Court itself has recognized, the scope of Clean Water Act jurisdiction does not easily lend itself to bright lines: "In sum, we recognize that a more absolute position . . . may be easier to administer. But, as we have said, those positions have consequences that are inconsistent with major congressional objectives, as revealed by the statute's language, structure, and purposes." *Maui*, 140 S. Ct. at 1477. Ultimately, for purposes of the final rule, the agencies concluded that there was not a reasoned basis, consistent with the text of the statute, to establish such a regulatory bright line.

Under the pre-2015 regulatory regime, the agencies have stated that a wetland's reasonably close proximity supports the science-based inference that such wetlands have an ecological interconnection with jurisdictional waters.³⁸ Because of the scientific basis for this inference, determining whether a wetland is reasonably close to a jurisdictional water does not generally require a case-specific demonstration of an ecological interconnection. In the case of a jurisdictional water and a reasonably close wetland, such implied ecological interconnectivity is neither speculative nor insubstantial. For example, species, such as amphibians or anadromous and catadromous fish, move between such waters for spawning and their life stage requirements. In assessing whether a wetland is reasonably close to a jurisdictional water, the proximity of the wetland (including all parts of a single wetland that has been divided by road crossings, ditches, berms, etc.) in question is evaluated. This is consistent with pre-2015 practice. *See Rapanos* Guidance at 6.

Based on a review of the scientific literature and the agencies' expertise and experience, the agencies determined that adjacent wetlands that are within reasonably close proximity to a jurisdictional water are integrally linked to the chemical, physical, or biological functions of waters to which they are adjacent and to the integrity of traditional navigable waters, the territorial seas, or interstate waters, where such wetlands meet the requirements to be jurisdictional under paragraph (a)(4) of the final rule.

Although determining whether a wetland is reasonably close to a jurisdictional water does not generally require a case-specific demonstration of an ecologic interconnection, there may be instances where the presence of such an ecological interconnection between the wetland and the jurisdictional waterbody demonstrates that the wetlands is reasonably close and therefore adjacent. For example, if resident aquatic species (*e.g.*, amphibians, aquatic turtles, fish, or ducks) rely on both the wetland and the jurisdictional waterbody for all or part of their life cycles (*e.g.*, nesting, rearing, or feeding), that may demonstrate that the wetland is within reasonably close proximity and thus adjacent. The agencies recognize that as the distance between the wetland and jurisdictional water increases, the potential ecological interconnection between the waters is likely to decrease. The distance between a jurisdictional

³⁸ See, e.g., *Riverside Bayview Homes*, 474 U.S. at 134 (". . .the Corps' ecological judgment about the relationship between waters and their adjacent wetlands provides an adequate basis for a legal judgment that adjacent wetlands may be defined as waters under the Act.")

water and its adjacent wetlands may vary by region, as well as based on site-specific factors within regions.

Wetlands that are reasonably close to a jurisdictional water perform a myriad of critical chemical, physical, and biological functions associated with a traditional navigable water, the territorial seas, or an interstate water. Such wetlands are often located within the riparian area or floodplain and are often connected via surface and shallow subsurface hydrology to the water to which they are adjacent. While the SAB in its review of the Science Report was clear that distance is not the only factor that influences connections and their effects downstream, due to their close proximity to jurisdictional waters, reasonably close wetlands are often located within a landscape position that allows for them to receive and process surface and shallow subsurface flows before they reach nearby streams and rivers. These waters can individually and collectively affect the integrity of traditional navigable waters, the territorial seas, and interstate waters by acting primarily as sinks that retain floodwaters, sediments, nutrients, and contaminants that could otherwise negatively impact the condition or function of such paragraph (a)(1)waters. Wetlands within close proximity of jurisdictional waters improve water quality through assimilation, transformation, or sequestration of nutrients, sediment, and other pollutants that can affect the integrity of traditional navigable waters, the territorial seas, or interstate waters. These waters, including wetlands, also provide important habitat for aquatic-associated species to forage, breed, and rest. Such species can travel between the wetlands and the nearby jurisdictional water, as well as to the traditional navigable waters, the territorial seas, and interstate waters.

As noted above, reasonably close wetlands may be within the riparian area even if the floodplain is limited. Riparian wetlands within floodplains and riparian wetlands that occur in systems that do not have floodplains are an important part of the overall riverine landscape. Science Report at 4-7 (citing Ward 1998). Wetlands within riparian areas are also connected to streams and rivers by a diverse set of hydrologic inputs and outputs. *Id.* (citing Junk *et al.* 1989; Winter and Rosenberry 1998; Benke *et al.* 2000; Tockner *et al.* 2000; Bunn *et al.* 2006). Waters in stream and river channels can readily reach wetlands in riparian areas via overbank flow, which occurs when floodwaters flow over stream and river channels. *Id.* at 2-12 (citing Mertes 1997).

Riparian areas can have a diverse array of hydrologic inputs and outputs, which, in turn influence riparian wetlands. *Id.* at 2-14. Riparian areas receive water from precipitation; overland flow from upland areas; local, intermediate, regional ground water; and hyporheic flows. *Id.* (citing National Research Council 2002; Richardson *et al.* 2005; Vidon *et al.* 2010). Water flowing over the land surface in many situations can infiltrate soils in riparian areas. If low permeability subsoils or impervious clay layers are present, water contact with the plant root zone is increased and the water is subject to ecological functions such as denitrification before it reaches the stream channel. *Id.* (citing National Research Council 2002; Naiman *et al.* 2005; Vidon *et al.* 2010). Riparian wetlands can have bidirectional, lateral hydrologic connections to the river network, either through overbank flooding (*i.e.*, lateral expansion of the network) or hyporheic flow, in addition to unidirectional flows from upland and ground-water sources. *Id.* at 2-20.

Floodplain wetlands that are reasonably close to jurisdictional waters perform important functions that improve downstream water quality, including the temporary storage and deposition of channeling-forming sediment and woody debris, temporary storage of local ground water that supports baseflow in rivers, and transformation and transport of stored organic matter. *Id.* at ES-2 to ES-3. Floodplain wetlands

improve water quality through the assimilation, transformation, or sequestration of pollutants, including excess nutrients and chemical contaminants such as pesticides and metals, that can degrade downstream water integrity. *Id.* at ES-3. In addition to providing effective buffers to protect downstream waters from point source and nonpoint source pollution, these systems form integral components of river food webs, providing nursery habitat for breeding fish and amphibians, colonization opportunities for stream invertebrates, and maturation habitat for stream insects. *Id.* Lateral expansion and contraction of the river in its floodplain result in an exchange of organic matter and organisms, including fish populations that are adapted to use floodplain habitats for feeding and spawning during high water, that are critical to river ecosystem function. *Id.* Floodplain wetlands also affect the integrity of downstream waters by subsequently releasing (desynchronizing) floodwaters and retaining large volumes of stormwater, sediment, and contaminants in runoff that could otherwise negatively affect the condition or function of downstream waters. *Id.*

As discussed in section I, riparian and floodplain wetlands lie within landscape settings that have bidirectional hydrological exchange with the jurisdictional waters. This can occur through overbank flooding (*i.e.*, lateral expansion of the network) or hyporheic flow or from unidirectional overflow from the wetland to the channel. *Id.* at 2-20. Although elevation is the primary factor determining areas that are inundated through overbank flooding, connectivity with the river generally will be higher for riparian/floodplain wetlands located near the river's edge compared with riparian/floodplain wetlands occurring near the floodplain edge. *Id.* at 2-39.

Science demonstrates that distance is a factor in the connectivity and the strength of connectivity of wetlands to downstream waters. Science Report at ES-4, ES-11, 4-2, 5-6-5. The Science Report states, "[s]patial proximity is one important determinant of the magnitude, frequency and duration of connections between wetlands and streams that will ultimately influence the fluxes of water, materials and biota between wetlands and downstream waters." Science Report at ES-11. Thus, waters that are more distant generally have less opportunity to be connected to and to effect downstream waters. Wetlands closer to the stream network or coastline generally will have greater hydrologic and biological connectivity than waters located farther from the same network. *See, e.g., id.* at 2-38. For instance, wetlands that are more closely proximate have a greater opportunity to contribute flow, as water is likely to be lost from the channel through evaporation or transpiration. *Id.* Via their hydrologic connectivity, proximate waters and are more likely to impact water quality due to their close distance. Waters more closely located to other water resources are also more likely to be biologically connected to such waters more frequently and by more species, including amphibians and other aquatic animals.

All factors being equal, wetlands closer to the tributary network will have greater hydrologic and biological connectivity than wetlands located farther from the same network. *Id.* at 2-40. Distance is a factor that is well known to have various effects on physical and biological processes within and between system components. *Id.* at 4-44. Sometimes this is due to the direct effect of distance. In some cases, there is an indirect effect due to distance controlling how long transport of a material will take. This fact is embedded in the time of concentration concept in hydrology, whereby under similar slope and velocities, water traveling from more distant points and with a longer flowpath will—because of the length of time in transit—have greater potential for evapotranspiration and soil infiltration losses before reaching a stream. *Id.* at 2-39; Blanco-Canqui and Lal 2008. There are many examples in the scientific literature of

distance effects from various factors with respect to chemical, physical, and biological processes. Graf 1984; Marron 1989; Leigh 1997; King *et al.* 2005; Alexander *et al.* 2007; Attum *et al.* 2007; Subalusky 2007; Van Sickle and Johnson 2008; Colvin *et al.* 2009; Flitcroft *et al.* 2012; Greathouse *et al.* 2014.

With respect to provision of water quality benefits downstream, non-floodplain wetlands within close proximity of the stream network often are able to have more water quality benefits than those located at a distance from the stream. Many studies indicate that the primary water quality and habitat benefits will generally occur within a several hundred foot zone of a water. *See, e.g.*, Peterjohn and Correll 1984; Hawes and Smith 2005. In addition, the scientific literature indicates that to be effective, contaminant removal needs to occur at a proximate distance prior to entry into stream network and subsequently into the downstream traditional navigable waters, the territorial seas, or interstate waters. Some studies also indicate that fish, amphibians (*e.g.*, frogs, toads), reptiles (*e.g.*, turtles), and small mammals (*e.g.*, otters, beavers, etc.) will use a reasonably close zone for foraging, breeding, nesting, and other life cycle needs. Dole 1965; Smith and Green 2005; Semlitsch 2008; Steen *et al.* 2012.

Based on a review of the scientific literature and the agencies' expertise and experience, there is clear evidence that the reasonably close wetlands adjacent to traditional navigable waters, the territorial seas, or interstate waters or reasonably close wetlands adjacent to jurisdictional impoundments or tributaries and that meet either the relatively permanent or significant standard under paragraph (a)(4) of the final rule perform critical processes and functions discussed in the sections below.

iii. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Physical Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

Scientific research shows that adjacent wetlands are important in protecting the physical integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. This includes adjacent wetlands in riparian areas and floodplains, which exhibit bidirectional exchange of water with the waters to which they are adjacent and that play an important role in determining the volume and duration of stream flow. Such adjacent wetlands also have an essential role in regulating and stabilizing sediment transport to downstream traditional navigable waters, the territorial seas, and interstate waters. These characteristics are fundamental to the physical integrity of streams as well as downstream traditional navigable waters, the territorial seas, and interstate waters.

Adjacent wetlands located in riparian areas and floodplains are important for the reduction or delay of floods. *Id*.at 2-21, 4-7 (citing Mertes *et al.* 1995; Walton *et al.* 1996; Bullock and Acreman 2003; Poole *et al.* 2006; Rassam *et al.* 2006). Such adjacent wetlands control flooding during times of high precipitation or snowmelt by capturing water from overbank flow and storing excess stream water. *Id.* at 4-7. One study found that peak flows in the Cache River in Arkansas decreased by 10-20% mainly because of floodplain water storage. *Id.* (citing Walton *et al.* 1996). Research has shown that floodplain wetlands in Ohio store about 40% of the flow of small streams. *Id.* (citing Gamble *et al.* 2007). These and similar findings point to the close hydrological influence that adjacent wetlands in riparian and floodplain areas have on streams and downstream traditional navigable waters, the territorial seas, and interstate waters.

Adjacent wetlands are bordering, contiguous, or neighboring jurisdictional waters. Because of their close physical proximity to nearby jurisdictional waters, they can readily exchange their waters through the saturated soils surrounding the stream or through surface exchange. This commingling of waters allows bordering, contiguous, or neighboring wetlands to both provide chemically transformed waters to streams and to absorb excess stream flow.

Flow between neighboring wetlands and streams tends to be more longitudinal (downslope) at headwaters and more lateral further downstream. Id. at 4-40, Table 4-3. These connections in part determine stream flow volume and duration. Adjacent wetlands in riparian areas connect to nearby water bodies through various surface and subsurface connections. See, e.g., id. at 2-4 (citing National Research Council 2002). Floodplains, similarly, are closely associated with the groundwater found beneath and beside river channels (which are considered shallow aquifers), and adjacent wetlands in floodplains readily exchange water with such aquifers. Id. at 2-12 (citing Stanford and Ward 1993; Amoros and Bornette 2002; Poole et al. 2006). Riparian and floodplain wetlands are frequently contiguous with streams and other water bodies and can significantly influence the physical form, hydrology, chemistry, and biology of such water bodies. Id. at 4-6 (citing Junk et al. 1989; Abbott et al. 2000; Tockner et al. 2000; Woessner 2000; Amoros and Bornette 2002; Ward et al. 2002; King et al. 2003; Naiman et al. 2005; Church 2006; Kondolf et al. 2006; Poole et al. 2006; Poole 2010; Tockner et al. 2010; Vidon et al. 2010; Helton et al. 2011; McLaughlin et al. 2011; Humphries et al. 2014). Adjacent wetlands located in floodplains are important for the reduction or delay of floods by capturing water from overbank flow and by storing excess water from the streams to which they are adjacent. Id. at 4-7 (citing Bullock and Acreman 2003).

Adjacent wetlands in riparian areas and floodplains filter sediment washed down from uplands and collect sediment from overbank flow as the river or stream floods. *Id.* at 4-8 (citing Boto and Patrick 1979; Whigham *et al.* 1988). For example, riparian areas were observed to collect 80-90% of the sediment from farmlands in a study in North Carolina. *Id.* (citing Cooper *et al.* 1987; Daniels and Gilliam 1996; Naiman and Decamps 1997). Maintaining the equilibrium between sediment deposition and sediment transport is important to maintain the physical shape and structure of stream channels, including of downstream traditional navigable waters, the territorial seas, and interstate waters. Significant changes to upstream channels can affect the chemical, physical, and biological condition of downstream traditional navigable waters, the territorial seas, and interstate waters.

The physical effects of excess sediment can impair chemical and ecological integrity in a variety of ways. *Id.* at 5-9 (citing Wood and Armitage 1997). Excess sediment is linked to increasing contaminant and nutrient concentrations, all of which tributaries can transmit downstream, affecting the water quality of traditional navigable waters, the territorial seas, and interstate waters. Excess sediment may block and absorb sunlight transmission through the water column, inhibiting plant photosynthesis and warming the water in the stream. Sediment may fill the interstitial spaces between rocks in a streambed, which many fish and aquatic species use for mating, reproduction, and shelter from predators. This kind of physical degradation of tributary streambeds results in less suitable habitat available for animals and fish that move between upstream and downstream waters. Adjacent wetlands located in riparian areas retain sediments and thus protect downstream traditional navigable waters, the territorial seas, and interstate waters from the effects of excess sediment.

Because adjacent wetlands support riparian vegetation, they affect the capacity of riparian vegetation to influence stream flow, morphology, temperature, and habitat provided in the nearby water body. Vegetation in adjacent wetlands located in riparian areas influences the amount of water in the stream by capturing and transpiring stream flow and intercepting groundwater and overland flow. Id. at 2-21, 4-8 (citing Meyboom 1964). Riparian vegetation in adjacent wetlands also reduces stream bank erosion, serving to maintain the physical integrity of the channel. See, e.g., id. at 4-8 to 4-9 (citing Beeson and Doyle 1995; Naiman and Decamps 1997; Burt et al. 2002; Zaimes et al. 2004). In addition, inputs of woody debris from aquatic vegetation or logs into waters make important contributions to the channel's geomorphology and the stream's aquatic habitat value. Id. at 4-9 (citing Anderson and Sedell 1979; Harmon et al. 1986; Nakamura and Swanson 1993; Abbe and Montgomery 1996; Naiman and Decamps 1997; Gurnell et al. 2002; Brummer et al. 2006; Sear et al. 2010; Collins et al. 2012). Also, the riparian vegetation in adjacent wetlands that overhangs streams provides shade, providing a critically important function of reducing fluctuations in water temperature helping to reduce excessive algal production and to maintain life-supporting oxygen levels in streams and other types of waters. Id. at 4-9 to 4-10 (citing Gregory et al. 1991; Volkmar and Dahlgren 2006). Even small changes in water temperature can have significant impacts on the type and number of species present in waters, with higher temperatures generally associated with degraded habitat which supports only those species that can tolerate higher temperatures and reduced levels of dissolved oxygen. Higher water temperatures are associated with streams and rivers with less valuable recreational and commercial fisheries. As discussed below, these physical characteristics of headwater streams influence what types of organisms live in the region.

Headwaters and nearby adjacent wetlands supply downstream traditional navigable waters, the territorial seas, and interstate waters with dissolved organic carbon as a result of decomposition processes from dead organic matter such as plants. Both production and consumption of organic and inorganic carbon occur in adjacent wetlands. Id. at 4-13. Adjacent wetlands are an important source of dissolved organic carbon (DOC) to downstream waters. Allochthonous inputs from adjacent wetlands to streams are important to aquatic food webs, particular in headwaters. Id. (citing Tank et al. 2010). Allochthonous inputs are terrestrial organic materials that enter the stream through vegetation litter (*i.e.*, woody debris, leaves, and partially decomposed plant parts), erosion, and hydrologic flows. Id. (citing Wetzel 1992). These inputs of organic matter are the primary source of energy flow into the food webs of streams. Id. Organic matter inputs are important because they affect food availability to aquatic organisms by releasing organic carbon and nitrogen into streams. Id. (citing Wetzel and Manny 1972; Mulholland and Hill 1997). This organic carbon contributes to the downstream foodweb and ultimately supports downstream fisheries, including in traditional navigable waters, the territorial seas, and interstate waters. See, e.g., id. at 4-16. Export of DOC to downstream waters supports primary productivity, effects pH and buffering capacity, and can protect aquatic organisms from the harmful effects of UV-B radiation. Id. at 4-28 (citing Eshelman and Hemond 1985; Hobbie and Wetzel 1992; Hedin et al. 1995; Schindler and Curtis 1997; Nuff and Asner 2001; Reddy and DeLaune 2008). However, too much organic matter downstream can have negative effects because contaminants, such as methylmercury and other trace metals, can be adsorbed to it. Id. (citing Thurman 1985; Driscoll et al. 1995).

iv. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Chemical Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

As stated above in section III.A on tributaries, pollutants such as petroleum waste products and other harmful pollutants dumped into any part of the tributary system are likely to flow downstream, or to be washed downstream, and thereby pollute traditional navigable waters, the territorial seas, and interstate waters from which American citizens take their drinking water, shellfish, fin fish, water-based recreation, and many other uses. Some wetlands perform the valuable function of trapping or filtering out pollutants (such as fertilizers, silt, and some pesticides), thereby reducing the likelihood that those pollutants will reach and pollute the tributaries of traditional navigable waters, the territorial seas, and interstate waters (and eventually pollute those larger waters themselves). However, many other pollutants (such as petroleum wastes and toxic chemical wastes), if dumped into wetlands that are adjacent to tributary streams or to jurisdictional impoundments, may reach those tributaries or impoundments themselves, and thereafter flow downstream to pollute traditional navigable waters, the territorial seas, and interstate waters waters, posing risks to the nation's drinking water supply, fisheries, and recreation areas.

Adjacent wetlands in riparian areas and floodplains play a critical role in controlling the chemicals that enter streams and other "waters of the United States" and as a result are vital in protecting the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters. Runoff (the water that has not evaporated or infiltrated into the groundwater) from uplands is a large source of pollution, but research has shown that adjacent wetlands in riparian areas trap and chemically transform a substantial amount of the nutrients, pesticides, and other pollutants before they enter streams, river, lakes, and other downstream waters.

Chemicals and other pollutants enter waters from point sources such as outfalls and pipes, nonpoint sources (*e.g.*, runoff from agricultural and urban fields and lawns), dry and wet (*e.g.*, rain, snow) atmospheric deposition, upstream reaches, and through the hyporheic zone, a region beneath and alongside a stream bed where surface water and shallow groundwater mix. *Id.* at 4-10 (citing Nixon and Lee 1986; Tiner 2003c; Whigham and Jordan 2003; Comer *et al.* 2005; Whitmire and Hamilton 2008). Throughout the stream network, but especially in headwater streams and their adjacent wetlands, chemicals are sequestered via sorption (adsorption and absorption) or sedimentation processes, assimilated into the flora and fauna, transformed into other compounds, or lost to the atmosphere through transformational processes performed by microbes, fungi, algae, and macrophytes present in riparian waters and soils. *Id.* (citing Nixon and Lee 1986; Johnston 1991; Boon 2006; Mitsch and Gosselink 2007; Reddy and DeLaune 2008). These chemical processes reduce or eliminate pollution that would otherwise enter streams, rivers, lakes, and other types of waters and subsequently downstream traditional navigable waters, the territorial seas, or interstate waters.

The removal of the nutrients nitrogen and phosphorus is a particularly important role for adjacent wetlands. As described previously, nutrients are necessary to support aquatic life, but the presence of excess nutrients can lead to eutrophication and the depletion of oxygen (hypoxia) in nearby waters and in waters far downstream. *See, e.g., id.* at ES-8. Eutrophication is a large problem in waters across the United States including such important ecosystems as the Chesapeake Bay and Lake Spokane in Washington. Kemp *et al* 2005; Moore and Ross 2010; Murphy *et al.* 2011. Eutrophication is the natural

or artificial enrichment of a water body by nutrients, typically phosphates and nitrates. Science Report at A-4. It can occur when plants and algae grow in waters to such an extent that the abundance of vegetation monopolizes the available oxygen, detrimentally affecting other aquatic organisms. Protection of these waters therefore helps maintain the chemical integrity of the nation's waters, including downstream traditional navigable waters, the territorial seas, and interstate waters.

The removal of nitrogen is an important function of adjacent wetlands. Adjacent wetlands located in riparian areas regularly remove more than half of dissolved nitrogen found in surface and subsurface water by plant uptake and microbial transformation. *Id.* at 4-11 (citing Vidon *et al.* 2010). Denitrification potential in surface and subsurface flows is highest where there is high organic matter and/or anoxic conditions. *Id.* at 4-12 (citing McClain *et al.* 2003; Orr *et al.* 2014). The highest denitrification potentials occur in adjacent wetlands located in floodplain and riparian areas where high organic matter, denitrifying microbes, and saturated soil conditions are present, and rates increase with proximity to streams. *Id.* (citing Gregory *et al.* 1991; Vidon *et al.* 2010). Adjacent wetlands are therefore important in maintaining the conditions important for denitrification, which in turn protects streams, rivers, lakes, and downstream traditional navigable waters, the territorial seas, and interstate waters from nitrogen pollution.

Plant uptake of dissolved nitrogen in subsurface flows through wetlands located in riparian areas also accounts for large quantities of nitrogen removal. *Id.* (citing Vidon *et al.* 2010). Riparian forests have been found to remove 75% of dissolved nitrate transported from agricultural fields to a Maryland river. *Id.* (citing Vidon *et al.* 2010). Likewise, riparian forests in Georgia remove 65% of nitrogen and 30% of phosphorus from agricultural sources. *Id.* (citing Vidon *et al.* 2010). A Pennsylvania forested riparian area removed 26% of the total nitrate input from the subsurface. *Id.* (citing Newbold *et al.* 2010). The vegetation associated with adjacent wetlands in riparian waters also removes nitrogen from subsurface flows. Therefore, the conservation of adjacent wetlands helps protect downstream traditional navigable waters, the territorial seas, and interstate waters from influxes of dissolved nitrogen.

Phosphorus is another potentially harmful nutrient that is captured and processed in adjacent wetlands, including those located in riparian areas. Id. at 4-12 to 4-13 (citing Dillaha and Inamdar 1997; Sharpley and Rekolainen 1997; Carlyle and Hill 2001). Biogeochemical processes, sedimentation, and plant uptake account for high rates of removal of particulate phosphorus in riparian areas. Id. at 4-12 (citing Hoffmann et al. 2009). The amount of contact the water has with nearby soils and the characteristics of that soil determine the ability of the riparian area to remove phosphorus. Id. Adjacent wetlands in riparian areas are phosphorus sinks in oxic soils (containing oxygen), while riparian soils generally can serve as sources of phosphorus when soils are anoxic (lacking oxygen) or when mineral dissolution releases phosphorus. Id. at 4-12 (citing Baldwin and Mitchell 2000; Carlyle and Hill 2001; Chacon et al. 2008). Adjacent wetlands in riparian areas where agricultural sediments are deposited are phosphorus sources to streams if the phosphorus is desorbed and leached but can be sinks by adsorbing dissolved phosphorus if sediment phosphorus concentrations are low. Id. at 4-12 to 4-13 (citing Dillaha and Inamdar 1997; Sharpley and Rekolainen 1997). Adjacent wetlands in riparian areas also serve as phosphorus sinks when upland surface runoff travels through the riparian area or when fine-grained sediment containing phosphorus is deposited overbank onto the riparian area. Id at 4-13 (citing Dillaha and Inamdar 1997). These sediments, however, can become sources of phosphorus if they are later saturated with water and iron and manganese are reductively dissolved during anoxic conditions, thus causing them to desorb phosphorus. Id. (citing Reddy and DeLaune 2008). The function of adjacent

wetlands to move and uptake phosphorus is crucial for maintaining the chemical and biological integrity of the waters to which they are adjacent, and for preventing eutrophication in downstream traditional navigable waters, the territorial seas, and interstate waters, thereby helping to restore and maintain the integrity of those larger fundamental waters. In the case where adjacent wetlands are acting as a source of phosphorus for traditional navigable waters, the territorial seas, and interstate waters, this also can significantly affect the chemical and biological integrity of these downstream waters.

v. Adjacent Wetlands Can Provide Functions that Restore and Maintain the Biological Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

Adjacent wetlands support the biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters in a variety of ways. They provide habitat for aquatic and water-tolerant plants, invertebrates (aquatic insects), and vertebrates, and provide feeding, refuge, and breeding areas for invertebrates and fish. Seeds, plants, and animals move between adjacent wetlands and the nearby streams, and from there colonize or utilize downstream waters, including traditional navigable waters, the territorial seas, and interstate waters.

Organic matter from adjacent wetlands is critical to aquatic food webs, particularly in headwaters, where it is the primary source of energy flow due to low light conditions that inhibit photosynthesis. See, e.g., id. at 4-13 (citing Tank et al. 2010). Headwater streams tend to be located in heavily vegetated areas compared to larger waters, so they are more likely to contain leaf litter, dead and decaying plants, and other organic matter that forms the basis of headwater food webs. The organic matter is processed by microbes and insects that make the energy available to higher levels of stream life such as amphibians and fish. Studies have shown that aquatic insects rely on leaf inputs in headwater streams and that excluding organic litter from a stream resulted in significant changes to the food web at multiple levels. Id. (citing Minshall 1967; Wallace and Webster 1996; Wallace et al. 1997; Meyer et al. 1998). Fish and amphibian species found in headwaters travel downstream and in turn become part of the food web for larger aquatic organisms in traditional navigable waters, the territorial seas, and interstate waters. Organic material provided by adjacent wetlands to small, headwater streams is therefore important not only to the small streams that directly utilize this source of energy to support their biological populations but also to the overall biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters that benefit from the movement of fish and other species that contribute to the food web of larger streams, rivers, estuaries, and the ocean.

Adjacent wetlands accumulate organic carbon and nitrogen, an important function influenced by the size and frequency of floods from rivers to which they are adjacent. *See, e.g., id.* at B-11 (citing Cabezas *et al.* 2009). These stored chemicals are available for exchange with river water when hydrological connections are present. Organic materials are the basis for the food web in stream reaches where photosynthetic production of energy is absent or limited, particularly in headwater systems where vegetative litter alone makes up the base of the aquatic food web. The maintenance of floodplain wetlands is therefore an important component of protecting the biological integrity of traditional navigable waters, the territorial seas, and interstate waters into which the headwaters flow.

Adjacent wetlands play an important role in the removal of pesticides. *Id.* at 4-14 (citing Vidon *et al.* 2010). Microbes near plant roots break down these pesticides. *See, e.g., id.* (citing Voos and Groffman 1996). Uptake by aquatic plants has also been shown to be an important mechanism of removal of the pesticides alachlor and atrazine. *Id.* (citing Paterson and Schnoor 1992). Adjacent wetlands also trap and hold pesticide contaminated runoff preventing it from harming neighboring waters.

Adjacent wetlands located in riparian areas and floodplains are dynamic places that support a diversity of aquatic, amphibious, and terrestrial species adapted to the unique habitat created by periodic or episodic flooding or inundation events. *Id.* at 4-15 (citing Power *et al.* 1995a; Power *et al.* 1995b; Galat *et al.* 1998; Robinson *et al.* 2002; Toth and van der Valk 2012; Rooney *et al.* 2013; Granado and Henry 2014). Plants, aquatic insects, and vertebrates use adjacent wetlands for habitat, nutrients, and breeding. As a result, adjacent wetlands act as sources of organisms, particularly during inundation events, replenishing the waters to which they are adjacent with organisms, seeds, and organic matter. Inundation and hydrological connectivity of adjacent wetlands to the tributary network greatly increase the area of aquatic habitats and species diversity. *Id.* at 4-15, 4-16 (citing Junk *et al.* 1989; Tockner *et al.* 2000; Jansson *et al.* 2005; Brooks and Serfass 2013). Aquatic animals, including amphibians and fish, take advantage of the adjacent wetlands, either inhabiting them or moving between the wetlands and the waters to which the wetlands are adjacent. *Id.* at 4-15, 4-17 through 4-19 (citing Copp 1989; Smock *et al.* 1992; Smock 1994; Robinson *et al.* 2002; Richardson *et al.* 2005; Ilg *et al.* 2008; Shoup and Wahl 2009). Likewise, seeds, plant fragments, and whole plants move between adjacent wetlands and the river network. *Id.* at 4-15 (citing Schneider and Sharitz 1988; Middleton 2000; Nilsson *et al.* 2010).

Hydrological connections are often drivers of biological connections, and flooding events enhance the existing connections between adjacent wetlands and the river network. As a result, adjacent wetlands have important functions for aquatic health. Many species have cycles timed to flooding events, particularly in circumstances where flooding is associated with annual spring snowmelt or high precipitation. Id. at 4-15 to 4-16, 4-19 (citing Thomas et al. 2006; Tronstad et al. 2007; Gurnell et al. 2008). Adjacent wetlands act as sinks of seeds, plant fragments, and invertebrate eggs and as sources of such biological material during times of periodic flooding, allowing for cross-breeding and resulting gene flow across time. Id. at 4-16, 4-19 to 4-20 (citing Middleton 2000; Jenkins and Boulton 2003; Frisch and Threlkeld 2005; Gurnell et al. 2008; Vanschoenwinkel et al. 2009). Stream macroinvertebrates (e.g., insects, crayfish, and mollusks) and microinvertebrates (e.g., zooplankton such as cladocerans, copepods, rotifers, and gastropods) colonize nutrient rich waters within riparian areas and floodplains in large numbers during periods of seasonal or episodic inundation, facilitating an increase in population and sustaining them though times of limited resources and population decline. Id. at 4-19 to 4-20 (citing Fisher and Willis 2000; Frisch and Threlkeld 2005; Junk et al. 1989; Malmqvist 2002; Ilg et al. 2008). Such animals are adapted to high floods, desiccation (drying out), or other stresses that come with these regular, systemic fluctuations. Id. at 4-19. Adjacent wetlands, including those in riparian areas and floodplains therefore maintain various biological populations, which periodically replenish jurisdictional waters to which they adjacent and to traditional navigable waters, the territorial seas, and interstate waters, serving to maintain their biological integrity.

Plants and animals use adjacent wetlands for habitat, food, and breeding. Adjacent wetlands also provide food sources for stream invertebrates, which colonize during inundation events. *Id.* at 4-19 (citing Junk *et al.* 1989; Ilg *et al.* 2008). Adjacent wetlands also form an integral part of the river food web,

linking primary producers and plants to higher animals. *Id.* (citing Malmqvist 2002; Woodward and Hildrew 2002; Stead *et al.* 2005; Woodford and McIntosh 2010). Likewise, adjacent wetlands located in floodplains are important foraging, hunting, and breeding sites for fish, amphibians, and aquatic macroinvertebrates. *Id.* at 4-15 (citing Copp 1989; Smock *et al.* 1992; Smock 1994; Bestgen *et al.* 2000; Richardson *et al.* 2005; Schramm and Eggleton 2006; Sullivan and Watzin 2009; Alford and Walker 2013; Magana 2013).

Plants and animals move back and forth between adjacent wetlands and the river network. This movement is assisted in some cases when flooding events create hydrological connections. For instance, these floodplain and riparian wetlands provide refuge, feeding, and rearing habitat for many fish species. *Id.* at 4-17 (citing Wharton *et al.* 1982; Boltz and Stauffer 1989; Matheney and Rabeni 1995; Pease *et al.* 2006; Henning *et al.* 2007; Jeffres *et al.* 2008). Seeds of aquatic and riparian plants ingested by animals such as carp are dispersed in stream channels and associated waters. *See, e.g., id.* at 4-16 (citing King *et al.* 2003; Pollux *et al.* 2007). Also, phytoplankton move between adjacent wetlands located in the floodplain and the river network. *Id.* at 4-16 (citing Angeler *et al.* 2010). In turn, the primary productivity conditions in such adjacent wetlands results in large populations of phytoplankton that enrich river networks when hydrological connections form. *Id.* at 4-16 to 4-17 (citing Lehman *et al.* 2008). This influx of carbon into the river system nourishes the aquatic food webs of downstream foundation waters, for example, by supporting fisheries.

However, even when hydrological connections are absent, some aquatic organisms can move between adjacent wetlands and their nearby tributaries by overland movement in order to complete their life cycle. River-dwelling mammals, such as river otters, move from the river to riparian/floodplain wetlands. Id. at 4-17 (citing Newman and Griffin 1994). In addition, both river otters and beavers have a strong preference for riparian areas that are pond- and lake-dominated (Swimley et al. 1999). Several species of amphibians and reptiles including frogs, snakes and turtles use both streams and neighboring wetlands. Id. at ES-10, 3-47 (Table 3-1), 4-15 (citing Richardson et al. 2005). Movement between adjacent wetlands and the river network also occurs by the dispersal of seed and plant fragments and the wind dispersal of invertebrates. Id. at 4-15 to 4-16, 4-20 (citing Schneider and Sharitz 1988; Middleton 2000; Gurnell 2007; Gurnell et al. 2008; Nilsson et al. 2010; Tronstad et al. 2007; Vanschoenwinkel et al. 2009). Animals, particularly migratory fish, can thus move between adjacent wetlands and traditional navigable waters, the territorial seas, and interstate waters. And even when some species do not traverse the entire distance from adjacent wetlands to traditional navigable waters, the territorial seas, and interstate waters, the larger downstream waters still benefit from the ecological integrity that persists because of the close relationship that adjacent wetlands have with the waters to which they are adjacent. This is because the chemical and biological properties that arise from interactions between adjacent wetlands and tributaries or impoundments move downstream and support the integrity of traditional navigable waters, the territorial seas, and interstate waters.

C. Impoundments

The final rule retains the provision in the 1986 regulations that defines "waters of the United States" to include impoundments of "waters of the United States." Under paragraph (a)(2) of the final rule, impoundments of traditional navigable waters, the territorial seas, interstate waters, jurisdictional

tributaries, and jurisdictional adjacent wetlands are "waters of the United States." As discussed in section IV.C.3 of the preamble of the final rule, the caselaw supports that damming or impounding a jurisdictional water does not make a water non-jurisdictional. Impoundments can be natural (like beaver ponds) or artificial (like reservoirs). Impoundments under this rule can be located off-channel or in-line with the channel. They vary in size and volume, from small ponds on headwater streams to large reservoirs, based in part on the water that is being impounded and type of structure that creates the impoundment.

Asserting Clean Water Act jurisdiction over impoundments also aligns with the scientific literature, as well as the agencies' scientific and technical expertise and experience, which confirm that impoundments have chemical, physical, and biological effects on downstream waters through surface or subsurface hydrologic connections. Impoundments do not sever the effects the impounded "waters of the United States" have on the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, and interstate waters, though the impounding of the jurisdictional water may change the nature of the effects that the impoundment has on downstream waters. For example, dams often remove much of the sediment from transport, whereas most streams naturally are sediment sources. Science Report at 3-16. Impoundments trap water, sediment, and particulate nutrients and result in downstream impacts on channel morphology and aquatic function in traditional navigable waters, the territorial seas, and interstate waters. *See, e.g., id.* at 5-8.

Berms, dikes, and similar features used to create impoundments typically do not block all water flow. Even dams, which are specifically designed and constructed to impound large amounts of water effectively and safely, generally do not prevent all water flow, but rather allow seepage under the foundation of the dam and through the dam itself. *See, e.g.*, International Atomic Energy Agency 2003 ("All dams are designed to lose some water through seepage"); U.S. Bureau of Reclamation ("All dams seep, but the key is to control the seepage through properly designed and constructed filters and drains"); Federal Energy Regulatory Commission 2005 ("Seepage through a dam or through the foundations or abutments of dams is a normal condition"). As discussed in section III.B.ii.2, seepage occurs not only for earthen dams but for concrete structures as well. *See, e.g.*, Texas Commission on Environmental Quality 2006.

Further, as an agency with expertise and responsibilities in engineering and public works, the Corps extensively studies water retention structures like berms, levees, and earth and rock-fill dams. The agency has found that all water retention structures are subject to seepage through their foundations and abutments. *See, e.g.*, U.S. Army Corps of Engineers 1992; U.S. Army Corps of Engineers 1993; U.S. Army Corps of Engineers 2004. The Supreme Court has recognized that a canal and an impoundment area separated by levees were hydrologically connected (and might even be considered a single water body) because, inter alia, the "levees continually leak." *South Florida Water Mgmt. District v. Miccosukee Tribe of Indians*, 541 U.S. 95, 110 (2004).

The inevitability of seepage is a consequence not of poor design, but of physics: water will flow downward where it can and thus will seep through small spaces in the structure and in the ground beneath it. *See, e.g.*, U.S. Army Corps of Engineers 1993; U.S. Army Corps of Engineers 2000. Thus, good engineering practices do not entail the prevention of all seepage; rather, they assume seepage and entail steps to manage it so that it will not compromise the integrity of berms, levees, and dams. *See, e.g.*, U.S.

Army Corps of Engineers 1993; U.S. Army Corps of Engineers 2004; U.S. Army Corps of Engineers 2000; U.S. Army Corps of Engineers 1992; U.S. Army Corps of Engineers 2005a; Federal Energy Regulatory Commission 2005.

Many tributary systems in the United States have impoundments located along their reach. There are more than 91,000 dams in the United States as documented in the National Inventory of Dams, with over 6,800 exceeding 50 feet (approximately 15 meters) in height. U.S. Army Corps of Engineers 2020a. Although, the National Inventory of Dams is comprehensive, it only includes approximately 1-2% of the impoundments in the country and likely excludes millions of small impoundments. Renwick et al. 2005. Nearly all river networks in prairie regions have been altered by impoundments for irrigation storage and flood control, from small farm ponds in headwaters to large reservoirs on river mainstems. Science Report at B-28 (Smith et al. 2002; Galat et al. 2005; Matthews et al. 2005). The purpose of a dam is to impound (store) water for any of several reasons (e.g., flood control, human water supply, irrigation, livestock water supply, energy generation, containment of mine tailings, recreation, erosion control, or pollution control such as sediment trapping). See Association of State Dam Safety Officials 2021; Field and Lichvar 2007; Renwick et al. 2005. Many dams fulfill a combination of the above functions. Because the purpose of a dam is to retain water effectively and safely, the water retention ability of a dam is of prime importance. Water may pass from the reservoir to the downstream side of a dam by: passing through the main spillway or outlet works; passing over an auxiliary spillway; overtopping the dam; seepage through the abutments; and seepage under the dam. Id. All water retention structures are subject to seepage through their foundations and abutments. U.S. Army Corps of Engineers 1992. Thus, waters behind a dam still maintain a hydrologic connection to downstream waters, though the presence of the dam can reduce the hydrological connectivity to downstream traditional navigable waters, the territorial seas, or interstate waters.

Impoundments store water and can have impacts on hydrology downstream. Dams alter the natural flow regime of the river, affecting movement of water and sediment. Science Report at 5-4. For example, impoundments for irrigation storage and flood control have altered flood magnitude, altered flow timing, and reduced flow variability and turbidity across the prairie regions. *Id.* at B-37 (citing Cross and Moss 1987; Hadley *et al.* 1987; Galat and Lipkin 2000). Under the Federal Flood Control Act of 1944, detention impoundments were extensively constructed on headwater streams in the Great Plains to retard flooding in downstream rivers. Science Report at B-29 (citing Schoof *et al.* 1978; Van Haveren 1986). Headwater impoundments reduced runoff to the Washita River in Oklahoma by 36%, but channel dredging of streams offset these reductions by increasing flow from ground water and reducing transmission loss. *Id.* (citing Schoof *et al.* 1978). Downstream, dams decrease peak stream volumes during the normal high-runoff seasons, while increasing minimum flows during normal low-flow seasons—an overall dampening of stream-flow variability *Id.* at 2-45 (citing Poff *et al.* 2007).

Numerous studies have shown that dams impede biotic movements, reducing biological connectivity between upstream and downstream locations. Science Report at 2-45 (citing Greathouse *et al.* 2006; Hall *et al.* 2011). They also form a discontinuity in the normal stream-order-related progression in stream ecosystem structure and function. *Id.* (citing Stanford and Ward 1984). Dams, however, can have the opposite effect with respect to natural lakes: increasing their biological connectivity with respect to invasive species by adding impoundments that decrease average distances between lakes and serving as stepping stone habitat. *Id.* (citing Johnson *et al.* 2008). Dams alter but typically do not sever the

hydrologic connection between upstream and downstream waters. Riparian areas are permanently inundated upstream of large dams, increasing hydrological connectivity. Downstream, peak flows decrease during normal high-runoff seasons, while minimum flows increase during normal low-flow seasons—an overall reduction of stream-flow variability. *Id.* (citing Poff *et al.* 2007). Many species that live in or near rivers are adapted (via life history, behavioral, and morphological characteristics) to the seasonality of natural flow regimes, so a reduction in flow variability can have harmful effects on the persistence of such species where dams have been built. *Id.* (citing Lytle and Poff 2004). This reduction in high flows also decreases the connectivity of riparian wetlands with the stream by reducing the potential for overbank lateral flow. Reducing overbank lateral flow can affect downstream water quality, because overbank flow deposits sediment and nutrients that otherwise remain entrained in the river. *Id.* (citing Hupp *et al.* 2009).

Dams also modify sediment dynamics within river networks. Science Report at 3-14. The reservoirs behind dams are very effective at retaining sediment (see, e.g., O'Melia 1998), which can reduce the amount of sediment delivered downstream and affect downstream waters. Kondolf et al. 2014. In the early 20th century, government agencies encouraged and funded various soil conservation practices and the construction of small impoundments on headwater streams to trap sediment and provide stable water supplies for livestock, irrigation, and recreation. Science Report at 3-15 (Person et al. 1936; Renwick *et al.* 2005). Although most such ponds are small (≤ 1 hectare or 2.5 acre) and represent only approximately 20% of the total impounded area (or 0.4% of the total watershed area), they can cumulatively have a significant effect on downstream waters. Id. For example, Smith and Kraft (2005) estimated that the approximately 2.3 million ponds distributed primarily on headwater streams of the Mississippi River network cumulatively captured 25–50% of the eroded soil from the landscape. Id. Similarly, Blum and Roberts (2009) estimated that the Mississippi River's natural sediment load has been reduced by an estimated 50% through dam construction in the Mississippi Basin. Sediment concentrations and suspended loads can be reduced for hundreds of kilometers downstream of dams, as is especially apparent in river networks in the semiarid and arid West. Science Report at 3-14 (citing Williams and Wolman 1984). The disruption of downstream sediment supply by dams alters the balance between sediment supply and transport capacity. Id. (citing Williams and Wolman 1984; Kondolf 1997). As described above in section III.A.i, sediment is a necessary material needed in river networks in certain quantities. Id. at 3-13. Too much sediment can impact downstream water integrity, but too little sediment can also impact downstream waters. Sediment helps structure stream and river channels by slowing the flow of water through channels and providing substrate and habitat for aquatic organisms. Id. at ES-8. At some point in the lower portions of river networks, sediment deposition becomes the dominant process and floodplains form. Id. at 2-4. Sediment also helps to build wetlands in coastal areas. Mitsch and Gosselink 2007. In coastal Louisiana, an estimated 25-38 square miles of wetlands are being lost each year to open water areas on the coastline due in part to the loss of sediment upstream behind the levee systems. Mitsch and Gosselink 2007. The river is no longer able to naturally replenish the sediment that rebuilds the marsh system. The disruption of downstream sediment supply by dams alters the balance between sediment supply and transport capacity. Science Report at 3-14 (citing Williams and Wolman 1984: Kondolf 1997). In addition, water released from dams lacks sediment load and thus has excess energy. This energy often downcuts channels downstream of dams, causing channel incision and streambed coarsening as finer gravels and sands are transported downstream over time. Id. (citing Williams and Wolman 1984; Kondolf 1997). The elimination of floods enables the encroachment of

terrestrial vegetation, resulting in channel narrowing and the conversion of complex, multithreaded channels into simple, single-thread channels.

Impoundments can also be created by natural processes, such as the construction of beaver dams. Factors such as hydrology, channel characteristics, topography, and building materials can influence the size and structure of beaver dams, as can ecological factors. Brazier *et al.* 2021. For example, primary beaver dams that maintained a lodge pond can be much larger than secondary dams, which are often used to improve mobility and the transport of woody material. Id. The size of individual beaver dams can be large, especially across wetland habitats. Larsen et al. 2021. The area impounded by beaver dams can be large: In the Kabetogama Peninsula of Minnesota, impounded area accounted for up to 13% of the landscape, with an average pond area of about four hectares. Science Report at 2-44 (citing Johnston and Naiman 1990a; Johnston and Naiman 1990b). Beaver dams do not completely restrict transport of water, sediment, and biology downstream. Id. at 3-18. Beaver dams typically reduce hydrologic connectivity downstream by impounding streams, increasing the extent of open water, and generally modifying conditions above the dam from lotic (flowing) to lentic (still). Id. at 2-44; Brazier et al. 2021; Larsen et al. 2021. Beaver dams, however, can increase connectivity laterally and vertically (e.g., hyporheic connectivity) by forcing water into nearby riparian areas, inundating floodplains, creating wetland and diverse habitats, and contributing to soil and groundwater recharge. Brazier et al. 2021; Larsen et al. 2021. These changes in hydrological connectivity are the bases for all subsequent impacts. Larsen et al. 2021. Beaver dams can hold large volumes of water in the impoundments they create and also expand riparian wetlands and water tables. Wegener et al. 2017. In a review of the effects of beaver dams on stream ecosystems, Collen and Gibson (2001) noted that, although the hydrologic effects of a single beaver dam can be small, the impact of a series of dams on streams can be significant; for example, up to 30% of the water in an Oregon catchment was impounded by beaver dams. Science Report at 2-44. Beaver dams can slow the flow of water, moderating peak flows downstream, and storing and slowly releasing waters during times of low flow or drought. Brazier et al. 2021. Westbrook et al. (2006) found that beaver dams in the Colorado River affected depth, extent, and duration of inundation resulting from a 10-year flood event. Science Report at 4-18. In addition, beaver dams attenuated declines in water tables during drier summer periods in 25% of their 58-hectare study area. Id. They concluded that the main hydrologic effects occurred downstream, however, rather than near the dam. Id. (citing Westbrook et al. 2006). The hydraulic head generated by the dam raised the water level above the banks, resulting in lateral and downstream spreading of flows during high- and low-flow periods; these effects extended over hundreds of meters. Id. For example, mottled soils occurred throughout the study area, suggesting that the beaver dams caused waterlogged soils for extended periods. Id. Increased overbank flooding increases hydrologic connectivity between riparian areas and streams. Id. In contrast, when no dams were present, flooding was limited to the area immediately near the stream channel. Id. Similar to human constructed dams, beaver dams can directly affect material transport (e.g., the ability of the stream to carry sediment is reduced) and alter biogeochemical characteristics. Science Report at 2-44 (citing Naiman et al. 1994; Collen and Gibson 2001) and 4-18. For example, beaver dams modify nutrient cycling and decomposition dynamics and can affect downstream transport of materials. Id. at 4-18 (citing Naiman et al. 1988; Naiman et al. 1994). Wegener et al. (2017) found that a segment of river network with beaver dams served as a sink for water, carbon, and nutrients during high flows, while subsequently becoming a source of these materials as flows decreased. Beaver dams slow the downstream transport of organic matter, enabling instream organisms to process the carbon and slowly leak material downstream Science Report

at 3-32 (citing Wohl and Beckman 2014). By slowing the flow of water, beaver dams and the ponds they create can result in the storage of sediment and nutrients that might otherwise be transported downstream. Brazier *et al.* 2021; Larsen *et al.* 2021. Beaver-dam wetlands can serve as a source of methylmercury Science Report at 4-18 (citing Roy *et al.* 2009). Beaver dams also can affect biological connectivity, for example, by obstructing upstream migration, and cause changes in fish distributions. Science Report at 2-44 (citing Collen and Gibson 2001); Brazier *et al.* 2021; Larsen *et al.* 2021. However, beaver dams can also increase habitat diversity due to the creation of beaver ponds and wetlands, which provide habitat for different life cycle needs for fish and other aquatic species that also utilize the water being impounded, including breeding, rearing, and feeding habitat. Brazier *et al.* 2021; Larsen *et al.* 2021. Beaver dams can also potentially stabilize downstream water temperatures. Larsen *et al.* 2021. Thus, impoundments of jurisdictional waters created by beaver dams can have chemical, physical, and biological effects on downstream traditional navigable waters, the territorial seas, and interstate waters.

Many adjacent wetlands are impounded for a variety of purposes, including for waterfowl habitat creation, aquaculture, agriculture, flood control, hurricane protection, mosquito control, and control of marsh subsidence and erosion. Day et al. 1990; U.S. Army Corps of Engineers b. For example, impoundments of wetlands in Louisiana were historically constructed to help control seasonal flooding of the Mississippi River or for agricultural purposes, later were constructed for wildlife management, and more recently have been constructed in coastal Louisiana to enclose wetlands in an attempt to reduce wetland loss. Day et al. 1990. Wetlands can be naturally impounded by beaver dams. Larsen et al. 2021. In addition, levees with gated culverts or pumps for controlling water levels can be used to impounded wetlands. Bryant et al. 1998. Dikes and constructed dams can also impound adjacent wetlands. One study that conducted an inventory of impounded wetlands on the Louisiana coast found that an area equal to approximately 30% of the total wetland area in the Louisiana coastal zone was currently or historically impounded. Day et al. 1990. Impoundments created in adjacent wetlands are similar to impoundments created on flowing bodies of water, like rivers that are traditional navigable waters or streams that are tributaries, and the impoundments can serve different functions than that of the wetlands that they are created from. Sometimes impoundments created from adjacent wetlands create open water, well other times wetland characteristics are maintained, depending on the structure that creates the impoundment. See, e.g., id. Levees that are constructed to impound adjacent wetlands restrict water movement between the impounded area and the wetland that is impounded and restrict sediment input to the impounded wetlands. Bryant et al. 1998. This may be a result of the prevention of delivery of floodwaters and associated sediments to the impounded areas, with the exception of overtopping of the levees during storm events. Id. Seepage can still occur through or under the levee. When adjacent wetlands are impounded by beaver dams, hydrologic connectivity is reduced, impacting the associated movement of materials like sediment. See, e.g., Larsen et al. 2021. Thus, the impoundment of adjacent wetlands could result in increased sediment deposition in downstream traditional navigable waters, the territorial seas, and interstate waters, due to the reduced ability of in-stream sediment to be deposited in the impounded portions of the adjacent wetlands. The scientific literature and the agencies' technical expertise and experience support that impoundments of adjacent wetlands can continue to have chemical, physical, and biological effects on traditional navigable waters, the territorial seas, and interstate waters.

Though the impoundment of traditional navigable waters, the territorial seas, interstate waters, jurisdictional tributaries, and jurisdictional adjacent wetlands can change the nature of the chemical, physical, and biological connections that such waters have downstream, it does not eliminate them. Thus,

impoundments continue to serve many important functions as an integral part of the tributary system, which in turn impact traditional navigable waters, the territorial seas, and interstate waters. *See, e.g.*, Science Report; Kondolf *et al.* 2014; Schmadel *et al.* 2019.

D. Intrastate Lakes and Ponds, Streams, or Wetlands Evaluated Under Paragraph (a)(5)

Paragraph (a)(5) of the final rule defines "waters of the United States" to include "intrastate lakes and ponds, streams, or wetlands not identified in paragraphs (a)(1) through (4)" that meet either the relatively permanent standard or the significant nexus standard. Thus, under paragraph (a)(5) of the final rule, jurisdiction over such waters would be based on the relatively permanent or significant nexus standards, not based on the interstate commerce test of the 1986 regulations.³⁹ The agencies have made important changes to the 1986 regulations to reflect the agencies' construction of the statutory limits on the scope of "waters of the United States" informed by the relevant provisions of the Clean Water Act and the statute as a whole, the scientific record, relevant Supreme Court precedent, and the agencies' experience and technical expertise after more than 45 years of implementing the longstanding pre-2015 regulations defining "waters of the United States." See section IV.C.6 of the final rule preamble. The final rule would provide for case-specific analysis of waters not addressed by any other provision of the definition to determine whether they are "waters of the United States" under the relatively permanent or significant nexus standards. As described in section III.D.i below, waters assessed under paragraph (a)(5) can provide functions that restore and maintain the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters. Therefore, the agencies have determined that including the category for paragraph (a)(5) waters in this rule best advances the objective of the Clean Water Act.

Under the final rule, such intrastate waters meet the relatively permanent standard when they are relatively permanent, standing or continuously flowing bodies of water with a continuous surface connection to traditional navigable waters, the territorial seas, interstate waters, and tributaries that meet the relatively permanent standard. For example, an intrastate lake with relatively permanent standing water that is not a traditional navigable water, is not a tributary, is not a jurisdictional impoundment, and is not an adjacent wetland may have a continuous surface connection to a traditional navigable water. Under paragraph (a)(5)(i) of the final rule such a water is evaluated for jurisdiction under the relatively permanent standard. Intrastate lakes and ponds, streams, or wetlands not identified in paragraphs (a)(1)through (4) of this rule that do not meet the relatively permanent standard are considered for jurisdiction under the significant nexus standard, where they are not excluded under paragraph (c) of the final rule. Waters assessed under paragraph (a)(5)(i) of the final rule can meet the continuous surface connection requirement if they are connected to a traditional navigable water, the territorial seas, or an interstate water or a tributary that is relatively permanent by a discrete feature like a non-jurisdictional ditch, swale, pipe, or culvert. Similarly, a natural berm, bank, dune, or similar natural landform between a water assessed under paragraph (a)(5) and a traditional navigable water, the territorial seas, or an interstate water or a tributary that is relatively permanent does not sever a continuous surface connection to the

 $^{^{39}}$ Under the 1986 regulations, such intrastate waters were evaluated under paragraph (a)(3) of that rule and were sometimes referred to as "(a)(3) waters" or "other waters."

extent it provides evidence of a continuous surface connection. *See* section IV.A of this document and section IV.C.5.c of the final rule preamble for a description of implementation tools that can be used to assess a continuous surface connection for a water assessed under paragraph (a)(5).

Under the significant nexus standard, waters assessed under paragraph (a)(5)(ii) are jurisdictional if they, either alone or in combination with similarly situated waters in the region, significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

The final rule contains an exclusive list of water types that could be jurisdictional under this provision if they are not jurisdictional under the other provisions of the definition: "[i]ntrastate lakes and ponds, streams, or wetlands." The list of water types does not reflect a conclusion that these waters are necessarily jurisdictional; rather the list is simply meant to inform the public of types of waters that can be jurisdictional if they meet the requisite test (either the relatively permanent standard or the significant nexus standard), even though they do not fall within the other provisions of the final rule. Though this list differs from the list in the 1986 regulations, this revision to the regulatory text is not meant to reflect a change in the types of waters that will be considered for jurisdiction under this provision; rather, based on public comment, the agencies believe that a streamlined list provides more clarity to the public. The agencies have identified the water types listed in the rule so that the more specific water types that were previously listed in paragraph (a)(3) of the 1986 regulations fall within one of the four water types in the final rule. For example, prairie potholes were in the list of water types in the 1986 regulations and, depending upon the characteristics of a particular prairie pothole, they may fall within the wetlands water type on the list (where they meet the regulatory definition of wetlands) or they may be lakes or ponds. Other examples include sloughs, as they typically fall within the wetlands water type or the streams water type, and playa lakes, which may fall within the lakes or ponds water type depending upon their size.

Intrastate waters that are not tributaries, jurisdictional paragraph (a)(2) impoundments, or adjacent wetlands under the final rule are sometimes referred to in scientific literature as "geographically isolated waters" and in policy as "isolated waters." Some geographically isolated wetlands also meet the definition of adjacent, such as wetlands behind berms and the like and wetlands within reasonably close proximity of other jurisdictional waters. The term "geographically isolated" should be used with caution, and cannot be used to infer a lack of connectivity to downstream waters, as these wetlands are often connected to downstream waters through deeper groundwater connections, biological connections, or spillage. The degree of connectivity of such wetlands will vary depending on landscape features such as distance from downstream waters and proximity to other wetlands of similar nature that as a group connect to jurisdictional downstream waters. Science Report at 3-43, 5-2.

The Science Report shows that intrastate waters evaluated under paragraph (a)(5) of the final rule—examples of which include, but are not limited to, depressional non-adjacent wetlands, non-tributary streams, open waters like lakes and ponds, and non-adjacent peatland wetlands—can provide important hydrologic (*e.g.*, flood control), water quality, and habitat functions which vary as a result of the diverse settings in which they exist across the country. *Id.* at 6-1. These functions are particularly valuable when considered cumulatively across the landscape or at the watershed scale and are often similar to the functions that adjacent wetlands provide, including water storage to control streamflow and mitigate or lessen downstream flooding; interruption and delay of the transport of water-borne pollutants

(such as excess nutrients and contaminants) over long distances; and retention of sediment. When there is a significant nexus, these functions restore and maintain the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. For non-floodplain wetlands and open waters lacking a channelized surface or regular shallow subsurface connection, generalizations from the literature available at the time of the Science Report's publication about their specific effects on downstream waters are difficult because information on both function and connectivity is needed (Science Report at ES-3) with notable exceptions (for example, the Report concluded that non-floodplain wetlands situated between a pollution source and a downstream water, intercepting the surface or shallow subsurface flowpath, will affect downstream waters through sink functions (Science Report at 60)), and thus case-specific analysis of significant nexus is appropriate from policy perspective to determine the specific effects that intrastate waters that do not meet the relatively permanent standard have on traditional navigable waters, the territorial seas, or interstate waters.

Intrastate waters evaluated under paragraph (a)(5) of the final rule individually span the gradient of hydrologic connectivity identified in the Science Report; they can be open waters located in the riparian area or floodplain of traditional navigable waters, the territorial seas, or interstate waters (e.g., oxbow lakes) and otherwise be physically proximate to the stream network (similar to adjacent wetlands) or they can be open waters or wetlands that are fairly distant from the network. See, e.g., Science Report at ES-12. They can be connected to traditional navigable waters, the territorial seas, or interstate waters via confined surface or subsurface connections (including channels, culverts, pipes, and tile drains), unconfined surface connections, shallow subsurface connections, deeper groundwater connections, biological connections, or spillage. They can also provide additional functions such as storage and mitigation of peak flows, natural filtration by biochemical uptake and/or breakdown of contaminants, and in some locations, high volume aquifer recharge that contributes to the baseflow in larger downstream waters. Id. at ES-10 to ES-11; McLaughlin et al. 2014; Lane et al. 2018; Neff and Rosenberry 2018. The strength of functions provided by such intrastate waters on traditional navigable waters, the territorial seas, and interstate waters will vary depending on the type (e.g., chemical, physical or hydrologic, or biological) and degree of connection (*i.e.*, from highly connected to highly isolated) to downstream waters and landscape features such as proximity to stream networks and similarly situated waters that function as a group to influence downstream traditional navigable waters, the territorial seas, or interstate waters. See, e.g., Science Report at ES-11.

Since the publication of the Science Report in 2015, the published literature has expanded scientific understanding and quantification of functions that such intrastate waters perform that affect the integrity of traditional navigable waters, the territorial seas, and interstate waters, particularly in the aggregate. As discussed in section I.C, the more recent literature (*i.e.*, 2014-present) has determined that non-floodplain wetlands can have demonstrable hydrologic and biogeochemical downstream effects, such as decreasing peak flows (Fossey and Rousseau 2016; Golden *et al.* 2016; Wang *et al.* 2019; Yeo *et al.* 2019; Rajib *et al.* 2020), maintaining baseflows (McLaughlin *et al.* 2014; Golden *et al.* 2019), and performing nitrate removal (Golden *et al.* 2019; Evenson *et al.* 2021), particularly when considered cumulatively.

Some intrastate waters considered under paragraph (a)(5) would meet the final rule's definition of "adjacent" if it applied to non-wetland waters like lakes and ponds. This would include, for example, lakes and ponds located behind berms but that are not impoundments of jurisdictional waters, non-

tributary lakes and ponds with an unbroken surface or shallow subsurface connection to jurisdictional waters, and lakes and ponds that are in close physical proximity to jurisdictional waters (e.g., oxbow lakes). As discussed in section III.B.ii.3.b, the science is clear that a water's proximity to downstream waters influences its impact on those waters. Open waters within close proximity of jurisdictional waters improve water quality through assimilation, transformation, or sequestration of nutrients, sediment, and other pollutants that can affect the integrity of traditional navigable waters, the territorial seas, or interstate waters. These waters also provide important habitat for aquatic-associated species to forage, breed, and rest. Some intrastate waters considered under paragraph (a)(5) have a shallow subsurface connection to jurisdictional waters. As discussed in section III.B.ii.3.a, the science demonstrates that waters with a shallow subsurface connection to jurisdictional waters can have important effects on traditional navigable waters, the territorial seas, and interstate waters. Some open waters are located within the riparian area or floodplain of the water to which they are proximate. Waters in stream and river channels can readily reach open waters in riparian areas via overbank flow, which occurs when floodwaters flow over stream and river channels. Science Report at 2-12 (citing Mertes 1997). The scientific literature, including the Science Report, supports that open waters in riparian areas and floodplains are chemically, physically, and biologically connected to traditional navigable waters, the territorial seas, or interstate waters and provide important functions that affect the integrity of such waters. See, e.g., ES-2 to ES-3.

Riparian and floodplain waters take many different forms. Some may be wetlands, while others may be ponds, oxbow lakes, or other types of open waters. Intrastate waters considered under paragraph (a)(5) of the final rule that are located in riparian areas or floodplains "are physically, chemically and biologically integrated with rivers via functions that improve downstream water quality, including the temporary storage and deposition of channel-forming sediment and woody debris, temporary storage of local ground water that supports baseflow in rivers, and transformation and transport of stored organic matter." Science Report at ES-2 to ES-3. Such wetlands and open waters act as an effective buffer to protect larger downstream waters from nonpoint source pollution (such as nitrogen and phosphorus), provide habitat for breeding fish and aquatic insects that also live in streams, and retain floodwaters, sediment, nutrients, and contaminants that could otherwise negatively impact the condition or function of traditional navigable waters, the territorial seas, and interstate waters. This inclusion of a case-specific significant analysis for such waters evaluated under paragraph (a)(5) that are located in the floodplain is supported by the SAB's review of a previous proposed rule. The SAB concluded that "distance should not be the sole indicator used to evaluate the connection of 'other waters' to jurisdictional waters." SAB 2014b at 3. In allowing the case-specific evaluation of waters that do not otherwise meet the definition of "waters of the United States" under the final rule's other categories, the agencies are allowing for the functional relationship of those floodplain waters to be considered regardless of proximity to the jurisdictional water. The SAB also supported the Science Report's conclusion that "the scientific literature strongly supports the conclusions that streams and 'bidirectional' floodplain wetlands are physically, chemically, and/or biologically connected to downstream navigable waters; however, these connections should be considered in terms of a connectivity gradient." SAB 2014b at 1. In addition, the SAB noted, "the literature review does substantiate the conclusion that floodplains and waters and wetlands in floodplain settings support the physical, chemical, and biological integrity of downstream waters." Id. at 3. By allowing for intrastate waters, including wetlands, that are located within the floodplain of a jurisdictional water to be considered for a case-specific analysis under the significant

nexus standard where the water does not meet the jurisdictional criteria under the rule's other categories of waters, the agencies are recognizing the science supporting the important effects that floodplain waters have on the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

Oxbow lakes and ponds (hereafter referred to as oxbow lakes), commonly found in floodplains of large rivers, are formed when river meanders (curves) are cutoff from the rest of the river. These waters, could be evaluated under paragraph (a)(5) of the final rule. Id. at 5-3. The Science Report presents a case study of these floodplain waters and concludes that the scientific evidence supports that oxbow lakes periodically connect to the active river channel and the connection between oxbow lakes and the active river channel provides for several ecological effects on the river ecosystem. Id. at B-8. Oxbow lakes and other lakes and ponds that are in close proximity to jurisdictional waters, that are located within floodplain or riparian areas, or that are connected via surface and shallow subsurface hydrology to the stream network or to other "waters of the United States" also perform critical chemical, physical, and biological functions that affect downstream traditional navigable waters, the territorial seas, and interstate waters. Like adjacent wetlands, these waters individually and collectively can affect the integrity of downstream waters by acting as sinks that retain floodwaters, sediments, nutrients, and contaminants that could otherwise negatively impact the condition or function of downstream waters. Id. at B-10, B-11. They also provide important habitat for aquatic species to forage, breed, and rest. Oxbow lakes play similar roles as floodplain wetlands as they are an integral part of alluvial floodplains of meandering rivers. Id. at B-8 (citing Winemiller et al. 2000; Glinska-Lewczuk 2009). They connect to rivers by periodic overland flow, typically from the river during flooding events, and bidirectional shallow subsurface flow through fine river soils (bidirectional means flow occurs both from the river to oxbow lake when the river has a high water stage and from the oxbow lake to the river at low water stage). Id. at B-9 to B-10. Oxbow lakes generally have an important influence on the chemical, physical, and biological condition and function of rivers. Id. at B-13 to B-14. That influence can vary with the distance from the river and the age of the oxbow, reflecting the frequency and nature of the exchange of materials that takes place between the two water bodies. Oxbow lakes also have high mineralization rates, suggesting that similar to adjacent wetlands they process and trap nutrients in runoff before the runoff reaches the river channel. Science Report at B-11 (citing Winemiller et al. 2000). Oxbow lakes in the floodplain provide critical fish habitat needed for feeding and rearing, leading researchers to conclude that the entire floodplain should be considered a single functional unit, essential to the river's biological integrity. Id. at 4-17 (citing Shoup and Wahl 2009). Since ponds that are near the tributary network are structurally and biologically similar to oxbow lakes, they serve similar functions relative to the nearby river or stream.

Waters evaluated under paragraph (a)(5) can be connected downstream through unidirectional flow from the wetland or open water to a nearby tributary. Such connections can occur through a surface or a shallow subsurface hydrologic connection. *Id.* at 2-7, 4-1 to 4-2, 4-22. Outside of the riparian zone and floodplain, surface hydrologic connections between waters assessed under paragraph (a)(5) of the final rule and jurisdictional waters can occur via confined flows (*e.g.*, a swale, gully, ditch, or other discrete feature). In some cases, these connections will be a result of "fill and spill" hydrology. A directional flowpath is a path where water flows repeatedly from the wetland or open water to the nearby jurisdictional water that at times contains water originating in the wetland or open water as opposed to just directly from precipitation. *Id.* at B-12 (citing Winter and Rosenberry 1998; Leibowitz and Vining 2003). Water connected through such flows originate from the "other water," travel to the downstream

jurisdictional water, and are connected to those downstream waters by swales or other directional flowpaths on the surface. Surface connections can also be unconfined (non-channelized flow), such as overland flow. *Id.* at 2-14.

A confined surface hydrologic connection, which may be perennial, intermittent, or ephemeral, supports periodic flows between the water assessed under paragraph (a)(5) and the jurisdictional water. For example, wetland seeps are likely to have perennial connections to streams that provide important sources of baseflow, particularly during summer. *Id.* at 4-21 (citing Morley *et al.* 2011). Some waters assessed under paragraph (a)(5) of the final rule are connected to streams via intermittent or ephemeral conveyances and can contribute flow to downstream waters via their surface hydrologic connection. *Id.* (citing Rains *et al.* 2006; Rains *et al.* 2008; McDonough *et al.* 2015). The surface hydrologic connection of the waters evaluated under paragraph (a)(5) to the jurisdictional water can contribute to the effects these waters have on traditional navigable waters, the territorial seas, or interstate waters. Waters evaluated under paragraph (a)(5) that are connected to jurisdictional waters through a confined surface hydrologic connection can have an impact on traditional navigable waters, the territorial seas, or e.g., *id.* at 4-40.

Waters evaluated under paragraph (a)(5) with confined surface connections can affect the physical integrity of waters to which they connect. Such waters can provide an important source of baseflow to nearby streams, helping to sustain the water levels in those streams, and ultimately in traditional navigable waters, the territorial seas, and interstate waters. Id. at 4-21 to 4-22 (citing Morley et al. 2011; Rains et al. 2006; Rains et al. 2008; Wilcox et al. 2011; McDonough et al. 2015); Lee et al. 2010. Waters evaluated under paragraph (a)(5) with a confined surface connection to downstream jurisdictional waters can affect streamflow by altering baseflow or stormflow through several mechanisms, including surface storage and groundwater recharge. Science Report at 4-24. Waters evaluated under paragraph (a)(5) with confined surface connections can affect water quality of jurisdictional waters through source and sink functions, often mediated by transformation of chemical constituents. The surface hydrologic connections to nearby jurisdictional waters provide pathways for materials transformed in the waters evaluated under paragraph (a)(5) (such as methylmercury or degraded organic matter) to reach and affect the nearby waters and the traditional navigable waters, the territorial seas, and interstate waters. Id. at 4-26 to 4-27. Waters evaluated under paragraph (a)(5) with confined surface connections also can affect the biological integrity of waters to which they connect. Movement of organisms between these waters and the nearby jurisdictional water is governed by many of the same factors that affect movement of organisms between adjacent wetlands and the river network. See, e.g., id. at 4-30. Because such waters evaluated under paragraph (a)(5) are at least periodically hydrologically connected to the nearby jurisdictional tributary network on the surface, dispersal of organisms can occur actively through the surface connection or via wind dispersal, hitchhiking, walking, crawling, or flying. See, e.g., id. at 4-30 to 4-31.

Biological connections between waters evaluated under paragraph (a)(5) and river systems do not always increase with hydrologic connections. In some cases, the lack of connection improves the biological contribution provided by such waters to nearby streams, rivers, and lakes and downstream traditional navigable waters, the territorial seas, and interstate waters. For instance, the periodic hydrologic disconnectedness of oxbow lakes is *necessary* for the accumulation of plankton, an important source of carbon more easily assimilated by the aquatic food chain than terrestrial forms of carbon. *Id.* at

B-11 to B-12 (citing Baranyi *et al.* 2002; Keckeis *et al.* 2003). Similarly, some degree of hydrological disconnectedness is important in increasing the number of mollusk species and macroinvertebrate diversity in oxbow lakes, which in turn support the diversity of mollusks throughout the aquatic system. *Id.* at B-12 (citing Reckendorfer *et al.* 2006; Obolewski *et al.* 2009).

Some waters assessed under paragraph (a)(5) are wetlands that are located too far from jurisdictional waters to be considered "adjacent" or are lakes and ponds that are not proximate to jurisdictional waters. The agencies have always recognized that adjacency is bounded by proximity. The science is clear that a water's proximity to downstream waters influences its impact on those waters. The Science Report states, "[s]patial proximity is one important determinant of the magnitude, frequency and duration of connections between wetlands and streams that will ultimately influence the fluxes of water, materials and biota between wetlands and downstream waters." Science Report at ES-11. Generally, waters that are closer to a jurisdictional water are more likely to be connected to that water than waters that are farther away.

The specific distance from jurisdictional waters may vary based on the characteristics of the aquatic resources being evaluated, but they are often located outside of the riparian area or floodplain, lack a confined surface or shallow subsurface hydrologic connection to jurisdictional waters, or exceed the minimum distances necessary for aquatic to utilize both the subject waters and the waters in the broader tributary network. Some intrastate waters considered under paragraph (a)(5) of the rule may be too removed from the stream network or from jurisdictional waters to have significant effects on traditional navigable waters, the territorial seas, or interstate waters. However, particularly when considered in the aggregate, some intrastate waters considered under paragraph (a)(5) can, in certain circumstances, have strong chemical, physical, and biological connections to and effects on traditional navigable waters, the territorial seas, or interstate waters. Sometimes it is their relative isolation from the stream network (e.g., lack of a hydrologic surface connection) that contributes to the important effect that they have downstream; for example, depressional non-floodplain wetlands lacking a confined surface outlet can function individually and cumulatively to retain and transform nutrients, retain sediment, provide habitat, and reduce or attenuate downstream flooding, depending on site-specific conditions such as landscape characteristics (e.g., slope of the terrain, soil permeability). Id. at 4-38; Lane et al. 2018; Golden et al. 2019; see section I.C.vii.

Some waters assessed under paragraph (a)(5) are located outside of the floodplain. Nonfloodplain waters perform many of the same functions as floodplain waters, but as discussed above, their connectivity to downstream waters varies. Generalizations about their effects on downstream waters can be difficult to ascertain from the available scientific literature. The functions of non-floodplain waters are discussed below.

Waters assessed under paragraph (a)(5) that are located outside the floodplain can affect streamflow by altering baseflow or storm flow through several mechanisms, including surface storage and groundwater recharge. Science Report at 4-24. Studies at the larger scale have shown that by storing water, wetlands, reduce peak flows and thus, downstream flooding. *Id.* at 4-25 (citing Jacques and Lorenz 1988; Vining 2002; McEachern *et al.* 2006; Gleason *et al.* 2007).

Non-floodplain waters evaluated under paragraph (a)(5), including wetlands, contain diverse microbial populations that perform various chemical transformations, acting as source of compounds and

potentially influencing the water quality downstream. *Id.* at 4-27 (citing Reddy and DeLaune 2008). Sulfate-reducing bacteria found in some non-floodplain wetlands produce methylated mercury, which is then transported downstream by surface flows. *Id.* (citing Linqvist *et al.* 1991; Mierle and Ingram 1991; Driscoll *et al.* 1995; Branfireun *et al.* 1999). Wetlands, including those that are waters assessed under paragraph (a)(5), are the principal sources of dissolved organic carbon (DOC) in forests to downstream waters. *Id.* at 4-28 (citing Mulholland and Kuenzler 1979; Urban *et al.* 1989; Eckhardt and Moore 1990; Koprivnjak and Moore 1992; Kortelainen 1993; Clair *et al.* 1994; Hope *et al.* 1994; Dillon and Molot 1997; Gergel *et al.* 1999). Export of DOC to traditional navigable waters, the territorial seas, and interstate waters from waters evaluated under paragraph (a)(5) can support primary productivity, affect pH and buffering capacity, and regulate exposure to UV-B radiation. *Id.* (citing Eshelman and Hemond 1985; Hedin *et al.* 1995; Schindler and Curtis 1997; Nuff and Asner 2001).

Non-floodplain waters assessed under paragraph (a)(5) also act as sinks and transformers for pollutants, including excess nutrients, through such processes as denitrification, ammonia volatilization, microbial and plant biomass assimilation, sedimentation, sorption and precipitation, biological uptake, and long-term storage of plant detritus. *Id.* at 4-29 (citing Ewel and Odum 1984; Nixon and Lee 1986; Johnston 1991; Detenbeck *et al.* 1993; Reddy *et al.* 1999; Mitsch and Gosselink 2007; Reddy and DeLaune 2008; Kadlec and Wallace 2009). Specifically, non-floodplain waters evaluated under paragraph (a)(5) can reduce phosphorus, nitrate, and ammonium by large percentages. *Id.* (citing Dierberg and Brezonik 1984; Dunne *et al.* 2006; Jordan *et al.* 2007; Cheesman *et al.* 2010). Wetland microbial processes reduce other pollutants, such as pesticides, hydrocarbons, heavy metals, and chlorinated solvents. *Id.* at 4-30 (citing Brooks *et al.* 1977; Kao *et al.* 2002; Boon 2006).

Non-floodplain waters considered under paragraph (a)(5) can have biological connections downstream that have the potential to impact the integrity of traditional navigable waters, the territorial seas, or interstate waters. Emergent and aquatic vegetation found in such non-floodplain waters disperse downstream by water, wind, and hitchhiking on (i.e., adhering to) migratory animals. Id. at 4-31 (citing Soons and Heil 2002; Soons 2006; Nilsson et al. 2010). Similarly, fish move between the river network and non-floodplain waters evaluated under paragraph (a)(5) during times of surface water connections. Id. at 4-34 (citing Snodgrass et al. 1996; Zimmer et al. 2001; Baber et al. 2002; Hanson et al. 2005; Herwig et al. 2010). Mammals that can disperse overland can also contribute to connectivity. Id. (citing Shanks and Arthur 1952; Roscher 1967; Serfass et al. 1999; Clark 2000; Spinola et al. 2008). Insects also hitchhike on birds and mammals from non-floodplain wetlands to the stream network, which can then serve as a food source for downstream waters. Id. at 4-31 (citing Figuerola and Green 2002; Figuerola et al. 2005). Insects that are flight-capable also use both the stream and non-floodplain waters moving from the stream to the wetland to find suitable habitat for overwintering, refuge from adverse conditions, hunting, foraging, or breeding. Id. at 4-34 (citing Williams 1996; Bohonak and Jenkins 2003). Amphibians and reptiles, including frogs, toads, and newts, also move between streams or rivers and nonfloodplain waters to satisfy part of their life history requirements, feed on aquatic insects, and avoid predators. Id. at 4-34 to 4-35 (citing Lamoureux and Madison 1999; Babbitt et al. 2003; Adams et al. 2005; Green 2005; Hunsinger and Lannoo 2005; Petranka and Holbrook 2006; Attum et al. 2007; Subalusky et al. 2009a; Subalusky et al. 2009).

The science itself does not establish bright lines for establishing where waters do not have a significant nexus to traditional navigable waters, the territorial seas, and interstate waters. For instance, as

noted above, the SAB concluded that distance should not be a sole factor used to evaluate the connection of waters to jurisdictional waters. SAB 2014b at 3. A case-specific analysis for waters assessed under paragraph (a)(5) of the final rule allows such waters to be considered jurisdictional only where they meet the relatively permanent or significant nexus standard.

The agencies emphasize that they fully support efforts by Tribes and States to protect under their own laws any additional waters, including locally important waters that may not be within the federal interests of the Clean Water Act as the agencies have interpreted its scope in this final rule. Indeed, the final rule and the definition of "waters of the United States" do not foreclose Tribes and States from acting consistent with their Tribal and State authorities to establish protection for waters that fall outside of the protection of the Clean Water Act.

Based on the functions that can be provided by intrastate lakes and ponds, streams, and wetlands that are not paragraph (a)(1) through (4) waters to traditional navigable waters, the territorial seas, and interstate waters, the final rule provides that such waters will be assessed to determine whether they meet either the relatively permanent standard or the significant nexus standard reflects proper consideration of the objective of the Act, relevant Supreme Court discussions, and the best available science.

i. Intrastate Waters Evaluated Under Paragraph (a)(5) Can Provide Functions that Restore and Maintain the Chemical, Physical, and Biological Integrity of Traditional Navigable Waters, the Territorial Seas, and Interstate Waters

Intrastate lakes and ponds, streams, and wetlands that do not meet the relatively permanent standard can provide functions that restore and maintain the chemical, physical, and biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters, and such waters will be evaluated on a case-specific basis under the final rule. This section will focus on intrastate waters evaluated under paragraph (a)(5) of the final rule and the functions they provide that benefit traditional navigable waters, the territorial seas, and interstate waters.

Though much of the literature cited in the Science Report relates to waters evaluated under paragraph (a)(5) that are streams or wetlands, the Science Report indicates that open waters also can have chemical, physical, or biological connections that significantly impact downstream waters. For instance, ponds or lakes that are not part of the tributary network can still be connected to downstream waters through chemical, physical, and biological connections. Lake storage has been found to attenuate peak streamflows in Minnesota. *Id.* at 4-25 (citing Jacques and Lorenz 1988; Lorenz *et al.* 2010). Similar to wetlands, ponds are often used by invertebrate, reptile, and amphibian species that also utilize downstream waters for various life history requirements, particularly because many ponds, particularly temporary ponds, are free of predators, such as fish, that prey on larvae. The American toad and Eastern newt, for example, are widespread habitat generalists that can move among streams, wetlands, and ponds to take advantage of each aquatic habitat, feeding on aquatic invertebrates, and avoiding larger predators. *See, e.g., Id.* at 4-35 (citing Babbitt *et al.* 2003; Green 2005; Hunsinger and Lannoo 2005; Petranka and Holbrook 2006).

The physical effect that intrastate waters evaluated under paragraph (a)(5) of the final rule have downstream can be less obvious than the physical connections of adjacent wetlands and of tributaries that do not meet the relatively permanent standard when such waters are physical distant from the stream network or from jurisdictional waters. Despite this physical distance, they are frequently connected in some degree through either surface water or groundwater systems; over time, impacts in one part of the hydrologic system will be felt in other parts. Winter and LaBaugh 2003. For example, waters assessed under paragraph (a)(5) that overspill into downstream water bodies during times of abundant precipitation are connected over the long term. Id. Wetlands that lack surface connectivity in a particular season or year can, nonetheless, be highly connected in wetter seasons or years. Science Report at 4-24. Floodplain waters and non-adjacent wetlands are connected to traditional navigable waters, the territorial seas, or interstate waters via both surface and subsurface hydrologic flowpaths and can reduce flood peaks by storing floodwaters. Id. at ES-9. Many waters assessed under paragraph (a)(5) interact with groundwater, either by receiving groundwater discharge (flow of groundwater to the case-specific water), contributing to groundwater recharge (flow of water from the case-specific water to groundwater), or both. Id. at 4-22 (citing Lide et al. 1995; Devito et al. 1996; Matheney and Gerla 1996; Rosenberry and Winter 1997; Pyzoha et al. 2008). Factors that determine whether a water recharges groundwater or is a site of groundwater discharge include topography, geology, soil features, and seasonal position of the water table relative to the water. Id. at 4-23 (citing Phillips and Shedlock 1993; Shedlock et al. 1993; Lide et al. 1995; Sun et al. 1995; Rosenberry and Winter 1997; Pyzoha et al. 2008; McLaughlin et al. 2014). Similarly, the magnitude and transit time of groundwater flow from a water assessed under paragraph (a)(5) to downstream traditional navigable waters, the territorial seas, or interstate waters depend on several factors, including the intervening distance and the properties of the rock or unconsolidated sediments between the water bodies (*i.e.*, the hydraulic conductivity of the material). Id. at 4-23. Surface and groundwater hydrological connections are those generating the capacity for waters assessed under paragraph (a)(5) to affect downstream waters, as water from the aquatic resource being assessed may contribute to baseflow or stormflow through groundwater recharge. Id. at 4-24. Contributions to baseflow are important for maintaining conditions that support aquatic life in downstream traditional navigable waters, the territorial seas, and interstate waters. As discussed further below, even in cases where waters assessed under paragraph (a)(5) lack a connection to downstream waters, they can influence downstream waters through water storage and mitigation of peak flows. Id. at 4-2, 4-42, 4-43.

The chemical effects that waters assessed under paragraph (a)(5) have on downstream waters are linked to their hydrologic connection downstream, though a surface connection is not needed for a water to influence the chemical integrity of the downstream traditional navigable water, the territorial seas, or interstate water. When waters assessed under paragraph (a)(5) are hydrologically connected to downstream waters via surface or groundwater connections, such waters can affect water quality downstream (although these connections do not meet the definition of adjacency for wetlands). Whigham and Jordan 2003. Waters assessed under paragraph (a)(5) can act as sinks and transformers for nitrogen and phosphorus, metals, pesticides, and other contaminants that could otherwise negatively impact downstream traditional navigable waters, the territorial seas, and interstate waters. Science Report at 4-29 to 4-30 (citing Brooks *et al.* 1977; Hemond 1980; Davis *et al.* 1981; Hemond 1983; Ewel and Odum 1984; Moraghan 1993; Craft and Chiang 2002; Kao *et al.* 2002; Boon 2006; Dunne *et al.* 2006; Cohen *et al.* 2007; Jordan *et al.* 2007; Whitmire and Hamilton 2008; Bhadha *et al.* 2011; Marton *et al.* 2014). *See also, e.g.*, Isenhart 1992. Schmadel *et al.* (2019) found that small ponds nearby the stream network are the

dominant nitrogen sinks in headwater catchments, while small ponds not near the stream network are the dominant phosphorus sinks. The body of published scientific literature and the Science Report indicate that sink removal of nutrients and other pollutants by waters assessed under paragraph (a)(5) is significant and geographically widespread. Science Report at 4-30. Such waters located on floodplains provide water quality benefits for the downstream traditional navigable waters, the territorial seas, and interstate waters, including retention of sediment and organic matter and retention, cycling, and transformation of pollutants like nutrients. *Id.* at ES-9. Water quality characteristics of waters assessed under paragraph (a)(5) are highly variable, depending primarily on the sources of water, characteristics of the substrate, and land uses within the watershed. Whigham and Jordan 2003. These variables inform whether a water assessed under paragraph (a)(5) has a significant nexus to a traditional navigable water, the territorial seas, or an interstate water. *See also* section III.E.ii. For instance, some prairie potholes may improve water quality and may efficiently retain nutrients that might otherwise cause water quality problems in larger downstream waters; in such systems it may be their lack of a direct hydrologic connection that enables the prairie potholes to retain nutrients more effectively. Whigham and Jordan 2003; Science Report at 2-28 to 2-29.

Waters assessed under paragraph (a)(5) can be biologically connected to each other and to downstream traditional navigable waters, the territorial seas, and interstate waters through the movement of seeds, macroinvertebrates, amphibians, reptiles, birds, and mammals. Science Report at 4-30 to 4-35; Leibowitz 2003. The movement of organisms between such waters and downstream waters is governed by many of the same factors that affect movement of organisms between adjacent wetlands and downstream waters. See section III.B; Science Report at 4-30. For example, "waters assessed under paragraph (a)(5) that are located in the floodplain of a jurisdictional water are hydrologically connected to traditional navigable waters, the territorial seas, or interstate waters by lateral expansion and contraction of the jurisdictional water in its floodplain, resulting in an exchange of matter and organisms with the jurisdictional water and further downstream to traditional navigable waters, the territorial seas, or interstate waters. One example of such a connection is fish populations that are adapted to use wetlands and open waters in the floodplain for feeding and spawning during high water. Id. at ES-9 to ES-10. Many waters assessed under paragraph (a)(5) are further away from stream channels than adjacent wetlands, making hydrologic connectivity less frequent, and increasing the number and variety of landscape barriers over which organisms must disperse. Id. Plants, though non-mobile, have evolved many adaptations to achieve dispersal over a variety of distances, including water-borne dispersal during periodic hydrologic connections, "hitchhiking" on or inside highly mobile animals, and more typically via wind dispersal of seeds and/or pollen. Id. at 4-31 (citing Galatowitsch and van der Valk 1996; Murkin and Caldwell 2000; Amezaga 2002; Figuerola and Green 2002; Soons and Heil 2002; Soons 2006; Haukos et al. 2006 and references therein; Nilsson et al. 2010). Mammals that disperse overland can also contribute to connectivity and can act as transport vectors for hitchhikers such as algae. Id. at 4-34 (citing Shanks and Arthur 1952; Roscher 1967; Serfass et al. 1999; Clark 2000; Spinola et al. 2008). Invertebrates also utilize birds and mammals to hitchhike, and these hitchhikers can be an important factor structuring invertebrate metapopulations in case-specific waters and in aquatic habitats separated by hundreds of kilometers. Id. at 4-31 through 4-32 (Figuerola and Green 2002; Figuerola et al. 2005; Allen 2007; Frisch 2007). Numerous flight-capable insects use both waters assessed under paragraph (a)(5) and downstream traditional navigable waters, the territorial seas, or interstate waters; these insects move outside the tributary network to find suitable habitat for overwintering, refuge from adverse conditions, hunting,

foraging, or breeding, and then can return back to the tributary network for other life cycle needs. *Id.* at 4-34 (citing Williams 1996; Bohonak and Jenkins 2003). Amphibians and reptiles also move between waters assessed under paragraph (a)(5) and downstream waters to satisfy part of their life history requirements. *Id.* at 4-34. Alligators in the Southeast, for instance, can move from tributaries to shallow, seasonal limesink wetlands for nesting, and also use these wetlands as nurseries for juveniles; sub-adults then shift back to the tributary network through overland movements. *Id.* (citing Subalusky *et al.* 2009a; Subalusky *et al.* 2009b). Similarly, amphibians and small reptile species, such as frogs, toads, and newts, commonly use both tributaries and waters assessed under paragraph (a)(5), during one or more stages of their life cycle, and can at times disperse over long distances. *Id.* at 4-34 to 4-35 (citing Knutson *et al.* 1999; Lamoureux and Madison 1999; Babbitt *et al.* 2003; Adams *et al.* 2005; Green 2005; Hunsinger and Lannoo 2005; Petranka and Holbrook 2006; Attum *et al.* 2007).

Even when a surface or groundwater hydrologic connection between a water assessed under paragraph (a)(5) of the final rule and a downstream traditional navigable water, the territorial sea, or interstate water is visibly absent, many waters still have the ability to have a material influence on the integrity of those downstream fundamental waters. Such circumstances would be uncommon, but can occur, for instance, where a wetland recharges a deep groundwater aquifer that does not feed surface waters, or it is located in a basin where evapotranspiration is the dominant form of water loss. Id. at 4-21 to 4-24. Aquatic systems that may seem disconnected hydrologically are often connected but at irregular timeframes or through subsurface flow, and perform important functions that can be vital to the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. Some wetlands that may be hydrologically disconnected most of the time but connected to the stream network during rare high-flow events or during wetter seasons or years. Although the Science Report focuses primarily on the benefits that connectivity can have on downstream systems, isolation also can have important positive effects on the condition and function of downstream traditional navigable waters, the territorial seas, and interstate waters. Id. at 2-28. For instance, the lack of a hydrologic connection allows for water storage in such waters, attenuating peak streamflows, and, thus, downstream flooding, and also reducing nutrient and soil pollution in downstream waters. Id. at 2-28 to 2-29, 4-2, 4-38. Prairie potholes a great distance from any tributary, for example, are thought to store significant amounts of runoff. Id. at 4-38 (citing Novitzki 1979; Hubbard and Linder 1986; Vining 2002; Bullock and Acreman 2003; McEachern et al. 2006; Gleason et al. 2007). Filling wetlands reduces water storage capacity in the landscape and causes runoff from rainstorms to overwhelm the remaining available water conveyance system. See, e.g., Johnston et al. 1990; Moscrip and Montgomery 1997; Detenbeck et al. 1999; Detenbeck et al. 2005. Wetlands, even when lacking a hydrologic connection downstream, improve downstream water quality by accumulating nutrients, trapping sediments, and transforming a variety of substances. See, e.g., National Research Council 1995.

The structure and function of a river are highly dependent on the constituent materials that are stored in, or transported through, the river. Most of the materials found in rivers originate outside of them. Thus, the fundamental way that waters evaluated under (a)(5) are able to affect river structure and function is by providing or altering the materials delivered to the river. Science Report at 1-13. Since the alteration of material fluxes depends on the functions within these waters evaluated under paragraph (a)(5) and the degree of connectivity, it is appropriate to consider both these factors for purposes of the significant nexus standard under the final rule.

Based on the functions that can be provided by waters evaluated under paragraph (a)(5) to traditional navigable waters, the territorial seas, and interstate waters, the agencies' assessment of such intrastate waters on a case-specific basis to determine whether they meet either the relatively permanent standard or the significant nexus standard in the final rule reflects proper consideration of the objective of the Clean Water Act and the best available science.

E. Significant Nexus Standard

i. Waters Subject to the Significant Nexus Standard

Under the final rule, several categories of waters require a case-specific analysis to determine if they meet the significant nexus standard. This includes tributaries not meeting the relatively permanent standard, wetlands adjacent to tributaries and impoundments and that do not meet the relatively permanent standard, and intrastate waters assessed under paragraph (a)(5) that do not meet the relatively permanent standard.

The agencies have concluded that the significant nexus standard is consistent with the statutory text, advances the objective of the Clean Water Act, is informed by the scientific record and Supreme Court case law, and appropriately considers the policies of the Act. The agencies have also concluded that the relatively permanent standard on its own identifies only a subset of the "waters of the United States," although it is being retained because it provides important administrative efficiencies. The terms in this rule are generally familiar and implementable as they reflect consideration of the agencies' experience and expertise, as well as updates in implementation tools and resources.

The scientific literature documents the functions of waters that do meet the final rule's relatively permanent standard, including the chemical, physical, and biological impact they can have downstream. Available literature indicates that such waters can have important hydrologic, water quality, and habitat functions that have the ability to affect downstream traditional navigable waters, the territorial seas, and interstate waters, if and when a connection exists between the water and those larger downstream waters. Science Report at 6-5. Connectivity of waters evaluated under the significant nexus standard to traditional navigable waters, the territorial seas, and interstate waters, and for waters evaluated under paragraph (a)(5) in particular, will vary within a watershed and over time. The agencies have determined that where such waters do not meet the relatively permanent standard, as a matter of policy that a case-specific significant nexus evaluation is appropriate. See, e.g., id. The types of chemical, physical, and biological connections between such waters and traditional navigable waters, the territorial seas, and interstate waters are described below for illustrative purposes. As described in the rule's preamble, when the agencies are conducting a case-specific evaluation for significant nexus, they examine the connections between the water (including any similarly situated waters in the region) and downstream traditional navigable waters, the territorial seas, or interstate waters, assess the relevant factors that affect connectivity and functions, and determine if those waters have a material influence on the chemical, physical, or biological integrity of the traditional navigable water, the territorial seas, or the interstate water, using any available site-information and field observations, relevant scientific studies or data, or other information.

The hydrologic connectivity of waters evaluated under the significant nexus standard to traditional navigable waters, the territorial seas, or interstate waters occurs on a gradient and can include waters in the floodplain, waters that have groundwater or occasional surface water connections (through overland flow) to the tributary network, and waters that have no hydrologic connection to the tributary network or to jurisdictional waters. *Id.* at 4-2. The connectivity of waters analyzed under the significant nexus standard to downstream traditional navigable waters, the territorial seas, or interstate waters will vary within a watershed as a function of local factors (*e.g.*, position, topography, and soil characteristics). *Id.* at 4-41. Connectivity also varies over time, as the tributary network and water table expand and contract in response to local climate. *Id.* Lack of connection does not necessarily translate to lack of impact; even when lacking connectivity, waters can still impact chemical, physical, and biological conditions downstream. *Id.* at 4-42 to 4-43.

Under the final rule, on a case-specific basis, waters that meet the significant nexus standard are "waters of the United States" under paragraphs (a)(3)(ii), (a)(4)(iii), and (a)(5)(ii). The scientific literature and data cited in the Science Report and this Technical Support Document demonstrate that these waters, along with any similarly situated waters in the region, significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

The agencies also rely on the extensive experience the Corps has gained in making significant nexus determinations since the *Rapanos* decision. Since the *Rapanos* decision, the agencies have developed extensive experience making significant nexus determinations. The agencies have made determinations in every state in the country, for a wide range of waters in a wide range of conditions.

The rule's requirements for these waters, coupled with those for waters meeting the relatively permanent standard, create an integrated approach that tailors the regulatory regime based on the Supreme Court decisions and the agencies' policy objectives and informed by science. Providing for case-specific significant nexus analysis for waters that do not meet the relatively permanent standard is consistent with science, agency experience, and longstanding pre-2015 practice and will ensure protection of the important waters whose protection will advance the goals of the Clean Water Act.

For these reasons, the agencies decided to allow case-specific determinations of significant nexus for tributaries not meeting the relatively permanent standard, wetlands adjacent to tributaries and impoundments and that do not meet the relatively permanent standard, and intrastate waters assessed under paragraph (a)(5) that do not meet the relatively permanent standard. Under the rule, these waters are jurisdictional only where they individually or cumulatively (if it is determined that there are other similarly situated waters in the region) have a significant nexus to traditional navigable waters, the territorial seas, or interstate waters.

ii. "Similarly Situated"

Science supports analyzing the contributions of similarly situated waters in combination with each other for their effect on downstream traditional navigable waters, the territorial seas, and interstate waters. The agencies are establishing an approach to "similarly situated" for adjacent wetlands and tributaries and a different approach to "similarly situated" for intrastate waters assessed under paragraph

(a)(5)(ii).

In implementing the final rule, the agencies consider tributaries and their adjacent wetlands to be "similarly situated" waters for purposes of the significant nexus standard. This is consistent with longstanding practice, and the agencies believe that this integrated approach to considering tributaries and their adjacent wetlands together when assessing the effects of these waters on the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters reflects that these waters act together to influence downstream waters, as discussed further in this section. In considering how to apply the significant nexus standard, the agencies have long focused on the integral relationship between the ecological characteristics of tributaries and those of their adjacent wetlands, which determines in part their contribution to restoring and maintaining the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

In implementing the significant nexus standard for waters assessed under paragraph (a)(5) of the final rule, the agencies generally intend to analyze such waters individually to determine if they significantly affect the chemical, physical, or biological integrity of a paragraph (a)(1) water. This approach reflects the agencies' consideration of public comments, as well as implementation considerations for waters assessed under paragraph (a)(5). While the agencies' regulations have long authorized the assertion of jurisdiction on a case-specific basis over waters that do not fall within the jurisdictional provisions by water type, since SWANCC and the issuance of the SWANCC guidance with its requirement of headquarters approval over determinations under that provision, the agencies have not in practice asserted jurisdiction over paragraph (a)(3) "other waters" under the pre-2015 regulatory regime.⁴⁰ The agencies addressed such waters individually on a case-specific basis under pre-2015 practice and have concluded at this time that individual assessments are practical and implementable for significant nexus determinations for waters assessed under paragraph (a)(5). The agencies note that generally assessing waters evaluated under paragraph (a)(5) on an individual basis represents a policy decision. One conclusion of Science Report was that the incremental effects of individual streams and wetlands are cumulative across entire watersheds and therefore should be evaluated in context with other streams and wetlands.

Streams, wetlands, and other surface waters interact with ground water and terrestrial environments throughout the landscape, "from the mountains to the oceans." *Id.* at 1-2 (citing Winter *et al.* 1998). Thus, an integrated perspective of the landscape, provides the appropriate scientific context for evaluating and interpreting evidence about the chemical, physical, and biological connectivity of streams, wetlands, and open waters to downstream traditional navigable waters, the territorial seas, and interstate waters.

Connectivity has long been a central tenet for the study of aquatic ecosystems. The River Continuum Concept viewed the entire length of rivers, from source to mouth, as a complex hydrologic

⁴⁰ Note that when the 2015 Clean Water Rule was in effect, the agencies did assert categorical jurisdiction over waters if they were adjacent waters as defined by that rule and asserted jurisdiction on a case-specific basis over waters that fell within the provisions requiring case-specific significant nexus determinations where such waters were determined to have a significant nexus. Under the pre-2015 regulatory regime, such waters would have been assessed under paragraph (a)(3) of the agencies' 1986 regulations. The 2020 NWPR also asserted jurisdiction over certain lakes and ponds that would have been considered under the paragraph (a)(3) "other waters" category under the pre-2015 regulatory regime.
gradient with predictable longitudinal patterns of ecological structure and function. Id. (citing Vannote et al. 1980). The key pattern is that downstream communities are organized, in large part, by upstream communities and processes. Id. (citing Vannote et al. 1980; Battin et al. 2009). The Serial Discontinuity Concept built on the River Continuum Concept to improve our understanding of how dams and impoundments disrupt the longitudinal patterns of flowing waters with predictable downstream effects. Id. (citing Ward and Stanford 1983). The Spiraling Concept described how river network connectivity can be evaluated and quantified as materials cycle from dissolved forms to transiently stored forms taken up by living organisms, then back to dissolved forms, as they are transported downstream. Id. at 1-3 (citing Webster and Patten 1979; Newbold et al. 1981; Elwood et al. 1983). These three conceptual frameworks focused on the longitudinal connections of river ecosystems, whereas the subsequent flood pulse concept examined the importance of lateral connectivity of river channels to floodplains, including wetlands and open waters, through seasonal expansion and contraction of river networks. Id. (citing Junk et al. 1989). Ward (1989) summarized the importance of connectivity to lotic ecosystems along four dimensions: longitudinal, lateral, vertical (surface-subsurface), and temporal connections; he concluded that running water ecosystems are open systems that are highly interactive with both contiguous habitats and other ecosystems in the surrounding landscape. Id. As these conceptual frameworks illustrate, scientists have long recognized the hydrologic connectivity the physical structure of river networks represents.

More recently, scientists have incorporated this connected network structure into conceptual frameworks describing ecological patterns in river ecosystems and the processes linking them to other watershed components, including wetlands and open waters. *Id.* (citing Power and Dietrich 2002; Benda *et al.* 2004b; Nadeau and Rains 2007; Rodriguez-Iturbe *et al.* 2009). Sheaves (2009) emphasized the key ecological connections—which include process-based connections that maintain habitat function (*e.g.*, nutrient dynamics, trophic function) and movements of individual organisms—throughout a complex of interlinked freshwater, tidal wetlands, and estuarine habitats as critical for the persistence of aquatic species, populations, and communities over the full range of time scales. *Id.*

Scientists routinely aggregate the effects of groups of waters, multiplying the known effect of one water by the number of similar waters in a specific geographic area, or to a certain scale. This kind of functional aggregation of non-adjacent (and other types of waters) is well-supported in the scientific literature. *See, e.g.*, Stevenson and Hauer 2002; Leibowitz 2003; Gamble *et al.* 2007; Lane and D'Amico 2010; Wilcox *et al.* 2011. Similarly, streams and rivers are routinely aggregated by scientists to estimate their combined effect on downstream waters in the same watershed. This is because chemical, physical, or biological integrity of downstream waters is directly related to the aggregate contribution of upstream waters that flow into them, including any tributaries and connected wetlands. As a result, the scientific literature and the Science Report consistently document that the health of larger downstream waters is directly related to the aggregate bealth of waters located upstream, including waters such as wetlands that may not be hydrologically connected but function together to prevent floodwaters and contaminants from reaching downstream traditional navigable waters, the territorial seas, and interstate waters.

Stream and wetland connectivity to downstream waters, and the resulting effects on the integrity of downstream traditional navigable waters, the territorial seas, and interstate waters, is best understood and assessed when considered cumulatively. Science Report at 1-10. First, when considering the effect of an individual stream or wetland, including the cumulative effect of all the contributions and functions that a stream or wetland provides is essential. For example, the same stream transports water, removes excess

nutrients, mitigates flooding, and provides refuge for fish when conditions downstream are unfavorable; ignoring any of these functions would underestimate the overall effect of that stream.

Secondly, and perhaps more importantly, stream channel networks and the watersheds they drain are fundamentally cumulative in how they are formed and maintained. Excess precipitation that is not evaporated, taken up by organisms, or stored in soils and geologic layers moves downgradient as overland flow or through channels, which concentrate flows and carry sediment, chemical constituents, and organisms. As flows from numerous headwater channels combine in larger channels, the volume and effects of those flows accumulate as they move through the river network. As a result, the incremental contributions of individual streams and wetlands accumulate in the downstream waters. Important cumulative effects are exemplified by ephemeral flows in arid landscapes, which are key sources of baseflow for downgradient waters, and by the high rates of denitrification in headwater streams. Id. (citing Schlesinger and Jones 1984; Baillie et al. 2007; Izbicki 2007). The amount of nutrients removed by any one stream over multiple years or by all headwater streams in a watershed in a given year can have substantial consequences for downstream waters. Id. (citing Alexander et al. 2007; Alexander et al. 2009; Böhlke et al. 2009; Helton et al. 2011). Similar cumulative effects on downstream waters have been documented for other material contributions from headwater streams in the Science Report. For example, although the probability of a large-magnitude transfer of organisms from any given headwater stream in a given year might be low (*i.e.*, a low-frequency connection when each stream is considered individually), headwater streams are the most abundant type of stream in most watersheds. Thus, the overall probability of a large-magnitude transfer of organisms is higher when considered for all headwater streams in a watershed—that is, there is a high-frequency connection when considered cumulatively at the watershed scale, compared with probabilities of transport for streams individually. Similarly, a single pollutant discharge might be negligible but the cumulative effect of multiple discharges could degrade the integrity of downstream traditional navigable waters, the territorial seas, or interstate waters.

Evaluating cumulative contributions over time is critical in streams and wetlands with variable degrees of connectivity. *Id.* at 1-11. For example, denitrification in a single headwater stream in any given year might affect downstream waters; over multiple years, however, this effect could accumulate. Western vernal pools provide another example of cumulative effects over time. These pools typically occur as complexes in which the hydrology and ecology are tightly coupled with the local and regional geological processes that formed them. When seasonal precipitation exceeds wetland storage capacity and wetlands overflow into the river network and generate stream discharge, the vernal pool basins, swales, and seasonal streams function as a single surface-water and shallow ground-water system connected to the river network.

In the aggregate, similarly situated waters may have significant effects on the quality of water many miles away, particularly in circumstances where numerous similarly situated waters are located in the region and are performing like functions that combine to influence downstream waters. *See, e.g.,* Jansson *et al.* 1998; Mitsch *et al.* 2001; Forbes *et al.* 2012. Cumulatively, many small wetlands can hold a large amount of snowmelt and precipitation, reducing the likelihood of flooding downstream. Science Report at 4-24 (citing Hubbard and Linder 1986).

Scientists can and do routinely classify similar waters and wetlands into groups for a number of different reasons; because of their inherent physical characteristics, because they provide similar

functions, because they were formed by similar geomorphic processes, and by their level of biological diversity, for example. Classifying wetlands based on their functions is also the basis for the U.S. Army Corps of Engineers hydrogeomorphic (HGM) classification of wetlands. Brinson 1993. The HGM method is a wetlands assessment approach pioneered by the Corps in the 1990s, and extensively applied via regional handbooks since then. The Corps HGM method uses a conceptual framework for identifying broad wetland classes based on common structural and functional features, which includes a method for using local attributes to further subdivide the broad classes into regional subclasses. Assessment methods like the HGM provide a basis for determining if waters provide similar functions based on their structural attributes and indicator species. Scientists also directly measure attributes and processes taking place in particular types of waters during in-depth field studies that provide reference information that informs the understanding of the functions performed by many types of aquatic systems nationwide.

Consideration of the aggregate effects of wetlands and other types of waters often gives the most complete information about how such waters influence the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, and interstate waters. In many watersheds, wetlands have a disproportionate effect on water quality relative to their surface area because wetland plants slow down water flow, allowing suspended sediments, nutrients, and pollutants to settle out. They filter these materials out of the water received from large areas, absorbing or processing them, and then releasing higher quality water. National Research Council 1995. For an individual wetland, this is most pronounced where it lies immediately upstream of a drinking water intake, for example. *See, e.g.*, Johnston *et al.* 1990. The cumulative influence of many individual wetlands within watersheds can strongly affect the spatial scale, magnitude, frequency, and duration of hydrologic, biological, and chemical fluxes or transfers of water and materials to downstream waters. Science Report at ES-11.

For example, as discussed in section III.A.ii, excess nutrients discharged into small tributary streams in the aggregate can cause algal blooms downstream that reduce dissolved oxygen levels and increase turbidity in traditional navigable waters, traditional navigable waters, the territorial seas, and interstate waters. This oxygen depletion in waters, known as hypoxia, has impacted commercial and recreational fisheries in the northern Gulf of Mexico, as water low in dissolved oxygen cannot support living aquatic organisms. Committee on Environment and Natural Resources 2000; Rabalais *et al.* 2002; Freeman *et al.* 2007; Colvin 2019. In this instance, the cumulative effects of nutrient export from the many small headwater streams of the Mississippi River have resulted in large-scale ecological and economically harmful impacts hundreds of miles downstream. *See, e.g.*, Goolsby *et al.* 1999; Rabotyagov *et al.* 2014; Colvin 2019.

In their review of the scientific and technical adequacy of a previous rulemaking effort, the SAB panel members "generally agreed that aggregating 'similarly situated' waters is scientifically justified, given that the combined effects of these waters on downstream waters are often only measurable in aggregate. Panelists also were generally comfortable with the idea of using "similarly situated" waters to guide aggregation." SAB 2014a at 4 to 5. One of the main conclusions of the Science Report is that the incremental contributions of individual streams and wetlands are cumulative across entire watersheds, and their effects on downstream waters should be evaluated within the context of other streams and wetlands in that watershed. For example, the Science Report finds, "[t]he amount of nutrients removed by any one stream over multiple years or by all headwater streams in a watershed in a given year can have substantial consequences for downstream waters." Science Report at 1-10. Cumulative effects of streams, wetlands,

and open waters across a watershed must be considered because "[t]he downstream consequences (*e.g.*, the amount and quality of materials that eventually reach a river) are determined by the aggregate effect of contributions and sequential alterations that begin at the source waters and function along continuous flowpaths to the watershed outlet." *Id.* at 1-19.

iii. "In the Region"

For the reasons discussed in section III.E.ii, the agencies believe that science supports analyzing the contributions of similarly situated waters "in the region" for their effect on traditional navigable waters, the territorial seas, or interstate waters because the incremental contributions of individual streams and wetlands are cumulative, and their effect on downstream waters should be evaluated within the context of other streams and wetlands in the appropriate region.

For tributaries and their adjacent wetlands, the agencies consider similarly situated waters to be "in the region" when they lie within the catchment area of the tributary of interest. Identifying the catchment area for purposes of this significant nexus analysis is described below. The agencies developed this updated evaluation method from the current pre-2015 implementation approach informed by their experience, the best available science, Supreme Court decisions, and public comments. Accordingly, in implementing the significant nexus standard under this rule, all tributaries and adjacent wetlands within the catchment area of the tributary of interest will be analyzed as part of the significant nexus analysis.⁴¹ This approach to significant nexus analysis recognizes the ecological relationship between the tributaries and their adjacent wetlands, and the role those similarly situated waters have in influencing the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, and instated waters.

In the case of wetlands adjacent to tributaries, science supports the consideration of the effects of not just a singular adjacent wetland, but any wetland adjacent to the tributary for assessing if there is significant effect on the integrity of downstream waters. Additionally, science supports that similarly situated waters are "in the region" when they lie within the catchment area of the tributary of interest because all the upstream tributaries and their adjacent wetlands in that catchment cumulatively have an impact on larger downstream waters. Because the adjacent wetlands are integrated with the tributary, science also supports the consideration of all wetlands adjacent to that tributary in combination with the tributary itself when conducting a significant nexus analysis.

Using a catchment (*i.e.*, the watershed of the tributary of interest) as the framework for conducting significant nexus evaluations for tributaries and their adjacent wetlands is also scientifically supportable. Watersheds are generally regarded as the most appropriate spatial unit for water resource management. *See, e.g.*, Omernik and Bailey 1997; Montgomery 1999; Winter 2001; Baron *et al.* 2002; Allan 2004; U.S. Environmental Protection Agency 2008; Wigington *et al.* 2013. The watershed framework is consistent with over two decades of practice by EPA and many other governmental, academic, and other entities which recognize that a watershed approach is generally the most effective

⁴¹ This implementation approach to the region for purposes of the significant nexus standard is a change from the *Rapanos* Guidance. *See* section IV.C.9.c of the preamble and section IV.B of this document for additional discussion on implementing the significant nexus analysis.

framework to address water resource challenges. See, e.g., U.S. Environmental Protection Agency 1996; U.S. Environmental Protection Agency 2010. In addition, the Science Report also supports evaluating waters on a watershed scale, concluding, "[c]umulative effects across a watershed must be considered when quantifying the frequency, duration, and magnitude of connectivity, to evaluate the downstream effects of streams and wetlands." Science Report at ES-14 (emphasis added). In addition, the Science Report notes, "[a] river is the time-integrated result of all waters contributing to it, and connectivity is the property that spatially integrates the individual components of the watershed. In discussions of connectivity, the watershed scale is the appropriate context for interpreting technical evidence about individual watershed components." Id. at 2-1 (citing Newbold et al. 1982b; Stanford and Ward 1993; Bunn and Arthington 2002; Power and Dietrich 2002; Benda et al. 2004b; Naiman et al. 2005; Nadeau and Rains 2007; Rodriguez-Iturbe et al. 2009). The Science Report also states that watersheds are integrated at multiple spatial and temporal scales by flows of surface water and ground water, transport and transformation of physical and chemical materials, and movements of organisms. Id. at 6-8. The movement of water from watershed drainage basins to river networks and lakes shapes the development and function of these systems in a way that is critical to their long-term health. See, e.g., Montgomery 1999. Anthropogenic actions and natural events can have widespread effects within the watershed that collectively impact the quality of the traditional navigable water, the territorial seas, or interstate water. Levick et al. 2008. From a water quality management perspective, science supports the evaluation of the effects of tributaries and their adjacent wetlands on a catchment scale as the integrity of downstream traditional navigable waters, the territorial seas, and interstate waters is dependent on the condition of the contributing upstream waters, including streams, lakes, and ponds connected to the tributary network and wetlands adjacent to such waters. The functions of the contributing waters are inextricably linked and have a cumulative effect on the integrity of the downstream traditional navigable water, the territorial sea, or interstate water. Thus, the watershed reflects specific water management objectives and can be scaled up or down as is appropriate to meet that objective or to meet specific policy needs. For purposes of implementing this final rule, the agencies have scaled the watershed to the "catchment level"—meaning, as described above, the catchment of the tributary of interest.

Because waters assessed under paragraph (a)(5) of the final rule will generally be evaluated individually for their effects on traditional navigable waters, the territorial seas, or interstate waters, the agencies have not established what would be considered "in the region" for this such waters, as it is not needed to implement the final rule, as such waters will generally be considered on an individually basis.

Other potential approaches were considered in the Technical Support Document for the Proposed Rule, including ecoregions (Omernik 1987; Omernik 1995; Omernik and Griffith 2014; U.S. EPA 2022j; U.S. EPA 2022k), hydrologic landscape units (Winter 2001; Wolock 2003), and physiographic divisions (Fenneman 1917; Fenneman and Johnson 1946), or a combination of hydrologic landscape regions and physiographic divisions for further refinement of regions (*e.g.*, Blackburn-Lynch *et al.* 2017). These methods could present implementation challenges, whereby the region is too large and obscures the measurable effects of single aquatic resources or is difficult to implement in the field. The agencies solicited comment on what constitutes an appropriate "region" for purposes of analyzing if waters significantly affect the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. Based on a review of the public comments, the best available science, and the agencies' technical expertise and experience, the agencies have determined that the final rule's approach to implementing "in the region" under the significant nexus standard for

tributaries and their adjacent wetlands is clear and implementable. The agencies have determined that the catchment of the tributary is a reasonable and technically appropriate scale. The catchment is an easily identified and scientifically defensible unit for identifying the scope of waters that together may have an effect on the chemical, physical, or biological integrity of a particular traditional navigable water, the territorial seas, or an interstate water.

iv. "Significantly Affect"

Paragraph (c)(6) of the final rule defines the term "significantly affect" for purposes of determining whether a water meets the significant nexus standard to mean "a material influence on the chemical, physical, or biological integrity of" traditional navigable waters, the territorial seas, or interstate waters. Under this rule, waters, including wetlands, are evaluated either alone, or in combination with other similarly situated waters in the region,⁴² based on the functions the evaluated waters perform. The final rule identifies specific functions that will be assessed and identifies specific factors that will be considered when assessing whether the functions provided by the water, alone or in combination, have a material influence on the chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water. The agencies are not requiring the use of "functional assessments" for significant nexus analyses under this rule; *see* section IV.C.9.c of the preamble for further discussion.

The functions in this rule are indicators that are tied to the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters. The functions include contribution of flow; trapping, transformation, filtering, and transport of materials (including nutrients, sediment, and other pollutants); retention and attenuation of floodwaters and runoff; modulation of temperature in traditional navigable waters, the territorial seas, or interstate waters; or provision of habitat and food resources for aquatic species located in traditional navigable waters, the territorial seas, or interstate waters.

The factors in this rule are readily understood criteria that influence the types and strength of chemical, physical, or biological connections and associated effects on those downstream traditional navigable waters, the territorial seas, or interstate waters. These factors include the distance from a paragraph (a)(1) water; hydrologic factors, such as the frequency, duration, magnitude, timing, and rate of hydrologic connections, including shallow subsurface flow; the size, density, or number of waters that have been determined to be similarly situated; landscape position and geomorphology; and climatological variables such as temperature, rainfall, and snowpack.

The definition of "significantly affect" is derived from the objective of the Clean Water Act and is informed by and consistent with Supreme Court case law. It is also informed by the agencies' technical and scientific judgment and supported by the best available science regarding the functions performed by upstream waters relevant to achieving the Clean Water Act's objective. The significant nexus standard established by the final rule is carefully constructed to fall within the bounds of the Clean Water Act. Not all waters subject to evaluation under the significant nexus standard will have a material influence on

⁴² See sections III.E.ii and IV.B.i for discussion of how the agencies' intend to implement "similarly situated" and sections III.E.iii and IV.B.ii for discussion of how the agencies' intend to implement "in the region" as discussed in the final rule's preamble.

traditional navigable waters, the territorial seas, or interstate waters sufficient to be determined jurisdictional. First, the standard is limited to consideration of effects on traditional navigable waters, the territorial seas, and interstate waters. Second, the standard is limited to effects only on the three statutorily identified aspects of those fundamental waters: chemical, physical, or biological integrity. Third, the standard cannot be met by merely speculative or insubstantial effects on those aspects of those traditional navigable waters, the territorial seas, and interstate waters, but rather requires the demonstration of a "material influence." In the final rule, the phrase "material influence" establishes that the agencies will be assessing the influence of the waters either alone or in combination have on the chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water and will provide qualitative and/or quantitative information (based on the factual record, relevant scientific data and information, and available tools) and articulate a reasoned basis for determining that the waters being assessed significantly affect that fundamental downstream water.

Under the significant nexus standard, to be jurisdictional, waters, alone or in combination with other similarly situated waters in the region, must have a material influence on the chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water. The final definition is generally consistent with the pre-2015 regulatory regime. Under the *Rapanos* Guidance, the agencies evaluate whether waters "are likely to have an effect that is more than speculative or insubstantial on the chemical, physical, and biological integrity of a traditional navigable water." *Rapanos* Guidance at 11. Under the final rule, a water may be determined to meet the definition of "waters of the United States" when it "significantly affects" any one form of chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water, consistent with the pre-2015 regulatory regime.

Significant nexus is not purely a scientific determination. Further, the opinions of the Supreme Court have noted that as the agencies charged with interpreting the statute, EPA and the Corps, must develop the outer bounds of the scope of the Clean Water Act, while science does not provide bright lines with respect to where "water ends" for purposes of the Clean Water Act. Therefore, the agencies' interpretation of the Clean Water Act is informed by science, but not dictated by it.

With this context, this section of the Technical Support Document addresses the factors that will be considered in a significant nexus evaluation as well as the relevant scientific functions that streams, wetlands, and open waters perform that will be evaluated to determine whether such waters, either alone or with similarly situated waters in the region, significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

The agencies are adding to the final rule's definition of "significantly affect" a specific list of functions to be assessed when making a significant nexus determination after soliciting comment on whether it would be useful to add such a list to the definition. The *Rapanos* Guidance identified some of the relevant functions upstream waters can provide for downstream traditional navigable waters, the territorial seas, or interstate waters including temperature regulation; sediment trapping and transport; nutrient recycling, retention, and export; pollutant trapping, transformation, filtering, and transport; retention and attenuation of floodwaters and runoff; contribution of flow; provision of habitat for aquatic species that also live in traditional navigable waters, the territorial seas, or interstate waters (*e.g.*, for refuge, feeding, nesting, spawning, or rearing young); and provision and export of food resources for

aquatic species located in traditional navigable waters, the territorial seas, or interstate waters. Evaluation of such functions is consistent with the agencies' implementation of the pre-2015 regulatory regime. *See Rapanos* Guidance at 8, 9.

Under the final rule, a water does not need to perform all of the listed functions. This is consistent with the pre-2015 regulatory regime. *See* U.S. Army Corps of Engineers 2007a. If a water, either alone or in combination with similarly situated waters, performs just one function, and that function has a more than speculative or insubstantial impact on the integrity of a traditional navigable water, the territorial seas, or an interstate water, that water has a significant nexus under the final rule, consistent with the pre-2015 regulatory regime.⁴³ The functions listed in the final rule's definition of "significantly affect" and how they can affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters are discussed below and throughout sections I and III of this document.

Contribution of flow. The contribution of flow downstream is an important role, as upstream waters can be a cumulative source of the majority of the total mean annual flow to larger downstream rivers and waters, including via the recharge of baseflow. Streams, wetlands, and open waters contribute surface and subsurface water downstream, and are the dominant sources of water in most rivers. Contribution of flow can significantly affect the physical integrity of traditional navigable waters, the territorial seas, and interstate waters, helping to sustain the volume of water in these larger waters.

Trapping, transformation, filtering, and transport of materials (including nutrients, sediment, and other pollutants). States identify sediment and nutrients as the primary contaminants in the nation's waters. Sediment storage and export via streams to downstream waters is critical for maintaining the river network, including the formation of channel features. Although sediment is essential to river systems, excess sediment can impair ecological integrity by filling interstitial spaces, reducing channel capacity, blocking sunlight transmission through the water column, and increasing contaminant and nutrient concentrations. Streams and wetlands can prevent excess deposits of sediment downstream and reduce pollutant concentrations in downstream waters. Thus, the function of trapping of excess sediment, along with export of sediment, can have a significant effect on the chemical, physical, and biological integrity of downstream waters.

Nutrient recycling results in the uptake and transformation of large quantities of nitrogen and other nutrients that otherwise would be transported directly downstream, thereby decreasing nutrient loads and associated impairments due to excess nutrients in downstream waters. Streams, wetlands and open waters improve water quality through the assimilation, transformation, or sequestration of pollutants, including excess nutrients and chemical contaminants such as pesticides and metals that can degrade downstream water integrity. Nutrient transport exports nutrients downstream and can degrade water quality and lead to stream impairments. Nutrients are necessary to support aquatic life, but excess nutrients lead to excessive plant growth and hypoxia, in which over-enrichment causes dissolved oxygen concentrations to fall below the level necessary to sustain most aquatic animal life in the downstream

⁴³ See, e.g., Memorandum to Assert Jurisdiction for SPL-2007-261-FBV (December 6, 2007), available at <u>https://usace.contentdm.oclc.org/utils/getfile/collection/p16021coll5/id/1433</u>.

waters. Nutrient recycling, retention, and export can significantly affect downstream chemical integrity by impacting downstream water quality.

Streams, wetlands, and open waters improve water quality through the assimilation and sequestration of pollutants, including excess nutrients and chemical contaminants such as pesticides and metals that can degrade downstream water integrity. In addition to nutrient recycling, removal of nutrients by streambed algal and microbial populations, subsequent feeding by fish and insects, and release by excretion or decomposition, delays the export of nutrients downstream. Riparian, floodplain and non-floodplain wetlands and open waters are active sites for aerobic and anaerobic microbial processes (*e.g.*, denitrification) and physical processes (*e.g.*, settling, photo-degradation) that enhance the assimilation and sequestration of pollutants, thereby limiting their transport to downstream waters. When pollutants reach a wetland, they can be trapped in sediments, assimilated into wetland plants and animals, biogeochemically transformed into less harmful or mobile forms or compounds, or lost to the atmosphere. Importantly, many of the functions performed by non-floodplain wetlands and open waters that affect downstream waters (*e.g.*, surface water storage) result from the disconnections (often hydrological) that create and maintain conditions conducive to the performance of beneficial functions (frequently hydrological and biogeochemical, *e.g.*, mitigation of flood peak flows, sequestration of contaminants).

Retention and attenuation of floodwaters and runoff. Small streams and wetlands are particularly effective at retaining and attenuating floodwaters. By subsequently releasing floodwaters and retaining large volumes of stormwater and runoff (*i.e.*, overland flow) that could otherwise negatively affect the condition or function of downstream waters, streams, wetlands, and open waters can affect the physical integrity of traditional navigable waters, the territorial seas, or interstate waters. This function can reduce flood peaks downstream and can also maintain downstream river baseflows by recharging alluvial aquifers. Runoff occurs when rain, snowmelt, or stormflow exceeds the infiltration rate of soils or storage capacity of soils, streams, wetlands, and open waters. Runoff is an important source of stream and river baseflow. Runoff also carries deposits of sediment and human-made contaminants such as petroleum, pesticides, and fertilizers and is therefore a major source of non-point pollution. Annual runoff generally reflects water surplus and varies widely across the United States. The flowing portions of river networks tend to have their maximum extent during seasons with the highest water surplus, when conditions for flooding are most likely. The Southwest experiences summer monsoonal rains, while the West Coast and Pacific Northwest receive most precipitation during the winter season. Throughout the West, winter precipitation in the mountains occurs as snowfall, where it is stored as seasonal snowpack and is released during the spring and summer melt seasons to sustain streamflow during late spring and summer months. Typically, the occurrence of ephemeral and intermittent streams is greatest in watersheds with low annual runoff and high water-surplus seasonality. Ephemeral tributaries in arid and semi-arid areas convey large volumes of stormflow into alluvial aquifer storage; runoff through ephemeral tributaries that is not stored as groundwater provides baseflow for rivers like the Rio Grande. Precipitation and water surplus in the Eastern United States is less seasonal than in Western states. In temperate climates, headwater tributaries store water and sediment from runoff, reducing the volume and velocity of flows that cause bank erosion, streambed down-cutting, and reduced infiltration to ground water. Riparian, floodplain, and nonfloodplain wetlands and open waters store and subsequently release stormflows, desynchronizing floodwaters and retaining large volumes of runoff (*i.e.*, stormwater, sediment, and contaminants) that could otherwise negatively affect the condition or function of downstream waters.

Modulation of temperature in traditional navigable waters, the territorial seas, or interstate waters. Tributaries can greatly influence water temperatures in tributary networks. See section III.A.i. This is important because water temperature is a critical factor governing the distribution and growth of aquatic life, both directly (through its effects on organisms) and indirectly (through its effects on other physiochemical properties, such as dissolved oxygen and suspended solids). For example, water temperature controls metabolism and level of activity in cold-blooded species like fish, amphibians, and aquatic invertebrates. Temperature can also control the amount of dissolved oxygen in streams, as colder water holds more dissolved oxygen, which fish and other fauna need to breathe. Tributaries provide both cold and warm water refuge habitats that are critical for protecting aquatic life. Because headwater tributaries often depend on groundwater inputs, temperatures in these systems tend to be warmer in the winter (when groundwater is warmer than ambient temperatures) and colder in the summer (when groundwater is colder than ambient temperatures) relative to downstream waters. Thus, tributaries provide organisms with both warm water and coldwater refuges at different times of the year. Id. at 3-42. For example, when temperature conditions in downstream waters are adverse, fish can travel upstream and use tributaries as refuge habitat. Tributaries also help buffer temperatures in downstream waters that are many kilometers away. Adjacent wetlands and open waters located in floodplains also exert substantial controls on water temperature in the downgradient tributary network and ultimately in the traditional navigable water, the territorial seas, or the interstate water.

Provision of habitat and food resources for aquatic species located in traditional navigable waters, the territorial seas, or interstate waters. Streams, wetlands, and open waters provide life-cycle dependent aquatic habitat (such as foraging, feeding, nesting, breeding, spawning, and use as a nursery area) for species located in traditional navigable waters, the territorial seas, or interstate waters. Many species require different habitats for different resources (*e.g.*, food, spawning habitat, overwintering habitat), and thus move throughout the river network over their life cycles. For example, headwater streams can provide refuge habitat under adverse conditions, enabling fish to persist and recolonize downstream areas once conditions have improved. These upstream systems form integral components of downstream food webs, providing nursery habitat for breeding fish and amphibians, colonization opportunities for stream invertebrates, and maturation habitat for stream insects, including for species that are critical to downstream ecosystem function. Streams, wetlands, and open waters can also serve as a refuge for aquatic species also located in traditional navigable waters, the territorial seas, and interstate waters. The provision of life-cycle dependent aquatic habitat for species located in such larger downstream waters significantly affects the biological integrity of those downstream waters.

Streams, wetlands, and open waters supply habitat and food resources for downstream waters, such as dissolved and particulate organic matter (*e.g.*, leaves, wood), which support biological activity throughout the river network. In addition to organic matter, streams, wetlands, and open waters can also export other food resources downstream, such as aquatic insects that are the food source for fish in downstream waters. The export of organic matter and food resources downstream is important to maintaining the food webs and thus the biological integrity of traditional navigable waters, the territorial seas, and interstate waters.

It is also important to note that the agencies' significant nexus standard in the final rule is carefully tailored so that only particular types of functions provided by upstream waters can be considered for their effects on traditional navigable waters, the territorial seas, and interstate waters. Wetlands, streams, and open waters are well-known to provide a wide variety of functions that translate into ecosystem services. *See* section I.H.i. A significant nexus analysis, however, is limited to an assessment of only those functions identified in the final rule that have a nexus to the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. Thus, there are some important functions provided by wetlands, tributaries, and waters evaluated under paragraph (a)(5) that will not be considered by the agencies when making jurisdictional decisions under the final rule.

For example, for purposes of a jurisdictional analysis under the significant nexus standard, the agencies will not be taking into account the carbon sequestration benefits that aquatic resources like wetlands provide. Provision of habitat for non-aquatic species, such as migratory birds, and endemic aquatic species would not be considered as part of a significant analysis under the final rule.⁴⁴ Furthermore, the agencies would not assess soil fertility in terrestrial systems, which is enhanced by processes in stream and wetland soils and non-floodplain wetlands that accumulate sediments, prevent or reduce soil erosion, and retain water on the landscape, benefiting soil quality and productivity in uplands. There are also a wide variety of functions that streams, wetlands, and open waters provide that translate into ecosystem services that benefit society that would not be assessed in a significant nexus analysis under the final rule. These include provision of areas for recreation and personal enjoyment (e.g., fishing, hunting, boating, and birdwatching areas); ceremonial or religious uses; production of fuel, forage, and fibers; extraction of materials (e.g., biofuels, food, such as shellfish, vegetables, seeds, nuts, rice); plants for clothes and other materials; and medical compounds from wetland and aquatic plants or animals. While these types of ecosystem services can contribute to the economy, they are not relevant to the chemical, physical, or biologic integrity of paragraph traditional navigable waters, the territorial seas, and interstate waters and would not be considered in a significant nexus analysis under the final rule.

In evaluating a water individually or in combination with other similarly situated waters for the presence of a significant nexus to a traditional navigable water, the territorial seas, or an interstate water, the agencies will consider factors that influence the types and strength of the chemical, physical, or biological connections and associated effects on those downstream waters. The agencies include in the definition of "significantly affect" the factors to be considered in assessing the strength of the effects: (1) the distance from the traditional navigable water, the territorial seas, or interstate water; (2) hydrologic factors, such as the frequency, duration, magnitude, timing, and rate of hydrologic connections, including shallow subsurface flow; (3) the size, density, or number of waters that have been determined to be similarly situated (and thus can be evaluated in combination); (4) landscape position and geomorphology; and (5) climatological variables such as temperature, rainfall, and snowpack.

These factors influence the strength of the connections and associated effects that streams, wetlands, and open waters have on the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters and are not the functions themselves that the agencies will

⁴⁴ As the agencies have discussed, consideration of biological functions such as provision of habitat is relevant for purposes of significant nexus determinations under the final rule only to the extent that the functions provided by tributaries, adjacent wetlands, and waters assessed under paragraph (a)(5) significantly affect the biological integrity of a traditional navigable water, the territorial seas, or an interstate water. For example, to protect Pacific and Atlantic salmon in traditional navigable waters (and their associated commercial and recreational fishing industries), protections must be provided from the headwater streams where the fish are born and spawn to the marine waters where they spend most of their lives. *See, e.g.*, Science Report at 2-40.

consider as part of the significant nexus standard. These factors also cannot be considered in isolation, but rather must be considered together and in the context of the case-specific analysis. For example, the likelihood of a connection with associated significant effects is generally greater with increasing number and size of the aquatic resource or resources being considered and decreasing distance from the identified traditional navigable water, the territorial seas, or interstate water as well as with increased density of the waters for such waters that can be considered in combination as similarly situated waters. However, the agencies also recognize that in watersheds with fewer aquatic resources, even a small number or low density of similarly situated waters can have disproportionate effects on traditional navigable waters, the territorial seas, or interstate waters. Hydrologic factors include volume (or magnitude), duration, timing, rate, and frequency of flow, size of the watershed or subwatershed, and surface and shallow subsurface hydrologic connections. The presence of a surface or shallow subsurface hydrologic connection, as well as increased frequency, volume, or duration of such connections, can increase the chemical, physical (*i.e.*, hydrologic), or biological impact that a water has on traditional navigable waters, the territorial seas, or interstate waters. In other situations, streams with low duration but a high volume of flow can significantly affect downstream traditional navigable waters, the territorial seas, or interstate waters by transporting large volumes of water, sediment, and woody debris that help maintain the integrity of those larger downstream waters. Science Report at ES-5, ES-8, 6-9. The lack of hydrologic connections can also contribute to the strength of effects for certain functions such as floodwater attenuation or the retention and transformation of pollutants. Id. at 4-26. Climatological factors like temperature, rainfall, and snowpack in a given region can influence the agencies' consideration of the effects of subject waters on downstream traditional navigable waters, the territorial seas, or interstate waters by providing information about expected hydrology and the expected seasonality of connections and associated effects.

The agencies have more than a decade of experience implementing the significant nexus standard by making determinations of whether a water alone or in combination with similarly situated waters in the region significantly affects the chemical, physical, and biological integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. The agencies under the pre-2015 regulatory regime routinely concluded that there was no significant nexus. Based on the agencies' experience, many waters under the final rule will not have a significant nexus to downstream traditional navigable waters, the territorial seas, or interstate waters, and thus will not be jurisdictional under the Act. The agencies also note that the vast majority of resources assessed in approved jurisdictional determinations under the *Rapanos* Guidance were not assessed under the significant nexus standard. Historically, roughly 11% of resources assessed in approved jurisdiction that the number of significant nexus analyses will increase under this rule due to the assessment of paragraph (a)(5) waters, but it is correspondingly expected that the percent of resources found to be jurisdictional under significant nexus analyses will decrease. *See* section IV below for more information on significant nexus determinations.

The scientific record demonstrates that the aquatic functions provided by smaller streams, ponds, wetlands, and other types of waters are important for protecting the chemical, physical, and biological integrity of larger downstream traditional navigable waters, the territorial seas, and interstate waters. For example, states have identified sediment and nutrients as the primary contaminants in the nation's waters. *See, e.g.,* U.S. Environmental Protection Agency 2003; U.S. Environmental Protection Agency 2008; U.S. Environmental Protection Agency 2017. Sediment storage and export via streams to downstream waters is critical for maintaining the river network, including the formation of channel features. Science

Report at 3-13. Although sediment is essential to river systems, excess sediment can impair ecological integrity by filling interstitial spaces (*e.g.*, burying gravel or cobble habitats in streambeds that are essential for the sustainability of aquatic insects and fish spawning sites), reducing channel capacity which can lead to increased flood risk downstream;, blocking sunlight transmission through the water column which impacts fish feeding and schooling practices and can deprive aquatic plants of light needed for photosynthesis, increasing water temperatures which can negatively impact fish and other aquatic species, clogging and irritating the gills of fish, and increasing contaminant and nutrient concentrations. *Id.* (citing Wood and Armitage 1997). Thus, too much sediment in streams can reduce water clarity, harm aquatic species, interfere with the recreational use of streams, and affect drinking water supplies (*see, e.g.*, Mukundan *et al.* 2013). Streams and wetlands can prevent excess deposits of sediment downstream and reduce pollutant concentrations in downstream waters. *Id.* at ES-2 to ES-3; ES-8. Degraded streams can be a source of excess sediment to downstream waters. The function of trapping excess sediment, along with export of sediment, has an effect on the chemical, physical, and biological integrity of downstream waters.

Streams, wetlands, and open waters improve water quality through the assimilation, transformation, or sequestration of pollutants, including excess nutrients and chemical contaminants such as pesticides and metals that can degrade downstream water integrity. *Id.* at ES-2 to ES-3, ES-13. Nutrients, for example, are necessary to support aquatic life, but excess nutrients lead to excessive plant growth and hypoxia or "dead zones," in which over-enrichment causes dissolved oxygen concentrations to fall below the level necessary to sustain most aquatic animal life in the downstream waters. *Id.* at ES-8. Nutrient recycling that occurs in streams, wetlands, and open waters results in the uptake and transformation of large quantities of nitrogen and other nutrients that otherwise would be transported directly downstream, thereby decreasing nutrient loads and associated impairments due to excess nutrients in downstream waters. *Id.* at ES-8. Streams can transport excess nutrients downstream, which can degrade water quality and lead to stream impairments. *Id.* at ES-2. Nutrient recycling, retention, and export thus can affect downstream chemical integrity by impacting downstream water quality.

The contribution of flow downstream is also an important function of streams, wetlands, and open waters, as upstream waters can be a cumulative source of the majority of the total mean annual flow to bigger downstream rivers and waters, including via the recharge of baseflow. *Id.* at ES-8. Streams, wetlands, and open waters contribute surface and subsurface water downstream, and are the dominant sources of water in most rivers. *Id.* at ES-2, ES-7, ES-9, ES-11. Contribution of flow can affect the physical integrity of downstream waters, helping to sustain the volume of water in larger waters.

Small streams and wetlands are particularly effective at retaining and attenuating floodwaters, thereby contributing to maintaining the integrity of downstream waters. *Id.* at ES-2, ES-8, ES-9, ES-10. By subsequently and slowly releasing (desynchronizing) floodwaters and retaining large volumes of stormwater that could otherwise negatively affect the condition or function of downstream waters, streams, wetlands, and open waters affect the physical integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. *Id.* at ES-3. This function can reduce flood peaks downstream and can also maintain downstream river baseflows by recharging alluvial aquifers (shallow aquifers near stream channels).

Streams, wetlands, and open waters supply downstream waters with dissolved and particulate organic matter (*e.g.*, leaves, wood), which support biological activity throughout the river network. *Id.* at ES-2, ES-3. In addition to organic matter, streams, wetlands, and open waters can also export other food resources downstream, such as aquatic insects that are the food source for fish in downstream waters. *Id.* The export of organic matter and food resources downstream is important to maintaining the food webs and thus the biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

Streams, wetlands, and open waters provide life-cycle dependent aquatic habitat (such as foraging, feeding, nesting, breeding, spawning, and use as a nursery area) for species that travel to and from traditional navigable waters, the territorial seas, or interstate waters. *Id.* at ES-2, ES-3, ES-8, ES-9, ES-11. Many species require different habitats for different resource needs (*e.g.*, food, spawning habitat, overwintering habitat), and thus move throughout the river network to over their life-cycles. *Id.* at ES-11, 3-38 (citing Schlosser 1991; Fausch *et al.* 2002). For example, headwater streams can provide refuge habitat when adverse conditions exist in the larger waterbodies downstream, enabling fish to persist and recolonize downstream areas once conditions have improved. *Id.* at 3-38 (citing Meyer and Wallace 2001; Meyer *et al.* 2004; Huryn *et al.* 2005). These upstream systems form integral components of downstream food webs, providing nursery habitat for breeding fish and amphibians, transport of aquatic insects and other food resources to downstream communities, colonization opportunities for stream invertebrates, and maturation habitat for stream insects, including for species that are critical to downstream ecosystem function. *Id.* at ES-3. The provision of life-cycle dependent aquatic habitat for species located in traditional navigable waters, the territorial seas, and interstate waters has important affects the biological integrity of those downstream waters.

To be clear, the agencies would consider biological functions for purposes of significant nexus determinations under the final rule only to the extent that the functions provided by tributaries, adjacent wetlands, and waters assessed under paragraph (a)(5) significantly affect the integrity of the downstream traditional navigable waters, the territorial seas, or interstate waters. For example, to protect Pacific and Atlantic salmon in traditional navigable waters (and their associated commercial and recreational fishing industries), headwater streams must be protected because Pacific and Atlantic salmon require both freshwater and marine habitats over their life cycles and therefore migrate along river networks, providing one of the clearest illustrations of biological connectivity. See, e.g., id. at 2-40. Many Pacific salmon species spawn in headwater streams, where their young grow for a year or more before migrating downstream, living their adult life stages in the ocean, and then migrating back upstream to spawn. Id. Even where they do not provide direct habitat for salmon themselves, ephemeral streams can contribute to the habitat needs of salmon by supplying sources of cold water that these species need to survive (*i.e.*, by providing appropriate physical conditions for cold water upwelling to occur at downstream confluences, see, e.g., Ebersole et al. 2015), by transporting sediment that supports fish habitat downstream (Marteau et al. 2020), and by providing and transporting food for juveniles and adults downstream. See, e.g., Science Report. These species thereby create a biological connection along the entire length of the river network and functionally help to maintain the biological integrity of the downstream traditional navigable water. Id. at 2-40. Many other species of anadromous fish like certain species of lamprey; American eels and other species of catadromous fish; and species of freshwater fish like rainbow trout and brook trout also require small headwater streams to carry out life cycle functions.

Streams, wetlands, and open waters can perform multiple functions, including functions that change depending upon the season. *Id.* at 2-24. For example, the same stream can contribute flow when evapotranspiration is low and can retain water when evapotranspiration is high. *Id.* These functions, particularly when considered in aggregate with the functions of similarly situated waters in the region, can significantly affect the chemical, physical, or biological integrity of a traditional navigable water, the territorial seas, or an interstate water. When considering the effect of an individual stream, wetland, or open water, the science supports that all contributions and functions that the water provides should be evaluated cumulatively. *Id.* at 6-10. For example, the same wetland retains sediment, removes excess nutrients, mitigates flooding, and provides habitat for amphibians that also live downstream; if any of these functions is ignored, the overall effect of that wetland would be underestimated. *See, e.g., id.* at ES-7, 6-10. It is important to note, however, that a water or wetland can provide just one function that may significantly affect the chemical, physical or biological integrity of the downstream water.

1. The significant nexus standard allows for consideration of the effects of climate change on water resources consistent with the best available science

There are ways the agencies can consider a changing climate under the significant nexus standard, but only to the extent it is relevant to the evaluation of whether upstream waters significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. For example, a lake that dries up from warming temperatures due to climate change and no longer has a surface hydrologic connection to downstream waters might become non-jurisdictional, whereas another lake that previously had limited surface hydrologic connectivity might have increased hydrologic connectivity with higher precipitation conditions under a changing climate.

In addition, under the significant nexus standard the agencies can consider the functions of streams, wetlands, and open waters that support the resilience of the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters to climate change. For example, as more intense and frequent storms and other shifts in precipitation cause floods to increase in frequency and volume in some areas of the United States, a significant nexus determination can evaluate the strength of the effect of runoff storage in wetlands, open waters, and headwater tributaries in mitigating increased flood risk associated with climate change in downstream traditional navigable waters, the territorial seas, and interstate waters. In addition, as drought leads to decreased baseflows in traditional navigable waters or interstate waters in other areas of the country, the transmission of flows into alluvial or regional aquifer storage through tributaries and wetlands can mitigate for these climate change-related conditions, and those benefits to downstream traditional navigable waters or interstate waters can be assessed as part of a significant nexus analysis. Changes in flow in tributaries caused by climate change will also be relevant to the relatively permanent standard, but that standard may not allow the agencies to take into account the contribution of upstream waters to the resilience of the integrity of downstream waters.

As discussed in section III.E.iv, the agencies believe that there are climate benefits that streams, wetlands, and open waters provide that are not related to restoring or maintaining the integrity of downstream traditional navigable waters, the territorial seas, or interstate waters, such as carbon sequestration. Those functions would not be considered under this rule because they are not directly

related to the chemical, physical, and biological integrity of downstream waters. However, considering a changing climate when conducting jurisdictional decisions by considering on a case-specific basis the functions of aquatic resources that contribute to the resilience of the integrity of downstream traditional navigable waters, the territorial seas, and interstate waters to climate change is consistent with the policy and goals of the Clean Water Act, case law, and the policy goals of this administration as articulated in Executive Order 13990.

F. The Relatively Permanent Standard

The agencies also conclude that federal protection is appropriate where a water meets the relatively permanent standard. Waters that are evaluated under the standard include tributaries, adjacent wetlands, and intrastate lakes and ponds, streams, and wetlands that do not meet the jurisdictional criteria under one of the final rule's other categories. Waters that meet the relatively permanent standard are an example of a subset of waters that will virtually always significantly affect traditional navigable waters, the territorial seas, or interstate waters, and therefore properly fall within the Clean Water Act's scope. As discussed in section IV.A.3.a.ii of the preamble to the final rule, the relatively permanent standard is administratively useful but is insufficient as the sole standard for geographic jurisdiction under the Clean Water Act because by itself it is inconsistent with the Act's text and objective. Protecting only waters that meet the relatively permanent standard also runs counter to the scientific principles underlying protection of water quality. The agencies are thus promulgating an approach to tributaries, adjacent wetlands, and waters evaluated under paragraph (a)(5) of the final rule that includes, but that is not limited, to the relatively permanent standard.

Under the final rule, tributaries meet the relatively permanent standard when they are relatively permanent, standing or continuously flowing bodies of water. This category includes surface waters that have flowing or standing water year-round or continuously during certain times of the year and more than in direct response to precipitation. The agencies have decided to implement this approach for tributaries under the relatively permanent standard because it is consistent with the *Rapanos* plurality opinion, it reflects and accommodates regional differences in hydrology and water management, and it can be implemented using available, easily accessible tools, as discussed further in section IV.A.ii.1.

The agencies decided not to establish a minimum flow duration for the relatively permanent standard for tributaries because flow duration varies extensively by region. Establishing a uniform number equally applicable to the deserts in the arid West, the Great Lakes region, and New England forests would not be scientifically sound. The agencies instead have chosen to establish a more flexible approach to implementing this rule that accounts for specific conditions in each region. Moreover, it would often be infeasible for the regulated community or agency staff to determine whether a stream ordinarily flows or whether a lake contains standing water, for example, 12 weeks as opposed to 11 weeks per year. Even if this determination was possible, such a bright line cutoff would not reflect hydrological diversity among different regions and alterations in flow regimes. The agencies' conclusion that a minimum duration is not feasible is consistent with the pre-2015 regulatory regime, which did not establish a bright line cutoff (though provided three months as an example of seasonal flow)⁴⁵ and with

⁴⁵ See, e.g., Memorandum to Assert Jurisdiction for NWP-2007-945 (January 23, 2008), *available at* <u>https://usace.contentdm.oclc.org/utils/getfile/collection/p16021coll5/id/1437</u> (joint EPA-Corps memorandum in

the approach of the 2020 NWPR. See 85 FR 22292.

Implementation of relatively permanent tributaries in this rule does not require that relatively permanent flow come from particular sources. This rule's approach is consistent with the plurality opinion in *Rapanos*, which lays out the relatively permanent standard and does not require that relatively permanent waters originate from any particular source. *See, e.g.*, 547 U.S. at 739. This rule's approach is also science-based, as the source of a tributary's flow does not influence its effect on downstream waters, including traditional navigable waters, the territorial seas, or interstate waters. This rule's approach is similar to the familiar approach taken in the *Rapanos* Guidance and the 2020 NWPR, which also did not specify that relatively permanent flow come from particular sources.

Sources of flow in relatively permanent tributaries may include an elevated groundwater table that provides baseflow to a channel bed. Relatively permanent flow could also result from upstream contributions of flow, effluent flow, or snowpack that melts slowly over time in certain geographic regions or at high elevations. In addition, in certain regions relatively permanent flow could result from a concentrated period of back-to-back precipitation events that leads to sustained flow through a combination of runoff and upstream contributions of flow or an elevated groundwater table that provides baseflow to the channel bed. In contrast, non-relatively permanent tributaries may flow only during or shortly after individual precipitation events (including rainfall or snowfall events). Non-relatively permanent flow may occur simply because it is raining or has very recently rained, or because a recent snow has melted.

Streamflow that occurs during the monsoon season in certain parts of the country (typically June through September in the arid West) may be relatively permanent or non-relatively permanent, depending on the conditions at the location. Many tributaries in the arid West are dominated by coarse, alluvial sediments and exhibit high transmission losses, resulting in streams that often dry rapidly following a storm event (*e.g.*, within minutes, hours, or days). These streams are not relatively permanent under this rule. However, relatively permanent flow may occur as a result of multiple back-to-back storm events throughout a watershed, during which the combination of runoff and upstream contributions of flow is high enough to exceed rates of transmission loss for an extended period of time. Relatively permanent flow may also follow one or more larger storm events, when floodwaters locally recharge the riparian aquifer through bank infiltration, which supplies sustained baseflow throughout the monsoon season.

As discussed in section III.A.vi, tributaries encompass lakes, ponds, and impoundments that are part of the tributary network, as such waters outlet to the tributary network and contribute flow downstream at the outlet point. In addition, "flowing water" under the final rule is meant to encompass those tributaries that are frozen for parts of the year. Such tributaries typically have flowing water underneath the frozen surface. Although some streams may appear to have standing water without a current, particularly at certain times of the year, such waters do contribute flow downstream, either slowly or episodically. The phrase "standing water" is intended to describe waters assessed under paragraph (a)(5) that are lentic or "still" systems, such as lakes, ponds, and impoundments, which are characterized

which the agencies found that two months of continuous flow, for example, is considered "seasonal" flow in certain regions of the country and could be sufficient to support a relatively permanent designation under the pre-2015 regulatory regime).

by standing water.

The phrase "certain times of the year" is intended to include extended periods of standing or continuously flowing water occurring in the same geographic feature year after year, except in times of drought. The defining characteristic of relatively permanent waters with flowing or standing water continuously during only certain times of the year is a temporary lack of surface flow, which may lead to isolated pools or dry channels during certain periods of the year. The phrase "direct response to precipitation" is intended to distinguish between episodic periods of flow associated with discrete precipitation events versus continuous flow for extended periods of time.

Tributaries under the final rule that are "relatively permanent" include those that are "relatively permanent" under the Rapanos Guidance, though the standard under the rule encompasses additional tributaries. Thus, tributaries that meet the relatively permanent standard under the final rule include those that flow year-round or at least seasonally (e.g., typically three months). See Rapanos Guidance at 6-7. While relatively permanent flow may occur seasonally, the phrase is also intended to encompass tributaries in which extended periods of standing or continuously flowing water are not linked to naturally recurring annual or seasonal cycles. Specifically, relatively permanent waters may include tributaries in which flow is driven more by various water management regimes and practices, such as tributaries with extensive flow alteration (e.g., diversions, bypass channels, water transfers) and effluent-dependent streams. For example, in areas of the West where water withdrawals or groundwater pumping can substantially modify flow characteristics, onset and cessation of streamflow in some tributaries may be more closely tied to changes in water use associated with irrigation than with seasons of the year. See, e.g., Kustu et al. 2010; Science Report at B-39; Vogl and Lopes 2009. In such flow-altered tributaries, streamflow may change abruptly throughout the year due to adjustments in facility operations or may vary from year to year due to changes in water rights or water management regimes. In addition, tributaries that typically flow throughout the spring may run dry in years following a drought while storage reservoirs are being refilled. When evaluating these types of artificially manipulated regimes, the agencies may consider information about the regular manipulation schedule and may potentially consider other remote resources or on-site information to assess flow frequency.

Under the relatively permanent standard for adjacent wetlands, wetlands meet the continuous surface connection requirement if they physically abut, or touch, a relatively permanent paragraph (a)(2) impoundment or a jurisdictional tributary when the jurisdictional tributary meets the relatively permanent standard, or if the wetlands are connected to these waters by a discrete feature like a non-jurisdictional ditch, swale, pipe, or culvert. A natural berm, bank, dune, or similar natural landform between an adjacent wetland and a relatively permanent water does not sever a continuous surface connection to the extent it provides evidence of a continuous surface connection. Wetlands that have a continuous surface connection to a relatively permanent paragraph (a)(2) impoundment or a jurisdictional tributary when the jurisdictional tributary meets the relatively permanent standard are a subset of adjacent wetlands. Wetlands that do not have a continuous surface connection but are adjacent to paragraph (a)(2) impoundments or jurisdictional tributaries will be evaluated for jurisdiction under the significant nexus standard.

Under the relatively permanent standard for adjacent wetlands and waters evaluated under paragraph (a)(5), the continuous surface connection is a physical connection that does not have to be

hydrologic—for example, a pond with relatively permanent standing water that touches a tributary of a traditional navigable water that meets the relatively permanent standard but that does not contribute flow downstream and thus is not a tributary could be considered under the standard. The continuous surface connection could include a confined surface connection like a swale or non-jurisdictional ditch that connects a pond with relatively permanent standing water to tributary that meets the relatively permanent standard. As discussed in section III.B.ii.3.a, waters with a visible (*e.g.*, channelized) surface-water connection to the river network clearly have effects on downstream waters. Science Report at ES-3. In addition, a natural berm, bank, dune, or similar natural landform between an adjacent wetland and a relatively permanent water does not sever a continuous surface connection to the extent it provides evidence of a continuous surface connection.

Waters that meet the relatively permanent standard are within the scope of the Clean Water Act because scientific evidence supports the conclusion that tributaries of traditional navigable waters, the territorial seas, and interstate waters with relatively permanent, standing, or continuously flowing water perform important functions that either individually, or cumulatively with similarly situated waters in the region, have a material influence on the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. See section III.A. For example, tributaries that meet the relatively permanent standard contribute consistent flow to downstream traditional navigable waters, the territorial seas, or interstate waters, and with that flow export nutrients, sediment, and food resources, contaminants, and other materials that can both positively (e.g., bycontributing to downstream baseflow, providing food for aquatic species, contributing to downstream aquatic habitat) and negatively (e.g., if exporting too much sediment, runoff, or nutrients or if exporting pollutants) affect the integrity, including the water quality, of those larger downstream waters. In addition, wetlands with a continuous surface connection to such relatively permanent waters can attenuate floodwaters, trap sediment, and process and transform nutrients that might otherwise reach downstream traditional navigable waters, the territorial seas, or interstate waters. The relatively permanent standard is useful because it generally requires less information gathering and assessment and because it focuses on flow and includes wetlands with a continuous surface connection. As such, while both the significant nexus and relatively permanent standards require case-specific, fact-specific inquiries before determining whether a water is meets the definition of "waters of the United States," the relatively permanent standard will generally require less assessment.

Standing alone as the sole test for Clean Water Act jurisdiction, the relatively permanent standard is insufficient. The standard's apparent exclusion of major categories of waters from the protections of the Clean Water Act, specifically with respect to tributaries that are not relatively permanent (such as ephemeral streams) and adjacent wetlands that do not have a continuous surface water connection to other jurisdictional waters, is inconsistent with the Act's text and objective and runs counter to the science demonstrating how such waters can affect the integrity of downstream waters, including traditional navigable waters, the territorial seas, and interstate waters. The 2020 NWPR, for example, excluded federal jurisdiction over the many ephemeral tributaries that regularly and directly provide sources of freshwater to the sparse traditional navigable waters in the arid Southwest, such as portions of the Gila River.

As discussed in sections III.A.v and III.B, there is overwhelming scientific information demonstrating the effects ephemeral streams can have on downstream waters and the effects wetlands can

have on downstream waters when they do not have a continuous surface connection. The science is clear that aggregate effects of ephemeral streams "can have substantial consequences on the integrity of the downstream waters" and that the evidence of such downstream effects is "strong and compelling." Science Report at 6-10, 6-13. The SAB explained that ephemeral streams "are no less important to the integrity of the downgradient waters" than perennial or intermittent streams. SAB 2014c at 22-23, 54 fig. 3.

The science is also clear that wetlands may significantly affect downstream waters when they have other types of surface connections, such as wetlands that overflow and flood jurisdictional waters or wetlands with less frequent surface water connections due to long-term drought; wetlands with shallow subsurface connections to other protected waters; or other wetlands proximate to jurisdictional waters. Such wetlands provide a number of functions, including water storage that can help reduce downstream flooding, recharging groundwater that contributes to baseflow of downstream rivers, improving water quality through processes that remove, store, or transform pollutants such as nitrogen, phosphorus, and metals, and serving as unique and important habitats including for aquatic species that also utilize larger downstream waters. *See, e.g.*, Science Report at 4-20 to 4-38. For example, adjacent, interdunal wetlands separated from the Atlantic Ocean only by beach dunes would not meet the relatively permanent standard, but provide numerous functions, including floodwater storage and attenuation, storage and transformation of sediments and pollutants, and important habitat for species that utilize both the wetlands and the ocean, that significantly affect the Atlantic Ocean (both a traditional navigable water and territorial sea).

The relatively permanent standard includes waters that will virtually always significantly affect traditional navigable waters, the territorial seas, or interstate waters, and thus waters that meet the standard properly fall within the Act's protections. As a result, the final rule's incorporation of jurisdictional limitations based upon the relatively permanent standard and the significant nexus standard reflect the agencies' careful consideration of the science and their technical expertise and experience. Waters that meet the relatively permanent standard under the final rule include certain tributaries, certain adjacent wetlands, and certain intrastate lakes and ponds assessed under paragraph (a)(5).⁴⁶ The many functions that such waters provide that can significantly affect the chemical, physical, and biological integrity of traditional navigable waters, the territorial seas, and interstate waters are discussed in more detail in sections III.A, III.B, and III.D above. For waters that meet the relatively permanent standard, their impacts on traditional navigable waters, the territorial seas, or interstate waters are facilitated by the surface connections that have to such paragraph (a)(1) waters. For example, tributaries that meet the relatively permanent standard have flowing or standing water year-round or continuously during certain times of the year and that flow directly or indirectly through another water or waters to a paragraph (a)(1) water or to a paragraph (a)(2)impoundment. Scientific evidence supports the conclusion that such tributaries perform important functions that either individually, or cumulatively with similarly situated waters in the region, have significant effects on the chemical, physical, or biological integrity of paragraph (a)(1) waters. See section III.A above. The same is true of adjacent wetlands and relatively permanent open waters with continuous surface connections to tributaries that meet the relatively permanent standard. See sections III.B and III.D. Tributaries that meet the relatively permanent standard contribute consistent flow to paragraph (a)(1)

 $^{^{46}}$ Streams that meet the relatively permanent standard under the final rule would be considered tributaries. Similarly, wetlands that meet the relatively permanent standard under the final rule would be considered adjacent wetlands. Thus, under paragraph (a)(5), waters that meet the relatively standard under the final rule will typically be lakes and ponds.

waters and, with that flow, export nutrients, sediment, food resources, contaminants, and other materials that can both positively (e.g., by contributing to downstream baseflow, providing food for aquatic species, and contributing to downstream aquatic habitat) and negatively (e.g., by exporting too much sediment, runoff, or nutrients or exporting pollutants) affect the integrity of those paragraph (a)(1) waters. In such cases, the frequency and duration of hydrologic connections help contribute to the influence that such tributaries have on paragraph (a)(1) waters, either alone or in combination with similarly situated waters in the region. In addition, wetlands with a continuous surface connection to tributaries that meet the relatively permanent standard can and do attenuate floodwaters, trap sediment, and process and transform nutrients that might otherwise reach traditional navigable waters, the territorial seas, or interstate waters. Similarly, lakes and ponds that meet the relatively permanent standard and that are assessed under paragraph (a)(5), such as oxbow lakes and ponds, are often physically proximate to a traditional navigable water, the territorial seas, an interstate water, or a relatively permanent tributary and have a continuous surface connection to such waters. The proximity and the continuous surface connection impact the frequency and nature of the exchange of materials that takes place between the lake or pond and the jurisdictional water. Intrastate lakes and ponds that meet the relatively permanent standard process and trap nutrients in runoff that might otherwise reach a paragraph (a)(1) water, provide critical habitat needed for feeding and rearing for semi-aquatic and aquatic species, contribute flow downstream, and serve other functions that are similar to adjacent wetlands. If the agencies assessed waters that meet the relatively permanent standard (e.g., tributaries that meet the relatively permanent standard or adjacent wetlands with a continuous surface connection to such tributaries) they would virtually always find evidence of strong factors, particularly hydrologic factors like flow frequency and duration, that lead to strong connections and associated effects on paragraph (a)(1) waters. Therefore, the agencies have concluded that waters that meet the relatively permanent standard will virtually always meet the significant nexus standard.

IV. Implementation of the Final Rule

The preamble to the final rule contains information on implementing the final rule. This Technical Support Document provides additional citations.

A. Resources for Making Jurisdictional Determinations

The final rule preamble provides implementation guidance informed by sound science, implementation tools, and other resources, drawing on more than a decade of post-*Rapanos* implementation experience. The following discussion is provided to clarify how available data, tools, and methods inform the agencies' determinations and to confirm that interested parties may use these same resources to inform their own siting decisions, if so desired.

Many field-based and remote tools and sources of data are available to determine Clean Water Act jurisdiction under the final rule. In some cases, a property owner may be able to determine whether a property includes a jurisdictional water based on observation or experience. In other cases, a property owner may seek assistance from a consultant to assess the jurisdictional status of features on their property. Property owners may also seek a jurisdictional determination from the Corps, which provides jurisdictional determinations as a public service. When conducting a jurisdictional determination, the Corps will review any documentation that a property owner, or consultant, provides to assist in making a jurisdictional determination. EPA staff also regularly assess the jurisdictional status of waters in implementing Clean Water Act programs. The agencies expect that EPA and Corps staff, as well as private consultants, would be the primary users of the tools and sources of remote data described below, and they have ample experience in using them from prior regulatory regimes.

The agencies utilize many tools and many sources of information to help support decisions on jurisdiction, including USGS and State and local topographic maps, geospatial datasets, aerial photography, satellite imagery, gage data, soil surveys, NWI maps, floodplain maps, watershed studies, modeling tools, scientific literature and references, and field work. As discussed further in section IV.G of the preamble for the final rule and section IV.A.i.3 of this document, these tools have undergone important technological advances, and have become increasingly available, since the *Rapanos* decision. For example, USGS, Tribal, State, and local stream maps and datasets, aerial photography, gage data, watershed assessments, monitoring data, and field observations are often used to help assess the flow contributions of tributaries, including intermittent and ephemeral streams, to downstream traditional navigable waters, the territorial seas, or interstate waters. Similarly, floodplain and topographic maps from federal, State, and local agencies, modeling tools, and field observations can be used to assess how wetlands are storing floodwaters that might otherwise affect the integrity of traditional navigable waters, the territorial seas, or interstate waters. Further, the agencies utilize the large body of scientific literature regarding the functions of tributaries, including tributaries with ephemeral, intermittent, and perennial flow, and of wetlands and open waters to inform their significant nexus analyses. In addition, the agencies have experience and expertise from decades of making decisions on jurisdiction that considered hydrology, ordinary high water mark and its associated indicators (see section IV.C.8.d of the preamble), biota, and other technical factors in implementing Clean Water Act programs. The agencies' immersion in the science, along with the practical expertise developed over more than decade of case-specific determinations across the country, have helped the agencies determine which waters have a significant nexus and where to draw boundaries demarking the "waters of the United States."

The resources covered in this section include tools for identifying tributaries, including tributaries that meet the relatively permanent standard; tools for identifying wetlands, including wetlands adjacent to traditional navigable waters, the territorial seas, interstate waters, impoundments, or tributaries; tools for identifying impoundments; and tools for applying a significant nexus standard. This section presents a non-exclusive list of tools that the agencies have used in the past and will continue to use to assist in making jurisdictional decisions, but other tools could also be used to determine jurisdiction. The agencies have also identified a number of recent advancements in the data, tools, and methods that can be used to make jurisdictional decisions (section IV.A.i.3).

i. Available Tools

In this section, tools and resources that can be used to support jurisdictional decisions are discussed more generally, with more detail on they can be used in implementation for different categories of the final rule in the sections that follow. Additional tools are also discussed in those sections.

1. Mapping and Remote Sensing

Multiple federal agencies provide data, maps, web-based viewers and tools that can help implement this rule. These include, but are not limited to, USGS, U.S. FWS, NRCS, NOAA, Federal Emergency Management Agency (FEMA), EPA, and the Corps.

The USGS provides publicly and freely available historic and recent topographic maps, aerial photography, the National Hydrography Dataset (NHD), and other data and applications which depict and classify many features relevant to identifying "waters of the United States." One of the most commonly used geospatial datasets from the USGS is the NHD, which was created to assist scientists in modeling hydrologic features and for cartographic mapping purposes. Simley 2018. The NHD depicts aquatic resources such as lakes, ponds, streams, rivers, wetlands, and oceans throughout the United States (including many canals and ditches). *Id.* NHD High Resolution is at the 1:24,000 scale⁴⁷ or higher. In Alaska, the NHD is available at the 1:63,360 scale. Stream and river "flowlines" in NHD are characterized as "ephemeral," "intermittent," or "perennial." This hydrographic categorization was initially based on the original pre-digital mapping effort of USGS topographic maps, with periodic updates from the USGS and data stewards. In NHD, perennial reaches are presumed to carry water throughout the year except during drought, whereas intermittent reaches are assumed to lack flow for some duration.⁴⁸ The NHD defines ephemeral as having water only during or after, a local rainstorm or heavy snowmelt, although the NHD did not start classifying some streams in the digital dataset as "ephemeral" until the 2000s. Simley 2006; Simley 2015; Dewald 2017. Although many ephemeral streams are not mapped, those that are mapped are primarily mapped in NHD at high resolution. That said, even in the high-resolution dataset, many ephemeral streams are included in the "intermittent" category, particularly those outside of the arid West. Many, but not all, canals and ditches, lakes and ponds, wetlands, and reservoirs are also mapped in the NHD. The high-resolution dataset is currently the most up-to-date and detailed hydrography dataset for the nation, mapping more streams and other aquatic resources than the medium resolution dataset.

In 2006, USGS and EPA developed the first medium-resolution version of the NHDPlus to support modeling the occurrence of water and to provide the ability to connect detailed information from the surrounding landscape to the stream network. Buto and Anderson 2020. The NHDPlus is a suite of geospatial products that that build upon and extend the capabilities of the NHD, the National Elevation Dataset, and the Watershed Boundary Dataset. The NHDPlus includes a stream network, catchments, and streamflow estimates, as well as other attributes that enable stream "navigation" (*e.g.*, allow users to "navigate" up- and downstream from a given point in the stream network).⁴⁹ An NHDPlus catchment is

https://nhd.usgs.gov/userGuide/Robohelpfiles/NHD_User_Guide/Feature_Catalog/NHD_Feature_Catalog.htm.

⁴⁷ Scale is the relationship between distance on the map and distance on the ground. If the scale were 1:24,000, for instance, then one inch on the map would represent 24,000 inches or 2,000 feet on the ground. If the scale were 1:63,360, then one inch on the map would represent 63,360 inches or one mile on the ground. *See* USGS 1992.
⁴⁸ Definitions of terms used in the NHD and additional information on NHD features are available in the National Hydrography Dataset Feature Catalog, available at

⁴⁹ "Navigate" and "navigation" in this context refer to the ability to trace a stream network upstream and downstream using GIS. The terms do not refer to actual navigability of a water and do not imply that a feature is or is not navigable.

the land surface area that flows directly to an NHDPlus feature (*e.g.*, a single stream (reach)) in the NHD stream network. Moore *et al.* 2019. For most features, the catchment represents the incremental area that drains directly to each feature or stream segment. The medium resolution NHDPlus is currently available for the conterminous United States, Hawaii, and the U.S. territories at

https://www.epa.gov/waterdata/nhdplus-national-hydrography-dataset-plus. USGS has also developed the NHDPlus High Resolution (NHDPlus HR) in an iterative fashion throughout the country using updated, high-resolution datasets (high-resolution NHD, Watershed Boundary Dataset (WBD) data, and 3D Elevation Program (3DEP) 10-meter digital elevation model (DEM)) to create a modern, scalable, and openly accessible hydrography framework for the inland waters of the nation. Id. Data are available for download and as web-based map services at https://www.usgs.gov/national-hydrography/access-nationalhydrography-products. An initial production of NHDPlus HR has been created for the conterminous United States, Hawaii, and several U.S. territories including Puerto Rico, Guam, and American Samoa. USGS 2021. USGS is in the process of creating a second or updated production in those areas. As part of a longer-term project to update the NHD and WBD in Alaska, NHDPlus HR is currently in production in that state, with some areas in Alaska are currently available. Id. NHDPlus HR contains a set of valueadded attributes, in addition to the standard NHD attributes, that enhance stream network navigation, analysis and display, including (a) catchment characteristics, including mean annual precipitation, mean annual temperature, and mean annual runoff, and mean latitude; (b) cumulative drainage area characteristics; and (c) mean annual flow (1971-2000) and velocity estimates for each flowline in the stream network. Id.; USGS d. NHDPlus HR will provide a consistent modeling framework to enable a better understanding of the water quality and contaminant transport in the nation's streams by models such as SPARROW (SPAtially Referenced Regression On Watershed attributes). Buto and Anderson 2020. NHDPlus HR will also provide the hydrography base for the National Water Model, which simulates streamflow volume and velocity over the entire continental United States to help forecasters predict when and where flooding can be expected. Id.; see also section IV.D.iii.2 for a discussion of models. The USGS is also starting to implement the new 3D Hydrography Program (3DHP), which will greatly improve the level of detail, currency, and content of hydrography data by deriving updated stream networks and watersheds from high-quality 3DEP data. USGS a. The USGS believes that 3DHP will better support hydrologic modeling and improve attribution of important hydrologic characteristics like streamflow permanence. Id.

Historic and recent USGS topographic maps can be found at <u>https://www.usgs.gov/the-national-map-data-delivery/topographic-maps</u>. In addition to the traditional topographic maps the USGS also provides map viewers, data download capability, and application tools relevant to implementing this rule. In addition to NHD data, USGS watershed and elevation data can viewed and downloaded via the National Map at <u>https://apps.nationalmap.gov/downloader/#/</u>.

The Watershed Boundary Dataset (WBD) is a seamless, national dataset depicting hydrologic units or watersheds—that is, the area of the landscape that drains to a portion of the stream network. USGS e. In the WBD, a hydrologic unit may represent all or only part of the total drainage area to an outlet point so that multiple hydrologic units may be required to define the entire drainage area at a given outlet. *Id.* Topographic, hydrologic, and other relevant landscape characteristics are used to delineate the hydrologic unit boundaries in the WBD. *Id.* The WBD's hydrologic units (HU) are arranged in a nested, hierarchical system with each HU in the system identified using a unique code called the hydrologic unit code (HUC). *Id.* WBD contains eight levels of progressive hydrologic units identified by unique 2- to 16-

digit codes. *Id.* The dataset is complete for the United States at the HUC-12 level, while the 14- and 16digit hydrologic units are optional and are not complete for the nation. *Id.*

The USGS has several elevation products and programs, including managing the interagency 3D Elevation Program (3DEP). The goal of the 3DEP is to complete nationwide three-dimensional (3D) elevation data acquisition by 2023 in the form of light detection and ranging (LIDAR) data to provide the first-ever national baseline of consistent high-resolution 3D data, including bare earth elevations and 3D point clouds, collected in a timeframe of less than a decade. Lukas and Baez 2021. The 3DEP products and services available through the National Map consist of standard digital elevation models (DEMs) at various horizontal resolutions (*e.g.*, 10 meter, 30 meter), and elevation source (*e.g.*, LIDAR) and associated datasets.

The USGS has several different portals for accessing remote sensing data and geospatioal information. The National Map (https://apps.nationalmap.gov/downloader/#/) includes high resolution aerial images from USDA's National Agriculture Imagery Program (NAIP), high-resolution elevation data from 3DEP, hydrography data from the NHD, wetland data from NWI, topographic maps, and landcover data from the NLCD. In addition to the online mapper, the National Map supports data download and geospatial data services. The EarthExplorer data portal (https://earthexplorer.usgs.gov/) allows the public to obtain earth imagery across available geospatial data types. USGS b. This includes Landsat satellite imagery, Radar data, unmanned aircraft systems (UAS) data, digital line graphs, digital elevation model data, aerial photography, Sentinel satellite data, some commercial satellite imagery including IKONOS and OrbView3, land cover data, digital map data from the National Map, and many other datasets. *Id.* Users can search by exact location via the interactive map or input specific coordinates to view what data types are available. *Id.* Date searches can also be made for specific date ranges. USGS 2012. The Global Visualization Viewer (GloVis) is another tool for accessing remote sensing data (https://glovis.usgs.gov/app).

The U.S. FWS established the National Wetlands Inventory (NWI) to conduct a nationwide inventory of wetlands to provide biologists and others with information on the distribution and type of wetlands to aid in conservation efforts. U.S. FWS a; U.S. FWS 2020. Today, the NWI Wetlands Data Layer is used for general mapping of wetlands⁵⁰ and deepwater habitats and for purposes of data analyses and modeling. *Id.* The NWI Wetlands Data Layer is a mapping dataset that provides detailed information on the extent, characteristics, functions, and distribution of wetlands and deepwater habitats across the United States. *Id.* These data are primarily derived from manual aerial image interpretation. The NWI is available as digital data at the 1:24,000 scale or higher throughout the country, except for portions of Alaska (data in Alaska are at the 1:63,360 scale). The NWI data are available for download (available at https://www.fws.gov/program/national-wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/wetlands-inventory/webmapping-services).

USDA, mainly through the NRCS, produces several products and online mappers relevant to evaluating aquatic resources. NAIP (<u>https://naip-usdaonline.hub.arcgis.com/</u>) is administered through the USDA's Farm Production and Conservation Business Center, and the imagery program acquires aerial

⁵⁰ The NWI uses the Cowardin (1979) definition of "wetlands."

imagery during the agricultural growing seasons in the United States (*i.e.*, "leaf-on" imagery). The imagery is available to the public through the NAIP website and through the National Map. The NRCS produces data and maps which contain information relative to implementing the final rule including, but not limited to, the potential presence of wetlands, the locations of tributaries and details on the properties of soils, including flood frequency and duration, ponding frequency and duration, hydric soils, and drainage class. These NRCS resources can be obtained through

https://websoilsurvey.sc.egov.usda.gov/App/WebSoilSurvey.aspx or via the NRCS Soil Survey Geographic Database (SSURGO) available at https://catalog.data.gov/dataset/soil-survey-geographicdatabase-ssurgo). The NRCS National Water and Climate Center is responsible for producing and disseminating accurate and reliable water supply forecasts and other climatic data. NRCS b. The Center administers the Snow Telemetry (SNOTEL) program which consists of over 900 automated and semiautomated data collection sites across the western United States and tracks snowpack depths, precipitation, and other climate information (available at https://www.wcc.nrcs.usda.gov/snow/). NRCS a. All the data collected at the National Water and Climate Center are placed in a comprehensive database known as the Water and Climate Information System (WCIS), available at https://www.nrcs.usda.gov/wps/portal/wcc/home/aboutUs/monitoringPrograms/wcis/ (see

https://www.nrcs.usda.gov/wps/portal/wcc/home/quicklinks/imap for an interactive map of WCIS data). NRCS b. In addition to the data collected through SNOTEL and manual snow data collection processes, the Center also incorporates precipitation, streamflow, and reservoir data from the Corps, the U.S. Bureau of Reclamation (BOR), the Applied Climate Information System (ACIS), the USGS, various water districts, and other entities into the WCIS database. *Id.* Information in the WCIS database which may be useful in estimating tributary runoff in the areas of the country where data are collected and available.

NOAA publishes a variety of climate information that may be relevant to accessing jurisdiction. The National Centers for Environmental Information, for example, publishes precipitation records (available at <u>https://www.ncdc.noaa.gov/cdo-web/</u>) and the National Weather Service publishes snow analysis maps (available at <u>https://www.nohrsc.noaa.gov/nsa/</u>). This information may be useful for understating tributary runoff frequency and duration.

FEMA produces flood zone or other floodplain maps which can give an estimate of frequency of flooding and other physical floodplain factors (FEMA 2021) that may have relevance in implementing this rule. These maps are publicly available and provide a readily accessible and transparent tool for locating the 100-year floodplain, where FEMA has mapped such information. The FEMA Map Service Center allows the public to search for flood maps by address or location (available at https://msc.fema.gov/portal/home), and the National Flood Hazard Layer Viewer is an interactive web mapping application that displays the current effective flood hazard data (available at https://msc.fema.gov/nfhl).

EPA also has several resources useful for the implementation of this rule. EPA's How's My Waterway (available at <u>https://www.epa.gov/waterdata/hows-my-waterway</u>) maps streams and identifies Clean Water Act section 303(d) listed waters, water quality impairments, and Total Maximum Daily Loads (TMDLs). EPA's NEPAssist (available at <u>https://www.epa.gov/nepa/nepassist</u>), provides locations and information on wastewater discharge facilities and hazardous-waste sites. EPA also publishes ecoregion maps (<u>https://www.epa.gov/eco-research/ecoregions-north-america</u>). The EPA Watershed Assessment, Tracking, and Environmental Results System (WATERS) GeoViewer is an example of a webmapping application that provides accessibility to many spatial dataset layers like NHDPlus and watershed reports for analysis and interpretation (<u>https://www.epa.gov/waterdata/waters-geoviewer</u>). This includes EPA's Stream-Catchment (StreamCat) Dataset, which provides metrics for approximately 2.65 million stream segments and their associated catchments (<u>https://www.epa.gov/national-aquatic-resource-surveys/streamcat-dataset-0</u>). *See* Hill *et al.* 2016. The GeoViewer also allows users to conduct upstream/downstream searches and to delineate watersheds. Although not available in WATERS, the EPA's Lake-Catchment (LakeCat) Dataset is similar to StreamCat, and currently contains over 300 metrics for over 375,000 lakes and their associated catchments. *See* Hill *et al.* 2018. Data are available at <u>https://www.epa.gov/national-aquatic-resource-surveys/lakecat-dataset</u>.

The Corps' Geospatial Platform provides Corps geospatial data, services, and applications for use by partner agencies and the public (<u>https://geospatial-usace.opendata.arcgis.com/</u>). For example, the Corps manages geospatial data related to navigation (*e.g.*, the Navigation Data Center) and reservoirs and lakes owned and operated by the Corps. In addition, the Corps manages the National Inventory of Dams (<u>https://nid.usace.army.mil/#/</u>), which contains information from states and federal agencies on the location of over 91,000 dams across the country. U.S. Army Corps of Engineers 2020a.

In addition to these federal agencies most states have geo-spatial data gateways for viewing and downloading both USGS and potentially other federal agency data but also sometimes state specific aerial photography and geo-spatial data. A web search of a specific state plus "geo-spatial" should provide information on available resources. Some Tribal, Territorial, and local governments may also have geospatial information.

2. Hydrologic Models

Golden et al. (2014) provided a review and critique of modeling approaches to quantify the connectivity and (cumulative) effects of geographically isolated wetlands on other types of waters. They include (listed in Table 1 of the paper) Soil and Water Assessment Tool (SWAT), Hydrologic Simulation Program - FORTRAN (HSPF), DRAINMOD for Watershed (DRAINWAT), TOPMODEL, Grid Based Mercury Model (GBMM), and Visualizing Ecosystems for Land Management Assessment (VELMA)). The authors also explored different groundwater (e.g., MODFLOW Wetlands Package; Table 2 of the paper) and coupled surface-subsurface flow models (e.g., GS-FIOW; Table 3 of the paper) for modeling geographically isolated wetland hydrologic connectivity. Jones et al. (2019) provided an update to modeling connectivity and effects between non-floodplain wetlands and other systems using processbased models (such as SWAT), providing case studies to inform best modeling to guide the model application. Additional models noted by Jones et al. (2019) as applicable include (but are not limited to) MODFLOW (Harbaugh et al. 2005; available at https://www.usgs.gov/mission-areas/waterresources/science/modflow-and-related-programs), the Wetland Hydrologic Capacitance model (Mclaughlin et al. 2014), the USGS model VS2DI (Hsieh et al. 2000; Rossi and Nimmo 1994), and fully distributed coupled surface and subsurface models (Ameli and Creed 2017). Model application examples in Jones et al. (2019) include Wetland Hydrologic Capacitance (e.g., Watts et al. 2015; Jones et al. 2018b), modifications to the SWAT model by Evenson et al. (2016; 2018), and Lee et al. (2018), and application of VS2DI by Neff and Rosenberry (2018) (available at https://www.usgs.gov/software/vs2diversion-13). From the case studies presented, Jones et al. (2019) provided a table summarizing the models, including their fidelity and resource requirements (Table 2 of the paper).

Additional application of hydrologic models and/or advancements to models (e.g., SWAT) include the hydrologic equivalent wetland approach (Golden et al. 2014, citing Wang et al. 2008; Yang et al. 2010; Neitsch et al. 2011), P2P (puddle-to-puddle) modeling (Chu et al. 2013), and a hybrid SWAT/empirical (Spatial Stream Network) modeling approach (Golden et al. 2016). EPA has developed a web-based interactive water quality and quantity modeling system (Hydrologic and Water Quality System, HAWQS, see U.S. EPA 20221) that is being used to assess the cumulative effects of wetlands on downgradient waters (G. Evenson, personal communication). HAWQS employs SWAT as its core modeling engine and is available at https://www.epa.gov/waterdata/hawqs-hydrologic-and-water-qualitysystem. U.S. EPA 20221. Recently, Driscoll et al. (2020) used the USGS's National Hydrologic Model to analyze the surface-depression storage. Rajib et al. (2020) applied a modified SWAT model to assess surface water storage and stream-effects of non-floodplain wetlands across the ~500,000 km² Upper Mississippi River basin, finding that including non-floodplain wetlands into the modeling domain improved the modeling accuracy of stream flows (e.g., the non-floodplain wetland storage effect, once quantified, increased model performance hence demonstrating the substantive effect of non-floodplain wetlands on downgradient systems). Further, Golden et al. (2021a) recently noted that the effects of nonfloodplain wetlands on water storage were substantive and important such that ignoring the systems would lead to erroneous model responses.

Additional approaches to quantifying hydrologic storage include statical models, such as including LIDAR-based topography with precipitation totals. *See, e.g.*, Green *et al.* 2019. Both statistical and process-based models have been used to quantify downgradient nutrient assimilation and mitigation effects of non-floodplain wetlands. Golden *et al.* 2019. These include applying SWAT residence times to assess potential (*e.g.*, Evenson *et al.* 2018), to incorporating non-floodplain wetlands into large basin-scale nutrient analyses (*e.g.*, Evenson *et al.* 2021). Golden *et al.* (2019) found incorporating non-floodplain wetlands into a SWAT watershed model for Iowa resulted in a 7% reduction in nitrate concentrations. Mengistu *et al.* (2020), which looked at how wetlands mediate the amount of nitrogen and phosphorus reaching streams and rivers, is an additional statistical model.

3. Advancements in Implementation Data, Tools, and Methods

Since the *Rapanos* decision, there have been dramatic advancements in the data, tools, and methods used to make jurisdictional determinations, including in the digital availability of information and data. In 2006, when the agencies began to implement the *Rapanos* and *Carabell* decisions, there were fewer implementation tools and support resources to guide staff in jurisdictional decision-making under the relatively permanent and significant nexus standards. Agency staff were forced to rely heavily on information provided in applicant submittals and available aerial imagery to make jurisdictional decisions or to schedule an in-person site visit to review the property themselves. The 2007 U.S. Army Corps of Engineers Jurisdictional Determination Form Instructional Guidebook encouraged practitioners to utilize maps, aerial photography, soil surveys, watershed studies, scientific literature, previous jurisdictional decisions involving significant nexus evaluations, the Guidebook encouraged practitioners to identify and evaluate the functions relevant to the significant nexus by incorporating literature citations and/or references from studies pertinent to the parameters being reviewed. *Id.* For significant nexus decisions specifically, the Guidebook instructed practitioners to consider all available hydrologic information (*e.g.*, gage data,

precipitation records, flood predictions, historic records of water flow, statistical data, personal observations/records, etc.) and physical indicators of flow including the presence and characteristics of a reliable OHWM. *Id*.

The Corps also issued Regulatory Guidance Letter (RGL) No. 07-01⁵¹ in 2007. U.S. Army Corps of Engineers 2007b. RGL No. 07-01 laid out principal considerations for evaluating the significant nexus of a tributary and its adjacent wetlands which included the volume, duration, and frequency of flow of water in the tributary, proximity of the tributary to a traditional navigable water, and functions performed by the tributary and its adjacent wetlands. *Id*. This RGL highlighted wetland delineation data sheets, delineation maps, and aerial photographs as important for adequate information to support all jurisdictional decision-making. *Id*. Gathering the data necessary to support preliminary or approved jurisdictional decisions was often time consuming for staff and the regulated public. There were not many nationally available repositories for much of the information that the agency staff utilized in decision-making, particularly during the first years of implementing of the guidance. Despite these challenges, the agencies and others in the practitioner community gained substantial collective experience implementing the relatively permanent and significant nexus standards from 2006 to 2015.

Since 2015, there have been dramatic improvements to the quantity and quality of water resource information available on the internet. The agencies and other practitioners can use online mapping tools to determine whether waters are connected or sufficiently close to "waters of the United States," and new user interfaces have been developed that make it easier and quicker to access information from a wide variety of sources. Furthermore, some information used to only be available in hard-copy paper files, including water resource inventories and habitat assessments, and many of these resources have been made available online or updated with new information.

The following overview of several tools and data that have been developed or improved since 2015 is intended to demonstrate how case-specific evaluations can be made more quickly and consistently than ever before. Advancements in geographic information systems (GIS) technology and cloud-hosting services have led to an evolution in user interfaces for publicly available datasets frequently used in jurisdictional decision-making such as the NWI, USGS NHD, soil surveys, aerial imagery, and other geospatial analysis tools like USGS StreamStats (https://streamstats.usgs.gov/ss/). Not only are the individual datasets more easily accessible to users, but it has also become much easier for users to quickly integrate these various datasets using desktop or online tools like map viewers to consolidate and evaluate the relevant data in one visual platform. Such map viewers can assist, for example, with considering the factors and assessing the functions in paragraph (c)(6) of the final rule that are relevant to the significant nexus analysis. The EPA WATERS GeoViewer (https://www.epa.gov/waterdata/waters-geoviewer) is an example of a web mapping application that provides accessibility to many spatial dataset layers like NHDPlus and watershed reports for analysis and interpretation. Another web mapping application is the EPA's EnviroAtlas (https://www.epa.gov/enviroatlas/enviroatlas-interactive-map), which provides information and interpretative tools to help facilitate surface water assessments using multiple data layers such as land cover, stream hydrography, soils, and topography. Several States also have state specific interactive online mapping tools called Water Resource Registries (WRRs). Watershed Resources

⁵¹ It should be noted that RGL No. 07-01 was later superseded by RGL 08-02 and RGL 16-01, neither of which addressed significant nexus evaluations.

Registry 2022. WRRs host publicly available GIS data layers providing various information such as the presence of wetlands, land use/cover, impaired waters, and waters of special concern. Other websites like the Corps' Jurisdictional Determinations and Permits Decision site

(https://permits.ops.usace.army.mil/orm-public) and webservices like EPA's Enforcement and Compliance History Online (ECHO) Map Services (https://echo.epa.gov/tools/map-service) allow users to find geospatial and technical information about Clean Water Act section 404 and NPDES permitted discharges. Information on approved jurisdictional determinations finalized by the Corps is also available on the Corps' Jurisdictional Determinations and Permit Decisions site (https://permits.ops.usace.army.mil/orm-public) and EPA's Clean Water Act Approved Jurisdictional Determinations website (https://watersgeo.epa.gov/cwa/CWA-JDs/).

The data that are available online have increased in quality as well as quantity. The NHD has undergone extensive improvements in data availability, reliability, and resolution since 2015, including the release of NHDPlus High Resolution datasets for the conterminous U.S. and Hawaii, with Alaska under development. Buto and Anderson 2020; USGS 2021. One notable improvement in NHD data quality is that the flow-direction network data are much more accurate than in the past. Improvements have also been made to the NWI website and geospatial database, which has served as the primary source of wetland information in the United States for many years. In 2016, NWI developed a more comprehensive dataset (NWI Version 2) that is inclusive of all surface water features in addition to wetlands. U.S. FWS a. This NWI Version 2 dataset provides more complete geospatial data on surface waters and wetlands than has been available in the past and provides a more efficient means to make determinations of flow and water movement in surface water basins and channels, as well as in wetlands. The agencies and other practitioners can use this dataset to help assess potential hydrologic connectivity between waterways and wetlands. For example, it can be used in part to help the agencies identify wetlands that do not meet the definition of "adjacent" (waters assessed under paragraph (a)(5) of the final rule).

The availability of aerial and satellite imagery has improved dramatically since 2015. This imagery is used to observe the presence or absence of flow and identify relatively permanent flow in tributary streams and hydrologic connections to waters. The agencies often use a series of aerial and satellite images, spanning multiple years and taken under normal climatic conditions, to determine the flow characteristics for a tributary, as a first step to determine if additional field-based information is needed to determine the flow characteristics, such as timing and duration of flow. Other practitioners may also use aerial and satellite images to identify aquatic resources and inform assessments of those aquatic resources. The growth of the satellite imagery industry through services such as DigitalGlobe (available at https://discover.maxar.com/) in addition to resources for aerial photography and imagery, such as USGS EarthExplorer (available at https://earthexplorer.usgs.gov/) and National Aeronautics and Space Administration (NASA) Earth Data (available at https://earthdata.nasa.gov/) have reduced the need to perform as many field investigations to verify Clean Water Act jurisdiction. Some of these services charge a fee for use, but others are freely available. The USGS Landsat Level-3 Dynamic Surface Water Extent (DSWE) product (available at https://www.usgs.gov/landsat-missions/landsat-dynamic-surfacewater-extent-science-products?qt-science_support_page_related_con=0#qtscience support page related con) is a specific example of a tool that may be useful for identifying surface water inundation on the landscape in certain geographic areas (e.g., large rivers in the arid

Southwest).⁵²

Similarly, LIDAR data have increased in availability and utility for informing decisions on Clean Water Act jurisdiction. LIDAR produces high-resolution elevation data (<1-3 meter) which can be used to create maps of local topography. The high-resolution maps can highlight the potential hydrologic connections and flowpaths at a site. See, e.g., Lang et al. 2012. Where LIDAR data have been processed to create a bare earth model, detailed depictions of the land surface reveal subtle elevation changes and characteristics of the land surface, including the identification of tributaries. See, e.g., id.; Grau et al. 2021. Hydrologists, for example, have long used digital elevation models of the earth's surface to model watershed dynamics (see, e.g., Tarboton 1997), and the agencies have used such information where available to help inform jurisdictional decisions. LIDAR-derived digital elevation models tend to be high resolution (<1-3 meter), so they are particularly helpful for identifying fine-scale surface features. For example, LIDAR-indicated tributaries can be correlated with aerial photography or other tools to help identify channels and to help determine flow permanence (e.g., relatively permanent flow) in the absence of a field visit. The agencies have been using such remote sensing and desktop tools to assist with identifying jurisdictional tributaries for many years, and such tools are particularly critical where data from the field are unavailable or a field visit is not possible. High-resolution LIDAR data are becoming more widespread for engineering and land use planning purposes. The USGS is in the process of collecting LIDAR data for the entire United States. USGS f. LIDAR data are available for download via the National Map Download Client (available at https://apps.nationalmap.gov/downloader/#/) and LIDAR-derived digital elevation models are available via the 3DEP LidarExplorer (available at https://apps.nationalmap.gov/lidar-explorer/#/). However, LIDAR-derived elevation maps are not always available, so the agencies use other elevation data, including digital elevation models derived from other sources (e.g., 10-meter digital elevation models) and topographic maps to help determine the elevation on a site and to assess the potential location of tributaries.

Since 2015, tools have been developed that automate some of the standard practices the agencies rely on to assist in jurisdictional determinations. One example of this automation is the Antecedent Precipitation Tool (APT), which was released to the public in 2020 and had been used internally by the agencies prior to its public release. The APT is a desktop tool developed by the Corps and is commonly used by the agencies to help determine whether field data collection and other site-specific observations occurred under normal climatic conditions. Gutenson and Deters 2022; Sparrow *et al.* 2022. In addition to providing a standardized methodology to evaluate normal precipitation conditions, ("precipitation normalcy"), the APT can also be used to assess the presence of drought conditions, as well as the approximate dates of the wet and dry seasons for a given location. Gutenson and Deters 2022. As discussed in section IV.B.3 of the final rule preamble and section II.B above, precipitation data are often not useful in providing evidence as to whether a surface water connection exists in a typical year, as required by the 2020 NWPR. *See also* Sparrow *et al.* 2022. However, the agencies have long used the methods employed in the APT to provide evidence that wetland delineations are made under normal circumstances or to account for abnormalities during interpretation of data. *See, e.g.*, Gutenson and Deters 2022; Sprecher and Warne 2000. The development and public release of the APT has accelerated the

⁵² Though DSWE is a useful product for many water-related applications (Jones 2019), due to its resolution (30 meters) and potential issues with canopy cover (*see, e.g.*, Taylor *et al.* 2022), it may be useful only in certain geographic areas for purposes of supporting jurisdictional decisions.

speed at which these analyses are completed; has standardized methods, which reduces errors; and has enabled more people to perform these analyses themselves, including members of the public. Automated tools like the APT will continue to be important for supporting jurisdictional decision-making.

Site visits are still sometimes needed to perform on-site observations of surface hydrology or collect regionally-specific field-based indicators of relatively permanent flow (e.g., the presence of riparian vegetation or certain aquatic macroinvertebrates). The methods and instruments used to collect field data have also improved since 2015, such as the development of rapid, field-based stream duration assessment methods (SDAMs) that use physical and biological indicators to determine the flow duration class of a stream reach. The agencies have previously used existing SDAMs developed by federal and State agencies to identify perennial, intermittent, or ephemeral streams. See Nadeau 2015; Fritz et al. 2020; Mazor et al. 2021. The agencies will continue to use these tools whenever they are determined to be a reliable source of information for the specific water feature of interest. See Nadeau 2015; Fritz et al. 2020; Mazor et al. 2021. The agencies are currently working to develop region-specific SDAMs for nationwide coverage (U.S. EPA 2022m), which will promote consistent implementation across the United States in a manner that accounts for differences between each ecoregion (see, e.g., Fritz et al. 2020). The region-specific SDAMs will be publicly available, with user manuals that will guide not only the agencies, but also other practitioners, in applying the methods to assess aquatic resources. Additional information on the agencies' efforts to develop SDAMs is available on the Regional Streamflow Duration Assessment Methods webpage, available at https://www.epa.gov/streamflow-duration-assessment. Consistent with longstanding practice, the agencies will make decisions based on the best available information.

ii. Identifying Tributaries

Tributaries, including those that are relatively permanent, under the final rule include rivers, streams, lakes, ponds, and impoundments that flow directly or indirectly through another water or waters to a traditional navigable water, the territorial seas, an interstate water, or a paragraph (a)(2) impoundment. A tributary may flow through a number of downstream waters, including non-jurisdictional features. The lateral limits of tributaries, in the absence of adjacent wetlands, is typically delineated by the ordinary high water mark. The final rule at paragraph (c)(4) defines ordinary high water mark, retaining the pre-2015 definition, as "that line on the shore established by the fluctuations of water and indicated by physical characteristics such as clear, natural line impressed on the bank, shelving, changes in the character of soil, destruction of terrestrial vegetation, the presence of litter and debris, or other appropriate means that consider the characteristics of the surrounding areas."

While EPA and Corps field staff must exercise judgment to identify the OHWM on a casespecific basis, the regulations at 33 CFR 328.3(e) and 329.11(a)(1) list the factors to be applied. Regulatory Guidance Letter (RGL) 05-05 further explains these regulations. U.S. Army Corps of Engineers 2005b. Delineation of an OHWM in tributaries relies on identification and interpretation of physical features, including topographic breaks in slope, changes in vegetation characteristics (*e.g.*, destruction of terrestrial vegetation and change in plant community), and changes in sediment characteristics (*e.g.*, sediment sorting and deposition). Gartner *et al.* 2016a (citing U.S. Army Corps of Engineers 2005b; Lichvar and McColley 2008; Mersel and Lichvar 2014). Field indicators, remote sensing, and mapping information can also help identify an OHWM. The Corps continues to improve regulatory practices across the country through ongoing research (*see, e.g.*, Lichvar *et al.* 2009; Gartner *et al.* 2016b; Wohl *et al.* 2016; Hamill and David 2021) and the development of regional and national ordinary high water mark delineation procedures (*see* Lichvar and McColley 2008; Mersel and Lichvar 2014). For example, the Corps has developed field indicators to help field staff identify the OHWM in common stream types in the arid West. Lichvar and McColley 2008. A series of Corps manuals and technical reports aims to clarify procedures and to reduce some of the difficulties in delineating the OHWM in rivers and streams (*e.g.*, Lichvar *et al.* 2006; Lichvar and McColley 2008; Curtis *et al.* 2011; Mersel and Lichvar 2014). These resources focus on the use of field evidence for delineating the OHWM, especially three primary indicators—lateral topographic breaks in slope, changes in vegetation characteristics, and changes in sediment characteristics.

While OHWM delineation is principally a field-based exercise, delineating the OHWM based on visual observations alone can be challenging in some circumstances. In such circumstances, hydraulic modeling, hydrologic modeling and flow frequency analysis, or a combination of the two can be used as an additional line of evidence to help support potential OHWM locations observed in the field. Gartner *et al.* 2016a; Gartner *et al.* 2016b; Gartner *et al.* 2016c. Hydraulic and hydrologic modeling each have advantages and disadvantages. Hydraulic modeling can be used to simulate the water surface elevation and lateral extent of water and other hydraulic parameters at a given location for a given discharge. Gartner *et al.* 2016a. Hydrologic modeling can be used to determine the amount of flow discharge at a given location for a given recurrence interval, and in this use, it is one of several methods used in flow frequency analysis. Gartner *et al.* 2016a; Gartner *et al.* 2016b. The scale of the two different types of modeling also differs. Hydraulic modeling focuses on the reach as a given input and then simulating the hydraulic properties of the water as it flows through a reach. Gartner *et al.* 2016a. Hydrologic modeling the amount of water delivered to a reach as a given input and then simulating the hydraulic properties of the water as it flows through a reach. Gartner *et al.* 2016a. Hydrologic modeling often is conducted on a basin-wide view because the conditions throughout a contributing area can affect the amount of water delivered to the location of interest. *Id.*

Computational hydraulic modeling can be helpful in OHWM delineations but can be misleading if performed or applied improperly. Id. Hydraulic models use a set of algebraic and differential equations based on fundamental physical processes, such as Newton's laws of motion, or well-established empirical relations, such as the Manning equation. Gartner et al. 2016c. The models convert a measured or estimated input (e.g., geometry, discharge rate, and channel roughness) into an output (e.g., water elevation and flow velocity). Id. These models usually make certain assumptions to simplify the calculations, such as assuming that the water flows only directly downstream and that the water surface elevation varies in only the downstream direction, not from one side of a channel to the other. Id. (citing Novak et al. 2010). With a hydraulic model, a user can test if a physical feature or potential OHWM location corresponds with flow levels that are reasonably associated with the OHWM. Gartner et al. 2016a. Gartner et al. (2016a) demonstrated that one-dimensional models, such as the Manning equation (it is well established and relatively simple), the Hydraulic Engineering Center River Analysis System (HEC-RAS) (it is a free and widely used hydraulic modeling program), and HEC-GeoRAS (it enhances visualization of the OHWM in concert with digital elevation models (DEMs) and remotely sensed imagery and allows interpolation of model results between surveyed cross sections), are the most suitable choices if one were to model the OHWM because one-dimensional models have reasonable data requirements and long-standing records of use in simulating water elevations.

The goal of flow frequency analysis is to determine how often flows of various magnitudes occur, for example, the peak flow that occurs once every 5 years on average. Gartner *et al.* 2016b. Each of the three primary ways to determine flood flow frequencies has benefits and limitations: (1) stream-gage analysis, (2) regression equations, and (3) rainfall-runoff modeling. *Id.* Broadly, these are all hydrologic models in that they use a set of inputs (such as stream- or rain-gage records or watershed characteristics) to mathematically compute a desired hydrologic result (in this case, recurrence-interval estimates of high flows). *Id.* Gartner *et al.* (2016b) found that in most cases, regional regression equations are the first choice for estimating flood recurrence intervals for OHWM purposes, unless there is a gage at the site, in which case Log-Pearson Type III analysis may be preferred because of the narrower confidence intervals.

In some circumstances, field observations of OHWM indicators may not be possible due to potential illegal activity (e.g., filling in a stream). In some cases, knowledge of the flow characteristics at a site can assist in OHWM delineation beyond the information provided by field evidence. Gartner et al. 2016b. The OHWM is not associated with a specific streamflow recurrence interval (or flow frequency); however, it is generally associated with streamflow levels well above mean discharge but less than extreme and infrequent flood events. Gartner et al. 2016a. (citing Lichvar and McColley 2008; Mersel and Lichvar 2014). In non-perennial arid streams, for instance, the OHWM signature has been associated with flows generally ranging from about the 1- to 15-year flood event. Id. (citing Curtis et al. 2011). Some streams may have conflicting field indicators or multiple potential OHWM locations based on the available field evidence, particularly in arid and semi-arid stream systems with one or more low-flow channels within a broader active channel. Gartner et al. 2016b. In challenging OHWM delineation situations, knowledge of the streamflow recurrence intervals associated with various field indicators can sometimes be used to rule out or support potential OHWM locations. Id. (citing Gartner et al. 2016a). For example, in a situation in the arid West, conflicting field indicators can be evaluated in the context of modeled recurrence-interval flows; the larger channel corresponds with approximately a 5- to 10- year flood event, which is consistent with the range of recurrence intervals associated with the OHWM in semi-arid systems. Id. (citing Lichvar et al. 2006; Curtis et al. 2011). Thus, in this example, quantitative analysis can effectively be used to rule out the field indicators associated with the smaller channel and to provide supporting evidence for the field indicators associated with the larger channel. Id.

In addition, the agencies will assess any discontinuity in the OHWM and, consistent with pre-2015 practice, a natural or human-made discontinuity in the OHWM does not necessarily sever jurisdiction upstream. A discontinuity may exist where the stream temporarily flows underground. Tributaries may temporarily flow underground in regions with karst geology or topography or lava tubes, for example, often maintaining similar flow characteristics underground and at the downstream point where they return to the surface. *See* section III.A. The agencies will also continue their familiar practice that a discontinuity in the OHWM also does not typically sever jurisdiction upstream where the OHWM has been removed by development, agriculture, or other land uses. For example, tributaries can be relocated below ground to allow reasonable development to occur. In urban areas, surface waters are often rerouted through an artificial tunnel system to facilitate development. *See, e.g.*, Science Report at 3– 3, and section III.A of this Technical Support Document. Underground streams are distinct from groundwater due to their very direct hydrologic connection to the portions of the tributaries that are or resurface above ground. Typically, groundwater connections would be much slower than connections via underground streams. Tributaries that have been rerouted underground are contained within a tunnel system or other similar channelized subsurface feature, while naturally occurring subterranean streams flow within natural conduits like karst formations or lava tubes. *See, e.g.*, Science Report at 3-3; U.S. EPA 2002; Weary and Doctor 2014. The agencies will look for indicators of flow both above and below the discontinuity. For example, a discontinuity in the OHWM may exist due to constructed breaks (*e.g.*, culverts, pipes, or dams)⁵³ or natural breaks (*e.g.*, debris piles or boulder fields). Site specific conditions will continue to determine the distance up the tributary network that is evaluated to see if the feature creates a temporary break or if it severs the upstream connection and constitutes the start of the tributary system.

The agencies will apply the regulations, RGL 05-05, and applicable OHWM delineation manuals and tools and take other steps as needed to ensure that the OHWM identification factors are applied consistently nationwide. *See Rapanos* Guidance at 10-11 n.36.

A variety of field and remote tools can be used to determine whether a water is a tributary. Due to limitations associated with some remote tools, field verification for accuracy may be necessary (*e.g.*, due to scale or vegetation cover, not all tributaries may be visible in aerial photographs, satellite imagery, or mapped in the NHD). Examples of field indicators will be discussed in more detail below.

Tributaries can be identified on the landscape using direct observation or various remote sensing resources such as USGS stream gage data (available at https://waterdata.usgs.gov/nwis/rt), USGS topographic maps (available at https://www.usgs.gov/the-national-map-data-delivery/topographic-maps), high-resolution elevation data (including digital terrain depictions created from LIDAR) and associated derivatives (*e.g.*, slope or curvature metrics), FEMA flood zone maps (available at https://msc.fema.gov/portal/home), NRCS soil maps (available at https://websoilsurvey.sc.egov.usda.gov/App/WebSoilSurvey.aspx), NHD data, NWI data, maps and geospatial datasets from Tribal, State, Territorial, or local governments, and/or aerial or satellite imagery. For example, tributaries are often observable in aerial imagery and high-resolution satellite imagery by their topographic expression, characteristic linear and curvilinear patterns, dark photographic tones, or the presence of riparian vegetation. USGS topographic maps often include different symbols to indicate mapped hydrographic features such as perennial, intermittent, and ephemeral streams and rivers. *See* USGS 2005. Due to limitations associated with some remote tools, field verification for accuracy may be necessary (*e.g.*, due to scale or vegetation cover, not all tributaries may be visible in satellite imagery and aerial photographs or mapped in the NHD).

Determining whether a waterbody is a tributary under the final rule includes identifying whether the waterbody is part of the tributary system of a traditional navigable water, the territorial seas, or an interstate water. The tributary must flow directly or indirectly through another water or waters to a traditional navigable water, the territorial seas, or interstate water, as discussed further in section IV.C.4.c.i of the preamble. Waters through which a tributary may flow indirectly include, for example, impoundments, wetlands, lakes, ponds, and streams. A tributary may flow through a number of downstream waters, including non-jurisdictional features and jurisdictional waters that are not tributaries,

⁵³ Under past practice, the agencies have sometimes characterized bridges as artificial breaks, such as under the 2015 Clean Water Rule. *See* 80 FR 37106. However, bridges do not necessarily create discontinuity in the OHWM, and the agencies recognize that tributaries flowing under bridges may still show evidence of an OHWM and in such circumstances would continue to be jurisdictional where they meet either the relatively permanent or significant nexus standard.

such as an adjacent wetland. But, the tributary must be part of a tributary system that eventually flows to a traditional navigable water, the territorial seas, or an interstate water to be jurisdictional. A tributary may flow through another stream that flows infrequently, and only in direct response to precipitation, and the presence of that stream is sufficient to demonstrate that the tributary flows to a traditional navigable water, the territorial seas, or an interstate water. Tributaries are not required to have a surface flowpath all the way down to the traditional navigable water, the territorial seas, or the interstate water. For example, tributaries can contribute flow through certain natural and artificial breaks (including certain non-jurisdictional features), some of which may involve subsurface flow.

In evaluating the flowpath from a water feature, the agencies can use USGS maps and hydrography data; NWI data; Tribal, State, Territorial, and local knowledge or maps; dye tests, tracers, or other on the ground tests; field observations; aerial photography; or other remote sensing information. The agencies can also use available models, including models developed by federal, Tribal, State, Territorial, and local governments, academia, and the regulated community. One such available GIS tool is the USGS StreamStats "Flow (Raindrop) Path," which allows the user to click a point on a map, after which a flow path is drawn to estimate where water may flow from that point to the stream network, eventually making its way to the ocean if the tributary network allows for it. The StreamStats "Flow (Raindrop) Path" tool may potentially be used to identify the flow path from the subject waters to the downstream traditional navigable water, the territorial seas, or interstate water. The tool is available on USGS's website at https://streamstats.usgs.gov/ss/. Similarly, via WATERS, EPA provides both its webbased mapping application (WATERS GeoViewer, available at https://www.epa.gov/waterdata/watersgeoviewer) and application programming interface (API) services (available at https://watersgeo.epa.gov/openapi/waters/#/Navigation) that can perform downstream network "navigation" to trace the flowpath downstream. See supra note 49. These tools could be used in conjunction with field observations, data, and other desktop tools to evaluate whether a tributary flows directly or indirectly to a traditional navigable water, the territorial seas, or an interstate water. For tributaries to impoundments of "waters of the United States," a flowpath to the impoundment and to a traditional navigable water, the territorial sea, or an interstate water can be identified using these same tools.

The following section discusses tools for identifying tributaries that meet the relatively permanent standard. Many of the tools discussed below can also be used for identifying tributaries more generally.

1. Identifying Tributaries That Meet the Relatively Permanent Standard

Under the final rule, tributaries meet the relatively permanent standard when they are relatively permanent, standing or continuously flowing bodies of water. Under the final rule, this category includes surface waters that have flowing or standing water year-round or continuously during certain times of the year and more than in direct response to precipitation. In contrast, tributaries not meeting the relatively permanent standard may flow only during or shortly after individual precipitation events (including rainfall or snowfall events). In implementing the relatively permanent standard, the agencies draw key concepts from the 2020 NWPR's interpretation, but modify that rule's approach to ensure the term can be practically implemented. The approach in the final rule would encompass tributaries considered relatively permanent under the 2020 NWPR, as well as those considered relatively permanent under the *Rapanos* Guidance, providing continuity in approach for the regulated community and other stakeholders. Under
the final rule, tributaries that do not meet the relatively permanent standard will be evaluated according to the significant nexus standard, as discussed in section IV.A.v of this document and section IV.C.4.c.iii of the final rule preamble.

A key factor that the agencies typically consider when assessing relatively permanent flows⁵⁴ is the geographic region. Many factors, including climate, hydrology, topography, soils, and other conditions, may affect the period in which relatively permanent flow may occur for those relatively permanent waters that do not have continuously flowing or standing water year-round. The factors which affect streamflow and flow cessation are climatically and geographically specific and therefore the periods during which a tributary might have relatively permanent flow vary by region. Non-relatively permanent tributaries are similarly diverse, and the mechanisms which differentiate relatively permanent flow from non-relatively permanent flow also vary by region.

For example, in parts of the Southeastern United States (Southeast), precipitation is distributed somewhat uniformly throughout the year, but increased evapotranspiration during the growing season can reduce surficial ground water levels and reduce or remove surface flows late in the growing season (*e.g.*, late summer or early autumn). *See, e.g.*, Anandhi *et al* 2018; Mulholland *et al.* 1997. Consequently, certain streams in the Southeast may flow primarily in the winter or early spring. Non-relatively permanent tributaries in the Southeast may often be characterized by the repeated sequence of streamflow, flow cessation, and channel drying throughout the year, where the onset of streamflow coincides with distinct rainfall events and is driven primarily by storm runoff. Streamflow in these systems may persist anywhere from a few hours to days at a time, where the cessation of flow is most often associated with termination of overland flow, hillslope runoff recession, and the depletion of water in saturated soils. Although streamflow in these tributaries may occur regularly, off and on, over the duration of a season or longer, they do not exhibit continuously flowing water for an extended period at any point during the year. In other areas, snowpack melt drives streamflow more than rainfall, and relatively permanent flow may therefore coincide with warming temperatures typically in the spring or early summer.

Many headwater streams in mountainous regions flow through channels incised in bedrock with no groundwater interface with the bed of the stream. Instead, these streams are often fed primarily by high elevation snowpack melt. The same scenario may also exist in Northern regions, where flows could be fed almost exclusively through melting snowpack absent elevated groundwater tables. In these regions, relatively permanent flows coincide with warming temperatures in the spring or early summer and may persist well into the summer until there are no longer enough inputs to sustain surface water, or later into autumn when more permanent sources of meltwater (*e.g.*, glaciers or snowfields) begin to freeze. Non-relatively permanent flows in these regions may occur in basins with thin layers of snow, where snow melts rapidly at the onset of spring thaw, and the snowmelt produced is not sufficient to sustain flows for an extended period and into the summer.

Sources of flow in tributaries that meet the relatively permanent standard may include an elevated groundwater table that provides baseflow to a channel bed. Relatively permanent flow could also result

⁵⁴ While "flow" is used throughout this section, it is meant to include circumstances where tributaries such as lakes, ponds, or impoundments have standing water during certain parts of the year, as such waters outlet to the tributary network and contribute flow downstream at the outlet point.

from upstream contributions of flow, effluent flow, or snowpack that melts slowly over time in certain geographic regions or at high elevations. In addition, relatively permanent flow could result from a concentrated period of back-to-back precipitation events in certain regions that leads to sustained flow through a combination of runoff and upstream contributions of flow or an elevated groundwater table that provides baseflow to the channel bed. In contrast, non-relatively permanent tributaries may flow only during or shortly after individual precipitation events (including rainfall or snowfall events). Non-relatively permanent flow may occur simply because it is raining or has very recently rained, or because a recent snow has melted.

Streamflow that occurs during the monsoon season in certain parts of the country (typically June through September in the arid West) may be relatively permanent or non-relatively permanent, depending on the conditions at the location. Many tributaries in the arid West are dominated by coarse, alluvial sediments and exhibit high transmission losses, resulting in streams that often dry rapidly following a storm event (*e.g.*, within minutes, hours, or days). Such streams are not relatively permanent under this rule. However, relatively permanent flow may occur as a result of multiple back-to-back storm events throughout a watershed, during which the combination of runoff and upstream contributions of flow is high enough to exceed rates of transmission loss for an extended period of time. Relatively permanent flow may also follow one or more larger storm events, when floodwaters locally recharge the riparian aquifer through bank infiltration, which supplies sustained baseflow throughout the monsoon season.

Similar to the 2020 NWPR's approach, the agencies will consider tributaries that flow in direct response to "snowfall" for only a short duration during or shortly after that snowfall event to be non-relatively permanent waters under this rule. Streams that flow as a result of "snowpack melt" will be considered relatively permanent waters under this rule, where snowpack is defined as "layers of snow that accumulate over extended periods of time in certain geographic regions or at high elevation (*e.g.*, in northern climes or mountainous regions)." *See* 85 FR at 22275.

As a general matter, the agencies will assess tributaries as they find them, based on current conditions. For example, if a tributary were not a relatively permanent water because of routine water withdrawals, the agencies would assess the tributary under the significant nexus standard. Conversely, if a tributary routinely receives effluent flow that makes it a relatively permanent water, the agencies will assess it under the relatively permanent standard, even if the tributary would otherwise be a non-relatively permanent water without the inputs of effluent.

To determine the flow characteristics of a tributary for purposes of implementing this rule, the agencies will evaluate the entire reach of the tributary that is of the same Strahler stream order (*i.e.*, from the point of confluence, where two lower order streams meet to form the tributary, downstream to the point such tributary enters a higher order stream). Science Report at 2-2; Strahler 1957; *see* Figure 12. The flow characteristics of lakes, ponds, and impoundments that are part of the tributary network will be assessed in conjunction with the stream they connect to. Consistent with the pre-2015 regulatory regime, the agencies will assess the flow characteristics of a particular tributary at the farthest downstream limit of such tributary (*i.e.*, the point the tributary enters a higher order stream). *Rapanos* Guidance at 6, n. 24. Where data indicate the flow characteristics that best characterize the entire tributary reach will be used.



Figure 12: A generalized example of a stream network to illustrate stream order. Blue lines illustrate the river network, within the light green area of its watershed. Numbers represent Strahler stream order, with streams increasing in order when two streams of equal order join. Blue squares indicate channel heads, and orange dots depict confluences. Source: Science Report (Figure 2-1).

Direct observations and various remote tools and resources can be used to identify tributary reaches based on stream order, and topographic characteristics can assist in determining stream order. USGS topographic map blue line symbology and contour line patterns can be used to interpret the connectivity and contribution of flow within a river network, as well as topography within an evaluation area. LIDAR-based elevation models may also illustrate tributary connectivity and flow patterns, as well as topography. In addition, aerial and satellite imagery along with maps or geospatial mapping products (*e.g.*, NHD, NWI, soil maps, and state, tribal, or local maps) can be used to help identify tributary reaches based on stream order. In addition to remote tools and resources, factors identified through field observations can also be used to help determine the extent of a tributary reach. For example, tributary systems can be traversed to identify and characterize the branches of the network that contribute flow to a particular evaluation area. Certain geographic features (*e.g.*, ditches, swales) may also be found to contribute to a tributary's surface hydrology. The agencies also have experience evaluating relatively permanent flow and will continue to use multiple tools, including direct observations and remote and field-based indicators to inform decisions.

Direct observation and various remote or desktop tools can help the agencies and the public better understand streamflow and inform determinations of relatively permanent flow. These tools include local maps, StreamStats by the USGS (available at <u>https://streamstats.usgs.gov/ss/</u>; *see* Ries *et al.* 2017), Probability of Streamflow Permanence (PROSPER) by the USGS, which provides streamflow permanence probabilities during the summer for stream reaches in the Pacific Northwest (available at <u>https://www.usgs.gov/centers/wy-mt-water/science/probability-streamflow-permanence-prosper?qt-science_center_objects=0#qt-science_center_objects</u>), and NRCS hydrologic tools and soil maps.

Other tools include regional desktop tools or models that provide for the hydrologic estimation of a discharge sufficient to generate intermittent or perennial flow (*e.g.*, a regional regression analysis or hydrologic modeling), or modeling tools using drainage area, precipitation data, climate, topography, land use, vegetation cover, geology, and/or other publicly available information. Some models that are developed for use at the reach scale may be localized in their geographic scope. The USGS has developed models for some states for estimating the probability of perennial or non-perennial (*e.g.*, intermittent) flow, including Arizona (Anning and Parker 2009), Idaho (Rea and Skinner 2009), Massachusetts (Bent and Steeves 2006), and Vermont (Olson and Brouillette 2006). Other similar models have been developed for different regions of the United States for classifying different flow classes, including for arid regions of the United States (Merritt *et al.* 2021), California (Lane *et al.* 2017), North Carolina (Russell *et al.* 2015), the northern Rocky Mountains (Sando and Blasch 2015), the upper Colorado River Basin (Reynolds *et al.* 2015). In addition, several models have been developed to classify ephemeral, intermittent, and perennial stream reaches in the Appalachian coal basin of eastern Kentucky. Villines *et al.* 2015; Williamson *et al.* 2015.

Remote or desktop tools can also help the agencies and the public better understand stream flow and whether tributaries have continuously flowing or standing water year-round or during certain times of the year for more than for a short duration and more than in direct response to precipitation. Aerial photographs showing visible water on multiple dates can provide evidence about the presence and duration of continuously flowing or standing water. Aerial photographs may also show other indicators commonly used to identify the presence of an OHWM (*see* definition of OHWM in section IV.A.ii and U.S. Army Corps of Engineers 2020b for additional information on OHWM). Lichvar and McColley 2008; Mershal and Lichvar 2014. These indicators may include the destruction of terrestrial vegetation, the absence of vegetation in a channel, and stream channel morphology with evidence of scour, material sorting, and deposition. These indicators from aerial photographs can be correlated to the presence of USGS stream data (*e.g.*, NHD or topographic maps) to support an assessment of the duration and timing of flow for a tributary.

In addition to aerial photographs, desktop tools, such as a regional regression analysis and the Hydrologic Modeling System (HEC-HMS), provide for the hydrologic estimation of stream discharge in tributaries under regional conditions. The increasing availability of LIDAR derived data can also be used to help implement this final rule. Where LIDAR data have been processed to create elevation data such as a bare earth model, detailed depictions of the land surface are available and subtle elevation changes can indicate a tributary's bed and banks and channel morphology. Visible linear and curvilinear incisions on a bare earth model can help inform the potential duration and timing of flow of a water in greater detail than aerial photography interpretation alone. Several tools (*e.g.*, TauDEM, Whitebox, GeoNet) can assist in developing potential stream networks based on contributing areas, curvature and flowpaths using GIS.

Potential LIDAR-indicated tributaries can be correlated with aerial photography or high-resolution satellite imagery interpretation and USGS stream gage data, to reasonably conclude the presence of an OHWM and shed light on the potential flow characteristics.

Sources of information that can facilitate the evaluation of relatively permanent flow from snowmelt are NOAA national snow analyses maps (available at <u>https://www.wcc.nrcs.usda.gov/snow/</u>), NRCS sources (available at <u>https://www.wcc.nrcs.usda.gov/snow/</u>), or use of hydrographs to indicate a large increase in stream discharge due to the late spring/early summer thaws of melting snow.

Regional field observations can be used to verify desktop assessments of whether a tributary meets the relatively permanent standard, when necessary. Geomorphic indicators could include active/relict floodplain, substrate sorting, clearly defined and continuous bed and banks, depositional bars and benches, and recent alluvial deposits. Hydrologic indicators might include wrack/drift deposits, hydric soils, or water-stained leaves. Biologic indicators could include aquatic mollusks, crayfish, benthic macroinvertebrates, algae, and wetland or submerged aquatic plants. The agencies have been working to develop regionalized streamflow duration assessment methods (SDAMs, see https://www.epa.gov/streamflow-duration-assessment) which are rapid field-based assessment methods that can be used to classify streamflow duration. Nadeau et al. 2015; Fritz et al. 2020; Mazor et al. 2021. These SDAMS can be used to assist in determining whether tributaries at the reach scale meet the relatively permanent standard. These methods rely on physical and/or biological field indicators, such as the presence of hydrophytic vegetation and benthic macroinvertebrates, that can be collected or observed in a single site visit to determine the flow duration of a tributary in a reliable and rapid way. EPA, the Corps, and the State of Oregon developed a regionalized SDAM that has been validated for use throughout the Pacific Northwest. Nadeau 2015. EPA and the Corps have also developed a beta SDAM for the arid West (Mazor et al. 2021a) and the Western Mountains (Mazor et al. 2021b). EPA and the Corps are working to develop additional regionalized SDAMs in other parts of the country (U.S. Environmental Protection Agency 20211). The agencies, co-regulators, and stakeholders can use the regionalized field indicators from SDAMs to quickly and easily identify "relatively permanent" tributaries as interpreted by the agencies under this rule. In addition, several states have developed their own methods for evaluating streamflow classifications, including Ohio (Ohio Environmental Protection Agency 2020), New Mexico (New Mexico 2020), North Carolina (North Carolina Division of Water Quality 2010), Tennessee (Tennessee DEC 2020), and Virginia (Virginia Department of Conservation and Recreation 2010). Some local governments have similarly developed their own methods. See, e.g., Fairfax County 2003; James City County 2009. Such state and local flow duration classifications may also be useful tools. Ultimately, multiple indicators, data points, and sources of information may be used to determine whether a water, including a tributary, is relatively permanent.

iii. Identifying Wetlands

Before determining if a wetland is jurisdictional, the agencies first determine if the wetland in question meets the definition of "wetlands." Wetlands have long been defined by the agencies as "those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include swamps, marshes, bogs, and similar areas."

33 CFR 328.3(b) (1987); 33 CFR 328.3(c)(16) (2021); 40 CFR 230.3(t) (1988); 40 CFR 120.2(3)(xvi) (2021). The final rule does not change the longstanding regulatory definition of "wetlands."

As under prior regimes, wetlands are identified in the field in accordance with Corps' 1987 Wetland Delineation Manual (Corps 1987) and applicable regional delineation manuals. The 1987 Wetland Delineation Manual was regionalized following recommendations from the National Research Council (1995) to improve sensitivity to regional differences in climate, hydrologic and geologic conditions, and other wetland characteristics. Wakeley *et al.* 2002. Between 2002 and 2012, the Corps developed ten regional supplements covering every region of the conterminous United States, as well as Alaska, the Caribbean Islands, and Hawaii and the Pacific Islands. U.S. Army Corps of Engineers c. Each of the regional supplements possesses uniform format and field methods, while accounting for regional variability through application of field indicators of hydrophytic vegetation, hydric soils, and wetland hydrology. Berkowitz 2011.

Field work is often necessary to confirm the presence of a wetland and to accurately delineate its boundaries. However, in addition to field observations on hydrology, vegetation, and soils, remote tools and resources can be used to support the identification of a wetland, including USGS topographic maps, NRCS soil maps and properties of soils including flood frequency and duration, ponding frequency and duration, hydric soils, and drainage class, aerial or high-resolution satellite imagery, high-resolution elevation data, and NWI maps.

Wetland mosaics are landscapes where wetland and non-wetland components are too closely associated to be easily delineated or mapped separately. These areas often have complex microtopography, with repeated small changes in elevation occurring over short distances. U.S. Army Corps of Engineers 2010. In certain regions where wetland mosaics are common, Corps regional wetland delineation manuals address how to delineate such wetlands. *Id.;* U.S. Army Corps of Engineers 2007c; U.S. Army Corps of Engineers 2011; U.S. Army Corps of Engineers 2012. Longstanding practice is that wetlands in the mosaic are not individually delineated, but that the Corps considers the entire mosaic and estimates percent wetland in the mosaic.

Under longstanding agency practice, a wetland is also delineated as a single wetland if a humanmade levee or similar artificial structure divides it, but a hydrologic connection is maintained between the divided wetlands. *See, e.g.*, U.S. EPA and U.S. Army Corps of Engineers 2008. One example of this concept is a wetland that is divided by a road or railway bed. In this example, evidence of a potential hydrologic connection via a shallow subsurface connection could be observed if the wetland continued to function similarly and retain similar species on either side of the human-made structure. The wetland should thus be delineated as a single wetland for the purposes of assessing wetland adjacency and, if required under the final rule, for assessing if the wetland has a significant nexus.

1. Identifying Wetlands Adjacent to Traditional Navigable Waters, the Territorial Seas, Interstate Waters, Impoundments, or Tributaries

Once a feature is identified as a wetland per the agencies' longstanding definition of that term, if the wetland itself is not a traditional navigable water (*i.e.*, it is not a tidal wetland) or an interstate water,

the agencies assess whether it is adjacent to a traditional navigable water, the territorial seas, an interstate water, a jurisdictional impoundment, or a tributary. Wetlands that are not a traditional navigable water or an interstate water or that are not adjacent to a traditional navigable water, the territorial seas, an interstate water, a jurisdictional impoundment, or a tributary are assessed under paragraph (a)(5) of the final rule. A variety of remote tools can help to assess adjacency, including maps, high-resolution elevation data, aerial photographs, and high-resolution satellite imagery. For example, USGS topographic maps, elevation data, and NHD data (including NHDPlus) may identify a physical barrier or illustrate the location of the traditional navigable water, the territorial seas, the interstate water, the jurisdictional impoundment, or the jurisdictional tributary; the wetland's proximity to the jurisdictional water; and the nature of topographic relief between the two aquatic resources. Aerial photographs or high-resolution satellite imagery may illustrate hydrophytic vegetation from the boundary (e.g., ordinary high water mark for non-tidal waters or high tide line for tidal waters) of the traditional navigable water, the territorial seas, the interstate water, the jurisdictional impoundment, or the jurisdictional tributary to the wetland boundary, or the presence of water or soil saturation. NRCS soil maps may identify the presence of hydric soil types, hydrologic soil groups based on the soil's runoff potential, soil saturation, and the occurrence of a high or seasonal water table, which can be used to help assess potential surface or shallow subsurface hydrologic connections. The soils information from the Natural Resources Conservation Service Soil Survey is available for nearly every county in the United States. See NRCS 2017. Additionally, methods that overlay depressions on the landscape with hydric soils and hydrophytic vegetation can be used to identify likely wetlands and hydrologic connections. Other indicators of a shallow subsurface connection include slope soil permeability, saturated hydraulic conductivity, the presence of an aquitard (confining layer), and permafrost. See, e.g., Science Report at 2-34. Direct visual observations on the ground, such as noting a change in vegetation or evidence of hillslope springs or seeps can be indicators, as can direct measurements of the water table. Location with a floodplain or riparian area is also an indicator of shallow subsurface connection, as wetlands located within a floodplain or riparian area of a water often have shallow subsurface flows to that water that contribute to connectivity and function. Science Report at ES-9.

NWI maps may identify that the wetlands are near the traditional navigable water, the territorial seas, the interstate water, the jurisdictional impoundment, or the jurisdictional tributary. Field work can help confirm the presence and location of the OHWM or high tide line of the traditional navigable water, the territorial seas, the interstate water, the jurisdictional impoundment, or the jurisdictional tributary and can provide additional information about the wetland's potential adjacency to that water (*e.g.*, by traversing the landscape from the traditional navigable water, the territorial seas, the interstate water, the jurisdictional tributary to the wetland and examining topographic and geomorphic features, as well as hydrologic and biologic indicators). Wetlands adjacent to the traditional navigable waters, the territorial seas, the territorial seas, or the interstate waters do not need further analysis to determine if they are "waters of the United States."

For a wetland adjacent to relatively permanent, non-navigable tributaries and relatively permanent impoundments of jurisdictional waters, similar remote tools and resources as those described above may be used to identify if the wetlands has a continuous surface connection to such waters. The tools and resources most useful for addressing this standard are those that reveal breaks in the surface connection between the wetland and the relatively permanent water, such as separations by uplands, or a berm, dike, or similar feature. For example, USGS topographic maps may show topographic highs between the two features, or simple indices can be calculated based on topography to indicate where these connectivity breaks occur. FEMA flood zone or other floodplain maps may indicate constricted floodplains along the length of the tributary channel with physical separation of flood waters that could indicate a break. High-resolution elevation data can illustrate topographic highs between the two features that extend along the tributary channel. Aerial photographs or high-resolution satellite imagery may illustrate upland vegetation along the tributary channel between the two features, or bright soil signatures indicative of higher ground. NRCS soil maps may identify mapped linear, upland soil types along the tributary channel. Field work may help to confirm the presence and location of the relatively permanent, non-navigable tributary's OHWM. In addition, field work may confirm whether there is a continuous physical connection between the wetland and the relatively permanent, non-navigable tributary, or identify breaks that may sever the continuous surface connection (*e.g.*, by traversing the landscape from the tributary to the wetland and examining topographic and geomorphic features, as well as hydrologic and biologic indicators).

For wetlands adjacent to jurisdictional tributaries or jurisdictional impoundments but that do not meet the relatively permanent standard, the agencies will conduct a significant nexus analysis to assess if the wetlands are jurisdictional. Tools to assess if the adjacent wetlands significant affect traditional navigable waters, the territorial seas, or interstate waters are discussed in section IV.A.v below.

All wetlands within a wetland mosaic are ordinarily considered collectively when determining adjacency. The agencies have long considered wetland mosaics to be delineated as one wetland, and this is consistent with pre-2015 practice. Wetlands present in such systems act generally as a single ecological unit. A "wetland mosaic" refers to a landscape where wetland and non-wetland components are too numerous and closely associated to be appropriately delineated or mapped separately. These areas often have complex microtopography, with repeated small changes in elevation occurring over short distances. Barrett 1979; U.S. Army Corps of Engineers 2007c; Michigan Natural Features Inventory 2010; Liljedahl et al. 2012; Lara et al. 2015. Tops of ridges and hummocks are often non-wetland but are interspersed with wetlands having hydrophytic vegetation, hydric soils, and wetland hydrology. Low-centered polygonal tundra and patterned ground bogs (also called strangmoor, string bogs, or patterned ground fens) are an example of wetland mosaics and are considered a single water for purposes of the final rule because their small, intermingled wetland and non-wetland components are physically and functionally integrated. See, e.g., U.S. Army Corps of Engineers 2007c. Science demonstrates that these wetlands function as a single wetland matrix and ecological unit having clearly hydrophytic vegetation, hydric soils, and wetland hydrology. Corps regional wetland delineation manuals address how to address wetland/non-wetland mosaics, that is a landscape where wetland and non-wetland components are too closely associated to be easily delineated or mapped separately. U.S. Army Corps of Engineers 2007c; U.S. Army Corps of Engineers 2012. For example, at Klatt Bog, one of the prominent patterned ground bogs in Anchorage, Alaska, the plant communities (and thus the wetland and non-wetland areas) intersperse more than can be mapped. Hogan and Tande 1983. Ridges and hummocks are often nonwetland but are interspersed throughout a wetland matrix having clearly hydrophytic vegetation, hydric soils, and wetland hydrology. Id. The agencies are continuing the longstanding practice that wetlands in the mosaic are not individually delineated, and that the agencies will consider the entire mosaic and estimate percent wetland in the mosaic. See, e.g., U.S. Army Corps of Engineers 2007c; U.S. Army Corps of Engineers 2012. As a result, the agencies will continue to evaluate these wetlands as a single water under the final rule. This applies for all aquatic resources that meet the regulatory definition of wetland,

including tidal wetlands that are traditional navigable waters, interstate wetlands, adjacent wetlands, and wetlands assessed under paragraph (a)(5) of the final rule.

iv. Impoundments

Impoundments of jurisdictional waters were not addressed in the *Rapanos* decision and thus were not directly addressed by the agencies in the *Rapanos* Guidance. In this rule, the paragraph (a)(2) impoundments⁵⁵ category provides that a jurisdictional water does not lose its jurisdictional status simply because it is impounded. However, in a change from the 1986 regulation, waters that are determined to be jurisdictional under paragraph (a)(5) and that are subsequently impounded do not retain their jurisdictional status as "waters of the United States" by rule under this provision. Such subsequently impounded jurisdictional paragraph (a)(5) waters may still be determined to be jurisdictional under one of the other categories of "waters of the United States" under this rule (*i.e.*, as a traditional navigable water, the territorial seas, an interstate water, a jurisdictional tributary, a jurisdictional adjacent wetland, or a paragraph (a)(5) water).

In implementing this rule, the agencies consider paragraph (a)(2) impoundments to include impoundments created in "waters of United States" that were jurisdictional at the time the impoundment was created, as well as impoundments of waters that are currently "waters of the United States" under paragraph (a)(1), (a)(3), or (a)(4) of this rule. The agencies also note that an impoundment of a water that does not initially meet the definition of "waters of the United States" can become jurisdictional under another provision of the regulation; for example, an impounded water could become navigable-in-fact and covered under paragraph (a)(1)(i) of the final rule. In addition, under the final rule impounding a water can create a relatively permanent water, even if the water that is being impounded is a non-relatively permanent water. For purposes of implementation, relatively permanent waters include waters where water is standing or ponded at least seasonally.

Consistent with the 1986 regulations, tributaries under this rule may be tributaries to traditional navigable waters, the territorial seas, or interstate waters or to jurisdictional impoundments. Therefore, tributaries to impoundments that are jurisdictional under paragraph (a)(2), wetlands adjacent to such tributaries, and wetlands adjacent to paragraph (a)(2) impoundments are jurisdictional if they meet either the relatively permanent standard or the significant nexus standard. For tributaries to paragraph (a)(2) impoundments to meet the relatively permanent standard, the agencies must be able to trace evidence of a flowpath (*e.g.*, physical features on the landscape, such as a channel, ditch, pipe, or swale) directly or indirectly through another water or waters downstream from the structure that creates the paragraph (a)(2) impoundment to a paragraph (a)(1) water. For wetlands adjacent to paragraph (a)(2) impoundments or to tributaries to paragraph (a)(2) impoundments that meet the relatively permanent standard, to determine if the wetlands meet the relatively permanent standard field staff would assess whether the impounded water

⁵⁵ Impounded waters may be jurisdictional under provisions other than the (a)(2) impoundments provision. For example, they may be impoundments that are traditional navigable waters and would be jurisdictional under paragraph (a)(1), or they may be impounded adjacent wetlands and meet the requirements to be jurisdictional under the paragraph (a)(4) adjacent wetlands provision. To provide clarity in the preamble of the final rule, when the agencies are discussing the subsection of impoundments that are jurisdictional under paragraph (a)(2) because they are impoundments of "waters of the United States," they will refer to "paragraph (a)(2) impoundments."

is relatively permanent, standing or continuously flowing and then determine whether the wetlands have a continuous surface connection to the impoundment or the tributary. *See* section IV.A.ii.1 and section IV.A.iii.1 of this document for additional information on evaluations under the relatively permanent standard for tributaries and adjacent wetlands. For tributaries to paragraph (a)(2) impoundments and for wetlands adjacent to either a paragraph (a)(2) impoundment or a tributary to a paragraph (a)(2) impoundment that are assessed under the significant nexus standard, the significant nexus must be to a paragraph (a)(1) water. *See* section IV.A.v for additional information on significant nexus evaluations for tributaries and adjacent wetlands.

Impoundments are distinguishable from natural lakes and ponds because they are created by discrete structures (often human-built) like dams or levees that typically have the effect of raising the water surface elevation, creating or expanding the area of open water, or both. Impoundments can vary in size, with some being very small, like many beaver ponds, and others being very large, like Lake Mead, a reservoir on the Colorado River that is created by the Hoover Dam. Paragraph (a)(2) impoundments under this rule can include both natural impoundments (like beaver ponds) and artificial impoundments (like reservoirs). Paragraph (a)(2) impoundments under this rule can be located off-channel or in-line with the channel.

An impoundment is jurisdictional under paragraph (a)(2) of the final rule if (1) the impounded water met the definition of "waters of the United States" based on the final rule's definition when the impoundment was created⁵⁶ (other than an impoundment of a paragraph (a)(5) water) or (2) the water that is being impounded is currently meets the definition of "water of the United States" under paragraph (a)(1), (a)(3), or (a)(4) of the final rule, regardless of the water's jurisdictional status at the time the impoundment was created. This approach to implementation of impoundments is generally consistent with pre-2015 practice.

The agencies can utilize a variety of tools to help identify impoundments, including many of the tools discussed in section IV.A.ii on identifying tributaries. In addition, as discussed in section IV.A.i, the Corps manages geospatial data related to reservoirs owned and operated by the Corps (*see* <u>https://geospatial-usace.opendata.arcgis.com/</u>) and also manages the National Inventory of Dams (<u>https://nid.usace.army.mil/#/</u>), which contains information from States and federal agencies on the location of over 91,000 dams across the country (U.S. Army Corps of Engineers 2020a.). In the field, agency staff can look for evidence that a water is being impounded (*e.g.*, evidence of a dam or similar structure that would create an impoundment) and then can assess if the water meets the criteria under paragraph (a)(2) of the rule to be considered jurisdictional as an impoundment. Additional information about implementation of paragraph (a)(2) impoundments is in section IV.C.3 of the final rule preamble.

v. Applying a Significant Nexus Standard

Categories of waters under the final rule that require an analysis of significant effects are tributaries that do not meet the relatively permanent standard, wetlands adjacent to tributaries and

⁵⁶ Note, however, if an impoundment is a waste treatment system constructed prior to the 1972 Clean Water Act amendments, it is eligible for the exclusion under paragraph (b) of this rule so long as the system is in compliance with currently applicable Clean Water Act requirements, such as treating water such that discharges, if any, from the system meet the Act's requirements. *See* section IV.C.7.b of the preamble.

impoundments and that do not meet the relatively permanent standard (*i.e.*, wetlands adjacent to tributaries that do not meet the relatively permanent standard or to non-relatively permanent impoundments of jurisdictional waters and wetlands adjacent to but not directly abutting tributaries that meet the relatively permanent standard or to non-relatively permanent impoundments of jurisdictional waters), or for waters assessed under paragraph (a)(5) that do not meet the relatively permanent standard.

For purposes of this rule for significant nexus analyses involving tributaries and their adjacent wetlands, the agencies will assess the tributaries and their adjacent wetlands in a catchment, as described in section III.E.iii. If the tributaries in the region, including the tributary under assessment, have no adjacent wetlands, the agencies consider only the factors and functions of the tributaries in determining whether there is a significant effect on the chemical, physical, or biological integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. If any of the tributaries in the region, including the tributary under assessment, have adjacent wetlands, the agencies will consider the factors and functions of the tributaries, including the tributary under assessment, have adjacent wetlands, the agencies will consider the factors and functions of the tributaries, including the tributary under assessment, together with the functions performed by the wetlands adjacent to the tributaries in the catchment, in evaluating whether a significant nexus is present. Tools relevant to implementing "similarly situated" and "in the region" relevant to tributaries and their adjacent wetlands is discussed in section IV.B. As discussed in the preamble and in section III.E.i, the agencies generally intend to analyze such waters individually to determine if they significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters.

In conducting a significant nexus analysis under the final rule, the agencies will evaluate available hydrologic information (*e.g.*, gage data, precipitation records, flood predictions, historic records of water flow, statistical data, personal observations/records, etc.) and physical indicators of flow including the presence and characteristics of a reliable OHWM. To understand the chemical, physical, and biological functions provided by tributaries and their adjacent wetlands or by waters evaluated under paragraph (a)(5), and the effects those functions have on traditional navigable waters, the territorial seas, or interstate waters, it is important to use relevant geographic water quality data in conjunction with site-specific data from field sampling and hydrologic modeling.

The agencies have used many tools and sources of information to assess significant effects on the chemical, physical, and biological integrity of downstream traditional navigable waters, the territorial seas, or interstate waters. Some tools and resources that the agencies have used to provide and evaluate evidence of a significant effect on the physical integrity of traditional navigable waters, the territorial seas, or interstate waters include USGS stream gage data, floodplain maps, statistical analyses, hydrologic models and modeling tools such as USGS's StreamStats (*see* USGS 2019) or the Corps' Hydrologic Engineering Centers River System Analysis System (HEC-RAS) (*see* U.S. Army Corps of Engineers a), physical indicators of flow such as the presence and characteristics of a reliable OHWM with a channel defined by bed and banks, or other physical indicators of flow including such characteristics as shelving, wracking, water staining, sediment sorting, and scour, information from NRCS soil surveys, precipitation and rainfall data, and NRCS SNOTEL data or NOAA national snow analyses maps.

To evaluate the evidence of a significant effect on the biological integrity of traditional navigable waters, the territorial seas, or interstate waters, the agencies and practitioners have used tools and resources such as: population survey data and reports from federal, Tribal, State, Territorial, and local

resource agencies, natural history museum collections databases, bioassessment program databases, fish passage inventories, U.S. FWS and NOAA Critical Habitat layers (U.S. FWS b; NOAA 2022a), species distribution models, and scientific literature and references from studies pertinent to the distribution and natural history of the species under consideration.

Tools and resources that provide and evaluate evidence of a significant effect on the chemical integrity of traditional navigable waters, the territorial seas, or interstate waters include data from USGS water quality monitoring stations; state, tribal, and local water quality reports; water quality monitoring and assessment databases; EPA's How's My Waterway, which identifies Clean Water Act section 303(d) listed waters, water quality impairments, and TMDLs (available at https://www.epa.gov/waterdata/howsmy-waterway); watershed studies; stormwater runoff data or models; EPA's NEPAssist, which provides locations and information on wastewater discharge facilities and hazardous-waste sites (available at https://www.epa.gov/nepa/nepassist); the National Land Cover Database (NLCD), which provides nationally consistent information on land cover classifications (available at https://www.usgs.gov/centers/eros/science/national-land-cover-database); and scientific literature and references from studies pertinent to the parameters being reviewed. As discussed in section IV.A.i.2, EPA's HAWQS model can be used to assess the cumulative effects of wetlands on the larger downstream waters into which they drain. Additional approaches to quantifying the hydrologic storage capacity of wetlands include statistical models, such as pairing LIDAR-based topography with precipitation totals (e.g., Lane and D'Amico 2010; Shook et al. 2013; Wu and Lane 2016). Both statistical and process-based models have been used to quantify the nutrient removal capacities of non-floodplain wetlands (e.g., Cheng et al. 2020), and in some cases to assess the effects of non-floodplain wetland nutrient removal, retention, or transformation on downstream water quality (e.g., Evenson et al. 2021). Evaluations of a significant effect on the chemical integrity of a traditional navigable water, the territorial seas, or an interstate water may include qualitative reviews of available information or incorporate quantitative analysis components including predictive transport modeling.

A variety of modeling approaches can be used to quantify the connectivity and cumulative effects of wetlands, including non-floodplain wetlands, on downstream waters, as discussed in section IV.A.i.2. Some examples include SWAT (available at https://swat.tamu.edu/), the Hydrologic Simulation Program in Fortran (available at https://www.epa.gov/ceam/hydrological-simulation-program-fortran-hspf), and DRAINWAT available at (https://www.bae.ncsu.edu/agricultural-water-management/drainmod/). Other examples of models applicable to identifying effects of wetlands on downstream waters include MODFLOW (available at https://www.usgs.gov/mission-areas/water-resources/science/modflow-and-related-programs?qt-science_center_objects=0#qt-science_center_objects) and VS2DI (available at https://www.usgs.gov/software/vs2di-version-13), as well as the hydrologic models discussed in section IV.A.i.2.

When assessing whether a water significantly affects traditional navigable waters, the territorial seas, or interstate waters, the significant nexus analysis can consider whether surface or shallow subsurface connections contribute to the type and strength of functions provided by a water either alone or with similarly situated waters. The SAB (2014b) has noted the importance of shallow subsurface connections and stated, "[t]he available science...shows that groundwater connections, particularly via shallow flow paths in unconfined aquifers, can be critical in support the hydrology and biogeochemical functions of wetlands and other waters." However, neither shallow subsurface connections nor any type

of groundwater, shallow or deep, are themselves "waters of the United States."

B. Case Specific Significant Nexus Analysis

i. Similarly Situated

As discussed in the preamble for the final rule and in section III.E.ii of this document, the agencies are taking an approach to which waters are "similarly situated" that is comparable to the *Rapanos* Guidance for tributaries and their adjacent wetlands. Like pre-2015 practice, the agencies consider tributaries and their adjacent wetlands to be "similarly situated" waters. The *Rapanos* Guidance also interpreted "similarly situated" to mean a tributary and its adjacent wetlands.

The agencies note that the best available science supports evaluating the connectivity and effects of streams, wetlands, and open waters to downstream waters in a cumulative manner in context with other streams, wetlands, and open waters. Science Report at ES-5 to ES-6, ES-13 to ES-14, 6-10 to 6-12.

Many of the tools described in section IV.B.ii below for delineating the catchment (*i.e.*, "the region") can also be used for helping to determine which tributaries should be considered as similarly situated. Such tools, when coupled with aerial photography, NWI maps, NHD, and other sources or methods for identifying adjacent wetlands (as discussed in section IV.A.iii.1), can also be used for identifying which adjacent wetlands should be considered as similarly situated.

In implementing the significant nexus standard for waters assessed under paragraph (a)(5) of the final rule, the agencies generally intend to analyze such waters individually to determine if they significantly affect the chemical, physical, or biological integrity of traditional navigable waters, the territorial seas, or interstate waters. This approach reflects the agencies' consideration of public comments, as well as implementation considerations for waters assessed under paragraph (a)(5). While the agencies' regulations have long authorized the assertion of jurisdiction on a case-specific basis over waters that do not fall within the jurisdictional provisions by water type, since *SWANCC* and the issuance of the *SWANCC* guidance with its requirement of headquarters approval over determinations under that provision, the agencies have not in practice asserted jurisdiction over paragraph (a)(3) "other waters" under the pre-2015 regulatory regime. The agencies addressed such waters individually on a case-specific basis under pre-2015 practice and have concluded at this time that individual assessments are practical and implementable for significant nexus determinations for waters assess under paragraph (a)(5). *See* also section III.E.ii.

ii. In the Region

In the final rule for tributaries and their adjacent wetlands, the agencies consider similarly situated waters to be "in the region" when they lie within the catchment area of the tributary of interest. Accordingly, in implementing the significant nexus standard under this rule, all tributaries and adjacent wetlands within the catchment area of the tributary of interest will be analyzed as part of the significant nexus analysis. For purposes of a significant nexus analysis for tributaries and adjacent wetlands, the

agencies will identify the "region" as the catchment that drains to and includes the tributary of interest. Catchments will be delineated from the downstream-most point of the tributary reach of interest and include the land uphill that drains to that point. For example, if the tributary of interest is a second order stream, the catchment would be delineated from the point that the second order stream enters a third order stream (including at the point where the second order stream confluences with another second order stream). This is a change from implementation under the *Rapanos* Guidance, which relied on a concept of a relevant "reach" of a tributary—defined as the entire reach of the stream that is of the same order (*i.e.*, from the point of confluence, where two lower order streams meet to form the tributary, downstream to the point such tributary enters a higher order stream). *Rapanos* Guidance at 10.

Because waters assessed under paragraph (a)(5) of the final rule will generally be evaluated individually for their effects on traditional navigable waters, the territorial seas, or interstate waters, the agencies have not established what would be considered "in the region" for this such waters, as it is not needed to implement the final rule, as such waters will generally be considered on an individually basis. *See also* section III.E.iii.

The agencies sought comment on a range of approaches for determining the "region" in which similarly situated waters lie. One such option was for a watershed to be delineated from the downstreammost point of the "relevant reach"—that is, the region would be the watershed that drains to and includes the relevant reach in question. This is the option that the agencies have chosen to implement in this final rule as discussed above, informed by their experience, the best available science, Supreme Court decisions, and public comments. Note that "catchment" as used in the final rule is the watershed that drains to the downstream-most point of the tributary of interest.

Many existing spatial analysis tools based on watershed frameworks and elevation models can be used to delineate catchments quickly and reliably in most parts of the country. USGS topographic maps can be manually interpreted to delineate catchments based on the location of the outlet point (the downstream-most point of the tributary of interest where the tributary enters a higher order stream), using calculations informed by topographic contours, the alignment of topographic high spots, and grouping of lower, valley bottoms. Various GIS tools, web applications, and automated modeling systems can also delineate catchments based on one of the more of the many factors that can influence drainage, including surface topography, climate, land use, the presence of hydrologic sinks, topology of sewer systems, and design of wastewater treatment plant (WWTP) service areas. For example, NHDPlus provides delineated catchments for individual stream segments, by linking the mapped stream network to the landscape. U.S. EPA 2022a. The WATERS GeoViewer, available at <u>https://www.epa.gov/waterdata/waters-geoviewer</u>, utilizes NHDPlus to provide interactive watershed delineation. U.S. EPA 2022n. This is also available through WATERS as an API service (available at

https://watersgeo.epa.gov/openapi/waters/#/Delineation). StreamStats by the USGS (available at https://streamstats.usgs.gov/ss/) is a map-based web tool that can delineate drainage areas for streams and estimate flow characteristics for selected sites, based on stream gage data, basin characteristics, climate, and other factors. USGS has also developed the NHD Watershed Tool (available at https://www.usgs.gov/national-hydrography/nhd-watershed-tool) that allows users to delineate a watershed from any point on any NHD reach in a fast, accurate, and reliable manner. EPA's EnviroAtlas Interactive Map (available at https://www.epa.gov/enviroatlas/enviroatlas-interactive-map) has a wide variety of tools that can help delineate catchments, including a tool that illustrates how precipitation will

flow over the land surface, mapped elevation profiles for selected tributaries, and designations of upstream and downstream watersheds within a stream network.





Appendices to the Technical Support Document for the "Revised Definition of 'Waters of the United States'" Rule

U.S. Environmental Protection Agency

and

Department of the Army

December 2022

Appendices Table of Contents

Appendices Table of Contents	269
Appendices List of Tables	269
Appendix A: Glossary	270
Appendix B: References	284
Appendix C: References from the Literature Update Screening and Public Comments on Literature Published Since 2014	362
Appendix C1: References Relevant to the Conclusions of the Science Report Published Since 2014	362
Appendix C2: Plain-Text Language from the Abstracts of Illustrative Scientific Papers	531
Ephemeral, Intermittent, and Perennial Streams	531
Floodplain Wetlands and Open Waters	541
Non-Floodplain Wetlands and Open Waters	546
Appendix C3: Additional References	554
Appendix C3i: Reference Review Process and Findings	554
Appendix C3ii: Additional References Not Captured During Screening Process (October 2021)	560

Appendices List of Tables

Cable C-1: References Relevant to the Conclusions of the Science Report (of the 37 Identified by the	
Agencies in October 2021 as Not Captured During the Screening Process)	555
Cable C-2: References Provided During the Public Comment Period Relevant to the Conclusions of the	e
cience Report	559

Appendix A: Glossary

Most of the terms in this glossary are derived directly from EPA's 2015 report *Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence* (hereafter the Science Report). Some terms are denoted by Clean Water Act, Final Rule, 2020 NWPR, or Science Report to indicate if they are being defined in the context of the Clean Water Act (including under the agencies' longstanding regulations and the final rule defining "waters of the United States"), the final rule, the 2020 Navigable Waters Protection Rule (2020 NWPR), or the Science Report, respectively.

Allochthonous: Describing organic material that originates from outside of streams, rivers, wetlands, or lakes (*e.g.*, terrestrial plant litter, soil).

Alluvial Aquifer: An aquifer with geologic materials deposited by a stream or river (alluvium) that retains a hydraulic connection with the depositing stream.

Alluvial Deposits: See Alluvium.

Alluvium: Deposits of clay, silt, sand, gravel, or other particulate materials that have been deposited by a stream or other body of running water in a streambed, on a flood plain, on a delta, or at the base of a mountain. *See* Colluvium.

Aquatic Ecosystem: Any aquatic environment, including all of the environment's living and nonliving constituents and the interactions among them.

Aquifer: A geologic formation (*e.g.*, soil, rock, alluvium) with permeable materials partially or fully saturated with ground water that yields ground water to a well, spring, or stream.

Autochthonous: Describing organic matter that originates from production within streams, rivers, wetlands, or lakes (*e.g.*, periphyton, macrophytes, phytoplankton).

Bank Storage: Storage of water that flows from a stream to an alluvial aquifer during a flood or period of high streamflow. The volume of water is stored and released after the high-water event over days to months. The volume of water stored and the timing of release depends on the hydraulic properties of the alluvial aquifer.

Baseflow: Sustained flow of a stream (or river) in the absence of stormflow (direct runoff). Natural baseflow is sustained by ground-water discharge in the stream network. Baseflow also can be sustained by human sources (e.g., irrigation recharges to ground water).

Carolina Bays: Elliptical, ponded, depressional wetlands that range along the Atlantic Coastal Plain from northern Florida to New Jersey. *See* Delmarva Bays.

Catadromous: Species that breed in the ocean and spend most of their lives in freshwater.

Catchment: The area drained by a stream, river, or other water body; typically defined by the topographic divides between one water body and another. *Synonymous with* Watershed *and* Drainage Basin.

Channel: A natural or constructed passageway or depression of perceptible linear extent that conveys water and associated material downgradient.

Channelization: A type of artificial drainage in which complex channels are straightened to increase the rate of water flow from an area.

Channelized Flow: Flow that occurs in a natural or artificial channel.

Colluvium: A layer of unconsolidated soils, sediment and rock fragments deposited by surface runoff and gravitational processes; colluvium generally occurs as a blanket of poorly sorted sediment and rock fragments on the lower parts of hillslopes underlain by bedrock. *See* Alluvium.

Condition: General health or quality of an ecosystem, typically assessed using one or more indicators.

Confined Aquifer: An aquifer bounded above and below by confining units of distinctly lower permeability than that of the aquifer itself.

Confluence: The point at which two stream channels intersect to form a single channel.

Connectivity: The degree to which components of a river system are joined, or connected, by various transport mechanisms; connectivity is determined by the characteristics of both the physical landscape and the biota of the specific system.

Connectivity Descriptors (for streams and wetlands): The frequency, duration, magnitude, timing, and rate of change of fluxes to and biological exchanges with downstream waters.

Contributing Area: Location within a watershed/river network that serves as a source of stream flow or material flux.

Contaminants: Any material that might be harmful to humans or other organisms when released to the environment.

Deep Ground Water: Ground-water flow systems having the deepest and longest flowpaths; also referred to as regional ground-water flow systems, they can occur beneath local and intermediate ground-water flow systems. *See* Local Ground Water, Regional Ground Water.

Delmarva Bays: Carolina bays that are geographically specific to the Delmarva Peninsula. These wetlands frequently have the same elliptical shape and orientation as Carolina bays. *See* Carolina Bays.

Dendritic Stream Network: A stream network pattern of branching tributaries (see Figure 2-19B).

Depressional Wetland: A wetland occupying a topographic low point that allows the accumulation of surface water. Depressional wetlands can have any combination of inlets and outlets or lack them completely. Examples include kettles, prairie potholes, and Carolina bays. This category also includes slope wetlands (wetlands associated with surface discharge of ground water or saturated overflow with no channel formation).

Diadromous: Migratory between fresh and salt waters.

Direct Runoff: Runoff that occurs in direct response to precipitation. See Stormflow.

Discharge (Science Report): The volume of water (surface water or ground water) that passes a given location over a given period of time; the rate of runoff. Often expressed as cubic feet per second ($\text{ft}^3 \text{ s}^{-1}$) or cubic meters per second ($\text{m}^3 \text{ s}^{-1}$).

Discharge (Clean Water Act): The term "discharge" when used without qualification includes a discharge of a pollutant, and a discharge of pollutants. Clean Water Act section 502(16).

Discharge of a pollutant (Clean Water Act): The term "discharge of a pollutant" and the term "discharge of pollutants" each means (A) any addition of any pollutant to navigable waters from any point source, (B) any addition of any pollutant to the waters of the contiguous zone or the ocean from any point source other than a vessel or other floating craft. Clean Water Act section 502(12).

Discontinuous Flow: Refers to stream and river reaches that have flow in one part of the reach but not another part of the reach. *See* Reach.

Dispersal: Movement from natal breeding sites to new breeding sites.

Drainage Area: The spatial extent of a drainage basin. Typically expressed in square miles (mi²) or square kilometers (km²).

Drainage Basin: The area drained by a stream, river, or other water body; typically defined by the topographic divides between one water body and another. *Synonymous with* Catchment *and* Watershed.

Drainage Density: The total length of stream channels per unit drainage area (e.g., per mi2 or km2).

Drainage Network: See River Network.

Egg Bank: Viable dormant eggs that accumulate in soil or in sediments under water. See Seed bank.

Endorheic Basins: A closed drainage basin with no outflows to other water bodies.

Endorheic Stream: A stream or river reach that experiences a net loss of water to a ground-water system. *See* Losing Stream or Wetland.

Ephemeral (NWPR): The term *ephemeral* means surface water flowing or pooling only in direct response to precipitation (*e.g.*, rain or snow fall). 85 FR 22338; 33 CFR 328.3(c)(3) (2021).

Ephemeral Stream (Science Report): A stream or river that flows briefly in direct response to precipitation; these channels are always above the water table.

Eutrophication: Natural or artificial enrichment of a water body by nutrients, typically phosphates and nitrates. If enrichment leads to impairment (*e.g.*, toxic algal blooms), eutrophication is a form of pollution.

Evapotranspiration: The combined loss of water to the atmosphere due to evaporation and transpiration losses. Transpiration is the loss of water vapor to air by plants.

Fen: A peat-accumulating wetland characterized by mineral-rich water inputs.

Flood: The occurrence of stream or river flow of such magnitude that it overtops the natural or artificial banks in a reach of the stream or river; where a floodplain exists, a flood is any flow that spreads over or inundates the floodplain. Floods also can result from rising stages in lakes and other water bodies.

Flood (100-year): Flood level (stage or discharge) with a 1% probability of being equaled or exceeded in a given year.

Flood Flows: Discharge or flow of sufficient (or greater) magnitude to cause a flood.

Flood Recurrence Interval: The average number of years between floods of a certain size is the recurrence interval or return period. The actual number of years between floods of any given size varies a lot because of the naturally changing climate. USGS.

Flood Stage: The stage at which streams or rivers overtop their natural or artificial banks.

Floodwater: Water associated with a flood event.

Floodplain: A level area bordering a stream or river channel that was built by sediment deposition from the stream or river under present climatic conditions and is inundated during moderate to high flow events. Floodplains formed under historic or prehistoric climatic conditions can be abandoned by rivers and form terraces.

Floodplain Wetland: Portions of floodplains that meet the Cowardin *et al.* (1979) three-attribute definition of a wetland (*i.e.*, having wetland hydrology, hydrophytic vegetation, or hydric soils). *See* Wetland.

Flow: Water movement above ground or below ground.

Flow Duration Class: A classification that assigns streamflow duration to ephemeral, intermittent, or perennial classes.

Flow Regime: Descriptor of flow types in a temporal or magnitude sense (i.e., slow-flow regime, low-flow regime)

Flowpath: See Hydrologic Flowpath.

Fluvial: Refers to or pertains to streams; *e.g.*, stream processes (fluvial processes), fluvial landforms, such as fluvial islands and bars, and biota living in and near stream channels.

Flux: Flow of materials between system components per unit time.

Gaining Stream or Wetland: A wetland or a stream or river reach that experiences a net gain of water from ground water (see Figure 2-5). In this situation, the water table elevation near the stream or wetland is higher than the stream or wetland water surface. Conditions conducive to losing or gaining streams and wetlands can change over short distances within river networks and river basins. *See* Losing Stream or Wetland.

Geographically Isolated Wetland: A wetland that is completely surrounded by uplands; for example, hydrophytic plant communities surrounded by terrestrial plant communities or undrained hydric soils surrounded by nonhydric soils. This term often is mistakenly understood to mean hydrologically isolated. Geographically isolated wetlands vary in their degree of hydrologic and biotic connectivity.

Ground Water: Any water that occurs and flows in the saturated zone. See Saturated Zone.

Ground-water Discharge: The flow of ground water to surface waters; discharge areas occur where the water tables intersect land surfaces. *See* Seep, Spring.

Ground-water Discharge Wetland: A wetland that receives ground-water discharge.

Ground-water Flow: Flow of water in the subsurface saturated zone.

Ground-water Recharge: The process by which ground water is replenished; a recharge area occurs where precipitation or surface water infiltrates and is transmitted downward to the saturated zone (aquifer). *See* Infiltration, Percolation, Transmission.

Ground-water Recharge Wetland: A wetland that recharges ground water.

Ground-water System: Reference to the ground water and geologic materials comprising the saturated zone; the ground-water system, as a whole, is a three-dimensional flow field.

Ground water–Surface water Interactions: Movement of water between surface-water bodies and ground-water systems. Flows can occur in either direction.

Ground-water Withdrawal: Pumping of water from aquifers for human uses.

Habitat: Environment (place and conditions) in which organisms reside.

Headwater: Areas from which water originates within a river or stream network. This term typically refers to stream channels but can also describe wetlands or open waters, such as ponds.

Headwater Stream: Headwater streams are first- to third-order streams. Headwater streams can be ephemeral, intermittent, or perennial. *See* Stream Order, Flow Duration Class.

Hillslope: A sloping segment of land surface.

Hydraulic Conductivity: A measure of the permeability of a porous medium. For a given hydraulic gradient, water moves more rapidly through media with high hydraulic conductivity than low hydraulic conductivity.

Hydraulic Gradient: Slope of the water table. See Water Table.

Hydraulic Head: The height above a standard datum of the surface of a column of water that can be supported by the static pressure at a given point; for a well, the hydraulic head is the height of the water level in the well compared to a datum elevation.

Hydraulics: The physics of water in its liquid state.

Hydric: An area, environment, or habitat that is generally very wet with plenty of moisture.

Hydrograph: A graph of stream or river discharge over time. Stage or water table elevation also can be plotted.

Hydrologic Event: An increase in streamflow resulting from precipitation or snowmelt.

Hydrologic Flowpath: The pathway that water follows as it moves over the watershed surface or through the subsurface environment.

Hydrology: The study of the properties, distribution, and effects of water as a liquid, solid, and gas on Earth's surface, in the soils and underlying rocks, and in the atmosphere.

Hydrologic Landscape: A landscape with a combination of geology, soils, topography, and climate that has characteristic influences on surface water and ground water.

Hydrologic Permanence: The frequency and duration of streamflow in channels or the frequency and duration of standing water in wetlands.

Hyporheic Flow: Water from a stream or river channel that enters subsurface materials of the streambed and bank and then returns to the stream or river.

Hyporheic Exchange: Water and solutes exchanged between a surface channel and the shallow subsurface. *See* Hyporheic Flow.

Hyporheic Zone: The area adjacent to and beneath a stream or river in which hyporheic flow occurs. The dimensions of the hyporheic zone are controlled by the distribution and characteristics of alluvium and hydraulic gradients between streams and local ground water.

Hypoxia: The condition in which dissolved oxygen is below the level necessary to sustain most animal life. *See* Anoxic Conditions.

Infiltration: The downward entry of water from the land surface into the subsurface.

Intermittent (2020 NWPR): The term *intermittent* means surface water flowing continuously during certain times of the year and more than in direct response to precipitation (*e.g.*, seasonally when the groundwater table is elevated or when snowpack melts).

Intermittent (Science Report): This term also can be applied to other surface-water bodies and ground-water flow or level. *See* Intermittent Stream.

Intermittent Stream: A stream or portion of a stream that flows continuously only at certain times of year; for example, when it receives water from a spring, ground-water source, or a surface source such as melting snow. At low flow, dry segments alternating with flowing segments can be present.

Inundation: To cover dry land with floodwaters.

Isolation: Condition defined by reduced or nonexistent transport mechanisms between system components.

Lag Function: Any function within a stream or wetland that provides temporary storage and subsequent release of materials without affecting cumulative flux (exports = imports); delivery is delayed and can be prolonged.

Lateral Source Stream: A first-order stream that flows into a higher order stream.

Lentic: Of, relating to, or living in still water. See Lotic.

Levee (Artificial): An engineered structure built next to a stream or river from various materials to prevent flooding of surrounding areas. The levee raises the elevation of the channel height to convey greater discharge of water without flooding.

Levee (Natural): A broad, low ridge or embankment of coarse silt and sand that is deposited by a stream on its floodplain and along either bank of its channel. Natural levees are formed by reduced velocity of flood flows as they spill onto floodplain surfaces and can no longer transport the coarse fraction of the suspended sediment load.

Local Ground Water: Ground water with a local flow system. Water that recharges at a high point in the water table that discharges to a nearby lowland. Local ground-water flow is the most dynamic and shallowest of ground-water flow systems. Therefore, it has the greatest interchange with surface water. Local flow systems can be underlain by intermediate and regional flow systems. Water in these deeper flow systems have longer flowpaths and longer contact time with subsurface materials. Deeper flow systems also eventually discharge to surface waters and influence their condition.

Losing Stream or Wetland: A stream, wetland, or river reach that experiences a net loss of water to a ground-water system (*see* Figure 2-5 of the Science Report). In this situation, the water table elevation near the stream or wetland is lower than the stream or wetland water surface. Conditions conducive to losing or gaining streams and wetlands can change over short distances within river networks and river basins. *See* Gaining Stream or Wetland.

Lotic: Of, relating to, or living in moving water. See Lentic.

Mainstem: Term used to distinguish the larger (in terms of discharge) of two intersecting channels in a river network.

Materials: Any physical, chemical, or biological entity, including but not limited to water, heat energy, sediment, wood, organic matter, nutrients, chemical contaminants, and organisms.

Meltwater: Liquid water that results from the melting of snow, snowpacks, ice, or glaciers.

Migration: Long-distance movements undertaken by organisms on a seasonal basis.

Non-floodplain Wetland: An area outside of the floodplain that meets the Cowardin *et al.* (1979) threeattribute definition of a wetland (i.e., having wetland hydrology, hydrophytic vegetation, or hydric soils). For the purposes of this report, riparian wetlands that occur outside of the floodplain are not included as non-floodplain wetlands, since these wetlands are subject to bidirectional, lateral hydrologic flows. *See* Floodplain, Wetland. **Nutrients (In Aquatic Systems):** Elemental forms of nitrogen, phosphorus, and trace elements, including sulfur, potassium, calcium, and magnesium, that are essential for the growth of organisms but can be contaminants when present in high concentrations.

Nutrient Spiraling: Longitudinal cycles ("spirals") of nutrient uptake and release along the stream or river continuum. The spirals are created as aquatic organisms consume, transform, and regenerate nutrients, altering the rates of nutrient transport to downstream waters.

Open-channel Flow: Water flowing within natural or artificial channels.

Open Waters: Nontidal lentic water bodies such as lakes and oxbow lakes that are frequently small or shallow.

Overbank Flow: Streamflow that overtops a stream or river channel.

Overland Flow: The portion of streamflow derived from net precipitation that fails to infiltrate the land surface at any point and runs over the surface to the nearest stream channel.

Oxbow Lakes: Water bodies that originate from the cutoff meanders of rivers; such lakes are common in floodplains of large rivers.

Peatland: A wetland that accumulates partially decayed organic matter. Fens and bogs are common examples.

Perched Ground Water: Unconfined ground water separated from an underlying body of ground water by an unsaturated zone; perched ground water is supported by a perching layer (bed) for which the permeability is so low that water percolating downward to the underlying unsaturated zone is restricted.

Percolation: The downward movement of water through soil or rock formations.

Perennial (2020 NWPR): The term *perennial* means surface water flowing continuously year-round.

Perennial (Science Report): *See* Perennial Stream. This term can be applied to other surface-water bodies and to ground-water flow or level.

Perennial Stream: A stream or portion of a stream that flows year-round and is maintained by local, intermediate, or regional ground-water discharge or flow from higher in the river network.

Permanent Waters: Water bodies that contain water year-round; perennial waters.

Permeability: Property of a porous medium that enables it to transmit fluids under a hydraulic gradient. For a given hydraulic gradient, water will move more rapidly through high permeability materials than low permeability materials.

Potential Evapotranspiration: The amount of water that would be lost to the atmosphere over a given area through evaporation and transpiration, assuming no limits on the water supply. *See* Evapotranspiration.

Prairie Potholes: Complex of glacially formed wetlands, usually lacking natural outlets, found in the central United States and Canada.

Precipitation: Water that condenses in the atmosphere and falls to a land surface. Common types include rain, snow, hail, and sleet.

Precipitation Intensity: The rate at which precipitation occurs; generally refers to rainfall intensity.

Primary Production: The fixation of inorganic carbon into organic carbon (*e.g.*, plant and algae biomass) through the process of photosynthesis. Primary production is the first level of the food web, and provides most of the autochthonous carbon produced in ecosystems. The rate of fixation is referred to as gross primary productivity (GPP) or net primary productivity (NPP), where NPP is equal to GPP minus respiration. *See* Respiration, Secondary Production.

Propagule: Any part of an organism that can give rise to a new individual organism. Seeds, eggs, and spores are propagules.

Reach (Science Report): A length of stream channel with relatively uniform discharge, depth, area, and slope.

Recession [of Flow]: Decrease in flow following a hydrologic event.

Recharge Area: An area in which water infiltrates the surface and reaches the zone of saturation.

Refuge Function: The protective function of a stream or wetland that allows an organism (or material) to avoid mortality (or loss) in a nearby sink area, thereby preventing the net decrease in material flux that otherwise would have occurred (exports = imports). This term typically refers to organisms but can be used for nonliving materials. *See* Sink Function.

Regional Ground Water: Ground water with a deep, regional-scale flow system; also referred to as deep ground water. These flow systems can occur beneath local and intermediate ground-water flow systems. *See* Local Ground Water, Deep Ground Water.

Respiration: The chemical process by which organisms break down organic matter and produce energy for growth, movement, and other biological processes. Aerobic respiration uses oxygen and produces carbon dioxide.

Return Flow: Water that infiltrates into a land surface and moves to the saturated zone and then returns to the land surface (or displaces water that returns to the soil surface).

Riparian Areas: Transition areas or zones between terrestrial and aquatic ecosystems that are distinguished by gradients in biophysical conditions, ecological processes, and organisms. They are areas through which surface hydrology and subsurface hydrology connect water bodies with their uplands. They include those portions of terrestrial ecosystems that significantly influence exchanges of energy and matter with aquatic ecosystems. Riparian areas are adjacent to perennial, intermittent, and ephemeral streams, lakes, and estuarine-marine shorelines. *See* Upland.

Riparian Wetland: Portions of riparian areas that meet the Cowardin *et al.* (1979) three-attribute definition of a wetland (i.e., having wetland hydrology, hydrophytic vegetation, hydric soils). *See* Wetland.

River: A relatively large volume of flowing water within a visible channel, including subsurface water moving in the same direction as the surface water, and lateral flows exchanged with associated floodplain and riparian areas. *See* Stream.

River Network: A hierarchical, interconnected population of channels or swales that drain water to a river. Flow through these channels can be perennial, intermittent, or ephemeral.

River Network Expansion/Contraction: The extent of flowing water in a river network increases during wet seasons and large precipitation events and decreases during dry periods. *See* Variable Source Area.

River System: A river and its entire drainage basin, including its river network, associated riparian areas, floodplains, alluvial aquifers, regional aquifers, connected water bodies, geographically isolated water, and terrestrial ecosystems.

Runoff: The part of precipitation, snowmelt, or other flow contributions (e.g., irrigation water) that appears in surface streams at the outlet of a drainage basin; it can originate from both above land surface (e.g., overland flow) and below land surface sources (e.g., ground water). Units of runoff are depth of water (similar to precipitation units, e.g., mm). This measurement is the depth of water if it were spread across the entire drainage basin. Can also be expressed as a volume of water (i.e., m3, feet3, acre-ft).

Saturated Zone: The zone below the land surface where the voids in soil and geologic material are completely filled with water. Water in the saturated zone is referred to as ground water. The upper surface of the saturated zone is referred to as the water table. *See* Ground Water, Unsaturated Zone, Water Table.

Saturation Overland Flow: Water that falls onto a saturated land surface and moves overland to the nearest stream or river.

Seasonality: Refers to the seasonal distribution of water surplus of a river system. See Water Surplus.

Secondary Production: The generation of biomass of consumer organisms that feed on organic material from primary producers (algae, microbes, aquatic and terrestrial plants), and biomass of predators that feed on consumer organisms. *See* Primary Production.

Seed Bank: Viable dormant seeds that accumulate in soil or in sediments under water. See Egg bank.

Seep: A small area where water slowly flows from the subsurface to the surface. A seep can also refer to a wetland formed by a seep; such a wetland is referred to as a ground-water slope wetland.

Seepage: Water that flows from a seep.

Shallow Ground Water: Ground water with shallow hydrologic flowpaths. See Local Ground Water.

Significantly Affect (Final Rule Paragraph (c)(6)): *Significantly affect* means a material influence on the chemical, physical, or biological integrity of waters identified in paragraph (a)(1) of this section. To determine whether waters, either alone or in combination with similarly situated waters in the region,

have a material influence on the chemical, physical, or biological integrity of waters identified in paragraph (a)(1) of this section, the functions identified in paragraph (c)(6)(i) below will be assessed and the factors identified in paragraph (c)(6)(ii) below will be considered:

(i) Functions to be assessed:

(A) Contribution of flow;

(B) Trapping, transformation, filtering, and transport of materials (including nutrients, sediment, and other pollutants);

(C) Retention and attenuation of floodwaters and runoff;

(D) Modulation of temperature in waters identified in paragraph (a)(1) of this section; or

(E) Provision of habitat and food resources for aquatic species located in waters

identified in paragraph (a)(1) of this section;

(ii) Factors to be considered:

(A) The distance from a water identified in paragraph (a)(1) of this section;

(B) Hydrologic factors, such as the frequency, duration, magnitude, timing, and rate of hydrologic connections, including shallow subsurface flow;

(C) The size, density, or number of waters that have been determined to be similarly situated;

(D) Landscape position and geomorphology; and

(E) Climatological variables such as temperature, rainfall, and snowpack.

Sink Function: Any function within a stream or wetland that causes a net decrease in material flux (imports exceed exports).

Snowpack (Science Report): Accumulation of snow during the winter season; an important source of water for streams and rivers in the western United States.

Snowmelt: The complete or partial melting and release of liquid water from seasonal snowpacks.

Solute: A substance that is dissolved in water.

Source Area: The originating location of water or other materials that move through a river system.

Source Function: Any function within a stream or wetland that causes a net increase in material flux (exports exceed imports).

Spillage: Overflow of water from a depressional wetland to a swale or channel.

Spring: A surface-water body formed when the side of a hill, a valley bottom, or other excavation intersects a flowing body of ground water at or below the local water table.

Stable Isotope Tracer: Certain elements such as oxygen, hydrogen, carbon, and nitrogen have multiple isotopes that occur in nature that do not undergo radioactive decay. These isotopes can be used to track the source and movement of water and other substances.

Stage: The elevation of the top of a water surface.

Stream: A relatively small volume of flowing water within a visible channel, including subsurface water moving in the same direction as the surface water, and lateral flows exchanged with associated floodplain and riparian areas. *See* River.

Stream Burial: The process of incorporating streams—particularly headwaters—into storm sewer systems, usually by routing through underground pipes.

Stream Power: A measure of the erosive capacity of flowing water in stream channels or the rate of energy dissipation against the stream bed or banks per unit of channel length that has the mathematical form: $\omega a = \rho g QS$ where ωa is the stream power, ρ is the density of water (1000 kg/m3), g is acceleration due to gravity (9.8 m/s2), Q is discharge (m3/s), and S is the channel slope.

Stream Network—*See* River Network. A stream network is the same as river network, but typically refers to a smaller spatial scale.

Stream Reach: See Reach.

Storm: A precipitation event that produces an increase in streamflow.

Stormflow: The part of flow through a channel that occurs in direct response to precipitation; it includes surface and subsurface sources of flow. *See* Direct Runoff.

Stream Order (Strahler): A method for stream classification based on relative position within a river network, when streams lacking upstream tributaries (*i.e.*, headwater streams) are first-order streams and the junction of two streams of the same order results in an increase in stream order (*i.e.*, two first-order streams join to form a second-order stream, two second-order streams join to form a third-order stream, and so on). When streams of different order join, the order of the larger stream is retained. Stream-order classifications can differ, depending on the map scale used to determine order.

Streamflow: Flow of water through a stream or river channel. See Discharge.

Subsurface Water: All water that occurs below the land surface.

Surface Runoff: See Overland Flow.

Surface Water: Water that occurs on Earth's surface (e.g., springs, streams, rivers, lakes, wetlands, estuaries, oceans).

Surface-water Bodies: Types of water bodies that comprise surface water. See Surface Water.

Swale (Science Report): A nonchannelized, shallow trough-like depression that carries water mainly during rainstorms or snowmelt. A swale might or might not be considered a wetland depending on whether it meets the Cowardin *et al.* (1979) three-attribute wetland criteria. *See* Wetland.

Symmetry Ratio: The size ratio of a minor tributary (*T*2) to a major tributary (*T*1) at a confluence. Discharge (Q2/Q1), drainage area (A2/A1), or channel width (W2/W1) can be used to characterize the ratio of tributary size.

Terminal Source Stream: A first-order stream that intersects another first-order stream.

Terrace: An historic or prehistoric floodplain that has been abandoned by its river and is not currently in the active floodplain. *See* Floodplain.

Tracer: A substance that can be used to track the source and movement of water and other substances.

Transformation Function: Any function within a stream or wetland that converts a material into a different form; the amount of the base material is unchanged (base exports equal base imports), but the mass of the different forms can vary.

Transmission Loss: The loss of runoff water by infiltration into stream and river channel beds as water moves downstream; this process is common in arid and semiarid environments.

Transport Mechanism: Any physical mechanism, such as moving water, wind, or movement of organisms, which can transport materials or energy. As used in this report, the term specifically refers to physical mechanisms that move material or energy between streams or wetlands and downstream waters.

Tributary (Science Report): A stream or river that flows into a higher order stream or river.

Unconfined Aquifer: An aquifer that has a water table; the aquifer is not bounded by lower permeability layers. *See* Confined Aquifer.

Unsaturated Zone: Also referred to as the vadose zone. The zone between land surface and the water table within which the moisture content is less than saturation and pressure is less than atmospheric. Soil pore spaces also typically contain air or other gases. *See* Saturated Zone.

Uplands: (1) Higher elevation lands surrounding streams and their floodplains. (2) Within the wetland literature, specifically refers to any area that is not a water body and does not meet the Cowardin *et al.* (1979)-attribute wetland definition. *See* Wetland.

Uptake Length (for dissolved nitrogen in streams): The distance traveled in the water column before algal and microbial assimilation occurs.

Valley: A depression of the earth's surface that drains water between two upland areas.

Variable Source Area: Neither stormflow nor baseflow is uniformly produced from the entire surface or subsurface area of a basin. Instead, the flow of water in a stream at any given moment is influenced by dynamic, expanding or shrinking source areas, normally representing only a few percent of the total basin areas. The source area is highly variable during stormflow. During large rainfall or snowmelt events, the flowing portions of the river network, and associated source areas, expand. As the event ends, the network and source areas contract.

Vernal Pool: Shallow seasonal wetlands that generally accumulate water during colder, wetter months and gradually dry down during warmer, dryer months.

Water Balance: The accounting of the volume of water that enters, leaves, and is stored in a hydrologic unit, area, or arbitrarily defined control volume, typically a drainage basin or aquifer, during a specified period of time.

Water Body: Any sizable accumulation of water on the land surface, including streams, rivers, lakes, and wetlands.

Water Surplus: Water that is available for streamflow or recharge of ground water; precipitation minus evapotranspiration.

Water Table: The top of the zone of saturation of an unconfined aquifer.

Watershed: The area drained by a stream, river, or other water body; typically defined by the topographic divides between one water body and another. *Synonymous with* Catchment *and* Drainage Basin.

Wet Channel: Channel with flowing or standing water.

Wetland (Science Report): An area that generally exhibits at least one of the following three attributes (Cowardin *et al.*, 1979): (1) is inundated or saturated at a frequency sufficient to support, at least periodically, plants adapted to a wet environment; (2) contains undrained hydric soil; or (3) contains nonsoil saturated by shallow water for part of the growing season.

Wetlands (Clean Water Act): The term *wetlands* means areas that are nundated or saturated by surface or ground water at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include swamps, marshes, bogs, and similar areas.

Wetland Storage: The capacity of a wetland to detain or retain water from various sources.

Appendix B: References

Note that the below references are only those that were cited in this technical support document. The references for the Science Report are available in that Report, which is available in the Docket for the proposed rule and on EPA's website at <u>http://cfpub.epa.gov/ncea/cfm/recordisplay.cfm?deid=296414</u>. In addition, EPA's Office of Research and Development considered additional peer-reviewed literature for the completion of the Science Report and in the review of literature published since the Science Report's publication. The references considered for the literature update are available in Appendix C. The agencies also solicited comment on whether additional citations published since the Science Report's publication (*i.e.*, since 2014) should be considered by the agencies. The agencies' review of those references is available in Appendix C3.

- Abbe, T.E., and D.R. Montgomery. 1996. "Large Woody Debris Jams, Channel Hydraulics and Habitat Formation in Large Rivers." *Regulated Rivers: Research & Management* 12:201-221.
- Abbott, B.W., *et al.* 2018. "Unexpected spatial stability of water chemistry in headwater stream networks." *Ecology Letters* 21: 296-308.
- Abbott, M.D., *et al.* 2000. "18O, D, and 3H Measurements Constrain Groundwater Recharge Patterns in an Upland Fractured Bedrock Aquifer, Vermont, USA." *Journal of Hydrology* 228:101-112.
- Acreman, M., *et al.* 2021. "Evidence for the effectiveness of nature-based solutions to water issues in Africa.: *Environmental Research Letters* 16: 063007.
- Adams, S.B., *et al.* 2001. "Geography of Invasion in Mountain Streams: Consequences of Headwater Lake Fish Introductions." *Ecosystems* 4(4): 296-307.
- Adams, S.B., *et al.* 2005. "Instream Movements by Boreal Toads (*Bufo boreas boreas*)." *Herpetological Review* 36:27–33.
- Addy, K., *et al.* 2019. "Connectivity and Nitrate Uptake Potential of Intermittent Streams in the Northeast USA." *Frontiers in Ecology and Evolution* 7: 225.
- Albert, D.A. 2000. *Borne of the Wind: An Introduction to the Ecology of Michigan Sand Dunes*. Michigan Natural Features Inventory, Lansing, MI.
- Albert, D.A. 2003. *Between Land and Lake: Michigan's Great Lakes Coastal Wetlands*. Bulletin E-2902. Michigan Natural Features Inventory, Michigan State University Extension, East Lansing, MI.
- Albert, D.A. 2007. *Natural Community Abstract for Interdunal Wetland*. Michigan Natural Features Inventory, Lansing, MI.
- Alexander, L.C., *et al.* 2018. "Featured Collection Introduction: Connectivity of Streams and Wetlands to Downstream Waters." *Journal of the American Water Resources Association* 54: 287-297.

- Alexander, R., *et al.* 2009. "Dynamic Modeling of Nitrogen Losses in River Networks Unravels the Coupled Effects of Hydrological and Biogeochemical Processes." *Biogeochemistry* 93:91-116.
- Alexander, R.B., et al. 2007. "The Role of Headwater Streams in Downstream Water Quality." Journal of the American Water Resources Association 43: 41-59.
- Alexander, R. B., *et al.* 2008. "Differences in phosphorus and nitrogen delivery to the Gulf of Mexico from the Mississippi River Basin." *Environmental Science & Technology* 42(3): 822-830.
- Alexander, R.G., *et al.* 2000. "Effect of Stream Channel Size on the Delivery of Nitrogen to the Gulf of Mexico." *Nature* 403: 758-761.
- Alford, J.D., and M.R. Walker. 2013. "Managing the Flood Pulse for Optimal Fisheries Production in the Atchafalaya River Basin, Louisiana (USA)." *River Research and Applications* 29: 279-296.
- Ali, G., and C. English. 2019. "Phytoplankton blooms in Lake Winnipeg linked to selective watergatekeeper connectivity." *Scientific Reports* 9(1): 8395. <u>https://doi.org/10.1038/s41598-019-44717-y</u>.
- Allan, J.D. 1995. *Ecology Structure and Function of Running Waters*. Chapman & Hall, New York, NY.
- Allan, J.D. 2004. "Landscapes and Riverscapes: The Influence of Land Use on Stream Ecosystems." Annual Review of Ecology Evolution and Systematics 35: 257-284.
- Allen, G.H., *et al.* 2018. "Similarity of stream width distributions across headwater systems." *Nature Communications* 9: 610.
- Allen, M.R. 2007. "Measuring and Modeling Dispersal of Adult Zooplankton." Oecologia 153: 135-143.
- Amezaga, J.M., *et al.* 2002. "Biotic Wetland Connectivity Supporting a New Approach for Wetland Policy." *Acta Oecologica-International Journal of Ecology* 23:213-222.
- Ameli, A.A., and I.F. Creed. 2017. "Quantifying hydrologic connectivity of wetlands to surface water systems." *Hydrology and Earth System Sciences*. 21: 1791-1808.
- Ameli, A.A., and I.F. Creed. 2019. "Does Wetland Location Matter When Managing Wetlands for Watershed-Scale Flood and Drought Resilience?" *Journal of the American Water Resources Association* 55: 529-542.
- American Society of Civil Engineers. 1996. *Hydrology Handbook*. 2nd Edition. ASCE Manuals and Reports on Engineering Practice No. 28. Task Committee on Hydrology Handbook. ASCE Publications, New York, NY.
- Amoros, C., and G. Bornette. 2002. "Connectivity and Biocomplexity in Waterbodies of Riverine Floodplains." *Freshwater Biology* 47:761-776.
- Anandhi, A., et al. 2018. "Hydrologic Characteristics of Streamflow in the Southeast Atlantic and Gulf Coast Hydrologic Region during 1939–2016 and Conceptual Map of Potential Impacts." Hydrology 5(3): 42. <u>https://doi.org/10.3390/hydrology5030042</u>.

- Anderson, C.D. 2010. "Considering Spatial and Temporal Scale in Landscape-Genetic Studies of Gene Flow." *Molecular Ecology* 19:3565-3575.
- Anderson, N.H., and J.R. Sedell. 1979. "Detritus Processing by Macroinvertebrates in Stream Ecosystems." *Annual Review of Entomology* 24:351-377.
- Anderson, R.L., *et al.* 2020. "Influence of a high-head dam as a dispersal barrier to fish community structure of the Upper Mississippi River." *River Research and Applications* 36(1): 47-56
- Angeler, D.G., *et al.* 2010. "Phytoplankton Community Similarity in a Semiarid Floodplain under Contrasting Hydrological Connectivity Regimes." *Ecological Research* 25:513-520.
- Anning, D.W., and J.T.C. Parker. 2009. Predictive models of the hydrological regime of unregulated streams in Arizona. U.S. Geological Survey Open-File Report 2009-1269. U.S. Geological Survey, Reston, Virginia. 33 p. and spreadsheets. Available at <u>https://pubs.usgs.gov/of/2009/1269/</u>.
- Arguez, A., et al. 2019. "ENSO Normals: A New U.S. Climate Normals Product Conditioned by ENSO Phase and Intensity and Accounting for Secular Trends." *Journal of Applied Meteorology and Climatology 58*(6): 1381-1397. <u>https://journals.ametsoc.org/view/journals/apme/58/6/jamc-d-18-0252.1.xml</u>.
- Arheimer, B., and B.C. Pers. 2017. "Lessons learned? Effects of nutrient reductions from constructing wetlands in 1996–2006 across Sweden." *Ecological Engineering* 103: 404-414.
- Association of State Dam Safety Officials. 2021. "Topic: Seepage through Earthen Dams." Dam Ownership Fact Sheet. 2 pp. Accessed at <u>https://damsafety-prod.s3.amazonaws.com/s3fs-public/files/2021_Seepage%20Through%20Earthen%20Dams.pdf</u>.
- Association of State Wetland Mangers. 2019. Comments of the Association of State Wetland Managers on the Proposed Rule Revised Definition of "Waters of the United States." Docket ID No. EPA-HQ-OW-2018-0149-4897. <u>https://www.regulations.gov/comment/EPA-HQ-OW-2018-0149-4897</u>.
- Attum, O., *et al.* 2007. "Upland–wetland Linkages: Relationship of Upland and Wetland Characteristics with Watersnake Abundance." *Journal of Zoology* 271:134-139.
- Augspurger, C., *et al.* 2008. "Tracking Carbon Flow in a 2-Week-Old and 6-Week-Old Stream Biofilm Food Web." *Limnology and Oceanography* 53:642-650.
- Axtmann, E.V., and S.N. Luoma. 1991. "Large-scale Distribution of Metal Contamination in the Finegrained Sediments of the Clark Fork River, Montana, USA," *Applied Geochemistry* 6:75-88.
- Azinheira, D.L., *et al.* 2014. "Comparison of effects of inset floodplains and hyporheic exchange induced by in-stream structures on solute retention." *Water Resources Research* 50: 6168-6190.
- Babbitt, K.J., and G.W. Tanner. 2000. "Use of Temporary Wetlands by Anurans in a Hydrologically Modified Landscape." *Wetlands* 20:313–322.
- Babbitt, K.J., *et al.* 2003. "Patterns of Larval Amphibian Distribution along a Wetland Hydroperiod Gradient." *Canadian Journal of Zoology-Revue Canadienne De Zoologie* 81:1539-1552.

- Baber, M.J., *et al.* 2002. "Controls on Fish Distribution and Abundance in Temporary Wetlands." *Canadian Journal of Fisheries and Aquatic Sciences* 59:1441-1450.
- Bachmair, S., and M. Weiler. 2014. "Interactions and connectivity between runoff generation processes of different spatial scales." *Hydrological Processes* 28: 1916-1930.
- Baillie, M., *et al.* 2007. "Quantifying Water Sources to a Semiarid Riparian Ecosystem, San Pedro River, Arizona." *Journal of Geophysical Research* 112: G03S02.
- Baldwin, D.S., and A.M. Mitchell. 2000. "The Effects of Drying and Re-flooding on the Sediment and Soil Nutrient Dynamics of Lowland River-Floodplain Systems: A Synthesis." *Regulated Rivers: Research & Management* 16:457-467.
- Bam, E., *et al.* 2020. "Ephemeral Ponds: Are They the Dominant Source of Depression-Focused Groundwater Recharge?" *Water Resources Research* 56.
- Baranyi, C., *et al.* 2002. "Zooplankton Biomass and Community Structure in a Danube River Floodplain System: Effects of Hydrology." *Freshwater Biology* 47:473-482.
- Baron, J. S., *et al.* 2002. "Meeting ecological and societal needs for freshwater." *Ecological Applications* 12:1247-1260.
- Baron, J.S., *et al.* 2013. "The interactive effects of excess reactive nitrogen and climate change on aquatic ecosystems and water resources of the United States." *Biogeochemistry* 114: 71-92.
- Barrett, P. 1979. "Interaction of Bryophytes and Thermal Cracking in the Genesis of Hummock and String-like Microtopography in High Arctic Tundra Meadows." *Polarforschung* 49 (1):70-79.
- Battauz, Y.S., *et al.* 2017. "Macrophytes as dispersal vectors of zooplankton resting stages in a subtropical riverine floodplain." *Aquatic Ecology* 51: 191-201.
- Battin, T.J., *et al.* 2009. "Biophysical Controls on Organic Carbon Fluxes in Fluvial Networks." *Nature Geoscience* 1:95-100.
- Baxter, C.V., and F.R. Hauer. 2000. "Geomorphology, Hyporheic Exchange and Selection of Spawning Habitat by Bull Trout (*Salvelinus confluentus*)." *Canadian Journal of Fisheries and Aquatic Sciences* 57: 1470-1481.
- Beesley, L.S., *et al.* 2020. "New insights into the food web of an Australian tropical river to inform water resource management." *Scientific Reports* 10: 14294.
- Beeson, C.E., and P.F. Doyle. 1995. "Comparison of bank erosion at vegetated and non-vegetated channel bends." *Journal of the American Water Resources Association* 31:983-990.
- Bencala, K.E. 1993. "A Perspective on Stream-Catchment Connections." *Journal of the North American Benthological Society* 12:44-47.
- Bencala, K.E. 2005. "Hyporheic Exchange Flows." Pages 1733-1740 in *Encyclopedia of Hydrological Sciences*. M.G. Anderson, editor. John Wiley and Son, Ltd., New York, NY.

- Bencala, K.E. 2011. "Stream-Groundwater Interactions." Pages 537-546 in *Treatise on Water Science*. P. Wilderer, editor. Academic Press, Oxford, UK.
- Bencala, K.E., *et al.* 2011. "Rethinking Hyporheic Flow and Transient Storage to Advance Understanding of Stream-Catchment Connections." *Water Resources Research* 47:W00H03.
- Benda, L. 2008. "Confluence Environments at the Scale of River Networks." Pages 271-300 in *River Confluences, Tributaries and the Fluvial Network*. S.P. Rice, *et al.*, editors. John Wiley & Sons, Chichester, UK.
- Benda, L., *et al.* 2004a. "Confluence effects in rivers: Interactions of basin scale, network geometry, and disturbance regimes." *Water Resources Research* 40.
- Benda, L., *et al.* 2004b. "The Network Dynamics Hypothesis: How Channel Networks Structure Riverine Habitats." *BioScience* 54: 413-427.
- Benda, L.E., and T.W. Cundy. 1990. "Predicting Deposition of Debris Flows in Mountain Channels." *Canadian Geotechnical Journal* 27:409-417.
- Benda, L.E., *et al.* 2005. "Geomorphology of Steepland Headwaters: The Transition from Hillslopes to Channels." *Journal of the American Water Resources Association* 41:835-851.
- Benke, A.C., *et al.* 2000. "Flood Pulse Dynamics of an Unregulated River Floodplain in the Southeastern US Coastal Plain." *Ecology* 81:2730-2741.
- Bent, G.C., and P.A. Steeves. 2006. A revised logistic regression equation and an automated procedure for mapping the probability of a stream flowing perennially in Massachusetts. U.S. Geological Survey Scientific Investigations Report 2006–5031. U.S. Geological Survey, Reston, Virginia. 107 p. Available at: https://pubs.usgs.gov/sir/2006/5031/pdfs/sir2006-5031_text-appendix1_508.pdf.
- Bergey, E.A., *et al.* 2008. "Springs in Time: Fish Fauna and Habitat Changes in Springs over a 20-Year Interval." *Aquatic Conservation: Marine and Freshwater Ecosystems* 18:829-838.
- Bergstrom, A., *et al.* 2016. "Spatiotemporal processes that contribute to hydrologic exchange between hillslopes, valley bottoms, and streams." *Water Resources Research* 52: 4628-4645.
- Berkowitz, J.F. 2011. Updating regional supplements to the Corps of Engineers Wetland Delineation Manual. WRAP Technical Notes Collection. ERDC TNWRAP-12-1. U.S. Army Engineer Research and Development Center, Vicksburg, MS.
- Bernal, B., and W. Mitsch. 2013. "Carbon sequestration in freshwater wetlands in Costa Rica and Botswana." *Biogeochemistry* 115: 77-93.
- Bernhardt, E.S., *et al.* 2002. "Whole-system Estimates of Nitrification and Nitrate Uptake in Streams of the Hubbard Brook Experimental Forest." *Ecosystems* 5:419-430.
- Bernhardt, E.S., *et al.* 2005. "Can't See the Forest for the Stream? In-stream Processing and Terrestrial Nitrogen Exports." *BioScience* 55: 219-230.
- Bestgen, A.C., *et al.* 2000. "Flood Pulse Dynamics of an Unregulated River floodplain in the Southeastern U.S. Coastal Plain." *Ecology* 81:2730-2741.
- Beven, K., and P. Germann. 1982. "Macropores and Water Flow in Soils." *Water Resources Research* 18:1311-1325.
- Bhadha, J., *et al.* 2011. "Phosphorus Mass Balance and Internal Load in an Impacted Subtropical Isolated Wetland." *Water, Air, & Soil Pollution* 218:619-632.
- Bigelow, P.E., *et al.* 2007. "On Debris Flows, River Networks, and the Spatial Structure of Channel Morphology." *Forest Science* 53:220-238.
- Biggs, J., *et al.* 2017. "The importance of small waterbodies for biodiversity and ecosystem services: implications for policy makers." *Hydrobiologia* 793: 3-39.
- Bilton, D.T., et al. 2001. "Dispersal in Freshwater Invertebrates" Annual Review of Ecology and Systematics 32:159-81.
- Blackburn-Lynch, W., et al. 2017. "Development of Regional Curves for Hydrologic Landscape Regions (HLR) in the Contiguous United States." Journal of the American Water Resources Association 53(4): 903-928. <u>https://doi.org/10.1111/1752-1688.12540</u>.
- Blanco-Canqui, H. and R. Lal. 2008. *Principles of Soil Conservation and Management*. Springer, Netherlands.
- Blann, K.L., et al. 2009. "Effects of Agricultural Drainage on Aquatic Ecosystems: A Review." Critical Reviews in Environmental Science and Technology 39:909-1001.
- Blinn, D.W., and N.L. Poff. 2005. "Colorado River Basin." Pages 483-526 in *Rivers of North America*.A.C. Benke and C. E. Cushing, editors. Elseviar Academic Press, Amsterdam, The Netherlands.
- Blum, M.D., and H. H. Roberts. 2009. "Drowning of the Mississippi Delta Due to Insufficient Sediment Supply and Global Sea-Level Rise." *Nature Geoscience* 2(7): 488-491.
- Bodamer, B.L., and J.M. Bossenbroek. 2008. "Wetlands as Barriers: Effects of Vegetated Waterways on Downstream Dispersal of Zebra Mussels." *Freshwater Biology* 53:2051-2060.
- Böhlke, J., et al. 2009. "Multi-Scale Measurements and Modeling of Denitrification in Streams with Varying Flow and Nitrate Concentration in the Upper Mississippi River Basin, USA." *Biogeochemistry* 93:117-141.
- Bohonak, A.J., and D.G. Jenkins. 2003. "Ecological and Evolutionary Significance of Dispersal by Freshwater Invertebrates." *Ecology Letters* 6:783-796.
- Boltz, J.M., and R.R.J. Stauffer. 1989. "Fish Assemblages of Pennsylvania Wetlands." Pages 158-170 in Wetland Ecology and Conservation: Emphasis in Pennsylvania. S.K. Majumdar, editor. The Pennsylvania Academy of Sciences, Lafayette College, Easton, PA.

- Boon, P.I. 2006. "Biogeochemistry and Bacterial Ecology of Hydrologically Dynamic Wetlands." Pages 115-176 in Ecology of Freshwater and Estuarine Wetlands. D.P. Batzer and R.R. Sharitz, ed. University of California Press, Berkeley, CA.
- Booth, D.B., *et al.* 2002. "Forest Cover, Impervious-Surface Area, and the Mitigation of Stormwater Impacts." *Journal of the American Water Resources Association* 38:835-845.
- Borchert, J.R. 1950. "The Climate of the Central North American Grassland." Annals of the Association of American Geographers 40:1-39.
- Borchert, S.M., *et al.* 2018. "Coastal wetland adaptation to sea level rise: Quantifying potential for landward migration and coastal squeeze." *Journal of Applied Ecology* 55: 2876-2887. https://doi.org/10.1111/1365-2664.13169.
- Borja, S., *et al.* 2020. "Global Wetting by Seasonal Surface Water Over the Last Decades." *Earth's Future* 8 (3): e2019EF001449. https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019EF001449.
- Boto, K.G., and W.H. Patrick. 1979. "Role of Wetlands in the Removal of Suspended Sediments." Pages 479-489 in Wetland Functions and Values: The State of Our Understanding. Proceedings of National Symposium on Wetlands. P.E. Greeson, et al, editors. American Water Resources Association, Minneapolis, MN.
- Boughton, E.H., *et al.* 2010. "Land-use and Isolation Interact to Affect Wetland Plant Assemblages." *Ecography* 33:461-470.
- Boulton, A.J., *et al.* 1998. "The Functional Significance of the Hyporheric Zone in Streams and Rivers." *Annual Review of Ecology and Systematics* 29:59-81.
- Bracken, L.J., and J. Croke. 2007. "The Concept of Hydrological Connectivity and Its Contribution to Understanding Runoff-Dominated Geomorphic Systems." *Hydrological Processes* 21:1749-1763.
- Bracken, L.J., *et al.* 2013. "Concepts of hydrological connectivity: Research approaches, pathways and future agendas." *Earth-Science Reviews* 119:17-34.
- Bradford, M.J., *et al.* 2001. "Ecology of Juvenile Chinook Salmon in a Small Non-natal Stream of the Yukon River Drainage and the Role of Ice Conditions on Their Distribution and Survival." *Canadian Journal of Zoology-Revue Canadienne De Zoologie* 79: 2043-2054.
- Brady, N.C., and R.R. Weil. 2002. *The Nature and Properties of Soils*. 13th Edition. Prentice Hall, Upper Saddle River, NJ.
- Bramblett, R.G., *et al.* 2002. "Seasonal use of small tributary and main-stem habitats by juvenile steelhead, coho salmon, and Dolly Varden in a southeastern Alaska drainage basin." *Transactions of the American Fisheries Society* 131: 498-506.

- Brander, L.M., *et al.* 2012a. "Ecosystem service values for mangroves in Southeast Asia: A meta-analysis and value transfer application." *Ecosystem Services* 1(1), 62-69. <u>https://doi.org/10.1016/j.ecoser.2012.06.003</u>.
- Brander, L.M., *et al.* 2012b. "Using meta-analysis and GIS for value transfer and scaling up: valuing climate change induced losses of European wetlands." *Environmental and resource economics* 52(3), 395-413. https://doi.org/10.1007/s10640-011-9535-1.
- Branfireun, B.A., *et al.* 1999. "In situ Sulphate Stimulation of Mercury Methylation in a Boreal Peatland: Toward a Link Between Acid Rain and Methylmercury Contamination in Remote Environments." *Global Biogeochemical Cycles* 13:743-750.
- Brazier, R.E., *et al.* 2021. "Beaver: Nature's ecosystem engineers." *WIREs Water* 8:e1494. <u>https://doi.org/10.1002/wat2.1494</u>.
- Brennan, S.R., *et al.* 2019. "Shifting habitat mosaics and fish production across river basins." *Science* 364: 783-786.
- Brinson, M.M. 1993. *A Hydrogeomorphic Classification for Wetlands*. U.S. Army Corps of Engineers, Washington, D.C.
- Brittain, J.E., and T.J. Eikeland. 1988. "Invertebrate Drift: A Review." Hydrobiologia 166:77-93.
- Brooks, B.W., *et al.* 2006. "Water quality of effluent-dominated ecosystems: Ecotoxicological, hydrological, and management considerations." *Hydrobiologia* 556:365-379.
- Brooks, J.R., *et al.* 2012. "Willamette River Basin Surface Water Isoscape (δ18O and δ2H): Temporal Changes of Source Water within the River." *Ecosphere* 3:39.
- Brooks, J.R., *et al.* 2018. "Estimating Wetland Connectivity to Streams in the Prairie Pothole Region: An Isotopic and Remote Sensing Approach." *Water Resources Research* 54: 955-977.
- Brooks, P.D., and M.M. Lemon. 2007. "Spatial Variability in Dissolved Organic Matter and Inorganic Nitrogen Concentrations in a Semiarid Stream, San Pedro River, Arizona." *Journal of Geophysical Research-Biogeosciences* 112:G03S05.
- Brooks, R.P., and T.L. Serfass. 2013. "Wetland-riparian wildlife of the Mid-Atlantic Region: An overview. Pages 259-268 in *Mid-Atlantic freshwater wetlands: Advances in wetlands science, management, policy, and practice*. R. P. Brooks and D. H. Wardrop, editors. Springer, New York, NY.
- Brooks, R.R., et al. 1977. "Cobalt and Nickel Uptake by the Nyssaceae." Taxon 26:197-201.
- Brown, A.V., and W.J. Matthews. 1995. "Stream Ecosystems of the Central United States." Pages 89-116 in *River and Stream Ecosystems*. C. E. Cushing, *et al.*, editors. Elsevier Science, Amsterdam, The Netherlands.
- Brummer, C.J., *et al.* 2006. "Influence of Vertical Channel Change Associated with Wood Accumulations on Delineating Channel Migration Zones, Washington, USA." *Geomorphology* 80:295-309.

- Brunke, M., and T.O.M. Gonser. 1997. "The ecological significance of exchange processes between rivers and groundwater." *Freshwater Biology* 37:1-33.
- Bryant, J.C., and R.H. Chabreck. 1998. "Effects of Impoundment on Vertical Accretion of Coastal Marsh." *Estuaries* 21(3): 416–422. <u>https://doi.org/10.2307/1352840</u>.
- Bugna, G.C., *et al.* 2020. "Sensitivity of using stable water isotopic tracers to study the hydrology of isolated wetlands in North Florida." *Journal of Hydrology* 580: 124321.
- Bukaveckas, P.A. 2007. "Effects of Channel Restoration on Water Velocity, Transient Storage, and Nutrient Uptake in a Channelized Stream." *Environmental Science & Technology* 41(5): 1570-1576. <u>https://doi.org/10.1021/es061618x</u>.
- Bullock, A., and M. Acreman. 2003. "The Role of Wetlands in the Hydrological Cycle." *Hydrology and Earth System Sciences* 7:358-389.
- Bunn, S.E., and A.H. Arthington. 2002. "Basic principles and ecological consequences of altered flow regimes for aquatic biodiversity." *Environmental Management* 30:492-507.
- Bunn, S.E., *et al.*, 2006. "Flow Variability in Dryland Rivers: Boom, Bust and the Bits in Between." *River Research and Applications* 22:179-186.
- Burkart, M.R., *et al.* 1999. "Tributary Stream Infiltration as a Source of Herbicides in an Alluvial Aquifer." *Journal of Environmental Quality* 28:69-74.
- Burt, T.P., *et al.* 2002. "Water Table Fluctuations in the Riparian Zone: Comparative Results from a Pan-European Experiment." *Journal of Hydrology* 265:129-148.
- Busch, M.H., *et al.* 2020. "What's in a name? Patterns, trends, and suggestions for defining non-perennial rivers and streams." *Water* 12: 1980. <u>https://doi.org/10.3390/w12071980</u>.
- Butcher, G.S., and B. Zimpel. 1991. "Habitat Value of Isolated Waters to Migratory Birds." Prepared by Cornell Laboratory of Ornithology and The Cadmus Group, Inc. for U.S. Environmental Protection Agency Office of Wetlands Protection. Cornell and Cadmus, Washington, D.C.
- Buto, S.G., and R.D. Anderson. 2020. *NHDPlus High Resolution (NHDPlus HR)—A Hydrography Framework for the Nation*. U.S. Geological Survey Fact Sheet 2020–3033. U.S. Geological Survey. https://doi.org/10.3133/fs20203033.
- Buttle, J. 2006. "Mapping first-order controls on streamflow from drainage basins: The T3 template." *Hydrological Processes* 20:3415-3422.
- Cabezas, A. *et al.* 2009. "Changing Patterns of Organic Carbon and Nitrogen Accretion on the Middle Ebro Floodplain (NE Spain)." *Ecological Engineering* 35:1547-1558.
- Cairns, M.A., *et al.* 2005. "Influence of Summer Stream Temperatures on Black Spot Infestation of Juvenile Coho Salmon in the Oregon Coast Range." *Transactions of the American Fisheries Society* 134: 1471-1479.

Caissie, D. 2006. "The Thermal Regime of Rivers: A Review." Freshwater Biology 51: 1389-1406.

- Calhoun, A.J.K., *et al.* 2017. "The Significant Surface-Water Connectivity of 'Geographically Isolated Wetlands'." *Wetlands* 37: 801-806.
- California Division of Safety of Dams. 1993. *Guidelines for the Design and Construction of Small Embankment Dams*. Department of Water Resources, Sacramento, CA. Accessed at https://www.countyofnapa.org/DocumentCenter/View/6157/California-Department-of-Water-Resources-Division-of-Safety-of-Dams-Guidelines-Small-Dams-PDF.
- Callegary, J.B., *et al.* 2007. "Rapid estimation of recharge potential in ephemeral-stream channels using electromagnetic methods, and measurements of channel and vegetation characteristics." *Journal of Hydrology* 344: 17-31.
- Carlyle, G.C., and A.R. Hill. 2001. "Groundwater Phosphate Dynamics in a River Riparian Zone: Effects of Hydrologic Flowpaths, Lithology, and Redox Chemistry." *Journal of Hydrology* 247: 151-168.
- Carlson, A.K., *et al.* 2016. "Effects of historic flooding on fishes and aquatic habitats in a Missouri River delta." *Journal of Freshwater Ecology* 31: 271-288.
- Cayan, D.R., and D.H. Peterson. 1989. "The Influence of North Pacific Atmospheric Circulation on Streamflow in the West." *Geophysical Monographs* 55:375-397.
- Cederholm, C.J., *et al.* 1999. "Pacific Salmon Carcasses: Essential Contributions of Nutrients and Energy for Aquatic and Terrestrial Ecosystems." *Fisheries* 24(10):6-15.
- Chacon, N., *et al.* 2008. "Seasonal Changes in Soil Phosphorus Dynamics and Root Mass along a Flooded Tropical Forest Gradient in the Lower Orinoco River, Venezuela." *Biogeochemistry* 87:157-168.
- Chaikumbung, M., *et al.* 2016. "The economic value of wetlands in developing countries: A metaregression analysis." *Ecological Economics* 124: 164-174.
- Chapra, S.C. 1996. Surface Water Quality Modeling. McGraw-Hill, New York, NY.
- Cheesman, A.W., *et al.* 2010. "Soil Phosphorus Forms in Hydrologically Isolated Wetlands and Surrounding Pasture Uplands." *Journal of Environmental Quality* 39:1517-1525.
- Chen, X., and X. Chen. 2003. "Stream Water Infiltration, Bank Storage, and Storage Zone Changes Due to Stream-Stage Fluctuations." *Journal of Hydrology* 280: 246-264.
- Cheng, F.Y., and N.B. Basu. 2017. "Biogeochemical hotspots: Role of small water bodies in landscape nutrient processing." *Water Resources Research* 53: 5038-5056.
- Cheng, F.Y., K.J. Van Meter, D.K. Byrnes, and N.B. Basu. 2020. "Maximizing US nitrate removal through wetland protection and restoration." *Nature* 588: 625-630. https://doi.org/10.1038/s41586-020-03042-5.
- Chezik, K.A., *et al.* 2017. "River networks dampen long-term hydrological signals of climate change." *Geophysical Research Letters* 44: 7256-7264.

- Chiu, M.C., *et al.* 2020. "Branching networks can have opposing influences on genetic variation in riverine metapopulations." *Diversity and Distributions* 26: 1813-1824.
- Christensen, J.R., *et al.* 2013. "Identifying Riparian Buffer Effects on Stream Nitrogen in Southeastern Coastal Plain Watersheds." *Environmental Management* 52: 1161-1176.
- Chu, X., *et al.* 2013. "Dynamic puddle delineation and modeling of puddle-to-puddle filling-spillingmerging-splitting overland flow processes." *Water Resources Research* 49: 3825–3829. <u>https://doi.org/10.1002/wrcr.20286</u>.
- Church, M. 2002. "Geomorphic Thresholds in Riverine Landscapes." Freshwater Biology 47: 541-557.
- Church, M. 2006. "Bed Material Transport and the Morphology of Alluvial River Channels." *Annual Review of Earth and Planetary Sciences*: 325-354.
- Clair, T.A., *et al.* 1994. "Exports of Carbon and Nitrogen from River Basins in Canada's Atlantic Provinces." *Global Biogeochemical Cycles* 8: 441-450.
- Clark, W.R. 2000. "Ecology of Muskrats in Prairie Wetlands." Pages 287-313 in Prairie Wetland Ecology: The Contribution of the Marsh Ecology Research Program. H. R. Murkin, et al., editors. Iowa State University Press, Ames, IA.
- Cohen, M., J. Paris, and M. Clark. 2007. "P-sorption capacity estimation in southeastern USA wetland soils using visible/near infrared (VNIR) reflectance spectroscopy." *Wetlands* 27: 1098-1111.
- Cohen, M.J., *et al.* 2016. "Do geographically isolated wetlands influence landscape functions?" *Proceedings of the National Academy of Sciences* 113: 1978-1986.
- Collen, P., and R.J. Gibson. 2001. "The general ecology of beavers (*Castor* spp.), as related to their influence on stream ecosystems and riparian habitats, and the subsequent effects on fish a review." *Reviews in Fish Biology and Fisheries* 10: 439-461.
- Collins, B.D., et al. 2012. "The Floodplain Large-Wood Cycle Hypothesis: A Mechanism for the Physical and Biotic Structuring of Temperature Forested Alluvial Valleys in the North Pacific Coastal Ecoregion. Geomorphology 139-140: 460-470.
- Colson, T.C., *et al.* 2008. "Topographic and Soil Maps Do Not Accurately Depict Headwater Stream Networks." *National Wetlands Newsletter* 30: 3.
- Colvin, R., *et al.* 2009. "Fish Use of Intermittent Watercourses Draining Agricultural Lands in the Upper Willamette River Valley, Oregon." *Transactions of the American Fisheries Society* 138: 1302-1313. https://doi.org/10.1577/T08-150.1.
- Colvin, S.A.R., *et al.* 2019. "Headwater Streams and Wetlands are Critical for Sustaining Fish, Fisheries, and Ecosystem Services." *Fisheries* 44: 73-91.
- Comer, P., et al. 2005. Biodiversity Values of Geographically Isolated Wetlands in the United States. NatureServe, Arlington, VA. Accessed at https://www.natureserve.org/sites/default/files/biodiversity_values_of_giw_in_the_us.pdf.

- Committee on Environment and Natural Resources. 2000. *Integrated Assessment of Hypoxia in the Northern Gulf of Mexico*. National Science and Technology Council, Washington, D.C.
- Congalton, R.G., and K. Green. 1990. Assessing the Accuracy of Remotely Sensed Data: Principles and Practices. Lewis Publishers, Washington, D.C.
- Consortium of Aquatic Science Societies. 2021. Pre-proposal Recommendation to Docket EPA-HQ-OW-2021-0328. Docket ID No. EPA-HQ-OW-2021-0328-0205. https://www.regulations.gov/comment/EPA-HQ-OW-2021-0328-0205. ("CASS 2021.")
- Convention on Wetlands. 2021. *Global Wetland Outlook: Special Edition 2021*. Secretariat of the Convention on Wetlands, Gland, Switzerland.
- Cook, B.J., and F.R. Hauer. 2007. "Effects of Hydrologic Connectivity on Water Chemistry, Soils, and Vegetation Structure and Function in an Intermontane Depressional Wetland Landscape." Wetlands 27: 719-38.
- Cooper, A., J. W. Gilliam, R. B. Daniels, and W. P. Robarge. 1987. "Riparian areas as filters for agricultural sediment." *Soil Science Society of America Journal* 51:416-420.
- Cooper, A.B. 1990. "Nitrate depletion in the riparian zone and stream channel of a small headwater catchment." *Hydrobiologia* 202: 13-26.
- Copp, G.H. 1989. "The Habitat Diversity and Fish Reproductive Function of Floodplain Ecosystems." *Environmental Biology of Fishes* 26:1-27.
- Corti, R., *et al.* 2011. "Natural Variation in Immersion and Emersion Affects Breakdown and Invertebrate Colonization of Leaf Litter in a Temporary River. *Aquatic Sciences* 73:537-550.
- Costanza, R., *et al.* 1997. "The value of the world's ecosystem services and natural capital." *Nature* 387: 253–260. <u>https://doi.org/10.1038/387253a0</u>.
- Costelloe, J.F., et.al. 2007. "Determining Loss Characteristics of Arid Zone River Waterbodies." River Research and Applications 23:715-731.
- Costigan, K.H., C.M. Ruffing, J.S. Perkin, and M.D. Daniels. 2016. "Rapid Response of a Sand-Dominated River to Installation and Removal of a Temporary Run-of-the-River Dam." *River Research and Applications* 32(1): 110-124.
- Covino, T. 2017. "Hydrologic connectivity as a framework for understanding biogeochemical flux through watersheds and along fluvial networks." *Geomorphology* 277: 133-144.
- Cowardin, L.M., et al. 1979. Classification of Wetlands and Deepwater Habitats of the United States. U.S. Department of the Interior, U.S. Fish and Wildlife Service, Office of Biological Services, Washington, DC.
- Cowdery, T.K., et al. 2019. The hydrologic benefits of wetland and prairie restoration in western Minnesota—Lessons learned at the Glacial Ridge National Wildlife Refuge, 2002–15 (2019–5041). https://pubs.usgs.gov/sir/2019/5041/sir20195041.pdf.

- Crabot, J., *et al.* 2021. "A global perspective on the functional responses of stream communities to flow intermittence." *Ecography* 44: 1511–1523. <u>https://doi.org/https://doi.org/10.1111/ecog.05697</u>.
- Craft, C.B., and C. Chiang. 2002. "Forms and Amounts of Soil Nitrogen and Phosphorus across a Longleaf Pine–Depressional Wetland Landscape." *Soil Science Society of America Journal* 66:1713-1721.
- Creed, I.F., and F.D. Beall. 2009. "Distributed topographic indicators for predicting nitrogen export from headwater catchments." *Water Resources Research* 45.
- Creed, I.F., *et al.* 2015. "The river as a chemostat: fresh perspectives on dissolved organic matter flowing down the river continuum." *Canadian Journal of Fisheries and Aquatic Sciences* 72: 1272-1285.
- Creed, I.F., et al. 2017. "Enhancing protection for vulnerable waters." Nature Geoscience 10: 809-815.
- Cross, F.B., and R.E. Moss. 1987. "Historic Changes in Fish Communities and Aquatic Habitats in Plains Streams of Kansas." Pages 155-165 in *Community and Evolutionary Ecology of North American Stream Fishes*. W.J. Matthews and D.C. Heins, editors. University of Oklahoma Press, Norman, OK.
- Curry, R.A., et al. 1997. "Use of Small Streams by Young Brook Trout Spawned in a Lake." *Transactions of the American Fisheries Society* 126:77-83.
- Curtis, K.E., et al. 2011. Ordinary High Flows and the Stage–Discharge Relationship in the Arid West Region. ERDC/CRREL TR-11-12. U.S. Army Engineer Research and Development Center, Hanover, NH.
- Dahl, T.E. 1990. *Wetlands Losses in the United States 1780's to 1980's*. U.S. Department of the Interior, U.S. Fish and Wildlife Service, Washington, D.C.
- Daniels, R.B., and J.G. Gilliam. 1996. "Sediment and chemical load reduction by grass and riparian filters." *Soil Science Society of America Journal* 60: 246-251.
- Datry T, *et al.* 2014. "Intermittent Rivers: A Challenge for Freshwater Ecology." *BioScience* 64(3): 229–235. https://doi.org/10.1093/biosci/bit027.
- David, M. B., et al. 2010. "Sources of nitrate yields in the Mississippi River Basin." Journal of Environmental Quality, 39(5): 1657-1667.
- Davis, C.L., *et al.* 2017. "Species interactions and the effects of climate variability on a wetland amphibian metacommunity." *Ecological Applications* 27: 285-296.
- Day, D.G. 1978. "Drainage Density Changes during Rainfall." *Earth Surface Processes and Landforms* 3:319-326.
- Day, J.W., et al. 2007. "Restoration of the Mississippi Delta: Lessons from Hurricanes Katrina and Rita." Science 315(5819): 1679-1684.
- Day, R.H., *et al.* 1990. "An inventory of wetland impoundments in the coastal zone of Louisiana, USA: Historical trends." *Environmental Management* 14: 229–240. <u>https://doi.org/10.1007/BF02394040</u>.

- De Giudici, G., *et al.* 2018. "Role of riparian wetlands and hydrological connectivity in the dynamics of stream thermal regimes." *Nordic Hydrology* 49: 634-647.
- De Groot, D., *et al.* 2018. "Wetland ecosystem services." Pages 323-333 in *The wetland book I: Structure and function, management and methods*. Springer, Dordrecht. https://doi.org/10.1007/978-90-481-9659-3_66.
- De Jong, G.D., *et al.* 2015. "Aquatic Invertebrate and Vertebrate Communities of Ephemeral Stream Ecosystems in the Arid Southwestern United States." *The Southwestern Naturalist* 60(4): 349-359.
- Denver, J.M., *et al.* 2014. "Nitrate fate and transport through current and former depressional wetlands in an agricultural landscape, Choptank Watershed, Maryland, United States." *Journal of Soil and Water Conservation* 69: 1-16.
- Detenbeck, N. E., *et al.* 1993. "Wetland Effects on Lake Water Quality in the Minneapolis/St. Paul Metropolitan Area." *Landscape Ecology* 8: 39-61.
- Detenbeck, N.E., *et al.* 1999. "Evaluating Perturbations and Developing Restoration Strategies for Inland Wetlands in the Great Lakes Basin." *Wetlands* 19(4): 789-820.
- Detenbeck, N.E., *et al.* 2005. "Relationship of Stream Flow Regime in the Western Lake Superior Basin to Watershed Type Characteristics." *Journal of Hydrology* 309(1-4): 258-276.
- Devito, K., *et al.* 2005. "A Framework for Broad-Scale Classification of Hydrologic Response Units on the Boreal Plain: Is Topography the Last Thing to Consider?" *Hydrological Processes* 19:1705-1714.
- Devito, K.J., *et al.* 1996. "Groundwater-Surface Water Interactions in Headwater Forested Wetlands of the Canadian Shield." *Journal of Hydrology* 181: 127-147.
- Dewald, T. 2017. Making the Digital Water Flow: The Evolution of Geospatial Surfacewater Frameworks. U.S. Environmental Protection Agency, Washington, D.C. Accessed at https://www.epa.gov/sites/default/files/2021-04/documents/making_the_digital_water_flow.pdf.
- Díaz, R.J., and R. Rosenberg. 2011. "Introduction to Environmental and Economic Consequences of Hypoxia." *International Journal of Water Resources Development* 27 (1):71-82.
- Dierberg, F.E., and P.L. Brezonik. "Nitrogen and Phosphorus Mass Balances in a Cypress Dome Receiving Wastewater." Pages 112-118 in *Cypress Swamps*. K.C. Ewel and H.T. Odum, ed., University Presses of Florida, Gainesville, FL.
- Dieter, *et al.* 2011. "Preconditioning Effects of Intermittent Stream Flow on Leaf Litter Decomposition." *Aquatic Sciences* 73:599-609.
- Dieterich, M., and N.H. Anderson. 1998. "Dynamics of Abiotic Parameters, Solute Removal and Sediment Retention in Summer-Dry Headwater Stream of Western Oregon." *Hydrobiologia* 379: 1-15.
- Dietrich, W.E., and T. Dunne. 1993. "The Channel Head." Pages 175-219 in *Channel Network Hydrology*. K. Beven and M. J. Kirby, editors. John Wiley & Sons, New York, NY.

- Dillaha, T.A., and S.P. Inamdar. 1997. "Buffer Zones as Sediment Traps or Sources." Pages 33-42 in *Buffer Zones: Their Processes and Potential in Water Protection*. N.E. Haycock, *et al.*, ed., Proceedings of the International Conference on Buffer Zones, September 1996. Quest Environmental, Hertfordshire, UK.
- Dillon, P.J., and L.A. Molot. 1997. "Effects of Landscape Form on Export of Dissolved Organic Carbon, Iron, and Phosphorus from Forested Stream Catchments." *Water Resources Research* 33:2591-2600.
- Dodds, W.K., *et al.* 1996a. "Nitrogen Transport from Tallgrass Prairie Watersheds." *Journal of Environmental Quality* 25:973-981.
- Dodds, W.K., *et al.* 1996b. "The Relationship of Floods, Drying, Flow and Light to Primary Production and Producer Biomass in a Prairie Stream." *Hydrobiologia* 333:151-159.
- Dodds, W.K., *et al.* 2004. "Life on the Edge: The Ecology of Great Plains Prairie Streams," *Bioscience* 54:205-216.
- Dole, J.W. 1965. "Summer Movements of Adult Leopard Frogs, *Rana Pipiens Schreber*, in Northern Michigan." *Ecology* 46 (3):236-255.
- Dollinger, J., *et al.* 2015. Managing ditches for agroecological engineering of landscape. A review. *Agronomy for Sustainable Development* 35(3): 999-1020.
- Downing D., *et al.* 2007. "Technical and Scientific Challenges in Implementing *Rapanos*" 'Water of the United States."" *Natural Resources & Environment* 22(1):42-45, 62-63.
- Driscoll, C.T., *et al.* 1995. "The Role of Dissolved Organic Carbon in the Chemistry and Bioavailability of Mercury in Remote Adirondack Lakes." *Water Air and Soil Pollution* 80:499-508.
- Driscoll, J.M., *et al.* 2020. "Spatiotemporal Variability of Modeled Watershed Scale Surface-Depression Storage and Runoff for the Conterminous United States.: *Journal of the American Water Resources Association* 56(1): 16-29.
- Duncan, S.H., *et al.* 1987. "Transport of Road-Surface Sediment through Ephemeral Stream Channels." *Water Resources Bulletin* 23(1): 113-119.
- Dunham, J.B., et al. 2002. "Alien Invasions in Aquatic Ecosystems: Toward an Understanding of Brook Trout Invasions and Potential Impacts on Inland Cutthroat Trout in Western North America." Reviews in Fish Biology and Fisheries 12(4): 373-391.
- Dunne, E.J., et al. 2006. "Phosphorus Release and Retention by Soils of Natural Isolated Wetlands." International Journal of Environment and Pollution 28:496-516.
- Dunne, T., and R.D. Black. 1970. "Partial Area Contributions to Storm Runoff in a Small New England Watershed." *Water Resources Research* 6:1296-1311.
- Dunne, T., and L.B. Leopold. 1978. *Water in Environmental Planning*. W.H. Freeman and Co., San Francisco, CA.

- Dunnivant, F.M., and E. Anders. 2006. A Basic Introduction to Pollutant Fate and Transport: An Integrated Approach with Chemistry, Modeling, Risk Assessment, and Environmental Legislation. John Wiley & Sons, Inc., Hoboken, NJ.
- Durham, B.W., and G.R. Wilde. 2008. "Composition and Abundance of Drifting Fish Larvae in the Canadian River, Texas." *Journal of Freshwater Ecology* 23:273-280.
- Dvorett, D., *et al.* 2012. "Developing a Hydrogeomorphic Wetland Inventory: Reclassifying National Wetlands Inventory Polygons in Geographic Information Systems." *Wetlands* 32(1):83-93.
- Dwivedi, D., *et al.* 2018. "Hot Spots and Hot Moments of Nitrogen in a Riparian Corridor." *Water Resources Research* 54: 205-222.
- Ebersole, J.L., *et. al.* 2003. "Cold Water Patches in Warm Streams: Physicochemical Characteristics and the Influence of Shading." *Journal of the American Water Resources Association* 39:355-368.
- Ebersole, J.L., *et al.* 2015. "Predicting the occurrence of cold-water patches at intermittent and ephemeral tributary confluences with warm rivers." *Freshwater Science* 34: 111-124.
- Ebersole, J.L., *et al.* 2020. "Managing climate refugia for freshwater fishes under an expanding human footprint." *Frontiers in Ecology and the Environment* 18(5): 271–280. https://doi.org/10.1002/fee.2206.
- Eckhardt, B.W., and T.R. Moore. 1990. "Controls on Dissolved Organic Carbon Concentrations in Streams of Southern Quebec." *Canadian Journal of Fisheries and Aquatic Sciences* 47: 1537-1544.
- Elliott, J.M. 1971. "Distances Travelled by Drifting Invertebrates in a Lake District Stream." *Oecologia* 6: 350-379.
- Elliott, J.M. 2003. "A Comparative Study of the Dispersal of 10 Species of Stream Invertebrates." *Freshwater Biology* 48: 1652-1668.
- Elwood, J., *et al.* 1983. "Resource Spiraling: An Operational Paradigm for Analyzing Lotic Ecosystems." Pages 3-23 in *Dynamics of Lotic Ecosystems*. T.D. Fontaine and S.M. Bartell, ed. Ann Arbor Science, Ann Arbor, MI.
- Enanga, E.M., et al. 2016. "Summer storms trigger soil N2O efflux episodes in forested catchments." Journal of Geophysical Research: Biogeosciences 121: 95-108.
- Enanga, E.M., *et al.* 2017. "Nitrous Oxide and Dinitrogen: The Missing Flux in Nitrogen Budgets of Forested Catchments?" *Environmental Science & Technology* 51: 6036-6043.
- Eng, K., *et al.* 2013. "River flow changes related to land and water management practices across the conterminous United States." *Science of the Total Environment* 463–464: 414–422. <u>https://doi.org/10.1016/j.scitotenv.2013.06.001</u>.
- Eng, K., *et al.* 2016. "Sensitivity of Intermittent Streams to Climate Variations in the USA." *River Research and Applications* 32: 885-895. <u>https://doi.org/10.1002/rra.2939</u>.

- Ensign, S.H., and M.W. Doyle. 2006. "Nutrient Spiraling in Streams and River Networks." *Journal of Geophysical Research-Biogeosciences* 111: G04009.
- Erman, D.C., and V.M. Hawthorne. 1976. "The Quantitative Importance of an Intermittent Stream in the Spawning of Rainbow Trout." *Transactions of the American Fisheries Society* 105:675-681.
- Eshelman, K.N., and H.F. Hemond. 1985. "The Role of Organic Acids in the Acid-base Status of Surface Waters at Bickford Watershed, Massachusetts." *Water Resources Research* 21:1503-1510.
- Everson, G. Personal communication. September 21, 2021.
- Evenson, G.R., et al. 2016. "An Improved Representation of Geographically Isolated Wetland in a Watershed-Scale Hydrologic Model." Hydrological Processes 30 (2): 4168-84. <u>https://doi.org/10.1002/hyp.10930</u>.
- Evenson, G.R., *et al.* 2018. "Depressional Wetlands Affect Watershed Hydrological, Biogeochemical, and Ecological Functions." *Ecological Applications* 28: 953-966.
- Evenson, G.R., *et al.* 2021. "Wetland restoration yields dynamic nitrate responses across the Upper Mississippi river basin." *Environmental Research Communications* 3(9): 095002.
- Ewel, K.C. 1997. Water Quality Improvement by Wetlands. Island Press, Washington, D.C.
- Ewel, K.C., and H.T. Odum, ed. 1984. *Cypress Swamps*. (Gainesville, Florida: University of Florida Press).
- Fairfax County. 2003. Perennial Stream Field Identification Protocol. Fairfax County Stormwater Planning Division, Fairfax County, Virginia. 16 pp. Available at: <u>https://www.fairfaxcounty.gov/publicworks/sites/publicworks/files/assets/documents/pdf/03_ps_prot_ocol_ada.pdf</u>.
- Falke, J.A., *et al.* 2011. "The Role of Groundwater Pumping and Drought in Shaping Ecological Futures for Stream Fishes in a Dryland River Basin of the Western Great Plains, USA." *Ecohydrology* 4:682-697.
- Fausch, K.D., and K.R. Bestgen. 1997. "Ecology of Fishes Indigenous to the Central and Southwestern Great Plains." Pages 131-166 in F.L. Knopf and F.B. Samson, ed. *Ecology and Conservation of Great Plains Vertebrates*. Springer-Verlag, New York, NY.
- Fausch, K.D., and R.G. Bramblett. 1991. "Disturbance and Fish Communities in Intermittent Tributaries of a Western Great Plains River." *Copeia* 1991:659-674.
- Fausch, K.D., *et al.* 2002. "Landscapes to Riverscapes: Bridging the Gap between Research and Conservation of Stream Fishes." *BioScience* 52:483-498.
- Federal Geographic Data Committee. 2013. Classification of Wetlands and Deepwater Habitats of the United States, FGDC-STD-004-2013. Second Edition. Federal Geographic Data Committee, Washington DC. Available at <u>https://www.fws.gov/wetlands/documents/Classification-of-Wetlandsand-Deepwater-Habitats- of-the-United-States-2013.pdf.</u>

- Federal Emergency Management Agency. 2020. "Flood Zones." Accessed at <u>https://www.fema.gov/glossary/flood-zones</u>. ("FEMA 2020.")
- Federal Emergency Management Agency. 2021. "Flood Map." Accessed at <u>https://www.fema.gov/flood-maps</u>. ("FEMA 2021.")
- Federal Emergency Management Agency. 1995. The Zone A Manual: Managing Floodplain Development in Approximate Zone A Areas – A Guide for Obtaining and Developing Base (100-Year) Flood Elevations. FEMA 265. <u>https://www.fema.gov/sites/default/files/documents/fema_approx-zone-a-guide.pdf</u>. ("FEMA 1995.")
- Federal Energy Regulatory Commission. 2005. "Chapter 14: Dam Safety Performance Monitoring Program." *Engineering Guidelines for the Evaluation of Hydropower Projects*. <u>https://www.ferc.gov/sites/default/files/2020-04/chap14.pdf</u>.
- Federal Interagency Stream Restoration Working Group. 1998. Stream Corridor Restoration: Principles, Processes and Practices. GPO Item No. 0120-A; SuDocs No. A 57.6/2:EN 3/PT.653, USDA National Engineering Handbook Part 653. <u>https://directives.sc.egov.usda.gov/OpenNonWebContent.aspx?content=34804.wba</u>.
- Fellman, J.B., *et al.* 2013. "Leaf Litter Age, Chemical Quality, and Photodegradation Control the Fate of Leachate Dissolved Organic Matter in a Dryland River." *Journal of Arid Environments* 89:30-37.
- Fenneman, N.M. 1917. "Physiographic Subdivision of the United States." *Proceedings of the National Academy of Sciences of the United States of America* 3(1), 17–22. <u>https://doi.org/10.1073/pnas.3.1.17</u>.
- Fenneman, N.M., and D.W. Johnson. 1946. *Physiographic Divisions of the United States*. U.S. Geological Survey, Washington, D.C. <u>http://water.usgs.gov/GIS/dsdl/physio.gz</u>.
- Fernald, A.F., *et al.* 2001. "Transient Storage and Hyporheic Flow along the Willamette River, Oregon: Field Measurements and Model Estimates. *Water Resources Research* 37:1681-1694.
- Ferrington, L.C. 1993. "Endangered Rivers: A Case History of the Arkansas River in the Central Plains." *Aquatic Conservation: Marine and Freshwater Ecosystems* 3:305-316.
- Fesenmyer, K., *et al.* 2021. "Large portion of USA streams lose protection with new interpretation of Clean Water Act." *Freshwater Science* 40: 252-258.
- Field, J.J., and R.W. Lichvar. 2007. Review and Synopsis of Natural and Human Controls on Fluvial Channel Processes in the Arid West. ERDC/CRREL TR-07-16. U.S. Army Engineer Research and Development Center, Cold Regions Research and Engineering Laboratory, Hanover, NH. <u>https://hdl.handle.net/11681/5291</u>.
- Figuerola, J., and A.J. Green. 2002. "Dispersal of Aquatic Organisms by Waterbirds: A Review of Past Research and Priorities for Future Studies." *Freshwater Biology* 47:483-494.
- Figuerola, J., *et al.* 2005. "Invertebrate Eggs Can Fly: Evidence of Waterfowl-Mediated Gene Flow in Aquatic Invertebrates." *American Naturalist* 165:274-280.

- Finkelstein, P.L., and L.E. Truppi. 1991. "Spatial Distribution of Precipitation Seasonality in the United States. *Journal of Climate* 4:373-385.
- Fisher, S.G. and G.E. Likens. 1973. "Energy Flow in Bear Brook, New Hampshire: An Integrative Approach to Stream Ecosystem Metabolism." *Ecological Monographs* 43: 421-439.
- Fisher, S.G., *et al.* 2001. "Landscape Challenges to Ecosystem Thinking: Creative Flood and Drought in the American Southwest." *Scientia Marina* **65**:181-192.
- Fisher, S.J., and D.W. Willis. 2000. "Seasonal Dynamics of Aquatic Fauna and Habitat Parameters in a Perched Upper Missouri River Wetland." *Wetlands* 20:470-478.
- Fleming, S.W., *et al.* 2007. "Regime-Dependent Streamflow Sensitivities to Pacific Climate Modes Cross the Georgia-Puget Transboundary Ecoregion. *Hydrological Processes* 21:3264-3287.
- Flint, S.A., and W.H. McDowell. 2015. "Effects of headwater wetlands on dissolved nitrogen and dissolved organic carbon concentrations in a suburban New Hampshire watershed." *Freshwater Science* 34: 456-471.
- Flitcroft, R.L., et al. 2012. "Do Network Relationships Matter? Comparing Network and Instream Habitat Variables to Explain Densities of Juvenile Coho Salmon (Oncorhynchus Kisutch) in Mid-Coastal Oregon, USA." Aquatic Conservation: Marine and Freshwater Ecosystems 22:288-302.
- Florsheim, J.L., and J.F. Mount. 2003. "Changes in Lowland Floodplain Sedimentation Processes: Predisturbance to Post-rehabilitation, Cosumnes River, CA." *Geomorphology* 56(3-4):305-323.
- Florsheim, J.L., et al. 2008. "Bank Erosion as a Desirable Attribute of Rivers." Bioscience 58: 519-529.
- Forbes, M.G., *et al.* 2012. "Nutrient Transformation and Retention by Coastal Prairie Wetlands, Upper Gulf Coast, Texas." *Wetlands* 32(4): 705-715.
- Fossey, M., and A.N. Rousseau. 2016. "Can isolated and riparian wetlands mitigate the impact of climate change on watershed hydrology? A case study approach." *Journal of Environmental Management* 184(2):327-339.
- Fossey, M., *et al.* 2016. "Assessment of the impact of spatio-temporal attributes of wetlands on stream flows using a hydrological modelling framework: a theoretical case study of a watershed under temperate climatic conditions." *Hydrological Processes* 30: 1768-1781.
- Fovet, O., *et al.* 2020. "Dynamics of dissolved organic matter in headwaters: comparison of headwater streams with contrasting DOM and nutrient composition." *Aquatic Sciences* 82: 29.
- Franklin, S.B., *et al.* 2009. "Complex Effects of Channelization and Levee Construction on Western Tennessee Floodplain Forest Function." *Wetlands* 29(2): 451-464.
- Franssen, N.R., *et al.* 2006. "Effects of Floods on Fish Assemblages in an Intermittent Prairie Stream." *Freshwater Biology* 51:2072-2086.

- Fraser, D.F., *et al.* 1995. "Predation As an Agent of Population Fragmentation in a Tropical Watershed." *Ecology* 76:1461-1472.
- Fraser, D.J., *et al.* 2011. "Extent and Scale of Local Adaptation in Salmonid Fishes: Review and Meta-Analysis." *Heredity* 106:404-420.
- Freeman, M.C., et al. 2007. "Hydrologic Connectivity and the Contribution of Stream Headwaters to Ecological Integrity at Regional Scales." *Journal of the American Water Resources Association* 43:5-14.
- French, D.W., *et al.* 2020. "Headwater Catchments Govern Biogeochemistry in America's Largest Free-Flowing River Network." *Journal of Geophysical Research: Biogeosciences* 125: e2020JG005851.
- Frisch, D., and S.T. Threlkeld. 2005. "Flood-Mediated Dispersal Versus Hatching: Early Recolonisation Strategies of Copepods in Floodplain Ponds." *Freshwater Biology* 50:323-330.
- Frisch, D., *et al.* 2007. "High Dispersal Capacity of a Broad Spectrum of Aquatic Invertebrates Via Waterbirds." *Aquatic Sciences* 69:568-574.
- Frissell, C.A., *et al.* 1986. "A hierarchical framework for stream habitat classification: Viewing streams in a watershed context." *Environmental Management* 10:199-214.
- Fritz, K.M., and W.K. Dodds. 2004. "Resistance and Resilience of Macroinvertebrate Assemblages to Drying and Flood in a Tallgrass Prairie Stream System." *Hydrobiologia* 527:99-112.
- Fritz, K.M., and W.K. Dodds. 2005. "Harshness: Characterization of Intermittent Stream Habitat over Space and Time." *Marine and Freshwater Research* 56:13-23.
- Fritz, K.M., et al. 2006. Field Operations Manual for Assessing the Hydrologic Permanence and Ecological Condition of Headwater Streams. EPA/600/R-06/126, U.S. Environmental Protection Agency, Office of Research and Development, National Exposure Research Laboratory, Washington, D.C. <u>https://www.epa.gov/sites/default/files/2015-</u> <u>11/documents/manual for assessing hydrologic permanence - headwater streams.pdf</u>.
- Fritz, K.M., et al. 2013. "Comparing the Extent and Permanence of Headwater Streams from Two Field Surveys to Values from Hydrographic Databases and Maps." Journal of the American Water Resources Association 49: 867-882.
- Fritz, K.M., et al. 2018. "Physical and Chemical Connectivity of Streams and Riparian Wetlands to Downstream Waters: A Synthesis." Journal of the American Water Resources Association 54: 323-345.
- Fritz, K.M., *et al.* 2020. "Classifying Streamflow Duration: The Scientific Basis and an Operational Framework for Method Development." *Water* 12(9): 2545. <u>https://doi.org/10.3390/w12092545</u>.
- Fryirs, K., and D.B Gore. 2014. "Geochemical insights to the formation of 'sedimentary buffers': Considering the role of tributary-trunk stream interactions on catchment-scale sediment flux and drainage network dynamics." *Geomorphology* 219: 1-9.

- Fullerton, A.H., *et al.* 2010. "Hydrological Connectivity for Riverine Fish: Measurement Challenges and Research Opportunities." *Freshwater Biology* 55:2215-2237.
- Galat, D.L., and R. Lipkin. 2000. "Restoring Ecological Integrity of Great Rivers: Historical Hydrographs Aid in Defining Reference Conditions for the Missouri River." *Hydrobiologia* 422/423:29-48.
- Galat, D.L., *et al.* 1998. "Flooding to restore connectivity of regulated, large-river wetlands." *BioScience* 48:721-733.
- Galat, D.L., *et al.* 2005. "Missouri River Basin." Pages 427-480 in *Rivers of North America*. A.C. Benke and C.E. Cushing, editors. Elsevier Academic Press, Burlington, MA.
- Galatowitsch, S.M., and A.G. van der Valk. 1996. "The Vegetation of Restored and Natural Prairie Wetlands," *Ecological Applications* 6:102-112.
- Gallo, E.L., *et al.* 2020. "Estimating Surface Water Presence and Infiltration in Ephemeral to Intermittent Streams in the Southwestern US." *Frontiers in Water* 2: 572950.
- Gamble, D., et al. 2007. An Ecological and Functional Assessment of Urban Wetlands in Central Ohio.
 Volume 2: Morphometric Surveys, Depth-Area-Volume Relationships and Flood Storage Function of Urban Wetlands in Central Ohio. Ohio EPA Technical Report WET/2007-3B. Ohio Environmental Protection Agency, Wetland Ecology Group, Division of Surface Water, Columbus, OH.
 https://epa.ohio.gov/static/Portals/35/wetlands/L919ReportVol_2_FunctionAssessUrbanWTLD.pdf.
- Gardner, B., and P.J. Sullivan. 2004. "Spatial and Temporal Stream Temperature Prediction: Modeling Nonstationary Temporal Covariance Structures." *Water Resources Research* 40(1): W01102.
- Gartner, J.D., et al. 2016a. The Benefits and Limitations of Hydraulic Modeling for Ordinary High Water Mark Delineation. ERDC/CRREL TR-16-1. U.S. Army Engineer Research and Development Center, Hanover, NH. Accessed at <u>https://hdl.handle.net/11681/21574</u>.
- Gartner, J.D., *et al.* 2016b. *Hydrologic modeling and flood frequency analysis for ordinary high water mark delineation*. ERDC/CRREL TR-16-2. U.S. Army Corps of Engineers, Engineering Research and Development Center, Hanover, NH. Accessed at <u>https://hdl.handle.net/11681/21573</u>.
- Gartner, J.D., et al. 2016c. Integrating Hydrologic Modeling, Hydraulic Modeling, and Field Data for Ordinary High Water Mark Delineation. ERDC/CRREL TR-16-3. U.S. Army Engineer Research and Development Center, Hanover, NH. Accessed at <u>https://hdl.handle.net/11681/21571</u>.
- Gergel, S.E. 2005. "Spatial and Non-spatial Factors: When Do They Affect Landscape Indicators of Watershed Loading?" *Landscape Ecology* 20:177-189.
- Gergel, S.E., *et al.* 1999. "Dissolved Organic Carbon as an Indicator of the Scale of Watershed Influence on Lakes and Rivers." *Ecological Applications* 9:1377-1390.
- Gergel, S.E., *et al.* 2002. "Consequences of Human-altered Floods: Levees, Floods, and Floodplain Forests along the Wisconsin River." *Ecological Applications* 12(6): 1755-1770.

- Gergel, S.E., *et al.* 2005. "Do Dams and Levees Impact Nitrogen Cycling? Simulating the Effects of Flood Alterations on Floodplain Denitrification." *Global Change Biology* 11(8): 1352-1367.
- Ghermandi, A., *et al.* 2010. "Values of natural and human-made wetlands: a meta-analysis." *Water Resources Research* 46: W12156.
- Gibbons, J.W., *et al.* 2006. "Remarkable Amphibian Biomass and Abundance in an Isolated Wetland: Implications for Wetland Conservation." *Conservation Biology* 20:1457-1465.
- Gillespie, J.L., *et al.* 2018. "Floodplain Trapping and Cycling Compared to Streambank Erosion of Sediment and Nutrients in an Agricultural Watershed." *Water Resources Bulletin* 54: 565.
- Gleason, R.A., *et al.* 2003. "Effects of sediment load on emergence of aquatic invertebrates and plants from wetland soil egg and seed banks." *Wetlands* 23:26-34.
- Gleason, R.A., et al. 2007. Estimating Water Storage Capacity of Existing and Potentially Restorable Wetland Depressions in a Subbasin of the Red River of the North. U.S. Geological Survey Open-File Report 2007-1159. U.S. Geological Survey, Reston, VA. https://pubs.usgs.gov/of/2007/1159/pdf/ofr2007-1159.pdf.
- Gleeson, T., *et al.* 2011. "Classifying the Water Table at Regional to Continental Scales." *Geophysical Research Letters* 38: L05401.
- Glinska-Lewczuk, K. 2009. "Water Quality Dynamics of Oxbow Lakes in Young Glacial Landscape of NE Poland in Relation to Their Hydrological Connectivity." *Ecological Engineering* 35: 25-37.
- Godsey, S.E., and J.W Kirchner. 2014. "Dynamic, discontinuous stream networks: hydrologically driven variations in active drainage density, flowing channels and stream order." *Hydrological Processes* 28: 5791-5803.
- Golden, H.E., *et al.* 2013. "Climate Change and Watershed Mercury Export: A Multiple Projection and Model Analysis." *Environmental Toxicology and Chemistry* 32:2165-2174.
- Golden, H.E., *et al.* 2014. "Hydrologic connectivity between geographically isolated wetlands and surface water systems: A review of select modeling methods." *Environmental Modelling & Software* 53: 190-206. https://doi.org/10.1016/j.envsoft.2013.12.004.
- Golden, H.E., *et al.* 2016. "Relative effects of geographically isolated wetlands on streamflow: a watershed-scale analysis." *Ecohydrology* 9: 21–38. <u>https://doi.org/10.1002/eco.1608</u>.
- Golden, H.E., *et al.* 2019. "Non-floodplain Wetlands Affect Watershed Nutrient Dynamics: A Critical Review." *Environmental Science & Technology* 53: 7203-7214.
- Golden, H.E., et al. 2021a. "Improving global flood and drought predictions: integrating non-floodplain wetlands into watershed hydrologic models." Environmental Research Letters 16(9): 091002. <u>https://doi.org/10.1088/1748-9326/ac1fbc</u>.
- Golden, H.E., *et al.* 2021b. "River drying influences genetic variation and population structure in an Arctic freshwater fish." *Conservation Genetics* 22(3): 369-382.

- Gomi, T., and R.C. Sidle. 2003. "Bed Load Transport in Managed Steep-gradient Headwater Streams of Southeastern Alaska." *Water Resources Research* 39: 1336.
- Gomi, T., et al. 2002. "Understanding Processes and Downstream Linkages of Headwater Systems." Bioscience 52: 905-916.
- Gooderham, J.P.R., *et al.* 2007. "Upstream Heterogeneous Zones: Small Stream Systems Structured by a Lack of Competence?" *Journal of the North American Benthological Society* 26: 365-374.
- Goodrich, D.C., *et al.* 1997. "Linearity of Basin Response as a Function of Scale in a Semiarid Watershed." *Water Resources Research* 33: 2951-2965.
- Goodrich, D.C., et al. 2018. "Southwestern Intermittent and Ephemeral Stream Connectivity." Journal of the American Water Resources Association 54(2): 400-422. <u>https://doi.org/10.1111/1752-1688.12636</u>.
- Goolsby, D., et al. 1999. Topic Report 3, Flux and Sources of Nutrients in the Mississippi-Atchafalaya River Basin. National Science and Technology Council Committee on Environment and Natural Resources, Washington, D.C. <u>https://repository.library.noaa.gov/view/noaa/21437</u>.
- Gooseff, M.N., *et al.* 2008. "Solute Transport along Stream and River Networks." Pages 395-417 in *River Confluences, Tributaries, and the Fluvial Network.* S.P. Rice, *et al.*, editors. John Wiley & Sons, Chichester, UK.
- Gordon, B.A., *et al.* 2020. "Nutrient Retention in Ecologically Functional Floodplains: A Review." *Water* 12: 2762.
- Gorman, O.T. 1986. "Assemblage Organization of Stream Fishes: The Effects of Rivers on Adventitious Streams." *American Naturalist* 128(4): 611-616.
- Goulden, T., *et al.* 2014. "Sensitivity of watershed attributes to spatial resolution and interpolation method of LiDAR DEMs in three distinct landscapes." *Water Resources Research* 50(3): 1908-1927.
- Graf, W.L. 1984. "A Probabilistic Approach to the Spatial Assessment of River Channel Instability." *Water Resources Research* 20(7): 953-962.
- Graf, W.L. 1994. *Plutonium and the Rio Grande: Environmental Change and Contamination in the Nuclear Age*. Oxford University Press, New York, NY.
- Graf, W.L. 2006. "Downstream hydrologic and geomorphic effects of large dams on American rivers." *Geomorphology* 79: 336–360.
- Granado, D.C., and R. Henry. 2014. "Phytoplankton Community to Hydrologic Variations in Oxbow Lakes with Different Levels of Connection to a Tropical River." *Hydrobiologia* 721: 223-238.
- Grant, E.H.C., *et al.* 2007. "Living in the Branches: Population Dynamics and Ecological Processes in Dendritic Networks." *Ecology Letters* 10: 165-175.
- Grau J., *et al.* 2021. "Using Unmanned Aerial Vehicle and LiDAR-Derived DEMs to Estimate Channels of Small Tributary Streams." *Remote Sensing* 13(17):3380. <u>https://doi.org/10.3390/rs13173380</u>.

- Greathouse, E.A., *et al.* 2006. "Indirect Upstream Effects Of Dams: Consequences Of Migratory Consumer Extirpation In Puerto Rico." *Ecological Applications* 16: 339-352.
- Greathouse, E.A., *et al.* 2014. "Linking Landscape Characteristics and High Stream Nitrogen in the Oregon Coast Range: Red Alder Complicates Use of Nutrient Criteria." *Journal of the American Water Resources Association* 50: 1383-1400.
- Green, D.I.S., *et al.* 2019. "Runoff Storage Potential of Drained Upland Depressions on the Des Moines Lobe of Iowa." *Journal of the American Water Resources Association* 55: 543-558.
- Green, D.M. 2005. "Bufo americanus, American Toad." Pages 692-704 in Amphibian Declines: The Conservation Status of United States Species. M. Lannoo, ed. University of California Press, Berkeley, CA.
- Gregory, K. J., and D.E. Walling. 1968. "The Variation of Drainage Density within a Catchment." *Bulletin of the International Association of Scientific Hydrology* 13: 61-68.
- Gregory, S.V., *et al.* 1991. "An Ecosystem Perspective of Riparian Zones: Focus on Links between Land and Water." *BioScience* 41: 540-551.
- Guinessey, Elizabeth, et al. 2019. A Literature Review: The Chemical, Physical and Biological Significance of Geographically Isolated Wetlands and Non-Perennial Streams in the Southeast. Accessed at <u>https://www.southernenvironment.org/wp-</u> content/uploads/legacy/words_docs/Exhibit_A.pdf.
- Gulbin, S., *et al.* 2019. "Wetland loss impact on long term flood risks in a closed watershed." *Environmental Science & Policy* 94: 112-122.
- Gurnell, A., *et al.* 2008. "Propagule Deposition along River Margins: Linking Hydrology and Ecology." *Journal of Ecology* 96: 553-565.
- Gurnell, A.M. 2007. "Analogies Between Mineral Sediment and Vegetative Particle Dynamics in Fluvial Systems." *Geomorphology* 89: 9-22.
- Gurnell, A.M *et al.* 1995. "The Role of Coarse Woody Debris in Forest Aquatic Habitats: Implications for Management." *Aquatic Conservation: Marine and Freshwater Ecosystems* 5: 143-166.
- Gutenson, J.L., and J.C. Deters. 2022. Antecedent Precipitation Tool (APT) Version 1.0: Technical and User Guide. ERDC/TN WRAP-22-1. U.S. Army Engineer Research and Development Center, Vicksburg, MS. <u>http://dx.doi.org/10.21079/11681/43160</u>.
- Hadley, R.F., *et al.* 1987. "Water Development and Associated Hydrologic Changes in the Platte River, Nebraska, U.S.A." *Regulated Rivers: Research & Management* 1:331-341.
- Hafen, K.C., *et al.* 2020. "The Influence of Climate Variability on the Accuracy of NHD Perennial and Nonperennial Stream Classifications." *Journal of the American Water Resources Association* 56(5): 903-916.

- Hall, B.R., *et al.* 2001. "Environmental Influences on Plant Species Composition in Ground-water Seeps in the Catskill Mountains of New York." *Wetlands* 21: 125-134.
- Hall, C.J., *et al.* 2011. "The Historic Influence of Dams on Diadromous Fish Habitat with a Focus on River Herring and Hydrologic Longitudinal Connectivity." *Landscape Ecology* 26: 95-107.
- Hall, R.O., and J.L. Meyer. 1998. "The Trophic Significance of Bacteria in a Detritus-Based Stream Food Web." *Ecology* 79: 1995-2012.
- Hall, R.O., *et al.* 2000. "Organic Matter Flow in Stream Food Webs with Reduced Detrital Resource Base." *Ecology* 81: 3445-3463.
- Hall, R.O., *et al.* 2009. "Hydrologic Control of Nitrogen Removal, Storage, and Export in a Mountain Stream." *Limnology and Oceanography* 54: 2128-2142.
- Hamill, D., and G.C.L. David. 2021. Hydrologic Analysis of Field Delineated Ordinary High Water Marks for Rivers and Streams. ERDC/CRREL TR-21-9. U.S. Army Corps of Engineers, Engineer Research and Development Center, Hanover, NH. 63 pp. Accessed at <u>https://erdclibrary.erdc.dren.mil/jspui/bitstream/11681/41681/1/ERDC-CRREL% 20TR-21-9.pdf</u>.
- Hamilton, S.K., *et al.* 2005. "Persistence of Aquatic Refugia between Flow Pulses in a Dryland River System (Cooper Creek, Australia)." *Limnology and Oceanography* 50: 743-754.
- Hammersmark, C.T., *et al.* 2008. "Quantifying the Hydrological Effects of Stream Restoration in a Montane Meadow Environment." *River Research and Applications* 24: 735–753.
- Hansen, W.F. 2001. "Identifying stream types and management implications." *Forest Ecology and Management* 143: 39-46.
- Hanson, M.A., *et al.* 2005. "Biotic Interactions as Determinants of Ecosystem Structure in Prairie Wetlands: An Example Using Fish." *Wetlands* 25: 764-775.
- Harbaugh, A.W. 2005. MODFLOW-2005, the US Geological Survey Modular Ground-Water Model the Ground-Water Flow Process. US Geological Survey Techniques and Methods 6-A16. U.S. Department of the Interior, U.S. Geological Survey, Reston, VA. 253 pp. <u>https://wwwbrr.cr.usgs.gov/hill_tiedeman_book/documentation/MODFLOW-MODPATH-ModelViewer/MF2005-tma6a16.pdf</u>.
- Harmon, M.E., *et al.* 1986. "Ecology of Coarse Woody Debris in Temperature Ecosystems." *Advances in Ecological Research* 15:133-302.
- Harvey, J., et al. 2019. "How Hydrologic Connectivity Regulates Water Quality in River Corridors." Journal of the American Water Resources Association 55(2): 369-381. <u>https://onlinelibrary.wiley.com/doi/abs/10.1111/1752-1688.12691</u>.
- Harvey, J.W., and N.M. Schmadel. 2021. "The River Corridor's Evolving Connectivity of Lotic and Lentic Waters." *Frontiers in Water* 2 (70). <u>https://www.frontiersin.org/article/10.3389/frwa.2020.580727.</u>

- Harvey, J.W., *et al.* 2004. "Ground Water Recharge and Discharge in the Central Everglades." *Groundwater* 42(7): 1090-1102.
- Hassan, M. A. 1990. "Observations of desert flood bores." *Earth Surface Processes and Landforms* 15: 481-485.
- Haukos, D.A., *et al.* 2006. "Spring migration of northern pintails from Texas and New Mexico, USA. *Waterbirds* 29: 127-136.
- Hawes, E., and M. Smith. 2005. *Riparian Buffer Zones: Functions and Recommended Widths*. Prepared for the Eightmile River Wild and Scenic Study Committee.
 http://eightmileriver.org/resources/digital_library/appendicies/09c3_Riparian%20Buffer%20Science_YALE.pdf. Accessed via Wayback Machine at https://web.archive.org/web/20210507013228/http://eightmileriver.org/resources/digital_library/appendicies/09c3_Riparian%20Buffer%20Science_YALE.pdf.
- He, S., and Y.J. Xu. 2015. "Three Decadal Inputs of Total Organic Carbon from Four Major Coastal River Basins to the Summer Hypoxic Zone of the Northern Gulf of Mexico." *Marine Pollution Bulletin* 90 (1–2): 121-128.
- Heckscher C.M., and C.R. Bartlett. 2004. "Rediscovery and Habitat Associations of *Photuris Bethaniensis* McDermott (Coleoptera: Lampyridae)." *The Coleopterists Bulletin* 58(3): 349-353.
- Hedin, L.O., *et al.* 1995. "Patterns of Nutrient Loss from Unpolluted Old-growth Temperate Forests: Evaluation of Biogeochemical Theory." *Ecology* 76: 493-509.
- Hedin, L.O., *et al.* 1998. "Thermodynamic constraints on nitrogen transformations and other biogeochemical processes at soil-stream interfaces." *Ecology* 79: 684-703.
- Hedman, E.R., and W.R. Osterkamp. 1982. Streamflow Characteristics Related to Channel Geometry of Streams in Western United States. United States Geological Survey Water-Supply Paper 2193. United States Government Printing Office, Washington D.C. <u>https://pubs.usgs.gov/wsp/2193/report.pdf</u>.
- Heine, R.A., *et al.* 2004. "Development and Comparison of Approaches for Automated Mapping of Stream Channel Networks." *Annals of the Association of American Geographers* 94: 477-490.
- Helton, A.M., *et al.* 2011. "Thinking Outside the Channel: Modeling Nitrogen Cycling in Networked River Ecosystems." *Frontiers in Ecology and the Environment* 9: 229-238.
- Hemond, H.F. 1980. "Biogeochemistry of Thoreau's Bog, Concord, Massachusetts." *Ecological Monographs* 50: 507-526.
- Hemond, H.F. 1983. "The Nitrogen Budget of Thoreau's Bog." Ecology 64:99-109.
- Henning, J.A., *et al.* 2007. "Use of Seasonal Freshwater Wetlands by Fishes in a Temperate River Floodplain." *Journal of Fish Biology* 71: 476-492.

- Herbert, E.R., *et al.* 2015. "A global perspective on wetland salinization: ecological consequences of a growing threat to freshwater wetlands." *Ecosphere* 6(10): 206. <u>http://dx.doi.org/10.1890/ES14-00534.1</u>.
- Herwig, B.R., *et al.* 2010. "Factors Influencing Fish Distributions in Shallow Lakes in Prairie and Prairieparkland Regions of Minnesota, USA." *Wetlands* 30: 609-619.
- Hess, G.R. 1996. "Linking Extinction to Connectivity and Habitat Destruction in Metapopulation Models." *The American Naturalist* 148: 226-236.
- Hewlett, J.D. 1982. Principles of Forest Hydrology. University of Georgia Press, Athens, GA.
- Hewlett, J.D., and A.R. Hibbert. 1967. "Factors Affecting the Response of Small Watersheds to Precipitation in Humid Areas. Pages 275-290 in *International Symposium on Forest Hydrology*. W.S. Sopper and H.W. Hull, editors. Pergamon Press, New York, NY.
- Hewlett, J.D., and W.L. Nutter. 1970. "The varying source area of streamflow from upland basins." American Society of Civil Engineers, Symposium on Interdisciplinary Aspects of Watershed Management. Montana State University, Bozeman, Montana.
- Hewlett, J.D., *et al.* 1977. "Predicting Stormflow and Peakflow from Small Basin in Humid Areas by the R-Index Method." *Water Resources Bulletin* 13: 231-253.
- Hill, B.H., *et al.* 1992. "Predictability of Streamflow and Particulate Organic Matter Concentration as Indicators of Stability in Prairie Streams." *Hydrobiologia* 242: 7-18.
- Hill, B.H., *et al.* 2014. "A synoptic survey of ecosystem services from headwater catchments in the United States." *Ecosystem Services* 7: 106-115.
- Hill, R.A., *et al.* 2016. "The Stream-Catchment (StreamCat) Dataset: A Database of Watershed Metrics for the Conterminous United States." *Journal of the American Water Resources Association* 52 (1): 120-128. <u>https://onlinelibrary.wiley.com/doi/abs/10.1111/1752-1688.12372</u>.
- Hill, R.A., et al. 2018. "The Lake-Catchment (LakeCat) Dataset: characterizing landscape features for lake basins within the conterminous USA." Freshwater Science 37(2): 208-221. https://www.journals.uchicago.edu/doi/full/10.1086/697966.
- Hitt, N.P., and P.L. Angermeier. 2008. "Evidence for Fish Dispersal from Spatial Analysis of Stream Network Topology." *Journal of the North American Benthological Society* 27: 304-320.
- Hobbie, J.E., and R.G. Wetzel. 1992. "Microbial Control of Dissolved Organic Carbon in Lakes: Research for the Future." *Hydrobiologia* 229: 169-180.
- Hoffmann, C.C., et al. 2009. "Phosphorus Retention in Riparian Buffers: Review of Their Efficiency." Journal of Environmental Quality 38: 1942-1955.
- Hogan, M.E., and G.F. Tande. 1983. Anchorage Wetlands Study: Special Report on Connors Bog and Klatt Bog: U.S. Fish and Wildlife Service, Special Studies.

- Hope, D., *et al.* 1994. "A Review of the Export of Carbon in River Water: Fluxes and Processes." *Environmental Pollution* 84: 301-324.
- Hopkins, K.G., *et al.* 2018. "A Method to Quantify and Value Floodplain Sediment and Nutrient Retention Ecosystem Services." *Journal of Environmental Management* 220: 65-76. <u>https://doi.org/10.1016/j.jenvman.2018.05.013</u>.
- Horton, R.E. 1945. "Erosional development of streams and their drainage basins: hydrophysical approach to quantitative morphology." *Bulletin of the Geological Society of America* 56: 275-370.
- Horwitz, R. J. 1978. "Temporal variability patterns and the distributional patterns of stream fishes." *Ecological Monographs* 48: 307-321.
- Howard, B.E., *et al.* 2020. "SWIFT-Active Screener: Accelerated document screening through active learning and integrated recall estimation." *Environment International* 138: 105623.
- Hsieh, P.A., et al. 2000. "VS2DI: A Graphical Software Package for Simulating Fluid Flow and Solute or Energy Transport in Variably Saturated Porous Media." U.S. Geological Survey Water-Resources Investigations Report 99-4130. U.S. Geological Survey, Denver, Colorado. 16pp. Available at: https://water.usgs.gov/water-resources/software/VS2DI/vs2di.pdf.
- Huang J, et al. 1996. "Long-lead seasonal temperature prediction using optimal climate normals." *Journal* of Climate 9: 809–817.
- Hubbard, D.E., and R.L. Linder. 1986. "Spring Runoff Retention in Prairie Pothole Wetlands." *Journal of Soil and Water Conservation* 41(2): 122-125.
- Hughes, J.M., *et al.* 2009. "Genes in Streams: Using DNA to Understand the Movement of Freshwater Fauna and Their Riverine Habitat." *BioScience* 59: 573-583.
- Humphries, P., *et al.* 2014. "The River Wave Concept: Integrating River Ecosystem Models." *BioScience* 64(10): 870-882. <u>https://doi.org/10.1093/biosci/biu130</u>.
- Hunsinger, T.W., and M.J. Lannoo. 2005. "Notophthalmus viridescens, Eastern Newt." Pages 912-914 in Amphibian Declines: The Conservation Status of United States Species. M. Lannoo, ed. University of California Press, Berkeley, CA.
- Hunter, M.A., *et al.* 2005. "Low Flow Spatial Characteristics in Forested Headwater Channels of Southwest Washington." *Journal of the American Water Resources Association* 41: 503-516.
- Hupp, C. R., *et al.* 2009. "Floodplain Geomorphic Processes and Environmental Impacts of Human Alteration along Coastal Plain Rivers, USA." *Wetlands* 29: 413-429.
- Huryn, A.D., and K.E. Gibbs. 1999. "Riparian Sedge Meadows in Maine. A Macroinvertebrate Community Structured by River-Floodplain Interaction." Pages 363-382 in *Invertebrates in Freshwater Wetlands of North America: Ecology and Management*. D. P. Batzer, *et al*, editors. John Wiley & Sons, New York, NY.

- Huryn, A.D., *et al.* 2005. "Landscape Heterogeneity and the Biodiversity of Arctic Stream Communities: A Habitat Template Analysis. *Canadian Journal of Fisheries and Aquatic Sciences* 62: 1905-1919.
- Ice, G.G. 2008. "Chapter 3: Stream Temperature and Dissolved Oxygen." Pages 37-54 in *Hydrologic and Biological Responses to Forest Practices*. J.D. Stednick, ed. Springer, New York, NY.
- Ilg, C., *et al.* 2008. "Long-term Reactions of Plants and Macroinvertebrates to Extreme Floods in Floodplain Grasslands." *Ecology* 89: 2392-2398. <u>https://doi.org/10.1890/08-0528.1</u>.
- Intergovernmental Panel on Climate Change. 2014: Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team, R.K. Pachauri and L.A. Meyer (eds.)]. IPCC, Geneva, Switzerland, 151 pp. Accessed at https://www.ipcc.ch/site/assets/uploads/2018/02/SYR_AR5_FINAL_full.pdf. ("IPCC 2014.")
- International Atomic Energy Agency. 2003. "Investigating Leaks in Dams & Reservoirs." INIS-XA-616. Vienna, Austria. <u>https://inis.iaea.org/collection/NCLCollectionStore/_Public/34/028/34028298.pdf</u>.
- Isenhart, T.M. 1992. Transformation and Fate of Nitrate in Northern Prairie Wetlands, Ph.D. Dissertation. Iowa State University, Ames, Iowa. https://dr.lib.iastate.edu/server/api/core/bitstreams/365dda7b-e91e-4f16-8e34-3a9948706326/content.
- Izbicki, J.A. 2007. "Physical and Temporal Isolation of Mountain Headwater Streams in the Western Mojave Desert, Southern California." *Journal of the American Water Resources Association* 43: 26-40.
- Jacques, J., and D.L. Lorenz. 1988. Techniques for Estimating the Magnitude and Frequency of Floods in Minnesota. USGS Water-Resources Investigations Report 87-4170. U.S. Geological Survey, Washington, D.C. <u>https://pubs.usgs.gov/wri/1987/4170/report.pdf</u>.
- Jaeger, K.L., et al. 2014. "Climate change poised to threaten hydrologic connectivity and endemic fishes in dryland streams." Proceedings of the National Academy of Sciences 111(38): 13894-13899. <u>https://doi.org/10.1073/pnas.1320890111</u>.
- Jaeger, K.L., *et al.* 2019. "Probability of Streamflow Permanence Model (PROSPER): A spatially continuous model of annual streamflow permanence throughout the Pacific Northwest." *Journal of Hydrology X* 2: 100005. <u>https://doi.org/10.1016/j.hydroa.2018.100005</u>.
- James City County. 2009. *Perennial Stream Protocol Guidance Manual*. James City County, Virginia. 18 pp. Available at: <u>https://jamescitycountyva.gov/DocumentCenter/View/2158/JCC-Perennial-Stream-Protocol-Manual-PDF?bidId=</u>.
- Jansson, A., *et al.* 1998. "Quantifying the Nitrogen Retention Capacity of Natural Wetlands in the Large-Scale Drainage Basin of the Baltic Sea." *Landscape Ecology* 13: 249-262.
- Jansson, R., *et al.* 2005. "Hydrochory Increases Riparian Plant Species Richness: A Comparison between a Free-Flowing and a Regulated River." *Journal of Ecology* 93: 1094-1103.

- Jeffres, C.A., *et al.* 2008. "Ephemeral Floodplain Habitats Provide Best Growth Conditions for Juvenile Chinook Salmon in a California River." *Environmental Biology of Fishes* 83: 449-458.
- Jencso, K.G., *et al.* 2009. "Hydrologic connectivity between landscapes and streams: Transferring reachand plot-scale understanding to the catchment scale." *Water Resources Research* 45: W04428.
- Jencso, K.G., and B.L. McGlynn. 2011. "Hierarchical Controls on Runoff Generation: Topographically Driven Hydrologic Connectivity, Geology, and Vegetation." *Water Resources Research* 47: W11527.
- Jenkins, K.M., and A.J. Boulton. 2003. "Connectivity in a Dryland River: Short-term Aquatic Microinvertebrate Recruitment following Floodplain Inundation." *Ecology* 84: 2708-2723.
- Jensen, A.K., and W.I. Ford. 2019. "Quantifying Nitrate Dynamics of a Confluence Floodplain Wetland in a Disturbed Appalachian Watershed: High-Resolution Sensing and Modeling." *Transactions of the ASABE* 62: 1545-1565.
- John, K.R. 1964. "Survival of Fish in Intermittent Streams of the Chirichua Mountains, Arizona." *Ecology* 45: 112-119.
- Johnson, B.R., *et al.* 2010. "Use of Spatially Explicit Physicochemical Data to Measure Downstream Impacts of Headwater Stream Disturbance." *Water Resources Research* 46: W09526.
- Johnson, P.T.J., *et al.* 2008. "Dam Invaders: Impoundments Facilitate Biological Invasions into Freshwaters." *Frontiers in Ecology and the Environment* 6: 357-363.
- Johnston, C. 1991. "Sediment and Nutrient Retention by Freshwater Wetlands: Effects on Surface Water Quality." *Critical Reviews in Environmental Control* 21: 491-565.
- Johnston, C.A., and N.E. McIntyre. 2019. "Effects of cropland encroachment on prairie pothole wetlands: numbers, density, size, shape, and structural connectivity." *Landscape Ecology* 34: 827-841.
- Johnston, C.A., and R.J. Naiman. 1990a. "Aquatic patch creation in relation to beaver population trends." *Ecology* 71: 1617-1621.
- Johnston, C.A., and R.J. Naiman. 1990b. "The use of a geographic information system to analyze long-term landscape alteration by beaver." *Landscape Ecology* 4: 5-19.
- Johnston, C.A., *et al.* 1990. "The Cumulative Effect of Wetlands on Stream Water Quality and Quantity." *Biogeochemistry* 10: 105-141.
- Johnston, C.A., et al., 1997. "The Potential Role of Riverine Wetlands as Buffer Zones." Pages 155-170 in Buffer Zones Their Processes and Potential in Water Protection. N.E. Haycock, et al., ed. Quest International, Harpenden, UK.
- Johnston, C.A., *et al.* 2001. "Nutrient Dynamics in Relation to Geomorphology of Riverine Wetlands." *Soil Science Society of America Journal* 65(2): 557-577.
- Jones, C.N., *et al.* 2018a. "Estimating restorable wetland water storage at landscape scales." *Hydrological Processes* 32: 305-313. <u>http://dx.doi.org/10.1002/hyp.11405</u>.

- Jones, C.N., *et al.* 2018b. "From Salamanders to Greenhouse Gases: Does Upland Management Affect Wetland Functions?" *Frontiers in Ecology and the Environment* 16: 14-19. <u>https://doi.org/10.1002/fee.1744</u>.
- Jones, C.N., *et al.* 2019. "Modeling Connectivity of Non-floodplain Wetlands: Insights, Approaches, and Recommendations." *Journal of the American Water Resources Association* 55(3): 559-577.
- Jones, C.S., et al. 2018c. "Iowa stream nitrate and the Gulf of Mexico." PloS ONE 13: e0195930.
- Jones, J.W. 2019. "Improved Automated Detection of Subpixel-Scale Inundation—Revised Dynamic Surface Water Extent (DSWE) Partial Surface Water Tests." *Remote Sensing* 11(4): 374. <u>https://doi.org/10.3390/rs11040374</u>.
- Jordan, T.E., *et al.* 2007. "Comparing Functional Assessments of Wetlands to Measurements of Soil Characteristics and Nitrogen Processing." *Wetlands* 27: 479-497.
- Junk, W.J., et al. 1989. "The Flood Pulse Concept in River-Floodplain Systems." Pages 110-127 in Proceedings of the International Large River Symposium Ottawa. D.P. Dodge, ed. Canadian Special Publication of Fisheries and Aquatic Sciences 106, Ottawa, Canada.
- Kadlec, R.H., and S.D. Wallace. 2009. Treatment Wetlands. 2nd edition. CRC Press, Boca Raton, FL.
- Kalinin, A., *et al.* 2016. "The influence of an in-network lake on the timing, form, and magnitude of downstream dissolved organic carbon and nutrient flux." *Water Resources Research* 52: 8668-8684.
- Kao, C.M., *et al.* 2002. "Non-point Source Pesticide Removal by a Mountainous Wetland." *Water Science and Technology* 46: 199-206.
- Kaplan, L.A., *et al.* 1980. "Patterns of Dissolved Organic Carbon in Transport." *Limnology and Oceanography* 25: 1034-1043.
- Kappas, I., *et al.* 2017. "Molecular and morphological data suggest weak phylogeographic structure in the fairy shrimp *Streptocephalus torvicornis* (Branchiopoda, Anostraca)." *Hydrobiologia* 801: 21-32.
- Karwan, D.L., and J.E. Saiers. 2012. "Hyporheic exchange and streambed filtration of suspended particles." *Water Resources Research* 48: W01519.
- Kasak, K., *et al.* 2018. "Efficiency of a newly established in-stream constructed wetland treating diffuse agricultural pollution." *Ecological Engineering* 119: 1-7.
- Keckeis, S., *et al.* 2003. "The Significance of Zooplankton Grazing in a Floodplain System of the River Danube." *Journal of Plankton Research* 25: 243-253.
- Keddy, P.A., *et al.* 2007. "The Wetlands of Lakes Pontchartrain and Maurepas: Past, Present and Future." *Environmental Reviews* 15: 43-77.
- Keiper, J.B., *et al.* 2002. "Biology and Ecology of Higher Diptera from Freshwater Wetlands." *Annual Review of Entomology* 47: 207-232.

- Kellman, L. 2004. "Nitrate removal in a first-order stream: reconciling laboratory and field measurements." *Biogeochemistry* 71: 89-105.
- Kelson, S.J., and S.M. Carlson. 2019. "Do precipitation extremes drive growth and migration timing of a Pacific salmonid fish in Mediterranean-climate streams?" *Ecosphere* 10: e02618.
- Kemp, W.M., *et al.* 2005. "Eutrophication of Chesapeake Bay: Historical Trends and Ecological Interactions." *Marine Ecology Progress Series* 303(21): 1-29.
- Kentula, M., *et al.* 2004. "Tracking Changes in Wetlands with Urbanization: Sixteen Years of Experience in Portland, Oregon, USA." *Wetlands* 24(4): 734-743.
- Kimball, B.A., *et al.* 1995. "Effects of Colloids on Metal Transport in a River Receiving Acid Mine Drainage, Upper Arkansas River, Colorado, USA." *Applied Geochemistry* 10:285-306.
- King, A.J., *et al.* 2003. "Fish Recruitment on Floodplains: The roles of Patterns of Flooding and Life History Characteristics." *Canadian Journal of Fisheries and Aquatic Sciences* 60:773-786.
- King, R.S., *et al.* 2005. "Spatial Considerations for Linking Watershed Land Cover to Ecological Indicators in Streams." *Ecological Applications* 15: 137-153.
- Kirwan, M., and J. Megonigal. 2013. Tidal wetland stability in the face of human impacts and sea-level rise. *Nature* 504: 53–60. <u>https://doi.org/10.1038/nature12856</u>.
- Klammler, H., et al. 2020. "Local Storage Dynamics of Individual Wetlands Predict Wetlandscape Discharge." Water Resources Research 56 (11): e2020WR027581. https://doi.org/10.1029/2020WR027581.
- Klaus, J., *et al.* 2015. "Where does streamwater come from in low-relief forested watersheds? A dualisotope approach." *Hydrology and Earth System Science* 19: 125-135.
- Knapp, R.A., and K.R. Matthews. 2000. "Effects on Nonnative Fishes on Wilderness Lake Ecosystems in the Sierra Nevada and Recommendations for Reducing Impacts." Pages 312-317 in Wilderness Science in a Time of Change Conference, Volume 5: Wilderness Ecosystems, Threats, and Management, Missoula, Montana, May 23-27, 1999. https://www.fs.usda.gov/rm/pubs/rmrs_p015_5/rmrs_p015_5_312_317.pdf.
- Knispel, S., and E. Castella. 2003. "Disruption of a Longitudinal Pattern in Environmental Factors and Benthic Fauna by a Glacial Tributary." *Freshwater Biology* 48: 604-618.
- Knutson, M.G., et al. 1999. "Effects of Landscape Composition and Wetland Fragmentation on Frog and Toad Abundance and Species Richness in Iowa and Wisconsin, U.S.A." Conservation Biology 13: 1437-1446.
- Koizumi, I., *et al.* 2017. "Mass immigration of juvenile fishes into a small, once-dried tributary demonstrates the importance of remnant tributaries as wintering habitats." *Ichthyological Research* 64: 353-356.

- Kondolf, G. M. 1997. "Hungry Water: Effects of Dams and Gravel Mining on River Channels." *Environmental Management* 21: 533-551.
- Kondolf, G.M., *et al.* 2006. "Process-Based Ecological River Restoration: Visualizing Three-Dimensional Connectivity and Dynamic Vectors to Recover Lost Linkages." *Ecology and Society* 11:5.
- Kondolf, G.M., *et al.* 2014. "Sustainable sediment management in reservoirs and regulated rivers: Experiences from five continents." *Earth's Future* 2(5): 256-280. <u>https://doi.org/https://doi.org/10.1002/2013EF000184</u>.
- Konrad, C.P. 2019. "Seasonal Precipitation Influences Streamflow Vulnerability to the 2015 Drought in the Western United States." *Journal of Hydrometeorology* 20 (7): 1261-1274. <u>https://doi.org/10.1175/jhm-d-18-0121.1</u>.
- Koprivnjak, J.-F., and T.R. Moore. 1992. "Sources, Sinks, and Fluxes of Dissolved Organic Carbon in Subarctic Fen Catchments." *Arctic and Alpine Research* 24: 204-210.
- Kortelainen, P. 1993. "Content of Total Organic Carbon in Finnish Lakes and Its Relationship to Catchment Characteristics." *Canadian Journal of Fisheries and Aquatic Sciences* 50:1477-1483.
- Kovacic, D.A., et al. 2000. "Effectiveness of Constructed Wetlands in Reducing Nitrogen and Phosphorus Export from Agricultural Tile Drainage." Journal of Environmental Quality 29(4): 1262-1274.
- Krabbenhoft, C.A. *et al.* 2022. "Assessing placement bias of the global gauge network." *Nature Sustainability* 5: 586–592. <u>https://doi.org/10.1038/s41893-022-00873-0</u>.
- Kudray, G.M., and M.R. Gale. 2000. "Evaluation of National Wetland Inventory maps in a heavily forested region in the Upper Great Lakes." *Wetlands* 20(4): 581–587.
- Kuenzler, E.J., *et al.* 1980. *Distributions and Budgets of Carbon, Phosphorus, Iron and Manganese in a Floodplain Swamp Ecosystem.* Water Resources Research Institute Report 157. University of North Carolina, Chapel Hill, NC. <u>https://repository.lib.ncsu.edu/bitstream/handle/1840.4/1721/NC-WRRI-157.pdf?sequence=1&isAllowed=y</u>.
- Kusler, J. 2006. *Common Questions: Wetland Definition, Delineation, and Mapping.* Association of State Wetland Managers. <u>https://www.nawm.org/pdf_lib/14_mapping_6_26_06.pdf</u>.
- Kustu, M.D., *et al.* 2010. "Large-scale water cycle perturbation due to irrigation pumping in the US High Plains: A synthesis of observed streamflow changes." *Journal of Hydrology* 390(3-4): 222-244.
- Labbe, T.R., and K.D. Fausch. 2000. "Dynamics of Intermittent Stream Habitat Regulate Persistence of a Threatened Fish at Multiple Scales." *Ecological Applications* 10: 1774-1791.
- Lamoureux, V.S. and D.M. Madison. 1999. "Overwintering Habitats of Radio-Implanted Green Frogs, *Rana clamitans.*" *Journal of Herpetology* 33: 430-435.
- Lande, R., and S. Shannon. 1996. "The Role of Genetic Variation in Adaptation and Population Persistence in a Changing Environment." *Evolution* 50: 434-437.

- Lane, B.A., *et al.* 2017. "Revealing the Diversity of Natural Hydrologic Regimes in California with Relevance for Environmental Flows Applications." *Journal of the American Water Resources Association* 53(2): 411-430. <u>https://doi.org/10.1111/1752-1688.12504</u>.
- Lane, C.R. and E. D'Amico. 2010. "Calculating the Ecosystem Service of Water Storage in Isolated Wetlands using LiDAR in North Central Florida, USA." *Wetlands* 30: 967–977.
- Lane, C.R., and E. D'Amico. 2016. "Identification of Putative Geographically Isolated Wetlands of the Conterminous United States." *Journal of the American Water Resources Association* 52: 705-722.
- Lane, C.R., et al. 2018. "Hydrological, Physical, and Chemical Functions and Connectivity of Non-Floodplain Wetlands to Downstream Waters: A Review." Journal of the American Water Resources Association 54: 346-371.
- Lang, M., *et al.* 2012. "Enhanced Detection of Wetland-Stream Connectivity using LiDAR." *Wetlands* 32: 461-473.
- Lara, M.J., *et al.* 2015. "Polygonal Tundra Geomorphological Change in Response to warming Alters Future CO2 and CH4 flux on the Barrow Peninsula." *Global Change Biology* 21: 1634–1651.
- Larned, S.T., *et al* 2010. "A Framework of Analyzing Longitudinal and Temporal Variation in River Flow and Developing Flow-Ecology Relationships." *Journal of the American Water Resources Association* 46: 541-553.
- Larsen, A., *et al.* 2021. "Dam builders and their works: Beaver influences on the structure and function of river corridor hydrology, geomorphology, biogeochemistry and ecosystems." *Earth-Science Reviews* 218. https://doi.org/10.1016/j.earscirev.2021.103623.
- Lauenroth, W.K., *et al.* 1999. "The Structure and Function of Ecosystems in the Central North American Grassland Region. *Great Plains Research* 9: 223-259.
- Laughlin, T.W., *et al.* 2016. "Recruitment Sources of Channel and Blue Catfishes Inhabiting the Middle Mississippi River." *River Research and Applications* 32: 1808-1818.
- Lawrence, J. E., *et al.* 2013. "Hyporheic zone in urban streams: A review and opportunities for enhancing water quality and improving aquatic habitat by active management." *Environmental Engineering Science* 30: 480-501.
- Lawrence, C. B., et al. 2019. "Valuation of the Flood Attenuation Ecosystem Service in Difficult Run, VA, USA." Journal of Environmental Management 231: 1056-1064. <u>https://doi.org/10.1016/j.jenvman.2018.10.023</u>.
- Lee, S., *et al.* 2018. "Assessing the Cumulative Impacts of Geographically Isolated Wetlands on Watershed Hydrology Using the SWAT Model Coupled with Improved Wetland Modules." *Journal of Environmental Management* 223: 37–48.
- Lee, T.M., et al. 2010. Effect of Groundwater Levels and Headwater Wetlands on Streamflow in the Charlie Creek Basin, Peace River Watershed, West-Central Florida. U.S. Geological Survey

Scientific Investigations Report 2010–5189. U.S. Department of the Interior, U.S. Geological Survey, Reston, VA. <u>https://pubs.usgs.gov/sir/2010/5189/pdf/sir2010-5189_Lee.pdf</u>.

- Lefebvre, L., et al. 2013. Channel Classification across Arid West Landscapes in Support of OHW Delineation. ERDC/CRREL TR-13-3. U.S. Army Corps of Engineers, Hanover, NH. <u>https://usace.contentdm.oclc.org/digital/api/collection/p266001coll1/id/4350/download</u>.
- Lehman, P.W., *et al.* 2008. "The Influence of Floodplain Habitat on the Quantity and Quality of Riverine Phytoplankton Carbon Produced During the Flood Season in San Francisco Estuary." *Aquatic Ecology* 42: 363-378.
- Leibowitz, S.G. 2003. "Isolated Wetlands and Their Functions: An Ecological Perspective." *Wetlands* 23: 517-531.
- Leibowitz, S.G., and K.C. Vining. 2003. "Temporal Connectivity in a Prairie Pothole Complex." *Wetlands* 23: 13-25.
- Leibowitz, S.G., *et al.* 2008. "Non-Navigable Streams and Adjacent Wetlands: Addressing Science Needs Following the Supreme Court's *Rapanos* Decision." *Frontiers in Ecology and the Environment* 6: 364-371.
- Leibowitz, S.G., *et al.* 2016. "Intermittent Surface Water Connectivity: Fill and Spill Vs. Fill and Merge Dynamics." *Wetlands* 36: 323-342.
- Leibowitz, S.G., *et al.* 2018. "Connectivity of Streams and Wetlands to Downstream Waters: An Integrated Systems Framework." *Journal of the American Water Resources Association* 54: 298-322.
- Leigh, D. S. 1997. "Mercury-Tainted Overbank Sediment from Past Gold Mining in North Georgia, USA." *Environmental Geology* 30: 244-251.
- Leopold, L.B. 1994. A View of the River. Harvard University Press, Cambridge, MA.
- Leopold, L.B., et al. 1964. Fluvial Processes in Geomorphology. General Publishing Co. Ltd., Toronto, Canada.
- Leslie, A.W., *et al.* 2012. "Environmental Factors Structuring Benthic Macroinvertebrate Communities of Agricultural Ditches in Maryland." *Environmental Entomology* 4: 802-812.
- Letcher, B.H., *et al.* 2007. "Population Response to Habitat Fragmentation in a Stream-Dwelling Brook Trout Population." *PLOS ONE* 2: e1139.
- Leuthold, S.J., *et al.* 2021. "Seasonal connections between meteoric water and streamflow generation along a mountain headwater stream." *Hydrological Processes* 35(2): e14029. <u>https://doi.org/10.1002/hyp.14029</u>.
- Levick, L., et al. 2008. The Ecological and Hydrological Significance of Ephemeral and Intermittent Streams in the Arid and Semi-arid American Southwest. EPA/600/R-08/134 and ARS/233046. U.S. Environmental Protection Agency, Office of Research and Development and U.S. Department of Agriculture/Agricultural Research Service. Southwest Watershed Research Center, Washington, D.C.

https://www.epa.gov/sites/default/files/2015-03/documents/ephemeral_streams_report_final_508-kepner.pdf.

- Lewis, D.B., and S.J. Feit. 2015. "Connecting carbon and nitrogen storage in rural wetland soil to groundwater abstraction for urban water supply." *Global Change Biology* 21: 1704-1714.
- Li, L. 2019. "Watershed Reactive Transport." Reviews in Mineralogy and Geochemistry 85: 381-418.
- Lichvar, R., et al. 2006. Distribution of Ordinary High Water Mark (OHWM) Indicators and Their Reliability in Identifying the Limits of "Waters of the United States" in Arid Southwestern Channels. ERDC/CRREL TR-06-05. U.S. Army Engineer Research and Development Center, Hanover, NH. <u>https://apps.dtic.mil/sti/pdfs/ADA444089.pdf</u>.
- Lichvar, R.W., and S.M. McColley. 2008. A Field Guide to the Identification of the Ordinary High Water Mark (OHWM) in the Arid West Region of the Western United States: A Delineation Manual. ERDC/CRREL TR-08-12. U.S. Army Engineer Research and Development Center, Hanover, NH. <u>http://hdl.handle.net/11681/5308</u>.
- Lichvar, R.W., et al. 2009. Vegetation and Channel Morphology Responses to Ordinary High Water Discharge Events in Arid West Stream Channels. ERDC/CRREL TR-09-5. U.S. Army Engineer Research and Development Center, Hanover, NH. <u>https://apps.dtic.mil/sti/pdfs/ADA508422.pdf</u>.
- Lide, R.F., *et al.* 1995. "Hydrology of a Carolina Bay Located on the Upper Coastal Plain of Western South Carolina." *Wetlands* 15: 47-57.
- Liljedahl, A.K., *et al.* 2012. "Ice-Wedge Polygon Type Controls Low-Gradient Watershed-Scale Hydrology." *Proceedings of the Tenth International Conference on Permafrost* 1: 231-236. Salekhard, Russia, June 25-29, 2012. <u>https://ine.uaf.edu/media/1166/liljedahl_et_al_2012.pdf</u>.
- Linqvist, O.K., *et al.* 1991. "Mercury in the Swedish Environment Recent Research on Causes, Consequences, and Remedial Measures." *Water Air and Soil Pollution* 55: xi-xiii.
- Livezey, R.E., *et al.* 2007. "Estimation and extrapolation of climate normals and climatic trends." *Journal of Applied Meteorology and Climatology* 46: 1759–1776.
- Lloyd, O.B., and W.L. Lyke. 1995. Ground Water Atlas of the United States, Segment 10, Illinois, Indiana, Kentucky, Ohio, Tennessee. USGS Hydrologic Investigations Atlas 730-K. U.S. Department of the Interior, U.S. Geological Survey, Reston, VA. <u>https://pubs.usgs.gov/ha/730k/report.pdf</u>.
- Lohse, Kathleen A., et al. 2020. "Influence of Climate and Duration of Stream Water Presence on Rates of Litter Decomposition and Nutrient Dynamics in Temporary Streams and Surrounding Environments of Southwestern USA." Frontiers in Water 2 (34). <u>https://doi.org/10.3389/frwa.2020.571044</u>.
- Lorenz, D.L., et al. 2010. Techniques for Estimating the Magnitude and Frequency of Peak Flows on Small Streams in Minnesota Based on Data through Water Year 2005. U.S. Geological Survey Scientific Investigations Report 2009-5250. U.S. Geological Survey, Reston, VA. <u>https://pubs.usgs.gov/sir/2009/5250/pdf/sir2009-5250.pdf</u>.

- Lukas, V., and V. Baez. 2021. "3D Elevation Program—Federal best practices." U.S. Geological Survey Fact Sheet 2020–3062. U.S. Geological Survey, Reston, VA. 2 pp. https://doi.org/10.3133/fs20203062.
- Lynch, L.M., *et al.* 2019. "River channel connectivity shifts metabolite composition and dissolved organic matter chemistry." *Nature Communications* 10: 459.
- Lytle, D.A., and N.L. Poff. 2004. "Adaptation to Natural Flow Regimes." *Trends in Ecology and Evolution* 19: 94-100.
- Lytle, D.A., *et al.* 2008. "Evolution of Aquatic Insect Behaviors across a Gradient of Disturbance Predictability." *Proceedings of the Royal Society Series B* 275: 453-462.
- Macdonald, D.M.J., *et al.* 2018. "Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications." *Ecological Engineering* 123: 226-237.
- Madalena, C. 2021. Comments of Pueblo of Jemez in Response to Request for Recommendations on Defining "Waters of the United States." Clarice Madalena, Interim Director, Natural Resources Department, to Administrator Michael Regan, Acting Assistant Secretary of the Army for Civil Works Jaime A. Pinkham, John Goodin, and Vance F. Stewart III. Pueblo of Jemez, October 4, 2021. 13 pp. <u>https://www.epa.gov/system/files/documents/2021-11/pueblo1_0.pdf</u>.
- Magana, H.A. 2013. "Flood Pulse Trophic Dynamics of Larval Fishes in a Restored Arid-Land River Floodplain, Middle Rio Grande, Los Lunas, New Mexico." *Reviews in Fish Biology and Fisheries* 23: 507-521. <u>https://doi.org/10.1007/s11160-013-9313-y</u>.
- Magliozzi, C., *et al.* 2018. "Toward a conceptual framework of hyporheic exchange across spatial scales." *Hydrology and Earth System Sciences* 22: 6163-6185.
- Maine Department of Environmental Protection. 2006. *Phosphorus Control Action Plan and Total Maximum Daily (Annual Phosphorous) Load Report, Daigle Pond, New Canada, Aroostook County, Maine*, Daigle Pond PCAP – TMDL Report, Maine DEPLW – 0789. <u>https://www.maine.gov/dep/water/monitoring/tmdl/2006/daigle_pond.pdf</u>.
- Malard, F., *et al.* 1999. "Shifting Dominance of Subcatchment Water Sources and Flow Paths in a Glacial Floodplain, Val Roseg, Switzerland." *Arctic, Antarctic, and Alpine Research* 31: 135-150.
- Malard, F., *et al.* 2002. "A Landscape Perspective on Surface-Subsurface Hydrological Exchanges in River Corridors." *Freshwater Biology* 47: 621-640.
- Malard, F., *et al.* 2006. "Flood-Pulse and Riverscape Dynamics in a Braided Glacier River." *Ecology* 87: 704-716.

Malmqvist, B. 2002. "Aquatic Invertebrates in Riverine Landscapes." Freshwater Biology 47: 679-694.

- Mantua, N., *et al.* 2010. "Climate change impacts on streamflow extremes and summertime stream temperature and their possible consequences for freshwater salmon habitat in Washington State." *Climatic Change* 102: 187–223. https://doi.org/10.1007/s10584-010-9845-2.
- Marcarelli, A.M., *et al.* 2019. "Of Small Streams and Great Lakes: Integrating Tributaries to Understand the Ecology and Biogeochemistry of Lake Superior." *Journal of the American Water Resources Association* 55: 442-458.
- Marron, D.C. 1989. "The Transport of Mine Tailings as Suspended Sediment in the Belle Fourche River, West-central South Dakota, USA." *International Association of Hydrologic Sciences* 184:19-26.
- Marteau, B., *et al.* 2017. "The importance of a small ephemeral tributary for fine sediment dynamics in a main-stem river." *River Research and Applications* 33: 1564-1574.
- Marteau, B., *et al.* 2020. "Geomorphological response to system-scale river rehabilitation I: Sediment supply from a reconnected tributary." *River Research and Applications* 36: 1488-1503.
- Martens, K.D., and P.J. Connolly. 2014. "Juvenile Anadromous Salmonid Production in Upper Columbia River Side Channels with Different Levels of Hydrological Connection." *Transactions of the American Fisheries Society* 143: 757-767.
- Martin, A.R., *et al.* 2019. "Seasonal and Intra-event Nutrient Levels in Farmed Prairie Potholes of the Des Moines Lobe." *Transactions of the ASABE* 62: 1607-1617.
- Martinez-Martinez, E., et al. 2014. "Modeling the Hydrological Significance of Wetland Restoration Scenarios." Journal of Environmental Management 133: 121-134. https://doi.org/10.1016/j.jenvman.2013.11.046.
- Marton, J.M., *et al.* 2014. "USDA Conservation Practices Increase Carbon Storage and Water Quality Improvement Functions: An Example from Ohio." *Restoration Ecology* 22:117-124.
- Marton, J.M., *et al.* 2015. "Geographically isolated wetlands are important biogeochemical reactors on the landscape." *BioScience* 65: 408-418.
- Matheney, R.K., and P.J. Gerla. 1996. "Environmental Isotopic Evidence for the Origins of Ground and Surface Water in a Prairie Discharge Wetland." *Wetlands* 16: 109-120.
- Matheney, M.P., and C.F. Rabeni. 1995. "Patterns of Movement and Habitat use by Northern Hogsuckers in an Ozark Stream." *Transactions of the American Fisheries Society* 124: 886-897.
- Matthai, H.F. 1969. *Floods of June 1965 in South Platte River Basin, Colorado*. Water Supply Paper 1850-B. U.S. Geological Survey, Washington, D.C. <u>https://pubs.usgs.gov/wsp/1850b/report.pdf</u>.
- Matthews J.W., *et al.* 2016. "Field Verification of Original and Updated National Wetlands Inventory Maps in three Metropolitan Areas in Illinois, USA." *Wetlands* 36: 1155–1165.
- Matthews, W.J. 1988. "North American Prairie Streams as Systems for Ecological Study." *Journal of the North American Benthological Society* 7: 387-409.

- Matthews, W.J., and H.W. Robinson. 1998. "Influence of Drainage Connectivity, Drainage Area and Regional Species Richness on Fishes of the Interior Highlands in Arkansas." *American Midland Naturalist* 139: 1-19.
- Matthews, J.W., *et al.* 2016. "Field Verification of Original and Updated National Wetlands Inventory Maps in three Metropolitan Areas in Illinois, USA." *Wetlands*. 36: 1155-1165
- Matthews, W.J., et al. 1985. "Fishes of Oklahoma springs." Southwestern Naturalist 30: 23-32.
- Matthews, W.J., *et al.* 2005. "Southern Plains Rivers." Pages 283-325 in *Rivers of North America*. A.C. Benke and C.E. Cushing, editors. Elsevier Academic Press, Burlington, MA.
- May, C.L., and R.E. Gresswell. 2003. "Processes and Rates of Sediment and Wood Accumulation in Headwater Streams of the Oregon Coast Range, USA." *Earth Surface Processes and Landforms* 28: 409-424.
- May, C.L., and D.C. Lee. 2004. "The Relationship among In-Channel Sediment Storage, Pool Depth, and Summer Survival of Juvenile Salmonids in Oregon Coast Range Streams." *North American Journal of Fisheries Management* 24: 761-774.
- Mazor, R.D., et al. 2021. User Manual for a Beta Streamflow Duration Assessment Method for the Arid West of the United States. Version 1.0. EPA-800-5-21001. U.S. Environmental Protection Agency, Washington, D.C. 83 pp. Available at: <u>https://www.epa.gov/sites/default/files/2021-03/documents/user_manual_beta_sdam_aw.pdf</u>.
- Mazor, R.D., et al. 2021. User Manual for a Beta Streamflow Duration Assessment Method for the Western Mountains of the United States. Version 1.0. EPA-840-B-21008. U.S. Environmental Protection Agency, Washington, D.C. 116 pp. Available at: <u>https://www.epa.gov/system/files/documents/2021-12/beta-sdam-for-the-wm-user-manual.pdf</u>.
- McClain, M.E., *et al.* 2003. "Biogeochemical Hot Spots and Hot Moments at the interface of Terrestrial and Aquatic Ecosystems." *Ecosystems* 6: 301-312.
- McDonough, O.T., *et al.* 2015. "Surface Hydrologic Connectivity between Delmarva Bay Wetlands and Nearby Streams Along a Gradient of Agricultural Alteration." *Wetlands* 35(1): 41-53.
- McEachern, P., *et al.* 2006. "Landscape Control of Water Chemistry in Northern Boreal Streams of Alberta." *Journal of Hydrology* 323: 303-324.
- McGlynn, B.L., and J.J. McDonnell. 2003. "Quantifying the Relative Contributions of Riparian and Hillslope Zones to Catchment Runoff." *Water Resources Research* 39: 1310. https://doi.org/10.1029/2003WR002091.
- McGlynn, B.L., *et al.* 2004. "Scale Effects on Headwater Catchment Runoff Timing, Flow Sources, and Groundwater-Streamflow Relations." *Water Resources Research* 40: W07504.
- McGuire, K.J., and J.J. McDonnell. 2010. "Hydrological connectivity of hillslopes and streams: Characteristic time scales and nonlinearities." *Water Resources Research* 46: W10543.

- McGuire, K.J., *et al.* 2005. "The Role of Topography on Catchment-Scale Water Residence Time." *Water Resources Research* 41: W05002.
- McLaughlin, D.L., *et al.* 2014. "A significant nexus: Geographically isolated wetlands influence landscape hydrology." *Water Resources Research* 50: 7153-7166.
- McLaughlin, J.W., *et al.* 2011. "Biogeochemical Cycling and Chemical Fluxes in a Managed Northern Forested Wetland, Michigan, USA." *Forest Ecology and Management* 261:649-661.
- McManamay, R., and C. DeRolph. 2019. "A stream classification system for the conterminous United States." *Scientific Data* 6: 190017. <u>https://doi.org/10.1038/sdata.2019.17</u>.
- Mekonnen, B.A., *et al.* 2016." Incorporating landscape depression heterogeneity into the Soil and Water Assessment Tool (SWAT) using a probability distribution." *Hydrological Processes* 30: 2373-2389.
- Meffe, G.K. 1984. "Effects of Abiotic Disturbance on Coexistence of Predator-Prey Fish Species." *Ecology* 65: 1525–1534.
- Mengistu, S.G., *et al.* 2013a. "Nutrient export from catchments on forested landscapes reveals complex nonstationary and stationary climate signals." *Water Resources Research* 49 (6): 3863-3880. https://doi.org/10.1002/wrcr.20302.
- Mengistu, S.G., *et al.* 2013b. "Russian nesting dolls effect Using wavelet analysis to reveal nonstationary and nested stationary signals in water yield from catchments on a northern forested landscape." *Hydrological Processes* 27: 669-686.
- Mengistu, S.G., et al. 2020. "Wetland Flowpaths Mediate Nitrogen and Phosphorus Concentrations across the Upper Mississippi River Basin." Journal of the American Water Resources Association: 1-18. https://doi.org/10.1111/1752-1688.12885.
- Merriam, J.L, *et al.* 2002. "Characterizing Nitrogen Dynamics, Retention and Transport in a Tropical Rainforest Stream Using an in situ N-15 Addition." *Freshwater Biology* 47: 143-160.
- Merritt, A.M., B. Lane, and C.P. Hawkins. 2021. "Classification and Prediction of Natural Streamflow Regimes in Arid Regions of the USA." *Water* 13(3): 380. <u>https://doi.org/10.3390/w13030380</u>.
- Mersel, M.K., and R.W. Lichvar. 2014. A Guide to Ordinary High Water Mark (OHWM) Delineation for Non-perennial Streams in the Western Mountains, Valleys, and Coast Region of the United States. ERDC/CRREL TR-14-13. U.S. Army Corps of Engineers, Hanover, NH. <u>https://usace.contentdm.oclc.org/digital/api/collection/p266001coll1/id/3691/download</u>.
- Mertes, L.A.K., *et al.* 1995. "Spatial Patterns of Hydrology, Geomorphology, and Vegetation on the Floodplain of the Amazon River in Brazil from a Remote Sensing Perspective." *Geomorphology* 13: 215-232.
- Mertes, L.A.K. 1997. "Documentation and Significance of the Perirheic Zone." *Water Resources Research* 33:1749-1762.

- Messager, M.L., *et al.* 2021. "Global prevalence of non-perennial rivers and streams." *Nature* 594: 391-397. https://doi.org/10.1038/s41586-021-03565-5.
- Meyer, A., *et al.* 2004. "The Effect of Low Flow and Stream Drying on the Distribution and Relative Abundance of the Alien Amphipod, *Echinogammarus berilloni* (Catta, 1878) in a Karstic Stream System (Westphalia, Germany)." *Crustaceana* 77: 909-922.
- Meyboom, P. 1964. "Three observations on streamflow depletion by phreatophytes." *Journal of Hydrology* 2: 248-261.
- Meyer, J.L. 1994. "The Microbial Loop in Flowing Waters." Microbial Ecology 28: 195-199.
- Meyer, J.L., and G.E. Likens. 1979. "Transport and Transformation of Phosphorus in a Forest Stream Ecosystem." *Ecology* 60: 1255-1269.
- Meyer, J.L., and J.B. Wallace. 2001. "Lost Linkages and Lotic Ecology: Rediscovering Small Streams." Pages 295-317 in *Ecology: Achievement and Challenge*. M. C. Press, N. J. Huntly, and S. Levin, editors. Blackwell Science, Oxford, UK.
- Meyer, J.L., *et al.* 1998. "Leaf Litter as a Source of Dissolved Organic Carbon in Streams." *Ecosystems* 1: 240-249.
- Meyer, J.L., *et al.* 2007. "The Contribution of Headwater Streams to Biodiversity in River Networks." *Journal of the American Water Resources Association* 43(1): 86-103.
- Michelson, C., *et al.* 2018. "Agricultural land cover does not affect the diet of Tree Swallows in wetlanddominated habitats." *The Condor* 120: 751-764.
- Michigan Natural Features Inventory. 2010. *Patterned Fen: Community Abstract*. Michigan State University Extension, Lansing, MI. <u>https://mnfi.anr.msu.edu/abstracts/ecology/Patterned_Fen.pdf</u>.
- Middleton, B. 2000. "Hydrochory, Seed Banks, and Regeneration Dynamics along the Landscape Boundaries of a Forested Wetland." *Plant Ecology* 146: 169-184.
- Mierle, G., and R. Ingram. 1991. "The Role of Humic Substances in the Mobilization of Mercury from Watersheds." *Water Air and Soil Pollution* 56: 349-357.
- Millennium Ecosystem Assessment. 2005. *Ecosystems and Human Well-being: Synthesis*. Island Press, Washington, D.C. <u>https://www.millenniumassessment.org/documents/document.356.aspx.pdf</u>.
- Miller, B., *et al.* 2013. "Case Studies of Fluorescent Groundwater Tracing in Recent Cave Research." *Proceedings of the 16th International Congress of Speleology, July 21–28, Brno* 3: 99-104. Czech Speleological Society, Praha, Czech Republic.
- Miller, E.L., *et al.* 1988. "Forest Harvest and Site Preparation Effects on Stormflow and Peakflow of Ephemeral Streams in the Ouachita Mountains." *Journal of Environmental Quality* 17: 212-218.
- Milner, V.S., *et al.* 2019. "The ecological importance of unregulated tributaries to macroinvertebrate diversity and community composition in a regulated river." *Hydrobiologia* 829: 291-305.
- Min, L., *et al.* 2020. "Numerical Approaches for Estimating Daily River Leakage from Arid Ephemeral Streams." *Water* 12(2): 499. <u>https://doi.org/10.3390/w12020499</u>.
- Minshall, G.W. 1967. "Role of Allochthonous Detritus in the Tropic Structure of a Woodland Springbrook Community." *Ecology* 48: 139-149.
- Minshall, G.W., *et al.* 1983. "Interbiome Comparison of Stream Ecosystem Dynamics." *Ecological Monographs* 53: 1-25.
- Mitsch, W.J., and J.G. Gosselink. 2007. Wetlands, 4th edition. John Wiley & Sons Inc., Hoboken, NJ.
- Mitsch, W.J., *et al.* 2001. "Reducing Nitrogen Loading to the Gulf of Mexico from the Mississippi River Basin: Strategies to Counter a Persistent Ecological Problem." *BioScience* 51(5): 373-388.
- Mitsch, W.J., et al. 2009. Wetland Ecosystems. 1st edition. John Wiley & Sons, Hoboken, NJ.
- Mitsch, W.J., et al. 2013. "Wetlands, Carbon, and Climate Change." Landscape Ecology 28(4), 583-597.
- Mitsch, W.J., et al. 2015. "Ecosystem Services of Wetlands." International Journal of Biodiversity Science, Ecosystem Services & Management 11:1-4. <u>https://doi.org/10.1080/21513732.2015.1006250</u>.
- Moll, D. 1990. "Population Sizes and Foraging Ecology in a Tropical Freshwater Stream Turtle Community." *Journal of Herpetology* 24: 48-53.
- Monteith, D.T., *et al.* 2007. "Dissolved organic carbon trends resulting from changes in atmospheric deposition chemistry." *Nature* 450: 537-540.
- Montgomery, D.R. 1999. "Process Domains and the River Continuum." *Journal of the American Water Resources Association* 35: 397-410.
- Mooney, R.J., *et al.* 2020. "Outsized nutrient contributions from small tributaries to a Great Lake." *Proceedings of the National Academy of Sciences of the United States of America* 117: 28175-28182. <u>https://doi.org/10.1073/pnas.2001376117</u>.

 Moore, D.J., and J. Ross. 2010. Spokane River and Lake Spokane Dissolved Oxygen Total Maximum Daily Load: Water Quality Improvement Report. Publication No. 07-10-073. Washington State Department of Ecology, Spokane, WA.
 <u>https://apps.ecology.wa.gov/publications/documents/0710073.pdf</u>. Accessed via Wayback Machine at <u>https://web.archive.org/web/20220120071032/https://apps.ecology.wa.gov/publications/documents/0710073.pdf</u>.

- Moore, J.W., et al. 2015. "Emergent stability in a large, free-flowing watershed." Ecology 96: 340-347.
- Moore, R.B., et al. 2019. User's guide for the national hydrography dataset plus (NHDPlus) high resolution. U.S. Geological Survey Open-File Report 2019–1096. U.S. Geological Survey, Reston, Virginia. 66 pp. <u>https://doi.org/10.3133/ofr20191096</u>.

- Moraghan, J.T. 1993. "Loss and Assimilation of 15N-nitrate Added to a North Dakota Cattail Marsh," *Aquatic Botany* 46:225-234.
- Moreno-Mateos, D., *et al.* 2008. "Relationships between Landscape Pattern, Wetland Characteristics, and Water Quality in Agricultural Catchments." *Journal of Environmental Quality* 37:2170-2180.
- Morita, K., and S. Yamamoto. 2002. "Effects of Habitat Fragmentation by Damming on the Persistence of Stream-dwelling Charr Populations." *Conservation Biology* 16:1318-1323.
- Morley, T.R., *et al.* 2011. "The Role of Headwater Wetlands in Altering Streamflow and Chemistry in a Maine, USA Catchment." *Journal of the American Water Resources Association* 47:337-349.
- Morrison, R.R., *et al.* 2018. "Spatial Relationships of Levees and Wetland Systems within Floodplains of the Wabash Basin, USA." *Journal of the American Water Resources Association* 54(4): 934-948. <u>https://doi.org/10.1111/1752-1688.12652</u>.
- Moscrip, A.L., and D.R. Montgomery. 1997. "Urbanization, Flood Frequency, and Salmon Abundance in Puget Lowland Streams." *Journal of the American Water Resources Association* 33:1289-1297 (1997).
- Mueller, N., et al. 2016. "Water observations from space: Mapping surface water from 25 years of Landsat imagery across Australia." *Remote Sensing of Environment* 174: 341-352. <u>https://www.sciencedirect.com/science/article/pii/S0034425715301929</u>.
- Mukundan, R., *et al.* 2013. "Suspended sediment source areas and future climate impact on soil erosion and sediment yield in a New York City water supply watershed, USA." *Geomorphology* 183: 110-119. https://www.sciencedirect.com/science/article/pii/S0169555X12003054.
- Mulholland, P.J., and W.R. Hill. 1997. "Seasonal Patterns in Streamwater Nutrient and Dissolved Organic Carbon Concentrations: Separating Catchment Flow Path and In-Stream Effects." *Water Resources Research* 33: 1297-1306.
- Mulholland, P.J. 2004. "The Importance of In-stream Uptake for Regulating Stream Concentrations and Outputs of N and P from a Forested Watershed: Evidence from Long-Term Chemistry Records for Walker Branch Watershed." *Biogeochemistry* 70: 403-426.
- Mulholland, P.J., and E.J. Kuenzler. 1979. "Organic Carbon Export from Upland and Forested Wetland Watersheds." *Limnology and Oceanography* 24: 960-966.
- Mulholland, P.J., *et al.* 1988. "Production of Soluble, High Molecular Weight Phosphorus and Its Subsequent Uptake by Stream Detritus." *Verhandlungen des Internationalen Verein Limnologie* 23: 1190-1197.
- Mulholland, P.J., *et al.* 1995. "Longitudinal Patterns of Nutrient Cycling and Periphyton Characteristics in Streams: a Test of Upstream-Downstream Linkage." *Journal of the North American Benthological Society* 14: 357-370.

- Mulholland, P.J., *et al.* 1997. "Effects of climate change on freshwater ecosystems of the south-eastern United States and the Gulf Coast of Mexico." *Hydrological Processes* 11: 949–970.
- Mulholland, P.J., *et al.* 2008. "Stream Denitrification across Biomes and Its Response to Anthropogenic Nitrate Loading." *Nature* 452: 202-205.
- Mulhouse, J.M., and S.M. Galatowitsch. 2003. "Revegetation of Prairie Pothole Wetlands in the Midcontinental US: Twelve Years Post-Reflooding." *Plant Ecology* 169: 143-159.
- Müller, K. 1982. "The Colonization Cycle of Insects." Oecologia 53: 202-207.
- Multi-Resolution Land Characteristics Consortium. "National Land Cover Database 2019 (NLCD2019) Statistics for 2019." Accessed at <u>https://www.mrlc.gov/data/statistics/national-land-cover-database-2019-nlcd2019-statistics-2019</u>. ("MRLC").
- Murkin, H.R., and P.J. Caldwell. 2000. "Avian Use of Prairie Wetlands." Pages. 249-286 in *Prairie Wetland Ecology: The Contribution of the Marsh Ecology Research Program.* H.R. Murkin, *et al.*, ed. Iowa State University Press, Ames, IA.
- Murphy, R.R., *et al.* 2011. "Long-Term Trends in Chesapeake Bay Seasonal Hypoxia, Stratification, and Nutrient Loading." *Estuaries and Coasts* 34(6): 1293-1309.
- Mushet, D., *et al.* 2013. "Complex spatial dynamics maintain northern leopard frog (*Lithobates pipiens*) genetic diversity in a temporally varying landscape." *Herpetological Conservation and Biology* 8: 163-175.
- Mushet, D., *et al.* 2015. "Geographically Isolated Wetlands: Rethinking a Misnomer. *Wetlands* 35: 423-431.
- Mushet, D.M., et al. 2019. "Differing Modes of Biotic Connectivity within Freshwater Ecosystem Mosaics." Journal of the American Water Resources Association 55 (2): 307-317. <u>https://doi.org/10.1111/1752-1688.12683</u>.
- Nadeau, T.-L. 2015. Streamflow Duration Assessment Method for the Pacific Northwest. EPA 910-K-14-001, U.S. Environmental Protection Agency, Region 10, Seattle, WA. 36 pp. Available at: <u>https://www.epa.gov/sites/default/files/2016-</u> 01/documents/streamflow duration assessment method pacific northwest 2015.pdf.
- Nadeau, T.-L., and M.C. Rains. 2007. "Hydrological Connectivity Between Headwater Streams and Downstream Waters: How Science Can Inform Policy." *Journal of the American Water Resources Association* 43: 118-133.
- Nadeau, T.-L., *et al.* 2015. "Validation of Rapid Assessment Methods to Determine Streamflow Duration Classes in the Pacific Northwest, USA." *Environmental Management* 56: 34–53. <u>https://doi.org/10.1007/s00267-015-0466-4</u>.
- Nag, S.K., *et al.* 2017. "Emission of Greenhouse Gases and Soil Carbon Sequestration in a Riparian Marsh Wetland in Central Ohio." *Environmental Monitoring and Assessment* 189(11): 580.

- Nahlik, A.M., and M.S. Fennessy. 2016. "Carbon Storage in US Wetlands." *Nature Communications* 7(1): 13835. <u>https://doi.org/10.1038/ncomms13835</u>.
- Naiman, R.J., and R.E. Bilby. 1998. *River ecology and management: Lessons from the Pacific coastal ecoregion*. Springer-Verlag, New York, NY.
- Naiman, R.J., and H. Decamps. 1997. "The Ecology of Interfaces: Riparian Zones." *Annual Review of Ecology and Systematics* 28:621-658.
- Naiman, R.J., et al. 1988. "Alteration of North-American streams by beaver." BioScience 38: 753-762.
- Naiman, R.J., *et al.* 1994. "Beaver influences on the long-term biogeochemical characteristics of boreal forest drainage networks." *Ecology* 75: 905-921.
- Naiman, R.J., et al. 2005. *Riparia: Ecology, conservation, and management of streamside communities.* Elsevier Academic Press, Burlington, MA.
- Nakamura, F., and F.J. Swanson. 1993. "Effects of Coarse Woody Debris on Morphology and Sediment Storage of a Mountain Stream System in Western Oregon." *Earth Surface Processes and Landforms* 18:43-61.
- Nakano, S., and M. Murakami. 2001. "Reciprocal Subsidies: Dynamic Interdependence between Terrestrial and Aquatic Food Webs." *Proceedings of the National Academy of Sciences USA* 98: 166-170.
- Nanson, G.C., and J.C. Croke. 1992. "A Genetic Classification of Floodplains." *Geomorphology* 4: 459-486.
- Narayan, Siddharth, *et al.* 2017. "The Value of Coastal Wetlands for Flood Damage Reduction in the Northeastern USA." *Scientific Reports* 7: 9463. <u>https://doi.org/10.1038/s41598-017-09269-z</u>.
- Nasab, M.T., and X. Chu. 2020. "Macro-HyProS: A new macro-scale hydrologic processes simulator for depression-dominated cold climate regions." *Journal of Hydrology* 580: 124366.
- Nasab, M.T., *et al.* 2017. "SWAT Modeling for Depression-Dominated Areas: How Do Depressions Manipulate Hydrologic Modeling?" *Water* 9: 58.
- National Association of Wetland Managers. 2022. Comments of the National Association of Wetland Managers on the Revised Definition of "Waters of the United States" (Docket # EPA-HQ-OW-2021-0602; FRL -6027.4-03-OW). Docket ID No. EPA-HQ-OW-2021-0602-0234. https://www.regulations.gov/comment/EPA-HQ-OW-2021-0602-0234.
- National Aeronautics and Space Administration. "Application Areas." Surface Water and Ocean Topography, Jet Propulsion Laboratory, California Institute of Technology. Accessed at <u>https://swot.jpl.nasa.gov/applications/applications-areas/</u>. ("NASA.")
- National Aeronautics and Space Administration. 2019. "Water: Vital for Life and Civilization." NISAR: The NASA-ISRO SAR Mission. Jet Propulsion Laboratory, California Institute of Technology, Pasadena, California. Accessed at

https://nisar.jpl.nasa.gov/system/documents/files/15_NISARApplications_SoilMoisture1.pdf. ("NASA 2019)."

- National Aeronautics and Space Administration. 2021. "How Does Climate Change Affect Precipitation?" Accessed at <u>https://gpm.nasa.gov/resources/faq/how-does-climate-change-affect-precipitation</u>. ("NASA 2021.")
- National Geographic Society. 2022. "Levee." Resource Library, Encyclopedic Entry. Accessed at <u>https://education.nationalgeographic.org/resource/levee</u>.
- National Oceanic and Atmospheric Administration. "National Trends." National Centers for Environmental Information. Accessed at <u>https://www.ncei.noaa.gov/access/monitoring/us-trends</u>. ("NOAA a.")
- National Oceanic and Atmospheric Administration. 2019. "Defining Climate Normals in New Ways." National Centers for Environmental Information. <u>https://www.ncdc.noaa.gov/news/defining-climate-normals-new-ways</u>. Accessed September 29, 2019 version via Wayback Machine at <u>https://web.archive.org/web/20190925192055/https://www.ncdc.noaa.gov/news/defining-climate-normals-new-ways</u>. ("NOAA 2019.")
- National Oceanic and Atmospheric Administration. 2021. "NOAA Delivers New U.S. Climate Normals." National Centers for Environmental Information. Accessed at <u>https://www.ncei.noaa.gov/news/noaa-delivers-new-us-climate-normals</u>. ("NOAA 2021.")
- National Oceanic and Atmospheric Administration. 2022. "Critical Habitat." <u>National Marine Fisheries</u> <u>Service.</u> Accessed at <u>https://www.fisheries.noaa.gov/national/endangered-species-</u> <u>conservation/critical-habitat</u>. ("NOAA 2022a.")
- National Oceanic and Atmospheric Administration. 2022. "U.S. Climate Normals." National Centers for Environmental Information. Accessed at https://www.ncei.noaa.gov/products/land-based-station/usclimate-normals. ("NOAA 2022b.")
- National Research Council. 1995. *Wetlands: Characteristics and Boundaries*. National Academy Press, Washington, D.C. <u>https://doi.org/10.17226/4766</u>.
- National Research Council. 1997. *Watershed Research in the U.S. Geological Survey*. National Academy Press, Washington, D.C. <u>https://doi.org/10.17226/5589</u>.
- National Research Council. 2002. *Riparian Areas: Functions and Strategies for Management*. National Academy Press, Washington, D.C. <u>https://doi.org/10.17226/10327</u>.
- Natural Resources Conservation Service. "Automated Snow Monitoring." U.S. Department of Agriculture, Natural Resources Conservation Service. Accessed at <u>https://www.nrcs.usda.gov/wps/portal/wcc/home/aboutUs/monitoringPrograms/automatedSnowMoni</u> <u>toring/</u>. ("NRCS a.")

- Natural Resources Conservation Service. "National Water and Climate Center Overview." U.S. Department of Agriculture, Natural Resources Conservation Service. Accessed at https://www.nrcs.usda.gov/wps/portal/wcc/home/aboutUs/nwccOverview/. ("NRCS b.")
- Natural Resources Conservation Service. 2019. "Web Soil Survey." Department of Agriculture, Natural Resources Conservation Service. Accessed at https://websoilsurvey.sc.egov.usda.gov/App/HomePage.htm. ("NRCS 2019.")
- Needelman, B. A., *et al.* 2007. "Drainage Ditches: Improved management of agricultural drainage ditches for water quality protection: An overview." *Journal of Soil and Water Conservation* 62(4): 171-178.
- Neff, B.P., and D.O. Rosenberry. 2018. "Groundwater Connectivity of Upland-Embedded Wetlands in the Prairie Pothole Region." *Wetlands* 38: 51-63.
- Neitsch, S.L., et al. 2011. Soil and Water Assessment Tool: Theoretical Documentation Version 2009. Texas Water Resources Institute Technical Report No. 406. Texas Water Resources Institute, College Station, Texas. Available at: <u>https://swat.tamu.edu/media/99192/swat2009-theory.pdf</u>.
- New Mexico. 2020. State of New Mexico Water Quality Management Plan & Continuing Planning Process, Appendix C: Hydrology Protocol for the Determination of Uses Supported by Ephemeral, Intermittent, and Perennial Waters. Approved Revision October 23, 2020. Available at: <u>https://www.env.nm.gov/wp-content/uploads/sites/25/2019/11/WQMP-CPP-Appendix-C-Hydrology-Protocol-20201023-APPROVED.pdf</u>.
- New York Department of Environmental Conservation. 2021. *Final Amended Phase III Watershed Implementation Plan.* Division of Water, Bureau of Watershed Resource Management. <u>https://www.dec.ny.gov/docs/water_pdf/amendedwipiii.pdf</u>.
- Newbold, J.D., et al. 1981. "Measuring Nutrient Spiraling in Streams." Canadian Journal of Fisheries and Aquatic Sciences 38:860-863.
- Newbold, J.D., et al. 1982a. "Organic carbon spiralling in stream ecosystems." Oikos 38: 266-272.
- Newbold, J.D., *et al.* 1982b. "Nutrient spiraling in streams: Implications for nutrient limitation and invertebrate activity." *The American Naturalist* 120: 628-665.
- Newbold, J.D., *et al.* 1983. "Phosphorus Dynamics in a Woodland Stream Ecosystem: a Study of Nutrient Spiraling." *Ecology* 64: 1249-1265.
- Newbold, J.D., et al. 2010. "Water Quality Functions of a 15-Year-Old Riparian Forest Buffer System." Journal of the American Water Resources Association 46: 299-310.
- Newman, D.G., and C.R. Griffin. 1994. "Wetland Use by River Otters in Massachusetts." *Journal of Wildlife Management* 58: 18-23.
- Nez, J., and M. Lizer. 2021. Re: Navajo Nation's Comments in Response to Letter to Tribal Leaders, "Notification of Consultation and Coordination on Revising the Definition of 'Waters of the United States'" (July 29, 2021) and Notice of Public Meetings Regarding "Waters of the United States";

Establishment of a Public Docket; Request for Recommendations, 86 Fed. Reg. 147 41911 (August 4, 2021)." President Jonathan Nez and Vice President Myron Lizer to Administrator Michael S. Regan and Acting Assistant Secretary of the Army for Civil Works Jaime A. Pinkham. Navajo Nation, October 4, 2021. 13 pp. <u>https://www.epa.gov/system/files/documents/2021-11/navajo1.pdf</u>.

NHD. 2008. National Hydrography Dataset. U.S. Geological Survey.

- Nichols, C. 1994. Map Accuracy of National Wetlands Inventory Maps for Areas Subject to Land Use Regulation Commission Jurisdiction. Ecological Services report R5-94/6. U.S. Fish and Wildlife Service, Hadley, MA. 14 pp. Accessed at <u>https://www.fws.gov/wetlands/Documents%5CMap-Accuracy-of-National-Wetlands-Inventory-Maps-for-Areas-Subject-to-Maine-Land-Use-Regulation-Commission-Jurisdiction.pdf.</u>
- Nilsson, C., *et al.* 2005. "Fragmentation and Flow Regulation of the World's Large River Systems." *Science* 308(5720):405-408.
- Nilsson, C., *et al.* 2010. "The Role of Hydrochory in Structuring Riparian and Wetland Vegetation." *Biological Reviews* 85:837-858.
- Nitzsche, K.N., *et al.* 2017. "Land-use and hydroperiod affect kettle hole sediment carbon and nitrogen biogeochemistry." *Science of The Total Environment* 574: 46-56.
- Nixon, S.J., and V.J. Lee. 1986. Wetlands and Water Quality: A Regional Review of Recent Research in the United States on the Role of Freshwater and Saltwater Wetlands as Sources, Sinks, and Transformers of Nitrogen, Phosphorus, and Various Heavy Metals. Technical Report Y-86-2. U.S. Army Corp of Engineers, Waterways Experiment Station, Vicksburg, MS. <u>https://hdl.handle.net/11681/6489</u>.
- Nolan K.M., *et al.* 1987. "Bankfull discharge and sediment transport in northwestern California." Pages 439-449 in *Erosion and Sedimentation in the Pacific Rim.* R Beschta, *et al.*, ed. International Association of Hydrological Sciences Pub. No. 165.
- North Carolina Division of Water Quality. 2010. *Methodology for Identification of Intermittent and Perennial Streams and Their Origins*. Version 4.11. North Carolina Department of Environment and Natural Resources, Division of Water Quality, Raleigh, NC. 43 pp. Available at: <u>https://files.nc.gov/ncdeq/Water%20Quality/Surface%20Water%20Protection/401/Policies_Guides_Manuals/StreamID_v_4point11_Final_sept_01_2010.pdf</u>.
- Novak, P., et al. 2010. Hydraulic Modeling—An Introduction: Principles, Methods and Applications. Taylor and Francis, New York, NY.
- Novitzki, R.P. 1979. "Hydrologic Characteristics of Wisconsin's Wetlands and Their Influence on Floods," *in* P. Greeson, *et al.*, ed., *Wetland Functions and Values: The State of Our Understanding*, Proceedings of the National Symposium on Wetlands (Minneapolis, MN: American Water Resources Association), pp. 377-388.
- Nuff, J.C., and G.P. Asner. 2001. "Dissolved Organic Carbon in Terrestrial Ecosystems: Synthesis and a Model." *Ecosystems* 4: 29-48.

- O'Connor, T., and D. Whitall. 2007. "Linking Hypoxia to Shrimp Catch in the Northern Gulf of Mexico." *Marine Pollution Bulletin* 54 (4): 460-463.
- O'Driscoll, M.A., and D.R. DeWalle. 2010. "Seeps Regulate Stream Nitrate Concentration in a Forested Appalachian Catchment." *Journal of Environmental Quality* 39: 420-431.
- O'Driscoll, Michael A., and Richard R. Parizek. 2003. "The Hydrologic Catchment Area of a Chain of Karst Wetlands in Central Pennsylvania, USA," *Wetlands* 23: 171-79.
- O'Melia, Charles R. 1998. "Coagulation and sedimentation in lakes, reservoirs and water treatment plants." *Water Science and Technology* 37(2): 129-135. <u>https://doi.org/https://doi.org/10.1016/S0273-1223(98)00018-3</u>.
- Obolewski, K., *et al.* 2009. "Effect of Hydrological Connectivity on the Molluscan Community Structure in Oxbow Lakes of the Lyna River." *Oceanological and Hydrobiological Studies* 38:75-88.
- Odum, W.E. 1988. "Non-Tidal Freshwater Wetlands in Virginia." Virginia Journal of Natural Resources Law 7: 421-434.
- Ohio Environmental Protection Agency. 2015. *Nonpoint Source Impacts on Primary Headwater Streams*. Ohio EPA Division of Surface Water, Columbus, OH. https://epa.ohio.gov/static/Portals/35/wqs/headwaters/HWH_nonpoint.pdf.
- Ohio Environmental Protection Agency. 2020. *Field Methods for Evaluating Primary Headwater Streams in Ohio*. Version 4.1. Ohio EPA Division of Surface Water, Columbus, Ohio. 130 pp. https://epa.ohio.gov/static/Portals/35/credibledata/PHWHManual_2020_Ver_4_1_May_2020_Final.p_df.
- Olson, S.A., and M.C. Brouillette. 2006. A logistic regression equation for estimating the probability of a stream in Vermont having intermittent flow. U.S. Geological Survey Scientific Investigations Report 2006–5217. U.S. Geological Survey, Reston, Virginia. 15 p. https://pubs.usgs.gov/sir/2006/5217/pdf/SIR2006-5217_report.pdf.
- Omernik, J.M. 1987. "Ecoregions of the conterminous United States." Annals of the Association of American Geographers 77(1): 118-125.
- Omernik, J.M. 1995. "Ecoregions: A spatial framework for environmental management." Pages 49-62 in Biological Assessment and Criteria: Tools for Water Resource Planning and Decision Making. W.S. Davis, and T.P. Simon, editors. Lewis Publishers, Boca Raton, FL.
- Omernik, J.M., and R.G. Bailey. 1997. "Distinguishing Between Watersheds and Ecoregions." *Journal of the American Water Resources Association* 33.5: 939-40.
- Omernik, J.M., and G.E. Griffith. 2014. "Ecoregions of the Conterminous United States: Evolution of a Hierarchical Spatial Framework." *Environmental Management* 54: 1249–1266. https://doi.org/10.1007/s00267-014-0364-1.

- Orr, C.H., *et al.* 2014. "Spatial Autocorrelation of Denitrification in a Restored and a Natural Floodplain." *Wetlands* 34: 89-100.
- Osborne, L.L., and D.A. Kovacic. 1993. "Riparian Vegetated Buffer Strips in Water-Quality Restoration and Stream Management." *Freshwater Biology* 29(2): 243-258.
- Osterkamp, W.R. 2008. Annotated Definitions of Selected Geomorphic Terms and Related Terms of Hydrology, Sedimentology, Soil Science and Ecology. USGS Open File Report 2008-1217. U.S. Department of the Interior, U.S. Geological Survey, Reston, VA. <u>https://doi.org/10.3133/ofr20081217</u>.
- Painter, K.J., *et al.* 2015. "Effects of in-channel beaver impoundments on mercury bioaccumulation in Rocky Mountain stream food webs." *Ecosphere* 6: art194.
- Park, J., *et al.* 2020. "Spatial and temporal variations in the groundwater contributing areas of inland wetlands." *Hydrological Processes* 34: 1117-1130.
- Paterson, K.G., and J.L. Schnoor. 1992. "Fate of Alachlor and Atrazine in a Riparian Zone Field Site." *Water Environment Research* 64: 274-283.
- Patten, D.T., *et al.* 2008. "Isolated Spring Wetlands in the Great Basin and Mojave Deserts, USA: Potential Response of Vegetation to Groundwater Withdrawal." *Environmental Management* 41: 398-413.
- Paul, M. J., et al. 2006. "Leaf Breakdown in Streams Differing in Catchment Land Use." Freshwater Biology 51: 1684-1695.
- Pease, A.A., *et al.* 2006. "Habitat and Resource Use by Larval and Juvenile Fishes in an Arid-land River (Rio Grande, New Mexico)." *Freshwater Biology* 51:475-486.
- Perkin, J.S., and K.B. Gido. 2011. "Stream Fragmentation Thresholds for a Reproductive Guild of Great Plains Fishes." *Fisheries* 36: 371-383.
- Person, H.S., et al. 1936. Little waters. A study of headwater streams and other little waters, their use and relations to the land. The Upstream Engineering Conference, The 75th Congress, 2nd Session. Senate Document No. 198. U.S. Government Printing Office, Washington, D.C.
- Peterjohn, W.T., and D.L. Correll. 1984. "Nutrient dynamics in an agricultural watershed: Observations on the role of a riparian forest." *Ecology* 65:1466-1475.
- Peterson, B.J., *et al.* 2001. "Control of Nitrogen Export from Watersheds by Headwater Streams." *Science* 292: 86-90.
- Peterson, E.E., *et al.* 2007. "Geostatistical Modelling on Stream Networks: Developing Valid Covariance Matrices Based on Hydrologic Distance and Stream Flow." *Freshwater Biology* 52:267-279.
- Petranka, J.W., and C.T. Holbrook. 2006. "Wetland Restoration for Amphibians: Should Local Sites Be Designed to Support Metapopulations or Patchy Populations?" *Restoration Ecology* 14: 404-411.

- Phillips, P.J., and R.J. Shedlock. 1993. "Hydrology and Chemistry of Groundwater and Seasonal Ponds in the Atlantic Coastal-Plain in Delaware, USA." *Journal of Hydrology* 141:157-178.
- Phillips, R.W., *et al.* 2011. "Connectivity and runoff dynamics in heterogeneous basins." *Hydrological Processes* 25: 3061-3075.
- Pires, A.M., *et al.* 1999. "Seasonal Changes in Fish Community Structure of Intermittent Streams in the Middle Reaches of the Guadiana Basin, Portugal." *Journal of Fish Biology* 54:235-249.
- Platania, S.P., and C.S. Altenbach. 1998. "Reproductive Strategies and Egg Types of Seven Rio Grande Basin Cyprinids." *Copeia* 1998: 559-569.
- Poff, N.L. 1996. "A Hydrogeography of Unregulated Streams in the United States and an Examination of Scale Dependence in Some Hydrological Descriptors." *Freshwater Biology* 36: 71-91.
- Poff, N.L., *et al.* 2006. "Hydrologic variation with land use across the contiguous United States: Geomorphic and ecological consequences for stream ecosystems." *Geomorphology* 79(3): 264-285.
- Poff, N.L., *et al.* 2007. "Homogenization of Regional River Dynamics by Dams and Global Biodiversity Implications." *Proceedings of the National Academy of Sciences of the United States of America* 104: 5732-5737.
- Pollux, B.J.A., et al. 2007. "Consequences of Intraspecific Seed-Size Variation in Sparganium emersum for Dispersal by Fish." Functional Ecology 21: 1084-1091.
- Poole, G.C. 2010. "Stream Hydrogeomorphology as a Physical Science Basis for Advances in Stream Ecology." *Journal of the North American Benthological Society* 29: 12-25.
- Poole, G.C., et al. 2006. "Multiscale Geomorphic Drivers of Groundwater Flow Paths: Subsurface Hydrologic Dynamics and Hyporheic Diversity." *Journal of the North American Benthological Society* 25: 288-303.
- Poppenga, S.K., et al. 2013. "Hydrography Change Detection: The Usefulness of Surface Channels Derived From LiDAR DEMs for Updating Mapped Hydrography." Journal of the American Water Resources Association 49: 371-389. <u>https://doi.org/10.1111/jawr.12027</u>.
- Power, G., et al. 1999. "Groundwater and Fish: Insights from Northern North America." *Hydrological Processes* 13: 401-422.
- Power, M.E., and W.E. Dietrich. 2002. "Food webs in river networks." *Ecological Research* 17: c451-471.
- Power, M.E., *et al.* 1995a. "How does floodplain width affect floodplain river ecology? A preliminary exploration using simulations." *Geomorphology* 13: 301-317.
- Power, M.E., et al. 1995b. "Hydraulic food-chain models." BioScience 45: 159-167.
- Powers, S.M., *et al.* 2012. "Nutrient Retention and the Problem of Hydrologic Disconnection in Streams and Wetlands." *Ecosystems* 15: 435-449.

- Preston, S.D., et al. 2011. "Factors Affecting Stream Nutrient Loads: A Synthesis of Regional SPARROW Model Results for the Continental United States." Journal of the American Water Resources Association 47: 891-915.
- Pringle, C.M. 2001. "Hydrologic Connectivity and the Management of Biological Reserves: A Global Perspective." *Ecological Applications* 11:981-998.
- Progar, D.J., and A.R. Modenke. 2002. "Insect Production from Temporary and Perennially Flowing Headwater Streams in Western Oregon." *Journal of Freshwater Ecology* 17:391-407.
- Puttock, A., *et al.* 2017. "Eurasian beaver activity increases water storage, attenuates flow and mitigates diffuse pollution from intensively-managed grasslands." *Science of the Total Environment* 576: 430-443.
- Pyron, M., et al. 2014. "Fish Assemblages of Floodplain Lakes in the Ohio River Basin." Northeastern Naturalist 21: 419-430.
- Pyzoha, J.E., *et al.* 2008. "A Conceptual Hydrologic Model for a Forested Carolina Bay Depressional Wetland on the Coastal Plain of South Carolina, USA." *Hydrological Processes* 22: 2689-2698.
- Quin, A. and G. Destouni. 2018. "Large-scale comparison of flow-variability dampening by lakes and wetlands in the landscape." *Land Degradation & Development* 29(10): 3617-3627.
- Quin, A., *et al.* 2015. "Dissecting the ecosystem service of large-scale pollutant retention: The role of wetlands and other landscape features." *Ambio* 44: S127-S137.
- Rabalais, N.N., et al. 2002. "Gulf of Mexico Hypoxia, a.k.a. 'the Dead Zone." Annual Review of Ecology and Systematics 33:235-263.
- Rabotyagov, S.S., *et al.* 2014. "The economics of dead zones: causes, impacts, policy challenges, and a model of the Gulf of Mexico hypoxic zone." *Review of Environmental Economics and Policy* 8(1): 58–79.
- Rahel, F.J. 2007. "Biogeographic Barriers, Connectivity and Homogenization of Freshwater Faunas: It's a Small World after All." *Freshwater Biology* 52(4): 696-710.
- Rains, M.C., *et al.* 2006. "The role of perched aquifers in hydrological connectivity and biogeochemical processes in vernal pool landscapes, Central Valley, California." *Hydrological Processes* 20: 1157-1175.
- Rains, M.C., *et al.* 2008. "Geological Control of Physical and Chemical Hydrology in California Vernal Pools." *Wetlands* 28:347-362.
- Rains, M.C., *et al.* 2016. "Geographically isolated wetlands are part of the hydrological landscape." *Hydrological Processes* 30: 153-160.
- Rajib, A., *et al.* 2020. "Surface depression and wetland water storage improves major river basin hydrologic predictions." *Water Resources Research* 56: e2019WR026561.

- Ranalli, A.J., and D.L. Macalady. 2010. "The importance of the riparian zone and in-stream processes in nitrate attenuation in undisturbed and agricultural watersheds – A review of the scientific literature." *Journal of Hydrology* 389: 406-415.
- Rankin Studio. 2019. *Amazing Flash Flood / Debris Flow Southern Utah HD* (July 19, 2019). https://www.youtube.com/watch?v=_yCnQuILmsM.
- Rassam, D.W., *et al.* 2006. "The Hydrology of Riparian Buffer Zones; Two Case Studies in an Ephemeral and a Perennial Stream." *Journal of Hydrology* 325:308-324.
- Rea, A., and K.D. Skinner. 2009. Estimated perennial streams of Idaho and related geospatial datasets. U.S. Geological Survey Data Series 412. U.S. Geological Survey, Reston, Virginia. 32 p. Available at <u>https://pubs.usgs.gov/ds/412/</u>.
- Reckendorfer, W., *et al.* 2006. "Floodplain Restoration by Reinforcing Hydrological Connectivity: Expected Effects on Aquatic Mollusc Communities." *Journal of Applied Ecology* 43:474-484.
- Redder, B.W., *et al.* 2021. "Groundwater contributions of flow and nitrogen in a headwater agricultural watershed." *Hydrological Processes* 35(5): e14179. <u>https://doi.org/10.1002/hyp.14179</u>.
- Reddy, K.R., and R.D. DeLaune. 2008. *Biogeochemistry of Wetlands: Science and Applications*. CRC Press: Boca Raton, FL.
- Reddy, K.R., et al. 1999. "Phosphorus Retention in Streams and Wetlands: A Review." Critical Reviews in Environmental Science and Technology 29: 83-146.
- Reed, D.J., *et al.* 2006. "Reducing the Effects of Dredged Material Levees on Coastal Marsh Function: Sediment Deposition and Nekton Utilization." *Environmental Management* 37(5): 671-685.
- Rees, G.N., *et al.* 2020. "Managed floodplain inundation maintains ecological function in lowland rivers." *Science of the Total Environment* 727: 138469.
- Reid, I., and J.B. Laronne. 1995. "Bed Load Sediment Transport in an Ephemeral Stream and a Comparison with Seasonal and Perennial Counterparts." *Water Resources Research* 31(3): 773–781. <u>https://doi.org/10.1029/94WR02233</u>.
- Reneau, S.L., *et al.* 2004. "Geomorphic Controls on Contaminant Distribution along an Ephemeral Stream." *Earth Surface Processes and Landforms* 29: 1209-1223.
- Renwick, W., *et al.* 2005. "The role of impoundments in the sediment budget of the conterminous United States." *Geomorphology* 71: 99-111.
- Rhoads, B.L. 1987. "Changes in Stream Channel Characteristics at Tributary Junctions." *Physical Geography* 8: 346-367.
- Richardson, C.J., *et al.* 2011. "Integrated stream and wetland restoration: A watershed approach to improved water quality on the landscape." *Ecological Engineering* 37(1): 25-39.
- Richardson, J.S. 2019. "Biological Diversity in Headwater Streams." Water 11: 366.

- Richardson, J.S., *et al.* 2005. "Riparian Communities Associated with Pacific Northwest Headwater Streams: Assemblages, Processes, and Uniqueness." *Journal of the American Water Resources Association* 41: 935-947.
- Rice, S.P., *et al.* 2008. "The Ecological Importance of Tributaries and Confluences." Pages 209-242 in *River Confluences, Tributaries and the Fluvial Network.* S.P. Rice, *et al.*, ed. John Wiley & Sons, Chichester, UK.
- Ries, K.G., III, *et al.* 2017. "StreamStats, Version 4." U.S. Geological Survey Fact Sheet 2017–3046. U.S. Geological Survey. 4 pp. Accessed at <u>https://pubs.usgs.gov/fs/2017/3046/fs20173046.pdf</u>.
- Rinne, J.N., and D. Miller. 2006. "Hydrology, Geomorphology and Management: Implications for Sustainability of Native Southwestern Fishes." *Reviews in Fisheries Science* 14: 91-110.
- Robinson, C.T., *et al.* 2002. "The Fauna of Dynamic Riverine Landscapes." *Freshwater Biology* 47: 661-677.
- Rodriguez-Iturbe, I., *et al.* 2009. "River networks as ecological corridors: A complex systems perspective for integrating hydrologic, geomorphologic, and ecologic dynamics." *Water Resources Research* 45: W01413.
- Rooney, R.C., *et al.* 2013. "River connectivity affects submerged and floating aquatic vegetation in floodplain wetlands." *Wetlands* 33: 1165-1177.
- Roscher, J.P. 1967. "Alga Dispersal by Muskrat Intestinal Contents." *Transactions of the American Microscopical Society* 86: 497-498.
- Rosenberry, D.O. and T.C. Winter. 1997. "Dynamics of Water-Table Fluctuations in an Upland between Two Prairie-Pothole Wetlands in North Dakota," *Journal of Hydrology* 191: 266-289.
- Rossi, C., and J.R. Nimmo. 1994. "Modeling of Soil Water Retention from Saturation to Oven Dryness." *Water Resources Research* 30: 701–08.
- Roy, A.G., and M.J. Woldenberg. 1986. "A Model for Changes in Channel Form at a River Confluence." *Journal of Geology* 94: 402-411.
- Roy, A.H., *et al.* 2009. "Urbanization affects the extent and hydrologic permanence of headwater streams in a midwestern US metropolitan area." *Journal of the North American Benthological Society* 28: 911-928.
- Royer, T.V., *et al.* 2004. "Transport and Fate of Nitrate in Headwater Agricultural Streams in Illinois. *Journal of Environmental Quality* 33: 1296-1304.
- Ruhi, A., *et al.* 2018. Tracking the pulse of the Earth's fresh waters. *Nature Sustainability* 1: 198–203. https://doi.org/10.1038/s41893-018-0047-7.
- Rupp, D.E., *et al.* 2021. "Changing River Network Synchrony Modulates Projected Increases in High Flows." *Water Resources Research* 57: e2020WR028713.

- Russell, P.P., *et al.* 2015. "A Spatially Explicit Model for Mapping Headwater Streams." *Journal of the American Water Resources Association* 51(1): 226-239. https://doi.org/10.1111/jawr.12250.
- Sabo, J.L., and E.M. Hagen. 2012. "A network theory for resource exchange between rivers and their watersheds." *Water Resources Research* 48: W04515.
- Saco, P.M., and P. Kumar. 2002. "Kinematic dispersion in stream networks 1. Coupling hydraulic and network geometry." *Water Resources Research* 38: 26-1-26-14.
- Samia, Y., and F. Lutscher. 2017. "Downstream flow and upstream movement determine the value of a stream reach for potadromous fish populations." *Theoretical Ecology* 10: 21-34.
- Sampath, P.V., *et al.* 2015. "Understanding the Groundwater Hydrology of a Geographically-Isolated Prairie Fen: Implications for Conservation." *PloS ONE* 10: e0140430.
- Sánchez-Carrillo, S., and M. Álvarez-Cobelas. 2018. "Stable isotopes as tracers in aquatic ecosystems. *Environmental Reviews* 26(1): 69-81. <u>https://doi.org/10.1139/er-2017-0040</u>.
- Sando, R., and K.W. Blasch. 2015. "Predicting alpine headwater stream intermittency: a case study in the northern Rocky Mountains." *Ecohydrology & Hydrobiology* 15 (2): 68-80. <u>https://doi.org/10.1016/j.ecohyd.2015.04.002</u>.
- Sanford, S.E., *et al.* 2007. "Scale-dependence of natural variability of flow regimes in a forested landscape." *Water Resources Research* 43: W08414.
- Santa Clara Pueblo Fire/Rescue/EMS Volunteer Department, Greg Lonewolf. 2017. #4 Santa Clara Pueblo Flash Flood Event 01 Sept 2013 (April 14, 2017). https://www.youtube.com/watch?v=nKOQzkRi4BQ.
- Sawin, R.S., et al. 1999. "Flint Hills Springs." Transactions of the Kansas Academy of Science 102: 1-31.
- Scanlon, T.M., *et al.* 2010. "Terrestrial and in-stream influences on the spatial variability of nitrate in a forested headwater catchment." *Journal of Geophysical Research: Biogeosciences* 115: G02022.
- Schalk, C.M., and T.M. Luhring. 2010. "Vagility of Aquatic Salamanders: Implications for Wetland Connectivity." *Journal of Herpetology* 44: 104-109.
- Schiller, K. E., *et al.* 2012. "Impact of artificial subsurface drainage on groundwater travel times and baseflow discharge in an agricultural watershed, Iowa (USA)." *Hydrological Processes* 26: 3092-3100.
- Schindler, D.E., *et al.* 2005. "Marine-Derived Nutrients, Commercial Fisheries, and Production of Salmon and Lake Algae in Alaska." *Ecology* 86: 3225-3231.
- Schindler, D.W., and P.J. Curtis. 1997. "The Role of DOC in Protecting Freshwaters Subjected to Climate Warming and Acidification from UV Exposure." *Biogeochemistry* 36: 1-8.
- Schlesinger, W. H., and C. S. Jones. 1984. "The comparative importance of overland runoff and mean annual rainfall to shrub communities of the Mojave Desert." *Botanical Gazette* 145:116-124.

Schlosser, I.J. 1991. "Stream Fish Ecology: A Landscape Perspective." *BioScience* 41(10):704-712.

- Schmadel, N.M., *et al.* 2019. "Small Ponds in Headwater Catchments Are a Dominant Influence on Regional Nutrient and Sediment Budgets." *Geophysical Research Letters* 46: 9669-9677.
- Schmidt, D.J., et al. 2018. "Genome-Wide SNPs Identify Limits to Connectivity in the Extreme Freshwater Disperser, Spangled Perch Leiopotherapon unicolor (Terapontidae)." Journal of Heredity 109: 320-325.
- Schmidt, J.P., *et al.* 2007. "Nitrogen Export from Coastal Plain Field Ditches." *Journal of Soil and Water Conservation* 62(4): 235-243.
- Schneider, R.L., and R.R. Sharitz. 1988. "Hydrochory and Regeneration in a Bald Cypress Water Tupelo Swamp Forest." *Ecology* 69: 1055-1063.
- Schofield, K.A., *et al.* 2018. "Biota Connect Aquatic Habitats throughout Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 54: 372-399.
- Schoof, R.R., *et al.* 1978. "Hydrologic effects of the Tonkawa Creek flood abatement program." *Water Resources Bulletin* 14: 629-639.
- Schramm, H.L., and M.A. Eggleton. 2006. "Applicability of the Flood-Pulse Concept in a Temporal Floodplain River Ecosystem: Thermal and Temporal Components." *River Research and Applications* 22: 543-553.
- Schroder, B. 2006. "Pattern, Process, and Function in Landscape Ecology and Catchment Hydrology— How Can Quantitative Landscape Ecology Support Predictions in Ungauged Basins? *Hydrology and Earth System Sciences* 10: 967-979.
- Schwalb, A.N., et al. 2010. "Dispersion of Freshwater Mussel Larvae in a Lowland River." *Limnology* and Oceanography 55: 628-638.
- Schwartz, S.S., and D.G. Jenkins. 2000. "Temporary Aquatic Habitats: Constraints and Opportunities." *Aquatic Ecology* 34:3-8.
- Scott, D.T., et al. 2014. "Floodplain biogeochemical processing of floodwaters in the Atchafalaya River Basin during the Mississippi River flood of 2011." *Journal of Geophysical Research: Biogeosciences* 119: 537-546.
- Scrivener, J.C., *et al.* 1994. "Juvenile Chinook salmon (*Oncorhynchus tshawytscha*) utilization of Hawks Creek, a small and nonnatal tributary of the Upper Fraser River." *Canadian Journal of Fisheries and Aquatic Sciences* 51: 1139-1146.
- Sear, D.A., *et al.* 2010. "Log Jam Controls on Channel: Floodplain Interactions in Wooded Catchments and Their Role in the Formation of Multi-Channel Patterns. *Geomorphology* 116: 305-319.
- Seitzinger, S.P., *et al.* 2002. "Bioavailability of DON from Natural and Anthropogenic Sources to Estuarine Plankton." *Limnology and Oceanography* 47: 353-366.

- Semlitsch, R.D. 2008. "Differentiating Migration and Dispersal Processes for Pond-Breeding Amphibians." *The Journal of Wildlife Management* 72:260-267.
- Semlitsch, R.D., and J.R. Bodie. 2003. "Biological Criteria for Buffer Zones around Wetlands and Riparian Habitats for Amphibians and Reptiles." *Conservation Biology* 17:1219-1228.
- Senar, O.E., et al. 2018. "Catchment-Scale Shifts in the Magnitude and Partitioning of Carbon Export in Response to Changing Hydrologic Connectivity in a Northern Hardwood Forest." Journal of Geophysical Research: Biogeosciences 123: 2337-2352.
- Serfass, T. L., *et al.* 1999. "Status and Distribution of River Otters in Pennsylvania following a Reintroduction Project." *Journal of the Pennsylvania Academy of Science* 73:10-14.
- Seymour, M., *et al.* 2015. "Dendritic network structure and dispersal affect temporal dynamics of diversity and species persistence." *Oikos* 124: 908-916.
- Shanks, C.E., and G.C. Arthur. 1952. "Muskrat Movements and Population Dynamics in Missouri Farm Ponds and Streams." *Journal of Wildlife Management* 16:138-148.
- Sharpe, P., *et al.* 2016. "Comparison of Alternative Approaches for Wetlands Mapping: A Case Study from three U.S. National Parks." *Wetlands* 36: 547-556.
- Sharpley, A.N., and S. Rekolainen. 1997. "Phosphorus in Agriculture and Its Environmental Implications." Pages 1-54 in *Phosphorus Losses from Soil to Water*. H. Tunney, *et al.*, ed. CAB International, Cambridge, UK.
- Sharpley, A.N., *et al.* 2007. "Managing Natural Processes in Drainage Ditches for Non-Point Source Phosphorus Control." *Journal of Soil and Water Conservation* 62(4): 197-206.
- Shavers, E., and L.V. Stanislawski. 2020. "Channel cross-section analysis for automated stream head identification." *Environmental Modelling & Software* 132: 104809. https://doi.org/https://doi.org/10.1016/j.envsoft.2020.104809.
- Shaw, D.A., *et al.* 2012. "The Fill–Spill Hydrology of Prairie Wetland Complexes during Drought and Deluge." *Hydrological Processes* 26: 3147-3156.
- Shaw, J.R., and D.J. Cooper. 2008. "Linkages among Watersheds, Stream Reaches, and Riparian Vegetation in Dryland Ephemeral Stream Networks." *Journal of Hydrology* 350: 68-82.
- Sheaves, M. 2009. "Consequences of ecological connectivity: the coastal ecosystem mosaic." *Marine Ecology Progress Series* 391: 107-115.
- Shedlock, R.J., *et al.* 1993. "Interactions between Ground-Water and Wetlands, Southern Shore of Lake-Michigan, USA," *Journal of Hydrology* 141:127-55.
- Sheldon, A. L. 1988. "Conservation of Stream Fishes: Patterns of Diversity, Rarity, and Risk." *Conservation Biology* 2: 149-156.

- Shogren, A.J., *et al.* 2019. "Revealing biogeochemical signatures of Arctic landscapes with river chemistry." *Scientific Reports* 9: 12894.
- Shook, K., *et al.* 2015. "The transformation of frequency distributions of winter precipitation to spring streamflow probabilities in cold regions; case studies from the Canadian Prairies." *Journal of Hydrology* 521: 395-409.
- Shook, K., *et al.* 2021. "Quantifying the effects of Prairie depressional storage complexes on drainage basin connectivity." *Journal of Hydrology* 593: 125846.
- Shook, K.R., and J.W. Pomeroy. 2011. "Memory Effects of Depressional Storage in Northern Prairie Hydrology." *Hydrological Processes* 25: 3890-3898.
- Shoup, D.E., and D. H. Wahl. 2009. "Fish diversity and abundance in relation to interannual and lakespecific variation in abiotic characteristics of floodplain lakes of the lower Kaskaskia River, Illinois." *Transactions of the American Fisheries Society* 138: 1076-1092.
- Simenstad, C., *et al.* 1999. "Preliminary Results from the Sacramento-San Joaquin Delta Breached Levee Wetland Study." *Interagency Ecological Program for the Sacramento-San Joaquin Estuary Newsletter* 12(4):15-21.
- Simley, J. 2006. "USGS National Hydrography Dataset Newsletter." Vol. 5, No. 4, February 2006. Available at <u>https://www.usgs.gov/national-hydrography/nhd-newsletter-archive</u>.
- Simley J. 2011. "USGS National Hydrography Dataset Newsletter." Vol. 10, No. 9, July 2011. Available at <u>https://www.usgs.gov/national-hydrography/nhd-newsletter-archive</u>.
- Simley, J. 2015. "USGS National Hydrography Dataset Newsletter." Vol. 14, No. 6, April 2015. Available at <u>https://www.usgs.gov/national-hydrography/nhd-newsletter-archive</u>.
- Simley, J. 2018. *GIS for Surface Water: Using the National Hydrography Dataset*. Redlands, CA: ESRI Press.
- Simmons, J.A. 2010. "Phosphorus Removal by Sediment in Streams Contaminated with Acid Mine Drainage." *Water Air and Soil Pollution* 209: 123-132.
- Sjodin, A., *et al.* 2001. "Analysis of Groundwater Exchange for a Large Plains River in Colorado (USA)." *Hydrological Processes* 15: 609-620.
- Smiley, P.C., Jr., *et al.* 2008. "Contribution of Habitat and Water Quality to the Integrity of Fish Communities in Agricultural Drainage Ditches." *Journal of Soil and Water Conservation* 63(6): 218A-219A.
- Smith, M.A., and D.M. Green. 2005. "Dispersal and the Metapopulation Paradigm in Amphibian Ecology and Conservation: Are All Amphibian Populations Metapopulations?" *Ecography* 28 (1): 110-128.
- Smith, R.D., and C.V. Klimas. 2002. A Regional Guidebook for Applying the Hydrogeomorphic Approach to Assessing Wetland Functions of Selected Regional Wetland Subclasses, Yazoo Basin,

Lower Mississippi River Alluvial Valley. ERDC/EL TR-02-4. U.S. Army Corps of Engineers, Vicksburg, MS. <u>https://wetlands.el.erdc.dren.mil/pdfs/trel02-4.pdf</u>.

- Smith, R.D., et al. 1995. An Approach for Assessing Wetland Functions Using Hydrogeomorphic Classification, Reference Wetlands, and Functional Indices. Wetlands Research Program Technical Report WRP-DE-9. U.S. Army Corps of Engineers, Waterways Experiment Station, Wetlands Research Program, Vicksburg, MS. https://wetlands.el.erdc.dren.mil/pdfs/wrpde9.pdf.
- Smith, S.M., *et al.* 2008. "Development of Vegetation in Dune Slack Wetlands of Cape Cod National Seashore (Massachusetts, USA)." *Plant Ecology* 194(2): 243-256. Smith, S.M., *et al.* 2008.
 "Development of Vegetation in Dune Slack Wetlands of Cape Cod National Seashore (Massachusetts, USA)." *Plant Ecology* 194(2): 243-256.
- Smith, S.V., *et al.* 2002. "Distribution and Significance of Small, Artificial Water Bodies Across the United States Landscape." *The Science of the Total Environment* 299:21-36.
- Smith, T.A., and C.E. Kraft. 2005. "Stream fish assemblages in relation to landscape position and local habitat variables." *Transactions of the American Fisheries Society* 134: 430-440.
- Smock, L.A. 1994. "Movements of invertebrates between stream channels and forested floodplains." *Journal of the North American Benthological Society* 13:524-531.
- Smock, L.A., *et al.* 1992. "Lotic Macroinvertebrate Production in Three Dimensions: Channel Surface, Hyporheic, and Floodplain Environments." *Ecology* 73:876-886.
- Snodgrass, J.W., et al. 1996. "Factors Affecting the Occurrence and Structure of Fish Assemblages in Isolated Wetlands of the Upper Coastal Plain, USA." Canadian Journal of Fisheries and Aquatic Sciences 53:443-454 (1996).
- Soons, M.B. 2006. "Wind Dispersal in Freshwater Wetlands: Knowledge for Conservation and Restoration." *Applied Vegetation Science* 9:271-278.
- Soons, M.B., and G.W. Heil. 2002. "Reduced Colonization Capacity in Fragmented Populations of Wind-Dispersed Grassland Forbs." *Journal of Ecology* 90:1033-1043.
- Sophocleous, M. 2010. "Review: Groundwater Management Practices, Challenges, and Innovations in the High Plains Aquifer, USA—Lessons and Recommended Actions. *Hydrogeology Journal* 18:559-575.
- Sparrow, K.H, et al. 2022. Evaluation of Climatic and Hydroclimatic Resources to Support the US Army Corps of Engineers Regulatory Program. ERDC/CHL TR-22-19. Department of the Army, Vicksburg, MS. https://erdc-library.erdc.dren.mil/jspui/handle/11681/45484.
- Spinola, R. M., *et. al.* 2008. "Survival and Post-Release Movements of River Otters Translocated to Western New York. *Northeastern Naturalist* 15:13-24.
- Sprecher, S.W., and A. G. Warne. 2000. Accessing and Using Meteorological Data to Evaluate Wetland Hydrology. ERDC/EL TR-WRAP-00-1. U.S. Army Engineer Research and Development Center, Vicksburg, MS. <u>http://hdl.handle.net/11681/4362</u>.

- Stanford, J.A., and J.V. Ward. 1984. "The Effects of Regulation on the Limnology of the Gunnison River: A North American Case History." Pages 467-480 in *Regulated Rivers*. Proceedings of the Second International Symposium on Regulated Streams, Oslo, August 1982. A. Lillehammer and S.J. Salveit, ed. Universitetsforlaget AS, Oslo, Norway.
- Stanford, J.A., and J.V. Ward. 1988. "The hyporheic habitat of river ecosystems." Nature 335: 64-66.
- Stanford, J.A., and J.V. Ward. 1993. "An ecosystem perspective of alluvial rivers: Connectivity and the hyporheic corridor." *Journal of the North American Benthological Society* 12:48-60.
- Stanislawski, L.V., *et al.* 2015. "A rapid approach for automated comparison of independently derived stream networks." *Cartography and Geographic Information Science* 42:5, 435-448.
- Stanislawski, L.V., et al. 2016. "High performance computing to support multiscale representation of hydrography for the conterminous United States." 19th ICA Workshop on Generalisation and Multiple Representation. June 14, 2016. Helsinki, Finland. http://pubs.er.usgs.gov/publication/70191228.
- Stanislawski, L.V., et al. 2017 "An Open Source Solution to High Performance Processing for Extracting Surface Water Drainage Networks from Diverse Terrain Conditions." Cartography and GIScience. <u>https://www.cartogis.org/docs/proceedings/2016/Stanislawski Survila Wendel Liu and Buttenfield.</u> <u>pdf</u>.
- Stanislawski, L.V., et al. 2018. "An open source high-performance solution to extract surface water drainage networks from diverse terrain conditions." Cartography and Geographic Information Science 45(4): 319-328. <u>https://doi.org/10.1080/15230406.2017.1337524</u>.
- Stanley, E.H., et al. 1997. "Ecosystem Expansion and Contraction in Streams." BioScience 47: 427-435.
- Stead, T.K., *et al.* 2005. "Secondary Production of a Stream Metazoan Community: Does the Meiofauna Make a Difference?" *Limnology and Oceanography* 50: 398-403.
- Steen, D.A., *et al.* 2012. "Terrestrial Habitat Requirements of Nesting Freshwater Turtles." *Biological Conservation* 150 (1): 121-128.
- Steinacker R. 2021. "How to correctly apply Gaussian statistics in a non-stationary climate?" *Theoretical* and Applied Climatology 144:1363–1374. <u>https://doi.org/10.1007/s00704-021-03601-4</u>.
- Stevenson, R.J. and F.R. Hauer. 2002. "Integrating Hydrogeomorphic and Index of Biotic Integrity Approaches for Environmental Assessment of Wetlands," *Journal of the North American Benthological Society* 21(3): 502-513.
- Steward, A.L., *et al.* 2012. "When the River Runs Dry: Human and Ecological Values of Dry Riverbeds." *Frontiers in Ecology and the Environment* 10:202-209.
- Stolt, M., and J. Baker. 1995. "Evaluation of national wetland inventory maps to inventory wetlands in the southern Blue Ridge of Virginia." *Wetlands* 15(4):346–353.

- Strahler, A. N. 1957. "Quantitative analysis of watershed geomorphology." *American Geophysical Union Transactions* 38: 913-920.
- Strock, J.S. *et al.* 2007. "Managing Natural Processes in Drainage Ditches for Nonpoint Source Nitrogen Control." *Journal of Soil and Water Conservation* 62(4): 188-196.
- Stromberg, J.C., *et al.* 2005. "Effects of Stream Flow Intermittency on Riparian Vegetation of a Semiarid Region River (San Pedro River, Arizona)." *River Research and Applications* 21:925-938.
- Subalusky, A.L. 2007. *The Role of Seasonal Wetlands in the Ecology of the American Alligator*. Texas A&M University, College Station, TX. <u>http://oaktrust.library.tamu.edu/bitstream/handle/1969.1/ETD-TAMU-1640/SUBALUSKY-THESIS.pdf?sequence=1</u>.
- Subalusky, A.L., *et al.* 2009a. "Ontogenetic Niche Shifts in the American Alligator Establish Functional Connectivity between Aquatic Systems," *Biological Conservation* 142:1507-1514.
- Subalusky, A.L. *et al.* 2009b. "Detection of American Alligators in Isolated, Seasonal Wetlands," *Applied Herpetology* 6:199-210.
- Sullivan, S.M.P., and M.C. Watzin. 2009. "Stream-Floodplain Connectivity and Fish Assemblage Diversity in the Champlain Valley, Vermont, U.S.A." *Journal of Fish Biology* 74:1394-1418.
- Sullivan, S.M.P., et al. 2019a. "Opinion: The proposed change to the definition of 'waters of the United States' flouts sound science." Proceedings of the National Academy of Sciences 116 (24): 11558-11561. https://doi.org/10.1073/pnas.1907489116.
- Sullivan, S.M.P., et al. 2019b. Re: Scientific Societies Comments on Proposed Rule Revised Definition of "Waters of the United States" (84 FR 4154; Docket ID No. EPA-HQ-OW-2018-0149). Docket ID No. EPA-HQ-OW-2018-0149-3825. <u>https://www.regulations.gov/comment/EPA-HQ-OW-2018-0149-3825</u>.
- Sullivan, S.M.P., *et al.* 2020. "Distorting science, putting water at risk." *Science* 369(6505): 766-768. <u>https://doi.org/10.1126/science.abb6899</u>.
- Sun, G.W., *et al.* 1995. "Shallow Groundwater Table Dynamics of Cypress Wetland Pine Upland Systems in Florida Flatwoods." *Soil and Crop Science Society of Florida Proceedings* 54:66-71.
- Suter, G. W., *et al.* 2002. "A methodology for inferring the causes of observed impairments in aquatic ecosystems." *Environmental Toxicology and Chemistry* 21: 1101-1111.
- Suter, G.W., and S.M. Cormier. 2011. "Why and how to combine evidence in environmental assessments: Weighing evidence and building cases." *Science of The Total Environment* 409: 1406-1417.
- Swimley, T.J., *et al.* 1999. "Otter and Beaver Interactions in the Delaware Water Gap National Recreation Area." *Journal of the Pennsylvania Academy of Science* 72:97-101.
- Tang, Y., *et al.* 2020. "Impact of Size and Location of Wetlands on Watershed-Scale Flood Control." *Water Resources Management* 34(5): 1693-1707.

- Tangen, B A., and S. Bansal. 2020. "Soil Organic Carbon Stocks and Sequestration Rates of Inland, Freshwater Wetlands: Sources of Variability and Uncertainty." *Science of The Total Environment* 749: 141444.
- Tank, J.L., *et al.* 2010. "A Review of Allochthonous Organic Matter Dynamics and Metabolism in Streams." *Journal of the North American Benthological Society* 29:118-146.
- Tarboton, D.G. 1997. "A new method for the determination of flow directions and upslope areas in grid digital elevation models." *Water Resources Research* 33: 309-319.
- Taylor, C.A., and H. Druckenmiller. 2022. "Wetlands, Flooding, and the Clean Water Act." *American Economic Review* 112(4): 1334-1363.
- Taylor, J.B., et al. 2022. "Comparing Landsat Dynamic Surface Water Extent to alternative methods of measuring inundation in developing waterbird habitats." *Remote Sensing Applications: Society and Environment* 28: 100845. <u>https://doi.org/10.1016/j.rsase.2022.100845</u>.
- Teachey, M.E., *et al.* 2019. "Rapid and Stable Microbial Community Assembly in the Headwaters of a Third-Order Stream." *Applied and Environmental Microbiology* 85: e00188-19.
- Tennessee Department of Environment and Conservation. 2020. *Guidance for Making Hydrologic Determinations*. Version 1.5. <u>https://www.tnhdt.org/PDF/HD%20Guidance.pdf</u>.
- Tetzlaff, D., and C. Soulsby. 2008. "Sources of Baseflow in Larger Catchments Using Tracers to Develop a Holistic Understanding of Runoff Generation." *Journal of Hydrology* 359:287-302.
- Tetzlaff, D., *et al.* 2014. "Storage dynamics in hydropedological units control hillslope connectivity, runoff generation, and the evolution of catchment transit time distributions." *Water Resources Research* 50(2): 969-985.
- Texas Commission on Environmental Quality. 2006. *Guidelines for Operation and Maintenance of Dams in Texas*. <u>https://www.tceq.texas.gov/assets/public/comm_exec/pubs/gi/gi357/gi-357.pdf</u>.
- Thomas, J.R., *et al.* 2006. "A Landscape Perspective of the Stream Corridor Invasion and Habitat Characteristics of an Exotic (*Dioscorea oppositifolia*) in a Pristine Watershed in Illinois." *Biological Invasions* 8:1103-1113.
- Thorp, J.H., *et al.* 2006. "The Riverine Ecosystem Synthesis: Biocomplexity in River Networks across Space and Time." *River Research and Applications* 22:123-147.
- Thorslund, J., *et al.* 2018. "Solute evidence for hydrological connectivity of geographically isolated wetlands." *Land Degradation & Development* 29: 3954-3962.
- Thurman, E.M. 1985. *Organic Geochemistry of Natural Waters*. Martinus Nijhoff/Dr. W. Junk Publishers, Boston, MA.
- Tihansky, A.B. 1999. "Sinkholes, West-Central Florida." Pages 121-140 in Land Subsidence in the United States. USGS Circular 1182. D. Galloway, et al., editors. U.S. Department of the Interior, U.S. Geological Survey, Reston, VA. <u>https://doi.org/10.3133/cir1182</u>.

Tiner R.W. 1997a. "NWI Maps: What They Tell Us." National Wetlands Newsletter 19(2):7-12.

Tiner RW. 1997b. "NWI Maps: Basic Information on the Nation's Wetlands." BioScience 47(5):269.

- Tiner, R.W. 2003a. *Dichotomous Keys and Mapping Codes for Wetland Landscape Position, Landform, Water Flow Path, and Waterbody Type Descriptors*. U.S Fish and Wildlife Service, National Wetlands Inventory Program, Northeast Region, Hadley, MA.
- Tiner, R.W. 2003b. "Estimated Extent of Geographically Isolated Wetlands in Selected Areas of the United States." *Wetlands* 23:636-652.
- Tiner, R.W. 2003c. "Geographically Isolated Wetlands of the United States." Wetlands 23(3):494-516.
- Tiner, R.W. 2011. Dichotomous Keys and Mapping Codes for Wetland Landscape Position, Landform, Water Flow Path, and Waterbody Type Descriptors: Version 2.0. U.S. Fish and Wildlife Service, National Wetlands Inventory Program, Northeast Region, Hadley, MA.
- Tockner, K., *et al.* 2000. "An Extension of the Flood Pulse Concept." *Hydrological Processes* 14:2861-2883.
- Tockner, K., *et al.* 2010. "Multiple Stressors in Coupled Riverfloodplain Ecosystems." *Freshwater Biology* 55 (Suppl. 1):135-151.
- Toth, L.A., and A. van der Valk. 2012. "Predictability of Flood Pulse Driven Assembly Rules for Restoration of a Floodplain Plan Community." *Wetlands Ecology and Management* 20:59-75.
- Trimble, S.W. 1999. "Decreased Rates of Alluvial Sediment Storage in the Coon Creek Basin, Wisconsin, 1975-93." *Science* 285:1244-1246.
- Tronstad, L.M., *et al.* 2007. "Aerial Colonization and Growth: Rapid Invertebrate Responses to Temporary Aquatic Habitats in a River Floodplain." *Journal of the North American Benthological Society* 26:460-471.
- Turner, R.E., *et al.* 2006. "Wetland Sedimentation from Hurricanes Katrina and Rita." *Science* 314(5798): 449-452.
- Turton, D.J., *et al.* 1992. "Subsurface Flow Responses of a Small Forested Catchment in the Ouachita Mountains." *Hydrological Processes* 6:111-125.
- U.S. Army Corps of Engineers. "HEC-RAS Features." U.S. Army Corps of Engineers Hydrologic Engineering Center. Accessed at <u>https://www.hec.usace.army.mil/software/hec-ras/features.aspx</u>. ("U.S. Army Corps of Engineers a.")
- U.S. Army Corps of Engineers. "Impoundments and Other Water Level Controls." Accessed at https://www.swg.usace.army.mil/Portals/26/docs/regulatory/Construction_Guidelines/impound.pdf. (U.S. Army Corps of Engineers b.")

- U.S. Army Corps of Engineers. "Regional Supplements to Corps Delineation Manual." Accessed at <u>https://www.usace.army.mil/Missions/Civil-Works/Regulatory-Program-and-Permits/reg_supp/</u>. (U.S. Army Corps of Engineers c.")
- U.S. Army Corps of Engineers. 1987. *Corps of Engineers Wetlands Delineation Manual*. Technical Report Y-87-1. U.S. Army Engineer Waterways Experiment Station, Vicksburg, MS. https://usace.contentdm.oclc.org/digital/collection/p266001coll1/id/4530.
- U.S. Army Corps of Engineers. 1992. Engineering and Design: Design, Construction, and Maintenance of Relief Wells. EM 1110-2-1914. Department of the Army, Washington, D.C. <u>http://www.publications.usace.army.mil/Portals/76/Publications/EngineerManuals/EM_1110-2-1914.pdf</u>.
- U.S. Army Corps of Engineers. 1993. Engineering and Design: Seepage Analysis and Control for Dams. Engineer Manual 1110-2-1901. Original 1986 – Revised 1993. Department of the Army, Washington, D.C. <u>http://www.publications.usace.army.mil/Portals/76/Publications/EngineerManuals/EM_1110-2-1901.pdf</u>.
- U.S. Army Corps of Engineers. 1995. *Construction Control for Earth and Rock-filled Dams*. EM 1110-2-1911. Department of the Army, Washington, D.C. https://usace.contentdm.oclc.org/digital/collection/p16021coll9/id/1933/.
- U.S. Army Corps of Engineers. 2000. *Engineering and Design: Design and Construction of Levees*. EM 1110-2-1913. Department of the Army, Washington, D.C. <u>https://www.publications.usace.army.mil/Portals/76/Publications/EngineerManuals/EM_1110-2-1913.pdf</u>.
- U.S. Army Corps of Engineers. 2004. Engineering and Design: General Design and Construction Considerations for Earth and Rock-Fill Dams. EM 1110-2-2300. Department of the Army, Washington, D.C. <u>http://www.publications.usace.army.mil/Portals/76/Publications/EngineerManuals/EM_1110-2-2300.pdf.</u>
- U.S. Army Corps of Engineers. 2005a. *Engineering and Design: Design Guidance for Levee Underseepage*. ETL 1110-2-569. Department of the Army, Washington, D.C. http://www.usace.army.mil/usace-docs/eng-tech-ltrs/etl1110-2-569/entire.pdf.
- U.S. Army Corps of Engineers. 2005b. *Regulatory Guidance Letter, Subject: Ordinary High Water Mark.* RGL 05-05. Department of the Army, Washington, D.C. <u>https://usace.contentdm.oclc.org/utils/getfile/collection/p16021coll9/id/1253</u>.
- U.S. Army Corps of Engineers. 2007a. *Jurisdictional Determination Form Instructional Guidebook*. Department of the Army, Washington, D.C. https://usace.contentdm.oclc.org/digital/collection/p16021coll11/id/2321/.
- U.S. Army Corps of Engineers. 2007b. Regulatory Guidance Letter, Subject: Practices for Documenting Jurisdiction under Section 404 of the Clean Water Act (CWA) and Sections 9 & 10 of the Rivers and

Harbors Act (RHA) of 1899. RGL 07-01. Department of the Army, Washington, D.C. https://cdm16021.contentdm.oclc.org/digital/collection/p16021coll9/id/1279/rec/1.

- U.S. Army Corps of Engineers. 2007c. *Regional Supplement to the Corps of Engineers Wetland Delineation Manual: Alaska Region*. Version 2.0. ERDC/EL TR-07-24. Department of the Army, Vicksburg, MS. <u>https://usace.contentdm.oclc.org/utils/getfile/collection/p266001coll1/id/7608</u>.
- U.S. Army Corps of Engineers. 2010. Regional Supplement to the Corps of Engineers Wetland Delineation Manual: Western Mountains, Valleys, and Coast Region. Version 2.0. ERDC/EL TR-10-3. Department of the Army, Vicksburg, MS. https://usace.contentdm.oclc.org/utils/getfile/collection/p266001coll1/id/7646.
- U.S. Army Corps of Engineers. 2012. Regional Supplement to the Corps of Engineers Wetland Delineation Manual: Northcentral and Northeast Region. Version 2.0. ERDC/EL TR-12-1.
 Department of the Army, Vicksburg, MS. https://usace.contentdm.oclc.org/utils/getfile/collection/p266001coll1/id/7640.
- U.S. Army Corps of Engineers. 2020a. "National Inventory of Dams." https://nid.usace.army.mil/#/.
- U.S. Army Corps of Engineers. 2020b. "Ordinary High Water Mark (OHWM) Research, Development, and Training: Improving Corps Regulatory Practices." U.S. Army Corps of Engineers, Engineering Research and Development Center. Accessed at <u>https://www.erdc.usace.army.mil/Media/Fact-Sheets/Fact-Sheet-Article-View/Article/486085/ordinary-high-water-mark-ohwm-research-development-and-training/</u>.
- U.S. Bureau of Reclamation. "Safety of Dams." Provo Area Office. Last updated May 11, 2022. Accessed at <u>http://www.usbr.gov/uc/provo/progact/damsafety.html</u>.
- U.S. Department of Agriculture, Agricultural Research Service. *Multiflume Runoff Event August 1, 1990*, Accessed at <u>https://www.tucson.ars.ag.gov/unit/WGWebcam/WalnutGulchWebcam.htm</u>.
- U.S. Department of Agriculture Forest Service. 2017. *Forest Inventory and Analysis: Fiscal Year 2016 Business Report.* FS-1075. Accessed at <u>https://www.fs.usda.gov/sites/default/files/fs_media/fs_document/publication-15817-usda-forest-</u> <u>service-fia-annual-report-508.pdf.</u> ("USFS 2017.")
- U.S. Department of Agriculture Forest Service. 2020. "Tree Canopy Cover Mapping Products." GTAC-10192-BRO1. Geospatial Technology and Applications Center, Salt Lake City, Utah. Accessed at <u>https://www.mrlc.gov/sites/default/files/TCC_Project_Overview_Brochure-MRLC_2020-06-05.pdf</u>. ("USFS 2020.")
- U.S. Environmental Protection Agency. 1996. *Why Watersheds?* EPA 800-F-96-001. Accessed at <u>https://nepis.epa.gov/Exe/ZyPURL.cgi?Dockey=20001P3C.txt</u>.
- U.S. Environmental Protection Agency. 2002. A Lexicon Of Cave And Karst Terminology With Special Reference To Environmental Karst Hydrology (2002 Edition). EPA/600/R-02/003. U.S. Environmental Protection Agency, Office of Research and Development, National Center for

Environmental Assessment, Washington, D.C. Accessed at <u>https://cfpub.epa.gov/ncea/risk/era/recordisplay.cfm?deid=54964</u>.

- U.S. Environmental Protection Agency. 2003. Setting and Allocating the Chesapeake Bay Basin Nutrient and Sediment Loads: The Collaborative Process, Technical Tools and Innovative Approaches. EPA 903-R-03-007. Region III, Chesapeake Bay Program Office, Washington, D.C. Accessed at https://d38c6ppuviqmfp.cloudfront.net/content/publications/cbp_19713.pdf.
- U.S. Environmental Protection Agency. 2006. Wetlands: Protecting Life and Property from Flooding. EPA843-F-06-001. Accessed at <u>https://www.epa.gov/sites/default/files/2016-02/documents/flooding.pdf</u>.
- U.S. Environmental Protection Agency. 2008. *Handbook for Developing Watershed Plans to Restore and Protect Our Waters: Planning & Implementation Steps*. EPA 841-B-08-002. U.S. EPA, Washington D.C. <u>https://www.epa.gov/nps/handbook-developing-watershed-plans-restore-and-protect-our-waters</u>.
- U.S. Environmental Protection Agency. 2010. FY 2011-2015 Strategic Plan: Achieving Our Vision. https://nepis.epa.gov/Exe/ZyPURL.cgi?Dockey=P1008YOS.txt.
- U.S. Environmental Protection Agency. 2012. "Section 6 Echo Park Lake TMDLs." Pages 6-1 to 6-66 in Los Angeles Area Lakes Total Maximum Daily Loads for Nitrogen, Phosphorus, Mercury, Trash, Organochlorine Pesticides and PCBs. March 2012. <u>https://attains.epa.gov/attainspublic/api/documents/actions/CA_SWRCB/42214/100132</u>.
- U.S. Environmental Protection Agency. 2015. Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence (Final Report). EPA/600/R-14/475F. U.S. Environmental Protection Agency, Washington, D.C. <u>https://cfpub.epa.gov/ncea/risk/recordisplay.cfm?deid=296414</u>. ("Science Report.")
- U.S. Environmental Protection Agency. 2017. *National Water Quality Inventory: Report to Congress*. EPA 841-R-16-011. <u>https://www.epa.gov/sites/default/files/2017-</u> 12/documents/305brtc_finalowow_08302017.pdf.
- U.S. Environmental Protection Agency. 2022a. Basic Information: NHDPlus (National Hydrography Dataset Plus). Accessed at <u>https://www.epa.gov/waterdata/basic-information</u>. ("U.S. EPA 2022a.")
- U.S. Environmental Protection Agency. 2022b. Climate Adaptation and Wetland Protection. Accessed at <u>https://www.epa.gov/arc-x/climate-adaptation-and-wetland-protection</u>. ("U.S. EPA 2022b.")
- U.S. Environmental Protection Agency. 2022c. Climate Change Indicators: Coastal Flooding. Accessed at https://www.epa.gov/climate-indicators/climate-change-indicators-coastal-flooding. ("U.S. EPA 2022c.")
- U.S. Environmental Protection Agency. 2022d. Climate Change Indicators: Drought. Accessed at https://www.epa.gov/climate-indicators/climate-change-indicators-drought. ("U.S. EPA 2022d.")

- U.S. Environmental Protection Agency. 2022e. Climate Change Indicators: Heavy Precipitation. Accessed at <u>https://www.epa.gov/climate-indicators/climate-change-indicators-heavy-precipitation</u>. ("U.S. EPA 2022e.")
- U.S. Environmental Protection Agency. 2022f. Climate Change Indicators: River Flooding. Accessed at https://www.epa.gov/climate-indicators/climate-change-indicators-river-flooding. ("U.S. EPA 2022f.")
- U.S. Environmental Protection Agency. 2022g. Climate Change Indicators: Snowpack. Accessed at https://www.epa.gov/climate-indicators/climate-change-indicators-snowpack. ("U.S. EPA 2022g.")
- U.S. Environmental Protection Agency. 2022h. Climate Change Indicators: Streamflow. Accessed at https://www.epa.gov/climate-indicators/climate-change-indicators-streamflow. ("U.S. EPA 2022h.")
- U.S. Environmental Protection Agency. 2022i. Climate Change Indicators: U.S. and Global Precipitation. Accessed at <u>https://www.epa.gov/climate-indicators/climate-change-indicators-us-and-global-precipitation</u>. ("U.S. EPA 2022i.")
- U.S. Environmental Protection Agency. 2022j. "Ecoregions." Accessed at <u>https://www.epa.gov/eco-research/ecoregions</u>. ("U.S. EPA 2022j.")
- U.S. Environment Protection Agency. 2022k. "Ecoregions of North America." Accessed at <u>https://www.epa.gov/eco-research/ecoregions-north-america</u>. ("U.S. EPA 2022k.")
- U.S. Environmental Protection Agency. 2022l. HAWQS (Hydrologic and Water Quality System). Accessed at <u>https://www.epa.gov/waterdata/hawqs-hydrologic-and-water-quality-system</u>. ("U.S. EPA 20211.")
- U.S. Environmental Protection Agency. 2022m. Streamflow Duration Assessment Method (SDAM) Status by Region. Accessed at <u>https://www.epa.gov/streamflow-duration-assessment/streamflow-duration-assessment-method-sdam-status-region</u>. ("U.S. EPA 2022m.")
- U.S. Environmental Protection Agency. 2022n. "WATERS GeoViewer Tutorial." Accessed at https://www.epa.gov/waterdata/waters-geoviewer-tutorial. ("U.S. EPA 2022n.")
- U.S. Environmental Protection Agency Science Advisory Board. 2014a. Memorandum from Dr. Amanda Rodewald to Dr. David Allen. Comments to the chartered SAB on the Adequacy of the Scientific and Technical Basis of the EPA's Proposed Rule titled "Definition of 'Waters of the United States' under the Clean Water Act." Accessed at https://sab.epa.gov/ords/sab/f?p=100:18:7003824515153:::RP.18:P18 ID:2352. ("SAB 2014a.")
- U.S. Environmental Protection Agency Science Advisory Board. 2014b. *SAB Consideration of the Adequacy of the Scientific and Technical Basis of the EPA's Proposed Rule titled "Definition of Waters of the United States under the Clean Water Act.*" EPA-SAB-14-007, U.S. Environmental Protection Agency, Washington, D.C. Accessed at https://sab.epa.gov/ords/sab/f?p=100:18:7003824515153:::RP.18:P18 ID:2352, ("SAB 2014b.")

U.S. Environmental Protection Agency Science Advisory Board. 2014c. SAB review of the draft EPA report Connectivity of Streams and Wetlands to Downstream Waters: A Review and Synthesis of the Scientific Evidence (October 17, 2014). EPA-SAB-15-001, U.S. Environmental Protection Agency, Washington, D.C. Accessed at

https://sab.epa.gov/ords/sab/f?p=100:18:15494586833200:::RP,18:P18 ID:2251. ("SAB 2014c.")

- U.S. Environmental Protection Agency Science Advisory Board. 2020. "Commentary on the Proposed Rule Defining the Scope of Waters Federally Regulated Under the Clean Water Act" (February 27, 2020). EPA-SAB-20-0002. U.S. Environmental Protection Agency, Washington, D.C. <u>https://yosemite.epa.gov/sab/sabproduct.nsf/LookupWebReportsARsLastMonthBOARD/729C61F75</u> <u>763B8878525851F00632D1C/\$File/EPA-SAB-20-002+.pdf</u>. ("SAB 2020.")
- U.S. Environmental Protection Agency Science Advisory Board. 2022. Memorandum from Dr. Allison Cullen to Dr. David Allen. "Recommendations of the SAB Work Group for Review of Science Supporting EPA Decisions Regarding Two Planned EPA Regulatory Actions" (February 7, 2022). EPA-SAB-20-0002. U.S. Environmental Protection Agency, Washington, D.C. *Available at* https://sab.epa.gov/ords/sab/f?p=114:19:32360986054023:::19:P19_ID:967. ("SAB 2022a").
- U.S. Environmental Protection Agency Science Advisory Board. 2022. Science Advisory Board (SAB) Consideration of the Scientific and Technical Basis of the EPA and Department of the Army's Proposed Rule titled "Revised Definition of Waters of the United States" (July 5, 2022). EPA-SAB-22-005. U.S. Environmental Protection Agency, Washington, D.C. Available at https://sab.epa.gov/ords/sab/f?p=114:18:32360986054023:::RP,18:P18_ID:2620 ("SAB 2022b.")
- U.S. Environmental Protection Agency, and U.S. Army Corps of Engineers. 2007. "Memorandum to Assert Jurisdiction for SPL-2007-261-FBV." December 6, 2007. *Available at* https://usace.contentdm.oclc.org/utils/getfile/collection/p16021coll5/id/1433.
- U.S. Environmental Protection Agency, and U.S. Army Corps of Engineers. 2008. "Memorandum to Reevaluate Jurisdiction for NWP-2007-428." February 26, 2008. *Available at* <u>https://usace.contentdm.oclc.org/utils/getfile/collection/p16021coll5/id/1435</u>.
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2008. "Clean Water Act Jurisdiction Following the U.S. Supreme Court's Decision in *Rapanos v. United States & Carabell v. United States*." Washington, D.C. <u>https://www.epa.gov/sites/default/files/2016-</u> 02/documents/cwa_jurisdiction_following_rapanos120208.pdf. ("*Rapanos* Guidance.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2015a. "Clean Water Rule: Definition of 'Waters of the United States.'" Docket ID No. EPA-HQ-OW-2011-0880-20862. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2011-0880-20862</u>. ("2015 Clean Water Rule.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2015b. Economic Analysis of the EPA-Army Clean Water Rule. 80 FR 37054 (June 29, 2015). Docket ID No. EPA-HQ-OW-2011-0880-20866. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2011-0880-20866</u>. ("2015 Clean Water Rule EA.")

- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2020c. "The Navigable Waters Protection Rule: Definition of 'Waters of the United States." 85 FR 22250 (April 21, 2020). Docket ID No. EPA-HQ-OW-2018-0149-11571. Washington, D.C. https://www.regulations.gov/document/EPA-HQ-OW-2018-0149-11571. ("2020 NWPR.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2020a. Economic Analysis for the Navigable Waters Protection Rule: Definition of "Waters of the United States." Docket ID No. EPA-HQ-OW-2018-0149-11572. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2018-0149-11572</u>. ("2020 NWPR EA.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2020b. "Limitations of the National Hydrography Dataset at High Resolution and the National Wetlands Inventory and their use for Determining the Scope of Waters Subject to Clean Water Act Jurisdiction." Docket ID No. EPA-HQ-OW-2018-0149-11585. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2018-0149-11585</u>. ("White Paper.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2020c. "The Navigable Waters Protection Rule: Definition of "Waters of the United States." 85 FR 22250 (April 21, 2020). Docket ID No. EPA-HQ-OW-2018-0149-11571. Washington, D.C. https://www.regulations.gov/document/EPA-HQ-OW-2018-0149-11571. ("2020 NWPR.")
- U.S. Environmental Protection Agency, and U.S. Department of the Army. 2020d. Resource and Programmatic Assessment for the Navigable Waters Protection Rule: Definition of "Waters of the United States" Final Rule. Docket ID No. EPA-HQ-OW-2018-0149-11573. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2018-0149-11573</u>. ("2020 NWPR RPA.")
- U.S. Environmental Protection Agency and U.S. Department of the Army. 2021. Economic Analysis for the Proposed "Revised Definition of 'Waters of the United States'" Rule. Docket ID No. EPA-HQ-OW-2021-0602-0083. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2021-0602-0083</u>. ("Economic Analysis for the Proposed Rule").
- U.S. Environmental Protection Agency and U.S. Department of the Army. 2021. *Technical Support* Document for the Proposed "Revised Definition of 'Waters of the United States'" Rule. Docket ID No. EPA-HQ-OW-2021-0602-0081. Washington, D.C. <u>https://www.regulations.gov/document/EPA-HQ-OW-2021-0602-0081</u>. ("Technical Support Document for the Proposed Rule").
- U.S. Environmental Protection Agency and U.S. Department of the Army. 2022. *Economic Analysis for the Final "Revised Definition of 'Waters of the United States'" Rule*. Washington, D.C. ("Economic Analysis for the Final Rule").
- U.S. Fish and Wildlife Service. "NWI Overview and Use Applications." Accessed at https://www.fws.gov/wetlands/documents/NWI-Overview-And-Use-Applications.pdf. ("U.S. FWS a.")
- U.S. Fish and Wildlife Service. "USFWS Threatened & Endangered Species Active Critical Habitat Report." Accessed at <u>https://ecos.fws.gov/ecp/report/table/critical-habitat.html</u>. ("U.S. FWS b.")

- U.S. Fish and Wildlife Service. 2010. "NWIPlus: Geospatial Database for Watershed-level Functional Assessment." <u>https://www.fws.gov/northeast/wetlands/factsheets/nwiplus_factsheet.pdf</u>. Accessed via Wayback Machine at https://www.fws.gov/northeast/wetlands/factsheets/nwiplus_factsheet.pdf. Accessed via Wayback Machine at <a href="https://www.fws.gov/northeast/wetlands/factsheets/Nwww.fws.gov/northeast/wetlan
- U.S. Fish and Wildlife Service. 2020. "National Wetlands Inventory Program Overview." U.S. Fish and Wildlife Service Ecological Services, Falls Church, VA. Accessed at https://www.fws.gov/sites/default/files/documents/national-wetlands-inventory-fact-sheet.pdf.
- U.S. Fish and Wildlife Service. 2021. "Wetlands Mapper." National Wetlands Inventory. Accessed at <u>https://www.fws.gov/program/national-wetlands-inventory/wetlands-mapper</u>.
- U.S. Geological Survey. "The 3D National Topography Model Call for Action Part 1: The 3D Hydrography Program." Accessed at <u>https://www.usgs.gov/national-hydrography/3d-national-topography-model-call-action-part-1-3d-hydrography-program</u>. ("USGS a.")
- U.S. Geological Survey. "Earth Explorer." Accessed at <u>https://www.usgs.gov/educational-</u> resources/earth-explorer. ("USGS b.")
- U.S. Geological Survey. "Floods: Recurrence intervals and 100-year floods." New Jersey Water Science Center. Accessed at <u>https://www.usgs.gov/centers/new-jersey-water-science-center/floods-</u> <u>recurrence-intervals-and-100-year-floods?qt-science_center_objects=0#qt-science_center_objects.</u> ("USGS c.")
- U.S. Geological Survey. "Value Added Attributes." Accessed at <u>https://www.usgs.gov/national-hydrography/value-added-attributes-vaas</u>. ("USGS d.")
- U.S. Geological Survey. "Watershed Boundary Dataset." Accessed at <u>https://www.usgs.gov/national-hydrography/watershed-boundary-dataset?qt-science_support_page_related_con=4#qt-science_support_page_related_con.</u> ("USGS e.")
- U.S. Geological Survey. "What is Lidar data and where can I download it?" *Available at* <u>https://www.usgs.gov/faqs/what-lidar-data-and-where-can-i-download-it</u>. ("USGS f.")
- U.S. Geological Survey. 1992. "Map Scales." 2 pp. Accessed at https://pubs.usgs.gov/unnumbered/70039582/report.pdf. ("USGS 1992.")
- U.S. Geological Survey. 2005. "Topographic Map Symbols." 4 pp. U.S. Geological Survey, Reston, VA. Accessed at <u>https://pubs.usgs.gov/gip/TopographicMapSymbols/topomapsymbols.pdf</u>. ("USGS 2005.")
- U.S. Geological Survey. 2011. *Post-fire Flash Flood in Coronado National Memorial, Arizona* (August 25, 2011). <u>https://www.youtube.com/watch?v=qJ8JxBZt6Ws</u>. ("USGS 2011.")
- U.S. Geological Survey. 2012. "EarthExplorer." U.S. Geological Survey General Information Product 136. Accessed at <u>https://pubs.usgs.gov/gip/136/pdf/gip136.pdf</u>. ("USGS 2012.")

- U.S. Geological Survey. 2014. "Frequently Asked Questions about the NHD & WBD Datasets." <u>https://nhd.usgs.gov/Frequently+Asked+Questions+about+the+NHD+&+WBD.htm</u>. Accessed via Wayback Machine at <u>https://web.archive.org/web/20171111232126/https://nhd.usgs.gov/Frequently+Asked+Questions+abo</u> <u>ut+the+NHD+&+WBD.htm</u>. ("USGS 2014.")
- U.S. Geological Survey. 2019. "StreamStats: Streamflow Statistics and Spatial Analysis Tools for Water-Resources Applications." Accessed at <u>https://www.usgs.gov/mission-areas/water-</u> <u>resources/science/streamstats-streamflow-statistics-and-spatial-analysis-tools?qt-</u> <u>science_center_objects=0#qt-science_center_objects.</u> ("USGS 2019.")
- U.S. Geological Survey. 2021. "NHDPlus High Resolution." Accessed at <u>https://www.usgs.gov/national-hydrography/nhdplus-high-resolution</u>. ("USGS 2021.")
- U.S. Global Change Research Program. 2017. *Climate Science Special Report: Fourth National Climate Assessment, Volume I.* [Wuebbles, D.J., D.W. Fahey, K.A. Hibbard, D.J. Dokken, B.C. Stewart, and T.K. Maycock (eds.)]. U.S. Global Change Research Program, Washington, D.C. 470 pp. Accessed at https://science2017.globalchange.gov/downloads/CSSR2017_FullReport.pdf. ("USGCRP 2017.")
- U.S. Global Change Research Program. 2018. Impacts, Risks, and Adaptation in the United States: Fourth National Climate Assessment, Volume II. [Reidmiller, D.R., C.W. Avery, D.R. Easterling, K.E. Kunkel, K.L.M. Lewis, T.K. Maycock, and B.C. Stewart (eds.)]. U.S. Global Change Research Program, Washington, D.C. 1515 pp. doi: 10.7930/NCA4.2018. Accessed at https://nca2018.globalchange.gov/downloads/NCA4_2018_FullReport.pdf. ("USGCRP 2018.")
- Uden, D.R., *et al.* 2014. "The role of reserves and anthropogenic habitats for functional connectivity and resilience of ephemeral wetlands." *Ecological Applications* 24: 1569-1582.
- University Corporation for Atmospheric Research (UCAR). 2022. "The Water Cycle and Climate Change." Accessed at <u>https://scied.ucar.edu/learning-zone/climate-change-impacts/water-cycle-climate-change</u>.
- Uno, H., and M.E. Power. 2015. "Mainstem-tributary linkages by mayfly migration help sustain salmonids in a warming river network." *Ecology Letters* 18(10): 1012-20. <u>https://doi.org/10.1111/ele.12483</u>.
- Urban, N.R., *et al.* 1989. "Export of Dissolved Organic Carbon and Acidity from Peatlands." *Water Resources Research* 25:1619-1628.
- Valett, H.M., *et. al.* 2005. "Biogeochemical and Metabolic Responses to the Flood Pulse in a Semiarid Floodplain." *Ecology* 86(1): 220-234.
- Van Haveren, B.P. 1986. "Management of instream flows through runoff detention and retention." *Water Resources Bulletin* 22: 399-404.
- van Meerveld, H.J., *et al.* 2019. "Expansion and contraction of the flowing stream network alter hillslope flowpath lengths and the shape of the travel time distribution." *Hydrology and Earth System Sciences* 23: 4825-4834.

- Van Meter, K.J., and N.B. Basu. 2015. "Signatures of Human Impact: Size Distributions and Spatial Organization of Wetlands in the Prairie Pothole Landscape." *Ecological Applications*: 25:451–465.
- Van Sickle, J., and C.B. Johnson. 2008. "Parametric Distance Weighting of Landscape Influence on Streams." *Landscape Ecology* 23:427-438.
- Vanderhoof, M.K., and L.C. Alexander. 2016. "The Role of Lake Expansion in Altering the Wetland Landscape of the Prairie Pothole Region, United States." *Wetlands* 36(2): 309-321.
- Vanderhoof, M.K., and C.R. Lane. 2019. "The potential role of very high-resolution imagery to characterise lake, wetland and stream systems across the Prairie Pothole Region, United States." *International Journal of Remote Sensing* 40 (15): 5768-5798. https://doi.org/10.1080/01431161.2019.1582112.
- Vanderhoof, M.K., et al. 2016. "Patterns and drivers for wetland connections in the Prairie Pothole Region, United States." Wetlands Ecology and Management 25: 1-23. <u>https://doi.org/10.1007/s11273-016-9516-9</u>.
- Vanderhoof, M.K., *et al.* 2017. "The influence of data characteristics on detecting wetland/stream surface-water connections in the Delmarva Peninsula, Maryland and Delaware." *Wetlands Ecology and Management* 26: 63–86.
- Vannote, R.L., et al. 1980. "The River Continuum Concept." Canadian Journal of Fisheries and Aquatic Science: 130-137.
- Vanschoenwinkel, B., *et al.* 2009. "Wind Mediated Dispersal of Freshwater Invertebrates in a Rock Pool Metacommunity: Differences in Dispersal Capacities and Modes." *Hydrobiologia* 635:363-372.
- Ver Hoef, J.M., and E.E. Peterson. 2010. "A Moving Average Approach for Spatial Statistical Models of Stream Networks." *Journal of the American Statistical Association* 105:6-18.
- Ver Hoef, J.M., *et al.* 2006. "Spatial Statistical Models that Use Flow and Stream Distance." *Environmental and Ecological Statistics* 13:449-464.
- Vidon, P., *et al.* 2010. "Hot Spots and Hot Moments in Riparian Zones: Potential for Improved Water Quality Management." *Journal of the American Water Resources Association* 46:278-298.
- Villines, J.A., et al. 2015. "Using GIS to Delineate Headwater Stream Origins in the Appalachian Coalfields of Kentucky." Journal of the American Water Resources Association 51(6): 1667-1687. <u>https://doi.org/10.1111/1752-1688.12350</u>.
- Vining, K.C. 2002. "Simulation of Streamflow and Wetland Storage, Starkweather Coulee Subbasin, North Dakota, Water Years 1981-98." *Water-Resources Investigations Report* 02-4113. U.S. Geological Survey, Bismarck, North Dakota. <u>https://doi.org/10.3133/wri024113</u>.
- Virginia Department of Conservation and Recreation. 2010. *Determinations of Water Bodies with Perennial Flow: Guidance on the Chesapeake Bay Preservation Area Designation and Management Regulations*. DCR-CBLAB-019. Available at:

https://townhall.virginia.gov/l/GetFile.cfm?File=C:%5CTownHall%5Cdocroot%5CGuidanceDocs%5 C440%5CGDoc_DEQ_5406_v1.pdf.

- Vivoni, E.R., *et al.* 2006. "Analysis of a Monsoon Flood Event in an Ephemeral Tributary and Its Downstream Hydrologic Effects." *Water Resources Research* 42: W03404.
- Vogl, A.L. and V.L. Lopes. 2009. "Impact of water resources development on flow regimes in the Brazos River." *Environmental Monitoring and Assessment* 157: 331–345. <u>https://doi.org/10.1007/s10661-008-0538-5</u>.
- Volkmar, E.C., and R.A. Dahlgren. 2006. Biological Oxygen Demand Dynamics in the Lower San Joaquin River, California." *Environmental Science & Technology* 40:5653-5660.
- Voos, G., and P.M. Groffman. 1996. "Relationships between Microbial Biomass and Dissipation of 2,4-D and Dicamba in Soil." *Biology and Fertility of Soils* 24:106-110.
- Wagener, T., *et al.* 2007. "Catchment Classification and Hydrologic Similarity." *Geography Compass* 1:901-931.
- Wakeley, J.S. 2002. Developing a 'Regionalized' Version of the Corps of Engineers Wetlands Delineation Manual: Issues and Recommendations. ERDC/EL TR-02-20. U.S. Army Engineer Research and Development Center, Vicksburg, MS. <u>https://hdl.handle.net/11681/7005</u>.
- Wallace, J.B., and J.R. Webster. 1996. "The Role of Macroinvertebrates in Stream Ecosystem Function." *Annual Review of Entomology* 41:115-139.
- Wallace, J.B., *et al.* 1997. "Multiple Trophic Levels of a Forest Stream Linked to Terrestrial Litter Inputs," *Science* 277:102-104.
- Walton, R., *et al.* 1996. "Hydrology of the Black Swamp Wetlands on the Cache River, Arkansas." *Wetlands* 16:279-287.
- Wang, N., et al. 2019. "New Model for Simulating Hydrologic Processes under Influence of Surface Depressions." *Journal of Hydrologic Engineering* 24(5): 04019008.
- Wang, X., *et al.* 2007. "Water Quality Changes as a Result of Coalbed Methane Development in a Rocky Mountain Watershed." *Journal of the American Water Resources Association* 43:1383-1399.
- Wang, X., *et al.* 2008. "Using hydrologic equivalent wetland concept within SWAT to estimate streamflow in watersheds with numerous wetlands." *Transactions of the ASABE* 51 (1): 55e72.
- Ward, A.S., *et al.* 2018. "Simulation of dynamic expansion, contraction, and connectivity in a mountain stream network." *Advances in Water Resources* 114: 64-82.
- Ward, J.V. 1989. "The Four-Dimensional Nature of Lotic Ecosystems." *Journal of the North American Benthological Society* 8:2-8.
- Ward, J.V. 1998. "Riverine Landscapes: Biodiversity Patterns, Disturbance Regimes, and Aquatic Conservation." *Biological Conservation* 83:269-278.

- Ward, J.V., and J.A. Stanford. 1983. "The Serial Discontinuity Concept of Lotic Ecosystems." Pages 29-42 in *Dynamics of Lotic Ecosystems*. T.D. Fontaine and S.M. Bartell, editors. Ann Arbor Science, Ann Arbor, MI.
- Ward, J.V., *et al.* 2002. "Applicability of Ecological Theory to Riverine Ecosystems." *Verhandlungen der Internationale Vereinigung für Theoretische und Angewandte Limnologie* 28:443-450.

Watershed Resources Registry. 2022. "Welcome." Accessed at https://watershedresourcesregistry.org/.

- Watson, K.B., et al. 2016. "Quantifying Flood Mitigation Services: The Economic Value of Otter Creek Wetlands and Floodplains to Middlebury, VT." *Ecological Economics* 130: 16-24. <u>https://doi.org/10.1016/j.ecolecon.2016.05.015</u>.
- Watts, A.C., *et al.* 2015. "Hydrologic Implications of Smoldering Fires in Wetland Landscapes." *Freshwater Science* 34: 1394–405.
- Weary, D.J., and D.H. Doctor. 2014. Karst in the United States: A digital map compilation and database. U.S. Geological Survey Open-File Report 2014–1156. 23 p. Accessed at <u>https://dx.doi.org/10.3133/ofr20141156</u>.
- Webb, J.R., *et al.* 2017. "Constraining the annual groundwater contribution to the water balance of an agricultural floodplain using radon: The importance of floods." *Water Resources Research* 53: 544-562.
- Webster, J.R., and B.C. Patten. 1979. "Effects of Watershed Perturbation on Stream Potassium and Calcium Dynamics." *Ecological Monographs* 49:51-72.
- Wegener, P., *et al.* 2017. "Beaver-mediated lateral hydrologic connectivity, fluvial carbon and nutrient flux, and aquatic ecosystem metabolism." *Water Resources Research* 53(6): 4606-4623.
- Wegener, P.A., et al. 2018. "Evaluating Controls on Nutrient Retention and Export in Wide and Narrow Valley Segments of a Mountain River Corridor." *Journal of Geophysical Research: Biogeosciences* 123: 1817-1826.
- Wendt, A., *et al.* 2021. "Metapopulation genetics of endangered reticulated flatwoods salamanders (*Ambystoma bishopi*) in a dynamic and fragmented landscape." *Conservation Genetics* 22: 551–567.
- Westbrook, C. J., *et al.* 2006. "Beaver dams and overbank floods influence groundwater-surface water interactions of a Rocky Mountain riparian area." *Water Resources Research* 42: W06404.
- Wetzel, R.G. 1992. "Gradient-Dominated Ecosystems: Sources and Regulatory Functions of Dissolved Organic Matter in Freshwater Ecosystems. *Hydrobiologia* 229:181-198.
- Wetzel, R.G., and B.A. Manny. 1972. "Decomposition of Dissolved Organic Carbon and Nitrogen Compounds from Leaves in an Experimental Hard-Water Stream." *Limnology and Oceanography* 17:927-931.

Wharton, Charles H., et al. 1982. The Ecology of Bottomland Hardwood Swamps of the Southeast: A Community Profile. FWS/OBS-81/37. U.S. Fish and Wildlife Service Washington, D.C. https://ecos.fws.gov/ServCat/DownloadFile/105358?Reference=65824.

Whigham, D.F., and T.E. Jordan. 2003. "Isolated Wetlands and Water Quality." Wetlands 23:541-549.

- Whigham, D.F., *et al.* 1988. "Impacts of Fresh-Water Wetlands on Water-Quality: A Landscape Perspective. *Environmental Management* 12:663-671.
- Whiles, M.R., and W.K. Dodds. 2002. "Relationships between Stream Size, Suspended Particles, and Filter-Feeding Macroinvertebrates in a Great Plains Drainage Network." *Journal of Environmental Quality* 31:1589-1600.
- White, W.R., and T.L. Crisman. 2016. "Headwater streams of Florida: Types, distribution, and a framework for conservation." *River Research and Applications* 32: 452-461.
- Whiting, P.J., and M. Pomeranets. 1997. "A Numerical Study of Bank Storage and Its Contribution to Streamflow." *Journal of Hydrology* 202:121-136.
- Whitmire, S.L., and S.K. Hamilton. 2008. "Rates of Anaerobic Microbial Metabolism in Wetlands of Divergent Hydrology on a Glacial Landscape." *Wetlands* 28:703-714.
- Wiens, J.A. 2002. "Riverine Landscapes: Taking Landscape Ecology into the Water." *Freshwater Biology* 47: 501-515.
- Wigington, P.J., *et al.* 2003. "Nitrate Removal Effectiveness of a Riparian Buffer along a Small Agricultural Stream in Western Oregon." *Journal of Environmental Quality* 32: 162-170.
- Wigington, P.J., et al. 2005. "Stream Network Expansion: A Riparian Water Quality Factor." *Hydrological Processes* 19: 1715-1721.
- Wigington, P.J., et al. 2006. "Coho Salmon Dependence on Intermittent Streams." Frontiers in Ecology and the Environment 4: 513-518.
- Wigington, P.J., et al. 2013. "Oregon Hydrologic Landscapes: A Classification Framework." Journal of the American Water Resources Association 49.1: 163-182.
- Wilcox, B.P., et al. 2011. "Evidence of surface connectivity for Texas Gulf Coast depressional wetlands." Wetlands 31: 451-458.
- Williams, C., and D. Tufford. 2015. "Groundwater Recharge Rates in Isolated and Riverine Wetlands: Influencing Factors." *Journal of South Carolina Water Resources* 2(1): 86-92. <u>https://doi.org/10.34068/JSCWR.02.10</u>.
- Williams, D.D. 1996. "Environmental Constraints in Temporary Fresh Waters and Their Consequences for the Insect Fauna." *Journal of the North American Benthological Society* 15:634-650.

- Williams, G.P., and M.G. Wolman. 1984. Downstream Effects of Dams on Alluvial Rivers. USGS Professional Paper 1286, U.S. Department of the Interior, U.S. Geological Survey, Washington, DC. <u>https://pubs.usgs.gov/pp/1286/report.pdf</u>.
- Williamson, T.N., *et al.* 2015. "Classification of Ephemeral, Intermittent, and Perennial Stream Reaches Using a TOPMODEL-Based Approach." *Journal of the American Water Resources Association* 51 (6): 1739-1759. <u>https://doi.org/10.1111/1752-1688.12352</u>.
- Willson, M.F., and K.C. Halupka. 1995. "Anadromous Fish as Keystone Species in Vertebrate Communities." *Conservation Biology* 9(3):489-497.
- Wilks, D.S. 2013: "Projecting 'normals' in a nonstationary climate." *Journal of Applied Meteorology and Climatology* 52:289–302.
- Winemiller, K.O., *et al.* "Fish Assemblage Structure in Relation to Environmental Variation among Brazos River Oxbow Lakes." Transactions of the American Fisheries Society 129:451-468.
- Winter, T.C., 2001. "The concept of hydrologic landscapes." *Journal of the American Water Resources Association* 37: 335-349.
- Winter, T.C. 2007. "The Role of Groundwater in Generating Streamflow in Headwater areas and in Maintaining Base Flow." *Journal of the American Water Resources Association* 43:15-25.
- Winter, T.C., and J.W. LaBaugh. 2003. "Hydrologic Considerations in Defining Isolated Wetlands." *Wetlands* 23: 532-540.
- Winter, T.C., and D.O. Rosenberry. 1998. "Hydrology of Prairie Pothole Wetlands during Drought and Deluge: A 17-Year Study of the Cottonwood Lake Wetland Complex in North Dakota in the Perspective of Longer Term Measured and Proxy Hydrological Records." *Climatic Change* 40:189-209.
- Winter, T.C., et al. 1998. Ground Water and Surface Water: A Single Resource. USGS Circular 1139, U.S. Department of the Interior, U.S. Geological Survey, Denver, CO. <u>https://doi.org/10.3133/cir1139</u>.
- Wipfli, M.S., and D.P. Gregovich. 2002. "Export of Invertebrates and Detritus from Fishless Headwater Streams in Southeastern Alaska: Implications for Downstream Salmonid Production." *Freshwater Biology* 47:957-969.
- Wiskow, E., and R.R. van der Ploeg. 2003. "Calculation of drain Spacings for Optimal Rainstorm Flood Control." *Journal of Hydrology* 272:163-174.
- Woessner, W.W. 2000. "Stream and Fluvial Plain Ground Water Interactions: Rescaling Hydrogeologic Thought." *Ground Water* 38:423-429.
- Wohl, E., and N.D. Beckman. 2014. "Leaky rivers: Implications of the loss of longitudinal fluvial disconnectivity in headwater streams." *Geomorphology* 205: 27-35.

- Wohl, E., et al. 2016. Synthesizing the scientific foundation for ordinary high water mark delineation in fluvial systems. ERDC/CRREL SR-16-5. U.S. Army Corps of Engineers, Engineer Research and Development Center, Washington, D.C. 217 pp. <u>https://hdl.handle.net/11681/20650</u>.
- Wolock, D. M. 2003. Hydrologic landscape regions of the United States. Open File Report No. 2003-145. U.S. Geological Service, Reston, Virginia. <u>https://doi.org/10.3133/ofr03145</u>.
- Wolock, D.M., et al. 2004. "Delineation and Evaluation of Hydrologic-Landscape Regions in the United States Using Geographic Information System Tools and Multivariate Statistical Analysis." *Environmental Management* 34: S71-S88.
- Wollheim, W.M., *et al.* 2008. "Dynamics of N removal over annual time periods in a suburban river network." *Journal of Geophysical Research: Biogeosciences* 113: G03038.
- Wolman, M.G., and J.P. Miller. 1960. "Magnitude and Frequency of Forces in Geomorphic Processes." *Journal of Geology* 68: 54-74.
- Wood, P.J., and P.D. Armitage.1997. "Biological Effects of Fine Sediment in the Lotic Environment." *Environmental Management* 21: 203-217.
- Woodford, D. J., and A. R. McIntosh. 2010. "Evidence of source-sink metapopulations in a vulnerable native galaxiid fish driven by introduced trout." *Ecological Applications* 20: 967-977.
- Woodward, G.U.Y., and A.G. Hildrew. 2002. "Food Web Structure in Riverine Landscapes." *Freshwater Biology* 47: 777-798.
- World Meteorological Organization. 2007. *The Role of Climatological Normals in a Changing Climate*. WCDMP-No. 61, WMO-TD/No. 1377. <u>https://library.wmo.int/doc_num.php?explnum_id=4546</u>. ("WMO 2007.")
- World Meteorological Organization. 2017. WMO Guidelines on the Calculation of Climate Normals. 2017 edition. WMO – No. 1203. World Meterological Organization, Geneva, Switzerland. Available at: <u>https://library.wmo.int/doc_num.php?explnum_id=4166</u>. ("WMO 2017.")
- Wu, K., and A. Johnston, C. 2008. "Hydrologic Comparison Between a Forested and a Wetland/Lake Dominated Watershed Using SWAT." *Hydrological Processes* 22(10): 1431-1442. <u>https://doi.org/10.1002/hyp.6695</u>.
- Wu, M., et al. 2014a. "Differential Assessment of Designations of Wetland Status Using Two Delineation Methods." Environmental Management. 54(1):22-29.
- Wu, Q., and C.R. Lane. 2016. "Delineation and Quantification of Wetland Depressions in the Prairie Pothole Region of North Dakota." *Wetlands* 36 (2): 215-227. <u>http://dx.doi.org/10.1007/s13157-015-0731-6</u>.
- Wu, Q., and C.R. Lane. 2017. "Delineating wetland catchments and modeling hydrologic connectivity using lidar data and aerial imagery." *Hydrol. Earth Syst. Sci.* 21 (7): 3579-3595. <u>https://doi.org/10.5194/hess-21-3579-2017</u>.
- Wu, Q., *et al.* 2014b. "An Effective Method for Detecting Potential Woodland Vernal Pools Using High-Resolution LiDAR Data and Aerial Imagery." *Remote Sensing* 6 (11): 11444-11467.
- Wu, Q., et al. 2015. "A localized contour tree method for deriving geometric and topological properties of complex surface depressions based on high-resolution topographical data." *International Journal of Geographical Information Science*: 1-20. <u>http://dx.doi.org/10.1080/13658816.2015.1038719</u>.
- Wu, T., et al. 2019. "High-efficient extraction of drainage networks from digital elevation models constrained by enhanced flow enforcement from known river maps." *Geomorphology* 340: 184-201. <u>https://doi.org/10.1016/j.geomorph.2019.04.022</u>.
- Xu, J. 2016. "Sediment jamming of a trunk stream by hyperconcentrated floods from small tributaries: case of the Upper Yellow River, China." *Hydrological Sciences Journal* 61: 1926-1940.
- Yamazaki, D., *et al.* 2015. "Development of a global ~90m water body map using multi-temporal Landsat images." *Remote Sensing of Environment* 171: 337-351. <u>https://doi.org/10.1016/j.rse.2015.10.014</u>.
- Yang, W., *et al.* 2010. "Simulated environmental effects of wetland restoration scenarios in a typical Canadian prairie watershed." *Wetlands Ecology and Management* 18(3): 269e279.
- Yeo, I.Y., et al. 2019. "Mapping landscape-level hydrological connectivity of headwater wetlands to downstream waters: A geospatial modeling approach - Part 1." Science of the Total Environment 653: 1546-1556.
- Yuan, F., and S. Miyamoto. 2008. "Characteristics of Oxygen-18 and Deuterium Composition in Waters from the Pecos River in American Southwest." *Chemical Geology* 255:220-230.
- Zaimes, G.N., *et al.* 2004. "Stream Bank Erosion Adjacent to Riparian Forest Buffers, Row-Crop Fields, and Continuously-Grazed Pastures along Bear Creek in Central Iowa." *Journal of Soil and Water Conservation* 59:19-27.
- Zale, A.V., *et al.* 1989. "The Physicochemistry, Flora, and Fauna of Intermittent Prairie Streams: A Review of the Literature." *United States Fish and Wildlife Service Biological Report* 89:1-44. Accessed at <u>https://apps.dtic.mil/sti/pdfs/ADA322854.pdf</u>.
- Zimmer, K.D., *et al.* 2001. "Effects of Fathead Minnow Colonization and Removal on a Prairie Wetland Ecosystem." *Ecosystems* 4:346-357.
- Zimmer, M.A., and B.L. McGlynn. 2017. "Bidirectional stream–groundwater flow in response to ephemeral and intermittent streamflow and groundwater seasonality." *Hydrological Processes* 31: 3871-3880. https://doi.org/10.1002/hyp.11301.

Appendix C: References from the Literature Update Screening and Public Comments on Literature Published Since 2014

As discussed in section I.C of the Technical Support Document, subject-matter experts from the U.S. Environmental Protection Agency's Office of Research and Development conducted a screening analysis to identify papers that were relevant to the major conclusions of the Science Report. Appendix C1 contains those references that the screeners believed were relevant to the conclusions of aquatic systems: (1) ephemeral, intermittent, and perennial streams; (2) floodplain wetlands and open waters; and (3) nonfloodplain wetlands and open waters and is broken out by aquatic system. Appendix C2 contains plaintext summaries of the abstracts for a sample of the relevant literature (note that Appendix C2 was numbered Appendix C3 in the Technical Support Document for the Proposed Rule, but no other changes have been made to this Appendix). Appendix C3 contains 37 additional references identified by the screeners as being relevant to the Science Report's major conclusions and the agencies' review of those 37 references (note that Appendix C3 was numbered Appendix C2 in the Technical Support Document for the Proposed Rule). The agencies solicited comment on the scientific literature contained in Appendix C of the Proposed Rule and on whether additional scientific literature and references published since 2014 are relevant to the Science Report's conclusions on the connectivity and effects of streams, wetlands, and open waters on the chemical, physical, and biological integrity of downstream water. Several commenters provided additional literature published since 2014, and the agencies' review of those citations is contained in Appendix C3.

Appendix C1: References Relevant to the Conclusions of the Science Report Published Since 2014

The agencies believe the below references are relevant to conclusions of the Science Report and have been published since 2014. The agencies are seeking comment on this list of references and if additional references are relevant to the report's conclusions but are not listed below. This list of references is derived mainly from the agencies' screening process, which is discussed in section I.C of this document. That screening process yielded 2,022 unique citations that were found to be relevant to the conclusions of the Report. The results of that screening process have been supplemented with additional literature published since 2014 that the agencies believe is relevant to the findings of the Report but that was not captured through the screening process. For purposes of the screening, the agencies conducted screenings for references that were relevant to the Science Reports conclusions on (1) streams; (2) riparian and floodplain wetlands and open waters, and (3) riparian and non-floodplain wetlands and open waters. Note that some references are relevant to more than one system but may not be denoted as such in the below appendix.

Ephemeral, Intermittent, or Perennial Streams

Abbott, B.W., G. Gruau, J.P. Zarnetske, F. Moatar, L. Barbe, Z. Thomas, O. Fovet, T. Kolbe, S. Gu, A.-C. Pierson-Wickmann, P. Davy, and G. Pinay. 2018. "Unexpected spatial stability of water chemistry in headwater stream networks." *Ecology Letters* 21 (2): 296-308. https://www.ncbi.nlm.nih.gov/pubmed/29282860.

Acre, M.R., T.B. Grabowski, D.J. Leavitt, N.G. Smith, A.A. Pease, and J.E. Pease. 2021. "Blue sucker

habitat use in a regulated Texas river: implications for conservation and restoration." *Environmental Biology of Fishes* 104(4): 501-516.

- Addy, K., A.J. Gold, M.K. Welsh, P.V. August, M.H. Stolt, C.P. Arango, and P.M. Groffman. 2019.
 "Connectivity and Nitrate Uptake Potential of Intermittent Streams in the Northeast USA." *Frontiers in Ecology and Evolution* 7: 225.
- Al-Chokhachy, R., S. Moran, P.A. McHugh, S. Bernall, W. Fredenberg, and J.M. Dossantos. 2015.
 "Consequences of Actively Managing a Small Bull Trout Population in a Fragmented Landscape." *Transactions of the American Fisheries Society* 144(3): 515-531.
- Alexander, L.C., K.M. Fritz, K.A. Schofield, B.C. Autrey, J.E. DeMeester, H.E. Golden, D.C. Goodrich, W.G. Kepner, H.R. Kiperwas, C.R. Lane, S.D. LeDuc, S.G. Leibowitz, M.G. McManus, A.I. Pollard, C.E. Ridley, M.K. Vanderhoof, and P.J. Wigington. 2018. "Featured Collection Introduction: Connectivity of Streams and Wetlands to Downstream Waters." *JAWRA Journal of the American Water Resources Association* 54(2): 287-297.
- Alexander, R.B., G.E. Schwarz, and E.W. Boyer. 2019. "Advances in Quantifying Streamflow Variability Across Continental Scales: 1. Identifying Natural and Anthropogenic Controlling Factors in the USA Using a Spatially Explicit Modeling Method." *Water Resources Research* 55(12): 10893-10917.
- Almeida, R.S., and M. Cetra. 2016. "Longitudinal gradient effects on the stream fish metacommunity." *Natureza & Conservacao* 14(2): 112-119.
- Altenritter, M.E., S.M. Pescitelli, A.L. Whitten, and A.F. Casper. 2019. "Implications of an invasive fish barrier for the long-term recovery of native fish assemblages in a previously degraded northeastern Illinois River system." *River Research and Applications* 35(7): 1044-1052.
- Altermatt, F., and E.A. Fronhofer. 2018. "Dispersal in dendritic networks: Ecological consequences on the spatial distribution of population densities." *Freshwater Biology* 63(1): 22-32.
- Anderson, J.H., P.L. Faulds, K.D. Burton, M.E. Koehler, W.I. Atlas, and T.P. Quinn. 2015. "Dispersal and productivity of Chinook (*Oncorhynchus tshawytscha*) and coho (*Oncorhynchus kisutch*) salmon colonizing newly accessible habitat." *Canadian Journal of Fisheries and Aquatic Sciences* 72(3): 454-465.
- Anderson, K.E., and S.M. Hayes. 2018. "The effects of dispersal and river spatial structure on asynchrony in consumer-resource metacommunities." *Freshwater Biology* 63(1): 100-113.
- Anderson, R.L., C.A. Anderson, J.H. Larson, B. Knights, J. Vallazza, S.E. Jenkins, and J.T. Lamer. 2020.
 "Influence of a high-head dam as a dispersal barrier to fish community structure of the Upper Mississippi River." *River Research and Applications* 36(1): 47-56.
- Andrade, I.C.P., T.K. Krolow, R. Boldrini, and F.M. Pelicice. 2020. "Diversity of EPT (Ephemeroptera, Plecoptera, Trichoptera) Along Streams Fragmented by Waterfalls in the Brazilian Savanna." *Neotropical Entomology* 49(2): 203-212.

Antonelli, M., B. Glaser, A.J. Teuling, J. Klaus, and L. Pfister. 2020. "Saturated areas through the lens: 2.

Spatio-temporal variability of streamflow generation and its relationship with surface saturation." *Hydrological Processes* 34(6): 1333-1349.

- Antonelli, M., J. Klaus, K. Smettem, A.J. Teuling, and L. Pfister. 2017. "Exploring Streamwater Mixing Dynamics via Handheld Thermal Infrared Imagery." *Water* 9(5).
- Arce, M.I., M. del Mar Sánchez-Montoya, M.R. Vidal-Abarca, M. Luisa Suárez, and R. Gómez. 2014.
 "Implications of flow intermittency on sediment nitrogen availability and processing rates in a Mediterranean headwater stream." *Aquatic Sciences* 76(2): 173-186.
- Archdeacon, T.P., S.R. Davenport, J.D. Grant, and E.B. Henry. 2018. "Mass upstream dispersal of pelagic-broadcast spawning cyprinids in the Rio Grande and Pecos River, New Mexico." Western North American Naturalist 78(1): 100-105.
- Arciniega, M., A.J. Clemento, M.R. Miller, M. Peterson, J.C. Garza, and D.E. Pearse. 2016. "Parallel evolution of the summer steelhead ecotype in multiple populations from Oregon and Northern California." *Conservation Genetics* 17(1): 165-175.
- Ardren, W.R., and S.R. Bernall. 2017. "Dams impact westslope cutthroat trout metapopulation structure and hybridization dynamics." *Conservation Genetics* 18(2): 297-312.
- Arheimer, B., and G. Lindstrom. 2019. "Detecting Changes in River Flow Caused by Wildfires, Storms, Urbanization, Regulation, and Climate Across Sweden." *Water Resources Research* 55(11): 8990-9005.
- Arkle, R.S., and D.S. Pilliod. 2015. "Persistence at distributional edges: Columbia spotted frog habitat in the arid Great Basin, USA." *Ecology and Evolution* 5(17): 3704-3724.
- Armstrong, J.B., A.H. Fullerton, C.E. Jordan, J.L. Ebersole, J.R. Bellmore, I. Arismendi, B.E. Penaluna, and G.H. Reeves. 2021. "The importance of warm habitat to the growth regime of cold-water fishes." *Nature Climate Change* 11(4): 354-361.
- Arroita, M., I. Aristi, J. Díez, M. Martinez, G. Oyarzun, and A. Elosegi. 2015. "Impact of water abstraction on storage and breakdown of coarse organic matter in mountain streams." *Science of the Total Environment*: 233-240.
- Atkinson, C.L., D.C. Allen, L. Davis, and Z.L. Nickerson. 2018. "Incorporating ecogeomorphic feedbacks to better understand resiliency in streams: A review and directions forward." *Geomorphology* 305: 123-140.
- Atkinson, S., M. Bruen, J.J. O' Sullivan, J.N. Turner, B. Ball, J. Carlsson, C. Bullock, C.M. Casserly, and M. Kelly-Quinn. 2020. "An inspection-based assessment of obstacles to salmon, trout, eel and lamprey migration and river channel connectivity in Ireland." *Science of the Total Environment* 719.
- Atkore, V., N. Kelkar, and J. Krishnaswamy. 2017. "Assessing the recovery of fish assemblages downstream of hydrological barriers in India's Western Ghats." *River Research and Applications* 33(7): 1026-1035.

- Aubeneau, A.F., J.D. Drummond, R. Schumer, D. Bolster, J.L. Tank, and A.I. Packman. 2015. "Effects of benthic and hyporheic reactive transport on breakthrough curves." *Freshwater Science* 34(1): 301-315.
- Aubert, A.H., M.C. Thrun, L. Breuer, and A. Ultsch. 2016. "Knowledge discovery from high-frequency stream nitrate concentrations: hydrology and biology contributions." *Scientific Reports* 6.
- Audet, J., D. Bastviken, M. Bundschuh, I. Buffam, A. Feckler, L. Klemedtsson, H. Laudon, S. Löfgren, S. Natchimuthu, M. Öquist, M. Peacock, and M.B. Wallin. 2020. "Forest streams are important sources for nitrous oxide emissions." *Global Change Biology* 26(2): 629-641.
- Audet, J., M.B. Wallin, K. Kyllmar, S. Andersson, and K. Bishop. 2017. "Nitrous oxide emissions from streams in a Swedish agricultural catchment." *Agriculture, Ecosystems and Environment* 236: 295-303.
- Augusto Medeiros, P.H., J.C. de Araujo, G.L. Mamede, B. Creutzfeldt, A. Guentner, and A. Bronstert. 2014. "Connectivity of sediment transport in a semiarid environment: a synthesis for the Upper Jaguaribe Basin, Brazil." *Journal of Soils and Sediments* 14(12): 1938-1948.
- Ayalew, T.B., and W.F. Krajewski. 2017. "Effect of River Network Geometry on Flood Frequency: A Tale of Two Watersheds in Iowa." *Journal of Hydrologic Engineering* 22(8).
- Ayles, C.P., and M. Church. 2015. "Downstream channel gradation in the regulated Peace River." *Regulation of Peace River: A Case Study for River Management*: 39-66.
- Azouaoui, O., and A.A. Assani. 2018. "The case of extreme hydrologic drought downstream from reservoirs in Quebec (Canada): The intermittent flow." *River Research and Applications* 34(2): 135-143.
- Baalousha, M., S. Stoll, M. Motelica-Heino, N. Guigues, G. Braibant, F. Huneau, and P. Le Coustumer. 2018. "Suspended particulate matter determines physical speciation of Fe, Mn, and trace metals in surface waters of Loire watershed." *Environmental Science and Pollution Research* 26(6): 5251-5266.
- Bae, M.-J., and Y.-S. Park. 2019. "Evaluation of precipitation impacts on benthic macroinvertebrate communities at three different stream types." *Ecological Indicators* 102: 446-456. https://doi.org/10.1016/j.ecolind.2019.02.060.
- Balamurugan, M., and S.M. Bhallamudi. 2016. "Flood routing in an ephemeral channel with compound cross-section." *Sadhana* 41(7): 771-785.
- Balestrini, R., C.A. Delconte, M.T. Palumbo, and A. Buffagni. 2018. "Biotic control of in-stream nutrient retention in nitrogen-rich springs (Po Valley, Northern Italy)." *Ecological Engineering* 122: 303-314.
- Barcala, V., J. Rozemeijer, L. Oste, B. van Der Grift, L. Gerner, and T. Behrends. 2021. "Processes controlling the flux of legacy phosphorus to surface waters at the farm scale." *Environmental Research Letters* 16(1).

- Barefoot, E., T.M. Pavelsky, G.H. Allen, M.A. Zimmer, and B.L. McGlynn. 2019. "Temporally Variable Stream Width and Surface Area Distributions in a Headwater Catchment." *Water Resources Research* 55 (8): 7166-7181.
- Barkwith, A., M.D. Hurst, C.R. Jackson, L. Wang, M.A. Ellis, and T.J. Coulthard. 2015. "Simulating the influences of groundwater on regional geomorphology using a distributed, dynamic, landscape evolution modelling platform." *Environmental Modelling and Software* 74: 1-20.
- Barnett, S.E., and D.A. Woolnough. 2021. "Variation in Assemblages of Freshwater Mussels Downstream of Dams and Dam Removals in the Lake Michigan Basin, Michigan, USA." *Diversity-Basel* 13 (3).
- Bartakova, V., M. Reichard, R. Blazek, M. Polacik, and J. Bryja. 2015. "Terrestrial fishes: rivers are barriers to gene flow in annual fishes from the African savanna." *Journal of Biogeography* 42(10): 1832-1844.
- Bateman, D.S., M.R. Sloat, R.E. Gresswell, A.M. Berger, D.P. Hockman-Wert, D.W. Leer, and A.E. Skaugset. 2016. "Effects of stream-adjacent logging in fishless headwaters on downstream coastal cutthroat trout." *Canadian Journal of Fisheries and Aquatic Sciences* 73(12): 1898-1913.
- Bean, J.R., A.C. Wilcox, W.W. Woessner, and C.C. Muhlfeld. 2015. "Multiscale hydrogeomorphic influences on bull trout (*Salvelinus confluentus*) spawning habitat." *Canadian Journal of Fisheries and Aquatic Sciences* 72(4): 514-526.
- Beatty, S., M. Allen, A. Lymbery, M.S. Jordaan, D. Morgan, D. Impson, S. Marr, B. Ebner, and O.L.F. Weyl. 2017. "Rethinking refuges: Implications of climate change for dam busting." *Biological Conservation* 209: 188-195.
- Beatty, S.J., D.L. Morgan, and A.J. Lymbery. 2014. "Implications of climate change for potamodromous fishes." *Global Change Biology* 20(6): 1794-1807.
- Beaulieu, J.J., H.E. Golden, C.D. Knightes, P.M. Mayer, S.S. Kaushal, M.J. Pennino, C.P. Arango, D.A. Balz, C.M. Elonen, K.M. Fritz, and B.H. Hill. 2015. "Urban Stream Burial Increases Watershed-Scale Nitrate Export." *PLoS ONE* 10(7).
- Beaulieu, J.J., P.M. Mayer, S.S. Kaushal, M.J. Pennino, C.P. Arango, D.A. Balz, T.J. Canfield, C.M. Elonen, K.M. Fritz, B.H. Hill, H. Ryu, and J.W.S. Domingo. 2014. "Effects of urban stream burial on organic matter dynamics and reach scale nitrate retention." *Biogeochemistry* 121(1): 107-126.
- Beck, W.J., P.L. Moore, K.E. Schilling, C.F. Wolter, T.M. Isenhart, K.J. Cole, and M.D. Tomer. 2019."Changes in lateral floodplain connectivity accompanying stream channel evolution: Implications for sediment and nutrient budgets." *Science of the Total Environment* 660: 1015-1028.
- Beer, S.D., M.L. Bartron, D.G. Argent, and W.G. Kimmel. 2019. "Genetic Assessment Reveals Population Fragmentation and Inbreeding in Populations of Brook Trout in the Laurel Hill of Pennsylvania." *Transactions of the American Fisheries Society* 148(3): 620-635.
- Bejar, M., D. Vericat, R.J. Batalla, and C.N. Gibbins. 2018. "Variation in flow and suspended sediment

transport in a montane river affected by hydropeaking and instream mining." *Geomorphology* 310: 69-83.

- Bendik, N.F., K.D. McEntire, and B.N. Sissel. 2016. "Movement, demographics, and occupancy dynamics of a federally-threatened salamander: evaluating the adequacy of critical habitat." *PeerJ* 4.
- Benettin, P., S.W. Bailey, J.L. Campbell, M.B. Green, A. Rinaldo, G.E. Likens, K.J. McGuire, and G. Botter. 2015. "Linking water age and solute dynamics in streamflow at the Hubbard Brook Experimental Forest, NH, USA." *Water Resources Research* 51(11): 9256-9272.
- Benettin, P., S.W. Bailey, A. Rinaldo, G.E. Likens, K.J. McGuire, and G. Botter. 2017. "Young runoff fractions control streamwater age and solute concentration dynamics." *Hydrological Processes* 31(16): 2982-2986.
- Benone, N.L., R. Ligeiro, L. Juen, and L.F.A. Montag. 2018. "Role of environmental and spatial processes structuring fish assemblages in streams of the eastern Amazon." *Marine and Freshwater Research* 69(2): 243-252.
- Benson, L.J., and R.G. Pearson. 2020. "Dynamics of organic material and invertebrates in a tropical headwater stream." *Hydrobiologia* 847(1): 121-136.
- Bentley, K.T., D.E. Schindler, J.B. Armstrong, T.J. Cline, and G.T. Brooks. 2015. "Inter-Tributary Movements by Resident Salmonids across a Boreal Riverscape." *PLoS ONE* 10(9).
- Bergman, N., O. Sholker, J. Roskin, and N. Greenbaum. 2014. "The Nahal Oz Reservoir dam-break flood: Geomorphic impact on a small ephemeral loess-channel in the semi-arid Negev Desert, Israel." *Geomorphology* 210: 83-97.
- Bergstrom, A., B. McGlynn, J. Mallard, and T. Covino. 2016. "Watershed structural influences on the distributions of stream network water and solute travel times under baseflow conditions." *Hydrological Processes* 30(15): 2671-2685.
- Bernal, S., A. Lupon, W.M. Wollheim, F. Sabater, S. Poblador, and E. Marti. 2019. "Supply, Demand, and In-Stream Retention of Dissolved Organic Carbon and Nitrate During Storms in Mediterranean Forested Headwater Streams." *Frontiers in Environmental Science* 7.
- Bertuzzo, E., A.M. Helton, and T.J. Battin. 2017. "Scaling of dissolved organic carbon removal in river networks." *Advances in Water Resources* 110: 136-146.
- Betts, E.D., and D.L. Kane. 2015. "Linking North Slope of Alaska climate, hydrology, and fish migration." *Nordic Hydrology* 46(4): 578-590.
- Beveridge, C., E. Istanbulluoglu, C. Bandaragoda, and A.M. Pfeiffer. 2020. "A Channel Network Model for Sediment Dynamics Over Watershed Management Time Scales." *Journal of Advances in Modeling Earth Systems* 12(6).
- Bhattacharya, R., and C.L. Osburn. 2021. "Chromophoric dissolved organic matter composition and load from a coastal river system under variable flow regimes." *Science of the Total Environment* 760.

- Bianchi, T.S., D. Butman, P.A. Raymond, N.D. Ward, R.J.S. Kates, K.W. Flessa, H. Zamora, A.R. Arellano, J. Ramirez, and E. Rodriguez. 2017. "The experimental flow to the Colorado River delta: Effects on carbon mobilization in a dry watercourse." *Journal of Geophysical Research: Biogeosciences* 122(3): 607-627. <u>https://doi.org/10.1002/2016JG003555</u>.
- Bigelow, P., L. Benda, and S. Pearce. 2016. "Delineating incised stream sediment sources within a San Francisco Bay tributary basin." *Earth Surface Dynamics* 4 (3): 531-547.
- Birkel, C., A. Correa-Barahona, M. Martinez-Martinez, S. Granados-Bolaños, N. Venegas-Cordero, K. Gutiérrez-García, S. Blanco-Ramírez, R. Quesada-Mora, V. Solano-Rivera, J. Mussio-Mora, A. Chavarría-Palma, K. Vargas-Arias, G.W. Moore, A.M. Durán-Quesada, J. Vasquez-Morera, C. Soulsby, D. Tetzlaff, E. Espinoza-Cisneros, and R. Sánchez-Murillo. 2020. "Headwaters drive streamflow and lowland tracer export in a large-scale humid tropical catchment." *Hydrological Processes* 34(18): 3824-3841. <u>https://doi.org/10.1002/hyp.13841</u>.
- Birkel, C., C. Soulsby, G. Ali, and D. Tetzlaff. 2014. "ASSESSING THE CUMULATIVE IMPACTS OF HYDROPOWER REGULATION ON THE FLOW CHARACTERISTICS OF A LARGE ATLANTIC SALMON RIVER SYSTEM." *River Research and Applications* 30(4): 456-475.
- Bishop-Taylor, R., M.G. Tulbure, and M. Broich. 2015. "Surface water network structure, landscape resistance to movement and flooding vital for maintaining ecological connectivity across Australia's largest river basin." *Landscape Ecology* 30(10): 2045-2065.
- Bishop-Taylor, R., M.G. Tulbure, and M. Broich. 2017. "Surface-water dynamics and land use influence landscape connectivity across a major dryland region." *Ecological Applications* 27(4): 1124-1137.
- Bishop-Taylor, R., M.G. Tulbure, and M. Broich. 2018. "Impact of hydroclimatic variability on regionalscale landscape connectivity across a dynamic dryland region." *Ecological Indicators* 94: 142-150.
- Bladon, K.D., C. Segura, N.A. Cook, S. Bywater-Reyes, and M. Reiter. 2018. "A multicatchment analysis of headwater and downstream temperature effects from contemporary forest harvesting." *Hydrological Processes* 32(2): 293-304.
- Blakney, J.R., J.L. Loxterman, and E.R. Keeley. 2014. "Range-wide comparisons of northern leatherside chub populations reveal historical and contemporary patterns of genetic variation." *Conservation Genetics* 15(4): 757-770.
- Blanpied, J., J.M. Carozza, and J.M. Antoine. 2018. "Sediment connectivity in the high Pyrenees mountain range by 2013 flood analysis: role of surficial sediment storages." *Geomorphologie* 24(4): 389-402.
- Blondel, L., L. Baillie, J. Quinton, J.B. Alemu, I. Paterson, A.P. Hendry, and P. Bentzen. 2019. "Evidence for contemporary and historical gene flow between guppy populations in different watersheds, with a test for associations with adaptive traits." *Ecology and Evolution* 9(8): 4504-4517.
- Boavida, I., A. Harby, K.D. Clarke, and J. Heggenes. 2017. "Move or stay: habitat use and movements by Atlantic salmon parr (*Salmo salar*) during induced rapid flow variations." *Hydrobiologia* 785(1): 261-275.

- Boddy, N.C., D.J. Booker, and A.R. McIntosh. 2019. "Confluence configuration of river networks controls spatial patterns in fish communities." *Landscape Ecology* 34(1): 187-201.
- Boddy, N.C., D.J. Booker, and A.R. McIntosh. 2020. "Heterogeneity in flow disturbance around river confluences influences spatial patterns in native and non-native species co-occurrence." *Biological Invasions* 22(11): 3457-3475.
- Boddy, N.C., and A.R. McIntosh. 2021. "Could spatial heterogeneity in flow disturbance drive temporal stability of native-invasive species co-occurrence in riverscapes?" *Freshwater Biology* 66(5): 902-913.
- Bogan, M.T., K.S. Boersma, and D.A. Lytle. 2015. "Resistance and resilience of invertebrate communities to seasonal and supraseasonal drought in arid-land headwater streams." *Freshwater Biology* 60(12): 2547-2558.
- Bogan, M.T., J.L. Hwan, and S.M. Carlson. 2015. "High Aquatic Biodiversity in an Intermittent Coastal Headwater Stream at Golden Gate National Recreation Area, California." *Northwest Science* 89(2): 188-197.
- Bohrman, K.J., and E.A. Strauss. 2018. "Macrophyte-driven transient storage and phosphorus uptake in a western Wisconsin stream." *Hydrological Processes* 32.
- Bokhutlo, T., E.R. Cunha, and K.O. Winemiller. 2021. "Ephemeral habitat supports high fish alphadiversity and beta-diversity during drought in a subtropical semiarid wetland." *Biotropica* 53(3): 778-785.
- Bolduc, C., S.F. Lamoureux, and J. Franssen. 2018. "Thermal and isotopic evidence for surface and subsurface water contributions to baseflow in a high Arctic river." *Hydrological Processes* 32(5): 602-616.
- Bond, M.H., T.G. Nodine, T.J. Beechie, and R.W. Zabel. 2019. "Estimating the benefits of widespread floodplain reconnection for Columbia River Chinook salmon." *Canadian Journal of Fisheries and Aquatic Sciences* 76(7): 1212-1226.
- Bond, N.R., S.R. Balcombe, D.A. Crook, J.C. Marshall, N. Menke, and J.S. Lobegeiger. 2015. "Fish population persistence in hydrologically variable landscapes." *Ecological Applications* 25(4): 901-913.
- Borthagaray, A.I., F. Teixeira-De Mello, G. Tesitore, E. Ortiz, M. Illarze, V. Pinelli, L. Urtado, P. Raftopulos, I. González-Bergonzoni, S. Abades, M. Loureiro, and M. Arim. 2020. "Community isolation drives lower fish biomass and species richness, but higher functional evenness, in a river metacommunity." *Freshwater Biology* 65(12): 2081-2095. <u>https://doi.org/10.1111/fwb.13603</u>.
- Botter, G., and N. Durighetto. 2020. "The Stream Length Duration Curve: A Tool for Characterizing the Time Variability of the Flowing Stream Length." *Water Resources Research* 56 (8): e2020WR027282. <u>https://www.ncbi.nlm.nih.gov/pubmed/33041380</u>.

Botwe, P.K., L.A. Barmuta, R. Magierowski, P. McEvoy, P. Goonan, and S. Carver. 2015. "Temporal

Patterns and Environmental Correlates of Macroinvertebrate Communities in Temporary Streams." *PLoS ONE* 10(11).

- Boulton, A.J., R.J. Rolls, K.L. Jaeger, and T. Datry. 2017. "Chapter 2.3 Hydrological Connectivity in Intermittent Rivers and Ephemeral Streams." Pages 79-108 in *Intermittent Rivers and Ephemeral Streams: Ecology and Management*. T. Datry, N. Bonada, and A. Boulton, eds. Academic Press.
- Boulton, A.J., R.J. Rolls, K.L. Jaeger, and T. Datry. 2017. "HYDROLOGICAL CONNECTIVITY IN INTERMITTENT RIVERS AND EPHEMERAL STREAMS." *Intermittent Rivers and Ephemeral Streams: Ecology and Management*: 79-108.
- Bourgeois, B., E. González, A. Vanasse, I. Aubin, and M. Poulin. 2016. "Spatial processes structuring riparian plant communities in agroecosystems: implications for restoration." *Ecological Applications* 26(7): 2103-2115.
- Bowen, H.L., and M.P. Marchetti. 2016. "Ecomorphological plasticity of juvenile fall-run chinook salmon (*Oncorhynchus tshawytscha*) in perennial and ephemeral streams." *Environmental Biology of Fishes* 99(1): 67-78.
- Bradbury, I.R., L.C. Hamilton, M.J. Robertson, C.E. Bourgeois, A. Mansour, and J.B. Dempson. 2014. "Landscape structure and climatic variation determine Atlantic salmon genetic connectivity in the Northwest Atlantic." *Canadian Journal of Fisheries and Aquatic Sciences* 71(2): 246-258. <u>https://doi.org/10.1139/cjfas-2013-0240</u>.
- Brauer, C.J., M.P. Hammer, and L.B. Beheregaray. 2016. "Riverscape genomics of a threatened fish across a hydroclimatically heterogeneous river basin." *Molecular Ecology* 25(20): 5093-5113.
- Brauer, C.J., P.J. Unmack, S. Smith, L. Bernatchez, and L.B. Beheregaray. 2018. "On the roles of landscape heterogeneity and environmental variation in determining population genomic structure in a dendritic system." *Molecular Ecology* 27(17): 3484-3497.
- Brendel, C.E., M.L. Soupir, L.A.M. Long, M.J. Helmers, C.D. Ikenberry, and A.L. Kaleita. 2019. "Catchment-scale Phosphorus Export through Surface and Drainage Pathways." *Journal of Environmental Quality* 48(1): 117-126.
- Brenkman, S.J., R.J. Peters, R.A. Tabor, J.J. Geffre, and K.T. Sutton. 2019. "Rapid Recolonization and Life History Responses of Bull Trout Following Dam Removal in Washington's Elwha River." *North American Journal of Fisheries Management* 39(3): 560-573.
- Brennan, S.R., D.E. Schindler, T.J. Cline, T.E. Walsworth, G. Buck, and D.P. Fernandez. 2019. "Shifting habitat mosaics and fish production across river basins." *Science* 364(6442): 783-786.
- Brennan, S.R., C.E. Torgersen, J.P. Hollenbeck, D.P. Fernandez, C.K. Jensen, and D.E. Schindler. 2016.
 "Dendritic network models: Improving isoscapes and quantifying influence of landscape and instream processes on strontium isotopes in rivers." *Geophysical Research Letters* 43(10): 5043-5051.
- Brintrup, K., C. Amigo, J. Fernandez, A. Hernandez, F. Perez, J. Felez-Bernal, A. Butturini, K. Saez-Carrillo, M.A. Yevenes, and R. Figueroa. 2019. "Comparison of organic matter in intermittent and

perennial rivers of Mediterranean Chile with the support of citizen science." *Revista Chilena de Historia Natural* 92.

- Brooks, A.J., J. Lancaster, B.J. Downes, and B. Wolfenden. 2020. "Just add water: rapid assembly of new communities in previously dry riverbeds, and limited long-distance effects on existing communities." *Oecologia* 194(4): 709-722.
- Brysiewicz, A., M. Bonislawska, P. Czerniejewski, and B. Kierasinski. 2019. "Quality Analysis of Waters from Selected Small Watercourses within the River Basins of Odra River and Wisla River." *Rocznik Ochrona Srodowiska* 21(2): 1202-1216.
- Buddendorf, W.B., F.L. Jackson, I.A. Malcolm, K.J. Millidine, J. Geris, M.E. Wilkinson, and C. Soulsby. 2019. "Integration of juvenile habitat quality and river connectivity models to understand and prioritise the management of barriers for Atlantic salmon populations across spatial scales." *Science* of the Total Environment 655: 557-566.
- Budy, P., P.D. Thompson, M.D. McKell, G.P. Thiede, T.E. Walsworth, and M.M. Conner. 2020. "A Multifaceted Reconstruction of the Population Structure and Life History Expressions of a Remnant Metapopulation of Bonneville Cutthroat Trout: Implications for Maintaining Intermittent Connectivity." *Transactions of the American Fisheries Society* 149(4): 443-461. <u>https://doi.org/10.1002/tafs.10240</u>.
- Bunte, K., K.W. Swingle, J.M. Turowski, S.R. Abt, and D.A. Cenderelli. 2016. "Measurements of coarse particulate organic matter transport in steep mountain streams and estimates of decadal CPOM exports." *Journal of Hydrology* 539: 162-176.
- Burndred, K.R., B.J. Cockayne, J.A. Donaldson, and B.C. Ebner. 2018. "Natural flow events influence the behaviour and movement patterns of eel-tailed catfish (*Tandanus tandanus*) in a subtropical Queensland river." *Australian Journal of Zoology* 66(3): 185-194.
- Buttle, J.M. 2018. "Mediating stream baseflow response to climate change: The role of basin storage." *Hydrological Processes* 32(3): 363-378.
- Cai, Y., M.-J. Shim, L. Guo, and A. Shiller. 2016. "Floodplain influence on carbon speciation and fluxes from the lower Pearl River, Mississippi." *Geochimica et Cosmochimica Acta* 186: 189-206. https://doi.org/10.1016/j.gca.2016.05.007.
- Cai, Y., J. Xu, M. Zhang, J. Wang, and J. Heino. 2019. "Different roles for geography, energy and environment in determining three facets of freshwater molluscan beta diversity at broad spatial scales." *Science of the Total Environment* 659: 451-462.
- Caillon, F., K. Besemer, P. Peduzzi, and J. Schelker. 2021. "Soil microbial inoculation during flood events shapes headwater stream microbial communities and diversity." *Microbial Ecology*.
- Cain, M.R., A.S. Ward, and M. Hrachowitz. 2019. "Ecohydrologic separation alters interpreted hydrologic stores and fluxes in a headwater mountain catchment." *Hydrological Processes* 33(20): 2658-2675.

- Call, B.C., P. Belmont, J.C. Schmidt, and P.R. Wilcock. 2017. "Changes in floodplain inundation under nonstationary hydrology for an adjustable, alluvial river channel." *Water Resources Research* 53(5): 3811-3834.
- Camana, M., R.B. Dala-Corte, and F.G. Becker. 2016. "Relation between species richness and stream slope in riffle fish assemblages is dependent on spatial scale." *Environmental Biology of Fishes* 99: 603-612.
- Camarasa-Belmonte, A.M. 2016. "Flash floods in Mediterranean ephemeral streams in Valencia Region (Spain)." *Journal of Hydrology* 541: 99-115. <u>https://doi.org/10.1016/j.jhydrol.2016.03.019</u>.
- Camp, C.D., T.L. Brock, T.W. Pierson, J.R. Milanovich, and J.C. Maerz. 2018. "LIFE HISTORY AND HABITAT OF THE RARE PATCH-NOSED SALAMANDER (URSPELERPES BRUCEI)." *Herpetological Conservation and Biology* 13(3): 609-616.
- Campbell, R.E., M.J. Winterbourn, T.A. Cochrane, and A.R. McIntosh. 2015. "Flow-related disturbance creates a gradient of metacommunity types within stream networks." *Landscape Ecology* 30(4): 667-680.
- Campbell, T., J. Simmons, J. Saenz, C.L. Jerde, W. Cowan, S. Chandra, and Z. Hogan. 2019. "Population connectivity of adfluvial and stream-resident Lahontan cutthroat trout: implications for resilience, management, and restoration." *Canadian Journal of Fisheries and Aquatic Sciences* 76(3): 426-437. <u>https://doi.org/10.1139/cjfas-2017-0483</u>.
- Canedo-Arguelles, M., K.S. Boersma, M.T. Bogan, J.D. Olden, I. Phillipsen, T.A. Schriever, and D.A. Lytle. 2015. "Dispersal strength determines meta-community structure in a dendritic riverine network." *Journal of Biogeography* 42(4): 778-790.
- Capell, R., D. Tetzlaff, R. Essery, and C. Soulsby. 2014. "Projecting climate change impacts on stream flow regimes with tracer-aided runoff models preliminary assessment of heterogeneity at the mesoscale." *Hydrological Processes* 28(3): 545-558.
- Carlisle, D.M., S.M. Nelson, and K. Eng. 2014. "MACROINVERTEBRATE COMMUNITY CONDITION ASSOCIATED WITH THE SEVERITY OF STREAMFLOW ALTERATION." *River Research and Applications* 30(1): 29-39.
- Carney, R.L., S.M. Mitrovic, T. Jeffries, D. Westhorpe, N. Curlevski, and J.R. Seymour, Jr. 2015. "River bacterioplankton community responses to a high inflow event." *Aquatic Microbial Ecology* 75(3): 187-205.
- Carrara, F., A. Rinaldo, A. Giometto, and F. Altermatt. 2014. "Complex interaction of dendritic connectivity and hierarchical patch size on biodiversity in river-like landscapes." *American Naturalist* 183(1): 13-25.
- Cartwright, I., D. Irvine, C. Burton, and U. Morgenstern. 2018. "Assessing the controls and uncertainties on mean transit times in contrasting headwater catchments." *Journal of Hydrology* 557: 16-29. https://doi.org/10.1016/j.jhydrol.2017.12.007.

- Caruso, B.S. 2015. "A hydrologic connectivity index for jurisdictional analysis of headwater streams in a montane watershed." *Environmental Monitoring and Assessment* 187(10).
- Carvalho, R.A., F.B. Teresa, and F.L. Tejerina-Garro. 2021. "The effect of riverine networks on fish betadiversity patterns in a Neotropical system." *Hydrobiologia* 848(2): 515-529.
- Casarim, R., I.G. Prado, R.C. Loures, and P.S. Pompeu. 2018. "Fish movement patterns in a Neotropical free-flowing tributary located downstream from a large dam." *Marine and Freshwater Research* 69(10): 1626-1634. <u>https://doi.org/10.1071/MF17216</u>.
- Casas-Ruiz, J.P., R.G.M. Spencer, F. Guillemette, D. von Schiller, B. Obrador, D.C. Podgorski, A.M. Kellerman, J. Hartmann, L. Gomez-Gener, S. Sabater, and R. Marce. 2020. "Delineating the Continuum of Dissolved Organic Matter in Temperate River Networks." *Global Biogeochemical Cycles* 34(8): e2019GB006495. <u>https://doi.org/10.1029/2019GB006495</u>.
- Castaldelli, G., E. Soana, E. Racchetti, F. Vincenzi, E.A. Fano, and M. Bartoli. 2015. "Vegetated canals mitigate nitrogen surplus in agricultural watersheds." *Agriculture, Ecosystems and Environment* 212: 253-262.
- Castillo-Escriva, A., J. Rueda, L. Zamora, R. Hernandez, M. del Moral, and F. Mesquita-Joanes. 2016."The role of watercourse versus overland dispersal and niche effects on ostracod distribution in Mediterranean streams (eastern Iberian Peninsula)." *Acta Oecologica* 73: 1-9.
- Castro, D.M.P., P. da Silva, R. Solar, and M. Callisto. 2020. "Unveiling patterns of taxonomic and functional diversities of stream insects across four spatial scales in the neotropical savanna." *Ecological Indicators* 118.
- Cataño-Alvarez, S. 2021. "Critical transition of incising gravel channel to evacuate alluvial lateral supply." *Physical Geography*. <u>https://doi.org/10.1080/02723646.2021.1923368</u>.
- Cathcart, C.N., K.B. Gido, and W.H. Brandenburg. 2019. "Spawning Locations within and among Tributaries Influence Flannelmouth Sucker Offspring Experience." *Transactions of the American Fisheries Society* 148(5): 963-977.
- Cathcart, C.N., K.B. Gido, and M.C. McKinstry. 2015. "Fish Community Distributions and Movements in Two Tributaries of the San Juan River, USA." *Transactions of the American Fisheries Society* 144(5): 1013-1028.
- Cathcart, C.N., K.B. Gido, M.C. McKinstry, and P.D. Mackinnon. 2018. "Patterns of fish movement at a desert river confluence." *Ecology of Freshwater Fish* 27(1): 492-505.
- Cathcart, C.N., M.C. McKinstry, P.D. Mackinnon, and C.M. Ruffing. 2019. "A Tribute to Tributaries: Endangered Fish Distributions within Critical Habitat of the San Juan River, USA." *North American Journal of Fisheries Management* 39(5): 1015-1025.
- Cathcart, C.N., C.A. Pennock, C.A. Cheek, M.C. McKinstry, P.D. Mackinnon, M.M. Conner, and K.B. Gido. 2018. "Waterfall formation at a desert river-reservoir delta isolates endangered fishes." *River Research and Applications* 34(8): 948-956.

- Cauvy-Fraunié, S., R. Espinosa, P. Andino, D. Jacobsen, and O. Dangles. 2015. "Invertebrate Metacommunity Structure and Dynamics in an Andean Glacial Stream Network Facing Climate Change." *PLoS ONE* 10(8).
- Cavalli, M., D. Vericat, and P. Pereira. 2019. "Mapping water and sediment connectivity." *Science of the Total Environment* 673: 763-767.
- Cayuela, H., L. Boualit, D. Arsovski, E. Bonnaire, J. Pichenot, A. Bellec, C. Miaud, J.P. Léna, P. Joly, and A. Besnard. 2016. "Does habitat unpredictability promote the evolution of a colonizer syndrome in amphibian metapopulations?" *Ecology* 97(10): 2658-2670.
- Chappell, J., S.K. McKay, M.C. Freeman, and C.M. Pringle. 2019. "Long-term (37 years) impacts of lowhead dams on freshwater shrimp habitat connectivity in northeastern Puerto Rico." *River Research and Applications* 35(7): 1034-1043.
- Chaput-Bardy, A., N. Alcala, J. Secondi, and S. Vuilleumier. 2017. "Network analysis for species management in rivers networks: Application to the Loire River." *Biological Conservation* 210: 26-36.
- Chará-Serna, A.M., and J.S. Richardson. 2021. "Multiple-Stressor Interactions in Tributaries Alter Downstream Ecosystems in Stream Mesocosm Networks." *Water* 13(9): 1194. <u>https://doi.org/10.3390/w13091194</u>.
- Chen, B., and W.F. Krajewski. 2015. "Recession analysis across scales: The impact of both random and nonrandom spatial variability on aggregated hydrologic response." *Journal of Hydrology* 523: 97-106.
- Chen, D., H. Wang, G. Gielen, Y. Shen, S. Chen, J. Chen, and J. Lu. 2014. "Effects of sediment texture on in-stream nitrogen uptake." *Environmental Earth Sciences* 72(1): 21-33.
- Chen, J., W. Cao, D. Cao, Z. Huang, and Y. Liang. 2015. "Nitrogen loading and nitrous oxide emissions from a river with multiple hydroelectric reservoirs." *Bulletin of Environmental Contamination and Toxicology* 94(5): 633-639.
- Chester, E.T., A.D. Miller, I. Valenzuela, S.J. Wickson, and B.J. Robson. 2015. "Drought survival strategies, dispersal potential and persistence of invertebrate species in an intermittent stream landscape." *Freshwater Biology* 60(10): 2066-2083.
- Chezik, K.A., S.C. Anderson, and J.W. Moore. 2017. "River networks dampen long-term hydrological signals of climate change." *Geophysical Research Letters* 44(14): 7256-7264.
- Childress, E.S., J.D. Allan, and P.B. McIntyre. 2014. "Nutrient Subsidies from Iteroparous Fish Migrations Can Enhance Stream Productivity." *Ecosystems* 17(3): 522-534.
- Chiu, M.C., C. Leigh, R. Mazor, N. Cid, and V. Resh. 2017. "ANTHROPOGENIC THREATS TO INTERMITTENT RIVERS AND EPHEMERAL STREAMS." Intermittent Rivers and Ephemeral Streams: Ecology and Management: 433-454.

- Chiu, M.C., B. Li, K. Nukazawa, V.H. Resh, T. Carvajal, and K. Watanabe. 2020. "Branching networks can have opposing influences on genetic variation in riverine metapopulations." *Diversity and Distributions* 26(12): 1813-1824.
- Chiu, M.C., K. Nukazawa, T. Carvajal, V.H. Resh, B. Li, and K. Watanabe. 2020. "Simulation modeling reveals the evolutionary role of landscape shape and species dispersal on genetic variation within a metapopulation." *Ecography* 43(12): 1891-1901.
- Cho, S.J., P. Wilcock, and B. Hobbs. 2018. "Topographic filtering simulation model for sediment source apportionment." *Geomorphology* 309: 1-19.
- Cho, S.J., C.A. Braudrick, C.L. Dolph, S.S. Day, B.J. Dalzell, and P.R. Wilcock. 2021. "Simulation of fluvial sediment dynamics through strategic assessment of stream gaging data: A targeted watershed sediment loading analysis." *Journal of Environmental Management* 277.
- Clarke, A.D., K.H. Telmer, and J.M. Shrimpton. 2015. "Movement patterns of fish revealed by otolith microchemistry: a comparison of putative migratory and resident species." *Environmental Biology of Fishes* 98(6): 1583-1597.
- Clay, P.A., J.D. Muehlbauer, and M.W. Doyle. 2015. "Effect of tributary and braided confluences on aquatic macroinvertebrate communities and geomorphology in an alpine river watershed." *Freshwater Science* 34(3): 845-856.
- Coble, A.A., A.M. Marcarelli, and E.S. Kane. 2019. "Year-round measurements reveal seasonal drivers of nutrient uptake in a snowmelt-driven headwater stream." *Freshwater Science* 38(1): 156-169.
- Coble, A.C., L.E. Koenig, J.D. Potter, L.M. Parham, and W.H. McDowell. 2019. "Homogenization of dissolved organic matter within a river network occurs in the smallest headwaters." *Biogeochemistry* 143(1): 85-104.
- Cojoc, G.M., G. Romanescu, and A. Tirnovan. 2015. "Exceptional floods on a developed river: case study for the Bistrita River from the Eastern Carpathians (Romania)." *Natural Hazards* 77(3): 1421-1451.
- Collins, S.D., S. Shukla, and N.K. Shrestha. 2016. "Drainage ditches have sufficient adsorption capacity but inadequate residence time for phosphorus retention in the Everglades." *Ecological Engineering* 92: 218-228.
- Comer-Warner, S.A., D.C. Gooddy, S. Ullah, L. Glover, N. Kettridge, S.K. Wexler, J. Kaiser, and S. Krause. 2020. "Seasonal variability of sediment controls of nitrogen cycling in an agricultural stream." *Biogeochemistry* 148(1): 31-48.
- Comer-Warner, S.A., D.C. Gooddy, S. Ullah, L. Glover, A. Percival, N. Kettridge, and S. Krause. 2019. "Seasonal variability of sediment controls of carbon cycling in an agricultural stream." *Science of the Total Environment* 688: 732-741.
- Comte, L., and J.D. Olden. 2018. "Fish dispersal in flowing waters: A synthesis of movement- and genetic-based studies." *Fish and Fisheries* 19(6): 1063-1077.

- Constanza Ranieri, M., P. Gantes, and F. Momo. 2015. "Diversity patterns of Pampean stream vegetation at different spatial scales." *Aquatic Botany* 126: 1-6.
- Copeland, T., D.A. Venditti, and B.R. Barnett. 2014. "The Importance of Juvenile Migration Tactics to Adult Recruitment in Stream-Type Chinook Salmon Populations." *Transactions of the American Fisheries Society* 143(6): 1460-1475.
- Cossart, E., and M. Fressard. 2017. "Assessment of structural sediment connectivity within catchments: insights from graph theory." *Earth Surface Dynamics* 5(2): 253-268.
- Costanzi, J.M., P. Mège, A. Boissinot, F. Isselin-Nondedeu, S. Guérin, O. Lourdais, A. Trochet, Q. Le Petitcorps, A. Legrand, F. Varenne, P. Grillet, S. Morin-Pinaud, and D. Picard. 2018. "Agricultural landscapes and the Loire River influence the genetic structure of the marbled newt in Western France." *Scientific Reports* 8(1).
- Costigan, K.H., M.D. Daniels, and W.K. Dodds. 2015. "Fundamental spatial and temporal disconnections in the hydrology of an intermittent prairie headwater network." *Journal of Hydrology* 522: 305-316.
- Costigan, K.H., M.D. Daniels, J.S. Perkin, and K.B. Gido. 2014. "Longitudinal variability in hydraulic geometry and substrate characteristics of a Great Plains sand-bed river." *Geomorphology* 210: 48-58.
- Costigan, K.H., C.M. Ruffing, J.S. Perkin, and M.D. Daniels. 2016. "Rapid Response of a Sand-Dominated River to Installation and Removal of a Temporary Run-of-the-River Dam." *River Research and Applications* 32(1): 110-124.
- Coughlan, N.E., T.C. Kelly, J. Davenport, and M.A.K. Jansen. 2017. "Up, up and away: bird-mediated ectozoochorous dispersal between aquatic environments." *Freshwater Biology* 62(4): 631-648.
- Couillard, C.M., P. Ouellet, G. Verreault, S. Senneville, S. St-Onge-Drouin, and D. Lefaivre. 2017."Effect of Decadal Changes in Freshwater Flows and Temperature on the Larvae of two Forage Fish Species in Coastal Nurseries of the St. Lawrence Estuary." *Estuaries and Coasts* 40(1): 268-285.
- Coulthard, T.J., and M.J. Van De Wiel. 2017. "Modelling long term basin scale sediment connectivity, driven by spatial land use changes." *Geomorphology* 277: 265-281. https://doi.org/10.1016/j.geomorph.2016.05.027.
- Covino, T. 2017. "Hydrologic connectivity as a framework for understanding biogeochemical flux through watersheds and along fluvial networks." *Geomorphology* 277: 133-144.
- Covino, T.P., A.N. Wlostowski, M.N. Gooseff, W.M. Wollheim, and W.B. Bowden. 2021. "The Seasonality of In-Stream Nutrient Concentrations and Uptake in Arctic Headwater Streams in the Northern Foothills of Alaska's Brooks Range." *Journal of Geophysical Research: Biogeosciences* 126(4).
- Crabot, J., M. Polasek, B. Launay, P. Paril, and T. Datry. 2021. "Drying in newly intermittent rivers leads to higher variability of invertebrate communities." *Freshwater Biology* 66(4): 730-744.

Crawford, J.T., E.H. Stanley, S.A. Spawn, J.C. Finlay, L.C. Loken, and R.G. Striegl. 2014. "Ebullitive

methane emissions from oxygenated wetland streams." Global Change Biology 20(11): 3408-3422.

- Crawford, J.T., E.G. Stets, and L.A. Sprague. 2019. "Network Controls on Mean and Variance of Nitrate Loads from the Mississippi River to the Gulf of Mexico." *Journal of Environmental Quality* 48(6): 1789-1799.
- Creed, I.F., D.M. McKnight, B.A. Pellerin, M.B. Green, B.A. Bergamaschi, G.R. Aiken, D.A. Burns, S.E.G. Findlay, J.B. Shanley, R.G. Striegl, B.T. Aulenbach, D.W. Clow, H. Laudon, B.L. McGlynn, K.J. McGuire, R.A. Smith, and S.M. Stackpoole. 2015. "The river as a chemostat: fresh perspectives on dissolved organic matter flowing down the river continuum." *Canadian Journal of Fisheries and Aquatic Sciences* 72(8): 1272-1285. <u>https://doi.org/10.1139/cjfas-2014-0400</u>.
- Crookes, S., and P.W. Shaw. 2016. "Isolation by distance and non-identical patterns of gene flow within two river populations of the freshwater fish *Rutilus rutilus* (L. 1758)." *Conservation Genetics* 17(4): 861-874. <u>https://doi.org/10.1007/s10592-016-0828-3</u>.
- Cubley, E.S., and R.L. Brown. 2016. "Restoration of Hydrochory Following Dam Removal on the Elwha River, Washington." *River Research and Applications* 32(7): 1566-1575.
- Cunha, E.J., and L. Juen. 2020. "Environmental drivers of the metacommunity structure of insects on the surface of tropical streams of the Amazon." *Austral Ecology* 45(5): 586-595.
- Czegledi, I., P. Saly, P. Takacs, A. Dolezsai, S.A. Nagy, and T. Eros. 2016. "The scales of variability of stream fish assemblages at tributary confluences." *Aquatic Sciences* 78(4): 641-654.
- Czuba, J.A. 2018. "A Lagrangian framework for exploring complexities of mixed-size sediment transport in gravel-bedded river networks." *Geomorphology* 321: 146-152.
- Czuba, J.A., and E. Foufoula-Georgiou. 2014. "A network-based framework for identifying potential synchronizations and amplifications of sediment delivery in river basins." *Water Resources Research* 50(5): 3826-3851.
- Czuba, J.A., and E. Foufoula-Georgiou. 2015. "Dynamic connectivity in a fluvial network for identifying hotspots of geomorphic change." *Water Resources Research* 51(3): 1401-1421.
- Czuba, J.A., E. Foufoula-Georgiou, K.B. Gran, P. Belmont, and P.R. Wilcock. 2017. "Interplay between spatially explicit sediment sourcing, hierarchical river-network structure, and in-channel bed material sediment transport and storage dynamics." *Journal of Geophysical Research-Earth Surface* 122(5): 1090-1120.
- Dala-Corte, R.B., F.G. Becker, and A.S. Melo. 2017. "The importance of metacommunity processes for long-term turnover of riffle-dwelling fish assemblages depends on spatial position within a dendritic network." *Canadian Journal of Fisheries and Aquatic Sciences* 74(1): 101-115.
- D'Ambrosio, J.L., L.R. Williams, M.G. Williams, J.D. Witter, and A.D. Ward. 2014. "Geomorphology, habitat, and spatial location influences on fish and macroinvertebrate communities in modified channels of an agriculturally-dominated watershed in Ohio, USA." *Ecological Engineering* 68: 32-46.

- Danczak, R.E., A.E. Goldman, R.K. Chu, J.G. Toyoda, V.A. Garayburu-Caruso, N. Tolić, E.B. Graham, J.W. Morad, L. Renteria, J.R. Wells, S.P. Herzog, A.S. Ward, and J.C. Stegen. 2021. "Ecological theory applied to environmental metabolomes reveals compositional divergence despite conserved molecular properties." *Science of the Total Environment* 788.
- Danehy, R.J., R.E. Bilby, S. Owen, S.D. Duke, and A. Farrand. 2017. "Interactions of baseflow habitat constraints: Macroinvertebrate drift, stream temperature, and physical habitat for anadromous salmon in the Calapooia River, Oregon." *Aquatic Conservation: Marine and Freshwater Ecosystems* 27(3): 653-662.
- Dare, G.C., R.G. Murray, D.M.M. Courcelles, J.M. Malt, and W.J. Palen. 2020. "Run-of-river dams as a barrier to the movement of a stream-dwelling amphibian." *Ecosphere* 11(8).
- Datry, T., N. Bonada, and A. Boulton. 2017. *Intermittent Rivers and Ephemeral Streams: Ecology and Management*. Academic Press.
- Datry, T., A.J. Boulton, N. Bonada, K. Fritz, C. Leigh, E. Sauquet, K. Tockner, B. Hugueny, and C.N. Dahm. 2018. "Flow intermittence and ecosystem services in rivers of the Anthropocene." *Journal of Applied Ecology* 55(1): 353-364. <u>https://www.ncbi.nlm.nih.gov/pubmed/29681651</u>.
- Datry, T., R. Corti, J. Heino, B. Hugueny, R.J. Rolls, and A. Ruhi. 2017. "HABITAT FRAGMENTATION AND METAPOPULATION, METACOMMUNITY, AND METAECOSYSTEM DYNAMICS IN INTERMITTENT RIVERS AND EPHEMERAL STREAMS." Intermittent Rivers and Ephemeral Streams: Ecology and Management: 377-403.
- Datry, T., K. Fritz, and C. Leigh. 2016. "Challenges, developments and perspectives in intermittent river ecology." *Freshwater Biology* 61(8): 1171-1180.
- Datry, T., H. Pella, C. Leigh, N. Bonada, and B. Hugueny. 2016. "A landscape approach to advance intermittent river ecology." *Freshwater Biology* 61(8): 1200-1213.
- David, S.R., Sr., J.A. Czuba, and D.A. Edmonds. 2019. "Channelization of meandering river floodplains by headcutting." *Geology* 47(1): 15-18.
- Davie, A.W., and S.M. Mitrovic. 2014. "Benthic algal biomass and assemblage changes following environmental flow releases and unregulated tributary flows downstream of a major storage." *Marine and Freshwater Research* 65(12): 1059-1071.
- Davis, B.M., and D.E. Schindler. 2021. "Effects of variability and synchrony in assessing contributions of individual streams to habitat portfolios of river basins." *Ecological Indicators* 124.
- Davis, J., L. Sim, R.M. Thompson, A. Pinder, J.B. Box, N.P. Murphy, F. Sheldon, A. Moran-Ordonez, and P. Sunnucks. 2018. "Patterns and drivers of aquatic invertebrate diversity across an arid biome." *Ecography* 41(2): 375-387.
- Davis, M.P., and M.B. David. 2018. "Nitrous Oxide Fluxes from Agricultural Streams in East-Central Illinois." *Water, Air, and Soil Pollution* 229(11).

- de Jager, M., B. Kaphingst, E.L. Janse, R. Buisman, S.G.T. Rinzema, and M.B. Soons. 2019. "Seed size regulates plant dispersal distances in flowing water." *Journal of Ecology* 107(1): 307-317.
- De Jong, G.D., and S.P. Canton. 2014. "INPUT OF TERRESTRIAL INVERTEBRATES TO STREAMS DURING MONSOON-RELATED FLASH FLOODS IN THE SOUTHWESTERN UNITED STATES." *Southwestern Naturalist* 59(2): 228-234.
- De Jong, R.G.D., S.P. Canton, J. Lynch, and M. Murphy. 2015. "Aquatic Invertebrate and Vertebrate Communities of Ephemeral Stream Ecosystems in the Arid Southwestern United States." *Southwestern Naturalist* 60(4): 349-359.
- De Miguel, R.J., L. Galvez-Bravo, F.J. Oliva-Paterna, L. Cayuela, and C. Fernandez-Delgado. 2016. "Recolonization Process and Fish Assemblage Dynamics in the Guadiamar River (SW Spain) After the Aznalcollar Mine Toxic Spill." *River Research and Applications* 32(6): 1196-1206.
- de Szoeke, S.M., T.L. Crisman, and P.E. Thurman. 2016. "Comparison of macroinvertebrate communities of intermittent and perennial streams in the dry forest of Guanacaste, Costa Rica." *Ecohydrology* 9(4): 659-672.
- Dedrick, A.G., and M.L. Baskett. 2018. "Integrating Genetic and Demographic Effects of Connectivity on Population Stability: The Case of Hatchery Trucking in Salmon." *American Naturalist* 192(2).
- Dehaan, P.W., J. Von Bargen, and P.D. Scheerer. 2017. "Genetic variation and the relationship between stream and lake ecotypes of a threatened desert Catostomid, the Warner sucker (*Catostomus warnerensis*)." *Ecology of Freshwater Fish* 26(4): 609-620.
- Deiner, K., and F. Altermatt. 2014. "Transport distance of invertebrate environmental DNA in a natural river." *PLoS ONE* 9(2).
- del Campo, R., R. Gomez, and G. Singer. 2019. "Dry phase conditions prime wet-phase dissolved organic matter dynamics in intermittent rivers." *Limnology and Oceanography* 64(5): 1966-1979.
- Delconte, C.A., E. Sacchi, E. Racchetti, M. Bartoli, J. Mas-Pla, and V. Re. 2014. "Nitrogen inputs to a river course in a heavily impacted watershed: a combined hydrochemical and isotopic evaluation (Oglio River Basin, N Italy)." *Science of the Total Environment*: 924-938.
- Della Croce, P., G.C. Poole, R.A. Payn, and C. Izurieta. 2014. "Simulating the effects of stream network topology on the spread of introgressive hybridization across fish populations." *Ecological Modelling* 279: 68-77.
- Dexter, T., N. Bond, R. Hale, and P. Reich. 2014. "Dispersal and recruitment of fish in an intermittent stream network." *Austral Ecology* 39(2): 225-235. <u>https://doi.org/10.1111/aec.12064</u>.
- Dias, M.S., T. Oberdorff, B. Hugueny, F. Leprieur, C. Jezequel, J.F. Cornu, S. Brosse, G. Grenouillet, and P.A. Tedesco. 2014. "Global imprint of historical connectivity on freshwater fish biodiversity." *Ecology Letters* 17(9): 1130-1140. <u>https://www.ncbi.nlm.nih.gov/pubmed/25039890</u>.
- Ding, C., X. Jiang, L. Wang, H. Fan, L. Chen, J. Hu, H. Wang, Y. Chen, X. Shi, H. Chen, B. Pan, L.

Ding, C. Zhang, and D. He. 2019. "Fish Assemblage Responses to a Low-head Dam Removal in the Lancang River." *Chinese Geographical Science* 29(1): 26-36.

- Dobos, M.E., B.J. Bowersox, T. Copeland, and E.J. Stark. 2020. "Understanding Life History Diversity of a Wild Steelhead Population and Managing for Resiliency." North American Journal of Fisheries Management 40(5): 1087-1099.
- Dobos, M.E., M.P. Corsi, D.J. Schill, J.M. Dupont, and M.C. Quist. 2016. "Influences of Summer Water Temperatures on the Movement, Distribution, and Resource Use of Fluvial Westslope Cutthroat Trout in the South Fork Clearwater River Basin." North American Journal of Fisheries Management 36(3): 549-567.
- Dong, G., B. Weng, T. Qin, D. Yan, H. Wang, B. Gong, W. Bi, and J. Wang. 2018. "Study on the stable isotopes in surface waters of the Naqu River basin, Tibetan Plateau." *Journal of the Science Society of Thailand* 44(6): 402-411.
- Dong, X., D.A. Lytle, J.D. Olden, T.A. Schriever, and R. Muneepeerakul. 2017. "Importance of neutral processes varies in time and space: Evidence from dryland stream ecosystems." *PLoS ONE* 12(5).
- Dong, X.Y., B. Li, F.Z. He, Y. Gu, M.Q. Sun, H.M. Zhang, L. Tan, W. Xiao, S.R. Liu, and Q.H. Cai. 2016. "Flow directionality, mountain barriers and functional traits determine diatom metacommunity structuring of high mountain streams." *Scientific Reports* 6.
- Donovan, M., P. Belmont, and Z. Sylvester. 2021. "Evaluating the Relationship Between Meander-Bend Curvature, Sediment Supply, and Migration Rates." *Journal of Geophysical Research-Earth Surface* 126(3).
- Doretto, A., E. Piano, E. Falasco, S. Fenoglio, M.C. Bruno, and F. Bona. 2018. "Investigating the role of refuges and drift on the resilience of macroinvertebrate communities to drying conditions: An experiment in artificial streams." *River Research and Applications* 34(7): 777-785.
- Doretto, A., E. Piano, and C.E. Larson. 2020. "The River Continuum Concept: lessons from the past and perspectives for the future." *Canadian Journal of Fisheries and Aquatic Sciences* 77(11): 1853-1864.
- Douglas, C.M., M. Mulligan, X.A. Harrison, J.R. Henschel, N. Pettorelli, and G. Cowlishaw. 2016. "Widespread dieback of riparian trees on a dammed ephemeral river and evidence of local mitigation by tributary flows." *PeerJ* 4.
- Douglas, C.M.S., G. Cowlishaw, X.A. Harrison, J. Henschel, N. Pettorelli, and M. Mulligan. 2018. "Identifying the determinants of tree distributions along a large ephemeral river." *Ecosphere* 9(6). <u>https://doi.org/10.1002/ecs2.2223</u>.
- Doulatyari, B., A. Betterle, D. Radny, E.A. Celegon, P. Fanton, M. Schirmer, and G. Botter. 2017."Patterns of streamflow regimes along the river network: The case of the Thur river." *Environmental Modelling and Software* 93: 42-58.

Downes, B.J., and J. Lancaster. 2018. "Itinerant, nomad or invader? A field experiment sheds light on the

characteristics of successful dispersers and colonists." Freshwater Biology 63(11): 1394-1406.

- Driver, L.J., and D.J. Hoeinghaus. 2016. "Fish metacommunity responses to experimental drought are determined by habitat heterogeneity and connectivity." *Freshwater Biology* 61(4): 533-548.
- Driver, L.J., and D.J. Hoeinghaus. 2016. "Spatiotemporal dynamics of intermittent stream fish metacommunities in response to prolonged drought and reconnectivity." *Marine and Freshwater Research* 67(11): 1667-1679.
- Du, Y., B. Lou, R. Peng, and M. Zhou. 2020. "The Fisher-KPP equation over simple graphs: varied persistence states in river networks." *Journal of Mathematical Biology* 80(5): 1559-1616.
- Duan, S., T. Newcomer-Johnson, P. Mayer, and S. Kaushal. 2016. "Phosphorus Retention in Stormwater Control Structures across Streamflow in Urban and Suburban Watersheds." *Water* 8(9): 390. <u>https://doi.org/10.3390/w8090390</u>.
- Dugdale, S.J., N.E. Bergeron, and A. St-Hilaire. 2015. "Spatial distribution of thermal refuges analysed in relation to riverscape hydromorphology using airborne thermal infrared imagery." *Remote Sensing of Environment* 160: 43-55.
- Duncan, M.B., R.G. Bramblett, and A. Zale. 2021. "Movements of selected minnows between the lower Yellowstone River and its tributaries." *Canadian Journal of Zoology* 99(1): 45-56.
- Durighetto, N., and G. Botter. 2021. "Time-lapse visualization of spatial and temporal patterns of stream network dynamics." *Hydrological Processes* 35(2): e14043. <u>https://doi.org/10.1002/hyp.14053</u>.
- Dyer, J.J., and S.K. Brewer. 2020. "Seasonal movements and tributary-specific fidelity of blue sucker Cycleptus elongatus in a Southern Plains riverscape." *Journal of Fish Biology* 97(1): 279-292.
- Dzul, M.C., C.B. Yackulic, D.M. Stone, and D.R. Van Haverbeke. 2016. "Survival, Growth, and Movement of Subadult Humpback Chub, Gila Cypha, in the Little Colorado River, Arizona." *River Research and Applications* 32(3): 373-382.
- Earnest, K., J. Scott, J. Schaefer, and D. Duvernell. 2014. "The landscape genetics of syntopic topminnows (*Fundulus notatus* and *F. olivaceus*) in a riverine contact zone." *Ecology of Freshwater Fish* 23(4): 572-580. <u>https://doi.org/10.1111/eff.12109</u>.
- Eastman, M., S. Parry, C. Sefton, J. Park, and J. England. 2021. "Reconstructing Spatiotemporal Dynamics in Hydrological State Along Intermittent Rivers." *Water* 13(4).
- Ebersole, J.L., P.J. Wigington, S.G. Leibowitz, R.L. Comeleo, and J. Van Sickle. 2015. "Predicting the occurrence of cold-water patches at intermittent and ephemeral tributary confluences with warm rivers." *Freshwater Science* 34(1): 111-124. <u>https://doi.org/10.1086/678127</u>.
- Edge, C.B., M.J. Fortin, D.A. Jackson, D. Lawrie, L. Stanfield, and N. Shrestha. 2017. "Habitat alteration and habitat fragmentation differentially affect beta diversity of stream fish communities." *Landscape Ecology* 32(3): 647-662.
- Eisenhauer, Z.J., P.M. Christman, J.M. Matte, W.R. Ardren, D.J. Fraser, and J.W.A. Grant. 2021.

"Revisiting the restricted movement paradigm: the dispersal of Atlantic salmon fry from artificial redds." *Canadian Journal of Fisheries and Aquatic Sciences* 78(4): 493-503. https://doi.org/10.1139/cjfas-2020-0162.

- Elias, D., and M.J. Bernot. 2017. "Pesticide and nitrate transport in an agriculturally influenced stream in Indiana." *Environmental Monitoring and Assessment* 189(4).
- Emelko, M.B., M. Stone, U. Silins, D. Allin, A.L. Collins, C.H. Williams, A.M. Martens, and K.D. Bladon. 2015. "Sediment-phosphorus dynamics can shift aquatic ecology and cause downstream legacy effects after wildfire in large river systems." *Global Change Biology* 22(3): 1168-1184.
- Eppehimer, D.E., H. Hamdhani, K.D. Hollien, and M.T. Bogan. 2020. "Evaluating the potential of treated effluent as novel habitats for aquatic invertebrates in arid regions." *Hydrobiologia* 847(16): 3381-3396.
- Erdozain, M., K.A. Kidd, E.J.S. Emilson, S.S. Capell, T. Luu, D.P. Kreutzweiser, and M.A. Gray. 2021."Forest management impacts on stream integrity at varying intensities and spatial scales: Do biological effects accumulate spatially?" *Science of the Total Environment* 763.
- Ertel, B.D., T.E. McMahon, T.M. Koel, R.E. Gresswell, and J.C. Burckhardt. 2017. "Life History Migrations of Adult Yellowstone Cutthroat Trout in the Upper Yellowstone River." North American Journal of Fisheries Management 37(4): 743-755.
- Evans, D.M., S.H. Schoenholtz, S.M. Griffith, and W.C. Floyd. 2014. "Spatial and temporal patterns of dissolved nitrogen and phosphorus in surface waters of a multi-land use basin." *Environmental Monitoring and Assessment* 186(2): 873-887.
- Falasco, E., F. Bona, C. Monauni, A. Zeni, and E. Piano. 2019. "Environmental and spatial factors drive diatom species distribution in Alpine streams: Implications for biomonitoring." *Ecological Indicators* 106.
- Fasching, C., A.J. Ulseth, J. Schelker, G. Steniczka, and T.J. Battin. 2016. "Hydrology controls dissolved organic matter export and composition in an Alpine stream and its hyporheic zone." *Limnology and Oceanography* 61(2): 558-571.
- Faukner, J., S. Sillowaway, M. Sparkman, and P. Drobny. 2017. "A previously undocumented life history behavior in juvenile coho salmon (*Oncorhynchus kisutch*) from the Klamath River, California." *California Fish and Game* 103(2): 72-78.
- Faustino-Queiroz, A.C., and B.F. Terra. 2020. "Ecological drivers of fish metacommunities: Environmental and spatial factors surpass predation in structuring metacommunities of intermittent rivers." *Ecology of Freshwater Fish* 29(1): 145-155. <u>https://doi.org/10.1111/eff.12502</u>.
- Favaro, E.A., and S.F. Lamoureux. 2015. "Downstream patterns of suspended sediment transport in a High Arctic river influenced by permafrost disturbance and recent climate change." *Geomorphology* 246: 359-369.
- Febria, C.M., J.D. Hosen, B.C. Crump, M.A. Palmer, and D.D. Williams. 2015. "Microbial responses to

changes in flow status in temporary headwater streams: a cross-system comparison." *Frontiers in Microbiology* 6.

- Feist, S.M., J.T. Briggler, J.B. Koppelman, and L.S. Eggert. 2014. "Within-river gene flow in the hellbender (*Cryptobranchus alleganiensis*) and implications for restorative release." *Conservation Genetics* 15(4): 953-966.
- Fencl, J.S., M.E. Mather, K.H. Costigan, and M.D. Daniels. 2015. "How Big of an Effect Do Small Dams Have? Using Geomorphological Footprints to Quantify Spatial Impact of Low-Head Dams and Identify Patterns of Across-Dam Variation." *PLoS ONE* 10(11).
- Ferencz, S.B., M.B. Cardenas, and B.T. Neilson. 2019. "Analysis of the Effects of Dam Release Properties and Ambient Groundwater Flow on Surface Water-Groundwater Exchange Over a 100km-Long Reach." Water Resources Research 55(11): 8526-8546.
- Fernandes Cunha, D.G., N.R. Finkler, N. Gomez, J. Cochero, J.L. Donadelli, W.A. Saltarelli, M.d. Calijuri, A.C. Poli Miwa, F. Tromboni, W.K. Dodds, I.G. Boechat, B. Gucker, and S.A. Thomas. 2020. "Agriculture influences ammonium and soluble reactive phosphorus retention in South American headwater streams." *Ecohydrology* 13(2).
- Ferreira, F.C., U.P. Souza, M. Cetra, and M. Petrere. 2019. "Rhithronic and potamonic fishes coexist in wadeable streams under distinct metacommunity processes." *Ecology of Freshwater Fish* 28(1): 85-96.
- Ferreira, K., T.M. Lopes, I.P. Affonso, A.A. Agostinho, and L.C. Gomes. 2020. "Dam reverse flow events influence limnological variables and fish assemblages of a downstream tributary in a Neotropical floodplain." *River Research and Applications* 36(2): 305-313. <u>https://doi.org/10.1002/rra.3584</u>.
- Ficklin, D.L., J.T. Abatzoglou, S.M. Robeson, S.E. Null, and J.H. Knouft. 2018. "Natural and managed watersheds show similar responses to recent climate change." *Proceedings of the National Academy of Sciences of the United States of America* 115(34): 8553-8557.
- Filipe, A.F., L. Quaglietta, M. Ferreira, M.F. Magalhães, and P. Beja. 2017. "Geostatistical distribution modelling of two invasive crayfish across dendritic stream networks." *Biological Invasions* 19(10): 2899-2912. <u>https://doi.org/10.1007/s10530-017-1492-3</u>.
- Finkler, N.R., F. Tromboni, I.G. Boechat, B. Gucker, and D.G. Fernandes Cunha. 2018. "Nitrogen and Phosphorus Uptake Dynamics in Tropical Cerrado Woodland Streams." *Water* 10(8).
- Fiori, A. 2014. "Channeling, channel density and mass recovery in aquifer transport, with application to the MADE experiment." *Water Resources Research* 50(12): 9148-9161.
- Fischer, J., C. Paukert, and M. Daniels. 2015. "Influence of Riparian and Watershed Alterations on Sandbars in a Great Plains River." *River Research and Applications* 31(9): 1140-1150.
- Fitzpatrick, S.W., H. Crockett, and W.C. Funk. 2014. "Water availability strongly impacts population genetic patterns of an imperiled Great Plains endemic fish." *Conservation Genetics* 15(4): 771-788.

- Fleming, B.P., and N.G. Smith. 2019. "Spatial Distribution and Hybridization Levels of Guadalupe Bass Five Years after Remedial Stocking." *American Fisheries Society. Symposium* 87: 231-244.
- Fletcher, D.E., B.E. Lindell, A.H. Lindell, P.T. Stankus, N.D. Fletcher, J.V. McArthur, and J.C. Seaman. 2019. "Basins, beaver ponds, and the storage and redistribution of trace elements in an industrially impacted coastal plain stream on the Savannah River Site, SC, USA." *Environment International* 133.
- Flores-Galicia, N., I. Trejo, and N. Ramírez-Marcial. 2021. "Environment-driven changes in diversity of riparian plant communities along a mountain river." *Ecology and Evolution* 11(10): 5690-5701.
- Foley, K.M., A. Rosenberger, and F.J. Mueter. 2018. "Longitudinal Patterns of Juvenile Coho Salmon Distribution and Densities in Headwater Streams of the Little Susitna River, Alaska." *Transactions of the American Fisheries Society* 147(2): 247-264.
- Fong, C.S., S.M. Yarnell, and J.H. Viers. 2016. "Pulsed Flow Wave Attenuation on a Regulated Montane River." *River Research and Applications* 32(5): 1047-1058.
- Ford, W.I., J.F. Fox, D.T. Mahoney, G. Degraves, A. Erhardt, and S. Yost. 2020. "Backwater Confluences of the Ohio River: Organic and Inorganic Fingerprints Explain Sediment Dynamics in Wetlands and Marinas." *Journal of the American Water Resources Association* 56(4): 692-711.
- Ford, W.I., J.F. Fox, E. Pollock, H. Rowe, and S. Chakraborty. 2015. "Testing assumptions for nitrogen transformation in a low-gradient agricultural stream." *Journal of Hydrology* 527: 908-922.
- Ford, W.I., A. Husic, A. Fogle, and J. Taraba. 2019. "Long-term assessment of nutrient flow pathway dynamics and in-stream fate in a temperate karst agroecosystem watershed." *Hydrological Processes* 33(11): 1610-1628.
- Ford, W.I., K. King, and M.R. Williams. 2018. "Upland and in-stream controls on baseflow nutrient dynamics in tile-drained agroecosystem watersheds." *Journal of Hydrology* 556: 800-812.
- Fortesa, J., G.F. Ricci, J. Garcia-Comendador, F. Gentile, J. Estrany, E. Sauquet, T. Datry, and A.M. De Girolamo. 2021. "Analysing hydrological and sediment transport regime in two Mediterranean intermittent rivers." *Catena* 196.
- Four, B., M. Thomas, E. Arce, A. Cebron, M. Danger, and D. Banas. 2017. "Fishpond dams affect leaflitter processing and associated detritivore communities along intermittent low-order streams." *Freshwater Biology* 62(10): 1741-1755.
- Fovet, O., D.M. Cooper, D.L. Jones, T.G. Jones, and C.D. Evans. 2020. "Dynamics of dissolved organic matter in headwaters: comparison of headwater streams with contrasting DOM and nutrient composition." *Aquatic Sciences* 82(2): 29. https://link.springer.com/article/10.1007/s00027-020-0704-6.
- Francisco de Oliveira, P.H., K.B. Machado, F.B. Teresa, J. Heino, and J.C. Nabout. 2020. "Spatial processes determine planktonic diatom metacommunity structure of headwater streams." *Limnologica* 84.

- Franssen, N.R., S.L. Durst, K.B. Gido, D.W. Ryden, V. Lamarra, and D.L. Propst. 2016. "Long-term Dynamics of Large-bodied Fishes Assessed from Spatially Intensive Monitoring of a Managed Desert River." *River Research and Applications* 32(3): 348-361.
- Fraser, G.S., D.L. Winkelman, K.R. Bestgen, and K.G. Thompson. 2017. "Tributary Use by Imperiled Flannelmouth and Bluehead Suckers in the Upper Colorado River Basin." *Transactions of the American Fisheries Society* 146(5): 858-870.
- French, D.W., D.E. Schindler, S.R. Brennan, and D. Whited. 2020. "Headwater Catchments Govern Biogeochemistry in America's Largest Free-Flowing River Network." *Journal of Geophysical Research: Biogeosciences* 125(12): e2020JG005851. <u>https://doi.org/10.1029/2020JG005851</u>.
- Frieden, J.C., E.E. Peterson, J.A. Webb, and P.M. Negus. 2014. "Improving the predictive power of spatial statistical models of stream macroinvertebrates using weighted autocovariance functions." *Environmental Modelling and Software* 60: 320-330.
- Fritz, K.M., G.J. Pond, B.R. Johnson, and C.D. Barton. 2019. "Coarse particulate organic matter dynamics in ephemeral tributaries of a Central Appalachian stream network." *Ecosphere* 10(3).
- Fritz, K.M., K.A. Schofield, L.C. Alexander, M.G. McManus, H.E. Golden, C.R. Lane, W.G. Kepner, S.D. LeDuc, J.E. DeMeester, and A.I. Pollard. 2018. "Physical and Chemical Connectivity of Streams and Riparian Wetlands to Downstream Waters: A Synthesis." *JAWRA Journal of the American Water Resources Association* 54(2): 323-345.
- Fryirs, K., and D.B. Gore. 2014. "Geochemical insights to the formation of 'sedimentary buffers': Considering the role of tributary-trunk stream interactions on catchment-scale sediment flux and drainage network dynamics." *Geomorphology* 219: 1-9.
- Fuller, M.R., M.W. Doyle, and D.L. Strayer. 2015. "Causes and consequences of habitat fragmentation in river networks." *Annals of the New York Academy of Sciences* 1355: 31-51.
- Fullerton, A.H., C.E. Torgersen, J.J. Lawler, R.N. Faux, E.A. Steel, T.J. Beechie, J.L. Ebersole, and S.G. Leibowitz. 2015. "Rethinking the longitudinal stream temperature paradigm: region-wide comparison of thermal infrared imagery reveals unexpected complexity of river temperatures." *Hydrological Processes* 29(22): 4719-4737.
- Gabbud, C., and S.N. Lane. 2016. "Ecosystem impacts of Alpine water intakes for hydropower: the challenge of sediment management." *WIREs Water* 3(1): 41-61.
- Galbraith, H.S., D.T. Zanatta, and C.C. Wilson. 2015. "Comparative analysis of riverscape genetic structure in rare, threatened and common freshwater mussels." *Conservation Genetics* 16(4): 845-857.
- Galia, T. 2018. "Controls on log step occurrence in steep headwater streams draining Carpathian managed forests." *Ecological Engineering* 120: 384-396.
- Galia, T., R. Tichavsky, V. Skarpich, and K. Silhan. 2018. "Characteristics of large wood in a headwater channel after an extraordinary event: The roles of transport agents and check dams." *Catena* 165:

537-550.

- Gallardo, B., S. Doledec, A. Paillex, D.B. Arscott, F. Sheldon, F. Zilli, S. Merigoux, E. Castella, and F.A. Comin. 2014. "Response of benthic macroinvertebrates to gradients in hydrological connectivity: a comparison of temperate, subtropical, Mediterranean and semiarid river floodplains." *Freshwater Biology* 59(3): 630-648.
- Gallo, E.L., K.A. Lohse, C.M. Ferlin, T. Meixner, and P.D. Brooks. 2014. "Physical and biological controls on trace gas fluxes in semi-arid urban ephemeral waterways." *Biogeochemistry* 121(1): 189-207.
- Gallo, E.L., T. Meixner, K.A. Lohse, and H. Nicholas. 2020. "Estimating Surface Water Presence and Infiltration in Ephemeral to Intermittent Streams in the Southwestern US." *Frontiers in Water* 2(47): 572950.
- Gangodagamage, C., E. Foufoula-Georgiou, and P. Belmont. 2014. "River basin organization around the main stem: Scale invariance in tributary branching and the incremental area function." *Journal of Geophysical Research-Earth Surface* 119(10): 2174-2193.
- Gansfort, B., and W. Traunspurger. 2019. "Environmental factors and river network position allow prediction of benthic community assemblies: A model of nematode metacommunities." *Scientific Reports* 9(1).
- Gao, P., and Z. Zhang. 2016. "Spatial patterns of sediment dynamics within a medium-sized watershed over an extreme storm event." *Geomorphology* 267: 25-36.
- Garbin, S., E.A. Celegon, P. Fanton, and G. Botter. 2019. "Hydrological controls on river network connectivity." *Royal Society Open Science* 6(2).
- Garcia, C., D.A. Schumann, J. Howell, B.D.S. Graeb, K.N. Bertrand, and R.A. Klumb. 2018."Seasonality, floods and droughts structure larval fish assemblages in prairie rivers." *Ecology of Freshwater Fish* 27(1): 389-397.
- Garcia, V.J., P. Gantes, L. Gimenez, C. Hegoburu, N. Ferreiro, F. Sabater, and C. Feijoo. 2017. "High nutrient retention in chronically nutrient-rich lowland streams." *Freshwater Science* 36(1): 26-40.
- Gasiunas, V., and J. Lysoviene. 2014. "Nutrient retention efficiency of small regulated streams during the season of low-flow regime in Central Lithuanian lowland." *Nordic Hydrology* 45(3): 357-367.
- Gateuille, D., P.N. Owens, E.L. Petticrew, B.P. Booth, T.D. French, and S.J. Dery. 2019. "Determining contemporary and historical sediment sources in a large drainage basin impacted by cumulative effects: the regulated Nechako River, British Columbia, Canada." *Journal of Soils and Sediments* 19(9): 3357-3373.
- Geismar, J., P. Haase, C. Nowak, J. Sauer, and S.U. Pauls. 2015. "Local population genetic structure of the montane caddisfly Drusus discolor is driven by overland dispersal and spatial scaling." *Freshwater Biology* 60(1): 209-221.

- Gellis, A.C., M.K. Myers, G.B. Noe, C.R. Hupp, E.R. Schenk, and L. Myers. 2017. "Storms, channel changes, and a sediment budget for an urban-suburban stream, Difficult Run, Virginia, USA." *Geomorphology* 278: 128-148.
- Gendaszek, A.S., J.B. Dunham, C.E. Torgersen, D.P. Hockman-Wert, M.P. Heck, J. Thorson, J. Mintz, and T. Allai. 2020. "Land-Cover and Climatic Controls on Water Temperature, Flow Permanence, and Fragmentation of Great Basin Stream Networks." *Water* 12(7).
- Ghosh, D.K., D. Wang, and T. Zhu. 2016. "On the transition of base flow recession from early stage to late stage." *Advances in Water Resources* 88: 8-13.
- Gibson, C.A., B.J. Koch, Z.G. Compson, B.A. Hungate, and J.C. Marks. 2018. "Ecosystem responses to restored flow in a travertine river." *Freshwater Science* 37(1): 169-177.
- Gibson, C.A., C.M. O'Reilly, A.L. Conine, and S.M. Lipshutz. 2015. "Nutrient uptake dynamics across a gradient of nutrient concentrations and ratios at the landscape scale." *Journal of Geophysical Research: Biogeosciences* 120(2): 326-340.
- Gido, K.B., J.E. Whitney, J.S. Perkin, and T.F. Turner. 2016. "Fragmentation, connectivity and fish species persistence in freshwater ecosystems." Pages 292-323 in *Conservation of Freshwater Fishes*. G. Closs, M. Krkosek, and J. Olden, eds. Conservation Biology. Cambridge University Press, Cambridge.
- Gilbert, J.T., and A.C. Wilcox. 2020. "Sediment Routing and Floodplain Exchange (SeRFE): A Spatially Explicit Model of Sediment Balance and Connectivity Through River Networks." *Journal of Advances in Modeling Earth Systems* 12(9).
- Gilbert, M.D., and A.A. Pease. 2019. "Use of fragmented reservoir habitats by larval fish assemblages across years with contrasting hydrological conditions." *Environmental Biology of Fishes* 102(6): 857-871.
- Gilbert-Horvath, E.A., K.A. Pipal, B.C. Spence, T.H. Williams, and J.C. Garza. 2016. "Hierarchical Phylogeographic Structure of Coho Salmon in California." *Transactions of the American Fisheries Society* 145(5): 1122-1138.
- Giling, D.P., R. Mac Nally, and R.M. Thompson. 2015. "How Might Cross-System Subsidies in Riverine Networks be Affected by Altered Flow Variability?" *Ecosystems* 18(7): 1151-1164.
- Ginson, R., R.P. Walter, N.E. Mandrak, C.L. Beneteau, and D.D. Heath. 2015. "Hierarchical analysis of genetic structure in the habitat-specialist Eastern Sand Darter (*Ammocrypta pellucida*)." *Ecology and Evolution* 5(3): 695-708. https://doi.org/10.1002/ece3.1392.
- Glassic, H.C., and J.W. Gaeta. 2020. "The influence of multiyear drought and associated reduction in tributary connectivity on an adfluvial fish species." *Ecology of Freshwater Fish* 29(4): 588-601.
- Glendell, M., and R.E. Brazier. 2014. "Accelerated export of sediment and carbon from a landscape under intensive agriculture." *Science of the Total Environment*: 643-656.

- Glińska-Lewczuk, K., I. Gołaś, J. Koc, A. Gotkowska-Płachta, M. Harnisz, and A. Rochwerger. 2016."The impact of urban areas on the water quality gradient along a lowland river." *Environmental Monitoring and Assessment* 188(11).
- Godsey, S.E., and J.W. Kirchner. 2014. "Dynamic, discontinuous stream networks: hydrologically driven variations in active drainage density, flowing channels and stream order." *Hydrological Processes* 28(23): 5791-5803.
- Goeller, B.C., C.M. Febria, H.J. Warburton, K.L. Hogsden, K.E. Collins, H.S. Devlin, J.S. Harding, and A.R. McIntosh. 2019. "Springs drive downstream nitrate export from artificially-drained agricultural headwater catchments." *Science of the Total Environment* 671: 119-128.
- Golden, H.E., K.E. Holsinger, L.A. Deegan, C.J.A. Mackenzie, and M.C. Urban. 2021. "River drying influences genetic variation and population structure in an Arctic freshwater fish." *Conservation Genetics* 22(3): 369-382.
- Gomez-Gener, L., B. Obrador, R. Marce, V. Acuna, N. Catalan, J. Pere Casas-Ruiz, S. Sabater, I. Munoz, and D. von Schiller. 2016. "When Water Vanishes: Magnitude and Regulation of Carbon Dioxide Emissions from Dry Temporary Streams." *Ecosystems* 19(4): 710-723.
- Gomez-Velez, J.D., and J.W. Harvey. 2014. "A hydrogeomorphic river network model predicts where and why hyporheic exchange is important in large basins." *Geophysical Research Letters* 41(18): 6403-6412.
- Gonzalez-Ferreras, A.M., E. Bertuzzo, J. Barquin, L. Carraro, C. Alonso, and A. Rinaldo. 2019. "Effects of altered river network connectivity on the distribution of Salmo trutta: Insights from a metapopulation model." *Freshwater Biology* 64(11): 1877-1895.
- Gonzalez-Pinzon, R., R. Haggerty, and A. Argerich. 2014. "Quantifying spatial differences in metabolism in headwater streams." *Freshwater Science* 33(3): 798-811. <u>https://doi.org/10.1086/677555</u>.
- Goodman, D.H., N.A. Som, and N.J. Hetrick. 2018. "Increasing the availability and spatial variation of spawning habitats through ascending baseflows." *River Research and Applications* 34(7): 844-853.
- Goodrich, D.C., W.G. Kepner, and L.R. Levick. 2018. "Southwestern intermittent and ephemeral stream connectivity." *Journal of the American Water Resources Association* 54(2): 400-422.
- Gothe, E., A. Baattrup-Pedersen, P. Wiberg-Larsen, D. Graeber, E.A. Kristensen, and N. Friberg. 2017.
 "Environmental and spatial controls of taxonomic versus trait composition of stream biota." *Freshwater Biology* 62(2): 397-413.
- Gothe, E., N. Friberg, M. Kahlert, J. Temnerud, and L. Sandin. 2014. "Headwater biodiversity among different levels of stream habitat hierarchy." *Biodiversity and Conservation* 23(1): 63-80.
- Goulsbra, C., M. Evans, and J. Lindsay. 2014. "Temporary streams in a peatland catchment: pattern, timing, and controls on stream network expansion and contraction." *Earth Surface Processes and Landforms* 39(6): 790-803.

- Gourdin, E., S. Huon, O. Evrard, O. Ribolzi, T. Bariac, O. Sengtaheuanghoung, and S. Ayrault. 2015. "Sources and export of particle-borne organic matter during a monsoon flood in a catchment of northern Laos." *Biogeosciences* 12(4): 1073-1089.
- Graham, S.E., R. Storey, and B. Smith. 2017. "Dispersal distances of aquatic insects: upstream crawling by benthic EPT larvae and flight of adult Trichoptera along valley floors." *New Zealand Journal of Marine and Freshwater Research* 51(1): 146-164.
- Grams, P.E., D.J. Dean, A.E. Walker, A. Kasprak, and J.C. Schmidt. 2020. "The roles of flood magnitude and duration in controlling channel width and complexity on the Green River in Canyonlands, Utah, USA." *Geomorphology* 371.
- Gran, K.B., and J.A. Czuba. 2017. "Sediment pulse evolution and the role of network structure." *Geomorphology* 277: 17-30.
- Grant, S.B., M. Azizian, P. Cook, F. Boano, and M.A. Rippy. 2018. "Factoring stream turbulence into global assessments of nitrogen pollution." *Science* 359(6381): 1266-1269.
- Guastini, E., G. Zuecco, A. Errico, G. Castelli, E. Bresci, F. Preti, and D. Penna. 2019. "How does streamflow response vary with spatial scale? Analysis of controls in three nested Alpine catchments." *Journal of Hydrology* 570: 705-718.
- Guillen-Ludena, S. 2015. "Hydro-Morphodynamic Differences Induced by Different Angles in an Experimental Discordant Confluence." *Proceedings of the 36th Iahr World Congress*: 2525-2535.
- Guillen-Ludena, S., M.J. Franca, F. Alegria, A.J. Schleiss, and A.H. Cardoso. 2017."Hydromorphodynamic effects of the width ratio and local tributary widening on discordant confluences." *Geomorphology* 293: 289-304.
- Guillen-Ludena, S., M.J. Franca, A.H. Cardoso, and A.J. Schleiss. 2016. "Evolution of the hydromorphodynamics of mountain river confluences for varying discharge ratios and junction angles." *Geomorphology* 255: 1-15.
- Gutierrez, R.R., J.D. Abad, M. Choi, and H. Montoro. 2014. "Characterization of confluences in free meandering rivers of the Amazon basin." *Geomorphology* 220: 1-14.
- Haddou, K., A. Bendaoud, N. Belaidi, and A. Taleb. 2018. "A large-scale study of hyporheic nitrate dynamics in a semi-arid catchment, the Tafna River, in Northwest Algeria." *Environmental Earth Sciences* 77(13).
- Hale, A.N., G. Noble, K. Piper, K. Garmire, and S.J. Tonsor. 2016. "Controlling for hydrologic connectivity to assess the importance of catchment- and reach-scale factors on macroinvertebrate community structure." *Hydrobiologia* 763(1): 285-299.
- Hale, R.L., and S.E. Godsey. 2019. "Dynamic stream network intermittence explains emergent dissolved organic carbon chemostasis in headwaters." *Hydrological Processes* 33(13): 1926-1936.

Hamdhani, H., D.E. Eppehimer, and M.T. Bogan. 2020. "Release of treated effluent into streams: A

global review of ecological impacts with a consideration of its potential use for environmental flows." *Freshwater Biology* 65(9): 1657-1670.

- Hance, D.J., L.M. Ganio, K.M. Burnett, and J.L. Ebersole. 2016. "Basin-Scale Variation in the Spatial Pattern of Fall Movement of Juvenile Coho Salmon in the West Fork Smith River, Oregon." *Transactions of the American Fisheries Society* 145(5): 1018-1034.
- Hansen, A., *et al.* 2018. "Contribution of wetlands to nitrate removal at the watershed scale." *Nature Geoscience* 11(2).
- Hardenbicker, P., M. Weitere, S. Ritz, F. Schoell, and H. Fischer. 2016. "Longitudinal Plankton Dynamics in the Rivers Rhine and Elbe." *River Research and Applications* 32(6): 1264-1278.
- Harding, D.J., D.T. Roberts, D. Sternberg, T.M. Mullins, M.J. Kennard, and Y.G. Dwyer. 2019. "Flowrelated migration, juvenile dispersal and gonad development in two co-occurring mullet species, *Mugil cephalus* and *Trachystoma petardi*, in a regulated river system." *Marine and Freshwater Research* 70(8): 1105-1115. <u>https://doi.org/10.1071/MF18365</u>.
- Harding, J.N., J.M.S. Harding, and J.D. Reynolds. 2014. "Movers and shakers: nutrient subsidies and benthic disturbance predict biofilm biomass and stable isotope signatures in coastal streams." *Freshwater Biology* 59(7): 1361-1377.
- Harjung, A., F. Sabater, and A. Butturini. 2018. "Hydrological connectivity drives dissolved organic matter processing in an intermittent stream." *Limnologica* 68: 71-81.
- Harman, C.J. 2015. "Time-variable transit time distributions and transport: Theory and application to storage-dependent transport of chloride in a watershed." *Water Resources Research* 51(1): 1-30.
- Harmon, R.S., D.L. Leslie, W.B. Lyons, K.A. Welch, and D.M. McKnight. 2021. "Geochemistry of contrasting stream types, Taylor Valley, Antarctica." *Geological Society of America Bulletin* 133: 425-448.
- Harrington, J.S., A. Mozil, M. Hayashi, and L.R. Bentley. 2018. "Groundwater flow and storage processes in an inactive rock glacier." *Hydrological Processes* 32(20): 3070-3088.
- Harrington, R.A., N.L. Poff, and B.C. Kondratieff. 2016. "Aquatic insect -diversity is not dependent on elevation in Southern Rocky Mountain streams." *Freshwater Biology* 61(2): 195-205.
- Harris, H.E., C.V. Baxter, and J.M. Davis. 2015. "Debris flows amplify effects of wildfire on magnitude and composition of tributary subsidies to mainstem habitats." *Freshwater Science* 34(4): 1457-1467.
- Harvey, J., J. Gomez-Velez, N. Schmadel, D. Scott, E. Boyer, R. Alexander, K. Eng, H. Golden, A. Kettner, C. Konrad, R. Moore, J. Pizzuto, G. Schwarz, C. Soulsby, and J. Choi. 2019. "How Hydrologic Connectivity Regulates Water Quality in River Corridors." *JAWRA Journal of the American Water Resources Association* 0(0).
- Harwood, L.A., and J.A. Babaluk. 2014. "Spawning, Overwintering and Summer Feeding Habitats Used by Anadromous Arctic Char (*Salvelinus alpinus*) of the Hornaday River, Northwest Territories,

Canada." Arctic 67(4): 449-461.

- Hasenmueller, E.A., and H.K. Robinson. 2016. "Hyporheic zone flow disruption from channel linings: Implications for the hydrology and geochemistry of an urban stream, St. Louis, Missouri, USA." *Journal of China University of Geosciences* 27.
- Hassell, N., K.A. Tinker, T. Moore, and E.A. Ottesen. 2018. "Temporal and spatial dynamics in microbial community composition within a temperate stream network." *Environmental Microbiology* 20(10): 3560-3572.
- Hatje, V., R.M.A. Pedreira, C.E. de Rezende, C.A.F. Schettini, G.C. de Souza, D.C. Marin, and P.C. Hackspacher. 2017. "The environmental impacts of one of the largest tailing dam failures worldwide." *Scientific Reports* 7(1).
- Hatley, J., and N.P. Murphy. 2016. "Trouble at the top? Restricted distribution and extreme population isolation in an alpine crustacean assemblage with unexpected lineage diversity." *Freshwater Biology* 61(11): 1891-1904.
- Haxton, T., S. Nienhuis, K. Punt, and T. Baker. 2015. "Assessing Walleye Movement among Reaches of a Large, Fragmented River." North American Journal of Fisheries Management 35(3): 537-550. <u>https://doi.org/10.1080/02755947.2015.1012278</u>.
- Hay, S.E., K.M. Jenkins, and R.T. Kingsford. 2018. "Diverse invertebrate fauna using dry sediment as a refuge in semi-arid and temperate Australian rivers." *Hydrobiologia* 806(1): 95-109.
- Hayakawa, A., Y. Shiraiwa, N. Murakami, Y. Murayama, T. Ishida, and Y. Ishikawa. 2021. "Influence of surface geology on phosphorus export in coastal forested headwater catchments in Akita, Japan." *Soil Science and Plant Nutrition.*
- Hayasaka, D., C.T.E. Ramos-Scharron, and M.C. Lafevor. 2018. "Effects of Forest Roads on Runoff Initiation in Low-Order Ephemeral Streams." *Water Resources Research* 54(11): 8613-8631.
- He, D., Z. Kang, J. Tao, C. Liu, J. Yang, and Y. Chen. 2017. "Hydrologic connectivity driven natural stream fish assemblages in mountain streams in the Yangtze River basin: implications for stream fish conservation in monsoonal East Asia." *Hydrobiologia* 785(1): 185-206.
- He, S., Y. Hao, J. Wu, and J. Lu. 2020. "Estimation of baseflow nitrate loads by a recursive tracing source algorithm in a rainy agricultural watershed." *Hydrological Processes* 34(2): 441-454.
- He, S., J. Soininen, G. Deng, and B. Wang. 2020. "Metacommunity Structure of Stream Insects across Three Hierarchical Spatial scales." *Ecology and Evolution* 10(6): 2874-2884.
- He, S.W., J. Soininen, K. Chen, and B.X. Wang. 2020. "Environmental Factors Override Dispersal-Related Factors in Shaping Diatom and Macroinvertebrate Communities Within Stream Networks in China." *Frontiers in Ecology and Evolution* 8.
- Hedden, S.C., and K.B. Gido. 2020. "Dispersal drives changes in fish community abundance in intermittent stream networks." *River Research and Applications* 36(5): 797-806.

- Hedden, S.C., E.A. Renner, K.B. Gido, and K.J. Hase. 2018. "IMPACTS OF SMALL IMPOUNDMENTS ON AN INTERMITTENT HEADWATER STREAM COMMUNITY." *Southwestern Naturalist* 63(1): 34-41.
- Hegeman, E.E., S.W. Miller, and K.E. Mock. 2014. "Modeling freshwater mussel distribution in relation to biotic and abiotic habitat variables at multiple spatial scales." *Canadian Journal of Fisheries and Aquatic Sciences* 71(10): 1483-1497.
- Heidbuechel, P., and A. Hussner. 2020. "Falling into pieces: In situ fragmentation rates of submerged aquatic plants and the influence of discharge in lowland streams." *Aquatic Botany* 160.
- Heidbuechel, P., M. Sachs, N. Hamzehian, and A. Hussner. 2020. "Go with the flow: Fragment retention patterns shape the vegetative dispersal of aquatic plants in lowland streams." *Freshwater Biology* 65(11): 1936-1949.
- Heidbuechel, P., M. Sachs, N. Stanik, and A. Hussner. 2019. "Species-specific fragmentation rate and colonization potential partly explain the successful spread of aquatic plants in lowland streams." *Hydrobiologia* 843(1): 107-123.
- Heim, K.C., C.D. Arp, M.S. Whitman, and M.S. Wipfli. 2019. "The complementary role of lentic and lotic habitats for Arctic grayling in a complex stream-lake network in Arctic Alaska." *Ecology of Freshwater Fish* 28(2): 209-221.
- Heim, K.C., M.S. Wipfli, M.S. Whitman, C.D. Arp, J. Adams, and J.A. Falke. 2015. "Seasonal cues of Arctic grayling movement in a small Arctic stream: the importance of surface water connectivity." *Environmental Biology of Fishes* 99(1): 49-65.
- Heitmuller, F.T. 2014. "Channel adjustments to historical disturbances along the lower Brazos and Sabine Rivers, south-central USA." *Geomorphology* 204: 382-398.
- Helton, A.M., and E. Bertuzzo. 2018. "How network structure can affect nitrogen removal by streams." *Freshwater Biology* 63(1): 128-140.
- Hemstrom, W., S. van De Wetering, and M. Banks. 2018. "Fish ladder installation across a historical barrier asymmetrically increased conspecific introgressive hybridization between wild winter and summer run steelhead salmon in the Siletz River, Oregon." *Canadian Journal of Fisheries and Aquatic Sciences* 75(9): 1383-1392.
- Hensley, R.T., M.J. Cohen, and J.W. Jawitz. 2018. "Channel Filtering Generates Multifractal Solute Signals." *Geophysical Research Letters* 45(21): 11722-11731.
- Hensley, R.T., M.J. Spangler, L.F. Devito, P.H. Decker, M.J. Cohen, and M.N. Gooseff. 2020."Evaluating spatiotemporal variation in water chemistry of the upper Colorado River using longitudinal profiling." *Hydrological Processes* 34(8): 1782-1793.
- Heritage, G., and N. Entwistle. 2020. "Impacts of River Engineering on River Channel Behaviour: Implications for Managing Downstream Flood Risk." *Water* 12(5).

- Hermoso, V., A.F. Filipe, P. Segurado, and P. Beja. 2018. "Freshwater conservation in a fragmented world: Dealing with barriers in a systematic planning framework." *Aquatic Conservation: Marine and Freshwater Ecosystems* 28(1): 17-25. <u>https://doi.org/10.1002/aqc.2826</u>.
- Herndon, E.M., G. Steinhoefel, A.L.D. Dere, and P.L. Sullivan. 2018. "Perennial flow through convergent hillslopes explains chemodynamic solute behavior in a shale headwater catchment." *Chemical Geology* 493: 413-425.
- Herrera-Perez, J., J.L. Parra, D. Restrepo-Santamaria, and L.F. Jimenez-Segura. 2019. "The Influence of Abiotic Environment and Connectivity on the Distribution of Diversity in an Andean Fish Fluvial Network." *Frontiers in Environmental Science* 7.
- Hill, B.H., C.M. Elonen, A.T. Herlihy, T.M. Jicha, and R.M. Mitchell. 2017. "A synoptic survey of microbial respiration, organic matter decomposition, and carbon efflux in US streams and rivers." *Limnology and Oceanography* 62(1).
- Hill, B.H., R.K. Kolka, F.H. McCormick, and M.A. Starry. 2014. "A synoptic survey of ecosystem services from headwater catchments in the United States." *Ecosystem Services* 7: 106-115.
- Hill, M.J., K.L. Mathers, S. Little, T. Worrall, J. Gunn, and P.J. Wood. 2019. "Ecological effects of a supra-seasonal drought on macroinvertebrate communities differ between near-perennial and ephemeral river reaches." *Aquatic Sciences* 81(4).
- Hine, C.S., H.M. Hagy, M.M. Horath, A.P. Yetter, R.V. Smith, D.F.M. Windsor, M.T. Grocott, and A.M. Milner. 2017. "An inter-catchment assessment of macroinvertebrate communities across groundwater-fed streams within Denali National Park, interior Alaska." *Hydrobiologia* 785(1): 373-384.
- Hipondoka, M.H.T., B.C.W. van Der Waal, M.H. Ndeutapo, and L. Hango. 2018. "Sources of fish in the ephemeral western iishana region of the Cuvelai-Etosha Basin in Angola and Namibia." *African Journal of Aquatic Science* 43(3): 199-214.
- Hitchman, S.M., M.E. Mather, J.M. Smith, and J.S. Fencl. 2018. "Identifying keystone habitats with a mosaic approach can improve biodiversity conservation in disturbed ecosystems." *Global Change Biology* 24(1): 308-321.
- Hoagstrom, C.W. 2015. "Habitat loss and subdivision are additive mechanisms of fish extinction in fragmented rivers." *Global Change Biology* 21(1): 4-5.
- Hocking, D.J., J.T. Thorson, K. O'Neil, and B.H. Letcher. 2018. "A geostatistical state-space model of animal densities for stream networks." *Ecological Applications* 28(7): 1782-1796.
- Hodge, B.W., K.D. Battige, and K.B. Rogers. 2017. "Seasonal and temperature-related movement of Colorado River cutthroat trout in a low-elevation, Rocky Mountain stream." *Ecology and Evolution* 7(7): 2346-2356.
- Hoellein, T.J., A.R. McCormick, J. Hittie, M.G. London, J.W. Scott, and J.J. Kelly. 2017. "Longitudinal patterns of microplastic concentration and bacterial assemblages in surface and benthic habitats of an

urban river." Freshwater Science 36(3): 491-507.

- Hofmann, H., I. Cartwright, and U. Morgenstern. 2018. "Estimating retention potential of headwater catchment using Tritium time series." *Journal of Hydrology* 561: 557-572. <u>https://doi.org/10.1016/j.jhydrol.2018.04.030</u>.
- Hofmeister, K.L., L.E. Nave, P. Drevnick, T. Veverica, R. Knudstrup, K.A. Heckman, S.J. Riha, R.L. Schneider, and M.T. Walter. 2019. "Seasonal dynamics and exports of elements from a first-order stream to a large inland lake in Michigan." *Hydrological Processes* 33(10): 1476-1491.
- Hogg, R.S., J. Zydlewski, and C. Gardner. 2015. "Fish Community Response to a Small-Stream Dam Removal in a Maine Coastal River Tributary." *Transactions of the American Fisheries Society* 144(3): 467-479.
- Hohensinner, S., C. Hauer, and S. Muhar. 2018. "Aquatic Ecology Series River Morphology, Channelization, and Habitat Restoration." Pages 41-65 in *Riverine Ecosystem Management: Science for Governing Towards a Sustainable Future*. S. Schmutz, and J. Sendzimir, eds. Aquatic Ecology Series. Springer, Cham, Switzerland.
- Hohl, A., T. Vaclavik, and R.K. Meentemeyer. 2014. "Go with the flow: geospatial analytics to quantify hydrologic landscape connectivity for passively dispersed microorganisms." *International Journal of Geographical Information Science* 28(8): 1626-1641.
- Holmes, R.W., D.E. Rankin, E. Ballard, and M. Gard. 2016. "Evaluation of Steelhead Passage Flows Using Hydraulic Modeling on an Unregulated Coastal California River." *River Research and Applications* 32(4): 697-710.
- Hood, J.M., L.M. Collis, J.D. Schade, R.A. Stark, and J.C. Finlay. 2021. "Longitudinal patterns and linkages in benthic fine particulate organic matter composition, respiration, and nutrient uptake." *Limnology and Oceanography*.
- Hooke, J.M. 2019. "Extreme sediment fluxes in a dryland flash flood." Scientific Reports 9(1).
- Hooley-Underwood, Z.E., S.B. Stevens, N.R. Salinas, and K.G. Thompson. 2019. "An Intermittent Stream Supports Extensive Spawning of Large-River Native Fishes." *Transactions of the American Fisheries Society* 148(2): 426-441.
- Hooley-Underwood, Z.E., K.G. Thompson, and K.R. Bestgen. 2021. "Razorback Sucker Spawning in an Intermittent Colorado Tributary." *North American Journal of Fisheries Management*.
- Hoppeler, F., R.D.T. Shah, D.N. Shah, S.C. Jähnig, J.D. Tonkin, S. Sharma, and S.U. Pauls. 2016.
 "Environmental and spatial characterisation of an unknown fauna using DNA sequencing an example with Himalayan Hydropsychidae (Insecta: Trichoptera)." *Freshwater Biology* 61(11): 1905-1920.
- Horgby, Å., P.L. Segatto, E. Bertuzzo, R. Lauerwald, B. Lehner, A.J. Ulseth, T.W. Vennemann, and T.J. Battin. 2019. "Unexpected large evasion fluxes of carbon dioxide from turbulent streams draining the world's mountains." *Nature Communications* 10(1).

- Horvath, Z., C.F. Vad, and R. Ptacnik. 2016. "Wind dispersal results in a gradient of dispersal limitation and environmental match among discrete aquatic habitats." *Ecography* 39(8): 726-732.
- Hosen, J.D., K.S. Aho, J.H. Fair, E.D. Kyzivat, S. Matt, J. Morrison, A. Stubbins, L.C. Weber, B. Yoon, and P.A. Raymond. 2021. "Source Switching Maintains Dissolved Organic Matter Chemostasis Across Discharge Levels in a Large Temperate River Network." *Ecosystems* 24(2): 227-247.
- Hosen, J.D., G.H. Allen, G. Amatuli, S. Breitmeyer, M.J. Cohen, B.C. Crump, Y.H. Lu, J.P. Payet, B.A. Poulin, A. Stubbins, B. Yoon, and P.A. Raymond. 2021. "River network travel time is correlated with dissolved organic matter composition in rivers of the contiguous United States." *Hydrological Processes* 35(5).
- Hosen, J.D., A.W. Armstrong, and M.A. Palmer. 2018. "Dissolved organic matter variations in coastal plain wetland watersheds: The integrated role of hydrological connectivity, land use, and seasonality." *Hydrological Processes* 32(11): 1664-1681.
- Hosen, J.D., C.M. Febria, B.C. Crump, and M.A. Palmer. 2017. "Watershed Urbanization Linked to Differences in Stream Bacterial Community Composition." *Frontiers in Microbiology* 8.
- Hrachowitz, M., P. Benettin, B.M. van Breukelen, O. Fovet, N.J.K. Howden, L. Ruiz, Y. van Der Velde, and A.J. Wade. 2016. "Transit times—the link between hydrology and water quality at the catchment scale." WIREs Water 3(5): 629-657. <u>https://doi.org/10.1002/wat2.1155</u>.
- Huang, J., H. Yin, S. Jomaa, M. Rode, and Q. Zhou. 2019. "Limited nitrogen retention in an urban river receiving raw sewage and wastewater treatment plant effluent." *Environmental Science: Processes* & *Impacts* 21(9): 1477-1488.
- Huntsman, B.M., and J.T. Petty. 2014. "Density-dependent regulation of brook trout population dynamics along a core-periphery distribution gradient in a central Appalachian watershed." *PLoS ONE* 9(3).
- Huntsman, B.M., J.T. Petty, S. Sharma, and E.R. Merriam. 2016. "More than a corridor: use of a main stem stream as supplemental foraging habitat by a brook trout metapopulation." *Oecologia* 182(2): 463-473.
- Huttunen, K.L., H. Mykra, A. Huusko, A. Maki-Petays, T. Vehanen, and T. Muotka. 2014. "Testing for temporal coherence across spatial extents: the roles of climate and local factors in regulating stream macroinvertebrate community dynamics." *Ecography* 37(6): 599-608. https://doi.org/10.1111/j.1600-0587.2013.00325.x.
- Hwan, J.L., and S.M. Carlson. 2016. "Fragmentation of an Intermittent Stream During Seasonal Drought: Intra-annual and Interannual Patterns and Biological Consequences." *River Research and Applications* 32(5): 856-870.
- Iannucci, F.M., J. Benes, A. Medvedeff, and W.B. Bowden. 2021. "Biogeochemical responses over 37 years to manipulation of phosphorus concentrations in an Arctic river: The Upper Kuparuk River Experiment." *Hydrological Processes* 35(3): e14075. <u>https://doi.org/10.1002/hyp.14075</u>.
- Ilheu, M., J. da Silva, M. Morais, P. Matono, and J.M. Bernardo. 2020. "Types of dry-season stream

pools: environmental drivers and fish assemblages." Inland Waters 10(4): 516-528.

- Inamdar, S., G. Dhillon, S. Singh, T. Parr, and Z. Qin. 2015. "Particulate nitrogen exports in stream runoff exceed dissolved nitrogen forms during large tropical storms in a temperate, headwater, forested watershed." *Journal of Geophysical Research: Biogeosciences* 120(8): 1548-1566.
- Irving, K., M. Kuemmerlen, J. Kiesel, K. Kakouei, S. Domisch, and S.C. Jähnig. 2018. "A highresolution streamflow and hydrological metrics dataset for ecological modeling using a regression model." *Scientific Data* 5: 180224. <u>https://www.ncbi.nlm.nih.gov/pubmed/30398476</u>.
- Istvanovics, V., M. Honti, A. Kovacs, G. Kocsis, and I. Stier. 2014. "Phytoplankton growth in relation to network topology: time-averaged catchment-scale modelling in a large lowland river." *Freshwater Biology* 59(9): 1856-1871.
- Jacobsen, D., S. Cauvy-Fraunie, P. Andino, R. Espinosa, D. Cueva, and O. Dangles. 2014. "Runoff and the longitudinal distribution of macroinvertebrates in a glacier-fed stream: implications for the effects of global warming." *Freshwater Biology* 59(10): 2038-2050.
- Jaeger, K.L., J.D. Olden, and N.A. Pelland. 2014. "Climate change poised to threaten hydrologic connectivity and endemic fishes in dryland streams." *Proceedings of the National Academy of Sciences* 111(38).
- Jaeger, K.L., N.A. Sutfin, S. Tooth, K. Michaelides, and M. Singer. 2017. "Geomorphology and Sediment Regimes of Intermittent Rivers and Ephemeral Streams." *Intermittent Rivers and Ephemeral Streams: Ecology and Management*: 21-49.
- James, L.A., C. Monohan, and B. Ertis. 2019. "Long-term hydraulic mining sediment budgets: Connectivity as a management tool." *Science of the Total Environment* 651: 2024-2035.
- Jamshidi, R., D. Dragovich, and A.A. Webb. 2014. "Distributed empirical algorithms to estimate catchment scale sediment connectivity and yield in a subtropical region." *Hydrological Processes* 28(4): 2671-2684.
- Jarihani, A.A., J.R. Larsen, J.N. Callow, T.R. McVicar, and K. Johansen. 2015. "Where does all the water go? Partitioning water transmission losses in a data-sparse, multi-channel and low-gradient dryland river system using modelling and remote sensing." *Journal of Hydrology* 529: 1511-1529. https://doi.org/10.1016/j.jhydrol.2015.08.030.
- Jarvie, H.P., A.N. Sharpley, T. Kresse, P.D. Hays, R.J. Williams, S.M. King, and L.G. Berry. 2018. "Coupling High-Frequency Stream Metabolism and Nutrient Monitoring to Explore Biogeochemical Controls on Downstream Nitrate Delivery." *Environmental Science and Technology* 52(23): 13708-13717.
- Jasechko, S., J.W. Kirchner, J.M. Welker, and J.J. McDonnell. 2016. "Substantial proportion of global streamflow less than three months old." *Nature Geoscience* 9(2): 126-129.
- Jensen, C.K., K.J. McGuire, D.L. McLaughlin, and D.T. Scott. 2019. "Quantifying spatiotemporal variation in headwater stream length using flow intermittency sensors." *Environmental Monitoring*
and Assessment 191(4).

- Jiang, R., R. Hatano, R. Hill, K. Kuramochi, T. Jiang, and Y. Zhao. 2015. "Water connectivity in hillslope of upland-riparian zone and the implication for stream nitrate-N export during rain events in an agricultural and forested watershed." *Environmental Earth Sciences* 74(5): 4535-4547. <u>https://doi.org/10.1007/s12665-015-4516-2</u>.
- Jiang, X., X. Xu, M. Tao, Z. Li, S. Zhang, and H. Li. 2021. "The effects of dispersal ability on metacommunity structure of macroinvertebrates in subtropical Chinese high-mountain streams: seasonal shifts in relative contribution of local environment and spatial processes." *Environmental Science and Pollution Research*.
- Johnson, B., E. Smith, J.W. Ackerman, S. Dye, R. Polinsky, E. Somerville, C. Decker, D. Little, G. Pond, and E. D'Amico. 2019. "Spatial Convergence in Major Dissolved Ion Concentrations and Implications of Headwater Mining for Downstream Water Quality." *Journal of the American Water Resources Association* 55(1): 247-258.
- Johnson, T.A.N., S.S. Kaushal, P.M. Mayer, R.M. Smith, and G.M. Sivirichi. 2016. "Nutrient Retention in Restored Streams and Rivers: A Global Review and Synthesis." *Water* 8(4).
- Jolley, J.C., G.S. Silver, J.E. Harris, and T.A. Whitesel. 2018. "Pacific lamprey recolonization of a Pacific Northwest river following dam removal." *River Research and Applications* 34(1): 44-51.
- Jones, C., S.W. Kim, and K. Schilling. 2017. "Use of continuous monitoring to assess stream nitrate flux and transformation patterns." *Environmental Monitoring and Assessment* 189(1).
- Jones, C.S., S.W. Kim, T.F. Wilton, K.E. Schilling, and C.A. Davis. 2018. "Nitrate uptake in an agricultural stream estimated from high-frequency, in-situ sensors." *Environmental Monitoring and Assessment* 190(4).
- Jones, C.S., J.K. Nielsen, K.E. Schilling, and L.J. Weber. 2018. "Iowa stream nitrate and the Gulf of Mexico." *PLoS ONE* 13(4).
- Jones, E.F., N. Griffin, J.E. Kelso, G.T. Carling, M.A. Baker, and Z.T. Aanderud. 2020. "Stream Microbial Community Structured by Trace Elements, Headwater Dispersal, and Large Reservoirs in Sub-Alpine and Urban Ecosystems." *Frontiers in Microbiology* 11.
- Jones, N.E., and R.W. Mackereth. 2016. "Resource subsidies from adfluvial fishes increase stream productivity." *Freshwater Biology* 61(6): 991-1005.
- Jones, N.E., and B.J. Schmidt. 2017. "Tributary effects in rivers: interactions of spatial scale, network structure, and landscape characteristics." *Canadian Journal of Fisheries and Aquatic Sciences* 74(4): 503-510.
- Jones, N.E., and B.J. Schmidt. 2018. "Influence of tributaries on the longitudinal patterns of benthic invertebrate communities." *River Research and Applications* 34(2): 165-173.

Jones, P.E., S. Consuegra, L. Borger, J. Jones, and C. Garcia de Leaniz. 2020. "Impacts of artificial

barriers on the connectivity and dispersal of vascular macrophytes in rivers: A critical review." *Freshwater Biology* 65(6): 1165-1180.

- Jovanovic, T., R.L. Hale, J. Gironas, and A. Mejia. 2019. "Hydrological Functioning of an Evolving Urban Stormwater Network." *Water Resources Research* 55(8): 6517-6533.
- Jung, S.P., Y.J. Kim, and H. Kang. 2014. "Denitrification Rates and Their Controlling Factors in Streams of the Han River Basin with Different Land-Use Patterns." *Pedosphere* 24(4): 516-528.
- Jyrkankallio-Mikkola, J., M. Siljander, V. Heikinheimo, P. Pellikka, and J. Soininen. 2018. "Tropical stream diatom communities - The importance of headwater streams for regional diversity." *Ecological Indicators* 95: 183-193.
- Kaase, C.T., and J.A. Kupfer. 2016. "Sedimentation patterns across a Coastal Plain floodplain: The importance of hydrogeomorphic influences and cross-floodplain connectivity." *Geomorphology* 269: 43-55.
- Kaiser, K., M. Canedo-Oropeza, R. McMahon, and R.M.W. Amon. 2017. "Origins and transformations of dissolved organic matter in large Arctic rivers." *Scientific Reports* 7(1).
- Kamjunke, N., N. Hertkorn, M. Harir, P. Schmitt-Kopplin, C. Griebler, M. Brauns, W. von Tümpling, M. Weitere, and P. Herzsprung. 2019. "Molecular change of dissolved organic matter and patterns of bacterial activity in a stream along a land-use gradient." *Water Research* 164.
- Kamjunke, N., M.R. Oosterwoud, P. Herzsprung, and J. Tittel. 2016. "Bacterial production and their role in the removal of dissolved organic matter from tributaries of drinking water reservoirs." *Science of the Total Environment* 548: 51-59.
- Kan, J. 2018. "Storm Events Restructured Bacterial Community and Their Biogeochemical Potentials." Journal of Geophysical Research: Biogeosciences 123(7): 2257-2269.
- Kanno, Y., B.H. Letcher, J.A. Coombs, K.H. Nislow, and A.R. Whiteley. 2014. "Linking movement and reproductive history of brook trout to assess habitat connectivity in a heterogeneous stream network." *Freshwater Biology* 59(1): 142-154.
- Karaouzas, I., C. Theodoropoulos, A. Vourka, K. Gritzalis, and N.T. Skoulikidis. 2019. "Stream invertebrate communities are primarily shaped by hydrological factors and ultimately fine-tuned by local habitat conditions." *Science of the Total Environment* 665: 290-299.
- Karki, S., M.J. Stewardson, J.A. Webb, K. Fowler, G.R. Kattel, and D.J. Gilvear. 2021. "Does the topology of the river network influence the delivery of riverine ecosystem services?" *River Research and Applications* 37(2): 256-269.
- Kärnä, O.M., M. Grönroos, H. Antikainen, J. Hjort, J. Ilmonen, L. Paasivirta, and J. Heino. 2015.
 "Inferring the effects of potential dispersal routes on the metacommunity structure of stream insects: as the crow flies, as the fish swims or as the fox runs?" *Journal of Animal Ecology* 84(5): 1342-1353.

- Karwan, D.L., J.E. Pizzuto, R. Aalto, J. Marquard, A. Harpold, K. Skalak, A. Benthem, D.F. Levia, C.M. Siegert, and A.K. Aufdenkampe. 2018. "Direct Channel Precipitation and Storm Characteristics Influence Short-Term Fallout Radionuclide Assessment of Sediment Source." *Water Resources Research* 54(7): 4579-4594.
- Kasurinen, V., K. Alfredsen, A. Ojala, J. Pumpanen, G.A. Weyhenmeyer, M.N. Futter, H. Laudon, and F. Berninger. 2016. "Modeling nonlinear responses of DOC transport in boreal catchments in Sweden." Water Resources Research 52(7): 4970-4989.
- Kaus, A., S. Michalski, B. Hänfling, D. Karthe, D. Borchardt, and W. Durka. 2019. "Fish conservation in the land of steppe and sky: Evolutionarily significant units of threatened salmonid species in Mongolia mirror major river basins." *Ecology and Evolution* 9(6): 3416-3433.
- Kavanagh, J.A., and S.S.C. Harrison. 2014. "The contribution of a drainage network to the spatial and temporal patterns of macroinvertebrate diversity across an agricultural headwater catchment." *Biology and Environment* 114(3): 181-197.
- Kaylor, M.J., C. Justice, J.B. Armstrong, B.A. Staton, L.A. Burns, E. Sedell, and S.M. White. 2021. "Temperature, emergence phenology and consumption drive seasonal shifts in fish growth and production across riverscapes." *Journal of Animal Ecology*.
- Keck, F., A. Franc, and M. Kahlert. 2018. "Disentangling the processes driving the biogeography of freshwater diatoms: A multiscale approach." *Journal of Biogeography* 45(7): 1582-1592.
- Keesstra, S.D., J. Davis, R.H. Masselink, J. Casali, E.T.H.M. Peeters, and R. Dijksma. 2019. "Coupling hysteresis analysis with sediment and hydrological connectivity in three agricultural catchments in Navarre, Spain." *Journal of Soils and Sediments* 19(3): 1598-1612.
- Kelly, B., K.E. Smokorowski, and M. Power. 2017. "Growth, Condition and Survival of Three Forage Fish Species Exposed to Two Different Experimental Hydropeaking Regimes in a Regulated River." *River Research and Applications* 33(1): 50-62.
- Kelson, S.J., and S.M. Carlson. 2019. "Do precipitation extremes drive growth and migration timing of a Pacific salmonid fish in Mediterranean-climate streams?" *Ecosphere* 10(3).
- Kelson, S.J., A.R. Kapuscinski, D. Timmins, and W.R. Ardren. 2015. "Fine-scale genetic structure of brook trout in a dendritic stream network." *Conservation Genetics* 16(1): 31-42.
- Kerezsy, A., K. Gido, M.F. Magalhaes, and P.H. Skelton. 2017. "The Biota of Intermittent Rivers and Ephemeral Streams: Fishes." *Intermittent Rivers and Ephemeral Streams: Ecology and Management*: 273-298.
- Kerr, J.R., P. Karageorgopoulos, and P.S. Kemp. 2015. "Efficacy of a side-mounted vertically oriented bristle pass for improving upstream passage of European eel (*Anguilla anguilla*) and river lamprey (*Lampetra fluviatilis*) at an experimental Crump weir." *Ecological Engineering* 85: 121-131.
- Kiaghadi, A., A. Govindarajan, R.S. Sobel, and H.S. Rifai. 2020. "Environmental damage associated with severe hydrologic events: a LiDAR-based geospatial modeling approach." *Natural Hazards* 103(3):

2711-2729.

- Kim, H., P.B.E. Sandersen, R. Jakobsen, A.J. Kallesae, N. Claes, G. Blicher-Mathiesen, N. Foged, J. Aamand, and B. Hansen. 2021. "A 3D hydrogeochemistry model of nitrate transport and fate in a glacial sediment catchment: A first step toward a numerical model." *Science of the Total Environment* 776.
- Kim, M.Y., and S.W. Lee. 2021. "Regression Tree Analysis for Stream Biological Indicators Considering Spatial Autocorrelation." *International Journal of Environmental Research and Public Health* 18(10).
- Kincaid, D.W., E.C. Seybold, E.C. Adair, W.B. Bowden, J.N. Perdrial, M.C.H. Vaughan, and A.W. Schroth. 2020. "Land Use and Season Influence Event-Scale Nitrate and Soluble Reactive Phosphorus Exports and Export Stoichiometry from Headwater Catchments." *Water Resources Research* 56(10).
- King, K.B.S., M.T. Bremigan, D. Infante, and K.S. Cheruvelil. 2021. "Surface water connectivity affects lake and stream fish species richness and composition." *Canadian Journal of Fisheries and Aquatic Sciences* 78(4): 433-443.
- King, K.W., M.R. Williams, M.L. Macrae, N.R. Fausey, J. Frankenberger, D.R. Smith, P.J.A. Kleinman, and L.C. Brown. 2015. "Phosphorus Transport in Agricultural Subsurface Drainage: A Review." *Journal of Environmental Quality* 44(2): 467-485.
- King, M.D., R.B. Bryant, L.S. Saporito, A.R. Buda, A.L. Allen, L.A. Hughes, F.M. Hashem, P.J. Kleinman, and E.B. May. 2017. "Urea Release by Intermittently Saturated Sediments from a Coastal Agricultural Landscape." *Journal of Environmental Quality* 46(2): 302-310.
- Kleine, L., D. Tetzlaff, A. Smith, T. Goldhammer, and C. Soulsby. 2021. "Using isotopes to understand landscape-scale connectivity in a groundwater-dominated, lowland catchment under drought conditions." *Hydrological Processes* 35(5).
- Kleyheeg, E., W. Fiedler, K. Safi, J. Waldenstrom, M. Wikelski, and M.L. van Toor. 2019. "A Comprehensive Model for the Quantitative Estimation of Seed Dispersal by Migratory Mallards." *Frontiers in Ecology and Evolution* 7.
- Kleyheeg, E., and C.H.A. van Leeuwen. 2015. "Regurgitation by waterfowl: An overlooked mechanism for long-distance dispersal of wetland plant seeds." *Aquatic Botany* 127: 1-5.
- Koenig, L.E., A.M. Helton, P. Savoy, E. Bertuzzo, J.B. Heffernan, and E.S. Bernhardt. 2019. "Emergent productivity regimes of river networks." *Limnology and Oceanography: Letters* 4(5): 173-181.
- Koenig, L.E., C. Song, W.M. Wollheim, J. Ruegg, and W.H. McDowell. 2017. "Nitrification increases nitrogen export from a tropical river network." *Freshwater Science* 36(4): 698-712.
- Koizumi, I., Y. Tanaka, and Y. Kanazawa. 2017. "Mass immigration of juvenile fishes into a small, oncedried tributary demonstrates the importance of remnant tributaries as wintering habitats." *Ichthyological Research* 64(3): 353-356.

- Kolmakova, O.V., M.I. Gladyshev, A.S. Rozanov, S.E. Peltek, and M.Y. Trusova. 2014. "Spatial biodiversity of bacteria along the largest Arctic river determined by next-generation sequencing." *FEMS Microbiology Ecology* 89(2): 442-450.
- Kominoski, J.S., A. Ruhí, M.M. Hagler, K. Petersen, J.L. Sabo, T. Sinha, A. Sankarasubramanian, and J.D. Olden. 2018. "Patterns and drivers of fish extirpations in rivers of the American Southwest and Southeast." *Global Change Biology* 24(3): 1175-1185.
- Koster, W.M., D.R. Dawson, C. Liu, P.D. Moloney, D.A. Crook, and J.R. Thomson. 2017. "Influence of streamflow on spawning-related movements of golden perch *Macquaria ambigua* in south-eastern Australia." *Journal of Fish Biology* 90(1): 93-108. <u>https://doi.org/10.1111/jfb.13160</u>.
- Koster, W.M., D.R. Dawson, D.J. O'Mahony, P.D. Moloney, and D.A. Crook. 2014. "Timing, frequency and environmental conditions associated with mainstem-tributary movement by a lowland river fish, golden perch (*Macquaria ambigua*)." *PLoS ONE* 9(5); e96044. https://doi.org/10.1371/journal.pone.00960.
- Kotzian, C.B., M.M. Pires, and L.U. Hepp. 2020. "Effects of spatial distances on the assemblage dissimilarity of macroinvertebrates with different dispersal pathways and abilities in southern Brazilian streams." *Ecological Research* 35(5): 826-837.
- Kramer, N., and E. Wohl. 2017. "Rules of the road: A qualitative and quantitative synthesis of large wood transport through drainage networks." *Geomorphology* 279: 74-97.
- Kreiling, R.M., W.B. Richardson, L.A. Bartsch, M.C. Thoms, and V.G. Christensen. 2019."Denitrification in the river network of a mixed land use watershed: unpacking the complexities." *Biogeochemistry* 143(3): 327-346.
- Kuglerová, L., R. Jansson, R.A. Sponseller, H. Laudon, and B. Malm-Renöfält. 2015. "Local and regional processes determine plant species richness in a river-network metacommunity." *Ecology* 96(2): 381-391.
- Kurthen, A.L., F.Z. He, X. Dong, A. Maasri, N.C. Wu, Q. Cai, and S.C. Jahnig. 2020. "Metacommunity Structures of Macroinvertebrates and Diatoms in High Mountain Streams, Yunnan, China." *Frontiers in Ecology and Evolution* 8.
- Kurz, M.J., J.F. Drummond, E. Marti, J.P. Zarnetske, J. Lee-Cullin, M.J. Klaar, S. Folegot, T. Keller, A.S. Ward, J.H. Fleckenstein, T. Datry, D.M. Hannah, and S. Krause. 2017. "Impacts of water level on metabolism and transient storage in vegetated lowland rivers: Insights from a mesocosm study." *Journal of Geophysical Research: Biogeosciences* 122(3): 628-644.
- Lagarde, R., N. Teichert, L. Faivre, H. Grondin, H. Magalon, A. Pirog, P. Valade, and D. Ponton. 2018."Artificial daily fluctuations of river discharge affect the larval drift and survival of a tropical amphidromous goby." *Ecology of Freshwater Fish* 27(3): 646-659.
- Lai, J.T., and A.M. Anders. 2018. "Modeled Postglacial Landscape Evolution at the Southern Margin of the Laurentide Ice Sheet: Hydrological Connection of Uplands Controls the Pace and Style of Fluvial Network Expansion." *Journal of Geophysical Research-Earth Surface* 123(5): 967-984.

- Lancaster, J., and B.J. Downes. 2017. "A landscape-scale field experiment reveals the importance of dispersal in a resource-limited metacommunity." *Ecology* 98(2): 565-575.
- Landguth, E.L., C.C. Muhlfeld, R.S. Waples, L. Jones, W.H. Lowe, D. Whited, J. Lucotch, H. Neville, and G. Luikart. 2014. "Combining demographic and genetic factors to assess population vulnerability in stream species." *Ecological Applications* 24(6): 1505-1524.
- Lannergard, E.E., O. Agstam-Norlin, B.J. Huser, S. Sandstrom, J. Rakovic, and M.N. Futter. 2020. "New Insights Into Legacy Phosphorus From Fractionation of Streambed Sediment." *Journal of Geophysical Research: Biogeosciences* 125(9).
- Lapides, D.A., C.D. Leclerc, H. Moidu, D.N. Dralle, and W.J. Hahm. 2021. "Variability of stream extents controlled by flow regime and network hydraulic scaling." *Hydrological Processes* 35(3).
- Larsen, L.G., and C. Woelfle-Erskine. 2018. "Groundwater Is Key to Salmonid Persistence and Recruitment in Intermittent Mediterranean-Climate Streams." *Water Resources Research* 54(11): 8909-8930.
- Larsen, S., M.C. Bruno, I.P. Vaughan, and G. Zolezzi. 2019. "Testing the River Continuum Concept with geostatistical stream-network models." *Ecological Complexity* 39: 100773. <u>https://doi.org/10.1016/j.ecocom.2019.100773</u>.
- Larsen, S., L. Comte, A. Filipa Filipe, M.-J. Fortin, C. Jacquet, R. Ryser, P.A. Tedesco, U. Brose, T. Erős, X. Giam, K. Irving, A. Ruhi, S. Sharma, and J.D. Olden. 2021. "The geography of metapopulation synchrony in dendritic river networks." *Ecology Letters* 24(4): 791-801. <u>https://doi.org/10.1111/ele.13699</u>.
- Larson, J.H., J.M. Vallazza, and B.C. Knights. 2019. "Estimating the degree to which distance and temperature differences drive changes in fish community composition over time in the upper Mississippi River." *PLoS ONE* 14(12).
- Laub, B.G., G.P. Thiede, W.W. Macfarlane, and P. Budy. 2018. "Evaluating the Conservation Potential of Tributaries for Native Fishes in the Upper Colorado River Basin." *Fisheries* 43(4): 194-206.
- Laudon, H., and R.A. Sponseller. 2018. "How landscape organization and scale shape catchment hydrology and biogeochemistry: insights from a long-term catchment study." *WIREs Water* 5(2).
- Laughlin, T.W., G.W. Whitledge, D.C. Oliver, and N.P. Rude. 2016. "Recruitment Sources of Channel and Blue Catfishes Inhabiting the Middle Mississippi River." *River Research and Applications* 32(8): 1808-1818.
- Lawson, K.M., and C.E. Johnston. 2016. "The role of flow dependency and water availability in fish assemblage homogenization in tributaries of the Chattahoochee River, Alabama, USA." *Ecology of Freshwater Fish* 25(4): 631-641.
- Lazzaro, G., C. Soulsby, D. Tetzlaff, and G. Botter. 2017. "A probabilistic approach to quantifying hydrologic thresholds regulating migration of adult Atlantic salmon into spawning streams." *Water Resources Research* 53(3): 2264-2277.

- Leboucher, T., W.R. Budnick, S. Passy, S. Boutry, A. Jannoneau, J. Soininen, W. Vyverman, and J. Tison-Rosebery. 2019. "Diatom beta-diversity in streams increases with spatial scale and decreases with nutrient enrichment across regional to sub-continental scales." *Journal of Biogeography* 46(4): 734-744.
- Lee, K.T., and J.K. Huang. 2016. "Influence of storm magnitude and watershed size on runoff nonlinearity." *Journal of Earth System Science* 125(4): 777-794.
- Lee, K.Y., R. van Geldern, and J.A.C. Barth. 2021. "Extreme gradients in CO2 losses downstream of karstic springs." *Science of the Total Environment* 778.
- Lee, M., C. Jung, E. Shevliakova, S. Malyshev, H. Han, S. Kim, K. Kim, and P.R. Jaffe. 2018. "Control of Nitrogen Exports From River Basins to the Coastal Ocean: Evaluation of Basin Management Strategies for Reducing Coastal Hypoxia." *Journal of Geophysical Research: Biogeosciences* 123(10): 3111-3123.
- Lehtinen, R.M., and J.R. Witter. 2014. "DETECTING FROGS AND DETECTING DECLINES: AN EXAMINATION OF OCCUPANCY AND TURNOVER PATTERNS AT THE RANGE EDGE OF BLANCHARD'S CRICKET FROG (ACRIS BLANCHARDI)." *Herpetological Conservation and Biology* 9(3): 502-515.
- Leibowitz, S.G., P.J. Wigington, K.A. Schofield, L.C. Alexander, M.K. Vanderhoof, and H.E. Golden. 2018. "Connectivity of Streams and Wetlands to Downstream Waters: An Integrated Systems Framework." *JAWRA Journal of the American Water Resources Association* 54(2): 298-322.
- Leigh, C., A.J. Boulton, J.L. Courtwright, K.M. Fritz, C.L. May, R.H. Walker, and T. Datry. 2016. "Ecological research and management of intermittent rivers: an historical review and future directions." *Freshwater Biology* 61(8): 1181-1199.
- Lemoine, M.T., L.A. Eby, C.G. Clancy, L.G. Nyce, M.J. Jakober, and D.J. Isaak. 2020. "Landscape resistance mediates native fish species distribution shifts and vulnerability to climate change in riverscapes." *Global Change Biology* 26(10): 5492-5508.
- Leuthold, S.J., S.A. Ewing, R.A. Payn, F.R. Miller, and S.G. Custer. 2021. "Seasonal connections between meteoric water and streamflow generation along a mountain headwater stream." *Hydrological Processes* 35(2). https://doi.org/10.1002/hyp.14029.
- Levesque, D., C. Hudon, P.M.A. James, and P. Legendre. 2017. "Environmental factors structuring benthic primary producers at different spatial scales in the St. Lawrence River (Canada)." *Aquatic Sciences* 79(2): 345-356.
- Levine, R., and G.A. Meyer. 2014. "Beaver dams and channel sediment dynamics on Odell Creek, Centennial Valley, Montana, USA." *Geomorphology* 205: 51-64.
- Li, Q., Y. Li, M. Jiang, Y. Wang, D. Xu, L. Chu, and Y. Yan. 2021. "Effects of low-head dams on fish assemblages in subtropical streams: Context dependence on local habitat and landscape conditions." *Ecological Indicators* 121.

- Li, R., Y. Wang, J. Qian, and Y. Fu. 2020. "Evaluating transient storage and associated nutrient retention in a nutrient-rich headwater stream: a case study in Lake Chaohu Basin, China." *Environmental Science and Pollution Research* 27(6): 6066-6077.
- Li, Z., X. Chen, X. Jiang, J.D. Tonkin, Z. Xie, and J. Heino. 2021. "Distance decay of benthic macroinvertebrate communities in a mountain river network: Do dispersal routes and dispersal ability matter?" *Science of the Total Environment* 758.
- Li, Z., P. Gao, and Y. You. 2018. "Characterizing Hydrological Connectivity of Artificial Ditches in Zoige Peatlands of Qinghai-Tibet Plateau." *Water* 10(10).
- Li, Z., J. Heino, X. Chen, Z. Liu, X. Meng, X. Jiang, Y. Ge, J. Chen, and Z. Xie. 2021. "Understanding macroinvertebrate metacommunity organization using a nested study design across a mountainous river network." *Ecological Indicators* 121.
- Li, Z., J. Heino, Z. Liu, X. Meng, X. Chen, Y. Ge, and Z. Xie. 2021. "The drivers of multiple dimensions of stream macroinvertebrate beta diversity across a large montane landscape." *Limnology and Oceanography* 66(1): 226-236.
- Li, Z., Y. Xing, Z. Liu, X. Chen, X. Jiang, Z. Xie, and J. Heino. 2020. "Seasonal changes in metacommunity assembly mechanisms of benthic macroinvertebrates in a subtropical river basin." *Science of the Total Environment* 729.
- Lin, A., M.S. Krause, E. Nkonya, B.L. Griess, A.S. Wymore, B.M. Rodriguez-Cardona, A. Herreid, and W.H. McDowell. 2019. "LINX I and II: Lessons Learned and Emerging Questions." *Frontiers in Environmental Science* 7.
- Lin, J., J.K. Bohlke, S. Huang, M. Gonzalez-Meler, and N.C. Sturchio. 2019. "Seasonality of nitrate sources and isotopic composition in the Upper Illinois River." *Journal of Hydrology* 568: 849-861.
- Lin, L., L. Davis, S. Cohen, E. Chapman, and J.W. Edmonds. 2016. "The influence of geomorphic unit spatial distribution on nitrogen retention and removal in a large river." *Ecological Modelling* 336: 26-35.
- Lisenby, P.E., and K.A. Fryirs. 2017. "Sedimentologically significant tributaries: catchment-scale controls on sediment (dis) connectivity in the Lockyer Valley, SEQ, Australia." *Earth Surface Processes and Landforms* 42(10): 1493-1504.
- Lisi, P.J., and D.E. Schindler. 2015. "Wind-driven upwelling in lakes destabilizes thermal regimes of downstream rivers." *Limnology and Oceanography* 60(1): 169-180.
- Liu, X., R. Wu, X. Chen, Y. Zhou, L. Yang, S. Ouyang, and X. Wu. 2020. "Effects of dams and their environmental impacts on the genetic diversity and connectivity of freshwater mussel populations in Poyang Lake Basin, China." *Freshwater Biology* 65(2): 264-277.
- Liu, Z.Y., T.T. Zhou, Y.D. Cui, Z.F. Li, W.M. Wang, Y.S. Chen, and Z.C. Xie. 2021. "Environmental filtering and spatial processes equally contributed to macroinvertebrate metacommunity dynamics in the highly urbanized river networks in Shenzhen, South China." *Ecological Processes* 10(1).

- Llang, Z., F. Wu, W. Yang, Y. Zhang, J. Wu, and F. Yang. 2019. "Headwater Streams Contain Amounts of Heavy Metal in an Alpine Forest in the Upper Reaches of the Yangtze River." *Sains Malaysiana* 48(8): 1565-1574.
- Lois, S., D.E. Cowley, A. Outeiro, E. San Miguel, R. Amaro, and P. Ondina. 2015. "Spatial extent of biotic interactions affects species distribution and abundance in river networks: the freshwater pearl mussel and its hosts." *Journal of Biogeography* 42(2): 229-240. <u>https://doi.org/10.1111/jbi.12435</u>.
- Looman, A., D.T. Maher, E. Pendall, A. Bass, and I.R. Santos. 2017. "The carbon dioxide evasion cycle of an intermittent first-order stream: contrasting water-air and soil-air exchange." *Biogeochemistry* 132: 87-102.
- Looman, A., I.R. Santos, D.R. Tait, J.R. Webb, C.A. Sullivan, and D.T. Maher. 2016. "Carbon cycling and exports over diel and flood-recovery timescales in a subtropical rainforest headwater stream." *Science of the Total Environment* 550: 645-657.
- López-Delgado, E.O., K.O. Winemiller, and F.A. Villa-Navarro. 2020. "Local environmental factors influence beta-diversity patterns of tropical fish assemblages more than spatial factors." *Ecology* 101(2).
- Lynch, L.M., N.A. Sutfin, T.S. Fegel, C.M. Boot, T.P. Covino, and M.D. Wallenstein. 2019. "River channel connectivity shifts metabolite composition and dissolved organic matter chemistry." *Nature Communications* 10(1): 459.
- Lyon, S.W., S.W. Ploum, Y. van Der Velde, G. Rocher-Ros, C.M. Morth, and R. Giesler. 2018. "Lessons learned from monitoring the stable water isotopic variability in precipitation and streamflow across a snow-dominated subarctic catchment." *Arctic, Antarctic, and Alpine Research* 50(1): e1454778. <u>https://doi.org/10.1080/15230430.2018.1454778</u>.
- Ma, C., Y. Shen, D. Bearup, W.F. Fagan, and J. Liao. 2020. "Spatial variation in branch size promotes metapopulation persistence in dendritic river networks." *Freshwater Biology* 65(3): 426-434.
- Maavara, T., Q. Chen, K. Van Meter, L.E. Brown, J.Y. Zhang, J.R. Ni, and C. Zarfl. 2020. "River dam impacts on biogeochemical cycling." *Nature Reviews Earth & Environment* 1(2): 103-116.
- Magdaleno, F., C. Donadio, and G.M. Kondolf. 2018. "30-year response to damming of a Mediterranean river in California, USA." *Physical Geography* 39(3): 197-215.
- Mahlum, S., D. Kehler, D. Cote, Y.F. Wiersma, and L. Stanfield. 2014. "Assessing the biological relevance of aquatic connectivity to stream fish communities." *Canadian Journal of Fisheries and Aquatic Sciences* 71(12): 1852-1863.
- Mahoney, D.T., J. Fox, N. Al-Aamery, and E. Clare. 2020. "Integrating connectivity theory within watershed modelling part II: Application and evaluating structural and functional connectivity." *Science of the Total Environment* 740.
- Major, J.J., D. Bertin, T.C. Pierson, A. Amigo, A. Iroume, H. Ulloa, and J. Castro. 2016. "Extraordinary sediment delivery and rapid geomorphic response following the 2008-2009 eruption of Chaiten

Volcano, Chile." Water Resources Research 52(7): 5075-5094.

- Mallard, J., B. McGlynn, and T. Covino. 2014. "Lateral inflows, stream-groundwater exchange, and network geometry influence stream water composition." *Water Resources Research* 50(6): 4603-4623.
- Razeng, E., A. Morán-Ordóñez, J.B. Box, R. Thompson, J. Davis, and P. Sunnucks. 2016. "A potential role for overland dispersal in shaping aquatic invertebrate communities in arid regions." *Freshwater Biology* 61(5): 745-757.
- Manis, E., T.V. Royer, L.T. Johnson, and L.G. Leff. 2014. "Denitrification in Agriculturally Impacted Streams: Seasonal Changes in Structure and Function of the Bacterial Community." *PLoS ONE* 9(8).
- Manning, A., J.P. Julian, and M.W. Doyle. 2020. "Riparian vegetation as an indicator of stream channel presence and connectivity in arid environments." *Journal of Arid Environments* 178.
- Manning, D.W.P., A. Rosemond, J.P. Benstead, P.M. Bumpers, and J.S. Kominoski. 2020. "Transport of N and P in US streams and rivers differs with land use and between dissolved and particulate forms." *Ecological Applications* 30(6).
- Mao, F., K.S. Richards, M. Toland, Y. Shi, D.M. Hannah, and S. Krause. 2019. "Rivervis: A tool for visualising river ecosystems." *Computers and Geosciences* 123: 59-64.
- Marcarelli, A.M., A.A. Coble, K.M. Meingast, E.S. Kane, C.N. Brooks, I. Buffam, S.A. Green, C.J. Huckins, D. Toczydlowski, and R. Stottlemyer. 2019. "Of Small Streams and Great Lakes: Integrating Tributaries to Understand the Ecology and Biogeochemistry of Lake Superior." *Journal* of the American Water Resources Association 55(2): 442-458.
- Marcé, R., S. Daniel von, R. Aguilera, E. Martí, and S. Bernal. 2018. "Contribution of Hydrologic Opportunity and Biogeochemical Reactivity to the Variability of Nutrient Retention in River Networks." *Global Biogeochemical Cycles* 32.
- Marshall, J.C., N. Menke, D.A. Crook, J.S. Lobegeiger, S.R. Balcombe, J.A. Huey, J.H. Fawcett, N.R. Bond, A.H. Starkey, D. Sternberg, S. Linke, and A.H. Arthington. 2016. "Go with the flow: the movement behaviour of fish from isolated waterhole refugia during connecting flow events in an intermittent dryland river." *Freshwater Biology* 61(8): 1242-1258.
- Marteau, B., R.J. Batalla, D. Vericat, and C. Gibbins. 2017. "The importance of a small ephemeral tributary for fine sediment dynamics in a main-stem river." *River Research and Applications* 33(10): 1564-1574.
- Marteau, B., C. Gibbins, D. Vericat, and R.J. Batalla. 2020. "Geomorphological response to system-scale river rehabilitation I: Sediment supply from a reconnected tributary." *River Research and Applications* 36(8): 1488-1503.
- Marteau, B., C. Gibbins, D. Vericat, and R.J. Batalla. 2020. "Geomorphological response to system-scale river rehabilitation II: Main-stem channel adjustments following reconnection of an ephemeral tributary." *River Research and Applications* 36(8): 1472-1487.

- Martin, K.R., P.T.J. Johnson, J. Bowerman, and J. Li. 2020. "Biogeography of the freshwater gastropod, *Planorbella trivolvis*, in the western United States." *PLoS ONE* 15(7): e0235989.
- Marzadri, A., G. Amatulli, D. Tonina, A. Bellin, L.Q. Shen, G.H. Allen, and P.A. Raymond. 2021. "Global riverine nitrous oxide emissions: The role of small streams and large rivers." *Science of the Total Environment* 776.
- Maxwell, M.F., E. Secretti, M.M. Pires, and C.B. Kotzian. 2021. "The relative importance of spatial and environmental processes in the assembly of larval Chironomidae (Insecta, Diptera) communities along a transition landscape in southern Brazilian streams." *Limnology* 22(2): 259-268.
- Mazgareanu, I., P.M. Biron, and T. Buffin-Belanger. 2020. "A fuzzy GIS model to determine confluence morphological sensitivity to tributary inputs at the watershed scale." *Geomorphology* 357.
- Mazzucco, R., N. Tuyen Van, D.-H. Kim, T.-S. Chon, and U. Dieckmann. 2015. "Adaptation of aquatic insects to the current flow in streams." *Ecological Modelling* 309: 143-152. <u>https://doi.org/10.1016/j.ecolmodel.2015.04.019</u>.
- McCartney, M.A., and S. Mallez. 2018. "The role of waterway connections and downstream drift of veliger larvae in the expanding invasion of inland lakes by zebra mussels in Minnesota, USA." *Aquatic Invasions* 13(3): 393-408.
- McDowell, R.W., N. Cox, and T.H. Snelder. 2017. "Assessing the Yield and Load of Contaminants with Stream Order: Would Policy Requiring Livestock to Be Fenced Out of High-Order Streams Decrease Catchment Contaminant Loads?" *Journal of Environmental Quality* 46(5): 1038-1047.
- McFarland, J.J., M.S. Wipfli, and M.S. Whitman. 2018. "Trophic pathways supporting Arctic grayling in a small stream on the Arctic Coastal Plain, Alaska." *Ecology of Freshwater Fish* 27(1): 184-197.
- McGill, L.M., J.R. Brooks, and E.A. Steel. 2021. "Spatiotemporal dynamics of water sources in a mountain river basin inferred through δ2H and δ18O of water." *Hydrological Processes* 35(3).
- McGill, L.M., E.A. Steel, J.R. Brooks, R.T. Edwards, and A.H. Fullerton. 2020. "Elevation and spatial structure explain most surface-water isotopic variation across five Pacific Coast basins." *Journal of Hydrology* 583.
- McGuire, K.J., C.E. Torgersen, G.E. Likens, D.C. Buso, W.H. Lowe, and S.W. Bailey. 2014. "Network analysis reveals multiscale controls on streamwater chemistry." *Proceedings of the National Academy of Sciences* 111(19).
- McIntyre, P.B., C.R. Liermann, E. Childress, E.J. Hamann, J.D. Hogan, S.R. Januchowski-Hartley, A.A. Koning, T.M. Neeson, D.L. Oele, and B.M. Pracheil. 2016. "Conservation of migratory fishes in freshwater ecosystems." Pages 324-360 in *Conservation of Freshwater Fishes*. G.P. Closs, M. Krkosek, and J.D. Olden, eds. Conservation Biology Series. Cambridge University Press, Cambridge.

McMahon, T.A., and R.J. Nathan. 2021. "Baseflow and transmission loss: A review." WIREs Water 8(4).

- Mee, J.A., G.L. Robins, and J.R. Post. 2018. "Patterns of fish species distributions replicated across three parallel rivers suggest biotic zonation in response to a longitudinal temperature gradient." *Ecology of Freshwater Fish* 27(1): 44-61.
- Meingast, K.M., E.S. Kane, A.A. Coble, A.M. Marcarelli, and D. Toczydlowski. 2020. "Climate, snowmelt dynamics and atmospheric deposition interact to control dissolved organic carbon export from a northern forest stream over 26 years." *Environmental Research Letters* 15(10): 104034. <u>https://doi.org/10.1088/1748-9326/ab9c4e</u>.
- Meißner, T., M. Schütt, B. Sures, and C.K. Feld. 2018. "Riverine regime shifts through reservoir dams reveal options for ecological management." *Ecological Applications* 28(7): 1897-1908.
- Meitzen, K.M., J.A. Kupfer, and P. Gao. 2018. "Modeling hydrologic connectivity and virtual fish movement across a large Southeastern floodplain, USA." *Aquatic Sciences* 80(1).
- Mejia, F.H., A.K. Fremier, J.R. Benjamin, J.R. Bellmore, A.Z. Grimm, G.A. Watson, and M. Newsom. 2019. "Stream metabolism increases with drainage area and peaks asynchronously across a stream network." *Aquatic Sciences* 81(1).
- Mejia, F.H., C.E. Torgersen, E.K. Berntsen, J.R. Maroney, J.M. Connor, A.H. Fullerton, J.L. Ebersole, and M.S. Lorang. 2020. "Longitudinal, lateral, vertical and temporal thermal heterogeneity in a large impounded river: implications for cold-water refuges." *Remote Sensing* 12(9): 1-1386.
- Mengistu, S.G., I.F. Creed, K.L. Webster, E. Enanga, and F.D. Beall. 2014. "Searching for similarity in topographic controls on carbon, nitrogen and phosphorus export from forested headwater catchments." *Hydrological Processes* 28(8): 3201-3216.
- Merbt, S.N., L. Proia, J.I. Prosser, E. Martí, E.O. Casamayor, and D. von Schiller. 2016. "Stream drying drives microbial ammonia oxidation and first-flush nitrate export." *Ecology* 97(9): 2192-2198.
- Merriam, E.R., R. Fernandez, J.T. Petty, and N. Zegre. 2017. "Can brook trout survive climate change in large rivers? If it rains." *Science of the Total Environment*: 1225-1236.
- Merriam, E.R., and J.T. Petty. 2016. "Under siege: Isolated tributaries are threatened by regionally impaired metacommunities." *Science of the Total Environment* 560: 170-178.
- Metcalfe, A.N., T.A. Kennedy, J.C. Marks, A.D. Smith, and J.D. Muehlbauer. 2020. "Spatial population structure of a widespread aquatic insect in the Colorado River Basin: Evidence for a *Hydropsyche* oslari species complex." *Freshwater Science* 39(2): 309-320. <u>https://doi.org/10.1086/709022</u>.
- Meulemans, M.J., C.S. Jones, K.E. Schilling, N.C. Young, and L.J. Weber. 2020. "Assessment of Spatial Nitrate Patterns in An Eastern Iowa Watershed Using Boat-Deployed Sensors." *Water* 12(1).
- Michaelides, K., and M.B. Singer. 2014. "Impact of coarse sediment supply from hillslopes to the channel in runoff-dominated, dryland fluvial systems." *Journal of Geophysical Research-Earth Surface* 119(6): 1205-1221.
- Michalek, A., A. Zarnaghsh, and A. Husic. 2021. "Modeling linkages between erosion and connectivity in

an urbanizing landscape." Science of the Total Environment 764.

- Milesi, S.V., and A.S. Melo. 2014. "Conditional effects of aquatic insects of small tributaries on mainstream assemblages: position within drainage network matters." *Canadian Journal of Fisheries and Aquatic Sciences* 71(1): 1-9.
- Miller, A.D., and S.K. Brewer. 2020. "Age-0 Smallmouth Bass abundance depends on physicochemical conditions and stream network position." *Ecosphere* 11(9).
- Miller, M.P., A.J. Tesoriero, P.D. Capel, B.A. Pellerin, K.E. Hyer, and D.A. Burns. 2016. "Quantifying watershed-scale groundwater loading and in-stream fate of nitrate using high-frequency water quality data." *Water Resources Research* 52(1): 330-347.
- Milner, V.S., S.M. Yarnell, and R.A. Peek. 2019. "The ecological importance of unregulated tributaries to macroinvertebrate diversity and community composition in a regulated river." *Hydrobiologia* 829(1): 291-305.
- Milt, A.W., M.W. Diebel, P.J. Doran, M.C. Ferris, M. Herbert, M.L. Khoury, A.T. Moody, T.M. Neeson, J. Ross, T. Treska, J.R. O'Hanley, L. Walter, S.R. Wangen, E. Yacobson, and P.B. McIntyre. 2018.
 "Minimizing opportunity costs to aquatic connectivity restoration while controlling an invasive species." *Conservation Biology* 32(4): 894-904.
- Mims, M.C., C.C. Day, J.J. Burkhart, M.R. Fuller, J. Hinkle, A. Bearlin, J.B. Dunham, P.W. Dehaan, Z.A. Holden, and E.E. Landguth. 2019. "Simulating demography, genetics, and spatially explicit processes to inform reintroduction of a threatened char." *Ecosphere* 10(2).
- Mineau, M.M., W.M. Wollheim, I. Buffam, S.E.G. Findlay, E.R. Hotchkiss, L.E. Koenig, W.H. McDowell, and T.B. Parr. 2016. "Dissolved organic carbon uptake in streams: A review and assessment of reach-scale measurements." *Journal of Geophysical Research: Biogeosciences* 121(8): 2019-2029.
- Mineau, M.M., W.M. Wollheim, and R.J. Stewart. 2015. "An index to characterize the spatial distribution of land use within watersheds and implications for river network nutrient removal and export." *Geophysical Research Letters* 42(16): 6688-6695.
- Miranda, L.E., and D.J. Dembkowski. 2016. "Evidence for Serial Discontinuity in the Fish Community of a Heavily Impounded River." *River Research and Applications* 32(6): 1187-1195.
- Miyazono, S., and C.M. Taylor. 2015. "LONG-TERM CHANGES IN SEASONAL FISH ASSEMBLAGE DYNAMICS IN AN ADVENTITIOUS DESERT STREAM." *Southwestern Naturalist* 60(1): 72-79.
- Moatar, F., M. Floury, A.J. Gold, M. Meybeck, B. Renard, M. Ferréol, A. Chandesris, C. Minaudo, K. Addy, J. Piffady, and G. Pinay. 2020. "Stream Solutes and Particulates Export Regimes: A New Framework to Optimize Their Monitoring." *Frontiers in Ecology and Evolution* 7(516).
- Mooney, R.J., E.H. Stanley, W.C. Rosenthal, P.C. Esselman, A.D. Kendall, and P.B. McIntyre. 2020. "Outsized nutrient contributions from small tributaries to a Great Lake." *Proceedings of the National*

Academy of Sciences of the United States of America 117(45): 28175-28182.

- Moore, J.W. 2015. "Bidirectional connectivity in rivers and implications for watershed stability and management." *Canadian Journal of Fisheries and Aquatic Sciences* 72(5): 785-795.
- Moore, J.W., M.P. Beakes, H.K. Nesbitt, J.D. Yeakel, D.A. Patterson, L.A. Thompson, C.C. Phillis, D.C. Braun, C. Favaro, D. Scott, C. Carr-Harris, and W.I. Atlas. 2015. "Emergent stability in a large, free-flowing watershed." *Ecology* 96(2): 340-347.
- Mor, J.R., A. Ruhi, E. Tornes, H. Valcarcel, I. Munoz, and S. Sabater. 2018. "Dam regulation and riverine food-web structure in a Mediterranean river." *Science of the Total Environment* 625: 301-310.
- Moran-Ordonez, A., A. Pavlova, A.M. Pinder, L. Sim, P. Sunnucks, R.M. Thompson, and J. Davis. 2015.
 "Aquatic communities in arid landscapes: local conditions, dispersal traits and landscape configuration determine local biodiversity." *Diversity and Distributions* 21(10): 1230-1241.
- Mori, T., Y. Kato, T. Takagi, Y. Onoda, and Y. Kayaba. 2018. "Turbid water induces refuge behaviour of a commercially important ayu: A field experiment for interstream movement using multiple artificial streams." *Ecology of Freshwater Fish* 27(4): 1015-1022.
- Morris, D.M., T.R. Gemeinhardt, N.J.C. Gosch, and D.E. Jensen. 2014. "Water Quality during Two High-Flow Years on the Lower Missouri River: The Effects of Reservoir and Tributary Contributions." *River Research and Applications* 30(8): 1024-1033. <u>https://doi.org/10.1002/rra.2693</u>.
- Morse, N.B., and W.M. Wollheim. 2014. "Climate variability masks the impacts of land use change on nutrient export in a suburbanizing watershed." *Biogeochemistry* 121(1): 45-59.
- Mortensen, J.G., R. González-Pinzón, C.N. Dahm, J. Wang, L.H. Zeglin, and D.J. Van Horn. 2016. "Advancing the Food-Energy-Water Nexus: Closing Nutrient Loops in Arid River Corridors." *Environmental Science and Technology* 50(16): 8485-8496.
- Mosesso, L.R., A. Buda, A. Collick, C. Kennedy, G. Folmar, and A. Shober. 2021. "Examining sources and pathways of phosphorus transfer in a ditch-drained field." *Journal of Environmental Quality* 50(3): 680-693.
- Mossop, K.D., M. Adams, P.J. Unmack, K.L.S. Date, B.B.M. Wong, and D.G. Chapple. 2015. "Dispersal in the desert: ephemeral water drives connectivity and phylogeography of an arid-adapted fish." *Journal of Biogeography* 42(12): 2374-2388.
- Mulder, A.J.E., R. Aalderen, and C.H.A. Leeuwen. 2021. "Tracking temperate fish reveals their relevance for plant seed dispersal." *Functional Ecology* 35(5): 1134-1144.
- Muraoka, K., I. Masumoto, Y. Agata, M. Ota, Y. Yamahara, Y. Seguchi, T. Asukagawa, and Y. Kayaba. 2015. "Molecular Biology-Based Evaluation of Fish Migration in Rivers." *Proceedings of the 36th Iahr World Congress*: 563-571.
- Murphy, A.L., A. Pavlova, R. Thompson, J. Davis, and P. Sunnucks. 2015. "Swimming through sand: connectivity of aquatic fauna in deserts." *Ecology and Evolution* 5(22): 5252-5264.

- Murray, B.F., M.A. Reid, S.J. Capon, M. Thoms, and S.-B. Wu. 2019. "Gene flow and genetic structure in Acacia stenophylla (Fabaceae): Effects of hydrological connectivity." Journal of Biogeography 46(6): 1138-1151. <u>https://doi.org/10.1111/jbi.13566</u>.
- Nakagawa, H. 2014. "Contribution of environmental and spatial factors to the structure of stream fish assemblages at different spatial scales." *Ecology of Freshwater Fish* 23(2): 208-223.
- Nakamoto, B.J., M.L. Fogel, C.A. Jeffres, and J.H. Viers. 2020. "Dynamic river processes drive variability in particulate organic matter over fine spatiotemporal scales." *Freshwater Biology* 65(9): 1569-1584.
- Nakayama, T. 2017. "Scaled-dependence and seasonal variations of carbon cycle through development of an advanced eco-hydrologic and biogeochemical coupling model." *Ecological Modelling* 356: 151-161.
- Nathan, L.R., Y. Kanno, and J.C. Vokoun. 2017. "Population demographics influence genetic responses to fragmentation: A demogenetic assessment of the 'one migrant per generation' rule of thumb." *Biological Conservation* 210: 261-272.
- Naus, C.J., and S.R. Adams. 2018. "Fish nursery habitat function of the main channel, floodplain tributaries and oxbow lakes of a medium-sized river." *Ecology of Freshwater Fish* 27(1): 4-18.
- Neilson, T.B., A.H. Schmidt, P.R. Bierman, D.H. Rood, and V.S. Gonzalez. 2017. "Efficacy of in situ and meteoric Be-10 mixing in fluvial sediment collected from small catchments in China." *Chemical Geology* 471: 119-130.
- Nelson, A., and K. Dube. 2016. "Channel response to an extreme flood and sediment pulse in a mixed bedrock and gravel-bed river." *Earth Surface Processes and Landforms* 41(2): 178-195.
- Nelson, P.A., A.K. Brew, and J.A. Morgan. 2015. "Morphodynamic response of a variable-width channel to changes in sediment supply." *Water Resources Research* 51(7): 5717-5734.
- Neufeld, K.R., M.S. Poesch, and D.A. Watkinson. 2016. "The Effect of Hydrologic Alteration on Capture Efficiency of Freshwater Fishes in a Highly Modified Prairie Stream." *River Research and Applications* 32(5): 975-983.
- Newton, M., J. Barry, J.A. Dodd, M.C. Lucas, P. Boylan, and C.E. Adams. 2019. "A test of the cumulative effect of river weirs on downstream migration success, speed and mortality of Atlantic salmon (Salmo salar) smolts: An empirical study." *Ecology of Freshwater Fish* 28(1): 176-186.
- Ngeve, M.N., T. Van Der Stocken, T. Sierens, N. Koedam, and L. Triest. 2017. "Bidirectional gene flow on a mangrove river landscape and between-catchment dispersal of *Rhizophora racemosa* (Rhizophoraceae)." *Hydrobiologia* 790(1): 93-108. <u>https://doi.org/10.1007/s10750-016-3021-2</u>.
- Nicacio, G., and L. Juen. 2018. "Relative roles of environmental and spatial constraints in assemblages of Chironomidae (Diptera) in Amazonian floodplain streams." *Hydrobiologia* 820(1): 201-213.

Nino-Garcia, J.P., C. Ruiz-Gonzalez, and P.A. Del Giorgio. 2016. "Interactions between hydrology and

water chemistry shape bacterioplankton biogeography across boreal freshwater networks." *ISME Journal: Multidisciplinary Journal of Microbial Ecology* 10(7): 1755-1766.

- Niño-García, J.P., C. Ruiz-González, and P.A. Del Giorgio. 2016. "Landscape-scale spatial abundance distributions discriminate core from random components of boreal lake bacterioplankton." *Ecology Letters* 19(12): 1506-1515.
- Nippgen, F., B.L. McGlynn, and R.E. Emanuel. 2015. "The spatial and temporal evolution of contributing areas." *Water Resources Research* 51(6): 4550-4573.
- Nippgen, F., M.R.V. Ross, E.S. Bernhardt, and B.L. McGlynn. 2017. "Creating a More Perennial Problem? Mountaintop Removal Coal Mining Enhances and Sustains Saline Baseflows of Appalachian Watersheds." *Environmental Science and Technology* 51(15): 8324-8334.
- Noe, G.B., M.J. Cashman, K. Skalak, A. Gellis, K.G. Hopkins, D. Moyer, J. Webber, A. Benthem, K. Maloney, J. Brakebill, A. Sekellick, M. Langland, Q. Zhang, G. Shenk, J. Keisman, and C. Hupp. 2020. "Sediment dynamics and implications for management: State of the science from long-term research in the Chesapeake Bay watershed, USA." *WIREs Water* 7(4).
- Novais, M.H., M.M. Morais, J. Rosado, L.S. Dias, L. Hoffmann, and L. Ector. 2014. "Diatoms of temporary and permanent watercourses in Southern Europe (Portugal)." *River Research and Applications* 30(10): 1216-1232.
- Obedzinski, M., S.N. Pierce, G.E. Horton, and M.J. Deitch. 2018. "Effects of Flow-Related Variables on Oversummer Survival of Juvenile Coho Salmon in Intermittent Streams." *Transactions of the American Fisheries Society* 147(3): 588-605.
- Ohira, M., and S. Fukuda. 2018. "Flow regime shapes seasonal patterns of fish species richness and abundance in main and branch channels of a rice irrigation system." *Paddy and Water Environment* 16(4): 783-793.
- Okano, J.I., J. Shibata, Y. Sakai, M. Yamaguchi, M. Ohishi, Y. Goda, S.I. Nakano, and N. Okuda. 2018. "The effect of human activities on benthic macroinvertebrate diversity in tributary lagoons surrounding Lake Biwa." *Limnology* 19(2): 199-207.
- O'Leary, N., R. Johnston, E.L. Gardner, S.M. Penningroth, and D.R. Bouldin. 2019. "Long-Term Study of Soluble Reactive Phosphorus Concentration in Fall Creek and Comparison to Northeastern Tributaries of Cayuga Lake, NY: Implications for Watershed Monitoring and Management." *Water* 11(10).
- Olson, D.H., and M.R. Kluber. 2014. "Plethodontid Salamander Distributions in Managed Forest Headwaters in Western Oregon, USA." *Herpetological Conservation and Biology* 9(1): 76-96.
- Omer, A.R., C. Aldridge, and B.H. Baker. 2020. "Outflow hydrology of tailwater recovery systems with comparisons to a control catchment*." *Irrigation and Drainage* 70(4): 786–796. https://doi.org/10.1002/ird.2551.

Oromeng, K.V., E.A. Atekwana, L. Molwalefhe, and G.J. Ramatlapeng. 2021. "Time-series variability of

solute transport and processes in rivers in semi-arid endorheic basins: The Okavango Delta, Botswana." *Science of the Total Environment* 759.

- Osawa, T., H. Mitsuhashi, and A. Ushimaru. 2014. "Plant species' coexistence relationships may shift according to life history traits and seasons." *Plant Ecology* 215(6): 597-612.
- Padgett, D.J., M. Joyal, S. Quirk, M. Laubi, and T.D. Surasinghe. 2018. "Evidence of aquatic plant seed dispersal by eastern painted turtles (*Chrysemys picta picta*) in Massachusetts, USA." *Aquatic Botany* 149: 40-45.
- Painter, K.J., R.B. Brua, P.A. Chambers, J.M. Culp, C.T. Chesworth, S.N. Cormier, C.D. Tyrrell, and A.G. Yates. 2020. "An ecological causal assessment of tributaries draining the Red River Valley, Manitoba." *Journal of Great Lakes Research* 47(3): 773-787.
- Painter, K.J., R.B. Brua, G. Koehler, J. Spoelstra, and A.G. Yates. 2021. "Contribution of nitrogen sources to streams in mixed-use catchments varies seasonally in a cold temperate region." *Science of the Total Environment* 764.
- Palmer, M., and A. Ruhi. 2019. "Linkages between flow regime, biota, and ecosystem processes: Implications for river restoration." *Science* 365(6459).
- Pan, S., Z. Bian, H. Tian, Y. Yao, R.G. Najjar, M.A.M. Friedrichs, E.E. Hofmann, R. Xu, and B. Zhang. 2021. "Impacts of Multiple Environmental Changes on Long-Term Nitrogen Loading From the Chesapeake Bay Watershed." *Journal of Geophysical Research: Biogeosciences* 126(5).
- Papangelakis, E., B. Macvicar, and P. Ashmore. 2019. "Bedload Sediment Transport Regimes of Semialluvial Rivers Conditioned by Urbanization and Stormwater Management." *Water Resources Research* 55(12): 10565-10587.
- Pattison, I., S.N. Lane, R.J. Hardy, and S.M. Reaney. 2014. "The role of tributary relative timing and sequencing in controlling large floods." *Water Resources Research* 50(7): 5444-5458. <u>https://doi.org/10.1002/2013WR014067</u>.
- Paz-Vinas, I., G. Loot, V.M. Stevens, and S. Blanchet. 2015. "Evolutionary processes driving spatial patterns of intraspecific genetic diversity in river ecosystems." *Molecular Ecology* 24(18): 4586-4604.
- Pelaez, O., and C.S. Pavanelli. 2019. "Environmental heterogeneity and dispersal limitation explain different aspects of beta-diversity in Neotropical fish assemblages." *Freshwater Biology* 64(3): 497-505.
- Pennino, M.J., S.S. Kaushal, J.J. Beaulieu, P.M. Mayer, and C.P. Arango. 2014. "Effects of urban stream burial on nitrogen uptake and ecosystem metabolism: implications for watershed nitrogen and carbon fluxes." *Biogeochemistry* 121(1): 247-269.
- Peoples, B.K., R.A. McManamay, D.A. Orth, and E.R. Frimpong. 2014. "Nesting habitat use by river chubs in a hydrologically variable Appalachian tailwater." *Ecology of Freshwater Fish* 23(2): 283-293.

- Pepino, M., M.A. Rodriguez, and P. Magnan. 2017. "Incorporating lakes in stream fish habitat models: are we missing a key landscape attribute?" *Canadian Journal of Fisheries and Aquatic Sciences* 74(5): 629-635.
- Peralta-Tapia, A., R.A. Sponseller, D. Tetzlaff, C. Soulsby, and H. Laudon. 2015. "Connecting precipitation inputs and soil flow pathways to stream water in contrasting boreal catchments." *Hydrological Processes* 29(16): 3546-3555.
- Perkin, J.S., K.B. Gido, A.R. Cooper, T.F. Turner, M.J. Osborne, E.R. Johnson, and K.B. Mayes. 2015. "Fragmentation and dewatering transform Great Plains stream fish communities." *Ecological Monographs* 85(1): 73-92.
- Perkin, J.S., K.B. Gido, J.A. Falke, K.D. Fausch, H. Crockett, E.R. Johnson, and J. Sanderson. 2017. "Groundwater declines are linked to changes in Great Plains stream fish assemblages." *Proceedings* of the National Academy of Sciences 114(28).
- Perkin, J.S., S.P. Murphy, C.M. Murray, W.K. Gibbs, and A.E. Gebhard. 2019. "If you build it, they will go: A case study of stream fish diversity loss in an urbanizing riverscape." *Aquatic Conservation: Marine and Freshwater Ecosystems* 29(4): 623-638.
- Perrier, C., J. Le Gentil, V. Ravigne, P. Gaudin, and J.C. Salvado. 2014. "Origins and genetic diversity among Atlantic salmon recolonizing upstream areas of a large South European river following restoration of connectivity and stocking." *Conservation Genetics* 15(5): 1095-1109.
- Peter, D.H., E. Castella, and V.I. Slaveykova. 2014. "Effects of a reservoir flushing on trace metal partitioning, speciation and benthic invertebrates in the floodplain." *Environmental Science: Processes & Impacts* 16(12): 2692-2702.
- Peter, H., G. Singer, A.J. Ulseth, T. Dittmar, Y.T. Prairie, and T.J. Battin. 2020. "Travel Time and Source Variation Explain the Molecular Transformation of Dissolved Organic Matter in an Alpine Stream Network." *Journal of Geophysical Research: Biogeosciences* 125(8).
- Peterson, D.P., B.E. Rieman, D.L. Horan, and M.K. Young. 2014. "Patch size but not short-term isolation influences occurrence of westslope cutthroat trout above human-made barriers." *Ecology of Freshwater Fish* 23(4): 556-571.
- Pfauserová, N., O. Slavík, P. Horký, J. Turek, and T. Randák. 2021. "Spatial distribution of native fish species in tributaries is altered by the dispersal of non-native species from reservoirs." *Science of the Total Environment* 755.
- Pfeiffer, A., and E. Wohl. 2018. "Where Does Wood Most Effectively Enhance Storage? Network-Scale Distribution of Sediment and Organic Matter Stored by Instream Wood." *Geophysical Research Letters* 45(1): 194-200.
- Phillipsen, I.C., E.H. Kirk, M.T. Bogan, M.C. Mims, J.D. Olden, and D.A. Lytle. 2015. "Dispersal ability and habitat requirements determine landscape-level genetic patterns in desert aquatic insects." *Molecular Ecology* 24(1): 54-69.

- Piano, E., A. Doretto, S. Mammola, E. Falasco, S. Fenoglio, and F. Bona. 2020. "Taxonomic and functional homogenisation of macroinvertebrate communities in recently intermittent Alpine watercourses." *Freshwater Biology* 65(12): 2096-2107.
- Pierce, C.L., N.L. Ahrens, A.K. Loan-Wilsey, G.A. Simmons, and G.T. Gelwicks. 2014. "Fish assemblage relationships with physical characteristics and presence of dams in three eastern Iowa rivers." *River Research and Applications* 30(4): 427-441. https://doi.org/10.1002/rra.2654.
- Pierce, R., C. Podner, T. Wendt, R. Shields, and K. Carim. 2014. "Westslope Cutthroat Trout Movements through Restored Habitat and Coanda Diversions in the Nevada Spring Creek Complex, Blackfoot Basin, Montana." *Transactions of the American Fisheries Society* 143(1): 230-239. <u>https://doi.org/10.1080/00028487.2013.839959</u>.
- Pilger, T.J., K.B. Gido, D.L. Propst, J.E. Whitney, and T.F. Turner. 2017. "River network architecture, genetic effective size and distributional patterns predict differences in genetic structure across species in a dryland stream fish community." *Molecular Ecology* 26(10): 2687-2697.
- Pilliod, D.S., R.S. Arkle, J.M. Robertson, M.A. Murphy, and W.C. Funk. 2015. "Effects of changing climate on aquatic habitat and connectivity for remnant populations of a wide-ranging frog species in an arid landscape." *Ecology and Evolution* 5(18): 3979-3994.
- Pitt, A.L., J.J. Tavano, R.F. Baldwin, and B.S. Stegenga. 2017. "Movement Ecology and Habitat Use of Three Sympatric Anuran Species." *Herpetological Conservation and Biology* 12(1): 212-224.
- Pizzuto, J. 2020. "Suspended sediment and contaminant routing with alluvial storage: New theory and applications." *Geomorphology* 352: 106983.
- Placzkowska, E., and K. Krzemien. 2018. "Natural conditions of coarse bedload transport in headwater catchments (Western Tatras, Poland)." *Geografiska Annaler Series A: Physical Geography* 100(4): 370-387.
- Placzkowska, E., K. Krzemien, E. Gorczyca, A. Bojarczuk, and M. Zelazny. 2020. "Disturbances in coarse bedload transport in a high-mountain stream channel system (Western Tatras, Poland)." *Geomorphology* 371.
- Platania, S.P., J.G. Mortensen, M.A. Farrington, W.H. Brandenburg, and R.K. Dudley. 2019. "Dispersal of Stocked Rio Grande Silvery Minnow (*Hybognathus amarus*) in the Middle Rio Grande, New Mexico." *Southwestern Naturalist* 64(1): 31-42.
- Plont, S., B.M. O'Donnell, M.T. Gallagher, and E.R. Hotchkiss. 2020. "Linking carbon and nitrogen spiraling in streams." *Freshwater Science* 39(1): 126-136.
- Pollux, B.J.A. 2017. "Consistent individual differences in seed disperser quality in a seed-eating fish." *Oecologia* 183(1): 81-91.
- Pond, G.J., K. Fritz, and B.R. Johnson. 2016. "Macroinvertebrate and organic matter export from headwater tributaries of a Central Appalachian stream." *Hydrobiologia* 779(1): 75-91.

- Pool, T.K., and J.D. Olden. 2015. "Assessing long-term fish responses and short-term solutions to flow regulation in a dryland river basin." *Ecology of Freshwater Fish* 24(1): 56-66.
- Portinho, J.L., G. Perbiche-Neves, and M.G. Nogueira. 2016. "Zooplankton community and tributary effects in free-flowing section downstream a large tropical reservoir." *International Review of Hydrobiology* 101: 48-56.
- Powers, S.M., J.L. Tank, and D.M. Robertson. 2015. "Control of nitrogen and phosphorus transport by reservoirs in agricultural landscapes." *Biogeochemistry* 124: 417-439.
- Prancevic, J.P., and J.W. Kirchner. 2019. "Topographic Controls on the Extension and Retraction of Flowing Streams." *Geophysical Research Letters* 46(4): 2084-2092.
- Prokop, P., L. Wiejaczka, S. Sarkar, T. Bryndal, A. Bucala-Hrabia, R. Kroczak, R. Soja, and E. Placzkowska. 2020. "Morphological and sedimentological responses of small stream channels to extreme rainfall and land use in the Darjeeling Himalayas." *Catena* 188.
- Prunier, J.G., V. Dubut, G. Loot, L. Tudesque, and S. Blanchet. 2018. "The relative contribution of river network structure and anthropogenic stressors to spatial patterns of genetic diversity in two freshwater fishes: A multiple-stressors approach." *Freshwater Biology* 63(1): 6-21.
- Ptatscheck, C., B. Gansfort, N. Majdi, and W. Traunspurger. 2020. "The influence of environmental and spatial factors on benthic invertebrate metacommunities differing in size and dispersal mode." *Aquatic Ecology* 54(2): 447-461.
- Pu, J., J. Li, T. Zhang, X. Xiong, and D. Yuan. 2019. "High spatial and seasonal heterogeneity of pCO₂ and CO₂ emissions in a karst groundwater-stream continuum, southern China." *Environmental Science and Pollution Research* 26(25): 25733-25748. <u>https://doi.org/10.1007/s11356-019-05820-9</u>.
- Pusey, B.J., M.J. Kennard, M. Douglas, and Q. Allsop. 2018. "Fish assemblage dynamics in an intermittent river of the northern Australian wet-dry tropics." *Ecology of Freshwater Fish* 27(1): 78-88.
- Qu, Y., N. Wu, B. Guse, and N. Fohrer. 2018. "Riverine phytoplankton shifting along a lentic-lotic continuum under hydrological, physiochemical conditions and species dispersal." *Science of the Total Environment*: 1628-1636.
- Quaglietta, L., J. Pauperio, F.M.S. Martins, P.C. Alves, and P. Beja. 2018. "Recent range contractions in the globally threatened Pyrenean desman highlight the importance of stream headwater refugia." *Animal Conservation* 21(6): 515-525.
- Quichimbo, E.A., M.B. Singer, and M.O. Cuthbert. 2020. "Characterising groundwater-surface water interactions in idealised ephemeral stream systems." *Hydrological Processes* 34(18): 3792-3806.
- Raabe, J.K., and J.E. Hightower. 2014. "Assessing Distribution of Migratory Fishes and Connectivity following Complete and Partial Dam Removals in a North Carolina River." *North American Journal of Fisheries Management* 34(5): 955-969.

- Radinger, J., F. Essl, F. Hölker, P. Horký, O. Slavík, and C. Wolter. 2017. "The future distribution of river fish: The complex interplay of climate and land use changes, species dispersal and movement barriers." *Global Change Biology* 23(11): 4970-4986.
- Radinger, J., and E. García-Berthou. 2020. "The role of connectivity in the interplay between climate change and the spread of alien fish in a large Mediterranean river." *Global Change Biology* 26(11): 6383-6398.
- Radinger, J., and C. Wolter. 2014. "Patterns and predictors of fish dispersal in rivers." *Fish and Fisheries* 15(3): 456-473.
- Radinger, J., and C. Wolter. 2015. "Disentangling the effects of habitat suitability, dispersal, and fragmentation on the distribution of river fishes." *Ecological Applications* 25(4): 914-927.
- Rahman, M.M., Z. Lin, X. Jia, D.D. Steele, and T.M. Desutter. 2014. "Impact of subsurface drainage on streamflows in the Red River of the North basin." *Journal of Hydrology* 511: 474-483.
- Ramulifho, P.A., N.A. Rivers-Moore, H.F. Dallas, and S.H. Foord. 2018. "A conceptual framework towards more holistic freshwater conservation planning through incorporation of stream connectivity and thermal vulnerability." *Journal of Hydrology* 556: 173-181.
- Ranjbar, S., A. Singh, and D. Wang. 2020. "Controls of the Topological Connectivity on the Structural and Functional Complexity of River Networks." *Geophysical Research Letters* 47(22).
- Rantala, H., D. Glover, J. Garvey, Q. Phelps, S. Tripp, D. Herzog, R. Hrabik, J. Crites, and M. Whiles. 2016. "Fish Assemblage and Ecosystem Metabolism Responses to Reconnection of the Bird's Point-New Madrid Floodway during the 2011 Mississippi River Flood." *River Research and Applications* 32(5): 1018-1029.
- Rathburn, S.L., G.L. Bennett, E.E. Wohl, C. Briles, B. McElroy, and N. Sutfin. 2017. "The fate of sediment, wood, and organic carbon eroded during an extreme flood, Colorado Front Range, USA." *Geology* 45(6): 499-502.
- Rattan, K.J., J.C. Corriveau, R.B. Brua, J.M. Culp, A.G. Yates, and P.A. Chambers. 2017. "Quantifying seasonal variation in total phosphorus and nitrogen from prairie streams in the Red River Basin, Manitoba Canada." *Science of the Total Environment* 575: 649-659.
- Raymond, P.A., J.E. Saiers, and W.V. Sobczak. 2016. "Hydrological and biogeochemical controls on watershed dissolved organic matter transport: pulse-shunt concept." *Ecology* 97(1): 5-16.
- Rebenack, J.J., S. Ricker, C. Anderson, M. Wallace, and D.M. Ward. 2015. "Early Emigration of Juvenile Coho Salmon: Implications for Population Monitoring." *Transactions of the American Fisheries Society* 144(1): 163-172.
- Records, R.M., E. Wohl, and M. Arabi. 2016. "Phosphorus in the river corridor." *Earth-Science Reviews* 158: 65-88.

Reisinger, A.J., J.L. Tank, E.J. Rosi-Marshall, and M.A. Baker. 2015. "The varying role of water column

nutrient uptake along river continua in contrasting landscapes." Biogeochemistry 125(1): 115-131.

- Reuter, C.N., M.D. Kaller, C.E. Walsh, and W.E. Kelso. 2019. "Fish Assemblage Response to Altered Dendritic Connectivity in the Red River Basin, Central Louisiana." *American Midland Naturalist* 181(1): 63-80.
- Ribeiro, M.L., K. Blanckaert, and A.J. Schleiss. 2016. "Local tributary widening for river rehabilitation." *Ecohydrology* 9(2): 204-217.
- Ribeiro, M.L., S. Wampfler, and A.J. Schleiss. 2014. "Morphodynamic changes in a natural river confluence due to a hydropower modified flow regime." *Swiss Competences in River Engineering and Restoration*: 191-199.
- Rice, J.S., R.E. Emanuel, J.M. Vose, and S.A.C. Nelson. 2015. "Continental US streamflow trends from 1940 to 2009 and their relationships with watershed spatial characteristics." *Water Resources Research* 51(8): 6262-6275.
- Rice, S.P. 2017. "Tributary connectivity, confluence aggradation and network biodiversity." *Geomorphology* 277: 6-16.
- Richards, G., T.E. Gilmore, A.R. Mittelstet, T.L. Messer, and D.D. Snow. 2021. "Baseflow nitrate dynamics within nested watersheds of an agricultural stream in Nebraska, USA." *Agriculture, Ecosystems and Environment* 308.
- Richardson, C.M., M.A. Zimmer, J.K. Fackrell, and A. Paytan. 2020. "Geologic Controls on Source Water Drive Baseflow Generation and Carbon Geochemistry: Evidence of Nonstationary Baseflow Sources Across Multiple Subwatersheds." *Water Resources Research* 56(7).
- Richardson, J.S. 2019. "Biological Diversity in Headwater Streams." Water 11(2).
- Richmond, J.Q., A.R. Backlin, C. Galst-Cavalcante, J.W. O'Brien, and R.N. Fisher. 2018. "Loss of dendritic connectivity in southern California's urban riverscape facilitates decline of an endemic freshwater fish." *Molecular Ecology* 27(2): 369-386.
- Ritter, T.D., A. Zale, G. Grisak, and M.J. Lance. 2020. "Groundwater Upwelling Regulates Thermal Hydrodynamics and Salmonid Movements during High-Temperature Events at a Montane Tributary Confluence." *Transactions of the American Fisheries Society* 149(5): 600-619.
- Robbins, J., and M. Pyron. 2021. "Geomorphological characteristics of the Wabash River, USA: Influence on fish assemblages." *Ecology and Evolution* 11(9): 4542-4549.
- Roberts, J.J., K.D. Fausch, M.B. Hooten, and D.P. Peterson. 2017. "Nonnative Trout Invasions Combined with Climate Change Threaten Persistence of Isolated Cutthroat Trout Populations in the Southern Rocky Mountains." *North American Journal of Fisheries Management* 37(2): 314-325.
- Robinson, C. 2015. "Review on groundwater as a source of nutrients to the Great Lakes and their tributaries." *Journal of Great Lakes Research* 41(4): 941-950.

Rodeles, A.A., D. Galicia, and R. Miranda. 2021. "A simple method to assess the fragmentation of

freshwater fish meta-populations: Implications for river management and conservation." *Ecological Indicators* 125.

- Rodeles, A.A., P.M. Leunda, J. Elso, J. Ardaiz, D. Galicia, and R. Miranda. 2019. "Consideration of habitat quality in a river connectivity index for anadromous fishes." *Inland Waters* 9(3): 278-288.
- Rodríguez-Blanco, M.L., M.M. Taboada-Castro, and M.T. Taboada-Castro. 2019. "An overview of patterns and dynamics of suspended sediment transport in an agroforest headwater system in humid climate: Results from a long-term monitoring." *Science of the Total Environment* 648: 33-43.
- Rodríguez-Cardona, B.M., A.A. Coble, A.S. Wymore, R. Kolosov, D.C. Podgorski, P. Zito, R.G.M. Spencer, A.S. Prokushkin, and W.H. McDowell. 2020. "Wildfires lead to decreased carbon and increased nitrogen concentrations in upland arctic streams." *Scientific Reports* 10(1).
- Rodríguez-González, P.M., C. García, A. Albuquerque, T. Monteiro-Henriques, C. Faria, J.B. Guimarães, D. Mendonça, F. Simões, M.T. Ferreira, A. Mendes, J. Matos, and M.H. Almeida. 2019. "A spatial stream-network approach assists in managing the remnant genetic diversity of riparian forests." *Scientific Reports* 9(1).
- Rodriguez-Lozano, P., R.A. Leidy, and S.M. Carlson. 2019. "Brook lamprey survival in the dry riverbed of an intermittent stream." *Journal of Arid Environments* 166: 83-85.
- Rogosch, J.S., and J.D. Olden. 2019. "Dynamic contributions of intermittent and perennial streams to fish beta diversity in dryland rivers." *Journal of Biogeography* 46(10): 2311-2322.
- Rohlfs, A.M., S.M. Mitrovic, S. Williams, and D. Coleman. 2016. "Can tributary in-flows improve the recovery of the dissolved organic carbon regime in a snowmelt river regulated by a large reservoir?" *Marine and Freshwater Research* 67(9): 1338-1345.
- Rohlfs, A.M., S.M. Mitrovic, S. Williams, J.N. Hitchcock, and G.N. Rees. 2016. "Dissolved organic carbon delivery from managed flow releases in a montane snowmelt river." *Aquatic Sciences* 78(4): 793-807.
- Rojan, E., M. Dluzewski, and K. Krzemien. 2020. "Sediment budget of high mountain stream channels in an arid zone (High Atlas mountains, Morocco)." *Catena* 190.
- Rolls, R.J., J. Heino, and B.C. Chessman. 2016. "Unravelling the joint effects of flow regime, climatic variability and dispersal mode on beta diversity of riverine communities." *Freshwater Biology* 61(8): 1350-1364.
- Romanescu, G., and C. Stoleriu. 2014. "An inter-basin backwater overflow (the Buhai Brook and the Ezer reservoir on the Jijia River, Romania)." *Hydrological Processes* 28(7): 3118-3131.
- Rosenvald, R., R. Järvekülg, and A. Lõhmus. 2014. "Fish assemblages in forest drainage ditches: Degraded small streams or novel habitats?" *Limnologica* 46: 37-44.
- Rougemont, Q., V. Dolo, A. Oger, A.L. Besnard, D. Huteau, M.A. Coutellec, C. Perrier, S. Launey, and G. Evanno. 2021. "Riverscape genetics in brook lamprey: genetic diversity is less influenced by

river fragmentation than by gene flow with the anadromous ecotype." Heredity 126(2): 235-250.

- Rowland, R., S. Inamdar, and T. Parr. 2017. "Evolution of particulate organic matter (POM) along a headwater drainage: role of sources, particle size class, and storm magnitude." *Biogeochemistry* 133(2): 181-200.
- Roy, S., and A.S. Sahu. 2017. "Potential interaction between transport and stream networks over the lowland rivers in Eastern India." *Journal of Environmental Management* 197: 316-330.
- Rudi, G., J.S. Bailly, G. Belaud, and F. Vinatier. 2018. "Characterization of the long-distance dispersal of Johnsongrass (*Sorghum halepense*) in a vegetated irrigation channel." *River Research and Applications* 34(9): 1219-1228.
- Rueegg, J., W.K. Dodds, M.D. Daniels, K.R. Sheehan, C.L. Baker, W.B. Bowden, K.J. Farrell, M.B.
 Flinn, T.K. Harms, J.B. Jones, L.E. Koenig, J.S. Kominoski, W.H. McDowell, S.P. Parker, A.D.
 Rosemond, M.T. Trentman, M. Whiles, and W.M. Wollheim. 2016. "Baseflow physical characteristics differ at multiple spatial scales in stream networks across diverse biomes." *Landscape Ecology* 31(1): 119-136.
- Ruhala, S.S., J.P. Zarnetske, D.T. Long, J.A. Lee-Cullin, S. Plont, and E.R. Wiewiora. 2018. "Exploring dissolved organic carbon cycling at the stream-groundwater interface across a third-order, lowland stream network." *Biogeochemistry* 137: 105-126.
- Ruhi, A., I. Munoz, E. Tornes, R.J. Batalla, D. Vericat, L. Ponsati, V. Acuna, D. von Schiller, R. Marce, G. Bussi, F. Frances, and S. Sabater. 2016. "Flow regulation increases food-chain length through omnivory mechanisms in a Mediterranean river network." *Freshwater Biology* 61(9): 1536-1549.
- Ruiz-Herrera, A. 2018. "Metapopulation dynamics and total biomass: Understanding the effects of diffusion in complex networks." *Theoretical Population Biology* 121: 1-11.
- Rupp, D.E., O.S. Chegwidden, B. Nijssen, and M.P. Clark. 2021. "Changing River Network Synchrony Modulates Projected Increases in High Flows." *Water Resources Research* 57(4).
- Sabo, J.L. 2014. "Predicting the river's blue line for fish conservation." *Proceedings of the National Academy of Sciences of the United States of America* 111(38): 13686-13687.
- Sabo, J.L., M. Caron, R. Doucett, K.L. Dibble, A. Ruhi, J.C. Marks, B.A. Hungate, and T.A. Kennedy. 2018. "Pulsed flows, tributary inputs and food-web structure in a highly regulated river." *Journal of Applied Ecology* 55(4): 1884-1895. <u>https://doi.org/10.1111/1365-2664.13109</u>.
- Saito, R., and K. Tojo. 2016. "Comparing spatial patterns of population density, biomass, and genetic diversity patterns of the habitat generalist mayfly *Isonychia japonica* Ulmer (Ephemeroptera:Isonychiidae) in the Chikuma-Shinano River basin." *Freshwater Science* 35(2): 724-737.
- Saito, V.S., N.E. Stoppa, E.M. Shimabukuro, M. Canedo-Arguelles, N. Bonada, and T. Siqueira. 2021."Stochastic colonisation dynamics can be a major driver of temporal beta diversity in Atlantic Forest coastal stream communities." *Freshwater Biology*.

- Samia, Y., and F. Lutscher. 2017. "Downstream flow and upstream movement determine the value of a stream reach for potadromous fish populations." *Theoretical Ecology* 10(1): 21-34.
- Sandlund, O.T., S. Karlsson, E.B. Thorstad, O.K. Berg, M.P. Kent, I.C. Norum, and K. Hindar. 2014. "Spatial and temporal genetic structure of a river-resident Atlantic salmon (Salmo salar) after millennia of isolation." *Ecology and Evolution* 4(9): 1538-1554.
- Sa-Oliveira, J.C., J.E. Hawes, V.J. Isaac-Nahum, and C.A. Peres. 2015. "Upstream and downstream responses of fish assemblages to an eastern Amazonian hydroelectric dam." *Freshwater Biology* 60(10): 2037-2050.
- Sarhad, J., R. Carlson, and K.E. Anderson. 2014. "Population persistence in river networks." *Journal of Mathematical Biology* 69(2): 401-448.
- Sarhad, J., S. Manifold, and K.E. Anderson. 2017. "Geometric indicators of population persistence in branching continuous-space networks." *Journal of Mathematical Biology* 74(4): 981-1009.
- Sarker, S., A. Veremyev, V. Boginski, and A. Singh. 2019. "Critical Nodes in River Networks." *Scientific Reports* 9(1).
- Savio, D., L. Sinclair, U.Z. Ijaz, J. Parajka, G.H. Reischer, P. Stadler, A.P. Blaschke, G. Blöschl, R.L. Mach, A.K. Kirschner, A.H. Farnleitner, and A. Eiler. 2015. "Bacterial diversity along a 2600 km river continuum." *Environmental Microbiology* 17(12): 4994-5007.
- Schall, M.K., T. Wertz, G.D. Smith, V.S. Blazer, and T. Wagner. 2019. "Movement dynamics of smallmouth bass (*Micropterus dolomieu*) in a large river-tributary system." *Fisheries Management* and Ecology 26(6): 590-599.
- Schelker, J., R. Sponseller, E. Ring, L. Hogbom, S. Lofgren, and H. Laudon. 2016. "Nitrogen export from a boreal stream network following forest harvesting: seasonal nitrate removal and conservative export of organic forms." *Biogeosciences* 13(1): 1-12.
- Schiller, D.v., R. Marcé, B. Obrador, L. Gómez-Gener, J.P. Casas-Ruiz, V. Acuña, and M. Koschorreck. 2014. "Carbon dioxide emissions from dry watercourses." *Inland Waters* 4(4): 377-382.
- Schilling, K.E., P.W. Gassman, A. Arenas-Amado, C.S. Jones, and J. Arnold. 2019. "Quantifying the contribution of tile drainage to basin-scale water yield using analytical and numerical models." *Science of the Total Environment* 657: 297-309.
- Schilling, K.E., and C.S. Jones. 2019. "Hydrograph separation of subsurface tile discharge." *Environmental Monitoring and Assessment* 191(4).
- Schilling, O.S., P.G. Cook, P.F. Grierson, S. Dogramaci, and C.T. Simmons. 2021. "Controls on Interactions Between Surface Water, Groundwater, and Riverine Vegetation Along Intermittent Rivers and Ephemeral Streams in Arid Regions." *Water Resources Research* 57(2).
- Schliemann, S.A., N. Grevstad, and R.H. Brazeau. 2021. "Water quality and spatio-temporal hot spots in an effluent-dominated urban river." *Hydrological Processes* 35(1).

- Schmadel, N.M., J.W. Harvey, R.B. Alexander, E.W. Boyer, G.E. Schwarz, J.D. Gomez-Velez, D. Scott, and C.P. Konrad. 2020. "Low threshold for nitrogen concentration saturation in headwaters increases regional and coastal delivery." *Environmental Research Letters* 15(4).
- Schmidt, B.V., and J. Schaefer. 2018. "Ecological and landscape effects on genetic distance in an assemblage of headwater fishes." *Ecology of Freshwater Fish* 27(2): 617-631.
- Schmidt, D.J., J.A. Huey, and J.M. Hughes. 2018. "Genome-Wide SNPs Identify Limits to Connectivity in the Extreme Freshwater Disperser, Spangled Perch Leiopotherapon unicolor (Terapontidae)." *Journal of Heredity* 109(3): 320-325.
- Schmitt, R.J.P., S. Bizzi, and A. Castelletti. 2016. "Tracking multiple sediment cascades at the river network scale identifies controls and emerging patterns of sediment connectivity." *Water Resources Research* 52(5): 3941-3965.
- Schmitt, R.J.P., S. Bizzi, A.F. Castelletti, and G.M. Kondolf. 2018. "Stochastic Modeling of Sediment Connectivity for Reconstructing Sand Fluxes and Origins in the Unmonitored Se Kong, Se San, and Sre Pok Tributaries of the Mekong River." *Journal of Geophysical Research-Earth Surface* 123(1): 2-25.
- Schofield, K.A., L.C. Alexander, C.E. Ridley, M.K. Vanderhoof, K.M. Fritz, B.C. Autrey, J.E. DeMeester, W.G. Kepner, C.R. Lane, S.G. Leibowitz, and A.I. Pollard. 2018. "Biota Connect Aquatic Habitats throughout Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 54(2): 372-399.
- Schook, D.M., E.A. Carlson, J.S. Sholtes, and D.J. Cooper. 2016. "Effects of Moderate and Extreme Flow Regulation on Populus Growth along the Green and Yampa Rivers, Colorado and Utah." *River Research and Applications* 32(8): 1698-1708.
- Schreiner-Mcgraw, A.P., H. Ajami, and E.R. Vivoni. 2019. "Extreme weather events and transmission losses in arid streams." *Environmental Research Letters* 14(8).
- Schreiner-Mcgraw, A.P., and E.R. Vivoni. 2018. "On the Sensitivity of Hillslope Runoff and Channel Transmission Losses in Arid Piedmont Slopes." *Water Resources Research* 54(7): 4498-4518.
- Schroeder, R.K., L.D. Whitman, B. Cannon, and P. Olmsted. 2016. "Juvenile life-history diversity and population stability of spring Chinook salmon in the Willamette River basin, Oregon." *Canadian Journal of Fisheries and Aquatic Sciences* 73(6): 921-934.
- Schuetz, T., C. Gascuel-Odoux, P. Durand, and M. Weiler. 2016. "Nitrate sinks and sources as controls of spatio-temporal water quality dynamics in an agricultural headwater catchment." *Hydrology and Earth System Sciences* 20(2): 843-857.
- Schultz, L.D. 2014. "USE OF SEASONALLY AVAILABLE HABITAT BY NORTHERN LEATHERSIDE IN WYOMING." *Southwestern Naturalist* 59(1): 118-120.
- Schwinn, M., H. Baktoft, K. Aarestrup, M.C. Lucas, and A. Koed. 2018. "Telemetry observations of predation and migration behaviour of brown trout (*Salmo trutta*) smolts negotiating an artificial

lake." River Research and Applications 34(8): 898-906. https://doi.org/10.1002/rra.3327.

- Scribner, K.T., C. Soiseth, J. McGuire, G.K. Sage, L. Thorsteinson, J.L. Nielsen, and E. Knudsen. 2017.
 "Genetic assessment of the effects of streamscape succession on coho salmon *Oncorhynchus kisutch* colonization in recently deglaciated streams." *Journal of Fish Biology* 91(1): 195-218.
- Segev, O., and L. Blaustein. 2014. "Influence of water velocity and predation risk on fire salamander (*Salamandra infraimmaculata*) larval drift among temporary pools in ephemeral streams." *Freshwater Science* 33(3): 950-957.
- Seliger, C., and B. Zeiringer. 2018. "River Connectivity, Habitat Fragmentation and Related Restoration Measures." Pages 171-186 in *Riverine Ecosystem Management*. S. Schmutz, and J. Sendzimir, eds. Aquatic Ecology Series. Springer, Cham, Switzerland.
- Sergeant, C.J., J.A. Falke, R.A. Bellmore, J.R. Bellmore, and R.L. Crumley. 2020. "A Classification of Streamflow Patterns Across the Coastal Gulf of Alaska." *Water Resources Research* 56(2).
- Seybold, E., and B. McGlynn. 2018. "Hydrologic and biogeochemical drivers of dissolved organic carbon and nitrate uptake in a headwater stream network." *Biogeochemistry* 138.
- Seymour, M., K. Deiner, and F. Altermatt. 2016. "Scale and scope matter when explaining varying patterns of community diversity in riverine metacommunities." *Basic and Applied Ecology* 17(2): 134-144.
- Seymour, M., E.A. Fronhofer, and F. Altermatt. 2015. "Dendritic network structure and dispersal affect temporal dynamics of diversity and species persistence." *Oikos* 124(7): 908-916.
- Shao, X., Y. Fang, J.W. Jawitz, J. Yan, and B. Cui. 2019. "River network connectivity and fish diversity." *Science of the Total Environment* 689: 21-30.
- Shaw, E.A., E. Lange, J.D. Shucksmith, and D.N. Lerner. 2016. "Importance of partial barriers and temporal variation in flow when modelling connectivity in fragmented river systems." *Ecological Engineering* 91: 515-528.
- Shaw, S.B., D.B. Bonville, and D.G. Chandler. 2017. "Combining observations of channel network contraction and spatial discharge variation to inform spatial controls on baseflow in Birch Creek, Catskill Mountains, USA." *Journal of Hydrology-Regional Studies* 12: 1-12.
- Shehata, M., and J. Petrie. 2016. "Hydrodynamics in the confluence of the Snake and Clearwater Rivers." *River Flow 2016*: 1690-1695.
- Sheibley, R.W., J.H. Duff, and A.J. Tesoriero. 2014. "Low Transient Storage and Uptake Efficiencies in Seven Agricultural Streams: Implications for Nutrient Demand." *Journal of Environmental Quality* 43: 1980-1990.
- Sherriff, S.C., J.S. Rowan, O. Fenton, P. Jordan, A.R. Melland, P.E. Mellander, and D.Ó. Huallacháin.
 2016. "Storm Event Suspended Sediment-Discharge Hysteresis and Controls in Agricultural
 Watersheds: Implications for Watershed Scale Sediment Management." *Environmental Science and*

Technology 50(4): 1769-1778.

- Shogren, A.J., J.L. Tank, E. Andruszkiewicz, B. Olds, A.R. Mahon, C.L. Jerde, and D. Bolster. 2017. "Controls on eDNA movement in streams: Transport, Retention, and Resuspension." *Scientific Reports* 7(1).
- Shogren, A.J., J.L. Tank, B.R. Hanrahan, and D. Bolster. 2020. "Controls on fine particle retention in experimental streams." *Freshwater Science* 39(1): 28-38.
- Shogren, A.J., J.P. Zarnetske, B.W. Abbott, F. Iannucci, R.J. Frei, N.A. Griffin, and W.B. Bowden. 2019. "Revealing biogeochemical signatures of Arctic landscapes with river chemistry." *Scientific Reports* 9(1).
- Siddiqui, E., K. Verma, U. Pandey, and J. Pandey. 2019. "Metal Contamination in Seven Tributaries of the Ganga River and Assessment of Human Health Risk from Fish Consumption." *Archives of Environmental Contamination and Toxicology* 77(2): 263-278.
- Siebers, A.R., A. Paillex, and C.T. Robinson. 2019. "Flow intermittency influences the trophic base, but not the overall diversity of alpine stream food webs." *Ecography* 42(9): 1523-1535.
- Silva, F.d.O., F.R.d. Andrade Neto, S. Hermes Silva, J.d.O.d. Silva, E. Zaniboni Filho, I.G. Prado, A. Peressin, and F.M. Pelicice. 2020. "Recruitment dynamics of a migratory fish in a semi-arid river system." *Inland Waters* 10(4): 529-541. <u>https://doi.org/10.1080/20442041.2020.1805977</u>.
- Simon, D.J., S.S.H. Gregory, B.L. James, N.P. Andrew, B.M. Jon, V.E. Mark, S.H. Maminul, and G. Steven. 2018. "The planform mobility of river channel confluences: Insights from analysis of remotely sensed imagery." *Earth-Science Reviews* 176: 1-18.
- Sinclair-Winters, C.M. 2014. "Upstream or downstream? Population structure of the land snail *Ventridens ligera* (Say, 1821) in the Potomac River drainage basin." *Journal of Molluscan Studies* 80: 280-285.
- Singer, G.P., E.D. Chapman, A.J. Ammann, A.P. Klimley, A.L. Rypel, and N.A. Fangue. 2020. "Historic drought influences outmigration dynamics of juvenile fall and spring-run Chinook Salmon." *Environmental Biology of Fishes* 103(5): 543-559.
- Škarpich, V., T. Galia, S. Ruman, and Z. Máčka. 2019. "Variations in bar material grain-size and hydraulic conditions of managed and re-naturalized reaches of the gravel-bed Bečva River (Czech Republic)." *Science of the Total Environment* 649: 672-685.
- Slater, L.J., A. Khouakhi, and R.L. Wilby. 2019. "River channel conveyance capacity adjusts to modes of climate variability." *Scientific Reports* 9(1).
- Sloat, M.R., G.H. Reeves, and K.R. Christiansen. 2017. "Stream network geomorphology mediates predicted vulnerability of anadromous fish habitat to hydrologic change in southeast Alaska." *Global Change Biology* 23(2): 604-620.
- Slone, L.A., M.J. McCarthy, J.A. Myers, C.R. Hammerschmidt, and S.E. Newell. 2018. "River sediment nitrogen removal and recycling within an agricultural Midwestern USA watershed." *Freshwater*

Science 37(1): 1-12.

- Smazak, I., and T. Galia. 2021. "Influence of tributaries on downstream bed sediment grain sizes under flysch conditions." *Journal of Mountain Science* 18(4): 847-862.
- Smith, R.F., P.D. Venugopal, M.E. Baker, and W.O. Lamp. 2015. "Habitat filtering and adult dispersal determine the taxonomic composition of stream insects in an urbanizing landscape." *Freshwater Biology* 60(9): 1740-1754.
- Smith, R.L., and J.K. Böhlke. 2019. "Methane and nitrous oxide temporal and spatial variability in two midwestern USA streams containing high nitrate concentrations." *Science of the Total Environment* 685: 574-588.
- Smith, R.M., S.S. Kaushal, J.J. Beaulieu, M.J. Pennino, and C. Welty. 2017. "Influence of infrastructure on water quality and greenhouse gas dynamics in urban streams." *Biogeosciences* 14(11): 2831-2849.
- Smits, A.P., C.M. Ruffing, T. Royer, A.P. Appling, N.A. Griffiths, R. Bellmore, M.D. Scheuerell, T.K. Harms, and J.B. Jones. 2019. "Detecting Signals of Large-Scale Climate Phenomena in Discharge and Nutrient Loads in the Mississippi-Atchafalaya River Basin." *Geophysical Research Letters* 46(7): 3791-3801.
- Snook, E.L., B.H. Letcher, T.L. Dubreuil, J. Zydlewski, M.J. O'Donnell, A.R. Whiteley, S.T. Hurley, and A.J. Danylchuk. 2016. "Movement patterns of Brook Trout in a restored coastal stream system in southern Massachusetts." *Ecology of Freshwater Fish* 25(3): 360-375.
- Soana, E., R. Balestrini, F. Vincenzi, M. Bartoli, and G. Castaldelli. 2017. "Mitigation of nitrogen pollution in vegetated ditches fed by nitrate-rich spring waters." *Agriculture, Ecosystems and Environment* 243: 74-82.
- Soana, E., M. Bartoli, M. Milardi, E.A. Fano, and G. Castaldelli. 2019. "An ounce of prevention is worth a pound of cure: Managing macrophytes for nitrate mitigation in irrigated agricultural watersheds." *Science of the Total Environment* 647: 301-312.
- Soares, A.R.A., J.F. Lapierre, B.P. Selvam, G. Lindström, and M. Berggren. 2019. "Controls on Dissolved Organic Carbon Bioreactivity in River Systems." *Scientific Reports* 9(1).
- Sorribas, M.V., D. da Motta Marques, N.M. dos Reis Castro, and F.M. Fan. 2017. "Fluvial carbon export and CO₂ efflux in representative nested headwater catchments of the eastern La Plata River Basin." *Hydrological Processes* 31(5): 995-1006.
- Sousa, R., A. Ferreira, F. Carvalho, M. Lopes-Lima, S. Varandas, A. Teixeira, and B. Gallardo. 2020. "Small hydropower plants as a threat to the endangered pearl mussel *Margaritifera margaritifera*." *Science of the Total Environment* 719.
- Spice, E.K., T.A. Whitesel, G.S. Silver, and M.F. Docker. 2019. "Contemporary and historical river connectivity influence population structure in western brook lamprey in the Columbia River Basin." *Conservation Genetics* 20(2): 299-314.

- Spurgeon, J.J., C.P. Paukert, B.D. Healy, M. Trammell, D. Speas, and E. Omana-Smith. 2015.
 "Translocation of Humpback Chub into Tributary Streams of the Colorado River: Implications for Conservation of Large-River Fishes." *Transactions of the American Fisheries Society* 144(3): 502-514.
- Spurgeon, J.J., M.A. Pegg, and N.M. Halden. 2018. "Mixed-origins of channel catfish in a large-river tributary." *Fisheries Research* 198: 195-202.
- Spurgeon, J.J., M.A. Pegg, and M.J. Hamel. 2016. "Multi-scale Approach to Hydrological Classification Provides Insight to Flow Structure in Altered River System." *River Research and Applications* 32(9): 1841-1852.
- Spurgeon, J.J., M.A. Pegg, M.J. Hamel, and K.D. Steffensen. 2018. "Spatial structure of large-river fish populations across main-stem and tributary habitats." *River Research and Applications* 34(7): 807-815.
- Stachelek, J., and P.A. Soranno. 2019. "Does freshwater connectivity influence phosphorus retention in lakes?" *Limnology and Oceanography* 64(4): 1586-1599.
- Stanley, E.H., R.A. Sponseller, and J.B. Heffernan. 2016. "Landscape and Regional Stream Ecology." *Stream Ecosystems in a Changing Environment*: 389-415.
- Stark, K., D. Cadol, D. Varyu, and J.B. Laronne. 2021. "Direct, continuous measurements of ultra-high sediment fluxes in a sandy gravel-bed ephemeral river." *Geomorphology* 382.
- Starkel, L., L. Wiejaczka, and K. Kiszka. 2017. "Role of tributaries in shaping the middle course of the Himalayan River Teesta after the 1968 extreme floods." *Current Science* 112(9): 1896-1903.
- Stegmann, L.F., R.P. Leitão, J. Zuanon, and W.E. Magnusson. 2019. "Distance to large rivers affects fish diversity patterns in highly dynamic streams of Central Amazonia." *PLoS ONE* 14(10).
- Stelzer, R.S., E.A. Strauss, and M. Coulibaly. 2017. "Assessing the importance of seepage and springs to nitrate flux in a stream network in the Wisconsin sand plains." *Hydrological Processes* 31(11): 2016-2028.
- Stickney, E., L. Kunza, G. Hoffman, and K. Chowanski. 2021. "Small land cover changes in the transboundary Kootenai River basin greatly alter water quality." *River Research and Applications* 37(4): 605-614.
- Stitz, L., L. Fabbro, and S. Kinnear. 2017. "Macroinvertebrate community succession under variable flow regimes in subtropical Australia." *Marine and Freshwater Research* 68(6): 1153-1164.
- Stone, P.A., K.J. Locey, S. Laverty, B.D. Stanila, and M.E.B. Stone. 2015. "CONSERVATION IMPLICATIONS OF MALE-BIASED MOVEMENTS IN SONORAN MUD TURTLES (KINOSTERNON SONORIENSE) INHABITING INTERMITTENT AQUATIC HABITATS." *Herpetological Conservation and Biology* 10(2): 728-739.

Stranko, S.A., M.J. Ashton, R.H. Hilderbrand, S.L. Weglein, D.C. Kazyak, and J.V. Kilian. 2014. "Fish

and Benthic Macroinvertebrate Densities in Small Streams with and without American Eels." *Transactions of the American Fisheries Society* 143(3): 700-708.

- Strickling, H.L., and D.R. Obenour. 2018. "Leveraging Spatial and Temporal Variability to Probabilistically Characterize Nutrient Sources and Export Rates in a Developing Watershed." *Water Resources Research* 54(7): 5143-5162.
- Stuart, I.G., and C.P. Sharpe. 2020. "Riverine spawning, long distance larval drift, and floodplain recruitment of a pelagophilic fish: A case study of golden perch (Macquaria ambigua) in the arid Darling River, Australia." *Aquatic Conservation: Marine and Freshwater Ecosystems* 30(4): 675-690.
- Sugimoto, R., A. Kasai, D.R. Tait, T. Rihei, T. Hirai, K. Asai, Y. Tamura, and Y. Yamashita. 2021. "Traditional land use effects on nutrient export from watersheds to coastal seas." *Nutrient Cycling in Agroecosystems* 119(1): 7-21.
- Sun, C., L. Chen, H. Zhu, H. Xie, S. Qi, and Z. Shen. 2021. "New framework for natural-artificial transport paths and hydrological connectivity analysis in an agriculture-intensive catchment." *Water Research* 196.
- Sun, H., T. Kasahara, K. Otsuki, T. Saito, and Y. Onda. 2017. "Spatio-temporal streamflow generation in a small, steep headwater catchment in western Japan." *Hydrological Sciences Journal* 62(5): 818-829.
- Sutfin, N.A., E. Wohl, T. Fegel, N. Day, and L. Lynch. 2021. "Logjams and Channel Morphology Influence Sediment Storage, Transformation of Organic Matter, and Carbon Storage Within Mountain Stream Corridors." *Water Resources Research* 57(5).
- Swan, C.M., and B.L. Brown. 2017. "Metacommunity theory meets restoration: isolation may mediate how ecological communities respond to stream restoration." *Ecological Applications* 27(7): 2209-2219.
- Swanson, B.J., and G. Meyer. 2014. "Tributary confluences and discontinuities in channel form and sediment texture: Rio Chama, NM." *Earth Surface Processes and Landforms* 39(14): 1927-1943.
- Sweka, J.A., S. Eyler, and M.J. Millard. 2014. "An Egg-Per-Recruit Model to Evaluate the Effects of Upstream Transport and Downstream Passage Mortality of American Eel in the Susquehanna River." North American Journal of Fisheries Management 34(4): 764-773.
- Takagi, M., and H. Haga. 2019. "Carbon and nitrogen exports from forested headwater catchment in southwestern Japan." *Biogeochemistry* 145: 35-46.
- Tangi, M., R. Schmitt, S. Bizzi, and A. Castelletti. 2019. "The CASCADE toolbox for analyzing river sediment connectivity and management." *Environmental Modelling and Software* 119: 400-406.
- Tank, J.L., E. Marti, T. Riis, D. von Schiller, A.J. Reisinger, W.K. Dodds, M.R. Whiles, L.R. Ashkenas,W.B. Bowden, S.M. Collins, C.L. Crenshaw, T.A. Crowl, N.A. Griffiths, N.B. Grimm, S.K.Hamilton, S.L. Johnson, W.H. McDowell, B.M. Norman, E.J. Rosi, K.S. Simon, S.A. Thomas, and

J.R. Webster, Jr. 2018. "Partitioning assimilatory nitrogen uptake in streams: an analysis of stable isotope tracer additions across continents." *Ecological Monographs* 88(1): 120-138.

- Tank, J.L., A.J. Reisinger, E.J. Rosi, G.A. Lamberti, and F.R. Hauer. 2017. "Nutrient Limitation and Uptake." *Methods in Stream Ecology, Vol 2: Ecosystem Function, 3rd Edition*: 147-171.
- Tavernini, D.A., and J.S. Richardson. 2020. "Effects of tributary size on the resource supply and physical habitat at tributary junctions along two mainstem rivers." *Canadian Journal of Fisheries and Aquatic Sciences* 77(8): 1393-1408.
- Taylor, J., M. Moore, and S.L. Speir. 2020. "Vegetated Ditch Habitats Provide Net Nitrogen Sink and Phosphorus Storage Capacity in Agricultural Drainage Networks Despite Senescent Plant Leaching." Water 12(3).
- Taylor, J.M., T.S. Seilheimer, and W.L. Fisher. 2014. "Downstream fish assemblage response to river impoundment varies with degree of hydrologic alteration." *Hydrobiologia* 728(1): 23-39.
- Teachey, M.E., J.M. McDonald, and E.A. Ottesen. 2019. "Rapid and Stable Microbial Community Assembly in the Headwaters of a Third-Order Stream." *Applied and Environmental Microbiology* 85(11).
- Temnerud, J., C. von Bromssen, J. Folster, I. Buffam, J.O. Andersson, L. Nyberg, and K. Bishop. 2016."Map-based prediction of organic carbon in headwater streams improved by downstream observations from the river outlet." *Biogeosciences* 13(2): 399-413.
- Tennant, L.B., R.E. Gresswell, C.S. Guy, and M.H. Meeuwig. 2016. "Spawning and rearing behavior of bull trout in a headwater lake ecosystem." *Environmental Biology of Fishes* 99(1): 117-131.
- Terui, A., N. Ishiyama, H. Urabe, S. Ono, J.C. Finlay, and F. Nakamura. 2018. "Metapopulation stability in branching river networks." *Proceedings of the National Academy of Sciences* 115(26).
- Terui, A., and Y. Miyazaki. 2015. "A 'parasite-tag' approach reveals long-distance dispersal of the riverine mussel *Margaritifera laevis* by its host fish." *Hydrobiologia* 760(1): 189-196.
- Terui, A., Y. Miyazaki, A. Yoshioka, T. Kadoya, F. Jopp, and I. Washitani. 2014. "Dispersal of larvae of Margaritifera laevis by its host fish." *Freshwater Science* 33(1): 112-123.
- Terui, A., Y. Miyazaki, A. Yoshioka, K. Kaifu, S.S. Matsuzaki, and I. Washitani. 2014. "Asymmetric dispersal structures a riverine metapopulation of the freshwater pearl mussel Margaritifera laevis." *Ecology and Evolution* 4(15): 3004-3014.
- Teshima, F.A., B.J.G. Mello, F.C. Ferreira, and M. Cetra. 2016. "High beta-diversity maintains regional diversity in Brazilian tropical coastal stream fish assemblages." *Fisheries Management and Ecology* 23(6): 531-539.
- Thomas, G. 2014. "Improving restoration practice by deriving appropriate techniques from analysing the spatial organization of river networks." *Limnologica* 45: 50-60.

Thomas, N.W., A.A. Arenas, K.E. Schilling, and L.J. Weber. 2016. "Numerical investigation of the

spatial scale and time dependency of tile drainage contribution to stream flow." *Journal of Hydrology* 538: 651-666.

- Thomaz, A.T., M.R. Christie, and L.L. Knowles. 2016. "The architecture of river networks can drive the evolutionary dynamics of aquatic populations." *Evolution* 70(3): 731-739.
- Thomaz, E.L., and G.T. Peretto. 2016. "Hydrogeomorphic connectivity on roads crossing in rural headwaters and its effect on stream dynamics." *Science of the Total Environment* 550: 547-555.
- Thompson, C.J., K. Fryirs, and J. Croke. 2016. "THE DISCONNECTED SEDIMENT CONVEYOR BELT: PATTERNS OF LONGITUDINAL AND LATERAL EROSION AND DEPOSITION DURING A CATASTROPHIC FLOOD IN THE LOCKYER VALLEY, SOUTH EAST QUEENSLAND, AUSTRALIA." *River Research and Applications* 32(4): 540-551.
- Thompson, P.L., B. Rayfield, and A. Gonzalez. 2017. "Loss of habitat and connectivity erodes species diversity, ecosystem functioning, and stability in metacommunity networks." *Ecography* 40(1): 98-108.
- Thorp, J.H. 2014. "Metamorphosis in river ecology: from reaches to macrosystems." *Freshwater Biology* 59(1): 200-210.
- Tian, L., B. Zhu, and H. Akiyama. 2017. "Seasonal variations in indirect N2O emissions from an agricultural headwater ditch." *Biology and Fertility of Soils* 53(6): 651-662.
- Timusk, E.R., K.E. Smokorowski, and N.E. Jones. 2016. "An experimental test of sub-hourly changes in macroinvertebrate drift density associated with hydropeaking in a regulated river." *Journal of Freshwater Ecology* 31(4): 555-570.
- Tittel, J., C. Müller, M. Schultze, A. Musolff, and K. Knöller. 2015. "Fluvial radiocarbon and its temporal variability during contrasting hydrological conditions." *Biogeochemistry* 126: 57-69. <u>https://doi.org/10.1007/s10533-015-0137-9</u>.
- Tiwari, T., I. Buffam, R.A. Sponseller, and K. Laudon. 2017. "Inferring scale-dependent processes influencing stream water biogeochemistry from headwater to sea." *Limnology and Oceanography* 62.
- Tiwari, T., H. Laudon, K. Beven, and A.M. Agren. 2014. "Downstream changes in DOC: Inferring contributions in the face of model uncertainties." *Water Resources Research* 50(1): 514-525.
- Toczydlowski, R.H., and D.M. Waller. 2019. "Drift happens: Molecular genetic diversity and differentiation among populations of jewelweed (*Impatiens capensis* Meerb.) reflect fragmentation of floodplain forests." *Molecular Ecology* 28(10): 2459-2475.
- Tonkin, J.D., F. Altermatt, D.S. Finn, J. Heino, J.D. Olden, S.U. Pauls, and D.A. Lytle. 2018. "The role of dispersal in river network metacommunities: Patterns, processes, and pathways." *Freshwater Biology* 63(1): 141-163.
- Tonkin, J.D., J. Heino, and F. Altermatt. 2018. "Metacommunities in river networks: The importance of

network structure and connectivity on patterns and processes." Freshwater Biology 63(1): 1-5.

- Tonkin, J.D., J. Heino, A. Sundermann, P. Haase, and S.C. Jaehnig. 2016. "Context dependency in biodiversity patterns of central German stream metacommunities." *Freshwater Biology* 61(5): 607-620.
- Tonkin, J.D., S. Stoll, A. Sundermann, and P. Haase. 2014. "Dispersal distance and the pool of taxa, but not barriers, determine the colonisation of restored river reaches by benthic invertebrates." *Freshwater Biology* 59(9): 1843-1855.
- Tonkin, J.D., A. Sundermann, S.C. Jähnig, and P. Haase. 2015. "Environmental Controls on River Assemblages at the Regional Scale: An Application of the Elements of Metacommunity Structure Framework." *PLoS ONE* 10(8).
- Tornabene, B.J., R.G. Bramblett, A.V. Zale, and S.A. Leathe. 2017. "SPATIOTEMPORAL ECOLOGY OF APALONE SPINIFERA IN A LARGE, GREAT PLAINS RIVER ECOSYSTEM." *Herpetological Conservation and Biology* 12(1): 252-271.
- Tornabene, B.J., M.E. Jaeger, R.G. Bramblett, M. Nelson, N. McClenning, T. Watson, A. Ankrum, K. Frazer, A.M. Reinhold, and A.V. Zale. 2019. "Riverine turtles select habitats maintained by natural discharge regimes in an unimpounded large river." *River Research and Applications* 35(9): 1489-1498. <u>https://doi.org/10.1002/rra.3496</u>.
- Tornwall, B.M., C.M. Swan, and B.L. Brown. 2017. "Manipulation of local environment produces different diversity outcomes depending on location within a river network." *Oecologia* 184(3): 663-674.
- Trauth, N., C. Schmidt, M. Vieweg, U. Maier, and J.H. Fleckenstein. 2014. "Hyporheic transport and biogeochemical reactions in pool-riffle systems under varying ambient groundwater flow conditions." *Journal of Geophysical Research: Biogeosciences* 119(5): 910-928.
- Tremblay, C.C., M. Botrel, J.F. Lapierre, J. Franssen, and R. Maranger. 2020. "Relative influence of watershed and geomorphic features on nutrient and carbon fluxes in a pristine and moderately urbanized stream." *Science of the Total Environment* 715.
- Trigal, C., and E. Degerman. 2015. "Multiple factors and thresholds explaining fish species distributions in lowland streams." *Global Ecology and Conservation* 4: 589-601.
- Troia, M.J., A.L. Kaz, J.C. Niemeyer, and X. Giam. 2019. "Species traits and reduced habitat suitability limit efficacy of climate change refugia in streams." *Nature Ecology & Evolution* 3(9): 1321-1330.
- Tsang, L.M., K.H. Tsoi, S.K.-F. Chan, T.K.-T. Chan, and K.H. Chu. 2017. "Strong genetic differentiation among populations of the freshwater shrimp *Caridina cantonensis* in Hong Kong: implications for conservation of freshwater fauna in urban areas." *Marine and Freshwater Research* 68(1): 187-194. <u>https://doi.org/10.1071/MF15377</u>.
- Tschaplinski, P.J., and R.G. Pike. 2017. "Carnation Creek watershed experiment—long-term responses of coho salmon populations to historic forest practices." *Ecohydrology* 10(2): e1812.

https://doi.org/10.1002/eco.1812.

- Tsuboi, J., K. Morita, Y. Koseki, S. Endo, G. Sahashi, D. Kishi, T. Kikko, D. Ishizaki, M. Nunokawa, and Y. Kanno. 2020. "Spatial covariation of fish population vital rates in a stream network." *Oikos* 129(6): 924-937.
- Tsuruta, K., M.A. Hassan, S.D. Donner, and Y. Alila. 2018. "Development and Application of a Large-Scale, Physically Based, Distributed Suspended Sediment Transport Model on the Fraser River Basin, British Columbia, Canada." *Journal of Geophysical Research: Earth Surface* 123(10): 2481-2508.
- Tunaley, C., D. Tetzlaff, and C. Soulsby. 2017. "Scaling effects of riparian peatlands on stable isotopes in runoff and DOC mobilisation." *Journal of Hydrology* 549: 220-235.
- Turner, P.A., T.J. Griffis, X. Lee, J.M. Baker, R.T. Venterea, and J.D. Wood. 2015. "Indirect nitrous oxide emissions from streams within the US Corn Belt scale with stream order." *Proceedings of the National Academy of Sciences of the United States of America* 112(32): 9839-9843.
- Turner, T.F., M.J. Osborne, M.V. McPhee, and C.G. Kruse. 2015. "High and dry: intermittent watersheds provide a test case for genetic response of desert fishes to climate change." *Conservation Genetics* 16(2): 399-410.
- Uieda, V.S., and C.Y. Fujihara. 2018. "Structure of the ichthyofauna of adventitious streams in the South Pantanal." *Iheringia. Serie Zoologia* 108.
- Uno, H. 2019. "Migratory life cycle of *Ephemerella maculata* (Traver, 1934) (Ephemerellidae)." *Aquatic Insects* 40(2): 123-136. <u>https://doi.org/10.1080/01650424.2018.1563700</u>.
- Uno, H., and M.E. Power. 2015. "Mainstem-tributary linkages by mayfly migration help sustain salmonids in a warming river network." *Ecology Letters* 18(10): 1012-1020.
- Urabe, H., M. Nakajima, M. Torao, and T. Aoyama. 2014. "Application of a bioenergetics model to estimate the influence of habitat degradation by check dams and potential recovery of masu salmon populations." *Environmental Biology of Fishes* 97(5): 587-598.
- Uthe, P., R. Al-Chokhachy, A.V. Zale, B.B. Shepard, T.E. McMahon, and T. Stephens. 2016. "Life History Characteristics and Vital Rates of Yellowstone Cutthroat Trout in Two Headwater Basins." *North American Journal of Fisheries Management* 36(6): 1240-1253.
- Van Looy, K., and J. Piffady. 2017. "Metapopulation modelling of riparian tree species persistence in river networks under climate change." *Journal of Environmental Management* 202: 437-446.
- van Meerveld, H.J.I., J.W. Kirchner, M.J.P. Vis, R.S. Assendelft, and J. Seibert. 2019. "Expansion and contraction of the flowing stream network alter hillslope flowpath lengths and the shape of the travel time distribution." *Hydrology and Earth System Sciences* 23(11): 4825-4834.
- Vander Vorste, R., M. Obedzinski, S. Nossaman Pierce, S.M. Carlson, and T.E. Grantham. 2020. "Refuges and ecological traps: Extreme drought threatens persistence of an endangered fish in

intermittent streams." Global Change Biology 26(7): 3834-3845.

- Vander Vorste, R., R. Stubbington, V. Acuna, M.T. Bogan, N. Bonada, N. Cid, T. Datry, R. Storey, P.J. Wood, and A. Ruhi. 2021. "Climatic aridity increases temporal nestedness of invertebrate communities in naturally drying rivers." *Ecography* 44(6): 860-869.
- Vardakas, L., E. Kalogianni, E. Smeti, A.N. Economou, N.T. Skoulikidis, D. Koutsoubas, C. Dimitriadis, and T. Datry. 2020. "Spatial factors control the structure of fish metacommunity in a Mediterranean intermittent river." *Ecohydrology & Hydrobiology* 20(3): 346-356. <u>https://doi.org/10.1016/j.ecohyd.2020.04.005</u>.
- Vega-Retter, C., P. Munoz-Rojas, N. Rojas-Hernandez, S. Copaja, L. Flores-Prado, and D. Veliz. 2020.
 "Dammed river: Short- and long-term consequences for fish species inhabiting a river in a Mediterranean climate in central Chile." *Aquatic Conservation: Marine and Freshwater Ecosystems* 30(12): 2254-2268.
- Vera-Escalona, I., S. Senthivasan, E. Habit, and D.E. Ruzzante. 2018. "Past, present, and future of a freshwater fish metapopulation in a threatened landscape." *Conservation Biology* 32(4): 849-859.
- Vervloet, L.S.C., P.J. Binning, C.D. Børgesen, and A.L. Højberg. 2018. "Delay in catchment nitrogen load to streams following restrictions on fertilizer application." *Science of the Total Environment* 627: 1154-1166.
- Viana, D.S., J. Figuerola, K. Schwenk, M. Manca, A. Hobaek, M. Mjelde, C.D. Preston, R.J. Gornall, J.M. Croft, R.A. King, A.J. Green, and L. Santamaria. 2016. "Assembly mechanisms determining high species turnover in aquatic communities over regional and continental scales." *Ecography* 39(3): 281-288.
- Vigiak, O., C. Beverly, A. Roberts, T. Thayalakumaran, M. Dickson, J. McInnes, and L. Borselli. 2016.
 "Detecting changes in sediment sources in drought periods: The Latrobe River case study." *Environmental Modelling and Software* 85: 42-55.
- Villatarco Vazquez, A.P., D. Alfredo Fernandez, and R. Javier Albarino. 2019. "Drivers of macroinvertebrate metacommunity structure in Tierra del Fuego rivers." *Acta Oecologica* 97: 6-13.
- Vitorino Junior, O.B., R. Fernandes, C.S. Agostinho, and F.M. Pelicice. 2016. "Riverine networks constrain beta-diversity patterns among fish assemblages in a large Neotropical river." *Freshwater Biology* 61(10): 1733-1745.
- Vivancos, A., K. Górski, A. Manosalva, B. Toledo, M. Reid, and E. Habit. 2021. "Hydrological connectivity drives longitudinal movement of endangered endemic Chilean darter *Percilia irwini* (Eigenmann, 1927)." *Journal of Fish Biology* 98(1): 33-43.
- von Schiller, D., S. Bernal, C.N. Dahm, and E. Marti. 2017. "Chapter 3.2 Nutrient and Organic Matter Dynamics in Intermittent Rivers and Ephemeral Streams." Pages 135-160 in *Intermittent Rivers and Ephemeral Streams: Ecology and Management*. T. Datry, N. Bonada, and A. Boulton, eds. Academic Press.
- von Schiller, D., A. Elosegi, C. Mendoza-Lera, G. Gessner, Q. Stubbington, R. Albarino, D.C. Allen, F. Altermatt, M. Arce, S. Arnon, D. Banas, T. Datry, A. Banegas-Medina, E. Beller, M.L. Blanchette, J.F. Blanco-Libreros, J. Blessing, I.G. Boechat, K.S. Boersma, M.T. Bogan, N. Bonada, N.R. Bond, R. Corti, K. Brintrup, A. Bruder, R.M. Burrows, T. Cancellario, S.M. Carlson, S. Cauvy-Fraunie, N. Cid, M. Danger, B. de Freitas Terra, A. Dehedin, A. Foulquier, A.M. De Girolamo, R. del Campo, V. Diaz-Villanueva, C.P. Duerdoth, F. Dyer, E. Faye, C. Febria, R. Figueroa, B. Four, S. Gafny, K. Tockner, R. Gomez, L. Gomez-Gener, M.A.S. Graca, S. Guareschi, B. Gucker, F. Hoppeler, J.L. Hwan, S. Kubheka, A. Laini, S.D. Langhans, R. Marce, C. Leigh, C.J. Little, S. Lorenz, J. Marshall, E.J. Martin, A. McIntosh, E. Meyer, M. Milisa, M.C. Mlambo, M. Moleon, G. Garcia-Baquero, M. Morais, P. Negus, D. Niyogi, A. Papatheodoulou, I. Pardo, P. Paril, V. Pesic, C. Piscart, M. Polasek, P. Rodriguez-Lozano, I. Odriozola, R.J. Rolls, M.M. Sanchez-Montoya, A. Savic, O. Shumilova, A. Steward, A. Taleb, A. Uzan, R. Vander Vorste, N. Waltham, C. Woelfle-Erskine, B. Obrador, D. Zak, C. Zarfl, and A. Zoppini. 2019. "Sediment Respiration Pulses in Intermittent Rivers and Ephemeral Streams." *Global Biogeochemical Cycles* 33(10): 1251-1263.
- von Schiller, D., D. Graeber, M. Ribot, X. Timoner, V. Acuna, E. Marti, S. Sabater, and K. Tockner. 2015. "Hydrological transitions drive dissolved organic matter quantity and composition in a temporary Mediterranean stream." *Biogeochemistry* 123(3): 429-446.
- Vyšniauskienė, R., V. Rančelienė, D. Naugžemys, E. Rudaitytė-Lukošienė, J. Patamsytė, D. Butkauskas, E. Kupčinskienė, and D. Žvingila. 2020. "Genetic diversity of *Nuphar lutea* in Lithuanian river populations." *Aquatic Botany* 161: 103173. https://www.sciencedirect.com/science/article/pii/S0304377019302360.
- Wallace, J.B., and S.L. Eggert. 2015. "Terrestrial and longitudinal linkages of headwater streams." *Southeastern Naturalist* 14: 65-86.
- Wallace, M., S. Ricker, J. Garwood, A. Frimodig, and S. Allen. 2015. "Importance of the stream-estuary ecotone to juvenile coho salmon (*Oncorhynchus kisutch*) in Humboldt Bay, California." *California Fish and Game* 101(4): 241-266.
- Walley, Y., J. Tunnicliffe, and G. Brierley. 2018. "The influence of network structure upon sediment routing in two disturbed catchments, East Cape, New Zealand." *Geomorphology* 307: 38-49.
- Wallin, M.B., J. Audet, M. Peacock, E. Sahlee, and M. Winterdahl. 2020. "Carbon dioxide dynamics in an agricultural headwater stream driven by hydrology and primary production." *Biogeosciences* 17(9): 2487-2498.
- Wallin, M.B., A. Campeau, J. Audet, D. Bastviken, K. Bishop, J. Kokic, H. Laudon, E. Lundin, S. Lofgren, S. Natchimuthu, S. Sobek, C. Teutschbein, G.A. Weyhenmeyer, and T. Grabs. 2018.
 "Carbon dioxide and methane emissions of Swedish low-order streams-a national estimate and lessons learnt from more than a decade of observations." *Limnology and Oceanography: Letters* 3(3): 156-167.
- W Wan, Y., J.Q. Yang, D.W. Zou, J.J. Li, Y.J. Qiao, S.Q. An, and X. Leng. 2018. "Effects of multiple dams on the metacommunity structure of stream macroinvertebrates." *Marine and Freshwater Research* 69(5): 721-729.

- Wan, Y., J. Hu, L.L. Xu, C. Xu, A. Wan, and Y.S. Chen. 2015. "Scale-Related Contribution of Environmental and Spatial Processes to Structuring Phytoplankton Assemblages." *CLEAN - Soil, Air, Water* 43(12): 1592-1597.
- Wan, Y., L.L. Xu, J. Hu, C. Xu, A. Wan, S.Q. An, and Y.S. Chen. 2015. "The Role of Environmental and Spatial Processes in Structuring Stream Macroinvertebrates Communities in a Large River Basin." *CLEAN - Soil, Air, Water* 43(12): 1633-1639.
- Wang, A., L.H. Tang, and D.W. Yang. 2016. "Spatial and temporal variability of nitrogen load from catchment and retention along a river network: a case study in the upper Xin'anjiang catchment of China." *Hydrology Research* 47(4): 869-887.
- Wang, C., J.D. Gomez-Velez, and J.L. Wilson. 2018. "The Importance of Capturing Topographic Features for Modeling Groundwater Flow and Transport in Mountainous Watersheds." *Water Resources Research* 54(12): 10313-10338.
- Wang, J., C. Ding, J. Tao, X. Jiang, J. Heino, L. Ding, W. Su, M. Chen, K. Zhang, and D. He. 2021."Damming affects riverine macroinvertebrate metacommunity dynamics: Insights from taxonomic and functional beta diversity." *Science of the Total Environment* 763.
- Wang, S., W. Hou, H. Jiang, L. Huang, H. Dong, S. Chen, B. Wang, Y. Chen, B. Lin, and Y. Deng. 2021. "Microbial diversity accumulates in a downstream direction in the Three Gorges Reservoir." *Journal of Environmental Sciences* 101: 156-167.
- Wang, S., L. Wang, H.Y. Chang, F. Li, J.P. Tang, X.A. Zhou, X. Li, S.M. Tian, H.J. Lin, and Y. Yang. 2018. "Longitudinal variation in energy flow networks along a large subtropical river, China." *Ecological Modelling* 387: 83-95.
- Wang, T., J. Hou, P. Li, J. Zhao, Z. Li, E. Matta, L. Ma, and R. Hinkelmann. 2021. "Quantitative assessment of check dam system impacts on catchment flood characteristics a case in hilly and gully area of the Loess Plateau, China." *Natural Hazards* 105(3): 3059-3077.
- Wang, T., S.J. Kelson, G. Greer, S.E. Thompson, and S.M. Carlson. 2020. "Tributary confluences are dynamic thermal refuges for a juvenile salmonid in a warming river network." *River Research and Applications* 36(7): 1076-1086.
- Wang, X., T. Liu, L. Wang, Z.G. Liu, E.X. Zhu, S.M. Wang, Y. Cai, S.S. Zhu, and X.J. Feng. 2021.
 "Spatial-temporal variations in riverine carbon strongly influenced by local hydrological events in an alpine catchment." *Biogeosciences* 18(10): 3015-3028.
- Wang, Y.C., X.W. Gao, T.J. Li, Y. Yue, H.Q. Fang, and J.R. Ni, Jr. 2018. "Geocode-Based Aquatic Habitats in Hierarchical System of the Yellow River Basin." *Journal of Environmental Informatics* 32(2): 69-81.
- Wang, Z., B. Su, X. Xu, D. Di, H. Huang, K. Mei, R.A. Dahlgren, M. Zhang, and X. Shang. 2018.
 "Preferential accumulation of small (<300 µm) microplastics in the sediments of a coastal plain river network in eastern China." *Water Research* 144: 393-401.

- Ward, A.S., N.M. Schmadel, S.M. Wondzell, C. Harman, M.N. Gooseff, and K. Singha. 2016.
 "Hydrogeomorphic controls on hyporheic and riparian transport in two headwater mountain streams during base flow recession." *Water Resources Research* 52(2): 1479-1497.
- Ward, A.S., S.M. Wondzell, N.M. Schmadel, S. Herzog, J.P. Zarnetske, V. Baranov, P.J. Blaen, N. Brekenfeld, R. Chu, R. Derelle, J. Drummond, J.H. Fleckenstein, V. Garayburu-Caruso, E. Graham, D. Hannah, C.J. Harman, J. Hixson, J.L.A. Knapp, S. Krause, M.J. Kurz, J. Lewandowski, A.G. Li, E. Marti, M. Miller, A.M. Milner, K. Neil, L. Orsini, A.I. Packman, S. Plont, L. Renteria, K. Roche, T. Royer, C. Segura, J. Stegen, J. Toyoda, J. Wells, and N.I. Wisnoski. 2019. "Spatial and temporal variation in river corridor exchange across a 5th-order mountain stream network." *Hydrology and Earth System Sciences* 23(12): 5199-5225.
- Washburn, B.A., M.F. Cashner, and R.E. Blanton. 2020. "Small fish, large river: Surprisingly minimal genetic structure in a dispersal-limited, habitat specialist fish." *Ecology and Evolution* 10(4): 2253-2268.
- Weaver, A.K., M. Schallenberg, and C.W. Burns. 2017. "Land use, soil properties and weather conditions influence nutrient fluxes into a deep oligotrophic lake." *Marine and Freshwater Research* 68(10): 1830-1844.
- Webster, J.R., R.M. Stewart, J.D. Knoepp, and C.R. Jackson. 2019. "Effects of instream processes, discharge, and land cover on nitrogen export from southern Appalachian Mountain catchments." *Hydrological Processes* 33(2): 283-304.
- Weigelhofer, G. 2017. "The potential of agricultural headwater streams to retain soluble reactive phosphorus." *Hydrobiologia* 793: 149-160.
- Weigelhofer, G., J.P. Ramiao, A. Puritscher, and T. Hein. 2018. "How do chronic nutrient loading and the duration of nutrient pulses affect nutrient uptake in headwater streams?" *Biogeochemistry* 141(2): 249-263.
- Wekesa, S.S., T.Y. Stigter, L.O. Olang, F. Oloo, K. Fouchy, and M.E. McClain. 2020. "Water Flow Behavior and Storage Potential of the Semi-Arid Ephemeral River System in the Mara Basin of Kenya." *Frontiers in Environmental Science* 8.
- Welling, R.T., A.C. Wilcox, and J.L. Dixon. 2021. "Large wood and sediment storage in a mixed bedrock-alluvial stream, western Montana, USA." *Geomorphology* 384.
- Wells, C., S. Ketcheson, and J. Price. 2017. "Hydrology of a wetland-dominated headwater basin in the Boreal Plain, Alberta, Canada." *Journal of Hydrology* 547: 168-183.
- Welsh, M.K., P.G. Vidon, and S.K. McMillan. 2019. "Changes in riparian hydrology and biogeochemistry following storm events at a restored agricultural stream." *Environmental Science: Processes & Impacts* 21(4): 677-691.
- Welter, J.R., and S.G. Fisher. 2016. "The influence of storm characteristics on hydrological connectivity in intermittent channel networks: implications for nitrogen transport and denitrification." *Freshwater Biology* 61(8): 1214-1227.

- Werth, S., and C. Scheidegger. 2014. "Gene flow within and between catchments in the threatened riparian plant *Myricaria germanica*." *PLoS ONE* 9(6).
- Westwood, C.G., J. England, C. Hayes, T. Johns, and R. Stubbington. 2021. "The Plant Flow Index: A new method to assess the hydroecological condition of temporary rivers and streams." *Ecological Indicators* 120.
- Wheeler, K., S.W. Miller, and T.A. Crowl. 2018. "Adfluvial salmonids have engineering but not fertilization impacts in tributaries of a central Utah reservoir." *Canadian Journal of Fisheries and Aquatic Sciences* 75(3): 389-401.
- Whelan, N.V., M.P. Galaska, B.N. Sipley, J.M. Weber, P.D. Johnson, K.M. Halanych, and B.S. Helms. 2019. "Riverscape genetic variation, migration patterns, and morphological variation of the threatened Round Rocksnail, *Leptoxis ampla*." *Molecular Ecology* 28(7): 1593-1610.
- White, S.L., and T. Wagner. 2021. "Behaviour at short temporal scales drives dispersal dynamics and survival in a metapopulation of brook trout (Salvelinus fontinalis)." *Freshwater Biology* 66(2): 278-285.
- Whiterod, N.S., S. Zukowski, M. Asmus, D. Gilligan, and A.D. Miller. 2017. "Genetic analyses reveal limited dispersal and recovery potential in the large freshwater crayfish *Euastacus armatus* from the southern Murray-Darling Basin." *Marine and Freshwater Research* 68(2): 213-225.
- Whiting, J.A., and S.E. Godsey. 2016. "Discontinuous headwater stream networks with stable flowheads, Salmon River basin, Idaho." *Hydrological Processes* 30(13): 2305-2316.
- Whitney, J.E., K.B. Gido, E.C. Martin, and K.J. Hase. 2016. "The first to arrive and the last to leave: colonisation and extinction dynamics of common and rare fishes in intermittent prairie streams." *Freshwater Biology* 61(8): 1321-1334.
- Widder, S., K. Besemer, G.A. Singer, S. Ceola, E. Bertuzzo, C. Quince, W.T. Sloan, A. Rinaldo, and T.J. Battin. 2014. "Fluvial network organization imprints on microbial co-occurrence networks." *Proceedings of the National Academy of Sciences of the United States of America* 111(35): 12799-12804.
- Wiens, L.N., R. Bajno, J.T. Detwiler, M.Y. Janjua, and R.F. Tallman. 2021. "Genetic Assessment of Inconnu (*Stenodus leucichthys*) in Great Slave Lake, Northwest Territories, Canada." *Fisheries Research* 234.
- Wilbur, N.M., A.M. O'Sullivan, K.T.B. Macquarrie, T. Linnansaari, and R.A. Curry. 2020.
 "Characterizing physical habitat preferences and thermal refuge occupancy of brook trout (*Salvelinus fontinalis*) and Atlantic salmon (*Salmo salar*) at high river temperatures." *River Research and Applications* 36(5): 769-783.
- Wilcox, T.M., M.K. Young, K.S. McKelvey, D.J. Isaak, D.L. Horan, and M.K. Schwartz. 2018. "Finescale environmental DNA sampling reveals climate-mediated interactions between native and invasive trout species." *Ecosphere* 9(11).

- Wilkinson, M.E., and J.C. Bathurst. 2018. "A multi-scale nested experiment for understanding flood wave generation across four orders of magnitude of catchment area." *Nordic Hydrology* 49(3): 597-615.
- Williams, J.M., H.R. Dodd, and D.S. Finn. 2020. "A low-water crossing impacts Northern Hog Sucker Hypentelium nigricans movement in an Ozark stream." Journal of Freshwater Ecology 35(1): 157-171.
- Williams, M.R., S.J. Livingston, C.J. Penn, D.R. Smith, K.W. King, and C. Huang. 2018. "Controls of event-based nutrient transport within nested headwater agricultural watersheds of the western Lake Erie basin." *Journal of Hydrology* 559: 749-761. <u>https://doi.org/10.1016/j.jhydrol.2018.02.079</u>.
- Wilson, M.J., and M.E. McTammany. 2014. "Tributary and mainstem benthic macroinvertebrate communities linked by direct dispersal and indirect habitat alteration." *Hydrobiologia* 738(1): 75-85.
- Wilson, M.J., and M.E. McTammany. 2016. "Spatial scale and dispersal influence metacommunity dynamics of benthic invertebrates in a large river." *Freshwater Science* 35(2): 738-747.
- Witteveen, N.H., A. Freixa, and S. Sabater. 2020. "Local and regional environmental factors drive the spatial distribution of phototrophic biofilm assemblages in Mediterranean streams." *Hydrobiologia* 847(10): 2321-2336.
- Wlostowski, A.N., M.N. Gooseff, W.B. Bowden, and W.M. Wollheim. 2017. "Stream tracer breakthrough curve decomposition into mass fractions: A simple framework to analyze and compare conservative solute transport processes." *Limnology and Oceanography: Methods* 15(2): 140-153.
- Wlostowski, A.N., M.N. Gooseff, D.M. McKnight, C. Jaros, and W.B. Lyons. 2016. "Patterns of hydrologic connectivity in the McMurdo Dry Valleys, Antarctica: a synthesis of 20years of hydrologic data." *Hydrological Processes* 30(17): 2958-2975.
- Wlostowski, A.N., M.N. Gooseff, D.M. McKnight, and W.B. Lyons. 2018. "Transit Times and Rapid Chemical Equilibrium Explain Chemostasis in Glacial Meltwater Streams in the McMurdo Dry Valleys, Antarctica." *Geophysical Research Letters* 45(24): 13322-13331.
- Wohl, E. 2015. "Legacy effects on sediments in river corridors." Earth-Science Reviews 147: 30-53.
- Wohl, E. 2017. "Connectivity in rivers." Progress in Physical Geography 41(3): 345-362.
- Wohl, E. 2017. "The significance of small streams." Frontiers of Earth Science 11(3): 447-456.
- Wohl, E., and N.D. Beckman. 2014. "Leaky rivers: Implications of the loss of longitudinal fluvial disconnectivity in headwater streams." *Geomorphology* 205: 27-35.
- Wohl, E., S. Rathburn, S. Chignell, K. Garrett, D. Laurel, B. Livers, A. Patton, R. Records, M. Richards, D.M. Schook, N.A. Sutfin, and P. Wegener. 2017. "Mapping longitudinal stream connectivity in the North St. Vrain Creek watershed of Colorado." *Geomorphology* 277: 171-181.
- Wollheim, W.M. 2016. "From Headwaters to Rivers to River Networks: Scaling in Stream Ecology." *Stream Ecosystems in a Changing Environment*: 349-388.

- Wollheim, W.M., S. Bernal, D.A. Burns, J.A. Czuba, C.T. Driscoll, A.T. Hansen, R.T. Hensley, J.D. Hosen, S. Inamdar, S.S. Kaushal, L.E. Koenig, Y.H. Lu, A. Marzadri, P.A. Raymond, D. Scott, R.J. Stewart, P.G. Vidon, and E. Wohl. 2018. "River network saturation concept: factors influencing the balance of biogeochemical supply and demand of river networks." *Biogeochemistry* 141(3).
- Wollheim, W.M., G.K. Mulukutla, C. Cook, and R.O. Carey. 2017. "Aquatic Nitrate Retention at River Network Scales Across Flow Conditions Determined Using Nested In Situ Sensors." *Water Resources Research* 53(11): 9740-9756.
- Wollheim, W.M., R.J. Stewart, G.R. Aiken, K.D. Butler, N.B. Morse, and J. Salisbury. 2015. "Removal of terrestrial DOC in aquatic ecosystems of a temperate river network." *Geophysical Research Letters* 42(16): 6671-6679.
- Worthington, T.A., S.K. Brewer, N. Farless, T.B. Grabowski, and M.S. Gregory. 2014. "Interacting Effects of Discharge and Channel Morphology on Transport of Semibuoyant Fish Eggs in Large, Altered River Systems." *PLoS ONE* 9(5).
- Worthington, T.A., S.K. Brewer, T.B. Grabowski, and J. Mueller. 2014. "Backcasting the decline of a vulnerable Great Plains reproductive ecotype: identifying threats and conservation priorities." *Global Change Biology* 20(1): 89-102.
- Wubs, E.R.J., R.G.A. Fraaije, G.A. de Groot, R.H.J. Erkens, A.G. Garssen, E. Kleyheeg, B.M. Raven, and M.B. Soons. 2016. "Going against the flow: a case for upstream dispersal and detection of uncommon dispersal events." *Freshwater Biology* 61(5): 580-595.
- Xia, X., S. Zhang, S. Li, L. Zhang, G. Wang, J. Wang, and Z. Li. 2018. "The cycle of nitrogen in river systems: sources, transformation, and flux." *Environmental Science: Processes & Impacts* 20(6): 863-891.
- Xu, J. 2016. "Sediment jamming of a trunk stream by hyperconcentrated floods from small tributaries: case of the Upper Yellow River, China." *Hydrological Sciences Journal* 61(10): 1926-1940.
- Yang, L., J. Smith, M.L. Baeck, E. Morin, and D.C. Goodrich. 2017. "Flash Flooding in Arid/Semiarid Regions: Dissecting the Hydrometeorology and Hydrology of the 19 August 2014 Storm and Flood Hydroclimatology in Arizona." *Journal of Hydrometeorology* 18(12): 3103-3123.
- Yao, Y., H. Tian, H. Shi, S. Pan, R. Xu, N. Pan, and J.G. Canadell. 2020. "Increased global nitrous oxide emissions from streams and rivers in the Anthropocene." *Nature Climate Change* 10(2).
- Yarra, A.N., and D.D. Magoulick. 2018. "Stream permanence is related to crayfish occupancy and abundance in the Ozark Highlands, USA." *Freshwater Science* 37(1): 54-63.
- Ye, S., A.J. Reisinger, J.L. Tank, M.A. Baker, R.O. Hall, E.J. Rosi, and M. Sivapalan. 2017. "Scaling Dissolved Nutrient Removal in River Networks: A Comparative Modeling Investigation." *Water Resources Research* 53(11): 9623-9641.
- Yeakel, J.D., J.W. Moore, P.R. Guimarães, and M.A. de Aguiar. 2014. "Synchronisation and stability in river metapopulation networks." *Ecology Letters* 17(3): 273-283.

- Young, M.K., D.J. Isaak, S. Spaulding, C.A. Thomas, S.A. Barndt, M.C. Groce, D. Horan, and D.E. Nagel. 2018. "Effects of Climate Change on Cold-Water Fish in the Northern Rockies." Pages 37-58 in *Climate Change and Rocky Mountain Ecosystems*. J.E. Halofsky, and D.L. Peterson, eds. Advances in Global Change Research. Springer International Publishing.
- Zarnetske, J.P., M. Bouda, B.W. Abbott, J. Saiers, and P.A. Raymond. 2018. "Generality of Hydrologic Transport Limitation of Watershed Organic Carbon Flux Across Ecoregions of the United States." *Geophysical Research Letters* 45(21): 11702-11711.
- Zawalski, R., W.H. Nowlin, K. Cottenie, A. Grubh, and A.N. Schwalb. 2019. "Distinctive macroinvertebrate communities in a subtropical river network." *Journal of Freshwater Ecology* 34(1): 135-150.
- Zeiringer, B., C. Seliger, F. Greimel, and S. Schmutz. 2018. "River Hydrology, Flow Alteration, and Environmental Flow." Pages 67-89 in *Riverine Ecosystem Management*. S. Schmutz, and J. Sendzimir, eds. Aquatic Ecology Series Springer, Cham, Switzerland.
- Zell, W.O., T.B. Culver, W.E. Sanford, and J.L. Goodall. 2020. "Quantifying background nitrate removal mechanisms in an agricultural watershed with contrasting subcatchment baseflow concentrations." *Journal of Environmental Quality* 49(2): 392-403.
- Zhang, J., J. Song, L. Cheng, H. Zheng, Y. Wang, B. Huai, W. Sun, S. Qi, P. Zhao, and Q. Li. 2019."Baseflow estimation for catchments in the Loess Plateau, China." *Journal of Environmental Management* 233: 264-270.
- Zhang, J., J. Song, Y. Long, Y. Zhang, B. Zhang, and Y. Wang. 2017. "Quantifying the Spatial Variations of Hyporheic Water Exchange at Catchment Scale Using the Thermal Method: A Case Study in the Weihe River, China." Advances in Meteorology 2017.
- Zhang, W., D. Chen, and H. Li. 2018. "Spatio-temporal dynamics of water quality and their linkages with the watershed landscape in highly disturbed headwater watersheds in China." *Environmental Science and Pollution Research* 25(35): 35287-35300.
- Zhang, W., S.G. Pueppke, H. Li, J. Geng, Y. Diao, and D.W. Hyndman. 2019. "Modeling phosphorus sources and transport in a headwater catchment with rapid agricultural expansion." *Environmental Pollution* 255.
- Zhang, W., D.P. Swaney, B. Hong, and R.W. Howarth. 2017. "Anthropogenic Phosphorus Inputs to a River Basin and Their Impacts on Phosphorus Fluxes Along Its Upstream-Downstream Continuum." *Journal of Geophysical Research: Biogeosciences* 122(12): 3273-3287.
- Zhang, Y., C. Peng, S. Huang, J. Wang, X. Xiong, and D. Li. 2019. "The relative role of spatial and environmental processes on seasonal variations of phytoplankton beta diversity along different anthropogenic disturbances of subtropical rivers in China." *Environmental Science and Pollution Research* 26(2): 1422-1434.
- Zhang, Y., J. Zhang, L.Z. Wang, D.Q. Lu, D.S. Cai, and B.X. Wang. 2014. "Influences of dispersal and local environmental factors on stream macroinvertebrate communities in Qinjiang River, Guangxi,

China." Aquatic Biology 20(3): 185-194.

- Zhang, Z., J. Gao, and Y. Cai. 2019. "The effects of environmental factors and geographic distance on species turnover in an agriculturally dominated river network." *Environmental Monitoring and Assessment* 191(4).
- Zhao, K., K. Song, Y. Pan, L. Wang, L. Da, and Q. Wang. 2017. "Metacommunity structure of zooplankton in river networks: Roles of environmental and spatial factors." *Ecological Indicators* 73: 96-104.
- Zhao, L., G. Liu, and Y. Luo. 2020. "Assessing Hydrological Connectivity Mitigated by Reservoirs, Vegetation Cover, and Climate in Yan River Watershed on the Loess Plateau, China: The Network Approach." Water 12(6): 1742. https://doi.org/10.3390/w12061742.
- Zhou, S., N. Wu, M. Zhang, W. Peng, F. He, K. Guo, S. Yan, Y. Zou, and X. Qu. 2020. "Local environmental, geo-climatic and spatial factors interact to drive community distributions and diversity patterns of stream benthic algae, macroinvertebrates and fishes in a large basin, Northeast China." *Ecological Indicators* 117. <u>https://doi.org/10.1016/j.ecolind.2020.106673</u>.
- Zhou, Y., M. Liu, L. Zhou, K.S. Jang, H. Xu, K. Shi, G. Zhu, J. Deng, Y. Zhang, R.G.M. Spencer, D.N. Kothawala, E. Jeppesen, and F. Wu. 2020. "Rainstorm events shift the molecular composition and export of dissolved organic matter in a large drinking water reservoir in China: High frequency buoys and field observations." *Water Research* 187.
- Zhu, R., Q. Li, W. Wang, L. Chu, and Y. Yan. 2017. "Effects of local, river-network and catchment factors on fish assemblages in the headwater streams of the Xin'an basin, China." *Journal of Freshwater Ecology* 32(1): 300-313. <u>https://doi.org/10.1080/02705060.2016.1278408</u>.
- Zimmer, M.A., and B.L. McGlynn. 2018. "Lateral, Vertical, and Longitudinal Source Area Connectivity Drive Runoff and Carbon Export Across Watershed Scales." *Water Resources Research* 54(3): 1576-1598.
- Zoppini, A., N. Ademollo, S. Amalfitano, P. Casella, L. Patrolecco, and S. Polesello. 2014. "Organic priority substances and microbial processes in river sediments subject to contrasting hydrological conditions." *Science of the Total Environment* 484: 74-83.

Floodplain Wetlands and Open Waters

- Abrial, E., L.A. Espinola, A.P. Rabuffetti, M.F. Eurich, A.R. Paira, M.C.M. Blettler, and M.L. Amsler. 2019. "Variability of hydrological connectivity and fish dynamics in a wide subtropical-temperate floodplain." *River Research and Applications* 35(9): 1520-1529. <u>https://doi.org/10.1002/rra.3517</u>.
- Acreman, M., A. Smith, L. Charters, D. Tickner, J. Opperman, S. Acreman, F. Edwards, P. Sayers, and F. Chivava. 2021. "Evidence for the effectiveness of nature-based solutions to water issues in Africa." *Environmental Research Letters* 16(6): 063007.
- Adame, M.F., R. Reef, V.N.L. Wong, S.R. Balcombe, Sr., M.P. Turschwell, E. Kavehei, D.C. Rodriguez, J.J. Kelleway, P. Masque, and M. Ronan. 2020. "Carbon and Nitrogen Sequestration of Melaleuca

Floodplain Wetlands in Tropical Australia." Ecosystems 23(2): 454-466.

- Adame, M.F., M.E. Roberts, D.P. Hamilton, C.E. Ndehedehe, V. Reis, J. Lu, M. Griffiths, G. Curwen, and M. Ronan. 2019. "Tropical Coastal Wetlands Ameliorate Nitrogen Export During Floods." *Frontiers in Marine Science* 6.
- Addy, S., and M.E. Wilkinson. 2021. "Embankment lowering and natural self-recovery improves riverfloodplain hydro-geomorphic connectivity of a gravel bed river." *Science of the Total Environment* 770.
- Åhlén, I., P. Hambäck, J. Thorslund, A. Frampton, G. Destouni, and J. Jarsjö. 2020. "Wetlandscape size thresholds for ecosystem service delivery: Evidence from the Norrström drainage basin, Sweden." *Science of the Total Environment* 704.
- Alexander, L.C. 2015. "Science at the boundaries: scientific support for the Clean Water Rule." *Freshwater Science* 34(4): 1588-1594.
- Alexander, L.C., K.M. Fritz, K.A. Schofield, B.C. Autrey, J.E. DeMeester, H.E. Golden, D.C. Goodrich, W.G. Kepner, H.R. Kiperwas, C.R. Lane, S.D. LeDuc, S.G. Leibowitz, M.G. McManus, A.I. Pollard, C.E. Ridley, M.K. Vanderhoof, and P.J. Wigington. 2018. "Featured Collection Introduction: Connectivity of Streams and Wetlands to Downstream Waters." *Journal of the American Water Resources Association* 54(2): 287-297.
- Ali, G., and C. English. 2019. "Phytoplankton blooms in Lake Winnipeg linked to selective watergatekeeper connectivity." *Scientific Reports* 9(1): 8395. <u>https://doi.org/10.1038/s41598-019-44717-y</u>.
- Allen, C., R. Gonzales, and L. Parrott. 2020. "Modelling the contribution of ephemeral wetlands to landscape connectivity." *Ecological Modelling* 419.
- Almeida, B.D., C.B. Silva, M.R. Gimenes, and L. dos Anjos. 2016. "Waterbirds in a floodplain: influence of spatial and environmental factors through time." *Revista Brasileira De Ornitologia* 24(4): 314-322.
- Amado, A.A., M. Politano, K. Schilling, and L. Weber. 2018. "Investigating Hydrologic Connectivity of a Drained Prairie Pothole Region Wetland Complex using a Fully Integrated, Physically-Based Model." Wetlands 38(2): 233-245.
- Ameli, A.A., and I.F. Creed. 2017. "Quantifying hydrologic connectivity of wetlands to surface water systems." *Hydrol. Earth Syst. Sci.* 21: 1791-1808.
- Ameli, A.A., and I.F. Creed. 2019. "Does Wetland Location Matter When Managing Wetlands for Watershed-Scale Flood and Drought Resilience?" *Journal of the American Water Resources Association* 55(3): 529-542.
- Amezcua, F., J. Rajnohova, F. Flores-De-Santiago, F. Flores-Verdugo, and F. Amezcua-Linares. 2019."The Effect of Hydrological Connectivity on Fish Assemblages in a Floodplain System From the South-East Gulf of California, Mexico." *Frontiers in Marine Science* 6.

Andrews, C.S., L.E. Miranda, D.B. Goetz, and R. Kroeger. 2014. "Spatial patterns of lacustrine fish

assemblages in a catchment of the Mississippi Alluvial Valley." *Aquatic Conservation: Marine and Freshwater Ecosystems* 24(5): 634-644.

- Angarita, H., A.J. Wickel, J. Sieber, J. Chavarro, J.A. Maldonado-Ocampo, G.A. Herrera-R, J. Delgado, and D. Purkey. 2018. "Basin-scale impacts of hydropower development on the Mompos Depression wetlands, Colombia." *Hydrology and Earth System Sciences* 22(5): 2839-2865.
- Antiguedad, I., A. Zabaleta, M. Martinez-Santos, E. Ruiz, J. Uriarte, T. Morales, F.A. Comin, F. Carranza, C. Espanol, E. Navarro, J. Maria Bodoque, J. Ladera, D. Brito, R. Neves, L. Bernard-Jannin, X. Sun, S. Teissier, S. Sauvage, and J.M. Sanchez-Perez. 2017. "A simple multi-criteria approach to delimitate nitrate attenuation zones in alluvial floodplains. Four cases in south-western Europe." *Ecological Engineering* 103: 315-331. <u>https://doi.org/10.1016/j.ecoleng.2016.09.007</u>.
- Appling, A.P., E.S. Bernhardt, and J.A. Stanford. 2014. "Floodplain biogeochemical mosaics: A multidimensional view of alluvial soils." *Journal of Geophysical Research: Biogeosciences* 119(8): 1538-1553.
- Arheimer, B., and B.C. Pers. 2017. "Lessons learned? Effects of nutrient reductions from constructing wetlands in 1996–2006 across Sweden." *Ecological Engineering* 103(B): 404-414.
- Arrieira, R.L., L.T. Fatoreto Schwind, C.C. Bonecker, and F.A. Lansac-Toha. 2017. "Environmental factors exert predominant effects on testate amoeba metacommunities during droughts in floodplains." *Austral Ecology* 42(2): 210-217.
- Arthington, A.H., and S.R. Balcombe. 2017. "Natural flows drive the 'boom and bust' ecology of fish in Cooper Creek, an arid-zone floodplain river." *Lake Eyre Basin Rivers: Environmental, Social and Economic Importance*: 43-54.
- Ashpole, S.L., C.A. Bishop, and S.D. Murphy. 2018. "Reconnecting Amphibian Habitat through Small Pond Construction and Enhancement, South Okanagan River Valley, British Columbia, Canada." *Diversity-Basel* 10(4).
- Atkinson, C.L., B.C. van Ee, Y. Lu, and W. Zhong. 2019. "Wetland floodplain flux: temporal and spatial availability of organic matter and dissolved nutrients in an unmodified river." *Biogeochemistry* 142(3): 395-411.
- Auerswald, K., P. Moyle, S.P. Seibert, and J. Geist. 2019. "HESS Opinions: Socio-economic and ecological trade-offs of flood management - benefits of a transdisciplinary approach." *Hydrology and Earth System Sciences* 23(2): 1035-1044.
- Azinheira, D.L., D.T. Scott, W. Hession, and E.T. Hester. 2014. "Comparison of effects of inset floodplains and hyporheic exchange induced by in-stream structures on solute retention." *Water Resources Research* 50(7): 6168-6190.
- Balkic, A.G., I. Ternjej, and N. Katanic. 2019. "Alteration in microcrustacean secondary production and herbivory effects between the River Danube and its floodplain lake." *Hydrobiologia* 836(1): 185-196.
- Balkic, A.G., I. Ternjej, and M. Spoljar. 2018. "Hydrology driven changes in the rotifer trophic structure

and implications for food web interactions." Ecohydrology 11(1).

- Bashinskiy, I., T.G. Stoyko, V.A. Senkevich, A.O. Svinin, E.A. Katsman, and V.V. Osipov. 2020.
 "Structure and Dynamics of Mollusk Communities of Small Oxbow Lakes and the Determining Factors (the Khoper River Valley, Penza Oblast)." *Contemporary Problems of Ecology* 13(6): 631-642.
- Bason, C.W., D.E. Kroes, and M.M. Brinson. 2017. "The Effect of Beaver Ponds on Water Quality in Rural Coastal Plain Streams." *Southeastern Naturalist* 16(4): 584-602.
- Battauz, Y.S., S.B. Jose de Paggi, and J.C. Paggi. 2014. "Passive zooplankton community in dry littoral sediment: Reservoir of diversity and potential source of dispersal in a subtropical floodplain lake of the Middle Parana River (Santa Fe, Argentina)." *International Review of Hydrobiology* 99(3): 277-286.
- Battauz, Y.S., S.B. Jose de Paggi, and J.C. Paggi. 2017. "Macrophytes as dispersal vectors of zooplankton resting stages in a subtropical riverine floodplain." *Aquatic Ecology* 51(2): 191-201.
- Batzer, D.P., G.B. Noe, L. Lee, and M. Galatowitsch. 2018. "A Floodplain Continuum for Atlantic Coast Rivers of the Southeastern US: Predictable Changes in Floodplain Biota along a River's Length." *Wetlands* 38(1): 1-13.
- Bazzuri, M.E., N.A. Gabellone, and L.C. Solari. 2020. "Zooplankton-population dynamics in the Salado-River basin (Buenos Aires, Argentina) in relation to hydraulic works and resulting wetland function." *Aquatic Sciences* 82(3).
- Beck, W.J., P.L. Moore, K.E. Schilling, C.F. Wolter, T.M. Isenhart, K.J. Cole, and M.D. Tomer. 2019."Changes in lateral floodplain connectivity accompanying stream channel evolution: Implications for sediment and nutrient budgets." *Science of the Total Environment* 660: 1015-1028.
- Beesley, L., A.J. King, B. Gawne, J.D. Koehn, A. Price, D. Nielsen, F. Amtstaetter, and S.N. Meredith. 2014. "Optimising environmental watering of floodplain wetlands for fish." *Freshwater Biology* 59(10): 2024-2037.
- Beesley, L.S., D.C. Gwinn, A. Price, A.J. King, B. Gawne, J.D. Koehn, and D.L. Nielsen. 2014. "Juvenile fish response to wetland inundation: how antecedent conditions can inform environmental flow policies for native fish." *Journal of Applied Ecology* 51(6): 1613-1621.
- Beesley, L.S., B.J. Pusey, M.M. Douglas, D.C. Gwinn, C.A. Canham, C.S. Keogh, O.P. Pratt, M.J. Kennard, and S.A. Setterfield. 2020. "New insights into the food web of an Australian tropical river to inform water resource management." *Scientific Reports* 10(1).
- Bellmore, R.A., J.E. Compton, J.R. Brooks, E.W. Fox, R.A. Hill, D.J. Sobota, D.J. Thornbrugh, and M.H. Weber. 2018. "Nitrogen inputs drive nitrogen concentrations in U.S. streams and rivers during summer low flow conditions." *Science of the Total Environment* 639: 1349-1359.
- Benone, N.L., R. Ligeiro, L. Juen, and L.F.A. Montag. 2018. "Role of environmental and spatial processes structuring fish assemblages in streams of the eastern Amazon." *Marine and Freshwater*

Research 69(2): 243-252.

- Benson, C.E., B. Carberry, and T.A. Langen. 2018. "Public-private partnership wetland restoration programs benefit Species of Greatest Conservation Need and other wetland-associated wildlife." *Wetlands Ecology and Management* 26(2): 195-211. <u>https://doi.org/10.1007/s11273-017-9565-8</u>.
- Berezowski, T., D. Partington, J. Chormanski, and O. Batelaan. 2019. "Spatiotemporal Dynamics of the Active Perirheic Zone in a Natural Wetland Floodplain." *Water Resources Research* 55(11): 9544-9562.
- Bernal, B., C.J. Anderson, and W.J. Mitsch. 2017. "Nitrogen Dynamics in Two Created Riparian Wetlands over Space and Time." *Journal of Hydrologic Engineering* 22(1).
- Bernard-Jannin, L., X. Sun, S. Teissier, S. Sauvage, and J.M. Sanchez-Perez. 2017. "Spatio-temporal analysis of factors controlling nitrate dynamics and potential denitrification hot spots and hot moments in groundwater of an alluvial floodplain." *Ecological Engineering* 103: 372-384.
- Bertoncin, A.P.S., G.D. Pinha, M.T. Baumgartner, and R.P. Mormul. 2019. "Extreme drought events can promote homogenization of benthic macroinvertebrate assemblages in a floodplain pond in Brazil." *Hydrobiologia* 826(1): 379-393. <u>https://doi.org/10.1007/s10750-018-3756-z</u>.
- Biederman, J.A., T. Meixner, A.A. Harpold, D.E. Reed, E.D. Gutmann, J.A. Gaun, and P.D. Brooks. 2016. "Riparian zones attenuate nitrogen loss following bark beetle-induced lodgepole pine mortality." *Journal of Geophysical Research: Biogeosciences* 121(3): 933-948.
- Biehler, A., G. Chaillou, T. Buffin-Belanger, and P. Baudron. 2020. "Hydrological connectivity in the aquifer-river continuum: Impact of river stages on the geochemistry of groundwater floodplains." *Journal of Hydrology* 590.
- Billen, G., A. Ramarson, V. Thieu, S. Thery, M. Silvestre, C. Pasquier, C. Henault, and J. Garnier. 2018. "Nitrate retention at the river-watershed interface: a new conceptual modeling approach." *Biogeochemistry* 139(1): 31-51.
- Bino, G., K. Brandis, R.T. Kingsford, and J. Porter. 2020. "Waterbird synchrony across Australia's highly variable dryland rivers Risks and opportunities for conservation." *Biological Conservation* 243.
- Bishop-Taylor, R., M.G. Tulbure, and M. Broich. 2015. "Surface water network structure, landscape resistance to movement and flooding vital for maintaining ecological connectivity across Australia's largest river basin." *Landscape Ecology* 30(10): 2045-2065.
- Blackburn, M., J.L.J. Ledesma, T. Naesholm, H. Laudon, and R.A. Sponseller. 2017. "Evaluating hillslope and riparian contributions to dissolved nitrogen (N) export from a boreal forest catchment." *Journal of Geophysical Research: Biogeosciences* 122(2): 324-339.
- Blakey, R.V., R.T. Kingsford, B.S. Law, and J. Stoklosa. 2017. "Floodplain habitat is disproportionately important for bats in a large river basin." *Biological Conservation* 215: 1-10.

Blanchette, M., A.N. Rousseau, É. Foulon, S. Savary, and M. Poulin. 2019. "What would have been the

impacts of wetlands on low flow support and high flow attenuation under steady state land cover conditions?" *Journal of Environmental Management* 234: 448-457.

- Blanpied, J., J.M. Carozza, and J.M. Antoine. 2018. "Sediment connectivity in the high Pyrenees mountain range by 2013 flood analysis: role of surficial sediment storages." *Geomorphologie* 24(4): 389-402.
- Blumstock, M., D. Tetzlaff, J.J. Dick, G. Nuetzmann, and C. Soulsby. 2016. "Spatial organization of groundwater dynamics and streamflow response from different hydropedological units in a montane catchment." *Hydrological Processes* 30(21): 3735-3753.
- Bodoque, J.M., J. Ladera, J.L. Yela, J. Alonso-Azcarate, D. Brito, I. Antiguedad, R. Duran, E. Attard, B. Lauga, and J.M. Sanchez-Perez. 2017. "Recovering hydromorphological functionality to improve natural purification capacity of a highly human-modified wetland." *Ecological Engineering* 103: 332-343.
- Bolland, J.D., A.D. Nunn, M.C. Lucas, and I.G. Cowx. 2015. "The habitat use of young-of-the-year fishes during and after floods of varying timing and magnitude in a constrained lowland river." *Ecological Engineering* 75: 434-440.
- Bolpagni, R., and A. Piotti. 2016. "The importance of being natural in a human-altered riverscape: role of wetland type in supporting habitat heterogeneity and the functional diversity of vegetation." *Aquatic Conservation: Marine and Freshwater Ecosystems* 26(6): 1168-1183.
- Bomfim, F.F., F.M. Lansac-Toha, C.C. Bonecker, and F.A. Lansac-Toha. 2021. "Determinants of zooplankton functional dissimilarity during years of El Nino and La Nina in floodplain shallow lakes." *Aquatic Sciences* 83(2): 41. <u>https://doi.org/10.1007/s00027-021-00796-6</u>.
- Bonjour, S.M., M.R. Whiles, and K.B. Gido. 2020. "Influence of fishes on macroinvertebrate communities and insect emergence production in intermittent stream refuges." *Freshwater Biology* 65(8): 1412-1428.
- Bortolini, J.C., A. Pineda, L.C. Rodrigues, S. Jati, and L.F. Machado Velho. 2017. "Environmental and spatial processes influencing phytoplankton biomass along a reservoirs-river-floodplain lakes gradient: A metacommunity approach." *Freshwater Biology* 62(10): 1756-1767.
- Boudell, J.A., M.D. Dixon, S.B. Rood, and J.C. Stromberg. 2015. "Restoring functional riparian ecosystems: concepts and applications Preface." *Ecohydrology* 8(5): 747-752.
- Boughton, E.H., P.F. Quintana-Ascencio, D.G. Jenkins, P.J. Bohlen, J.E. Fauth, A. Engel, S. Shukla, G. Kiker, G. Hendricks, and H.M. Swain. 2019. "Trade-offs and synergies in a payment-for-ecosystem services program on ranchlands in the Everglades headwaters." *Ecosphere* 10(5).
- Bourgault, M.A., M. Larocque, and M. Roy. 2014. "Simulation of aquifer-peatland-river interactions under climate change." *Nordic Hydrology* 45(3): 425-440.
- Bouska, K.L., J.N. Houser, N.R. De Jager, and J. Hendrickson. 2018. "Developing a shared understanding of the Upper Mississippi River: the foundation of an ecological resilience assessment." *Ecology and*

Society 23(2).

- Bouska, K.L., J.N. Houser, N.R. De Jager, M. Van Appledorn, and J.T. Rogala. 2019. "Applying concepts of general resilience to large river ecosystems: A case study from the Upper Mississippi and Illinois rivers." *Ecological Indicators* 101: 1094-1110.
- Bouska, K.L., and T.J. Stoebner. 2015. "Characterizing Geomorphic Change from Anthropogenic Disturbances to Inform Restoration in the Upper Cache River, Illinois." *Journal of the American Water Resources Association* 51(3): 734-745.
- Bousquin, J., and K. Hychka. 2019. "A Geospatial Assessment of Flood Vulnerability Reduction by Freshwater Wetlands–A Benefit Indicators Approach." *Frontiers in Environmental Science* 7(54).
- Bregnballe, T., O. Amstrup, T.E. Holm, P. Clausen, and A.D. Fox. 2014. "Skjern River Valley, Northern Europe's most expensive wetland restoration project: benefits to breeding waterbirds." *Ornis Fennica* 91(4): 231-243.
- Brody, S.D. 2014. "The Role of Natural Functions in Shaping Community Resiliency to Floods." Pages 201-212 in *Disaster and Development*. Springer International Publishing.
- Buckmeier, D.L., N.G. Smith, B.P. Fleming, and K.A. Bodine. 2014. "Intra-annual variation in riverreservoir interface fish assemblages: Implications for fish conservation and management in regulated rivers." *River Research and Applications* 30(6): 780-790.
- Burchsted, D., and M.D. Daniels. 2014. "Classification of the alterations of beaver dams to headwater streams in northeastern Connecticut, USA." *Geomorphology* 205: 36-50.
- Byun, E., S.A. Finkelstein, S.A. Cowling, and P. Badiou. 2018. "Potential carbon loss associated with post-settlement wetland conversion in southern Ontario, Canada." *Carbon Balance and Management* 13(1).
- Cabral, A.F., L.R.P. Utz, and L.F.M. Velho. 2016. "Structure and distribution of ciliate epibiont communities in a tropical floodplain." *Hydrobiologia* 787(1): 167-180.
- Cai, Y., M.-J. Shim, L. Guo, and A. Shiller. 2016. "Floodplain influence on carbon speciation and fluxes from the lower Pearl River, Mississippi." *Geochimica et Cosmochimica Acta* 186: 189-206. <u>https://doi.org/10.1016/j.gca.2016.05.007</u>.
- Call, B.C., P. Belmont, J.C. Schmidt, and P.R. Wilcock. 2017. "Changes in floodplain inundation under nonstationary hydrology for an adjustable, alluvial river channel." *Water Resources Research* 53(5): 3811-3834.
- Callahan, M.K., D.F. Whigham, M.C. Rains, K.C. Rains, R.S. King, C.M. Walker, J.R. Maurer, and S.J. Baird. 2017. "Nitrogen Subsidies from Hillslope Alder Stands to Streamside Wetlands and Headwater Streams, Kenai Peninsula, Alaska." *Journal of the American Water Resources Association* 53(2): 478-492. <u>https://doi.org/10.1111/1752-1688.12508</u>.

Campbell, D., P. Humphries, N. McCasker, and D. Nielsen. 2018. "Subfossil chironomid head capsules

reveal assemblage differences in permanent and temporary wetlands of south-eastern Australia." *Hydrobiologia* 809(1): 91-110.

- Carli, C.M., and S.E. Bayley. 2015. "River connectivity and road crossing effects on floodplain vegetation of the upper Columbia River, Canada." *Ecoscience* 22: 97-107.
- Carlson, A.K., M.J. Fincel, C.M. Longhenry, and B.D.S. Graeb. 2016. "Effects of historic flooding on fishes and aquatic habitats in a Missouri River delta." *Journal of Freshwater Ecology* 31(2): 271-288.
- Carroll, R.W.H., L.A. Bearup, W. Brown, W. Dong, M. Bill, and K.H. Willlams. 2018. "Factors controlling seasonal groundwater and solute flux from snow-dominated basins." *Hydrological Processes* 32(14): 2187-2202.
- Carvalho, D.A., V. Williner, F. Giri, C. Vaccari, and P.A. Collins. 2017. "Quantitative food webs and invertebrate assemblages of a large River: a spatiotemporal approach in floodplain shallow lakes." *Marine and Freshwater Research* 68(2): 293-307.
- Casson, N.J., M.C. Eimers, S.A. Watmough, and M.C. Richardson. 2019. "The role of wetland coverage within the near-stream zone in predicting of seasonal stream export chemistry from forested headwater catchments." *Hydrological Processes* 33(10): 1465-1475.
- Castro, D.M.P., Y. Xuan, C. Tang, and Y. Cao. 2020. "Mechanisms of nitrate accumulation in highly urbanized rivers: Evidence from multi-isotopes in the Pearl River Delta, China." *Journal of Hydrology* 587.
- Catlin, A.K., K.J. Collier, and I.C. Duggan. 2017. "Zooplankton generation following inundation of floodplain soils: effects of vegetation type and riverine connectivity." *Marine and Freshwater Research* 68(1): 76-86. <u>https://doi.org/10.1071/MF15273</u>.
- Catlin, A.K., K.J. Collier, and I.C. Duggan. 2019. "Diet of juvenile *Galaxias maculatus* (Galaxiidae) during the upstream migration period in the lower Waikato River, New Zealand." *Marine and Freshwater Research* 70(6): 816-823. https://doi.org/10.1071/MF18305.
- Chaki, N., M. Reid, and D.L. Nielsen. 2021. "The influence of flood frequency and duration on microcrustacean egg bank composition in dryland river floodplain sediments." *Freshwater Biology* 66(7): 1382-1394.
- Chang, H., Y. Makido, and E. Foster. 2021. "Effects of land use change, wetland fragmentation, and best management practices on total suspended sediment concentrations in an urbanizing Oregon watershed, USA." *Journal of Environmental Management* 282.
- Chanut, P.C.M., T. Datry, C. Gabbud, and C.T. Robinson. 2019. "Direct and indirect effects of flood regime on macroinvertebrate assemblages in a floodplain riverscape." *Ecohydrology* 12(5).
- Chaparro, G., Z. Horváth, I. O'Farrell, R. Ptacnik, and T. Hein. 2018. "Plankton metacommunities in floodplain wetlands under contrasting hydrological conditions." *Freshwater Biology* 63(4): 380-391.

Chaparro, G., M. Soledad Fontanarrosa, D. Cataldo, and I. O'Farrell. 2015. "Hydrology driven factors

might weaken fish predation effects on zooplankton structure in a vegetated warm temperate floodplain lake." *Hydrobiologia* 752(1): 187-202.

- Chaparro, G., M. Soledad Fontanarrosa, M. Romina Schiaffino, P. de Tezanos Pinto, and I. O'Farrell. 2014. "Seasonal-dependence in the responses of biological communities to flood pulses in warm temperate floodplain lakes: implications for the 'alternative stable states' model." *Aquatic Sciences* 76(4): 579-594.
- Chembolu, V., A.K. Dubey, P.K. Gupta, S. Dutta, and R.P. Singh. 2019. "Application of satellite altimetry in understanding river-wetland flow interactions of Kosi river." *Journal of Earth System Science* 128(4).
- Chen, X., L. Chen, J. Zhao, and Z. Yu. 2015. "Modeling the hydrodynamic interactions between the main channel and the floodplain at McCarran Ranch in the lower Truckee River, Nevada." *Natural Hazards and Earth System Sciences* 15(9): 2161-2172.
- Chen, X.B., L. Chen, M.C. Stone, and K. Acharya. 2020. "Assessing connectivity between the river channel and floodplains during high flows using hydrodynamic modeling and particle tracking analysis." *Journal of Hydrology* 583.
- Chen, X., S. McGowan, L.H. Zeng, L. Xu, and X.D. Yang. 2017. "Changes in carbon and nitrogen cycling in a floodplain lake over recent decades linked to littoral expansion, declining riverine influx, and eutrophication." *Hydrological Processes* 31(17): 3110-3121.
- Cheng, C.F., M. Li, Z.S. Xue, Z.S. Zhang, X.G. Lyu, M. Jiang, and H.R. Zhang. 2020. "Impacts of Climate and Nutrients on Carbon Sequestration Rate by Wetlands: A Meta-analysis." *Chinese Geographical Science* 30(3): 483-492.
- Chizinski, C.J., P.G. Bajer, M.E. Headrick, and P.W. Sorensen. 2016. "Different Migratory Strategies of Invasive Common Carp and Native Northern Pike in the American Midwest Suggest an Opportunity for Selective Management Strategies." *North American Journal of Fisheries Management* 36(4): 769-779.
- Choi, J., and J.W. Harvey. 2014. "Relative Significance of Microtopography and Vegetation as Controls on Surface Water Flow on a Low-Gradient Floodplain." *Wetlands* 34(1): 101-115.
- Christensen, A., R.R. Twilley, C.S. Willson, and E. Castaneda-Moya. 2020. "Simulating hydrological connectivity and water age within a coastal deltaic floodplain of the Mississippi River Delta." *Estuarine, Coastal and Shelf Science* 245.
- Christian, J.M., and G.L. Adams. 2014. "Effects of pool isolation on trophic ecology of fishes in a highland stream." *Journal of Fish Biology* 85(3): 752-772.
- Christopher, S.F., J.L. Tank, U.H. Mahl, H. Yen, J.G. Arnold, M.T. Trentman, S.P. Sowa, M.E. Herbert, J.A. Ross, M.J. White, and T.V. Royer. 2017. "Modeling nutrient removal using watershed-scale implementation of the two-stage ditch." *Ecological Engineering* 108: 358-369.

Ciazela, J., M. Siepak, and P. Wojtowicz. 2018. "Tracking heavy metal contamination in a complex river-

oxbow lake system: Middle Odra Valley, Germany/Poland." *Science of the Total Environment*: 996-1006.

- Cienciala, P., and G.B. Pasternack. 2017. "Floodplain inundation response to climate, valley form, and flow regulation on a gravel-bed river in a Mediterranean-climate region." *Geomorphology* 282: 1-17.
- Clark, E., C.E. Zipper, W.L. Daniels, A.A. Prokin, and D.G. Seleznev. 2018. "Interannual Variations in Species Richness and Quantitative Parameters of Macrozoobenthos in Floodplain Lakes of the Khoper Nature Reserve." *Inland Water Biology* 11(1): 56-64.
- Clilverd, H.M., J.R. Thompson, Jr., C.M. Heppell, C.D. Sayer, and J.C. Axmacher. 2016. "Coupled Hydrological/Hydraulic Modelling of River Restoration Impacts and Floodplain Hydrodynamics." *River Research and Applications* 32(9): 1927-1948.
- Colloff, M.J., S. Lavorel, R.M. Wise, M. Dunlop, I. Overton, and K.J. Williams. 2016. "Adaptation services of floodplains and wetlands under transformational climate change." *Ecological Applications* 26(4): 1003-1017.
- Colvin, S.A.R., S.M.P. Sullivan, P.D. Shirey, R.W. Colvin, K.O. Winemiller, R.M. Hughes, K.D. Fausch, D.M. Infante, J.D. Olden, K.R. Bestgen, R.J. Danehy, and L. Eby. 2019. "Headwater Streams and Wetlands are Critical for Sustaining Fish, Fisheries, and Ecosystem Services." *Fisheries* 44(2): 73-91.
- Comin, F.A., J.M. Sánchez-Pérez, C. Español, F. Carranza, S. Sauvage, I. Antiguedad, A. Zabaleta, M. Martinez-Santos, M. Gerino, J.M. Yao, J.M. Bodoque, J. Ladera, J.L. Yela, S. Teissier, L. Bernard-Jannin, X. Sun, E. Navarro, E. Pinelli, O. Chamsi, R. Neves, D. Brito, E. Ruiz, J. Uriarte, J.J. Jiménez, M. García, A. Barcos, and R. Sorando. 2017. "Floodplain capacity to depollute water in relation to the structure of biological communities." *Ecological Engineering* 103: 301-314. https://www.sciencedirect.com/science/article/pii/S0925857416303329.
- Corline, N.J., T. Sommer, C.A. Jeffres, and J. Katz. 2017. "Zooplankton ecology and trophic resources for rearing native fish on an agricultural floodplain in the Yolo Bypass California, USA." *Wetlands Ecology and Management* 25(5): 533-545.
- Correa, A., D. Windhorst, D. Tetzlaff, P. Crespo, R. Celleri, J. Feyen, and L. Breuer. 2017. "Temporal dynamics in dominant runoff sources and flow paths in the Andean Paramo." *Water Resources Research* 53(7): 5998-6017.
- Correa, S.B., and K.O. Winemiller. 2014. "Niche partitioning among frugivorous fishes in response to fluctuating resources in the Amazonian floodplain forest." *Ecology* 95(1): 210-224.
- Corti, R., and T. Datry. 2014. "Drying of a temperate, intermittent river has little effect on adjacent riparian arthropod communities." *Freshwater Biology* 59(4): 666-678.
- Costigan, K.H., and J.E. Gerken. 2016. "Channel morphology and flow structure of an abandoned channel under varying stages." *Water Resources Research* 52(7): 5458-5472.
- Covino, T. 2017. "Hydrologic connectivity as a framework for understanding biogeochemical flux through watersheds and along fluvial networks." *Geomorphology* 277: 133-144.

- Cowley, K., A. Looman, D.T. Maher, and K. Fryirs. 2018. "Geomorphic controls on fluvial carbon exports and emissions from upland swamps in eastern Australia." *Science of the Total Environment* 618: 765-776.
- Craft, C., J. Vymazal, and L. Kropfelova. 2018. "Carbon sequestration and nutrient accumulation in floodplain and depressional wetlands." *Ecological Engineering* 114: 137-145.
- Crumpton, W.G., G.A. Stenback, S.W. Fisher, J.Z. Stenback, and D.I.S. Green. 2020. "Water quality performance of wetlands receiving nonpoint-source nitrogen loads: Nitrate and total nitrogen removal efficiency and controlling factors." *Journal of Environmental Quality* 49(3): 735-744.
- Cruz-Ramirez, A.K., M. Angel Salcedo, A.J. Sanchez, J. de Dios Mendoza, E. Barba, N. Alvarez-Pliego, and R. Florido. 2019. "Intra-annual variation of chlorophyll-a and nutrients in a hydraulically perturbed river-floodplain system in the Grijalva River basin." *Hidrobiologica* 29(3): 163-170.
- Cuevas-Lara, D., J. Alcocer, D. Cortes-Guzman, I.F. Soria-Reinoso, F. Garcia-Oliva, S. Sanchez-Carrillo, and L.A. Oseguera. 2021. "Particulate Organic Carbon in the Tropical Usumacinta River, Southeast Mexico: Concentration, Flux, and Sources." *Water* 13(11).
- Czerniawski, R., L. Slugocki, and M. Kowalska-Goralska. 2017. "Effects of beaver dams on the zooplankton assemblages in four temperate lowland streams (NW Poland)." *Biologia* 72(4): 417-430.
- Czuba, J.A., S.R. David, D.A. Edmonds, and A.S. Ward. 2019. "Dynamics of Surface-Water Connectivity in a Low-Gradient Meandering River Floodplain." *Water Resources Research* 55 (3): 1849-1870.
- da Conceicao, E.O., J. Higuti, R. de Campos, and K. Martens. 2018. "Effects of flood pulses on persistence and variability of pleuston communities in a tropical floodplain lake." *Hydrobiologia* 807(1): 175-188.
- da Cunha-Santino, M.B. 2018. "Contribution of humic substances as a sink and source of carbon in tropical floodplain lagoons." *Journal of Soils and Sediments* 18(4): 1232-1241.
- D'Ambrosio, J.L., L.R. Williams, M.G. Williams, J.D. Witter, and A.D. Ward. 2014. "Geomorphology, habitat, and spatial location influences on fish and macroinvertebrate communities in modified channels of an agriculturally-dominated watershed in Ohio, USA." *Ecological Engineering* 68: 32-46.
- Daneshvar, F., A.P. Nejadhashemi, U. Adhikari, B. Elahi, M. Abouali, M.R. Herman, E. Martinez-Martinez, T.J. Calappi, and B.G. Rohn. 2017. "Evaluating the significance of wetland restoration scenarios on phosphorus removal." *Journal of Environmental Management* 192: 184-196.
- D'Araujo Couto, T.B., J. Zuanon, J.D. Olden, and G. Ferraz. 2018. "Longitudinal variability in lateral hydrologic connectivity shapes fish occurrence in temporary floodplain ponds." *Canadian Journal of Fisheries and Aquatic Sciences* 75(2): 319-328.
- Darwiche-Criado, N., F.A. Comín, A. Masip, M. García, S.G. Eismann, and R. Sorando. 2017. "Effects of wetland restoration on nitrate removal in an irrigated agricultural area: The role of in-stream and offstream wetlands." *Ecological Engineering* 103: 426-435.

- Davenport, J.M., M.E. Hampson, A.B. King, and S.C. Bishir. 2017. "The effects of sunfish on spotted salamander oviposition, hatching time, and larval survival." *Amphibia Reptilia* 38(3): 327-337.
- Davidson, N.C., A.A. van Dam, C.M. Finlayson, and R.J. McInnes. 2019. "Worth of wetlands: revised global monetary values of coastal and inland wetland ecosystem services." *Marine and Freshwater Research* 70 (8): 1189-1194.
- Davis, A.M., B.J. Pusey, and R.G. Pearson. 2018. "Big floods, big knowledge gap: Food web dynamics in a variable river system." *Ecology of Freshwater Fish* 27(4): 898-909.
- Davis, B., C. Mattone, and M. Sheaves. 2014. "Bottom-up control regulates patterns of fish connectivity and assemblage structure in coastal wetlands." *Marine Ecology Progress Series* 500: 175-186.
- Davis, R.T., J.L. Tank, U.H. Mahl, S.G. Winikoff, and S.S. Roley. 2015. "The Influence of Two-Stage Ditches with Constructed Floodplains on Water Column Nutrients and Sediments in Agricultural Streams." *Journal of the American Water Resources Association* 51(4): 941-955.
- Dawidek, J., and B. Ferencz. 2014. "Water balance of selected floodplain lake basins in the Middle Bug River valley." *Hydrology and Earth System Sciences* 18(4): 1457-1465.
- de Amo, V.E., J. Ernandes-Silva, D.A. Moi, and R.P. Mormul. 2021. "Hydrological connectivity drives the propagule pressure of *Limnoperna fortunei* (Dunker, 1857) in a tropical river-floodplain system." *Hydrobiologia* 848(9): 2043-2053.
- De Jager, N.R., and J.N. Houser. 2016. "Patchiness in a Large Floodplain River: Associations Among Hydrology, Nutrients, and Fish Communities." *River Research and Applications* 32(9): 1915-1926.
- Dee, M.M., and J.L. Tank. 2020. "Inundation time mediates denitrification end products and carbon limitation in constructed floodplains of an agricultural stream." *Biogeochemistry* 149(2): 141-158.
- D'Elia, A.H., G.C. Liles, J.H. Viers, and D.R. Smart. 2017. "Deep carbon storage potential of buried floodplain soils." *Scientific Reports* 7(1).
- Dembowska, E.A., and P. Napiorkowski. 2015. "A case study of the planktonic communities in two hydrologically different oxbow lakes, Vistula River, Central Poland." *Journal of Limnology* 74(2): 346-357. <u>https://doi.org/10.4081/jlimnol.2014.1057</u>.
- Deng, D., Y. Pan, G. Liu, W. Liu, and L. Ma. 2020. "Seeking the hotspots of nitrogen removal: A comparison of sediment denitrification rate and denitrifier abundance among wetland types with different hydrological conditions." *Science of the Total Environment* 737.
- Denoël, M., P. Scimè, and N. Zambelli. 2016. "Newt life after fish introduction: extirpation of paedomorphosis in a mountain fish lake and newt use of satellite pools." *Current Zoology* 62(1): 61-69.
- Deoniziak, K., A. Hermaniuk, and A. Wereszczuk. 2017. "Effects of wetland restoration on the amphibian community in the Narew River Valley (Northeast Poland)." *Salamandra* 53(1): 50-58.

Depetris, P.J. 2021. "Revisiting the effect of extreme flooding on the chemistry of a large floodplain river:

the example of Argentina's middle Parana." Environmental Earth Sciences 80(2).

- Dias, J.D., C.C. Bonecker, and M.R. Miracle. 2014. "The rotifer community and its functional role in lakes of a neotropical floodplain." *International Review of Hydrobiology* 99: 72-83.
- Dias, J.D., M.R. Miracle, and C.C. Bonecker. 2017. "Do water levels control zooplankton secondary production in Neotropical floodplain lakes?" *Fundamental and Applied Limnology* 190(1): 49-62.
- Dick, J.J., D. Tetzlaff, and C. Soulsby. 2018. "Role of riparian wetlands and hydrological connectivity in the dynamics of stream thermal regimes." *Nordic Hydrology* 49(3): 634-647. <u>https://doi.org/10.2166/nh.2017.066</u>.
- Diniz, L.P., D.K. Petsch, and C.C. Bonecker. 2021. "Zooplankton beta diversity dynamics and metacommunity structure depend on spatial and temporal scales in a Neotropical floodplain." *Freshwater Biology* 66(7): 1328-1342.
- Dittrich, J., J.D. Dias, C.C. Bonecker, F.A. Lansac-Toha, and A.A. Padial. 2016. "Importance of temporal variability at different spatial scales for diversity of floodplain aquatic communities." *Freshwater Biology* 61(3): 316-327.
- Dixon, S.J., D.A. Sear, N.A. Odoni, T. Sykes, and S.N. Lane. 2016. "The effects of river restoration on catchment scale flood risk and flood hydrology." *Earth Surface Processes and Landforms* 41(7): 997-1008. <u>https://doi.org/10.1002/esp.3919</u>.
- Dodrill, M.J., C.B. Yackulic, B. Gerig, J. Korman, and C. Finch. 2015. "Do Management Actions to Restore Rare Habitat Benefit Native Fish Conservation? Distribution of Juvenile Native Fish Among Shoreline Habitats of the Colorado River." *River Research and Applications* 31(10): 1203-1217. <u>https://doi.org/10.1002/rra.2842</u>.
- Dodson, J.C., N.J. Moy, and L.P. Bulluck. 2016. "Prothonotary warbler nestling growth and condition in response to variation in aquatic and terrestrial prey availability." *Ecology and Evolution* 6(20): 7462-7474.
- Dolph, C.L., E. Boardman, M. Danesh-Yazdi, J.C. Finlay, A. Hansen, A.C. Baker, and B. Dalzell. 2019. "Phosphorus Transport in Intensively Managed Watersheds." *Water Resources Research* 55(11): 9148-9172.
- Donnelly, J.P., S.L. King, N.L. Silverman, D.P. Collins, E.M. Carrera-Gonzalez, A. Lafón-Terrazas, and J.N. Moore. 2020. "Climate and human water use diminish wetland networks supporting continental waterbird migration." *Global Change Biology*.
- Dorn, N.J., and M.I. Cook. 2015. "Hydrological disturbance diminishes predator control in wetlands." *Ecology* 96(11): 2984-2993.
- Dou, P., B. Cui, T. Xie, D. Dong, and B. Gu. 2016. "Macrobenthos Diversity Response to Hydrological Connectivity Gradient." *Wetlands* 36.
- Drexler, J.Z., C.C. Fuller, J. Orlando, A. Salas, F.C. Wurster, and J.A. Duberstein. 2017. "Estimation and

Uncertainty of Recent Carbon Accumulation and Vertical Accretion in Drained and Undrained Forested Peatlands of the Southeastern USA." *Journal of Geophysical Research: Biogeosciences* 122(10): 2563-2579.

- Dubea, T., L. Denecker, J.H.J. Van Vuren, V. Wepener, N.J. Smit, and L. Brendonck. 2017. "Spatial and temporal variation of invertebrate community structure in flood-controlled tropical floodplain wetlands." *Journal of Freshwater Ecology* 32(1): 1-15.
- Dunck, B., V.M. Algarte, M.V. Cianciaruso, and L. Rodrigues. 2016. "Functional diversity and traitenvironment relationships of periphytic algae in subtropical floodplain lakes." *Ecological Indicators* 67: 257-266.
- Dupler, K., M. Guidugli-Cook, D.R. Brown, and S.C. Richter. 2020. "Rapid Assessment of Wetland Condition Reflects Amphibian Community Composition." *Wetlands* 40(3): 451-464.
- Duvert, C., L.B. Hutley, C. Birkel, M. Rudge, N.C. Munksgaard, J.G. Wynn, S.A. Setterfield, D. Cendon, and M. Bird. 2020. "Seasonal Shift From Biogenic to Geogenic Fluvial Carbon Caused by Changing Water Sources in the Wet-Dry Tropics." *Journal of Geophysical Research: Biogeosciences* 125(2).
- Dwivedi, D., B. Arora, C.I. Steefel, B. Dafflon, and R. Versteeg. 2018. "Hot Spots and Hot Moments of Nitrogen in a Riparian Corridor." *Water Resources Research* 54 (1): 205-222.
- Ellis, I., N. Whiterod, D. Linklater, D. Bogenhuber, P. Brown, and D. Gilligan. 2015. "Spangled perch (Leiopotherapon unicolor) in the southern Murray-Darling Basin: Flood dispersal and short-term persistence outside its core range." *Austral Ecology* 40(5): 591-600.
- Escalera-Vazquez, Q.L.H., J.E. Garcia-Lopez, A. Sosa-Lopez, N. Calderon-Cortes, and D. Hinojosa-Garro. 2019. "Impact of the non-native locariid fish Pterygoplichthys pardalis in native fish community on a seasonal tropical floodplain in Mexico." *Aquatic Ecosystem Health and Management* 22(4): 462-472.
- Español, C., F.A. Comín, B. Gallardo, J. Yao, J.L. Yela, F. Carranza, A. Zabaleta, J. Ladera, M. Martínez-Santos, M. Gerino, S. Sauvage, and J.M. Sánchez-Pérez. 2017. "Does land use impact on groundwater invertebrate diversity and functionality in floodplains?" *Ecological Engineering* 103: 394-403. <u>https://doi.org/10.1016/j.ecoleng.2016.11.061</u>.
- Espinola, L.A., A.P. Rabuffetti, E. Abrial, M.L. Amsler, M.C.A. Blettler, A.R. Paira, N.R. Simoes, and L.N. Santos. 2017. "Response of fish assemblage structure to changing flood and flow pulses in a large subtropical river." *Marine and Freshwater Research* 68(2): 319-330.
- Evenson, G.R., H.E. Golden, C.R. Lane, D.L. McLaughlin, and E. D'Amico. 2018. "Depressional Wetlands Affect Watershed Hydrological, Biogeochemical, and Ecological Functions." *Ecological Applications* 28 (4): 953-966. <u>https://www.ncbi.nlm.nih.gov/pubmed/29437239</u>.
- Fabre, C., S. Sauvage, J. Guilhen, R. Cakir, M. Gerino, and J.M. Sanchez-Perez. 2020. "Daily denitrification rates in floodplains under contrasting pedo-climatic and anthropogenic contexts: modelling at the watershed scale." *Biogeochemistry* 149(3): 317-336.

- Farly, L., C. Hudon, A. Cattaneo, and G. Cabana. 2019. "Seasonality of a Floodplain Subsidy to the Fish Community of a Large Temperate River." *Ecosystems* 22(8): 1823-1837.
- Faulkner, B.R., S.G. Leibowitz, T.J. Canfield, and J.F. Groves. 2016. "Quantifying groundwater dependency of riparian surface hydrologic features using the exit gradient." *Hydrological Processes* 30(13): 2167-2177.
- Fehling, A., S. Gaffield, and S. Laubach. 2014. "Using enhanced wetlands for nitrogen removal in an agricultural watershed." *Journal of Soil and Water Conservation* 69(5).
- Fennessy, M.S., D.H. Wardrop, J.B. Moon, S. Wilson, and C. Craft. 2018. "Soil carbon sequestration in freshwater wetlands varies across a gradient of ecological condition and by ecoregion." *Ecological Engineering* 114: 129-136.
- Ferencz, B., J. Dawidek, M. Toporowska, and K. Raczyński. 2020. "Environmental implications of potamophases duration and concentration period in the floodplain lakes of the Bug River valley." *Science of the Total Environment* 746.
- Ferguson, B., W.E. Lukens, B. El Masri, and G.E. Stinchcomb. 2020. "Alluvial landform and the occurrence of paleosols in a humid-subtropical climate have an effect on long-term soil organic carbon storage." *Geoderma* 371.
- Finlay, J.C. 2020. "Targeted wetland restoration could greatly reduce nitrogen pollution." *Nature* 588(7839): 592-593.
- Fischer, J.R., B.D. Bakevich, C.P. Shea, C.L. Pierce, and M.C. Quist. 2018. "Floods, drying, habitat connectivity, and fish occupancy dynamics in restored and unrestored oxbows of West Central Iowa, USA." *Aquatic Conservation: Marine and Freshwater Ecosystems* 28(3): 630-640.
- Fleischmann, A.S., J.P.F. Breda, O.A. Passaia, S.C. Wongchuig, F.M. Fan, R.C.D. Paiva, G.F. Marques, and W. Collischonn. 2021. "Regional scale hydrodynamic modeling of the river-floodplain-reservoir continuum." *Journal of Hydrology* 596.
- Fleischmann, A.S., R.C.D. Paiva, W. Collischonn, M.V. Sorribas, and P.R.M. Pontes. 2016. "On river-floodplain interaction and hydrograph skewness." *Water Resources Research* 52(10): 7615-7630.
- Fletcher, D.E., B.E. Lindell, A.H. Lindell, P.T. Stankus, N.D. Fletcher, J.V. McArthur, and J.C. Seaman. 2019. "Basins, beaver ponds, and the storage and redistribution of trace elements in an industrially impacted coastal plain stream on the Savannah River Site, SC, USA." *Environment International* 133.
- Flint, S.A., and W.H. McDowell. 2015. "Effects of headwater wetlands on dissolved nitrogen and dissolved organic carbon concentrations in a suburban New Hampshire watershed." *Freshwater Science* 34(2): 456-471.
- Florentino, A.C., M. Petrere, C.E.D. Freitas, J.J. Toledo, L. Mateus, Y.R. Suarez, and J. Penha. 2016. "Determinants of changes in fish diversity and composition in floodplain lakes in two basins in the Pantanal wetlands, Brazil." *Environmental Biology of Fishes* 99(2-3): 265-274.

- Fossey, M., and A.N. Rousseau. 2016. "Assessing the long-term hydrological services provided by wetlands under changing climate conditions: A case study approach of a Canadian watershed." *Journal of Hydrology* 541: 1287-1302.
- Fossey, M., and A.N. Rousseau. 2016. "Can isolated and riparian wetlands mitigate the impact of climate change on watershed hydrology? A case study approach." *Journal of Environmental Management* 184: 327-339.
- Fossey, M., A.N. Rousseau, and S. Savary. 2016. "Assessment of the impact of spatio-temporal attributes of wetlands on stream flows using a hydrological modelling framework: a theoretical case study of a watershed under temperate climatic conditions." *Hydrological Processes* 30 (11): 1768-1781.
- Foubert, A., F. Lecomte, P. Brodeur, C. Le Pichon, and M. Mingelbier. 2020. "How intensive agricultural practices and flow regulation are threatening fish spawning habitats and their connectivity in the St. Lawrence River floodplain, Canada." *Landscape Ecology* 35(5): 1229-1247.
- Fraaije, R.G.A., C. Poupin, J.T.A. Verhoeven, and M.B. Soons. 2019. "Functional responses of aquatic and riparian vegetation to hydrogeomorphic restoration of channelized lowland streams and their valleys." *Journal of Applied Ecology* 56(4): 1007-1018.
- Frappart, F., F. Papa, Y. Malbeteau, J. Gabriel Leon, G. Ramillien, C. Prigent, L. Seoane, F. Seyler, and S. Calmant. 2015. "Surface Freshwater Storage Variations in the Orinoco Floodplains Using Multi-Satellite Observations." *Remote Sensing* 7(1): 89-110.
- Fritz, K.M., K.A. Schofield, L.C. Alexander, M.G. McManus, H.E. Golden, C.R. Lane, W.G. Kepner, S.D. LeDuc, J.E. DeMeester, and A.I. Pollard. 2018. "Physical and Chemical Connectivity of Streams and Riparian Wetlands to Downstream Waters: A Synthesis." *Journal of the American Water Resources Association* 54(2): 323-345.
- Fu, Y., Y. Dong, Y. Xie, Z. Xu, and L. Wang. 2020. "Impacts of Regional Groundwater Flow and River Fluctuation on Floodplain Wetlands in the Middle Reach of the Yellow River." *Water* 12(7).
- Fuller, I.C., and R.G. Death. 2018. "The science of connected ecosystems: What is the role of catchment-scale connectivity for healthy river ecology?" *Land Degradation and Development* 29(5): 1413-1426.
- Funk, A., M. Tschikof, B. Gruener, K. Boeck, T. Hein, and E. Bondar-Kunze. 2021. "Analysing the potential to restore the multi-functionality of floodplain systems by considering ecosystem service quality, quantity and trade-offs." *River Research and Applications* 37(2): 221-232.
- Furst, D., K.T. Aldridge, R.J. Shiel, G.G. Ganf, S. Mills, and J.D. Brookes. 2014. "Floodplain connectivity facilitates significant export of zooplankton to the main River Murray channel during a flood event." *Inland Waters* 4(4): 413-424.
- Gaillard, J., M. Thomas, A. Iuretig, C. Pallez, C. Feidt, X. Dauchy, and D. Banas. 2016. "Barrage fishponds: Reduction of pesticide concentration peaks and associated risk of adverse ecological effects in headwater streams." *Journal of Environmental Management* 169: 261-271.
- Gao, J., J. Holden, and M. Kirkby. 2016. "The impact of land-cover change on flood peaks in peatland

basins." Water Resources Research 52(5): 3477-3492.

- Garcia, A.M., P. Abreu, C. Ducatti, K.O. Winemiller, D.J. Hoeinghaus, M.C. Claudino, R. Bastos, F. Correa, S. Huckembeck, J. Vieira, and D. Loebmann. 2017. "Hydrologic pulsing promotes spatial connectivity and food web subsidies in a subtropical coastal ecosystem." *Marine Ecology Progress Series* 567: 17-28.
- Gardner, R.C., E. Okuno, S. Tai, M.S. Fennessy, C.A. Johnston, M.L. Otte, M. Palmer, J.E. Perry, C. Simenstad, B.R. Tanner, D. Tufford, R.E. Turner, K. Work, S.C. Yaich, and J.B. Zedler. 2019.
 "Advocating for Science: Amici Curiae Brief of Wetland and Water Scientists in Support of the Clean Water Rule." *Wetlands* 39(3): 403-414.
- Garnett, J.A., and D.P. Batzer. 2014. "Longitudinal variation in community structure of floodplain fishes along two rivers of the southeastern USA." *Canadian Journal of Fisheries and Aquatic Sciences* 71(9): 1291-1302.
- Gilbert, M.D., and A.A. Pease. 2019. "Use of fragmented reservoir habitats by larval fish assemblages across years with contrasting hydrological conditions." *Environmental Biology of Fishes* 102(6): 857-871.
- Gillespie, J.L., G.B. Noe, C.R. Hupp, A.C. Gellis, and E.R. Schenk. 2018. "Floodplain Trapping and Cycling Compared to Streambank Erosion of Sediment and Nutrients in an Agricultural Watershed." *Water Resources Bulletin* 54.
- Ginders, M.A., K.J. Collier, I.C. Duggan, and D.P. Hamilton. 2016. "Influence of Hydrological Connectivity on Plankton Communities in Natural and Reconstructed Side-Arms of a Large New Zealand River." *River Research and Applications* 32(8): 1675-1686.
- Ginders, M.A., K.J. Collier, and D.P. Hamilton. 2016. "Spatial and temporal changes in water quality along natural and restored side-arms of a large New Zealand river." *Freshwater Biology* 61(4): 411-425.
- Glinska-Lewczuk, K., P. Burandt, R. Kujawa, S. Kobus, K. Obolewski, J. Dunalska, M. Grabowska, S. Lew, and J. Chormanski. 2016. "Environmental Factors Structuring Fish Communities in Floodplain Lakes of the Undisturbed System of the Biebrza River." *Water* 8.
- Godfrey, P.C., A.H. Arthington, R.G. Pearson, F. Karim, and J. Wallace. 2017. "Fish larvae and recruitment patterns in floodplain lagoons of the Australian Wet Tropics." *Marine and Freshwater Research* 68(5): 964-979.
- Goertler, P.A.L., T.R. Sommer, W.H. Satterthwaite, and B.M. Schreier. 2018. "Seasonal floodplain-tidal slough complex supports size variation for juvenile Chinook salmon (*Oncorhynchus tshawytscha*)." *Ecology of Freshwater Fish* 27(2): 580-593.
- Gonzales, E.J., D. Tave, and G.M. Haggerty. 2014. "Endangered Rio Grande silvery minnow use constructed floodplain habitat." *Ecohydrology* 7(4): 1087-1093.
- Good, M., R. Smith, N. Pettit, and D.A. Keith. 2017. "Forests and Woodlands of Australia's Rivers and

Floodplains." Australian Vegetation, 3rd Edition: 516-543.

- Gordon, B.A., O. Dorothy, and C.F. Lenhart. 2020. "Nutrient Retention in Ecologically Functional Floodplains: A Review." *Water* 12(10): 2762.
- Gorski, K., K.J. Collier, D.P. Hamilton, and B.J. Hicks. 2014. "Effects of flow on lateral interactions of fish and shrimps with off-channel habitats in a large river-floodplain system." *Hydrobiologia* 729(1): 161-174.
- Gourdin, E., S. Huon, O. Evrard, O. Ribolzi, T. Bariac, O. Sengtaheuanghoung, and S. Ayrault. 2015. "Sources and export of particle-borne organic matter during a monsoon flood in a catchment of northern Laos." *Biogeosciences* 12(4): 1073-1089.
- Gozdziejewska, A., K. Glinska-Lewczuk, K. Obolewski, M. Grzybowski, R. Kujawa, S. Lew, and M. Grabowska. 2016. "Effects of lateral connectivity on zooplankton community structure in floodplain lakes." *Hydrobiologia* 774(1): 7-21.
- Gracz, M.B., M.F. Moffett, D.I. Siegel, and P.H. Glaser. 2015. "Analyzing peatland discharge to streams in an Alaskan watershed: An integration of end-member mixing analysis and a water balance approach." *Journal of Hydrology* 530: 667-676.
- Granado, D.C., and R. Henry. 2014. "Phytoplankton community response to hydrologic variations in oxbow lakes with different levels of connection to a tropical river." *Hydrobiologia* 721(1): 223-238.
- Granzotti, R.V., R.W. Tavares, A.C. Rodrigues, T.M. Lopes, and L.C. Gomes. 2019. "Environmental and geographic distance determining fish assemblage similarity in a floodplain: role of flow and macrophyte presence." *Environmental Biology of Fishes* 102(5): 747-757.
- Griffiths, L.N., and W.J. Mitsch. 2020. "Nutrient retention via sedimentation in a created urban stormwater treatment wetland." *Science of the Total Environment* 727.
- Grimm, K., and X. Chu. 2018. "Modeling of spatiotemporal variations in runoff contribution areas and analysis of hydrologic connectivity." *Land Degradation and Development* 29(8): 2629-2643.
- Guo, X., Y. Tang, Y. Xu, S. Zhang, J. Ma, S. Xiao, D. Ji, Z. Yang, and D. Liu. 2020. "Using stable nitrogen and oxygen isotopes to identify nitrate sources in the Lancang River, upper Mekong." *Journal of Environmental Management* 274.
- Guida, R.J., J.W.F. Remo, and S. Secchi. 2016. "Tradeoffs of strategically reconnecting rivers to their floodplains: The case of the Lower Illinois River (USA)." *Science of the Total Environment* 572: 43-55.
- Gulbin, S., A.P. Kirilenko, G. Kharel, and X. Zhang. 2019. "Wetland loss impact on long term flood risks in a closed watershed." *Environmental Science & Policy* 94: 112-122.
- Gutierrez-C, A.L., and G.A. Pinilla-A. 2016. "Effects of the local connectivity on fish assemblages in a tropical floodplain." *Caldasia* 38(2): 300-313.
- Guzy, J.C., E.A. Eskew, B.J. Halstead, and S.J. Price. 2018. "Influence of damming on anuran species

richness in riparian areas: A test of the serial discontinuity concept." *Ecology and Evolution* 8(4): 2268-2279.

- Hagy, H.M., C.S. Hine, M.M. Horath, A.P. Yetter, R.V. Smith, and J.D. Stafford. 2017. "Waterbird response indicates floodplain wetland restoration." *Hydrobiologia* 804(1): 119-137.
- Hanrahan, B.R., J.L. Tank, M.M. Dee, M.T. Trentman, E.M. Berg, and S.K. McMillan. 2018. "Restored floodplains enhance denitrification compared to naturalized floodplains in agricultural streams." *Biogeochemistry* 141(3): 419-437.
- Hansen, A., C.L. Dolph, and J.C. Finlay. 2016. "Do wetlands enhance downstream denitrification in agricultural landscapes?" *Ecosphere* 7(10).
- Hansen, A., C.L. Dolph, E. Foufoula-Georgiou, and J.C. Finlay. 2018. "Contribution of wetlands to nitrate removal at the watershed scale." *Nature Geoscience* 11(2).
- Haque, A., G. Ali, and P. Badiou. 2018. "Hydrological dynamics of prairie pothole wetlands: Dominant processes and landscape controls under contrasted conditions." *Hydrological Processes* 32(15): 2405-2422.
- Harrison, L.R., A. Pike, and D.A. Boughton. 2017. "Coupled geomorphic and habitat response to a flood pulse revealed by remote sensing." *Ecohydrology* 10(5).
- Harvey, J.W., and N.M. Schmadel. 2021. "The River Corridor's Evolving Connectivity of Lotic and Lentic Waters." *Frontiers in Water* 2(70).
- Hatten, J., J. Sloan, B. Frey, J. Straub, R. Kaminski, and A. Ezell. 2014. "Soil and Sediment Carbon and Nitrogen in Mississippi Alluvial Valley and Interior Flatwoods Bottomlands." Soil Science Society of America Journal 78.
- Hayes, D.S., J.M. Brändle, C. Seliger, B. Zeiringer, T. Ferreira, and S. Schmutz. 2018. "Advancing towards functional environmental flows for temperate floodplain rivers." *Science of the Total Environment* 633: 1089-1104.
- He, S., J. Soininen, G. Deng, and B. Wang. 2020. "Metacommunity Structure of Stream Insects across Three Hierarchical Spatial scales." *Ecology and Evolution* 10(6): 2874-2884.
- Heim, K.C., C.D. Arp, M.S. Whitman, and M.S. Wipfli. 2019. "The complementary role of lentic and lotic habitats for Arctic grayling in a complex stream-lake network in Arctic Alaska." *Ecology of Freshwater Fish* 28(2): 209-221.
- Heimhuber, V., M.G. Tulbure, and M. Broich. 2017. "Modeling multidecadal surface water inundation dynamics and key drivers on large river basin scale using multiple time series of Earth-observation and river flow data." *Water Resources Research* 53(2): 1251-1269.
- Heino, J., J. Alahuhta, L.M. Bini, Y. Cai, A.S. Heiskanen, S. Hellsten, P. Kortelainen, N. Kotamäki, K.T. Tolonen, P. Vihervaara, A. Vilmi, and D.G. Angeler. 2021. "Lakes in the era of global change: moving beyond single-lake thinking in maintaining biodiversity and ecosystem services." *Biological*

Reviews 96(1): 89-106.

- Helton, A.M., G.C. Poole, R.A. Payn, C. Izurieta, and J.A. Stanford. 2014. "Relative influences of the river channel, floodplain surface, and alluvial aquifer on simulated hydrologic residence time in a montane river floodplain." *Geomorphology* 205: 17-26.
- Helton, A.M., M.S. Wright, E.S. Bernhardt, G.C. Poole, R.M. Cory, and J.A. Stanford. 2015. "Dissolved organic carbon lability increases with water residence time in the alluvial aquifer of a river floodplain ecosystem." *Journal of Geophysical Research: Biogeosciences* 120(4): 693-706.
- Hester, E.T., C.R. Guth, D.T. Scott, and C.N. Jones. 2016. "Vertical surface water-groundwater exchange processes within a headwater floodplain induced by experimental floods." *Hydrological Processes* 30(21): 3770-3787.
- Hill, A.R. 2019. "Groundwater nitrate removal in riparian buffer zones: a review of research progress in the past 20 years." *Biogeochemistry* 143(3): 347-369.
- Hill, M.J., R.G. Death, K.L. Mathers, D.B. Ryves, J.C. White, and P.J. Wood. 2017. "Macroinvertebrate community composition and diversity in ephemeral and perennial ponds on unregulated floodplain meadows in the UK." *Hydrobiologia* 793(1): 95-108.
- Hinderer, R.K., A.R. Litt, and M. McCaffery. 2021. "Habitat selection by a threatened desert amphibian." *Ecology and Evolution* 11(1): 536-546.
- Hinson, A.L., R.A. Feagin, M. Eriksson, R.G. Najjar, M. Herrmann, T.S. Bianchi, M. Kemp, J.A. Hutchings, S. Crooks, and T. Boutton. 2017. "The spatial distribution of soil organic carbon in tidal wetland soils of the continental United States." *Global Change Biology* 23(12): 5468-5480.
- Hipondoka, M.H.T., B.C.W. van Der Waal, M.H. Ndeutapo, and L. Hango. 2018. "Sources of fish in the ephemeral western iishana region of the Cuvelai-Etosha Basin in Angola and Namibia." *African Journal of Aquatic Science* 43(3): 199-214.
- Hoagland, B., C. Schmidt, T.A. Russo, R. Adams, and J. Kaye. 2019. "Controls on nitrogen transformation rates on restored floodplains along the Cosumnes River, California." *Science of the Total Environment* 649: 979-994.
- Hoang, L., A. van Griensven, and A. Mynett. 2017. "Enhancing the SWAT model for simulating denitrification in riparian zones at the river basin scale." *Environmental Modelling and Software* 93: 163-179. http://dx.doi.org/10.1016/j.envsoft.2017.03.017.
- Hoffmann, E.P. 2018. "Environmental watering triggers rapid frog breeding in temporary wetlands within a regulated river system." *Wetlands Ecology and Management* 26(6): 1073-1087.
- Holgerson, M.A., A. Duarte, M.P. Hayes, M.J. Adams, J.A. Tyson, K.A. Douville, and A.L. Strecker. 2019. "Floodplains provide important amphibian habitat despite multiple ecological threats." *Ecosphere* 10(9).
- Hopkins, K.G., G.B. Noe, F. Franco, E.J. Pindilli, S. Gordon, M.J. Metes, P.R. Claggett, A.C. Gellis, C.R.

Hupp, and D.M. Hogan. 2018. "A method to quantify and value floodplain sediment and nutrient retention ecosystem services." *Journal of Environmental Management* 220: 65-76.

- Hosen, J.D., A.W. Armstrong, and M.A. Palmer. 2018. "Dissolved organic matter variations in coastal plain wetland watersheds: The integrated role of hydrological connectivity, land use, and seasonality." *Hydrological Processes* 32(11): 1664-1681.
- Hou, C., Y. Li, Y. Huang, H. Zhu, J. Ma, F. Yu, and X. Zhang. 2021. "Reclamation substantially increases soil organic and inorganic carbon stock in riparian floodplains." *Journal of Soils and Sediments* 21(2): 957-966.
- Houser, J.N. 2016. "Contrasts between channels and backwaters in a large, floodplain river: testing our understanding of nutrient cycling, phytoplankton abundance, and suspended solids dynamics." *Freshwater Science* 35(2): 457-473.
- Houser, J.N., L.A. Bartsch, W.B. Richardson, J.T. Rogala, and J.F. Sullivan. 2015. "Ecosystem metabolism and nutrient dynamics in the main channel and backwaters of the Upper Mississippi River." *Freshwater Biology* 60(9): 1863-1879.
- Hu, B., X. Hu, X. Nie, X. Zhang, N. Wu, Y. Hong, and H.M. Qin. 2019. "Seasonal and inter-annual community structure characteristics of zooplankton driven by water environment factors in a sub-lake of Lake Poyang, China." *PeerJ* 7.
- Hug Peter, D., E. Castella, and V.I. Slaveykova. 2017. "Lateral and longitudinal patterns of water physico-chemistry and trace metal distribution and partitioning in a large river floodplain." *Science of the Total Environment*: 248-257.
- Hupp, C.R., E.R. Schenk, D.E. Kroes, D.A. Willard, P.A. Townsend, and R.K. Peet. 2015. "Patterns of floodplain sediment deposition along the regulated lower Roanoke River, North Carolina: Annual, decadal, centennial scales." *Geomorphology* 228: 666-680.
- Hutson, A.M., L.A. Toya, and D. Tave. 2018. "Determining preferred spawning habitat of the endangered Rio Grande silvery minnow by hydrological manipulation of a conservation aquaculture facility and the implications for management." *Ecohydrology* 11(5).
- Ishiyama, N., T. Akasaka, and F. Nakamura. 2014. "Mobility-dependent response of aquatic animal species richness to a wetland network in an agricultural landscape." *Aquatic Sciences* 76(3): 437-449.
- Ishiyama, N., I. Koizumi, T. Yuta, and F. Nakamura. 2015. "Differential effects of spatial network structure and scale on population size and genetic diversity of the ninespine stickleback in a remnant wetland system." *Freshwater Biology* 60(4): 733-744.
- Ishiyama, N., M. Sueyoshi, and F. Nakamura. 2015. "To what extent do human-altered landscapes retain population connectivity? Historical changes in gene flow of wetland fish *Pungitius pungitius*." *Royal Society Open Science* 2(7).
- Jablonska, E., M. Winkowska, M. Wisniewska, J. Geurts, D. Zak, and W. Kotowski. 2021. "Impact of vegetation harvesting on nutrient removal and plant biomass quality in wetland buffer zones."

Hydrobiologia 848(14): 3273-3289.

- Jackson, A.K., C.A. Eagles-Smith, and C. Emery. 2019. "Spatial variation in aquatic invertebrate and riparian songbird mercury exposure across a river-reservoir system with a legacy of mercury contamination." *Ecotoxicology* 29(8): 1195-1204.
- Jardine, T.D., T.S. Rayner, N.E. Pettit, D. Valdez, D.P. Ward, G. Lindner, M.M. Douglas, and S.E. Bunn. 2017. "Body size drives allochthony in food webs of tropical rivers." *Oecologia* 183(2): 505-517.
- Jardine, T.D., R. Woods, J. Marshall, J. Fawcett, J. Lobegeiger, D. Valdez, and M.J. Kainz. 2015. "Reconciling the role of organic matter pathways in aquatic food webs by measuring multiple tracers in individuals." *Ecology* 96(12): 3257-3269.
- Jarihani, A.A., J.R. Larsen, J.N. Callow, T.R. McVicar, and K. Johansen. 2015. "Where does all the water go? Partitioning water transmission losses in a data-sparse, multi-channel and low-gradient dryland river system using modelling and remote sensing." *Journal of Hydrology* 529: 1511-1529. <u>https://doi.org/10.1016/j.jhydrol.2015.08.030</u>.
- Javaheri, A., and M. Babbar-Sebens. 2014. "On comparison of peak flow reductions, flood inundation maps, and velocity maps in evaluating effects of restored wetlands on channel flooding." *Ecological Engineering* 73: 132-145.
- Jaynes, D.B., and T.M. Isenhart. 2019. "Performance of Saturated Riparian Buffers in Iowa, USA." Journal of Environmental Quality 48(2): 289-296.
- Jensen, A.K., and W.I. Ford. 2019. "Quantifying Nitrate Dynamics of a Confluence Floodplain Wetland in a Disturbed Appalachian Watershed: High-Resolution Sensing and Modeling." *Transactions of the ASABE* 62(6): 1545-1565.
- Jensen, J.K., P. Engesgaard, A.R. Johnsen, V. Marti, and B. Nilsson. 2017. "Hydrological mediated denitrification in groundwater below a seasonal flooded restored riparian zone." *Water Resources Research* 53(3): 2074-2094.
- Jensen, J.K., B. Nilsson, and P. Engesgaard. 2021. "Numerical Modeling of Nitrate Removal in Anoxic Groundwater during River Flooding of Riparian Zones." *Ground Water*.
- Jeremiason, J.D., E.I. Baumann, S.D. Sebestyen, A.M. Agather, E.A. Seelen, B.J. Carlson-Stehlin, M.M. Funke, and J.B. Cotner. 2018. "Contemporary mobilization of legacy Pb stores by DOM in a boreal peatland." *Environmental Science and Technology* 52(6): 3375-3383.
- Jessop, J., G. Spyreas, G.E. Pociask, T.J. Benson, M.P. Ward, A.D. Kent, and J.W. Matthews. 2015. "Tradeoffs among ecosystem services in restored wetlands." *Biological Conservation* 191: 341-348.
- Ji, X., L.F.W. Lesack, J.M. Melack, S. Wang, W.J. Riley, and C. Shen. 2019. "Seasonal and Interannual Patterns and Controls of Hydrological Fluxes in an Amazon Floodplain Lake With a Surface-Subsurface Process Model." *Water Resources Research* 55(4): 3056-3075.
- Jiang, X., P. Zheng, L. Cao, and B. Pan. 2021. "Effects of long-term floodplain disconnection on multiple

facets of lake fish biodiversity: Decline of alpha diversity leads to a regional differentiation through time." *Science of the Total Environment* 763.

- Jin, B.S., K.O. Winemiller, B. Shao, J.K. Si, J.F. Jin, and G. Ge. 2019. "Fish assemblage structure in relation to seasonal environmental variation in sub-lakes of the Poyang Lake floodplain, China." *Fisheries Management and Ecology* 26(2): 131-140.
- Johnes, P.J., D.C. Gooddy, T.H.E. Heaton, A. Binley, M.P. Kennedy, P. Shand, and H. Prior. 2020. "Determining the Impact of Riparian Wetlands on Nutrient Cycling, Storage and Export in Permeable Agricultural Catchments." *Water* 12(1).
- Johnson, S.E., K.L. Amatangelo, P.A. Townsend, and D.M. Waller. 2016. "Large, connected floodplain forests prone to flooding best sustain plant diversity." *Ecology* 97(11): 3019-3030.
- Jones, C.N., N.G. Nelson, and L.L. Smith. 2019. "Featured Collection Introduction: The Emerging Science of Aquatic System Connectivity I." *Journal of the American Water Resources Association* 55(2): 287-293.
- Jones, C.N., D.T. Scott, B.L. Edwards, and R.F. Keim. 2014. "Perirheic mixing and biogeochemical processing in flow-through and backwater floodplain wetlands." *Water Resources Research* 50(9): 7394-7405.
- Joniak, T., and N. Kuczynska-Kippen. 2016. "Habitat features and zooplankton community structure of oxbows in the limnophase: reference to transitional phase between flooding and stabilization." *Limnetica* 35(1): 37-48.
- Juen, L., L.S. Brasil, F.F. Salles, J.D. Batista, and H.S. Ramos Cabette. 2017. "Mayfly assemblage structure of the Pantanal Mortes-Araguaia flood plain." *Marine and Freshwater Research* 68(11): 2156-2162.
- Juez, C., C. Schaerer, H. Jenny, A.J. Schleiss, and M.J. Franca. 2019. "Floodplain Land Cover and Flow Hydrodynamic Control of Overbank Sedimentation in Compound Channel Flows." *Water Resources Research* 55(11): 9072-9091.
- Kaase, C.T., and J.A. Kupfer. 2016. "Sedimentation patterns across a Coastal Plain floodplain: The importance of hydrogeomorphic influences and cross-floodplain connectivity." *Geomorphology* 269: 43-55.
- Kadykalo, A.N., and C.S. Findlay. 2016. "The flow regulation services of wetlands." *Ecosystem Services* 20: 91-103.
- Karim, F., D. Dutta, S. Marvanek, C. Petheram, C. Ticehurst, J. Lerata, S. Kim, and A. Yang. 2015. "Assessing the impacts of climate change and dams on floodplain inundation and wetland connectivity in the wet-dry tropics of northern Australia." *Journal of Hydrology* 522: 80-94. <u>https://doi.org/10.1016/j.jhydrol.2014.12.005</u>.
- Karim, F., A. Kinsey-Henderson, J. Wallace, P. Godfrey, A.H. Arthington, and R.G. Pearson. 2014. "Modelling hydrological connectivity of tropical floodplain wetlands via a combined natural and

artificial stream network." Hydrological Processes 28(23): 5696-5710.

- Karim, F., S. Marvanek, L.E. Merrin, D. Nielsen, J. Hughes, D. Stratford, and C. Pollino. 2020. "Modelling Flood-Induced Wetland Connectivity and Impacts of Climate Change and Dam." *Water* 12(5).
- Karim, F., C. Petheram, S. Marvanek, C. Ticehurst, J. Wallace, and M. Hasan. 2016. "Impact of climate change on floodplain inundation and hydrological connectivity between wetlands and rivers in a tropical river catchment." *Hydrological Processes* 30(10): 1574-1593.
- Karpack, M.N., R.R. Morrison, and R.A. McManamay. 2020. "Quantitative assessment of floodplain functionality using an index of integrity." *Ecological Indicators* 111.
- Karran, D.J., C.J. Westbrook, and A. Bedard-Haughn. 2018. "Beaver-mediated water table dynamics in a Rocky Mountain fen." *Ecohydrology* 11(2).
- Kasak, K., K. Kill, J. Parn, and U. Mander. 2018. "Efficiency of a newly established in-stream constructed wetland treating diffuse agricultural pollution." *Ecological Engineering* 119: 1-7.
- Kaushal, S.S., P.M. Mayer, P.G. Vidon, R.M. Smith, M.J. Pennino, T.A. Newcomer, S. Duan, C. Welty, and K.T. Belt. 2014. "Land use and climate variability amplify carbon, nutrient, and contaminant pulses: A review with management implications." *Journal of the American Water Resources Association* 50(3): 585-614.
- Kautza, A., S. Mazeika, and P. Sullivan. 2016. "The energetic contributions of aquatic primary producers to terrestrial food webs in a mid-size river system." *Ecology* 97(3): 694-705.
- Keith, M.K., S. Sobieszczyk, J.H. Goldman, and S.A. Rounds. 2014. "Investigating organic matter in Fanno Creek, Oregon, Part 2 of 3: Sources, sinks, and transport of organic matter with fine sediment." *Journal of Hydrology* 519: 3010-3027.
- Keizer, F.M., P.P. Schot, T. Okruszko, J. Chormanski, I. Kardel, and M.J. Wassen. 2014. "A new look at the Flood Pulse Concept: The (ir)relevance of the moving littoral in temperate zone rivers." *Ecological Engineering* 64: 85-99.
- Keizer, F.M., G.H. Van Der Lee, P.P. Schot, I. Kardel, A. Barendregt, and M.J. Wassen. 2018."Floodplain plant productivity is better predicted by particulate nutrients than by dissolved nutrients in floodwater." *Ecological Engineering* 119: 54-63.
- Keppeler, F.W., A.C. de Souza, G. Hallwass, A. Begossi, M.C. de Almeida, V.J. Isaac, and R.A. Matias Silvano. 2018. "Ecological influences of human population size and distance to urban centres on fish communities in tropical lakes." *Aquatic Conservation: Marine and Freshwater Ecosystems* 28(5): 1030-1043.
- Keys, T.A., M.F. Caudill, and D.T. Scott. 2019. "Storm effects on nitrogen flux and longitudinal variability in a river-reservoir system." *River Research and Applications* 35(6): 577-586.
- Kingsford, R.T., A. Basset, and L. Jackson. 2016. "Wetlands: conservation's poor cousins." Aquatic

Conservation: Marine and Freshwater Ecosystems 26(5): 892-916.

- Kleine, L., D. Tetzlaff, A. Smith, T. Goldhammer, and C. Soulsby. 2021. "Using isotopes to understand landscape-scale connectivity in a groundwater-dominated, lowland catchment under drought conditions." *Hydrological Processes* 35(5).
- Knutson, M.G., J.H. Herner-Thogmartin, W.E. Thogmartin, J.M. Kapfer, and J.C. Nelson. 2018. "Habitat Selection, Movement Patterns, and Hazards Encountered by Northern Leopard Frogs (*Lithobates pipiens*) in an Agricultural Landscape." *Herpetological Conservation and Biology* 13(1): 113-130.
- Kobus, S., K. Glinska-Lewczuk, M. Sidoruk, A. Skwierawski, K. Obolewski, C.M. Timofte, and P. Sowinski. 2016. "Effect of Hydrological Connectivity on Physico-Chemical Properties of Bottom Sediments of Floodplain Lakes A Case Study of the Lyna River, Northern Poland." *Environmental Engineering and Management Journal* 15(6): 1237-1246.
- Koontz, M.B., S.R. Pezeshki, and R.D. Delaune. 2014. "Nutrient Dynamics in a Restored Wetland." *Communications in Soil Science and Plant Analysis* 45(5): 609-623.
- Koster, W.M., D.R. Dawson, P. Clunie, F. Hames, J. McKenzie, P.D. Moloney, and D.A. Crook. 2015. "Movement and habitat use of the freshwater catfish (*Tandanus tandanus*) in a remnant floodplain wetland." *Ecology of Freshwater Fish* 24(3): 443-455. <u>https://doi.org/10.1111/eff.12159</u>.
- Kraus, J.M., T.S. Schmidt, D.M. Walters, R.B. Wanty, R.E. Zuellig, and R.E. Wolf. 2014. "Crossecosystem impacts of stream pollution reduce resource and contaminant flux to riparian food webs." *Ecological Applications* 24(2): 235-243.
- Kreiling, R.M., L.A. Bartsch, P.M. Perner, E.J. Hlavacek, and V.G. Christensen. 2021. "Riparian Forest Cover Modulates Phosphorus Storage and Nitrogen Cycling in Agricultural Stream Sediments." *Environmental Management* 68(2): 279-293.
- Kristensena, E.A., B. Kronvang, P. Wiberg-Larsen, H. Thodsen, C. Nielsen, E. Amor, N. Friberg, M.L. Pedersen, and A. Baattrup-Pedersen. 2014. "10 years after the largest river restoration project in Northern Europe: Hydromorphological changes on multiple scales in River Skjern." *Ecological Engineering* 66: 141-149.
- Kroes, D.E., E.R. Schenk, G.B. Noe, and A.J. Benthem. 2015. "Sediment and nutrient trapping as a result of a temporary Mississippi River floodplain restoration: The Morganza Spillway during the 2011 Mississippi River Flood." *Ecological Engineering* 82: 91-102.
- Kuiper, J.J., J.H. Janse, S. Teurlincx, J.T.A. Verhoeven, and R. Alkemade. 2014. "The impact of river regulation on the biodiversity intactness of floodplain wetlands." *Wetlands Ecology and Management* 22(6): 647-658.
- Kupfer, J.A., K.M. Meitzen, and P. Gao. 2015. "FLOODING AND SURFACE CONNECTIVITY OF TAXODIUM-NYSSA STANDS IN A SOUTHERN FLOODPLAIN FOREST ECOSYSTEM." *River Research and Applications* 31(10): 1299-1310.

Kupilas, B., F.J. Burdon, J. Thaulow, J. Hall, P.T. Mutinova, M.A.E. Forio, F. Witing, G. Risnoveanu, P.

Goethals, B.G. McKie, and N. Friberg. 2021. "Forested Riparian Zones Provide Important Habitat for Fish in Urban Streams." *Water* 13(6).

- Lacroix, R.E., M.M. Tfaily, M. McCreight, M.E. Jones, L. Spokas, and M. Keiluweit. 2019. "Shifting mineral and redox controls on carbon cycling in seasonally flooded mineral soils." *Biogeosciences* 16(13): 2573-2589.
- Lafage, D., E. Bergman, R.L. Eckstein, E.M. Osterling, J.P. Sadler, and J.J. Piccolo. 2019. "Local and landscape drivers of aquatic-to-terrestrial subsidies in riparian ecosystems: a worldwide meta-analysis." *Ecosphere* 10(4).
- Landress, C.M. 2016. "Fish Assemblage Associations with Floodplain Connectivity Following Restoration to Benefit an Endangered Catostomid." *Transactions of the American Fisheries Society* 145(1): 83-93.
- Lane, C.R., A. Hall, E. D'Amico, N. Sangwan, and V. Merwade. 2017. "Characterizing the Extent of Spatially Integrated Floodplain and Wetland Systems in the White River, Indiana, USA." *Journal of the American Water Resources Association*.
- Larocque, M., P.M. Biron, T. Buffin-Belanger, M. Needelman, C.A. Cloutier, and J.M. McKenzie. 2016.
 "Role of the geomorphic setting in controlling groundwater-surface water exchanges in riverine wetlands: A case study from two southern Quebec rivers (Canada)." *Canadian Water Resources Journal* 41(4): 528-542.
- Larsen, A., J.R. Larsen, and S.N. Lane. 2021. "Dam builders and their works: Beaver influences on the structure and function of river corridor hydrology, geomorphology, biogeochemistry and ecosystems." *Earth-Science Reviews* 218. <u>https://doi.org/10.1016/j.earscirev.2021.103623</u>.
- Larsen, L.G., J.W. Harvey, and M.M. Maglio. 2015. "Mechanisms of nutrient retention and its relation to flow connectivity in river-floodplain corridors." *Freshwater Science* 34(1): 187-205.
- Larsen, L.G., J. Ma, and D. Kaplan. 2017. "How Important Is Connectivity for Surface Water Fluxes? A Generalized Expression for Flow Through Heterogeneous Landscapes." *Geophysical Research Letters* 44(20): 10349-10358.
- Larsen, S., U. Karaus, C. Claret, F. Sporka, L. Hamerlík, and K. Tockner. 2019. "Flooding and hydrologic connectivity modulate community assembly in a dynamic river-floodplain ecosystem." *PLoS ONE* 14(4).
- Laurel, D., and E. Wohl. 2019. "The persistence of beaver-induced geomorphic heterogeneity and organic carbon stock in river corridors." *Earth Surface Processes and Landforms* 44(1): 342-353.
- Lavrnic, S., X. Nan, S. Blasioli, I. Braschi, S. Anconelli, and A. Toscano. 2020. "Performance of a full scale constructed wetland as ecological practice for agricultural drainage water treatment in Northern Italy." *Ecological Engineering* 154.
- Lawrence, C.B., E.J. Pindilli, and D.M. Hogan. 2019. "Valuation of the flood attenuation ecosystem service in Difficult Run, VA, USA." *Journal of Environmental Management* 231: 1056-1064.

- Lazar, J.G., K. Addy, A.J. Gold, P.M. Groffman, R.A. McKinney, and D.Q. Kellogg. 2015. "Beaver Ponds: Resurgent Nitrogen Sinks for Rural Watersheds in the Northeastern United States." *Journal of Environmental Quality* 44(5): 1684-1693.
- Le Cor, F., S. Slaby, J. Gaillard, X. Dauchy, C. Feidt, and D. Banas. 2020. "Barrage fishponds, a funnel effect for metal contaminants on headwater streams." *Environmental Science and Pollution Research* 27(6): 6228-6238.
- Ledesma, J.L.J., M.N. Futter, M. Blackburn, F. Lidman, T. Grabs, R.A. Sponseller, H. Laudon, K.H. Bishop, and S.J. Kohler. 2018. "Towards an Improved Conceptualization of Riparian Zones in Boreal Forest Headwaters." *Ecosystems* 21(2): 297-315.
- Lee, S., I.Y. Yeo, M.W. Lang, G.W. McCarty, A.M. Sadeghi, A. Sharifi, H. Jin, and Y. Liu. 2019. "Improving the catchment scale wetland modeling using remotely sensed data." *Environmental Modelling and Software* 122.
- Lee, S., I.Y. Yeo, M.W. Lang, A.M. Sadeghi, G.W. McCarty, G.E. Moglen, and G.R. Evenson. 2018. "Assessing the cumulative impacts of geographically isolated wetlands on watershed hydrology using the SWAT model coupled with improved wetland modules." *Journal of Environmental Management* 223: 37-48.
- Leibowitz, S.G., D.M. Mushet, and W.E. Newton. 2016. "Intermittent Surface Water Connectivity: Fill and Spill Vs. Fill and Merge Dynamics." *Wetlands* 36(2): 323-342.
- Leibowitz, S.G., P.J. Wigington, K.A. Schofield, L.C. Alexander, M.K. Vanderhoof, and H.E. Golden. 2018. "Connectivity of Streams and Wetlands to Downstream Waters: An Integrated Systems Framework." *Journal of the American Water Resources Association* 54(2): 298-322.
- Lemke, M.J., S.F. Paver, K.E. Dungey, L.F.M. Velho, A.D. Kent, L.C. Rodrigues, D.M. Kellerhals, and M.R. Randle. 2017. "Diversity and succession of pelagic microorganism communities in a newly restored Illinois River floodplain lake." *Hydrobiologia* 804(1): 35-58.
- Leuthold, S.J., S.A. Ewing, R.A. Payn, F.R. Miller, and S.G. Custer. 2021. "Seasonal connections between meteoric water and streamflow generation along a mountain headwater stream." *Hydrological Processes* 35(2).
- Levine, R., and G.A. Meyer. 2014. "Beaver dams and channel sediment dynamics on Odell Creek, Centennial Valley, Montana, USA." *Geomorphology* 205: 51-64.
- Lew, S., K. Glinska-Lewczuk, P. Burandt, K. Obolewski, A. Godziejewska, M. Lew, and J. Dunalska. 2016. "Impact of environmental factors on bacterial communities in floodplain lakes differed by hydrological connectivity." *Limnologica* 58: 20-29.
- Li, S., A. Christensen, and R.R. Twilley. 2020. "Benthic fluxes of dissolved oxygen and nutrients across hydrogeomorphic zones in a coastal deltaic floodplain within the Mississippi River delta plain." *Biogeochemistry* 149(2): 115-140.
- Li, W., Z. Su, D.S. van Maren, Z. Wang, and H.J. de Vriend. 2017. "Mechanisms of hyperconcentrated

flood propagation in a dynamic channel-floodplain system." *Advances in Water Resources* 107: 470-489.

- Li, Y., Q. Zhang, Y. Cai, Z. Tan, H. Wu, X. Liu, and J. Yao. 2019. "Hydrodynamic investigation of surface hydrological connectivity and its effects on the water quality of seasonal lakes: Insights from a complex floodplain setting (Poyang Lake, China)." *Science of the Total Environment* 660: 245-259.
- Li, Y., Q. Zhang, J. Lu, J. Yao, and Z. Tan. 2019. "Assessing surface water-groundwater interactions in a complex river-floodplain wetland-isolated lake system." *River Research and Applications* 35(1): 25-36.
- Li, Y., Q. Zhang, J. Yao, Z. Tan, and X. Liu. 2019. "Assessment of water storage response to surface hydrological connectivity in a large floodplain system (Poyang Lake, China) using hydrodynamic and geostatistical analysis." *Stochastic Environmental Research and Risk Assessment* 33: 2071-2088.
- Li, Y., E. Boswell, and A. Thompson. 2021. "Correlations between land use and stream nitrate-nitrite concentrations in the Yahara River Watershed in south-central Wisconsin." *Journal of Environmental Management* 278.
- Lind, L., E.M. Hasselquist, and H. Laudon. 2019. "Towards ecologically functional riparian zones: A meta-analysis to develop guidelines for protecting ecosystem functions and biodiversity in agricultural landscapes." *Journal of Environmental Management* 249.
- Lininger, K.B., and E. Wohl. 2019. "Floodplain dynamics in North American permafrost regions under a warming climate and implications for organic carbon stocks: A review and synthesis." *Earth-Science Reviews* 193: 24-44.
- Lininger, K.B., E. Wohl, and J.R. Rose, Jr. 2018. "Geomorphic Controls on Floodplain Soil Organic Carbon in the Yukon Flats, Interior Alaska, From Reach to River Basin Scales." *Water Resources Research* 54(3): 1934-1951.
- Liu, B., J. Liu, E. Jeppesen, Y. Chen, X. Liu, and W. Zhang. 2019. "Horizontal distribution of pelagic crustacean zooplankton biomass and body size in contrasting habitat types in Lake Poyang, China." *Environmental Science and Pollution Research* 26(3): 2270-2280.
- Liu, M., Z. Zhang, Q. He, H. Wang, X. Li, and J. Schoer. 2014. "Exogenous phosphorus inputs alter complexity of soil-dissolved organic carbon in agricultural riparian wetlands." *Chemosphere* 95: 572-580.
- Liu, W., L. Yao, Z. Wang, Z. Xiong, and G. Liu. 2015. "Human land uses enhance sediment denitrification and N2O production in Yangtze lakes primarily by influencing lake water quality." *Biogeosciences* 12(20): 6059-6070.
- Liu, X., and H. Wang. 2018. "Effects of loss of lateral hydrological connectivity on fish functional diversity." *Conservation Biology* 32(6): 1336-1345.
- Liu, X., S. Yuan, and H. Wang. 2020. "Periodic inundations drive community assembly of amphibious plants in floodplain lakes." *Hydrobiologia* 847(20): 4207-4217.

- Liu, X., Q. Zhang, Y. Li, Z. Tan, and A.D. Werner. 2020. "Satellite image-based investigation of the seasonal variations in the hydrological connectivity of a large floodplain (Poyang Lake, China)." *Journal of Hydrology* 585.
- Liu, Y., B. Cui, J. Du, Q. Wang, S. Yu, Y. Wu, G. Zhang, and A.N. Rousseau. 2020. "Quantitative assessment on basin-scale hydrological services of wetlands." *Science China Earth Sciences* 63(2): 279-291.
- Liu, Y., B. Cui, J. Du, Q. Wang, S. Yu, and W. Yang. 2020. "A method for evaluating the longitudinal functional connectivity of a river-lake-marsh system and its application in China." *Hydrological Processes* 34(26): 5278-5297.
- Liu, Y., L. Wang, H. Liu, W. Wang, C. Liang, J. Yang, and J.T.A. Verhoeven. 2015. "Comparison of Carbon Sequestration Ability and Effect of Elevation in Fenced Wetland Plant Communities of the Xilin River Floodplains: A Model Case Study." *River Research and Applications* 31(7): 858-866.
- Loczy, D., J. Dezso, P. Gyenizse, S. Czigany, and G. Toth. 2019. "Oxbow Lakes: Hydromorphology." Pages 177-198 in *The Drava River: Environmental Problems and Solutions*. D. Lóczy, ed. Spinger Geography Springer, Cham, Switzerland.
- Loken, L.C., J.T. Crawford, M.M. Dornblaser, R.G. Striegl, J.N. Houser, P.A. Turner, and E.H. Stanley. 2018. "Limited nitrate retention capacity in the Upper Mississippi River." *Environmental Research Letters* 13(7).
- Lopes, P.M., L.M. Bini, S.A. Declerck, V.F. Farjalla, L.C. Vieira, C.C. Bonecker, F.A. Lansac-Toha, F.A. Esteves, and R.L. Bozelli. 2014. "Correlates of zooplankton beta diversity in tropical lake systems." *PLoS ONE* 9(10).
- Lu, X., K. Zhong, C. Sun, and X. Li. 2016. "The correlation between land use and water quality in the typical urbanized region of China: A case study in the river net area of Pearl River delta." 2016 4th International Workshop on Earth Observation and Remote Sensing Applications (EORSA): 410-414.
- Luis Marin-Muniz, J., M.E. Hernandez, and P. Moreno-Casasola. 2014. "Comparing soil carbon sequestration in coastal freshwater wetlands with various geomorphic features and plant communities in Veracruz, Mexico." *Plant and Soil* 378: 189-203.
- Lupon, A., S. Bernal, S. Poblador, E. Marti, and F. Sabater. 2016. "The influence of riparian evapotranspiration on stream hydrology and nitrogen retention in a subhumid Mediterranean catchment." *Hydrology and Earth System Sciences* 20(9): 3831-3842.
- Lutz, S.R., N. Trauth, A. Musolff, B.M. Van Breukelen, K. Knoeller, and J.H. Fleckenstein. 2020. "How Important is Denitrification in Riparian Zones? Combining End-Member Mixing and Isotope Modeling to Quantify Nitrate Removal from Riparian Groundwater." *Water Resources Research* 56(1): e2019WR025528.
- Lynch, L.M., N.A. Sutfin, T.S. Fegel, C.M. Boot, T.P. Covino, and M.D. Wallenstein. 2019. "River channel connectivity shifts metabolite composition and dissolved organic matter chemistry." *Nature Communications* 10(1): 459. https://www.ncbi.nlm.nih.gov/pubmed/30692547.
- Macdonald, D.M.J., A.J. Dixon, and D.C. Gooddy. 2018. "Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications." *Ecological Engineering* 123: 226-237.
- Mackinnon, B.D., J. Sagin, H.M. Baulch, K.E. Lindenschmidt, and T.D. Jardine. 2016. "Influence of hydrological connectivity on winter limnology in floodplain lakes of the Saskatchewan River Delta, Saskatchewan." *Canadian Journal of Fisheries and Aquatic Sciences* 73(1): 140-152.
- Mahardja, B., J.A. Hobbs, N. Ikemiyagi, A. Benjamin, and A.J. Finger. 2019. "Role of freshwater floodplain-tidal slough complex in the persistence of the endangered delta smelt." *PLoS ONE* 14(1).
- Mahl, U.H., J.L. Tank, S.S. Roley, and R.T. Davis. 2015. "Two-stage ditch floodplains enhance nremoval capacity and reduce turbidity and dissolved P in agricultural streams." *Journal of the American Water Resources Association* 51(4): 923-940.
- Malison, R.L., B.K. Ellis, A.G. Delvecchia, H. Jacobson, B.K. Hand, G. Luikart, H.A. Woods, M. Gamboa, K. Watanabe, and J.A. Stanford. 2020. "Remarkable anoxia tolerance by stoneflies from a floodplain aquifer." *Ecology* 101(10).
- Malison, R.L., K.V. Kuzishchin, and J.A. Stanford. 2016. "Do beaver dams reduce habitat connectivity and salmon productivity in expansive river floodplains?" *PeerJ* 4.
- Malison, R.L., M.S. Lorang, D.C. Whited, and J.A. Stanford. 2014. "Beavers (Castor canadensis) influence habitat for juvenile salmon in a large Alaskan river floodplain." *Freshwater Biology* 59(6): 1229-1246.
- Manfrin, A., M. Bunzel-Drüke, A.W. Lorenz, A. Maire, M. Scharf, O. Zimball, and S. Stoll. 2020. "The effect of lateral connectedness on the taxonomic and functional structure of fish communities in a lowland river floodplain." *Science of the Total Environment* 719.
- Marks, C.O., B.C. Yellen, S.A. Wood, E.H. Martin, and K.H. Nislow. 2020. "Variation in Tree Growth along Soil Formation and Microtopographic Gradients in Riparian Forests." *Wetlands* 40(6): 1909-1922.
- Marle, P., J. Riquier, P. Timoner, H. Mayor, V.I. Slaveykova, and E. Castella. 2021. "The interplay of flow processes shapes aquatic invertebrate successions in floodplain channels A modelling applied to restoration scenarios." *Science of the Total Environment* 750.
- Leite, G.F.M., R.d.S. Rezende, H. Pains da Silva, and C.C. Muniz. 2018. "Effects of flood pulse on the community of Loricariidae (Pisces, Siluriformes) in oxbow lakes of the Pantanal, Brazil." *Boletim do Instituto de Pesca Sao Paulo* 44 (1): 35-43. https://www.pesca.sp.gov.br/boletim/index.php/bip/article/view/1276.
- Martens, K.D., and P.J. Connolly. 2014. "Juvenile Anadromous Salmonid Production in Upper Columbia River Side Channels with Different Levels of Hydrological Connection." *Transactions of the American Fisheries Society* 143(3): 757-767.

Marwick, T.R., A.V. Borges, K. Van Acker, F. Darchambeau, and S. Bouillon. 2014. "Disproportionate

Contribution of Riparian Inputs to Organic Carbon Pools in Freshwater Systems." *Ecosystems* 17(6): 974-989.

- Matteson, C.T., C.R. Jackson, D.P. Batzer, S.B. Wilde, and J.B. Jeffers. 2020. "Nitrogen and Phosphorus Gradients from a Working Farm through Wetlands to Streams in the Georgia Piedmont, USA." *Wetlands* 40(6): 2139-2149.
- Maxwell, C.M., A.G. Fernald, D. Cadol, A.M. Faist, and J.P. King. 2021. "Managing flood flow connectivity to landscapes to build buffering capacity to disturbances: An ecohydrologic modeling framework for drylands." *Journal of Environmental Management* 278.
- Mayora, G., P. Scarabotti, B. Schneider, P. Alvarenga, and M. Marchese. 2020. "Multiscale environmental heterogeneity in a large river-floodplain system." *Journal of South American Earth Sciences* 100.
- Mayr, M.J., K. Besemer, A. Sieczko, K. Demeter, and P. Peduzzi. 2020. "Bacterial community composition and function along spatiotemporal connectivity gradients in the Danube floodplain (Vienna, Austria)." *Aquatic Sciences* 82(2).
- McAlhaney, A.L., R.F. Keim, and S.T. Allen. 2020. "Species-specific growth capacity for floodplain forest trees inferred from sapwood efficiency and individual tree competition." *Forest Ecology and Management* 476.
- McCreesh, R.K., K. Fox-Dobbs, P. Wimberger, K. Woodruff, G. Holtgrieve, and T.K. Pool. 2019. "Reintroduced Beavers Rapidly Influence the Storage and Biogeochemistry of Sediments in Headwater Streams (Methow River, Washington)." *Northwest Science* 93(2): 112-121.
- McDonough, K.R., S.L. Hutchinson, J. Liang, T. Hefley, and J.M.S. Hutchinson. 2020. "Spatial Configurations of Land Cover Influence Flood Regulation Ecosystem Services." *Journal of Water Resources Planning and Management* 146(11).
- McGinness, H.M., A.D. Arthur, and M. Davies. 2015. "Does landscape water availability influence floodplain woodland bird density?" *Ecohydrology* 8(6): 977-987.
- McInerney, P.J., G. Holt, R.E. Lester, R.M. Thompson, B. Robson, D.S. Ryder, N.R. Bond, D.S. Baldwin, B. Gawne, and R. Petrie. 2020. "Basal resource quality and energy sources in three habitats of a lowland river ecosystem." *Limnology and Oceanography* 65(11): 2757-2771.
- McLean, K.I., D.M. Mushet, J.N. Sweetman, M.J. Anteau, and M.T. Wiltermuth. 2020. "Invertebrate communities of Prairie-Pothole wetlands in the age of the aquatic Homogenocene." *Hydrobiologia* 847(18): 3773-3793.
- McLean, L., L. Beevers, M. Wilkinson, G. Starrs, and G. Pender. 2015. "Riparian buffer hydrology: representing catchment-wide implementation and the influence on flood risk." *Proceedings of the 36th IAHR World Congress*: 5295-5303.
- McMillan, S.K., and G.B. Noe. 2017. "Increasing floodplain connectivity through urban stream restoration increases nutrient and sediment retention." *Ecological Engineering* 108: 284-295.

- McPhillips, L.E., P.M. Groffman, C.L. Goodale, and M.T. Walter. 2015. "Hydrologic and Biogeochemical Drivers of Riparian Denitrification in an Agricultural Watershed." *Water, Air, and Soil Pollution* 226(6).
- Meitzen, K.M., J.A. Kupfer, and P. Gao. 2018. "Modeling hydrologic connectivity and virtual fish movement across a large Southeastern floodplain, USA." *Aquatic Sciences* 80(1).
- Mendoza, M.U., K.L. Legaspi, M.G. Acojido, A.C. Cabais, J.L.E. de Guzman, A.M. Favila, S.M. Lazo, J.B. Rivera, J.C.A. Briones, and R.D.S. Papa. 2015. "Dietary Habits and Distribution of Some Fish Species in the Pansipit River-Lake Taal Connection, Luzon Island, Philippines." *Journal of Environmental Science and Management* 18(2): 1-9.
- Meng, Y., Z. He, B. Liu, L. Chen, P. Lin, and W. Luo. 2020. "Soil Salinity and Moisture Control the Processes of Soil Nitrification and Denitrification in a Riparian Wetlands in an Extremely Arid Regions in Northwestern China." *Water* 12(10).
- Mengistu, S.G., H.E. Golden, C.R. Lane, J.R. Christensen, M.L. Wine, E. D'Amico, A. Prues, S.G. Leibowitz, J.E. Compton, M.H. Weber, and R.A. Hill. 2020. "Wetland Flowpaths Mediate Nitrogen and Phosphorus Concentrations across the Upper Mississippi River Basin." *Journal of the American Water Resources Association*: 1-18.
- Mereta, S.T., L. De Meester, P. Lemmens, W. Legesse, P.L.M. Goethals, and P. Boets. 2020. "Sediment and Nutrient Retention Capacity of Natural Riverine Wetlands in Southwest Ethiopia." *Frontiers in Environmental Science* 8: 122.
- Merz, J.E., D.G. Delaney, J.D. Setka, and M.L. Workman. 2016. "Seasonal Rearing Habitat in a Large Mediterranean-Climate River: Management Implications at the Southern Extent of Pacific Salmon (Oncorhynchus spp.)." River Research and Applications 32(6): 1220-1231.
- Middleton, B.A. 2020. "Trends of litter decomposition and soil organic matter stocks across forested swamp environments of the southeastern US." *PLoS ONE* 15(1).
- Mihaljevic, M., F. Stevic, D. Spoljaric, and T.Z. Pfeiffer. 2015. "Spatial Pattern of Phytoplankton Based on the Morphology-Based Functional Approach along a River–Floodplain Gradient." *River Research and Applications* 31(2): 228-238. <u>https://doi.org/10.1002/rra.2739</u>.
- Mihelcic, J.R., and M. Rains. 2020. "Where's the Science? Recent Changes to Clean Water Act Threaten Wetlands and Thousands of Miles of Our Nation's Rivers and Streams." *Environmental Engineering Science* 37(3): 173-177.
- Milde, A.S., W.B. Richardson, E.A. Strauss, J.H. Larson, J.M. Vallazza, and B.C. Knights. 2017. "Spatial and Temporal Dynamics of Suspended Particle Characteristics and Composition in Navigation Pool 19 of the Upper Mississippi River." *River Research and Applications* 33(5): 740-752.
- Miller, R.B., D.M. Heeren, G.A. Fox, T. Halihan, D.E. Storm, and A.R. Mittelstet. 2014. "The hydraulic conductivity structure of gravel-dominated vadose zones within alluvial floodplains." *Journal of Hydrology* 513: 229-240.

- Miranda, L.E., and D.J. Dembkowski. 2016. "Evidence for Serial Discontinuity in the Fish Community of a Heavily Impounded River." *River Research and Applications* 32(6): 1187-1195.
- Miranda, L.E., S.L. Wigen, and J.D. Dagel. 2014. "RESERVOIR FLOODPLAINS SUPPORT DISTINCT FISH ASSEMBLAGES." *River Research and Applications* 30(3): 338-346.
- Modrak, P., S. Brunzel, and A.W. Lorenz. 2017. "Riparian plant species preferences indicate diversification of site conditions after river restoration." *Ecohydrology* 10(5).
- Moody, E.K., and J.L. Sabo. 2017. "Dissimilarity in the riparian arthropod communities along surface water permanence gradients in aridland streams." *Ecohydrology* 10(4).
- Moon, J.B., D.H. Wardrop, M.S. Fennessy, H.M. Ingram, A. Britson, and M.H. Okoro. 2020. "Variation in surface and subsurface nitrogen cycling in headwater floodplain wetlands due to soil type and wetland condition." *Wetlands Ecology and Management* 28(5): 727-751.
- Morat, F., P. Gibert, N. Reynaud, B. Testi, P. Favriou, V. Raymond, G. Carrel, and A. Maire. 2018.
 "Spatial distribution, total length frequencies and otolith morphometry as tools to analyse the effects of a flash flood on populations of roach (Rutilus rutilus)." *Ecology of Freshwater Fish* 27(1): 421-432.
- Mori, N., T. Simcic, A. Brancelj, C.T. Robinson, and M. Doering. 2017. "Spatiotemporal heterogeneity of actual and potential respiration in two contrasting floodplains." *Hydrological Processes* 31(14): 2622-2636.
- Morrison, A.C., A.J. Gold, and M.C. Pelletier. 2016. "Evaluating Key Watershed Components of Low Flow Regimes in New England Streams." *Journal of Environmental Quality* 45(3): 1021-1028.
- Mosquera, G.M., R. Celleri, P.X. Lazo, K.B. Vache, S.S. Perakis, and P. Crespo. 2016. "Combined use of isotopic and hydrometric data to conceptualize ecohydrological processes in a high-elevation tropical ecosystem." *Hydrological Processes* 30(17): 2930-2947.
- Mueller, M., J. Pander, J. Knott, and J. Geist. 2017. "Comparison of nine different methods to assess fish communities in lentic flood-plain habitats." *Journal of Fish Biology* 91(1): 144-174.
- Mueller, M.H., A. Alaoui, and C. Alewell. 2016. "Water and solute dynamics during rainfall events in headwater catchments in the Central Swiss Alps under the influence of green alder shrubs and wetland soils." *Ecohydrology* 9(6): 950-963.
- Mukherjee, K., and S. Pal. 2021. "Hydrological and landscape dynamics of floodplain wetlands of the Diara region, Eastern India." *Ecological Indicators* 121.
- Munir, T.M., and C.J. Westbrook. 2021. "Beaver dam analogue configurations influence stream and riparian water table dynamics of a degraded spring-fed creek in the Canadian Rockies." *River Research and Applications* 37(3): 330-342.
- Murray, B., M. Reid, S. Capon, and S.B. Wu. 2019. "Genetic analysis suggests extensive gene flow within and between catchments in a common and ecologically significant dryland river shrub species;

Duma florulenta (Polygonaceae)." Ecology and Evolution 9(13): 7613-7627.

- Murray, D., B.T. Neilson, and J. Brahney. 2021. "Source or sink? Quantifying beaver pond influence on non-point source pollutant transport in the Intermountain West." *Journal of Environmental Management* 285.
- Mushet, D.M., L.C. Alexander, M. Bennett, K. Schofield, J.R. Christensen, G. Ali, A. Pollard, K. Fritz, and M.W. Lang. 2019. "Differing Modes of Biotic Connectivity within Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 55(2): 307-317.
- Musolff, A., J.H. Fleckenstein, M. Opitz, O. Buettner, R. Kumar, and J. Tittel. 2018. "Spatio-temporal controls of dissolved organic carbon stream water concentrations." *Journal of Hydrology* 566: 205-215.
- Nahlik, A.M., and M.S. Fennessy. 2016. "Carbon storage in US wetlands." *Nature Communications* 7: 13835. https://www.ncbi.nlm.nih.gov/pubmed/27958272.
- Namaalwa, S., A.A. van Dam, G.M. Gettel, R.C. Kaggwa, I. Zsuffa, and K. Irvine. 2020. "The Impact of Wastewater Discharge and Agriculture on Water Quality and Nutrient Retention of Namatala Wetland, Eastern Uganda." *Frontiers in Environmental Science* 8.
- Napiorkowski, P., M. Bakowska, N. Mrozinska, M. Szymanska, N. Kolarova, and K. Obolewski. 2019.
 "The Effect of Hydrological Connectivity on the Zooplankton Structure in Floodplain Lakes of a Regulated Large River (the Lower Vistula, Poland)." *Water* 11(9).
- Narr, C.F., H. Singh, P. Mayer, A. Keeley, B. Faulkner, D. Beak, and K.J. Forshay. 2019. "Quantifying the effects of surface conveyance of treated wastewater effluent on groundwater, surface water, and nutrient dynamics in a large river floodplain." *Ecological Engineering* 129: 123-133.
- Naskar, M., K. Roy, G. Karnatak, S.K. Nandy, and A. Roy. 2018. "Quantifying climate change induced threats to wetland fisheries: a stakeholder-driven approach." *Environment, Development and Sustainability* 20(6): 2811-2830.
- Natho, S., M. Tschikof, E. Bondar-Kunze, and T. Hein. 2020. "Modeling the Effect of Enhanced Lateral Connectivity on Nutrient Retention Capacity in Large River Floodplains: How Much Connected Floodplain Do We Need?" *Frontiers in Environmental Science* 8.
- Natho, S., and M. Venohr. 2014. "Active versus potential floodplains-the effect of small flood events on nutrient retention along the river Elbe corridor (Germany)." *Aquatic Sciences* 76(4): 633-642.
- Naus, C.J., and S.R. Adams. 2018. "Fish nursery habitat function of the main channel, floodplain tributaries and oxbow lakes of a medium-sized river." *Ecology of Freshwater Fish* 27(1): 4-18.
- Ndehedehe, C.E., A.O. Onojeghuo, B. Stewart-Koster, S.E. Bunn, and V.G. Ferreira. 2021. "Upstream flows drive the productivity of floodplain ecosystems in tropical Queensland." *Ecological Indicators* 125.
- Neff, B.P., and D.O. Rosenberry. 2017. "Groundwater Connectivity of Upland-Embedded Wetlands in

the Prairie Pothole Region." Wetlands 38(1): 51-63.

- Nelson, A., and K. Dube. 2016. "Channel response to an extreme flood and sediment pulse in a mixed bedrock and gravel-bed river." *Earth Surface Processes and Landforms* 41(2): 178-195.
- Neri-Flores, I., P. Moreno-Casasola, L. Alberto Peralta-Pelaez, and R. Monroy. 2019. "Groundwater and River Flooding: The Importance of Wetlands in Coastal Zones." *Journal of Coastal Research*: 44-54.
- Ngor, P.B., G. Grenouillet, S. Phem, N. So, and S. Lek. 2018. "Spatial and temporal variation in fish community structure and diversity in the largest tropical flood-pulse system of South-East Asia." *Ecology of Freshwater Fish* 27(4): 1087-1100.
- Nielsen, D.L., R.A. Cook, N. Ning, B. Gawne, and R. Petrie. 2016. "Carbon and nutrient subsidies to a lowland river following floodplain inundation." *Marine and Freshwater Research* 67(9): 1302-1312.
- Nobrega, R.L.B., T. Ziembowicz, G.N. Torres, A.C. Guzha, R.S.S. Amorim, D. Cardoso, M.S. Johnson, T.G. Santos, E. Couto, and G. Gerold. 2020. "Ecosystem services of a functionally diverse riparian zone in the Amazon-Cerrado agricultural frontier." *Global Ecology and Conservation* 21.
- Noe, G.B., K. Boomer, J.L. Gillespie, C.R. Hupp, M. Martin-Alciati, K. Floro, E.R. Schenk, A. Jacobs, and S. Strano. 2019. "The effects of restored hydrologic connectivity on floodplain trapping vs. release of phosphorus, nitrogen, and sediment along the Pocomoke River, Maryland USA." *Ecological Engineering* 138: 334-352.
- Nowacki, D.J., A.S. Ogston, C.A. Nittrouer, A.T. Fricke, N.E. Asp, and P.W.M. Souza Filho. 2019. "Seasonal, tidal, and geomorphic controls on sediment export to Amazon River tidal floodplains." *Earth Surface Processes and Landforms* 44(9): 1846-1859.

Nummi, P., W. Liao, J. van Der Schoor, and J. Loehr. 2021. "Beaver creates early successional hotspots for water beetles." *Biodiversity and Conservation*.

- Nummi, P., M. Vehkaoja, J. Pumpanen, and A. Ojala. 2018. "Beavers affect carbon biogeochemistry: both short-term and long-term processes are involved." *Mammal Review* 48(4): 298-311.
- Obolewski, K., K. Glinska-Lewczuk, M. Ozgo, and A. Astel. 2016. "Connectivity restoration of floodplain lakes: an assessment based on macroinvertebrate communities." *Hydrobiologia* 774(1): 23-37.
- Obolewski, K., K. Glinska-Lewczuk, and A. Strzelczak. 2015. "Does hydrological connectivity determine the benthic macroinvertebrate structure in oxbow lakes?" *Ecohydrology* 8(8): 1488-1502.
- Obolewski, K., K. Glinska-Lewczuk, A. Strzelczak, and P. Burandt. 2014. "Effects of a floodplain lake restoration on macroinvertebrate assemblages a case study of the lowland river (the Slupia River, N Poland)." *Polish Journal of Ecology* 62(3): 557-573.
- O'Briain, R., S. Shephard, and B. Coghlan. 2017. "River reaches with impaired riparian tree cover and channel morphology have reduced thermal resilience." *Ecohydrology* 10(8).

Ocock, J.F., R.T. Kingsford, T.D. Penman, and J.J.L. Rowley. 2016. "Amphibian Abundance and

Detection Trends During a Large Flood in a Semi-Arid Floodplain Wetland." *Herpetological Conservation and Biology* 11: 408-425.

- Oestreicher, J.S., M. Lucotte, M. Moingt, É. Bélanger, C. Rozon, R. Davidson, F. Mertens, and C.A. Romaña. 2017. "Environmental and Anthropogenic Factors Influencing Mercury Dynamics During the Past Century in Floodplain Lakes of the Tapajós River, Brazilian Amazon." Archives of Environmental Contamination and Toxicology 72(1): 11-30.
- Ogston, L., S. Gidora, M. Foy, and J. Rosenfeld. 2015. "Watershed-scale effectiveness of floodplain habitat restoration for juvenile coho salmon in the Chilliwack River, British Columbia." *Canadian Journal of Fisheries and Aquatic Sciences* 72(4): 479-490. <u>https://doi.org/10.1139/cjfas-2014-0189</u>.
- Omengo, F.O., T. Alleman, N. Geeraert, S. Bouillon, and G. Govers. 2016. "Sediment deposition patterns in a tropical floodplain, Tana River, Kenya." *Catena* 143: 57-69.
- Ondruch, J., Z. Macka, M.S. Michalkova, R. Putiska, M. Knot, P. Holik, J. Mirijovsky, and M. Jenco. 2018. "Response of channel dynamics to recent meander neck cut-off in a lowland meandering river with artificial training history: the Morava River, Czech Republic." *Hydrological Sciences Journal* 63(8): 1236-1254.
- Oromeng, K.V., E.A. Atekwana, L. Molwalefhe, and G.J. Ramatlapeng. 2021. "Time-series variability of solute transport and processes in rivers in semi-arid endorheic basins: The Okavango Delta, Botswana." *Science of the Total Environment* 759.
- Painter, K.J., C.J. Westbrook, B.D. Hall, N.J. O'Driscoll, and T.D. Jardine. 2015. "Effects of in-channel beaver impoundments on mercury bioaccumulation in Rocky Mountain stream food webs." *Ecosphere* 6(10).
- Palijan, G. 2015. "Towards deconfounding hydrological and seasonal temperature variability in the determination of selected limnological variables of a temperate floodplain ecosystem." *Ecohydrology* 8(2): 325-339.
- Pallud, C., C.C. Rhoades, L. Schneider, P. Dwivedi, and T. Borch. 2020. "Temperature-induced iron (III) reduction results in decreased dissolved organic carbon export in subalpine wetland soils, Colorado, USA." *Geochimica et Cosmochimica Acta* 280: 148-160.
- Pander, J., M. Mueller, and J. Geist. 2015. "Succession of fish diversity after reconnecting a large floodplain to the upper Danube River." *Ecological Engineering* 75: 41-50.
- Park, E., and E.M. Latrubesse. 2017. "The hydro-geomorphologic complexity of the lower Amazon River floodplain and hydrological connectivity assessed by remote sensing and field control." *Remote Sensing of Environment* 198: 321-332.
- Patricia Gayol, M., N. Soledad Morandeira, and P. Kandus. 2019. "Dynamics of shallow lake cover types in relation to Parana River flood pulses: assessment with multitemporal Landsat data." *Hydrobiologia* 833(1): 9-24.
- Peel, R.A., J.M. Hill, G.C. Taylor, D. Tweddle, and O.L.F. Weyl. 2019. "Species succession and the

development of a lacustrine fish community in an ephemeral lake." *Journal of Fish Biology* 95(3): 855-869.

- Peipoch, M., S.R. Miller, T.R. Antao, and H.M. Valett. 2019. "Niche partitioning of microbial communities in riverine floodplains." *Scientific Reports* 9.
- Pendleton, M.C., S. Sedgwick, K.M. Kettenring, and T.B. Atwood. 2020. "Ecosystem Functioning of Great Salt Lake Wetlands." *Wetlands* 40(6): 2163-2177.
- Pendleton, R.M., D.J. Hoeinghaus, L.C. Gomes, and A.A. Agostinho. 2014. "Loss of rare fish species from tropical floodplain food webs affects community structure and ecosystem multifunctionality in a mesocosm experiment." *PLoS ONE* 9(1).
- Penha, J., V.L. Landeiro, J.C.G. Ortega, and L. Mateus. 2017. "Interchange between flooding and drying, and spatial connectivity control the fish metacommunity structure in lakes of the Pantanal wetland." *Hydrobiologia* 797(1): 115-126.
- Perez-Rojas, J., F. Moreno, J. Cesar Quevedo, and J. Villa. 2019. "Soil organic carbon stocks in fluvial and isolated tropical wetlands from Colombia." *Catena* 179: 139-148.
- Petersen, C.R., N.Z. Jovanovic, and M.C. Grenfell. 2020. "The effectiveness of riparian zones in mitigating water quality impacts in an agriculturally dominated river system in South Africa." *African Journal of Aquatic Science* 45(3): 336-349.
- Petersen, R.J., C. Prinds, S. Jessen, B.V. Iversen, and C. Kjaergaard. 2020. "Riparian Lowlands in Clay Till Landscapes Part II: Nitrogen Reduction and Release Along Variable Flow Paths." *Water Resources Research* 56(4).
- Pinay, G., and N.E. Haycock. 2019. "Diffuse nitrogen pollution control: Moving from riparian zone to headwater catchment approach-A tribute to the influence of Professor Geoff Petts." *River Research and Applications* 35(8): 1203-1211.
- Pinay, G., S. Peiffer, J.R. De Dreuzy, S. Krause, D.M. Hannah, J.H. Fleckenstein, M. Sebilo, K. Bishop, and L. Hubert-Moy. 2015. "Upscaling Nitrogen Removal Capacity from Local Hotspots to Low Stream Orders' Drainage Basins." *Ecosystems* 18(6): 1101-1120.
- Piney, G., S. Bernal, B.W. Abbott, A. Lupon, E. Marti, F. Sabater, and S. Krause. 2018. "Riparian Corridors: A New Conceptual Framework for Assessing Nitrogen Buffering Across Biomes." *Frontiers in Environmental Science* 6.
- Ploum, S.W., H. Laudon, A.P. Tapia, and L. Kuglerov. 2020. "Are dissolved organic carbon concentrations in riparian groundwater linked to hydrological pathways in the boreal forest?" *Hydrology and Earth System Sciences* 24(4): 1709-1720.
- Pongruktham, O., and C. Ochs. 2015. "The rise and fall of the Lower Mississippi: effects of hydrologic connection on floodplain backwaters." *Hydrobiologia* 742(1): 169-183.

Pool, T., G. Holtgrieve, V. Elliott, K. McCann, B. McMeans, N. Rooney, A. Smits, T. Phanara, M.

Cooperman, S. Clark, C. Phen, and S. Chhuoy. 2017. "Seasonal increases in fish trophic niche plasticity within a flood-pulse river ecosystem (Tonle Sap Lake, Cambodia)." *Ecosphere* 8(7).

- Poulsen, J.B., F. Hansen, N.B. Ovesen, S.E. Larsen, and B. Kronvang. 2014. "Linking floodplain hydraulics and sedimentation patterns along a restored river channel: River Odense, Denmark." *Ecological Engineering* 66: 120-128.
- Powers, S.M., J.L. Tank, and D.M. Robertson. 2015. "Control of nitrogen and phosphorus transport by reservoirs in agricultural landscapes." *Biogeochemistry* 124: 417-439.
- Preiner, S., E. Bondar-Kunze, B. Pitzl, G. Weigelhofer, and T. Hein. 2020. "Effect of Hydrological Connectivity on the Phosphorus Buffering Capacity of an Urban Floodplain." *Frontiers in Environmental Science* 8: 147. https://doi.org/10.3389/fenvs.2020.00147.
- Prem, M., H.C.B. Hansen, W. Wenzel, L. Heiberg, H. Sorensen, and O.K. Borggaard. 2015. "High Spatial and Fast Changes of Iron Redox State and Phosphorus Solubility in a Seasonally Flooded Temperate Wetland Soil." *Wetlands* 35(2): 237-246.
- Puttock, A., H.A. Graham, D. Carless, and R.E. Brazier. 2018. "Sediment and nutrient storage in a beaver engineered wetland." *Earth Surface Processes and Landforms* 43(11): 2358-2370.
- Puttock, A., H.A. Graham, A.M. Cunliffe, M. Elliott, and R.E. Brazier. 2017. "Eurasian beaver activity increases water storage, attenuates flow and mitigates diffuse pollution from intensively-managed grasslands." *Science of the Total Environment* 576: 430-443.
- Pyron, M., L. Etchison, and J. Backus. 2014. "Fish Assemblages of Floodplain Lakes in the Ohio River Basin." Northeastern Naturalist 21(3): 419-430.
- Qin, Y., Z. Xin, and D. Wang. 2016. "Comparison of topsoil organic carbon and total nitrogen in different flood-risk riparian zones in a Chinese Karst area." *Environmental Earth Sciences* 75(12).
- Quillet, A., M. Larocque, S. Pellerin, V. Cloutier, M. Ferlatte, C. Paniconi, and M.A. Bourgault. 2017."The role of hydrogeological setting in two Canadian peatlands investigated through 2D steady-state groundwater flow modelling." *Hydrological Sciences Journal* 62(15): 2541-2557.
- Quin, A., and G. Destouni. 2018. "Large-scale comparison of flow-variability dampening by lakes and wetlands in the landscape." *Land Degradation & Development* 0(0).
- Quin, A., F. Jaramillo, and G. Destouni. 2015. "Dissecting the ecosystem service of large-scale pollutant retention: The role of wetlands and other landscape features." *Ambio* 44.
- Rabuffetti, A.P., L.A. Espínola, E. Abrial, M.L. Amsler, M.C.M. Blettler, M.F. Eurich, and E.G. Eberle.
 2020. "Commercial fisheries in a mega unregulated floodplain river: Assessment of the most favourable hydrological conditions for its preservation." *Journal of Fish Biology* 96(1): 59-73.
- Radigan, W.J., A.K. Carlson, M.J. Fincel, and B.D.S. Graeb. 2018. "Otolith chemistry as a fisheries management tool after flooding: The case of Missouri River gizzard shad." *River Research and Applications* 34(3): 270-278.

- Rahman, M.A.T.M., J.N. Negishi, T. Akasaka, and F. Nakamura. 2021. "Estimates of resource transfer via winged adult insects from the hyporheic zone in a gravel-bed river." *Ecology and Evolution* 11(9): 4656-4669.
- Rahman, M.M., J.R. Thompson, and R.J. Flower. 2020. "Hydrological impacts of climate change on rice cultivated riparian wetlands in the Upper Meghna River Basin (Bangladesh and India)." *Hydrological Sciences Journal* 65(1): 33-56.
- Ramesh, R., C.J. Anderson, and L. Kalin. 2020. "Characterizing nitrogen attenuation by headwater slope wetlands across different land uses." *Ecological Engineering* 149: 105833. <u>https://doi.org/10.1016/j.ecoleng.2020.105833</u>.
- Ramey, T.L., and J.S. Richardson. 2018. "Experimental test of water, nutrients, and microclimate on leaf litter mass loss in headwater riparian forests." *Ecosphere* 9(10).
- Ramirez-Hernandez, J., J. Eliana Rodriguez-Burgueno, E. Kendy, A. Salcedo-Peredia, and M.A. Lomeli. 2017. "Hydrological response to an environmental flood: Pulse flow 2014 on the Colorado River Delta." *Ecological Engineering* 106: 633-644.
- Rana, A., T. Kobayashi, and T.J. Ralph. 2021. "Planktonic Metabolism and Microbial Carbon Substrate Utilization in Response to Inundation in Semi-Arid Floodplain Wetlands." *Journal of Geophysical Research: Biogeosciences* 126(4).
- Rantala, H., D. Glover, J. Garvey, Q. Phelps, S. Tripp, D. Herzog, R. Hrabik, J. Crites, and M. Whiles. 2016. "Fish Assemblage and Ecosystem Metabolism Responses to Reconnection of the Bird's Point-New Madrid Floodway during the 2011 Mississippi River Flood." *River Research and Applications* 32(5): 1018-1029.
- Rayner, T.S., R.T. Kingsford, I.M. Suthers, and D.O. Cruz. 2015. "Regulated recruitment: Native and alien fish responses to widespread floodplain inundation in the Macquarie Marshes, arid Australia." *Ecohydrology* 8(1): 148-159.
- Records, R.M., M. Arabi, S.R. Fassnacht, Sr., W.G. Duffy, M. Ahmadi, and K.C. Hegewisch. 2014. "Climate change and wetland loss impacts on a western river's water quality." *Hydrology and Earth System Sciences* 18(11): 4509-4527.
- Redder, B.W., C.D. Kennedy, A.R. Buda, G. Folmar, and E.W. Boyer. 2021. "Groundwater contributions of flow and nitrogen in a headwater agricultural watershed." *Hydrological Processes* 35(5).
- Rees, G.N., R.A. Cook, N.S.P. Ning, P.J. McInerney, R.T. Petrie, and D.L. Nielsen. 2020. "Managed floodplain inundation maintains ecological function in lowland rivers." *Science of the Total Environment* 727.
- Reid, M.A., M.C. Reid, and M.C. Thoms. 2015. "Ecological significance of hydrological connectivity for wetland plant communities on a dryland floodplain river, MacIntyre River, Australia." *Aquatic Sciences* 78 (1): 139-158.
- Reis, V., V. Hermoso, S.K. Hamilton, S.E. Bunn, E. Fluet-Chouinard, B. Venables, and S. Linke. 2019.

"Characterizing seasonal dynamics of Amazonian wetlands for conservation and decision making." *Aquatic Conservation: Marine and Freshwater Ecosystems* 29(7): 1073-1082.

- Reis, V., V. Hermoso, S.K. Hamilton, S.E. Bunn, and S. Linke. 2019. "Conservation planning for riverwetland mosaics: A flexible spatial approach to integrate floodplain and upstream catchment connectivity." *Biological Conservation* 236: 356-365.
- Renik, K.M., and A.W. Hafs. 2020. "Effect of Beaver on Brook Trout Habitat in North Shore, Lake Superior, Streams." *North American Journal of Fisheries Management* 40(2): 427-445.
- Renshaw, C.E., K. Abengoza, F.J. Magilligan, W.B. Dade, and J.D. Landis. 2014. "Impact of flow regulation on near-channel floodplain sedimentation." *Geomorphology* 205: 120-127.
- Rhodes, K.A., T. Proffitt, T. Rowley, P.S.K. Knappett, D. Montiel, N. Dimova, D. Tebo, and G.R. Miller.
 2017. "The Importance of Bank Storage in Supplying Baseflow to Rivers Flowing Through Compartmentalized, Alluvial Aquifers." *Water Resources Research* 53(12): 10539-10557.
- Richards, D.R., H.L. Moggridge, P.H. Warren, and L. Maltby. 2020. "Impacts of hydrological restoration on lowland river floodplain plant communities." *Wetlands Ecology and Management* 28(3): 403-417.
- Richardson, B.L., and K.S. Herrman. 2020. "Nitrogen Removal via Denitrification in Two Small Reservoirs in Central Wisconsin, USA." *American Midland Naturalist* 184(1): 73-86.
- Richardson, J.S., and T. Sato. 2015. "Resource subsidy flows across freshwater-terrestrial boundaries and influence on processes linking adjacent ecosystems." *Ecohydrology* 8(3): 406-415.
- Ricker, M.C., G.D. Blosser, W.H. Conner, and G. Lockaby. 2019. "Wood Biomass and Carbon Pools within a Floodplain Forest of the Congaree River, South Carolina, USA." *Wetlands* 39(5): 1003-1013.
- Ricker, M.C., and B.G. Lockaby. 2015. "Soil Organic Carbon Stocks in a Large Eutrophic Floodplain Forest of the Southeastern Atlantic Coastal Plain, USA." *Wetlands* 35(2): 291-301.
- Ricker, M.C., M.H. Stolt, and M.S. Zavada. 2014. "Comparison of Soil Organic Carbon Dynamics in Forested Riparian Wetlands and Adjacent Uplands." *Soil Science Society of America Journal* 78(5): 1817-1827.
- Riedl, H.L., L. Stinson, L. Pejchar, and W.H. Clements. 2018. "An introduced plant affects aquaticderived carbon in the diets of riparian birds." *PLoS ONE* 13(11).
- Roni, P., J.E. Hall, S.M. Drenner, and D. Arterburn. 2019. "Monitoring the effectiveness of floodplain habitat restoration: A review of methods and recommendations for future monitoring." WIREs Water 6(4).
- Rood, S.B., M.L. Scott, M. Dixon, E. Gonzalez, C.O. Marks, P.B. Shafroth, and M.A. Volke. 2020.
 "Ecological Interfaces between Land and Flowing Water: Themes and Trends in Riparian Research and Management." *Wetlands* 40(6): 1801-1811.

Rosen, T., and Y.J. Xu. 2015. "Estimation of sedimentation rates in the distributary basin of the

Mississippi River, the Atchafalaya River Basin, USA." Nordic Hydrology 46(2): 244-257.

- Rouillard, A., G. Skrzypek, S. Dogramaci, C. Turney, and P.F. Grierson. 2015. "Impacts of high interannual variability of rainfall on a century of extreme hydrologic regime of northwest Australia." *Hydrology and Earth System Sciences* 19(4): 2057-2078.
- Rowinski, P.M., K. Vastila, J. Aberle, J. Jarvela, and M.B. Kalinowska. 2018. "How vegetation can aid in coping with river management challenges: A brief review." *Ecohydrology & Hydrobiology* 18(4): 345-354.
- Roy, M.B., A. Ghosh, A. Kumar, and P.K. Roy. 2021. "Assessing the nature of seasonal meteorological change in people's dependency on wetland: a case study of Bhagirathi-Hooghly floodplain system." *Environment, Development and Sustainability.*
- Ru, H., Y. Li, Q. Sheng, L. Zhong, and Z. Ni. 2020. "River damming affects energy flow and food web structure: a case study from a subtropical large river." *Hydrobiologia* 847(3): 679-695.
- Rude, N.P., J.T. Trushenski, and G.W. Whitledge. 2016. "Fatty acid profiles are biomarkers of fish habitat use in a river-floodplain ecosystem." *Hydrobiologia* 773(1): 63-75.
- Rudman, S.M., M. Kreitzman, M.F. Adame, M. Ronan, H. Franklin, N.J. Waltham, S. Rodriguez, E. Kavehei, M.P. Turschwell, S.R. Balcombe, P. Kaniewska, and M.A. Burford. 2019. "Nitrogen removal by tropical floodplain wetlands through denitrification." *Marine and Freshwater Research* 70(11): 1513-1521.
- Rudolph, J.C., C.A. Arendt, A.G. Hounshell, H.W. Paerl, and C.L. Osburn. 2020. "Use of Geospatial, Hydrologic, and Geochemical Modeling to Determine the Influence of Wetland-Derived Organic Matter in Coastal Waters in Response to Extreme Weather Events." *Frontiers in Marine Science* 7.
- Rumm, A., F. Foeckler, F. Dziock, C. Ilg, M. Scholz, R.M.B. Harris, and M. Gerisch. 2018. "Shifts in mollusc traits following floodplain reconnection: Testing the response of functional diversity components." *Freshwater Biology* 63(6): 505-517.
- Sadinski, W., M. Roth, T. Hayes, P. Jones, and A. Gallant. 2014. "Indicators of the statuses of amphibian populations and their potential for exposure to atrazine in four midwestern U.S. conservation areas." *PLoS ONE* 9(9).
- Salgado, J., C.D. Sayer, S.J. Brooks, T.A. Davidson, A.G. Baker, N. Willby, I.R. Patmore, B. Goldsmith, H. Bennion, and B. Okamura. 2019. "Connectivity and zebra mussel invasion offer short-term buffering of eutrophication impacts on floodplain lake landscape biodiversity." *Diversity and Distributions* 25(8): 1334-1347.
- Sanchez, A.J., N. Alvarez-Pliego, H. Espinosa-Perez, R. Florido, A. Macossay-Cortez, E. Barba, M.A. Salcedo, and A. Garrido-Mora. 2019. "Species richness of urban and rural fish assemblages in the Grijalva Basin floodplain, southern Gulf of Mexico." *Cybium* 43(3): 239-254.
- Santos, R.C.L., C.T. Callil, and V.L. Landeiro. 2020. "Unraveling the effects of water-sediment conditions and spatial patterns on Unionida assemblages in seasonally connected floodplain lakes."

Hydrobiologia 847(13): 2909-2922.

- Sawyer, A.H., L.A. Kaplan, O. Lazareva, and H.A. Michael. 2014. "Hydrologic dynamics and geochemical responses within a floodplain aquifer and hyporheic zone during Hurricane Sandy." *Water Resources Research* 50(6): 4877-4892.
- Scamardo, J., and E. Wohl. 2020. "Sediment storage and shallow groundwater response to beaver dam analogues in the Colorado Front Range, USA." *River Research and Applications* 36(3): 398-409.
- Scheel, K., R.R. Morrison, A. Annis, and F. Nardi. 2019. "Understanding the Large-Scale Influence of Levees on Floodplain Connectivity Using a Hydrogeomorphic Approach." *Journal of the American Water Resources Association* 55(2): 413-429.
- Schilling, K.E., B.J. Haines, C.S. Jones, and M. St Clair. 2018. "Effectiveness of a newly reconstructed floodplain oxbow to reduce NO3-N loads from a spring flood." *Journal of Environmental Management* 215: 385-393.
- Schilling, K.E., P.J. Jacobson, and J.A. Vogelgesang. 2015. "Agricultural conversion of floodplain ecosystems: implications for groundwater quality." *Journal of Environmental Management* 153: 74-83.
- Schilling, K.E., K. Kult, A. Seemon, G.K. Wilke, and L.C.S. Jones. 2018. "Nitrate-N load reduction measured in a central Iowa restored oxbow." *Ecological Engineering* 124: 19-22.
- Schilling, K.E., K. Kult, K. Wilke, M. Streeter, and J. Vogelgesang. 2017. "Nitrate reduction in a reconstructed floodplain oxbow fed by tile drainage." *Ecological Engineering* 102: 98-107.
- Schilling, K.E., K. Wilke, C.L. Pierce, K. Kult, and A. Kenny. 2019. "Multipurpose Oxbows as a Nitrate Export Reduction Practice in the Agricultural Midwest." *Agricultural & Environmental Letters* 4(1).
- Schmadel, N.M., J.W. Harvey, R.B. Alexander, G.E. Schwarz, R.B. Moore, K. Eng, J.D. Gomez-Velez, E.W. Boyer, and D. Scott. 2018. "Thresholds of lake and reservoir connectivity in river networks control nitrogen removal." *Nature Communications* 9(1).
- Schmidt-Mumm, U., and G. Janauer. 2014. "Seasonal dynamics of the shoreline vegetation in the Zapatosa floodplain lake complex, Colombia." *Revista de Biologia Tropical* 62(3): 1073-1097. <u>http://www.scielo.sa.cr/scielo.php?script=sci_arttext&pid=S0034-</u> <u>77442014000300021&lng=en&nrm=iso</u>.
- Schofield, K.A., L.C. Alexander, C.E. Ridley, M.K. Vanderhoof, K.M. Fritz, B.C. Autrey, J.E. DeMeester, W.G. Kepner, C.R. Lane, S.G. Leibowitz, and A.I. Pollard. 2018. "Biota Connect Aquatic Habitats throughout Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 54(2): 372-399.
- Schwab, A., and K. Kiehl. 2017. "Analysis of soil seed bank patterns in an oxbow system of a disconnected floodplain." *Ecological Engineering* 100: 46-55.
- Schwerdtfeger, J., M. Weiler, M.S. Johnson, and E.G. Couto. 2014. "Estimating water balance

components of tropical wetland lakes in the Pantanal dry season, Brazil." *Hydrological Sciences Journal* 59(12): 2158-2172.

- Schwinn, M., H. Baktoft, K. Aarestrup, M.C. Lucas, and A. Koed. 2018. "Telemetry observations of predation and migration behaviour of brown trout (*Salmo trutta*) smolts negotiating an artificial lake." *River Research and Applications* 34(8): 898-906. <u>https://doi.org/10.1002/tra.3327</u>.
- Scott, D.N., and E. Wohl. 2020. "Geomorphology and climate interact to control organic carbon stock and age in mountain river valley bottoms." *Earth Surface Processes and Landforms* 45(9): 1911-1925.
- Scott, D.T., J.D. Gomez-Velez, C.N. Jones, and J.W. Harvey. 2019. "Floodplain inundation spectrum across the United States." *Nature Communications* 10(1): 5194. https://www.ncbi.nlm.nih.gov/pubmed/31729364.
- Scott, D.T., R.F. Keim, B.L. Edwards, C.N. Jones, and D.E. Kroes. 2014. "Floodplain biogeochemical processing of floodwaters in the Atchafalaya River Basin during the Mississippi River flood of 2011." *Journal of Geophysical Research: Biogeosciences* 119(4): 537-546.
- Shim, M.J., Y. Cai, L. Guo, and A.M. Shiller. 2017. "Floodplain effects on the transport of dissolved and colloidal trace elements in the East Pearl River, Mississippi." *Hydrological Processes* 31(5): 1086-1099. <u>https://doi.org/10.1002/hyp.11093</u>.
- Shrestha, J., P.A. Niklaus, N. Pasquale, B. Huber, R.L. Barnard, E. Frossard, P. Schleppi, K. Tockner, and J. Luster. 2014. "Flood pulses control soil nitrogen cycling in a dynamic river floodplain." *Geoderma* 228: 14-24.
- Shuai, P., M.B. Cardenas, P.S.K. Knappett, P.C. Bennett, and B.T. Neilson. 2017. "Denitrification in the banks of fluctuating rivers: The effects of river stage amplitude, sediment hydraulic conductivity and dispersivity, and ambient groundwater flow." *Water Resources Research* 53(9): 7951-7967. <u>https://doi.org/10.1002/2017WR020610</u>.
- Sica, Y.V., G.I. Gavier-Pizarro, A.M. Pidgeon, A. Travaini, J. Bustamante, V.C. Radeloff, and R.D. Quintana. 2018. "Changes in bird assemblages in a wetland ecosystem after 14years of intensified cattle farming." *Austral Ecology* 43(7): 786-797.
- Sieczko, A., and P. Peduzzi. 2014. "Origin, enzymatic response and fate of dissolved organic matter during flood and non-flood conditions in a river-floodplain system of the Danube (Austria)." *Aquatic Sciences* 76(1): 115-129.
- Siev, S., E.C. Paringit, C. Yoshimura, and S. Hul. 2019. "Modelling inundation patterns and sediment dynamics in the extensive floodplain along the Tonle Sap River." *River Research and Applications* 35(9): 1387-1401.
- Silio-Calzada, A., J. Barquín, V.L.M. Huszar, N. Mazzeo, F. Méndez, and J.M. Álvarez-Martínez. 2017. "Long-term dynamics of a floodplain shallow lake in the Pantanal wetland: Is it all about climate?" *Science of the Total Environment*: 527-540.

Simpson, N.T., A.P. Bybe, M.J. Weber, C.L. Pierce, and K.J. Roe. 2019. "Occurrence, abundance and

associations of Topeka shiners (*Notropis topeka*) in restored and unrestored oxbows in Iowa and Minnesota, USA." *Aquatic Conservation: Marine and Freshwater Ecosystems* 29(10): 1735-1748.

- Singer, M.B., L.R. Harrison, P.M. Donovan, J.D. Blum, and M. Marvin-Dipasquale. 2016. "Hydrologic indicators of hot spots and hot moments of mercury methylation potential along river corridors." *Science of the Total Environment* 568: 697-711.
- Singh, H.V., B.R. Faulkner, A.A. Keeley, J. Freudenthal, and K.J. Forshay. 2018. "Floodplain restoration increases hyporheic flow in the Yakima River Watershed, Washington." *Ecological Engineering* 116: 110-120.
- Singh, N.K., J.D. Gourevitch, B.C. Wemple, K.B. Watson, D.M. Rizzo, S. Polasky, and T.H. Ricketts. 2019. "Optimizing wetland restoration to improve water quality at a regional scale." *Environmental Research Letters* 14(6).
- Sloat, M.R., G.H. Reeves, and K.R. Christiansen. 2017. "Stream network geomorphology mediates predicted vulnerability of anadromous fish habitat to hydrologic change in southeast Alaska." *Global Change Biology* 23(2): 604-620.
- Smith, L.L., C.N. Jones, and N.G. Nelson. 2019. "Featured Collection Introduction: The Emerging Science of Aquatic System Connectivity II." *Journal of the American Water Resources Association* 0(0): 526-528.
- Smith, L.L., A.L. Subalusky, C.L. Atkinson, J.E. Earl, D.M. Mushet, D.E. Scott, S.L. Lance, and S.A. Johnson. 2019. "Biological Connectivity of Seasonally Ponded Wetlands across Spatial and Temporal Scales." *Journal of the American Water Resources Association* 55(2): 334-353.
- Solari, L.C., K.P. Quaíni, and N.A. Gabellone. 2018. "Succession of microconsumers in waterlogged pampean soils (Buenos aires, Argentina) and its significance for nearby wetlands." *Aquatic Sciences* 80(4): 1-16.
- Soledad Morandeira, N., and P. Kandus. 2017. "Do taxonomic, phylogenetic and functional plant alphaand beta-diversity reflect environmental patterns in the Lower Parana River floodplain?" *Plant Ecology and Diversity* 10: 153-165.
- Song, S., B. Schmalz, and N. Fohrer. 2015. "Simulation, quantification and comparison of in-channel and floodplain sediment processes in a lowland area - A case study of the Upper Stor catchment in northern Germany." *Ecological Indicators* 57: 118-127.
- Speir, S.L., J.L. Tank, and U.H. Mahl. 2020. "Quantifying denitrification following floodplain restoration via the two-stage ditch in an agricultural watershed." *Ecological Engineering* 155.
- Spence, C., K. Beaty, P.J. Blanchfield, L. Hrenchuk, and M.D. Mackay. 2018. "The impact of a loss of hydrologic connectivity on boreal lake thermal and evaporative regimes." *Limnology and Oceanography* 63(5): 2028-2044.
- Spurgeon, J., M. Rhodes, J.W. Neal, and K. Evans. 2021. "Aquatic habitat changes within the channelized and impounded Arkansas River, Arkansas, USA." *River Research and Applications* 37(3): 462-474.

- Starks, T.A., and J.M. Long. 2017. "Diet Composition of Age-0 Fishes in Created Habitats of the Lower Missouri River." *American Midland Naturalist* 178(1): 112-122.
- Starks, T.A., J.M. Long, and A.R. Dzialowski. 2016. "Community Structure of Age-0 Fishes in Paired Mainstem and Created Shallow-water Habitats in the Lower Missouri River." *River Research and Applications* 32(4): 753-762.
- Starr, S.M., J.P. Benstead, and R.A. Sponseller. 2014. "Spatial and temporal organization of macroinvertebrate assemblages in a lowland floodplain ecosystem." *Landscape Ecology* 29(6): 1017-1031.
- Stegmann, L.F., R.P. Leitão, J. Zuanon, and W.E. Magnusson. 2019. "Distance to large rivers affects fish diversity patterns in highly dynamic streams of Central Amazonia." *PLoS ONE* 14(10).
- Sterzynska, M., V. Pizl, K. Tajovsky, M. Stelmaszczyk, and T. Okruszko. 2015. "Soil Fauna of Peat-Forming Wetlands in a Natural River Floodplain." *Wetlands* 35(4): 815-829.
- Stoffels, R.J., K.R. Clarke, and D.S. Linklater. 2015. "Temporal dynamics of a local fish community are strongly affected by immigration from the surrounding metacommunity." *Ecology and Evolution* 5(1): 200-212.
- Stoffels, R.J., K.R. Clarke, R.A. Rehwinkel, and B.J. McCarthy. 2014. "Response of a floodplain fish community to river-floodplain connectivity: natural versus managed reconnection." *Canadian Journal of Fisheries and Aquatic Sciences* 71(2): 236-245.
- Stoffels, R.J., R.A. Rehwinkel, A.E. Price, and W.F. Fagan. 2016. "Dynamics of fish dispersal during river-floodplain connectivity and its implications for community assembly." *Aquatic Sciences* 78(2): 355-365.
- Stone, M.C., C.F. Byrne, and R.R. Morrison. 2017. "Evaluating the impacts of hydrologic and geomorphic alterations on floodplain connectivity." *Ecohydrology* 10(5).
- Stuart, I.G., and C.P. Sharpe. 2020. "Riverine spawning, long distance larval drift, and floodplain recruitment of a pelagophilic fish: A case study of golden perch (Macquaria ambigua) in the arid Darling River, Australia." *Aquatic Conservation: Marine and Freshwater Ecosystems* 30(4): 675-690.
- Stunkle, C.R., A.T. Davidson, W.J. Shuart, M.W. McCoy, and J.R. Vonesh. 2021. "Taxa-specific responses to flooding shape patterns of abundance in river rock pools." *Freshwater Science* 40(2): 397-406.
- Stutter, M., N. Baggaley, D. Ó hUallacháin, and C. Wang. 2021. "The utility of spatial data to delineate river riparian functions and management zones: A review." *Science of the Total Environment* 757: 143982. <u>https://doi.org/10.1016/j.scitotenv.2020.143982</u>.
- Sueyoshi, M., D. Nakano, and F. Nakamura. 2014. "The relative contributions of refugium types to the persistence of benthic invertebrates in a seasonal snowmelt flood." *Freshwater Biology* 59(2): 257-271.

- Sullivan, S.M.P., M.C. Rains, and A.D. Rodewald. 2019. "Opinion: The proposed change to the definition of "waters of the United States" flouts sound science." *Proceedings of the National Academy of Sciences* 116(24).
- Sumaiya, S., J.A. Czuba, J.T. Schubert, S.R. David, G.H. Johnston, and D.A. Edmonds. 2021. "Sediment Transport Potential in a Hydraulically Connected River and Floodplain-Channel System." *Water Resources Research* 57(5).
- Sun, M., X. Tian, Y. Zou, and M. Jiang. 2019. "Ecological aesthetic assessment of a rebuilt wetland restored from farmland and management implications for National Wetland Parks." *PLoS ONE* 14(10).
- Sun, X., L. Bernard-Jannin, S. Sauvage, C. Garneau, J.G. Arnold, R. Srinivasan, and J.M. Sanchez-Perez. 2017. "Assessment of the denitrification process in alluvial wetlands at floodplain scale using the SWAT model." *Ecological Engineering* 103: 344-358.
- Sutfin, N.A., and E. Wohl. 2017. "Substantial soil organic carbon retention along floodplains of mountain streams." *Journal of Geophysical Research-Earth Surface* 122(7): 1325-1338.
- Sutfin, N.A., E.E. Wohl, and K.A. Dwire. 2016. "Banking carbon: a review of organic carbon storage and physical factors influencing retention in floodplains and riparian ecosystems." *Earth Surface Processes and Landforms* 41(1): 38-60.
- Tamooh, F., F.J.R. Meysman, A.V. Borges, T.R. Marwick, K. Van Den Meersche, F. Dehairs, R. Merckx, and S. Bouillon. 2014. "Sediment and carbon fluxes along a longitudinal gradient in the lower Tana River (Kenya)." *Journal of Geophysical Research: Biogeosciences* 119(7): 1340-1353.
- Tan, Z., X. Wang, B. Chen, X. Liu, and Q. Zhang. 2019. "Surface water connectivity of seasonal isolated lakes in a dynamic lake-floodplain system." *Journal of Hydrology* 579.
- Taylor, C.M., C. Prigent, and S.J. Dadson. 2018. "Mesoscale rainfall patterns observed around wetlands in sub-Saharan Africa." *Quarterly Journal of the Royal Meteorological Society* 144(716): 2118-2132.
- Terrado, M., R. Tauler, and E.M. Bennett. 2015. "Landscape and local factors influence water purification in the Monteregian agroecosystem in Québec, Canada." *Regional Environmental Change* 15(8): 1743-1755. <u>https://doi.org/10.1007/s10113-014-0733-6</u>.
- Tetzlaff, D., C. Birkel, J. Dick, J. Geris, and C. Soulsby. 2014. "Storage dynamics in hydropedological units control hillslope connectivity, runoff generation, and the evolution of catchment transit time distributions." *Water Resources Research* 50(2): 969-985.
- Thom, R.M., S.A. Breithaupt, H.L. Diefenderfer, A.B. Borde, G.C. Roegner, G.E. Johnson, and D.L. Woodruff. 2018. "Storm-driven particulate organic matter flux connects a tidal tributary floodplain wetland, mainstem river, and estuary." *Ecological Applications* 28(6): 1420-1434.
- Thornton, S.A., E. Setiana, K. Yoyo, M.E. Harrison, and S.E. Page. 2020. "Towards biocultural approaches to peatland conservation: The case for fish and livelihoods in Indonesia." *Environmental Science and Policy* 114: 341-351.

- Tiwari, T., J. Lundstrom, L. Kuglerova, H. Laudon, K. Ohman, and A.M. Agren. 2016. "Cost of riparian buffer zones: A comparison of hydrologically adapted site-specific riparian buffers with traditional fixed widths." *Water Resources Research* 52(2): 1056-1069.
- Todhunter, P.E. 2018. "A volumetric water budget of Devils Lake (USA): non-stationary precipitationrunoff relationships in an amplifier terminal lake." *Hydrological Sciences Journal* 63(9): 1275-1291.
- Tomasek, A.A., M. Hondzo, J.L. Kozarek, C. Staley, P. Wang, N. Lurndahl, and M.J. Sadowsky. 2019. "Intermittent flooding of organic-rich soil promotes the formation of denitrification hot moments and hot spots." *Ecosphere* 10(1).
- Tomscha, S.A., J.R. Deslippe, M. de Roiste, S. Hartley, and B. Jackson. 2019. "Uncovering the ecosystem service legacies of wetland loss using high-resolution models." *Ecosphere* 10(10).
- Tornabene, B.J., M.E. Jaeger, R.G. Bramblett, M. Nelson, N. McClenning, T. Watson, A. Ankrum, K. Frazer, A.M. Reinhold, and A.V. Zale. 2019. "Riverine turtles select habitats maintained by natural discharge regimes in an unimpounded large river." *River Research and Applications* 35(9): 1489-1498. https://doi.org/10.1002/rra.3496.
- Tournebize, J., C. Chaumont, and U. Mander. 2017. "Implications for constructed wetlands to mitigate nitrate and pesticide pollution in agricultural drained watersheds." *Ecological Engineering* 103: 415-425.
- Trentman, M.T., J.L. Tank, S.E. Jones, S.K. McMillan, and T.V. Royer. 2020. "Seasonal evaluation of biotic and abiotic factors suggests phosphorus retention in constructed floodplains in three agricultural streams." *Science of the Total Environment* 729.
- Trevelline, B.K., T. Nuttle, B.D. Hoenig, N.L. Brouwer, B.A. Porter, and S.C. Latta. 2018. "DNA metabarcoding of nestling feces reveals provisioning of aquatic prey and resource partitioning among Neotropical migratory songbirds in a riparian habitat." *Oecologia* 187(1): 85-98.
- Tumushimire, L., M. Mindje, U. Sinsch, and J.M. Dehling. 2020. "Anuran diversity of cultivated wetlands in Rwanda: melting pot of generalists?" *Salamandra* 56(2): 99-112.
- Tunaley, C., D. Tetzlaff, and C. Soulsby. 2017. "Scaling effects of riparian peatlands on stable isotopes in runoff and DOC mobilisation." *Journal of Hydrology* 549: 220-235.
- Uuemaa, E., C.C. Palliser, A.O. Hughes, and C.C. Tanner. 2018. "Effectiveness of a Natural Headwater Wetland for Reducing Agricultural Nitrogen Loads." *Water* 10(3).
- Valdez, R.A., G.M. Haggerty, K. Richard, and D. Klobucar. 2019. "Managed spring runoff to improve nursery floodplain habitat for endangered Rio Grande silvery minnow." *Ecohydrology* 12(7).
- van Der Merwe, J., and E.C. Hellgren. 2016. "Spatial variation in trophic ecology of small mammals in wetlands: support for hydrological drivers." *Ecosphere* 7(11).
- Vanderhoof, M.K., and L.C. Alexander. 2016. "The Role of Lake Expansion in Altering the Wetland Landscape of the Prairie Pothole Region, United States." *Wetlands* 36(2): 309-321.

- Vanderhoof, M.K., L.C. Alexander, and M.J. Todd. 2016. "Temporal and spatial patterns of wetland extent influence variability of surface water connectivity in the Prairie Pothole Region, United States." *Landscape Ecology* 31(4): 805-824.
- Vanderhoof, M.K., J.R. Christensen, and L.C. Alexander. 2016. "Patterns and drivers for wetland connections in the Prairie Pothole Region, United States." *Wetlands Ecology and Management* 25(3): 1-23.
- Vanderhoof, M.K., J.R. Christensen, and L.C. Alexander. 2019. "Influence of multi-decadal land use, irrigation practices and climate on riparian corridors across the Upper Missouri River headwaters basin, Montana." *Hydrology and Earth System Sciences* 23(10): 4269-4292.
- Vanderhoof, M.K., H.E. Distler, M.W. Lang, and L.C. Alexander. 2017. "The influence of data characteristics on detecting wetland/stream surface-water connections in the Delmarva Peninsula, Maryland and Delaware." Wetlands Ecology and Management 26: 63–86.
- Vanderhoof, M.K., C.R. Lane, M.G. McManus, L.C. Alexander, and J.R. Christensen. 2018. "Wetlands inform how climate extremes influence surface water expansion and contraction." *Hydrology and Earth System Sciences* 22(3): 1851-1873.
- Vehkaoja, M., and P. Nummi. 2015. "Beaver facilitation in the conservation of boreal anuran communities." *Herpetozoa* 28 (1-2): 75-87.
- Venarsky, M.P., B. Stewart-Koster, R.J. Hunt, T.D. Jardine, and S.E. Bunn. 2020. "Spatial and temporal variation of fish community biomass and energy flow throughout a tropical river network." *Freshwater Biology* 65(10): 1782-1792.
- Victoria Torres, M., F. Giri, and P. Agustin Collins. 2017. "'La Nina' phenomenon and the relationship between decapod populations and fishes in temporarily isolated shallow lakes." *Marine and Freshwater Research* 68(6): 1010-1022.
- Vilizzi, L., L.A. Thwaites, and B.B. Smith. 2014. "EXPLOITATION BY COMMON CARP (CYPRINUS CARPIO) OF A FLOODPLAIN WETLAND OF THE LOWER RIVER MURRAY UNDER DROUGHT AND FLOODING CONDITIONS." *Royal Society of South Australia. Transactions* 138: 113-123.
- Vilizzi, L., L.A. Thwaites, B.B. Smith, J.M. Nicol, and C.P. Madden. 2014. "Ecological effects of common carp (*Cyprinus carpio*) in a semi-arid floodplain wetland." *Marine and Freshwater Research* 65(9): 802-817.
- von Freyberg, J., D. Radny, H.E. Gall, and M. Schirmer. 2014. "Implications of hydrologic connectivity between hillslopes and riparian zones on streamflow composition." *Journal of Contaminant Hydrology* 169: 62-74.
- Von Korff, B.H., M.F. Piehler, and S.H. Ensign. 2014. "Comparison of Denitrification Between River Channels and Their Adjoining Tidal Freshwater Wetlands." *Wetlands* 34(6): 1047-1060.

von Schiller, D., S. Bernal, C.N. Dahm, and E. Marti. 2017. "Chapter 3.2 - Nutrient and Organic Matter

Dynamics in Intermittent Rivers and Ephemeral Streams." Pages 135-160 in *Intermittent Rivers and Ephemeral Streams: Ecology and Management*. T. Datry, N. Bonada, and A. Boulton, eds. Academic Press.

- Vonbank, J.A., H.M. Hagy, and A.F. Casper. 2016. "Energetic Carrying Capacity of Riverine and Connected Wetlands of the Upper Illinois River for Fall-Migrating Waterfowl." *American Midland Naturalist* 176(2): 210-221.
- Walalite, T., S.C. Dekker, F.M. Keizer, I. Kardel, P.P. Schot, S.M. Dejong, and M.J. Wassen. 2016.
 "Flood Water Hydrochemistry Patterns Suggest Floodplain Sink Function for Dissolved Solids from the Songkhram Monsoon River (Thailand)." *Wetlands* 36(6): 995-1008.
- Walcker, R., D. Corenblit, F. Julien, J.M. Martinez, and J. Steiger. 2021. "Contribution of meandering rivers to natural carbon fluxes: Evidence from the Ucayali River, Peruvian Amazonia." *Science of the Total Environment* 776.
- Walker, C.M., D.F. Whigham, I.S. Bentz, J.M. Argueta, R.S. King, M.C. Rains, C.A. Simenstad, C. Guo, S.J. Baird, and C.J. Field. 2021. "Linking landscape attributes to salmon and decision-making in the southern Kenai Lowlands, Alaska, USA." *Ecology and Society* 26(1).
- Walker, R.H., B.M. Maitland, T.N. Lasharr, M.N. Rosing, and M. Ben-David. 2018. "Fate of juvenile salmonids stranded in off-channel pools: implications for nutrient transfers." *Aquatic Sciences* 80(1).
- Wallace, T.A., and D. Furst. 2016. "Open water metabolism and dissolved organic carbon in response to environmental watering in a lowland river-floodplain complex." *Marine and Freshwater Research* 67(9): 1346-1361.
- Walsh, R., and A.S. Ward. 2019. "Redefining Clean Water Regulations Reduces Protections for Wetlands and Jurisdictional Uncertainty." *Frontiers in Water* 1(1).
- Walters, K.M., and M. Babbar-Sebens. 2016. "Using climate change scenarios to evaluate future effectiveness of potential wetlands in mitigating high flows in a Midwestern US watershed." *Ecological Engineering* 89: 80-102.
- Wang, D.L. 2020. "Environmental gradient relative to oxbow lake-meandering river connectivity in Zoige Basin of the Tibetan Plateau." *Ecological Engineering* 156: 1-1.
- Wang, H.Z., X.Q. Liu, and H.J. Wang. 2016. "The Yangtze River Floodplain: Threats and Rehabilitation." American Fisheries Society. Symposium 84: 263-291.
- Wang, M., L. Duan, J. Wang, J. Peng, and B. Zheng. 2020. "Determining the width of lake riparian buffer zones for improving water quality base on adjustment of land use structure." *Ecological Engineering* 158: 106001. <u>https://doi.org/10.1016/j.ecoleng.2020.106001</u>.
- Wang, P., W. Ouyang, Z. Wu, X. Cui, W. Zhu, R. Jin, and C. Lin. 2020. "Diffuse nitrogen pollution in a forest-dominated watershed: Source, transport and removal." *Journal of Hydrology* 585.
- Wang, X., E.L. Shaw, C.J. Westbrook, and A. Bedard-Haughn. 2018. "Beaver Dams Induce Hyporheic

and Biogeochemical Changes in Riparian Areas in a Mountain Peatland." *Wetlands* 38(5): 1017-1032.

- Webb, J.R., I.R. Santos, B. Robson, B. Macdonald, L. Jeffrey, and D.T. Maher. 2017. "Constraining the annual groundwater contribution to the water balance of an agricultural floodplain using radon: The importance of floods." *Water Resources Research* 53(1): 544-562.
- Webster, A.J., and M.L. Cadenasso. 2020. "Cross-scale controls on the in-stream dynamics of nitrate and turbidity in semiarid agricultural waterway networks." *Journal of Environmental Management* 262.
- Webster, A.J., P.M. Groffman, and M.L. Cadenasso. 2018. "Controls on denitrification potential in nitrate-rich waterways and riparian zones of an irrigated agricultural setting." *Ecological Applications* 28(4): 1055-1067. https://www.ncbi.nlm.nih.gov/pubmed/29465768.
- Wegener, P., T. Covino, and C. Rhoades. 2018. "Evaluating Controls on Nutrient Retention and Export in Wide and Narrow Valley Segments of a Mountain River Corridor." *Journal of Geophysical Research: Biogeosciences* 123(6): 1817-1826.
- Wegener, P., T. Covino, and E. Wohl. 2017. "Beaver-mediated lateral hydrologic connectivity, fluvial carbon and nutrient flux, and aquatic ecosystem metabolism." *Water Resources Research* 53(6): 4606-4623.
- Weigelhofer, G., S. Preiner, A. Funk, E. Bondar-Kunze, and T. Hein. 2015. "The hydrochemical response of small and shallow floodplain water bodies to temporary surface water connections with the main river." *Freshwater Biology* 60: 781-793.
- Welsh, M.K., S.K. McMillan, and P.G. Vidon. 2017. "Denitrification along the Stream-Riparian Continuum in Restored and Unrestored Agricultural Streams." *Journal of Environmental Quality* 46(5): 1010-1019.
- Welter, J.R., and S.G. Fisher. 2016. "The influence of storm characteristics on hydrological connectivity in intermittent channel networks: implications for nitrogen transport and denitrification." *Freshwater Biology* 61(8): 1214-1227.
- Were, D., F. Kansiime, T. Fetahi, and T. Hein. 2021. "Soil organic carbon storage in a tropical freshwater wetland: the influence of vegetation type." *African Journal of Aquatic Science* 46(2): 161-172.
- Wesner, J.S., D.L. Swanson, M.D. Dixon, D.A. Soluk, D.J. Quist, L.A. Yager, J.W. Warmbold, E. Oddy, and T.C. Seidel. 2020. "Loss of Potential Aquatic-Terrestrial Subsidies Along the Missouri River Floodplain." *Ecosystems* 23(1): 111-123.
- Wesner, J.S., D.M. Walters, and R.E. Zuellig. 2019. "Pulsed salmonfly emergence and its potential contribution to terrestrial detrital pools." *Food Webs* 18.
- Whigham, D.F., C.M. Walker, J. Maurer, R.S. King, W. Hauser, S. Baird, J.A. Keuskamp, and P.J. Neale. 2017. "Watershed influences on the structure and function of riparian wetlands associated with headwater streams - Kenai Peninsula, Alaska." *Science of the Total Environment*: 124-134.

- Whipple, A.A., J.H. Viers, and H.E. Dahlke. 2017. "Flood regime typology for floodplain ecosystem management as applied to the unregulated Cosumnes River of California, United States." *Ecohydrology* 10(5).
- Whitten, A.L., O.M. Mendenhall, L.E. Solomon, and A.F. Casper. 2021. "Operational Impacts of a Water Management Structure on the Surrounding Fish Assemblages in a Restored Backwater and a Large Floodplain River." *American Midland Naturalist* 185(1): 120-138.
- Williams, J.E., and S. Gregory. 2018. "Fish Occurrence in a Seasonally Inundated Floodplain of the Willamette River, Oregon." *Northwest Science* 92(3): 191-202.
- Williams, M.R., A.R. Buda, H.A. Elliott, K. Singha, and J. Hamlett. 2015. "Influence of Riparian Seepage Zones on Nitrate Variability in Two Agricultural Headwater Streams." *Journal of the American Water Resources Association* 51(4): 883-897.
- Wohl, E. 2017. "Connectivity in rivers." Progress in Physical Geography 41(3): 345-362.
- Wohl, E. 2021. "Legacy effects of loss of beavers in the continental United States." *Environmental Research Letters* 16(2).
- Wohl, E., R.O. Hall, K.B. Lininger, N.A. Sutfin, and D.M. Walters. 2017. "Carbon dynamics of river corridors and the effects of human alterations." *Ecological Monographs* 87(3): 379-409.
- Wohl, E., and A. Pfeiffer. 2018. "Organic carbon storage in floodplain soils of the US prairies." *River Research and Applications* 34(5): 406-416.
- Wollheim, W.M., T.K. Harms, B.J. Peterson, K. Morkeski, C.S. Hopkinson, R.J. Stewart, M.N. Gooseff, and M.A. Briggs. 2014. "Nitrate uptake dynamics of surface transient storage in stream channels and fluvial wetlands." *Biogeochemistry* 120: 239-257.
- Wong, V.N., C. McNaughton, and A. Pearson. 2016. "Changes in soil organic carbon fractions after remediation of a coastal floodplain soil." *Journal of Environmental Management* 168: 280-287.
- Woznicki, S.A., J. Baynes, S. Panlasigui, M. Mehaffey, and A. Neale. 2019. "Development of a spatially complete floodplain map of the conterminous United States using random forest." *Science of the Total Environment* 647: 942-953.
- Wu, G., and Y. Liu. 2017. "Seasonal Water Exchanges between China's Poyang Lake and Its Saucer-Shaped Depressions on River Deltas." *Water* 9(11).
- Wu, H., Q. Guan, K. Lu, G. Han, B. Li, and M. Yang. 2020. "Effects of Hydrological Connectivity on Snail Assemblages in the Intertidal Zone of Coastal Wetlands." *Wetlands* 40(5): 1627-1634.
- Wu, Y., G. Zhang, A.N. Rousseau, and Y.J. Xu. 2020. "Quantifying streamflow regulation services of wetlands with an emphasis on quickflow and baseflow responses in the Upper Nenjiang River Basin, Northeast China." *Journal of Hydrology* 583.
- Wu, Y., G. Zhang, A.N. Rousseau, Y.J. Xu, and E. Foulon. 2020. "On how wetlands can provide flood resilience in a large river basin: A case study in Nenjiang river Basin, China." *Journal of Hydrology*

587.

- Wu, Y., Y. Zhang, L. Dai, L. Xie, S. Zhao, Y. Liu, and Z. Zhang. 2021. "Hydrological connectivity improves soil nutrients and root architecture at the soil profile scale in a wetland ecosystem." *Science* of the Total Environment 762.
- Xia, S., Y. Liu, Y. Wang, B. Chen, Y. Jia, G. Liu, X. Yu, and L. Wen. 2016. "Wintering waterbirds in a large river floodplain: Hydrological connectivity is the key for reconciling development and conservation." *Science of the Total Environment* 573: 645-660.
- Xia, X., S. Zhang, S. Li, L. Zhang, G. Wang, J. Wang, and Z. Li. 2018. "The cycle of nitrogen in river systems: sources, transformation, and flux." *Environmental Science: Processes & Impacts* 20(6): 863-891.
- Xiao, R., Q. Wang, M.X. Zhang, W.B. Pan, and J.J. Wang. 2020. "Plankton distribution patterns and the relationship with environmental gradients and hydrological connectivity of wetlands in the Yellow River Delta." *Ecohydrology & Hydrobiology* 20(4): 584-596.
- Nath, S., A.J. Nath, G.W. Sileshi, and A.K. Das. 2017. "Biomass stocks and carbon storage in Barringtonia acutangula floodplain forests in North East India." *Biomass and Bioenergy* 98: 37-42.
- Yabusaki, S.B., A.N. Myers-Pigg, N.D. Ward, S.R. Waichler, A. Sengupta, Z. Hou, X. Chen, Y. Fang, Z. Duan, J.A. Serkowski, J. Indivero, C.W. Moore, and C.M. Gunn. 2020. "Floodplain Inundation and Salinization From a Recently Restored First-Order Tidal Stream." *Water Resources Research* 56(7).
- Yang, W., Y. Liu, C. Ou, and S. Gabor. 2016. "Examining water quality effects of riparian wetland loss and restoration scenarios in a southern Ontario watershed." *Journal of Environmental Management*.
- Yao, J., J.M. Sanchez-Perez, S. Sauvage, S. Teissier, E. Attard, B. Lauga, R. Duran, F. Julien, L. Bernard-Jannin, H. Ramburn, and M. Gerino. 2017. "Biodiversity and ecosystem purification service in an alluvial wetland." *Ecological Engineering* 103: 359-371.
- Yao, X., L. Zhang, Y. Zhang, H. Xu, and X. Jiang. 2016. "Denitrification occurring on suspended sediment in a large, shallow, subtropical lake (Poyang Lake, China)." *Environmental Pollution* 219: 501-511.
- Yu, H., X. Wang, L. Chu, G. Wang, G. Sun, M. Sun, J. Wang, and M. Jiang. 2019. "Is There Any Correlation Between Landscape Characteristics and Total Nitrogen in Wetlands Receiving Agricultural Drainages?" *Chinese Geographical Science* 29(4): 712-724.
- Yuan, T., H. Lee, H.C. Jung, A. Aierken, E. Beighley, D.E. Alsdorf, R.M. Tshimanga, and D. Kim. 2017."Absolute water storages in the Congo River floodplains from integration of InSAR and satellite radar altimetry." *Remote Sensing of Environment* 201: 57-72.
- Yurek, S., D.L. Deangelis, J.C. Trexler, J.A. Klassen, and L.G. Larsen. 2016. "Persistence and diversity of directional landscape connectivity improves biomass pulsing in simulations of expanding and contracting wetlands." *Ecological Complexity* 28: 1-11.

- Zaffaroni, M., P. Zamberletti, I.F. Creed, F. Accatino, C. De Michele, and B. Devries. 2019.
 "Safeguarding Wetlands and Their Connections within Wetlandscapes to Improve Conservation Outcomes for Threatened Amphibian Species." *Journal of the American Water Resources Association* 55(3): 641-656.
- Zehe, E., R. Loritz, C. Jackisch, M. Westhoff, A. Kleidon, T. Blume, S.K. Hassler, and H.H. Savenije. 2019. "Energy states of soil water - a thermodynamic perspective on soil water dynamics and storagecontrolled streamflow generation in different landscapes." *Hydrology and Earth System Sciences* 23(2): 971-987.
- Zero, V.H., and M.A. Murphy. 2016. "An amphibian species of concern prefers breeding in active beaver ponds." *Ecosphere* 7(5).
- Zhang, P., J. Mao, T. Hu, L. Dai, D. Xu, and H. Dai. 2020. "Water exchange in a large river-lake system: Modeling, characteristics and causes." *River Research and Applications* 36(5): 697-708.
- Zhang, X.L., Q. Zhang, A.D. Werner, and Z.Q. Tan. 2017. "Characteristics and causal factors of hysteresis in the hydrodynamics of a large floodplain system: Poyang Lake (China)." *Journal of Hydrology* 553: 574-583.
- Zhao, S., B. Zhang, X. Sun, and L. Yang. 2021. "Hot spots and hot moments of nitrogen removal from hyporheic and riparian zones: A review." *Science of the Total Environment* 762.
- Zhu, H., D. Liu, W. Shi, and Q. Chen. 2019. "Nitrogen cycles within the stream-to-riparian continuum under flood waves." *Episodes* 42(4): 333-341.

Non-floodplain Wetlands and Open Waters

- "Valuing wetlands." 2021. Nature Geoscience 14(3): 111-111.
- Abernethy, E.F., K.L. Turner, and J.C. Beasley. 2017. "Scavenging along an ecological interface: utilization of amphibian and reptile carcasses around isolated wetlands." *Ecosphere* 8(11).
- Abouali, M., A.P. Nejadhashemi, F. Daneshvar, U. Adhikari, M.R. Herman, T.J. Calappi, and B.G. Rohn. 2017. "Evaluation of wetland implementation strategies on phosphorus reduction at a watershed scale." *Journal of Hydrology* 552: 105-120.
- Acreman, M., A. Smith, L. Charters, D. Tickner, J. Opperman, S. Acreman, F. Edwards, P. Sayers, and F. Chivava. 2021. "Evidence for the effectiveness of nature-based solutions to water issues in Africa." *Environmental Research Letters* 16(6): 063007. https://doi.org/10.1088/1748-9326/ac0210.
- Aguilar, A., A.M. Maeda-Martinez, G. Murugan, H. Obregon-Barboza, D.C. Rogers, K. McClintock, and J.L. Krumm. 2017. "High intraspecific genetic divergence in the versatile fairy shrimp *Branchinecta lindahli* with a comment on cryptic species in the genus Branchinecta (Crustacea: Anostraca)." *Hydrobiologia* 801(1): 59-69.

- Aguirrezabala-Campano, T., O. Gerardo-Nieto, R. Gonzalez-Valencia, V. Souza, and F. Thalasso. 2019.
 "Methane dynamics in the subsaline ponds of the Chihuahuan Desert: A first assessment." *Science of the Total Environment* 666: 1255-1264.
- Åhlén, I., P. Hambäck, J. Thorslund, A. Frampton, G. Destouni, and J. Jarsjö. 2020. "Wetlandscape size thresholds for ecosystem service delivery: Evidence from the Norrström drainage basin, Sweden." *Science of the Total Environment* 704.
- Ahmed, F. 2014. "Cumulative Hydrologic Impact of Wetland Loss: Numerical Modeling Study of the Rideau River Watershed, Canada." *Journal of Hydrologic Engineering* 19(3): 593-606.
- Ahmed, F. 2017. "Influence of Wetlands on Black-Creek Hydraulics." *Journal of Hydrologic Engineering* 22(1).
- Al Sayah, M.J., R. Nedjai, K. Kaffas, C. Abdallah, and M. Khouri. 2019. "Assessing the Impact of Man-Made Ponds on Soil Erosion and Sediment Transport in Limnological Basins." *Water* 11(12).
- Ala-Aho, P., C. Soulsby, O.S. Pokrovsky, S.N. Kirpotin, J. Karlsson, S. Serikova, S.N. Vorobyev, R.M. Manasypov, S. Loiko, and D. Tetzlaff. 2018. "Using stable isotopes to assess surface water source dynamics and hydrological connectivity in a high-latitude wetland and permafrost influenced landscape." *Journal of Hydrology* 556: 279-293.
- Albanese, G., and C.A. Davis. 2015. "Characteristics within and around stopover wetlands used by migratory shorebirds: Is the neighborhood important?" *The Condor* 117(3): 328-340.
- Albanese, G., and D.A. Haukos. 2017. "A network model framework for prioritizing wetland conservation in the Great Plains." *Landscape Ecology* 32(1): 115-130.
- Alderton, E., C.D. Sayer, R. Davies, S.J. Lambert, and J.C. Axmacher. 2017. "Buried alive: Aquatic plants survive in 'ghost ponds' under agricultural fields." *Biological Conservation* 212: 105-110.
- Alexander, L.C. 2015. "Science at the boundaries: scientific support for the Clean Water Rule." *Freshwater Science* 34(4): 1588-1594.
- Alexander, L.C., K.M. Fritz, K.A. Schofield, B.C. Autrey, J.E. DeMeester, H.E. Golden, D.C. Goodrich, W.G. Kepner, H.R. Kiperwas, C.R. Lane, S.D. LeDuc, S.G. Leibowitz, M.G. McManus, A.I. Pollard, C.E. Ridley, M.K. Vanderhoof, and P.J. Wigington. 2018. "Featured Collection Introduction: Connectivity of Streams and Wetlands to Downstream Waters." *Journal of the American Water Resources Association* 54(2): 287-297.
- Alfsnes, K., A. Hobaek, L.J. Weider, and D.O. Hessen. 2016. "Birds, nutrients, and climate change: mtDNA haplotype diversity of Arctic Daphnia on Svalbard revisited." *Polar Biology* 39(8): 1425-1437.
- Ali, G., and C. English. 2019. "Phytoplankton blooms in Lake Winnipeg linked to selective watergatekeeper connectivity." *Scientific Reports* 9(1): 8395. <u>https://doi.org/10.1038/s41598-019-44717-y</u>.
- Ali, G., A. Haque, N.B. Basu, P. Badiou, and H. Wilson. 2017. "Groundwater-Driven Wetland-Stream

Connectivity in the Prairie Pothole Region: Inferences Based on Electrical Conductivity Data." *Wetlands* 37(4): 773-785.

- Allen, C., R. Gonzales, and L. Parrott. 2020. "Modelling the contribution of ephemeral wetlands to landscape connectivity." *Ecological Modelling* 419.
- Amado, A.A., M. Politano, K. Schilling, and L. Weber. 2018. "Investigating Hydrologic Connectivity of a Drained Prairie Pothole Region Wetland Complex using a Fully Integrated, Physically-Based Model." Wetlands 38(2): 233-245.
- Ameli, A., and I.F. Creed. 2019. "Groundwaters at Risk: Wetland Loss Changes Sources, Lengthens Pathways, and Decelerates Rejuvenation of Groundwater Resources." *Journal of the American Water Resources Association* 55(2): 294-306.
- Ameli, A.A., and I.F. Creed. 2017. "Quantifying hydrologic connectivity of wetlands to surface water systems." *Hydrol. Earth Syst. Sci.* 21: 1791-1808.
- Ameli, A.A., and I.F. Creed. 2019. "Does Wetland Location Matter When Managing Wetlands for Watershed-Scale Flood and Drought Resilience?" *Journal of the American Water Resources Association* 55(3): 529-542.
- Anderson, C.D. 2018. "AMPHIBIAN IMMIGRATION AND EMIGRATION AT A TEMPORARY POND IN THE FLORIDA SANDHILLS, USA: IMPLICATIONS FOR CONSERVATION." *Herpetological Conservation and Biology* 13(1): 131-145.
- Andresen, C.G., M.J. Lara, C.E. Tweedie, and V.L. Lougheed. 2017. "Rising plant-mediated methane emissions from arctic wetlands." *Global Change Biology* 23(3): 1128-1139.
- Anteau, M.J., M.T. Wiltermuth, M. P. van Der Burg, and A.T. Pearse. 2016. "Prerequisites for Understanding Climate-Change Impacts on Northern Prairie Wetlands." Wetlands 36: S299-S307. <u>https://doi.org/10.1007/s13157-016-0811-2</u>.
- Antolini, F., E. Tate, B. Dalzell, N. Young, K. Johnson, and P.L. Hawthorne. 2019. "Flood Risk Reduction from Agricultural Best Management Practices." *Journal of the American Water Resources Association* 56(1): 161-179.
- Arheimer, B., and B.C. Pers. 2017. "Lessons learned? Effects of nutrient reductions from constructing wetlands in 1996–2006 across Sweden." *Ecological Engineering* 103: 404-414.
- Aticho, A., D.O. Gemeda, D.H. Feyssa, D.B. Jiru, A. Beyene, D. Seyoum, D.J. Snelder, G.L. Feyisa, S. Aynalem, G. Archibald, and T.M. Gutema. 2018. "Assessment of black crowned crane and wattled crane population and spatiotemporal distribution in Jimma Zone, Southwest Ethiopia." *Global Ecology and Conservation* 16.
- Badiou, P., B. Page, and W. Akinremi. 2018. "Phosphorus Retention in Intact and Drained Prairie Wetland Basins: Implications for Nutrient Export." *Journal of Environmental Quality* 47(4): 902-913. https://www.ncbi.nlm.nih.gov/pubmed/30025042.

- Bagella, S., S. Gascon, R. Filigheddu, A. Cogoni, and D. Boix. 2016. "Mediterranean Temporary Ponds: new challenges from a neglected habitat." *Hydrobiologia* 782(1): 1-10.
- Balke, R., C.O. Koning, and B. Lundsted. 2018. 2018. "Watershed model identifies historic wastewater discharge to wetlands as major phosphorus source to a eutrophic lake, NH USA." *Lake and Reservoir Management* 34(4): 349-364.
- Ballantine, K.A., T.R. Anderson, E.A. Pierce, and P.M. Groffman. 2017. "Restoration of denitrification in agricultural wetlands." *Ecological Engineering* 106: 570-577.
- Bam, E., A.M. Ireson, G. Kamp, and J. Hendry. 2020. "Ephemeral Ponds: Are They the Dominant Source of Depression-Focused Groundwater Recharge?" *Water Resources Research* 56(3).
- Bam, E.K.P.A., and K.A.M. Ireson. 2019. "Quantifying the wetland water balance: A new isotope-based approach that includes precipitation and infiltration." *Journal of Hydrology* 570: 185-200.
- Bansal, S., O.F. Johnson, J. Meier, and X. Zhu. 2020. "Vegetation Affects Timing and Location of Wetland Methane Emissions." *Journal of Geophysical Research: Biogeosciences* 125(9).
- Barbaree, B.A., M.E. Reiter, C.M. Hickey, N.K. Elliott, D. Schaffer-Smith, M.D. Reynolds, and G.W. Page. 2018. "Dynamic surface water distributions influence wetland connectivity within a highly modified interior landscape." *Landscape Ecology* 33(5): 829-844.
- Bartelt, P.E., and R.W. Klaver. 2017. "Response of Anurans to Wetland Restoration on a Midwestern Agricultural Landscape." *Journal of Herpetology* 51(4): 504-514.
- Bartolucci, N.N., T.R. Anderson, and K.A. Ballantine. 2021. "Restoration of retired agricultural land to wetland mitigates greenhouse gas emissions." *Restoration Ecology* 29(3).
- Baulch, H.M., J.A. Elliott, M.R.C. Cordeiro, D.N. Flaten, D.A. Lobb, and H.F. Wilson. 2019. "Soil and water management: opportunities to mitigate nutrient losses to surface waters in the Northern Great Plains." *Environmental Reviews* 27(4): 447-477.
- Beatty, W.S., D.C. Kesler, E.B. Webb, A.H. Raedeke, L.W. Naylor, and D.D. Humburg. 2014. "The role of protected area wetlands in waterfowl habitat conservation: Implications for protected area network design." *Biological Conservation* 176: 144-152.
- Benson, C.E., B. Carberry, and T.A. Langen. 2018. "Public-private partnership wetland restoration programs benefit Species of Greatest Conservation Need and other wetland-associated wildlife." *Wetlands Ecology and Management* 26(2): 195-211. https://doi.org/10.1007/s11273-017-9565-8.
- Bertassello, L.E., A.F. Aubeneau, G. Botter, J.W. Jawitz, and P.S.C. Rao. 2020. "Emergent dispersal networks in dynamic wetlandscapes." *Scientific Reports* 10(1).
- Bertassello, L.E., P.S.C. Rao, J.W. Jawitz, A.F. Aubeneau, and G. Botter. 2020. "Wetlandscape hydrologic dynamics driven by shallow groundwater and landscape topography." *Hydrological Processes* 34(6): 1460-1474.

Bertassello, L.E., P.S.C. Rao, J. Park, J.W. Jawitz, and G. Botter. 2018. "Stochastic modeling of wetland-

groundwater systems." Advances in Water Resources 112: 214-223.

- Biggs, J., S. von Fumetti, and M. Kelly-Quinn. 2017. "The importance of small waterbodies for biodiversity and ecosystem services: implications for policy makers." *Hydrobiologia* 793(1): 3-39.
- Birkel, C., C. Duvert, A. Correa, N.C. Munksgaard, D.T. Maher, and L.B. Hutley. 2020. "Tracer-Aided Modeling in the Low-Relief, Wet-Dry Tropics Suggests Water Ages and DOC Export Are Driven by Seasonal Wetlands and Deep Groundwater." *Water Resources Research* 56(4).
- Bishop-Taylor, R., M.G. Tulbure, and M. Broich. 2018. "Impact of hydroclimatic variability on regionalscale landscape connectivity across a dynamic dryland region." *Ecological Indicators* 94: 142-150.
- Blake-Bradshaw, A.G., J.D. Lancaster, J.R. O'Connell, J.W. Matthews, M.W. Eichholz, and H.M. Hagy. 2020. "Suitability of Wetlands for Migrating and Breeding Waterbirds in Illinois." *Wetlands* 40(6): 1993-2010.
- Boda, P., A. Móra, G. Várbíró, and Z. Csabai. 2018. "Livin' on the edge: the importance of adjacent intermittent habitats in maintaining macroinvertebrate diversity of permanent freshwater marsh systems." *Inland Waters* 8(3): 312-321. <u>https://doi.org/10.1080/20442041.2018.1461969</u>.
- Bolpagni, R. 2020. "Linking vegetation patterns, wetlands conservation, and ecosystem services provision: From publication to application." *Aquatic Conservation: Marine and Freshwater Ecosystems* 30(9): 1734-1740.
- Bourgault, M.A., M. Larocque, and M. Garneau. 2017. "Quantification of peatland water storage capacity using the water table fluctuation method." *Hydrological Processes* 31(5): 1184-1195.
- Brandis, K.J., D. Mazumder, P. Gadd, B. Ji, R.T. Kingsford, and D. Ramp. 2021. "Using feathers to map continental-scale movements of waterbirds and wetland importance." *Conservation Letters*.
- Brannen, R., C. Spence, and A. Ireson. 2015. "Influence of shallow groundwater-surface water interactions on the hydrological connectivity and water budget of a wetland complex." *Hydrological Processes* 29(18): 3862-3877.
- Breen, S.P.W., P.A. Loring, and H. Baulch. 2018. "When a Water Problem Is More Than a Water Problem: Fragmentation, Framing, and the Case of Agricultural Wetland Drainage." *Frontiers in Environmental Science* 6.
- Brooks, G.C., J.A. Smith, E.A. Frimpong, T.A. Gorman, H.C. Chandler, and C.A. Haas. 2019. "Indirect connectivity estimates of amphibian breeding wetlands from spatially explicit occupancy models." *Aquatic Conservation: Marine and Freshwater Ecosystems* 29(11): 1815-1825.
- Brooks, R.J., D. M. Mushet, M. Vanderhoof, S. G. Leibowitz, J. R. Christensen, B. P. Neff, D. Rosenberry, W. D. Rugh, and L. C. Alexander. 2018. "Estimating Wetland Connectivity to Streams in the Prairie Pothole Region: An Isotopic and Remote Sensing Approach." *Water Resources Research*(2): 955-977.

Buchanan, S.W., B. Buffum, G. Puggioni, and N.E. Karraker. 2019. "Occupancy of freshwater turtles

across a gradient of altered landscapes." Journal of Wildlife Management 83(2): 435-445.

- Bugna, G.C., J.M. Grace, and Y.P. Hsieh. 2020. "Sensitivity of using stable water isotopic tracers to study the hydrology of isolated wetlands in North Florida." *Journal of Hydrology* 580: 124321.
- Buttle, J.M. 2018. "Mediating stream baseflow response to climate change: The role of basin storage." *Hydrological Processes* 32(3): 363-378.
- Byun, E., S.A. Finkelstein, S.A. Cowling, and P. Badiou. 2018. "Potential carbon loss associated with post-settlement wetland conversion in southern Ontario, Canada." *Carbon Balance and Management* 13(1).
- Cabezas, A., M. Pallasch, I. Schoenfelder, J. Gelbrecht, and D. Zak. 2014. "Carbon, nitrogen, and phosphorus accumulation in novel ecosystems: Shallow lakes in degraded fen areas." *Ecological Engineering* 66: 63-71.
- Calhoun, A.J.K., D.M. Mushet, L.C. Alexander, E.S. DeKeyser, L. Fowler, C.R. Lane, M.W. Lang, M.C. Rains, S.C. Richter, and S.C. Walls. 2017. "The Significant Surface-Water Connectivity of "Geographically Isolated Wetlands"." *Wetlands* 37(4): 801-806.
- Calhoun, A.J.K., D.M. Mushet, K.P. Bell, D. Boix, J.A. Fitzsimons, and F. Isselin-Nondedeu. 2017."Temporary wetlands: challenges and solutions to conserving a 'disappearing' ecosystem." *Biological Conservation* 211: 3-11.
- Cameron, C.R., M.C. Hancock, D.W. Carr, M.K. Hurst, D.E. Campbell, T.J. Venning, P.D. Tara, and K.R. Holzwart. 2020. "Hydroperiods of Cypress Domes in West-Central Florida, USA." *Wetlands* 40(6): 2225-2234. <u>https://doi.org/10.1007/s13157-020-01329-x</u>.
- Carnell, P.E., S.M. Windecker, M. Brenker, J. Baldock, P. Masque, K. Brunt, and P.I. Macreadie. 2018. "Carbon stocks, sequestration, and emissions of wetlands in south eastern Australia." *Global Change Biology* 24(9): 4173-4184.
- Carvajal-Quintero, J.D., S.R. Januchowski-Hartley, J.A. Maldonado-Ocampo, C. Jezequel, J. Delgado, D.K. Hare, D.F. Boutt, W.P. Clement, C.E. Hatch, G. Davenport, and A. Hackman. 2017.
 "Hydrogeological controls on spatial patterns of groundwater discharge in peatlands." *Hydrology and Earth System Sciences* 21(12): 6031-6048.
- Casson, N.J., M.C. Eimers, S.A. Watmough, and M.C. Richardson. 2019. "The role of wetland coverage within the near-stream zone in predicting of seasonal stream export chemistry from forested headwater catchments." *Hydrological Processes* 33(10): 1465-1475.
- Catford, J.A., J. Roberts, S.J. Capon, R.H. Froend, S.M. Windecker, and M.M. Douglas. 2017. "Wetland Vegetation of Inland Australia." *Australian Vegetation*, *3rd Edition*: 490-515.
- Chandler, H.C., D.L. McLaughlin, T.A. Gorman, K.J. McGuire, J.B. Feaga, and C.A. Haas. 2017."Drying Rates of Ephemeral Wetlands: Implications for Breeding Amphibians." *Wetlands* 37(3): 545-557.

- Chandler, H.C., B.S. Stegenga, and D.J. Stevenson. 2019. "Movement and Space Use in Southern Populations of Spotted Turtles (Clemmys guttata)." *Southeastern Naturalist* 18(4): 602-618.
- Chang, H., Y. Makido, and E. Foster. 2021. "Effects of land use change, wetland fragmentation, and best management practices on total suspended sediment concentrations in an urbanizing Oregon watershed, USA." *Journal of Environmental Management* 282.
- Chen, C., C.D. Meurk, Z. Jia, M. Lv, S. Wu, and J. Jia. 2017. "Incorporating landscape connectivity into household pond configuration in a hilly agricultural landscape." *Landscape and Ecological Engineering* 13(1): 189-204.
- Chen, W., D. Nover, H. Yen, Y. Xia, B. He, W. Sun, and J. Viers. 2020. "Exploring the multiscale hydrologic regulation of multipond systems in a humid agricultural catchment." *Water Research* 184.
- Cheng, F.Y., and N.B. Basu. 2017. "Biogeochemical hotspots: Role of small water bodies in landscape nutrient processing." *Water Resources Research* 53(6): 5038-5056.
- Cheng, F.Y., K.J. Van Meter, D.K. Byrnes, and N.B. Basu. 2020. "Maximizing US nitrate removal through wetland protection and restoration." *Nature* 588: 625-630.
- Chow, A.T., A.L. Pitt, R.F. Baldwin, D. Suhre, and J.J. Wang. 2016. "Water quality dynamics of ephemeral wetlands in the Piedmont ecoregion, South Carolina, USA." *Ecological Engineering* 94: 555-563.
- Chu, X. 2018. "Quantifying Discontinuity, Connectivity, Variability, and Hierarchy in Overland Flow Generation: Comparison of Different Modeling Methods." Pages 587-603 in *Hydrologic Modeling: Select Proceedings of ICWEES-2016.* V.P. Singh, S. Yadav, and R.N. Yadava, eds. Water Science and Technology Library. Springer, Singapore.
- Cieslinski, R., M. Major, and L. Pietruszynski. 2020. "Seasonal variability in the supply of dissolved matter to catchments of basins without outlets in northern Poland, measured by selected physiochemical indicators." *Baltica* 33(1): 21-34.
- Clements, A.R., P.J. Suter, M. Fussell, and E. Silvester. 2016. "Macroinvertebrate communities in spring-fed alpine source pools." *Hydrobiologia* 777(1): 119-138.
- Cohen, M.J., I.F. Creed, L. Alexander, N.B. Basu, A.J.K. Calhoun, C. Craft, E. D'Amico, E. Dekeyser, L. Fowler, H.E. Golden, J.W. Jawitz, P. Kalla, L.K. Kirkman, C.R. Lane, M. Lang, S.G. Leibowitz, D.B. Lewis, J. Marton, D.L. McLaughlin, D.M. Mushet, H. Raanan-Kiperwas, M.C. Rains, L. Smith, and S.C. Walls. 2016. "Do geographically isolated wetlands influence landscape functions?" *Proceedings of the National Academy of Sciences of the United States of America* 113(8): 1978-1986.
- Colas, F., J.M. Baudoin, P. Bonin, L. Cabrol, M. Daufresne, R. Lassus, and J. Cucherousset. 2021. "Ecosystem maturity modulates greenhouse gases fluxes from artificial lakes." *Science of the Total Environment* 760.
- Conde-Porcuna, J.M., C. Perez-Martinez, and E. Moreno. 2018. "Variations in the hatching response of rotifers to salinity and waterbird ingestion." *Journal of Plankton Research* 40(3): 326-341.

- Connon, R.F., W.L. Quinton, J.R. Craig, Jr., J. Hanisch, and O. Sonnentag. 2015. "The hydrology of interconnected bog complexes in discontinuous permafrost terrains." *Hydrological Processes* 29(18): 3831-3847.
- Coster, S.S., K.J. Babbitt, and A.I. Kovach. 2015. "High Genetic Connectivity in Wood Frogs (*Lithobates sylvaticus*) and Spotted Salamanders (*Ambystoma maculatum*) in a Commercial Forest." *Herpetological Conservation and Biology* 10(1): 64-89.
- Craft, C., J. Vymazal, and L. Kropfelova. 2018. "Carbon sequestration and nutrient accumulation in floodplain and depressional wetlands." *Ecological Engineering* 114: 137-145.
- Crawford, J.A., W.E. Peterman, A.R. Kuhns, and L.S. Eggert. 2016. "Altered functional connectivity and genetic diversity of a threatened salamander in an agroecosystem." *Landscape Ecology* 31(10): 2231-2244.
- Creed, I.F., C.R. Lane, J.N. Serran, L.C. Alexander, N.B. Basu, A.J.K. Calhoun, J.R. Christensen, M.J. Cohen, C. Craft, E. D'Amico, E. DeKeyser, L. Fowler, H.E. Golden, J.W. Jawitz, P. Kalla, L.K. Kirkman, M. Lang, S.G. Leibowitz, D.B. Lewis, J. Marton, D.L. McLaughlin, H. Raanan-Kiperwas, M.C. Rains, K.C. Rains, and L. Smith. 2017. "Enhancing protection for vulnerable waters." *Nature Geoscience* 10 (11): 809-815. <u>https://www.ncbi.nlm.nih.gov/pubmed/30079098</u>.
- Cui, Q., M.E. Ammar, M. Iravani, J. Kariyeva, and M. Faramarzi. 2021. "Regional wetland water storage changes: The influence of future climate on geographically isolated wetlands." *Ecological Indicators* 120.
- Czuba, J.A., A. Hansen, E. Foufoula-Georgiou, and J.C. Finlay. 2018. "Contextualizing Wetlands Within a River Network to Assess Nitrate Removal and Inform Watershed Management." *Water Resources Research* 54(2): 1312-1337.
- Dalcin Martins, P., R.E. Danczak, S. Roux, J. Frank, M.A. Borton, R.A. Wolfe, M.N. Burris, and M.J. Wilkins. 2018. "Viral and metabolic controls on high rates of microbial sulfur and carbon cycling in wetland ecosystems." *Microbiome* 6(1): 138. <u>https://www.ncbi.nlm.nih.gov/pubmed/30086797</u>.
- Daneshvar, F., A.P. Nejadhashemi, U. Adhikari, B. Elahi, M. Abouali, M.R. Herman, E. Martinez-Martinez, T.J. Calappi, and B.G. Rohn. 2017. "Evaluating the significance of wetland restoration scenarios on phosphorus removal." *Journal of Environmental Management* 192: 184-196.
- Danesh-Yazdi, M., E. Foufoula-Georgiou, D.L. Karwan, and G. Botter. 2016. "Inferring changes in water cycle dynamics of intensively managed landscapes via the theory of time-variant travel time distributions." *Water Resources Research* 52(10): 7593-7614.
- Daniel, D.W., L.M. Smith, and S.T. McMurry. 2017. "Effects of sediment removal and surrounding land use on carbon and nitrogen storage in playas and watersheds in the Rainwater Basin region of Nebraska." Soil & Tillage Research 174: 169-176.
- Darwiche-Criado, N., F.A. Comín, A. Masip, M. García, S.G. Eismann, and R. Sorando. 2017. "Effects of wetland restoration on nitrate removal in an irrigated agricultural area: The role of in-stream and offstream wetlands." *Ecological Engineering* 103: 426-435.

- Daugherty, E.E., G.A. McKee, R. Bergstrom, S. Burton, C. Pallud, R.M. Hubbard, E.F. Kelly, C.C. Rhoades, and T. Borch. 2019. "Hydrogeomorphic controls on soil carbon composition in two classes of subalpine wetlands." *Biogeochemistry* 145: 161-175.
- Davila, D.C., I. Sanchez, J. Vallejos, R.D. Appleton, and R.P. Piana. 2019. "Diversity and Seasonality of Birds in Permanent Waterholes in the Peruvian Northwest." *Ornitologia Neotropical* 30: 1-7.
- Davis, C.L., D.A. Miller, S.C. Walls, W.J. Barichivich, J.W. Riley, and M.E. Brown. 2017. "Species interactions and the effects of climate variability on a wetland amphibian metacommunity." *Ecological Applications* 27(1): 285-296.
- Davis, C.L., D.A.W. Miller, E.H.C. Grant, B.J. Halstead, P.M. Kleeman, S.C. Walls, and W.J. Barichivich. 2019. "Linking variability in climate to wetland habitat suitability: is it possible to forecast regional responses from simple climate measures?" Wetlands Ecology and Management 27(1): 39-53.
- De Morais-Junior, C.S., M. De Melo-Junior, T. Goncalves-Souza, and R.M. De Lyra-Neves. 2019. "Zoochory of zooplankton: seasonality and bird morphological diversity can influence metacommunity dynamics of temporary ponds." *Journal of Plankton Research* 41(4): 465-477.
- de Necker, L., M. Ferreira, J.H.J. van Vuren, and W. Malherbe. 2016. "Aquatic invertebrate community structure of selected endorheic wetlands (pans) in South Africa." *Inland Waters* 6(3): 303-313.
- Deane, D.C., D.A. Fordham, F. He, and C.J.A. Bradshaw. 2017. "Future extinction risk of wetland plants is higher from individual patch loss than total area reduction." *Biological Conservation* 209: 27-33.
- Deane, D.C., and F. He. 2018. "Loss of only the smallest patches will reduce species diversity in most discrete habitat networks." *Global Change Biology* 24(12): 5802-5814.
- Deboo, M.L., N.P. Murphy, A.D. Austin, C.H.S. Watts, and M.T. Guzik. 2019. "Arid zone island hopping: the impact of dispersal on endemism in hydraenid beetles (Coleoptera: Hydraenidae) found in isolated desert springs." *Austral Entomology* 58(4): 886-896.
- Deemy, J.B., and T.C. Rasmussen. 2017. "Hydrology and water quality of isolated wetlands: Stormflow changes along two episodic flowpaths." *Journal of Hydrology-Regional Studies* 14: 23-36.
- Deiner, K., J.M. Hull, and B. May. 2017. "Range-wide phylogeographic structure of the vernal pool fairy shrimp (*Branchinecta lynchi*)." *PLoS ONE* 12(5).
- Dell, A.I., R.A. Alford, and R.G. Pearson. 2014. "Intermittent pool beds are permanent cyclic habitats with distinct wet, moist and dry phases." *PLoS ONE* 9(9). <u>https://doi.org/10.1371/journal.pone.0108203</u>.
- Delvecchia, A.G., S. Gougherty, B.W. Taylor, and S.A. Wissinger. 2021. "Biogeochemical characteristics and hydroperiod affect carbon dioxide flux rates from exposed high-elevation pond sediments." *Limnology and Oceanography* 66(4): 1050-1067. <u>https://doi.org/10.1002/lno.11663</u>.
- Denoel, M., S. Dalleur, E. Langrand, A. Besnard, and H. Cayuela. 2018. "Dispersal and alternative

breeding site fidelity strategies in an amphibian." *Ecography* 41(9): 1543-1555.

- Denoël, M., P. Scimè, and N. Zambelli. 2016. "Newt life after fish introduction: extirpation of paedomorphosis in a mountain fish lake and newt use of satellite pools." *Current Zoology* 62(1): 61-69.
- Denver, J.M., S.W. Ator, M.W. Lang, T.R. Fisher, A.B. Gustafson, R. Fox, J.W. Clune, and G.W. McCarty. 2014. "Nitrate fate and transport through current and former depressional wetlands in an agricultural landscape, Choptank Watershed, Maryland, United States." *Journal of Soil and Water Conservation* 69(1): 1-16.
- Devito, K.J., K.J. Hokanson, P.A. Moore, N. Kettridge, A.E. Anderson, L. Chasmer, C. Hopkinson, M.C. Lukenbach, C.A. Mendoza, J. Morissette, D.L. Peters, R.M. Petrone, U. Silins, B. Smerdon, and J.M. Waddington. 2017. "Landscape controls on long-term runoff in subhumid heterogeneous Boreal Plains catchments." *Hydrological Processes* 31(15): 2737-2751.
- Dibike, Y., A. Muhammad, R.R. Shrestha, C. Spence, B. Bonsal, L. de Rham, J. Rowley, G. Evenson, and T. Stadnyk. 2021. "Application of dynamic contributing area for modelling the hydrologic response of the Assiniboine River basin to a changing climate." *Journal of Great Lakes Research* 47(3): 663-676.
- Dolph, C.L., E. Boardman, M. Danesh-Yazdi, J.C. Finlay, A. Hansen, A.C. Baker, and B. Dalzell. 2019. "Phosphorus Transport in Intensively Managed Watersheds." *Water Resources Research* 55(11): 9148-9172.
- Donnelly, J.P., S.L. King, N.L. Silverman, D.P. Collins, E.M. Carrera-Gonzalez, A. Lafón-Terrazas, and J.N. Moore. 2020. "Climate and human water use diminish wetland networks supporting continental waterbird migration." *Global Change Biology*.
- Donnelly, J.P., D.E. Naugle, D.P. Collins, B.D. Dugger, B.W. Allred, J.D. Tack, and V.J. Dreitz. 2019. "Synchronizing conservation to seasonal wetland hydrology and waterbird migration in semi-arid landscapes." *Ecosphere* 10(6).
- Drayer, A.N., and S.C. Richter. 2016. "Physical wetland characteristics influence amphibian community composition differently in constructed wetlands and natural wetlands." *Ecological Engineering* 93: 166-174.
- Driscoll, J.M., L.E. Hay, M.K. Vanderhoof, and R.J. Viger. 2020. "Spatiotemporal Variability of Modeled Watershed Scale Surface-Depression Storage and Runoff for the Conterminous United States." *Journal of the American Water Resources Association* 56(1): 16-29.
- Du, E., C.R. Jackson, J. Klaus, J.J. McDonnell, N.A. Griffiths, M.F. Williamson, J.L. Greco, and M. Bitew. 2016. "Interflow dynamics on a low relief forested hillslope: Lots of fill, little spill." *Journal* of Hydrology 534: 648-658.
- Duvert, C., L.B. Hutley, J. Beringer, M.I. Bird, C. Birkel, D.T. Maher, M. Northwood, M. Rudge, S.A. Setterfield, and J.G. Wynn. 2020. "Net landscape carbon balance of a tropical savanna: Relative importance of fire and aquatic export in offsetting terrestrial production." *Global Change Biology*

26(10): 5899-5913.

- Dvorett, D., C. Davis, and M. Papes. 2016. "Mapping and Hydrologic Attribution of Temporary Wetlands Using Recurrent Landsat Imagery." *Wetlands* 36(3): 431-443.
- Edokpa, D.A., M.G. Evans, T.E.H. Allott, M. Pilkington, and J.J. Rothwell. 2017. "Peatland restoration and the dynamics of dissolved nitrogen in upland freshwaters." *Ecological Engineering* 106: 44-54.
- Ehsanzadeh, E., G. van Der Kamp, and C. Spence. 2016. "On the changes in long-term streamflow regimes in the North American Prairies." *Hydrological Sciences Journal* 61(1): 64-78.
- Endter-Wada, J., K.M. Kettenring, and A. Sutton-Grier. 2020. "Protecting wetlands for people: Strategic policy action can help wetlands mitigate risks and enhance resilience." *Environmental Science and Policy* 108: 37-44.
- Epele, L.B., C. Brand, and M.L. Miserendino. 2019. "Ecological drivers of alpha and beta diversity of freshwater invertebrates in arid and semiarid Patagonia (Argentina)." *Science of the Total Environment* 678: 62-73.
- Epele, L.B., D.A. Dos Santos, R. Sarremejane, M.G. Grech, P.A. Macchi, L.M. Manzo, M.L. Miserendino, N. Bonada, and M. Canedo-Arguelles. 2021. "Blowin' in the wind: Wind directionality affects wetland invertebrate metacommunities in Patagonia." *Global Ecology and Biogeography* 30(6): 1191-1203.
- Epele, L.B., and M. Laura Miserendino. 2016. "Temporal dynamics of invertebrate and aquatic plant communities at three intermittent ponds in livestock grazed Patagonian wetlands." *Journal of Natural History* 50: 711-730.
- Euliss, N.H., D.M. Mushet, W.E. Newton, C.R.V. Otto, R.D. Nelson, J.W. Labaugh, E.J. Scherff, and D.O. Rosenberry. 2014. "Placing prairie pothole wetlands along spatial and temporal continua to improve integration of wetland function in ecological investigations." *Journal of Hydrology* 513: 490-503.
- Evans, J.P., K.K. Cecala, B.R. Scheffers, C.A. Oldfield, N.A. Hollingshead, D.G. Haskell, and B.A. McKenzie. 2017. "Widespread Degradation of a Vernal Pool Network in the Southeastern United States: Challenges to Current and Future Management." *Wetlands* 37(6): 1093-1103.
- Evenson, G.R., H.E. Golden, C.R. Lane, and E. D'Amico. 2015. "Geographically isolated wetlands and watershed hydrology: A modified model analysis." *Journal of Hydrology* 529: 240-256.
- Evenson, G.R., H.E. Golden, C.R. Lane, and E. D'Amico. 2016. "An improved representation of geographically isolated wetlands in a watershed-scale hydrologic model." *Hydrological Processes* 30 (22): 4168-4184.
- Evenson, G.R., H.E. Golden, C.R. Lane, D.L. McLaughlin, and E. D'Amico. 2018. "Depressional Wetlands Affect Watershed Hydrological, Biogeochemical, and Ecological Functions." *Ecological Applications* 28 (4): 953-966.

- Evenson, G.R., C.N. Jones, D.L. McLaughlin, H.E. Golden, C.R. Lane, B. DeVries, L.C. Alexander, M.W. Lang, G.W. McCarty, and A. Sharifi. 2018. "A watershed-scale model for depressional wetland-rich landscapes." *Journal of Hydrology X* 1.
- Fehling, A., S. Gaffield, and S. Laubach. 2014. "Using enhanced wetlands for nitrogen removal in an agricultural watershed." *Journal of Soil and Water Conservation* 69(5).
- Fergus, C., J.F. Lapierre, S.K. Oliver, N.K. Skaff, K.S. Cheruvelil, K. Webster, C. Scott, and P. Soranno. 2017. "The freshwater landscape: lake, wetland, and stream abundance and connectivity at macroscales." *Ecosphere* 8(8).
- Ferlatte, M., A. Quillet, M. Larocque, V. Cloutier, S. Pellerin, and C. Paniconi. 2015. "Aquifer-peatland connectivity in southern Quebec (Canada)." *Hydrological Processes* 29(11): 2600-2612.
- Finlayson, C.M., and R.C. Gardner. 2021. "Ten key issues from the Global Wetland Outlook for decision makers." *Marine and Freshwater Research* 72(3): 301-310.
- Fissore, C., E.A. Nater, K.J. McFarlane, and A.S. Klein. 2019. "Decadal carbon decomposition dynamics in three peatlands in Northern Minnesota." *Biogeochemistry* 145: 63-79.
- Flint, S.A., and W.H. McDowell. 2015. "Effects of headwater wetlands on dissolved nitrogen and dissolved organic carbon concentrations in a suburban New Hampshire watershed." *Freshwater Science* 34(2): 456-471.
- Florencio, M., L. Serrano, P. Siljestrom, R. Fernandez-Zamudio, P. Garcia-Murillo, and C. Diaz-Paniagua. 2014. "The influence of geomorphology on the composition of aquatic flora and fauna within a temporary pond network." *Limnetica* 33(2): 327-339.
- Fossey, M., and A.N. Rousseau. 2016. "Assessing the long-term hydrological services provided by wetlands under changing climate conditions: A case study approach of a Canadian watershed." *Journal of Hydrology* 541: 1287-1302.
- Fossey, M., and A.N. Rousseau. 2016. "Can isolated and riparian wetlands mitigate the impact of climate change on watershed hydrology? A case study approach." *Journal of Environmental Management* 184(2): 327-339.
- Fossey, M., A.N. Rousseau, and S. Savary. 2016. "Assessment of the impact of spatio-temporal attributes of wetlands on stream flows using a hydrological modelling framework: a theoretical case study of a watershed under temperate climatic conditions." *Hydrological Processes* 30 (11): 1768-1781.
- Gardner, R.C., E. Okuno, S. Tai, M.S. Fennessy, C.A. Johnston, M.L. Otte, M. Palmer, J.E. Perry, C. Simenstad, B.R. Tanner, D. Tufford, R.E. Turner, K. Work, S.C. Yaich, and J.B. Zedler. 2019.
 "Advocating for Science: Amici Curiae Brief of Wetland and Water Scientists in Support of the Clean Water Rule." *Wetlands* 39(3): 403-414.
- Gillespie, C.R., and J.J. Fontaine. 2017. "Shorebird Stopover Habitat Decisions in a Changing Landscape." *Journal of Wildlife Management* 81(6): 1051-1062.

- Giosa, E., C. Mammides, and S. Zotos. 2018. "The importance of artificial wetlands for birds: A case study from Cyprus." *PLoS ONE* 13(5).
- Glennon, M.J., S.F. Langdon, M.A. Rubenstein, and M.S. Cross. 2019. "Relative contribution of climate and non-climate drivers in determining dynamic rates of boreal birds at the edge of their range." *PLoS ONE* 14(10).
- Golden, H.E., I.F. Creed, G. Ali, N.B. Basu, B.P. Neff, M.C. Rains, D.L. McLaughlin, L.C. Alexander, A.A. Ameli, J.R. Christensen, G.R. Evenson, C.N. Jones, C.R. Lane, and M.G. Lang. 2017.
 "Integrating geographically isolated wetlands into land management decisions." *Frontiers in Ecology and the Environment* 15(6): 319-327.
- Golden, H.E., C.R. Lane, D.M. Amatya, K.W. Bandilla, H.R. Kiperwas, C.D. Knightes, and H. Ssegane.
 2014. "Hydrologic connectivity between geographically isolated wetlands and surface water systems: A review of select modeling methods." *Environmental Modelling and Software* 53: 190-206.
- Golden, H.E., A. Rajib, C.R. Lane, J.R. Christensen, Q. Wu, and S. Mengistu. 2019. "Non-floodplain Wetlands Affect Watershed Nutrient Dynamics: A Critical Review." *Environmental Science & Technology* 53(13): 7203-7214.
- Golden, H.E., H.A. Sander, C.R. Lane, C. Zhao, K. Price, E. D'Amico, and J.R. Christensen. 2016. "Relative effects of geographically isolated wetlands on streamflow: a watershed-scale analysis." *Ecohydrology* 9(1): 21-38.
- Gould, W.R., A.M. Ray, L.L. Bailey, D. Thoma, R. Daley, and K. Legg. 2019. "Multistate occupancy modeling improves understanding of amphibian breeding dynamics in the Greater Yellowstone Area." *Ecological Applications* 29(1).
- Gramlich, A., S. Stoll, C. Stamm, T. Walter, and V. Prasuhn. 2018. "Effects of artificial land drainage on hydrology, nutrient and pesticide fluxes from agricultural fields A review." *Agriculture, Ecosystems & Environment* 266: 84-99.
- Grand, L.A., M.P. Hayes, K.A. Vogt, D.J. Vogt, P.R. Yarnold, K.O. Richter, C.D. Anderson, E.C. Ostergaard, and J.O. Wilhelm. 2017. "Identification of habitat controls on northern red-legged frog populations: implications for habitat conservation on an urbanizing landscape in the Pacific Northwest." *Ecological Processes* 6.
- Green, D.I.S., S.M. McDeid, and W.G. Crumpton. 2019. "Runoff Storage Potential of Drained Upland Depressions on the Des Moines Lobe of Iowa." *Journal of the American Water Resources Association* 55 (3): 543-558.
- Greenberg, C.H., S. Goodrick, J.D. Austin, and B.R. Parresol. 2015. "Hydroregime Prediction Models for Ephemeral Groundwater-Driven Sinkhole Wetlands: a Planning Tool for Climate Change and Amphibian Conservation." *Wetlands* 35(5): 899-911.
- Greenberg, C.H., S.J. Zarnoch, and J.D. Austin. 2017. "Weather, hydroregime, and breeding effort influence juvenile recruitment of anurans: implications for climate change." *Ecosphere* 8(5).
- Grimm, K., and X. Chu. 2018. "Modeling of spatiotemporal variations in runoff contribution areas and analysis of hydrologic connectivity." *Land Degradation and Development* 29(8): 2629-2643.
- Grimm, K., and X. Chu. 2020. "Depression threshold control proxy to improve HEC-HMS modeling of depression-dominated watersheds." *Hydrological Sciences Journal* 65(2): 200-211.
- Groff, L.A., C.S. Loftin, and A.J.K. Calhoun. 2017. "Predictors of breeding site occupancy by amphibians in montane landscapes." *The Journal of Wildlife Management* 81(2): 269-278.
- Guan, Q., J. Liu, D.P. Batzer, X. Lu, and H. Wu. 2018. "Snails (Mollusca: Gastropoda) as potential surrogates of overall aquatic invertebrate assemblage in wetlands of Northeastern China." *Ecological Indicators* 90: 193-200.
- Guareschi, S., A. Laini, P. Viaroli, and R. Bolpagni. 2020. "Integrating habitat- and species-based perspectives for wetland conservation in lowland agricultural landscapes." *Biodiversity and Conservation* 29(1): 153-171.
- Gulbin, S., A.P. Kirilenko, G. Kharel, and X. Zhang. 2019. "Wetland loss impact on long term flood risks in a closed watershed." *Environmental Science & Policy* 94: 112-122.
- Habtezion, N., M.T. Nasab, and X. Chu. 2016. "How does DEM resolution affect microtopographic characteristics, hydrologic connectivity, and modelling of hydrologic processes?" *Hydrological Processes* 30(25): 4870-4892.
- Haig, S.M., S.P. Murphy, J.H. Matthews, I. Arismendi, and M. Safeeq. 2019. "Climate-Altered Wetlands Challenge Waterbird Use and Migratory Connectivity in Arid Landscapes." *Scientific Reports* 9(1).
- Hall, B.D., R.H. Hesslein, C.A. Emmerton, S.N. Higgins, P. Ramlal, and M.J. Paterson. 2019. "Multidecadal carbon sequestration in a headwater boreal lake." *Limnology and Oceanography* 64.
- Hansen, A., C.L. Dolph, and J.C. Finlay. 2016. "Do wetlands enhance downstream denitrification in agricultural landscapes?" *Ecosphere* 7(10).
- Hansen, A., C.L. Dolph, E. Foufoula-Georgiou, and J.C. Finlay. 2018. "Contribution of wetlands to nitrate removal at the watershed scale." *Nature Geoscience* 11(2).
- Haque, A., G. Ali, and P. Badiou. 2018. "Hydrological dynamics of prairie pothole wetlands: Dominant processes and landscape controls under contrasted conditions." *Hydrological Processes* 32(15): 2405-2422.
- Haque, A., G. Ali, M. Macrae, P. Badiou, and D. Lobb. 2018. "Hydroclimatic influences and physiographic controls on phosphorus dynamics in prairie pothole wetlands." *Science of the Total Environment* 645: 1410-1424.
- Hayashi, M., G. van der Kamp, and D.O. Rosenberry. 2016. "Hydrology of Prairie Wetlands: Understanding the Integrated Surface-Water and Groundwater Processes." *Wetlands* 36(2): 237-254.
- Henry, B.L., J.S. Wesner, and J.L. Kerby. 2020. "Cross-Ecosystem Effects of Agricultural Tile Drainage, Surface Runoff, and Selenium in the Prairie Pothole Region." *Wetlands* 40(3): 527-538.

- Herteux, C.E., D.E. Gawlik, and L.L. Smith. 2020. "Habitat Characteristics Affecting Wading Bird Use of Geographically Isolated Wetlands in the US Southeastern Coastal Plain." *Wetlands* 40(5): 1149-1159.
- Herve, P., S. Grellier, S.D. Tiegs, K.M. Wantzen, and F. Isselin-Nondedeu. 2020. "Monitoring organicmatter decomposition and environmental drivers in restored vernal pools." *Wetlands Ecology and Management* 28(6): 937-952.
- Herve, P., S.D. Tiegs, S. Grellier, K.M. Wantzen, and F. Isselin-Nondedeu. 2019. "Combined Effects of Vegetation and Drought on Organic-Matter Decomposition in Vernal Pool Soils." *Wetlands* 39(2): 321-327.
- Hidayat, H., A.J. Teuling, B. Vermeulen, M. Taufik, K. Kastner, T.J. Geertsema, D.C.C. Bol, D.H.
 Hoekman, G.S. Haryani, H.A.J. Van Lanen, R.M. Delinom, R. Dijksma, G.Z. Anshari, N.S. Ningsih,
 R. Uijlenhoet, and A.J.F. Hoitink. 2017. "Hydrology of inland tropical lowlands: the Kapuas and
 Mahakam wetlands." *Hydrology and Earth System Sciences* 21(5): 2579-2594.
 https://doi.org/10.5194/hess-21-2579-2017.
- Hill, B.H., T.M. Jicha, L.L.P. Lehto, C.M. Elonen, S.D. Sebestyen, and R.K. Kolka. 2016. "Comparisons of soil nitrogen mass balances for an ombrotrophic bog and a minerotrophic fen in northern Minnesota." *Science of the Total Environment* 550: 880-892.
- Hill, M.J., J. Biggs, I. Thornhill, R.A. Briers, M. Ledger, D.G. Gledhill, P.J. Wood, and C. Hassall. 2018.
 "Community heterogeneity of aquatic macroinvertebrates in urban ponds at a multi-city scale." *Landscape Ecology* 33(3): 389-405.
- Ho, J., E.H. Boughton, D.G. Jenkins, G. Sonnier, P.J. Bohlen, and L.G. Chambers. 2018. "Ranching practices interactively affect soil nutrients in subtropical wetlands." *Agriculture, Ecosystems and Environment*.
- Hokanson, K.J., E.S. Peterson, K.J. Devito, and C.A. Mendoza. 2020. "Forestland-peatland hydrologic connectivity in water-limited environments: hydraulic gradients often oppose topography." *Environmental Research Letters* 15(3).
- Holgerson, M.A., D.M. Post, and D.K. Skelly. 2016. "Reconciling the role of terrestrial leaves in pond food webs: a whole-ecosystem experiment." *Ecology* 97(7): 1771-1782.
- Hood, G.A., and D.G. Larson. 2015. "Ecological engineering and aquatic connectivity: a new perspective from beaver-modified wetlands." *Freshwater Biology* 60(1): 198-208.
- Hornbach, D.J., M.C. Hove, M.W. Ensley-Field, M.R. Glasenapp, I. Goodbar, J.D. Harman, B.D. Huber, E.A. Kangas, K.X. Liu, M. Stark-Ragsdale, and L.K. Tran. 2017. "Comparison of ecosystem processes in a woodland and prairie pond with different hydroperiods." *Journal of Freshwater Ecology* 32.
- Hosen, J.D., A.W. Armstrong, and M.A. Palmer. 2018. "Dissolved organic matter variations in coastal plain wetland watersheds: The integrated role of hydrological connectivity, land use, and seasonality." *Hydrological Processes* 32(11): 1664-1681.

- Howell, N.L., E.B. Butler, and B. Guerrero. 2019. "Water quality variation with storm runoff and evaporation in playa wetlands." *Science of the Total Environment* 652: 583-592.
- Ishiyama, N., I. Koizumi, T. Yuta, and F. Nakamura. 2015. "Differential effects of spatial network structure and scale on population size and genetic diversity of the ninespine stickleback in a remnant wetland system." *Freshwater Biology* 60(4): 733-744.
- Janke, A.K., M.J. Anteau, and J.D. Stafford. 2019. "Prairie wetlands confer consistent migrant refueling conditions across a gradient of agricultural land use intensities." *Biological Conservation* 229: 99-112.
- Javaheri, A., and M. Babbar-Sebens. 2014. "On comparison of peak flow reductions, flood inundation maps, and velocity maps in evaluating effects of restored wetlands on channel flooding." *Ecological Engineering* 73: 132-145.
- Jeliazkov, A., F. Chiron, J. Garnier, A. Besnard, M. Silvestre, and F. Jiguet. 2014. "Level-dependence of the relationships between amphibian biodiversity and environment in pond systems within an intensive agricultural landscape." *Hydrobiologia* 723(1): 7-23.
- Johnston, C.A., and N.E. McIntyre. 2019. "Effects of cropland encroachment on prairie pothole wetlands: numbers, density, size, shape, and structural connectivity." *Landscape Ecology* 34(4): 827-841.
- Jones, C.N., A. Ameli, B.P. Neff, G.R. Evenson, D.L. McLaughlin, H.E. Golden, and C.R. Lane. 2019. "Modeling Connectivity of Non-floodplain Wetlands: Insights, Approaches, and Recommendations." *Journal of the American Water Resources Association* 55(3): 559-577.
- Jones, C.N., G.R. Evenson, D.L. McLaughlin, M.K. Vanderhoof, M.W. Lang, G.W. McCarty, H.E. Golden, C.R. Lane, and L.C. Alexander. 2018. "Estimating restorable wetland water storage at landscape scales." *Hydrological Processes* 32 (2): 305-313.
- Jones, C.N., D.L. McLaughlin, K. Henson, C.A. Haas, and D.A. Kaplan. 2018. "From salamanders to greenhouse gases: does upland management affect wetland functions?" *Frontiers in Ecology and the Environment* 16 (1): 14-19.
- Kadykalo, A.N., and C.S. Findlay. 2016. "The flow regulation services of wetlands." *Ecosystem Services* 20: 91-103.
- Kalcic, M., W. Crumpton, X. Liu, J. D'Ambrosio, A. Ward, and J. Witter. 2018. "Assessment of beyondthe-field nutrient management practices for agricultural crop systems with subsurface drainage." *Journal of Soil and Water Conservation* 73(1): 62-74.
- Kappas, I., G. Mura, D. Synefiaridou, F. Marrone, G. Alfonso, M. Alonso, and T.J. Abatzopoulos. 2017.
 "Molecular and morphological data suggest weak phylogeographic structure in the fairy shrimp *Streptocephalus torvicornis* (Branchiopoda, Anostraca)." *Hydrobiologia* 801(1): 21-32.
- Karaouzas, I., E. Dimitriou, A. Lampou, and E. Colombari. 2015. "Seasonal and spatial patterns of macroinvertebrate assemblages and environmental conditions in Mediterranean temporary ponds in Greece." *Limnology* 16(1): 41-53. <u>https://doi.org/10.1007/s10201-014-0437-0</u>.

- Karim, F., S. Marvanek, L.E. Merrin, D. Nielsen, J. Hughes, D. Stratford, and C. Pollino. 2020. "Modelling Flood-Induced Wetland Connectivity and Impacts of Climate Change and Dam." *Water* 12(5).
- Kayler, Z.E., M. Badrian, A. Frackowski, H. Rieckh, K.N. Nitzsche, T. Kalettka, C. Merz, and A. Gessler. 2018. "Ephemeral kettle hole water and sediment temporal and spatial dynamics within an agricultural catchment." *Ecohydrology* 11(2). <u>https://doi.org/10.1002/eco.1929</u>.
- Kazanjian, G., S. Flury, K. Attermeyer, T. Kalettka, A. Kleeberg, K. Premke, J. Köhler, and S. Hilt. 2018. "Primary production in nutrient-rich kettle holes and consequences for nutrient and carbon cycling." *Hydrobiologia* 806: 77-93. <u>https://doi.org/10.1007/s10750-017-3337-6</u>.
- Kessler, A.C., and S.C. Gupta. 2015. "Drainage Impacts on Surficial Water Retention Capacity of a Prairie Pothole Watershed." *Journal of the American Water Resources Association* 51(4): 1101-1113.
- Kifner, L.H., A.J.K. Calhoun, S.A. Norton, K.E. Hoffmann, and A. Amirbahman. 2018. "Methane and carbon dioxide dynamics within four vernal pools in Maine, USA." *Biogeochemistry* 139.
- Kim, B., and J. Park. 2021. "Random ecological networks that depend on ephemeral wetland complexes (vol 156, 105972, 2020)." *Ecological Engineering* 159.
- King, S.L., M.K. Laubhan, P. Tashjian, J. Vradenburg, and L. Fredrickson. 2021. "Wetland Conservation: Challenges Related to Water Law and Farm Policy." *Wetlands* 41(5): 54. <u>https://doi.org/10.1007/s13157-021-01449-y</u>.
- Kingsford, R.T., A. Basset, and L. Jackson. 2016. "Wetlands: conservation's poor cousins." *Aquatic Conservation: Marine and Freshwater Ecosystems* 26(5): 892-916.
- Kiss, J., and A. Bedard-Haughn. 2021. "Predictive Mapping of Solute-rich Wetlands in the Canadian Prairie Pothole Region Through High-resolution Digital Elevation Model Analyses." *Wetlands* 41(3).
- Kissel, A.M., M. Halabisky, R.D. Scherer, M.E. Ryan, and E.C. Hansen. 2020. "Expanding wetland hydroperiod data via satellite imagery for ecological applications." *Frontiers in Ecology and the Environment* 18(8): 432-438.
- Klammler, H., C.J. Quintero, J.W. Jawitz, D.L. McLaughlin, and M.J. Cohen. 2020. "Local Storage Dynamics of Individual Wetlands Predict Wetlandscape Discharge." *Water Resources Research* 56(11).
- Kohnert, K., B. Juhls, S. Muster, S. Antonova, A. Serafimovich, S. Metzger, J. Hartmann, and T. Sachs. 2018. "Toward understanding the contribution of waterbodies to the methane emissions of a permafrost landscape on a regional scale—A case study from the Mackenzie Delta, Canada." *Global Change Biology* 24(9): 3976-3989.
- Kroeger, I., M. Liess, and S. Duquesne. 2014. "Temporal and spatial habitat preferences and biotic interactions between mosquito larvae and antagonistic crustaceans in the field." *Journal of Vector Ecology* 39(1): 103-11. https://www.ncbi.nlm.nih.gov/pubmed/24820562.

- LaBaugh, J.W., D.M. Mushet, D.O. Rosenberry, N.H. Euliss, M.B. Goldhaber, C.T. Mills, and R.D. Nelson. 2016. "Changes in Pond Water Levels and Surface Extent Due to Climate Variability Alter Solute Sources to Closed-Basin Prairie-Pothole Wetland Ponds, 1979 to 2012." Wetlands 36(2): 343-355.
- LaBaugh, J.W., D.O. Rosenberry, D.M. Mushet, B.P. Neff, R.D. Nelson, and N.H. Euliss. 2018. "Longterm changes in pond permanence, size, and salinity in Prairie Pothole Region wetlands: The role of groundwater-pond interaction." *Journal of Hydrology: Regional Studies* 17: 1-23.
- Lane, C.R., and B.C. Autrey. 2016. "Phosphorus retention of forested and emergent marsh depressional wetlands in differing land uses in Florida, USA." *Wetlands Ecology and Management* 24(1): 45-60.
- Lane, C.R., and B.C. Autrey. 2017. "Sediment accretion and accumulation of P, N and organic C in depressional wetlands of three ecoregions of the United States." *Marine & Freshwater Research*. 68(12): 2253-2265. https://www.ncbi.nlm.nih.gov/pubmed/30505203.
- Lane, C.R., B.C. Autrey, T. Jicha, L. Lehto, C. Elonen, and L. Seifert-Monson. 2015. "Denitrification Potential in Geographically Isolated Wetlands of North Carolina and Florida, USA." *Wetlands* 35(3): 459-471.
- Lane, C.R., and E. D'Amico. 2016. "Identification of Putative Geographically Isolated Wetlands of the Conterminous United States." *Journal of the American Water Resources Association* 52(3): 705-722.
- Lane, C.R., S.G. Leibowitz, B.C. Autrey, S.D. LeDuc, and L.C. Alexander. 2018. "Hydrological, Physical, and Chemical Functions and Connectivity of Non-Floodplain Wetlands to Downstream Waters: A Review." *Journal of the American Water Resources Association* 54(2): 346-371.
- Lane, D., C.P.R. McCarter, M. Richardson, C. McConnell, T. Field, H.X. Yao, G. Arhonditsis, and C.P.J. Mitchell. 2020. "Wetlands and low-gradient topography are associated with longer hydrologic transit times in Precambrian Shield headwater catchments." *Hydrological Processes* 34(3): 598-614.
- Lange, C.J., D.P. Collins, K.L. Metzger, and B.M. Ballard. 2018. "Predicting impacts of sea level rise on wintering redhead ducks along the lower Texas Coast." *Global Ecology and Conservation* 16.
- Lannoo, M.J., R.M. Stiles, M.A. Sisson, J.W. Swan, V.C.K. Terrell, and K.E. Robinson. 2017. "Patch Dynamics Inform Management Decisions in a Threatened Frog Species." *Copeia* 105(1): 53-63.
- Le Cor, F., S. Slaby, V. Dufour, A. Iuretig, C. Feidt, X. Dauchy, and D. Banas. 2021. "Occurrence of pesticides and their transformation products in headwater streams: Contamination status and effect of ponds on contaminant concentrations." *Science of the Total Environment* 788.
- Le Lay, G., S. Angelone, R. Holderegger, C. Flory, and J. Bolliger. 2015. "Increasing Pond Density to Maintain a Patchy Habitat Network of the European Treefrog (*Hyla arborea*)." *Journal of Herpetology* 49(2): 217-221.
- Lee, S., G.W. McCarty, G.E. Moglen, M.W. Lang, C. Nathan Jones, M. Palmer, I.Y. Yeo, M. Anderson, A.M. Sadeghi, and M.C. Rabenhorst. 2020. "Seasonal drivers of geographically isolated wetland hydrology in a low-gradient, Coastal Plain landscape." *Journal of Hydrology* 583.

- Lee, S., G.W. McCarty, G.E. Moglen, M.W. Lang, A.M. Sadeghi, T.R. Green, I.Y. Yeo, and M.C. Rabenhorst. 2019. "Effects of subsurface soil characteristics on wetland-groundwater interaction in the coastal plain of the Chesapeake Bay watershed." *Hydrological Processes* 33(2): 305-315.
- Lee, S., H. Yen, I.Y. Yeo, G.E. Moglen, M.C. Rabenhorst, and G.W. McCarty. 2020. "Use of multiple modules and Bayesian Model Averaging to assess structural uncertainty of catchment-scale wetland modeling in a Coastal Plain landscape." *Journal of Hydrology* 582: 124544.
- Lee, S., I.Y. Yeo, M.W. Lang, A.M. Sadeghi, G.W. McCarty, G.E. Moglen, and G.R. Evenson. 2018. "Assessing the cumulative impacts of geographically isolated wetlands on watershed hydrology using the SWAT model coupled with improved wetland modules." *Journal of Environmental Management* 223: 37-48.
- Lee, S.Y., M.E. Ryan, A.F. Hamlet, W.J. Palen, J.J. Lawler, and M. Halabisky. 2015. "Projecting the Hydrologic Impacts of Climate Change on Montane Wetlands." *PLoS ONE* 10(9).
- Leibowitz, S.G. 2015. "Geographically isolated wetlands: why we should keep the term." *Wetlands* 35: 997-1003.
- Leibowitz, S.G., D.M. Mushet, and W.E. Newton. 2016. "Intermittent Surface Water Connectivity: Fill and Spill Vs. Fill and Merge Dynamics." *Wetlands* 36(2): 323-342.
- Leibowitz, S.G., P.J. Wigington, K.A. Schofield, L.C. Alexander, M.K. Vanderhoof, and H.E. Golden.
 2018. "Connectivity of Streams and Wetlands to Downstream Waters: An Integrated Systems Framework." *Journal of the American Water Resources Association* 54(2): 298-322.
- Leon, A.S., Y. Tang, D. Chen, A. Yolcu, C. Glennie, and S.C. Pennings. 2018. "Dynamic Management of Water Storage for Flood Control in a Wetland System: A Case Study in Texas." *Water* 10(3).
- Levy, Z.F., D.O. Rosenberry, R. Moucha, D.M. Mushet, M.B. Goldhaber, J.W. Labaugh, A.J. Fiorentino, and D.I. Siegel. 2018. "Drought-induced recharge promotes long-term storage of porewater salinity beneath a prairie wetland." *Journal of Hydrology* 557: 391-406.
- Lewis, D.B., and S.J. Feit. 2015. "Connecting carbon and nitrogen storage in rural wetland soil to groundwater abstraction for urban water supply." *Global Change Biology* 21(4): 1704-1714.
- Lewis-Phillips, J., S.J. Brooks, C.D. Sayer, I.R. Patmore, G.M. Hilton, A. Harrison, H. Robson, and J.C. Axmacher. 2020. "Ponds as insect chimneys: Restoring overgrown farmland ponds benefits birds through elevated productivity of emerging aquatic insects." *Biological Conservation* 241.
- Li, Y., Y. Wu, A. Wright, J. Xu, H. Liu, G. Wang, and C. Wang. 2020. "Integrated factor analysis of water level variation in geographically isolated ponds." *Environmental Science and Pollution Research* 27(31): 38861-38870.
- Lischeid, G., T. Kalettka, M. Hollaender, J. Steidl, C. Merz, R. Dannowski, T. Hohenbrink, C. Lehr, G. Onandia, F. Reverey, and M. Paetzig. 2018. "Natural ponds in an agricultural landscape: External drivers, internal processes, and the role of the terrestrial-aquatic interface." *Limnologica* 68: 5-16.

- Liu, G., F.W. Schwartz, C.K. Wright, and N.E. McIntyre. 2016. "Characterizing the Climate-Driven Collapses and Expansions of Wetland Habitats with a Fully Integrated Surface-Subsurface Hydrologic Model." Wetlands 36.
- Liu, Y., B. Cui, J. Du, Q. Wang, S. Yu, Y. Wu, G. Zhang, and A.N. Rousseau. 2020. "Quantitative assessment on basin-scale hydrological services of wetlands." *Science China Earth Sciences* 63(2): 279-291.
- Liu, Y., W. Yang, L. Leon, I. Wong, C. McCrimmon, A. Dove, and P. Fong. 2016. "Hydrologic modeling and evaluation of Best Management Practice scenarios for the Grand River watershed in Southern Ontario." *Journal of Great Lakes Research* 42(6): 1289-1301.
- Lobato-De Magalhaes, T., Y. Rico, D. Cabrera-Toledo, and M. Martinez. 2020. "Plant functional connectivity of *Nymphoides fallax* in geographically isolated temporary wetlands in Mexican highlands." *Aquatic Botany* 164.
- Lozada-Gobilard, S., C. Schwarzer, R. Dyer, R. Tiedemann, and J. Joshi. 2021. "Genetic Diversity and Connectivity in Plant Species Differing in Clonality and Dispersal Mechanisms in Wetland Island Habitats." *Journal of Heredity* 112(1): 108-121.
- Lupon, A., B.A. Denfeld, H. Laudon, J. Leach, J. Karlsson, and R.A. Sponseller. 2019. "Groundwater inflows control patterns and sources of greenhouse gas emissions from streams." *Limnology and Oceanography* 64(4): 1545-1557.
- Maas, C.M., and K. Cockerill. 2021. "Managing Stormwater by Accident: A Conceptual Study." *Water* 13(11).
- Mabidi, A., M.S. Bird, and R. Perissinotto. 2017. "Distribution and diversity of aquatic macroinvertebrate assemblages in a semi-arid region earmarked for shale gas exploration (Eastern Cape Karoo, South Africa)." *PLoS ONE* 12(6).
- Mabidi, A., M.S. Bird, R. Perissinotto, and D.C. Rogers. 2016. "Ecology and distribution of large branchiopods (Crustacea, Branchiopoda, Anostraca, Notostraca, Laevicaudata, Spinicaudata) of the Eastern Cape Karoo, South Africa." *ZooKeys* (618): 15-38.
- Macingo, S.C., K.A. Kormas, A. Oikonomou, and H. Karayanni. 2019. "Taxa-area and distance-decay relationships of unicellular eukaryotes along an elevation gradient of mountainous freshwater ecosystems." *Journal of Plankton Research* 41(6): 821-834.
- Mahoney, D.T., J.F. Fox, and N. Al Aamery. 2018. "Watershed erosion modeling using the probability of sediment connectivity in a gently rolling system." *Journal of Hydrology* 561: 862-883.
- Main, A.R., N.L. Michel, M.C. Cavallaro, J.V. Headley, K.M. Peru, and C.A. Morrissey. 2016.
 "Snowmelt transport of neonicotinoid insecticides to Canadian Prairie wetlands." *Agriculture, Ecosystems and Environment* 215: 76-84.
- Malaj, E., K. Liber, and C.A. Morrissey. 2020. "Spatial distribution of agricultural pesticide use and predicted wetland exposure in the Canadian Prairie Pothole Region." *Science of the Total*

Environment 718.

- Malzone, J.M., E.G. Sweet, A.C. Bell, and G.L. Minzenberger. 2020. "Geomorphic controls of perched groundwater interaction with natural ridge-top depressional wetlands." *Hydrological Processes* 34(4): 1089-1100.
- Martin, A.R., M.L. Soupir, and A.L. Kaleita. 2019. "SEASONAL AND INTRA-EVENT NUTRIENT LEVELS IN FARMED PRAIRIE POTHOLES OF THE DES MOINES LOBE." *Transactions of the ASABE* 62(6): 1607-1617.
- Martina, J.P., S.K. Hamilton, M.R. Turetsky, and C.J. Phillippo. 2014. "Organic matter stocks increase with degree of invasion in temperate inland wetlands." *Plant and Soil* 385: 107-123.
- Marton, J.M., I.F. Creed, D.B. Lewis, C.R. Lane, N.B. Basu, M.J. Cohen, and C.B. Craft. 2015. "Geographically Isolated Wetlands are Important Biogeochemical Reactors on the Landscape." *BioScience* 65(4): 408-418.
- Marton, J.M., M.S. Fennessy, and C.B. Craft. 2014. "Functional Differences between Natural and Restored Wetlands in the Glaciated Interior Plains." *Journal of Environmental Quality* 43(1): 409-417.
- Marton, J.M., M.S. Fennessy, and C.B. Craft. 2014. "USDA conservation practices increase carbon storage and water quality improvement functions: An example from Ohio." *Restoration Ecology* 22(1): 117-124.
- Matos, C., S.O. Petrovan, P.M. Wheeler, and A. Ward. 2019. "Landscape connectivity and spatial prioritization in an urbanising world: A network analysis approach for a threatened amphibian." *Biological Conservation* 237: 238-247.
- Matsuzaki, S.I.S., A. Kohzu, T. Kadoya, M. Watanabe, T. Osawa, K. Fukaya, K. Komatsu, N. Kondo, H. Yamaguchi, H. Ando, K. Shimotori, M. Nakagawa, T. Kizuka, A. Yoshioka, T. Sasai, N. Saigusa, B. Matsushita, and N. Takamura. 2019. "Role of wetlands in mitigating the trade-off between crop production and water quality in agricultural landscapes." *Ecosphere* 10(11).
- Matveev, A., I. Laurion, B.N. Deshpande, N. Bhiry, and W.F. Vincent. 2016. "High methane emissions from thermokarst lakes in subarctic peatlands." *Limnology and Oceanography* 61.
- Matveev, A., I. Laurion, and W.F. Vincent. 2018. "Methane and carbon dioxide emissions from thermokarst lakes on mineral soils." *Arctic Science* 4(4): 584-604.
- Matveev, A., I. Laurion, and W.F. Vincent. 2019. "Winter Accumulation of Methane and its Variable Timing of Release from Thermokarst Lakes in Subarctic Peatlands." *Journal of Geophysical Research: Biogeosciences* 124(11): 3521-3535.
- Mazurczyk, T., and R.P. Brooks. 2018. "Carbon storage dynamics of temperate freshwater wetlands in Pennsylvania." *Wetlands Ecology and Management* 26(5): 893-914.

McCaffery, R.M., L.A. Eby, B.A. Maxell, and P.S. Corn. 2014. "Breeding site heterogeneity reduces

variability in frog recruitment and population dynamics." Biological Conservation 170: 169-176.

- McCauley, L.A., M.J. Anteau, and M. Post van Der Burg. 2015. "Consolidation Drainage and Climate Change May Reduce Piping Plover Habitat in the Great Plains." *Journal of Fish and Wildlife Management* 7(1): 4-13. https://doi.org/10.3996/072015-JFWM-068.
- McCauley, L.A., M.J. Anteau, M. Post van Der Burg, and M.T. Wiltermuth. 2015. "Land use and wetland drainage affect water levels and dynamics of remaining wetlands." *Ecosphere* 6(6): 92. <u>https://doi.org/10.1890/ES14-00494.1</u>.
- McClellan, M., X. Comas, B. Benscoter, R. Hinkle, and D. Sumner. 2017. "Estimating Belowground Carbon Stocks in Isolated Wetlands of the Northern Everglades Watershed, Central Florida, Using Ground Penetrating Radar and Aerial Imagery." *Journal of Geophysical Research: Biogeosciences* 122(11): 2804-2816.
- McDonough, O.T., M.W. Lang, J.D. Hosen, and M.A. Palmer. 2015. "Surface hydrologic connectivity between Delmarva bay wetlands and nearby streams along a gradient of agricultural alteration." *Wetlands* 35(1): 41-53.
- McFarland, E.K., M. Laforgia, M. Yepsen, D.F. Whigham, A.H. Baldwin, and M. Lang. 2016. "Plant biomass and nutrients (C, N and P) in natural, restored and prior converted depressional wetlands in the Mid-Atlantic Coastal Plain, U. S." *Folia Geobotanica* 51(3): 267-283.
- McIntyre, N.E., S.D. Collins, L.J. Heintzman, S.M. Starr, and N. van Gestel. 2018. "The challenge of assaying landscape connectivity in a changing world: A 27-year case study in the southern Great Plains (USA) playa network." *Ecological Indicators* 91: 607-616.
- McIntyre, N.E., G. Liu, J. Gorzo, C.K. Wright, G.R. Guntenspergen, and F. Schwartz. 2019. "Simulating the effects of climate variability on waterbodies and wetland-dependent birds in the Prairie Pothole Region." *Ecosphere* 10(4).
- McIntyre, N.E., C.K. Wright, S. Swain, K. Hayhoe, G. Liu, F.W. Schwartz, and G.M. Henebry. 2014. "Climate forcing of wetland landscape connectivity in the Great Plains." *Frontiers in Ecology and the Environment* 12(1): 59-64.
- McKee, A.M., J.C. Maerz, L.L. Smith, and T.C. Glenn. 2017. "Habitat predictors of genetic diversity for two sympatric wetland-breeding amphibian species." *Ecology and Evolution* 7(16): 6271-6283.
- McKenna, O.P., S.R. Kucia, D.M. Mushet, M.J. Anteau, and M.T. Wiltermuth. 2019. "Synergistic Interaction of Climate and Land-Use Drivers Alter the Function of North American, Prairie-Pothole Wetlands." *Sustainability* 11(23).
- McKenna, O.P., D.M. Mushet, D.O. Rosenberry, and J.W. LaBaugh. 2017. "Evidence for a climateinduced ecohydrological state shift in wetland ecosystems of the southern Prairie Pothole Region." *Climatic Change* 145: 273-287.
- McKenna, O.P., J.M. Osorio, K.D. Behrman, L. Doro, and D.M. Mushet. 2020. "Development of a novel framework for modeling field-scale conservation effects of depressional wetlands in agricultural

landscapes." Journal of Soil and Water Conservation 75(6): 695-703.

- McKenna, O.P., D.A. Renton, D.M. Mushet, and E.S. Dekeyser. 2021. "Upland burning and grazing as strategies to offset climate-change effects on wetlands." *Wetlands Ecology and Management* 29(2): 193-208.
- McKenna, O.P., and O.E. Sala. 2016. "Biophysical controls over concentration and depth distribution of soil organic carbon and nitrogen in desert playas." *Journal of Geophysical Research: Biogeosciences* 121(12): 3019-3029.
- McKenna, O.P., and O.E. Sala. 2018. "Playa-Wetlands Effects on Dryland Biogeochemistry: Space and Time Interactions." *Journal of Geophysical Research: Biogeosciences* 123(6): 1879-1887.
- McLaughlin, D.L., J.S. Diamond, C. Quintero, J. Heffernan, and M.J. Cohen. 2019. "Wetland Connectivity Thresholds and Flow Dynamics From Stage Measurements." *Water Resources Research* 55 (7): 6018-6032.
- McLaughlin, D.L., D.A. Kaplan, and M.J. Cohen. 2014. "A significant nexus: Geographically isolated wetlands influence landscape hydrology." *Water Resources Research* 50(9): 7153-7166.
- McLean, K.I., D.M. Mushet, and C.A. Stockwell. 2016. "From 'Duck Factory' to 'Fish Factory': Climate Induced Changes in Vertebrate Communities of Prairie Pothole Wetlands and Small Lakes." *Wetlands* 36.
- Mekonnen, B.A., K.A. Mazurek, and G. Putz. 2016. "Incorporating landscape depression heterogeneity into the Soil and Water Assessment Tool (SWAT) using a probability distribution." *Hydrological Processes* 30(13): 2373-2389.
- Mekonnen, B.A., K.A. Mazurek, and G. Putz. 2017. "Modeling of nutrient export and effects of management practices in a cold-climate prairie watershed: Assiniboine River watershed, Canada." *Agricultural Water Management* 180: 235-251.
- Mekonnen, B.A., A. Nazemi, K.A. Mazurek, A. Elshorbagy, and G. Putz. 2015. "Hybrid modelling approach to prairie hydrology: fusing data-driven and process-based hydrological models." *Hydrological Sciences Journal* 60(9): 1473-1489.
- Meng, B., J.L. Liu, K. Bao, and B. Sun. 2020. "Methodologies and Management Framework for Restoration of Wetland Hydrologic Connectivity: A Synthesis." *Integrated Environmental Assessment and Management* 16(4): 438-451.
- Mengistu, S.G., I.F. Creed, K.L. Webster, E. Enanga, and F.D. Beall. 2014. "Searching for similarity in topographic controls on carbon, nitrogen and phosphorus export from forested headwater catchments." *Hydrological Processes* 28(8): 3201-3216.
- Mengistu, S.G., H.E. Golden, C.R. Lane, J.R. Christensen, M.L. Wine, E. D'Amico, A. Prues, S.G. Leibowitz, J.E. Compton, M.H. Weber, and R.A. Hill. 2020. "Wetland Flowpaths Mediate Nitrogen and Phosphorus Concentrations across the Upper Mississippi River Basin." *Journal of the American Water Resources Association*: 1-18.

- Mengistu, S.G., and C. Spence. 2016. "Testing the ability of a semidistributed hydrological model to simulate contributing area." *Water Resources Research* 52(6): 4399-4415.
- Merken, R., E. Deboelpaep, J. Teunen, S. Saura, and N. Koedam. 2015. "Wetland suitability and connectivity for trans-Saharan migratory waterbirds." *PLoS ONE* 10(8).
- Messer, T.L., T.L. Moore, N. Nelson, L. Ahiablame, E.Z. Bean, C. Boles, S.L. Cook, S.G. Hall, J. McMaine, and D. Schlea. 2021. "CONSTRUCTED WETLANDS FOR WATER QUALITY IMPROVEMENT: A SYNTHESIS ON NUTRIENT REDUCTION FROM AGRICULTURAL EFFLUENTS." *Transactions of the ASABE* 64(2): 625-639.
- Michelson, C., R.G. Clark, and C.A. Morrissey. 2018. "Agricultural land cover does not affect the diet of Tree Swallows in wetland-dominated habitats." *The Condor* 120(4): 751-764.
- Miguel-Chinchilla, L., E. Gonzalez, and F.A. Comin. 2014. "Assessing metal pollution in ponds constructed for controlling runoff from reclaimed coal mines." *Environmental Monitoring and Assessment* 186(8): 5247-5259.
- Miller, J.O., T.F. Ducey, P.W. Brigman, C.O. Ogg, and P.G. Hunt. 2017. "Greenhouse Gas Emissions and Denitrification within Depressional Wetlands of the Southeastern US Coastal Plain in an Agricultural Landscape." *Wetlands* 37(1): 33-43.
- Mimbs, W.H., J.P. Cusaac, L.M. Smith, S.T. McMurry, and J.B. Belden. 2016. "Occurrence of currentuse fungicides and bifenthrin in Rainwater Basin wetlands." *Chemosphere* 159: 275-281.
- Minot, M., A. Besnard, and A. Huste. 2021. "Habitat use and movements of a large dragonfly (Odonata:Anax imperator) in a pond network." *Freshwater Biology* 66(2): 241-255.
- Mischler, J.A., P.G. Taylor, and A.R. Townsend. 2014. "Nitrogen Limitation of Pond Ecosystems on the Plains of Eastern Colorado." *PLoS ONE* 9(5).
- Miyashita, T., M. Yamanaka, and M.H. Tsutsui. 2014. "Distribution and Abundance of Organisms in Paddy-Dominated Landscapes with Implications for Wildlife-Friendly Farming." Pages 45-65 in Social-Ecological Restoration in Paddy-Dominated Landscapes. N. Usio, and T. Miyashita, eds. Ecological Research Monographs. Springer, Tokyo, Japan.
- Moreno, E., C. Perez-Martinez, and J.M. Conde-Porcuna. 2019. "Dispersal of rotifers and cladocerans by waterbirds: seasonal changes and hatching success." *Hydrobiologia* 834(1): 145-162.
- Morison, M.Q., M.L. Macrae, R.M. Petrone, and L. Fishback. 2017. "Seasonal dynamics in shallow freshwater pond-peatland hydrochemical interactions in a subarctic permafrost environment." *Hydrological Processes* 31(2): 462-475.
- Morrison, A.C., A.J. Gold, and M.C. Pelletier. 2016. "Evaluating Key Watershed Components of Low Flow Regimes in New England Streams." *Journal of Environmental Quality* 45(3): 1021-1028.
- Mosquera, G.M., R. Celleri, P.X. Lazo, K.B. Vache, S.S. Perakis, and P. Crespo. 2016. "Combined use of isotopic and hydrometric data to conceptualize ecohydrological processes in a high-elevation tropical

ecosystem." Hydrological Processes 30(17): 2930-2947.

- Muhammad, A., G.R. Evenson, T.A. Stadnyk, A. Boluwade, S.K. Jha, and P. Coulibaly. 2018. "Assessing the Importance of Potholes in the Canadian Prairie Region under Future Climate Change Scenarios." *Water* 10(11).
- Muhammad, A., G.R. Evenson, T.A. Stadnyk, A. Boluwade, S.K. Jha, and P. Coulibaly. 2019. "Impact of model structure on the accuracy of hydrological modeling of a Canadian Prairie watershed." *Journal* of Hydrology-Regional Studies 21: 40-56.
- Muhammad, A., G.R. Evenson, F. Unduche, and T.A. Stadnyk. 2020. "Climate Change Impacts on Reservoir Inflow in the Prairie Pothole Region: A Watershed Model Analysis." *Water* 12(1).
- Murphy, C.M., T.D. Tuberville, J.C. Maerz, and K.M. Andrews. 2016. "Evaporative Water Loss Rates of Four Species of Aquatic Turtles from the Coastal Plain of the Southeastern United States." *Journal of Herpetology* 50(3): 457-463.
- Murphy, K.T., and S.J. Dinsmore. 2018. "Waterbird Use of Sheetwater Wetlands in Iowa's Prairie Pothole Region." *Wetlands* 38(2): 211-219.
- Mushet, D.M., L.C. Alexander, M. Bennett, K. Schofield, J.R. Christensen, G. Ali, A. Pollard, K. Fritz, and M.W. Lang. 2019. "Differing Modes of Biotic Connectivity within Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 55(2): 307-317.
- Mushet, D.M., A.J.K. Calhoun, L.C. Alexander, M.J. Cohen, E.S. DeKeyser, L. Fowler, C.R. Lane, M.W. Lang, M.C. Rains, and S.C. Walls. 2015. "Geographically isolated wetlands: rethinking a misnomer." Wetlands 35.
- Musseau, R., M. Bastianelli, C. Bely, C. Rousselle, and O. Dehorter. 2021. "Using miniaturized GPS archival tags to assess home range features of a small plunge-diving bird: the European Kingfisher (*Alcedo atthis*)." *Avian Research* 12(1).
- Muths, E., L.L. Bailey, B.A. Lambert, and S.C. Schneider. 2018. "Estimating the probability of movement and partitioning seasonal survival in an amphibian metapopulation." *Ecosphere* 9(10).
- Nachshon, U., A. Ireson, G. van Der Kamp, S.R. Davies, Sr., and H.S. Wheater. 2014. "Impacts of climate variability on wetland salinization in the North American prairies." *Hydrology and Earth System Sciences* 18(4): 1251-1263.
- Nagel, L.D., S.A. McNulty, M.D. Schlesinger, and J.P. Gibbs. 2021. "Breeding Effort and Hydroperiod Indicate Habitat Quality of Small, Isolated Wetlands for Amphibians Under Climate Extremes." *Wetlands* 41(2).
- Nasab, M.T., and X. Chu. 2020. "Macro-HyProS: A new macro-scale hydrologic processes simulator for depression-dominated cold climate regions." *Journal of Hydrology* 580.
- Nasab, M.T., J. Zhang, and X. Chu. 2017. "A new depression-dominated delineation (D-cubed) method for improved watershed modelling." *Hydrological Processes* 31(19): 3364-3378.

- Neff, B.P., and D.O. Rosenberry. 2017. "Groundwater Connectivity of Upland-Embedded Wetlands in the Prairie Pothole Region." *Wetlands* 38(1): 51-63.
- Neff, B.P., D.O. Rosenberry, S.G. Leibowitz, D.M. Mushet, H.E. Golden, M.C. Rains, J.R. Brooks, and C.R. Lane. 2020. "A Hydrologic Landscapes Perspective on Groundwater Connectivity of Depressional Wetlands." *Water* 12(1).
- Nitzsche, K.N., T. Kalettka, K. Premke, G. Lischeid, A. Gessler, and Z.E. Kayler. 2017. "Land-use and hydroperiod affect kettle hole sediment carbon and nitrogen biogeochemistry." *Science of The Total Environment* 574: 46-56. <u>https://www.ncbi.nlm.nih.gov/pubmed/27623526</u>.
- Nolan, R.H., P.A. Vesk, and D. Robinson. 2018. "Recovery potential of microwetlands from agricultural land uses." *Ecological Management & Restoration* 19(1): 81-84.
- Noorduijn, S.L., M. Hayashi, G.A. Mohammed, and A.A. Mohammed. 2018. "A Coupled Soil Water Balance Model for Simulating Depression-Focused Groundwater Recharge." *Vadose Zone Journal* 17(1).
- Obrador, B., D. von Schiller, R. Marcé, L. Gómez-Gener, M. Koschorreck, C. Borrego, and N. Catalán. 2018. "Dry habitats sustain high CO2 emissions from temporary ponds across seasons." *Scientific Reports* 8(1).
- O'Bryan, C.J., J.A. Homyack, R.F. Baldwin, Y. Kanno, and A.L. Harrison. 2016. "Novel habitat use supports population maintenance in a reconfigured landscape." *Ecosphere* 7(3).
- Olid, C., A. Zannella, and D.C.P. Lau. 2021. "The Role of Methane Transport From the Active Layer in Sustaining Methane Emissions and Food Chains in Subarctic Ponds." *Journal of Geophysical Research: Biogeosciences* 126(3).
- on, J.B., D.H. Wardrop, E.A.H. Smithwick, K.J. Naithani, M. Clara Arteaga, N. Escudero, J. Gasca-Pineda, R. Bello-Bedoy, and C.M. Guilliams. 2019. "Genetic and phenotypic diversity of *Branchinecta sandiegonensis* (Crustacea: Anostraca) in the vernal pools of Baja California, Mexico." *Zootaxa* 4646(1): 145-163.
- O'Neill, B.J., and J.H. Thorp. 2014. "Untangling food-web structure in an ephemeral ecosystem." *Freshwater Biology* 59(7): 1462-1473.
- Osborn, J.M., H.M. Hagy, M.D. McClanahan, J.B. Davis, and M.J. Gray. 2017. "Habitat Selection and Activities of Dabbling Ducks during Non-Breeding Periods." *Journal of Wildlife Management* 81(8): 1482-1493.
- Owen, R.K., E.B. Webb, D.A. Haukos, and K.W. Goyne. 2020. "Projected climate and land use changes drive plant community composition in agricultural wetlands." *Environmental and Experimental Botany* 175.
- Palta, M.M., N.B. Grimm, and P.M. Groffman. 2017. "Accidental' urban wetlands: ecosystem functions in unexpected places." *Frontiers in Ecology and the Environment* 15(5): 248-256.

- Park, J., D. Wang, and M. Kumar. 2020. "Spatial and temporal variations in the groundwater contributing areas of inland wetlands." *Hydrological Processes* 34(5).
- Pattison-Williams, J.K., J.W. Pomeroy, P. Badiou, and S. Gabor. 2018. "Wetlands, Flood Control and Ecosystem Services in the Smith Creek Drainage Basin: A Case Study in Saskatchewan, Canada." *Ecological Economics* 147: 36-47.
- Pattison-Williams, J.K., W. Yang, Y. Liu, and S. Gabor. 2017. "Riparian wetland conservation: A case study of phosphorous and social return on investment in the Black River watershed." *Ecosystem Services* 26: 400-410.
- Pavlovskii, I., L. Noorduijn, N.J.E. Liggett, J. Klassen, and A. Diaz. 2020. "Quantifying terrain controls on runoff retention and routing in the Northern Prairies." *Hydrological Processes* 34(2): 473-484.
- Pearl, C.A., B. McCreary, J.C. Rowe, and M.J. Adams. 2018. "Late-Season Movement and Habitat Use by Oregon Spotted Frog (*Rana pretiosa*) in Oregon, USA." *Copeia* 106(3): 539-549.
- Peralta Pelaez, L.A., P. Moreno-Casasola, and H. Lopez Rosas. 2014. "Hydrophyte composition of dune lakes and its relationship to land-use and water physicochemistry in Veracruz, Mexico." *Marine and Freshwater Research* 65(4): 312-326.
- Pereyra, A.S., and W.J. Mitsch. 2018. "Methane emissions from freshwater cypress (Taxodium distichum) swamp soils with natural and impacted hydroperiods in Southwest Florida." *Ecological Engineering* 114: 46-56.
- Perez-Valdivia, C., B. Cade-Menun, and D.W. McMartin. 2017. "Hydrological modeling of the pipestone creek watershed using the Soil Water Assessment Tool (SWAT): Assessing impacts of wetland drainage on hydrology." *Journal of Hydrology-Regional Studies* 14: 109-129.
- Peterman, W.E., T.L. Anderson, B.H. Ousterhout, D.L. Drake, J.J. Burkhart, F. Rowland, and R.D. Semlitsch. 2018. "Using spatial demographic network models to optimize habitat management decisions." *Journal of Wildlife Management* 82(3): 649-659.
- Pfeifer-Meister, L., L.G. Gayton, B.A. Roy, B.R. Johnson, and S.D. Bridgham. 2018. "Greenhouse gas emissions limited by low nitrogen and carbon availability in natural, restored, and agricultural Oregon seasonal wetlands." *PeerJ* 6.
- Pierre, S.M., P.F. Quintana-Ascencio, E.H. Boughton, and D.G. Jenkins. 2017. "Dispersal and local environment affect the spread of an invasive apple snail (*Pomacea maculata*) in Florida, USA." *Biological Invasions* 19(9): 2647-2661.
- Pilliod, D.S., R.S. Arkle, J.M. Robertson, M.A. Murphy, and W.C. Funk. 2015. "Effects of changing climate on aquatic habitat and connectivity for remnant populations of a wide-ranging frog species in an arid landscape." *Ecology and Evolution* 5(18): 3979-3994.
- Porcel, S., G. Chaparro, M. Cristina Marinone, J.F. Saad, J. Lancelotti, and I. Izaguirre. 2020. "The role of environmental, geographical, morphometric and spatial variables on plankton communities in lakes of the arid Patagonian plateaus." *Journal of Plankton Research* 42(2): 173-187.

https://doi.org/10.1093/plankt/fbaa004.

- Powell, J.S., and K.J. Babbitt. 2015. "An Experimental Test of Buffer Utility as a Technique for Managing Pool-Breeding Amphibians." *PLoS ONE* 10(7).
- Praetzel, L.S.E., N. Plenter, S. Schilling, M. Schmiedeskamp, G. Broll, and K.H. Knorr. 2020. "Organic matter and sediment properties determine in-lake variability of sediment CO2 and CH4 production and emissions of a small and shallow lake." *Biogeosciences* 17(20): 5057-5078.
- Praetzel, L.S.E., M. Schmiedeskamp, and K.H. Knorr. 2021. "Temperature and sediment properties drive spatiotemporal variability of methane ebullition in a small and shallow temperate lake." *Limnology and Oceanography*.
- Preskienis, V., I. Laurion, F. Bouchard, P.M.J. Douglas, M.F. Billett, D. Fortier, and X. Xu. 2021. "Seasonal patterns in greenhouse gas emissions from lakes and ponds in a High Arctic polygonal landscape." *Limnology and Oceanography* 66.
- Qiu, J., and M.G. Turner. 2015. "Importance of landscape heterogeneity in sustaining hydrologic ecosystem services in an agricultural watershed." *Ecosphere* 6(11).
- Quesnelle, P.E., K.E. Lindsay, and L. Fahrig. 2014. "Low reproductive rate predicts species sensitivity to habitat loss: a meta-analysis of wetland vertebrates." *PLoS ONE* 9(3).
- Quesnelle, P.E., K.E. Lindsay, and L. Fahrigi. 2015. "Relative effects of landscape-scale wetland amount and landscape matrix quality on wetland vertebrates: a 'meta-analysis." *Ecological Applications* 25(3): 812-825.
- Quillet, A., M. Larocque, S. Pellerin, V. Cloutier, M. Ferlatte, C. Paniconi, and M.A. Bourgault. 2017."The role of hydrogeological setting in two Canadian peatlands investigated through 2D steady-state groundwater flow modelling." *Hydrological Sciences Journal* 62(15): 2541-2557.
- Quin, A., F. Jaramillo, and G. Destouni. 2015. "Dissecting the ecosystem service of large-scale pollutant retention: The role of wetlands and other landscape features." *Ambio* 44.
- Rahman, M.M., Z. Lin, X. Jia, D.D. Steele, and T.M. Desutter. 2014. "Impact of subsurface drainage on streamflows in the Red River of the North basin." *Journal of Hydrology* 511: 474-483.
- Rains, M.C., S.G. Leibowitz, M.J. Cohen, I.F. Creed, H.E. Golden, J.W. Jawitz, P. Kalla, C.R. Lane, M.W. Lang, and D.L. McLaughlin. 2016. "Geographically isolated wetlands are part of the hydrological landscape." *Hydrological Processes* 30(1): 153-160.
- Rajib, A., H.E. Golden, C.R. Lane, and Q. Wu. 2020. "Surface Depression and Wetland Water Storage Improves Major River Basin Hydrologic Predictions." *Water Resources Research* 56(7): e2019WR026561. <u>https://www.ncbi.nlm.nih.gov/pubmed/33364639</u>.
- Ramesh, R., C.J. Anderson, and L. Kalin. 2020. "Characterizing nitrogen attenuation by headwater slope wetlands across different land uses." *Ecological Engineering* 149: 105833. https://doi.org/10.1016/j.ecoleng.2020.105833.

- Ray, A.M., W.R. Gould, B.R. Hossack, A.J. Sepulveda, D.P. Thoma, D.A. Patla, R. Daley, and R. Al-Chokhachy. 2016. "Influence of climate drivers on colonization and extinction dynamics of wetlanddependent species." *Ecosphere* 7(7): e01409. <u>https://doi.org/10.1002/ecs2.1409</u>.
- Regensburg, T.H., P.J. Chapman, M.G. Pilkington, D.M. Chandler, M.G. Evans, and J. Holden. 2021."Effects of pipe outlet blocking on hydrological functioning in a degraded blanket peatland." *Hydrological Processes* 35(3).
- Reverey, F., L. Ganzert, G. Lischeid, A. Ulrich, K. Premke, and H.P. Grossart. 2018. "Dry-wet cycles of kettle hole sediments leave a microbial and biogeochemical legacy." *Science of the Total Environment* 627: 985-996.
- Reverey, F., H.P. Grossart, K. Premke, and G. Lischeid. 2016. "Carbon and nutrient cycling in kettle hole sediments depending on hydrological dynamics: a review." *Hydrobiologia* 775(1): 1-20.
- Reynolds, C., and G.S. Cumming. 2016. "Seed dispersal by waterbirds in southern Africa: comparing the roles of ectozoochory and endozoochory." *Freshwater Biology* 61(4): 349-361.
- Reynolds, C., N.A.F. Miranda, and G.S. Cumming. 2015. "The role of waterbirds in the dispersal of aquatic alien and invasive species." *Diversity and Distributions* 21(7): 744-754.
- Richardson, S.J., R. Clayton, B.D. Rance, H. Broadbent, M.S. McGlone, and J.M. Wilmshurst. 2015. "Small wetlands are critical for safeguarding rare and threatened plant species." *Applied Vegetation Science* 18(2): 230-241.
- Rivas, J.A., J. Mohl, R.S. Van Pelt, M.Y. Leung, R.L. Wallace, T.E. Gill, and E.J. Walsh. 2018.
 "Evidence for regional aeolian transport of freshwater micrometazoans in arid regions." *Limnology* and Oceanography: Letters 3(4): 320-330.
- Robertson, H.L., M.T. Guzik, and N.P. Murphy. 2014. "Persistence in the desert: ephemeral waterways and small-scale gene flow in the desert spring amphipod, *Wangiannachiltonia guzikae*." *Freshwater Biology* 59(4): 653-665. <u>https://doi.org/10.1111/fwb.12293</u>.
- Roland, F.A., F. Darchambeau, C. Morana, S. Bouillon, and A.V. Borges. 2017. "Emission and oxidation of methane in a meromictic, eutrophic and temperate lake (Dendre, Belgium)." *Chemosphere* 168: 756-764.
- Roley, S.S., J.L. Tank, J.C. Tyndall, and J.D. Witter. 2016. "How cost-effective are cover crops, wetlands, and two-stage ditches for nitrogen removal in the Mississippi River Basin?" *Water Resources and Economics* 15: 43-56.
- Rosset, E., A. Ruhi, M.T. Bogan, and T. Datry. 2017. "Do lentic and lotic communities respond similarly to drying?" *Ecosphere* 8(7).
- Rouissi, M., D. Boix, S.D. Muller, S. Gascón, A. Ruhí, J. Sala, A. Bouattour, I. Ben Haj Jilani, Z. Ghrabi-Gammar, S. Ben Saad-Limam, and A. Daoud-Bouattour. 2014. "Spatio-temporal variability of faunal and floral assemblages in Mediterranean temporary wetlands." *Comptes Rendus Biologies* 337 (12): 695-708.

- Roy, C. 2015. "Quantifying geographic variation in the climatic drivers of midcontinent wetlands with a spatially varying coefficient model." *PLoS ONE* 10(4).
- Ruiz, L., N. Parikh, L.J. Heintzman, S.D. Collins, S.M. Starr, C.K. Wright, G.M. Henebry, N. van Gestel, and N.E. McIntyre. 2014. "Dynamic connectivity of temporary wetlands in the southern Great Plains." *Landscape Ecology* 29(3): 507-516.
- Ruso, G.E., C.A. Morrissey, N.S. Hogan, C. Sheedy, M.J. Gallant, and T.D. Jardine. 2019. "Detecting Amphibians in Agricultural Landscapes Using Environmental DNA Reveals the Importance of Wetland Condition." *Environmental Toxicology and Chemistry* 38 (12): 2750-2763.
- Ryan, M.A. 2017. "WATER POLICY Science versus political realities." *Nature Geoscience* 10(11): 807-808.
- Ryan, M.E., W.J. Palen, M.J. Adams, and R.M. Rochefort. 2014. "Amphibians in the climate vice: loss and restoration of resilience of montane wetland ecosystems in the western US." *Frontiers in Ecology and the Environment* 12(4): 232-240.
- Sampath, P.V., H. Liao, Z.K. Curtis, P.J. Doran, M.E. Herbert, C.A. May, and S. Li. 2015. "Understanding the Groundwater Hydrology of a Geographically-Isolated Prairie Fen: Implications for Conservation." *PLoS ONE* 10(10).
- Santoro, A., J.M. Chambers, B.J. Robson, and S.J. Beatty. 2020. "Land use surrounding wetlands influences urban populations of a freshwater turtle." *Aquatic Conservation: Marine and Freshwater Ecosystems* 30(5): 1050-1060.
- Saunders, M.I., C.J. Brown, M.M. Foley, C.M. Febria, R. Albright, M.G. Mehling, M.T. Kavanaugh, and D.D. Burfeind. 2016. "Human impacts on connectivity in marine and freshwater ecosystems assessed using graph theory: a review." *Marine and Freshwater Research* 67(3): 277-290.
- Scheliga, B., D. Tetzlaff, G. Nuetzmann, and C. Soulsby. 2017. "Groundwater isoscapes in a montane headwater catchment show dominance of well-mixed storage." *Hydrological Processes* 31(20): 3504-3519.
- Schepker, T.J., T. Lagrange, and E.B. Webb. 2019. "Are Waterfowl Food Resources Limited during Spring Migration? A Bioenergetic Assessment of Playas in Nebraska's Rainwater Basin." Wetlands 39(1): 173-184. <u>https://doi.org/10.1007/s13157-018-1047-0</u>.
- Schmadel, N.M., J.W. Harvey, R.B. Alexander, G.E. Schwarz, R.B. Moore, K. Eng, J.D. Gomez-Velez, E.W. Boyer, and D. Scott. 2018. "Thresholds of lake and reservoir connectivity in river networks control nitrogen removal." *Nature Communications* 9(1).
- Schmadel, N.M., J.W. Harvey, G.E. Schwarz, R.B. Alexander, J.D. Gomez-Velez, D. Scott, and S.W. Ator. 2019. "Small Ponds in Headwater Catchments Are a Dominant Influence on Regional Nutrient and Sediment Budgets." *Geophysical Research Letters* 46 (16): 9669-9677.
- Schmiedeskamp, M., L.S.E. Praetzel, D. Bastviken, and K.H. Knorr. 2021. "Whole-lake methane emissions from two temperate shallow lakes with fluctuating water levels: Relevance of

spatiotemporal patterns." Limnology and Oceanography 66(6): 2455-2469.

- Schofield, K.A., L.C. Alexander, C.E. Ridley, M.K. Vanderhoof, K.M. Fritz, B.C. Autrey, J.E. DeMeester, W.G. Kepner, C.R. Lane, S.G. Leibowitz, and A.I. Pollard. 2018. "Biota Connect Aquatic Habitats throughout Freshwater Ecosystem Mosaics." *Journal of the American Water Resources Association* 54(2): 372-399.
- Schottler, S.P., J. Ulrich, P. Belmont, R. Moore, J.W. Lauer, D.R. Engstrom, and J.E. Almendinger. 2014. "Twentieth century agricultural drainage creates more erosive rivers." *Hydrological Processes* 28(4): 1951-1961.
- Schriever, T.A., and D.A. Lytle. 2016. "Convergent diversity and trait composition in temporary streams and ponds." *Ecosphere* 7(5).
- Semlitsch, R.D., W.E. Peterman, T.L. Anderson, D.L. Drake, and B.H. Ousterhout. 2015. "Intermediate pond sizes contain the highest density, richness, and diversity of pond-breeding amphibians." *PLoS ONE* 10(4).
- Serran, J.N., and I.F. Creed. 2016. "New mapping techniques to estimate the preferential loss of small wetlands on prairie landscapes." *Hydrological Processes* 30(3): 396-409.
- Serrano, F., R. Pita, M. Mota-Ferreira, P. Beja, and P. Segurado. 2020. "Landscape connectivity affects individual survival in unstable patch networks: The case of a freshwater turtle inhabiting temporary ponds." *Freshwater Biology* 65(3): 540-551.
- Shen, W., S. Li, M. Mi, Y. Zhuang, and L. Zhang. 2021. "What makes ditches and ponds more efficient in nitrogen control?" *Agriculture, Ecosystems and Environment* 314.
- Shook, K., S. Papalexiou, and J.W. Pomeroy. 2021. "Quantifying the effects of Prairie depressional storage complexes on drainage basin connectivity." *Journal of Hydrology* 593.
- Shook, K., J. Pomeroy, and G. van Der Kamp. 2015. "The transformation of frequency distributions of winter precipitation to spring streamflow probabilities in cold regions; case studies from the Canadian Prairies." *Journal of Hydrology* 521: 395-409.
- Simonis, J.L., and J.C. Ellis. 2014. "Bathing birds bias beta-diversity: frequent dispersal by gulls homogenizes fauna in a rock-pool metacommunity." *Ecology* 95(6): 1545-1555.
- Simpson, D.S., D.C. Forester, J.W. Snodgrass, and S.A. Smith. 2021. "Relationships among Amphibian Assemblage Structure, Wetland pH, and Forest Cover." *Journal of Wildlife Management* 85(3): 569-581.
- Singh, N.K., J.D. Gourevitch, B.C. Wemple, K.B. Watson, D.M. Rizzo, S. Polasky, and T.H. Ricketts. 2019. "Optimizing wetland restoration to improve water quality at a regional scale." *Environmental Research Letters* 14(6).
- Skagen, S.K., L.E. Burris, and D.A. Granfors. 2016. "Sediment Accumulation in Prairie Wetlands under a Changing Climate: the Relative Roles of Landscape and Precipitation." *Wetlands* 36.

- Smith, L.L., C.N. Jones, and N.G. Nelson. 2019. "Featured Collection Introduction: The Emerging Science of Aquatic System Connectivity II." *Journal of the American Water Resources Association* 0(0): 526-528.
- Smith, L.L., A.L. Subalusky, C.L. Atkinson, J.E. Earl, D.M. Mushet, D.E. Scott, S.L. Lance, and S.A. Johnson. 2019. "Biological Connectivity of Seasonally Ponded Wetlands across Spatial and Temporal Scales." *Journal of the American Water Resources Association* 55(2): 334-353.
- Smith, L.L., A.L. Subalusky, C.L. Atkinson, and L.K. Kirkman. 2017. "Geographically isolated wetlands: embedded habitats." Pages 153-172 in *Ecological Restoration and Management of Longleaf Pine Forests.* L.K. Kirkman, and S.B. Jack, eds. CRC Press, Boca Raton, FL.
- Sofaer, H.R., S.K. Skagen, J.J. Barsugli, B.S. Rashford, G.C. Reese, J.A. Hoeting, A.W. Wood, and B.R. Noon. 2016. "Projected wetland densities under climate change: habitat loss but little geographic shift in conservation strategy." *Ecological Applications* 26 (6): 1677-1692.
- Speldrich, B., P. Gerla, and G.E. Tschann. 2021. "Characterizing Groundwater Interaction with Lakes and Wetlands Using GIS Modeling and Natural Water Quality Measurements." *Water* 13(7).
- Spyra, A., and M. Strzelec. 2014. "Identifying factors linked to the occurrence of alien gastropods in isolated woodland water bodies." *Naturwissenschaften* 101(3): 229-239.
- Steen, V., S.K. Skagen, and B.R. Noon. 2018. "Preparing for an uncertain future: migrating shorebird response to past climatic fluctuations in the Prairie Potholes." *Ecosphere* 9(2).
- Steen, V., H.R. Sofaer, S.K. Skagen, A.J. Ray, and B.R. Noon. 2017. "Projecting species' vulnerability to climate change: Which uncertainty sources matter most and extrapolate best?" *Ecology and Evolution* 7(21): 8841-8851.
- Steen, V.A., S.K. Skagen, and C.P. Melcher. 2016. "Implications of Climate Change for Wetland-Dependent Birds in the Prairie Pothole Region." *Wetlands* 36.
- Stengard, E., A. Rasanen, C.S.S. Ferreira, and Z. Kalantari. 2020. "Inventory and Connectivity Assessment of Wetlands in Northern Landscapes with a Depression-Based DEM Method." *Water* 12(12).
- Streich, S.C., and C.J. Westbrook. 2020. "HYDROLOGICAL FUNCTION OF A MOUNTAIN FEN AT LOW ELEVATION UNDER DRY CONDITIONS." *Hydrological Processes* 34(2): 244-257.
- Sullivan, D.G., J.G. White, and M. Vepraskas. 2019. "Assessing Carolina Bay Wetland Restoration Risks to Downstream Water Quality by Characterizing Land Use and Stream Proximity." *Wetlands* 39(3): 495-506.
- Sullivan, S.M.P., M.C. Rains, and A.D. Rodewald. 2019. "Opinion: The proposed change to the definition of "waters of the United States" flouts sound science." *Proceedings of the National Academy of Sciences* 116(24).
- Swartz, T.M., and J.R. Miller. 2019. "Managing farm ponds as breeding sites for amphibians: key trade-

offs in agricultural function and habitat conservation." Ecological Applications 29(7).

- Swidnicki, K., A.M. Basinska, M. Pronin, and N. Kuczynska-Kippen. 2016. "Meteorite crater ponds as source of high zooplankton biodiversity." *Biologia* 71(12): 1361-1368.
- Tahmasebi Nasab, M., K. Grimm, M.H. Bazrkar, L. Zeng, A. Shabani, X. Zhang, and X. Chu. 2018. "SWAT Modeling of Non-Point Source Pollution in Depression-Dominated Basins under Varying Hydroclimatic Conditions." *International Journal of Environmental Research and Public Health* 15(11).
- Tahmasebi Nasab, M., V. Singh, and X. Chu. 2017. "SWAT Modeling for Depression-Dominated Areas: How Do Depressions Manipulate Hydrologic Modeling?" *Water* 9(1).
- Tang, Y., A.S. Leon, and M.L. Kavvas. 2020. "Impact of Size and Location of Wetlands on Watershed-Scale Flood Control." *Water Resources Management* 34(5): 1693-1707.
- Tang, Z., Y. Gu, W. Jiang, Y. Xue, A. Bishop, T. Lagrange, and E. Nugent. 2016. "Use RUSLE2 model to assess the impact of soil erosion on playa inundation and hydrophyte conditions in the Rainwater Basin, Nebraska." *Environmental Monitoring and Assessment* 188(6).
- Tangen, B.A., and S. Bansal. 2019. "Hydrologic Lag Effects on Wetland Greenhouse Gas Fluxes." *Atmosphere* 10(5).
- Tangen, B.A., and S. Bansal. 2020. "Soil organic carbon stocks and sequestration rates of inland, freshwater wetlands: Sources of variability and uncertainty." *Science of the Total Environment* 749.
- Tangen, B.A., and R.G. Finocchiaro. 2017. "A Case Study Examining the Efficacy of Drainage Setbacks for Limiting Effects to Wetlands in the Prairie Pothole Region, USA." *Journal of Fish and Wildlife Management* 8(2): 513-529.
- Tangen, B.A., and M.T. Wiltermuth. 2018. "Prairie Pothole Region Wetlands and Subsurface Drainage Systems: Key Factors for Determining Drainage Setback Distances." *Journal of Fish and Wildlife Management* 9(1): 274-284.
- Taylor, S., P.J. Gilbert, D.A. Cooke, M.E. Deary, and M.J. Jeffries. 2019. "High carbon burial rates by small ponds in the landscape." *Frontiers in Ecology and the Environment* 17(1): 25-30. <u>https://doi.org/10.1002/fee.1988</u>.
- Terer, T., A.M. Muasya, and L. Triest. 2015. "Strong isolation by distance revealed among Cyperus papyrus populations in the Rift Valley lakes, Lake Victoria, and isolated wetlands of Kenya." *Aquatic Botany* 121: 57-66.
- Thapa, A., L. Bradford, G. Strickert, X. Yu, A. Johnston, and K. Watson-Daniels. 2019. "Garbage in, Garbage Out' Does Not Hold True for Indigenous Community Flood Extent Modeling in the Prairie Pothole Region." Water 11(12).
- Thobaben, E.T., and S.K. Hamilton. 2014. "The Relative Importance of Groundwater and its Ecological Implications in Diverse Glacial Wetlands." *American Midland Naturalist* 172(2): 205-218.

- Thornhill, I., L. Batty, M. Hewitt, N.R. Friberg, and M.E. Ledger. 2018. "The application of graph theory and percolation analysis for assessing change in the spatial configuration of pond networks." *Urban Ecosystems* 21(2): 213-225.
- Thorslund, J., M.J. Cohen, J.W. Jawitz, G. Destouni, I.F. Creed, M.C. Rains, P. Badiou, and J. Jarsjö. 2018. "Solute evidence for hydrological connectivity of geographically isolated wetlands." *Land Degradation & Development* 29(11): 3954-3962.
- Thorslund, J., J. Jarsjo, F. Jaramillo, J.W. Jawitz, S. Manzoni, N.B. Basu, S.R. Chalov, M.J. Cohen, I.F. Creed, R. Goldenberg, A. Hylin, Z. Kalantari, A.D. Koussis, S.W. Lyon, K. Mazi, J. Mard, K. Persson, J. Pietro, C. Prieto, A. Quin, K. Van Meter, and G. Destouni. 2017. "Wetlands as large-scale nature-based solutions: Status and challenges for research, engineering and management." *Ecological Engineering* 108: 489-497.
- Todhunter, P.E. 2018. "A volumetric water budget of Devils Lake (USA): non-stationary precipitationrunoff relationships in an amplifier terminal lake." *Hydrological Sciences Journal* 63(9): 1275-1291.
- Todhunter, P.E. 2021. "Hydrological basis of the Devils Lake, North Dakota (USA), terminal lake flood disaster." *Natural Hazards* 106(3): 2797-2824.
- Tomer, M.D., and J.A. Nelson. 2020. "Measurements of landscape capacity for water detention and wetland restoration practices can inform watershed planning goals and implementation strategies." *Journal of Soil and Water Conservation* 75(4): 434-443.
- Tornwall, B.M., A.L. Pitt, B.L. Brown, J. Hawley-Howard, and R.F. Baldwin. 2020. "Diversity Patterns Associated with Varying Dispersal Capabilities as a Function of Spatial and Local Environmental Variables in Small Wetlands in Forested Ecosystems." *Forests* 11(11).
- Trigg, M.A., P.G. Cook, and P. Brunner. 2014. "Groundwater fluxes in a shallow seasonal wetland pond: The effect of bathymetric uncertainty on predicted water and solute balances." *Journal of Hydrology* 517: 901-912.
- Uden, D.R., C.R. Allen, A.A. Bishop, R. Grosse, C.F. Jorgensen, T.G. Lagrange, R.G. Stutheit, and M.P. Vrtiska. 2015. "Predictions of future ephemeral springtime waterbird stopover habitat availability under global change." *Ecosphere* 6(11).
- Uden, D.R., M.L. Hellman, D.G. Angeler, and C.R. Allen. 2014. "The role of reserves and anthropogenic habitats for functional connectivity and resilience of ephemeral wetlands." *Ecological Applications* 24 (7): 1569-1582.
- Ulrich, U., G. Hörmann, M. Unger, M. Pfannerstill, F. Steinmann, and N. Fohrer. 2018. "Lentic small water bodies: Variability of pesticide transport and transformation patterns." *Science of the Total Environment* 618: 26-38.
- Upadhyay, P., L.O.S. Pruski, A.L. Kaleita, and M.L. Soupir. 2019. "Effects of land management on inundation of prairie pothole wetlands in the Des Moines Lobe using AnnAGNPS." *Agricultural Water Management* 213: 947-956.

- Uuemaa, E., C.C. Palliser, A.O. Hughes, and C.C. Tanner. 2018. "Effectiveness of a Natural Headwater Wetland for Reducing Agricultural Nitrogen Loads." *Water* 10(3).
- Valls, L., A. Castillo-Escriva, L. Barrera, E. Gomez, J. Antonio Gil-Delgado, F. Mesquita-Joanes, and X. Armengol. 2017. "Differential endozoochory of aquatic invertebrates by two duck species in shallow lakes." Acta Oecologica 80: 39-46.
- van Bellen, S., W. Shotyk, G. Magnan, L. Davies, T. Nason, G. Mullan-Boudreau, M. Garneau, T. Noernberg, L. Bragazza, and C. Zaccone. 2020. "Carbon and nitrogen accumulation rates in ombrotrophic peatlands of central and northern Alberta, Canada, during the last millennium." *Biogeochemistry* 151: 251-272. <u>https://doi.org/10.1007/s10533-020-00724-0</u>.
- Van Meter, K.J., and N.B. Basu. 2015. "Signatures of human impact: Size distributions and spatial organization of wetlands in the Prairie Pothole landscape." *Ecological Applications* 25(2): 451-465.
- Vanderhoof, M.K., and L.C. Alexander. 2016. "The Role of Lake Expansion in Altering the Wetland Landscape of the Prairie Pothole Region, United States." *Wetlands* 36(2): 309-321.
- Vanderhoof, M.K., L.C. Alexander, and M.J. Todd. 2016. "Temporal and spatial patterns of wetland extent influence variability of surface water connectivity in the Prairie Pothole Region, United States." *Landscape Ecology* 31 (4): 805-824.
- Vanderhoof, M.K., J.R. Christensen, and L.C. Alexander. 2016. "Patterns and drivers for wetland connections in the Prairie Pothole Region, United States." *Wetlands Ecology and Management* 25(3): 1-23.
- Vanderhoof, M.K., H.E. Distler, M.W. Lang, and L.C. Alexander. 2017. "The influence of data characteristics on detecting wetland/stream surface-water connections in the Delmarva Peninsula, Maryland and Delaware." Wetlands Ecology and Management 26: 63–86.
- Vanderhoof, M.K., and C.R. Lane. 2019. "The potential role of very high-resolution imagery to characterise lake, wetland and stream systems across the Prairie Pothole Region, United States." *International Journal of Remote Sensing* 40(15): 5768-5798.
- Vanderhoof, M.K., C.R. Lane, M.G. McManus, L.C. Alexander, and J.R. Christensen. 2018. "Wetlands inform how climate extremes influence surface water expansion and contraction." *Hydrology and Earth System Sciences* 22(3): 1851-1873.
- Vasic, F., C. Paul, V. Strauss, and K. Helming. 2020. "Ecosystem Services of Kettle Holes in Agricultural Landscapes." Agronomy 10 (9): 1326.
- Verheijen, B.H.F., D.M. Varner, and D.A. Haukos. 2018. "Effects of large-scale wetland loss on network connectivity of the Rainwater Basin, Nebraska." *Landscape Ecology* 33(11): 1939-1951.
- Verheijen, B.H.F., D.M. Varner, and D.A. Haukos. 2020. "Future losses of playa wetlands decrease network structure and connectivity of the Rainwater Basin, Nebraska." *Landscape Ecology* 35(2): 453-467.

- Vickruck, J.L., L.R. Best, M.P. Gavin, J.H. Devries, and P. Galpern. 2019. "Pothole wetlands provide reservoir habitat for native bees in prairie croplands." *Biological Conservation* 232: 43-50.
- Vimercati, G., M. Labadesse, T. Dejean, and J. Secondi. 2020. "Assessing the effect of landscape features on pond colonisation by an elusive amphibian invader using environmental DNA." *Freshwater Biology* 65(3): 502-513.
- von Fumetti, S., and L. Blattner. 2017. "Faunistic assemblages of natural springs in different areas in the Swiss National Park: a small-scale comparison." *Hydrobiologia* 793(1): 175-184.
- Vyse, S.A., M. Taie Semiromi, G. Lischeid, and C. Merz. 2020. "Characterizing hydrological processes within kettle holes using stable water isotopes in the Uckermark of northern Brandenburg, Germany." *Hydrological Processes* 34(8): 1868-1887.
- Walsh, R., and A.S. Ward. 2019. "Redefining Clean Water Regulations Reduces Protections for Wetlands and Jurisdictional Uncertainty." *Frontiers in Water* 1(1).
- Walters, K.M., and M. Babbar-Sebens. 2016. "Using climate change scenarios to evaluate future effectiveness of potential wetlands in mitigating high flows in a Midwestern US watershed." *Ecological Engineering* 89: 80-102.
- Wang, N., X. Chu, and X. Zhang. 2021. "Functionalities of surface depressions in runoff routing and hydrologic connectivity modeling." *Journal of Hydrology* 593.
- Wang, N., X. Zhang, and X. Chu. 2019. "New Model for Simulating Hydrologic Processes under Influence of Surface Depressions." *Journal of Hydrologic Engineering* 24(5).
- Watkins, S.C., D.S. Baldwin, H.P. Waudby, and S.E.M.A. Ning. 2017. "Managing rain-filled wetlands for carbon sequestration: a synthesis." *The Rangeland Journal* 39(2): 145-152.
- Watson, K.B., T. Ricketts, G. Galford, S. Polasky, and J. O'Niel-Dunne. 2016. "Quantifying flood mitigation services: The economic value of Otter Creek wetlands and floodplains to Middlebury, VT." *Ecological Economics* 130: 16-24.
- Waz, A., and I.F. Creed. 2017. "Automated Techniques to Identify Lost and Restorable Wetlands in the Prairie Pothole Region." *Wetlands* 37(6): 1079-1091. <u>https://doi.org/10.1007/s13157-017-0942-0</u>.
- Welch, C., A.A. Smith, and T.A. Stadnyk. 2018. "Linking physiography and evaporation using the isotopic composition of river water in 16 Canadian boreal catchments." *Hydrological Processes* 32(2): 170-184.
- Wells, C., S. Ketcheson, and J. Price. 2017. "Hydrology of a wetland-dominated headwater basin in the Boreal Plain, Alberta, Canada." *Journal of Hydrology* 547: 168-183.
- Wendt, A., C.A. Haas, T. Gorman, and J.H. Roberts. 2021. "Metapopulation genetics of endangered reticulated flatwoods salamanders (*Ambystoma bishopi*) in a dynamic and fragmented landscape." *Conservation Genetics*.

Werner, B., J. Tracy, W.C. Johnson, R.A. Voldseth, G.R. Guntenspergen, and B. Millett. 2016.

"MODELING THE EFFECTS OF TILE DRAIN PLACEMENT ON THE HYDROLOGIC FUNCTION OF FARMED PRAIRIE WETLANDS." *Journal of the American Water Resources Association* 52(6): 1482-1492.

- Widney, S., A. Kanabrocki Klein, J. Ehman, C. Hackney, and C. Craft. 2018. "The value of wetlands for water quality improvement: an example from the St. Johns River watershed, Florida." Wetlands Ecology and Management 26(3): 265-276.
- Wijayarathne, D.B., and E. Gomezdelcampo. 2019. "Using shallow groundwater modeling to frame the restoration of a wet prairie in the Oak Openings Region, Ohio, USA: GSSHA model implementation." *Environmental Earth Sciences* 78(6).
- Williams, M.R., K.W. King, W. Ford, and N.R. Fausey. 2016. "Edge-of-field research to quantify the impacts of agricultural practices on water quality in Ohio." *Journal of Soil and Water Conservation* 71.
- Williams, M.R., S.J. Livingston, G.C. Heathman, and S.J. McAfee. 2019. "Thresholds for run-off generation in a drained closed depression." *Hydrological Processes* 33(18): 2408-2421.
- Williams, M.R., S.J. Livingston, C.J. Penn, D.R. Smith, K.W. King, and C. Huang. 2018. "Controls of event-based nutrient transport within nested headwater agricultural watersheds of the western Lake Erie basin." *Journal of Hydrology* 559: 749-761. <u>https://doi.org/10.1016/j.jhydrol.2018.02.079</u>.
- Williams, T.M., K.W. Krauss, and T. Okruszko. 2016. "Hydrology of Flooded and Wetland Forests." *Forest Hydrology: Processes, Management and Assessment:* 103-123.
- Willson, J.D., and C.T. Winne. 2016. "Evaluating the functional importance of secretive species: A case study of aquatic snake predators in isolated wetlands." *Journal of Zoology* 298(4): 266-273.
- Willson, M.F., and K.C. Halupka. 1995. "Anadromous Fish as Keystone Species in Vertebrate Communities." *Conservation Biology* 9(3):489-497.
- Wilson, H.F., N.J. Casson, A.J. Glenn, P. Badiou, and L. Boychuk. 2019. "Landscape controls on nutrient export during snowmelt and an extreme rainfall runoff event in northern agricultural watersheds." *Journal of Environmental Quality* 48(4): 841-849.
- Wu, L., Y. Chen, G. Zhang, Y.J. Xu, and Z. Tan. 2021. "Integrating the JRC Monthly Water History Dataset and Geostatistical Analysis Approach to Quantify Surface Hydrological Connectivity Dynamics in an Ungauged Multi-Lake System." *Water* 13(4).
- Wu, Q., and C.R. Lane. 2016. "Delineation and Quantification of Wetland Depressions in the Prairie Pothole Region of North Dakota." *Wetlands* 36(2): 215-227.
- Wu, Q., and C.R. Lane. 2017. "Delineating wetland catchments and modeling hydrologic connectivity using lidar data and aerial imagery." *Hydrology and Earth System Sciences* 21(7): 3579-3595.
- Ximenez, S.S., M. Oliveira, M.B. Santos, and A.M. Tozetti. 2014. "The Influence of Habitat Configuration on Anuran Species Composition in Subtemperate Wetlands of Southernmost Brazil."

South American Journal of Herpetology 9(3): 235-243.

- Xue, Z.S., G.L. Hou, Z.S. Zhang, X.G. Lyu, M. Jiang, Y.C. Zou, X.J. Shen, J. Wang, and X.H. Liu. 2019.
 "Quantifying the cooling-effects of urban and peri-urban wetlands using remote sensing data: Case study of cities of Northeast China." *Landscape and Urban Planning* 182: 92-100.
- Yao, Y., L. Wang, X. Lv, H. Yu, and G. Li. 2014. "Changes in stream peak flow and regulation in Naoli River watershed as a result of wetland loss." *The Scientific World Journal* 2014.
- Yeo, I.Y., M.W. Lang, S. Lee, G.W. McCarty, A.M. Sadeghi, O. Yetemen, and C. Huang. 2019.
 "Mapping landscape-level hydrological connectivity of headwater wetlands to downstream waters: A geospatial modeling approach Part 1." *Science of the Total Environment* 653: 1546-1556.
- Yeo, I.Y., S. Lee, M.W. Lang, O. Yetemen, G.W. McCarty, A.M. Sadeghi, and G. Evenson. 2019.
 "Mapping landscape-level hydrological connectivity of headwater wetlands to downstream waters: A catchment modeling approach Part 2." *Science of the Total Environment* 653: 1557-1570.
- Yu, F., and J.M. Harbor. 2019. "The effects of topographic depressions on multiscale overland flow connectivity: A high-resolution spatiotemporal pattern analysis approach based on connectivity statistics." *Hydrological Processes* 33(10): 1403-1419.
- Yu, X., J. Hawley-Howard, A.L. Pitt, J.J. Wang, R.F. Baldwin, and A.T. Chow. 2015. "Water quality of small seasonal wetlands in the Piedmont ecoregion, South Carolina, USA: Effects of land use and hydrological connectivity." *Water Research* 73: 98-108.
- Zaffaroni, M., P. Zamberletti, I.F. Creed, F. Accatino, C. De Michele, and B. Devries. 2019.
 "Safeguarding Wetlands and Their Connections within Wetlandscapes to Improve Conservation Outcomes for Threatened Amphibian Species." *Journal of the American Water Resources Association* 55(3): 641-656.
- Zamberletti, P., M. Zaffaroni, F. Accatino, I.F. Creed, and C. De Michele. 2018. "Connectivity among wetlands matters for vulnerable amphibian populations in wetlandscapes." *Ecological Modelling* 384: 119-127.
- Zammali, M.K., E. Hassanzadeh, E. Shupena-Soulodre, and K.E. Lindenschmidt. 2021. "A generic approach to evaluate costs and effectiveness of agricultural Beneficial Management Practices to improve water quality management." *Journal of Environmental Management* 287.
- Zamora-Marin, J.M., A. Zamora-Lopez, M. Jimenez-Franco, J.F. Calvo, and F.J. Oliva-Paterna. 2021.
 "Small ponds support high terrestrial bird species richness in a Mediterranean semiarid region." *Hydrobiologia* 848(7): 1623-1638.
- Zaragoza, G., J.P. Rose, K. Purcell, and B.D. Todd. 2015. "Terrestrial Habitat use by Western Pond Turtles (*Actinemys marmorata*) in the Sierra Foothills." *Journal of Herpetology* 49(3): 437-441.
- Zeng, L., and X.F. Chu. 2021. "Integrating depression storages and their spatial distribution in watershedscale hydrologic modeling." *Advances in Water Resources* 151.

- Zeng, L., J.L. Shao, and X.F. Chu. 2020. "Improved hydrologic modeling for depression-dominated areas." *Journal of Hydrology* 590.
- Zhang, W., H. Li, A.D. Kendall, D.W. Hyndman, Y. Diao, J. Geng, and J. Pang. 2019. "Nitrogen transport and retention in a headwater catchment with dense distributions of lowland ponds." *Science of the Total Environment* 683: 37-48.

Zhang, Z., X. Chen, Q. Cheng, and C. Soulsby. 2019. "Storage dynamics, hydrological connectivity and flux ages in a karst catchment: conceptual modelling using stable isotopes." *Hydrology and Earth System Sciences* 23(1): 51-71.

- Zhang, Z., Y.P. Li, M. Barlage, F. Chen, G. Miguez-Macho, A. Ireson, and Z.H. Li. 2020. "Modeling groundwater responses to climate change in the Prairie Pothole Region." *Hydrology and Earth System Sciences* 24(2): 655-672.
- Zhao, Q., T.W. Arnold, J.H. Devries, D.W. Howerter, R.G. Clark, and M.D. Weegman. 2019. "Land-use change increases climatic vulnerability of migratory birds: Insights from integrated population modelling." *Journal of Animal Ecology* 88(10): 1625-1637.
- Zhao, Q., T.W. Arnold, J.H. Devries, D.W. Howerter, R.G. Clark, and M.D. Weegman. 2020. "Using integrated population models to prioritize region-specific conservation strategies under global change." *Biological Conservation* 252.
- Zhu, J., G. Sun, W.H. Li, Y. Zhang, G.F. Miao, A. Noormets, S.G. McNulty, J.S. King, M. Kumar, and X. Wang. 2017. "Modeling the potential impacts of climate change on the water table level of selected forested wetlands in the southeastern United States." *Hydrology and Earth System Sciences* 21(12): 6289-6305.
- Zhu, X., Y. Yuan, D.M. Mushet, and M.L. Otte. 2019. "Can Multi-Element Fingerprinting of Soils Inform Assessments of Chemical Connectivity Between Depressional Wetlands?" Wetlands 39(5): 1015-1027.

Appendix C2: Plain-Text Language from the Abstracts of Illustrative Scientific Papers

The screening process identified illustrative peer-reviewed papers for each of the three aquatic systems (as discussed in section I.C of the Technical Support Document). Screeners further provided a "Plain Text" summary of the content based on the abstract. The papers are not ordered or prioritized and represent a sample of the references screened. This Appendix has been renumbered and was Appendix C3 in the Technical Support Document for the Proposed Rule. No other changes have been made to this Appendix.

Ephemeral, Intermittent, and Perennial Streams

Reference	Plain Text Summary
Ebersole, J. L., P. J. Wigington, S. G. Leibowitz, R. L. Comeleo and J. Van Sickle (2015). "Predicting the occurrence of cold- water patches at intermittent and ephemeral tributary confluences with warm rivers." <i>Freshwater Science</i> 34(1): 111-124	Small tributary streams, including ephemeral channels when they do not contain surface flow, serve as sources of cold water to warmer downstream waters. Tributary basins with higher water surpluses at the end of the preceding wet season were more likely to serve as summer cold water sources; basin area and the presence of surface flow at the time of sampling were not strong predictors. Continued release of groundwater from tributary basins creates refuges for cold- water taxa.
Chara-Serna, A. N. and J. S. Richardson (2021). "Multiple-Stressor Interactions in Tributaries Alter Downstream Ecosystems in Stream Mesocosm Networks." <i>Water</i> 13(9)	Using a mesocosm study, researchers examined how stressor interactions in tributaries affected downstream second-order channels. Results showed that (1) Ephemeroptera, Plecoptera, and Trichoptera (EPT) density and richness were higher in downstream channels when stressors were applied separately in tributaries, rather than in combination, and (2) combined stressors within a tributary reduced macroinvertebrate drift into downstream channel. These results support the hypothesis that cumulative upstream disturbance can influence downstream systems.
 Shogren, A. J., J. P. Zarnetske, B. W. Abbott, F. Iannucci, R. J. Frei, N. A. Griffin and W. B. Bowden (2019). "Revealing biogeochemical signatures of Arctic landscapes with river chemistry." <i>Scientific</i> <i>Reports</i> 9(1): 12894. 	The dominant spatial scale controlling organic carbon and inorganic nutrient concentrations within three Alaska watersheds was 3-30 km2, indicating that fine scale landscape patches and a continuum of diffuse and discrete sourcing and processing dynamics are driving solute generation and transport.
Chiu, M. C., B. i. Li, K. e. Nukazawa, V. H. Resh, T. Carvajal and K. Watanabe (2020). "Branching networks can have opposing influences on genetic variation in riverine metapopulations." <i>Diversity and</i> <i>Distributions</i> 26(12): 1813-1824	This study was designed to examine how branching complexity within stream networks can simultaneously increase and decrease genetic divergence of macroinvertebrate metapopulations. Simulation experiments showed that more branched stream networks had both greater landscape connectivity (resulting from shorter watercourse distance) and greater isolation of headwater streams. These two spatial features have negative and positive influences on genetic divergence, with their relative importance varying among species and dispersal characteristics.

Reference	Plain Text Summary
Seymour, M., E. A. Fronhofer and F. Altermatt (2015). "Dendritic network structure and dispersal affect temporal dynamics of diversity and species persistence." <i>Oikos</i> 124(7): 908-916	This study examined the effect of dendritic versus linear network structures on local (alpha), regional (beta) and total (gamma) diversity, using protist and rotifer assemblages as a test community. Local diversity remained higher in dendritic networks over time, especially at highly connected sites. Regional diversity was initially greater in linear networks due to dispersal limitation, but over time became more similar to regional diversity in dendritic networks. Results indicate that dispersal and network connectivity alone may, to a large extent, explain diversity dynamics.
Brennan, S. R., D. E. Schindler, T. J. Cline, T. E. Walsworth, G. Buck and D. P. Fernandez (2019). "Shifting habitat mosaics and fish production across river basins." <i>Science</i> 364(6442): 783-786	Researchers quantified how habitat mosaics (including headwaters) are expressed across a range of spatial scales within a large, free- flowing river in Alaska. The relative productivity of locations across the river network varies widely among years and across a broad range of spatial scales, and these shifts in natal and juvenile rearing habitat help stabilize interannual Pacific salmon production at the scale of the entire basin.
Sarker, S., A. Veremyev, V. Boginski and A. Singh (2019). "Critical Nodes in River Networks." <i>Scientific Reports</i> 9(1): 11178	In this study, researchers used an algorithm to determine the set of critical nodes (channel intersections) along river networks whose removal results in maximum network fragmentation. Results based on both simulated and natural basins in the US indicated a power-law relationship between the number of connected node pairs in the remaining river network and the number of removed critical nodes (i.e., one varies as a power of the other).
Teachey, M. E., J. M. McDonald and E. A. Ottesen (2019). "Rapid and Stable Microbial Community Assembly in the Headwaters of a Third-Order Stream." <i>Applied and</i> <i>Environmental Microbiology</i> 85(11)	This study examined the development and stability of microbial communities along a first- through third order stream in Georgia. Results show that the bacterioplankton community develops rapidly and predictably from the headwater population with increasing total stream length. Along the length of the stream, the microbial community exhibits substantial diversity loss and enriches repeatedly for select taxa across days and years, although the relative abundances of individual taxa vary over time and space. This repeated enrichment of a stable stream community likely contributes to the stability and flexibility of downstream communities.
Samia, Y. and F. Lutscher (2017). "Downstream flow and upstream movement determine the value of a stream reach for potadromous fish populations." <i>Theoretical</i> <i>Ecology</i> 10(1): 21-34	Because water flow transports certain local conditions downstream and individuals move upstream and downstream through river networks, the overall effects of disturbances should be examined at the scale of the entire network. Results from a fish population model show that upper stream reaches can be highly significant for population persistence if downstream transport of abiotic conditions or upstream movement of individuals is strong.

Reference	Plain Text Summary
Chezik, K. A., S. C. Anderson and J. W. Moore (2017). "River networks dampen long-term hydrological signals of climate change." <i>Geophysical Research Letters</i> 44(14): 7256-7264	Trends over 37 years between climate and daily flow data from 55 river gauging stations within the Fraser River network in British Columbia, Canada were examined to see if flow trends diminish with increasing river size or aggregation of tributary contributions. Long- term changes in discharge variability was dampened by >91% in larger rivers than in smaller tributaries and was >3.1 times the dampening when accounting for differences in sample size (more small tributaries than large rivers in a river network). The authors suggest their findings show that integration of the contributions in a river network (i.e., river network portfolio) has a stabilizing influence on long-term hydrologic trends of downstream rivers.
Rupp, D. E., O. S. Chegwidden, B. Nijssen and M. P. Clark (2021). "Changing River Network Synchrony Modulates Projected Increases in High Flows." <i>Water Resources</i> <i>Research</i> 57(4): e2020WR028713	Daily streamflow along the Columbia River and its tributaries were simulated (without dams and irrigation) to understand how climate change scenarios could influence downstream flood magnitude. One mechanism that affects flood magnitude is timing or synchrony of flooding between on a river and its branches or tributaries. Under moderate warming scenarios, synchrony and flooding was predicted to be lower for coldwater tributaries. However, under sufficient warming the main flow source is expected to transition from mixture of snowmelt and rain to rain-dominated which leads to higher synchrony and downstream flood magnitudes.
Jaeger, K. L., J. D. Olden and N. A. Pelland (2014). "Climate change poised to threaten hydrologic connectivity and endemic fishes in dryland streams." <i>Proceedings of the</i> <i>National Academy of Sciences</i> 111(38): 13894	The authors used a surface water model to forecast future streamflows within Verde River Basin and characterized the change in temporal and spatial dimensions of streamflow or hydrologic connectivity or fragmentation throughout the river network. The model predicted that the number of days which the river stops flowing to increase by 27% in 2050 and the frequency of river drying events to increase by 17%. The overall length of flowing stretches within the Verde River network was predicted to drop between 8% and 20% in spring and early summer with greater declines during the drier portions of the year. This will result in less spawning habitat and refuge from seasonal drying. Using dispersal models to contextualize the impact from climate change projections, the authors estimated the Verde River network will have 6-9% and 12-18% lower hydrologic connectivity during the year and spring spawning months, respectively. This finding has strong implications on the persistence of endemic fish fauna under climate change.

Reference	Plain Text Summary
Marcarelli, A. M., A. A. Coble, K. M. Meingast, E. S. Kane, C. N. Brooks, I. Buffam, S. A. Green, C. J. Huckins, D. Toczydlowski and R. Stottlemyer (2019). "Of Small Streams and Great Lakes: Integrating Tributaries to Understand the Ecology and Biogeochemistry of Lake Superior." <i>Journal of the American Water</i> <i>Resources Association</i> 55(2): 442-458	Approximately 2,800 tributaries flow into and contribute nutrients and dissolved organic matter to the nearshore areas of Lake Superior. Tributaries contribute bulk of these materials to Lake Superior during snowmelt-driven flows in the spring and rain-driven flows following rain during other times of the year. Temporary storage and transformations of these material occur during transport in the tributaries prior to entering Lake Superior. Despite being such a large water body, distinct physical and chemistry signals are detected where tributaries enter Lake Superior during periods of high runoff but are quickly transported and mixed with the bulk of the lake volume. The use of different technologies (e.g., automated sampling, remote imagery, drones) will enhance the monitoring and understanding of tributary-lake connections.
Kelson, S. J. and S. M. Carlson (2019). "Do precipitation extremes drive growth and migration timing of a Pacific salmonid fish in Mediterranean-climate streams?" <i>Ecosphere</i> 10(3)	Climate change is expected to cause more frequent weather extremes leading to more severe droughts and floods. Steelhead are migratory trout that live in the South Fork Eel River and its tributaries in California. This study examined extremely wet and dry years over the period of 2015-2018 to see how stream flow affected the steelhead growth, health, and migration timing. Despite strong differences in the timing and magnitude of winter-spring floods and summer low- flows between years, the growth, health, migration timing was not affected. The authors attributed the lack of impact on steelhead detected between extremes was due to the high quality of habitat provided by groundwater-fed tributaries that provided cool and stable base flows even in the driest years.
Marteau, B., R. J. Batalla, D. Vericat and C. Gibbins (2017). "The importance of a small ephemeral tributary for fine sediment dynamics in a main-stem river." <i>River</i> <i>Research and Applications</i> 33(10): 1564- 1574	An ephemeral tributary in the United Kingdom had key moments of influence on a downstream river (River Ehen) through the temporal mismatch between sediment transport from an ephemeral tributary and flooding in a mainstem river. Despite draining only 1.2% of the river catchment and flowing only ephemerally, the recently reconnected ephemeral tributary increased annual sediment yield in the downstream river by 65%.
Xu, J. (2016). "Sediment jamming of a trunk stream by hyperconcentrated floods from small tributaries: case of the Upper Yellow River, China." <i>Hydrological Sciences</i> <i>Journal</i> 61(10): 1926-1940	The study describes floods in ten small desert tributaries that transport large amounts of sediment which exacerbates downstream flooding in the Yellow River in China. The authors investigated in detail one such flooding event and developed a tool to understand the downstream influence of flood-driven sediment from tributaries over immediate (days to weeks) and longer time periods (decades).

Reference	Plain Text Summary
Swanson, B. J. and G. Meyer (2014). "Tributary confluences and discontinuities in channel form and sediment texture: Rio Chama, NM." <i>Earth Surface Processes and</i> <i>Landforms</i> 39(14): 1927-1943	Tributaries (arroyos) periodically deliver sediment downstream to the Rio Chama in northern New Mexico. The sediment is delivered by floods induced by summer thunderstorms. Channel measurements were collected from 203 cross-sections located upstream and downstream from 26 tributary confluences over a 17-km reach of the Rio Chama. The slope, bed sediment size, and cross-sectional area of the river channel was affected by tributaries and influence the transport and storage of sediment along the river. On a larger scale, tributaries have a stronger influence on the Rio Charma in the upper two-thirds of the 17-km reach than the lower third which had fewer tributaries and was dominated by canyon narrows. Tributaries and their associated watershed characteristics (e.g., geology) contribute the morphology of downstream rivers.
Marteau, B., C. Gibbins, D. Vericat and R. J. Batalla (2020). "Geomorphological response to system-scale river rehabilitation I: Sediment supply from a reconnected tributary." <i>River Research and Applications</i> 36(8): 1488-1503	An ephemeral tributary (Ben Gill) in the United Kingdom was reconnected to its sediment-limited downstream river (River Ehen) and the subsequent 2 years sediment transport from the tributary to the river was measured. Sufficient coarse sediment is critical to maintaining economically and culturally important salmonid habitat. An estimated minimum of 384 m3 of coarse sediment was exported to the downstream river and contributed to the habitat formation. The small, ephemeral stream (0.55 km2 or 1.2% of the river's drainage area) which flows only ~20% of the year approximately doubled the volume of coarse sediment estimated to occur in the confluence area prior to reconnection and so is providing needed critical material for salmonid habitat in the downstream river.
French, D. W., D. E. Schindler, S. R. Brennan and D. Whited (2020). "Headwater Catchments Govern Biogeochemistry in America's Largest Free-Flowing River Network." <i>Journal of Geophysical Research:</i> <i>Biogeosciences</i> 125(12)	Water chemistry samples collected from the Kuskokwim River (largest U.S. river without dams), Alaska was studied to understand the influence of the surrounding watershed and instream conditions from different parts of the river network. The conditions in small, headwater streams play a disproportionately important role in predicting the streamwater chemistry throughout the river network. Nutrients that are rapidly used by algae and microbes are spatially more variable in the river network when compared to chemicals that have lower biological demand.
Koizumi, I., Y. Tanaka and Y. Kanazawa (2017). "Mass immigration of juvenile fishes into a small, once-dried tributary demonstrates the importance of remnant tributaries as wintering habitats." <i>Ichthyological Research</i> 64(3): 353-356	A small tributary of the Otofuke River in northern Japan went dry during the summer. Four months after resuming flow more than 10,000 immature fish of three species, including rainbow trout used the tributary for wintering habitat.

Reference	Plain Text Summary
Fovet, O., D. M. Cooper, D. L. Jones, T. G. Jones and C. D. Evans (2020). "Dynamics of dissolved organic matter in headwaters: comparison of headwater streams with contrasting DOM and nutrient composition." <i>Aquatic Sciences</i> 82(2)	The processes by which headwater streams functionally alter terrestrial dissolved organic matter (carbon and nutrients) are influenced by local factors, including soils, land-use, and human pressures. This study compared the effects of sunlight, presence/absence of aquatic biota, and nutrient supplementation on DOM processing in two contrasting stream types – one, a headwater with low inorganic nutrient loadings (peatland stream) and the other a headwater with high nutrient loadings (an agricultural grassland stream). Exposure to sunlight resulted in net abiotic organic matter loss (removal) in the peatland stream but net biological production (increase) of organic matter in the agricultural streams. These results show that the quantity and quality of net DOM exported from headwaters are influenced by the composition of terrestrial DOM inputs, landscape setting, and exposure to sunlight. The author suggest that these results indicate that headwaters may be more active processors of carbon and nutrients than previously thought
Gallo, E. L., T. Meixner, K. A. Lohse and H. Nicholas (2020). "Estimating Surface Water Presence and Infiltration in Ephemeral to Intermittent Streams in the Southwestern US." <i>Frontiers in Water</i> 2(47)	Streamflow in arid and semi-arid regions is predominantly temporary, and of significant importance for groundwater recharge and biogeochemical processes. However, temporary streamflow, especially ephemeral flows, remain poorly quantified. The authors used in-stream streamflow data loggers and USGS stream gauge data in 15 southern Arizona streams spanning a climate gradient (mean annual precipitation from 160 to 750 mm) to quantify temporary streamflow as (a) streamflow presence and (b) water presence, which included streamflow, ponding and soil moisture. In addition, stream channel sediment data were used to estimate saturated hydraulic conductivity and potential annual infiltration. Annual streamflow ranged 0.6–82.4% or 2–301 days; while water presence ranged from 2.6 to 82.4% or 10 to over 301 days, or 4–33 times longer than streamflow. These data were used to develop 5 statistically distinct flow regimes based on the annual percent streamflow and water presence: (1) dry-ephemeral, (2) wet-ephemeral, (3) dry-intermittent, (4) wet-intermittent, and (5) seasonally-intermittent. Stream channel density was a better predictor of annual streamflow and water presence than annual rainfall alone. The dry-ephemeral and wet- ephemeral flow regimes varied with seasonal precipitation, while the dry-intermittent, wet intermittent and seasonally-intermittent flow regimes did not. These results coupled with the potential infiltration estimates indicate that streamflow at the driest sites occurs in response to rainfall and overland flow while groundwater discharge and vadose zone contributions enhance streamflow at the wetter sites. Flow regime classifications that include both stream flow and water presence, rather than on stream flow alone, may be important for predicting thresholds in ecological functions and refugia in these dryland systems.

Reference	Plain Text Summary
Hill, B. H., R. K. Kolka, F. H. McCormick and M. A. Starry (2014). "A synoptic survey of ecosystem services from headwater catchments in the United States." <i>Ecosystem</i> <i>Services</i> 7: 106-115	"Ecosystem production functions for water supply, climate regulation, and water purification were estimated for 568 headwater streams and their catchments. Results are reported for nine USA ecoregions. Headwater streams represented 74-80% of total catchment stream length. Water supply per unit catchment area was highest in the Northern Appalachian Mountains ecoregion and lowest in the Northern Plains. C, N, and P sequestered in trees were highest in Northern and Southern Appalachian and Western Mountain catchments, but C, N, and P sequestered in soils were highest in the Upper Midwest ecoregion. Catchment denitrification was highest in the Upper Midwest ecoregion. Catchment denitrification was highest in the Temperate Plains. Ecological production functions paired with published economic values for these services revealed the importance of mountain catchments for water supply, climate regulation, and water purification per unit catchment area. The larger catchment sizes of the plains ecoregions resulted in their higher economic value compared to the other ecoregions. The combined potential economic value across headwater catchments was INT \$14,000 ha(-1) yr(-1), or INT \$30 million yr(-1) per catchment. The economic importance of headwater catchments is even greater considering that our study catchments statistically represent more than 2 million headwater eatchments in the continented United States "
Jones, C. S., J. K. Nielsen, K. E. Schilling and L. J. Weber (2018). "Iowa stream nitrate and the Gulf of Mexico." <i>PLoS ONE</i> 13(4): e0195930	The objective of this study was to quantify and update Iowa's contribution of nitrate-nitrogen to the Mississippi River stream network against the backdrop of Gulf of Mexico hypoxia. Stream nitrate and discharge data collected from 1999 until 2016 at 23 Iowa stream sites near watershed outlets, along with publicly available data for sites downstream of Iowa on the Missouri and Mississippi Rivers shows that Iowa contributes between 11 and 52% of the long-term nitrate load to the Mississippi-Atchafalaya Basin, 20 to 63% to the Upper Mississippi River Basin, and 20 to 89% to the Missouri River Basin, with averages of 29, 45 and 55% respectively. Since 1999, nitrate loads in the Iowa inclusive basins have increased and these increases do not appear to be driven by changes in discharge and cropping intensity unique to Iowa. The 5-year running annual average of Iowa nitrate loading has been above the 2003 level for ten consecutive years.

Reference	Plain Text Summary
Jones, E. F., N. Griffin, J. E. Kelso, G. T. Carling, M. A. Baker and Z. T. Aanderud (2020). "Stream Microbial Community Structured by Trace Elements, Headwater Dispersal, and Large Reservoirs in Sub- Alpine and Urban Ecosystems." <i>Frontiers in</i> <i>Microbiology</i> 11: 491425	Functional stream bacterioplankton communities are needed to maintain surface water quality and other aquatic ecosystem services. The diversity and composition of stream bacterioplankton communities influence their function. This study quantified the role of environmental conditions, bacterioplankton dispersal, and human infrastructure (dams) on community composition in rivers from sub- alpine to urban environments in three watersheds (Utah, United States) across three seasons. Bacterioplankton community diversity decreased downstream along parts of the stream continuum but was disrupted where large reservoirs increased water residence time by orders of magnitude, potentially indicating a shift in the relative importance of environmental selection and dispersal at these sites. Reservoirs also had substantial effects on community composition, similarity, and species interactions. Communities downstream of reservoirs were enriched with anaerobic Sporichthyaceae, methanotrophic Methylococcaceae, and iron-transforming Acidimicrobiales, suggesting alternative metabolic pathways became active in the hypolimnion of large reservoirs. The results identify that human activity affects river microbial communities, with potential impacts on water quality through modified biogeochemical cycling
Larson, J. H., J. M. Vallazza and B. C. Knights (2019). "Estimating the degree to which distance and temperature differences drive changes in fish community composition over time in the upper Mississippi River." <i>PLoS ONE</i> 14(12): e0225630	Similarity in aquatic communities often declines with increasing distance between habitat locations. In addition to spatial separation, distance-dissimilarity relationships are driven by the presence of environmental gradients that alter habitat suitability for particular species. The Mississippi River is aligned mostly north-to-south so greater distances along the river roughly correspond to differences in latitude, which in turn correspond to different thermal regimes, which are important determinants of fish community structure. The authors of this study used a 21-year dataset of fish communities in the upper Mississippi River to examine the effect of distance on variation in community composition and to assess whether the effect of distance is primarily due to its effect on thermal regime. The results showed a moderate distance-similarity relationship, suggesting greater distance leads to less similarity, which appeared to increase slightly over time. Using a subset of data for which air temperature was available, models that incorporated both difference among sites in degree days (a surrogate for thermal regime) and physical distance (river km) found that temperature alone appears to be more strongly associated with differences in the Mississippi River fish community than spatial distance alone.

Reference	Plain Text Summary
Mooney, R. J., E. H. Stanley, W. C. Rosenthal, P. C. Esselman, A. D. Kendall and P. B. McIntyre (2020). "Outsized nutrient contributions from small tributaries to a Great Lake." <i>Proceedings of the</i> <i>National Academy of Sciences of the United</i> <i>States of America</i> 117(45): 28175-28182	For lakes across the United States, eutrophication is driven largely by nonpoint nutrient sources from tributaries that drain surrounding watersheds, which are relatively understudied in lake systems despite their ubiquity and potential importance to lake water quality. The authors of this study quantified a 'snapshot' of nutrient inputs from nearly all tributaries of Lake Michigan – the world's fifth largest freshwater lake by volume – to determine how land cover and dams alter nutrient inputs across different watershed sizes. Loads, concentrations, stoichiometry, and bioavailability (percentage dissolved inorganic nutrients) varied by orders of magnitude among tributaries, creating a mosaic of coastal nutrient inputs. The six largest of 235 tributaries accounted for approximately 70% of the daily nitrogen and phosphorus delivered to Lake Michigan. However, small tributaries exhibited nutrient loads that were high for their size and biased toward dissolved inorganic forms. Higher bioavailability of nutrients from small watersheds suggests greater potential to fuel algal blooms in coastal areas, especially given the likelihood that their plumes become trapped and then overlap in the nearshore zone. The findings reveal an underappreciated role that small streams may play in driving coastal eutrophication in large water bodies.
Schilling, K. E. and C. S. Jones (2019). "Hydrograph separation of subsurface tile discharge." <i>Environmental Monitoring and</i> <i>Assessment</i> 191(4): 231	Baseflow is an important component of streamflow and watershed hydrologic budgets. The fraction of baseflow contributed by tile drainage has rarely been reported. The authors of this study quantified baseflow discharge from three central Iowa drainage district tile mains using two different hydrograph separation methods and found that baseflow comprised approximately 60% of the annual flow for a 5-year period (2009–2013). The results of this study provide methods to better quantify hydrologic pathways throughout tiled landscapes.

Reference	Plain Text Summary
van Meerveld, H. J. I., J. W. Kirchner, M. J. P. Vis, R. S. Assendelft and J. Seibert (2019). "Expansion and contraction of the flowing stream network alter hillslope flowpath lengths and the shape of the travel time distribution." <i>Hydrology and Earth</i> <i>System Sciences</i> 23(11): 4825-4834	"Flowing stream networks dynamically extend and retract, both seasonally and in response to precipitation events. These network dynamics can dramatically alter the drainage density and thus the length of subsurface flow pathways to flowing streams. We mapped flowing stream networks in a small Swiss headwater catchment during different wetness conditions and estimated their effects on the distribution of travel times to the catchment outlet. For each point in the catchment, we determined the subsurface transport distance to the flowing stream based on the surface topography and determined the surface transport distance along the flowing stream to the outlet. We combined the distributions of these travel distances with assumed surface and subsurface flow velocities to estimate the distribution of travel times to the outlet. These calculations show that the extension and retraction of the stream network can substantially change the mean travel time and the shape of the travel time distribution. During wet conditions with a fully extended flowing stream network, the travel time distribution was strongly skewed to short travel times, but as the network retracted during dry conditions, the distribution of the travel times became more uniform. Stream network dynamics are widely ignored in catchment models, but our results show that they need to be taken into account when modeling solute transport and interpreting travel time distributions."
Wilkinson, M. E. and J. C. Bathurst (2018). "A multi-scale nested experiment for understanding flood wave generation across four orders of magnitude of catchment area." <i>Nordic Hydrology</i> 49(3): 597-615	Current understanding of flood response is deficient concerning the variation of flood generation as a function at different spatial scales as a result of spatial and temporal variations in storm rainfall. This study investigates flood response to spatially variable rainfall through a multi-scale nested experiment. Hydrological data from an extensive network in the Eden catchment, UK, were collected for a range of flood events over varying scales from 1.1 km ² to 2,286 km ² . Peak specific discharge for winter events appears to remain constant for areas up to 20-30 km ² , corresponding to upland headwater catchments. The flood response to the convective storms depends on the location of the rainfall, and the downstream rates of change of runoff and peak discharge can vary significantly from the winter storm relationships. Particularly for large synoptic storms, average scaling laws for peak discharge have been quantified (exponents ranging between 0.75 and 0.86), illustrating the non-linear nature of the cross-scale variations.
Floodplain Wetlands and Open Waters

Reference	Plain Text Summary
Thom, R. M., S. A. Breithaupt, H. L. Diefenderfer, A. B. Borde, G. C. Roegner, G. E. Johnson and D. L. Woodruff (2018). "Storm-driven particulate organic matter flux connects a tidal tributary floodplain wetland, mainstem river, and estuary." <i>Ecological</i> <i>Applications</i> 28(6): 1420-1434	The authors used a multi-model approach to simulate organic matter transport from a recently connected and restored tidal emergent marsh in the Grays River tributary to the Columbia River estuary. They found that "restored floodplain wetlands can contribute significant amounts of organic matter to the estuarine ecosystem and thereby contribute to the restoration of historical trophic structure."
Yang, W., Y. Liu, C. Ou and S. Gabor (2016). "Examining water quality effects of riparian wetland loss and restoration scenarios in a southern ontario watershed." <i>Journal of Environmental Management</i> 174: 26	"The purpose of the study [was] to develop [watershed-scale] wetland modelling to examine water quality effects of riparian wetland loss and restoration scenarios in the 323-km Black River watershed in southern Ontario, Canada." The model was applied to examine various riparian wetland loss scenarios on sediment and nutrient loads to the river network. The model outputs suggest that as riparian wetland loss increases, environmental functional losses increase at an accelerated rate. For example, sediment, total nitrogen, and total phosphorous loads to the river increased between by 2-, 3-, and 9- fold, respectively, with 100% riparian wetland loss, compared to current conditions. "The results further demonstrate the importance of targeting priority areas for stopping riparian wetland loss and initiating riparian wetland restoration based on scientific understanding of watershed wetland effects."
Pyron, M., L. Etchison and J. Backus (2014). "Fish Assemblages of Floodplain Lakes in the Ohio River Basin." <i>Northeastern</i> <i>Naturalist</i> 21(3): 419-430	The authors sampled and examined fish assemblages in 41 floodplain lakes [wetlands] in the Ohio River Basin (summer 2012). Their results demonstrated "that floodplain lakes in the Ohio River basin contain high species richness and are important habitats to conserve because they have the potential to act as source pools for river fish populations."
Mengistu, S. G., H. E. Golden, C. R. Lane, J. R. Christensen, M. L. Wine, E. D'Amico, A. Prues, S. G. Leibowitz, J. E. Compton, M. H. Weber and R. A. Hill (2020). "Wetland Flowpaths Mediate Nitrogen and Phosphorus Concentrations across the Upper Mississippi River Basin." <i>Journal of the American Water</i> <i>Resources Association:</i> 1-18	The authors developed a large, novel set of spatial variables characterizing hydrological connectivity from wetlands (floodplain and non-floodplain) to streams across the ~0.5 million km ² Upper Mississippi River Basin. They found that wetland connectivity variables provided insights into "processes governing how wetlands influence watershed-scale TN and TP concentrations". For example, they demonstrated that wetland connectivity variables describing how water transport slows along the flowpath from the wetland to the stream (e.g., in flowpaths with high soil porosity, which slows water via infiltration into the soils) were statistically related to lower total nitrogen and total phosphorus concentrations. This means that it is not just the wetlands, but the flowpaths/connectivity between wetlands and streams, that control their water quality effects on downstream surface waters.

Reference	Plain Text Summary
Jensen, A. K. and W. I. Ford (2019). "Quantifying nitrate dynamics of a confluence floodplain wetland in a disturbed Appalachia watershed: High-resolution sensing and modeling." <i>Transactions of the</i> <i>ASABE</i> 62(6): 1545-1565	The authors coupled high-resolution water quality data and simulation modeling to assess what physical (hydrologic, hydraulic) or biogeochemical processes affect nitrate cycling in a confluence floodplain wetland along the Ohio River (June 2017-June 2018). Despite the wetland comprising only 0.42% of the overall watershed drainage area, 2.6% to 58.5% of the annual nitrate loads entering the wetland were removed by it. Longer water storage times in the wetlands and less frequent connectivity with the river allowed nitrate removal to occur at higher rates. The findings therefore "demonstrate the significance of [wetland] connectivity [and disconnectivity] on watershed nitrate loadings to floodplain wetland soils".
Hansen, A., C. L. Dolph, E. f. Foufoula- Georgiou and J. C. Finlay (2018). "Contribution of wetlands to nitrate removal at the watershed scale." <i>Nature Geoscience</i> 11(2): 127	The authors evaluate how existing wetlands (floodplain and non- floodplain) across the landscape of the Minnesota River Basin affect in-stream nitrate concentrations. They found that "under moderate- high streamflow, wetlands are five times more efficient per unit area at reducing riverine nitrate concentration than the most effective land- based nitrogen mitigation strategies, which include cover crops and land retirement". Their results suggest that "wetland restorations that account for the effects of spatial position in stream networks could provide a much greater benefit to water quality then previously assumed."
Fossey, M., A. N. Rousseau and S. Savary (2016). "Assessment of the impact of spatio- temporal attributes of wetlands on stream flows using a hydrological modelling framework: a theoretical case study of a watershed under temperate climatic conditions." <i>Hydrological Processes</i> 30: 1768-1781	The authors applied a hydrological model to assess how floodplain and non-floodplain wetlands affect streamflow in the Becancour River watershed of the St Lawrence Lowlands, Quebec, Canada. Their model simulations suggested that the more often floodplain wetlands are connected to the main stem channel, the greater their effects on moderating high flows and providing baseflow support. They suggest that wetland effects on streamflow depends on the "combined effect of wetland and landscape attributes".
Daneshvar, F., A. P. Nejadhashemi, U. Adhikari, B. Elahi, M. Abouali, M. R. Herman, E. Martinez-Martinez, T. J. Calappi and B. G. Rohn (2017). "Evaluating the significance of wetland restoration scenarios on phosphorus removal." <i>Journal of</i> <i>Environmental Management</i> 192: 184-196	The authors used a calibrated Soil and Water Assessment Tool (SWAT) model to assess what areas of the Saginaw River Watershed, Michigan, had the highest potential for successful (floodplain and non-floodplain) wetland restoration related to decreasing downstream phosphorous loads. They found that "wetlands located in headwaters and downstream had significantly higher phosphorus reduction than the ones located in the middle of the watershed. More specifically, wetlands implemented at distances ranging from 200 to 250 km and 50-100 km from the outlet had the highest impact on phosphorus reduction at the subwatershed and watershed levels, respectively".
Blanchette, M., A. N. Rousseau, É. Foulon, S. Savary and M. Poulin (2019). "What would have been the impacts of wetlands on low flow support and high flow attenuation under steady state land cover conditions?" <i>Journal of Environmental Management</i> 234: 448-457	The authors use a physically based hydrological model to quantify how land cover change, particularly for wetlands both inside and outside floodplains, affect streamflow in the St. Charles River, Quebec, Canada. They found that with 15% loss of wetlands in the watershed area, baseflow decreased and peak (or flood) flows increased. Their results suggested that "the loss of wetland areas generally leads to a loss of hydrological services and highlighted the need for wetland conservation programs and restoration actions."

Reference	Plain Text Summary
Bellmore, R. A., J. E. Compton, J. R. Brooks, E. W. Fox, R. A. Hill, D. J. Sobota, D. J. Thornbrugh and M. H. Weber (2018). "Nitrogen inputs drive nitrogen concentrations in U.S. streams and rivers during summer low flow conditions." <i>Science of the Total Environment</i> 639: 1349- 1359	The authors examined how in-stream nitrogen (N) concentrations are related to N inputs to watersheds (e.g., atmospheric deposition, synthetic fertilizer), land cover characteristics (e.g., wetland presence), and stream network characteristics across the United States. They found that (floodplain and non-floodplain) wetlands mediated. i.e., lowered, N concentrations in streams with watersheds draining areas of high agricultural N inputs across the US.
Carlson, A. K., M. J. Fincel, C. M. Longhenry and B. D. S. Graeb (2016). "Effects of historic flooding on fishes and aquatic habitats in a Missouri River delta." <i>Journal of Freshwater Ecology</i> 31(2): 271- 288	Fish assemblages were studied pre and post flood in the Missouri River, Lewis and Clark delta in SD and NE. Findings suggest that backwater habitats in the delta provided refuge from floodwaters during the disturbance. Maintaining habitat connectivity in deltas during and after floods is particularly important for fisheries conservation.
Tetzlaff, D., C. Birkel, J. Dick, J. Geris and C. Soulsby (2014). "Storage dynamics in hydropedological units control hillslope connectivity, runoff generation, and the evolution of catchment transit time distributions." <i>Water Resources Research</i> 50(2): 969-985	In a montane catchment in NE Scotland, storage dynamics and isotopic analysis of riparian peatlands showed that water stored in the peats were typically >80% of flow, including base flow and storm flow. The riparian areas were a key zone, acting as a regulator of stream water composition and transit time.
Rudolph, J. C., C. A. Arendt, A. G. Hounshell, H. W. Paerl and C. L. Osburn (2020). "Use of Geospatial, Hydrologic, and Geochemical Modeling to Determine the Influence of Wetland-Derived Organic Matter in Coastal Waters in Response to Extreme Weather Events." <i>Frontiers in</i> <i>Marine Science</i> 7	Using a flood model, the contribution of Organic Matter (OM) from wetlands from a recent hurricane was quantified in Neuse River Estuary-Pamlico Sound (NRE-PS), in eastern North Carolina. The hurricane created a pulse of OM, with wetland contributing 48% and 18% of annual DOC loads to the NRE and PS respectively. The study highlights the importance of rare large events on the transport of materials within the connected riverine-estuarine system.
Bourgault, M. A., M. Larocque and M. Roy (2014). "Simulation of aquifer-peatland-river interactions under climate change." <i>Nordic Hydrology</i> 45(3): 425-440	In a peatland complex within Quebec Canada, a ground water (GW) model was used to investigate the role of the peatlands in supporting river baseflows. The model estimated that on average 77% of the annual river base flow originates from the peatland. Future climate scenarios both indicated reductions to the peat from the surrounding GW and then from the peat to the river.
Vanderhoof, M. K., H. E. Distler, M. W. Lang and L. C. Alexander (2018). "The influence of data characteristics on detecting wetland/stream surface-water connections in the Delmarva Peninsula, Maryland and Delaware." <i>Wetlands Ecology and</i> <i>Management</i> 26: 63-86	Remote sensing techniques indicated springtime connections of wetlands and streams in a Delmarva Peninsula river. Using Lidar, Landsat and Worldview imagery, between 12-60% of wetlands connected to streams, indicating that 50-94% of the watershed contributed direct surface water runoff to stream flows.

Reference	Plain Text Summary
Manfrin, A., M. Bunzel-Drüke, A. W. Lorenz, A. Maire, M. Scharf, O. Zimball and S. Stoll (2020). "The effect of lateral connectedness on the taxonomic and functional structure of fish communities in a lowland river floodplain." <i>Science of the</i> <i>Total Environment</i> 719: 137169	Using an 18-year data set of fish abundance, researchers in Germany compared fish communities in main channels and floodplain habitats. While the overall diversity and relative abundance of species decreased from the main channel to more isolated floodplain wetlands, the floodplain waterbodies showed distinct assemblages with different life histories and feeding strategies. This highlights the importance of including the complete spectrum of connected floodplains in conservation.
Gordon, B. A., O. Dorothy and C. F. Lenhart (2020). "Nutrient Retention in Ecologically Functional Floodplains: A Review." <i>Water</i> 12(10)	In a review of North American and European literature, the authors quantify the removal of nitrogen (N) and phosphorus (P) within floodplains. The review found that floodplains remove an average of 200 kg-N/ha/yr of nitrate and 21 kg-P/ha/yr of total or particulate P and floodplain wetlands are most effective when located within river systems with higher nutrient loads.
Atkinson, C. L., B. C. van Ee, Y. Lu and W. Zhong (2019). "Wetland floodplain flux: temporal and spatial availability of organic matter and dissolved nutrients in an unmodified river." <i>Biogeochemistry</i> 142(3): 395-411	Within an unregulated, low gradient river in Alabama, extensive floodplains including the Sipsey Swamp were found to exert strong controls of organic materials and nutrients. Over two years at 10 sites, nutrients declined through the floodplains while the same floodplains supplied large amounts of organic material downstream. This research highlights the importance of floodplain complexes on the transport of organics and nutrients.
Abrial, E., L. A. Espinola, A. n. Rabuffetti, M. F. Eurich, A. R. Paira, M. C. M. Blettler and M. L. Amsler (2019). "Variability of hydrological connectivity and fish dynamics in a wide subtropical-temperate floodplain." <i>River Research and Applications</i> 35(9): 1520-1529	Researchers analyzed fish abundance and richness within a large floodplain system in the Parana River of Brazil over seven years. Fish dispersal, abundance and migration patterns during the different wet and dry seasons could largely be explained by the variation in connectivity of the floodplain habitats. The study demonstrates the importance of dynamic connection and isolation of floodplains on the makeup of the fish community.
Martens, K. D. and P. J. Connolly (2014). "Juvenile Anadromous Salmonid Production in Upper Columbia River Side Channels with Different Levels of Hydrological Connection." <i>Transactions of the American</i> <i>Fisheries Society</i> 143(3): 757-767	Young salmon use side or off channel habitat that can connect and disconnect from the main river channel. This study looked at seasonally and continually connected side channels on the Upper Columbia river and measured salmon survival. Seasonally disconnected side channels resulted in improved survival for juvenile salmon during periods of disconnection. Upon reconnection with the main channel, the previous cohort would rejoin the main population while new young of year salmon would move into the side channels.
De Giudici, G., D. Medas, R. Cidu, P. Lattanzi, F. Podda, F. Frau, J. C. I. Dick, D. Tetzlaff and C. Soulsby (2018). "Role of riparian wetlands and hydrological connectivity in the dynamics of stream thermal regimes." <i>Nordic Hydrology</i> 49(3): 634-647	Riparian wetlands were analyzed for their important in determining stream temperatures. The authors found that in periods of high river and riparian wetland connectivity, the coupled saturation and connectivity decreased the relative importance of the riparian wetland for temperature regulation. Conversely, dry periods with less river and riparian floodplain hydrologic connectivity were found to be important periods of distinctions between river water and riparian wetland temperatures (e.g., lower temperature waters were coming from the riparian wetland to the riverine system).

Reference	Plain Text Summary
Battauz, Y. S., S. B. Jose de Paggi and J. C. Paggi (2017). "Macrophytes as dispersal vectors of zooplankton resting stages in a subtropical riverine floodplain." <i>Aquatic</i> <i>Ecology</i> 51(2): 191-201	Aquatic plants located in floodplain open waters were analyzed as potential sources for zooplankton dispersal into the riverine food web. Six plant species from Brazilian floodplain lakes (at least three of which occur in North American open waters) were analyzed as sources of passive zooplankton dispersal. The roots and submerged parts of the plants were found to host 70 different zooplankton taxa in resting stage (i.e., awaiting the proper environmental cues to emerge). The authors concluded that aquatic plants in floodplain open waters are important downstream dispersers of zooplankton, which are important base components of riverine food webs.
Vanderhoof, M. K., J. R. Christensen and L. C. Alexander (2016). "Patterns and drivers for wetland connections in the Prairie Pothole Region, United States." <i>Wetlands</i> <i>Ecology and Management</i> 25(3): 1-23	A remote sensing study in the Prairie Pothole Region of the US noted that precipitation-based expansion of surface waters connected wetlands and stream networks across a wide range, averaging from 90-1400 m, depending on the ecoregion studied. Most of the wetland to stream connections occurred first through consolidation, where clusters of wetlands connected to each other, followed by the stream connection which occurred most frequently through a riparian wetland.
Rees, G. N., R. A. Cook, N. S. P. Ning, P. J. McInerney, R. T. Petrie and D. L. Nielsen (2020). "Managed floodplain inundation maintains ecological function in lowland rivers." <i>Science of the Total Environment</i> 727: 138469	An inundated floodplain riparian zone was shown to be highly connected to the river food web, with substantively higher dissolved organic carbon, seston carbon, nutrients (nitrogen), and chlorophyll levels downstream of where the flood waters reentered the river. Isotopic analyses demonstrated that floodplain-derived carbon was incorporated into the riverine food webs and was measurably found for up to four months following the flood peak.
Macdonald, D. M. J., A. J. Dixon and D. C. Gooddy (2018). "Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications." <i>Ecological</i> <i>Engineering</i> 123: 226-237	A study in the United Kingdom measured river water with high nitrate levels, which was noted to move into the riparian floodplain and recharge the local aquifer during overbank floods. The authors reported substantial microbially mediated denitrification occurred in the shallow riparian groundwater table due to carbon availability, oxygen-free conditions, and high nitrate concentrations. The lowland floodplain studied was able to annual remove substantial amounts of nitrate, though this was estimated to be three orders of magnitude less than the annual flux within the river. The floodplain nitrate reduction was noted to be locally important (e.g., for local drinking water supplies sourced from the river's alluvial aquifer).
Gillespie, J. L., G. B. Noe, C. R. Hupp, A. C. Gellis and E. R. Schenk (2018). "Floodplain Trapping and Cycling Compared to Streambank Erosion of Sediment and Nutrients in an Agricultural Watershed." <i>Water Resources Bulletin</i> 54(2 (Apr 2018)): 565	The authors analyzed sedimentation, nutrient loads, and mineralization of a floodplain in an agricultural watershed in the Valley and Ridge physiographic province of the US. All study reaches were areas of net sediment deposition (e.g., river-borne sediments were deposited), had high nitrogen (N) and phosphorus (P) deposited by the river, high mineralization of N and P, and high concentrations of N and P in the floodplain soils. They conclude that the net sediment and nutrient trapping functions of their study watershed floodplains (Smith Creek) benefit downgradient water quality.

Reference	Plain Text Summary
Webb, J. R., I. R. Santos, B. Robson, B. e. Macdonald, L. Jeffrey and D. T. Maher (2017). "Constraining the annual groundwater contribution to the water balance of an agricultural floodplain using radon: The importance of floods." <i>Water</i> <i>Resources Research</i> 53(1): 544-562	Groundwater dynamics of an extensively drained agricultural floodplain was analyzed using radon to create a water budget. Flooding in the riparian zone occurred only 12% of the study period but contributed 72-76% of the groundwater discharge (to the river network). Annually, groundwater discharge contributed 30-80% of the total surface water discharged to the river system, which the authors propose is related to the high density of constructed drainage features (12.4 km per km ²) altering the hydrology of the floodplain.
Scott, D. T., R. F. Keim, B. L. Edwards, C. N. Jones and D. E. Kroes (2014). "Floodplain biogeochemical processing of floodwaters in the Atchafalaya River Basin during the Mississippi River flood of 2011." <i>Journal of Geophysical Research:</i> <i>Biogeosciences</i> 119(4): 537-546	The Atchafalaya River and floodplain was extensively flooded during the 2011 Lower Mississippi River flood event, with up to half of the water in the channel moving into the floodplain. The authors analyzed river water over the flood event and found that significant nitrate reduction (around 75%) occurred within the floodplain. The floodplain system was found to reduce total nitrate by 16.6% over the course of the flood event.
Quin, A. and G. Destouni (2018). "Large- scale comparison of flow-variability dampening by lakes and wetlands in the landscape." <i>Land Degradation &</i> <i>Development</i> 29(10): 3617-3627	A modeling study in Sweden used data from 82 catchments from 1984-2013 to investigate the floodwater attenuation capacity of floodplain wetlands and lakes. The storage function of lakes and floodplain wetlands lakes was responsible for decreasing the variability (e.g., "flashiness") of streamflow. Watersheds comprised of approximately 15% lakes and 0.5% floodplain wetlands decreased the streamflow variability to around 10-15%, compared to areas without lakes or floodplain wetlands, which had approximately 20-25% higher streamflow variabilities due to low landscape water storages.
Dwivedi, D., B. Arora, C. I. Steefel, B. Dafflon and R. Versteeg (2018). "Hot Spots and Hot Moments of Nitrogen in a Riparian Corridor." <i>Water Resources Research</i> 54(1): 205-222	Groundwater within the Colorado River riparian zone was modeled to typically flow towards the river, except during flood stages when oxygenated and nitrate-laden river water overtops the banks and then infiltrates downwards in the riparian area. Riparian area sediments differ in their nitrate removal capacity, and reduced zones along the river (or areas of low oxygen, as found in most wetland soils) were found to have approximately 70% greater nitrate removal capacity than non-reduced zones. However, the nitrate removal capacity of the reduced zones varied based on the oxygen content of the infiltrating water from 70% greater than non-reduced zones to ~5% greater than non-reduced zones.

Non-Floodplain Wetlands and Open Waters

Reference	Plain Text Summary
Vanderhoof, M. K., H. E. Distler, M. W. Lang and L. C. Alexander (2017). "The influence of data characteristics on detecting wetland/stream surface-water connections in the Delmarva Peninsula, Maryland and Delaware." <i>Wetlands Ecology and</i> <i>Management</i> 26: 63-86	A remote-sensing study in Maryland and Delaware found that streams and depressional wetlands were surface-water connected in spring 2015. The range reported in the large studied watershed, 12-60% of wetlands by count and 21-93% of wetlands by area, varied due to the spatial and temporal wetland and stream characteristics and the accuracy and resolution of the input remote-sensing datasets.

Reference	Plain Text Summary
Park, J., D. Wang and M. Kumar (2020). "Spatial and temporal variations in the groundwater contributing areas of inland wetlands." <i>Hydrological Processes</i> 34(5)	Hydrologic connections between groundwater and wetlands were measured in a study in the southern US. The study noted that a) groundwater contributing areas to wetlands often have a different extent and shape than topographic contributing areas, and b) groundwater-fed wetlands in the study area were found to expand their groundwater contributing area (and received greater groundwater) during dry periods which could influence baseflow in downstream waters (depending on hydraulic gradients).
Wang, N., X. Zhang and X. Chu (2019). "New Model for Simulating Hydrologic Processes under Influence of Surface Depressions." <i>Journal of Hydrologic</i> <i>Engineering</i> 24(5): https://doi.org/10.1061/(ASCE)HE.1943- 5584.0001772	In a North Dakota study, a model was developed to assess the effects of depressions on rainfall-runoff processes and mechanisms of dynamic water release from depressions affecting down-gradient systems. The hydrologic model demonstrated that depressions captured and held back precipitation from down-gradient systems such that the majority of the study area did not contribute water directly to the stream system. This finding supports the hydrologic lag and sink functions of depressions (wetlands) in attenuating storm flows and maintaining baseflows.
Bugna, G. C., J. M. Grace and Y. P. Hsieh (2020). "Sensitivity of using stable water isotopic tracers to study the hydrology of isolated wetlands in North Florida." <i>Journal</i> <i>of Hydrology</i> 580: 124321	In a North Florida water isotopic study of "isolated wetlands," the authors found that so-called isolated wetlands stored and evaporated rainwater (thereby retaining water and performing hydrological lag and sink functions). A sinkhole (pond) was found to connect to both groundwater and precipitation. Wetlands and sinkholes were measurably important to quantifying and determining hydrological budgets for forested watersheds.
Lewis, D. B. and S. J. Feit (2015). "Connecting carbon and nitrogen storage in rural wetland soil to groundwater abstraction for urban water supply." <i>Global Change</i> <i>Biology</i> 21(4): 1704-1714	Depressional wetlands were hydrologically affected by groundwater withdrawals for urban use (e.g., drinking water). Groundwater withdrawals diminished wetland hydroperiods, which in turn negatively affected carbon, nitrogen, and soil organic matter stocks in the wetlands (e.g., carbon sink functions of wetlands were diminished).
Shook, K., S. Papalexiou and J. W. Pomeroy (2021). "Quantifying the effects of Prairie depressional storage complexes on drainage basin connectivity." <i>Journal of Hydrology</i> 593: 125846	In a Canadian hydrologic modeling study, Prairie Pothole depression water storage was found to control the fraction of the watershed that contributes flow to down-gradient stream systems. The effects of depressions varied: when there were few extant depressions, their size and location on the landscape was most important. In systems with greater depression abundance, depressions still controlled the relationship between water storage and the fraction of the watershed contributing surface flow down-gradient but the spatial location within the watershed decreased in importance.
Vasic, F., C. Paul, V. Strauss and K. Helming (2020). "Ecosystem Services of Kettle Holes in Agricultural Landscapes." 10(9)	Kettle holes are glacially formed wetlands in Europe similar to the North American prairie potholes. A European literature review determined kettle hole wetlands provided important ecosystem services, including flood control and hydrological cycling, biogeochemical functions, and habitat. Agricultural activities around kettle hole wetlands were noted to potentially affect the provisioning of wetland ecosystem services.

Reference	Plain Text Summary
Nasab, M. T. and X. Chu (2020). "Macro- HyProS: A new macro-scale hydrologic processes simulator for depression- dominated cold climate regions." <i>Journal of</i> <i>Hydrology</i> 580: 124366	A hydrological modeling study of the Red River of the North (northern Great Plains) found that (wetland) depression-dominated areas controlled (or regulated) the connectivity of large areas of the basin via storage affecting surface-driven runoff. This was particularly important in the early spring months (i.e., during periods of rain-on-snow events, snowmelt, etc.).
Michelson, C., R. G. Clark and C. A. Morrissey (2018). "Agricultural land cover does not affect the diet of Tree Swallows in wetland-dominated habitats." <i>The Condor</i> 120(4): 751-764	Feeding habits of tree swallows, an insectivorous bird, were analyzed in a Canadian prairie landscape. Tree swallows were found to specialize in feeding on aquatic insects in wetland-dominated habitats (i.e., those insects emerging from wetlands). Agricultural land cover (e.g., grasslands, crops) within the study area did not affect tree swallow foraging success, though tree swallows were larger and in better condition in grasslands than cropped landscapes.
Martin, A. R., M. L. Soupir and A. L. Kaleita (2019). "Seasonal and intra-event nutrient levels in farmed Prairie Potholes of the Des Moines Lobe." <i>Transactions of the ASABE</i> 62(6): 1607-1617	Farmed wetlands (drained and under corn-soybean rotation) in the Des Moines Lobe of Iowa were found to reduce nitrate that entered into the wetland in 85% of the multi-day inundation events. Phosphorous was found to increase in the wetland water column over the inundation period (e.g., possibly through release from phosphorus sorbed onto soil particles), meaning that in addition to serving as nitrate removal areas, farmed wetlands were sources of total and soluble reactive phosphorus.
Shook, K., J. Pomeroy and G. van Der Kamp (2015). "The transformation of frequency distributions of winter precipitation to spring streamflow probabilities in cold regions; case studies from the Canadian Prairies." <i>Journal of Hydrology</i> 521: 395-409	Large-scale analyses of climate and basin processes contributing to streamflow in the Canadian Prairies were investigated. The authors found three major controls on streamflow, including a) the creation and distribution of the spring snowpack, b) the spring melt of the snowpack over frozen ground, and c) the subsequent filling and spilling of depressional (wetland) storage that connected fields, ponds, wetlands, and down-gradient lake systems.
Al Sayah, M. J., R. Nedjai, K. Kaffas, C. Abdallah and M. Khouri (2019). "Assessing the Impact of Man-Made Ponds on Soil Erosion and Sediment Transport in Limnological Basins." <i>Water</i> 11(12): 2526	Ponds, open-water systems in a French study, were analyzed in a modeling study to determine their effects on erosion and sediment transport at the watershed scale. The presence of ponds controlled sediment transport and erosion risk, with ~78% of the basin corresponding to no- or low-erosion risk zones and 22% noted as moderate to high-erosion risk. Without ponds, <2% of the basin was determined to be no- or low-erosion risk, while 98% was modeled as moderate to high erosion risk. Without ponded systems, the sediment pattern completely shifted to zones of higher sediment yields.
Kappas, I., G. Mura, D. Synefiaridou, F. Marrone, G. Alfonso, M. Alonso and T. J. Abatzopoulos (2017). "Molecular and morphological data suggest weak phylogeographic structure in the fairy shrimp <i>Streptocephalus torvicornis</i> (Branchiopoda, Anostraca)." <i>Hydrobiologia</i> 801(1): 21-32	A pan-European study analyzed the genetic structure of a fairy shrimp found in temporary ponds. Their results demonstrated there was unhindered gene flow and widespread connectivity between populations across the study area. One of five hypothesized reasons includes frequent dispersal by avian species throughout the range.

Reference	Plain Text Summary
Wendt, A., C. A. Haas, T. Gorman and J. H. Roberts (2021). "Metapopulation genetics of endangered reticulated flatwoods salamanders (<i>Ambystoma bishopi</i>) in a dynamic and fragmented landscape." <i>Conservation Genetics</i>	Population dynamics of the reticulated flatwoods salamander, found in forested ponds and riparian zones in the southeastern U.S., were analyzed. Distance between ponds was found to be an important factor controlling metapopulation dynamics, with very low migration among ponds further than 400 m.
Yeo, I. Y., M. W. Lang, S. Lee, G. W. McCarty, A. M. Sadeghi, O. Yetemen and C. Huang (2019). "Mapping landscape-level hydrological connectivity of headwater wetlands to downstream waters: A geospatial modeling approach - Part 1." <i>Science of the</i> <i>Total Environment</i> 653: 1546-1556	A modeling study analyzed the hydrologic connectivity of so-called geographically isolated wetlands of the Mid-Atlantic region of the US. Wetland inundation and stream flows were well correlated, demonstrating a similar relationship. Wetlands with longer flooding duration were more strongly correlated with stream discharge than shorter-duration inundated wetlands. The authors conclude that the wetlands function in aggregate and that both the streams and the wetlands of their 300 km2 study area were connected via groundwater pathways.
Vanderhoof, M. K., J. R. Christensen and L. C. Alexander (2016). "Patterns and drivers for wetland connections in the Prairie Pothole Region, United States." <i>Wetlands</i> <i>Ecology and Management</i> 25(3): 1-23	A remote-sensing study (1990-2011) quantified surface-water connections between streams and wetlands across the US Prairie Pothole Region. They reported surface-water connections varied across ecoregions of the Prairie Pothole Region, averaging 90-1400 m. Connections were controlled by the arrangement and abundance of wetlands and surface-water expansion.
Neff, B. P. and D. O. Rosenberry (2017). "Groundwater Connectivity of Upland- Embedded Wetlands in the Prairie Pothole Region." <i>Wetlands</i> 38(1): 51-63	Local to regional groundwater connectivity between wetlands and other waters in the Prairie Pothole Region of North Dakota was modeled. Sand layers were found to facilitate wetland connectivity through groundwater, whereas water-table mounds were found to retard connectivity if completely surrounding the wetland or wetland complex. In the absence of restricting water-table mounds, connectivity was modeled to occur.
Golden, H. E., H. A. Sander, C. R. Lane, C. Zhao, K. Price, E. D'Amico and J. R. Christensen (2015). "Relative effects of geographically isolated wetlands on streamflow: a watershed-scale analysis." <i>Ecohydrology</i> 9(1): 21-38	A modeling study in North Carolina found that geographically isolated wetlands influenced streamflow. Increased water storage associated with increased geographically isolated wetland extent decreased streamflow. The distance of geographically isolated wetlands was positively associated with streamflow, affecting (or reflecting) the movement of water across the landscape.
Brooks, R. J., D. M. Mushet, M. Vanderhoof, S. G. Leibowitz, J. R. Christensen, B. P. Neff, D. Rosenberry, W. D. Rugh and L. C. Alexander (2018). "Estimating Wetland Connectivity to Streams in the Prairie Pothole Region: An Isotopic and Remote Sensing Approach." <i>Water Resources</i> <i>Research</i> 54(2): 955-977.	A water isotope study in a North Dakota watershed found that Prairie Pothole wetlands had high evaporation rates, and that groundwater typified winter precipitation-based recharge. However, the evaporative isotopic signal in the steam indicated that surficial flow from wetlands contributed to and connected to the stream network throughout the summer.

Reference	Plain Text Summary
Ameli, A. A. and I. F. Creed (2019). "Does Wetland Location Matter When Managing Wetlands for Watershed-Scale Flood and Drought Resilience?" <i>Journal of the</i> <i>American Water Resources Association</i> 55(3): 529-542	A Canadian modeling study in the Prairie Pothole Region found that wetland loss affected streamflow, increasing peak flows from storm events that led to major down-gradient flooding in cities. Concurrently, wetland losses decreased base flow. Wetlands closer to the stream network were found to be disproportionately important to peak flow attenuation, while wetlands were found to be important controllers of baseflow regardless of their location vis-à-vis the stream network.
McKenna, O. P., S. R. Kucia, D. M. Mushet, M. J. Anteau and M. T. Wiltermuth (2019). "Synergistic Interaction of Climate and Land-Use Drivers Alter the Function of North American, Prairie-Pothole Wetlands." <i>Sustainability</i> 11(23): 6581	Twenty-five wetland basins were modeled over a 70-year period to ascertain the influence of both climate and land-use drivers on floods. During an extremely wet period (1993-2000), Prairie Pothole wetland drainage decreased watershed-scale water storage, resulting in 10 times the volume of water transiting towards local stream networks.
Bam, E., A. M. Ireson, G. Kamp and J. Hendry (2020). "Ephemeral Ponds: Are They the Dominant Source of Depression-Focused Groundwater Recharge?" <i>Water Resources</i> <i>Research</i> 56(3): e2019WR026640	Prairie wetland ponds within a Canadian study area are sources of recharge to confined groundwater aquifers providing water to farm and rural communities. An isotopic analysis found that permanently inundated wetland ponds were not the dominant groundwater recharge source. Ephemerally inundated wetlands were found to have identical isotopic signatures as the groundwater in aquifers. Ephemeral wetlands were the dominant source of groundwater recharge.
Ameli, A. and I. F. Creed (2019). "Groundwaters at Risk: Wetland Loss Changes Sources, Lengthens Pathways, and Decelerates Rejuvenation of Groundwater Resources." <i>Journal of the American Water</i> <i>Resources Association</i> 55(2): 294-306	A modeling analysis of a Canadian study area explored the connections and effects of wetlands, and wetland losses, on local and regional waters. Wetland losses decreased contributing areas affecting baseflows of local surface waters. Modeled wetland losses increased the contributions to regional surface waters through subsurface hydrologic connections, which increased regional baseflows.
Sampath, P. V., H. Liao, Z. K. Curtis, P. J. Doran, M. E. Herbert, C. A. May and S. Li (2015). "Understanding the Groundwater Hydrology of a Geographically-Isolated Prairie Fen: Implications for Conservation." <i>PLoS ONE</i> 10(10): e0140430	An analysis of a geographically isolated Michigan fen wetland determined the wetland was groundwater-fed by four sources: local recharge, regional recharge, a regional groundwater mound, and a nearby pond. The authors conclude a 3-dimensional groundwater 'pipeline' connects this fen to other fen cluster throughout southern Michigan, an interconnected and larger network of fens.
Marton, J. M., I. F. Creed, D. B. Lewis, C. R. Lane, N. B. Basu, M. J. Cohen and C. B. Craft (2015). "Geographically Isolated Wetlands are Important Biogeochemical Reactors on the Landscape." <i>BioScience</i> 65(4): 408-418	The authors conducted a literature review of biogeochemical functions performed by geographically isolated wetlands. They found these wetlands provided biogeochemically mediated ecosystem services such as sediment and carbon retention, nutrient transformations, and water quality improvement that maintain the integrity of US waters.

Reference	Plain Text Summary
Cohen, M. J., I. F. Creed, L. Alexander, N. B. Basu, A. J. K. Calhoun, C. Craft, E. D'Amico, E. Dekeyser, L. Fowler, H. E. Golden, J. W. Jawitz, P. Kalla, L. K. Kirkman, C. R. Lane, M. Lang, S. G. Leibowitz, D. B. Lewis, J. Marton, D. L. McLaughlin, D. M. Mushet, H. Raanan- Kiperwas, M. C. Rains, L. Smith and S. C. Walls (2016). "Do geographically isolated wetlands influence landscape functions?" <i>Proceedings of the National Academy of</i> <i>Sciences of the United States of America</i> 113(8): 1978-1986	Geographically isolated wetlands across the conterminous US were described as existing along a connectivity continuum. They were found to provide a disproportionately large fraction of wetland edges where many biogeochemical functions were enhanced. They also found that the slow (e.g., through groundwater) or episodic (e.g., through surface water) nature of wetland hydrologic connectivity to other waters provided the conditions for biogeochemical processing, sediment retention, and both biological and hydrological functioning.
Flint, S. A. and W. H. McDowell (2015). "Effects of headwater wetlands on dissolved nitrogen and dissolved organic carbon concentrations in a suburban New Hampshire watershed." <i>Freshwater Science</i> 34(2): 456-471	Ten headwater wetlands, a possible type of non-floodplain wetland system, were analyzed in New Hampshire. The headwater wetlands were found to decrease nitrate and increase dissolved organic carbon and nitrogen concentrations, and to vary the seasonal values of total dissolved nitrogen. These functions would affect the downgradient system.
Denver, J. M., S. W. Ator, M. W. Lang, T. R. Fisher, A. B. Gustafson, R. Fox, J. W. Clune and G. W. McCarty (2014). "Nitrate fate and transport through current and former depressional wetlands in an agricultural landscape, Choptank Watershed, Maryland, United States." <i>Journal of Soil and Water</i> <i>Conservation</i> 69(1): 1-16	Depressional wetlands in the Choptank River, a tributary to the Chesapeake Bay, were analyzed for biogeochemical processing. Natural wetlands had conditions conducive to nitrogen pollution removal for longer than farmed wetlands or restored wetlands but were generally groundwater-connected and were not exposed to nitrogen-laden waters; they were found to provide water to streams that diluted pollution concentrations. Farmed wetlands and restored wetlands that were exposed to nitrate pollution through groundwater removed substantial amounts of nitrate, but contributions to water quality improvement hinged on exposure to polluted waters.
Rajib, A., H. E. Golden, C. R. Lane and Q. Wu (2020). "Surface depression and wetland water storage improves major river basin hydrologic predictions." <i>Water Resources</i> <i>Research</i> 56(7): e2019WR026561	In an Upper Mississippi River Basin study, 455,000 depressional wetlands and open waters were incorporated into a hydrologic model and significantly affected and improved streamflow measures; these waters further improved the remotely sensed water yield across 70% of the study area. These results demonstrate the significant influence of wetlands and open waters on stream flow, and that wetlands and open waters affect landscape-scale hydrological conditions (e.g., rootzone wetness).
Schmadel, N. M., J. W. Harvey, G. E. Schwarz, R. B. Alexander, J. D. Gomez- Velez, D. Scott and S. W. Ator (2019). "Small Ponds in Headwater Catchments Are a Dominant Influence on Regional Nutrient and Sediment Budgets." <i>Geophysical</i> <i>Research Letters</i> 46(16): 9669-9677	Small ponds and impoundments across the Northeastern US were found to be important biogeochemical and physical sinks, retaining 34% of nitrogen, 69% of all phosphorus, and 12% of sediments in the study area. Their influence was dominant in headwater catchments, where they contained 54% of nitrogen, 85% of phosphorus, and 50% of sediments decreasing loads and thereby affecting downstream waters.

Reference	Plain Text Summary
Nasab, M. T., V. Singh and X. Chu (2017). "SWAT Modeling for Depression- Dominated Areas: How Do Depressions Manipulate Hydrologic Modeling?" <i>Water</i> 9(1): w9010058	Hydrologic modeling in a large watershed in North Dakota found that depression (e.g., Prairie Potholes) storage exerted a gate-keeper effect on downstream flows, decreasing surface runoff contributing to stream flow during low-flow scenarios. Conversely, during high precipitation scenarios, depressions (hydrologic gatekeepers) increased surface runoff peak flows.
Green, D. I. S., S. M. McDeid and W. G. Crumpton (2019). "Runoff Storage Potential of Drained Upland Depressions on the Des Moines Lobe of Iowa." <i>JAWRA Journal of</i> <i>the American Water Resources Association</i> 55(3): 543-558	Drained depressions (possibly Prairie Pothole wetlands in farmed landscapes) in Iowa were found to store up to 903.5 million m3 of runoff. Runoff from a 1-yr, 24-hr event would likely exhaust this storage, while rainfall runoff from a 5-yr, 24-hr event would exceed the capacity of the drained depressional storage.
Thorslund, J., M. J. Cohen, J. W. Jawitz, G. Destouni, I. F. Creed, M. C. Rains, P. Badiou and J. Jarsjö (2018). "Solute evidence for hydrological connectivity of geographically isolated wetlands." <i>Land Degradation &</i> <i>Development</i> 29(11): 3954-3962	In an analysis across North America, geographically isolated wetlands were found to generate runoff at 1.2x the mean catchment rate which implies they are well-connected watershed-scale sources of water to down-gradient systems.
McKenna, O. P., D. M. Mushet, D. O. Rosenberry and J. W. LaBaugh (2017). "Evidence for a climate-induced ecohydrological state shift in wetland ecosystems of the southern Prairie Pothole Region." <i>Climatic Change</i> 145(3): 273-287	Changing precipitation patterning in the Prairie Pothole Region has resulted in an increased number of ponded wetlands and open waters, increases in runoff to wetlands altering wetland solute concentrations, tile drainage connecting wetlands to down-stream systems, and increases in stream flows.
Evenson, G. R., H. E. Golden, C. R. Lane, D. L. McLaughlin and E. D'Amico (2018). "Depressional Wetlands Affect Watershed Hydrological, Biogeochemical, and Ecological Functions." <i>Ecological</i> <i>Applications</i> 28(4): 953-966	In a North Dakota hydrologic modeling study, the influence of non- floodplain wetlands on down-gradient stream characteristics was analyzed. Small depressional wetlands (<3.0 ha) significantly affected inundation characteristics (inundated area, hydrologic residence time, and inundation heterogeneity). Larger non-floodplain wetlands were gatekeepers, controlling flows to the stream network. Scenarios of wetland loss based on distance (30-m and 450-m from streams) decreased inundated areas and inundation residence time (for biogeochemical processing). Depression wetlands also attenuated peak flows, decreasing the probability of downstream flooding.
Schofield, K. A., L. C. Alexander, C. E. Ridley, M. K. Vanderhoof, K. M. Fritz, B. C. Autrey, J. E. DeMeester, W. G. Kepner, C. R. Lane, S. G. Leibowitz and A. I. Pollard (2018). "Biota Connect Aquatic Habitats throughout Freshwater Ecosystem Mosaics." <i>Journal of the American Water Resources</i> <i>Association</i> 54(2): 372-399	A review paper wherein freshwater streams, rivers, and wetlands were noted to form a 'freshwater ecosystem mosaic' that sustained aquatic life through connectivity and biotic linkages of organisms within and among these components. The biotic connectivity of these aquatic systems was considered to be critical to the ecological integrity of freshwaters.

Reference	Plain Text Summary
Ameli, A. A. and I. F. Creed (2017). "Quantifying hydrologic connectivity of wetlands to surface water systems." <i>Hydrol.</i> <i>Earth Syst. Sci.</i> 21: 1791-1808	In a modeled analysis in the North American Prairie Pothole Region, geographically isolated wetlands were found to have both "fast" surface-water connections and "slow" subsurface water connections to downgradient systems. Subsurface connections linked geographically isolated wetlands from throughout the watershed to the flowing water network, while surface water connections mainly emanated from large precipitation events and originated from wetlands closer to the stream network.
Mekonnen, B. A., K. A. Mazurek and G. Putz (2016). "Incorporating landscape depression heterogeneity into the Soil and Water Assessment Tool (SWAT) using a probability distribution." <i>Hydrological</i> <i>Processes</i> 30(13): 2373-2389	A hydrologic modeling study in the Canadian Prairie Pothole Region contrasted the effects of models with multiple small landscape (wetland) depressions versus a more typical and coarser modeling approach. Incorporating landscape depressions into the model improved streamflow estimates, suggesting that the hydrologic effects of depressions have an effect on down-gradient stream system characteristics.
Nitzsche, K. N., T. Kalettka, K. Premke, G. Lischeid, A. Gessler and Z. E. Kayler (2017). "Land-use and hydroperiod affect kettle hole sediment carbon and nitrogen biogeochemistry." <i>Science of The Total</i> <i>Environment</i> 574: 46-56	European kettle holes, like Prairie Potholes in North America, are wetland systems formed by past glaciation. An isotopic analysis of 51 kettle holes in Germany determined that in addition to processing organic matter from the surrounding area, kettle holes were coupled with shallow groundwater and were not closed hydrological systems but systems that connected with groundwater.
Golden, H. E., A. Rajib, C. R. Lane, J. R. Christensen, Q. Wu and S. Mengistu (2019). "Non-floodplain Wetlands Affect Watershed Nutrient Dynamics: A Critical Review." <i>Environmental Science & Technology</i> 53(13): 7203-7214	A review paper on the influence of non-floodplain wetlands on water quality at watershed scales. Includes analysis of how including non- floodplain wetlands in a Midwestern-US watershed-scale nutrient model markedly changes the predicted nutrient levels in down- gradient systems.
Evenson, G. R., H. E. Golden, C. R. Lane and E. D'Amico (2015). "Geographically isolated wetlands and watershed hydrology: A modified model analysis." <i>Journal of</i> <i>Hydrology</i> 529, Part 1: 240-256	In a hydrologic modeling study in a ~202 km2 North Carolina watershed, the authors directly analyzed the watershed-scale hydrologic effects of geographically isolated wetlands on stream characteristics. They found that geographically isolated wetlands: 1) seasonally affected stream baseflow, 2) decreased or attenuated storm peak flows, and 3) quantifiably affected the water balance (or the movement of water through various parts of the water cycle) at the watershed scale.

Appendix C3: Additional References

Appendix C3i: Reference Review Process and Findings

As discussed in section I.C.v of this Technical Support Document, reviewers from the U.S. Environmental Protection Agency's Office of Research and Development (ORD) identified an additional 37 papers in October 2021 that were not captured during the initial screening process. Additionally, the agencies solicited for and received additional scientific literature and references published since 2014 from the public during the notice and comment process. The agencies reviewed the 37 papers noted in section I.C.v (and cited in Appendix C3ii of the Technical Support Document for the Proposed Rule) as well as the literature submitted by the public, assessing first whether the literature was peer-reviewed and published in or after 2014, then (for the submitted literature) if it had already been included in Section I.C. (Updates to the Literature since Publication of the Science Report). Then, the agencies read each of the additional submitted papers that had been peer-reviewed and published in or after 2014 to discern if the paper provided updates on the "…scientific understanding about the connectivity and mechanisms by which streams and wetlands, singly or in aggregate, affect the physical, chemical, and biological integrity of downstream waters." Science Report at ES-1. To wit, the agencies assessed each paper for the charge questions listed in section I.C.i. (Update Process):

A. What are the physical, chemical, and biological connections to and effects of ephemeral, intermittent, and perennial streams on downstream waters (*e.g.*, rivers, lakes, reservoirs, estuaries)?

B. What are the physical, chemical, and biological connections to and effects of riparian or floodplain wetlands and open waters (*e.g.*, riverine wetlands, oxbow lakes) on downstream waters?

C. What are the physical, chemical, and biological connections to and effects of wetlands and open waters in non-floodplain settings (*e.g.*, most prairie potholes, vernal pools) on downstream waters?

Subsequently, the agencies assessed whether each paper provided findings relevant to the major conclusions of the Science Report (see sections I.A.i and I.A.ii), and determined this by aquatic system type as reported in the paper.

Streams: The scientific literature unequivocally demonstrates that streams, individually or cumulatively, exert a strong influence on the integrity of downstream waters. All tributaries, regardless of size or flow duration, are physically, chemically, and biologically connected to downstream waters and strongly influence their function.

Floodplain Wetlands and Open Waters: Wetlands and open waters in riparian areas and floodplains are physically, chemically, and biologically integrated with rivers via functions that improve downstream water quality. These systems buffer downstream waters from pollution and are essential components of river food webs.

Non-Floodplain Wetlands and Open Waters: Wetlands and open waters located outside of riparian areas and floodplains, even when lacking surface water connections, provide numerous

functions that could affect the integrity of downstream waters. Some benefits of these wetlands are due to their relative isolation rather than their connections.

The agencies chose from the following review responses: "supports findings," "refutes findings," or "cannot be discerned."

The agencies reviewed and assessed the citations they received during the public comment period. After assessing the publicly submitted citations as well as those 37 citations described in section I.C.v (*i.e.*, those added to the Technical Support Document in October 2021), the agencies determined that 80 new peer-reviewed references from the scientific literature had been published in or after 2014 needed to be assessed for their findings relevant to the Science Report's conclusions. The agencies did not include those papers which had not been peer-reviewed (or for which peer-review could not be ascertained) or those papers that were published prior to 2014 in this review, as they did not meet the criteria established for inclusion as part of the initial screening process. However, such papers were generally considered for relevance for other aspects of the Technical Support Document for the final rule. Some citations provided to the agencies were already part of the agencies' initial screening process (as described in section I.C of this Technical Support Document) and thus were not included in this review of additional references.

Specific to the 37 citations in Appendix C3ii (and noted below), the agencies were able to discern the relevant typology addressed in the paper in 31 cases (19 stream systems, five floodplain wetlands and open waters, seven non-floodplain wetlands and open waters). Seventeen of the papers with a determined typology from Appendix C3ii had sufficient information in the citation for the agencies to characterize a conclusion regarding the findings of the Science Report. In each case where a conclusion was reached (*i.e.*, all 17 citations), the agencies determined that the Science Report findings were supported by the literature (11 stream systems, three floodplain wetlands and open waters, and three non-floodplain wetlands and open waters papers; see Table 1).

Table C-8: References Relevant to the Conclusions of the Science Report (of the 37 Identified by the Agencies in October 2021 as Not Captured During the Screening Process). Seventeen of the 37 citations noted by ORD scientists as potentially relevant but that were not captured during the initial screening process (*see* section I.C.iv) were found to have sufficient information to discern system typology (*e.g.*, stream; non-floodplain wetlands and open waters; or floodplain wetlands and open waters) as well as a determination of whether the paper supports, refutes, or neither supports nor refutes the findings of the Science Report. In all 17 cases, the papers were found by ORD reviewers to support the findings of the Science Report.

Citation	Aquatic System Type	Science Report Findings
Acworth, R.I., G.C. Rau, M.O. Cuthbert, K. Leggett, and M.S. Andersen. 2021. "Runoff and focused groundwater-recharge response to flooding rains in the arid zone of Australia." <i>Hydrogeology Journal</i> 29 (2): 737-764. <u>https://doi.org/10.1007/s10040-020-02284-x</u> .	Streams	Supports

Citation	Aquatic System Type	Science Report Findings
 Arce, M.I., M.d.M. Sánchez-Montoya, and R. Gómez. 2015. "Nitrogen processing following experimental sediment rewetting in isolated pools in an agricultural stream of a semiarid region." <i>Ecological Engineering</i> 77: 233-241. <u>https://doi.org/10.1016/j.ecoleng.2015.01.035</u>. 	Streams	Supports
 Arora, B., M. Burrus, M. Newcomer, C.I. Steefel, R.W.H. Carroll, D. Dwivedi, W. Dong, K.H. Williams, and S.S. Hubbard. 2020. "Differential C-Q Analysis: A New Approach to Inferring Lateral Transport and Hydrologic Transients Within Multiple Reaches of a Mountainous Headwater Catchment." <i>Frontiers in Water</i> 2 (24). https://www.frontiersin.org/article/10.3389/frwa.2020.00024. 	Streams	Supports
 Baulch, H., C. Whitfield, J. Wolfe, N. Basu, A. Bedard-Haughn, K. Belcher, R. Clark, G. Ferguson, M. Hayashi, A. Ireson, P. Lloyd-Smith, P. Loring, J.W. Pomeroy, K. Shook, and C. Spence. 2021. "Synthesis of science: findings on Canadian Prairie wetland drainage." <i>Canadian Water Resources Journal / Revue canadienne des ressources hydriques</i>: 1-13. https://doi.org/10.1080/07011784.2021.1973911. 	Streams	Supports
Carroll, R.W.H., D. Gochis, and K.H. Williams. 2020. "Efficiency of the Summer Monsoon in Generating Streamflow Within a Snow- Dominated Headwater Basin of the Colorado River." Geophysical Research Letters 47 (23): e2020GL090856. <u>https://doi.org/10.1029/2020GL090856</u> .	Streams	Supports
Crabot, J., C.P. Mondy, P. Usseglio-Polatera, K.M. Fritz, P.J. Wood, M.J. Greenwood, M.T. Bogan, E.I. Meyer, and T. Datry. 2021. "A global perspective on the functional responses of stream communities to flow intermittence." <i>Ecography</i> 44 (10): 1511-1523. https://onlinelibrary.wiley.com/doi/abs/10.1111/ecog.05697.	Streams	Supports
 Evenson, G.R., H.E. Golden, J.R. Christensen, C.R. Lane, A. Rajib, E. D'Amico, D.T. Mahoney, E. White, and Q. Wu. 2021. "Wetland restoration yields dynamic nitrate responses across the Upper Mississippi river basin." <i>Environmental Research Communications</i> 3 (9): 095002. <u>http://dx.doi.org/10.1088/2515-7620/ac2125</u>. 	Non- floodplain Wetlands and Open Waters	Supports

Citation	Aquatic System Type	Science Report Findings
 Gall, H.E., S.A. Sassman, B. Jenkinson, L.S. Lee, and C.T. Jafvert. 2015. "Comparison of export dynamics of nutrients and animal-borne estrogens from a tile-drained Midwestern agroecosystem." <i>Water</i> <i>Research</i> 72: 162-173. <u>https://www.sciencedirect.com/science/article/pii/S004313541400608</u> <u>3</u>. 	Streams	Supports
Gallo, E.L., K.A. Lohse, C.M. Ferlin, T. Meixner, and P.D. Brooks. 2014. "Physical and biological controls on trace gas fluxes in semi- arid urban ephemeral waterways." <i>Biogeochemistry</i> 121 (1): 189-207. <u>https://doi.org/10.1007/s10533-013-9927-0</u> .	Streams	Supports
Golden, H.E., C.R. Lane, A. Rajib, and Q. Wu. 2021. "Improving global flood and drought predictions: integrating non-floodplain wetlands into watershed hydrologic models." <i>Environmental Research Letters 16</i> (9): 091002. <u>http://dx.doi.org/10.1088/1748-9326/ac1fbc</u> .	Non- floodplain Wetlands and Open Waters	Supports
 Gómez-Gener, L., A.R. Siebers, M.I. Arce, S. Arnon, S. Bernal, R. Bolpagni, T. Datry, G. Gionchetta, HP. Grossart, C. Mendoza-Lera, V. Pohl, U. Risse-Buhl, O. Shumilova, O. Tzoraki, D. von Schiller, A. Weigand, G. Weigelhofer, D. Zak, and A. Zoppini. 2021. "Towards an improved understanding of biogeochemical processes across surface-groundwater interactions in intermittent rivers and ephemeral streams." <i>Earth-Science Reviews</i> 220: 103724. https://www.sciencedirect.com/science/article/pii/S001282522100225 Z. 	Streams	Supports
 Hansen, A.T., T. Campbell, S.J. Cho, J.A. Czuba, B.J. Dalzell, C.L. Dolph, P.L. Hawthorne, S. Rabotyagov, Z. Lang, K. Kumarasamy, P. Belmont, J.C. Finlay, E. Foufoula-Georgiou, K.B. Gran, C.L. Kling, and P. Wilcock. 2021. "Integrated assessment modeling reveals near-channel management as cost-effective to improve water quality in agricultural watersheds." <i>Proceedings of the National Academy of Sciences</i> 118 (28): e2024912118. <u>http://www.pnas.org/content/118/28/e2024912118.abstract</u>. 	Floodplain Wetlands and Open Waters	Supports
Jakubínský, J., M. Prokopová, P. Raška, L. Salvati, N. Bezak, O. Cudlín, P. Cudlín, J. Purkyt, P. Vezza, C. Camporeale, J. Daněk, M. Pástor, and T. Lepeška. 2021. "Managing floodplains using nature-based solutions to support multiple ecosystem functions and services." <i>WIREs Water</i> 8 (5): e1545. <u>https://doi.org/10.1002/wat2.1545</u> .	Floodplain Wetlands and Open Waters	Supports

Citation	Aquatic System Type	Science Report Findings
 Kavehei, A., D.B. Gore, A.A. Chariton, and G.C. Hose. 2021. "Impact assessment of ephemeral discharge of contamination downstream of two legacy base metal mines using environmental DNA." <i>Journal of Hazardous Materials</i> 419: 126483. <u>https://www.sciencedirect.com/science/article/pii/S030438942101448</u> <u>5</u>. 	Streams	Supports
 Klammler, H., C.J. Quintero, J.W. Jawitz, D.L. McLaughlin, and M.J. Cohen. 2020. "Local Storage Dynamics of Individual Wetlands Predict Wetlandscape Discharge." <i>Water Resources Research</i> 56 (11): e2020WR027581. https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020WR027 581. 	Non- floodplain Wetlands and Open Waters	Supports
 Rau, G.C., L.J.S. Halloran, M.O. Cuthbert, M.S. Andersen, R.I. Acworth, and J.H. Tellam. 2017. "Characterising the dynamics of surface water-groundwater interactions in intermittent and ephemeral streams using streambed thermal signatures." <i>Advances in Water Resources</i> 107: 354-369. <u>https://www.sciencedirect.com/science/article/pii/S030917081730089</u> <u>1</u>. 	Streams	Supports
Schulz-Zunkel, C., M. Baborowski, T. Ehlert, H.D. Kasperidus, F. Krüger, P. Horchler, B. Neukirchen, H. Rupp, M. Scholz, L. Symmank, and S. Natho. 2021. "Simple modelling for a large-scale assessment of total phosphorus retention in the floodplains of large rivers." <i>Wetlands</i> 41 (6): 68. <u>https://doi.org/10.1007/s13157-021-01458-x</u> .	Floodplain Wetlands and Open Waters	Supports

Of the 43 citations received by the agencies during the public comment period that were found to be peerreviewed and published during or after 2014, the agencies were able to discern the appropriate and relevant aquatic system typology in 22 cases (15 stream systems, one floodplain wetlands and open water, three non-floodplain wetlands and open waters, and three papers that addressed both headwater streams and non-floodplain wetlands and open waters). The agencies were able to assess a conclusion on findings relevant to the Science Report in 13 of the 22 papers (seven stream-system focused papers, two papers that focused on both streams and non-floodplain wetlands and open waters; see Table 2). All 13 papers were found to support the conclusions of the Science Report. **Table C-9: Peer-Reviewed References Published Since 2014 Provided During the Public Comment Period Relevant to the Conclusions of the Science Report**. Thirteen of the 43 peer-reviewed citations (published in or after 2014) provided to the agencies during the public comment period were found to have sufficient information to discern system typology (*e.g.*, stream, non-floodplain wetlands and open waters, or floodplain wetlands and open waters) as well as a determination of whether the paper supports, refutes, or neither supports nor refutes the findings of the Science Report. In all 13 cases, the papers were found by ORD reviewers to support the findings of the Science Report.

Citation	Aquatic System Type	Science Report Findings
 Callahan, M. K., Rains, M. C., Bellino, J. C., Walker, C. M., Baird, S. J., Whigham, D. F., & King, R. S. 2015. "Controls on temperature in salmonid-bearing headwater streams in two common hydrogeologic settings, Kenai Peninsula, Alaska." <i>Journal of the American Water</i> <i>Resources Association</i> 51(1): 84-98. 	Streams	Supports
Epting, S. M., Hosen, J. D., Alexander, L. C., Lang, M. W., Armstrong, A. W., & Palmer, M. A. 2018. "Landscape metrics as predictors of hydrologic connectivity between Coastal Plain forested wetlands and streams." <i>Hydrological Processes</i> 32(4): 516-532.	Non- floodplain wetlands and Open Waters; Streams	Supports
Fesenmyer, K., <i>et al.</i> 2021. "Large portion of USA streams lose protection with new interpretation of Clean Water Act." <i>Freshwater</i> <i>Science</i> 40: 252-258.	Streams	Supports
Jackson, C. R., Sytsma, C., Sutter, L. A., & Batzer, D. P. 2021. "Redefining Waters of the US: a Case Study from the Edge of the Okefenokee Swamp." <i>Wetlands</i> 41(8): 1-10.	Non- floodplain Wetlands and Open Waters	Supports
 Lane, C. R., I. F. Creed, H. E. Golden, S. G. Leibowitz, D. M. Mushet, M. C. Rains, Q. Wu, E. D'Amico, L. C. Alexander, G. A. Ali, N. B. Basu, M. G. Bennett, J. R. Christensen, M. J. Cohen, T. P. Covino, B. DeVries, R. A. Hill, K. Jencso, M. W. Lang, D. L. McLaughlin, D. O. Rosenberry, J. Rover and M. K. Vanderhoof. 2022. "Vulnerable Waters are Essential to Watershed Resilience." <i>Ecosystems</i>. <u>https://doi.org.10.1007/s10021-021-00737-2</u>. 	Non- floodplain wetlands and Open Waters; Streams	Supports
Merritt, A. M., Lane, B., & Hawkins, C. P. 2021. "Classification and Prediction of Natural Streamflow Regimes in Arid Regions of the USA." <i>Water</i> 13(3): 380.	Streams	Supports

Citation	Aquatic System Type	Science Report Findings
Miller, O. L., Putman, A. L., Alder, J., Miller, M., Jones, D. K., & Wise, D. R. 2021. "Changing climate drives future streamflow declines and challenges in meeting water demand across the southwestern United States." <i>Journal of Hydrology</i> 11: 100074.	Streams	Supports
New Mexico Bureau of Geology and Mineral Resources. 2022. <i>Climate change in New Mexico over the next 50 years: Impacts on water resources</i> . New Mexico Bureau of Geology and Mineral Resources, Bulletin 164. <u>https://geoinfo.nmt.edu/ClimatePanel/report/</u> .	Streams	Supports
 Paller, M. H., Prusha, B. A., Fletcher, D. E., Kosnicki, E., Sefick, S. A., Jarrell, M. S., Sterrett, S. C., Grosse, A. M., Tuberville, T. D., & Feminella, J. W. 2016. "Factors influencing stream fish species composition and functional properties at multiple spatial scales in the Sand Hills of the southeastern United States." <i>Transactions of the American Fisheries Society</i> 145(3): 545-562. 	Streams	Supports
Price, S. J., Muncy, B. L., Bonner, S. J., Drayer, A. N., & Barton, C. D. 2016. "Effects of mountaintop removal mining and valley filling on the occupancy and abundance of stream salamanders." <i>Journal of</i> <i>Applied Ecology</i> 53(2): 459-468.	Streams	Supports
 Speir, S. L., Tank, J. L., Bieroza, M., Mahl, U. H., & Royer, T. V. 2021. "Storm size and hydrologic modification influence nitrate mobilization and transport in agricultural watersheds." <i>Biogeochemistry</i> 156(3): 319-334. 	Streams	Supports
Sullivan, S.M.P., <i>et al.</i> 2020. "Distorting science, putting water at risk." <i>Science</i> 369(6505): 766-768.	Stream and Wetlands	Supports
Zimmer, M. A., & McGlynn, B. L. (2017). "Bidirectional stream- groundwater flow in response to ephemeral and intermittent streamflow and groundwater seasonality." <i>Hydrological</i> <i>Processes</i> 31(22): 3871-3880.	Streams	Supports

Appendix C3ii: Additional References Not Captured During Screening Process (October 2021)

The below references were identified by ORD reviewers in late October 2021 as potentially relevant to the effort but not captured in the original screening began in mid-June 2021 (*e.g.*, they were scientific manuscripts accepted or published since the screening process began or were missed in the original screening). These references were included as Appendix C2 in the Technical Support Document for the

Proposed Rule. As described above, these, in addition to the literature received during the public comment period, were reviewed for their relevance to the Science Report conclusions.

- Acworth, R.I., G.C. Rau, M.O. Cuthbert, K. Leggett, and M.S. Andersen. 2021. "Runoff and focused groundwater-recharge response to flooding rains in the arid zone of Australia." *Hydrogeology Journal* 29(2): 737-764. <u>https://doi.org/10.1007/s10040-020-02284-x</u>.
- Arce, M.I., C. Mendoza-Lera, M. Almagro, N. Catalán, A.M. Romaní, E. Martí, R. Gómez, S. Bernal, A. Foulquier, M. Mutz, R. Marcé, A. Zoppini, G. Gionchetta, G. Weigelhofer, R. del Campo, C.T. Robinson, A. Gilmer, M. Rulik, B. Obrador, O. Shumilova, S. Zlatanović, S. Arnon, P. Baldrian, G. Singer, T. Datry, N. Skoulikidis, B. Tietjen, and D. von Schiller. 2019. "A conceptual framework for understanding the biogeochemistry of dry riverbeds through the lens of soil science." *Earth-Science Reviews* 188: 441-453. <u>https://www.sciencedirect.com/science/article/pii/S001282521830059X</u>.
- Arce, M.I., M.d.M. Sánchez-Montoya, and R. Gómez. 2015. "Nitrogen processing following experimental sediment rewetting in isolated pools in an agricultural stream of a semiarid region." *Ecological Engineering* 77: 233-241. <u>https://doi.org/10.1016/j.ecoleng.2015.01.035</u>.
- Arora, B., M. Burrus, M. Newcomer, C.I. Steefel, R.W.H. Carroll, D. Dwivedi, W. Dong, K.H. Williams, and S.S. Hubbard. 2020. "Differential C-Q Analysis: A New Approach to Inferring Lateral Transport and Hydrologic Transients Within Multiple Reaches of a Mountainous Headwater Catchment." *Frontiers in Water* 2 (24). <u>https://www.frontiersin.org/article/10.3389/frwa.2020.00024</u>.
- Baulch, H., C. Whitfield, J. Wolfe, N. Basu, A. Bedard-Haughn, K. Belcher, R. Clark, G. Ferguson, M. Hayashi, A. Ireson, P. Lloyd-Smith, P. Loring, J.W. Pomeroy, K. Shook, and C. Spence. 2021.
 "Synthesis of science: findings on Canadian Prairie wetland drainage." *Canadian Water Resources Journal / Revue canadienne des ressources hydriques*: 1-13. https://doi.org/10.1080/07011784.2021.1973911.
- Beck, H.E., M. Pan, P. Lin, J. Seibert, A.I.J.M. van Dijk, and E.F. Wood. 2020. "Global Fully Distributed Parameter Regionalization Based on Observed Streamflow From 4,229 Headwater Catchments." *Journal of Geophysical Research: Atmospheres* 125 (17): e2019JD031485. <u>https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019JD031485</u>.
- Buttle, J.M. 2016. "Dynamic storage: a potential metric of inter-basin differences in storage properties." *Hydrological Processes* 30 (24): 4644-4653. <u>https://doi.org/10.1002/hyp.10931</u>.
- Carroll, R.W.H., D. Gochis, and K.H. Williams. 2020. "Efficiency of the Summer Monsoon in Generating Streamflow Within a Snow-Dominated Headwater Basin of the Colorado River." *Geophysical Research Letters* 47 (23): e2020GL090856. <u>https://doi.org/10.1029/2020GL090856</u>.
- Crabot, J., C.P. Mondy, P. Usseglio-Polatera, K.M. Fritz, P.J. Wood, M.J. Greenwood, M.T. Bogan, E.I. Meyer, and T. Datry. 2021. "A global perspective on the functional responses of stream communities to flow intermittence." *Ecography* 44 (10): 1511-1523. https://onlinelibrary.wiley.com/doi/abs/10.1111/ecog.05697.

- Dages, C., A. Samouëlian, S. Negro, V. Storck, O. Huttel, and M. Voltz. 2015. "Seepage patterns of Diuron in a ditch bed during a sequence of flood events." *Science of The Total Environment* 537: 120-128. https://www.sciencedirect.com/science/article/pii/S0048969715304721.
- Dollinger, J., C. Dagès, S. Negro, J.-S. Bailly, and M. Voltz. 2016. "Variability of glyphosate and diuron sorption capacities of ditch beds determined using new indicator-based methods." *Science of The Total Environment* 573: 716-726. https://www.sciencedirect.com/science/article/pii/S0048969716318708.
- Dollinger, J., A. Samouëlian, C. Dagès, M. Lanoix, Y. Blanca, M. Voltz, and G. Coulouma. 2018. "Contrasting soil property patterns between ditch bed and neighbouring field profiles evidence the need of specific approaches when assessing water and pesticide fate in farmed landscapes." *Geoderma* 309: 50-59.
- Evenson, G.R., H.E. Golden, J.R. Christensen, C.R. Lane, A. Rajib, E. D'Amico, D.T. Mahoney, E. White, and Q. Wu. 2021. "Wetland restoration yields dynamic nitrate responses across the Upper Mississippi river basin." *Environmental Research Communications* 3 (9): 095002. http://dx.doi.org/10.1088/2515-7620/ac2125.
- Fakir, Y., H. Bouimouass, and J. Constantz. 2021. "Seasonality in Intermittent Streamflow Losses Beneath a Semiarid Mediterranean Wadi." *Water Resources Research* 57 (6): e2021WR029743. https://doi.org/10.1029/2021WR029743.
- Gala, T.S., and D. Young. 2021. "Geographically Isolated Depressional Wetlands Hydrodynamics, Ecosystem Functions and Conditions." *Applied Ecology and Environmental Sciences* 3 (4): 108-116. http://pubs.sciepub.com/.
- Gall, H.E., S.A. Sassman, B. Jenkinson, L.S. Lee, and C.T. Jafvert. 2015. "Comparison of export dynamics of nutrients and animal-borne estrogens from a tile-drained Midwestern agroecosystem." *Water Research* 72: 162-173. <u>https://www.sciencedirect.com/science/article/pii/S0043135414006083</u>.
- Gallo, E.L., K.A. Lohse, C.M. Ferlin, T. Meixner, and P.D. Brooks. 2014. "Physical and biological controls on trace gas fluxes in semi-arid urban ephemeral waterways." *Biogeochemistry* 121 (1): 189-207. <u>https://doi.org/10.1007/s10533-013-9927-0</u>.
- Golden, H.E., C.R. Lane, A. Rajib, and Q. Wu. 2021. "Improving global flood and drought predictions: integrating non-floodplain wetlands into watershed hydrologic models." *Environmental Research Letters* 16 (9): 091002. <u>http://dx.doi.org/10.1088/1748-9326/ac1fbc</u>.
- Gómez-Gener, L., A.R. Siebers, M.I. Arce, S. Arnon, S. Bernal, R. Bolpagni, T. Datry, G. Gionchetta, H.-P. Grossart, C. Mendoza-Lera, V. Pohl, U. Risse-Buhl, O. Shumilova, O. Tzoraki, D. von Schiller, A. Weigand, G. Weigelhofer, D. Zak, and A. Zoppini. 2021. "Towards an improved understanding of biogeochemical processes across surface-groundwater interactions in intermittent rivers and ephemeral streams." *Earth-Science Reviews* 220: 103724. https://www.sciencedirect.com/science/article/pii/S0012825221002257.

Gutierrez-Jurado, K.Y., D. Partington, and M. Shanafield. 2021. "Taking theory to the field: streamflow generation mechanisms in an intermittent Mediterranean catchment." *Hydrol. Earth Syst. Sci.* 25 (8): 4299-4317. https://hess.copernicus.org/articles/25/4299/2021/.

Hansen, A.T., T. Campbell, S.J. Cho, J.A. Czuba, B.J. Dalzell, C.L. Dolph, P.L. Hawthorne, S. Rabotyagov, Z. Lang, K. Kumarasamy, P. Belmont, J.C. Finlay, E. Foufoula-Georgiou, K.B. Gran, C.L. Kling, and P. Wilcock. 2021. "Integrated assessment modeling reveals near-channel management as cost-effective to improve water quality in agricultural watersheds." *Proceedings of the National Academy of Sciences* 118 (28): e2024912118. http://www.pnas.org/content/118/28/e2024912118.abstract.

- Jakubínský, J., M. Prokopová, P. Raška, L. Salvati, N. Bezak, O. Cudlín, P. Cudlín, J. Purkyt, P. Vezza, C. Camporeale, J. Daněk, M. Pástor, and T. Lepeška. 2021. "Managing floodplains using naturebased solutions to support multiple ecosystem functions and services." WIREs Water 8 (5): e1545. <u>https://doi.org/10.1002/wat2.1545</u>.
- Kampf, S.K., K.A. Dwire, M.P. Fairchild, J. Dunham, C.D. Snyder, K.L. Jaeger, C.H. Luce, J.C. Hammond, C. Wilson, M.A. Zimmer, and M. Sidell. 2021. "Managing nonperennial headwater streams in temperate forests of the United States." *Forest Ecology and Management* 497: 119523. <u>https://www.sciencedirect.com/science/article/pii/S0378112721006137</u>.
- Kavehei, A., D.B. Gore, A.A. Chariton, and G.C. Hose. 2021. "Impact assessment of ephemeral discharge of contamination downstream of two legacy base metal mines using environmental DNA." *Journal of Hazardous Materials* 419: 126483. https://www.sciencedirect.com/science/article/pii/S0304389421014485.
- Klammler, H., C.J. Quintero, J.W. Jawitz, D.L. McLaughlin, and M.J. Cohen. 2020. "Local Storage Dynamics of Individual Wetlands Predict Wetlandscape Discharge." *Water Resources Research* 56 (11): e2020WR027581. <u>https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020WR027581</u>.
- Köhn, D., C. Welpelo, A. Günther, and G. Jurasinski. 2021. "Drainage Ditches Contribute Considerably to the CH4 Budget of a Drained and a Rewetted Temperate Fen." *Wetlands* 41 (6): 71. <u>https://doi.org/10.1007/s13157-021-01465-y</u>.
- Land, M., W. Granéli, A. Grimvall, C.C. Hoffmann, W.J. Mitsch, K.S. Tonderski, and J.T.A. Verhoeven. 2016. "How effective are created or restored freshwater wetlands for nitrogen and phosphorus removal? A systematic review." *Environmental Evidence* 5 (1): 9. <u>https://doi.org/10.1186/s13750-016-0060-0</u>.
- Min, L., P.Y. Vasilevskiy, P. Wang, S.P. Pozdniakov, and J. Yu. 2020. "Numerical Approaches for Estimating Daily River Leakage from Arid Ephemeral Streams." *Water* 12 (2).
- Nahkala, B.A., A.L. Kaleita, and M.L. Soupir. 2021. "Characterization of prairie pothole inundation using AnnAGNPS under varying management and drainage scenarios." *Agricultural Water Management* 255: 107002. <u>https://www.sciencedirect.com/science/article/pii/S0378377421002675</u>.
- Natho, S. 2021. "How Flood Hazard Maps Improve the Understanding of Ecologically Active Floodplains." *Water* 13 (7).

- Price, A.N., C.N. Jones, J.C. Hammond, M.A. Zimmer, and S.C. Zipper. 2021. "The Drying Regimes of Non-Perennial Rivers and Streams." *Geophysical Research Letters* 48 (14): e2021GL093298. <u>https://doi.org/10.1029/2021GL093298</u>.
- Price, J.J., and J.F. Berkowitz. 2020. "Wetland Functional Responses to Prolonged Inundation in the Active Mississippi River Floodplain." *Wetlands* 40 (6): 1949-1956. <u>https://doi.org/10.1007/s13157-020-01309-1</u>.
- Rau, G.C., L.J.S. Halloran, M.O. Cuthbert, M.S. Andersen, R.I. Acworth, and J.H. Tellam. 2017.
 "Characterising the dynamics of surface water-groundwater interactions in intermittent and ephemeral streams using streambed thermal signatures." *Advances in Water Resources* 107: 354-369. <u>https://www.sciencedirect.com/science/article/pii/S0309170817300891</u>.
- Rodríguez-Rodríguez, M., H. Aguilera, C. Guardiola-Albert, and A. Fernández-Ayuso. 2021. "Climate Influence Vs. Local Drivers in Surface Water-Groundwater Interactions in Eight Ponds of Doñana National Park (Southern Spain)." Wetlands 41 (2): 25. <u>https://doi.org/10.1007/s13157-021-01425-6</u>.
- Schulz-Zunkel, C., M. Baborowski, T. Ehlert, H.D. Kasperidus, F. Krüger, P. Horchler, B. Neukirchen, H. Rupp, M. Scholz, L. Symmank, and S. Natho. 2021. "Simple modelling for a large-scale assessment of total phosphorus retention in the floodplains of large rivers." *Wetlands* 41 (6): 68. <u>https://doi.org/10.1007/s13157-021-01458-x</u>.
- Tripathi, M., P.K. Yadav, B.R. Chahar, and P. Dietrich. 2021. "A review on groundwater–surface water interaction highlighting the significance of streambed and aquifer properties on the exchanging flux." *Environmental Earth Sciences* 80 (17): 604. <u>https://doi.org/10.1007/s12665-021-09897-9</u>.
- Zhang, Z., L.E. Bortolotti, Z. Li, L.M. Armstrong, T.W. Bell, and Y. Li. 2021. "Heterogeneous Changes to Wetlands in the Canadian Prairies Under Future Climate." *Water Resources Research* 57 (7): e2020WR028727. <u>https://doi.org/10.1029/2020WR028727</u>.